# BELLSOUTH® / CLEC Agreement

# Customer Name: NewSouth Communications, Corp.

NewSouth Communications Corp.	2
Title Page and Table of Contents	3
General Terms and Conditions	4
Attachment 1	29
Attachment 1 Rates	68
Attachment 2	69
Attachment 2 Rates	147
Attachment 3	219
Attachment 3 Rates	243
Attachment 4	250
Attachment 5	316
Attachment 5 Rates	324
Attachment 6	325
Attachment 7	335
Attachment 7 Rates	354
Attachment 8	355
Attachment 9	357
Attachment 10	359
Attachment 11	369
Attachment 12	380
Amendment 6/27/01-Lata Wide	383
Interim Rates Amendment	386
Permanent Rates Amendment	390
Amendment 9-24-01 Attachment 2 Rates	427
Amendment (new combos/rates LA; new combos KY) 11/14/01	632
NewSouth - Florida Rates Amendment	672
Trunk Port Amendment	718

Note: This page is not part of the actual signed contract/amendment, but is present for record keeping purposes only.

# By and Between

BellSouth Telecommunications, Inc.

# And

**NewSouth Communications, Corp.** 

# TABLE OF CONTENTS

#### **General Terms and Conditions**

- 1. Definitions
- 2. Term of the Agreement
- 3. Operational Support Systems
- 4. Parity
- 5. White Pages Listings
- 6. Bona Fide Request/New Business Request Process for Further Unbundling
- 7. Court Ordered Requests for Call Detail Records and Other Subscriber Information
- 8. Liability and Indemnification
- 9. Intellectual Property Rights and Indemnification
- 10. Proprietary and Confidential Information
- 11. Assignments
- 12. Resolution of Disputes
- 13. Taxes
- 14. Force Majeure
- 15. Adoption of Agreements
- 16. Modification of Agreement
- 17. Non-waiver of Legal Rights
- 18. Severability
- 19. Waivers
- 20. Governing Law
- 21. Arm's Length Negotiations
- 22. Notices
- 23. Rule of Construction
- 24. Headings of No Force or Effect
- 25. Multiple Counterparts
- 26. Implementation of Agreement
- 27. Filing of Agreement
- 28. Compliance with Applicable Law
- 29. Necessary Approvals
- 30. Good Faith Performance
- 31. Nonexclusive Dealings
- 32. Survival
- 33. Entire Agreement
- Attachment 1 Resale
- **Attachment 2 Network Elements and Other Services**
- **Attachment 3 Network Interconnection**
- **Attachment 4 Physical Collocation**
- **Attachment 5 Access to Numbers and Number Portability**
- Attachment 6 Pre-Ordering, Ordering and Provisioning, Maintenance and Repair
- **Attachment 7 Billing and Billing Accuracy Certification**
- Attachment 8 Rights-of-Way, Conduits and Pole Attachments
- **Attachment 9 Performance Measurements**
- **Attachment 10- Agreement Implementation Template**
- **Attachment 11- BellSouth Disaster Recovery Plan**
- **Attachment 12-Bona Fide Request and New Business Requests Process**

#### **AGREEMENT**

**THIS AGREEMENT** is made by and between BellSouth Telecommunications, Inc., ("BellSouth"), a Georgia corporation, and NewSouth Communications, Corp., ("NewSouth") a Delaware corporation, and shall be deemed effective as of the date of the last signature of both Parties ("Effective Date"). This Agreement may refer to either BellSouth or NewSouth or both as a "Party" or "Parties."

#### WITNESSETH

WHEREAS, BellSouth is an Incumbent Local Exchange Telecommunications Company (ILEC) authorized to provide telecommunications services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee; and

WHEREAS, NewSouth is or seeks to become a Competitive Local Exchange Telecommunications Company ("CLEC") authorized to provide telecommunications services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee; and

WHEREAS, the Parties wish to resell BellSouth's telecommunications services and/or interconnect their facilities, purchase network elements and other services, and exchange traffic specifically for the purposes of fulfilling their obligations pursuant to sections 251 and 252 of the Telecommunications Act of 1996 ("the Act").

**NOW THEREFORE**, in consideration of the mutual agreements contained herein, BellSouth and NewSouth agree as follows:

# 1. Purpose

The Parties agree that the rates, terms and conditions contained within this Agreement, including all Attachments, comply and conform with each Parties' obligations under sections 251 and 252 of the Act. The resale, access and interconnection obligations contained herein enable NewSouth to provide competing telephone exchange service to residential and business subscribers within the territory of BellSouth. The Parties agree that NewSouth will not be considered to have offered telecommunications services to the public in any state within BellSouth's region until such time as it has ordered services for resale or interconnection facilities for the purposes of providing business and/or residential local exchange service to customers.

#### 2. Term of the Agreement

2.1 The term of this Agreement shall be two years, beginning on the Effective Date and shall apply to the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee. If as of the expiration of this Agreement, a Subsequent Agreement (as defined in Section 2.2 below) has not been executed by the Parties, this Agreement shall continue on a month-to-

Page 2

month basis while a Subsequent Agreement is being negotiated. The Parties' rights and obligations with respect to this Agreement after expiration shall be as set forth in Section 2.4 below.

- 2.2 The Parties agree that by no later than one hundred and eighty (180) days prior to the expiration of this Agreement, they shall commence negotiations with regard to the terms, conditions and prices of resale and/or local interconnection to be effective beginning on the expiration date of this Agreement ("Subsequent Agreement").
- If, within one hundred and thirty-five (135) days of commencing the negotiation referred to in Section 2.2 above, the Parties are unable to satisfactorily negotiate new resale and/or local interconnection terms, conditions and prices, either Party may petition the Commission to establish appropriate local interconnection and/or resale arrangements pursuant to 47 U.S.C. 252. The Parties agree that, in such event, they shall encourage the Commission to issue its order regarding the appropriate local interconnection and/or resale arrangements no later than the expiration date of this Agreement. The Parties further agree that in the event the Commission does not issue its order prior to the expiration date of this Agreement, or if the Parties continue beyond the expiration date of this Agreement to negotiate the local interconnection and/or resale arrangements without Commission intervention, the terms, conditions and prices ultimately ordered by the Commission, or negotiated by the Parties, will be effective retroactive to the day following the expiration date of this Agreement.
- 2.4 Notwithstanding the foregoing, in the event that as of the date of expiration of this Agreement and conversion of this Agreement to a month-to-month term, the Parties have not entered into a Subsequent Agreement and either no arbitration proceeding has been filed in accordance with Section 2.3 above, or the Parties have not mutually agreed (where permissible) to extend the arbitration window for petitioning the applicable Commission(s) for resolution of those terms upon which the Parties have not agreed, then either Party may terminate this Agreement upon sixty (60) days notice to the other Party. In the event that BellSouth terminates this Agreement as provided above, BellSouth shall continue to offer services to NewSouth pursuant to the terms, conditions and rates set forth in BellSouth's Statement of Generally Available Terms (SGAT) to the extent an SGAT has been approved by the applicable Commission(s). If any state Commission has not approved a BellSouth SGAT, then upon BellSouth's termination of this Agreement as provided herein, BellSouth will continue to provide services to NewSouth pursuant to BellSouth's then current standard interconnection agreement. In the event that the SGAT or BellSouth's standard interconnection agreement becomes effective as between the Parties, the Parties may continue to negotiate a Subsequent Agreement, and the terms of such Subsequent Agreement shall be effective retroactive to the day following expiration of this Agreement.

#### 3. Ordering Procedures

- 3.1 NewSouth shall provide BellSouth its Carrier Identification Code (CIC), Operating Company Number (OCN), Group Access Code (GAC) and Access Customer Name and Address (ACNA) code as applicable prior to placing its first order.
- 3.2 The Parties agree to adhere to the BellSouth Local Interconnection and Facility Based Ordering Guide and Resale Ordering Guide, as appropriate for the services ordered.
- NewSouth shall pay charges for Operational Support Systems (OSS) as set forth in this Agreement in Attachment 1 and/or in Attachment 2, 3, 5 and 7 as applicable.

# 4. Parity

When NewSouth purchases, pursuant to Attachment 1 of this Agreement, telecommunications services from BellSouth for the purposes of resale to end users, BellSouth shall provide said services so that the services are equal in quality, subject to the same conditions, and provided within the same provisioning time intervals that BellSouth provides to its affiliates, subsidiaries and end users. To the extent technically feasible, the quality of a Network Element, as well as the quality of the access to such Network Element provided by BellSouth to NewSouth shall be at least equal in quality to that which BellSouth provides to itself. The quality of the interconnection between the networks of BellSouth and the network of NewSouth shall be at a level that is equal to that which BellSouth provides itself, a subsidiary, an Affiliate, or any other party. The interconnection facilities shall be designed to meet the same technical criteria and service standards that are used within BellSouth's network and shall extend to a consideration of service quality as perceived by end users and service quality as perceived by NewSouth.

#### 5. White Pages Listings

BellSouth shall provide NewSouth and their customers access to white pages directory listings under the following terms:

- 5.1 <u>Listings</u>. NewSouth shall provide all new, changed and deleted listings on a timely basis and BellSouth or its agent will include NewSouth residential and business customer listings in the appropriate White Pages (residential and business) or alphabetical directories. Directory listings will make no distinction between NewSouth and BellSouth subscribers.
- 5.2 <u>Rates.</u> BellSouth and NewSouth will provide to each other subscriber primary listing information in the White Pages for a non-recurring charge.
- 5.3 Procedures for Submitting NewSouth Subscriber Information are found in BellSouth's Ordering Guide for manually processed listings and in the Local Exchange Ordering Guide for mechanically submitted listings.

- 5.3.1 Notwithstanding any provision(s) to the contrary, NewSouth agrees to provide to BellSouth, and BellSouth agrees to accept, NewSouth's Subscriber Listing Information (SLI) relating to NewSouth's customers in the geographic area(s) covered by this Interconnection Agreement. NewSouth authorizes BellSouth to release all such NewSouth SLI provided to BellSouth by NewSouth to qualifying third parties via either license agreement or BellSouth's Directory Publishers Database Service (DPDS), General Subscriber Services Tariff, Section A38.2, as the same may be amended from time to time. Such CLEC SLI shall be intermingled with BellSouth's own customer listings of any other CLEC that has authorized a similar release of SLI. Where necessary, BellSouth will use good faith efforts to obtain state commission approval of any necessary modifications to Section A38.2 of its tariff to provide for release of third party directory listings, including modifications regarding listings to be released pursuant to such tariff and BellSouth's liability therunder. BellSouth's obligation pursuant to this Section shall not arise in any particular state until the commission of such state has approved modifications to such tariff.
- 5.3.2 No compensation shall be paid to NewSouth for BellSouth's receipt of NewSouth SLI, or for the subsequent release to third parties of such SLI. In addition, to the extent BellSouth incurs costs to modify its systems to enable the release of NewSouth's SLI, or costs on an ongoing basis to administer the release of NewSouth SLI, NewSouth shall pay to BellSouth its proportionate share of the reasonable costs associated therewith.
- 5.3.3 BellSouth shall not be liable for the content or accuracy of any SLI provided by NewSouth under this Agreement. NewSouth shall indemnify, hold harmless and defend BellSouth from and against any damages, losses, liabilities, demands claims, suits, judgments, costs and expenses (including but not limited to reasonable attorneys' fees and expenses) arising from BellSouth's tariff obligations or otherwise and resulting from or arising out of any third party's claim of inaccurate NewSouth listings or use of the SLI provided pursuant to this Agreement. BellSouth shall forward to NewSouth any complaints received by BellSouth relating to the accuracy or quality of NewSouth listings.
- 5.3.4 Listings and subsequent updates will be released consistent with BellSouth system changes and/or update scheduling requirements.
- 5.4 <u>Unlisted/Non-Published Subscribers</u>. NewSouth will be required to provide to BellSouth the names, addresses and telephone numbers of all NewSouth customers that wish to be omitted from directories.
- 5.5 <u>Inclusion of NewSouth Customers in Directory Assistance Database</u>. BellSouth will include and maintain NewSouth subscriber listings in BellSouth's Directory Assistance databases at no recurring charge and NewSouth shall provide such Directory Assistance listings at no recurring charge. BellSouth and NewSouth will formulate appropriate procedures regarding lead-time, timeliness, format and content of listing information.

- Listing Information Confidentiality. BellSouth will accord NewSouth's directory listing information the same level of confidentiality that BellSouth accords its own directory listing information, and BellSouth shall limit access to NewSouth's customer proprietary confidential directory information to those BellSouth employees who are involved in the preparation of listings.
- 5.7 <u>Optional Listings</u>. Additional listings and optional listings will be offered by BellSouth at tariffed rates as set forth in the General Subscriber Services Tariff.
- 5.8 <u>Delivery.</u> BellSouth or its agent shall deliver White Pages directories to NewSouth subscribers at no charge or as specified in a separate BAPCO agreement.

### 6. Bona Fide Request/New Business Request Process for Further Unbundling

If NewSouth is a facilities based provider or a facilities based and resale provider, this section shall apply. BellSouth shall, upon request of NewSouth, provide to NewSouth access to its network elements at any technically feasible point for the provision of NewSouth's telecommunications service where such access is necessary and failure to provide access would impair the ability of NewSouth to provide services that it seeks to offer. Any request by NewSouth for access to a network element, interconnection option, or for the provisioning of any service or product that is not already available shall be treated as a Bona Fide Request/New Business Request, and shall be submitted to BellSouth pursuant to the Bona Fide Request/New Business Request process set forth in Attachment 12 of this Agreement

# 7. Court Ordered Requests for Call Detail Records and Other Subscriber Information

- 7.1 To the extent technically feasible, BellSouth maintains call detail records for NewSouth end users for limited time periods and can respond to subpoenas and court ordered requests for this information. BellSouth shall maintain such information for NewSouth end users for the same length of time it maintains such information for its own end users.
- NewSouth agrees that BellSouth will respond to subpoenas and court ordered requests delivered directly to BellSouth for the purpose of providing call detail records when the targeted telephone numbers belong to NewSouth end users. Billing for such requests will be generated by BellSouth and directed to the law enforcement agency initiating the request.
- 7.3 Where BellSouth is providing to NewSouth telecommunications services for resale or providing to NewSouth the local switching function, then NewSouth agrees that in those cases where NewSouth receives subpoenas or court ordered requests regarding

targeted telephone numbers belonging to NewSouth end users, if NewSouth does not have the requested information, NewSouth will advise the law enforcement agency initiating the request to redirect the subpoena or court ordered request to BellSouth. Where the request has been forwarded to BellSouth, billing for call detail information will be generated by BellSouth and directed to the law enforcement agency initiating the request.

In all other instances, NewSouth will provide NewSouth end user and/or other customer information that is available to NewSouth in response to subpoenas and court orders for their own customer records. When BellSouth receives subpoenas or court ordered requests regarding targeted telephone numbers belonging to NewSouth end users, BellSouth will advise the law enforcement agency initiating the request to redirect the subpoena or court ordered request to NewSouth.

#### 8. Liability and Indemnification

- 8.1 <u>BellSouth Liability</u>. BellSouth shall take financial responsibility for its own actions in causing or its lack of action in preventing, unbillable or uncollectible NewSouth revenues.
- 8.2 NewSouth Liability. In the event that NewSouth consists of two (2) or more separate entities as set forth in the preamble to this Agreement, all such entities shall be jointly and severally liable for the obligations of NewSouth under this Agreement.
- 8.3 <u>Liability for Acts or Omissions of Third Parties</u>. Neither BellSouth nor NewSouth shall be liable for any act or omission of another telecommunications company providing a portion of the services provided under this Agreement.
- 8.4 <u>Limitation of Liability.</u>
- 8.4.1 Each Party's liability to the other for any loss, cost, claim, injury or liability or expense, including reasonable attorney's fees relating to or arising out of any negligent act or omission in its performance of this Agreement whether in contract or in tort, shall be limited to a credit for the actual cost of the services or functions not performed or improperly performed.
- 8.4.2 <u>Limitations in Tariffs</u>. A Party may, in its sole discretion, provide in its tariffs and contracts with its Customer and third parties that relate to any service, product or function provided or contemplated under this Agreement, that to the maximum extent permitted by Applicable Law, such Party shall not be liable to Customer or third Party for (i) any Loss relating to or arising out of this Agreement, whether in contract, tort or otherwise, that exceeds the amount such Party would have charged that applicable person for the service, product or function that gave rise to such Loss and (ii) Consequential Damages. To the extent that a Party elects not to place in its tariffs or contracts such limitations of liability, and the other Party incurs a Loss as a result thereof, such Party shall indemnify and reimburse the

other Party for that portion of the Loss that would have been limited had the first Party included in its tariffs and contracts the limitations of liability that such other Party included in its own tariffs at the time of such Loss.

- 8.4.3 Neither BellSouth nor NewSouth shall be liable for damages to the other's terminal location, POI or other company's customers' premises resulting from the furnishing of a service, including, but not limited to, the installation and removal of equipment or associated wiring, except to the extent caused by a company's negligence or willful misconduct or by a company's failure to properly ground a local loop after disconnection.
- 8.4.4 Except in cases of gross negligence, willful or intentional misconduct, under no circumstance shall a Party be responsible or liable for indirect, incidental, or consequential damages, including, but not limited to, economic loss or lost business or profits, damages arising from the use or performance of equipment or software, or the loss of use of software or equipment, or accessories attached thereto, delay, error, or loss of data. In connection with this limitation of liability, each Party recognizes that the other Party may, from time to time, provide advice, make recommendations, or supply other analyses related to the Services, or facilities described in this Agreement, and, while each Party shall use diligent efforts in this regard, the Parties acknowledge and agree that this limitation of liability shall apply to provision of such advice, recommendations, and analyses.
- 8.5 <u>Indemnification for Certain Claims</u>. The Party providing services hereunder, its affiliates and its parent company, shall be indemnified, defended and held harmless by the Party receiving services hereunder against any claim, loss or damage arising from the receiving company's use of the services provided under this Agreement pertaining to (1) claims for libel, slander or invasion of privacy arising from the content of the receiving company's own communications, or (2) any claim, loss or damage claimed by the customer of the Party receiving services arising from such company's use or reliance on the providing company's services, actions, duties, or obligations arising out of this Agreement.
- 8.6 <u>Disclaimer</u>. EXCEPT AS SPECIFICALLY PROVIDED TO THE CONTRARY IN THIS AGREEMENT, NEITHER PARTY MAKES ANY REPRESENTATIONS OR WARRANTIES TO THE OTHER PARTY CONCERNING THE SPECIFIC QUALITY OF ANY SERVICES, OR FACILITIES PROVIDED UNDER THIS AGREEMENT. THE PARTIES DISCLAIM, WITHOUT LIMITATION, ANY WARRANTY OR GUARANTEE OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, ARISING FROM COURSE OF PERFORMANCE, COURSE OF DEALING, OR FROM USAGES OF TRADE.

#### 9. Intellectual Property Rights and Indemnification

9.1 <u>No License.</u> No patent, copyright, trademark or other proprietary right is licensed, granted or otherwise transferred by this Agreement. NewSouth is strictly

prohibited from any use, including but not limited to in sales, in marketing or advertising of telecommunications services, of any BellSouth name, service mark or trademark.

- 9.2 Ownership of Intellectual Property. Any intellectual property which originates from or is developed by a Party shall remain in the exclusive ownership of that Party. Except for a limited license to use patents or copyrights to the extent necessary for the Parties to use any facilities or equipment (including software) or to receive any service solely as provided under this Agreement, no license in patent, copyright, trademark or trade secret, or other proprietary or intellectual property right now or hereafter owned, controlled or licensable by a Party, is granted to the other Party or shall be implied or arise by estoppel. It is the responsibility of each Party to ensure at no additional cost to the other Party that it has obtained any necessary licenses in relation to intellectual property of third Parties used in its network that may be required to enable the other Party to use any facilities or equipment (including software), to receive any service, or to perform its respective obligations under this Agreement.
- 9.3 <u>Indemnification</u>. The Party providing a service pursuant to this Agreement will defend the Party receiving such service or data provided as a result of such service against claims of infringement arising solely from the use by the receiving Party of such service and will indemnify the receiving Party for any damages awarded based solely on such claims in accordance with Section 8 of this Agreement.
- 9.4 <u>Claim of Infringement</u>. In the event that use of any facilities or equipment (including software), becomes, or in reasonable judgment of the Party who owns the affected network is likely to become, the subject of a claim, action, suit, or proceeding based on intellectual property infringement, then said Party shall promptly and at its sole expense, but subject to the limitations of liability set forth below:
- 9.4.1 modify or replace the applicable facilities or equipment (including software) while maintaining form and function, or
- 9.4.2 obtain a license sufficient to allow such use to continue.
- 9.4.3 In the event 9.4.1 or 9.4.2 are commercially unreasonable, then said Party may, terminate, upon reasonable notice, this contract with respect to use of, or services provided through use of, the affected facilities or equipment (including software), but solely to the extent required to avoid the infringement claim.
- 9.5 <u>Exception to Obligations</u>. Neither Party's obligations under this Section shall apply to the extent the infringement is caused by: (i) modification of the facilities or equipment (including software) by the indemnitee; (ii) use by the indemnitee of the facilities or equipment (including software) in combination with equipment or facilities (including software) not provided or authorized by the indemnitor provided the facilities or equipment (including software) would not be infringing if

used alone; (iii) conformance to specifications of the indemnitee which would necessarily result in infringement; or (iv) continued use by the indemnitee of the affected facilities or equipment (including software) after being placed on notice to discontinue use as set forth herein.

9.6 <u>Exclusive Remedy</u>. The foregoing shall constitute the Parties' sole and exclusive remedies and obligations with respect to a third party claim of intellectual property infringement arising out of the conduct of business under this Agreement.

# 10. Proprietary and Confidential Information

- 10.1 Proprietary and Confidential Information: It may be necessary for BellSouth and NewSouth, each as the "Discloser," to provide to the other party, as "Recipient," certain proprietary and confidential information(including trade secret information) including but not limited to technical, financial, marketing, staffing and business plans and information, strategic information, proposals, request for proposals, specifications, drawings, prices, costs, procedures, processes, business systems, software programs, techniques, customer account data, call detail records and like information (collectively the Discloser's "Information"). All Information shall be provided to Recipient in written or other tangible or electronic form, clearly marked with a confidential and, proprietary notice. Information orally or visually provided to Recipient must be designated by Discloser as confidential and proprietary at the time of such disclosure and must be reduced to writing marked with a confidential and proprietary notice and provided to Recipient within thirty (30) calendar days after such oral or visual disclosure.
- 10.2 Use and Protection of Information. Recipient shall use the Information solely for the purpose(s) of performing this Agreement, and Recipient shall protect Information from any use, distribution or disclosure except as permitted hereunder. Recipient will use the same standard of care to protect Information as Recipient uses to protect its own similar confidential and proprietary information, but not less than a reasonable standard of care. Recipient may disclose Information solely to the Authorized Representatives of the Recipient who (a) have a substantive need to know such Information in connection with performance of the Agreement; (b) have been advised of the confidential and proprietary nature of the Information; and (c) have personally agreed in writing to protect from unauthorized disclosure all confidential and proprietary information, of whatever source, to which they have access in the course of their employment. "Authorized Representatives" are the officers, directors and employees of Recipient and its Affiliates, as well as Recipient's and its Affiliates' consultants, contractors, counsel and agents. " Affiliates" means any company that is owned in whole or in part, now or in the future, directly or indirectly through a subsidiary, by a party hereto.
- 10.3 Ownership, Copying & Return of Information. Information remains at all times the property of Discloser. Recipient may make tangible or electronic copies, notes, summaries or extracts of Information only as necessary for use as authorized herein. All such tangible or electronic copies, notes, summaries or

extracts must be marked with the same confidential and proprietary notice as appears on the original. Upon Discloser's request, all or any requested portion of the Information (including, but not limited to, tangible and electronic copies, notes, summaries or extracts of any information) will be promptly returned to Discloser or destroyed, and Recipient will provide Discloser with written certification stating that such Information has been returned or destroyed.

- 10.4 Exceptions. Discloser's Information does not include: (a) any information publicly disclosed by Discloser; (b) any information Discloser in writing authorizes Recipient to disclose without restriction; (c) any information already lawfully known to Recipient at the time it is disclosed by the Discloser, without an obligation to keep confidential; or (d) any information Recipient lawfully obtains from any source other than Discloser, provided that such source lawfully disclosed and/or independently developed such information. If Recipient is required to provide Information to any court or government agency pursuant to written court order, subpoena, regulation or process of law, Recipient must first provided Discloser with prompt written notice of such requirement and cooperate with Discloser to appropriately protect against or limit the scope of such disclosure. To the fullest extent permitted by law, Recipient will continue to protect as confidential and proprietary all Information disclosed in response to a written court order, subpoena, regulation or process of law.
- 10.5 Equitable Relief. Recipient acknowledges and agrees that any breach or threatened breach of this Agreement is likely to cause Discloser irreparable harm for which money damages may not be an appropriate or sufficient remedy. Recipient therefore agrees that Discloser or its Affiliates, as the case may be, are entitled to receive injunctive or other equitable relief to remedy or prevent any breach or threatened breach of this Agreement. Such remedy is not the exclusive remedy for any breach or threatened breach of this Agreement, but is in addition to all other rights and remedies available at law or in equity.
- 10.6 <u>Survival of Confidentiality Obligations.</u> The parties' rights and obligations under this Section 10 shall survive and continue in effect until two (2) years after the expiration or termination date of this Agreement with regard to all Information exchanged during the term of this Agreement. Thereafter, the parties' rights and obligations hereunder survive and continue in effect with respect to any Information that is a trade secret under applicable law.

#### 11. Assignments

Any assignment by either Party to any non-affiliated entity of any right, obligation or duty, or of any other interest hereunder, in whole or in part, without the prior written consent of the other Party shall be void, and such consent shall not be unreasonably withheld. A Party may assign this Agreement or any right, obligation, duty or other interest hereunder to an Affiliate company of the Party without the consent of the other Party. All obligations and duties of any Party under this Agreement shall be binding on all successors in interest and assigns of such Party. No assignment or delegation hereof shall relieve the assignor of its obligations under this Agreement in the event that the assignee fails to perform such obligations.

## 12. Resolution of Disputes

Except as otherwise stated in this Agreement, the Parties agree that if any dispute arises as to the interpretation of any provision of this Agreement or as to the proper implementation of this Agreement, either Party may petition the Commission for a resolution of the dispute. However, each Party reserves any rights it may have to seek judicial review of any ruling made by the Commission concerning this Agreement.

#### 13. Taxes

- Definition. For purposes of this Section, the terms "taxes" and "fees" shall include but not limited to federal, state or local sales, use, excise, gross receipts or other taxes or tax-like fees of whatever nature and however designated (including tariff surcharges and any fees, charges or other payments, contractual or otherwise, for the use of public streets or rights of way, whether designated as franchise fees or otherwise) imposed, or sought to be imposed, on or with respect to the services furnished hereunder or measured by the charges or payments therefore, excluding any taxes levied on income.
- 13.2 Taxes and Fees Imposed Directly On Either Providing Party or Purchasing Party.
- Taxes and fees imposed on the providing Party, which are not permitted or required to be passed on by the providing Party to its customer, shall be borne and paid by the providing Party.
- Taxes and fees imposed on the purchasing Party, which are not required to be collected and/or remitted by the providing Party, shall be borne and paid by the purchasing Party.
- 13.3 <u>Taxes and Fees Imposed on Purchasing Party But Collected And Remitted By Providing Party.</u>

- Taxes and fees imposed on the purchasing Party shall be borne by the purchasing Party, even if the obligation to collect and/or remit such taxes or fees is placed on the providing Party.
- To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.
- 13.3.3 If the purchasing Party determines that in its opinion any such taxes or fees are not payable, the providing Party shall not bill such taxes or fees to the purchasing Party if the purchasing Party provides written certification, reasonably satisfactory to the providing Party, stating that it is exempt or otherwise not subject to the tax or fee, setting forth the basis therefor, and satisfying any other requirements under applicable law. If any authority seeks to collect any such tax or fee that the purchasing Party has determined and certified not to be payable, or any such tax or fee that was not billed by the providing Party, the purchasing Party may contest the same in good faith, at its own expense. In any such contest, the purchasing Party shall promptly furnish the providing Party with copies of all filings in any proceeding, protest, or legal challenge, all rulings issued in connection therewith, and all correspondence between the purchasing Party and the taxing authority.
- In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.
- 13.3.5 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.
- 13.3.6 Notwithstanding any provision to the contrary, the purchasing Party shall protect, indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other charges or payable expenses (including reasonable attorney fees) with respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.
- 13.3.7 Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a taxing authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.

- Taxes and Fees Imposed on Providing Party But Passed On To Purchasing Party.
- Taxes and fees imposed on the providing Party, which are permitted or required to be passed on by the providing Party to its customer, shall be borne by the purchasing Party.
- To the extent permitted by applicable law, any such taxes and/or fees shall be shown as separate items on applicable billing documents between the Parties. Notwithstanding the foregoing, the purchasing Party shall remain liable for any such taxes and fees regardless of whether they are actually billed by the providing Party at the time that the respective service is billed.
- 13.4.3 If the purchasing Party disagrees with the providing Party's determination as to the application or basis for any such tax or fee, the Parties shall consult with respect to the imposition and billing of such tax or fee. Notwithstanding the foregoing, the providing Party shall retain ultimate responsibility for determining whether and to what extent any such taxes or fees are applicable, and the purchasing Party shall abide by such determination and pay such taxes or fees to the providing Party. The providing Party shall further retain ultimate responsibility for determining whether and how to contest the imposition of such taxes and fees; provided, however, that any such contest undertaken at the request of the purchasing Party shall be at the purchasing Party's expense.
- In the event that all or any portion of an amount sought to be collected must be paid in order to contest the imposition of any such tax or fee, or to avoid the existence of a lien on the assets of the providing Party during the pendency of such contest, the purchasing Party shall be responsible for such payment and shall be entitled to the benefit of any refund or recovery.
- 13.4.5 If it is ultimately determined that any additional amount of such a tax or fee is due to the imposing authority, the purchasing Party shall pay such additional amount, including any interest and penalties thereon.
- 13.4.6 Notwithstanding any provision to the contrary, the purchasing Party shall protect indemnify and hold harmless (and defend at the purchasing Party's expense) the providing Party from and against any such tax or fee, interest or penalties thereon, or other reasonable charges or payable expenses (including reasonable attorney fees) with respect thereto, which are incurred by the providing Party in connection with any claim for or contest of any such tax or fee.
- Each Party shall notify the other Party in writing of any assessment, proposed assessment or other claim for any additional amount of such a tax or fee by a taxing authority; such notice to be provided, if possible, at least ten (10) days prior to the date by which a response, protest or other appeal must be filed, but in no event later than thirty (30) days after receipt of such assessment, proposed assessment or claim.

Mutual Cooperation. In any contest of a tax or fee by one Party, the other Party shall cooperate fully by providing records, testimony and such additional information or assistance as may reasonably be necessary to pursue the contest. Further, the other Party shall be reimbursed for any reasonable and necessary out-of-pocket copying and travel expenses incurred in assisting in such contest.

## 14. Force Majeure

In the event performance of this Agreement, or any obligation hereunder, is either directly or indirectly prevented, restricted, or interfered with by reason of fire, flood, earthquake or like acts of God, wars, revolution, civil commotion, explosion, acts of public enemy, embargo, acts of the government in its sovereign capacity, labor difficulties, including without limitation, strikes, slowdowns, picketing, or boycotts, unavailability of equipment from vendor, changes requested by Customer, or any other circumstances beyond the reasonable control and without the fault or negligence of the Party affected, the Party affected, upon giving prompt notice to the other Party, shall be excused from such performance on a day-to-day basis to the extent of such prevention, restriction, or interference (and the other Party shall likewise be excused from performance of its obligations on a day-to-day basis until the delay, restriction or interference has ceased); provided however, that the Party so affected shall use diligent efforts to avoid or remove such causes of non-performance and both Parties shall proceed whenever such causes are removed or cease.

### 15. Network Maintenance and Management

- The Parties shall work cooperatively to implement this Agreement. The Parties shall exchange appropriate information (e.g., maintenance contact numbers, network information, information required to comply with law enforcement and other security agencies of the Government, etc.) as reasonably required to implement and perform this Agreement.
- Each Party hereto shall design, maintain and operate their respective networks as necessary to ensure that the other Party hereto receives service quality which is consistent with generally accepted industry standards at least at parity with the network service quality given to itself, its Affiliates, its End Users or any other Telecommunications Carrier.
- Neither Party shall use any service or facility provided under this Agreement in a manner that impairs the quality of service to other Telecommunications Carriers' or to either Party's End Users. Each Party will provide the other Party notice of any such impairment at the earliest practicable time.
- BellSouth agrees to provide NewSouth prior notice consistent with applicable FCC rules and the Act of changes in the information necessary for the transmission

and routing of services using BellSouth's facilities or networks, as well as other changes that affect the interoperability of those respective facilities and networks. This Agreement is not intended to limit BellSouth's ability to upgrade its network through the incorporation of new equipment, new software or otherwise so long as such upgrades are not inconsistent with BellSouth's obligations to NewSouth under the terms of this Agreement.

### 16. Modification of Agreement

- BellSouth shall make available, pursuant to 47 USC § 252(i), and the FCC rules and regulations and Court Orders regarding such availability, to NewSouth any interconnection, service, or network element provided under any other agreement filed and approved pursuant to 47 USC § 252 (e).
- If NewSouth changes its name or makes changes to its company structure or identity due to a merger, acquisition, transfer or any other reason, it is the responsibility of NewSouth to notify BellSouth of said change and request that an amendment to this Agreement, if necessary, be executed to reflect said change.
- No modification, amendment, supplement to, or waiver of the Agreement or any of its provisions shall be effective and binding upon the Parties unless it is made in writing and duly signed by the Parties.
- Execution of this Agreement by either Party does not confirm or infer that the executing Party agrees with any decision(s) issued pursuant to the Telecommunications Act of 1996 and the consequences of those decisions on specific language in this Agreement. Neither Party waives its rights to appeal or otherwise challenge any such decision(s) and each Party reserves all of its rights to pursue any and all legal and/or equitable remedies, including appeals of any such decision(s).
- In the event that any effective legislative, regulatory, judicial or other legal action materially affects any material terms of this Agreement, or the ability of NewSouth or BellSouth to perform any material terms of this Agreement, NewSouth or BellSouth may, on thirty (30) days' written notice require that such terms be renegotiated, and the Parties shall renegotiate in good faith such mutually acceptable new terms as may be required. In the event that such new terms are not renegotiated within ninety (90) days after such notice, the Dispute shall be referred to the Dispute Resolution procedure set forth in Section 12.
- If any provision of this Agreement, or the application of such provision to either Party or circumstance, shall be held invalid, the remainder of the Agreement, or the application of any such provision to the Parties or circumstances other than those to which it is held invalid, shall not be effective thereby, provided that the Parties shall attempt to reformulate such invalid provision to give effect to such portions thereof as may be valid without defeating the intent of such provision.

#### 17. Waivers

A failure or delay of either Party to enforce any of the provisions hereof, to exercise any option which is herein provided, or to require performance of any of the provisions hereof shall in no way be construed to be a waiver of such provisions or options, and each Party, notwithstanding such failure, shall have the right thereafter to insist upon the specific performance of any and all of the provisions of this Agreement.

#### 18. Governing Law

This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Georgia, without regard to its conflict of laws principles.

### 19. Arm's Length Negotiations

This Agreement was executed after arm's length negotiations between the undersigned Parties and reflects the conclusion of the undersigned that this Agreement is in the best interests of all Parties.

#### 20. Notices

20.1 Every notice, consent, approval, or other communications required or contemplated by this Agreement shall be in writing and shall be delivered in person or given by postage prepaid mail, address to:

#### **BellSouth Telecommunications, Inc.**

CLEC Account Team 9<sup>th</sup> Floor 600 North 19<sup>th</sup> Street Birmingham, Alabama 35203

and

General Attorney - COU Suite 4300 675 W. Peachtree St. Atlanta, GA 30375

#### **NewSouth Communications, Corp.**

Senior Vice President of Network Planning & Provisioning NewSouth Center Two N. Main Street Greenville, SC 29601

and

Vice President of Regulatory Affairs NewSouth Center Two N. Main Street Greenville, SC 29601

or at such other address as the intended recipient previously shall have designated by written notice to the other Party.

- Where specifically required, notices shall be by certified or registered mail. Unless otherwise provided in this Agreement, notice by mail shall be effective on the date it is officially recorded as delivered by return receipt or equivalent, and in the absence of such record of delivery, it shall be presumed to have been delivered the fifth day, or next business day after the fifth day, after it was deposited in the mails.
- 20.3 BellSouth shall provide NewSouth notice via Internet posting of price changes and of changes to the terms and conditions of services available for resale.

#### 21. Rule of Construction

No rule of construction requiring interpretation against the drafting Party hereof shall apply in the interpretation of this Agreement.

#### 22. Headings of No Force or Effect

The headings of Articles and Sections of this Agreement are for convenience of reference only, and shall in no way define, modify or restrict the meaning or interpretation of the terms or provisions of this Agreement.

### 23. Multiple Counterparts

This Agreement may be executed multiple counterparts, each of which shall be deemed an original, but all of which shall together constitute but one and the same document.

#### 24. Implementation of Agreement

If NewSouth is a facilities based provider or a facilities based and resale provider, this section shall apply. Within 60 days of the execution of this Agreement, the Parties will adopt a schedule for the implementation of the Agreement. The schedule shall state with specificity time frames for submission of including but not limited to, network design, interconnection points, collocation arrangement requests, pre-sales testing and full operational time frames for the business and residential markets. An implementation template to be used for the implementation schedule is contained in Attachment 10 of this Agreement.

### 25. Filing of Agreement

- 25.1 Provided that NewSouth is certified as a CLEC in all applicable states, upon execution of this Agreement it shall be filed with the appropriate state regulatory agency pursuant to the requirements of Section 252 of the Act. If the regulatory agency imposes any filing or public interest notice fees regarding the filing or approval of the Agreement, NewSouth shall be responsible for publishing the required notice and the publication and/or notice costs shall be borne by NewSouth.
- For electronic filing purposes in the State of Louisiana, the CLEC Louisiana Certification Number is required and must be provided by NewSouth prior to execution of the Agreement. The CLEC Louisiana Certification Number for NewSouth is TSP00231.

# **26.** Changes In Subscriber Carrier Selection

- Both Parties hereto shall apply all of the principles set forth in 47 C.F.R. § 64.1100 to the process for End User selection of a primary Local Exchange Carrier. BellSouth shall not require a disconnect order from an NewSouth Customer or another LEC in order to process an NewSouth order for Resale Service for an NewSouth End User. Until the FCC or the Commission adopts final rules and procedures regarding a Customer's selection of a primary Local Exchange Carrier, unless already done so, NewSouth shall deliver to BellSouth a Blanket Representation of Authorization that applies to all orders submitted by NewSouth under this Agreement that require a primary Local Exchange Carrier change. Both Parties hereto shall retain on file all applicable documentation of authorization, including letters of authorization, relating to their End User's selection as its primary Local Exchange Carrier, which documentation shall be available for inspection by the other Party hereto upon reasonable request during normal business hours.
- If an End User denies authorizing a change in his or her primary Local Exchange Carrier selection to a different local exchange carrier ("Unauthorized Switching"), the Party receiving the End User complaint shall switch or caused to be switched that End User back to his preferred carrier in accordance with Applicable Law.

### 27. Additional Fair Competition Requirements

- In the event that either Party transfers facilities or other assets to an Affiliate which are necessary to comply with its obligations under this Agreement, the obligations hereunder shall survive and transfer to such Affiliate.
- BellSouth shall allow local exchange customers of NewSouth to select BellSouth for the provision of intraLATA toll services on a nondiscriminatory basis; provided, however, that prior to establishment of BellSouth as the intraLATA toll carrier for NewSouth local exchange customers, the Parties shall negotiate a billing and collections agreement on commercially reasonable terms whereby NewSouth shall bill the customer on BellSouth's behalf and shall collect from the customer and remit to BellSouth intraLATA toll revenues. NewSouth agrees to bill its customers on BellSouth's behalf for both presubscribed and "dial around" intraLATA toll traffic. The Parties shall exchange customer record data on a timely basis as necessary to bill such customers for intraLATA toll usage.
- 27.3 BellSouth shall not use information derived from providing services or facilities to NewSouth to create a lead or other information base for a "winback" sales program.

### 28. Operational Support Systems (OSS) Rates

BellSouth has developed and made available the following mechanized systems by which NewSouth may submit LSRs electronically.

LENS Local Exchange Navigation System

EDI Electronic Data Interchange

TAG Telecommunications Access Gateway

RoboTAG

or such other mechanical systems BellSouth may support for LSRs

LSRs submitted by means of one of these interactive interfaces will incur an OSS electronic ordering charge as specified in the Table below. An individual LSR will be identified for billing purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (mail, fax, courier, etc.) will incur a manual order charge as specified in the table below:

<b>OPERATIONAL</b>	<b>Electronic</b>	Manual
SUPPORT	Per LSR received from the	Per LSR received from
SYSTEMS (OSS)	CLEC by one of the OSS	the CLEC by means
RATES	interactive interfaces	other than one of the
		OSS interactive
		interfaces
OSS LSR Charge	\$3.50	\$19.99
USOC	SOMEC	SOMAN

Note: In addition to the OSS charges, applicable discounted service order and related discounted charges apply per the tariff.

#### 28.1 Denial/Restoral OSS Charge

In the event NewSouth provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and, therefore will be billed as one LSR per location.

#### 28.2 Cancellation OSS Charge

NewSouth will incur an OSS charge for an accepted LSR that is later canceled by NewSouth.

Note: Supplements or clarifications to a previously billed LSR will not incur another OSS charge.

#### 28.3 Threshold Billing Plan (Resale and Number Portability only)

The Parties agree that NewSouth will incur the mechanized rate for all LSRs, both mechanized and manual, if the percentage of mechanized LSRs to total LSRs meets or exceeds the threshold percentages shown below:

Year	Ratio: Mechanized/Total LSRs
2000	80%
2001	90%

The threshold plan will be discontinued in 2002.

BellSouth will track the total LSR volume for each CLEC for each quarter. At the end of that time period, a Percent Electronic LSR calculation will be made for that quarter based on the LSR data tracked in the LCSC. If this percentage exceeds the threshold volume, all of that CLEC's future manual LSRs will be billed at the mechanized LSR rate. To allow time for obtaining and analyzing the data and updating the billing system, this billing change will take place on the first day of the second month following the end of the quarter (e.g. May 1 for 1Q, Aug 1 for 2Q, etc.). There will be no adjustments to the amount billed for previously billed LSRs.

#### 28.4 Network Elements and Other Services Manual Additives

The Commissions in some states have ordered per-element manual additive non-recurring charges (NRC) for Network Elements and Other Services ordered by means other than one of the interactive interfaces. These ordered Network Elements and Other Services manual additive NRCs will apply in these states, rather than the charge per LSR. The per-element charges are listed on the Rate Tables in Attachment 2 of this agreement.

### 29. Entire Agreement

This Agreement and its Attachments, incorporated herein by this reference, sets forth the entire understanding and supersedes prior Agreements between the Parties relating to the subject matter contained herein and merges all prior discussions between them, and neither Party shall be bound by any definition, condition, provision, representation, warranty, covenant or promise other than as expressly stated in this Agreement or as is contemporaneously or subsequently set forth in writing and executed by a duly authorized officer or representative of the Party to be bound thereby.

# This Agreement may include attachments with provisions for the following services:

Network Elements and Other Services Local Interconnection Resale Collocation

The following services are included as options for purchase by NewSouth. NewSouth shall elect said services by written request to its Account Manager if applicable.

Optional Daily Usage File (ODUF) Enhanced Optional Daily Usage File (EODUF) Access Daily Usage File (ADUF) Line Information Database (LIDB) Storage Centralized Message Distribution Service (CMDS) Calling Name (CNAM) IN WITNESS WHEREOF, the Parties have executed this Agreement the day and year above first written.

BellSouth Telecommunications, Inc.	NewSouth Communications, Corp.	
Signature	Signature	
Greg Follensbee Name	Jake E. Jennings Name	
Senior Director Title	Vice President of Regulatory Affairs Title	
Date		

#### **Definitions**

**Affiliate** is defined as a person that (directly or indirectly) owns or controls, is owned or controlled by, or is under common ownership or control with, another person. For purposes of this paragraph, the term "own" means to own an equity interest (or equivalent thereof) of more than 10 percent.

**Centralized Message Distribution System** is the Telcordia (formerly BellCore) administered national system, based in Kansas City, Missouri, used to exchange Exchange Message Interface (EMI) formatted data among host companies.

**Commission** is defined as the appropriate regulatory agency in each of BellSouth's nine state region, Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee.

**Daily Usage File** is the compilation of messages or copies of messages in standard Exchange Message Interface (EMI) format exchanged from BellSouth to a CLEC.

**Exchange Message Interface** is the nationally administered standard format for the exchange of data among the Exchange Carriers within the telecommunications industry.

**Information Service** means the offering of a capability for generating, acquiring, storing, transforming, processing, retrieving, utilizing, or making available information via telecommunications, and includes electronic publishing, but does not include any use of any such capability for the management, control, or operation of a telecommunications system or the management of a telecommunications service.

**Intercompany Settlements (ICS)** is the revenue associated with charges billed by a company other than the company in whose service area such charges were incurred. ICS on a national level includes third number and credit card calls and is administered by Telcordia (formerly BellCore)'s Calling Card and Third Number Settlement System (CATS). Included is traffic that originates in one Regional Bell Operating Company's (RBOC) territory and bills in another RBOC's territory.

**Intermediary function** is defined as the delivery of traffic from NewSouth; a CLEC other than NewSouth or another telecommunications carrier through the network of BellSouth or NewSouth to an end user of NewSouth; a CLEC other than NewSouth or another telecommunications carrier.

**Local Interconnection** is defined as 1) the delivery of local traffic to be terminated on each Party's local network so that end users of either Party have the ability to reach end users of the other Party without the use of any access code or substantial delay in the processing of the call; 2) the LEC network features, functions, and capabilities set forth in this Agreement; and 3) Service Provider Number Portability sometimes referred to as temporary telephone number portability to be implemented pursuant to the terms of this Agreement.

**Local Traffic** is defined in Attachment 3.

**Message Distribution** is routing determination and subsequent delivery of message data from one company to another. Also included is the interface function with CMDS, where appropriate.

Multiple Exchange Carrier Access Billing ("MECAB") means the document prepared by the Billing Committee of the Ordering and Billing Forum ("OBF:), which functions under the auspices of the Carrier Liaison Committee of the Alliance for Telecommunications Industry Solutions ("ATIS") and by Telcordia (formerly BellCore) as Special Report SR-BDS-000983, Containing the recommended guidelines for the billing of Exchange Service access provided by two or more LECs and/or CLECs or by one LEC in two or more states within a single LATA.

**Network Element** is defined to mean a facility or equipment used in the provision of a telecommunications service. Such term may include, but is not limited to, features, functions, and capabilities that are provided by means of such facility or equipment, including but not limited to, subscriber numbers, databases, signaling systems, and information sufficient for billing and collection or used in the transmission, routing, or other provision of a telecommunications service. BellSouth offers access to the Network Elements, unbundled loops; network interface device; sub-loop elements; local switching; transport; tandem switching; operator systems; signaling; access to call-related databases; dark fiber as set forth in Attachment 2 of this Agreement.

**Non-Intercompany Settlement System (NICS)** is the Telcordia (formerly BellCore) system that calculates non-intercompany settlements amounts due from one company to another within the same RBOC region. It includes credit card, third number and collect messages.

Percent of Interstate Usage (PIU) is defined as a factor to be applied to terminating access services minutes of use to obtain those minutes that should be rated as interstate access services minutes of use. The numerator includes all interstate "non-intermediary" minutes of use, including interstate minutes of use that are forwarded due to service provider number portability less any interstate minutes of use for Terminating Party Pays services, such as 800 Services. The denominator includes all "non-intermediary", local, interstate, intrastate, toll and access minutes of use adjusted for service provider number portability less all minutes attributable to terminating Party pays services.

**Percent Local Usage (PLU)** is defined as a factor to be applied to intrastate terminating minutes of use. The numerator shall include all "non-intermediary" local minutes of use adjusted for those minutes of use that only apply local due to Service Provider Number Portability. The denominator is the total intrastate minutes of use including local, intrastate toll, and access, adjusted for Service Provider Number Portability less intrastate terminating Party pays minutes of use.

**Revenue Accounting Office (RAO) Status Company** is a local exchange company/alternate local exchange company that has been assigned a unique RAO code. Message data exchanged among RAO status companies is grouped (i.e. packed) according to From/To/Bill RAO combinations.

**Service Control Points** ("SCPs") are defined as databases that store information and have the ability to manipulate data required to offer particular services.

**Signal Transfer Points** ("STPs") are signaling message switches that interconnect Signaling Links to route signaling messages between switches and databases. STPs enable the exchange of Signaling System 7 ("SS7") messages between switching elements, database elements and STPs. STPs provide access to various BellSouth and third party network elements such as local switching and databases.

**Signaling links** are dedicated transmission paths carrying signaling messages between carrier switches and signaling networks. Signal Link Transport is a set of two or four dedicated 56 kbps transmission paths between NewSouth designated Signaling Points of Interconnection that provide a diverse transmission path and cross connect to a BellSouth Signal Transfer Point.

**Telecommunications** means the transmission, between or among points specified by the user, of information of the user's choosing, without change in the form or content of the information as sent and received.

**Telecommunications Service** means the offering of telecommunications for a fee directly to the public, or to such classes of users as to be effectively available directly to the public, regardless of the facilities used.

**Telecommunications Act of 1996** ("Act") means Public Law 104-104 of the United States Congress effective February 8, 1996. The Act amended the Communications Act of 1934 (47, U.S.C. Section 1 et. seq.).

Attachment 1 Page 1

# **Attachment 1**

Resale

# TABLE OF CONTENTS

1.	Discount Rates3
2.	Definition of Terms3
3.	General Provisions4
4.	Bellsouth's Provision of Services To NewSouth8
5.	Maintenance of Services9
6.	Establishment of Service
7.	Payment and Billing Arrangements12
8.	Discontinuance of Service14
9.	Line Information Database (LIDB)15
10.	RAO Hosting15
11.	Optional Daily Usage File (ODUF)16
12.	Enhanced Optional Daily Usage File (EODUF)16
Exhib	it A – Applicable Discounts/OSS Rates17
Exhib	it B – Resale Restrictions20
Exhib	it C – Line Information Database (LIDB) Storage Agreement22
Exhib	it D – CMDS/ROA Hosting28
Exhib	it E – Optional Daily Usage File ODUF)33
Exhib	it F – Enhanced Option Daily Usage File (EODUF)37
Exhib	it G – ODUF/EODUF/CMDS RatesRate Table

#### **RESALE**

#### 1. Discount Rates

The discount rates applied to NewSouth purchases of BellSouth Telecommunications Services for the purpose of resale shall be as set forth in Exhibit A. Such discount shall reflect the costs avoided by BellSouth when selling a service for wholesale purposes.

#### 2. Definition of Terms

- 2.1 COMPETITIVE LOCAL EXCHANGE COMPANY (CLEC) means a telephone company certificated by the public service commissions of BellSouth's franchised area to provide local exchange service within BellSouth's franchised area.
- 2.2 CUSTOMER OF RECORD means the entity responsible for placing application for service; requesting additions, rearrangements, maintenance or discontinuance of service; payment in full of charges incurred such as non-recurring, monthly recurring, toll, directory assistance, etc.
- 2.3 DEPOSIT means assurance provided by a customer in the form of cash, surety bond or bank letter of credit to be held by BellSouth.
- 2.4 END USER means the ultimate user of the telecommunications services.
- 2.5 END USER CUSTOMER LOCATION means the physical location of the premises where an end user makes use of the telecommunications services.
- 2.6 NEW SERVICES means functions, features or capabilities that are not currently offered by BellSouth. This includes packaging of existing services or combining a new function, feature or capability with an existing service.
- 2.7 RESALE means an activity wherein a certificated CLEC, such as NewSouth subscribes to the telecommunications services of BellSouth and then offers those telecommunications services to the public
- 2.8 RESALE SERVICE AREA means the area, as defined in a public service commission approved certificate of operation, within which a CLEC, such as NewSouth, may offer resold local exchange telecommunications service.

#### 3. General Provisions

- 3.1 NewSouth may resell the tariffed local exchange and toll telecommunications services of BellSouth contained in the General Subscriber Service Tariff and Private Line Service Tariff subject to the terms, and conditions specifically set forth herein. Notwithstanding the foregoing, the exclusions and limitations on services available for resale will be as set forth in Exhibit B, attached hereto and incorporated herein by this reference.
- 3.2 All of the negotiated rates, terms and conditions set forth in this Attachment pertain to the resale of BellSouth's retail telecommunications services and other services specified in this Attachment. BellSouth shall make available telecommunications services for resale at the discount rates set forth in Exhibit A to this Agreement and subject to the exclusions and limitations set forth in Exhibit B to this Agreement. BellSouth does not however waive its rights to appeal or otherwise challenge any decision regarding resale that resulted in the discount rates contained in Exhibit A or the exclusions and limitations contained in Exhibit B. BellSouth reserves the right to pursue any and all legal and/or equitable remedies, including appeals of any decisions. If such appeals or challenges result in changes in the discount rates or exclusions and limitations, the parties agree that appropriate modifications to this Agreement will be made promptly to make its terms consistent with the outcome of the appeal.
- 3.3 NewSouth may purchase resale services from BellSouth for their own use in operating their business. The resale discount will apply to those services under the following conditions:
- 3.3.1 NewSouth must resell services to other end users.
- 3.3.2 NewSouth must order services through resale interfaces, i.e., the Local Carrier Service Center (LCSC) and/or appropriate Resale Account Teams pursuant to Section 3 of the General Terms and Conditions.
- 3.3.3 NewSouth cannot be a competitive local exchange telecommunications company for the single purpose of selling to themselves.
- The provision of services by BellSouth to NewSouth does not constitute a joint undertaking for the furnishing of any service.
- 3.5 NewSouth will be the customer of record for all services purchased from BellSouth. Except as specified herein, BellSouth will take orders from, bill and expect payment from NewSouth for said services.

- 3.6 NewSouth will be BellSouth's single point of contact for all services purchased pursuant to this Agreement. BellSouth shall have no contact with the end user except to the extent provided for herein.
- 3.7 BellSouth will continue to bill the end user for any services that the end user specifies it wishes to receive directly from BellSouth.
- 3.8 BellSouth maintains the right to serve directly any end user within the service area of NewSouth. BellSouth will continue to directly market its own telecommunications products and services and in doing so may establish independent relationships with end users of NewSouth.
- 3.9 Neither Party shall interfere with the right of any person or entity to obtain service directly from the other Party.
- 3.10 Current telephone numbers may normally be retained by the end user and are assigned to the service furnished. However, neither Party nor the end user has a property right to the telephone number or any other call number designation associated with services furnished by BellSouth, and no right to the continuance of service through any particular central office. BellSouth reserves the right to change such numbers, or the central office designation associated with such numbers, or both, whenever BellSouth deems it necessary to do so in the conduct of its business and in accordance with BellSouth practices and procedures on a nondiscriminatory basis.
- 3.11 For the purpose of the resale of BellSouth's telecommunications services by NewSouth, BellSouth will provide NewSouth with an on line access to telephone numbers for reservation on a first come first serve basis. Such reservations of telephone numbers, on a pre-ordering basis shall be for a period of forty-five (45) days. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular Common Language Location Identifier Code (CLLIC) and in such instances BellSouth may request that NewSouth cancel its reservations of numbers. Any such request shall be made at parity and in a non-discriminatory manner. NewSouth shall comply with such request.
- Further, upon NewSouth's request, and for the purpose of the resale of BellSouth's telecommunications services by NewSouth, BellSouth will reserve up to 100 telephone numbers per CLLIC, for NewSouth's sole use. Such telephone number reservations shall be valid for forty-five (45) days from the reservation date. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth shall use its best efforts to reserve for a forty-five (45) day period a sufficient quantity of NewSouth's reasonable need in that particular CLLIC.

- 3.13 Service is furnished subject to the condition that it will not be used for any unlawful purpose.
- 3.14 Service will be discontinued if any law enforcement agency advises that the service being used is in violation of the law.
- 3.15 BellSouth can refuse service when it has grounds to believe that service will be used in violation of the law.
- 3.16 BellSouth accepts no responsibility to any person for any unlawful act committed by NewSouth or its end users as part of providing service to NewSouth for purposes of resale or otherwise.
- 3.17 BellSouth will cooperate fully with law enforcement agencies with subpoenas and court orders for assistance with BellSouth's end users, pursuant to Section 7 of the General Terms and Conditions
- 3.18 The characteristics and methods of operation of any circuits, facilities or equipment provided by any person or entity other than BellSouth shall not:
- 3.18.1 Interfere with or impair service over any facilities of BellSouth, its affiliates, or its connecting and concurring carriers involved in its service; or
- 3.18.2 Cause damage to BellSouth's plant;
- 3.18.3 Impair the privacy of any communications; or
- 3.18.4 Create hazards to any BellSouth employees or the public.
- 3.19 Facilities and/or equipment utilized by BellSouth to provide service to NewSouth remain the property of BellSouth.
- 3.20 White page directory listings will be provided in accordance with Section 5 of the General Terms and Conditions.
- 3.21 BellSouth provides electronic access to customer record information. Access is provided through the Local Exchange Navigation System (LENS), the Telecommunications Access Gateway (TAG), and RoboTAG. Customer Record Information includes but is not limited to, customer specific information in CRIS and RSAG. In addition, NewSouth shall provide to BellSouth access to customer record information including electronic access where available. Otherwise, upon request by BellSouth, NewSouth shall provide paper copies of customer record information within a reasonable period of time. Customer Record Information is equivalent to but not limited to the type of customer specific information contained in CRIS and RSAG.

The Parties agree not to view, copy, or otherwise obtain access to the customer record information of any customer without that customer's permission, and further agrees that NewSouth and BellSouth will obtain access to customer record information only in strict compliance with applicable laws, rules, or regulations of the State in which the service is provided.

- 3.22 All costs incurred by BellSouth to develop and implement operational interfaces shall be recovered from Resellers who utilize the services. Charges for use of Operational Support Systems (OSS) shall be as set forth in Exhibit A of this Attachment.
- 3.23 Where available to BellSouth's end users, BellSouth shall provide the following telecommunications services at a discount to allow for voice mail services:
  - Message Waiting Indicator ("MWI") stutter dialtone and message waiting light feature capabilities
  - Call Forward on Busy ("CF/B")
  - Call Forward Don't Answer ("CF/DA")

Further, BellSouth messaging services set forth in BellSouth's Messaging Service Information Package shall be made available for resale without the wholesale discount.

- 3.23.1 BellSouth shall provide branding for, or shall unbrand, voice mail services to NewSouth per the Bona Fide Request/New Business Request process as set forth in Attachment 12 of this Agreement.
- 3.24 BellSouth's Inside Wire Maintenance Service Plans may be made available for resale at rates, terms and conditions as set forth by BellSouth and without the wholesale discount.
- 3.25 If NewSouth requires a special assembly NewSouth agrees to pay the costs incurred by BellSouth for providing the requested special assembly. The costs will be provided to NewSouth prior to providing the service. Such costs could include both recurring and non-recurring charges and shall exclude any cost attributable to any marketing, billing collection or other costs that will be avoided by BellSouth in providing service to NewSouth.
- 3.26 Recovery of charges associated with implementing Number Portability through monthly charges assessed to end-users has been authorized by the FCC. This end user line charge will be billed to Resellers of BellSouth's telecommunications services and will be as filed in FCC No. 1. This charge is not discounted.
- 3.27 BellSouth shall provide 911/E911 for NewSouth customers in the same manner that it is provided to BellSouth customers. BellSouth shall provide and validate NewSouth

- customer information to the PSAP. BellSouth shall use its service order process to update and maintain, on the same schedule that it uses for its customers, the NewSouth customer service information in the ALI/DMS (Automatic Location Identification/Location Information) databases used to support 911/E911 services.
- 3.28 Pursuant to 47 CFR Section 51.617, BellSouth will bill NewSouth end users common line charges identical to the end user common line charges BellSouth bills its end users.
- CSAs shall be available for resale at the wholesale discount set forth in Exhibit A of this Attachment; provided, however, that in the event the Commission establishes a specific discount for CSAs such discount shall apply thereafter. NewSouth may resell a CSA to the end user for whom the CSA was constructed or to end users similarly situated to the specific end user for whom the CSA was constructed. Customers shall be deemed to be similarly situated when the quantity of use; time of use; manner of service; and costs of rendering the service are the same. In cases where NewSouth resells an existing CSA, no termination or rollover charges shall apply to the assignment of the CSA to NewSouth provided that NewSouth assumes the obligations set forth within the CSA. Notwithstanding the foregoing, BellSouth may impose a single service order charge (not to exceed the level of tariffed service order charges for comparable services) to recover the cost of changing the billing name on the account.
- 3.30 BellSouth shall notify NewSouth in advance of long term promotions (offered for longer than ninety (90) days) by posting a notice on its website.
- 3.31 New Resale Services; Changes in Provision of Resale Services. BellSouth shall use best efforts to provide NewSouth forty-five (45) days advance notice via Internet posting of changes to the prices, terms or conditions of services available for Resale. To the extent that revisions occur between the time BellSouth notifies NewSouth of changes under this Agreement and the time the changes are scheduled to be implemented, BellSouth will notify NewSouth of such revisions consistent with its internal notification process; provided that, NewSouth shall not utilize any notice given under this subsection to market resold offerings of that service in advance of BellSouth. In addition, upon request BellSouth shall furnish NewSouth with copies of publicly available service descriptions regarding the Resale Services. Notwithstanding the foregoing, NewSouth shall not utilize any such BellSouth service descriptions as part of its own sales or marketing efforts.

# 4. BellSouth's Provision of Services to NewSouth

4.1 NewSouth agrees that its resale of BellSouth services shall be as follows:

- 4.1.1 The resale of telecommunications services shall be limited to users and uses conforming to the class of service restrictions.
- 4.1.2 Hotel and Hospital PBX services are the only telecommunications services available for resale to Hotel/Motel and Hospital end users, respectively. Similarly, Access Line Service for Customer Provided Coin Telephones is the only local service available for resale to Independent Payphone Provider (IPP) customers. Shared Tenant Service customers can only be sold those local exchange access services available in BellSouth's A23 Shared Tenant Service Tariff in the states of Florida, Georgia, North Carolina and South Carolina, and in A27 in the states of Alabama, Kentucky, Louisiana, Mississippi and Tennessee.
- 4.1.3 BellSouth reserves the right to periodically audit services purchased by NewSouth to establish authenticity of use. Such audit shall not occur more than once in a calendar year. NewSouth shall make any and all records and data available to BellSouth or BellSouth's auditors on a reasonable basis. BellSouth shall bear the cost of said audit.
- 4.2 Resold services can only be used in the same manner as specified in BellSouth's Tariffs. Resold services are subject to the same terms and conditions as are specified for such services when furnished to an individual end user of BellSouth in the appropriate section of BellSouth's Tariffs. Specific tariff features (e.g. a usage allowance per month), shall not be aggregated across multiple resold services.
- 4.3 NewSouth may resell services only within the specific resale service area as defined in its certificate.
- 4.4 Telephone numbers transmitted via any resold service feature are intended solely for the use of the end user of the feature. Resale of this information is prohibited.

### 5. Maintenance of Services

- NewSouth will adopt and adhere to the standards contained in the applicable reasonable and non-discriminatory CLEC Work Center Operational Understanding Agreement regarding maintenance and installation of service.
- 5.2 Services resold pursuant to this Attachment and BellSouth's General Subscriber Service Tariff and Private Line Service Tariff and facilities and equipment provided by BellSouth shall be maintained by BellSouth.
- NewSouth or its end users may not rearrange, move, disconnect, remove or attempt to repair any facilities owned by BellSouth, other than by connection or disconnection to any interface means used, except with the written consent of BellSouth.

- NewSouth accepts responsibility to notify BellSouth of situations that arise that may result in a service problem.
- NewSouth will be BellSouth's single point of contact for all repair calls on behalf of NewSouth's end users. The parties agree to provide one another with toll-free contact numbers for such purposes.
- NewSouth will contact the appropriate repair centers in accordance with procedures established by BellSouth.
- 5.7 For all repair requests, NewSouth accepts responsibility for adhering to BellSouth's prescreening guidelines prior to referring the trouble to BellSouth.
- 5.8 BellSouth will bill NewSouth for handling troubles that are found not to be in BellSouth's network pursuant to its standard time and material charges. The standard time and material charges will be no more than what BellSouth charges to its retail customers for the same services.
- 5.9 BellSouth reserves the right to contact NewSouth's end users, if deemed necessary, for maintenance purposes.

### 6. Establishment of Service

- After receiving certification as a local exchange company from the appropriate regulatory agency, NewSouth will provide the appropriate BellSouth service center the necessary documentation to enable BellSouth to establish a master account for NewSouth's resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable. When necessary deposit requirements are met, BellSouth will begin taking orders for the resale of service.
- 6.2 Service orders will be in a standard format designated by BellSouth.
- 6.3 When notification is received from NewSouth that a current end user of BellSouth will subscribe to NewSouth's service, standard service order intervals for the appropriate class of service will apply.
- 6.4 BellSouth will not require end user confirmation prior to establishing service for NewSouth's end user customer. NewSouth must, however, be able to demonstrate end user authorization upon request.

- NewSouth will be the single point of contact with BellSouth for all subsequent ordering activity resulting in additions or changes to resold services except that BellSouth will accept a request directly from the end user for conversion of the end user's service from NewSouth to BellSouth or will accept a request from another CLEC for conversion of the end user's service from NewSouth to the other LEC. BellSouth will notify in writing NewSouth within five (5) business days that such a request has been processed.
- 6.6 If BellSouth determines that an unauthorized change in local service to NewSouth has occurred, BellSouth will reestablish service with the appropriate local service provider and will assess NewSouth as the CLEC initiating the unauthorized change, the unauthorized change charge described in F.C.C. Tariff No. 1, Section 13 or applicable state tariff. Appropriate nonrecurring charges, as set forth in Section A4 of the General Subscriber Service Tariff, will also be assessed to NewSouth. These charges can be adjusted if NewSouth provides satisfactory proof of authorization.
- 6.7 In order to safeguard its interest, BellSouth reserves the right to secure the account with a suitable form of security deposit, unless satisfactory credit has already been established.
- 6.7.1 Such security deposit shall take the form of an irrevocable Letter of Credit or other forms of security acceptable to BellSouth. Any such security deposit may be held during the continuance of the service as security for the payment of any and all amounts accruing for the service.
- 6.7.2 If a security deposit is required, such security deposit shall be made prior to the inauguration of service.
- 6.7.3 Such security deposit may not exceed two months' estimated billing.
- 6.7.4 The fact that a security deposit has been made in no way relieves NewSouth from complying with BellSouth's regulations as to advance payments and the prompt payment of bills on presentation nor does it constitute a waiver or modification of the regular practices of BellSouth providing for the discontinuance of service for non-payment of any sums due BellSouth.
- 6.7.5 BellSouth reserves the right to increase the security deposit requirements when, in its reasonable and non-discriminatory judgment, circumstances so warrant and/or gross monthly billing has increased beyond the level initially used to determine the security deposit.
- 6.7.6 In the event that NewSouth defaults on its account, service to NewSouth will be terminated and any security deposits held will be applied to its account.

- 6.7.7 Interest on a security deposit shall accrue and be paid in accordance with the terms in the appropriate BellSouth tariff.
- Orders to switch services "as is" shall be treated as a change of service and shall not be treated as a disconnection and subsequent reconnection of service.

### 7. Payment And Billing Arrangements

- 7.1 Prior to submitting orders to BellSouth for local service, a master account must be established for NewSouth. NewSouth is required to provide the following before a master account is established: proof of PSC/PUC certification, the Application for Master Account, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable.
- 7.2 BellSouth shall bill NewSouth on a current basis all applicable charges and credits.
- Payment of all charges will be the responsibility of NewSouth. NewSouth shall make payment to BellSouth for all services provided. BellSouth is not responsible for payments not received by NewSouth from NewSouth's end user. BellSouth will not become involved in billing disputes that may arise between NewSouth and its end user. Payments made to BellSouth as payment on account will be credited to an accounts receivable master account and not to an end user's account.
- 7.4 BellSouth will render bills each month on established bill days for each of NewSouth's accounts.
- 7.5 BellSouth will bill NewSouth in advance charges for all services to be provided during the ensuing billing period except charges associated with service usage, which will be billed in arrears. Charges will be calculated on an individual end user account level, including, if applicable, any charge for usage or usage allowances. BellSouth will also bill NewSouth, and NewSouth will be responsible for and remit to BellSouth, all charges applicable to resold services including but not limited to 911 and E911 charges, telecommunications relay charges (TRS), and franchise fees.
- 7.6 The payment will be due by the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by BellSouth.
- 7.6.1 If the payment due date falls on a Sunday or on a Holiday which is observed on a Monday, the payment due date shall be the first non-Holiday day following such Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If payment is not

- received by the payment due date, a late payment penalty, as set forth in section 7.8 following, shall apply.
- 7.6.2 If NewSouth requests multiple billing media or additional copies of bills, BellSouth will provide these at an appropriate charge to NewSouth.
- 7.6.3 Billing Disputes
- 7.6.3.1 Each Party agrees to notify the other Party upon the discovery of a billing dispute. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the Bill Date on which such disputed charges appear. Resolution of the dispute is expected to occur at the first level of management resulting in a recommendation for settlement of the dispute and closure of a specific billing period. If the issues are not resolved within the allotted time frame, the following resolution procedure will begin:
- 7.6.3.2 If the dispute is not resolved within sixty (60) days of the Bill Date, the dispute will be escalated to the second level of management for each of the respective Parties for resolution. If the dispute is not resolved within ninety (90) days of the Bill Date, the dispute will be escalated to the third level of management for each of the respective Parties for resolution
- 7.6.3.3 If the dispute is not resolved within one hundred and twenty (120) days of the Bill Date, the dispute will be escalated to the fourth level of management for each of the respective Parties for resolution.
- 7.6.3.4 If a Party disputes a charge and does not pay such charge by the payment due date, such charges shall be subject to late payment charges as set forth in the Late Payment Charges provision of this Attachment. If a Party disputes charges and the dispute is resolved in favor of such Party, the other Party shall credit the bill of the disputing Party for the amount of the disputed charges along with any late payment charges assessed no later than the second Bill Date after the resolution of the dispute. Accordingly, if a Party disputes charges and the dispute is resolved in favor of the other Party, the disputing Party shall pay the other Party the amount of the disputed charges and any associated late payment charges assessed no later than the second bill payment due date after the resolution of the dispute. BellSouth shall only assess interest on previously assessed late payment charges in a state where it has authority pursuant to its tariffs.
- 7.7 Upon proof of tax exempt certification from NewSouth, the total amount billed to NewSouth will not include any taxes due from the end user to reflect the tax exempt certification and local tax laws. NewSouth will be solely responsible for the

computation, tracking, reporting, and payment of taxes applicable to NewSouth's end user.

- 7.8 If any portion of the payment is received by BellSouth after the payment due date as set forth preceding, or if any portion of the payment is received by BellSouth in funds that are not immediately available to BellSouth, then a late payment penalty shall be due to BellSouth. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor and will be applied on a per bill basis. The late factor shall be as set forth in Section A2 of the General Subscriber Services Tariff and Section B2 of the Private Line Service Tariff. NewSouth will be charged a fee for all returned checks as set forth in Section to A2 of the General Subscriber Services Tariff or in applicable state law.
- 7.9 Any switched access charges associated with interexchange carrier access to the resold local exchange lines will be billed by, and due to, BellSouth. No additional charges are to be assessed to NewSouth.
- 7.10 BellSouth will not perform billing and collection services for NewSouth as a result of the execution of this Agreement. All requests for billing services should be referred to the appropriate entity or operational group within BellSouth.
- 7.11 In general, BellSouth will not become involved in disputes between NewSouth and NewSouth's end user customers over resold services. If a dispute does arise that cannot be settled without the involvement of BellSouth, NewSouth shall contact the designated Service Center for resolution. BellSouth will make every effort to assist in the resolution of the dispute and will work with NewSouth to resolve the matter in as timely a manner as possible. NewSouth may be required to submit documentation to substantiate the claim.

### 8. Discontinuance of Service

- 8.1 The procedures for discontinuing service to an end user are as follows:
- 8.1.1 Where possible, BellSouth will deny service to NewSouth's end user on behalf of, and at the request of, NewSouth. Upon restoration of the end user's service, restoral charges will apply and will be the responsibility of NewSouth.
- 8.1.2 At the request of NewSouth, BellSouth will disconnect a NewSouth end user customer.
- 8.1.3 All requests by NewSouth for denial or disconnection of an end user for nonpayment must be in writing.

- 8.1.4 NewSouth will be made solely responsible for notifying the end user of the proposed disconnection of the service.
- 8.1.5 BellSouth will continue to process calls made to the Annoyance Call Center and will advise NewSouth when it is determined that annoyance calls are originated from one of their end user's locations. BellSouth shall be indemnified, defended and held harmless by NewSouth and/or the end user against any claim, loss or damage arising from providing this information to NewSouth. It is the responsibility of NewSouth to take the corrective action necessary with its end users who make annoying calls. Failure to do so will result in BellSouth's disconnecting the end user's service.
- 8.1.6 BellSouth may disconnect and reuse facilities when the facility is in a denied state and BellSouth has received an order to establish new service or transfer of service from an end user or an end user's CLEC at the same address served by the denied facility.
- 8.2 The procedures for discontinuing service to NewSouth are as follows:
- 8.2.1 BellSouth reserves the right to suspend or terminate service in the event of prohibited, unlawful or improper use of the facilities or service, abuse of the facilities, or any other violation or noncompliance by NewSouth of the rules and regulations of BellSouth's Tariffs. BellSouth shall provide notice and an opportunity to cure, not to exceed five business days.
- 8.2.2 BellSouth reserves the right to suspend or terminate service for nonpayment if payment of account is not received by the bill day in the month after the original bill day, BellSouth may provide written notice to NewSouth, that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition BellSouth may, at the same time, give thirty days notice to the person designated by NewSouth to receive notices of noncompliance, and discontinue the provision of existing services to NewSouth at any time thereafter.
- 8.2.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
- 8.2.4 If BellSouth does not discontinue the provision of the services involved on the date specified in the thirty days notice and NewSouth's noncompliance continues, nothing contained herein shall preclude BellSouth's right to discontinue the provision of the services to NewSouth without further notice.
- 8.2.5 If payment is not received or arrangements made for payment by the date given in the written notification, NewSouth's services will be discontinued. Upon discontinuance of service on a NewSouth's account, service to NewSouth's end users will be denied. BellSouth will also reestablish service at the request of the end user or NewSouth

upon payment of the appropriate connection fee and subject to BellSouth's normal application procedures. NewSouth is solely responsible for notifying the end user of the proposed disconnection of the service.

8.2.6 If within fifteen days after an end user's service has been denied no contact has been made in reference to restoring service, the end user's service will be disconnected.

### 9. Line Information Database (LIDB)

- 9.1 BellSouth will store in its Line Information Database (LIDB) records relating to service only in the BellSouth region. The LIDB Storage Agreement is included in this Attachment as Exhibit C.
- 9.2 BellSouth will provide LIDB Storage upon written request to NewSouth Account Manager stating requested activation date.

### 10. RAO Hosting

- 10.1 The RAO Hosting Agreement is included in this Attachment as Exhibit D. Rates for BellSouth's Centralized Message Distribution System (CMDS) are as set forth in Exhibit G of this Attachment.
- BellSouth will provide RAO Hosting upon written request to its Account Manager stating requested activation date.

### 11. Optional Daily Usage File (ODUF)

- 11.1 The Optional Daily Usage File (ODUF) Agreement with terms and conditions is included in this Attachment as Exhibit E. Rates for ODUF are as set forth in Exhibit G of this Attachment.
- BellSouth will provide Optional Daily Usage File (ODUF) service upon written request to its Account Manager stating requested activation date.

### 12. Enhanced Optional Daily Usage File (EODUF)

- 12.1 The Enhanced Optional Daily Usage File (EODUF) service Agreement with terms and conditions is included in this Attachment as Exhibit F. Rates for EODUF are as set forth in Exhibit H of this Attachment.
- BellSouth will provide Enhanced Optional Daily Usage File (EODUF) service upon written request to its Account Manager stating requested activation date.

Attachment 1 Page 17

EXHIBIT A Page 1

### APPLICABLE DISCOUNTS

The telecommunications services available for purchase by NewSouth for the purposes of resale to NewSouth end users shall be available at the following discount off of the retail rate. If NewSouth cancels an order for telecommunications services for the purpose of resale, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with the applicable sections of the GSST and the PLST.

### **DISCOUNT\***

218000111											
<u>STATE</u>	RESIDENCE	BUSINESS	CSAs***								
ALABAMA	16.3%	16.3%									
FLORIDA	21.83%	16.81%									
GEORGIA	20.3%	17.3%									
KENTUCKY	16.79%	15.54%									
LOUISIANA	20.72%	20.72%	9.05%								
MISSISSIPPI	15.75%	15.75%									
NORTH CAROLINA	21.5%	17.6%									
SOUTH CAROLINA	14.8%	14.8%	8.98%								
TENNESSEE**	16%	16%									

- \* When a CLEC provides Resale service in a cross boundary area (areas that are part of the local serving area of another state's exchange) the rates, regulations and discounts for the tariffing state will apply. Billing will be from the serving state.
- \*\* In Tennessee, if a CLEC provides its own operator services and directory services, the discount shall be 21.56%. CLEC must provide written notification to BellSouth within 30 days prior to providing its own operator services and directory services to qualify for the higher discount rate of 21.56%.
- \*\*\* Unless noted in this column, the discount for Business will be the applicable discount rate for CSAs.

# Exclusions and Limitations On Services Available for Resale

Type of Service	AL		FL		GA		KY		LA		MS		NC		SC		,	ΓN
	Resale	Discount																
1 Grandfathered Services (Note 1)	Yes	Yes																
2 Contract Service Arrangements	Yes	Yes																
3 Promotions - > 90 Days(Note 2)	Yes	Yes	Yes	Note 3														
4 Promotions - < 90 Days (Note 2)	Yes	No																
5 Lifeline/Link Up Services	Yes	Yes	Yes	Yes	Yes	Yes	Note 4	Note 4	Yes	Yes								
6 911/E911 Services	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Note 7	Note 7	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
7 N11 Services	Yes	Yes	Yes	Yes	Yes	Yes	Note 8	Note 8	Note 8	Note 8	Yes	Yes	Yes	Yes	Note 8	Note 8	Yes	Yes
8 AdWatch <sup>SM</sup> Svc (See Note 6)	Yes	Yes																
9 MemoryCall® Service	Yes	No																
10 Mobile Services	Yes	No																
11 Federal Subscriber Line Charges	Yes	No																
12 Non-Recurring Charges	Yes	Yes	Yes	No														
13 End User Line Charge – Number Portability	Yes	No																
14 Public Telephone Access Service (PTAS)	Yes	Yes	Yes	No	Yes	Yes												

Attachment 1
Page 20
EXHIBIT B

# **Exclusions and Limitations On Services Available for Resale**

### **Applicable**

### **Notes:**

- 1. **Grandfathered services** can be resold only to existing subscribers of the grandfathered service.
- 2. Where available for resale, **promotions** will be made available only to end users who would have qualified for the promotion had it been provided by BellSouth directly.
- 3. In Tennessee, long-term **promotions** (offered for more than ninety (90) days) may be obtained at one of the following rates:
  - (a) the stated tariff rate, less the wholesale discount;
  - (b) the promotional rate (the promotional rate offered by BellSouth will not be discounted further by the wholesale discount rate)
- 4. **Lifeline/Link Up** services may be offered only to those subscribers who meet the criteria that BellSouth currently applies to subscribers of these services as set forth in Sections A3 and A4 of the BellSouth General Subscriber Services Tariff.
- 5. Some of BellSouth's local exchange and toll telecommunications services are not available in certain central offices and areas.
- 6. AdWatch<sup>SM</sup> Service is tariffed as BellSouth<sup>®</sup> AIN Virtual Number Call Detail Service.
- 7. In Louisiana 911/E911 services are not available to be resold. NewSouth's customers, however, will be able to contact the appropriate 911/E911 service provider when service is provided on a resold basis.
- 8. N11 services are not available to be resold, however, NewSouth's customers will be able to make calls to N11 providers.

# LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

### I. SCOPE

- A. This Agreement sets forth the terms and conditions pursuant to which BellSouth agrees to store in its LIDB certain information at the request of NewSouth and pursuant to which BellSouth, its LIDB customers and NewSouth shall have access to such information. NewSouth understands that BellSouth provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of NewSouth, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained in the attached Addendum(s) are hereby made a part of this Agreement as if fully incorporated herein.
- B. LIDB is accessed for the following purposes:
  - 1. Billed Number Screening
  - 2. Calling Card Validation
  - 3. Fraud Control
- C. BellSouth will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BellSouth's LIDB, provided that such information is included in the LIDB query. BellSouth will establish fraud alert thresholds and will notify NewSouth of fraud alerts so that NewSouth may take action it deems appropriate. NewSouth understands and agrees BellSouth will administer all data stored in the LIDB, including the data provided by NewSouth pursuant to this Agreement, in the same manner as BellSouth's data for BellSouth's end user customers. BellSouth shall not be responsible to NewSouth for any lost revenue which may result from BellSouth's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BellSouth in its sole discretion from time to time.

NewSouth understands that BellSouth currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearing houses. NewSouth further understands that these billing and collection customers of BellSouth query BellSouth's LIDB to determine whether to accept various billing options from end users. Additionally, NewSouth understands that presently BellSouth has no method to differentiate between BellSouth's own billing and line data in the LIDB and such data which it includes in the LIDB on NewSouth's behalf pursuant to this Agreement. Therefore, until such time as BellSouth can and does implement in its LIDB and its supporting systems the means to differentiate NewSouth's data from

BellSouth's data and the Parties to this Agreement execute appropriate amendments hereto, the following terms and conditions shall apply:

- (a) NewSouth agrees that it will accept responsibility for telecommunications services billed by BellSouth for its billing and collection customers for NewSouth's end user accounts which are resident in LIDB pursuant to this Agreement. NewSouth authorizes BellSouth to place such charges on NewSouth's bill from BellSouth and agrees that it shall pay all such charges. Charges for which NewSouth hereby takes responsibility include, but are not limited to, collect and third number calls.
- (b) Charges for such services shall appear on a separate BellSouth bill page identified with the name of the entity for which BellSouth is billing the charge.
- (c) NewSouth shall have the responsibility to render a billing statement to its end users for these charges, but NewSouth's obligation to pay BellSouth for the charges billed shall be independent of whether NewSouth is able or not to collect from NewSouth's end users.
- (d) BellSouth shall not become involved in any disputes between NewSouth and the entities for which BellSouth performs billing and collection. BellSouth will not issue adjustments for charges billed on behalf of an entity to NewSouth. It shall be the responsibility of NewSouth and the other entity to negotiate and arrange for any appropriate adjustments.

### II. TERM

This Agreement will be effective as of \_\_\_\_\_\_, and will continue in effect for one year, and thereafter may be continued until terminated by either Party upon thirty (30) days written notice to the other Party.

### III. FEES FOR SERVICE AND TAXES

- A. NewSouth will not be charged a fee for storage services provided by BellSouth to NewSouth, as described in Section I of this Agreement.
- B. Sales, use and all other taxes (excluding taxes on BellSouth's income) determined by BellSouth or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by NewSouth. NewSouth shall have the right to have BellSouth contest with the imposing jurisdiction, at NewSouth's expense, any such taxes that NewSouth deems are improperly levied.

### IV. INDEMNIFICATION

To the extent not prohibited by law, each Party will indemnify the other and hold the other harmless against any loss, cost, claim, injury, or liability relating to or arising out of negligence or willful misconduct by the indemnifying Party or its agents or contractors in connection with the indemnifying Party's provision of services, provided, however, that any indemnity for any loss, cost, claim, injury or liability arising out of or relating to errors or omissions in the provision of services under this Agreement shall be limited as otherwise specified in this Agreement. The indemnifying Party under this Section agrees to defend any suit brought against the other Party for any such loss, cost, claim, injury or liability. The indemnified Party agrees to notify the other Party promptly, in writing, of any written claims, lawsuits, or demands for which the other Party is responsible under this Section and to cooperate in every reasonable way to facilitate defense or settlement of claims. The indemnifying Party shall not be liable under this Section for settlement by the indemnified Party of any claim, lawsuit, or demand unless the defense of the claim, lawsuit, or demand has been tendered to it in writing and the indemnifying Party has unreasonably failed to assume such defense.

### V. LIMITATION OF LIABILITY

Neither Party shall be liable to the other Party for any lost profits or revenues or for any indirect, incidental or consequential damages incurred by the other Party arising from this Agreement or the services performed or not performed hereunder, regardless of the cause of such loss or damage.

### VI. MISCELLANEOUS

- A. It is understood and agreed to by the Parties that BellSouth may provide similar services to other companies.
- B. All terms, conditions and operations under this Agreement shall be performed in accordance with, and subject to, all applicable local, state or federal legal and regulatory tariffs, rulings, and other requirements of the federal courts, the U. S. Department of Justice and state and federal regulatory agencies. Nothing in this Agreement shall be construed to cause either Party to violate any such legal or regulatory requirement and either Party's obligation to perform shall be subject to all such requirements.
- C. NewSouth agrees to submit to BellSouth all advertising, sales promotion, press releases, and other publicity matters relating to this Agreement wherein BellSouth's corporate or trade names, logos, trademarks or service marks or those of BellSouth's affiliated companies are mentioned or language from which the connection of said names or trademarks therewith may be inferred or implied; and NewSouth further agrees not to publish or use advertising, sales promotions, press releases, or publicity matters without BellSouth's prior written approval.

- D. This Agreement constitutes the entire Agreement between NewSouth and BellSouth which supersedes all prior Agreements or contracts, oral or written representations, statements, negotiations, understandings, proposals and undertakings with respect to the subject matter hereof.
- E. Except as expressly provided in this Agreement, if any part of this Agreement is held or construed to be invalid or unenforceable, the validity of any other Section of this Agreement shall remain in full force and effect to the extent permissible or appropriate in furtherance of the intent of this Agreement.
- F. Neither Party shall be held liable for any delay or failure in performance of any part of this Agreement for any cause beyond its control and without its fault or negligence, such as acts of God, acts of civil or military authority, government regulations, embargoes, epidemics, war, terrorist acts, riots, insurrections, fires, explosions, earthquakes, nuclear accidents, floods, strikes, power blackouts, volcanic action, other major environmental disturbances, unusually severe weather conditions, inability to secure products or services of other persons or transportation facilities, or acts or omissions of transportation common carriers.
- G. This Agreement shall be deemed to be a contract made under the laws of the State of Georgia, and the construction, interpretation and performance of this Agreement and all transactions hereunder shall be governed by the domestic law of such State.

# RESALE ADDENDUM TO LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

Т	This is a Resale Addendum to the Line Information Data Base Storage Agreement dated, 2000, between BellSouth Telecommunications, Inc.
("BellS 2000.	outh"), and NewSouth ("NewSouth"), effective the day of,
I.	GENERAL
	This Addendum sets forth the terms and conditions for NewSouth's provision of billing number information to BellSouth for inclusion in BellSouth's LIDB. BellSouth will store in its LIDB the billing number information provided by NewSouth, and BellSouth will provide responses to on-line, call-by-call queries to this information for purposes specified in Section I.B. of the Agreement.
II.	DEFINITIONS
A.	Billing number - a number used by BellSouth for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.
B.	Line number - a ten-digit number assigned by BellSouth that identifies a telephone line associated with a resold local exchange service, or with a SPNP arrangement.
C.	Special billing number - a ten-digit number that identifies a billing account established by BellSouth in connection with a resold local exchange service or with a SPNP arrangement.
D.	Calling Card number - a billing number plus PIN number assigned by BellSouth.
E.	PIN number - a four digit security code assigned by BellSouth which is added to a billing number to compose a fourteen digit calling card number.
F.	Toll billing exception indicator - associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by the NewSouth.
G.	Billed Number Screening - refers to the activity of determining whether a toll billing

exception indicator is present for a particular billing number.

- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- I. Billing number information information about billing number or Calling Card number as assigned by BellSouth and toll billing exception indicator provided to BellSouth by the NewSouth.

### III. RESPONSIBILITIES OF PARTIES

- A. BellSouth will include billing number information associated with resold exchange lines or SPNP arrangements in its LIDB. The NewSouth will request any toll billing exceptions via the Local Service Request (LSR) form used to order resold exchange lines, or the SPNP service request form used to order SPNP arrangements.
- B. Under normal operating conditions, BellSouth shall include the billing number information in its LIDB upon completion of the service order establishing either the resold local exchange service or the SPNP arrangement, provided that BellSouth shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BellSouth's reasonable control. BellSouth will store in its LIDB an unlimited volume of the working telephone numbers associated with either the resold local exchange lines or the SPNP arrangements. For resold local exchange lines or for SPNP arrangements, BellSouth will issue line-based calling cards only in the name of NewSouth. BellSouth will not issue line-based calling cards in the name of NewSouth's individual end users. In the event that NewSouth wants to include calling card numbers assigned by the NewSouth in the BellSouth LIDB, a separate agreement is required.
- C. BellSouth will provide responses to on-line, call-by-call queries to the stored information for the specific purposes listed in the next paragraph.
- D. BellSouth is authorized to use the billing number information to perform the following functions for authorized users on an on-line basis:
- 1. Validate a 14 digit Calling Card number where the first 10 digits are a line number or special billing number assigned by BellSouth, and where the last four digits (PIN) are a security code assigned by BellSouth.
- 2. Determine whether the NewSouth has identified the billing number as one which should not be billed for collect or third number calls, or both.

### **RAO Hosting**

- 1. RAO Hosting, Calling Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to NewSouth by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this Agreement, including such revisions as may be made from time to time by BellSouth.
- 2. NewSouth shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- 3. Applicable compensation amounts will be billed by BellSouth to NewSouth on a monthly basis in arrears. Amounts due from one Party to the other (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- 4. NewSouth must have its own unique RAO code. Requests for establishment of RAO status where BellSouth is the selected Centralized Message Distribution System (CMDS) interfacing host, require written notification from NewSouth to the BellSouth RAO Hosting coordinator at least eight (8) weeks prior to the proposed effective date. The proposed effective date will be mutually agreed upon between the Parties with consideration given to time necessary for the completion of required Telcordia (formerly BellCore) functions. BellSouth will request the assignment of an RAO code from its connecting contractor, currently Telcordia (formerly BellCore), on behalf of NewSouth and will coordinate all associated conversion activities.
- 5. BellSouth will receive messages from NewSouth that are to be processed by BellSouth, another LEC or CLEC in the BellSouth region or a LEC outside the BellSouth region.
- 6. BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from NewSouth.
- 7. All data received from NewSouth that is to be processed or billed by another LEC or CLEC within the BellSouth region will be distributed to that LEC or CLEC in accordance with the agreement(s) which may be in effect between BellSouth and the involved LEC or CLEC.
- 8. All data received from NewSouth that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the

agreement(s) which may be in effect between BellSouth and its connecting contractor (currently Telcordia (formerly BellCore)).

- 9. BellSouth will receive messages from the CMDS network that are destined to be processed by NewSouth and will forward them to NewSouth on a daily basis.
- 10. Transmission of message data between BellSouth and NewSouth will be via CONNECT:Direct.
- 11. All messages and related data exchanged between BellSouth and NewSouth will be formatted in accordance with accepted industry standards for EMI formatted records and packed between appropriate EMI header and trailer records, also in accordance with accepted industry standards.
- 12. NewSouth will ensure that the recorded message detail necessary to recreate files provided to BellSouth will be maintained for back-up purposes for a period of three (3) calendar months beyond the related message dates.
- 13. Should it become necessary for NewSouth to send data to BellSouth more than sixty (60) days past the message date(s), NewSouth will notify BellSouth in advance of the transmission of the data. If there will be impacts outside the BellSouth region, BellSouth will work with its connecting contractor and NewSouth to notify all affected Parties.
- In the event that data to be exchanged between the two Parties should become lost or destroyed, both Parties will work together to determine the source of the problem. Once the cause of the problem has been jointly determined and the responsible Party (BellSouth or NewSouth) identified and agreed to, the company responsible for creating the data (BellSouth or NewSouth) will make every effort to have the affected data restored and retransmitted. If the data cannot be retrieved, the responsible Party will be liable to the other Party for any resulting lost revenue. Lost revenue may be a combination of revenues that could not be billed to the end users and associated access revenues. Both Parties will work together to estimate the revenue amount based upon historical data through a method mutually agreed upon. The resulting estimated revenue loss will be paid by the responsible Party to the other Party within three (3) calendar months of the date of problem resolution, or as mutually agreed upon by the Parties.
- 15. Should an error be detected by the EMI format edits performed by BellSouth on data received from NewSouth, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify NewSouth of the error condition.

NewSouth will correct the error(s) and will resend the entire pack to BellSouth for processing. In the event that an out-of-sequence condition occurs on subsequent packs, NewSouth will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth.

- 16. In association with message distribution service, BellSouth will provide NewSouth with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 17. In no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this agreement.
- 18. RAO Compensation
- 18.1 Rates for message distribution service provided by BellSouth for NewSouth are as set forth in Exhibit A to this Attachment.
- 18.2 Rates for data transmission associated with message distribution service are as set forth in Exhibit A to this Attachment.
- Data circuits (private line or dial-up) will be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties.
- All equipment, including modems and software, that is required on the NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.
- 19. Intercompany Settlements Messages
- This Section addresses the settlement of revenues associated with traffic originated from or billed by NewSouth as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one Bell operating territory and bills in another Bell operating territory is included.

Traffic that originates and bills within the same Bell operating territory will be settled on a local basis between NewSouth and the involved company(ies), unless that company is participating in NICS.

- 19.2 Both traffic that originates outside the BellSouth region by NewSouth and is billed within the BellSouth region, and traffic that originates within the BellSouth region and is billed outside the BellSouth region by NewSouth, is covered by this Agreement (CATS). Also covered is traffic that either is originated by or billed by NewSouth, involves a company other than NewSouth, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).
- 19.3 Once NewSouth is operating within the BellSouth territory, revenues associated with calls originated and billed within the BellSouth region will be settled via Telcordia (formerly BellCore)'s, its successor or assign, NICS system.
- 19.4 BellSouth will receive the monthly NICS reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- 19.5 BellSouth will receive the monthly Calling Card and Third Number Settlement System (CATS) reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- BellSouth will collect the revenue earned by NewSouth from the Bell operating company in whose territory the messages are billed (CATS), less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth to the Bell operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf on NewSouth. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.
- 19.7 BellSouth will collect the revenue earned by NewSouth within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit

Attachment 1
Page 31
EXHIBIT D

issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

BellSouth and NewSouth agree that monthly netted amounts of less than fifty dollars (\$50.00) will not be settled.

### **Optional Daily Usage File**

- 1. Upon written request from NewSouth, BellSouth will provide the Optional Daily Usage File (ODUF) service to NewSouth pursuant to the terms and conditions set forth in this section.
- 2. NewSouth shall furnish all relevant information required by BellSouth for the provision of the Optional Daily Usage File.
- 3. The Optional Daily Usage Feed will contain billable messages that were carried over the BellSouth Network and processed in the BellSouth Billing System, but billed to a NewSouth customer.

Charges for delivery of the Optional Daily Usage File will appear on NewSouths' monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- 4. The Optional Daily Usage Feed will contain both rated and unrated messages. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- 5. Messages that error in NewSouth's billing system will be the responsibility of NewSouth. If, however, NewSouth should encounter significant volumes of errored messages that prevent processing by NewSouth within its systems, BellSouth will work with the to determine the source of the errors and the appropriate resolution.
- 6. The following specifications shall apply to the Optional Daily Usage Feed.
- 6.1 Usage To Be Transmitted
- 6.1.1 The following messages recorded by BellSouth will be transmitted to NewSouth:
  - Message recording for per use/per activation type services (examples: Three Way Calling, Verify, Interrupt, Call Return, ETC.)
  - Measured billable Local
  - Directory Assistance messages
  - IntraLATA Toll
  - WATS & 800 Service
  - N11

- Information Service Provider Messages
- Operator Services Messages
- Operator Services Message Attempted Calls (UNE only)
- Credit/Cancel Records
- Usage for Voice Mail Message Service
- Rated Incollects (originated in BellSouth and from other companies) can also be on Optional Daily Usage File. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.
- 6.1.3 BellSouth will perform duplicate record checks on records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 6.1.4 In the event that NewSouth detects a duplicate on Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).
- 6.2 Physical File Characteristics
- 6.2.1 The Optional Daily Usage File will be distributed to NewSouth via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a variable block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use

of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including modems and software, that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

### 6.3 <u>Packing Specifications</u>

- 6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 6.3.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

### THE DATA WILL BE PACKED USING ATIS EMI RECORDS.

### 6.4 Pack Rejection

NewSouth will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. NewSouth will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to NewSouth by BellSouth.

### 6.5 Control Data

NewSouth will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate NewSouth received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by NewSouth for reasons stated in the above section.

### 6.6 Testing

6.6.1 Upon request from NewSouth, BellSouth shall send test files to NewSouth for the Optional Daily Usage File. The parties agree to review and discuss the file's content and/or format. For testing of usage results, BellSouth shall request that NewSouth set

Attachment 1
Page 36
EXHIBIT E

up a production (LIVE) file. The live test may consist of NewSouth's employees making test calls for the types of services NewSouth requests on the Optional Daily Usage File. These test calls are logged by NewSouth, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be completed within 30 calendar days from the date on which the initial test file was sent.

### **Enhanced Optional Daily Usage File**

- 1. Upon written request from NewSouth, BellSouth will provide the Enhanced Optional Daily Usage File (EODUF) service to NewSouth pursuant to the terms and conditions set forth in this section. EODUF will only be sent to existing ODUF subscribers who request the EODUF option.
- 2. The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Enhanced Optional Daily Usage File.
- 3. The Enhanced Optional Daily Usage File (EODUF) will provide usage data for local calls originating from resold Flat Rate Business and Residential Lines.
- 4. Charges for delivery of the Enhanced Optional Daily Usage File will appear on NewSouths' monthly bills. The charges are as set forth in Exhibit A to this Attachment.
- 5. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- 6. Messages that error in the billing system of NewSouth will be the responsibility of NewSouth. If, however, NewSouth should encounter significant volumes of errored messages that prevent processing by NewSouth within its systems, BellSouth will work with NewSouth to determine the source of the errors and the appropriate resolution.
- 7. The following specifications shall apply to the Optional Daily Usage Feed.
- 7.1 Usage To Be Transmitted
- 7.1.1 The following messages recorded by BellSouth will be transmitted to NewSouth:

Customer usage data for flat rated local call originating from NewSouth's end user lines (1FB or 1FR). The EODUF record for flat rate messages will include:

Date of Call
From Number
To Number
Connect Time
Conversation Time
Method of Recording
From RAO
Rate Class
Message Type

Billing Indicators Bill to Number

- 7.1.2 BellSouth will perform duplicate record checks on EODUF records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 7.1.3 In the event that NewSouth detects a duplicate on Enhanced Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).
- 7.2 <u>Physical File Characteristics</u>
- 7.2.1 The Enhanced Optional Daily Usage Feed will be distributed to NewSouth over their existing Optional Daily Usage File (ODUF) feed. The EODUF messages will be intermingled among NewSouth's Optional Daily Usage File (ODUF) messages. The EODUF will be a variable block format (2476) with an LRECL of 2472. The data on the EODUF will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays).
- 7.2.2 Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including modems and software, that is required on NewSouth's end for the purpose of data transmission will be the responsibility of NewSouth.

### 7.3 <u>Packing Specifications</u>

- 7.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 7.3.2 The Operating Company Number (OCN), From Revenue Accounting Office (RAO), and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth

Attachment 1 Page 39 EXHIBIT F

and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

THE DATA WILL BE PACKED USING ATIS EMI RECORDS.

## BELLSOUTH/NEWSOUTH RATES ODUF/EDOUF/CMDS

#### RATES BY STATE

DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
ODUF/EODUF/CMDS										
ODUF: Recording, per message	N/A	\$0.0002	\$0.008	\$0.008	\$0.0008611	\$0.00019	\$0.0001179	\$0.008	\$0.0002862	\$0.008
ODUF: Message Processing, per message	N/A	\$0.0033	\$0.004	\$0.004	\$0.0032357	\$0.0024	\$0.0032089	\$0.004	\$0.0032344	\$0.004
EODUF: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
CMDS: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
ODUF: Message Processing, per magnetic tape provisioned	N/A	\$55.19	\$54.95	\$54.95	\$55.68	\$47.30	\$54.62	\$54.95	\$54.72	\$54.95
EODUF: Message Processing, per magnetic tape provisioned	N/A	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30
ODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.00004	\$0.001	\$0.001	\$0.0000365	\$0.00003	\$0.0000354	\$0.001	\$0.0000357	\$0.001
EODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364
CMDS: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001
* Volume and term arrangements are also available.										

### NOTES:

If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the parties upon request by either party.

## **Attachment 2**

**Network Elements and Other Services** 

### **TABLE OF CONTENTS**

1.	INTRODUCTION	3
2.	UNBUNDLED LOOPS, INTEGRATED DIGITAL LOOP CARRIERS, NETWORK INTERFACES DEVICE, UNBUNDLED LOOP CONCENTRATION (ULC) SYSTEM, SUB LOOPS AND DARK FIBER	
3.	SWITCHING	. 22
4.	ENHANCED EXTENDED LINK (EEL)	. 32
5.	PORT/LOOP COMBINATIONS	. 38
6	TRANSPORT AND DARK FIBER	. 41
7	BELLSOUTH SWA 8XX TOLL FREE DIALING TEN DIGIT SCREENING SERVICE	. 47
8	LINE INFORMATION DATABASE (LIDB)	. 48
9	SIGNALING	. 50
10.	OPERATOR CALL PROCESSING, INWARD OPERATOR SERVICES AND DIRECTORY ASSISTANCE SERVICES	. 59
11.	CALLING NAME (CNAM) DATABASE SERVICE	. 65
12.	BASIC 911 AND E911	. 66
13.	TRUE-UP	. 68
LII	OB Storage AgreementExhibi	t
	$\mathbf{A}$	
CN	AM Database ServicesExhib	it
В		
Rat	tesExhi	bit
C		

### ACCESS TO NETWORK ELEMENTS AND OTHER SERVICES

### 1. Introduction

- 1.1. This Attachment sets forth the unbundled network elements and combinations of unbundled network elements that BellSouth agrees to offer to NewSouth in accordance with its obligations under Section 251(c)(3) of the Act. The specific terms and conditions that apply to the unbundled network elements are described below in this Attachment 2. The price for each unbundled network element and combination of unbundled Network Elements are set forth in Exhibit A of this Agreement. As an option, deaveraged rates, where available, are included in Exhibit A.
- 1.2. For purposes of this Agreement, "Network Element" is defined to mean a facility or equipment provided by BellSouth on an unbundled basis as is used by the CLEC in the provision of a telecommunications service. These unbundled network elements will be consistent with the requirements of the FCC 319 rule. For purposes of this Agreement, combinations of Network Elements shall be referred to as "Combinations."
- 1.2.1. Except as otherwise permitted by law, BellSouth shall not impose limitation restrictions or requirements or request for the use of the network elements or combinations that would impair the ability of NewSouth to offer telecommunications service in the manner NewSouth intends.
- 1.2.2. Except upon request by NewSouth, BellSouth shall not separate requested network elements that BellSouth currently combines.
- 1.3. BellSouth shall, upon request of NewSouth, and to the extent technically feasible, provide to NewSouth access to its network elements for the provision of NewSouth's telecommunications services. If no rate is identified in the contract, the rate for the specific service or function will be as ordered by the Commission. If the Commission has not ordered a rate then the rates will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.
- 1.4. NewSouth may purchase network elements and other services from BellSouth for the purpose of combining such network elements in any manner NewSouth chooses to provide telecommunication services to its intended users, including recreating existing BellSouth services. With the exception of the sub-loop elements, which are located outside of the central office, BellSouth shall deliver the network elements purchased by NewSouth for combining to the designated NewSouth collocation space or any other technically feasible point. The network elements shall be provided as set forth in this Attachment.

- 1.5. Subject to applicable and effective FCC Rules and Orders as well as effective State Commission Orders, BellSouth will offer combinations of network elements pursuant to such orders. In addition to the combinations set forth in Sections 4 and 5 BellSouth will provide the following combined network elements for purchase by NewSouth. The rate of the following combined network elements is the sum of the individual element prices as set forth in this Attachment. Except as specified below, Order Coordination as defined in Section 2 of Attachment 2 of this Agreement is available for each of these combinations:
  - SL1 Loop and cross connect
  - SL2 loop and cross connect
  - Port and cross connect
  - Port and cross connect and common (shared) transport
  - Port and vertical features
  - SL2 Loop with loop concentration
  - Port and common (shared) transport
  - SL1 Loop and LNP
  - SL2 Loop and LNP
- 1.6. NewSouth will adopt and adhere to the reasonable and non-discriminatory standards contained in the applicable CLEC Work Center Operational Understanding Agreement regarding maintenance and installation of service. Provided, however, nothing herein, shall override the Parties rights or obligations under this agreement.
- 1.7. Standards for Network Elements
- 1.7.1 BellSouth shall comply with the requirements set forth in the technical references, as well as any performance or other requirements identified in this Agreement, to the extent that they are consistent with the greater of BellSouth's actual performance or applicable industry standards.
- 1.7.2 If one or more of the requirements set forth in this Agreement are in conflict, the parties shall mutually agree on which requirement shall apply. If the parties cannot reach agreement, the dispute resolution process set forth in Section 12 of the General Terms and Conditions of this Agreement, incorporated herein by this reference, shall apply.
- 2. Unbundled Loops, Integrated Digital Loop Carriers, Network Interfaces Device, Unbundled Loop Concentration (ULC) System, Sub loops and Dark Fiber
- 2.1 Unbundled Loops

### 2.1.1 Definition

- 2.1.2 The local loop network element ("Loop(s)") is defined as a transmission facility between a distribution frame (or its equivalent) in BellSouth's central office and the loop demarcation point at an end-user customer premises, including inside wire owned by BellSouth. The local loop network element includes all features, functions, and capabilities of the transmission facilities, including dark fiber and attached electronics (except those used for the provision of advanced services, such as Digital Subscriber Line Access Multiplexers) and line conditioning. The loop shall include the use of all test access functionality, including without limitation, smart jacks, for both voice and data. NewSouth shall be entitled to order all loops set forth in Exhibit C of this Attachment. Unless otherwise requested, all loops will be provisioned with the appropriate Network Interface Device (NID).
- 2.1.3 The provisioning of service to NewSouth will require cross-office cabling and cross-connections within the central office to connect the loop to a local switch, or to other transmission equipment. in collocation space. These cross-connects are a separate element and are not considered a part of the loop.
- 2.1.4 BellSouth Order Coordination referenced in Attachment 2 includes two types: "Order Coordination" and "Order Coordination Time Specific."
- 2.1.5 "Order Coordination" refers to standard BellSouth service order coordination involving the reuse of facilities where NewSouth is requesting that their loop order be provisioned over an existing circuit that is currently providing service to the end-user. Order coordination for physical conversions will be scheduled at BellSouth's discretion during normal working hours on the committed due date and NewSouth will be advised. OC will be provided as a standard item on SL2 voice grade loops and all Unbundled Digital Loops (UDLs). OC will be provided as a chargeable option on SL1 voice grade loops, and all Unbundled Copper Loops.
- 2.1.5.1 For a coordinated conversion, i.e., stand alone INP, INP or LNP with loop, or stand alone loop where order coordination is provided for in this agreement, BellSouth shall verbally coordinate the disconnect with NewSouth and perform any switch translations so as to limit end user service outage. BellSouth and NewSouth will mutually agree upon a cut-over time 24 to 48 hours prior to the actual conversion. NewSouth may designate the conversion time when the conversion involves a loop with ILNP or LNP by ordering time specific conversion at rates designated in this agreement. For time specific conversions, BellSouth will verify the cut-over time designated by NewSouth 24 to 48 hours in advance to ensure that the conversion is to be completed as ordered. Both parties will use best efforts to ensure mutually agreed to conversion times, as identified in this paragraph, will commence within 15 minutes of the agreed time. For coordinated conversions, BellSouth's target intervals for service disruption to the enduser is 15 minutes or less for each loop.

2.1.6 "Order Coordination – Time Specific" refers to service order coordination in which NewSouth requests a specific time for a service order conversion to take place. BellSouth will make every effort to accommodate NewSouth's specific conversion time request. However, BellSouth reserves the right to negotiate with NewSouth a conversion time based on load and appointment control when necessary. Loops on a single service order of 14 or more loops will be provisioned on a project basis. This is a chargeable option for any coordinated order and is billed in addition to the OC charge. NewSouth may specify a time between 9:00 a.m. and 4:00 p.m. (location time) Monday through Friday (excluding holidays). If NewSouth specifies a time outside this window, or selects a time or quantity of loops that requires BellSouth technicians to work outside normal work hours (as specified in Section 1.2.1 of Attachment 6 to this Agreement), overtime charges will apply in addition to the OC and OC-TS charges. Overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked, and any special circumstances.

If NewSouth requests work to be done for any UNE loop that requires BellSouth technicians to work outside normal work hours, overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances

	Order Coordinatio n (OC)	Order Coordination – Time Specific (OC-TS)	<b>Test Points</b>	DLR	Charge for Dispatch and Testing if No Trouble Found
SL-1	Chargeable option	Chargeable Option*	Not available	Chargeable Option – ordered as Engineering Information Document	Charged for Dispatch inside & outside Central Office
SL-2	Included	Chargeable Option*	Included	Included	Charged for Dispatch outside Central Office
Unbundled Digital Loop	Included	Chargeable Option* (except on Universal Digital Channel)	Included (where appropriate)	Included	Charged for Dispatch outside Central Office
Unbundled Copper Loop	Chargeable Option	Not available	Included	Included	Charged for Dispatch outside Central Office

\*Order Coordination-Time Specific charge for orders due on same day at same location will be applied on a per LSR basis. For UVL-SL1, NewSouth must order OC when requesting OC-TS.

- 2.1.7 Where facilities are available, BellSouth's targeted installation interval, for up to 5 DS1 Loops on a single order, will be 5 business days. Such interval shall include BellSouth's return of the Firm Order Confirmation. For 6 to 14 DS1 Loops on a single order, BellSouth's targeted installation interval will be 7 business days. Such interval shall include BellSouth's return of the Firm Order Confirmation. For orders of 14 or more DS1 loops, the installation will be handled, in a reasonable and non-discriminatory manner, on a project basis and the intervals will be set by the BellSouth project manager for that order. All other Loops will be provisioned in accordance with the intervals set forth in BellSouth's Products and Services Interval Guide. Some loops require a Service Inquiry (SI) to determine if facilities are available prior to issuing the order. The interval for the SI process is separate from the installation interval. For expedite requests by NewSouth, expedite charges will apply for intervals less than 5 days. The charges outlined in Exhibit C, will apply. If NewSouth cancels an order for network elements and other services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with Attachment 6 of this Agreement.
- 2.1.8 If NewSouth modifies an order after being sent a Firm Order Confirmation (FOC) from BellSouth, any costs incurred by BellSouth to accommodate the modification will be reimbursed by NewSouth.
- 2.1.9 BellSouth will offer Unbundled Voice Loops (UVL) in two different service levels Service Level One (SL1) and Service Level Two (SL2).
- 2.1.10 SL1 loops are 2-wire loop start circuits, and will be non-designed and will not have test points. OC will be offered as a chargeable option on SLI loops when reuse of existing facilities has been requested by NewSouth. NewSouth may also order OC-TS when a specified conversion time is requested. OC-TS is a chargeable option for any coordinated order and is billed in addition to the OC charge. An Engineering Information (EI) document can be ordered as chargeable option. The EI document provides loop make up information which is similar to the information normally provided in a Design Layout Record. Upon issuance of a non-coordinated order in the service order system, SL1 loops will be activated on the due date in the same manner and time frames that BellSouth normally activates POTS-type loops for its customers. If NewSouth requests work to be done for SL1s that requires BellSouth technicians to work outside normal work hours, overtime charges will be applied according to actual costs based on type of force group required to perform the work, overtime hours worked and any special circumstances. Such overtime charges shall be as set forth in BellSouth's applicable Tariff.

- 2.1.11 SL2 loops are either 2-wire or 4-wire circuits, with test points, with or without conditioning, and will be designed with a design layout record provided to NewSouth. SL2 circuits can be provisioned with loop start, ground start or reverse battery signaling. OC is provided as a standard feature on SL2 loops. The OC feature will allow NewSouth to coordinate the installation of the loop with the disconnect of an existing customer's service and/or number portability service. In these cases, BellSouth will perform the order conversion with standard order coordination at the time designated by BellSouth during normal work hours.
- 2.1.12 BellSouth will also offer Unbundled Digital Loops (UDL). They will be designed, will be provisioned with test points (where appropriate), and will come standard with Order Coordination and a Design Layout Record (DLR).
- 2.1.13 As a chargeable option on all loops except UVL-SL1 and UCL, BellSouth will offer Order Coordination Time Specific (OC-TS), as described in Section 2.1.6 of this Attachment. This will allow NewSouth the ability to specify the time that the coordinated conversion takes place. The OC-TS charge for orders due on the same day at the same location will be applied on a per Local Service Request (LSR) basis.
- 2.1.14 NewSouth will be responsible for testing and isolating troubles on the loops. Once NewSouth has isolated a trouble to the BellSouth provided loop, NewSouth will issue a trouble to BellSouth on the loop. BellSouth will take the actions necessary to repair the loop if a trouble actually exists. BellSouth will repair these loops in the same time frames that BellSouth repairs similarly situated loops to its customers.
- 2.1.15 If NewSouth reports a trouble on SL1 loops and no trouble actually exists, BellSouth will charge NewSouth for any dispatching and testing (both inside and outside the CO) required by BellSouth in order to confirm the loop's working status.
- 2.1.16 If NewSouth reports a trouble on SL2 loops and no trouble actually exists, BellSouth will charge NewSouth for any dispatching and testing (outside the CO) required by BellSouth in order to confirm the loop's working status. There will be no charges in the event BellSouth fails to isolate the trouble in the initial dispatch, but isolates the trouble on a subsequent dispatch.
- 2.1.17 In addition to the UVLs and UDLs, BellSouth shall make available an Unbundled Copper Loop (UCL). The UCL will be a copper twisted pair loop that is unencumbered by any intervening equipment (e.g., filters, load coils, range extenders, digital loop carrier, or repeaters). The UCL will be offered in two versions Short and Long. A short UCL (18 kft or less) will be provisioned according to Resistance Design parameters. The long UCL (beyond 18kft) will be used when a CLEC wants to condition copper loops longer than 18kft by removing load coils and other intervening equipment. BST will only ensure electrical continuity and balance relative to tip and ring on UCLs.

- 2.1.18 The UCL will be a designed circuit, with or without conditioning, provisioned with a test point and come standard with a DLR. OC will be offered as a chargeable option on all UCL loops. Order Coordination Time Specific (OC-TS) will not be offered on UCLs.
- 2.1.19 The UCL is a dry cooper loop and is not intended to support any particular telecommunications service. NewSouth may use the UCL loop for a variety of services, including xDSL (e.g., ADSL and HDSL) services, by attaching appropriate terminal equipment of NewSouth's choosing. NewSouth will determine the type of service that will be provided over the loop.
- 2.1.20 Because the UCL loop is an unbundled loop offering that is separate and distinct from BellSouth's ADSL and HDSL capable loop offerings, NewSouth agrees that BellSouth's UCL loop will not be held to the service level and performance expectations that apply to its ADSL and HDSL unbundled loop offerings. BellSouth shall only be obligated to maintain copper continuity and provide balance relative to tip and ring on UCL loops.
- 2.1.21 The UCL loop shall be provided to CLEC in accordance with BellSouth's Technical Reference 73600, or other industry guidelines.
- 2.1.22 <u>Technical Requirements</u>
- 2.1.22.1 To the extent available within BellSouth's Network at a particular location, BellSouth will offer loops capable of supporting telecommunications services such as: POTS, Centrex, basic rate ISDN, analog PBX, voice grade private line, ADSL, HDSL, DS1 and digital data (up to 64 kb/s). If a requested loop type is not available, then the CLEC can use the Special Construction process to request that BellSouth place facilities or otherwise modify facilities in order to meet NewSouth's request.
- 2.1.22.2 NewSouth will be responsible for providing BellSouth with a Service Profile Identifier (SPID) associated with a particular ISDN-capable loop and end user. With the SPID, BellSouth will be able to adequately test the circuit and ensure that it properly supports ISDN service.
- 2.1.22.3 The loop will support the transmission, signaling, performance and interface requirements of the services described in 2.1.3 above. It is recognized that the requirements of different services are different, and that a number of types or grades of loops are required to support these services.
- 2.1.22.4 NewSouth may utilize the unbundled loops to provide any telecommunication service it wishes. However, BellSouth will only provision, maintain and repair the loops to the standards that are consistent with the type of loop ordered. For example, if NewSouth orders an ISDN-capable loop but wants to use the loop for a service other than ISDN, BellSouth will only support that the loop is capable of providing ISDN service. For non-service specific loops (e.g. UCL, loops modified by NewSouth using

- the Special Construction process), BellSouth will only support that the loop has copper continuity and balanced tip-and-ring.
- 2.1.22.5 In those cases where NewSouth has requested that BellSouth modify a loop so that it no longer meets the technical parameters of the original loop type (e.g., voice grade, ISDN, ADSL, etc.) the resulting modified loop will be ordered and maintained as a UCL.
- 2.1.22.9 The loop shall be provided to NewSouth in accordance with BellSouth's TR73600 Unbundled Local Loop Technical Specification and applicable industry standard technical references.
- 2.1.23 Universal Digital Channel (UDC) Loop
- 2.1.23.1 Due to technical limitations associated with certain DLC systems, some ISDN-capable loops that are provisioned using DLC systems may not support IDSL service. Effective with this agreement, BellSouth will no longer reconfigure its ISDN-capable loop to support IDSL service.
- 2.1.23.2 Instead, BellSouth agrees to offer the Universal Digital Channel (UDC) loop as a part of their Unbundled Digital Loop offerings. The UDC loop is intended to be compatible with IDSL service and has the same physical characteristics and transmission specifications as BellSouth's ISDN-capable loop. These specifications are listed in BellSouth's TR73600.
- 2.1.23.3 Like the ISDN-capable loop, the UDC loop may be provisioned on copper or through a DLC system. However, when UDC loops are provisioned using a DLC system, BellSouth will ensure that they are only provisioned on time slots that are compatible with data-only services such as IDSL.

#### 2.1.24. <u>Testing</u>

2.1.24.1 BellSouth will perform the appropriate pre-service tests to ensure NewSouth dial tone is delivered to the appropriate connecting point. The timing of the test is based on the overall interval and type of the service being provisioned. Under normal intervals, testing for designed services are normally completed 24 hours in advance of the conversion. For non-designed services, dial tone is verified in time frames consistent with the same time frames that BellSouth uses to activate POTS services for it's own end users. In any event, BellSouth will advise NewSouth whenever connectivity cannot be verified with NewSouth and will work cooperatively with NewSouth to correct the problem. BellSouth will advise NewSouth at completion of the conversion or turn up of new services in order for NewSouth to accept, reject or open a maintenance ticket to BellSouth on the services being provisioned. BellSouth will work cooperatively with NewSouth to ensure end user service outage is minimal.

- 2.1.24.2 Where a field visit is required to provision the loop, BellSouth will test the loop ordered by NewSouth to the NID. Testing requested by NewSouth to points beyond the NID will be billed a time and material charge at the same increments BellSouth charges it's own end users. Requests for field-testing where a dispatch is not required may be made by NewSouth and where mutually agreed to, BellSouth will dispatch to perform additional field testing at rates billed on a time and material basis as mentioned in the previous paragraph.
- 2.1.24.3 BellSouth will place a tag on all unbundled loops that require a technician to be dispatched to the end user's premises during the provisioning process. The loop tag will include the CLEC's name and the circuit ID number. Otherwise, the loop will be tagged by BellSouth during the next scheduled maintenance or repair visit to the customer's location for that loop; or the loop may be tagged by the CLEC during their dispatch to that customer's location.
- 2.1.24.4 Cut-over intervals for ILNP, ILNP with loop and LNP with loop will be at parity with the intervals experienced by BellSouth end users, BellSouth itself or any other NewSouth as indicated in the results of the Service Quality Measurements published by BellSouth. In any event, BellSouth will use best efforts to convert each loop within fifteen (15) minutes.
- 2.1.24.5 BellSouth and NewSouth will jointly develop additional processes or procedures as the need arises to improve service delivery during the life of the agreement.

### 2.2 Unbundled Loop Modifications (ULM) / Loop Conditioning

- 2.2.1 Subject to applicable and effective FCC rules and orders, BellSouth shall condition loops, as requested by NewSouth, whether or not BellSouth offers advanced services to the End User on that loop.
- 2.2.2 Loop conditioning is defined as the removal from the loop of any devices that may diminish the capability of the loop to deliver high-speed switched wireline telecommunications capability, including xDSL service. Such devices include, but are not limited to, bridge taps, load coils, repeaters, low pass filters, and range extenders.
- 2.2.3 BellSouth shall recover the cost of line conditioning requested by NewSouth through a recurring charge and/or nonrecurring charge(s) in accordance with the FCC's forward-looking pricing principles promulgated pursuant to section 252 (d) (1) of the Act and in compliance with FCC Rule 52.507 (e).

### 2.3. Integrated Digital Loop Carriers

2.3.1 Where BellSouth uses Integrated Digital Loop Carrier (IDLC) systems to provide the local loop and BellSouth has a suitable alternate facility available, BellSouth will make

arrangements to permit NewSouth to order a contiguous local loop. To the extent it is technically feasible, these arrangements will provide NewSouth with the capability to serve end users at a level that is at parity with the level of service BellSouth provides its customers. If no alternate facility is available, BellSouth will utilize its Special Construction (SC) process to determine the additional costs required to provision the loop facilities. NewSouth will then have the option of paying the one-time SC rates to place the loop facilities or NewSouth may chose some other method of providing service to the end-user (e.g., Resale, private facilities, etc.).

#### 2.4 Network Interface Device

#### 2.4.1 Definition

The NID is defined as any means of interconnection of end-user customer premises wiring to BellSouth's distribution plant, such as a cross-connect device used for that purpose. The NID is a single-line termination device or that portion of a multiple-line termination device required to terminate a single line or circuit at the premises. The NID features two independent chambers or divisions that separate the service provider's network from the End User's on-premises wiring. Each chamber or division contains the appropriate connection points or posts to which the service provider and the End User each make their connections. The NID provides a protective ground connection and is capable of terminating cables such as twisted pair cable.

- 2.4.2. BellSouth shall permit NewSouth to connect NewSouth's loop facilities to onpremises wiring through the BellSouth NID or at any other technically feasible point.
- 2.4.3 Access to Network Interface Device (NID)
- 2.4.3.1. Due to the wide variety of NIDs utilized by BellSouth (based on subscriber size and environmental considerations), NewSouth may access the on-premises wiring by any of the following means: BellSouth shall allow NewSouth to connect its loops directly to BellSouth's multi-line residential NID enclosures that have additional space and are not used by BellSouth or any other telecommunications carriers to provide service to the premise. NewSouth agrees to install compatible protectors and test jacks and to maintain the protection system and equipment and to indemnify BellSouth pursuant to Section 8 of the General Terms and Conditions of this Agreement.
- 2.4.3.2. Where an adequate length of on-premises wiring is present and environmental conditions permit, either Party may remove the on-premises wiring from the other Party's NID and connect that wire to that Party's own NID; or
- 2.4.3.3. Enter the subscriber access chamber or "side" of "dual chamber" NID enclosures for the purpose of extending a connecterized or spliced jumper wire from the on-premises wiring through a suitable "punch-out" hole of such NID enclosures; or

- 2.4.3.4. Request BellSouth to make other rearrangements to the on-premises wiring terminations or terminal enclosure on a time and materials cost basis to be charged to the requesting Party (i.e., NewSouth, its agent, the building owner or the subscriber). Such charges will be billed to the requesting Party.
- 2.4.3.5. In no case shall either Party remove or disconnect the other Party's loop facilities from either Party's NIDs, enclosures, or protectors, without state regulatory requirement, without providing prior notice to the other Party, and without appropriately capping off and guarding the other Party's loop. In such cases, it shall be the responsibility of the disconnecting party to properly ground the other party's loop, maintain the NID, and assume full liability for its action and any adverse consequences.
- 2.4.3.6. In no case shall either Party remove or disconnect ground wires from BellSouth's NIDs, enclosures, or protectors.
- 2.4.3.7. In no case shall either Party remove or disconnect NID modules, protectors, or terminals from BellSouth's NID enclosures.
- 2.4.3.8. Due to the wide variety of NID enclosures and outside plant environments BellSouth will work with NewSouth to develop specific procedures to establish the most effective means of implementing this Section 2.4.3.

# 2.4.4 Technical Requirements

- 2.4.4.1 The NID shall provide an accessible point of interconnection and shall maintain a connection to ground.
- 2.4.4.2 The NID shall be capable of transferring electrical analog or digital signals between the subscriber's inside wiring and the Distribution Media and/or cross connect to NewSouth's NID, consistent with the NID's function at the Effective Date of this Agreement.
- 2.4.4.3 When NewSouth orders a NID at a particular location, NewSouth will get the NID that is currently located at such location. If NewSouth requires additional work or modification to the NID then BellSouth shall do such additional work or modification to the NID in accordance with Section 2.4.3.
- 2.4.4.4 When NewSouth deploys its own local loops with respect to multiple-line termination devices, NewSouth shall specify the quantity of NIDs connections that it requires within such device.
- 2.4.5 Interface Requirements
- 2.4.5.1 The NID shall be equal to or better than all of the requirements for NIDs set forth in the applicable industry standard technical references.

# 2.5 Unbundled Loop Concentration (ULC) System

- 2.5.1 BellSouth will provide to NewSouth Unbundled Loop Concentration (ULC). Loop concentration systems in the central office concentrate the signals transmitted over local loops onto a digital loop carrier system. The concentration device is placed inside a BellSouth central office. BellSouth will offer ULC with a TR008 interface or a TR303 interface.
- 2.5.2 ULC will be offered in two sizes. System A will allow up to 96 BellSouth loops to be concentrated onto multiple DS1s. The high-speed connection from the concentrator will be at the electrical DS1 level and may connect to NewSouth at NewSouth's collocation site. System B will allow up to 192 BellSouth loops to be concentrated onto multiple DS1s. System A may be upgraded to a System B. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). All DS1 interfaces will terminate to the CLEC's collocation space. ULC service is offered with or without concentration and with or without protection. A Line Interface element will be required for each loop that is terminated onto the ULC system. Rates for ULC are as set forth in this Attachment.

## 2.6 Sub-loop Elements

- 2.6.1 Where facilities permit and subject to applicable and effective FCC rules and orders, BellSouth shall offer access to its Unbundled Sub Loop (USL), Unbundled Subloop Concentration (USLC) System and Unbundled Network Terminating Wire (UNTW) elements. BellSouth shall provide non-discriminatory access on a unbundled basis to the subloop, in accordance with 47 C.F.R. 51.311 and section 251(c) (3) of the Act, and pursuant to the following terms and conditions and the rates approved by the Commission and set forth in this Attachment.
- 2.6.2 Subloop components include but are not limited to the following:
- 2.6.2.1 Unbundled Sub-Loop Distribution;
- 2.6.2.2 Unbundled Sub-Loop Concentration/Multiplexing Functionality; and
- 2.6.2.3 Unbundled Network Terminating Wire; and
- 2.6.2.4 Unbundled Sub-Loop Feeder.

# 2.6.3 Unbundled Sub-Loop (distribution facilities)

#### 2.6.3.1 Definition

- 2.6.3.2 Subject to applicable and effective FCC rules and orders, the unbundled sub-loop distribution facility is dedicated transmission facility that BellSouth provides from a customer's point of demarcation to a BellSouth cross-connect device. The BellSouth cross-connect device may be located within a remote terminal (RT) or a stand-alone cross-box in the field or in the equipment room of a building. There are two offerings available for Unbundled Sub-Loops (USL):
- 2.6.3.3 Unbundled Sub-Loop Distribution (USL-D) will include the sub-loop facility from the cross-box in the field up to and including the point of demarcation.
- 2.6.3.4 BellSouth will also provide sub-loop interconnection to the intrabuilding network cable (INC) (riser cable). INC is the distribution facility inside a subscriber's building or between buildings on one customer's same premises (continuous property not separated by a public street or road). USL-INC (riser cable) will include the facility from the cross-connect device in the building equipment room up to and including the point of demarcation.
- 2.6.4. Requirements for Unbundled Sub-Loop Distribution Facilities
- 2.6.4.1 Unbundled Sub-Loop distribution facilities were originally built as part of the entire voice grade loop from the BellSouth central office to the customer network interface. Therefore, the Unbundled Sub-Loop may have load coils, which are necessary for transmission of voice grade services. The Unbundled Sub-Loops will be provided in accordance with technical reference TR73600. However, NewSouth may request such sub-loop be conditioned in accordance with Section 2.2 of this Attachment.
- 2.6.4.2 Unbundled Sub-Loop distribution facilities shall support functions associated with provisioning, maintenance and testing of the Unbundled Sub-Loop. In a scenario that involves connection at a BellSouth cross-box located in the field, NewSouth would be required to deliver a cable to the BellSouth remote terminal or cross-box to provide continuity to NewSouth's feeder facilities. This cable would be connected, by a BellSouth technician, to a cross-connect panel within the BellSouth RT/cross-box. NewSouth's cable pairs can then be connected to BellSouth's USL within the BellSouth cross-box by the BellSouth technician. In a scenario that requires connection in a building equipment room, BellSouth will install a cross connect panel on which access to the requested sub-loops will be connected. The CLEC's cable pairs can then be connected to the Unbundled Sub-Loop pairs on this cross-connect panel by the BellSouth technician.
- 2.6.4.3 BellSouth will provide Unbundled Sub-Loops where technically feasible. Through the firm order Service Inquiry (SI) process, BellSouth will determine if it is feasible to place the required facilities where NewSouth has requested access to Unbundled Sub-Loops. If existing capacity is sufficient to meet the CLEC demand, then BellSouth will perform the set-up work as described in the next section 2.6.4.4. If any work must be done to modify existing BellSouth facilities or add new facilities (other than

adding the cross-connect panel in a building equipment room as noted in 2.6.4.4) to accommodate NewSouth's request for Unbundled Sub-Loops, BellSouth will use its Special Construction (SC) process to determine the additional costs required to provision the Unbundled Sub-Loops. NewSouth will then have the option of paying the one-time SC charge to modify the facilities to meet NewSouth's request.

- 2.6.4.4 During the initial set-up in a BellSouth cross-connect box in the field, the BellSouth technician will perform the necessary work to splice the CLEC's cable into the cross-connect box. For the set-up inside a building equipment room, BellSouth will perform the necessary work to install the cross-connect panel that will be used to provide access to the requested USLs. Once the set-up is complete, the CLEC requested sub-loop pairs would be provisioned through the service order process based on the submission of a LSR to the LCSC.
- 2.6.5 <u>Interface Requirements</u>
- 2.6.5.1 Unbundled Sub-Loop shall be equal to or better than each of the applicable requirements set forth in the applicable industry standard technical references.
- 2.6.6 **Unbundled Sub-Loop Concentration System (USLC)**
- 2.6.6.1 Where facilities permit, BellSouth will provide to NewSouth with the ability to concentrate its sub-loops onto multiple DS1s back to the BellSouth Central Office. The DS1s will then be terminated into NewSouth's collocation space. TR-008 and TR303 interface standards are available.
- USLC, using the Lucent Series 5 equipment, will be offered in two different systems. System A will allow up to 96 of NewSouth's sub-loops to be concentrated onto multiple DS1s. System B will allow an additional 96 of NewSouth's sub-loops to be concentrated onto multiple DS1s. One System A may be supplemented with one System B and they both must be physically located in a single Series 5 dual channel bank. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). The DS1 level facility that connects the RT site with the serving wire center is known as a Feeder Interface. All DS1 Feeder Interfaces will terminate to the CLEC's collocation space within the SWC that serves the RT where the CLEC's sub-loops are connected. USLC service is offered with or without concentration and with or without a protection DS1.
- 2.6.6.3 In these scenarios NewSouth would be required to place a cross-box, remote terminal (RT), or other similar device and deliver a cable to the BellSouth remote terminal. This cable would be connected, by a BellSouth technician, to a cross-connect panel within the BellSouth RT/cross-box and would allow NewSouth's sub-loops to then be placed on the ULSC and transported to their collocation space at a DS1 level.

# 2.6.7 Unbundled Network Terminating Wire (UNTW)

2.6.7.1 BellSouth agrees to offer its Unbundled Network Terminating Wire (UNTW) to NewSouth pursuant to the following terms and conditions at rates as set forth in this Attachment.

# 2.6.7.2 <u>Definition</u>

2.6.7.2.1 Subject to applicable and effective FCC rules and orders, UNTW is a dedicated transmission facility that BellSouth provides from the Wiring Closet /Garden Terminal (or other type of cross-connect point) at the point of termination of BellSouth's loop distribution facilities to the end user's point of demarcation.

# 2.6.7.3 <u>Requirements</u>

- 2.6.7.3.1 BellSouth will offer spare pairs that are available to an end user's premises to NewSouth. Available spare pairs are defined as pairs that are not being utilized by BellSouth or by a third party to provide an end user with working service at the time of NewSouth's request for UNTW. If no spare pairs are available and the end user is no longer using BellSouth's local service, BellSouth will relinquish the first pair to NewSouth. If after BellSouth has relinquished the first pair to NewSouth and the end user decides to change local service providers to BellSouth, NewSouth will relinquish the first pair back to BellSouth.
- 2.6.7.3.2 Notwithstanding the foregoing, should BellSouth subsequently require the use of additional pair(s) to provide for the activation of additional lines in an end users premises in response to a request from such end user, NewSouth agrees to surrender their spare pair(s) upon request by BellSouth.
- 2.6.7.3.3 If an end user of NewSouth desires to receive local exchange service from a service provider who is not a Party to this Agreement, and such third party service provider needs access to the BellSouth UNTW to provide local exchange service to the end user, then NewSouth agrees to surrender the requisite number of its inactive spare pair(s) if no other spare pair is available and upon request by BellSouth.
- 2.6.7.3.4 If NewSouth has placed NTW at a location and an end user desires to receive local exchange service from BellSouth and BellSouth needs access to NewSouth's NTW to provide local exchange service to the end user, then NewSouth agrees to make available the requisite number of its spare pair(s) upon request by BellSouth, at rates, terms and conditions to be negotiated by the Parties.
- 2.6.7.3.5 In new construction, where possible, both Parties may at their option and with the property owner's agreement install their own NTW. In existing construction, BellSouth shall not be required to install new or additional NTW beyond existing NTW to provision the services of NewSouth.
- 2.6.8 Technical Requirements

2.6.8.1 In these scenarios, BellSouth will connect the requested UNTW pairs to a single point of interconnection (SPOI) designed for NewSouth access to BellSouth's NTW. The SPOI will be installed either near BellSouth's garden terminal or wiring closet. NewSouth will be required to place a cross-box, terminal or other similar device and deliver a cable to this SPOI. NewSouth will then connect their cable to the cross-connect panel to access the requested UNTW pairs.

#### 2.7 Dark Fiber

#### 2.7.1 Definition

Dark Fiber is optical transmission facilities without attached multiplexing, aggregation or other electronics that connects two points within BellSouth's network. Dark Fiber is unused strands of optical fiber. It may be strands of optical fiber existing in aerial or underground structure. No line terminating elements terminated to such strands to operationalize its transmission capabilities will be available. No regeneration or optical amplification will be included with this element.

# 2.7.2 <u>Requirements</u>

- 2.7.2.1 BellSouth shall make available in a reasonable and non-discriminatory manner, Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. If BellSouth has bona fide plans to use the fiber within a two year planning period, there is no requirement to provide said fiber to NewSouth. BellSouth shall provide access to Dark Fiber at any technically feasible point.
- 2.7.2.2 If the requested dark fiber has any lightwave repeater equipment interspliced to it, BellSouth will remove such equipment at NewSouth's request subject to time and materials charges.
- 2.7.2.3 NewSouth may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- 2.7.2.4 BellSouth shall use its best efforts to provide to NewSouth information regarding the location, availability and performance of Dark Fiber within ten (10) business days for a records based answer and twenty (20) calendar days for a field based answer, after receiving a request from NewSouth ("Request"). Such request shall not be denied based on the fact that designated locations are not BellSouth end-offices or NewSouth's collocation space. Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation"). From the time of the Request to forty-five (45) days after Confirmation, BellSouth shall hold such requested Dark Fiber for NewSouth's use and may not allow any other party to use such media, including BellSouth.
- 2.7.2.5 BellSouth shall use its best efforts to make Dark Fiber available to NewSouth within thirty (30) business days after it receives written confirmation from NewSouth that the Dark Fiber previously deemed available by BellSouth is wanted for use by NewSouth.

This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable NewSouth to connect or splice NewSouth provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.

- 2.7.2.6 Dark Fiber shall meet the manufacturer's design specifications.
- 2.7.2.7 NewSouth may splice and test Dark Fiber obtained from BellSouth using NewSouth or NewSouth designated personnel. BellSouth shall provide appropriate interfaces to allow splicing and testing of Dark Fiber. BellSouth shall provide an excess cable length of 25 feet minimum (for fiber in underground conduit) to allow the uncoiled fiber to reach from the manhole to a splicing van.

## 2.8 Preordering Loop Makeup (LMU)

- 2.8.1 Description of Service
- 2.8.1.1 BellSouth shall make available to NewSouth loop makeup (LMU) data for BellSouth's network facilities. This section addresses LMU as a *preordering* transaction, distinct from NewSouth ordering any other service(s). Loop Makeup *Service Inquiries* (LMUSI) for preordering loop makeup are likewise unique from other preordering functions with associated service inquiries (SI) as described in this Agreement.
- 2.8.1.2 BellSouth will provide NewSouth with loop makeup information consisting of the composition of the loop material (copper/fiber); the existence, location and type of equipment on the loop, including but not limited to digital loop carrier or other remote concentration devises, feeder/distribution interfaces, bridge taps, load coils, pair-gain devices; the loop length; and the wire gauge. The LMUSI may be utilized by NewSouth for the purpose of determining whether the loop requested is capable of supporting DSL service or other advanced data services. The determination shall be made solely by NewSouth and BellSouth shall not be liable in any way for the performance of the advanced data services provisioned over said loop.
- 2.8.1.3 BellSouth's LMU information is provided to NewSouth as it exists either in BellSouth's databases or in its hard copy facility records. BellSouth does not guarantee accuracy or reliability of the LMU information provided.
- 2.8.1.4 BellSouth offers LMU information for the sole purpose of allowing NewSouth to determine whether, in NewSouth's judgment, BellSouth's loops will support the specific services that NewSouth wishes to provide over those loops. NewSouth may choose to use equipment that it deems will enable it to provide a certain type and level of service over a particular BellSouth loop; however, such configurations may not match BellSouth's or the industry's standards and specifications for the intended type and level of service. Accordingly, NewSouth shall be responsible for insuring that the

specific loop type (ADSL, HDSL, or otherwise) ordered on the LSR matches the LMU of the facility requested. NewSouth bears full responsibility for being knowledgeable of BellSouth's technical standards and the specifications of BellSouth's loops. NewSouth bears full responsibility for making the appropriate ordering decisions of matching BellSouth loops with NewSouth's equipment for accomplishing NewSouth's end goal for the intended service it wishes to provide its end-user(s). NewSouth is fully responsible for any of its service configurations that may differ from BellSouth's technical standard for the loop type ordered.

# 2.8.2 <u>Submitting Loop Makeup Service Inquiries</u>

- 2.8.2.1 NewSouth will be able to obtain LMU information by submitting a LMUSI mechanically or manually. **Mechanized** LMUSIs should be submitted through BellSouth's Operational Support Systems interfaces. After obtaining the resulting loop data from the mechanized LMUSI process, if NewSouth determines that it needs further loop data information in order to make a determination of loop service capability, NewSouth may initiate a separate manual LMUSI for a separate nonrecurring charge as set forth in Exhibit C hereto.
- 2.8.2.2 Manual LMUSIs shall be submitted on the preordering manual LMUSI form by means of fax or electronic-mail to BellSouth's Complex Resale Support Group (CRSG)/Account Team utilizing the Preordering Loop Makeup Service Inquiry form. The standard service interval for the return of a Loop Makeup Manual Service Inquiry is seven business days. This service interval is distinct from the interval applied to the subsequent service order. Manual LMUSIs are not subject to expedite requests.

## 2.8.3 <u>LMUSI Types & Associated Charges</u>

NewSouth may request LMU information by submitting LMUSIs in accordance with the rate elements in Exhibit C. LMU information is available for "working" loops (i.e., those loops that are currently serving a particular end user) and "spare" loops (i.e., those loops that are available to serve a particular end user but are not currently in service). NewSouth may request LMU information for up to three (3) spare facilities per Manual LMUSI and ten (10) spare facilities per Mechanized LMUSI. NewSouth shall submit its loop criteria when placing the LMUSI, and the response shall provide NewSouth with information of each loop (up to the total number of facilities queried) that meet the criteria specified by NewSouth.

- 2.8.3.1 NewSouth will be assessed a nonrecurring charge for each facility queried as specified in Exhibit C. Rates for all states are interim and subject to true-up pending approval of final rates by the respective State Commissions. True-ups will be retroactive to the effective date of this Agreement.
- 2.8.3.2 NewSouth may reserve spare facilities for up to four (4) days in connection with a LMUSI. Reservations are not available for working facilities. Reserved facilities for which NewSouth does not plan to place a UNE local service request (LSR) should be

- cancelled by NewSouth. Should NewSouth wish to cancel a reservation on a spare facility, the cancellation will require a facility reservation number (RESID/FRN).
- 2.8.3.3 The reservation holding timeframe is a maximum of four days from the time that BellSouth's LMU data is returned to NewSouth for the facility queried. During this holding time and prior to NewSouth's placing an LSR, the reserved facilities are rendered unavailable to other customers, whether for CLEC(s) or for BellSouth. Notwithstanding the foregoing, if multiple loops meet NewSouth's criteria as specified in the LMUSI and NewSouth does not order all of such loops, NewSouth shall not be entitled to specify which of the loops contained in the query response BellSouth will actually provision to complete NewSouth's order.
- 2.8.3.4 If NewSouth does not submit an LSR for a UNE service order on a reserved facility within the four-day reservation timeframe, the reservation of that spare facility will become invalid and the facility will be released.
- 2.8.3.5 Charges for preordering LMUSI are separate from any charges associated with ordering other services from BellSouth.
- 2.8.4 Ordering of Other UNE Services
- 2.8.4.1 Whenever NewSouth has reserved a facility through BellSouth's preordering LMU service, should NewSouth seek to place a subsequent UNE LSR on a reserved facility, NewSouth shall provide BellSouth the RESID/FRN of the single spare facility on the appropriate UNE LSR., NewSouth will be billed the appropriate rate element for the specific type UNE loop ordered by NewSouth as set forth in this Attachment. NewSouth will not be billed any additional Loop Makeup charges for the loop so ordered. Should NewSouth choose to place a UNE LSR having previously submitted a request for *preordering LMU without a reservation*, NewSouth will be billed the appropriate rate element for the specific UNE loop ordered as well as additional Loop Makeup charges as set forth in this Attachment. Rates are provided in the UNE Rate Exhibits for Attachment 2.
- 2.8.4.2 Where NewSouth submits an LSR to order facilities reserved during the LMUSI process, BellSouth will use its best efforts to assign to NewSouth the facility reserved as indicated on the return of the LMU. Multi-facility reservations per single RESID/FRN as provided with the mechanized LMUSI process are less likely to result in the specific assignment requested by NewSouth. For those occasions when BellSouth's assignment system cannot assign the specific facility reserved by NewSouth during the LMU pre-ordering transaction, BellSouth will assign to NewSouth, subject to availability, a facility that meets the BellSouth technical standards of the BellSouth type loop as ordered by NewSouth. If the ordered loop type is not available, NewSouth may utilize the Unbundled Loop Modification process or the Special Construction process, as applicable, to obtain the loop type ordered.

2.8.4.3 BellSouth offers LMU information for the sole purpose of allowing NewSouth to determine whether, in CLEC's judgment, BellSouth's loops will support the specific services that NewSouth wishes to provide over those loops. NewSouth may choose to use equipment that it deems will enable it to provide a certain type and level of service over a particular BellSouth loop; however, such configurations may not match BellSouth's or the industry's standards and specifications for the intended type and level of service. Accordingly, NewSouth shall be responsible for insuring that the specific loop type (ADSL, HDSL, or otherwise) ordered on the LSR matches the LMU of the facility requested. NewSouth bears full responsibility for being knowledgeable of BellSouth's technical standards and the specifications of BellSouth's loops. NewSouth bears full responsibility for making the appropriate ordering decisions of matching BellSouth loops with NewSouth's equipment for accomplishing NewSouth's end goal for the intended service it wishes to provide its end-user(s). NewSouth is fully responsible for any of its service configurations that may differ from BellSouth's technical standard for the loop type ordered.

### 3. Switching

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of local and tandem switching.

## 3.1 **Local Switching**

BellSouth shall provide non-discriminatory access to local circuit switching capability, and local tandem switching capability, on an unbundled basis, except as set forth below in Section 3.1.3 to NewSouth for the provision of a telecommunications service. BellSouth shall provide non-discriminatory access to packet switching capability on an unbundled basis to NewSouth for the provision of a telecommunications service only in the limited circumstance described below in Section 3.5.6

- 3.1.1. Except as otherwise provided herein, BellSouth shall not impose any restrictions on NewSouth regarding the use of Switching Capabilities purchased from BellSouth provided such use does not result in demonstrable harm to either the BellSouth network or personnel or the use of the BellSouth network by BellSouth or any other telecommunication carrier.
- 3.1.2. Local Circuit Switching Capability, including Tandem Switching Capability

### 3.1.2.1 Definition

Local Circuit Switching Capability is defined as: (A) line-side facilities, which include, but are not limited to, the connection between a loop termination at a main distribution frame and a switch line card; (B) trunk-side facilities, which include, but are not limited to, the connection between trunk termination at a trunk-side cross-connect panel and a switch trunk card; and (C) All features, functions, and capabilities of the switch, which include, but are not limited to: (1) the basic switching function of

connecting lines to lines, line to trunks, trunks to lines, and trunks to trunks, as well as the same basic capabilities made available to BellSouth's customers, such as a telephone number, white page listings, and dial tone; and (2) all other features that the switch is capable of providing, including but not limited to customer calling, customer local area signaling service features, and Centrex, as well as any technically feasible customized routing functions provided by the switch; (D) switching provided by remote switching modules.

- 3.1.2.2 When utilizing BellSouth's local circuit switching capability, local traffic shall be defined as set forth in Attachment 3 of this Agreement.
- 3.1.3 Notwithstanding BellSouth's general duty to unbundle local circuit switching, BellSouth shall not be required to unbundle local circuit switching for NewSouth when NewSouth serves end-users with four (4) or more voice-grade (DS-0) equivalents or lines in locations served by BellSouth's local circuit switches, which are in the following MSAs: Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, and BellSouth has provided non-discriminatory cost based access to the Enhanced Extended Link (EEL) throughout Density Zone 1 as determined by NECA Tariff No. 4 as in effect on January 1, 1999.
- 3.1.4 In the event that NewSouth orders local circuit switching for a single end user account name at a single physical end user location with four (4) or more two (2) wire voice-grade loops from a BellSouth central office in Density Zone 1 as determined by NECA Tariff No. 4 as in effect on January 1, 1999 that is also located in one of the MSAs listed above in Section 3.1.3, BellSouth shall charge NewSouth the market-based switching rates set forth in Exhibit C or, to the extent a particular rate is not set forth in Exhibit C, the parties shall negotiate such rate for use of the local circuit switching functionality for the affected facilities.
- 3.1.5 A featureless port is one that has a line port, switching facilities, and an interoffice port. A featured port is a port that includes all features then capable or a number of then capable features specifically requested by NewSouth. Any features that are not currently then capable but are technically feasible through the switch can be requested through the BFR process.
- 3.1.6 BellSouth will provide to NewSouth customized routing of calls: (i) to a requested directory assistance services platform; (ii) to an operator services platform pursuant to Section 10 of Attachment 2; (iii) for NewSouth's PIC'ed toll traffic in a two (2) PIC environment to an alternative OS/DA platform designated by NewSouth. NewSouth customers may use the same dialing arrangements as BellSouth customers.
- 3.1.7 Remote Switching Module functionality is included in Switching Capability. The switching capabilities used will be based on the line side features they support.

- 3.1.8 Switching Capability will also be capable of routing local, intraLATA, interLATA, and calls to international customer's preferred carrier; call features (e.g. call forwarding) and Centrex capabilities.
- 3.1.9 Where required to do so in order to comply with an effective Commission order, BellSouth will provide to NewSouth purchasing local BellSouth switching and reselling BellSouth local exchange service under Attachment 1, selective routing of calls to a requested directory assistance services platform or operator services platform. NewSouth customers may use the same dialing arrangements as BellSouth customers, but obtain a NewSouth branded service.

# 3.2 <u>Technical Requirements</u>

- 3.2.1 The requirements set forth in this Section apply to Local Switching, but not to the Data Switching function of Local Switching.
- 3.2.1.1 Local Switching shall be equal to or better than the requirements for Local Switching set forth in the applicable industry standard technical references.
- 3.2.1.2 When applicable, BellSouth shall route calls to the appropriate trunk or lines for call origination or termination.
- 3.2.1.3 Subject to this section, BellSouth shall route calls on a per line or per screening class basis to (1) BellSouth platforms providing Network Elements or additional requirements (2) Operator Services platforms, (3) Directory Assistance platforms, and (4) Repair Centers. Any other routing requests by NewSouth will be made pursuant to the Bona Fide Request/ New Business Request Process as set forth in General Terms and Conditions.
- 3.2.1.4 BellSouth shall provide unbranded recorded announcements and call progress tones to alert callers of call progress and disposition.
- 3.2.1.5 BellSouth shall activate service for an NewSouth customer or network interconnection on any of the Local Switching interfaces. This includes provisioning changes to change a customer from BellSouth's services to NewSouth's services without loss of switch feature functionality as defined in this Agreement.
- 3.2.1.6 BellSouth shall perform routine testing (e.g., Mechanized Loop Tests (MLT) and test calls such as 105, 107 and 108 type calls) and fault isolation on a mutually agreed upon schedule.
- 3.2.1.7 BellSouth shall repair and restore any equipment or any other maintainable component that may adversely impact Local Switching.
- 3.2.1.8 BellSouth shall control congestion points such as those caused by radio station callins, and network routing abnormalities. All traffic shall be restricted in a non-discriminatory manner.

- 3.2.1.9 BellSouth shall perform manual call trace and permit customer originated call trace.
- 3.2.1.10 Special Services provided by BellSouth will include the following:
- 3.2.1.10.1 Telephone Service Prioritization;
- 3.2.1.10.2 Related services for handicapped;
- 3.2.1.10.3 Soft dial tone where required by law; and
- 3.2.1.10.4 Any other service required by law.
- 3.2.1.11 BellSouth shall provide Switching Service Point (SSP) capabilities and signaling software to interconnect the signaling links destined to the Signaling Transfer Point Switch (STPS). These capabilities shall adhere to the technical specifications set forth in the applicable industry standard technical references.
- 3.2.1.12 BellSouth shall provide interfaces to adjuncts through Telcordia (formerly BellCore) standard interfaces. These adjuncts can include, but are not limited to, the Service Circuit Node and Automatic Call Distributors.
- 3.2.1.13 BellSouth shall provide performance data regarding a customer line, traffic characteristics or other measurable elements to NewSouth, upon a reasonable request from NewSouth. CLEC will pay BellSouth for all costs incurred to provide such performance data through the Business Opportunity Request process.
- 3.2.1.14 BellSouth shall offer Local Switching that provides feature offerings at parity to those provided by BellSouth to itself or any other Party. Such feature offerings shall include but are not limited to:
- 3.2.1.14.1 Basic and primary rate ISDN;
- 3.2.1.14.2 Residential features;
- 3.2.1.14.3 Customer Local Area Signaling Services (CLASS/LASS);
- 3.2.1.14.4 CENTREX (including equivalent administrative capabilities, such as customer accessible reconfiguration and detailed message recording); and
- 3.2.1.14.5 Advanced intelligent network triggers supporting NewSouth and BellSouth service applications.
- 3.2.2 BellSouth shall offer to NewSouth all AIN triggers in connection with its SMS/SCE offering which are supported by BellSouth for offering AIN-based services. Triggers that are currently available are:
- 3.2.2.1 Off-Hook Immediate

3.2.2.2 Off-Hook Delay 3.2.2.3 **Termination Attempt** 3.2.2.4 6/10 Public Office Dialing Plan 3.2.2.5 Feature Code Dialing 3.2.2.6 Customer Dialing Plan 3.2.3 When the following triggers are supported by BellSouth, BellSouth will make these triggers available to NewSouth: 3.2.3.1 Private EAMF Trunk 3.2.3.2 Shared Interoffice Trunk (EAMF, SS7) 3.2.3.3 N11 3.2.3.4 **Automatic Route Selection** 3.2.3.5 9XX Blocking 3.2.3.6 Toll Blocking 3.2.4 Where capacity exists, BellSouth shall assign each NewSouth customer line the class of service designated by NewSouth (e.g., using line class codes or other switch specific provisioning methods), and shall route directory assistance calls from NewSouth customers to NewSouth directory assistance operators at NewSouth's option. 3.2.5 Where capacity exists, BellSouth shall assign each NewSouth customer line the class of services designated by NewSouth (e.g., using line class codes or other switch specific provisioning methods) and shall route operator calls from NewSouth customers to NewSouth operators at NewSouth's option. For example, BellSouth may translate 0- and 0+ intraLATA traffic, and route the call through appropriate trunks to an NewSouth Operator Services Position System (OSPS). Calls from Local Switching must pass the ANI-II digits unchanged. 3.2.6 Local Switching shall be offered in accordance with the technical specifications set forth in the applicable industry standard references. 3.2.7 Interface Requirements

3.2.7.1

3.2.7.1.1

Standard Tip/Ring interface including loopstart or groundstart, on-hook signaling

BellSouth shall provide the following interfaces to loops:

(e.g., for calling number, calling name and message waiting lamp);

- 3.2.7.1.2 Coin phone signaling;
- 3.2.7.1.3 Basic Rate Interface ISDN adhering to appropriate Telcordia (formerly BellCore) Technical Requirements;
- 3.2.7.1.4 Two-wire analog interface to PBX;
- 3.2.7.1.5 Four-wire analog interface to PBX;
- 3.2.7.1.6 Four-wire DS1 interface to PBX or customer provided equipment (e.g. computers and voice response systems);
- 3.2.7.1.7 Primary Rate ISDN to PBX adhering to ANSI standards Q.931, Q.932 and appropriate Telcordia (formerly BellCore) Technical Requirements;
- 3.2.7.1.8 Switched Fractional DS1 with capabilities to configure Nx64 channels (where N=1 to 24); and
- 3.2.7.1.9 Loops adhering to Telcordia (formerly BellCore) TR-NWT-08 and TR-NWT-303 specifications to interconnect Digital Loop Carriers.
- 3.2.7.2 BellSouth shall provide access to the following but not limited to:
- 3.2.7.2.1 SS7 Signaling Network or Multi-Frequency trunking if requested by NewSouth;
- 3.2.7.2.2 Interface to NewSouth operator services systems or Operator Services through appropriate trunk interconnections for the system; and
- 3.2.7.2.3 Interface to NewSouth Directory Assistance Services through the NewSouth switched network or to Directory Assistance Services through the appropriate trunk interconnections for the system; and 950 access or other NewSouth required access to interexchange carriers as requested through appropriate trunk interfaces.

# 3.3 Tandem Switching

#### 3.3.1 Definition

Tandem Switching is the function that establishes a communications path between two switching offices through a third switching office (the Tandem switch).

- 3.3.2 Technical Requirements
- 3.3.2.1 Tandem Switching shall have the same capabilities or equivalent capabilities as those described in Bell Communications Research TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90. The requirements for Tandem Switching include, but are not limited to the following:
- 3.3.2.1.1 Tandem Switching shall provide signaling to establish a tandem connection;

- 3.3.2.1.2 Tandem Switching will provide screening as jointly agreed to by NewSouth and BellSouth;
- 3.3.2.1.3 Tandem Switching shall provide Advanced Intelligent Network triggers supporting AIN features where such routing is not available from the originating end office switch, to the extent such Tandem switch has such capability;
- 3.3.2.1.4 Tandem Switching shall provide access to Toll Free number portability database as designated by NewSouth;
- 3.3.2.1.5 Tandem Switching shall provide all trunk interconnections discussed under the "Network Interconnection" section (e.g., SS7, MF, DTMF, DialPulse, PRI-ISDN, DID, and CAMA-ANI (if appropriate for 911));
- 3.3.2.1.5.1 Tandem Switching shall provide connectivity to PSAPs where 911 solutions are deployed and the tandem is used for 911; and
- 3.3.2.1.5.2 Where appropriate, Tandem Switching shall provide connectivity to transit traffic to and from other carriers.
- 3.3.2.1.6 Tandem Switching shall accept connections (including the necessary signaling and trunking interconnections) between end offices, other tandems, IXCs, ICOs, CAPs and CLEC switches.
- 3.3.2.1.7 Tandem Switching shall provide local tandeming functionality between two end offices including two offices belonging to different CLEC's (e.g., between a CLEC end office and the end office of another CLEC).
- 3.3.2.1.8 Tandem Switching shall preserve CLASS/LASS features and Caller ID as traffic is processed.
- 3.3.2.1.9 Tandem Switching shall record billable events and send them to the area billing centers designated by NewSouth. Tandem Switching will provide recording of all billable events as jointly agreed to by NewSouth and BellSouth.
- 3.3.2.1.10 Upon a reasonable request from NewSouth, BellSouth shall perform routine testing and fault isolation on the underlying switch that is providing Tandem Switching and all its interconnections. The results and reports of the testing shall be made immediately available to NewSouth.
- 3.3.2.1.11 BellSouth shall maintain NewSouth's trunks and interconnections associated with Tandem Switching at least at parity to its own trunks and interconnections.
- 3.3.2.1.12 BellSouth shall control congestion points and network abnormalities. All traffic will be restricted in a non-discriminatory manner.

- 3.3.2.1.13 Selective Call Routing through the use of line class codes is not available through the use of tandem switching. Selective Call Routing through the use of line class codes is an end office capability only. Detailed primary and overflow routing plans for all interfaces available within BellSouth's switching network shall be mutually agreed to by NewSouth and BellSouth.
- 3.3.2.1.14 Tandem Switching shall process originating toll-free traffic received from NewSouth's local switch.
- 3.3.2.1.15 In support of AIN triggers and features, Tandem Switching shall provide SSP capabilities when these capabilities are not available from the Local Switching Network Element, to the extent such Tandem Switch has such capability.
- 3.3.2.2 Interface Requirements
- 3.3.2.2.1 Tandem Switching shall provide interconnection to the E911 PSAP where the underlying Tandem is acting as the E911 Tandem.
- 3.3.2.2.2 Tandem Switching shall interconnect, with direct trunks, to all carriers with which BellSouth interconnects.
- 3.3.2.2.3 BellSouth shall provide all signaling necessary to provide Tandem Switching with no loss of feature functionality.
- 3.3.2.2.4 Tandem Switching shall interconnect with NewSouth's switch, using two-way trunks, for traffic that is transiting via BellSouth's network to interLATA or intraLATA carriers. At NewSouth's request, Tandem Switching shall record and keep records of traffic for billing.
- 3.3.2.2.5 Tandem Switching shall provide an alternate final routing pattern for NewSouth's traffic overflowing from direct end office high usage trunk groups.
- 3.3.2.2.6 Tandem Switching shall be equal or better than the requirements for Tandem Switching set forth in the applicable technical references.
- 3.4 AIN Selective Carrier Routing for Operator Services, Directory Assistance and Repair Centers
- 3.4.1 BellSouth will provide AIN Selective Carrier Routing at the request of NewSouth. AIN Selective Carrier Routing will provide NewSouth with the capability of routing operator calls, 0+ and 0- and 0+ NPA (LNPA) 555-1212 directory assistance, 1+411 directory assistance and 611 repair center calls to pre-selected destinations.

- 3.4.2 NewSouth shall order AIN Selective Carrier Routing through its Account Team. AIN Selective Carrier Routing must first be established regionally and then on a per central office, per state basis.
- 3.4.3 AIN Selective Carrier Routing is not available in DMS 10 switches.
- 3.4.4 Where AIN Selective Carrier Routing is utilized by NewSouth, the routing of NewSouth's end user calls shall be pursuant to information provided by NewSouth and stored in BellSouth's AIN Selective Carrier Routing Service Control Point database. AIN Selective Carrier Routing shall utilize a set of Line Class Codes (LCCs) unique to a basic class of service assigned on an "as needed" basis. The same LCCs will be assigned in each central office where AIN Selective Carrier Routing is established.
- 3.4.5 Upon ordering of AIN Selective Carrier Routing Regional Service, NewSouth shall remit to BellSouth the Regional Service Order non-recurring charges set forth in Exhibit A of this Attachment. There shall be a non-recurring End Office Establishment Charge per office due at the addition of each central office where AIN Selective Carrier Routing will be utilized. Said non-recurring charge shall be as set forth in Exhibit A of this Attachment. For each NewSouth end user activated, there shall be a non-recurring End User Establishment charge as set forth in Exhibit A of this Attachment, payable to BellSouth pursuant to the terms of the General Terms and Conditions, incorporated herein by this reference. NewSouth shall pay the AIN Selective Carrier Routing Per Query Charge set forth in Exhibit A of this Attachment.
- 3.4.6 This Regional Service Order non-recurring charge will be non-refundable and will be paid with 1/2 coming up-front with the submission of all fully completed required forms, including: Regional Selective Carrier Routing (SCR) Order Request-Form A, Central Office AIN Selective Carrier Routing (SCR) Order Request Form B, AIN\_SCR Central Office Identification Form Form C, AIN\_SCR Routing Options Selection Form Form D, and Routing Combinations Table Form E. BellSouth has 30 days to respond to the client's fully completed firm order as a Regional Service Order. With the delivery of this firm order response to the client, BellSouth considers that the delivery schedule of this service commences. The remaining 1/2 of the Regional Service Order payment must be paid when at least 100% of the Central Offices listed on the original order have been turned up for the service.
- 3.4.7 The non-recurring End Office Establishment Charge will be billed to the client following our normal monthly billing cycle for this type of order.
- 3.4.8 End-User Establishment Orders will not be turned-up until the 2<sup>nd</sup> payment is received for the Regional Service Order. The non-recurring End-User Establishment Charges will be billed to the client following our normal monthly billing cycle for this type of order.

- 3.4.9 Additionally, the AIN Selective Carrier Routing Per Query Charge will be billed to the client following the normal billing cycle for per query charges.
- 3.4.10 All other network components needed, for example, unbundled switching and unbundled local transport, etc, will be billed according per contracted rates.

## 3.5 Packet Switching Capability

#### 3.5.1 Definition

Packet Switching Capability. The packet switching capability network element is defined as the basic packet switching function of routing or forwarding packets, frames, cells or other data units based on address or other routing information contained in the packets, frames, cells or other data units, and the functions that are performed by Digital Subscriber Line Access Mulitplexers, including but not limited to:

- 3.5.2 The ability to terminate copper customer loops (which includes both a low band voice channel and a high-band data channel, or solely a data channel);
- 3.5.3 The ability to forward the voice channels, if present, to a circuit switch or multiple circuit switches;
- 3.5.4 The ability to extract data units from the data channels on the loops, and
- 3.5.5 The ability to combine data units from multiple loops onto one or more trunks connecting to a packet switch or packet switches.
- 3.5.6 BellSouth shall be required to provide non-discriminatory access to unbundled packet switching capability only where each of the following conditions are satisfied:
- 3.5.6.1 BellSouth has deployed digital loop carrier systems, including but not limited to, integrated digital loop carrier or universal digital loop carrier systems; or has deployed any other system in which fiber optic facilities replace copper facilities in the distribution section (e.g., end office to remote terminal, pedestal or environmentally controlled vault);
- 3.5.6.2 There are no spare copper loops capable of supporting the xDSL services NewSouth seeks to offer:
- 3.5.6.3 BellSouth has not permitted NewSouth to deploy a Digital Subscriber Line Access Multiplexer at the remote terminal, pedestal or environmentally controlled vault or other interconnection point, nor has the NewSouth obtained a virtual collocation arrangement at these subloop interconnection points as defined by 47 C.F.R. § 51.319 (b); and
- 3.5.6.4 BellSouth has deployed packet switching capability for its own use.

3.5.7 If there is a dispute as to whether BellSouth must provide Packet Switching, such dispute will be resolved according tot the dispute resolution process set forth in Section 12 of the General Terms and Conditions of this Agreement, incorporated herein by this reference.

## 4. Enhanced Extended Link (EEL)

- 4.1 For purposes of this Section, references to "Already Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.
- Where necessary to comply with an effective FCC and/or State Commission order, or as otherwise mutually agreed by the Parties, BellSouth shall offer access to loop and transport combinations, also known as the Enhanced Extended Link ("EEL") as defined in Section 4.3 below.
- 4.2.2 Subject to Section 4.2.3 below, BellSouth will provide access to the EEL in the combinations set forth in 4.3 following. This offering is intended to provide connectivity from an end user's location through that end user's SWC to NewSouth's POP serving wire center. The circuit must be used for the purpose of provisioning telecommunications services, including telephone exchange service, to NewSouth's end-user customers. Except as provided for in paragraph 22 of the FCC's Supplemental Order Clarification, released June 2, 2000, in CC Docket No. 96-98 ("June 2, 2000 Order"), the EEL will be connected to NewSouth's facilities in NewSouth's collocation space at the POP SWC. NewSouth may purchase BellSouth's access facilities between NewSouth's POP and NewSouth's collocation space at the POP SWC.
- 4.2.3 BellSouth shall provide EEL combinations to NewSouth in the state of Georgia regardless of whether or not such EELs are Already Combined. In all other states, BellSouth shall make available to NewSouth those EEL combinations described in Section 4.3 below only to the extent such combinations are Already Combined.
- 4.2.4 BellSouth will make available EEL combinations to NewSouth in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs, regardless of whether or not such EELs are Already Combined.
- 4.2.5 Additionally, BellSouth shall make available to NewSouth a combination of an unbundled loop and tariffed special access interoffice facilities. To the extent NewSouth will require multiplexing functionality in connection with such combination, BellSouth will provide access to multiplexing within the central office pursuant to the terms, conditions and rates set forth in its Access Services Tariffs. The combination of an unbundled loop and tariffed special access interoffice facilities and any associated

tariffed services, including but not limited to multiplexing, shall not be eligible for conversion to UNEs as described in Section 4.5 below. Where multiplexing functionality is required in connection with loop and transport combinations, such multiplexing will be provided at the rates and on the terms set forth in this Agreement.

#### 4.3 EEL Combinations

- 4.3.1 DS1 Interoffice Channel + DS1 Channelization + 2-wire VG Local Loop
- 4.3.2 DS1 Interoffice Channel + DS1 Channelization + 4-wire VG Local Loop
- 4.3.3 DS1 Interoffice Channel + DS1 Channelization + 2-wire ISDN Local Loop
- 4.3.4 DS1 Interoffice Channel + DS1 Channelization + 4-wire 56 kbps Local Loop
- 4.3.5 DS1 Interoffice Channel + DS1 Channelization + 4-wire 64 kbps Local Loop
- 4.3.6 DS1 Interoffice Channel + DS1 Local Loop
- 4.3.7 DS3 Interoffice Channel + DS3 Local Loop
- 4.3.8 STS-1 Interoffice Channel + STS-1 Local Loop
- 4.3.9 DS3 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 4.3.10 STS-1 Interoffice Channel + DS3 Channelization + DS1 Local Loop
- 4.3.11 2-wire VG Interoffice Channel + 2-wire VG Local Loop
- 4.3.12 4-wire VG Interoffice Channel + 4-wire VG Local Loop
- 4.3.13 4-wire 56 kbps Interoffice Channel + 4-wire 56 kbps Local Loop
- 4.3.14 4-wire 64 kbps Interoffice Channel + 4-wire 64 kbps Local Loop

### 4.4 Other Network Element Combinations

In the state of Georgia, BellSouth shall make available to NewSouth, in accordance with Section 4.6 below: (1) combinations of network elements other than EELs that are Already Combined; and (2) combinations of network elements other than EELs that are not Already Combined but that BellSouth ordinarily combines in its network. In all other states, BellSouth shall make available to NewSouth, in accordance with Section 4.5 below, combinations of network elements other than EELs only to the extent such combinations are Already Combined.

- 4.5 Special Access Service Conversions
- 4.5.1 NewSouth may not convert special access services to combinations of loop and transport network elements, whether or not NewSouth self-provides its entrance facilities (or obtains entrance facilities from a third party), unless NewSouth uses the combination to provide a "significant amount of local exchange service" (as described in Section 4.5.2 below), in addition to exchange access service, to a particular customer. Such conversions of existing special access services pursuant to this section may include facilities within a single density zone (as described in 47 C. F. R. 69.123) or across Density Zones.
- 4.5.1.2 For the purpose of special access conversions under Section 4.5.1, a "significant amount of local exchange service" is as defined in the FCC's June 2, 2000 Order. The Parties agree to incorporate by reference paragraph 22 of the June 2, 2000 Order. When NewSouth requests conversion of special access circuits, NewSouth will self-certify to BellSouth in the manner specified in paragraph 29 of the June 2, 2000 Order that the circuits to be converted qualify for conversion. In addition there may be extraordinary circumstances where NewSouth is providing a significant amount of local exchange service, but does not qualify under any of the three options set forth in paragraph 22 of June 2, 2000 Order, or under a fourth option set forth below in Section 4.5.2. In such case, NewSouth may petition the FCC for a waiver of the local usage options set forth in the June 2, 2000 Order. If a waiver is granted, then upon NewSouth's request the Parties shall amend this Agreement to the extent necessary to incorporate the terms of such waiver for such extraordinary circumstance.
- 4.5.1.3 The recurring charges for such combinations shall be the sum of the recurring charge for the applicable UNE loop and transport segments (including multiplexing, if applicable), as set forth in Exhibit C to this Attachment. The nonrecurring charges for such combinations shall be an amount equal to all applicable conversion charges set forth in Exhibit C to this Attachment for conversion of special access circuits to EELs, plus all applicable nonrecurring cross connect charges (set forth in Attachment 4 to this Agreement) required to connect the facility to NewSouth's collocation arrangement. EELs that terminate in NewSouth collocation arrangements may be connected by NewSouth via cross-connects to BellSouth services used by NewSouth to transport traffic between NewSouth's collocation space and NewSouth's POP.
- 4.5.1.4 Upon request for conversions of up to 15 circuits from special access to EELs, BellSouth shall perform such conversions within seven (7) days from BellSouth's receipt of a valid, error free service order from NewSouth. Requests for conversions of fifteen (15) or more circuits from special access to EELs will be provisioned on a project basis. Except as set forth in Section 4.5.3 below, conversions should not require the special access circuit to be disconnected and reconnected because only the billing information or other administrative information associated with the circuit will change when NewSouth requests a conversion. Submission of a spreadsheet

identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR), only until such time as the LSR process is modified to accommodate such requests.

- 4.5.1.5 BellSouth may, at its sole expense, and upon thirty (30) days notice to NewSouth, audit NewSouth's records not more than once in any twelve month period, unless an audit finds non-compliance with the local usage options referenced in the June 2, 2000 Order, in order to verify the type of traffic being transmitted over combinations of loop and transport network elements. If, based on its audits, BellSouth concludes that NewSouth is not providing a significant amount of local exchange traffic over the combinations of loop and transport network elements, BellSouth may file a complaint with the appropriate Commission, pursuant to the dispute resolution process set forth in this Agreement. In the event that BellSouth prevails, BellSouth may convert such combinations of loop and transport network elements to special access services and may seek appropriate retroactive reimbursement from NewSouth.
- 4.5.2 In addition to the circumstances under which NewSouth may identify special access circuits that qualify for conversions to EELs (referenced in Section 4.5.1.2 above), NewSouth also shall be entitled to convert special access circuits to unbundled network elements pursuant to the terms of this section 4.5.2 et seq.
- 4.5.2.1 Upon request by NewSouth, BellSouth will convert special access circuits to combinations of an unbundled loop connected to special access transport provided that: (1) the combination terminates to a NewSouth collocation arrangement; and (2) NewSouth certifies, in the manner set forth in Section 4.5.2 above, that at least 75% of the unbundled network element(s) component of the facility is used to provide originating and terminating local voice traffic. The recurring charges for such combinations shall be the sum of the recurring charge for the applicable UNE loop, as set forth in Exhibit C to this Attachment, and all applicable recurring charges for the special access transport facility, as set forth in the BellSouth tariff under which such facilities were ordered. The nonrecurring charges for such combinations shall be an amount equal to all applicable conversion charges set forth in Exhibit C to this Attachment for conversion of special access circuits to EELs, plus the applicable nonrecurring cross connect charges (set forth in Attachment 4 to this Agreement) required to connect the facility to NewSouth's collocation arrangement. Such combinations that terminate in NewSouth collocation arrangements may be connected by NewSouth via cross-connects to BellSouth services used by NewSouth to transport traffic between NewSouth's collocation space and NewSouth's POP.
- 4.5.2.2 Upon request from NewSouth to convert special access circuits pursuant to Section 4.5.2, BellSouth shall have the right, upon 10 business days notice, to conduct an audit prior to any such conversion to determine whether the subject facilities meet local usage requirements set forth in Section 4.5.2. An audit conducted pursuant to this Section shall take into account a usage period of the past three (3) consecutive

- months, and shall be subject to the requirements for audits as set forth in the June 2, 2000 Order, except as expressly modified herein.
- 4.5.3 In consideration of Section 4.5.2.1 above, and subject to Section 4.5.7 below, for those special access circuits identified by NewSouth in writing as of January 19, 2001 as being eligible for conversion pursuant to the terms of this Agreement, BellSouth will provide to NewSouth a credit in an amount equal to three times the difference between the monthly special access rates for such circuits and the monthly rates for the combinations to which those circuits are converted.
- 4.5.3.1 For circuits converted pursuant to one of the three options made available to NewSouth in Section 4.5.1, the credit will be in an amount equal to three times the difference between the monthly special access rates for such circuits and the monthly UNE recurring charges for the loop, transport and multiplexing (if applicable), as set forth in Exhibit C to this Attachment, that, in combination, form an EEL.
- 4.5.3.2 For circuits converted pursuant to the fourth option made available to NewSouth in Section 4.5.2, the credit will be in an amount equal to three times the difference between the monthly special access rates for such circuits and the sum of the monthly UNE recurring charges for the loop, as set forth in Exhibit C to this Attachment, and the monthly recurring charge for the special access transport facility, as set forth in the BellSouth tariff under which such facility was ordered.
- 4.5.3.3 Such credits will be applied to NewSouth's bill within sixty (60) days following execution of this Agreement.
- 4.5.3.4 Within ten (10) days following execution of this Agreement, NewSouth shall certify to BellSouth in writing that the circuits designated as of January 19, 2001 meet significant local use requirements of one of the four conversion options set forth above. Such certification shall include a designation by NewSouth of which of the particular four conversion options specified herein is applicable to each of the individual circuits designated as of January 19, 2001.
- 4.5.3.5 BellSouth shall assign a project management team and designate a project manager to facilitate the timely conversion of special access circuits. BellSouth and NewSouth will participate in a joint implementation meeting within fifteen (15) days following execution of this Agreement, or within 15 days of any subsequent request for conversion, to establish a schedule for conversion of the identified special access circuits. BellSouth shall complete conversions of all circuits identified by NewSouth as of January 19, 2001 within 3 months of the joint implementation meeting, unless an alternative completion date is agreed to by the Parties. For purposes of conversion of the circuits identified by NewSouth as of January 19, 2001, NewSouth's spreadsheet identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR). For subsequent conversion requests pursuant to Sections 4.5.1 and 4.5.2 above, submission of a spreadsheet identifying the circuits to be

- converted shall serve as a substitute for submission of a local service request (LSR), only until such time as the LSR process is modified to accommodate such requests.
- 4.5.4 For all special access circuits converted under this Agreement, NewSouth shall pay BellSouth any termination charges applicable to the special access circuits converted, as specified in BellSouth's tariffs.
- 4.5.5 The Parties acknowledge that the conversion option described in Section 4.5.2 and the credits offered NewSouth in Section 4.5.3 constitute a reasonable negotiated alternative to those developed by the FCC in the June 2, 2000 Order. However, BellSouth has agreed to the terms of Sections 4.5.2 and 4.5.3 based upon the assumption that the FCC's current rules regarding special access conversions will remain in effect throughout the 2001 calendar year. In the event that the FCC modifies its rules regarding conversion of special access circuits in a manner that is inconsistent with BellSouth's stated position on the issue, then BellSouth cannot realize the value of the alternative option made available to NewSouth hereunder. In the event that the FCC rules regarding special access conversions are modified in the manner described herein with an effective date prior to January 1, 2002, NewSouth will reimburse BellSouth one-seventh of the credits extended to NewSouth under Section 4.5.3 above for each month or portion thereof prior to January 1, 2002, that such modified FCC rules are in effect.
- 4.6 Rates
- 4.6.1 Georgia
- 4.6.1.1 The non-recurring and recurring rates for the EEL Combinations of network elements set forth in 4.3, whether Already Combined or new, are as set forth in this Attachment.
- 4.6.1.2 On an interim basis, for combinations of loop and transport network elements not set forth in Section 4.3, where the elements are not Already Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.
- 4.6.1.3 To the extent that NewSouth seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, NewSouth, at its option, can request that such rates be determined pursuant to the Bona Fide Request/New Business Request (NBR) process set forth in this Agreement.
- 4.6.2 All Other States
- 4.6.2.1 Subject to Section 4.2.3 and 4.4 preceding, all other states, the rates for (1) Already Combined EEL combinations set forth in Section 4.3, and (2) other combinations of network elements that are Already Combined in the network will be the sum of the

- recurring rates for the individual network elements plus a nonrecurring charge as specified in Exhibit C of this Attachment.
- 4.6.2.2 Rates for new EEL combinations in Density Zone 1 in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs shall be as set forth in Exhibit C hereto; provided, however, that to the extent a rate is not established in Exhibit C, the rate shall be the sum of the recurring and nonrecurring charges for the individual network elements as set forth in Exhibit C to this Attachment, unless otherwise established by the Commission.

## 5. Port/Loop Combinations

- 5.1 For purposes of this Section, references to "Already Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location. For purposes of this Section, "soft dial tone" (i.e., where network elements are connected through from the end user premises to the BellSouth end office and no dispatch is required to initiate service) shall be considered "Already Combined".
- 5.2 At NewSouth's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in Section 5.5 below, that are Already Combined in BellSouth's network except as specified in Sections 5.2.1 and 5.2.2 below, consistent with the requirements of 47 C.F.R. 315(b) and all applicable FCC and Commission rules and policies.
- 5.2.1 BellSouth shall not provide access to combinations of unbundled port and loop network elements in locations where, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.
- 5.2.2 In accordance with effective and applicable FCC rules, BellSouth shall not provide unbundled circuit switching in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, of the Atlanta, Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs to NewSouth if NewSouth's customer has 4 or more DS0 equivalent lines.
- 5.3 Combinations of port and loop network elements provide local exchange service for the origination or termination of calls. BellSouth shall make available the following loop and port combinations at the terms and at the rates set forth below:
- 5.3.2.1 In Georgia, BellSouth shall provide to NewSouth combinations of port and loop network elements to NewSouth on an unbundled basis regardless of whether or not such combinations are Currently Combined except in those locations where BellSouth is not required to provide circuit switching, as set forth in Section 5.2.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit C of this Attachment.

- 5.3.2.2 In all other states, BellSouth shall provide to NewSouth combinations of port and loop network elements on an unbundled basis if such combinations are Currently Combined, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit C of this Attachment.
- 5.3.2.3 In all states other than Georgia, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2, BellSouth shall provide to NewSouth combinations of port and loop network elements that are not Currently Combined. The rates for such combinations shall be negotiated by the Parties.
- 5.3.2.4 In those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2, BellSouth shall provide to NewSouth combinations of port and loop network elements whether or not such combinations are Currently Combined. The rates for Currently Combined combinations are the market based rates as set forth in Exhibit C. The rates for not Currently Combined combinations shall be negotiated by the Parties.
- 5.4 When NewSouth orders loop/port combinations, and identifies to BellSouth the type of telecommunications service it intends to deliver to its end user customer through that combination (e.g., POTS, ISDN), BellSouth will provide the requested elements with all the functionality, and with at least the same quality of performance and operations systems support (ordering, provisioning, maintenance, billing and recording), that BellSouth provides through its own network to its local exchange service customers receiving equivalent service, unless NewSouth requests a lesser or greater quality of performance through the Bona Fide Request process. BellSouth will provide ordering, provisioning and maintenance services, including intervals, at parity with the same services BellSouth provides to its own end users or resold services as measured in Attachment 9 Performance Measures. The intervals that BellSouth provides for its products and services are as set forth in the Products and Services Interval Guide which can be found on the BellSouth Interconnection website at www.interconnection.BellSouth.com. The Products and Services Interval Guide may be amended from time to time. Any intervals contained in The Products and Service Interval Guide will not be increased unless ordered to do otherwise by the appropriate regulatory or judicial body. BellSouth's intervals begin with the receipt of an error free local service request (LSR). At the time of this interconnection agreement, not all combinations can be ordered electronically. All residence, business, and PBX port loop services can be electronically ordered. BellSouth will provide manual ordering processes for loop port combinations which cannot be electronically processed. BellSouth will provide notice of additional electronic ordering functionality via the Change Control Process.

- 5.4 Rates for Combinations of Loop and Port Network Elements
- Rates for combinations of loop and port network elements, as set forth in Section 5.4, are provided in Exhibit A of this Attachment.
- 5.4.2 Rates for Circuit Switching
- 5.4.2.1 Rates for circuit switching, where BellSouth is not required, pursuant to Sections 5.1.1 and 5.1.2, to provide circuit switching are as set forth in Exhibit A of this Attachment.
- 5.5 Port/Loop Combination Offerings
- 5.5.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.2 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.3 2-wire CENTREX port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.4. 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.5 2-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.6 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 4-wire DS1Loop with normal serving wire center channelization interface, 2 wire voice grade ports (PBX), 2-wire DID ports, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

# 6. Transport and Dark Fiber

BellSouth shall provide nondiscriminatory access, in accordance with FCC Rule 51.311 and Section 251(c)(3) of the Act, to interoffice transmission facilities on an unbundled basis to NewSouth for the provision of a telecommunications service. All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of unbundled transport and dark fiber.

# 6.1. **Transport**

6.1.1 <u>Definition of Common (Shared) Transport</u>

Common (Shared) Transport is an interoffice transmission path between two BellSouth end-offices, BellSouth end-office and a local tandem, or between two local tandems. Where BellSouth Network Elements are connected by intra-office wiring, such wiring is provided as a part of the Network Elements and is not Common (Shared) Transport. Common (Shared) Transport consists of BellSouth inter-office transport facilities and is unbundled from local switching.

- 6.1.2 Technical Requirements of Common (Shared) Transport
- 6.1.2.1 Common (Shared) Transport provided on DS1 or VT1.5 circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Central Office to Central Office ("CO to CO") connections in the appropriate industry standards.
- 6.1.2.2 Common (Shared) Transport provided on DS3 circuits, STS-1 circuits, and higher transmission bit rate circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CO to CO connections in the appropriate industry standards.
- 6.1.2.3 BellSouth shall be responsible for the engineering, provisioning, and maintenance of the underlying equipment and facilities that are used to provide Common (Shared) Transport.
- 6.1.2.4 At a minimum, Common (Shared) Transport shall meet all of the requirements set forth in the applicable industry standard technical references.
- 6.2 Interoffice transmission facility network elements include:
- 6.2.1 Dedicated transport, defined as BellSouth's transmission facilities, including all technically feasible capacity-related services including, but not limited to, DS1, DS3 and OCn levels, dedicated to a particular customer or carrier, that provide telecommunications between wire centers or switches owned by BellSouth, or between wire centers and switches owned by BellSouth and NewSouth.
- Dark Fiber transport, defined as BellSouth's optical transmission facilities without attached multiplexing, aggregation or other electronics;

- 6.2.3 Shared transport, defined as transmission facilities shared by more than one carrier, including BellSouth, between end office switches, between end office switches and tandem switches, and between tandem switches, in BellSouth's network.
- 6.2.4 BellSouth shall:
- 6.2.4.1 Provide NewSouth exclusive use of interoffice transmission facilities dedicated to a particular customer or carrier, or shared use of the features, functions, and capabilities of interoffice transmission facilities shared by more than one customer or carrier;
- 6.2.4.2 Provide all technically feasible transmission facilities, features, functions, and capabilities that NewSouth could use to provide telecommunications services;
- 6.2.4.3 Permit, to the extent technically feasible, NewSouth to connect such interoffice facilities to equipment designated by NewSouth, including but not limited to, NewSouth's collocated facilities; and
- 6.2.4.4 Permit, to the extent technically feasible, NewSouth to obtain the functionality provided by BellSouth's digital cross-connect systems in the same manner that BellSouth provides such functionality to interexchange carriers.

# **6.3** Dedicated Transport

- 6.3.1 <u>Definitions</u>
- 6.3.2 Dedicated Transport is defined as BellSouth transmission facilities dedicated to a particular customer or carrier that provide telecommunications between wire centers owned by BellSouth or requesting telecommunications carriers, or between switches owned by BellSouth or requesting telecommunications carriers.
- 6.3.3 Unbundled Local Channel
- 6.3.4 Unbundled Local Channel is the dedicated transmission path between NewSouth's Point of Presence and the BellSouth Serving Wire Center's collocation.
- 6.3.5 Unbundled Interoffice Channel.
- 6.3.6 Unbundled Interoffice Channel is the dedicated transmission path that provides telecommunication between BellSouth's Serving Wire Centers' collocations.
- 6.3.7 BellSouth shall offer Dedicated Transport in each of the following ways:
- 6.3.7.1 As capacity on a shared UNE facility.
- As a circuit (e.g., DS0, DS1, DS3, OCn) dedicated to NewSouth. This circuit shall consist of an Unbundled Local Channel or an Unbundled Interoffice Channel or both.

- 6.3.8 When Dedicated Transport is provided it shall include:
- 6.3.8.1 Transmission equipment such as, line terminating equipment, amplifiers, and regenerators;
- 6.3.8.2 Inter-office transmission facilities such as optical fiber, copper twisted pair, and coaxial cable.
- Rates for Dedicated Transport are listed in this Attachment. For those states that do not contain rates in this Attachment the rates in the applicable State Access Tariff will apply as interim rates. When final rates are developed, these interim rates will be subject to true up, and the Parties will amend the Agreement to reflect the new rates.
- 6.3.10 <u>Technical Requirements</u>
- 6.3.10.1 This Section sets forth technical requirements for all Dedicated Transport.
- 6.3.10.2 When BellSouth provides Dedicated Transport, the entire designated transmission service (e.g., DS0, DS1, DS3, and OCn) shall be dedicated to NewSouth designated traffic.
- 6.3.10.3 BellSouth shall offer Dedicated Transport in all technologies that become available including, but not limited to, (1) DS0, DS1, DS3, and OCn transport services, and (2) SONET at available transmission bit rates.
- 6.3.10.4 For DS1 or VT1.5 circuits, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Customer Interface to Central Office ("CI to CO") connections in the appropriate industry standards.
- 6.3.10.5 Where applicable, for DS3, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the appropriate industry standards.
- 6.3.10.6 SONET, OC-3, OC-12, and OC-48 Dedicated Transport shall, at a minimum meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the appropriate industry standards.
- 6.3.10.7 BellSouth shall offer the following interface transmission rates for Dedicated Transport:
- 6.3.10.7.1 DS0 Equivalent;
- 6.3.10.7.2 DS1
- 6.3.10.7.3 DS3
- 6.3.10.7.4 OC-3,

- 6.3.10.7.5 OC-12,
- 6.3.10.7.6 OC-48
- 6.3.10.7.6 SDH (Synchronous Digital Hierarchy) Standard interface rates in accordance with International Telecommunications Union (ITU) Recommendation G.707 and Plesiochronous Digital Hierarchy (PDH) rates per ITU Recommendation G.704.
- 6.3.10.7.8 When Dedicated Transport is provided, BellSouth shall design it according to BellSouth's network infrastructure to allow for the termination points specified by NewSouth.
- 6.3.11 At a minimum, Dedicated Transport shall meet each of the requirements set forth in the applicable industry technical references.
- 6.3.11.1 BellSouth Technical References:
- 6.3.11.2 TR-TSY-000191 Alarm Indication Signals Requirements and Objectives, Issue 1, May 1986.
- 6.3.11.3 TR 73501 LightGate<sup>®</sup> Service Interface and Performance Specifications, Issue D, June 1995.
- 6.3.11.4 TR 73525 MegaLink® Service, MegaLink Channel Service & MegaLink Plus Service Interface and Performance Specifications, Issue C, May 1996.

## 6.4 Unbundled Channelization

- 6.4.1 BellSouth agrees to offer access to Unbundled Channelization when available pursuant to following terms and conditions and at the rates set forth in the Attachment.
- 6.4.2 Definition
- Unbundled Channelization (UC) provides the multiplexing capability that will allow a DS1 (1.544 Mbps) or DS3 (44.736 Mbps) or STS-1 Unbundled Network Element (UNE) or collocation cross-connect to be multiplexed or channelized at a BellSouth central office. This can be accomplished through the use of a stand-alone multiplexer or a digital cross-connect system at the discretion of BellSouth. Once UC has been installed, NewSouth can have channels activated on an as-needed basis by having BellSouth connect lower level UNEs via Central Office Channel Interfaces (COCIs).
- 6.4.3 Channelization capabilities will be as follows:

- 6.4.3.1 DS3 Channelization System: An element that channelizes a DS3 signal into 28 DS1s/STS-1s.
- 6.4.3.2 DS1 Channelization System: An element that channelizes a DS1 signal into 24 DS0s.
- 6.4.3.3 Central Office Channel Interfaces (COCI): Elements that can be activated on a channelization system.
- 6.4.4 DS1 Central Office Channel Interface elements can be activated on a DS3 Channelization System.
- Voice Grade and Digital Data Central Office Channel Interfaces can be activated on a DS1 Channelization System.
- 6.4.6 AMI and B8ZS line coding with either Super Frame (SF) and Extended Super Frame (ESF) framing formats will be supported as options.
- 6.4.7 COCI will be billed on the lower level UNE order that is interfacing with the UC arrangement and will have to be compatible with those UNEs.
- 6.4.8 Channelization may be incorporated within dedicated transport or ordered as a standalone capability, which requires either the high or low speed side to be connected to collocation.
- 6.4.9 Technical Requirements
- 6.4.9.1 In order to assure proper operation with BST provided central office multiplexing functionality, the customer's channelization equipment must adhere strictly to form and protocol standards. Separate standards exist for the multiplex channel bank, for voice frequency encoding, for various signaling schemes, and for subrate digital access.
- 6.4.9.2 DS0 to DS1 Channelization
- 6.4.9.2.1 The DS1 signal must be framed utilizing the framing structure defined in ANSI T1.107, Digital Hierarchy Formats Specifications and ANSI T1.403.02, DS1 Robbed-bit Signaling State Definitions. DS0 to DS1 Channelization requirements are essential the same as defined in BellSouth Technical Reference 73525, MegaLink® Service, MegaLink® Channel Service, MegaLink® Plus Service, and MegaLink® Light Service Interface and Performance Specification.
- 6.4.9.3 DS1 to DS3 Channelization
- 6.4.9.3.1 The DS3 signal must be framed utilizing the framing structure define in ANSI T1.107, Digital Hierarchy Formats Specifications. DS1 to DS3 Channelization requirements are essentially the same as defined in BellSouth Technical Reference 73501,

LightGate<sup>®</sup> Service Interface and Performance Specifications. The asynchronous M13 multiplex format (combination of M12 and M23 formats) is specified for terminal equipment that multiplexes 28 DS1s into a DS3.

#### 6.4.9.4 DS1 to STS Channelization

6.4.9.4.1 The STS-1 signal must be framed utilizing the framing structure define in ANSI T1.105, Synchronous Optical Network (SONET) – Basic Description Including Multiplex Structure, Rates and Formats and T1.105.02, Synchronous Optical Network (SONET) – Payload Mappings. DS1 to STS Channelization requirements are essentially the same as defined in BellSouth Technical Reference TR 73501, LightGate® Service Interface and Performance Specifications

# 6.5 Dark Fiber

## 6.5.1 Definition

Dark Fiber is optical transmission facilities without attached multiplexing, aggregation or other electronics that connects two points within BellSouth's network. Dark Fiber is unused strands of optical fiber. It may be strands of optical fiber existing in aerial or underground structure. No line terminating elements terminated to such strands to operationalize its transmission capabilities will be available. No regeneration or optical amplification will be included with this element.

## 6.5.3 Requirements

- 6.5.3.1 BellSouth, on a reasonable and non-discriminatory basis, shall make available Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. If BellSouth has bona fide plans to use the fiber within a two-year period, there is no requirement to provide said fiber to NewSouth. BellSouth shall provide access to Dark Fiber at any technically feasible point.
- 6.5.3.2 If the requested dark fiber has any lightwave repeater equipment interspliced to it, BellSouth will remove such equipment at NewSouth's request subject to time and materials charges.
- 6.5.3.3 NewSouth may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- 6.5.3.4 BellSouth shall use its best efforts to provide to NewSouth information regarding the location, availability and performance of Dark Fiber within ten (10) business days for a records based answer and twenty (20) calendar days for a field based answer, after receiving a request from NewSouth ("Request"). Such request shall not be denied based on the fact that designated locations are not BellSouth end-offices or NewSouth's collocation space. Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation"). From the time of the

Request to forty-five (45) days after Confirmation, BellSouth shall hold such requested Dark Fiber for NewSouth's use an may not allow any other party to use such media, including BellSouth.

- 6.5.3.5 BellSouth shall use its best efforts to make Dark Fiber available to NewSouth within thirty (30) business days after it receives written confirmation from NewSouth that the Dark Fiber previously deemed available by BellSouth is wanted for use by NewSouth. This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable NewSouth to connect or splice NewSouth provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.
- 6.5.3.6 Dark Fiber shall meet the manufacturer's design specifications.
- 6.5.3.7 NewSouth may splice and test Dark Fiber obtained from BellSouth using NewSouth or NewSouth designated personnel. BellSouth shall provide appropriate interfaces to allow splicing and testing of Dark Fiber. BellSouth shall provide an excess cable length of 25 feet minimum (for fiber in underground conduit) to allow the uncoiled fiber to reach from the manhole to a splicing van.
- 7. BellSouth Switched Access (SWA) 8XX Toll Free Dialing Ten Digit Screening Service

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of 8XX Access Ten Digit Screening Services.

- 7.1 BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database
- 7.1.1 The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database (herein known as 8XX SCP) is a SCP that contains customer record information and functionality to provide call-handling instructions for 8XX calls. The 8XX SCP IN software stores data downloaded from the national SMS and provides the routing instructions in response to queries from the SSP or tandem. The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service (herein know as 8XX TFD), utilizes the 8XX SCP to provide identification and routing of the 8XX calls, based on the ten digits dialed. 8XX TFD is provided with or without POTS number delivery, dialing number delivery, and other optional complex features as selected by NewSouth. BellSouth shall provide 8XX TFD in accordance with the following:

# 7.1.2 <u>Technical Requirements</u>

7.1.2.1 BellSouth shall provide NewSouth with access to the 8XX record information located in the 8XX SCP. The 8XX SCP contains current records as received from the national SMS and will provide for routing 8XX originating calls based on the dialed ten digit 8XX number.

- 7.1.2.2 The 8XX SCP is designated to receive and respond to queries using the American National Standard Specification of Signaling System Seven (SS7) protocol. The 8XX SCP shall determine the carrier identification based on all ten digits of the dialed number and route calls to the carrier, POTS number, dialing number and/or other optional feature selected by NewSouth.
- 7.1.2.3 The SCP shall also provide, at NewSouth's option, such additional feature as described in SR-TSV-002275 (BOC Notes on BellSouth Networks, SR-TSV-002275, Issue 2, (Telcordia (formerly BellCore), April 1994)) as are available to BellSouth. These may include but are not limited to:
- 7.1.2.3.1 Network Management;
- 7.1.2.3.2 Customer Sample Collection; and
- 7.1.2.3.3 Service Maintenance.

# **8** Line Information Database (LIDB)

- 8.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of LIDB.
- 8.2 BellSouth will store in its LIDB only records relating to service in the BellSouth region. The LIDB Storage Agreement is included in this Attachment.

# 8.2.1 Definition

8.2.2 The Line Information Database (LIDB) is a transaction-oriented database accessible through Common Channel Signaling (CCS) networks. It contains records associated with end user Line Numbers and Special Billing Numbers. LIDB accepts queries from other Network Elements and provides appropriate responses. The query originator need not be the owner of LIDB data. LIDB queries include functions such as screening billed numbers that provides the ability to accept Collect or Third Number Billing calls and validation of Telephone Line Number based non-proprietary calling cards. The interface for the LIDB functionality is the interface between BellSouth's CCS network and other CCS networks. LIDB also interfaces to administrative systems.

## 8.2.3 <u>Technical Requirements</u>

- 8.2.4 BellSouth will offer to NewSouth any additional capabilities that are developed for LIDB during the life of this Agreement.
- 8.2.4.1 BellSouth shall process NewSouth's Customer records in LIDB at least at parity with BellSouth customer records, with respect to other LIDB functions. BellSouth shall indicate to NewSouth what additional functions (if any) are performed by LIDB in the BellSouth network.

- 8.2.4.2 Within two (2) weeks after a request by NewSouth, BellSouth shall provide NewSouth with a list of the customer data items, which NewSouth would have to provide in order to support each required LIDB function. The list shall indicate which data items are essential to LIDB function, and which are required only to support certain services. For each data item, the list shall show the data formats, the acceptable values of the data item and the meaning of those values.
- 8.2.4.3 BellSouth shall provide LIDB systems for which operating deficiencies that would result in calls being blocked shall not exceed 30 minutes per year.
- 8.2.4.4 BellSouth shall provide LIDB systems for which operating deficiencies that would not result in calls being blocked shall not exceed 12 hours per year.
- 8.2.4.5 BellSouth shall provide LIDB systems for which the LIDB function shall be in overload no more than 12 hours per year.
- 8.2.4.6 All additions, updates and deletions of NewSouth data to the LIDB shall be solely at the direction of NewSouth. Such direction from NewSouth will not be required where the addition, update or deletion is necessary to perform standard fraud control measures (e.g., calling card auto-deactivation).
- 8.2.4.7 BellSouth shall provide priority updates to LIDB for NewSouth data upon NewSouth's request (e.g., to support fraud detection), via password-protected telephone card, facsimile, or electronic mail within one hour of notice from the established BellSouth contact.
- 8.2.4.8 BellSouth shall provide LIDB systems such that no more than 0.01% of NewSouth customer records will be missing from LIDB, as measured by NewSouth audits. BellSouth will audit NewSouth records in LIDB against DBAS to identify record mismatches and provide this data to a designated NewSouth contact person to resolve the status of the records and BellSouth will update system appropriately. BellSouth will refer record of mis-matches to NewSouth within one business day of audit. Once reconciled records are received back from NewSouth, BellSouth will update LIDB the same business day if less than 500 records are received before 1:00PM Central Time. If more than 500 records are received, BellSouth will contact NewSouth to negotiate a time frame for the updates, not to exceed three business days.
- 8.2.4.9 BellSouth shall perform backup and recovery of all of NewSouth's data in LIDB including sending to LIDB all changes made since the date of the most recent backup copy, in at least the same time frame BellSouth performs backup and recovery of BellSouth data in LIDB for itself. Currently, BellSouth performs backups of the LIDB for itself on a weekly basis and when a new software release is scheduled, a backup is performed prior to loading the new release.

- 8.2.4.10 BellSouth shall provide NewSouth with LIDB reports of data, which are missing or contain errors, as well as any misrouted errors, within a reasonable time period as negotiated between NewSouth and BellSouth.
- 8.2.4.11 BellSouth shall prevent any access to or use of NewSouth data in LIDB by BellSouth personnel that are outside of established administrative and fraud control personnel, or by any other Party that is not authorized by NewSouth in writing.
- 8.2.4.12 BellSouth shall provide NewSouth performance of the LIDB Data Screening function, which allows a LIDB to completely or partially deny specific query originators access to LIDB data owned by specific data owners, for Customer Data that is part of an NPA-NXX or RAO-0/1XX wholly or partially owned by NewSouth at least at parity with BellSouth Customer Data. BellSouth shall obtain from NewSouth the screening information associated with LIDB Data Screening of NewSouth data in accordance with this requirement. BellSouth currently does not have LIDB Data Screening capabilities. When such capability is available, BellSouth shall offer it to NewSouth under the Bona Fide Request/New Business Process as set forth in General Terms and Conditions.
- 8.2.4.13 BellSouth shall accept queries to LIDB associated with NewSouth customer records, and shall return responses in accordance with industry standards.
- 8.2.4.14 BellSouth shall provide mean processing time at the LIDB within 0.50 seconds under normal conditions as defined in industry standards.
- 8.2.4.15 BellSouth shall provide processing time at the LIDB within 1 second for 99% of all messages under normal conditions as defined in industry standards.
- 8.2.5 <u>Interface Requirements</u>
- 8.2.6 BellSouth shall offer LIDB in accordance with the requirements of this subsection.
- 8.2.6.1 The interface to LIDB shall be in accordance with the technical references contained within.
- 8.2.6.2 The CCS interface to LIDB shall be the standard interface described herein.
- 8.2.6.3 The LIDB Data Base interpretation of the ANSI-TCAP messages shall comply with the technical reference herein. Global Title Translation shall be maintained in the signaling network in order to support signaling network routing to the LIDB.

# 9 Signaling

9.1 All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of Signaling Transport Services.

9.2 BellSouth agrees to offer access to signaling and access to BellSouth's signaling databases subject to compatibility testing and at the rates set forth in this Attachment. BellSouth may provide mediated access to BellSouth signaling systems and databases. Available signaling elements include signaling links, signal transfer points and service control points. Signaling functionality will be available with both A-link and B-link connectivity.

# 9.3 Signaling Link Transport

- 9.3.1 Definition Signaling Link Transport is a set of two or four dedicated 56 Kbps. transmission paths between CLEC-designated Signaling Points of Interconnection (SPOI) that provides appropriate physical diversity.
- 9.3.2 <u>Technical Requirements</u>
- 9.3.2.1 Signaling Link Transport shall consist of full duplex mode 56 kbps transmission paths.
- 9.3.3 Of the various options available, Signaling Link Transport shall perform in the following two ways:
- 9.3.3.1 As an "A-link" which is a connection between a switch or SCP and a home Signaling Transfer Point Switch (STP) pair; and
- 9.3.3.2 As a "B-link" which is a connection between two STP pairs in different company networks (e.g., between two STP pairs for two Competitive Local Exchange Carriers (CLECs)).
- 9.3.4 Signaling Link Transport shall consist of two or more signaling link layers as follows:
- 9.3.4.1 An A-link layer shall consist of two links.
- 9.3.4.2 A B-link layer shall consist of four links.
- 9.3.5 A signaling link layer shall satisfy a performance objective such that:
- 9.3.5.1 There shall be no more than two minutes down time per year for an A-link layer; and
- 9.3.5.2 There shall be negligible (less than 2 seconds) down time per year for a B-link layer.
- 9.3.5.3 A signaling link layer shall satisfy interoffice and intraoffice diversity of facilities and equipment, such that:
- 9.3.5.3.1 No single failure of facilities or equipment causes the failure of both links in an A-link layer (i.e., the links should be provided on a minimum of two separate physical paths end-to-end); and

- 9.3.5.3.2 No two concurrent failures of facilities or equipment shall cause the failure of all four links in a B-link layer (i.e., the links should be provided on a minimum of three separate physical paths end-to-end).
- 9.3.5.4 <u>Interface Requirements</u>
- 9.3.5.4.1 There shall be a DS1 (1.544 Mbps) interface at the NewSouth designated SPOIs. Each 56 kbps transmission path shall appear as a DS0 channel within the DS1 interface.
- 9.4 Signaling Transfer Points (STPs)
- 9.4.1 <u>Definition</u> Signaling Transfer Points is a signaling network function that includes all of the capabilities provided by the signaling transfer point switches (STPs) and their associated signaling links which enable the exchange of SS7 messages among and between switching elements, database elements and signaling transfer point switches.
- 9.4.2 <u>Technical Requirements</u>
- 9.4.2.1 STPs shall provide access to Network Elements connected to BellSouth SS7 network. These include:
- 9.4.2.1.1 BellSouth Local Switching or Tandem Switching;
- 9.4.2.1.2 BellSouth Service Control Points/DataBases;
- 9.4.2.1.3 Third-party local or tandem switching;
- 9.4.2.1.4 Third-party-provided STPs.
- 9.4.2.2 The connectivity provided by STPs shall fully support the functions of all other Network Elements connected to the BellSouth SS7 network. This explicitly includes the use of the BellSouth SS7 network to convey messages which neither originate nor terminate at a signaling end point directly connected to the BellSouth SS7 network (i.e., transient messages). When the BellSouth SS7 network is used to convey transient messages, there shall be no alteration of the Integrated Services Digital Network User Part (ISDNUP) or Transaction Capabilities Application Part (TCAP) user data that constitutes the content of the message.
- 9.4.2.3 If a BellSouth tandem switch routes calling traffic, based on dialed or translated digits, on SS7 trunks between an NewSouth local switch and third party local switch, the BellSouth SS7 network shall convey the TCAP messages that are necessary to provide Call Management features (Automatic Callback, Automatic Recall, and Screening List Editing) between NewSouth local STPs and the STPs that provide connectivity with the third party local switch, even if the third party local switch is not directly connected to BellSouth STPs.

- 9.4.2.4 STPs shall provide all functions of the MTP as defined in the applicable industry standard technical references.
- 9.4.2.5 STPs shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as defined in Telcordia (formerly BellCore) ANSI Interconnection Requirements. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. In cases where the destination signaling point is a NewSouth or third party local or tandem switching system directly connected to BellSouth SS7 network, BellSouth shall perform final GTT of messages to the destination and SCCP Subsystem Management of the destination. In all other cases, BellSouth shall perform intermediate GTT of messages to a gateway pair of STPs in an SS7 network connected with BellSouth SS7 network, and shall not perform SCCP Subsystem Management of the destination. If BellSouth performs final GTT to a NewSouth database, then NewSouth agrees to provide BellSouth with the Destination Point Code for the NewSouth database.
- 9.4.2.6 STPs shall provide on a non-discriminatory basis all functions of the OMAP commonly provided by STPs, as specified in the reference in Section 12.4.5 of this Attachment. All OMAP functions will be on a "where available" basis and can include:
- 9.4.2.6.1 MTP Routing Verification Test (MRVT); and
- 9.4.2.6.2 SCCP Routing Verification Test (SRVT).
- 9.4.2.7 In cases where the destination signaling point is a BellSouth local or tandem switching system or database, or is an NewSouth or third party local or tandem switching system directly connected to the BellSouth SS7 network, STPs shall perform MRVT and SRVT to the destination signaling point. In all other cases, STPs shall perform MRVT and SRVT to a gateway pair of STPs in an SS7 network connected with the BellSouth SS7 network. This requirement shall be superseded by the specifications for Internetwork MRVT and SRVT if and when these become approved ANSI standards and available capabilities of BellSouth STPs, and if mutually agreed upon by NewSouth and BellSouth.
- 9.4.2.8 STPs shall be on parity with BellSouth.
- 9.4.2.9 SS7 Advanced Intelligent Network (AIN) Access
- 9.4.2.9.1 When technically feasible and upon request by NewSouth, SS7 Access shall be made available in association with switching. SS7 AIN Access is the provisioning of AIN 0.1 triggers in an equipped BellSouth local switch and interconnection of the BellSouth SS7 network with the NewSouth SS7 network to exchange TCAP queries and responses with an NewSouth SCP.
- 9.4.2.9.2 SS7 AIN Access shall provide NewSouth SCP access to BellSouth local switch in association with switching via interconnection of BellSouth SS7 and NewSouth SS7

Networks. BellSouth shall offer SS7 access through its STPs. If BellSouth requires a mediation device on any part of its network specific to this form of access, BellSouth must route its messages in the same manner. The interconnection arrangement shall result in the BellSouth local switch recognizing the NewSouth SCP as at least at parity with BellSouth's SCP's in terms of interfaces, performance and capabilities.

- 9.4.3 <u>Interface Requirements</u>
- 9.4.3.1 BellSouth shall provide the following STPs options to connect NewSouth or NewSouth-designated local switching systems or STPs to the BellSouth SS7 network:
- 9.4.3.1.1 An A-link interface from NewSouth local switching systems; and,
- 9.4.3.1.2 A B-link interface from NewSouth local STPs.
- 9.4.3.2 Each type of interface shall be provided by one or more sets (layers) of signaling links.
- 9.4.3.3 The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling for interconnecting NewSouth local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOIs.
- 9.4.3.4 BellSouth CO shall provide intraoffice diversity between the SPOIs and BellSouth STPs, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOIs.
- 9.4.3.5 STPs shall provide all functions of the MTP as defined in the applicable industry standard technical references.
- 9.4.3.6 Message Screening
- 9.4.3.6.1 BellSouth shall set message screening parameters so as to accept valid messages from NewSouth local or tandem switching systems destined to any signaling point within BellSouth's SS7 network where the NewSouth switching system has a legitimate signaling relation.
- 9.4.3.6.2 BellSouth shall set message screening parameters so as to pass valid messages from NewSouth local or tandem switching systems destined to any signaling point or network accessed through BellSouth's SS7 network where the NewSouth switching system has a legitimate signaling relation.

- 9.4.3.6.3 BellSouth shall set message screening parameters so as to accept and pass/send valid messages destined to and from NewSouth from any signaling point or network interconnected through BellSouth's SS7 network where the NewSouth SCP has a legitimate signaling relation.
- 9.4.4 STPs shall be equal to or better than all of the requirements for STPs set forth in the applicable industry standard technical references.

#### 9.5 Service Control Points/Databases

# 9.5.1 <u>Definition</u>

- 9.5.1.1 Databases are the Network Elements that provide the functionality for storage of, access to, and manipulation of information required to offer a particular service and/or capability. Databases include, but are not limited to: Local Number Portability, LIDB, Toll Free Number Database, Automatic Location Identification/Data Management System, Calling Name Database, access to Service Creation Environment and Service Management System (SCE/SMS) application databases and Directory Assistance.
- 9.5.2 A Service Control Point (SCP) is a specific type of Database functionality deployed in a Signaling System 7 (SS7) network that executes service application logic in response to SS7 queries sent to it by a switching system also connected to the SS7 network. Service Management Systems provide operational interfaces to allow for provisioning, administration and maintenance of subscriber data and service application data stored in SCPs.
- 9.5.3 Technical Requirements for SCPs/Databases
- 9.5.3.1 Requirements for SCPs/Databases within this section address storage of information, access to information (e.g. signaling protocols, response times), and administration of information (e.g., provisioning, administration, and maintenance). All SCPs/Databases shall be provided to NewSouth in accordance with the following requirements.
- 9.5.3.2 BellSouth shall provide physical access to SCPs through the SS7 network and protocols with TCAP as the application layer protocol.
- 9.5.3.3 BellSouth shall provide physical interconnection to databases via industry standard interfaces and protocols (e.g. SS7, ISDN and X.25).
- 9.5.3.4 The reliability of interconnection options shall be consistent with requirements for diversity and survivability.
- 9.5.4 <u>Database Availability</u>

- 9.5.4.1 Call processing databases shall have a maximum unscheduled availability of 30 minutes per year. Unavailability due to software and hardware upgrades shall be scheduled during minimal usage periods and only be undertaken upon proper notification to providers, which might be impacted. Any downtime associated with the provision of call processing related databases will impact all service providers, including BellSouth, equally.
- 9.5.4.2 The operational interface provided by BellSouth shall complete Database transactions (i.e., add, modify, delete) for NewSouth customer records stored in BellSouth databases within 3 days, or sooner where BellSouth provisions its own customer records within a shorter interval.

# 9.6 Local Number Portability Database

- 9.6.1 <u>Definition</u>
- 9.6.2 The Permanent Number Portability (PNP) database supplies routing numbers for calls involving numbers that have been ported from one local service provider to another. PNP is currently being worked in industry forums. The results of these forums will dictate the industry direction of PNP. BellSouth agrees to provide access to the PNP database at rates, terms and conditions as set forth by BellSouth and in accordance with an effective FCC or Commission directive.

## 9.7 SS7 Network Interconnection

- 9.7.1 Definition.
- 9.7.2 SS7 Network Interconnection is the interconnection of NewSouth local Signaling Transfer Point Switches (STP) and NewSouth local or tandem switching systems with BellSouth STPs. This interconnection provides connectivity that enables the exchange of SS7 messages among BellSouth switching systems and databases (DBs), NewSouth local or tandem switching systems, and other third-party switching systems directly connected to the BellSouth SS7 network.
- 9.7.3 Technical Requirements
- 9.7.3.1 SS7 Network Interconnection shall provide connectivity to all components of the BellSouth SS7 network. These include:
- 9.7.3.1.1 BellSouth local or tandem switching systems;
- 9.7.3.1.2 BellSouth DBs; and
- 9.7.3.1.3 Other third-party local or tandem switching systems.

- 9.7.4 The connectivity provided by SS7 Network Interconnection shall fully support the functions of BellSouth switching systems and DBs and NewSouth or other third-party switching systems with A-link access to the BellSouth SS7 network.
- 9.7.5 If traffic is routed based on dialed or translated digits between an NewSouth local switching system and a BellSouth or other third-party local switching system, either directly or via a BellSouth tandem switching system, then it is a requirement that the BellSouth SS7 network convey via SS7 Network Interconnection the TCAP messages that are necessary to provide Call Management services (Automatic Callback, Automatic Recall, and Screening List Editing) between the NewSouth local STPs and BellSouth or other third-party local switch.
- 9.7.6 When the capability to route messages based on Intermediate Signaling Network Identifier (ISNI) is generally available on BellSouth STPs, the BellSouth SS7 Network shall also convey TCAP messages using SS7 Network Interconnection in similar circumstances where the BellSouth switch routes traffic based on a Carrier Identification Code (CIC).
- 9.7.7 SS7 Network Interconnection shall provide all functions of the MTP as specified in ANSI T1.111. This includes:
- 9.7.7.1 Signaling Data Link functions, as specified in ANSI T1.111.2;
- 9.7.7.2 Signaling Link functions, as specified in ANSI T1.111.3; and
- 9.7.7.3 Signaling Network Management functions, as specified in ANSI T1.111.4.
- 9.7.8 SS7 Network Interconnection shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as specified in ANSI T1.112. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. Where the destination signaling point is a BellSouth switching system or DB, or is another third-party local or tandem switching system directly connected to the BellSouth SS7 network, SS7 Network Interconnection shall include final GTT of messages to the destination and SCCP Subsystem Management of the destination. Where the destination signaling point is an NewSouth local or tandem switching system, SS7 Network Interconnection shall include intermediate GTT of messages to a gateway pair of NewSouth local STPs, and shall not include SCCP Subsystem Management of the destination.
- 9.7.9 SS7 Network Interconnection shall provide all functions of the Integrated Services Digital Network User Part (ISDNUP), as specified in ANSI T1.113.
- 9.7.10 SS7 Network Interconnection shall provide all functions of the TCAP, as specified in ANSI T1.114.
- 9.7.11 If and when Internetwork MTP Routing Verification Test (MRVT) and SCCP Routing Verification Test (SRVT) become approved ANSI standards and available

- capabilities of BellSouth STPs, SS7 Network Interconnection shall provide these functions of the OMAP.
- 9.7.12 SS7 Network Interconnection shall be equal to or better than the following performance requirements:
- 9.7.12.1 MTP Performance, as specified in ANSI T1.111.6;
- 9.7.12.2 SCCP Performance, as specified in ANSI T1.112.5; and
- 9.7.12.3 ISDNUP Performance, as specified in ANSI T1.113.5.
- 9.7.13 <u>Interface Requirements</u>
- 9.7.13.1 BellSouth shall offer the following SS7 Network Interconnection options to connect NewSouth or NewSouth-designated local or tandem switching systems or STPs to the BellSouth SS7 network:
- 9.7.13.1.1 A-link interface from NewSouth local or tandem switching systems; and
- 9.7.13.1.2 B-link interface from NewSouth STPs.
- 9.7.13.2 The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where the BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling links for interconnecting NewSouth local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOI.
- 9.7.13.3 BellSouth CO shall provide intraoffice diversity between the SPOIs and the BellSouth STP, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and NewSouth will work jointly to establish mutually acceptable SPOI.
- 9.7.13.4 The protocol interface requirements for SS7 Network Interconnection include the MTP, ISDNUP, SCCP, and TCAP. These protocol interfaces shall conform to the applicable industry standard technical references.
- 9.7.13.5 BellSouth shall set message screening parameters to accept messages from NewSouth local or tandem switching systems destined to any signaling point in the BellSouth SS7 network with which the NewSouth switching system has a legitimate signaling relation.

9.7.13.6 SS7 Network Interconnection shall be equal to or better than all of the requirements for SS7 Network Interconnection set forth in the applicable industry standard technical references.

# 10. Operator Call Processing, Inward Operator Services and Directory Assistance Services

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of Operator Call Processing, Inward Operator Services and Directory Assistance Services.

# 10.2 Operator Systems

10.2.1 <u>Definition.</u> Operator Systems is the Network Element that provides operator and automated call handling and billing, special services, end user telephone listings and optional call completion services. The Operator Systems, Network Element provides two types of functions: Operator Service functions and Directory Assistance Service functions, each of which are described in detail below.

# 10.3 Operator Service

10.3.1 <u>Definition</u>. Operator Service provides: (1) operator handling for call completion (for example, collect, third number billing, and manual credit card calls), (2) operator or automated assistance for billing after the end user has dialed the called number (for example, credit card calls); and (3) special services including but not limited to Busy Line Verification and Emergency Line Interrupt (ELI), Emergency Agency Call, Operator-assisted Directory Assistance, and Rate Quotes.

# 10.3.2 Requirements

- 10.3.2.1 When NewSouth requests BellSouth to provide Operator Services, the following requirements apply:
- 10.3.2.1.1 BellSouth shall complete 0+ and 0- dialed local calls.
- 10.3.2.1.2 BellSouth shall complete 0+ intraLATA toll calls.
- 10.3.2.1.3 BellSouth shall process calls that are billed to NewSouth end user's calling card that can be validated by BellSouth.
- 10.3.2.1.4 BellSouth shall complete person-to-person calls.
- 10.3.2.1.5 BellSouth shall complete collect calls.
- 10.3.2.1.6 BellSouth shall provide the capability for callers to bill to a third party and complete such calls.

- 10.3.2.1.7 BellSouth shall complete station-to-station calls.
- 10.3.2.1.8 BellSouth shall process emergency calls.
- 10.3.2.1.9 BellSouth shall process Busy Line Verify and Emergency Line Interrupt requests.
- 10.3.2.1.10 BellSouth shall process emergency call trace, as they do for their End users prior to the Effective Date. Call must originate from a 911 provider.
- 10.3.2.1.11 BellSouth shall process operator-assisted directory assistance calls.
- 10.3.2.1.12 BellSouth shall adhere to equal access requirements, providing NewSouth local end users the same IXC access as provided to BellSouth end users.
- 10.3.2.1.13 BellSouth shall exercise at least the same level of fraud control in providing Operator Service to NewSouth that BellSouth provides for its own operator service.
- 10.3.2.1.14 BellSouth shall perform Billed Number Screening when handling Collect, Personto-Person, and Billed-to-Third-Party calls.
- 10.3.2.1.15 BellSouth shall direct customer account and other similar inquiries to the customer service center designated by NewSouth.
- 10.3.2.1.16 BellSouth shall provide a feed of customer call records in "EMI" format to NewSouth in accordance with CLEC ODUF standards specified in Attachment 7.

# 10.3.3 <u>Interface Requirements</u>

10.3.3.1 With respect to Operator Services for calls that originate on local switching capability provided by or on behalf of NewSouth, the interface requirements shall conform to the then current established system interface specifications for the platform used to provide Operator Service and the interface shall conform to industry standards.

## 10.4 Directory Assistance Service

10.4.1 <u>Definition.</u> Directory Assistance Service provides local end user telephone number listings with the option to complete the call at the callers direction separate and distinct from local switching.

# 10.4.2 Requirements

10.4.3 Directory Assistance Service shall provide up to two listing requests per call. If available and if requested by NewSouth's end user, BellSouth shall provide caller-optional directory assistance call completion service at rates contained in this Attachment to one of the provided listings, equal to that which BellSouth provides its end users. If not available, NewSouth may request such requirement pursuant to the Bona Fide Request/New Business Process as set forth in General Terms and Conditions.

- 10.4.4 Directory Assistance Service Updates
- 10.4.4.1 BellSouth shall update end user listings changes daily. These changes include:
- 10.4.4.1.1 New end user connections: BellSouth will provide service to NewSouth that is equal to the service it provides to itself and its end users;
- 10.4.4.1.2 End user disconnections: BellSouth will provide service to NewSouth that is equal to the service it provides to itself and its end users; and
- 10.4.4.1.3 End user address changes: BellSouth will provide service to NewSouth that is equal to the service it provides to itself and its end users;
- 10.4.4.1.4 These updates shall also be provided for non-listed and non-published numbers for use in emergencies.
- 10.4.5 <u>Branding for Operator Call Processing and Directory Assistance</u>
- 10.4.5.1 The BellSouth Operator Systems Branding Feature provides a definable announcement to NewSouth end users using Directory Assistance (DA)/Operator Call Processing (OCP) prior to placing them in queue or connecting them to an available operator or automated operator system. This feature allows NewSouth to have its calls custom branded with NewSouth's name on whose behalf BellSouth is providing Directory Assistance and/or Operator Call Processing. Rates for Custom Branding, Operator Call Process and Directory Assistance are set forth in this Attachment.
- 10.4.5.2 BellSouth offers four service levels of branding to NewSouth when ordering Directory Assistance and/or Operator Call Processing.
- 10.4.5.2.1 Service Level 1 BellSouth Branding
- 10.4.5.2.2 Service Level 2 Unbranded
- 10.4.5.2.3 Service Level 3 Custom Branding
- 10.4.5.2.4 Service Level 4 Self Branding (applicable only to NewSouth for Resale or use with an Unbundled Port when routing to an operator service provider other than BellSouth).
- 10.4.6 For Resellers and Use with an Unbundled Port
- 10.4.6.1 BellSouth Branding is the Default Service Level.
- 10.4.6.2 Unbranding, Custom Branding, and Self Branding require NewSouth to order selective routing for each originating BellSouth end office identified by NewSouth. Rates for Selective Routing are set forth in this Attachment.

- 10.4.6.3 Customer Branding and Self Branding require NewSouth to order dedicated trunking from each BellSouth end office identified by NewSouth, to either the BellSouth Traffic Operator Position System (TOPS) or NewSouth Operator Service Provider. Rates for trunks are set forth in applicable BellSouth tariffs.
- 10.4.6.4 Unbranding Unbranded Directory Assistance and/or Operator Call Processing calls ride common trunk groups provisioned by BellSouth from those end offices identified by NewSouth to the BellSouth TOPS. These calls are routed to "No Announcement."
- 10.4.7 For Facilities Based Carriers
- 10.4.7.1 All Service Levels require NewSouth to order dedicated trunking from their end office(s) point of interface to the BellSouth TOPS Switches. Rates for trunks are set forth in applicable BellSouth tariffs.
- 10.4.7.2 Customized Branding includes charges for the recording of the branding announcement and the loading of the audio units in each TOPS Switch, IVS and NAV equipment for which NewSouth requires service.
- 10.4.8 Directory Assistance customized branding uses:
- 10.4.8.1 the recording of the name;
- 10.4.8.2 the front-end loading of the Digital Recorded Announcement Machine (DRAM) in each TOPS switch.
- 10.4.9 Operator Call Processing customized branding uses:
- 10.4.9.1 the recording of the name;
- 10.4.9.2 the front-end loading of the DRAM in the TOPS Switch;
- 10.4.9.3 the back-end loading in the audio units in the Automated Alternate Billing System (AABS) in the Interactive Voice Subsystem (IVS);
- 10.4.9.4 the 0- automation loading for the audio units in the Enhanced Billing and Access Service (EBAS) in the Network Applications Vehicle (NAV).
- 10.4.9.5 BellSouth will provide to NewSouth purchasing local BellSouth switching and reselling BellSouth local exchange service, selective routing of calls to a requested directory assistance services platform or operator services platform. NewSouth end users may use the same dialing arrangements as BellSouth end users, but obtain a NewSouth branded service.
- 10.5 Directory Assistance Database Service (DADS)

- 10.5.1 BellSouth shall make its Directory Assistance Database Service (DADS) available solely for the expressed purpose of providing Directory Assistance type services to NewSouth end users. The term "end user" denotes any entity which obtains Directory Assistance type services for its own use from a DADS customer. Directory Assistance type service is defined as Voice Directory Assistance (DA Operator assisted and Electronic Directory Assistance (Data System assisted)). NewSouth agrees that Directory Assistance Database Service (DADS) will not be used for any purpose which violates federal or state laws, statutes, regulatory orders or tariffs. Except for the permitted users, NewSouth agrees not to disclose DADS to others and shall provide due care in providing for the security and confidentiality of DADS. Further, NewSouth authorizes the inclusion of NewSouth Directory Assistance listings in the BellSouth Directory Assistance products.
- BellSouth shall provide NewSouth initially with a base file of subscriber listings which reflect all listing change activity occurring since NewSouth's most recent update via magnetic tape, and subsequently using electronic connectivity such as Network Data Mover to be developed mutually by NewSouth and BellSouth. NewSouth agrees to assume the costs associated with CONNECT: Direct TM connectivity, which will vary depending upon volume and mileage.
- BellSouth will require approximately one month after receiving an order to prepare the Base File. BellSouth will provide daily updates which will reflect all listing change activity occurring since CLEC's most recent update. BellSouth shall provide updates to NewSouth on a Business, Residence, or combined Business and Residence basis. NewSouth agrees that the updates shall be used solely to keep the information current. Delivery of Daily Updates will commence the day after NewSouth receives the Base File.
- 10.5.4 BellSouth is authorized to include NewSouth Directory Assistance Listing Information in its Directory Assistance Database Service (DADS). Any other use by BellSouth of NewSouth Directory Assistance Listing Information is not authorized and with the exception of a request for DADS, BellSouth shall refer any request for such information to NewSouth.
- 10.5.5 Rates for DADS are as set forth in this Attachment.

# 10.6 Direct Access to Directory Assistance Service

- 10.6.1 Direct Access to Directory Assistance Service (DADAS) will provide NewSouth's directory assistance operators with the ability to search all available BellSouth's subscriber listings using the Directory Assistance search format. Subscription to DADAS will allow NewSouth to utilize its own switch, operator workstations and optional audio subsystems.
- 10.6.2 BellSouth will provide DADAS from its DA location. NewSouth will access the DADAS system via a telephone company provided point of availability. NewSouth

- has the responsibility of providing the physical links required to connect to the point of availability. These facilities may be purchased from the telephone company as rates and charges billed separately from the charges associated with this offering.
- A specified interface to each NewSouth subsystem will be provided by BellSouth. Interconnection between NewSouth's system and a specified BellSouth location will be pursuant to the use of NewSouth owned or NewSouth leased facilities and shall be appropriate sized based upon the volume of queries being generated by NewSouth.
- 10.6.4 The specifications for the three interfaces necessary for interconnection are available in the following documents:
- 10.6.4.1 DADAS to Subscriber Operator Position System—Northern Telecom Document CSI-2300-07; Universal Gateway/ Position Message Interface Format Specification;
- 10.6.4.2 DADAS to Subscriber Switch—Northern Telecom Document Q210-1 Version A107; NTDMS/CCIDAS System Application Protocol; and AT&T Document 250-900-535 Operator Services Position System Listing Service and Application Call Processing Data Link Interface Specification;
- 10.6.4.3 DADAS to Audio Subsystem (Optional)—Directory One Call Control to Audio Response Unit system interface specifications are available through Northern Telecom as a licensed access protocol—Northern Telecom Document 355-004424 and Gateway/Interactive Voice subsystem Protocol Specification.
- 10.6.5 Rates for DADAS are as set forth in this Attachment.
- 10.7 Automatic Location Identification/Data Management System (ALI/DMS)
- The ALI/DMS Database contains end user information (including name, address, telephone information, and sometimes special information from the local service provider or end user) used to determine to which Public Safety Answering Point (PSAP) to route the call. The ALI/DMS database is used to provide more routing flexibility for E911 calls than Basic 911. BellSouth shall provide the Emergency Services Database in accordance with the following:
- 10.7.2 <u>Technical Requirements</u>
- 10.7.2.1 BellSouth shall offer NewSouth a data link to the ALI/DMS database or permit NewSouth to provide its own data link to the ALI/DMS database. BellSouth shall provide error reports from the ALI/DMS database to NewSouth immediately after NewSouth inputs information into the ALI/DMS database. Alternately, NewSouth may utilize BellSouth, to enter end user information into the data base on a demand basis, and validate end user information on a demand basis.
- 10.7.2.2 The ALI/DMS database shall contain the following end user information:

- 10.7.2.2.1 Name;
- 10.7.2.2.2 Address;
- 10.7.2.2.3 Telephone number; and
- 10.7.2.2.4 Other information as appropriate (e.g., whether a end user is blind or deaf or has another disability).
- 10.7.2.3 When BellSouth is responsible for administering the ALI/DMS database in its entirety, ported number NXXs entries for the ported numbers should be maintained unless NewSouth requests otherwise and shall be updated if NewSouth requests, provided NewSouth supplies BellSouth with the updates.
- 10.7.2.4 When Remote Call Forwarding (RCF) is used to provide number portability to the local end user and a remark or other appropriate field information is available in the database, the shadow or "forwarded-to" number and an indication that the number is ported shall be added to the customer record.
- 10.7.2.5 If BellSouth is responsible for configuring PSAP features (for cases when the PSAP or BellSouth supports an ISDN interface) it shall ensure that CLASS Automatic Recall (Call Return) is not used to call back to the ported number. Although BellSouth currently does not have ISDN interface, BellSouth agrees to comply with this requirement once ISDN interfaces are in place.
- 10.7.3 Interface Requirements

The interface between the E911 Switch or Tandem and the ALI/DMS database for NewSouth end users shall meet industry standards.

# 11. Calling Name (CNAM) Database Service

- 11.1 The Agreement for Calling Name (CNAM) with standard pricing is included as Exhibit B to this Attachment. NewSouth must provide to its account manager a written request with a requested activation date to activate this service. If NewSouth is interested in requesting CNAM with volume and term pricing, NewSouth must contact its account manager to request a separate CNAM volume and term Agreement. BellSouth provisioning of CNAM shall be compliant with all applicable industry standard technical references.
- SCPs/Databases shall be equal to or better than all of the requirements for SCPs/Databases set forth in the applicable industry standard technical references.
- 11.3 Service Creation Environment and Service Management System (SCE/SMS)
  Advanced Intelligent Network (AIN) Access

- 11.3.1 BellSouth's Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access shall provide NewSouth the capability that will allow NewSouth and other third parties to create service applications in a BellSouth Service Creation Environment and deploy those applications in a BellSouth SMS to a BellSouth SCP. The third party service applications interact with AIN triggers provisioned on a BellSouth SSP.
- BellSouth's SCE/SMS AIN Access shall provide access to SCE hardware, software, testing and technical support (e.g., help desk, system administrator) resources available to NewSouth. Scheduling procedures shall provide NewSouth equivalent priority to these resources.
- BellSouth SCP shall partition and protect NewSouth service logic and data from unauthorized access, execution or other types of compromise.
- When NewSouth selects SCE/SMS AIN Access, BellSouth shall provide training, documentation, and technical support to enable NewSouth to use BellSouth's SCE/SMS AIN Access to create and administer applications. Training, documentation, and technical support will address use of SCE and SMS access and administrative functions, but will not include support for the creation of a specific service application.
- When NewSouth selects SCE/SMS AIN Access, BellSouth shall provide for a secure, controlled access environment in association with its internal use of AIN components. NewSouth access will be provided via remote data connection (e.g., dial-in, ISDN).
- 11.3.5 When NewSouth selects SCE/SMS AIN Access, BellSouth shall allow NewSouth to download data forms and/or tables to BellSouth SCP via BellSouth SMS without intervention from BellSouth (e.g., service customization and end user subscription).

## 12. Basic 911 and E911

- All of the negotiated terms and conditions set forth in this Section pertain to the provision of Basic 911 and E911.
- 12.2 If NewSouth orders network elements and other services, then NewSouth is also responsible for providing E911 to its end users. BellSouth agrees to offer access to the 911/E911 network pursuant to the following terms and conditions set forth in this Attachment.

## 12.3 Definition

Basic 911 and E911 is an additional requirement that provides a caller access to the applicable emergency service bureau by dialing a 3-digit universal telephone number (911).

# 12.5 <u>Requirements</u>

- 12.5.1 <u>Basic 911 Service Provisioning.</u> For Basic 911 service, BellSouth will provide to NewSouth a list consisting of each municipality that subscribes to Basic 911 service. The list will also provide, if known, the E911 conversion date for each municipality and, for network routing purposes, a ten-digit directory number representing the appropriate emergency answering position for each municipality subscribing to 911. NewSouth will be required to arrange to accept 911 calls from its end users in municipalities that subscribe to Basic 911 service and translate the 911 call to the appropriate 10-digit directory number as stated on the list provided by BellSouth. NewSouth will be required to route that call to the appropriate tandem or end office. When a municipality converts to E911 service, NewSouth will be required to discontinue the Basic 911 procedures and being using E911 procedures.
- 12.5.2 E911 Service Provisioning. For E911 service, NewSouth will be required to install a minimum of two dedicated trunks originating from the NewSouth serving wire center and terminating to the appropriate E911 tandem. The dedicated trunks shall be, at a minimum, DS-0 level trunks configured either as a 2-wire analog interface or as part of a digital (1.544 Mb/s) interface. Either configuration shall use CAMA-type signaling with multifrequency ("MF") pulsing that will deliver automatic number identification ("ANI") with the voice portion of the call. If the user interface is digital, MF pulses, as well as other AC signals, shall be encoded per the u-255 Law convention. NewSouth will be required to provide BellSouth or the appropriate designated vendor daily updates to the E911 database. NewSouth will be required to forward 911 calls to the appropriate E911 tandem, along with ANI, based upon the current E911 end office to tandem homing arrangement as provided by BellSouth. If the E911 tandem trunks are not available. NewSouth will be required to route the call to a designated 7digit local number residing in the appropriate Public Service Answering Point ("PSAP"). This call will be transported over BellSouth's interoffice network and will not carry the ANI of the calling party. NewSouth shall be responsible for providing BellSouth with complete and accurate data for submission to the 911/E911 database for the purpose of providing 911/E911 to its end users.
- 12.5.3 <u>Rates.</u> Charges for 911/E911 service are borne by the municipality purchasing the service. BellSouth will impose no charge on NewSouth beyond applicable charges for BellSouth trunking arrangements.
- 12.5.4 Basic 911 and E911 functions provided to NewSouth shall be at least at parity with the support and services that BellSouth provides to its end users for such similar functionality.
- Detailed Practices and Procedures. The detailed practices and procedures contained in the E911 Local Exchange Carrier Guide For Facility-Based Providers as amended from time to time during the term of this Agreement will determine the appropriate practices and procedures for BellSouth and NewSouth to follow in providing 911/E911 services.

12.5.6 NewSouth shall have access to BellSouth's 911/E911 records database so that it can update its end users' 911 records in order to ensure accuracy. BellSouth will generate the disconnect order for the porting telephone number(s) within 36 hours, excluding weekends and published holidays, after the porting has been activated by NewSouth. The completion of the disconnect order will unlock the 911/E911 record to allow NewSouth to facilitate such updates, BellSouth will not be obligated to meet such interval as described in this paragraph as a result of a failure on NewSouth's part. Such failures shall include: (1) NewSouth's failure to port all numbers on the LSR resulting in an incomplete port; (2) instances where NewSouth activates a port without an FOC; (3) instances of NewSouth porting a number without an FOC, where a LSR was submitted to BellSouth and subsequently clarified to NewSouth for inaccurate or incomplete information; (4) NewSouth stops porting activity for less than all numbers required to be ported and subsequently reinstates porting activity at a later time or date; and (5) failure on New South's part to adhere to NPAC or other applicable industry standard processes for porting telephone numbers.

#### 13 Rates

- The prices that NewSouth shall pay to BellSouth for Network Elements and Other Services are as set forth in Exhibit C to this Attachment. It is the intent of the Parties that where applicable state commissions have approved rates for network elements and other services set forth in this Agreement as of the date of the date hereof, such rates have been included in Exhibit C.
- The prices that NewSouth shall pay to BellSouth for Operational Support Systems are as set forth in General Terms and Conditions of this Agreement.

# 14. True-Up

This section applies only to Tennessee and other rates that are interim or expressly subject to true-up under this attachment.

- 14.1 The interim prices for Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:
- 14.2 The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by an effective order of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the

records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions of Section 12 of the General Terms and Conditions.

- 14.3 The Parties may continue to negotiate toward final prices, but in the event that no such Agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in Section 12 of the General Terms and Conditions.
- 14.4 A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
  - (a) BellSouth and NewSouth are entitled to be a full Party to the proceeding;
  - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
  - (c) It shall include as an issue the geographic deaveraging of network element and other services prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

## **EXHIBIT A**

# LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

## I. SCOPE

- A. This Agreement sets forth the terms and conditions pursuant to which BellSouth agrees to store in its LIDB certain information at the request of NewSouth and pursuant to which BellSouth, its LIDB customers and NewSouth shall have access to such information. NewSouth understands that BellSouth provides access to information in its LIDB to various telecommunications service providers pursuant to applicable tariffs and agrees that information stored at the request of NewSouth, pursuant to this Agreement, shall be available to those telecommunications service providers. The terms and conditions contained in the attached Addendum(s) are hereby made a part of this Agreement as if fully incorporated herein.
- B. LIDB is accessed for the following purposes:
  - 1. Billed Number Screening
  - 2. Calling Card Validation
  - 3. Fraud Control
- C. BellSouth will provide seven days per week, 24-hours per day, fraud monitoring on Calling Cards, bill-to-third and collect calls made to numbers in BellSouth's LIDB, provided that such information is included in the LIDB query. BellSouth will establish fraud alert thresholds and will notify NewSouth of fraud alerts so that NewSouth may take action it deems appropriate. NewSouth understands and agrees BellSouth will administer all data stored in the LIDB, including the data provided by NewSouth pursuant to this Agreement, in the same manner as BellSouth's data for BellSouth's end user customers. BellSouth shall not be responsible to NewSouth for any lost revenue which may result from BellSouth's administration of the LIDB pursuant to its established practices and procedures as they exist and as they may be changed by BellSouth in its sole discretion from time to time.

NewSouth understands that BellSouth currently has in effect numerous billing and collection agreements with various interexchange carriers and billing clearing houses. NewSouth further understands that these billing and collection customers of BellSouth query BellSouth's LIDB to determine whether to accept various billing options from end users. Additionally, NewSouth understands that presently BellSouth has no method to differentiate between BellSouth's own billing and line data in the LIDB and such data which it includes in the LIDB on NewSouth's behalf pursuant to this Agreement. Therefore, until such time as BellSouth can and does implement in its LIDB and its supporting systems the means to differentiate NewSouth's data from BellSouth's data and the Parties to this Agreement execute appropriate amendments hereto, the following terms and conditions shall apply:

- (a) NewSouth agrees that it will accept responsibility for telecommunications services billed by BellSouth for its billing and collection customers for NewSouth's end user accounts which are resident in LIDB pursuant to this Agreement. NewSouth authorizes BellSouth to place such charges on NewSouth's bill from BellSouth and agrees that it shall pay all such charges. Charges for which NewSouth hereby takes responsibility include, but are not limited to, collect and third number calls.
- (b) Charges for such services shall appear on a separate BellSouth bill page identified with the name of the entity for which BellSouth is billing the charge.
- (c) NewSouth shall have the responsibility to render a billing statement to its end users for these charges, but NewSouth's obligation to pay BellSouth for the charges billed shall be independent of whether NewSouth is able or not to collect from NewSouth's end users.
- (d) BellSouth shall not become involved in any disputes between NewSouth and the entities for which BellSouth performs billing and collection. BellSouth will not issue adjustments for charges billed on behalf of an entity to NewSouth. It shall be the responsibility of NewSouth and the other entity to negotiate and arrange for any appropriate adjustments.

#### II. TERM

This Agreement will be effective as of \_\_\_\_\_\_, and will continue in effect for one year, and thereafter may be continued until terminated by either Party upon thirty (30) days written notice to the other Party.

## III. FEES FOR SERVICE AND TAXES

- A. NewSouth will not be charged a fee for storage services provided by BellSouth to NewSouth, as described in Section I of this Agreement.
- B. Sales, use and all other taxes (excluding taxes on BellSouth's income) determined by BellSouth or any taxing authority to be due to any federal, state or local taxing jurisdiction with respect to the provision of the service set forth herein will be paid by NewSouth. NewSouth shall have the right to have BellSouth contest with the imposing jurisdiction, at NewSouth's expense, any such taxes that NewSouth deems are improperly levied.

# IV. INDEMNIFICATION

To the extent not prohibited by law, each Party will indemnify the other and hold the other harmless against any loss, cost, claim, injury, or liability relating to or arising out

of negligence or willful misconduct by the indemnifying Party or its agents or contractors in connection with the indemnifying Party's provision of services, provided, however, that any indemnity for any loss, cost, claim, injury or liability arising out of or relating to errors or omissions in the provision of services under this Agreement shall be limited as otherwise specified in this Agreement. The indemnifying Party under this Section agrees to defend any suit brought against the other Party for any such loss, cost, claim, injury or liability. The indemnified Party agrees to notify the other Party promptly, in writing, of any written claims, lawsuits, or demands for which the other Party is responsible under this Section and to cooperate in every reasonable way to facilitate defense or settlement of claims. The indemnifying Party shall not be liable under this Section for settlement by the indemnified Party of any claim, lawsuit, or demand unless the defense of the claim, lawsuit, or demand has been tendered to it in writing and the indemnifying Party has unreasonably failed to assume such defense.

## V. LIMITATION OF LIABILITY

Except in cases of gross negligence, willful or intentional misconduct, neither Party shall be liable to the other Party for any lost profits or revenues or for any indirect, incidental or consequential damages incurred by the other Party arising from this Agreement or the services performed or not performed hereunder, regardless of the cause of such loss or damage.

## VI. MISCELLANEOUS

- A. It is understood and agreed to by the Parties that BellSouth may provide similar services to other companies.
- B. All terms, conditions and operations under this Agreement shall be performed in accordance with, and subject to, all applicable local, state or federal legal and regulatory tariffs, rulings, and other requirements of the federal courts, the U. S. Department of Justice and state and federal regulatory agencies. Nothing in this Agreement shall be construed to cause either Party to violate any such legal or regulatory requirement and either Party's obligation to perform shall be subject to all such requirements.
- C. NewSouth agrees to submit to BellSouth all advertising, sales promotion, press releases, and other publicity matters relating to this Agreement wherein BellSouth's corporate or trade names, logos, trademarks or service marks or those of BellSouth's affiliated companies are mentioned or language from which the connection of said names or trademarks therewith may be inferred or implied; and NewSouth further agrees not to publish or use advertising, sales promotions, press releases, or publicity matters without BellSouth's prior written approval.
- D. This Agreement constitutes the entire Agreement between NewSouth and BellSouth which supersedes all prior Agreements or contracts, oral or written representations,

- statements, negotiations, understandings, proposals and undertakings with respect to the subject matter hereof.
- E. Except as expressly provided in this Agreement, if any part of this Agreement is held or construed to be invalid or unenforceable, the validity of any other Section of this Agreement shall remain in full force and effect to the extent permissible or appropriate in furtherance of the intent of this Agreement.
- F. Neither Party shall be held liable for any delay or failure in performance of any part of this Agreement for any cause beyond its control and without its fault or negligence, such as acts of God, acts of civil or military authority, government regulations, embargoes, epidemics, war, terrorist acts, riots, insurrections, fires, explosions, earthquakes, nuclear accidents, floods, strikes, power blackouts, volcanic action, other major environmental disturbances, unusually severe weather conditions, inability to secure products or services of other persons or transportation facilities, or acts or omissions of transportation common carriers.
- G. This Agreement shall be deemed to be a contract made under the laws of the State of Georgia, and the construction, interpretation and performance of this Agreement and all transactions hereunder shall be governed by the domestic law of such State.

# FACILITIES BASED ADDENDUM TO LINE INFORMATION DATA BASE (LIDB) STORAGE AGREEMENT

This is a Facilities Based Addendum to the Line Information Data Base Storage

Agree	ment dated, between BellSouth
Teleco	ommunications, Inc. ("BellSouth"), and
	("NewSouth"), effective the day of
I.	GENERAL
	This Addendum sets forth the terms and conditions for NewSouth's provision of billing number information to BellSouth for inclusion in BellSouth's LIDB. BellSouth will store in its LIDB the billing number information provided by NewSouth, and BellSouth will provide responses to on-line, call-by-call queries to this information for purposes specified in Section I.B. of the Agreement.
II.	DEFINITIONS
A.	Billing number - a number that NewSouth creates for the purpose of identifying an account liable for charges. This number may be a line or a special billing number.
B.	Line number - a ten digit number that identifies a telephone line administered by NewSouth.
C.	Special billing number - a ten digit number that identifies a billing account established by NewSouth.
D.	Calling Card number - a billing number plus PIN number.
E.	PIN number - a four digit security code assigned by NewSouth which is added to a billing number to compose a fourteen digit calling card number.
F.	Toll billing exception indicator - associated with a billing number to indicate that it is considered invalid for billing of collect calls or third number calls or both, by NewSouth.
G.	Billed Number Screening - refers to the activity of determining whether a toll billing exception indicator is present for a particular billing number.

- H. Calling Card Validation refers to the activity of determining whether a particular calling card number exists as stated or otherwise provided by a caller.
- I. Billing number information information about billing number, Calling Card number and toll billing exception indicator provided to BellSouth by NewSouth.

## III. RESPONSIBILITIES OF PARTIES

- A. NewSouth will provide its billing number information to BellSouth's LIDB each business day by a method that has been mutually agreed upon by both Parties.
- B. BellSouth will store in its LIDB the billing number information provided by NewSouth. Under normal operating conditions, BellSouth shall include NewSouth's billing number information in its LIDB no later than two business days following BellSouth's receipt of such billing number information, provided that BellSouth shall not be held responsible for any delay or failure in performance to the extent such delay or failure is caused by circumstances or conditions beyond BellSouth's reasonable control. BellSouth will store in its LIDB an unlimited volume of NewSouth's working telephone numbers.
- C. BellSouth will provide responses to on-line, call-by-call queries to the stored information for the specific purposes listed in the next paragraph.
- D. BellSouth is authorized to use the billing number information provided by NewSouth to perform the following functions for authorized users on an on-line basis:
  - 1. Validate a 14 digit Calling Card number where the first 10 digits are a line number or special billing number assigned by NewSouth, and where the last four digits (PIN) are a security code assigned by NewSouth.
  - 2. Determine whether NewSouth or the subscriber has identified the billing number as one which should not be billed for collect or third number calls, or both.
- E. NewSouth will provide its own billing number information to BellSouth for storage and to be used for Billed Number Screening and Calling Card Validation. NewSouth will arrange and pay for transport of updates to BellSouth.

# IV. COMPLIANCE

Unless expressly authorized in writing by NewSouth, all billing number information provided pursuant to this Addendum shall be used for no purposes other than those set forth in this Addendum.

# CALLING NAME DELIVERY (CNAM) DATABASE SERVICES

#### 1. **Definitions**

For the purpose of this Attachment, the following terms shall be defined as:

**CALLING NAME DELIVERY DATABASE SERVICE (CNAM)** - The ability to associate a name with the calling party number, allowing the end user subscriber (to which a call is being terminated) to view the calling party's name before the call is answered. This service also provides NewSouth the opportunity to load and store its subscriber names in the BellSouth CNAM SCPs.

**CALLING PARTY NUMBER (CPN)** - The number of the calling party that is delivered to the terminating switch using common channel signaling system 7 (CCS7) technology, and that is contained in the Initial Address Message (IAM) portion of the CCS7 call setup.

**COMMON CHANNEL SIGNALING SYSTEM 7 (CCS7) -** A network signaling technology in which all signaling information between two or more nodes is transmitted over high-speed data links, rather than over voice circuits.

**SERVICE CONTROL POINTs (SCPs)** - The real-time data base systems that contain the names to be provided in response to queries received from CNAM SSPs.

**SERVICE MANAGEMENT SYSTEM (SMS)** - The main operations support system of CNAM DATABASE SERVICE. CNAM records are loaded into the SMS, which in turn downloads into the CNAM SCP.

**SERVICE SWITCHING POINTs (SSPs) -** Features of computerized switches in the telephone network that determine that a terminating line has subscribed to CNAM service, and then communicate with CNAM SCPs in order to provide the name associated with the calling party number.

**SUBSYSTEM NUMBER (SSN)** - The address used in the Signaling Connection Control Part (SCCP) layer of the SS7 protocol to designate an application at an end signaling point. A SSN for CNAM at the end office designates the CNAM application within the end office. BellSouth uses the CNAM SSN of 232.

## 2. Attachment

2.1 This Attachment contains the terms and conditions where BellSouth will provide to the NewSouth access to the BellSouth CNAM SCP for query or record storage purposes.

2.2 NewSouth shall submit to BellSouth a notice of its intent to access and utilize BellSouth CNAM Database Services pursuant to the terms and conditions of this Attachment. Said notice shall be in writing, no less than 60 days prior to NewSouth's access to BellSouth's CNAM Database Services and shall be addressed to NewSouth's Account Manager.

### 3. Physical Connection and Compensation

- 3.1 BellSouth's provision of CNAM Database Services to NewSouth requires interconnection from NewSouth to BellSouth CNAM Service Control Points (SCPs). Such interconnections shall be established pursuant to Attachment 3 of this Agreement . The appropriate charge for access to and use of the BellSouth CNAM Database service shall be as set forth in this Attachment.
- 3.2 In order to formulate a CNAM query to be sent to the BellSouth CNAM SCP, NewSouth shall provide its own CNAM SSP. NewSouth's CNAM SSPs must be compliant with TR-NWT-001188, "CLASS Calling Name Delivery Generic Requirements".
- 3.3 If NewSouth elects to access the BellSouth CNAM SCP via a third party CCS7 transport provider, the third party CCS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia (formerly BellCore)'s CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points (LSTPs) serving the BellSouth CNAM SCPs that NewSouth desires to query.

### 3.4 Out-Of-Region Customers

If the customer queries the BellSouth CNAM SCP via a third party national SS7 transport provider, the third party SS7 provider shall interconnect with the BellSouth CCS7 network according to BellSouth's Common Channel Signaling Interconnection Guidelines and Telcordia's (formerly BellCore's) CCS Network Interface Specification document, TR-TSV-000905. In addition, the third party provider shall establish SS7 interconnection at one or more of the BellSouth Gateway Signal Transfer Points (STPs). The payment of all costs associated with the transport of SS7 signals via a third party will be established by mutual agreement of the Parties in writing and shall, by this reference become an integral part of this Agreement.

### 4. CNAM Record Initial Load and Updates

4.1 The mechanism to be used by NewSouth for initial CNAM record load and/or updates shall be determined by mutual agreement. The initial load and all updates shall be provided by NewSouth in the BellSouth specified format and shall contain records for every working telephone number that can originate phone calls. It is the responsibility of NewSouth to provide accurate information to BellSouth on a current basis.

- 4.2 Updates to the SMS shall occur no less than once a week, reflect service order activity affecting either name or telephone number, and involve only record additions, deletions or changes.
- 4.3 NewSouth CNAM records provided for storage in the BellSouth CNAM SCP shall be available, on a SCP query basis only, to all Parties querying the BellSouth CNAM SCP. Further, CNAM service shall be provided by each Party consistent with state and/or federal regulation.

					R.	RATES BY STATE	ΤE			
DESCRIPTION	USOC	A	2	GA	শ	F	SW	NC	SC	ĭ
NID to NID Cross Connect, 2-Wire or 4-Wire, NRC	UNDC2	\$11.63	\$6.15	¥.	\$11.79	\$11.72	NA.	\$11.68	₹	¥
NID to NID Cross Connect, 2-Wire or 4-Wire, NRC	UNDC4	\$11.63	\$6.15	NA.	\$11.79	\$11.72	NA.	\$11.68	NA.	¥
NID, 1-z lines, per monun	UND12	TBD ₩	\$94.50	TBD	\$94.56	\$93.90	TBD	TBD	TBD	TBD №
NRC - Add'l	UND12	TBD	\$57.22	TBD	\$57.28	\$56.67	TBD	TBD	TBD	TBD
NRC - Disconnect Charge - 1st	UND12	# B	N X	TBD	N N	× ×	TBD	TBD	TBD	룡룡
NRC - Disconnect Charge - Add'i	SOMEC SOMEC	\$3.50	\$2 75 NA	NA IBD	\$3 50 NA	\$3.50	NA	\$3 ⊼0	NA TBD	<b>B</b>
NRC - Service Order submitted Electronically, per LSR - Disconnect	SOMEC	NA 30	\$0.42	₹ ₹	NA 00	NA S	<b>₹</b>	NA S	₹ ₹	围
NRC - Service Order submitted Manually, per LSR	SOMAN	¥ :	\$21.56	¥ :	\$29.24	NA :	¥.	¥.	<b>⊼</b> :	авт
NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	¥	\$3.84	Α	\$3.94	NA	A	¥	¥	TBD
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	행명	× ×	TBD	X X	×	TBD	TBD	TBD	룡
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN		X §	TBD	N S	NA S	TBD	TBD	TBD	d de
NID. 1-6 lines, per month	UND16	¥ į	X ;	N :	N :	NA :	¥ į	¥ į	¥ į	ĕį
1 1	UND16	TBD	\$136.75	TBD	\$136.91	\$135.29	TBD	TBD	TBD	TBD :
NRC - Add'l	UND16	TBD	\$99.47	TBD	\$99.63	\$98.07	TBD	TBD	TBD	ТВО
NRC - Disconnect Charge - 1st	UND16	TBD	NA NA	TBD	<b>X</b>	NA NA	TBD	TBD	TBD	國
NRC - Service Order submitted Electronically per LSR	SOMEC	\$3.50	\$2.75	N -	\$3.50	\$3 50	NA C	\$3.50	N -	d 5
NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA S	\$0.42	NA :	NA	NA	NA :	NA S	NA :	TBD
	SOMAN	¥	\$21.56	¥	\$29.24	NA NA	¥	¥	¥	TBD
NRC - Service Order submitted Manually, per LSR, Disconnect	SOMAN	# ₹	\$3.84	5 ₹	\$3.94	\$ ₹	T N	7 ₹	# ₹	3 8
NRC - Incremental Charge - Manual Service Order - Ist	SOMAN	TBD E	AN Oc. 12¢	TBD	<b>₹</b>	X §	TBD	TBD	TBD	큠
NRC - Incremental Charge - Manual Service Order - Disconnect	SOMAN	TBD	\$3.84	TBD	AN	NA	TBD	TBD	TBD	TBD
Nonrecurring Charge - customer transfer, feature additions, changes (1)		\$5.00	¥	¥	N.	NA.	\$5.00	⋠	₹	₹
LOOP, EXCLUDING NID										
2-Wire Analog VG Loop (Standard), per month		Z Z	N N	Z Z	\$18.20	NA NA	NA NA	NA NA	X X	N N
NRC - Add'l		<b>₹</b>	<b>₹</b>	<b>₹</b>	\$58.57	¥ \$	<b>₹</b>	₹ ₹	₹ ₹	<b>₹</b>
2-Wire Analog VG Loop (Customized), per month		NA	NA	NA	\$21.41	NA	NA	NA	NA	NA
NRC - 1st		¥	¥	¥	\$236.75	NA NA	¥	¥	¥	¥
1.Wire Analog VG I con (Standard) per month		8 8	N X	Z Z	\$26.38	NA NA	N N	NA KA	N K	N X
NRC - 1st		¥ ₹	<b>¥</b> §	¥ 3	\$457.14	¥ 3	¥ §	<b>₹</b>	¥ ₹	<b>₹</b>
NRC - Add'I		NA	NA AN	NA	\$348.83	NA	AN	NA	NA	NA
2-Wire ISDN Digital Grade Loop (Standard), per month		X	N N	X X	\$29.65	NA	NA NA	N A	\$ \$	8
NRC - Add'I		₹ ₹	<b>₹</b> 5	₹ ₹	\$431.61	<b>₹</b> 5	<b>₹</b> 5	₹ 5	<b>₹</b> §	₹ 5
2-Wire ADSL Loop (Standard), per month		NA	NA AN	NA	\$10.63	NA	AN	NA	NA	NA
NRC - 1st		N X	N X	N X	\$713.50	N N	N A	N X	N X	8
2-Wire HDSL Loop (Standard), per month		¥ :	¥ :	¥ ;	\$7.40	NA S	¥.	¥.	<b>₹</b>	₹ :
NRC - 1st		NA	AN	NA	\$713.50	NA	AN	AN	NA	A
NRC - Add'I		<b>×</b> ×	× ×	× ×	\$609.44	<b>X X</b>	× ×	<b>×</b> ¥	<b>×</b> ×	<b>₹</b>
NRC - 1st		₹ ₹	<b>₹</b> §	₹ ₹	\$748.93	<b>₹</b> §	<b>₹</b> §	₹ ₹	₹ ₹	<b>₹</b> §
NRC - Add'I		NA	NA AN	NA	\$646.17	NA	AN	NA	NA	NA
LOOP, INCLUDING ND										
Z-Wire Analog VG Loop-SL1	I IF AI 2	N	NA	NA	N	NA	NA	\$15.88	NA	NA
RC - Zone 1, per month (Note 2)	UEAL2	\$15.24	\$13.75	\$14.21	\$14.79	\$14.96	\$15.58	TBD	\$18.48	\$15.92
RC - Zone 2, per month (Note 2)	UEAL2	\$24.75	\$20.13	\$16.41	\$27.68	\$25.69	\$20.65	TBD	\$27.87	\$20.79
RC - Zone 3, per month (Note 2)	UEAL2	\$44.85	\$44.40	\$26.08	\$47.78	\$52.47	\$29.51	TBD	\$36.91	\$27.18
NRC - 1st	UEAL2	\$59.03	\$83.20	\$42.54	\$86.08	\$40.69	\$59.25	\$57.99	\$70.44	\$78.93
NRC - Add'l	UEAL2	\$43.14	\$35.12	\$31.33	\$58.57	\$29.96	\$43.67	\$42.37	\$44.05	\$50.98
NRC - Disconnect Charge - 1st	UEAL2	\$15.21	\$55.97	<b>₹</b>	<b>X</b>	\$11.48	\$16.35	X X	₹ ₹	₹ ₹
NRC - Service Order submitted Electronically, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
NRC - Setvice Order submitted Electronically, per LSR - Disconnect	SOMEC	NA	\$0.42	NA	AN	NA	AN	NA	NA	TBD
NRC - Service Order submitted Manually, per LSR	SOMAN	× ×	\$21.56	X X	\$29.24	S A	N X	×	<b>8</b>	\$19.99
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA OF	\$18.94	N.94	\$18.14	\$25.52	\$26.94	\$44.22	<b>₹</b>
THE HOLDING CHAIGO MAINGA COLLACT TOL	00000	.0.	101	6.0.0		6.0	10:01	0.01	11:11	

NKC - Disconnect Charge - Add i	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	RC - Zone 3, per month (Note 2)	RC - Zone 2, per month (Note 2)		RC - Statewide, per month	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR Disconnect			NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'I	RC - Zone 4, per month (Note 2)	RC - Zone 3, per month (Note 2)	RC - Zone 2, per month (Note 2)	RC - Zone 1, per month (Note 2)	-Wire Analog VG Loop	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Addil	NRC - Addit	NRC - 1st	RC - Zone 4, per month (Note 2)	RC - Zone 3, per month (Note 2)	RC - Zone 1, per month (Note 2)	RC - Statewide, per month	-Wire Analog VG Loop-SL2 w reverse battery signaling	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - 1st	RC - Zone 4, per month (Note 2)	RC - Zone 3, per month (Note 2)	RC - Zone 1, per month (Note 2)	RC - Statewide, per month	-Wire Analog VG Loop-SL2 wloop or ground start signaling	NRC - Inon Make-I In	NRC - Incremental Charge - Manual Service Order - Add'l	ESCRIPTION	
01127	U1L2X	U1L2X	U1L2X	U1L2X	U1L2X	U1L2X	U1L2X	ocost	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UEAL4	UEAL4		UEAL4	UEAL4	UEAL4	UEAL4	- in ∧ .	OCOCL	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UFAR2	UEAR2	UEAR2	UEAR2	UEAR2	UEAR2	UEAR2		OCOSL	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UEAL2	UEAL 2	UEAL2	UEAL2	UEAL2	UEAL2	UEAL2	OF TAKE	SOMAN	SOMAN	USOC	
\$57.01	\$108.95	\$255.87	\$331.85	\$68.38	\$37.74	\$23.23	NA	\$45.99	\$17.77	\$12.97	\$27.37	N N	<b>X X</b>	\$3.50	\$57.01	\$108.96	\$241.76	NA NA	\$70.67	\$39.00	\$24.01	20	\$45.99	\$17.77	\$12.97	\$27.37 AN	NA NA	NA	\$3.50	\$26.01	\$108.40	\$145.46	NA	\$52.84	\$17.95	NA	4	\$45.99	\$12.97 \$17.77	\$27.37	<b>⊼</b> ₹	NA NA	\$3.50	\$26.01	\$4031	\$145.46	NA	\$52.84	\$17.95	NA		\$17.77 TBD	\$12.97	AL	
\$10.ZO	\$111.10	\$283.00	\$306.00	\$104.47	\$47.35	\$32.34	NA	\$36.18	¥.	NA S	NA S	\$2 F.2	\$0.42	\$2.75	\$27.42	\$122.15	\$43.00	NA NA	\$78.35	\$35.51	\$24.26	N	\$36.18	¥	¥ §	\$3.84	\$21.56	\$0.42	\$2.75	\$20.58	\$136.44	\$218.96	NA	\$27.87	\$18.48	NA.	************	\$36.18	N N	NA	\$3.84	\$0.42	\$2.75	\$20.58	\$135.44	\$218.96	NA	\$27.87	\$18.48	NA.	ē	∄ ⊼	<b>X</b>	핃	
5	NA NA	\$180.35	\$233.38	\$40.17	\$25.27	\$21.89	NA	\$34.22	NA	\$8.42	\$18.94	N N	N A	\$3.50	NA	NA	\$170.57	NA	\$40.85	\$25.70	\$22.26	NIA	\$34.22	NA	\$8.42	\$18 94	NA	NA	\$3.50	NA 5	\$/8.10	\$104.17	NA	\$30.92	\$16.84	NA	40	\$34.22	\$8.42	\$18.94	¥ ₹	NA	\$3.50	N S	\$/8.10	\$104.17	NA	\$30.92	\$16.84	NA	-	TB)	\$8.42	GA	
5	NA NA	\$431.61	\$541.28	\$76.42	\$44.28	\$23.66	NA	\$36.18	¥	NA S	NA.ST	\$3.04	NA NA	\$3.50	NA	NA	\$348.83	NA	\$67.56	\$39.14	\$20.92	NIA	\$36.18	NA.	<b>₹</b> 5	\$3.94	\$29.24	NA	\$3.50	NA 5	\$1//.10	\$236.75	NA	\$55.78	\$17.27	NA	400110	\$36.18	¥	NA	\$3.94	NA Page 24	\$3.50	N S	\$1//:10	\$236.75	NA	\$55.78	\$17.27	NA	- 00	TBD	NA NA		₹.
ф39. <del>44</del>	\$74.27	\$172.63	\$223.27	\$74.19	\$36.22	\$21.15	NA	\$32.77	\$11.41	\$8.06	\$18.14	N N	<b>8</b>	\$3.50	\$39.44	\$74.27	\$163.26	NA	\$85.47	\$41.85	\$24.36	N	\$32.77	\$11.41	\$8.06	\$18 14	¥	NA	\$3.50	\$18.87	\$74.73	\$99.69	NA	\$61.93	\$17.65	NA.	******	\$32.77	\$8.06	\$18.14	<b>₹</b> §	NA NA	\$3.50	\$18.87	\$28.73	\$99.69	NA	\$61.93	\$17.65	NA	ē	\$11.41 TBD	\$8.06	F	RATES BY STATE
\$37.Z	\$108.14	\$252.00	\$326.38	\$41.40	\$28.97	\$21.86	NA	\$45.27	\$16.06	\$11.34	\$25.52	Z Z	<b>X X</b>	\$3.50	\$57.28	\$108.14	\$238 19	\$55.96	\$42.40	\$29.67	\$22.38	25	\$45.27	\$26.95	\$11.34	\$25.52 NA	¥	NA	\$3.50	\$26.95	\$107.70	\$144.01	\$45.88	\$34.77	\$18.35	NA	4 .0.00	\$45.27	\$11.34	\$25.52	<b>₹</b> ₹	<b>₹</b>	\$3.50	\$26.95	\$40 98	\$144.01	\$45.88	\$34.77	\$18.35	NA	-	\$16.06 TBD	\$11.34	MS	₩
5	X X	\$251.31	\$325.91	TBD	TBD	TBD	\$24.98	\$45.34	¥	\$12.76	\$26.94	Z Z	<b>₹</b>	\$3.50	NA	A	\$237.45	NA A	TBD	TBD	TBD	627 40	\$45.34	¥	\$12.76	\$26 94 AN	¥	NA	\$3.50	N 5	\$106.56	\$142.97	NA	TBO	T TB	\$19.50	4 : 0 : 0	\$45.34	\$12.76	\$26.94	<b>₹</b> 5	X X	\$3.50	≰ 5	\$106.56	\$142.97	NA	TB G	TBD	\$19.50	ē	∄ ₹	\$12.76	R	
N.	X X	\$301.75	\$423.04	\$53.29	\$40.24	\$26.68	NA	\$45.43	¥	\$13.55	\$44.06	Z Z	<b>8</b>	\$3.50	NA	NA :	\$286.77	NA S	\$58.85	\$44.44	\$29.47	N	\$45.43	¥	\$13.55	\$44 42	₹	NA	\$3.50	X 5	\$128.80	\$178.12	NA	\$43.08	\$21.57	NA		\$45.43	\$13.55	\$44.42	<b>₹</b> 5	X X	\$3.50	₹ 5	\$128.80	\$178.12	NA.	\$43.08	\$21.57	¥	ē	₹ 5	\$13.55	SC	
5	X X	\$31.00	\$58.50	\$27.18	\$20.79	\$15.92	NA	\$55.00	₩	NA S	N G	#19.99 TBD	TBD	\$3.50	NA	NA	\$31.00	S NA	\$27.18	\$20.79	\$15.92	N/S	\$55.00	¥	X 5	NA	\$19.99	TBD	\$3.50	8 3	\$140.72	\$192.97	NA	\$27.18	\$15.92	¥		\$55.00	××	NA	TBD	TBD	\$3.50	¥ 5	\$140.72	\$192.97	NA.	\$27.18	\$15.92	A	ē	∄ ₹	<b>X</b>	¥	

=										State	4-Win															A-Wir												State	2-Win										DESC	
NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Electronically, per LSR - Disconnect		NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	Statewide, per month  Zone 1 per month	4-Wire HDSL Compatible Loop, without Man Svc Inquiry & Fac Res	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Add'l	NRC - 1st	RC - Zone 4, per month (Note 2)	RC - Zone 3 per month (Note 2)	RC - Zone 1, per month (Note 2)	RC - Statewide, per month	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Disconnect Charge - Add'i	NRC - Disconnect Charge - 1st	NRC - Add'l	Zone 4, per month NRC - 1st	Zone 3, per month	Zone 2, per month	Zone 1 per month	2-Wire HDSL Compatible Loop, without Man Svc Inquiry & Fac Reserv	NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Addit	NRC - 1st	DESCRIPTION	
SOMAN	SOMAN	SOMEC	SOMEC	UHL4W	UHL4W	UHL4W	UHL4W	UHL4W	UHL4W	UHL4W		OCOSL	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UHL4X	UHL4X	UHL4X	UHL4X	UHL4X	UHL4X	UHL4X	OCOSL	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UHL2W	UHL2W	UHL2W	UHL2W	UHL2W	WCIHI I		OCOSL	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UHL2X	UHL2X	UHL2X	USOC	
\$27.37	¥ ₹	N A	\$3.50	\$56.98	\$106.65	\$352.50	NA NA	\$33.90	\$18.71	\$11.52	;	\$45.99	\$17.77	\$12.37	NA NA	NA	<b>∀</b>	\$3.50	\$56.98	\$491.50	\$541.13	NA	\$33.90	\$11.52	NA	\$45.99	\$17.77	\$12.97	\$27.37 AN	¥	NA S	\$56.98	\$106.65	\$325.58	\$375.21	\$27.70	\$15.29	\$0.41	;	\$45.99	\$12.97	\$27.37	<b>₹</b> §	<b>8</b>	\$3.50	\$56.98	\$106.65	\$514.21	P	
¥	\$3.84	\$0.42	\$2.75	\$19.58	\$114.30	\$250.01	NA S	\$47.64	\$21.59	\$14.75		\$36.18	<b>₹</b>	N NA	\$3.84	\$21.56	\$0.42	\$2.75	\$26.10	\$101.71	\$116.91	NA	\$47.64	\$14.75	NA	\$36.18	AN	N S	\$3.84	\$21.56	\$0.42	\$15.46	\$108.29	\$192.81	\$276.19	\$31.65	\$14.35	\$0.80		\$36.18	X X	NA	\$3.84	\$0.42	\$2.75	\$35.23	\$154.23	\$113.85	핃	
\$18.94	<b>₹</b> 5	N N	\$3.50	NA	NA	\$205.28	8	\$19.07	\$12.00	\$10.30		\$34.22	NA	\$18.94	Ž K	NA	¥	\$3.50	X X	\$344.28	\$378.86	NA	\$19.07	\$10.39	NA	\$34.22	NA AN	\$8.42	\$18 Q4	¥	NA S	\$3.50 NA	¥	\$186.15	\$20073 AN	\$14.48	\$9.09	\$7 88		\$34.22	\$8.42	\$18.94	<b>₹</b> 5	<b>₹</b>	\$3.50	¥.	\$325.15 NA	\$359.73	GA	
₹	\$3.94	\$20 2A	\$3.50	\$20.64	\$122.17	\$192.02	NA NA	\$24.82	\$14.38	\$7 68 NA		\$36.18	<b>₹</b> §	K X	\$3.94	\$29.24	Z	\$3.50	X §	\$646.17	\$748.93	NA	\$14.38	\$7.68	NA	\$36.18	NA.	<b>₹</b>	\$3.94	\$29.24	NA S	\$15.88	\$112.86	\$193.54	\$278 28	\$20.33	\$11.78	\$6.20		\$36.18	<b>₹</b>	W	\$3.94	\$30 NA	\$3.50	X S	\$609.44	\$713.50		_
\$18.14	<b>₹</b>	N N	\$3.50	\$20.98	\$124.19	\$245.15	N N	\$44.44	\$21.76	\$12 Q7		\$32.77	\$11.41	\$18.14	ž ×	NA.	<b>⊼</b>	\$3.50	\$39.42	\$328.35	\$361.45	NA :	\$44.44	\$12.97	¥	\$32.77	\$11.41	\$8.06	\$18 14	¥	NA S	\$15.81	\$112.32	\$188.02	\$272 64	\$31.48	\$15.41	\$8 Q7		\$32.77	\$8.06	\$18.14	¥ ₹	<b>X X</b>	\$3.50	\$39.42	\$310.03	\$343.13	FA :	RATES BY STA
\$25.52	¥ §	N N	\$3.50	\$57.25	\$105.86	\$392.21	\$25.90	\$19.62	\$13.73	\$10.36		\$45.27	\$16.06	\$25.52	AN AN	NA	Z F	\$3.50	\$57.25	\$482.63	\$531.21	\$25.90	\$19.62	\$10.36	NA	\$45.27	\$16.06	\$11.34	\$25.52 AN	NA	NA S	\$3.50	\$105.86	\$317.24	\$21.25	\$16.10	\$11.26	\$8 50 NA		\$45.27	\$11.34	\$25.52	<b>₹</b>	× ×	\$3.50	\$57.25	\$105.86	\$504.82		Ħ
\$26.94	<b>₹</b>	N N	\$3.50	ΝĄ	NA	\$343.62	NA NA	TBD	TBD 6	\$13.97		\$45.34	NA S	\$26.94	S NA	NA.	×.	\$3.50	N S	\$482.62	\$531.35	NA .	<b>H E</b>	TBD	\$13.97	\$45.34	NA	\$12.76	\$26 QA	N <sub></sub>	NA S	\$3.50	\	\$317.17	\$365 an	TBD	TBD	\$11.98		\$45.34	\$12.76	\$26.94	<b>₹</b> 5	× ×	\$3.50	¥ ;	\$456.17	\$504.90	NC	
\$44.06	¥ §	N N	\$3.50	NA	NA	\$393.78	NA.	\$32.38	\$24.45	\$16.21		\$45.43	NA C	\$13.55	NA NA	NA.	×.	\$3.50	N S	\$532.78	\$625.11	NA	\$3238	\$16.21	N	\$45.43	NA	\$13.55	\$44.06	NA.	NA S	\$3.50	NA.	\$368.33	\$461 60	\$24.39	\$18.41	\$12.21		\$45.43	\$13.55	\$44.06	¥ §	8	\$3.50	X S	\$507.33	\$600.61	SC	
¥	TBD	\$19 99	\$3.50	N <sub>A</sub>	NA	\$429.86	A	\$27.88	\$19.46	\$17.91		\$55.00	<b>₹</b> 5	N N	TBD	\$19.99	TBD	\$3.50	X X	\$568.86	\$666.70	NA S	\$27.88	\$15.46	\$17.91	\$55.00	NA AN	N S	N IBU	\$19.99	TBD	\$3 50 NA	¥	\$402.94	\$501 79	\$20.96	\$14.62	\$11 63		\$55.00	X X	NA	TBD	1BD	\$3.50	X S	\$541.94 NA	\$640.79	ī	

2-Wire Unb Copper Loop/Short(< or = 18kft), incl Man Sl & Fac Res\*
RC - Statewide, per month

NRC - Disconnect Charge - 1st

NRC - Disconnect Charge - Add'

NRC - Service Order submitted Electronically, per LSR - Disconnect

NRC - Service Order submitted Electronically, per LSR - Disconnect

NRC - Service Order submitted Manually, per LSR

NRC - Service Order submitted Manually, per LSR

NRC - Incremental Charge - Manual Service Order - 1st

NRC - Incremental Charge - Manual Service Order - Add' |

NRC - Incremental Charge - Manual Service Order - Service Order - Incremental Charge - Manual Service Order - Insconnect |

NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)

UDL64
UDL64
SOMEC
SOMEC
SOMAN
SOMAN
SOMAN

\$129.62 \$64.25 \$3.50 NA NA NA S27.37 \$12.97 \$17.77 \$45.99

\$122.15 \$27.42 \$2.75 \$0.42 \$21.56 \$3.84 NA NA NA NA

\$3.50 NA NA NA NA NA \$18.94

\$127.74 \$27.90 \$3.50 NA \$29.24 \$3.94 NA

\$87.99 \$44.24 \$3.50 NA NA NA NA \$18.14 \$8.06 \$11.41

\$128.36 \$64.35 \$3.50 NA NA NA NA \$25.52 \$11.34 \$16.06 \$45.27

\$44.06 \$13.55 \$3.50 NA NA NA

NA \$3.50 NA \$19.99 NA NA NA NA NA NA NA

X X

\$3.50 NA NA NA NA \$26.94 \$12.76 NA \$45.34

NA NA \$45.43

SOMAN

NA \$34.22

NA \$36.18

₹

₹

¥

¥

₹

₹

₹

4-Wire 64 Kbps Dig Grade Loop

RC - Statewide, per month

NRC - Service Order submitted Manually, per LSR, Disconnect
NRC - Service Order submitted Manually, per LSR, Disconnect
NRC - Incremental Charge - Manual Service Order - 1st
NRC - Incremental Charge - Manual Service Order - Add'l
NRC - Incremental Charge - Manual Service Order - Disconnect
NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)

SOMAN SOMAN SOMAN OCOSL

NA NA \$27.37 \$12.97 \$17.77

\$3.84 NA NA NA S36.18

\$8.42 NA NA NA \$18.94

\$29.24 \$3.94 NA

NA NA \$18.14 \$8.06 \$11.41 \$32.77

NA NA \$25.52 \$11.34 \$16.06 \$45.27

NA NA \$26.94 \$12.76 NA \$45.34

\$45.43

₹ ₹ ¥ ¥ ¥

TBD NA NA NA \$55.00

\$34.22

\$36.18

\$45.99

RC - Statewide, per month
RC - Zone 1, per month (Note 2)
RC - Zone 3, per month (Note 2)
RC - Zone 3, per month (Note 2)
RC - Zone 4, per month (Note 2)
NRC - State 4, per month (Note 2)
NRC - Add1

UDL64 UDL64 UDL64 UDL64 UDL64 UDL64 UDL64

NA \$27.33 \$44.40 \$80.45

\$498.05 \$343.70

₹

NA \$39.08 \$57.21 \$126.22 NA \$654.72 \$428.45

NA \$25.75 \$29.74 \$47.27 NA \$348.55 \$241.20

NA \$35.92 \$40.32 \$37.90 NA \$258.62 \$176.11

NA \$27.50 \$47.24 \$96.48

\$25.61 \$33.94 \$48.51 \$64.02 \$489.00 \$337.93

\$32.67 TBD TBD TBD TBD NA \$489.04 \$337.51

\$41.70 \$34.26 \$51.67 \$68.43 NA \$602.73 \$393.50

\$42.23 \$36.45 \$45.87 \$65.75 NA \$643.00 \$421.26

\$333.28 \$230.50

DESCRIPTION

| INRC - Incremental Charge - Manual Service Order - Add'|
| INRC - Incremental Charge - Manual Service Order - Disconnect
| INRC - Incremental Charge - Order Coordination - Time Specific (per LSR)
| A-Wire DS1 Digital Loop
| RC - Starwide, per month (Note 2)

SOMAN

\$45.99 \$12.97 \$17.77

NA NA \$36.18

\$8.42 NA \$34.22

₹ ₹ **₹** 

\$11.34 \$16.06 \$45.27

\$12.76 NA \$45.34

≰ ₹ **≵** 

Attachment 2 Exhibit C Rates - Page 5

\$45.43 \$13.55 NA

SOMAN

USLXX

\$51.74 ¥

NA \$64.69

NA \$55.53

NA \$50.26 \$36.18

\$56.32 ¥

NA \$50.99

\$62.78 TBD

NA \$59.61

NA \$57.73 \$55.00

NRC - Incremental Charge - Order Coordination - Time Specific (per LSR)
4-Wire 56 Kbps Dig Grade Loop
RC - Statewide, per month

OCOSL

\$49.18

\$36.18

\$34.52

\$36.18

\$33.05

\$48.17

\$45.34

\$48.47

\$55.00

¥

¥

¥

¥

¥

¥

¥

NRC - Service Order submitted Electronically, per LSR
NRC - Service Order submitted Electronically, per LSR - Disconnect
NRC - Service Order submitted Menually, per LSR, Disconnect
NRC - Service Order submitted Menually, per LSR, Disconnect
NRC - Service Order submitted Menually, per LSR, Disconnect
NRC - Incremental Charge - Manual Service Order - Add'l
NRC - Incremental Charge - Manual Service Order - Disconnect
NRC - Incremental Charge - Manual Service Order - Disconnect

SOMEC SOMAN SOMAN

\$2.75 \$0.42 \$21.56 \$3.84

\$3.50 NA \$29.24 \$3.94

\$3.50 NA

\$3.50 NA

\$3.50 NA NA

\$3.50 NA

\$3.50

¥ ¥

₹ ₹

₹ ₹

\$75.40 \$98.59 \$13.08 \$219.72 \$96.86 \$40.45 \$40.45 \$19.99 TBD \$19.99 TBD NA

NA \$410.38 \$255.48 \$92.35 \$38.44

\$67.58 \$96.58 \$127.47 \$599.09 \$373.90 \$133.53 \$56.25

TBD TBD NA \$714.84 \$421.47

\$89.90 \$119.06 NA \$715.77 \$421.50

₹ ₹ ₹

\$18.94 \$8.42 NA

\$18.14 \$8.06 \$11.41

\$25.52 \$11.34 \$16.06

\$26.94 \$12.76 NA

\$43.77 \$13.55 NA

USLXX USLXX

RELXX RELXX RELXX RELXX

\$84.05 \$152.29 NA \$610.13 \$380.26 \$134.77 \$55.97

\$94.71 \$208.93 NA \$540.00 \$465.00 \$82.85 \$21.69

\$64.13 \$101.93 NA \$429.98 \$268.18 NA

\$94.06 \$162.34 NA \$849.80 \$523.27

\$96.73 \$197.57

RC - Zone 2, per month (Note 2)
RC - Zone 3, per month (Note 2)
RC - Zone 4, per month (Note 2)
NRC - 1st
NRC - 1st
NRC - Disconnect Charge - 1st
NRC - Disconnect Charge - Add¹

RC - Zone 1, per month (Note 2)
RC - Zone 2, per month (Note 2)
RC - Zone 3, per month (Note 2)
RC - Zone 4, per month (Note 2)
RC - Zone 4, per month (Note 2)
RC - Sone 4, per month (Note 2)
NRC - 1st
NRC - Add1
NRC - Disconnect Charge - 1st
NRC - Disconnect Charge - Add1
NRC - Service Order submitted Electronically, per LSR - Disconnect
NRC - Service Order submitted Electronically, per LSR - Disconnect

UDL56
UDL56
UDL56
UDL56
UDL56
SOMEC
SOMEC

\$129.62

\$348.55 \$241.20 NA NA \$3.50

NA \$258.62 \$176.11 \$127.74 \$27.90 \$3.50 NA

NA \$333.28 \$230.50 \$87.99 \$44.24 \$3.50 NA

\$25.61 \$33.94 \$48.51 \$64.02 \$489.00 \$337.93 \$128.36 \$64.35 \$84.35

TBD TBD TBD NA \$489.04 \$337.51 NA NA NA

\$42.23 \$36.45 \$45.87 \$65.75 NA \$643.00 \$421.26 NA NA NA S3.50 TBD

NA \$602.73 \$393.50 \$44.06 \$13.55 \$3.50 NA

\$64.25 \$3.50 NA

NA \$498.05 \$343.70

\$39.08 \$57.21 \$126.22 NA \$654.72 \$428.45 \$122.15 \$27.42 \$2.75 \$0.42

UDL56

\$27.33 \$44.40 \$80.45

\$25.75 \$29.74 \$47.27

\$35.92 \$40.32 \$37.90

\$27.50 \$47.24 \$96.48

\$34.26 \$51.67 \$68.43

UDL56 UDL56

	l	l						RC-	2-W						Ī								F	I	1					2-W																		RC-	2-Wir																		DES	
NRC - Disconnect Charge - Add'l	NRC - Add	NRC - ISE	NDC 15t	Zone A per month	Zone 3 per month	Zone 2 per month	Zone 1, per month	Statewide, per month	2-Wire Unb Copper Loop/Long (> 18kft), without Man SI & Fac Res	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	<ul> <li>Incremental Charge - Manual Service Order - Disconnect -</li> </ul>	NRC - Incremental Charge - Manual Service Order - Addit	NAC - Inclemental Charge - Manual Service Order - 18t	NDO Incremental Observation Order Act	NRC - Service Order submitted Manually per LSR Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st		NRC - 1st	NC - Zone 4, per monum (Note 2)	DC - Zone A per month (Note 2)	RC - Zone 3. per month (Note 2)	RC - Zone 2, per month (Note 2)	RC - Zone 1. per month (Note 2)	RC - Statewide, per month	2-Wire Unb Copper Loop/Long (> 18kft), incl Man SI & Fac Res	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'I	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	Zone 1, per month	Statewide, per moth	2-Wire Unb Copper Loop/Short (< or = 18kft), without Man SI & Fac Res	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	RC - Zone 4, per month (Note 2)	RC - Zone 3, per month (Note 2)	RC - Zone 2, per month (Note 2)	RC - Zone 1, per month (Note 2)	DESCRIPTION	
UCL2W	OCL2W	OCL2W	IICI 3W	We IOII	Mc DII	Mc DII	UCL2W	UCL2W		UCLMC	SOMAN	SOMAN	SOMAN	SOMAN	COMAN	NAMOS	SOMAN	SOMEC	SOMEC	UCL2L	UCL2L	UCLZL	UCLZL	300	10131	UCL2L	UCL2L	UCL2L	UCL2L		UCLMC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UCLPW	UCLPW	UCLPW	UCLPW	UCLPW	UCLPW	UCLPW	UCLPW	UCLPW		UCLMC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UCLPB	UCLPB	UCLPB	UCLPB	UCLPB	UCLPB	UCLPB	UCLPB	USOC	
N S	\$00.00	\$30F E8	6375.31	NA G	H I	TBD	TBD	\$40.00		\$16.00	\$17.77	\$17.77	\$21.00	\$47.00	94700	NA :	NA	N.	\$3.50	¥	NA	\$464.58	\$514.21	3	N i	TBD	ТВD	TBD	\$40.00		\$16.00	\$17.77	\$17.77	\$21.00	\$47.00	NA.	NA	NA	\$3.50	TBD	TBD	\$325.58	\$375.21	2	UBD	OBI	TBD	ļ. Ķ		\$16.00	\$17.77	\$17.77	\$21.00	\$47.00	AN	AN	NA	\$3.50	TBD	TBD	\$464.58	\$514.21	NA	TBD	TBD	TBD	AL	
\$15.46	\$108.00	\$145.63	610001	No.	\$71.17	\$58 13	\$48.79	¥		\$16.31	NA.	¥	¥	3		\$2.84	\$21.56	\$0.42	\$2.75	¥	\$154.23	\$193.27	\$331.86	3	NA.	\$71.17	\$58.13	\$48.79	¥		\$16.31	¥	¥	NA	¥	\$3.84	\$21.56	\$0.42	\$2.75	\$15.46	\$108.29	\$173.62	\$257.00	Ž X	\$60.07	\$27.23	\$18.60	NA.		\$16.31	NA	¥	¥	¥	\$3.84	\$21.56	\$0.42	\$2.75	\$35.23	\$154.23	\$251.26	\$389.84	NA	\$60.07	\$27.23	\$18.60	2	
TBD	\$139.75	\$130.75	9151	NA G	TB ig	TBD	TBD	\$37.00		\$36.46	\$37.86	\$142.27	\$8.42	\$10.94	9400	NA :	NA	N N	\$3.50	\$37.86	\$142.27	\$217.39	\$395.16	5	NA.	\$36.34	\$22.86	\$19.80	\$41.61		\$36.46	¥	¥	NA	¥	NA	NA	¥	\$3.50	\$37.45	\$140.73	\$139.75	\$154.13	2 8	\$21.83	\$13.74	\$11.90	¥		\$36.46	\$37.86	\$142.27	\$8.42	\$18.94	AN	AN	NA	\$3.50	\$37.86	\$142.27	\$217.39	\$395.16	NA	\$36.34	\$22.86	\$19.80	GA	
\$15.88	\$110.00	\$445.62	20027	NO.	\$80.78	\$49.31	\$36.19	ΝĀ		\$16.31	¥	¥	¥	\$47.00	\$47.00	\$3.04	\$29.24	NA A	\$3.50	\$36.20	\$160.06	\$193.26	\$333.21	3	NA C	\$80.78	\$49.31	\$36.19	¥		\$16.31	¥	₹	¥	¥	\$3.94	\$29.24	NA	\$3.50	\$15.88	\$112.86	\$174.35	\$259.09	\$ Z	\$15.73	\$15.15	\$14.94	NA.		\$16.31	NA	₹	₹	₹	\$3.94	\$29.24	NA	\$3.50	\$36.20	\$160.06	\$251.98	\$391.93	AN	\$15.73	\$15.15	\$14.94	হ	
\$15.81	\$110.40	\$200.00	\$200 000	47.401 #	\$104.74	92 65\$	\$43.92	NΑ		\$32.77	\$11.41	\$11.41	\$8.06	\$10.14	94044	NA.	AN	ΑN	\$3.50	\$36.02	\$159.29	\$192.99	\$332.73	NA.	VIN.	\$104.74	\$59.76	\$43.92	¥		\$32.77	\$11.41	\$11.41	\$8.06	\$18.14	NA	NA	NΑ	\$3.50	\$15.81	\$112.32	\$168.86	\$253.48	NA NA	\$18.83	\$17.99	\$16.34	AN		\$32.77	\$11.41	\$11.41	\$8.06	\$18.14	٨N	٨N	AN	\$3.50	\$36.02	\$159.29	\$246.38	\$386.13	AN	\$18.83	\$17.99	\$16.34	F	RATES BY STATE
\$57.25	\$105.06	\$347.04	446.10	\$40.13	\$31 92	\$22.34	\$16.85	¥		\$45.27	\$16.06	\$16.06	\$11.34	\$20.02	900	NA :	NA A	¥	\$3.50	\$57.25	\$105.86	\$456.24	\$504.82	φ42.13	£42.13	\$31.92	\$22.34	\$16.85	¥		\$45.27	\$16.06	\$16.06	\$11.34	\$25.52	NA NA	NA	¥	\$3.50	\$57.25	\$105.86	\$317.24	\$365.82	\$42.13	\$31.92	\$22.34	\$16.85	¥		\$45.27	\$16.06	\$16.06	\$11.34	\$25.52	AN	AN	NA	\$3.50	\$57.25	\$105.86	\$456.24	\$504.82	\$42.13	\$31.92	\$22.34	\$16.85	MS	
N S	\$10.00	\$351.00	6311 00	NA O	\$73.80	\$64.03	\$38.66	¥		\$16.00	¥	¥	\$12.76	\$20.94	9000	NA :	NA	¥	\$3.50	₹	¥	\$390.00	\$450.00	5	NIA C	\$73.89	\$64.03	\$38.66	¥		\$16.00	Ā	¥	\$12.76	\$26.94	NA.	NA	¥	\$3.50	NA A	¥	\$251.00	\$311.00	S A	\$23.88	\$20.63	\$12.27	¥		\$16.00	NA	¥	\$12.76	\$26.94	NA	NA	NA	\$3.50	Ą	NA	\$390.00	\$450.00	NA	\$23.88	\$20.63	\$12.27	R	
N S	\$100.00	9269 22	6464 64	NA . T	\$37.75	\$28.50	\$18.90	¥		\$45.43	\$21.00	\$21.00	\$25.52	\$47.00	94700	NA :	NA	¥	\$3.50	¥	NA	\$507.33	\$600.61	5	NIA C	\$37.75	\$28.50	\$18.90	¥		\$45.43	\$21.00	\$21.00	\$25.52	\$47.00	T					T	T	\$461.61	t	Т	T	\$18.90	T	T	\$45.43	\$21.00	\$21.00	\$25.52	\$47.00	NA	NA	NA	\$3.50	Ą	NA	\$507.33	\$600.61	NA	\$37.75	\$28.50	\$18.90	SC	
\$39.14	\$74.E4	\$05.63	6131 01	NA.O.	\$35.81	\$24.98	\$19.85	¥		\$34.29	NA	NA	NA	- N	5	# H	\$19.99	dan	\$3.50	\$39.14	\$74.54	\$234.63	\$270.01		NIA C	\$35.81	\$24.98	\$19.85	Š		\$34.29	NA	¥	NA	NA	NA	\$19.99	ΝĀ	\$3.50	\$39.14	\$74.54	\$95.63	\$131.01	2	\$35.81	\$24.98	\$19.85	NA.		\$34.29	NA	¥	Ā	Ā	AN	\$19.99	NA	\$3.50	\$39.14	\$74.54	\$234.63	\$270.01	NA	\$35.81	\$24.98	\$19.85	Į	

$\vdash$	+	+	+	+		$\vdash$		$\vdash$	+	+	+	-	$\vdash$		$\dashv$	+	+	+			$\dashv$	$\dashv$	+	+	+	H	+	+	+-	-	H	+	+	+	+	$\dashv$	+	+	Н	$\dashv$	+	+	+	++	+	+	Н	$\dashv$	+	+	+	+	+	-
NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSK - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Add I  NRC - Disconnect Charge - 1st	, , , , , , , , , , , , , , , , , , ,	NRC - 1st	Zone 4, per month	Zone 3 per month	Zone 2 per month	Statewide, Per month	4-Wire Unb Copper Loop/Long (>18kft), incl Man Svc Ing & Fac Res	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	Zone 1 per month	4-Wire Unb Copper Loop/Short (< or = 18kft), without Man SI & Fac Res	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - 18t	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically per LSR	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	Zone 4 per month	Zone 2, per month	Zone 1, per month	Statewide, per month*	4-Wire Unb Copper Loop/Short (< or = 18kft), incl Man SI & Fac Res	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'L	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	DESCRIPTION	
SOMAN	SOMAN	SCMAN	SOMEC	SOMEC	UCL4L	UCL4L	5	UCL4L	UCL4L	1014	UCL4L	UCL4L		UCLMC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UCL4W	UCL4W	UCL4W	UCL4W	UCL4W	UCL4W	UCL4W		UCLMC	SOMAN	SOMAN	NAMOS	SOMAN	SOMAN	SOMEC	SOMEC.	UCL4S	UCL4S	UCL4S	UCL4S	CL4S	UCL4S	UCL4S	UCLIVIC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	USOC	
TBD	BB ≸	: ₹	: ₹	\$3.50	TBD	TBD	d d	TBO S	¥ 5	<b>7 8</b>	<b>3 8</b>	曹	!	TBD	i	<b>3</b> 8	d E	-	¥	NA	\$3.50	룡	룡	를 돈	₹ ₩	ПВD	OBI	를 돈		TBD		TB G		∃ ≼	NA	¥ S	\$3.50	間	ТВD	TB :	NA E	# E	TBD	TBD	\$10.00	\$17.77	\$17.77	\$21.00	\$47.00	× ×	<b>X</b>	\$3.50	A	_
NA	\$3.84 NA	\$21.56	\$0.42	\$2.75	\$39.76	\$161.19	3	\$380.29	NA ST	\$147.54	\$119.02	AN AN		\$16.31	NA :	Z 3	S &	\$3.84	\$21.56	\$0.42	\$2.75	\$19.58	\$114.30	\$305.43	NA NA	\$32.24	\$30.53	\$25.56 NA		\$16.31		¥ ₹	N S	\$3.84	\$21.56	\$0.42	\$39.75	\$161.19	\$299.68	\$438.27	\$32.24 NA	\$30.53	\$25.56	NA	\$16.31	e NA	NA.	¥ ;	\$3.84 NA	\$21.56	\$0.42	\$2.75	P	
NA	N X	3 3	: ₹	\$3.50	\$41.96	\$156.25	3	\$397.06	NA S	30.75%	\$47.56	\$55.86		\$36.46		N S	× ×	Z	¥.	NA.	\$3.50	\$41.96	\$156.25	\$16261	NA NA	\$30.55	\$19.22	\$19.34		\$36.46		₹ ₹	N S	₹ ₹	NA	N S	\$3.50	\$156.25	\$162.61	\$353.80	\$30.55	\$19.22	\$16.65	\$19.34	\$36.46	TBD	TBD	TBD	TRO S	× ×	X X	\$3.50	GA	
¥	NA 34	\$29.24	8	\$3.50	\$41.90	\$171.58	2	\$381.63	NA.	\$88.97	\$55.02	2 X	:	\$16.31	¥.	₹ ₹	<b>S §</b>	\$3.94	\$29.24	NA.	\$3.50	\$20.64	\$122.17	\$307.51	NA NA	\$19.08	\$23.00	\$25.26 NA		\$16.31	¥	₹ ₹	N ₹	\$3.94	\$29.24	N S	\$3.50	\$171.58	\$300.41	\$440.35	\$19.08	\$23.00	\$25.26	NA	\$16.31	\$16.21	₹	₹ :	\$3.94 AA	\$29.24	NA NA	\$3.50		
\$8.06	\$18.14	₹ ₹	: ₹	\$3.50	\$42.60	\$174.43	2	\$381.09	NA	\$143.00	\$115.84	7 K	:	\$32.77	\$11.41	\$11.41	\$18.14	NA NA	¥.	NA	\$3.50	\$20.98	\$124.19	\$301.83	NA NA	\$28.75	\$26.62	\$29 14		\$32.77	\$11.41	\$11.41	\$10.14	e N	NA.	N S	\$3.50	\$174.43	\$294.74	\$434.49	\$28.75	\$26.62	\$29.14	NA	\$32.11	\$11.41	\$11.41	\$8.06	\$18 14	× ×	¥	\$3.50	LA LA	7 A T C C T
\$11.34	\$25.52	₹ ₹	: ₹	\$3.50	\$40.00	\$150.00	3	\$400.00	TBD	\$90.00	\$60.00	P NA		\$45.27	\$16.06	\$16.06	\$25.52	AN AN	¥.	×.	\$3.50	\$40.00	\$150.00	\$261.00	\$35.00	\$30.00	\$27.50	\$25 00 NA		\$45.27	\$16.06	\$16.06	\$11.34	AN A	NA	N S	\$3.50	\$150.00	\$300.00	\$400.00	\$35.00	\$27.50	\$25.00	NA	\$40.21	\$16.06	\$16.06	\$11.34	\$25.52 NA	× ×	X X	\$3.50	MS	i
$\vdash$	+	+	+	+	$\vdash$	\$150.00		\$400.00	NA S	\$105.09	\$90.93	e N		\$16.00	NA :	NA	\$26.94	NA NA	¥.	NA.	\$3.50	\$40.00	\$150.00	\$261.00	NA NA	\$34.14	\$29.75	\$18.40	:	\$16.00	NA	W	\$20.94	A A	NA	N S	\$3.50	\$150.00	\$300.00	\$400.00	\$34.14 NA	\$29.75	\$18.49	NA	\$10.00	NA NA	A	\$12.76	\$26 94	<b>X X</b>	<b>X</b>	\$3.50	NC	
$\vdash$	+	╁	+	+		\$150.00		\$400.00	NA O	\$90.00	\$50.00	P N		\$45.43	\$21.00	\$21.00	\$47.00	AN AN	¥			-	+	+	NA NA		\$27.50	+	+	\$45.43	\$21.00	\$21.00	\$35.50	\$ A	NA	Z S	\$3.50	\$150.00	\$300.00	\$400.00	\$30.00	\$27.50	\$25.00	NA.	\$40.40	\$21.00	\$21.00	\$25.52	\$47.00	<b>₹</b>	X X	\$3.50	SC	
	N S	+	+	+		\$150.00		\$400.00	NA O	\$90.00	\$50.00	P N		\$34.29	NA :	Z 3	3	₹ ₩	\$19.99	NA.	\$3.50	\$40.00	\$150.00	\$261.00	NA S	\$30.00	\$27.50	\$25.00 NA		\$34.29	NA	₹ ₹	N S	<b>X X</b>	\$19.99	Z S	\$3.50	\$150.00	\$300.00	\$400.00	\$30.00	\$27.50	\$25.00	NA	\$34.28	AN AN	¥	¥.	2 2	\$19.99	NA NA	\$3.50	TN	

					R.	RATES BY STATE	Æ		
	USOC	AL	F	GA	ব	LA	MS	NC	
Manual Svc Order - Add'l	NAMOS	\$38.48	N	\$37 AA	\$03.10	#50 OF	\$68 63	\$60.34	e A

			Loop				cr	SUB-	UNB									Unbu																				00-1												OC -1	H	#	DEac	ק ק
Zone 3, per month	Zone 2, per month	Zone 1, per month	Loop Distribution per 2-Wire Analog VG Sub-Loop, per month	NRC - Set-Thing Full bring Equipment Room - per 25 pair panel set-up	NRC - Set-Up per Building Equipment Room - CLEC Feeder Facility set-up	NRC - Set-Up per Cross Box location in the field - CLEC Feeder Facility set-up	Cross-Box Set-Up	SUB-LOOP DISTRIBUTION	NDLED SUB-LOOPS	NRC - Bridge Tap Removal per pair unloaded **	NRC - Load Coil/Equipment Removal per 4 Wire pair - Loops greater than 18kft -Add" **	Inno Loda Comedulation remedial por 4 mino pari Lodas greater man rom for	NRC - Load Coil/Equipment Removal per 4 Wire pair - Loops greater than 18kft - 1st **	NRC - Load Coil/Equipment Removal per 4 Wire pair - Loops less than or equal to 18kft **	NRC - Load Coil/Equipment Removal per 2 Wire pair - Loops greater than 18kft -Add" **	NRC - Load Coil/Equipment Removal per 2 Wire pair - Loops greater than 18kft - 1st **	NRC - Load Coil/Equipment Removal per 2 Wire pair - Loops less than or equal to 18kft **	Unbundled Loop Modification/Conditioning	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-Add'i	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-1st	NRC - OC-48 - Incremental ChargeManual Svc Order-Add'l	NRC - OC-48 - Incremental ChargeManual Svc Order-1st	NBC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-1st	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-Add'l	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - OC48 - Interface OC12 on OC48 - Disconnect - Add"l	NRC - OC48 - Facility Termination - Disconnect - Add I	NRC - OC48 - Facility Termination - Disconnect - 1st	NRC - OC48 - Interface OC12 on OC48 - Add'l	NRC - OC48 - Interface OC12 on OC48 - 1st	NRC - OC46 - Facility Termination - Add'i	Local Loop - OC12 interface on OC48 Facility	Local Loop - OC48 - per Facility Termination	Local Loop - OC48 - per Mile	48 Local Loop	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-1st	NRC - OC12 - Incremental Charge - Manual Svc Order - Add'l	NRC -OC12 - Incremental Charge - Manual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - OC12 - Facility Termination - Disconnect - Add'l	NRC - OC12 - Facility Termination - Disconnect - 1st	NRC - OC12 - Facility Termination - Add'l	NRC - OC12 - Facility Termination - 1st	Local Loop - OC12 - per Mile	OC -12 Local Loop	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	NRC - OC3 - Incremental Charge-Manual Svc Order - Add'i	A TOPPICAL
USBN2	USBN2	USBN2	USBN2	USBSD	USBSC	USBSA				ULMBT	ULM4G	Cristia	ULM4G	ULM4L	ULM2G	ULM2G	ULM2L		SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMAN	SOMAN									1L5ND	OCIONA	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN					1L5ND	:	SOMAN	SOMAN	SOMAN	
ТВО	TBD	TBD	\$9.05	\$107.63	\$311.77	\$517.43				\$65.28	\$23.55	0.70	\$716.70	\$65.23	\$23.55	\$716.70	\$65.23		\$38.48	\$38.48	\$38.48	\$38.48	\$19.03	\$19.03	\$19.03	NA 0	Ž A	N.	\$118.54	\$121.72	\$121.72	\$118.54	\$121.72	\$413.38	\$736.71	\$1,713	\$33.22	<b>\$</b> 19.00	\$19.03	\$38.48	\$38.48	N.	\$3.50	Z X	\$118.54	\$121.72	\$413.38	\$1.165	\$10.13		\$19.03	\$19.03	\$38.48	2
\$16.13	\$12.49	\$9.36	N S	\$109.85	\$333.44	\$711.78	1			\$65.44	\$23.77	9	\$710.71	\$65.40	\$23.77	\$710.71	\$65.40		NA.	N <sub>A</sub>	¥	X 5	N N	¥	NA :	\$0.42	\$3.84	\$21.56	\$108.34	\$111.56	\$111.56	\$312.05	\$543.72	\$408.85	\$587.71	\$1,685.97	\$36.04	5	₹ ₹	¥	¥	\$0.42	\$2.84	\$21.56	\$108.34	\$111.56	\$408.85	\$1.183.46	\$10.99		NA :	¥ :		2
TBD	TBD	TBD	\$9.12	\$154.57	\$394.74	\$421.08				\$79.99	\$23.49	9	\$757.04	\$69.28	\$23.49	\$757.04	\$69.28		\$37.55	\$37.55	\$37.55	\$37.55	\$18.03	\$18.03	\$18.03	NA 00.50	Ž X	¥	\$119.14	\$122.31	\$122.31	\$317.38	\$539.36	\$413.00	\$594.80	\$1,598.00	\$27.25	6.00	\$18.03	\$37.55	\$37.55	Z.	\$3.50	Z X	\$1	\$122.31	_	\$1.162.00	\$8.31		\$18.03	\$18.03	\$37.55	2
\$16.71	\$12.25	\$9.03	N :	\$111.55	\$407.02	\$627.16				\$65.44	\$23.77		\$710.73	\$65.40	\$23.77	\$710.73	\$65.40		NA A	NA	\$93.12	\$93.12	N N	\$93.12	\$93.12	NA 00.50	Š	\$19.99	¥	<b>¥</b> ₹	. ₩	\$516.89	\$844.21	\$661.23	\$725.77	\$2,129	\$133.84	5	X X	\$93.12	\$93.12	N.	\$3.50	\$19.99	A	×.	\$661.23		\$40.80		NA :	N.	\$93.12	
\$21.11	\$14.43	\$10.33	NA S	\$104.26	\$289.90	\$639.68				\$65.35	\$23.74	00.7	\$709.71	\$65.30	\$23.74	\$709.71	\$65.30		\$50.25	\$50.25	\$50.25	\$50.25	\$20.94	\$20.94	\$20.94	NA 30.50	S A	¥	\$99.46	\$102.16	\$102.16	\$304.90	\$532.13	\$402.63	\$723.29	\$2,268	\$119.40	\$20.5 <del>4</del>	\$20.94	\$50.25	\$50.25	N.	\$3.50	×	\$99.46	\$102.16	\$402.63	\$1.245	\$36.40		\$20.94	\$20.94	\$50.25	RATES BY STATE
TBD	TBD	TBD	TBD	TBD 18	TBD	TBD TBD				\$121.14	\$27.30	<b>#000.00</b>	\$880.00	\$80.55	\$27.30	\$880.00	\$80.55		\$68.62	\$68.62	\$68.62	\$68.62	\$28.59	\$28.59	\$28.59	NA AN	Ž A	×.	\$130.59	\$134.07	\$134.07	\$404.94	\$729.04	\$549.17	\$667.00	\$1,753	\$166.59	ψ	\$28.59	\$68.62	\$68.62	Z.	\$3.50	¥	\$130.59	\$134.07	\$549.17	\$1.742	\$50.79		\$28.59	\$28.59	\$68.62	ATE
\$14.43	\$12.63	\$7.99	NA S	\$108.06	\$313.01	\$498.09				\$65.64	\$23.65	÷ 10:00	\$719.55	\$65.49	\$23.65	\$719.55	\$65.49		\$69.34	\$69.34	\$69.34	\$69.34	N N	N <sub>A</sub>	NA :	NA SO	Ž A	×	\$128.19	\$131.65	\$131.65	\$400.38	\$720.81	\$542.73	\$582.66	\$1,677	\$120.02	\$20.00	\$28.59	\$68.62	\$68.62	¥.	\$3.50	¥	\$128.19	\$131.65	\$542.73	\$1.722	\$30.38		\$29.76	\$29.76	\$69.34	5
H		NA.	+	+	+	\$510.15	+			\$65.24	\$23.54	0.00	\$716.32	\$65.20	\$23.54	\$716.32	\$65.20		NA	NA.	\$54.26					\$3.50 A			H	X X	+	ω.	+	+	+	+	H		<b>X X</b>				+	-				\$1,259			NA	N.	\$54.26	3
H	Н		+	+	+		╫			\$65.44	\$23.77	•	\$710.71	\$65.40	\$23.77	\$710.71	\$65.40		NA	W	¥	¥ 3	N N	N <sub>A</sub>	NA :	\$3.50 AN	3 X	\$19.99	\$100.59	\$100.59	-	H	+		+	-		3	<b>X X</b>	NA.	¥	N.	+	╁			-	+	\$28.51		NA.	¥	<b>₹</b>	į

	- g				l	L								L	L	Loop			Ⅎ							t	t			-000	E			1		L		1		L			t	_oop			l		╅				$\pm$		DESC	
NRC - Add'l	Sub-Loop-Intrabidg Ntwk Cable (aka riser cable), 2W analog, per mo	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Bervice Order submitted Manually, per LSK, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add"	Zone 4, per month	Zone 3, per month	Zone 2, per month	Zone 1, per month	undled Copp	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - ISt	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - 1St	Zone 4, per month	Zone 3, per month	Zone 2, per month		NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'I	NRC - Jervice Order sublitated Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Addi	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 1, per month	oop Distribution per 4-Wire Analog VG Sub-Loop, per month	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Service Order submitted Electronically, per LSR		NRC - Disconnect Charge - 1st	NRC - 1st	Zone 4, per month	DESCRIPTION 7 TO A DOCUMENT OF THE PROPERTY OF	
USBR2	USBR2	USBMC	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UCS4X	UCS4X	INSAX INSAX	UCS4X	UCS4X	UCS4X	UCS4X	UCS4X	USBMC	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	UCS2X	UCS2X	UCSZX	UCS2X	UCS2X	UCS2X	UCSZX	USBMC	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	USBN4	COBN4	USBN4	USBN4	USBN4	USBN4	USBN4	USBMC	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	USBN2	USBN2	USBN2	USBNZ	USOC	
\$50.44	\$1.33	\$51.29	TBD :	큠	B ₹	5 ₹	NA	\$3.50	\$18.46	\$107.24	\$85.04	\$161.50	NA	тво	TBD	\$8.06	\$51.29	OBT		# ₹	<b>X X</b>	₹	\$3.50	\$14.05	\$99.54	\$136.55	NA NA	TBD	명당	#B.55	\$51.29	TBD	TBD	T S	Z Z	K	\$3.50	\$18.46	\$79.35	\$155.90	NA NA	den den	曹围	\$10.56	\$51.29	TE S	# #B	NA	N 5	\$3.50	\$13.21	\$92.45	\$125.53 \$54.32	6105.53	ž A	
\$36.36	\$3.87	\$16.31	N.	<b>₹</b>	\$3.84	\$21.56	\$0.42	\$2.75	\$17.15	\$104.31	\$88.40	\$165.68	\$16.92	\$11.26	\$7.11	NA	\$16.31	¥ ;	X S	\$3.84	\$21.56	\$0.42	\$2.75	\$13.08	\$98.49	\$139.20	NA S	\$12.76	\$10.37	\$7 91	\$16.31	NA A	NA.	AN.04	\$21.56	\$0.42	\$2.75	\$17.15	\$88.42	\$165.68	NA	\$26.09	\$10.12	AN	\$16.31	<b>₹</b>	× ×	\$3.84	\$21.56	\$2.75	\$13.08	\$98.49	\$139.20	e130 30	- : P	!
\$41.59	\$1.61	TBD	ТВО	TBD	TB N	<b>X X</b>	NA	\$3.50	TBD	E G	T 5	TBD IS	NA CIBIT	TBD	TBD	TBD	TBD	OBT.	TB B	5 €	5 ₹	¥	\$3.50	TBD	TBO	TEN IE	ī K	TBD	TBD	TE E	\$34.22	NA	¥.	N S	X X	¥	\$3.50	\$28.77	\$12372	\$219.35	N.	TB G	# E	\$8.32	\$34.22	TBD	\$18.94	NA	₹ 5	\$3.50	TBD	TBD	\$207.01	\$207.01	GA	
\$36.35	\$3.23	\$16.31	×.	X S	\$3.94 NA	\$29.24	NA	\$3.50	\$18.08	\$109.94	\$88 41	\$165.67	\$8.45	\$9.71	\$10.65	NA	\$16.31	<b>⊼</b> :	<b>₹</b>	\$3.94	\$29.24	NA.	\$3.50	\$13.44	\$101.18	\$139.19	Z NA	\$11.02	\$9.18	\$8.01	\$16.31	NA	NA.	VA.94	\$29.24	×	\$3.50	\$18.08	\$88.41	\$165.67	NA	\$13.38	\$10.18	NA	\$16.31	¥ ₹	X X	\$3.94	\$29.24	\$3.50	\$13.44	\$101.18	\$139.19	\$130 10	ই	
\$35.12	\$1.59	\$16.29	\$11.41	\$8.06	\$18 14	¥ ×	NA A	\$3.50	\$18.38	\$111.76	\$8200.40	\$165.43	\$17.66	\$13.52	\$10.48	×	\$16.29	\$11.41	\$8.06	STO A	₹ ₹	¥	\$3.50	\$13.37	\$100.70	\$138.99	NA NA	\$14.08	\$11.43	\$8.70	\$16.29	\$11.41	\$8.06	\$18 14	8 8	¥	\$3.50	\$18.38	\$88.29	\$165.43	N N	\$24.93	\$15.67	NA AN	\$16.29	\$11.41	\$18.14	NA	¥ §	\$3.50	\$13.37	\$100.70	\$138.99	\$138 00	<b>5 5</b>	RATES BY STATE
TBD	TBD	TBD	TBD	TB 6	<b>B B</b>	TBD	TBD	TBD	TBD	TBD 6	# E	TBD IBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD I	3 8	TBD	TBD	TBD	OBT	TBD I	# E	TBD	TBD	OBT.	H E	TBD	TBD	TBD	T E	T 180	TBD	TBD	TBD I	TBD IBD	TBD	TBD	TBD	T 180	TBD	TBD	TBD 5	TBD	TBD	TB G	TBD	TBD	TBD	TBD	TBD	MS	
\$37.20	\$3.50	\$45.34	Z.	X S	NA NA	Z X	NA	\$3.50	\$13.53	\$78.56	\$2.20	\$162.24	\$12.63	\$11.09	\$7.14	NA	\$45.34	¥ ;	<b>₹</b>	N N	₹ ₹	N.	\$3.50	\$10.81	\$76.58	\$137.10	Z NA	\$12.36	\$10.95	\$7.33	\$45.34	NA	NA :	N 3	8 8	₹	\$3.50	\$13.53	\$79.56	\$156.52	NA	\$16.73	\$9.23	NA	\$45.34	\$15.12	\$26.94	NA	X 3	\$3.50	\$10.16	\$71.13	\$126.03	e126.03	: K	
\$47.09	\$1.60	\$45.43	X.	X S	NA NA	<b>X</b>	NA	\$3.50	¥.	NA S	\$07.61	\$238 46	\$ \$	X.	¥	\$8.86	\$45.43	¥ ;	¥ ₹	S &	₹ ₹	×.	\$3.50	<b>₹</b>	NA 0.00	\$212.46	S NA	NA	<b>₹</b>	\$6.8Z	\$45.43	TBD	NA :	N S	8 8	¥	\$3.50	¥ ₹	\$91.9Z	\$232.76	NA	<b>₹</b>	<b>5</b>	\$10.22	\$45.43	X 5	<b>₹</b>	NA	₹ 3	\$3.50	NA	NA S	\$195.98	e ine ng	SC	
\$45.63	\$1.47	TBD	TBD	명	룡	병명	ТВО	TBD	ТВD	TB G	Ħ 6	<b>4 5</b>		TBD	TBD	NA A	OBT.	TBD	DB D		曹围	I BD	TBD	OBT OBT		# E	TBD	TBD	OBT.	∄ ₹	TBD	TBD	ТВО	Ħ.				OBT OBT	<b>#</b>	TBO	OBT	OBT	# B	AN	TBD	dau	# B	TBD	TBI S	TBD	OBT	OBT		₫ 🕏	ž	

Zone 1, per month	2-Wire Analog VG Reverse Battery Unb Sub-Loop Feeder, per mo	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	Zone 1, per month	2-Wire Analog VG Loop-Start Unbundled Sub-Loop Feeder, per month	NVC - III eli eli al Cialge - Malidal Selvice Oldel - Discolliect	NRC - Incremental Charge - Manual Service Order - Add I	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add"	NRC - Disconnect Charge - 1st	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	Zone 1, per month	2-Wire Analog VG Ground-Start Unbundled Sub-Loop Feeder, per month	NRC - DS1 Set-Up per Cross Box location - per pair panel set-up	NRC - DS1 Set-Up per Cross Box location - CLEC Distribution Facility set-up	NRC - DS0 Set-Up per Cross Box location - per 25 pair panel set-up	NRC - DS0 Set-Up per Cross Box location - CLEC Distribution Facility set-up	Cross-Box Set-Up	SUB-LOOP FEEDER	NRC - Incremental Charge - Imanual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Disconnect Charge - Add	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	Sub-Loop-Intrabidg Ntwk Cable (aka riser cable), 4W analog, per mo	NRC - Incremental Charge - Manual Order Coordination - per loop	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charne - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	DESCRIPTION	
USBFC	USBFC	SOMAN	SOMAN	SOMAN	NAMOS	SOMEC	SOMEC	USBFB	USBFB	USBFB	BHBSU	USBFB	USBFB	USBFB	USBFB	NEWNOO	NAMOS	NAMOS	SOMAN	SOMAN	SOMEC	SOMEC	ABBAI	COBFA	USBFA	USBFA	USBFA	NSBFA	USBFA	USBFA	USBFZ	USBFY	USBFX	USBFW			COBINIC	NAMOS	SOMAN	SOMAN	SOMAN	NAMOS	SOMEC	SOMEC SOMEC	USBR4	USBR4	USBR4	USBR4	USBMC	SOMAN	SOMAN	SOMAN	NAMOS	SOMEC	SOMEC	USBR2	USBR2	USOC	
TBD	\$13.68	TBD	TBD	3 ₹	¥	NA	\$3.50	\$27.04	\$119.95	\$116.66	23.5	TBD	TBD	TBD	\$13.68	100	TBD	TBD	NA	NA NA	NA	\$3.50	\$27.04	\$116.66	\$198.42	NA	TBD	TBD	TBD	\$13.68	44.87	517.43	44.87	517.43			\$51.29	180	TBD	TBD	NA	NA	NA S	\$3.50	\$107.45	\$50.62	\$127.17	\$2.17	\$51.29	TBD	TBD	# F	\$ \$	X X	\$3.50	\$14.05	\$99.54	P	
\$10.75	×	N S	8 8	\$3.84	\$21.56	\$0.42	\$2.75	\$26.70	\$116.59	\$113.00	200	\$13.51	\$11.57	\$10.75	NA	3	NA NA	¥	\$3.84	\$21.56	\$0.42	\$2.75	\$26.70	\$116.50	\$193.62	NA	\$13.51	\$11.57	\$10.75	¥	45.28	711.78	45.28	711.78			0.01	e A NA	X X	NA A	\$3.84	\$21.56	\$0.42	\$17.15	\$104.31	\$48.84	\$126.10	\$7.20	\$16.31	¥.	NA S	\$3.04	\$21.56	\$0.42	\$2.75	\$13.08	\$98.49	핃	
TBD	\$8.58	TBD	\$18.94	200	¥	NA	\$3.50	TBD	TBD	\$170.05	\$ 2	TBD	TBD	TBD	\$8.58	- 00	\$8.42	\$18.94	A	¥	NA	\$3.50	TBD	\$170.05	\$206.44	NA	TBD	TBD	TBD	\$8.58	\$67.10	\$421.08	\$67.10	\$421.08			27.400	AN AN	X X	NA.	NA	Ā	NA S	\$3.50	\$122.17	\$55.11	\$176.46	\$2.96	\$34.22	NA :	NA S	Z 3	\$ \$	X.	\$3.50	\$19.17	\$115.85	GA	
\$10.36	¥.	N S	N N	\$3.94	\$29.24	NA	\$3.50	\$26.79	\$119.14	\$111.96	201	\$19.69	\$13.62	\$10.36	NA	3	N A	¥	\$3.94	\$29.24	NA	\$3.50	\$26.79	\$111.96	\$192.57	NA	\$19.69	\$13.62	\$10.36	¥	45.28	627.16	45.28	627.16			\$10.31	\$16.31	. ₹	NA.	\$3.94	\$29.24	NA S	\$18.08	\$109.94	\$48.84	\$126.10	\$6.29	\$16.31	¥ :	N S	\$3.94	\$29.24	NA NA	\$3.50	\$13.44	\$101.18	ক	77
\$11.01	¥.	\$11.41	\$18.14	201	<b>X</b>	NA	\$3.50	\$26.66	\$118.57	\$111.80	\$ X	\$21.56	\$13.36	\$11.01	NA	41.4	\$8.06	\$18.14	NA	¥	NA	\$3.50	\$26.66	\$111.80	\$192.30	NA	\$21.56	\$13.36	\$11.01	¥	\$45.22	\$639.68	\$45.22	\$639.68			\$10.29	\$17.41	\$8.06	\$18.14	NA A	¥	N S	\$3.50	\$101.65	\$47.70	\$118.69	\$2.83	\$16.29	\$11.41	\$8.06	\$18 14	¥ ×	¥	\$3.50	\$13.92	\$93.19	LA	RATES BY STATE
TBD	TBD	OBT		ġ =	TBD	TBD	TBD	OBT.	TBD G	B B	9 8	TBD	TBD	TBD	TBD	100	3 5	TBD	TBD	TBD	TBD	TBD	7 6		TBD	NA	TBD	TBD	OBT IS	dan	TBD	TBD	TBD	TBD			100	180	TBD	TBD	TBD	TBD	TBD	TBD I	TBD	TBD	TBD	TBD	TBD	TBD	TBD	# E	TBD	TBD	TBD	TBD	TBD		Ħ
\$8.92	N.	\$15.12	\$12 76	38 N	¥	NA.	\$3.50	\$14.68	\$77.90	\$46.61	3	\$16.11	\$14.10	\$8.92	NA	φ13.12	\$12.76	\$26.94	¥	¥	NA	\$3.50	\$14.68	\$46.61	\$122.52	NA	\$16.11	\$14.10	\$8.92	¥	45.04	498.09	45.04	498.09			\$45.34	NA NA	<b>X</b>	¥	N <sub>P</sub>	¥	N S	\$3.50	\$78.71	\$50.82	\$127.67	\$3.75	\$45.34	¥ :	N S	Z 3	\$ \$	¥	\$3.50	\$10.81	\$76.58	S	
NA	\$10.06	NA S	NA NA	\$ ₹	¥	NA	\$3.50	NA S	NA S	\$60.33	282	×	¥	NA	\$10.06	5	N K	. ₹	¥	¥	NA	\$3.50	NA 5	\$60.33	\$199.27	NA	NA	¥	NA S	\$10.06	44.84	510.15	44.84	510.15			\$45.43	NA AS	<b>X</b>	NA.	NA	Ā	NA S	\$3.50 NA	₹ ₹	\$60.40	\$204.20	\$2.78	\$45.43	¥.	N S	Z S	5 ₹	<b>X</b>	\$3.50	¥.	¥	SC	
TBD	N N	TBD		ġ @	TBD	TBD	TBD	agi d	den den	TBD IBD	d E	TBD	TBD	TBD	N.	ē	3	ТВD	ТВD	TBD	TBD	TBD I	3 8	<b>B B</b>	TBD	TBD	ТВD	TBD	큠	X.	TBD	TBD	TBD	ТВО			\$35.00	e e	<b>X</b>	NA A	TBD	\$19.99	TBD CO	\$2.70	\$135.88	\$60.47	\$193.62	\$2.55	\$55.00	<b>∑</b>	N S	Z 0	\$19.99	TBD	\$3.50	\$21.32	\$128.85	¥	

4-Wire DSI Unbundled Sub-Loop Feeder, per month	- Manual Service Order -	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	Zone 1, per month	2-Wire ISDN Unbundled Sub-Loop Feeder, per month	NKC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Addil	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	ectronically,	NRC - Disconnect Charge - Add"l	NRC - Disconnect Charge - 1st	NRC - Add'l	NDC 10t	Zone 3, per month	Zone 2, per month	Zone 1, per month	4-Wire Analog VG Loop-Start Unbundled Sub-Loop Feeder, per month	NRC - II CI el le lla l'Ciai ge - Mai tudi Service Oldei - Discoillect	NRC - Incremental Charge - Manual Service Order - Add I	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Manually, per LSR	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add"	NRC - Discorport Chargo - 1st	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	Zone 1. per month	4.Wire Analog VG Ground-Start Inhundled Sub-Loon Feeder per month	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSK	NRC - Setvice Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - 1st	Zone 4, per month	Zone 3, per month	Zone 2, per month	DESCRIPTION
USBFG	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	USBFF	USBFF	USBFF	USBFF	USBFF	USBFF	USBFF	USBFF	USBFF	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	USBFE	USBFE	USBFE	LIGHT	COBTE	USBFE	USBFE	USBFE	OCIVIAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	USBED	USBFD	USBFD	USBFD	USBFD	USBFD	USBFD	LISBED	SOMAN	SOMAN	NAMOS	SOMAN	SOMEC	SOMEC	USBFC	USBFC	USBFC	USBFC	USBFC	USBFC	USOC
\$49.54	TBD	TBD	TBD	NA	¥	¥	\$3.50	\$27.04	\$119.95	\$104.51	\$200.26	NA	TBD	TBD	TBD	\$23.66	IBU	d 8	TBD	¥	¥	NA.	\$3.50	\$34.66	\$131.15	\$142.45	\$334.34	3 5	TBD	TBD	\$24.94	100	3 8	TBD	NA	NA	¥.	\$3.50	\$34.66	\$142.45	\$224.21	NA	TBD	ТВD	TBD	\$24.94	TBD	TBD	TRO S	N N		\$3.50	\$27.04	\$119.95	\$198.42	NA	TBD	ТВD	<u> </u>
NA	AN	NA	NA	\$3.84	\$21.56	\$0.42	\$2.75	\$32.91	\$127.64	\$140.22	\$222.74	NA	\$26.12	\$25.85	\$22.39	¥	NA	₹ ₹	X.	\$3.84	\$21.56	\$0.42	\$2.75	\$32.91	\$127.64	\$140.22	\$222 \$4	\$40.51	\$27.94	\$23.35	NA	3	S X	X.	\$3.84	\$21.56	\$0.42	\$2.75	\$32.04	\$140.22	\$222.74	NA	\$40.51	\$27.94	\$23.35	NA	N S	NA .	NA OF	\$21.56	\$0.42	\$2.75	\$26.70	\$116.59	\$193.62	NA	\$13.51	\$11.57	<u>n</u>
TBD	AN	NA	NA	NA	¥	¥	\$3.50	\$29.58	\$119.68	\$62.31	\$208.50	NA	TBD	TBD	TBD	\$17.73	N	₹ ₹	NA.	NA.	Ā	NA	\$3.50	\$33.93	\$134.77	\$81.32	\$243.41	180	TBD	TBD	\$19.91	3	S N	NA NA	NA	NA	¥.	\$3.50	\$33.03	\$81.32	\$243.41	NA	TBD	TBD	TBD	\$1991	TBD	\$8.42	\$18.94	NA NA	X.	\$3.50	TBD	TBD	\$206.44	NA	TBD	TBD	GA
NA	NA	NA	NA	\$3.94	\$29.24	¥	\$3.50	\$26.04	\$121.40	\$136.39	\$218.90	NA	\$29.90	\$23.67	\$17.75	¥	N	₹ ₹	X.	\$3.94	\$29.24	NA	\$3.50	\$33.69	\$133.53	\$138.67	\$221.10	\$22.90	\$36.12	\$30.69	NA	3	¥ ¥	¥	\$3.94	\$29.24	Z.	\$3.50	\$33.50	\$138.67	\$221.19	NA	\$22.90	\$36.12	\$30.69	NA	N S	NA .	ψ3.9 <del>4</del>	\$29.24	NA.	\$3.50	\$26.79	\$119.14	\$192.57	NA		\$13.62	۲ ج
NA	\$11.41	\$8.06	\$18.14	NA	¥	Ā	\$3.50	\$25.91	\$120.82	\$136.19	\$218.59	NA	\$32.27	\$24.09	\$19.34	NA	\$11.41	\$8.06	\$18.14	NA	NA	NA	\$3.50	\$34.25	\$135.74	\$138.47	\$220.87	\$25.61	\$27.00	\$25.90	NA	φ11.41	\$8.06	\$18.14	NA	NA	NA.	\$3.50	\$34.25	\$138.47	\$220.87	NA	\$25.61	\$27.00	\$25.90	NA	\$11.41	\$8.06	\$18 14	NA NA	¥	\$3.50	\$26.66	\$118.57	\$192.30	NA	\$21.56	\$13.36	RATES BY STATE
NA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	A	IBU	d 8	ТВО	TBD	TBD	TBD	TBD	TBD	TBD		<b>B</b> 8	<b>3</b> B	TBD	TBD	TBD	-	3 8	тво	TBD	TBD	TBD	TBD			TBD	TBD	TBD	ПВП	TBJ .	NA	OBI	Ħ i	<b>B B</b>	TBD	TBD	TBD	OBIT	围	曹	TBD	TBD	ПВD	
NA	AN	NA	AN	NA	ΑN	Ā	\$3.50	\$21.09	\$92.57	\$105.88	\$202.01	NA	36,27	\$31.61	\$19.63	AN	Ä	3	×	NA	NA	NA	\$3.50	\$25.82	.50.50	\$144.28	\$2.36.28 	\$41.37	\$35.92	\$21.91	NA	AN	₹ ₹	NA NA	NA	NA	N.	\$3.50	\$25.50	\$144.28	\$226.36	NA	\$41.37	\$35.92	\$21.91	NA	\$15.12	\$12.76	\$26.04	NA AN	¥.	\$3.50	\$14.68	10.04¢	\$122.52	NA	\$16.11	\$14.10	N
NA	NA	NA	NA	NA	¥	¥	\$3.50	NA	NA	\$126.38	\$293.73	NA	¥	¥	NA	\$24.33	Ä	₹ ₹	¥	NA.	NA.	NA	\$3.50	Ā	NA.	\$91.92	\$23.78 	\$	¥	NA	\$10.22	3	₹ ₹	X X	NA	NA	¥.	\$3.50	NA A	\$91.92	\$232.76	NA	¥	¥	NA i	\$10.22	NA S	NA S	Z 3	NA NA	X.	\$3.50	¥.	NA S	\$199.27	NA.	NA	¥	SC
NA	TBD	TBD	TBD	TBD	TBD	ТВО	TBD	TBD	TBD	TBD	TBD	NA A	TBD	ТВО	TBD	¥	<u> </u>	3 8	ТВО	TBD	TBD	TBD	TBD	TBD	TBD	TEN CO	<b>B 5</b>	3	ТВО	TBD	NA.	0	3 8	ТВО	ТВО	TBD	da i	B B		<b>B B</b>	TBD	NA	TBD	TBD	TBD .	Z	ᇜ	H i	3 8		ТВО	TBD	UBD C	<b>B B</b>		NA	TBD	ТВD	Ź

4-Wire 2.4 KBPS Digital Unbundled Sub-Loop Feeder, per month

SOMAN SOMAN SOMAN

TBD TBD NA

TBD TBD NA

\$29.24 \$3.94 NA NA

NA NA \$18.14 \$8.06 \$11.41 \$16.29

NA \$209.61 \$127.09 \$119.80 \$25.07 \$25.07 \$2.75 \$0.42 \$21.56 \$3.84 NA NA NA

N

\$126.27 \$26.43 \$3.50 \$209.61 \$127.09

\$209.31 \$126.91 \$128.36 \$26.87 \$3.50 NA

\$90.12 \$19.44 \$3.50

₹ ₹

\$3.50

USBFK USBFK

TBD
TBD
TBD
NA
\$212.90
\$131.14
\$131.15
\$34.66

NA \$209.77 \$127.26 \$133.53 \$33.69

\$27.16 \$24.93 \$25.05 NA \$209.47 \$127.07 \$135.74 \$34.25 \$3.50 NA

NA NA NA NA \$309.57 \$157.93 NA

\$3.50 NA \$29.24

\$3.94

₹

\$27.38 \$33.41 \$24.47

¥

₹

₹

\$32.47

\$16.31

\$45.34

\$45.43

¥¥¥

USBFJ USBFJ USBFJ SOMEC SOMEC

NA \$206.32 \$134.23 \$123.01 \$26.53 \$3.50

USBFJ

¥

₹

NA \$207.14 \$134.77

B B 5.01

\$20.59 \$21.48 \$17.70

\$16.55 \$15.35 \$12.52

NA \$20.58 \$14.96 \$13.15

\$14.68 \$23.74 \$27.26

\$15.63 NA NA NA NA NA \$294.59 \$152.62

NRC Disconnect Charge - 1st

NRC Disconnect Charge - Add'I

NRC Service Order submitted Electronically, per LSR

NRC Service Order submitted Electronically, per LSR - Disconnect

NRC Service Order submitted Manually, per LSR - Disconnect

NRC Service Order submitted Manually, per LSR Disconnect

NRC Service Order submitted Manually, per LSR Disconnect

NRC Incremental Charge - Manual Service Order - 1st

NRC - Incremental Charge - Manual Service Order - Add'I

NRC - Incremental Charge - Manual Service Order - Disconnect

NRC - Incremental Charge - Manual Service Order - Disconnect

NRC - Incremental Charge - Manual Service Order - Disconnect

NRC - Incremental Charge - Manual Service Order - Disconnect

NRC - Incremental Charge - Manual Service Order - Disconnect

NRC - Incremental Charge - Manual Service Order - Disconnect

Wire Copper Unbundled Sub-Loop Feeder, per month
Zone 1, per month
Zone 2, per month
Zone 3, per month
Zone 4, per month
NRC 1st
NRC - Addfl

Zone 1, per month

Zone 2, per month

Zone 3, per month

Zone 3, per month

Zone 4, per month

Zone 4, per month

Zone 4, per month

RC - Disconnect Charge - 1st

NRC - Disconnect Charge - 1st

NRC - Service Order submitted Electronically, per LSR

NRC - Service Order submitted Electronically, per LSR

NRC - Service Order submitted Manually, per LSR

NRC - Service Order submitted Manually, per LSR

NRC - Service Order submitted Manually, per LSR

NRC - Nervice Order submitted Manually, per LSR

NRC - Incremental Charge - Manual Service Order - 1st

NRC - Incremental Charge - Manual Service Order - 4dd1

Attachment 2 Exhibit C Rates - Page 13

2-Wire Copper Unbundled Sub-Loop Feeder, per month

Zone 1, per month

NRC - Incremental Charge - Manual Service Order - Disconnect

SOMAN SOMAN SOMAN SOMAN

TBD

₹

TBD

¥

\$11.41 NA \$18.14 \$8.06

TBD

K

\$12.76

\$21.56 \$3.84 NA

\$29.24 \$3.94 NA

Disconnect Charge - Add'l
 Service Order submitted Electronically, per LSR
 Service Order submitted Electronically, per LSR - Disconnect
 Service Order submitted Manually, per LSR
 Service Order submitted Manually, per LSR, Disconnect
 Incremental Charge - Manual Service Order - 1st
 Incremental Charge - Manual Service Order - 4st
 Incremental Charge - Manual Service Order - 4st

SOMEC SOMEC

\$211.55 \$129.04 \$127.78 \$133.06 \$2.75 \$0.42

\$209.47 \$127.07 \$135.74 \$309.79 NA

\$393.01 \$153.37 NA

\$34.25 \$3.50 NA

\$3.50 NA NA NA \$42.19

\$3.50 NA

Zone 2, per month
Zone 3, per month
Zone 4, per month

NRC - 1st

NRC - 1st

NRC - Service Order submitted Electronically, per LSR
NRC - Service Order submitted Electronically, per LSR - Disconnect
NRC - Service Order submitted Electronically, per LSR - Disconnect
NRC - Service Order submitted Manually, per LSR - Disconnect
NRC - Service Order submitted Manually, per LSR, Disconnect
NRC - Service Order submitted Manually, per LSR, Disconnect
NRC - Incremental Charge - Manual Service Order - 4Adf I
NRC - Incremental Charge - Manual Service Order - Disconnect
NRC - Incremental Charge - Manual Service Order - Disconnect
NRC - Incremental Charge - Manual Service Order - Disconnect
NRC - Incremental Charge - Manual Service Order - Disconnect

Disconnect

USBFH

\$10.59
TBD
TBD
TBD
TBD
NA
\$172.20
\$90.45
\$114.73
\$21.82
\$3.50

NA \$11.01 \$9.78 \$7.83 NA \$175.18 \$92.66 \$113.67 \$20.84 \$2.75 \$0.42

\$8.29 \$7.30 \$6.03

NA \$8.81 \$7.72 \$5.93 \$174.93 \$92.53

\$10.78 NA NA

\$175.18 \$92.66 \$116.78

\$21.41 \$3.50 NA

NA \$10.66 \$16.46 \$18.69 NA \$172.89 \$90.81 \$89.0.81 \$88.27 \$16.79 \$16.79 \$16.79 \$3.50 NA NA NA

\$3.50

\$259.80 \$106.45 NA

\$116.22 \$21.31 \$21.13 \$21.13 \$3.50 NA NA NA NA NA NA S18.14 \$8.06 \$11.41

USBMC SOMAN SOMAN SOMAN

\$51.29

\$16.31

\$16.29

\$45.34

\$45.53

**XXXX** 

\$21.56 \$3.84 NA

TBD TBD NA NA

\$29.24 \$3.94 NA NA

SOMEC SOMAN SOMAN

4-Wire 4.8 KBPS Digital Unbundled Sub-Loop Feeder, per month

Zone 1, per month

Zone 2, per month

Zone 3, per month

Zone 4, per month

Zone 2, per month
Zone 3, per month
Zone 4, per month

NRC - Incremental Charge - Manual Service Order

Disconnect

SOMAN

₽

₹ ₽

TBD GA

≨ ঽ

\$11.41

B S

N SC

룡컬

RATES BY STATE

USBFL

\$212.90 \$131.14 \$131.15

TBD GBT

USBFL USBFL

1BD 1BD 1BD 1BD

\$24.89 \$28.83 \$29.16 NA

\$27.38 \$33.41 \$24.47

\$27.16 \$24.93 \$25.05

\$26.71 \$44.07 \$50.83

\$32.47 NA NA

\$309.57 \$157.93 NA

annel capacity - channels 1-96), per month	NRC-Disconnect Addi ICT8B			TR008 - System B (96 channel capacity - channels 97-192), per month	Add'l	Disconnect, 1st	- Add'l	NRC - 1st UCT8A		NRC - Incremental Chartes - Manual Service Order - Add'i		Disconnect	Ciacci III acc	NRC - Service Order submitted Electronically, per LSR - Disconnect SOMEO	NDC Society Cranton - Channelization bys (Outside CO)	Prit I am Company of the section (Contains (Co.)	NRC - Service Order submitted Manually, per LSR, Disconnect SOMAN	NRC - Service Order submitted Manually, per LSR SOMAN	- Disconnect	NRC - Service Order submitted Electronically, per LSR SOMEC	per pair	NRC - UNTW Pair, per pair UENPP	Unbundled Network Terminating Wire, per pair, per month	number of spare facilities per mechanized LMUSI is 10.**	number of spare facilities per mechanized LMUSI is 10.**	(Mechanized) **  Loop Makeup - Preordering Without Reservation, per spare facility guerned (Mechanized) Max	NKC-Loop Makeup - reproteing with Reservation, per spare facility queried (Manual) max  NMCLP  NMC1 - Loop Makeup - Preordering Without Reservation nor working facility quarter  NMC1 - Loop Makeup - Preordering Without Reservation nor working facility quarter  NMC1 - Loop Makeup - Preordering Without Reservation nor working facility quarter  UMKLP	number of spare facilities per manual LMUSI is 3. **  UMKLW	NRC - Loop Makeup - Preordering Without Reservation, per working facility queried (Manual) ** UMKLW	- Bridge Tap Removal per pair unbaded - Add'l	NRC - Bridge Tap Removal per pair unbaded - 1st  ULMBT	NRC - Load CoilEquipment Removal per 4 Wire pair - 1st ULM4X		Unbundled Sub-Loop Modification  NRC - Load Coil/Equipment Removal per 2 Wire pair - 1st  ULN2X		onnect	NRC - Incremental Charge - Manual Service Order - ISt  NRC - Incremental Charge - Manual Service Order - Add'I SOMANI	Disconnect		LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR SOMEC			NRC - 1st USBFP		Zone 3, per month		Inbundled Sub-Loop Feeder, per month		onnect	remental Charge - Manual Service Order - Add'l	DECORPTION
T3A	T8B	T8B -85	T8B	T8B	T8A	T8A	T8A	TBA	TRA	MAN	MAN	SOMAN	MAN	S C	0		MAN	MAN	SOMEC	MEC	UENPP	NPP	NPP				Σ P	KLW KLW	XL W	MBT	WBT	W4X	V2X	V2X		SOMAN	MAN	SOMAN	MAN	MEC	SOMEC		BFP	BFP	BFP	USBFP	ק קק	USBFP		MAN	MAN S	3
\$868.77	\$74.90	\$226.32	\$404.46	\$87.34	\$74.90	\$236.23	\$220.32	\$404.46	\$82976	3 8	g ;	₹ 5	N 5	\$3.50	9		¥	¥	NA	\$3.50	\$0.87	\$40.02	\$0.49	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	\$10.09	\$561.91	\$358.90	\$8.08	\$358.90		TBD	3 5	; ≼	¥	NA.	\$3.50	\$34.66	\$131.14	\$212.90	¥			\$30.97		TBD I	E P	≥
\$512.86	\$74.84	\$226.02	\$408.22	\$85.12	\$74.84	\$236.02	\$222.37	\$408.22	\$477.76	NA S	NA C	\$3.84	\$21.56	\$0.75	25 CE		\$3.84	\$21.56	\$0.42	\$2.75	¥	\$65.35	\$0.46	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	\$10.19	\$562.71	\$357.81	\$8.15	\$357.81		X S	N A	\$3.84	\$21.56	\$0.42	\$2.75	\$32.01	\$128.81	\$211.32	NA.	\$29.16	\$28.83	NA NA		¥ ;	× ;	<u>n</u>
\$764.42	N N	\$310.8Z	\$632.36	\$92.91	NA	NA.	\$310.82	\$632.36	\$72479	\$8.42	\$18.94	N S	NA S	\$3.50	70		NA	N <sub>A</sub>	NA	\$3.50	\$1.74	\$2.48	\$1.37	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	TBD	TBD	TBD	TBD	TBD		TBD	# E	Z N	NA	NA	N i	TB E	TBD	TBD	NA	E G	T 0	₹ 	i	TBO	TBD §	5
\$530.75	\$80.50	\$252.37	\$408.23	\$82.15	\$80.50	\$253.87	\$222.37	\$408.23	\$493.89	TBO	TRO.	\$3.94	\$29.24	\$3.50	3		\$3.94	\$29.24	NA	\$3.50	NA	\$65.35	\$0.64	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	\$10.19	\$562.73	\$357.82	\$8.15	\$357.82		¥ 5	N K	\$3.94	\$29.24	NA	\$3.50	\$133.53	\$127.26	\$209.77	NA.	\$24.47	\$33.41	82 7 C3		¥ ;	¥ 3	<b>5</b>
\$584.28	\$81.83	\$258.08	\$407.65	\$87.89	\$81.83	\$258.08	\$222.06	\$407.65	\$546.21	TBIO	TBO:	\$3.94	\$29.24	\$3.50	93		\$3.94	\$29.24	NA	\$3.50	NA	\$64.77	\$0.35	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	\$9.36	\$521.03	\$332.79	\$7.49	\$332.79		\$11.41	\$18.14	\$3.94	\$29.24	NA	\$3.50	\$135.74	\$127.07	\$209.47	NA NA	\$25.05	\$24.03	%37.16		\$11.41	\$8.06	RATES BY STATE
TBD	TBD	T E	TBD	TBD	TBD	TBD	TBD	TBD	TB IS	TB IS	TB is	TBD	T R	T B	3		TBD	TBD	TBD	TBD	TBD	OBT	Œ	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	ТВО	围	間	TBD	JBD		TBD	<b>B B</b>	TBD	TBD	TBD	TBD	<b>4 5</b>	曹	TBD	TBD	<b>B</b>	Ħ 8	5 €		TBO	图	
\$615.62	N S	\$2/8.83	\$587.83	\$85.59	¥	¥	\$278.83	\$587.83	\$579.03	NA 5	NA C	TBD	TBD	TRD	9		TBD	TBD	TBD	\$3.50	NA	\$65.82	\$0.44	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	\$10.14	\$564.15	\$360.33	\$8.11	\$360.33		X S	N A	¥	NA	NA	\$3.50	\$96.50	\$132.92	\$215.00	NA.	\$50.83	\$44.07	NA NA		Z.	¥ 8	5
\$611.62	N S	\$277.97	\$586.42	\$87.65	N <sub>A</sub>	¥	\$277.97	\$586.42	\$571.42	NA S	NA I	TBI C	d d	TRD	9		TBD	TBD	TBD	\$3.50	NA	\$60.93	\$0.46	\$1.08	\$1.08	\$1.08	\$140.00	\$134.00	\$134.00	\$8.07	\$561.62	\$358.71	\$8.07	\$358.71	+	¥ 5	$^{+}$	+	+	H	+	+	+			$^{+}$	+	+	Ħ	<b>⊼</b>	N S	ŝ
\$726.87	N S	\$311.78	\$634.31	\$102.12	N <sub>A</sub>	¥.	\$311.78	\$634.31	\$683.79	NA 5	NA S	NA S	\$19.99	\$3.50	3		TBD	TBD	OBT	OBT	TBD	OBT.	ПВП	\$0.6888	\$0.6888	\$0.6888	\$100.00	\$100.00	\$100.00	ТВО	間	TBO	OBT	TBD		TBD	<b>3 3</b>	TBD	TBD	TBD	TBD	<b>3 5</b>	d de	TBD	NA	da Ga	Ħ 0	5 ₹	i	TBD is	d a	TN

-		П						_	⇉				#			Ш							_		$\Box$			#			1						П	_ _				ш	$\Box$			_	_			4		Д.	
NRC - 1st	TR008 -System B (96 channel capacity - channels 97-192), per month	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC - 1st	TR008 -System A (96 channel capacity - channels 1-96), per month	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Service Order submitted Electronically, per LSR - Disconnect	NRC - Service Order submitted Electronically, per LSR	Loop Concentration System (Inside C.O.)	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC Addi	Channel Interface - Digital 64Kbps, per month	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC Add'l	Channel Interface - Digital 56Kbps, per month	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC Ist	Test Circuit, per month	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC 1st	erface -	NRC-Disconnect, Add'l	NRC Add'i	. NRC 1st		NRC-Disconnect, 1st	NRC Add'l	NRC 1st	NRC-Disconnect, Addii	NRC-Disconnect, 1st	NRC Add'l	Channel Interface-2 Wire Voice-Loop Start or Ground Start, per mo	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC 1st	Zone 4, per month	Zone 3, per month	Zone 2 per month	DS1 Feeder Interface, per month	NRC-Disconnect, Add'l	NRC - Add'l	NRC - 1st	TR303 - System B (96 channel capacity - channels 97-192), per month		NRC - Add'l	NRC - 1st	DESCRIPTION
UCI8B	UCT8B	UCT8A	UCT8A	UCTBA	UCT8A	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC		ULCC6	ULCC6	ULCC6	ULCC6	ULCC5	ULCC5	ULCC5	ULCC5	UCTTC	UCTTC	CTTC	UCTTC	ULCC4	ULCC4	ULCC4	ULCC4	ULCCR	ULCCR	ULCCR	ULCCR	ULCC1	ULCC1	ULCC1	ULCC2	ULCC2	ULCC2	ULCCZ	UCTFS	UCTFS	UCTES	UCTFS	UCTFS	UCIFS	UCTFS	UCT3B	UCT3B	<b>UCT3B</b>	UCT3B	UCT3A	UCT3A	UCT3A	USOC
\$464.57	\$67.41	NA :	<b>₹</b> ₹	\$1,115.10	\$327.44	TBD	TBD	¥:	<b>₹</b>	\$3.50	:	\$9.94	\$10.00	\$20.88	\$12.49	\$9.94	\$10.00	\$20.77	\$12.49	\$9.94	\$10.00	\$20.77	\$41.21	\$9.94	\$10.00	\$20.88	\$8.43	\$9.94	\$20.77	\$20.88	\$14.14	\$9.94	\$20.77	\$20.88	\$9.94	\$10.00	\$20.77	\$20.88	\$12.81	\$56.94	\$192.51	NA	X S	N N	\$67.03	\$74.90	\$220.32	\$404.46	\$126.35	\$236.23	\$220.32	\$404.46	A
\$271.27	\$55.96	NA :	<b>₹</b> ₹	\$651.05	\$470.73	NA	NA.	\$3.84	\$21.56	\$2.75		\$9.93	\$9.99	\$20.96	\$11.14	\$9.93	\$9.99	\$20.96	\$11.14	\$9.93	\$9.99	\$20.96	\$36.76	\$9.93	\$9.99	\$20.96	\$7.52	\$9.93	\$20.96	\$21.07	\$12.61	\$9.93	\$20.96	\$21.07	\$9.93	\$9.99	\$20.96	\$2.12	\$33.06	\$127.78	\$211.55	NA	\$107.08	\$65.86	NA	\$74.84	\$222.37	\$408.22	\$120.21	\$236.02	\$222.37	\$408.22	_ 
\$463.37	\$65.27	NA :	<b>₹</b> 5	\$1,111.95	\$316.63	\$8.42	\$18.94	¥.	<b>X</b>	\$3.50	:	NA	NA S	\$41.82	\$12.51	¥	NA	\$41.58	\$12.51	NA NA	NA	\$41.58	\$41.30	¥	¥	\$41.82	\$8.45	<b>₹</b> 5	\$41.58	\$41.82	\$14.17	¥ ¥	\$41.58	\$41.82	\$0 53	NA	\$41.58	\$2.38	NA	NA SO	\$425.74	NA	TB G	# E	\$72.12	<b>₹</b> 5	\$310.82	\$632.36	\$132.54	X X	\$310.82	\$632.36	GA
\$271.27	\$63.59	NA :	<b>₹</b> 5	\$651.04	\$522.17	TBD	TBD	\$3.94	\$29.24	\$3.50	:	\$10.68	\$10.75	\$20.96	\$14.08	\$10.68	\$10.75	\$20.96	\$11.79	\$10.68	\$10.75	\$20.96	\$38.90	\$10.68	\$10.75	\$21.08	\$7.96	\$10.68	\$20.96	\$21.08	\$13.35	\$10.75	\$20.96	\$21.08	\$8 08	\$10.75	\$20.96	\$21.08	\$33.59	\$133.43	\$209.62	NA	\$106.11	\$60.01	AN	\$80.50	\$222.37	\$408.23	\$119.01	\$253.87	\$222.37	\$408.23	۲ ج
\$270.88	\$61.71	NA :	<b>₹</b> 5	\$650.11	\$432.54	\$8.06	\$18.14	\$3.94	\$29.24	\$3.50	:	\$10.86	\$10.92	\$20.05	\$12.27	\$10.86	\$10.92	\$20.93	\$12.27	\$10.86	\$10.92	\$20.93	\$40.49	\$10.86	\$10.92	\$21.05	\$8.29	\$10.86	\$20.93	\$21.05	\$13.89	\$10.92	\$20.93	\$21.05	\$10.86	\$10.92	\$20.93	\$21.05	\$34.25	\$135.74	\$209.47	NA	\$187.11	\$88.79	AN	\$81.83	\$222.06	\$407.65	\$125.95	\$258.08	\$222.06	\$407.65	RATES BY STATE
\$464.71	\$73.30	NA :	<b>₹</b> 5	\$1,115.10	\$454.79	\$11.34	\$25.52	¥.	¥ ¥	\$3.50	:	TBD	TBD	TBN E	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD CBT	TBD	TBD	TBD	TBD IBD	TBD	OBI	TBI IBI	TBD	TBD	TBD	TBD	TBD	TBD	TBD	OBT.	TB IB	TBD	TBD CBT	TBD	TBD	TBD	TBD	TBD	UBI	
									<b>X</b> ¥			NA	NA S	\$28.50	\$11.66	¥	NA	\$28.50	\$11.66	N N	NA	\$28.50	\$38.47	¥	¥	\$28.56	\$7.87	<b>₹</b> §	\$28.50	\$28.66	\$13.20	<b>₹</b>	\$21.13	\$21.24	\$8.88 NA	NA	\$28.50	\$28.66	NA	NA S	\$237.09	NA	OBT.	TB IB	\$64.63	<b>₹</b> §	\$278.83	\$587.83	\$122.18	X X	\$278.83	\$587.83	N.C.
	_	_	-	_	_	-		-	<b>8</b>	_	_	NA	NA ST	\$28.58	\$12.77	¥	NA	\$28.42	\$12.77	ž K	N.	\$28.42	\$42.14	¥	¥	\$28.58	\$8.62	<b>₹</b> §	\$28.42	\$28.58	\$14.46	<b>X</b> X	\$28.42	\$28.58	\$0.72	NA	\$28.42	\$28.58	NA	NA S	\$236.32	NA	TBO	<b>H</b>	\$69.24	<b>₹</b> §	\$277.97	\$586.42	\$127.76	<b>X X</b>	\$277.97	\$586.42	ŝ
\$464.21	\$68.71	NA :	<b>₹</b> 3	\$1,114.05	\$380.06	TBD	TBD	Z.	\$19.99	\$3.50	:	NA	¥ :	\$41.95	\$13.71	NA.	NA :	\$41.71	\$13.71	NA.	NA	\$41.71	\$45.22	¥	¥.	\$41.95	\$9.26	<b>₹</b> 5	\$41.71	\$41.95	\$15.51	¥ ¥	\$41.71	\$41.95	\$10.43		\$41.71	\$2.61	NA	NA SO	\$418.37	AN	OBT.	<b>4 5</b>	\$76.73	¥ 5	\$311.78	\$634.31	\$145.21	X X	\$311.78	\$634.31	Ĭ

				Ħ	l			Ę		Ė				$\dagger$	Ė						1		H									L						L			l			1	Ė				L	Ħ	DE.
NRC - Disconnect Add'l **	NRC - Disconnect 1st **	NRC - Addl **	System Splitter - 96 Line Capacity  R.C Per month **		NRC - Additivators 3) ***	RC - per month (Note 3) **	2-Wire analog VG (SL1) for Line Sharing	LINE SHARING	NRC-Disconnect, Addi	NRC-Disconnect, 1st	NRC Add'l	NRC 1st	Channel Interface - Digital 64Kbps, per month	NRC-Disconnect Add'l	NRC Add'I	NRC 1st	Channel Interface - Digital 56Kbps, per month	NRC-Disconnect, 1st	NRC Add'l	NRC 1st	Test Circuit, per month	NRC-Disconnect, 1st	NRC Add'I	NRC 1st	Channel Interface - 4 Wire Voice, per month	NRC-Disconnect, 1st	NRC Add'l	Channel Interface - 2 Wire Voice - Reverse Battery, per month	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC 1st	Channel Interface - 2 Wire ISDN, per month	NRC-Disconnect, Add'l	NRC Add'I	NRC 1st		NRC-Disconnect, 1st	NRC Add'I	NRC 1st	DS1 Interface, per month	NRC-Disconnect, 1st	NRC - Add'I	NRC - 1st	TR303 - System B (96 channel capacity - channels 97-192), per month	NRC-Disconnect, 1st	NRC - Add'l	NRC - 1st	NRC-Disconnect, Add'l	NRC-Disconnect, 1st	NRC - Add'l	DESCRIPTION
		ULSDA							ULCC6	ULCC6	ULCC6	ULCC6	ULCC6	ULCC5	ULCC5	ULCC5		CTC				ULCC4			ULCCA		ULCCR							ULCC2			UCTCO						UCT3B		UCT3A		UCT3A \$		UCT8B	UCT8B	USOC
\$0.00	\$150.00	\$150.00	\$100.00	-		ТВD			BU	TBD	TBD	da G	명	<b>B B</b>	曹	TBD	TBD	\$35.55	\$35.55	\$35.77	\$44.16	<b>₹</b>	\$35.55	\$35.77	\$9.04	¥	\$35.55	\$35.77	Ž.	N.	\$35.77	\$10.19	¥.	\$35.55	\$35.77	\$2.55	¥ ₹	\$132.03	\$367.70	\$6.42	8 8	¥	\$464.57	\$111.30	<b>X X</b>	NA	\$1,115.10	NA NA	¥	NA A	AL
\$0.00	\$150.00	\$150.00	\$100.00	100		ПВD			\$9.93	\$9.99	\$20.96	\$21.07	\$11.01	\$9.99	\$20.96	\$21.07	\$11.01	\$9.99	\$20.96	\$21.07	36.31	\$9.99	\$20.96	\$21.07	\$7.43	\$9.99	\$20.96	\$12.46	\$9.93	\$9.99	\$20.96	\$8.38	\$9.93	\$20.96	\$21.07	\$2.10	\$8.71	\$92.17	\$126.61	\$5.28	N X	¥	\$271.27	\$94.30	X X	NA	\$651.05	% NA	NA.	NA	7
\$0.00	\$150.00	\$150.00	\$100.00	9	T BD	TBD			NA	. A	TBD	TBD	TBD :	NA KA	TBD	TBD	TBD	X X	\$35.48	\$35.68	\$42.30	X X	\$35.48	\$35.68	\$8.65	X X	\$35.48	\$35.68	Z N	NA	\$35.68	\$9.76	NA :	\$35.48 NA	\$35.68	\$2.44	N N	\$130.63	\$366.72	\$6.15	N N	¥.	\$463.37	\$110.02	3 X	NA	\$1,111.95	NA AN	NA	NA	GA
\$0.00	NA S	\$300.00	\$100.00	-	TBD	TBD			\$10.68	\$10.75	\$20.96	\$21.08	\$12.60	\$10.75	\$20.96	\$21.08	\$12.60	\$10.75	\$20.96	\$21.08	\$41.58	\$10.75	\$20.96	\$21.08	\$8.51	\$10.75	\$20.96	\$14.26	\$10.68	\$10.75	\$20.96	\$9.59	\$10.68	\$10.75	\$21.08	\$2.40	\$9.37	\$92.17	\$126.61	\$6.04	¥ ¥	\	\$271.27	\$107.16	5 ₹	NA	\$651.04	SEC Z 24	¥	NA	~
\$0.00	\$150.00	\$150.00	\$100.00	-	TBD	TBD			\$10.86	\$10.92	\$20.93	\$21.05	\$12.33	\$10.92	\$20.93	\$21.05	\$12.33	\$10.92	\$20.93	\$21.05	\$40.67	\$10.92	\$20.93	\$21.05	\$8.32	\$10.92	\$20.93	\$13.95	\$10.86	\$10.92	\$20.93	\$9.39	\$10.86	\$10.93	\$21.05	\$2.35	\$9.52	\$92.04	\$126.43	\$5.91	N X	¥	\$270.88	\$103.99	X X	NA	\$650.11	AN AN	NA.	NA	Ā
\$0.00	NA S	\$300.00	\$100.00	-	TBD	TBD			IBD	TBD	TBD	TBD	TBD	<b>3 8</b>	TBD	TBD	TBD	<b>×</b> ×	\$35.37	\$35.78	\$47.85	<b>X</b>	\$35.37	\$35.78	\$9.83	<b>X X</b>	\$35.37	\$35.78	NA NA	NA.	\$35.78	\$11.10	X.	\$35.37	\$35.78	\$2.77	N X	\$132.07	\$367.80	\$6.99	X X	¥	\$464.71	\$123.52	X X	NA	\$1,115.10	NA NA	¥	NA	W
\$0.00	NA S	\$300.00	\$100.00	-	TBD	TBD			T			T					TBD §		+	H	+	+		+	+	+	\$35.51	+	+		+	+	H	+	+	H	+	+		$\dashv$	+		$\vdash$	+	<b>3</b>		\$1,113.00				
_	+	\$300.00			TBD	TBD			IBU	TBD	TBD	TBD	TBD	E E	TBD	TBD	TBD	X X	+	ш	_	+		-	-	+	\$35.71	_	-		_	+-	$\vdash$	_	_		-	+		-	_	_	H	_	¥ ¥	-	\$1,119.30	_	_	$\vdash$	_
\$0.00	\$150.00	\$150.00	\$100.00	\$20.02	\$31.99	\$12.16			B	TBD	TBD	TBD	OBT OBT	<b>B B</b>	問問	TBD	TBD §	3 8		\$35.74	+		Н				\$35.54				_							+	$\vdash$	+	+	+	H	\$115.79	<b>.</b>		\$1,				_

	DESCRIPTION	RC - Per	NRC - 1st **	NRC - Addl **	NRC - D	NRC - D	Loop Ca	RC - Per	NRC - 1st **	NRC - Addl **		NRC - S	NRC - S	NRC - S	NRC - S NRC - S	NRC - S NRC - S NRC - S	NRC - S NRC - S NRC - S NRC - S NRC - In	NRC - S NRC - S NRC - S NRC - In NRC - In	NRC - S NRC - S NRC - In NRC - In NRC - In	NRC - Service NRC - Service NRC - Service NRC - Service NRC - Increa	NRC - Servicion NRC - Servicion NRC - Servicion NRC - Servicion NRC - Increm NRC - Increm NRC - Increm NRC - Add **	NRC - S. NRC - I. I. I. NRC - I. I. NRC - I.	NRC S NRC S NRC S NRC S NRC S NRC IN NRC IN	NRC - S NRC - S NRC - S NRC - IN NRC - IN NRC - IN NRC - IN NRC - A NRC - TN INT INT INT INT INT INT INT INT INT I	NRC - S NRC - S NRC - S NRC - IN NRC -	NRC. S. NRC. S. NRC. S. NRC. S. NRC. S. NRC. In	NRC -S NRC -S NRC -S NRC -S NRC -S NRC -S NRC - In NRC -	NRC - S NRC - S NRC - S NRC - I NRC -
	RIPTION	RC - Per month **	31 **	ddl **	NRC - Disconnect 1st **	NRC - Disconnect Add'l **	_oop Capacity, Line Activation Per Occurrence	RC - Per Month **	31 ***	odi **	NDC - Service Order submitted Electronically por LSB	NDC - Setvice Order submitted Electronically per Los	NRC - Service Order submitted Manually per I SR	NRC - Service Order submitted Manually, per LSR, Disconnect	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - Disconnect	CONTROL CHARLES THE CONTROL CONTROL CONTROL	Subsequent Activity - Per Occurrence	ent Activity - Per Occurrence	ent Activity - Per Occurrence st **	ent Activity - Per Occurrence 1 ** 3d **  Rates subject to true-up	Subsequent Activity - Per Occurrence NRC - Addi ** ** Tit rains Rates subject to true-up. *** Tit raits are interim and subject to true-up.	Subsequent Activity - Per Occurrence NRC - 1st ** NRC - Add1 **  *Interim Rates subject to true-up  ** TN rates are interim and subject to true-up. S:	Subsequent Activity - Per Occurrence NRC - 1st ** NRC - Add **  *Interim Rates subject to true-up  ** TN rates are interim and subject to true-up.  Ss.  Is states, the applicable NRC from the appropriate tariff applies.	Subsequent Activity - Per Occurrence  NRC - 1st **  NRC - Add! **  ** Interim Rates subject to true-up  ** TN rates are interim and subject to true-up.  ** TN rates are interim and subject to true-up.  ** TN rates are interim and subject to true-up.  S.  S.  S.  S.  S.  S.  S.  S.  S.	Subsequent Activity - Per Occurrence  NRC - Add!**  Interim Rates subject to true-up.  **TN rates are interim and subject to true-up.  **TN rates are interim and subject to true-up.  **TN rates are interim and subject to true-up.  St.  St.  St.  St.  St.  St.  St.  S	Subsequent Activity - Per Occurrence  NRC - 1st **  NRC - Add **  **Interim Rates subject to true-up.  **Interim Rates subject to true-up.  **Thates are interim and subject to true-up.  **Thates are publicable interim and changes is not standard the applicable interiment and policially Deaveraged UNE Zones and applicable rates have been established. Statewide rates are bosolete. Further, BelSouth is in the process of enhancing its billing systems in order to accompdate this Geographically Deaveraged UNE Zone Rate Structure. Until these enhancements are accomplished, estimate to be made. Once billing enhancements are complete, all applicable. Further, BelSouth is in the be billed. Once billing enhancements are complete, all applicable UNE Zone a state reflected in this Agreement will be billed. Reference Internet Website in this Agreement will be billed. Reference Internet Website in this Agreement will be billed. Reference Internet Website in this Agreement will be billed. Reference Internet Website UNE Zone areas reflected in this Agreement will be billed. Reference Internet Website  That recurring interim and nomecurring interim rates in TN for 2-Wire analog VG (SL1) for Line Sharing is for a stand-alone loop purchased by NewSouth to provide both analog voice service and xDSL services or in the event NewSouth wishes to continue providing xDSL services to an end-user who terminates is BelSouth-provided voice service. These rates apply when NewSouth purchases the splitter from BelSouth.
	USOC	ULSDB	ULSDB	ULSDB	ULSDB	ULSDB		ULSDC	ULSDC	ULSDC	SOMEO	SOMEO	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN			ULSDS	ULSDS	ULSDS	ULSDS	ULSDS	ULSDS			5 0 5 0
	Ą	\$25.00	\$150.00	\$0.00	\$150.00	\$0.00		\$6.00	\$40.00	\$22.00	\$ 550 500	NIA	NA	¥ :	\$27.37	\$12.97	\$17.77		\$30.00		\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00	\$15.00
	7	\$25.00	\$150.00	\$0.00	\$150.00	\$0.00		\$6.00	\$40.00	\$22.00	\$2.75	\$0.42	\$21.56	\$3.84	AN	NA	NA		\$30.00	\$15.00								
	GA	\$25.00	\$150.00	\$0.00	\$150.00	\$0.00		\$6.00	\$40.00	\$22.00	\$3 AO	NIA	NA	¥ :	\$18.94	\$8.42	NA		\$30.00	\$15.00								
	ŔΥ	\$25.00	\$300.00	\$0.00	NA	\$0.00		\$6.00	\$40.00	\$22.00	\$3 AO	NIA	\$29 24	\$3.94	¥	NA	NA		\$30.00	\$15.00								
RATES BY STATE	F	\$25.00	\$150.00	\$0.00	\$150.00	\$0.00		\$6.00	\$40.00	\$22.00	#3 AO	NIA	NA	Z.	\$18.14	\$8.06	\$11.41		\$30.00	\$15.00								
TE	MS	\$25.00	\$300.00	\$0.00	NA	\$0.00		\$6.00	\$40.00	\$22.00	\$3 50 50	NIA	NA	N.	\$25.52	\$11.34	\$16.06		\$30.00	\$15.00								
	S	\$25.00	\$300.00	\$0.00	N	\$0.00		\$6.00	\$40.00	\$22.00	#2 70	NIA	NA .	¥ :	\$26.94	\$12.76	NA		\$30.00	\$15.00								
	SC	\$25.00	\$300.00	\$0.00	NA	\$0.00		\$6.00	\$40.00	\$22.00	#3 70	NIA	NA .	¥ :	\$44.22	\$13.55	NA		\$30.00	\$15.00								
	ĭ	\$25.00	\$150.00	\$0.00	\$150.00	\$0.00		\$3.48	\$40.00	\$21.39	\$3 AO	TBD (S	\$19 99	OBL	AN	AN	NA		\$30.00	\$15.00								

### NETWORK ELEMENTS AND OTHER SERVICES

																																				2-W	LOCA	DES
2-wire voice unbunded port with caller ID - residence 2-wire voice unbunded port outgoing only - residence	Non-Recurring Charges (NRC) - 1st (Residence)	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	(B2F)	2-wire voice unbundled TN Bus 2-Way Area Calling Port Standard Option (TACC2)	2-wire voice unbundled TN Bus 2-Way Area Calling Port Economy Option (TACC1)	2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	2-wire voice unbundled outgoing only port	2-wire voice unbundled port with unbundled port with Caller+E484 ID	2-Wire Voice Inhundled nort without Caller ID	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	E THE VIOLE INSTITUTE OF THE WASSE THE POINT THE COURT OF LOTH	(2MR)	(1MF2X)  2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID		2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	2-wire voice unbundled port outgoing only - residence	2-wire voice unbundled port vith caller ID - residence	2-Wire Voice Grade Line Port (Residence), per month	DOCAL EXCHANGE SWITCHING (PORTS)	CRIPTION
UEPRO	I IE PRI	LNPCX	UEPAE	UEPAD	UEPAC	UEPAB	UEPAA	UEPAV	UEPAZ	UEPAY	UEPAX	UEPBM	UEPAW	UEPBO	UEPBC	I IF PR	LNPCX		UEPAO	UEPAN	i j	UEPAM	UEPAL	UEPAK	UEPAJ	UEPAH	UEPAG	UEPAQ	UEPAU	UEPAT	UEPAS	UEPRM	UEPAR	UEPRO	UEPRC		!	USOC
\$21.93	\$21 93		¥	NA	NA	NA	NA 7	Ž ×	NA	NA	¥	NA	\$2.07	\$2.07	\$2.07	\$2.07		<b>#</b> 1:01	\$2.07 NA	×		Z S	Z F	¥	¥	NA	Z 3	¥	¥	NA	NA	NA	\$2.07	\$2.07	\$2.07			٨
\$4.76	\$4.76		NA	NA	NA.	NA	NA 71.62	N N	N <sub>A</sub>	N <sub>A</sub>	NA	N <sub>A</sub>	N <sub>A</sub>	\$1.62	\$1.62	\$1.60			\$1.62	¥		Z	¥	NA	N <sub>A</sub>	AN	NA .02	NA NA	NA	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	\$1.62	\$1.62 \$1.62		,	2
\$17.16	\$17.16		NA	NA	NA.	NA	\$1.85 AN	N N	N <sub>A</sub>	N <sub>A</sub>	NA	N <sub>A</sub>	N <sub>A</sub>	\$1.85	\$1.85	\$1.85		::0	\$1 85	¥		Z	¥	NA	N <sub>A</sub>	AN	X X	N.	NA	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	\$1.85	1.85 - Note			GA
\$37.78	\$37 78		¥	NA	N	NA	NA 7	Ž X	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	\$2.61	N <sub>A</sub>	\$2.61	\$2.61	\$2.61		9	\$2.61	¥		Z F	Z F	¥	N N	NA	X 3	¥	N <sub>A</sub>	NA	NA	\$2.61	N <sub>A</sub>	\$2.61	1 2.61 - Note \$2.61			হ
\$16.43	\$16.43		¥	NA	NA	NA	\$2.20	Ŝ.	NA	NA	\$2.20	NA	NA	\$2.20	\$2.20	\$2 20			\$2 20 NA	×		Z	Z F	N N	NA	\$2.20	\$2.20	NA NA	¥	NA	\$2.20	NA	NA	\$2.20	\$2.20			5
\$22.98	\$22 Q8		¥	NA	NA	NA	NA 7.11	Ž.≱	NA	\$2.11	NA	NA	NA	\$2.11	\$2.11	\$2 11		11	\$2 11	×		Z	N A	¥ Ā	N <sub>A</sub>	NA	¥ §	NA	N <sub>A</sub>	\$2.11	NA	NA	NA	\$2.11	\$2.11			SW
\$24.04	\$21.60		NA	N <sub>A</sub>	N <sub>A</sub>	NA	NA 00	NA NA	N <sub>A</sub>	N <sub>A</sub>	NA	NA	N <sub>A</sub>	\$2.00	\$2.00	\$2 00		ØE:00	\$2 NA	S		N N	₹	NA	NA	NA	X X	N <sub>A</sub>	NA	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	\$2.00	\$2.00			š
\$24.98 \$24.98	\$24 Q8		NA	N <sub>A</sub>	N <sub>A</sub>	\$2.35	NA 35	NA NA	\$2.35	N <sub>A</sub>	NA	NA	N <sub>A</sub>	\$2.35	\$2.35	\$2.35		£	%2 35 NA	S		N N	₹	NA	\$2.35	NA	X X	N <sub>A</sub>	\$2.35	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	\$2.35	\$2.35			SC
\$9.93	\$0.03		\$4.73	\$4.73	\$4.73	NA	NA 3	\$4.73	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	N <sub>A</sub>	\$4.73	\$4.73	\$4.73			\$4.73	\$4.73	) 	\$4.73	\$4.73	\$4.73	N <sub>A</sub>	NA.	8	\$4.73	N <sub>A</sub>	X.	N <sub>A</sub>	X.	N <sub>A</sub>	\$4.73	4.73 - Note 1 \$4.73			뒫

### NETWORK ELEMENTS AND OTHER SERVICES

																				1																			DES
2-wire voice unbundled outgoing only port	2-wire Voice Unbundled Port with Caller ID	NRC - 1st (Business)	NRC - Subsequent Activity	2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	unbundled Louisiana Area Plus with caller ID - residen	2-wire voice unbundled Horida area calling with caller ID - residence (RUL)	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	outgoing only -	unbundled port w	NRC - Add'I (Residence)	THE THE PERSON NAMED IN TH	2-wire voice unbundled Res Low Usage Line Port with Caller+E563 ID (LLM)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	2-wire voice unbundled Florida area calling with caller ID - residence	2-wire voice grade unbundled Tennessee extended local dialing parity port with	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	DESCRIPTION
UEPBO	UEPBL		USASC	UEPAP	UEPAO	UEPAN	UEPAM	UEPAL	UEPAK	UEPAJ	UEPAH	UEPAG	UEPAQ	UEPAU	UEPAT	UEPAS	UEPRM	UEPAR	UEPRO	UEPRC	- III	0	UEPAP	UEPAO	UEPAN	UEPAM	UEPAL	UEPAK	UEPAJ	UEPAH	UEPAG	UEPAF	1000	UEPAU	UEPAT	UEPAS	UEPRM	_	USOC
	\$21.93		\$10.00	\$21.93	¥	NA	NA	N	NA	¥	NA	¥ ¥	₹	NA	NA	NA	NA	\$21.93	\$21.93	\$21.93	\$21.03		<b>X</b>	N A	Z F	N <sub>A</sub>	NA	NA	NA	¥	NA.	N S	20	Z A	NA	₹	NA	\$21.93	AL
\$4.76	\$4.76		\$10.00	\$4.54	X A	¥	¥	¥	¥	N <sub>A</sub>	NA	\$4.54 NA	N N	¥	¥	¥	NA	X A	\$4.54	\$4.54	\$4 54		Z F	N <sub>A</sub>	¥	¥	¥	¥	¥	NA	NA	\$4.76		₹	NA	¥	¥	N <sub>A</sub>	7
\$17.16	\$17.16		\$10.00	\$17.16	N <sub>A</sub>	N <sub>A</sub>	N.	N N	NA	N <sub>A</sub>	AN	¥ ¥	¥.	N <sub>A</sub>	N N	×.	NA	N <sub>A</sub>		\$17.16	¢17 16		¥	N <sub>A</sub>	¥	N N	×.	N N	N <sub>A</sub>	A	NA.	X X	NIA	¥	NA	¥	¥	NA.	GA
\$37.55	\$37.55		\$10.00	\$37.78	N <sub>A</sub>	N <sub>A</sub>	N N	N N	NA	N <sub>A</sub>	AN	¥ ¥	¥.	N <sub>A</sub>	N N	×.	\$37.78	N <sub>A</sub>	\$37.78	\$37.78	\$37 78		¥	N <sub>A</sub>	¥	N N	×.	N N	N <sub>A</sub>	A	NA.	X X	NIA	¥	NA	¥	\$37.78	NA.	ক
\$16.43	\$16.43		\$10.00	\$16.43	NA	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	NA	\$16.43	\$16.43	X.	N <sub>A</sub>	N <sub>A</sub>	\$16.43	NA	NA	\$16.43	\$16.43	\$16.43		¥	N <sub>A</sub>	₹	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	\$16.43	\$16.43	\$16.43	N X	NIA.	Š	NA	\$16.43	N <sub>A</sub>	NA.	5
\$22.98	\$22.98		\$10.00	\$22.98	N <sub>A</sub>	NA	NA	NA	NA	N <sub>A</sub>	NA	X	¥	NA	\$22.98	NA	NA	N <sub>A</sub>		\$22.98	\$22 Q8		Z.	N.	₹	NA	NA	NA	NA	NA	NA.	¥ ₹		₹	\$22.98	¥	N <sub>A</sub>	¥.	MS
\$24.04	\$21.60		\$10.00	\$9.08	NA	N <sub>A</sub>	NA	N <sub>A</sub>	NA	NA	NA	¥ ¥	X.	N <sub>A</sub>	NA	N <sub>A</sub>	NA	NA	\$9.08	\$9.08	\$31.80		¥	NA.	¥	NA	N <sub>A</sub>	NA	N <sub>A</sub>	NA.	A	X X	NIA	Š	NA	₹	N	N	NC
\$24.98	\$24.98		\$10.00	\$24.98	NA	NA	NA	NA	NA	\$24.98	NA	X	¥	\$24.98	NA	NA	NA	N <sub>A</sub>	\$24.98	\$24.98	80 VC\$	1 41 4	¥.	N.	₹	NA	NA	\$24.98	NA	¥	NA.	¥ ₹		\$24.98	NA	¥	N <sub>A</sub>	N.	SC
\$9.93	\$9.93		\$10.00	\$9.19	\$9.19	\$9.19	\$9.19	\$9.19	\$9.19	N <sub>A</sub>	AN	¥ ¥	\$9.19	N <sub>A</sub>	N N	×.	NA	N <sub>A</sub>	\$9.19	\$9.19	\$0.10	40.00	\$9.93	\$9.93	\$9.93	\$9.93	\$9.93	N A	N <sub>A</sub>	A	¥	NA 99.93	200	¥	NA	¥	¥	NA.	T

F									ł																							H					Ŧ		DESC
z-wile voice unbulided codisiana Area Flus with caller ID - lesidence (AC/)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	2-wire voice unbundled Florida area calling with caller ID - residence	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	2-wire voice grade unbundied kemucky extended local draiing party port with caller	ID	2-wire voice unbundled port outgoing only - residence	2-wire voice unbundled port with caller ID - residence	NRC - Disconnect Charge - 1st	NRC - Subsequent Activity	2-wire voice unbunded i'N bus 2-way collenville and memphis cocali caling Fort (B2F)	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	2-wire voice uphundled incoming only nort with Caller ID	z2-wire voice grade unbundled South Carolina extended local dialing parity port with	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	2-wire voice unbundled outgoing only port	2-wire voice unbundled port with Caller ID  2-wire voice unbundled port with Caller ID	NRC - Add'l (Business)	(B2F)	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2) 2-wire voice unbundled TN Bus 2-way Collierville and Memphis Local Calling Port	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	, K	2-wire voice unbundled SC Bus Area Calling Port with Caller ID+E587 (LMB)	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	awire voice glade unbunden territessee extended local daring party port with caller ID	zawire voice grade unbunded sourn carolina extended local daling panty bott with	z-wire voice grade unbundled Mississippi extended local drailing parity port with caller ID	2-wire voice grade undurined coursiana extended local draining parity bott with caller	Z-wire voice grade unbundled Kentucky extended local dialing parity port with caller in	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	DESCRIPTION
OETAI	UEPAG	UEPAF	UEPAQ	UEPAU	UEPAT	UEPAS	UEPRM	UEPAR	UEPRO	UEPRC	LEPRI	USASC	UEPAE	UEPAD	UEPAC	BAABN	UEPAA	UEPAV	UEPAZ	UEPAY	UEPAX	UEPBM	UEPAW	OBPBO	OEPBC		UEPAE	UEPAD	UEPAC		UEPAB	UEPAA	UEPAV	UEPAZ	UEPAY	UEPAX	UEPBM	UEPAW	osu
3	3 3	NA	¥	¥	NA	NA	NA	\$6.21	\$6.21	\$6.21	\$6.21	\$10.00	¥	NA	¥	NA	¥	\$21 93	NA	×.	N N	NA	\$21.93	\$21.93	\$21.93		NA	X	×		¥	N S	S NA	NA	N	¥	NA	\$21.93	A
N.	¥ ¥	\$2.76	₹	N <sub>A</sub>	N <sub>A</sub>	N A	NA	NA	\$2.76	\$2.76	\$2.76	\$10.00	NA	N <sub>A</sub>	N <sub>A</sub>	AN	NA.	\$4.54	NA	NA.	N <sub>A</sub>	N <sub>A</sub>	NA	\$4.54	\$4.54	)  -	NA	¥	¥		¥	NA S	NA YE	N <sub>A</sub>	NA.	NA.	N <sub>A</sub>	NA.	2
N.	¥ ¥	NA.	₹	N <sub>A</sub>	N <sub>A</sub>	N A	NA	NA	A	NA :	NA	\$10.00	NA	N <sub>A</sub>	N <sub>A</sub>	AN	N .	\$17.16	NA	NA.	N <sub>A</sub>	N <sub>A</sub>	NA	\$17.16	\$17.16		NA	¥	¥		¥	N.	NA NA	N <sub>A</sub>	NA	NA.	N <sub>A</sub>	NA.	GA
NA.	₹ ₹	¥	¥	N N	N N	N N	NA	N <sub>A</sub>	¥	NA.	NA	\$10.00	NA.	×	N <sub>A</sub>	AN	¥	\$37.55	NA.	N <sub>A</sub>	N <sub>A</sub>	\$37.78	¥	\$37.55	\$37.55		NA	×	¥		¥	N S	NA AN	N <sub>A</sub>	N <sub>A</sub>	¥	\$37.78	NA NA	ক্
\$4.30	\$4.38	W	₹	NA	N	\$4.38	NA	NA	\$4.38	\$4.38	\$4.38	\$10.00	NA	N <sub>A</sub>	N <sub>A</sub>	AN	\$16.43	NA \$16.43	NA	NA	\$16.43	N <sub>A</sub>	N <sub>A</sub>	\$16.43	\$16.43		NA	NA.	¥		¥	\$16.43	NA NA	N <sub>A</sub>	NA	\$16.43	NA	NA	F
¥	¥	¥	₹	N <sub>A</sub>	\$6.56	N A	NA	NA	\$6.56	\$6.56	\$6.56	\$10.00	NA	N <sub>A</sub>	N <sub>A</sub>	AN	NA.	\$22 98	NA	\$22.98	NA	N <sub>A</sub>	NA.	\$22.98	\$22.98		NA	X	¥		¥	NA S	NA A	N <sub>A</sub>	\$22.98	N <sub>A</sub>	NA	NA	NS
N.	<b>₹</b>	¥	¥	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	NA	NA	NA :	NA A	\$10.00	NA	N <sub>A</sub>	NA	NA	¥	\$9.08 NA	NA	NA.	N <sub>A</sub>	N <sub>A</sub>	NA	\$9.08	\$9.08		NA.	¥	¥	:	¥	A S	NA NA	N <sub>A</sub>	NA.	N <sub>A</sub>	N <sub>A</sub>	NA.	NC
Ä	X X	NA.	₹	NA	N	N	NA	NA	NA	NA.	NA	\$10.00	¥	N <sub>A</sub>	NA	\$24.98	NA.	NA \$24.98	\$24.98	NA	N <sub>A</sub>	N <sub>A</sub>	N.	\$24.98	\$24.98		NA	X.	¥		\$24.98	NA S	NA AN	\$24.98	NA	NA	N <sub>A</sub>	NA	SC
×	X X	NA.	\$3.66	NA	N	N	NA	NA	\$3.66	\$3.66	\$3.66	\$10.00	\$9.19	\$9.19	\$9.19	NA	N.	\$9.19	NA	NA	N <sub>A</sub>	N <sub>A</sub>	NA	\$9.19	\$9.19		\$9.93	\$9.93	\$9.93		¥	NA S	\$9.93	N <sub>A</sub>	NA	NA	N <sub>A</sub>	NA.	TN

Exhibit C Rates - Page 21

	L	L			L							L					Н		L	L	П		I			L		L	EF		EF	L		L	L	E	Ε	PE
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	2-wire voice unbundled Florida area calling with caller ID - residence	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with calle ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	2-wire voice unbundled port outgoing only - residence	2-wire voice unbundled port with caller ID - residence	NRC - Disconnect Charge - Add'l	(BZF)	2-wire voice unbundled TN Bus 2-way Area Calling Port Standard Option (TACC2)	2-wire voice unbundled TN Bus 2-way Area Calling Port Economy Option (TACC1)	2-wire voice unbundles SC Bus Area Calling Port with Caller ID (LMB)	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	2-wire voice grade unbundled rennessee extended local dating parity port with caller ID  2-wire voice unbundled incoming only Port with Caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller I D	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with calle ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	2-wire voice unbundled outgoing only Port	2-wire voice unbundled port with Caller ID	Z-WIE VOICE UIDUINIEU RES LOW USAGE LITE POIL WILL CAIRL ID (LOW)	(2MR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence ((MF2X)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)		(LW8)	DESCRIPTION
UEPAO	UEPAN	UEPAM	UEPAL	UEPAK	UEPAJ	UEPAH	UEPAG	UEPAF	UEPAQ	UEPAU	UEPAT	UEPAS		UEPAR	UEPRO	UEPRC		UEPAE	UEPAD	UEPAC	UEPAB	UEPAA	UEPAV UEPB1	UEPAZ	UEPAY	UEPAX	UEPBM		UEPBO	UEPBL	0072	UEPAO	UEPAN	UEPAM	UEPAL	UEPAK	UEPAJ	USOC
NA	N <sub>A</sub>	NA	NA	NA	N <sub>A</sub>	NA.	NA	NA	NA	N <sub>A</sub>	NA	N <sub>A</sub>	NA	\$6.21	\$6.21	\$6.21		NA	N <sub>A</sub>	NA	NA	X.	\$6.21	NA	N <sub>A</sub>	NA	NA	\$6.21	\$6.21	\$6.21	12.0¢	N	NA	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	AL
N <sub>A</sub>	N <sub>A</sub>	NA	N <sub>A</sub>	NA	N <sub>A</sub>	¥	NA	\$2.59	NA	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	\$2.59	\$2.59	9	NA	N <sub>A</sub>	NA	NA	¥.	\$2.76	NA	N <sub>A</sub>	NA	N <sub>A</sub>	N <sub>A</sub>	\$2.76	\$2.76	\$2.70	NA NA	NA.	N <sub>A</sub>	N <sub>A</sub>	NA	NA	7
NA	NA	NA	NA	NA	NA	NA.	NA	NA	NA	NA	NA	NA	NA	NA	NA	₹ ₹		NA	NA	NA	NA	¥ ;	N N	NA	NA	NA	NA	NA	¥ ₹	N N	3	¥	N <sub>A</sub>	NA	N <sub>A</sub>	NA	NA	GA
NA	NA	NA	NA	NA	NA	NA	AN	NA	N <sub>A</sub>	NA	NA	NA	N	NA	AN	<b>₹</b>	;	NA	NA	NA	NA	¥ :	X X	N <sub>A</sub>	NA	NA	NA	NA	¥ ₹	N N	3	× ×	N <sub>A</sub>	NA	NA	NA	NA	ব
\$4.38	NA	NA	NA	NA	NA	\$4.38	\$4.38	NA	NA	NA	NA	\$4.38	NA	NA	\$4.38	\$4.38		NA	NA	NA	NA	\$4.38	NA \$4.38	NA	NA	\$4.38	NA	NA	\$4.38	\$4.38	\$4.30	NA NA	NA	NA	NA	NA	NA	LA
\$6.56	N	N <sub>N</sub>	N <sub>A</sub>	N <sub>A</sub>	NA	NA	AN	AN	NA	N <sub>A</sub>	\$6.56	Ą	NA	N <sub>A</sub>	\$6.56	\$6.56	2	AN	NA	NA	AN	AN	\$6.56	NA	\$6.56	NA	N	N.	\$6.56	\$6.56	φο.οο	AN	NA	N	NA.	NA	Ä	SM
NA	NA	NA	NA	¥	NA	NA	AN	AN	N	NA	NA	NA	NA	NA	AN	N N	;	NA	NA	NA	NA	N :	N N	NA	NA	NA	NA	NA	N 3	N A	3	NA.	NA	NA	NA	NA	NA	NC
¥	¥	¥	¥	¥	¥	¥	NΑ	NA	X.	¥	¥	¥	N A	¥	NA.	¥ ¥		Ä	¥	Ą	NA	¥ ;	X X	×	N <sub>A</sub>	X N	×	×	¥ ₹	N A	5	¥	N <sub>N</sub>	¥	¥	¥	¥	SC
\$2.92	\$2.92	\$2.92	\$2.92	\$2.92	Ą	NA	AN	AN	\$2.92	N <sub>A</sub>	NA.	Ą	NA A	¥	\$2.92	\$2.92	3	\$3.66	\$3.66	\$3.66	AN	NA S	\$3.66	NA.	N <sub>A</sub>	NA.	NA.	A	\$3.66	\$3.66	\$3.00	\$3.66	\$3.66	\$3.66	\$3.66	\$3.66	NA	NL

H	-	-	N	T		Н	-	-	Ŧ	Ŧ	4	Ŧ				H	1	+	_	H	-	+		Ŧ	H		>	T		Н			-					H					_	$\vdash$	F	T		H	H
NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	2-Wire DID Port, per month	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	Incremental Charge - Manual Service Order -	Incrementa	1	NRC - Add'l	NRC - ISI	0	NRC - Incremental Charge - Manual Service Order - Disconnect - Addi	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	- Disconnect Charge -	NRC - Disconnect Charge - 1st	NRC - Add'l (all types)	Three available feature, per month	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'i	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add"		Disconnect Charge -		NRC - 1st (all types)	All available features, per month	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces	NRC - OSS	2-wire voice unbundled IN Bus 2-way Collierville and Memphis Local Calling Port (B2F)	unbundled TN Bus 2-way Area Callir	2-wire voice unbundled TNBus 2-way Area Calling Port Economy Option (TACC1)	2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	2-wire voice unbundled incoming only port with Caller ID	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with calle ID	z-wire voice grade undurined Nettucky exterioed local draining harity port with caries	ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller	2-wire voice unbundled outcoing only port	2-wire voice unbundled port without Caller ID	2-wire voice unbundled Res Low Usage Line Port with Caller ID (LUM)
UEPP2	UEPP2	UEPP2	UEPP2	SOMAN	SOMAN	SOMAN	SOMAN	BFR :	UEP4A BFR	OEP4A	UEP4A	OCIVAN	SOMAN	SOMAN	SOMAN				UEPVF	SOMAN	SOMAN	SOMAN					UEPVF	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC		UEPAE	UEPAD	UEPAC	UEPAB	UEPAA	UEPB1	UEPAV	UEPAZ	UEPAY	UEPAX	UEPBM	UEPAW	OEFBO	UEPBC	UEPBL	UEPAP
A	\$18.00	\$50.00	\$12.08	NA	NA	¥	¥	¥ ;	X X	<b>S</b>	X A	Ä	<b>3 3</b>	¥	¥	NA	¥ :	¥ §	× ×	\$1.44	\$17.77	\$12.37	\$18.41	\$18.41	\$24.72	\$24.72	\$5.55	\$1.44	\$17.77	\$12.97	\$27.37	\$3.50		N <sub>A</sub>	NA	N	A	NA	\$6.21	<b>⊼</b>	NA	NA	N	NA	\$6.21	40.0	\$6.21	\$6.21	\$6.21
\$113.28	\$37.49	\$248.44	\$9.38	\$3.84	\$3.84	\$21.56	\$21.56	\$2.64	\$2.82	\$4.75	\$8.74	N.	<b>₹</b>	¥	NA.	NA	¥ :	¥ §	¥ ¥	NA	¥ :	8 8	. ₹	₹ ₩	NA	¥	\$3.40	¥	¥	N <sub>A</sub>	NA	\$3.50		¥	NA	N	NA.	NA	\$2.59	N A	NA	NA	N <sub>A</sub>	NA	N.	ψ 202	\$2.59	\$2.59	\$2.59
NA.	\$61.91	\$61.91	\$11.35	NA	NA	\$8.42	\$18.94	¥.	\$17.16	\$17.16	\$8.47	Ä	<b>X X</b>	¥	Ā	NA	X :	8 5	× ×	NA.	¥ :	X X	<b>X</b>	¥	NA	¥	NA A	A	Ā	\$8.42	\$18.94	\$3.50		NA.	NA	NA	A	AN	NA	Z S	NA	N <sub>A</sub>	NA	NA.	A	5	N X	<b>X</b>	NA :
¥	X.	N .	¥	NA	NA	NA A	¥	X :	X X	3	X X	NA.	<b>X X</b>	¥	Ä	NA	X :	¥ §	<b>X</b>	NA.	¥ ;	8	<b>X</b>	<b>X</b>	AN	¥	NA.	¥	¥	NA.	NA	\$3.50		NA	NA	NA	Ą	AN	NA	Z >	NA	N <sub>A</sub>	NA	NA.	A	5	X X	<b>X</b>	NA:
\$9.20	\$59.28	\$59.28	\$13.12	\$8.94	\$8.94	\$8.06	\$18.14	\$3.77	\$16.43	\$16.43	\$10.13	N	\$ \$	¥	NA	NA	¥ :	¥ §	\$8.28	NA	¥ :	X X	<b>X</b>	<b>.</b> ¥	NA	¥.	\$8.28	¥	\$10.39	\$8.06	\$18.14	\$3.50		NA	NA	NA	NA.	\$4.38	\$4.38	¥	NA	NA	\$4.38	NA	NA.	ψ#.JO	\$4.38	\$4.38	\$4.38
\$13.48	\$83.09	\$83.09	\$14.63	\$16.06	\$16.06	\$11.34	\$25.52	\$6.56	\$6.56	\$22.98	\$9.60	Ä	\$16.06	\$11.34	\$25.52	\$8.20	\$8.20	\$3.06	\$3.31	NA.	\$16.06	\$11.34	\$19.68	\$19.68	\$21.42	\$21.42	\$6.75	A	\$16.06	\$11.34	\$25.52	\$3.50		NA.	NA	NA	¥	AN	\$6.56	Z F	NA	\$6.56	NA	NA.	A	ψ0.50	\$6.56	\$6.56	\$6.56
NA	\$81.84	\$81.84	\$12.36	NA A	NA	\$12.67	\$26.85	¥ ;	\$21.69	\$21.69	\$8.69	NA	<b>3 3</b>	N.	NA	NA	X.	₹ 5	<b>X</b>	NA	<b>∑</b>	N X	<b>X</b>	<b>X</b>	NA	¥	N.	Ą	Ā	\$12.76	\$26.94	\$3.50		¥	NA	NA	¥	NA	NA	₹	NA	¥	NA.	¥	Ą	3	8 8	<b>X</b>	NA I
¥	\$50.00	\$50.00	\$12.08	NA	NA	¥	¥	¥.	\$3.50	\$3.50	\$2.28	¥	<b>3 3</b>	\$14.63	\$44.42	NA	N S	\$4.53	\$3.03	NA.	¥ is	\$14.42	. X	¥	\$36.24	\$36.24	\$6.29	¥	¥	\$14.63	\$44.42	\$3.50		N <sub>A</sub>	NA	NA	A	NA	NA	Z N	NA	NA	NA	NA	A	5	X X	: 🛪	NA
\$9.21	\$47.01	\$47.75	\$8.97	\$1.40	\$13.32	\$10.54	\$20.35	\$2.92	\$3.66	\$9.93	\$11.11	×	<b>X</b>	A	¥	NA	N :	¥ 3	₹ ₹	¥	<b>₹</b>	X X	\$	¥	NA	¥	¥	\$1.40	\$13.32	\$10.54	\$20.35	\$3.50		\$2.92	\$2.92	\$2.92	¥	NA	\$2.92	\$2.92	NA	N <sub>A</sub>	NA.	N <sub>N</sub>	¥	76.70	\$2.92	\$2.92	\$2.92

TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	RMINAL PBX TRUN	NULEU INC	LINE SIDE CUBUNETU COL WARD TOX TRUNK - BUSINESS	LINE SIDE CUBUNCED COMBINATION 2-WAY PBX TRUNK - BUSINESS	2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	 NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st		z	NKC - IICI eli el la i Cialge - Malidal Selvice Oldei - Discolli ect - Addi	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	4-Wire ISDN DS1 Port, per month	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Add"		THE INDIVIDUAL CITATION FRANCE CONTROL	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'i	NRC - Incremental Charge - Manual Service Order - Add'l	- Manual Service Order -	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	3.Miro (SDN Bost/3) (3) including all available features per month	NRC - User Profile per B Channel (4)	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'i	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'i		MIVO - IIIVI BIIIBI IIGI MIBING AMAING OFFICE OFFICE - DISVIII IBVI - AVAI	NRC - Incremental Charge - Manual Service Order - Disconnect - Add!	Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'I	NRC - 1st	4-Wire DS1 Port w/DID capability, per month	MINO - II Matinatinat Otaliga - Inatinat Otalino Otali - Diavolliavi - Mari	NRC - Incremental Charge - Manual Service Order - Disconnect - Addil	=	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Disconnect Charge - Add'l	DECOBIDATION
UEPTO	UEPLD	UEPP1	OEPPO	OEPPC:	UEPRD	SOMAN	SOMAN	OEPEX	UEPEX	OCKEN	SOMAN	SOMAN	SOMAN	UEPEX	UEPEX	UEPEX	UEPEX	LIEPEX	SOMAN	SOMAN	LITEMA	U1PMA		SOMAN	SOMAN	SOMAN	U1PMA	U1PMA	U1PMA	I I PMA	- IADMA	U1UMA	SOMAN	SOMAN	SOMAN	U1PMA	U1PMA	U1PMA	U1PMA	OCIEDIA	NAMOS	SOMAN	SOMAN	UEPDD	UEPDD	UEPDD	UEPDD	UEPDD	Constitution	SOMAN	SOMAN	SOMAN	UEPP2	1600
\$2.07	\$2.07	\$2.07	\$2.07	\$2.07	\$2.07	N S	Z Z	Z N	NA.	\$11.00	911.53	\$54.75	\$54.75	\$51.19	\$51.19	\$244.85	\$244.85	\$186.02	¥.	N S	N N	X X		N S	<b>X</b>	NA	Ā	Ā	¥ :	Z 3	N.	NA .	\$12.97	\$12.07	\$56.19	\$5.69	\$5.69	\$63.24	\$16.42	3	N A	₹ ₹	×.	N.	NA	\$18.00	\$50.00	\$130.23	5	8 8	×.	NA	Z P	2
\$1.62	\$1.62	\$1.62	\$1.62	\$1.62	\$1.62	N S	Z Z	<b>S S</b>	AN	\$0.04	\$3.84	\$21.56	\$21.56	\$37.93	\$149.75	\$203.18	\$417.51	\$95.39	¥.	N S	Z Z	AN AN		Z Z	X X	¥	Ā	Ā	Z.	Z S	N	NA.	\$3.84	\$21.56	\$21.56	\$20.98	\$93.37	\$106.00	\$10.20	ψ.J.O <del>.</del>	\$3.04	\$21.56	\$21.56	\$4.65	\$137.29	\$191.44	\$413.93	\$63.31	<b>\$0.0</b>	\$3.84	\$21.56	\$21.56	\$7.12	0
\$1.85	\$1.85	\$1.85	\$1.85	\$1.85	\$1.85	N S	Z Z	<b>S S</b>	NA.	5	\$	\$37.88	\$37.88	¥	NA	\$186.80	\$186.80	\$163 16	NA :	N S	Z X	AN AN		N X	X X	NA.	¥	¥	¥.	N S	NIA	NA	N S	\$39.98	\$39.98	AN	¥	\$47.37	\$13.47	5		\$8.42	\$18.94	¥	NA A	\$52.46	\$89.44	\$120.80	5	N X	\$8.42	\$18.94	N S	7.0
\$2.61	\$2.61	\$2.61	\$2.61	\$2.61	\$2.61	N S	\$116.42	\$181.27	\$275.48	5	N A	<b>X X</b>	×	¥	Ą	NA	¥ ₹	NA	¥.	N S	N K	AN AN		N K	X	¥	¥	¥	¥.	N S	NIA	\$5.61	¥ §	N A	X X	AN	¥	\$84.53	\$12.33	5	N A	¥ ¥	¥	¥	NA	NA	NA	NA	5	N X	<b>X</b>	¥	N Z	5
\$2.20	\$2.20	\$2.20	\$2.20	\$2.20	\$2.20	NA S	N K	i N	NA.	\$1.13	\$7.73	\$33.18	\$33.18	\$27.11	\$27.11	\$181.89	\$181.89	\$19472	NA :	N S	N K	AN A		N K	¥ ¥	NA	¥	¥	Z.	Z 5	N	NA	\$6.65	\$38.29	\$38.29	\$4.31	\$4.31	\$45.35	\$23.33	<b>\$10.00</b>	\$10.39	\$8.06	\$18.14	\$8.82	\$8.82	\$50.23	\$85.63	\$149.27	6.00	\$10.39	\$8.06	\$18.14	\$9.20	- 0
\$2.11	\$2.11	\$2.11	\$2.11	\$2.11	\$2.11	NA S	N X	Z N	NA.	фо.от	\$0.51	\$51.03	\$51.03	\$53.32	\$53.32	\$244.12	\$244.12	\$213.21	NA :	NA S	N X	N N		N X	¥	NA.	Ā	Ā	Z,	Z 3	25	NA	\$11.34	\$53.87	\$53.87	\$7.04	\$7.04	\$63.59	\$51.91	#10.00	\$16.06	\$11.34	\$25.52	\$12.94	\$12.94	\$71.18	\$117.81	\$146.46	6.00	\$16.07	\$11.34	\$25.52	\$13.48	Mo
\$2.00	\$2.00	\$2.00	\$2.00	\$2.00	\$2.18	NA 5	N S	X X	NA.	3	N N	\$53.89	\$53.89	¥	NA	\$241.63	\$241.63	\$17975	N :	NA S	N N	N N		N S	¥ ¥	NA	¥	¥	¥.	N S	NIA	NA :	¥ §	\$55.30	\$55.30	NA	¥	\$62.29	\$24.50	5		\$12.76	\$26.94	NA	NA	\$69.92	\$116.59	\$123.65	5	8 8	\$12.76	\$26.94	NA &	Ś
\$2.35	\$2.35	\$2.35	\$2.35	\$2.35	\$2.35	\$65.48	\$65.48	\$311.73	\$251.00	3	NA	\$65.48	\$65.48	¥	NA	\$278.37	\$278.37	\$214 79	\$67.52	\$67.52	\$70.32	\$36.01		N S	\$67.52	\$67.52	NA.	NA.	\$106.40	\$106.00	630 60	NA	₹ ₹	\$67.5Z	\$67.52	NA	¥.	\$65.79	\$33.74	5	NA NA	\$ \$	¥	NA	NA	\$60.00	\$60.00	\$130.23	3	N N	X.	NA	NA 6	ŝ
\$4.63	\$4.63	\$4.63	\$4.63	\$4.63	\$4.63	NA 5	N N	X X	NA	\$10.54	\$9.07	\$42.17	\$40.69	\$36.98	\$38.46	\$147.18	\$148.66	\$308 00	NA :	NA S	Z X	N N	0.00	\$9.80	\$42.17	\$41.43	\$4.10	\$4.10	\$29.49	\$30.40	¢10 40	NA	<b>₹</b>	N X	<b>X</b>	NA	¥	<b>₹</b>	X X	5	N N	¥ ¥	¥	¥	NA	negotiated	negotiated	\$120.00	-1-0	\$13.32	\$10.54	\$20.35	\$8.47	į

				I	H																														H	=		DESC
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	PORT	TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS	LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	LINE SIDE LINBLINDLED OUTWARD PRY TRUNK - BUSINESS	SIDE LINBLINDI ED COMBINATION 2-WAY	NRC - 1st	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	UNBUNDLED LOOP BILLING USOC (REQUIRES ONE PER PORT)	2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING PORT	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT	2-VIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTELHOSPITAL ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT WITHOUT LUD	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING PORT WITHOUT LUD	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT 2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	2-WIRE VOICE UNBUNDLED 1:WAY OUTGOING PBX TENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT	DESCRIPTION
UEPT2	UEPLD	UEPL2	UEPA2	UEPTO	UEPT2	UEPLD	UEPP1	Oddali	CEPRU	UEPPC	LNPCP	UEPLX	UEPXV	UEPXU	UEPXT	UEPXS	UEPXR	UEPXQ	UEPXP	UEPXO	UEPXN	UEPXM	UEPXL	UEPXK	UEPXJ	UEPXH	UEPXG	UEPXF	UEPXE	UEPXD	UEPXC	UEPXA	UEPTO	UEPT2	UEPLD	UEPL2	UEPA2	USOC
NA	\$21.93	¥.	\$21.93	\$21.93	\$21.93	\$21.93	\$21.93	\$21.93	\$21.93	\$21.93			NA	N	N <sub>A</sub>	\$2.07	N <sub></sub>	NA	NA	\$2.07	×.	\$2.07	\$2.07	NA	NA	NA	N <sub>A</sub>	N <sub>A</sub>	\$2.07	\$2.07	\$2.07	\$2.07	NA	NA	\$2.07	NA	\$2.07	AL
NA	\$62.56	Z F	NA.	\$62.56	\$62.56	\$62.56	\$62.56	\$62.56	\$62.56	\$62.56			NA	N <sub>A</sub>	×	\$1.62	¥.	N <sub>A</sub>	N <sub>A</sub>	\$1.62	¥.	\$1.62	\$1.62	N <sub>A</sub>	N <sub>A</sub>	NA	¥.	N <sub>A</sub>	\$1.62	\$1.62	\$1.62	\$1.62	NA	NA.	\$1.62	NA	N <sub>A</sub>	7
NA	\$17.16	¥.	NA	\$17.16	\$17.16	\$17.16	\$17.16	\$17.16	\$17.16	\$17.16			NA	N <sub>A</sub>	N <sub>A</sub>	\$1.85	N <sub></sub>	NA	NA	\$1.85	×.	\$1.85	\$1.85	NA	NA	NA	N <sub>A</sub>	N <sub>A</sub>	\$1.85	\$1.85	\$1.85	\$1.85	NA	NA	\$1.85	NA	NA	GA
NA	\$36.47	Z F	NA	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47			NA	N <sub>A</sub>	N <sub>A</sub>	\$2.61	×	N <sub>A</sub>	N <sub>A</sub>	\$2.61	¥.	\$2.61	\$2.61	N N	\$2.61	\$2.61	\$2.61	\$2.61	\$2.61	\$2.61	\$2.61	\$2.61	NA	N <sub>A</sub>	\$2.61	NA	N <sub>A</sub>	ξ
NA	\$16.43	\$16.43	NA	\$16.43	\$16.43	\$16.43	\$16.43	\$16.43	\$16.43	\$16.43			NA	NA	N	\$2.20	N <sub>A</sub>	NA	\$2.20	\$2.20	×.	\$2.20	\$2.20	\$2.20	NA	NA	N <sub>A</sub>	N	\$2.20	\$2.20	\$2.20	\$2.20	NA	NA	\$2.20	\$2.20	NA	5
N <sub></sub>	\$22.98	¥	NA	\$22.98	\$22.98	\$22.98	\$22.98	\$22.90	\$22.98	\$22.98			NA	N <sub>A</sub>	N <sub>A</sub>	\$2.11	\$2.11	\$2.11	NA	\$2.11	¥	\$2.11	\$2.11	NA	NA	NA	N <sub>A</sub>	N <sub>A</sub>	\$2.11	\$2.11	\$2.11	\$2.11	NA	NA	\$2.11	NA	NA	MS
NA	\$24.04	Z F	NA	\$24.04	\$24.04	\$24.04	\$24.04	\$24.04	\$21.60	\$24.04			NA	NA	N	\$2.00	N.	NA	NA	\$2.00	N <sub>A</sub>	\$2.00	\$2.00	NA	NA	NA	N	N	\$2.00	\$2.00	\$2.00	\$2.00	NA	NA	\$2.00	N	NA	NC
NA	\$24.36	Z F	NA	\$24.36	\$24.36	\$24.36	\$24.36	\$24.36	\$24.36	\$24.36			NA	NA	\$2.35	\$2.35	N.	NA	NA	\$2.35	N <sub>A</sub>	\$2.35	\$2.35	NA	NA	NA	N	N	\$2.35	\$2.35	\$2.35	\$2.35	NA	NA	\$2.35	N	NA	SC
\$9.93	NA	¥	N <sub>A</sub>	\$9.93	\$9.93	\$9.93	\$9.93	\$9.93	\$9.93	\$9.93			\$4.63	\$4.63	¥	\$4.63	×.	N <sub>A</sub>	×.	\$4.63	\$4.63	\$4.63	\$4.63	NA.	N <sub>A</sub>	NA	N <sub>A</sub>	N <sub>A</sub>	\$4.63	\$4.63	\$4.63	\$4.63	\$4.63	\$4.63	\$4.63	NA	N <sub>A</sub>	N

											1																									Ŧ	F	DES
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	2-WIRE VOICE INBUINDLED 2-WAY COMBINATION DRY LISAGE DORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT	TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	LINE SIDE ONBONDEED INCOMING PBX IRONK - BOSINESS	LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS	NK - BUS	2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	Subsequent Activity	2-WIRE VOICE UNBUNDED 2-WAY PBX TENNESSEE REGIONSERV CALLING	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT WITHOUT LUD	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING PORT WITHOUT LUD	2-WIRE VOICE UNBUNDED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT		2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING	DESCRIPTION
UEPXE	UEPXD	UEPXC	UEPXB		ii pio	UEPT2	UEPLD	UEPL2	UEPA2	UEPTO	UEPT2		UEPPO	UEPPC	UEPRD	USASC	UEPXV	UEPXU	UEPXT	UEPXS	UEPXR	UEPXQ	UEPXP	UEPXO	UEPXN	UEPXM	UEPXL	UEPXK	UEPXJ	UEPXH	UEPXG	UEPXF	UEPXE	UEPXD	UEPXC	UEPXA	UEPTO	USOC
\$21.93	\$21.93	\$21.93	\$21.93	\$21 Q3	5	×	\$21.93	N A	\$21.93	\$21.93	\$21.93	\$21.93	\$21.93	\$21.93	\$21.93	\$10.00	N <sub>A</sub>	N <sub>A</sub>	NA.	\$21.93	NA.	N <sub>A</sub>	¥	\$21.93	NA	\$21.93	\$21.93	NA	N <sub>A</sub>	N	NA	×	\$21.93	\$21.93	\$21.93	\$21.93	Ş Ş	AL
\$29.70	\$29.70	\$29.70	\$29.70	07 PC\$	5	×	\$29.70	N A	¥	\$29.70	\$29.70	\$29.70	\$29.70	\$29.70	\$29.70	\$10.00	N <sub>A</sub>	N <sub>A</sub>	NA.	\$62.56	NA.	N <sub>A</sub>	¥	\$62.56	NA	\$62.56	\$62.56	NA	N <sub>A</sub>	N	NA	×	\$62.56	\$62.56	\$62.56	\$62.56	N N	7
\$17.16	\$17.16	\$17.16	\$17.16	\$17.16	NIA	N N	\$17.16	NA	N <sub>A</sub>	\$17.16	\$17.16	\$17.16	\$17.16	\$17.16	\$17.16	\$10.00	NA A	NA A	NA.	\$17.16	NA	NA	¥	\$17.16	NA	\$17.16	\$17.16	N <sub>A</sub>	NA	N <sub>A</sub>	N	N <sub>A</sub>	\$17.16	\$17.16	\$17.16	\$17.16	NA NA	GA
\$36.47	\$36.47	\$36.47	\$36.47	\$36.47		NA	\$36.47	AN	N <sub>A</sub>	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	\$10.00	NA	NA	NA	\$36.47	NA	NA	N	\$36.47	N <sub>A</sub>	\$36.47	\$36.47	NA	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	\$36.47	NA	ম
\$16.43	\$16.43	\$16.43	\$16.43	\$16.43	25	×	\$16.43	\$16.43	N.	\$16.43	\$16.43	\$16.43	\$16.43	\$16.43	\$16.43	\$10.00	¥	¥	Ä	\$16.43	NA	NA	\$16.43	\$16.43	¥	\$16.43	\$16.43	\$16.43	¥	N <sub>A</sub>	NA	N N	\$16.43	\$16.43	\$16.43	\$16.43 \$16.43	ž ×	F
\$22.98	\$22.98	\$22.98	\$22.98	\$22.08	25	N	\$22.98	N	NA	\$22.98	\$22.98	\$22.98	\$22.98	\$22.98	\$22.98	\$10.00	NA	NA	NA	\$22.98	\$22.98	\$22.98	N	\$22.98	N <sub>A</sub>	\$22.98	\$22.98	N	NA	NA	NA	NA	\$22.98	\$22.98	\$22.98	\$22.98	NA	MS
\$9.05	\$9.05	\$9.05	\$9.05	\$0.05	N	N N	\$9.05	N <sub>A</sub>	NA	\$9.05	\$9.05	\$9.U5	\$9.05	\$9.05	\$21.60	\$10.00	NA	NA	NA	\$24.04	NA	NA	NA	\$24.04	NA	\$24.04	\$24.04	NA	N <sub>A</sub>	NA	NA	NA	\$24.04	\$24.04	\$24.04	\$24.04	NA	NC
\$24.36	\$24.36	\$24.36	\$24.36	\$24.36	25	N	\$24.36	NA	N <sub>A</sub>	\$24.36	\$24.36	\$24.30	\$24.36	\$24.36	\$24.36	\$10.00	NA	NA	\$24.36	\$24.36	N	NA	¥.	\$24.36	N <sub>A</sub>	\$24.36	\$24.36	N	NA	N <sub>A</sub>	N	N	\$24.36	\$24.36	\$24.36	\$24.36	NA	SC
\$9.19	\$9.19	\$9.19	\$9.19	\$9.19	50	\$9.19	\$9.19	N	N <sub>A</sub>	\$9.19	\$9.19	\$9.19	\$9.19	\$9.19	\$9.19	\$10.00	\$9.93	\$9.93	NA	\$9.93	NA	NA	NA	\$9.93	\$9.93	\$9.93	\$9.93	N	N <sub>A</sub>	N N	N	N	\$9.93	\$9.93	\$9.93	\$9.93	\$9.93	TV

																																				E		DES
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT WITHOUT LUD	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING PORT WITHOUT LUD	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT	2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT  A MUDE VOICE UNBUNDLED A WAY OCCUPANTION DBX LE VOE BOOT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT	TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	LONG DISTANCE TERMINAL PBX TRUNK-BUSINESS	LINE SIDE LINBUNDLED ON WARD PBX TRUNK - BUSINESS	LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	NRC - Disconnect Charge - 1st	2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING PORT WITHOUT LUD	DESCRIPTION
UEPXL	UEPXK	UEPXJ	UEPXH	UEPXG	UEPXF	UEPXE	UEPXD	UEPXC	UEPXB	UEPTO	UEPT2	UEPLD	UEPL2	UEPA2	UEPTO	UEPT2	UEPLD	UEPP1	UEPPC	UEPRD		UEPXV	UEPXU	UEPXT	UEPXS	UEPXR	UEPXQ	UEPXP	UEPXO	UEPXN	UEPXM	UEPXL	UEPXK	UEPXJ	UEPXH	UEPXG	UEPXF	USOC
\$6.21	N <sub>A</sub>	NA	NA	NA	¥.	\$6.21	\$6.21	\$6.21	\$6.21	S NA	NA	\$6.21	₹	\$6.21	\$6.21	\$6.21	\$6.21	\$6.21	\$6.21	\$6.21		N <sub>A</sub>	NA	NA	\$21.93	NA	¥	NA	\$21.93	¥	\$21.93	\$21.93	N <sub>A</sub>	N <sub>A</sub>	NA	N <sub>A</sub>	¥.	AL
\$26.37	N <sub>A</sub>	N N	¥	N <sub>A</sub>	¥	\$26.37	\$26.37	\$26.37	\$26.37	NA A	N <sub>A</sub>	\$26.37	Z A	NA	\$26.37	\$26.37	\$26.37	\$26.37	\$26.37	\$26.37		Z ≽	NA	NA	\$29.70	¥	N <sub>A</sub>	NA	\$29.70	NA	\$29.70	\$29.70	NA	NA	N <sub>A</sub>	¥.	¥	2
N <sub>A</sub>	N <sub>A</sub>	N N	¥	N <sub>A</sub>	¥	NA	¥.	NA	¥ §	<b>X X</b>	N <sub>A</sub>	NA	Z A	NA	NA	¥.	¥ :	Z	. ₹	NA		Z ≽	NA	NA	\$17.16	¥	N <sub>A</sub>	NA	\$17.16	NA	\$17.16	\$17.16	NA	NA	N <sub>A</sub>	¥	¥	GA
NA	N <sub>A</sub>	NA	N <sub>A</sub>	NA	N <sub>A</sub>	NA	N N	NA	¥ §	<b>X X</b>	N <sub>A</sub>	NA	Z Ā	NA	NA	¥.	¥.	2 2	. ₹	NA		Z N	NA	NA	\$36.47	¥.	N	NA	\$36.47	NA	\$36.47	\$36.47	NA	\$39.47	\$38.47	\$37.47	\$36.47	হ
\$3.77	\$3.77	¥	¥	¥	N <sub>A</sub>	\$3.77	\$3.77	\$3.77	\$3.77	Ž N	N <sub>A</sub>	\$3.77	\$3.77	NA	\$3.77	\$3.77	\$3.77	\$3.77	\$3.77	\$3.77		×.	NA	¥	\$16.43	×	N <sub>A</sub>	\$16.43	\$16.43	N <sub>A</sub>	\$16.43	\$16.43	\$16.43	¥	N <sub>A</sub>	×	×	5
\$6.56	NA	NA	NA	NA	N <sub>A</sub>	\$6.56	\$6.56	\$6.56	\$6.56	NA	NA	\$6.56	₹	NA	\$6.56	\$6.56	\$6.56	\$6.56	\$6.56	\$6.56		NA	Ą	NA	\$22.98	\$22.98	\$22.98	NA	\$22.98	¥	\$22.98	\$22.98	NA	N <sub>A</sub>	NA	NA	NA	MS
NA	N <sub>A</sub>	N <sub>A</sub>	NA.	NA.	¥.	N <sub>A</sub>	N <sub>A</sub>	NA	× 5	X X	N <sub>A</sub>	NA	₹	NA	NA	¥ :	¥ :	N S	X X	Ą		NA.	NA.	NA.	\$9.05	NA.	N	N <sub>A</sub>	\$9.05	N	\$9.05	\$9.05	N <sub>A</sub>	×	NA	N <sub>A</sub>	N.	NC
NA	N.	NA	NA	NA	N	NA	N	NA	N 5	NA NA	N.	NA	¥	NA	NA	<b>∑</b>	¥ :	N N	. ₩	NA		N.	NA	\$24.36	\$24.36	N N	N	NA	\$24.36	NA	\$24.36	\$24.36	NA	N	NA	N	N	SC
\$3.66	¥.	N <sub>A</sub>	NA	NA	N <sub>A</sub>	\$3.66	\$3.66	\$3.66	\$3.66	\$3.66	\$3.66	\$3.66	Š	NA	\$3.66	\$3.66	\$3.66	\$3.66	\$3.66	\$3.66		\$9.19	\$9.19	N <sub>A</sub>	\$9.19	NA.	NA	N <sub>A</sub>	\$9.19	\$9.19	\$9.19	\$9.19	N <sub>A</sub>	×.	NA	N <sub>A</sub>	N <sub>A</sub>	¥

																																					DES
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING PORT WITHOUT LUD	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT	VOICE	2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	PORT	PORT STANSFORD ED 1-WAY OF TROONS DRY TENNESSEE CALLING	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT	TN OUTWARD CALLING PLAN PBX TRUNK - BUSINESS	TN 2-WAY CALLING PLAN PBX TRUNK - BUSINESS	3 DISTANCE TERMINAL PBX TRUNK-BUSINESS	LINE SIDE LINBLINDLED OUTWARD PBX TRUNK - BUSINESS	LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence	NRC - Disconnect Charge - Add'l	2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING PORT	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL CALLING PORT	2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	CALLING PORT	DESCRIPTION
UEPXQ	UEPXP	UEPXO	UEPXN	UEPXM	UEPXL	UEPXK	UEPXJ	UEPXH	UEPXG	UEPXF	UEPXE	UEPXD	UEPXC	UEPXB	UEPTO	UEPT2	UEPLD	UEPL2	UEPA2	UEPTO	UEPT2	UEPLD	UEPPO	UEPPC	UEPRD		UEPXV	UEPXU	UEPXT	UEPXS	UEPXR	UEPXQ	UEPXP	UEPXO	UEPXN	UEPXM	USOC
NA	\$6.21	\$6.21	NA	\$6.21	\$6.21	NA	N <sub>A</sub>	NA	NA	NA	\$6.21	\$6.21	\$6.21	\$6.21	Ž X	NA	\$6.21	N	\$6.21	\$6.21	\$6.21	\$6.21	\$6.21	\$6.21	\$6.21		NA	NA	N.	\$6.21	Z <sub>N</sub>	N <sub>A</sub>	\$6.21	\$6.21	NA	\$6.21	A
NA	N <sub>A</sub>	\$1.69	NA	\$1.69	\$1.69	NA	N <sub>A</sub>	NA	NA	NA	\$1.69	\$1.69	\$1.69	\$1.69	Ž NA	NA	\$1.69	N	N.	\$1.69	\$1.69	\$1.69	\$1.69	\$1.69	\$1.69		NA	NA	N.	\$26.37	Z <sub>N</sub>	N <sub>A</sub>	N <sub>A</sub>	\$26.37	\$26.37	\$26.37	P
N <sub>A</sub>	NA	N <sub>A</sub>	NA	N <sub>A</sub>	NA	NA	N <sub>A</sub>	NA	NA	NA NA	N <sub>A</sub>	N <sub>A</sub>	NA.	₹ ₹	<b>5</b> ≽	NA	NA.	NA	NA	NA	NA	¥ ;	N N	₹	NA.		NA	NA	NA	NA	NA	N <sub>A</sub>	NA	NA	NA	NA	GA
N <sub>A</sub>	¥	NA	N.	N <sub>A</sub>	NA	N N	NA.	¥	¥	N <sub>A</sub>	N <sub>A</sub>	N <sub>A</sub>	NA.	<b>₹</b> §	¥ ¥	¥	Ä	N <sub>A</sub>	NA	NA	NA	¥ ;	N N	¥	NA		Š	¥	NA	N <sub>A</sub>	NA	¥	NA.	NA	N <sub>A</sub>	¥	ক
NA	\$3.77	\$3.77	N <sub>A</sub>	\$3.77	\$3.77	\$3.77	NA	N <sub>A</sub>	NA.	N <sub>A</sub>	\$3.77	\$3.77	\$3.77	\$3.77	ž N	NA	\$3.77	\$3.77	NA	\$3.77	\$3.77	\$3.77	\$3.77	\$3.77	\$3.77		NA	N N	NA	\$3.77	NA	N <sub>A</sub>	\$3.77	\$3.77	N <sub>A</sub>	\$3.77	5
\$6.56	\$6.56	\$6.56	N <sub>A</sub>	\$6.56	\$6.56	×	NA.	₹	¥	N <sub>A</sub>	\$6.56	\$6.56	\$6.56	\$6.56	S N	¥	\$6.56	¥	NA	\$6.56	\$6.56	\$6.56	\$6.56	\$6.56	\$6.56		¥	¥	NA	\$6.56	\$6.56	\$6.56	\$6.56	\$6.56	N <sub>A</sub>	\$6.56	MS
¥	NA	NA	NA	¥.	NA	N <sub>A</sub>	¥	NA	NA	¥	NA	¥	NA.	₹ ₹	<b>5</b> ₽	NA	NA	NA	N <sub>A</sub>	NA	NA	¥ ;	N X	₹	NA		NA	NA	N <sub>A</sub>	NA	N	NA	NA	NA	NA	NA	NC
¥	NA	NA	NA	¥.	NA	N <sub>A</sub>	¥	NA	NA	¥	NA	¥	NA.	₹ <del>§</del>	S	NA	NA	NA	N <sub>A</sub>	NA	NA	¥ ;	N X	₹	NA		NA	NA	N <sub>A</sub>	NA	N	NA	NA	NA	NA	NA	SC
NA	N <sub>A</sub>	\$2.92	\$2.92	\$2.92	\$2.92	NA	NA	NA	NA	NA	\$2.92	\$2.92	\$2.92	\$2.92	\$2.92	\$2.92	\$2.92	NA	NA	\$2.92	\$2.92	\$2.92	\$2.92	\$2.92	\$2.92		\$3.66	\$3.66	NA	\$3.66	NA	NA	NA	\$3.66	\$3.66	\$3.66	TN

13 73 F

Attachment 2 Exhibit C Rates - Page 28

П	g	ç		ဂ		-	ဂ္ဂ	1	ç	?	-	>	>	-	Þ		F	S		-	Z.		F	ဂ္ဂ		H	ဂ	-	-	<b>⊒</b> !		<u> </u>	-			Ŧ			H	H	ဂ		F	2-		Н		-		Н	H				Ŧ		
NRC - Disconnect	NRC	INCC - Disconnect	NRC	Customer Originated Trace, per month	NRC - Disconnect	NRC	Calling Number Delivery Blocking, per month	NRC - Disconnect	NDC	NRC - Disconnect	NRC Disconnect	Automatic Recall, per month	NRC - Disconnect	NRC	Automatic Callback, per month	NRC - Disconnect	NRC	Cancel Call Waiting, per month	NRC - Disconnect	NRC	Remote Activation of Call Fordwarding, per month	NRC - Disconnect	NRC	Call Waiting	NRC - Disconnect	NRC	Customer Changeable Speed Calling, per month	NRC - Disconnect	NRC	Three-Way Calling, per month	ocal Switching Features offered with Port. Per month	VERTICAL FEATURES		NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Isc	NRC - Disconnect Charge - Addil	NRC - Disconnect Charge - 1st	NRC - Add'I	NRC - 1st	Coin Port, per month	NRC - Add'l	NRC - 1st	2-Wire Analog Hunting, per line per month	NRC - Incremental Charge - Manual Service Order - Disconnect - Add'i	NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces	NRC - OSS		2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING	2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	CALLING PORT	2-WIRE VOICE INBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	CALLING PORT	DESCRIPTION    SAMPE VOICE INDITION ED 3.WAY DBY MISSISSIDELLOCAL OBTIONAL
																														1000	N/A			NAMOS	SOMAN	NAMOS						HTGUX	XUBIH	жизтн	SOMAN	NAMOS	NAMOS	NAMOS	SOMEC		OFFAV	- - - - - - - - - - - - - - - - - - -	UEPXU	UEPXT	UEPXS	UEPXR	OSOC
\$0.55	\$1.03	\$0.55	\$1.03	\$0.14	\$0.55	\$1.03	\$1.17	\$0 55 50 55	\$1.02	\$0.55	\$1.03	\$0.28	\$0.55	\$1.03	\$0.29	\$0.55	\$1.03	\$0.01	\$0.55	\$1.03	\$0.18	\$0.55	\$1.03	\$0.03	\$0.55	\$1.03	\$0.08	\$0.55	\$1.03	\$1.12	¥.			\$0.48	\$16.33	\$12.97	\$5.21	\$5.21	\$21.93	\$21.93	\$2.34	See features	See features	See features	\$0.48	\$17.77	\$12.97	\$27.37	\$3.50		3	2	NA	NA	\$6.21	NA	AL
NA :	X ;	N N	5 ₹	¥	NA.	¥	<b>₹</b>	2 3	5	N A	\$	S 8	S 8	\$ \$	×	. ₹	¥	¥	X	¥	¥	NA	NA	¥	Ā	NA	NA A	¥	¥ :	N S	charge	NI D		\$3.84	\$3.84	\$21.56	\$2.59	\$2.76	\$4.54	\$4.76	\$1.62			X	\$3.84	\$3.84	\$21.56	\$21.56	\$2.75		3	5	¥	¥	\$1.69	¥	2
NA :	¥ ;	NA NA	X X	¥	NA	¥.	<b>₹</b> 5	2 3	5	N A	N A	N A	N A	NA NA	¥	×	N N	¥	¥	¥	¥	NA NA	NA	¥	¥	NA.	AN AN	¥	¥.	¥ ;	¥.			NA	NA	\$8.42	Ž.	¥	\$17.16	\$17.16	\$2.05	A	×.	X X	A	NA	\$8.42	\$18.94	\$3.50		3	5	NA	NA	NA	NA	GA
NA :	¥ ;	N N	<u> </u>	¥	NA.	¥ :	<b>₹</b> 5	3	3	2 2	X X	- N	X X	X X	¥	X	¥	¥	¥	¥	¥	NA NA	NA A	¥	¥	ΝA	Ą	¥	¥.	NA S	charge			AN	NA	<b>₹</b> §	<b>X</b>	¥	\$40.71	\$40.71	\$3.04	\$2.14	\$2.14	\$0.29	NA.	AN	¥	NA.	\$3.50		3	5	NA	NA	NA	NA	ĸ
NA	X ;	NA NA	S &	¥	NA	N.	<b>₹</b> 5	Z 3	× 5	N N	S S	NA NA	Z Z	S N	X.	X.	X.	NA.	Ā	¥	NA.	NA	NA	N <sub>A</sub>	NA A	NA	NA NA	¥	¥.	NA S	\$8.28			AN	\$9.86	\$8.06	\$4.15	\$4.15	\$16.43	\$16.43	\$2.50				Ā	\$8.94	\$8.06	\$18.14	\$3.50		3	Š	NA	NA	\$3.77	NA	F
\$0.5466	\$1.02	\$0.5466	\$1.02	\$0.1918	\$0.5466	\$1.02	\$0.9913	\$0.5466	\$1.00	\$0.5466	\$1.02	\$1.3164	\$0.5466	\$1.02	\$0.9977	\$0.5466	\$1.02	\$0.0082	\$0.5466	\$1.02	\$0.4859	\$0.5466	\$1.02	\$0.033	\$0.5466	\$1.02	\$0.0755	\$0.5466	\$1.02	\$1.32	Š			NA	\$16.06	\$11.34	\$6.56	\$6.56	\$22.98	\$22.98	\$2.32	See features	See features	See features	¥	\$16.06	\$11.34	\$25.52	\$3.50		3	Š	NA	NA	\$6.56	\$6.56	SW
NA	\$1.51	\$0 13	\$1.51	\$0.14	NA.	\$1.51	\$0.02	NA SI	61.51	\$0 33	\$1.51	\$0.29	en se	\$1.51	\$0.66	Ž.	\$1.51	\$0.01	¥	\$1.51	\$0.85	NA	\$1.51	\$0.09	Ā	\$1.51	\$0.17	¥	\$1.51	\$0.89	N N			NA	NA	¥ ₹	. ₹	¥	NA	NA	¥	¥	Ā	X X	¥	NA	\$12.76	\$26.94	\$3.50		3	5	¥	¥	N <sub>A</sub>	¥	NC
NA.	\$1.51	\$0.1528	\$1.51	\$0.1402	¥	\$1.51	\$0.3684	NA C	61.51	NA 02272	\$1.51	\$0.3102	\$0.3403	\$1.51	\$0.8015	\$ N	\$1.51	\$0.0099	¥	\$1.51	\$0.3743	NA	\$1.51	\$0.0665	¥	\$1.51	\$0.1247	¥	\$1.51	\$1.10	See above			NA	NA	\$14.57	NA NA	<b>X</b>	\$24.75	\$24.75	\$2.77	See features	See features	See features	¥	NA	\$14.46	\$41.86	\$3.50		5	5	¥	¥	X.	¥	SC
NA .	¥ ;	NA NA	X X	¥	NA NA	N.	<b>₹</b> 5	N 3	2 3	N X	N N	NA NA	N N	N N	×	<b>X</b>	Z X	Z.	¥	NA.	¥	NA NA	NA A	¥	¥	NA	NA A	¥	¥ :	¥.	×.			\$1.40	\$13.32	\$10.54	\$2.92	\$3.66	\$9.19	\$9.93	\$2.16			X X	\$1.40	\$13.32	\$10.54	\$20.35	\$3.50		\$2.3¢	3	\$2.92	N N	\$2.92	N <sub>A</sub>	N

	A Inform		Call P	2	l	Code	E		Teen :			Make			Multi /			Privac			Call by			ISDNE			Multip			Share			Anony		E	Messa			Toll R			Call H			Call T			Remo		-	Call Fo		9	Call Fr		Call	2	I	Servic	Multilli	E		Select		F	Select	DESC
NRC	NRC - Disconnect	NRC Disconnect	Call Park, per month	NRC - Disconnect	NRC:	Code Restriction and Diversion, per month	NRC - Disconnect	NRC	Service (Res. Dist. Alerting Service), per month	NRC - Disconnect	NRC	Make Set Busy, per month	NRC - Disconnect	NRC	Multi Appearance Directory Number Calls, per month	NRC - Disconnect	NRC	ര	NRC - Disconnect	NRC	Call Access, per month	NRC - Disconnect	NRC	ISDN Bridged Call Exclusion, per month	NRC - Disconnect	NRC	Multiple Call Appearances, per month	NRC - Disconnect	NRC	Shared Call Appearances of a DN, per month	NRC - Disconnect	NRC	Anonymous Call Rejection, per month	NRC - Disconnect	NRC	Message Waiting Indicator – Stutter Dial Tone, per month	NRC - Disconnect	NRC	ě	NRC - Disconnect	NRC	old, per month	NRC - Disconnect	NRC	Call Transfer, per month	NRC - Disconnect	NRC	Remote Call Forwarding, per month	NRC - Disconnect	NRC		NRC - Disconnect	NRC STREET	rwarding Busy I ine per month	NRC - Disconnect	Call Foliwarding variable, per illoitti	NAC - Discollinect	NRC Discorport	Service per line, (in addition to port), per month	Multiline Hunt Service (Rotary)	NRC - Disconnect	NRC .	Selective Call Acceptance, per month	NRC - Disconnect	NRC	Selective Call Forwarding, per month	DESCRIPTION
\$1.03	\$0.00	\$0.55 \$0.55	\$103	\$0.55	\$1.03	\$0.04	\$0.55	\$1.03	\$0.15	\$0.55	\$1.03	\$0.01	\$0.55	\$1.03	\$0.10	\$0.55	\$1.03	\$0.01	\$5.22	\$28.94	\$28.29	\$0.55	\$1.03	\$0.00	\$0.55	\$1.03	\$0.09	\$0.55	\$1.03	\$0.41	\$0.55	\$1.03	\$0.93	\$0.55	\$1.03	\$0.03	\$0.55	\$1.03	\$0.04	\$0.55	\$1.03	\$0.03	\$0.55	\$1.03	\$0.12	\$0.55	\$1.03	\$1.36	\$0.55	\$1.03	\$0.03	\$0.55	\$1.03	\$0.03	\$0.50 50.50	\$0.00	\$0.05	\$1.03	\$0.11	9)	\$0.55	\$1.03	\$0.29	\$0.55	\$1.03		USOC AL
NA :	2 3	× ×	× ×	<b>X</b>	×	X	¥	¥	¥	N.	¥	¥	¥	¥.	¥.	N.	¥.	N.	N.	NA	N.	N.	NA NA	¥	¥	¥	¥.	₹	AN	AN	AN	NA	N.	NA	NA	N.	¥	¥	¥	¥.	¥ :	¥.	¥ :	¥.	¥	¥.	N.	¥.	¥ ;	N S	N S	NA S	N S	NA .	NA 5	5	3	× ×	<b>₹</b>	;	NA.	<b>X</b>	₹	NA	¥	NA	P
NA :	N S	8	× ×	<b>₹</b>	<b>X</b>	X	¥	¥	N <sub>A</sub>	¥.	¥	¥	¥	NA NA	¥ :	NA.	NA.	N.	NA.	NA	N.	NA :	NA NA	¥	¥	¥	NA NA	¥	AN	AN	NA AN	NA	NA.	NA	NA	N	¥	¥	¥	¥.	₩ :	NA.	¥ :	¥ :	¥.	¥.	NA A	¥.	¥ ;	X S	X S	NA S	N S	NA S	N 5	5	3	× ×	: ₹	;	×	<b>X</b>	. ₹	NA	¥	NA	GA
NA S	2 3	× ×	¥	\$ ₹	×	¥	¥	NA.	N <sub>A</sub>	¥.	¥	¥	¥	N <sub>A</sub>	¥.	NA NA	NA.	¥.	NA.	NA A	¥.	NA.	N.	N <sub>A</sub>	N.	¥	NA NA	¥	AN	AN	AN	NA	NA NA	NA	NA	NA.	N <sub>A</sub>	¥	¥	¥	¥ :	NA.	¥ :	¥.	¥	¥.	NA NA	¥.	¥ ;	N S	N S	NA S	N S	NA S	NA S	3	5		; ₹	;	NA.	¥	₹	NA	NA	NA	KY
NA S	Z 3	N A	N A	3 3	×.	¥	¥	NA NA	NA	¥.	¥	¥	¥	N.	¥.	¥.	¥.	¥.	N.	NA A	¥.	N.	N.	¥.	¥	¥	N.	₹	AN	NA	NA.	NA	NA A	NA	NA	¥.	¥.	¥	¥	¥	<b>X</b> :	¥.	¥ :	¥ :	¥	¥.	N <sub>A</sub>	¥.	¥ :	N S	X S	NA S	N S	NA .	NA S	3	3	- N	₹ ₹	;	NA.	¥	₹	NA	NA	NA	LA
\$1.02	\$0.5466	\$1.02	\$0.0443	\$0.5466	\$1.02	\$0.0464	\$0.5466	\$1.02	\$0.1071	\$0.5466	\$1.02	\$0.0013	\$0.5466	\$1.02	\$0.1115	\$0.5466	\$1.02	\$0.0030	\$5.16	\$28.61	\$50.89	\$0.5466	\$1.02	\$0.0013	\$0.5466	\$1.02	\$0.0932	\$0.5466	\$1.02	\$0.5015	\$0.5466	\$1.02	\$0.9519	\$0.5466	\$1.02	\$0.0356	\$0.5466	\$1.02	\$0.0387	\$0.5466	\$1.02	\$0.0190	\$0.5466	\$1.02	\$0.1404	\$0.5466	\$1.02	\$1.47	\$0.5466	\$1.02	\$0.0308	\$0.5466	\$1.02	\$0.0279	\$0.5466	\$1.0474	\$0.0474	\$1.02	\$0.1271	***	\$0.5466	\$1.02	\$0.4010	\$0.5466	\$1.02	\$0.1050	SW
\$1.51	\$0.14	\$1.51	\$0.09	\$ ₹	\$1.51	\$0.09	¥	\$1.51	\$0.26	×.	\$1.51	\$0.0020	¥	\$1.51	\$0.13	NA.	\$1.51	\$0.0041	NA.	\$33.33	\$19.83	NA.	\$1.47	\$0.0011	NA.	\$1.47	\$0.07	¥	\$1.47	\$0.29	NA	\$1.51	\$1.29	NA	\$1.51	\$0.03	¥	\$1.51	\$0.10	×	\$1.51	\$0.15	X S	\$1.51	\$0.14	×	\$1.51	\$0.95	N S	\$1.51	\$0.09	NA S	\$1.51	\$0.08	NA .	61.51	20.40	91.51	\$0.14	*	×	\$1.51	\$0.33	NA	\$1.51	\$0.28	NC
\$1.51	\$0 1179	\$1.51	\$0.0694	PO OSO	\$1.51	\$0.0708	¥	\$1.51	\$0.2149	¥.	\$1.51	\$0.0101	¥	\$1.51	\$0.1048	NA.	\$1.51	\$0.0116	NA.	\$33.36	\$0.3621	NA.	\$1.47	\$0.0013	¥	\$1.47	\$0.0891	¥	\$1.47	\$0.3513	NA.	\$1.51	\$1.13	NA	\$1.51	\$0.0318	¥	\$1.51	\$0.0743	¥.	\$1.51	\$0.0677	N S	\$1.51	\$0.1392	¥	\$1.51	\$1.41	X S	\$1.51	\$0.0655	NA S	\$1.51	\$0.0603	NA .	\$1.0700	20020	ψ1.5.T	\$0.1301	,	×	\$1.51	\$0.3283	NA	\$1.51	\$0.1287	SC
NA S	N S	N K	× ×	¥ ¥	×	¥	¥	Ą	NA.	¥.	¥	¥	¥	¥.	¥.	N.	N.	NA.	NA.	NA A	NA.	N.	N.	¥	¥	₹	¥.	₹	NA.	NA	NA.	NA	NA	NA	NA	NA.	¥	₹	¥	N.	¥ :	N.	¥ :	¥.	¥	¥.	N.	¥.	¥ ;	N S	X S	NA S	N ;	NA .	NA 5	5	3	× ×	X X	;	NA	¥	₩	NA	NA	NA	TN

NRC - Manual - 1st	NRC - Electronic - Add'l	NRC - Flectronic - 1st	NRC - Disconnect	NRC		NRC - Disconnect	ISDN Message Waiting Indication-Lamp, per month	All Selective Class Of Call Screening		Non-List Listing No Rate	Non-List Listing	Non-Pub Listing No Rate	Cross Reference Listing	Additional Listings	All Customized Code Restrictions	Make Set Busy – Intragroup	Call Park/Call Retrieve	Selective Call Acceptance		Automatic Line/Direct Connect	Call Return Denial Of, Per Activation	Per Line Blocking For Non-Pub Customers	Per Line Blocking For Non-Pub Customers	Per Line Blocking For Non-Pub, And Non-Listed Customer	Per Line Blocking For Non-Pub Customers	Per Line Blocking For Agencies/Law Enforcement	Repeat Dialing	Preferred Call Forwarding	Call Return	Call Tracing	Additional Call Appearance, PDN Or DN	Audible Message Waiting Indicator	Speed Calling Visual Message Westing Indicator	Multi-Line Hunt Group – Data	Multi-Line Hunt Group – Voice Or Voice/Data	Conference, Drop, Hold And Transfer	2	Call Forwarding Don't Answer – Programmable - Data	n't Answer-Pro	Call Forwarding Don't Answer – Data	. 5	Call Frwdng Busy Line-Prgrmmbl-Voice Or Voice/Data	Call Forwarding Busy Line – Voice Of Voice/Data	Call Forwarding Variable – Feature Button – Data	Call Forwarding Variable - Feature Button - Voice	Call Forwarding Variable – Data	Call Forwarding Variable-Voice Or Voice/Data	Privacy Release	Shared Non-ISDN DN	Shared Secondary Only Dn-First Appr On Each Add'l Term	(Shared/Non-Shared) First	1	MICO - Discolling of	DESCRIPTION   NPC - Discognost
								SRG++	FNA	I Z	NLT	NP3	= =	CLT	CREX+	M6MGD	M6HP6	M6K16	M6MPD	M6GN9	BCR	NOBNR	NOBNP	NOBPP	NOBNN	NOB	NSQ	NO.	NSS	NST	DS1FG	MWW	LIZSU	HTGSD	HG	DS1FN	M6CD5	M6CV5	M6BVA	LLUCD	M6ADF	M6AVA	LLRCD	LLPCD	GJXCF	LLOCD	LLNCV	DS1FU	DOE	DS1F1	LLDSF	7		USOC
\$4.80	\$0.96	\$2 88 A	\$0.55	\$1.03	NA S	\$0.55	\$0.01	TBD IBD	TBD	TBD	TBD	TBD	T E	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD ED	TBD	TBD	TBD	TBD	TBD	180	TBD	TBD	TBD IBD	TBD	TBD	TBD	TBD ED	1	60.00	\$0 55 50 55
NA	X.	Z Z	X X	NA	¥ ₹	Z Z	S A	TBD	TBD	TBD	TBD	TBD	THE S	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD IBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD IBD	TBD	TBD	TBD	TBD IBD	1	3	N FL
¥	<b>¥</b> :	N X	₹	NA	₹ 5	N S	¥	룡		TBD	TBD	TBD	<b>#</b>	룡룡	TBD	큠		ТВО	TBD	國區	룡룡	TBD	TBD	围	間	TBD	國		曹	ТВD	dar dar		围	TBD (	國	間	TBD (	西西	TBD	큠	围	ТВО			TBD	ТВD	围	TBD	TBD	TB G	西西	d	5	GA NA
NA	X.	Z Z	¥	NA	₹ ₹	Z Z	<b>X X</b>	國	费图	ТВО	TBD	TBD	# E	룡룡	TBD	큠	# E	TBD	TBD	間回	룡룡	TBD	TBD	BB B	間	TBD	TBO E		間	TBD	명		間	TBO		間間	TBD G		TBD	TBD F	間	TBD	B 0		TBD	TBD		TBD	TBD	TBIO C		j	5	N X
NA	X.	NA A	¥	NA	¥ 3	NA S	NA A	TBD	TBD	TBD	TBD	TBD	T 5	TBD	TBD	TBD 18	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	<b>5 8</b>	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD E	TBD	TBD	TBD	TBD	TBD	T 180	TBD	TBD	TBD IBD	TBD	TBD	TBD	TBD E	1	5	LA NA
\$4.73	\$0.95	\$2.84	\$0.5466	\$1.02	NA NA	\$0.5466	\$0.0105	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD IBD	1	\$0.0400	<b>MS</b>
\$1.89	\$0.95	\$5.42	¥	\$1.51	<b>₹</b>	\$1.4/	\$0.0107	TBO	TBD	TBD	TBD	TBD	TB E	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD IBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD IBD	1	3	NC NC
\$7.35	\$0.71	\$1.36	¥	\$1.51	5	\$1.4/	\$0.0138	TB)	TBD	TBD	TBD	TBD	TBD E	TBD	TBD	TBD	TBD	TBD	TBD	TBD E	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD E	TBD	TBD	TBD E	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD IBD	TBD	TBD	TBD	TBD E	1	5	SC
¥	¥ :	NA	¥	NA	5	N N	<b>₹</b>	TBD B	TBD	TBD	TBD	TBD	# E	TBD	TBD	TBD 68	# E	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD I	TBD	TBD	TBD I	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD IBD	TBD	TBD	TBD	TBD ED	1	3	N T

### NRC - Manual - Add'l Tandem Switching (Port Usage) (Local or Access Tandem) Tandem Switching Function per mou Tandem Interoffice Trunk Port - Shared per mou Inhundled Port Usage Charges End Office Switching (Fort Usage) End Office Switching Function, per mou End Office Interoffice Trunk Port—Shared, per mou Common (Shared) Transport 5 This rate element is for use in those states with a different rate for additional minutes of use. Interim rates subject to true-up. 1 Port rate includes all available features. 2 Transmission/us age charges associated with POTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by B-Channels associated with 2-wire ISDN ports. This rate element is for those states which have a specific rate for User Profile per B Channel. Access to B Channel or D Channel Packet capabilities will be avail- able only through BFR/New Business Request Process. Rates for the packet capabilities will be determined via the Bona Fide Request/New Business Request Process. Common (Shared) Transport per mile per mou Common (Shared) Transport Facilities Termination per mou USOC N/A N N X X \$0.00001 \$0.00045 \$0.0018 \$0.0002 \$0.00063 \$0.96 \$2.88 \$0.0001522 \$0.0002713 \$0.000039 \$0.00008 \$0.000049 \$0.000083 \$0.000091 \$0.0004579 \$0.0004152 \$0.000426 \$0.00047 \$0.0004281 \$0.0008846 \$0.0016333 \$0.0001893 \$0.0001564 ₹ ₹ ₹ \$0.0006757 \$0.0002126 ≅ ≅ **g** \$0.002562 NA \$0.001096 NA ₹ ₹ ₹ \$0.0021 \$0.0002 \$0.0008 ₹ ₹ ⋝ \$0.0023771 \$0.0001927 \$0.0007834 \$0.0002834 \$0.95 \$2.84 1 \$0.00001 9 1 \$0.0017000 7 NA \$0.0009 NA \$0.0019295 \$0.0002581 \$0.0000121 \$0.0000064 \$0.0004672 \$0.0003871 \$0.0006843 \$0.0004034 \$0.95

\$0.000978 NA

\$0.0008 NA

₹ ₹ **₹** 

NOTES:

6 Rates in TN and FL are interim and shall be trued-up when final rates are ordered

Where the state Commission has adopted rates for the rate elements contained herein, it is the intent of the parties to reflect such rates in this exhibit and to apply the same consistent with applicable FCC and Commission rules and orders.

NRC - DS3 - Facility Termination - Disconnect - 1st NRC - DS3 - Facility Termination - Disconnect - Add'l	NRC - DS3 - Facility Termination - Add'l	NRC-	DS3 - Facility Termination per month	- per mile	Local Channel - Dedicated - DS3		NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect	NRC - DS1 - Incremental ChargeManual Svc Order - Add'i						NRC - Dol - Disconnect Crig - Add I	NBC - Dat - Disconnect Chg - Ndd'i	NDO DOL Disconnect Obs. 15t	NBC - DS1 - Add'll	NRC - DS1 - 1st	DS1 per month	DS1 per mile	Local Channel - Dedicated - DS1	NSC - +-wille vo - ilicielle iliai chaige - ivalidai ovo ordei-biscollieco	NIDO A Miro VO Incremento Chargo Manual Res Order Disconnect	14/C - +- Wile & G - Ilicielle Iliai Charge - Ivarida Ovc Older - Cod	4	Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - 4-WITE VG - DISCONNECT CNg - Add I	NRC - 4-Wife VG - Disconnect Crig - 1st	NTC - 4-Wile VG - Add	NEO 4 Wile VO - 13t	NRC - 4-Wire VG - 1st	A-wire VC Monthly Requiring per month	4-wire VG per mile	Local Channel - Dedicated - 4-Wire VG	NRC - 2-Wire VG - Incremental ChargeManual Svc Order-Disconnect		NRC - 2-Wire VG - Incremental ChargeManual Svc Order - Add'l						NRC - Manual Svc Order per I SR			NRC - 2-wire VG -Add'l	NRC - 2-wire VG - 1st	2-write VG Monthly Recurring per month	2-wire VG per mile	Top-Observed Delicated a Wiss NO Day Del	NRC - 2-Wire VG - Incremental ChargeManual Svc Order-Disconnect			NRC - Electronic Svc Order, per LSR disconnect					)hg - 1st		NRC - 2-wire VG - 1st	2-wire VG Monthly Recurring per month	2-wire VG per mile	Local Channel - Dedicated - 2-Wire VG	UNBUNDLED DEDICATED TRANSPORT - Local Channel	
ULDF3	ULDF3	ULDF3	ULDF3	1L5NC			SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN	OLDFI			III DE1	ULDF1	ULDF1	1L5NC		COMPA	SOMAN	CONTO	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN	OLDV4						1L5NC		SOMAN		SOMAN	SOMAN	SOMEC	SOMEC	CONTRA	SOMAN	SOMAN	III DRO	ULDR2	ULDR2	ULDR2	ULDR2	1L5NC		SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN	ULDV2	ULDV2	ULDV2	ULDV2	ULDV2	1L5NC			USOC
\$118.54	\$426.28	\$640.54	\$535.92	\$8.44		-	\$29.27	\$0.00	\$61.95	₹	\$3.50	X N	¥	\$32.10	\$20.20	646.20	\$442.84	\$503.57	\$35.52	\$0.00		\$17.70	¢17 76	Ø10.70	\$18.73	\$27.37	\$3.50	¥	¥	NA	\$8.53	\$/8./1	\$00.00	900000	\$502.43	\$15.77	\$0.00		\$17.75	• !	\$18.37	\$27.37	NA	\$3.50		NA S	NA.	\$7.63	\$77.81	\$84.44	\$494.65	\$14.61	\$0.00		\$17.75	\$18.37	\$27.37	NA NA	\$3.50	NA.	NA.	\$7.63	\$77.81	\$84.44	\$494.65	\$14.61	\$0.00		i	₽
\$154.90	\$528.05	\$903.37	\$556.27	\$9.16			X :	N :	NA	\$0.42	\$2.75	\$3.84	\$21.56	\$20.20	\$29.79	£44 43	\$226.44	\$242.45	\$43.53	\$0.00		5	25	5	NA S	NA	\$0.42	\$2.75	\$3.84	\$21.56	\$6.79	\$63.78	\$07.22	2007	\$387.21	\$30.50	\$0.00		¥		X	¥	\$0.42	\$2.75	\$3.04	\$2.50	\$21.56	\$5 Q2	\$67.91	\$66.36	\$3,865.34	\$29.33	\$0.00		NA	¥	N .	\$0.42	\$2.75	\$3.84	\$21.56	\$5.92	\$67.91	\$66.36	\$386.34	\$29.33	\$0.00			<b>P</b>
\$112.31	\$426.31	\$639.50	\$515.91	\$6.92		4	\$18.03	N I	\$44.22	¥.	\$3.50	Ā	¥	\$119.14	\$110.11	64000	\$312.89	\$356.15	\$38.36	\$0.00		Š	5	40.44	\$8.42	\$18.94	¥	\$3.50	¥	Ā	Š	\$	5 5	\$100.44	\$368 44	\$14.99	\$0.00		¥		\$8.42	\$18.94	NA	\$3.50		N 5	N S	NA .	NA	\$62.40	\$382.95	\$13.91	\$0.00		NA.	\$8.42	\$18.94	NA S	\$3.50	N S	NA	¥.	¥	\$62.40	\$382.95	\$13.91	\$0.00		9	GA
NA A	\$661.23	\$1,091	\$635.09	\$34.00			X.	NA:	\$87.71	N.	\$3.50	NA NA	\$19.99	NA.	2 2	NIA C	\$464.94	\$538.95	\$43.80	\$0.00		3	2	<b>4</b> 1.33	\$11.00	\$41.46	¥	\$3.50	¥	\$19.99	¥.	· ·	φ90.33	\$00 E3	\$585.15	\$23.38	\$0.00		X	:	\$11.99	\$41.46	Ä	\$3.50		WIO.	\$19.99	NA.	NA.	\$98.53	\$585.15	\$22.26	\$0.00		NA	\$11.99	\$41.46	NA CO	\$3.50	NA O	\$19.99	¥.	¥	\$98.53	\$585.15	\$22.26	\$0.00			₹
\$99.46	\$402.63	\$709.14	10.699\$	\$30.34			\$19.48	N .	\$42.34	AN AN	\$3.50	Ā	X	\$21.31	\$24.15	37.70	\$300.30	\$348.56	\$43.80	\$0.00		\$1.1.40	611 10	<b>#0.00</b>	30.83	\$1814	Ā	\$3.50	Ā	NA	\$1.28	\$54.30	00.100	604.33	37 CYES	\$16.21	\$0.00		\$11.40	•	\$8.06	\$18.14	NA	\$3.50	7	N 5	AN OO:OO	09.93	\$53.68	\$59.75	\$347.49	\$14.94	\$0.00		\$11.40	\$8.06	\$18.14	AN OC.CO	04.53	NA.	NA	\$6.60	\$53.68	\$59.75	\$347.49	\$14.94	\$0.00			5
\$42.41	\$493.71	\$526.67	\$533.33	NA		-	\$27.40	NA S	\$59.58	N.	\$3.50	¥	¥	\$33.02	\$33.03	90 90 9	\$435.28	\$494.83	\$38.91	\$0.00		\$17.23	647 25		\$11.34	\$25.52	¥	\$3.50	¥	NA	\$9.84	\$78.58	\$20.00	\$00 EC	\$495.25	\$19.00	\$0.00		\$16.05	,	\$11.34	\$25.50	Ä	\$3.50		N 5	NA C	\$8 Q5	\$77.69	\$84.35	\$487.62	\$17.83	\$0.00		\$16.05	\$11.34	\$25.50	NA C	\$3.50	NA.	NA	\$8.95	\$77.69	\$84.35	\$487.62	\$17.83	\$0.00			NS
X X	\$527.88	\$562.25	\$498.87	NA			¥.	\$1.77	\$86.15	¥	\$3.50	¥	¥	3	N	VIV.	\$462.69	\$534.48	\$35.68	\$0.00	Ī	5	<u>-</u>	⊕12.70	\$12.76	\$4217	₹	\$3.50	¥	X	3	3	ψ92.07	23.CO	\$562.23	\$15.87	\$0.00		¥	:	\$12.76	\$42.17	NA	\$3.50	200	NA 5	NA S	NA.	N.	\$89.69	\$553.80	\$14.83	\$0.00		¥	\$12.76	\$42.17	NA C	\$3.50	NA.	NA.	¥.	¥	\$89.69	\$553.80	\$14.83	\$0.00		i	ਨ
NA NA	\$519.31	\$735.42	\$493.31	\$12.08			NA:	\$3.11	\$87.99	NA.	\$3.50	¥	¥	3	2 2	WIA C	\$462.81	\$534.81	\$37.20	\$0.00		5	5	6 0.00	\$13.55	\$43.64	¥	\$3.50	¥	NA.	3	NA.	70.180	\$04.57	\$562.46	\$18.05	\$0.00		¥		\$13.55	\$43.75	NA	\$3.50		No 5	NA	No.	NA.	\$88.58	\$554.00	\$16.83	\$0.00		M	\$13.55	\$43.75	AN OC.OC	\$3.50	No.	NA	¥.	¥	\$88.58	\$554.00	\$16.83	\$0.00			SC
\$100.59	\$411.64	\$726.16	\$607.28	\$23.76			≸ :	≸ :	₹	₹	\$3.50	¥	\$19.99	\$22.30	\$33.10	#100:H0	\$233.26	\$277.35	\$40.27	\$0.00		3	5	3	K 5	¥.	¥	\$3.50	₹	\$19.99	\$0.0T	\$55.52	924.00	2000	\$201.53	\$20.14	\$0.00		₹	:	₹	₹	¥	\$3.50	9	NA CO	\$1999	\$4.80	\$54.81	\$24.16	\$199.33	\$19.02	\$0.00		₹	₹	₹:	₹ 5	\$3.50	NA CO	\$19.99	\$4.80	\$54.81	\$24.16	\$199.33	\$19.02	\$0.00		1	TN

								I	Ī					1														T				H													1		H		=
NRC - OC48 - Facility Termination -Add I  NRC - OC48 - Interface OC12 on OC48 - 1st	NRC - OC48 - Facility Termination - 1st	OC48 - Interface OC12 on OC48 per month	OC48 Facility Termination per month	Local Channel - Dedicated - OC48	INIVO - OCIT - Ilivialialial Olia Re-livialial Ose Oldal-Giscolliaet-Ameri	NRC - OC12 - Incremental ChargeManual Svc Order-Disconnect-Addi	NRC - OC12 - Incremental ChargeManual Svc Order-Disconnect-1st	NRC - OC12 - Incremental ChargeManual Svc Order - Add'l	NRC - CC12 - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - OC12 - Facility Termination - Disconnect Chg - Add'l	NRC - OC12 - Facility Termination - Disconnect Cha - 1st	NRC - OC12 - Facility Termination - 1st	OC12 Facility Termination per month	Local Channel - Dedicated - OC12 OC12 per mile per month	NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-Add1	NRC - OC3 - Incremental ChargeManual Svc Order-Disconnect-1st	NRC - OC3 - Incremental ChargeManual Svc Order - Add'l	NRC - OC3 - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - OC3 - Facility Termination - Disconnect Cha - Add'l	NRC - OC3 - Facility Termination - Add'I	NRC - OC3 - Facility Termination - 1st	OC3 Facility Termination per month	Local Channel - Dedicated - OC3	NRC - STS-1 - Incremental ChargeManual Svc Order-Disconnect-Add'l	NRC - STS-1 - Incremental Charge Manual Svc Order-Disconnect -1st	- STS-1 - Incremental ChargeManual Svc Order -	- STS-1 -Incremental C	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order per LSR	- STS-1 - Facility Termination - Disconnect -	NRC - STS-1 - Facility Termination - Add'l	NRC - STS-1 - Facility Termination - 1st	STS-1 - per mile per month	Local Channel - Dedicated - STS-1	NRC - DS3 - Incremental Charge Manual Svc Order-Disconnect-Addl	NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect -1st	NRC - DS3 - Incremental ChargeManual Svc Order - Add'l	NRC - DS3 -Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	
				1 500	OCIVICA	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMAN	SOMAN					1L5NC	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMAN	SOMAN				LONC		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMAN	SOMAN	ULDFS	ULDFS	ULDFS	1L5NC		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMAN	SOMAN	5
\$540.10	\$1,165	\$699.62	\$1,947	\$33 99	φ1σ.00	\$19.03	\$19.03	\$38.48	\$38.48	\$3.50	¥	AN	\$118.54	\$121.72	\$1,165	\$5,630	\$10.13	\$19.03	\$19.03	\$38.48	\$38.48	\$3.50	S A	NA	\$118.54	\$413.38	\$949.63	\$1.123		\$19.03	\$19.03	\$38.48	\$38.48	\$3.50	¥ :	\$118.54	\$121.72	\$426.82	\$640.54	\$8.44		\$19.03	\$19.03	\$38.48	\$38.48	\$3.50	N	¥ ¥	•
\$543.72	\$1,183.46	\$587.71	\$1,685.97	\$36.04	5	N <sub>A</sub>	¥.	¥ :	NA 54.5	\$0.42	\$2.75	\$3.84	\$21.56	\$108.34	\$408.85	\$1,183.46	\$2,053.06	NA.	¥	Ą	NA.	\$0.75	\$3.84	\$21.56	\$108.34	\$408.85	\$966.45	\$648.60	80	N <sub>A</sub>	NA	W	W 25:	\$2.75	\$3.84	\$754.90	\$221.46	\$528.05	\$903.37	\$9.16		NA	NA	NA.	AN -5-	\$2.75	\$3.84	\$21.56	1
\$539.36	\$1,162.00	\$558.07	\$1,671	\$27.25	6.00	\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	¥	NA	\$119.14	\$122.31	\$1,162.00	\$3,185.00	\$8.31	\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	NA NA	NA	\$119.14	\$413.00	\$947.69	\$914.22	9000	\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	¥.	\$119.14 NA	\$122.31	\$426.40	\$639.50	\$6.92		\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	NA	Z §	)
\$844.21	\$1,858	\$728.81	\$2,156	\$133 84	3	N A	¥	\$93.12	\$93.12	\$3.50	¥	\$19.99	NA :	NA S	\$1,858	\$4,492	\$40.80	NA.	NA.	\$93.12	\$93.12	NA 3.50	NA NA	\$19.99	<b>₹</b>	\$661.23	\$1,543	\$1,493	900	¥	¥	\$93.12	\$93.12	\$3.50	NA S	\$19.99	¥	\$661.23	\$1.091	\$34.00		NA	NA	\$93.12	\$93.12	\$3.50	N.	\$19.99	
\$532.13	\$1,245	\$706.85	\$2,311	\$110 40	920.94	\$20.94	\$20.94	\$50.25	\$50.25	\$3.50	×	NA	\$99.46	\$102.16	\$1,245	\$3,895	\$36.40	\$20.94	\$20.94	\$50.25	\$50.25	\$3.50	S A	NA	\$99.46	\$402.63	\$1,025	\$1,179	900	\$16.77	\$16.77	\$34.92	\$34.92	\$3.50	¥ :	\$110.80	\$113.75	\$396.54	\$594.71	\$8.77		\$20.94	\$20.94	\$50.25	\$50.25	\$3.50	¥	<b>₹</b>	, ,
\$729.04	\$1,742.00	\$668.36	\$1,768.00	\$166.50	ψ20.00	\$28.59	\$28.59	\$68.62	\$68.62	\$3.50	¥	NA	\$130.59	\$134.07	\$1,742.00	\$3,414.00	\$50.79	\$28.59	\$28.59	\$68.62	\$68.62	\$3.50	S A	NA	\$130.59	\$549.17	\$1,427.00	\$873.23	900	\$25.35	\$25.35	\$96.10	\$96.10	\$3.50	¥ :	\$40.87	\$42.41	\$682.13	\$1.084.33	\$38.98		\$25.35	\$25.35	\$31.49	\$31.49	\$3.50	N	N N	
\$852.47			\$1,837	\$00.66	5	N A	¥	\$99.10	\$99.10	\$3.50	¥	NA	NA :	NA COL	\$1,853	\$3,316	\$30.38	NA.	¥	\$99.10	\$99.10	\$3.50	Š K	NA	T	-	H	\$914.18	224	¥	¥	\$56.25	\$56.25	\$3.50	¥ :	8	¥	\$527.88	\$562.25	N N		NA	NA	\$56.25	\$56.25	\$3.50	N.	₹ ₹	ś
\$635.04	\$1,259	\$773.40	\$1,842	\$47.57	5	N.	N.	\$54.26	\$54.26	\$3.50	×	NA	NA :	NA S	\$1,259	\$4,414	\$14.50	NA	NA.	\$54.26	\$54.26	\$3.50	S A	NA	X 5	\$519.31	\$735.42	\$493.31	9	N <sub>A</sub>	NA	\$54.26	\$54.26	\$3.50	¥ :	X X	¥	\$519.31	\$735.42	\$12.08		¥.	NA	\$54.26	\$54.26	\$3.50	N	გ გ	3
\$544.55	\$1,276	\$572.61	\$1,853	\$03.50	3	₹	₹	₹ :	<b>₹</b> ₹	\$3.50	₹	\$19.99	\$100.59	\$103.36	\$1,276	\$7,158	\$28.51	₹	₹	¥	<b>X</b> :	\$3.50	Ž ¥	\$19.99	\$100.59	\$411.64	\$1,050	\$1 263	94000	₹	¥	¥	₹ 3	\$3.50	≰ 8	\$19.99	\$103.36	\$683.01	\$1,085.73	\$25.11		¥	N	¥	₹ 5	\$3.50	₹	\$19.99	1

			F																																										ŧ						Ħ
NRC - Electronic Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - 4-wire VG - Facility Termination - Disconnect Charge -Add'l	NRC - 4-wire VG -Facility Termination - Disconnect Charge -1st	NRC - 4-wire VG - Facility Termination - Add'l	NRC - 4-wire VG - Facility Termination -1st	4-Wire VG - Facility Termination per month	4-Wire VG - per mile per month	Interoffice Transport - Dedicated - 4-wire VG	Continon (Shared) Harisbort Facilities Termination ber mod	Common (Shared) Transport Escilitios Termination per mon	Common (Shared) Transport for mile per mou	NRC - 2-wire VG - Incremental ChargeManual Svc Order-DisconnectAddl	NRC - 2-wire VG - Incremental ChargeManual Svc Order-Disconnect1st	MINO - 7-MILE AO - Illore Herital Orial Barrana Osto Olde - Oral	NRC - 2-wire VG - Incremental ChargeManual Svc Order - 1st	R disconnect	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - 2-wire VG - Facility Termination - Disconnect Charge -1st	NRC - 2-wire VG - Facility Termination - Add'l	NRC - 2-wire VG - Facility Termination -1st	2-Wire VG - Per mile per month	Interoffice Transport - Dedicated - 2-wire VG - Rev Bat.	NRC - 2-wire VG - Incremental ChargeManual Svc Order-DisconnectAdd'i	NRC - 2-wire VG - Incremental ChargeManual Svc Order-Disconnect1st	NRC - 2-wire VG - Incremental ChargeManual Svc Order - Add'l	NRC - 2-wire VG - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - 2-wire VG - Facility Termination - Disconnect Charge -Add'l	NRC - 2-wire VG - Facility Termination - Disconnect Charge -1st	NRC - 2-wire VG - Facility Termination - 1st	2-Wire VG - Facility Termination per month	2-Wire VG - per mile per month	Interoffice Transport - Dedicated - 2-wire VG	Disconnect-Add'l	NRC - OC48 - Incremental ChargeManual Svc Order-Disconnect-Add'l	NRC - OC48 - Incremental ChargeManual Svc Order-Disconnect-1st	Add'l		NRC - OC48 - Incremental ChargeManual Svc Order - Add'l	NRC - OC48 - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR		NRC - Manual Svc Order, per LSR	NRC - OC48 - Interface OC12 on OC48 - Disconnect Chg - 13t	NRC - OC48 - Facility Termination - Disconnect Chg - Add1	NRC - OC48 - Facility Termination - Disconnect Chg - 1st	NRC -OC48 - Interface OC12 on OC48 -Add'l
SOMEC	SOMAN	U1TV4	U1TV4	U1TV4	U1TV4	U1TV4	1L5XX		ĺ	2 5		SOMAN	SOMAN	C	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN	U1TR2	U1TR2	U1TR2	1L5XX		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEO	SOMAN	U1TV2	U1TV2	U1IV2	U1TV2	1L5XX		SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMAN	SOMAN				USOC
<b>₹</b> ₹	¥	¥	¥	NA	NA	NA	¥		\$0.00045	\$0.00001	en 00001	\$12.97	\$12.97	10.73	\$27.37	K	\$3.50	N :	NA 00	\$37.16	\$48.27	\$107.11	\$18.49		\$12.97	\$12.97	\$27.57	\$27.37	NA S	\$ NA	<b>X</b>	\$5.88	\$37.16	\$107.11	\$18.49	\$0.0339		\$19.03	\$19.03	\$19.03	\$38.48	\$38.48	\$38.48	\$38.48	\$3.50	N N	X.	\$118.54	\$178.54	\$121.72	<b>AL</b> \$317.48
\$3.84	\$21.56	\$12.78	\$31.01	\$54.63	\$81.09	\$23.64	\$0.0098		\$0.000	\$0.000012	en 000013	NA	NA	5	N N	\$0.42	\$2.75	\$3.84	\$21.56	\$31.01	\$54.83	\$81.09	\$0.0098		NA	NA.	Ą	NA	\$0.42	\$3.84	\$21.56	\$12.78	\$31.01	\$54.83	\$26.52	\$0.0098		¥ ¥	¥	NA	×	. ₹	NA	NA 24:04	\$2.75	\$3.84	\$21.56	\$108.34	\$108.34	\$111.56	\$312.05
<b>₹</b>	¥	¥	¥	¥.	¥	N	₹		_	\$0.00000	_	NA	NA	6.0.04	\$18.94	NA.	\$3.50	¥ :	₹ 5	<b>X</b> X	\$36.08	\$79.61	\$17.0222		¥	×	\$18.94	\$18.94	N S	\$3.50	\$ \$	NA	NA O	\$79.61	\$17.07	\$0.0222		\$18.03	\$18.03	\$18.03	\$37.55	\$37.55	\$37.55	\$37.55	\$3.50	N.	NA	\$119.14	\$119.14	\$122.31	<b>GA</b> \$317.38
<b>₹</b> ₹	¥	¥	NA.	NA	NA NA	NA	¥		\$0.000428	\$0.0000049	80,0000000	NA	NA	42.72	\$37.21	¥	\$3.50	N S	\$19.99	X X	\$56.21	\$142.31	\$0.0301	2	NA	N	\$37.21	\$37.21	NA S	\$3 50 NA	\$19.99	NA	W S	\$142.31	\$27.66	\$0.0301		¥ ¥	N N	NA	\$93.12	\$93.12	\$93.12	\$93.12	\$3.50	NA.	\$19.99	¥ 5	NA NA	¥	\$516.89
<b>₹</b>	¥	¥	¥	ΝA	NA NA	NA AN	¥			\$0.000003		\$8.06	\$8.06	÷	\$18.14	¥	\$3.50	¥.	NA S	\$28.03	\$34.54	\$76.20	\$1910	2	\$8.06	\$8.06	\$18.14	\$18.14	NA S	\$3.50 NA	¥ ×	\$5.37	\$28.03	\$76.20	\$19.10	\$0.0384		\$20.94	\$20.94	\$20.94	\$50.25	\$50.25	\$50.25	\$50.25	\$3.50	NA.	¥	\$99.46	\$102.46	\$102.16	\$304.90
<b>₹</b> ₹	×	¥	¥	NA.	NA NA	NA	¥		\$0.0004201	\$0.0000091	\$0,000,0001	\$11.34	\$11.34	\$CU.UZ	\$25.52	×	\$3.50	¥ :	NA S	\$38.05	\$48.83	\$106.72	\$0.0323	9	\$11.34	\$11.34	\$25.52	\$25.52	N S	\$3.50 NA	<b>X</b>	\$7.23	\$38.05	\$106.72	\$21.33	\$0.0323		\$28.59	\$28.59	\$28.59	\$68.62	\$68.62	\$68.62	\$68.62	\$3.50	N N	N.	\$130.59	\$130.59	\$134.07	\$404.94
X X	¥	¥	¥	NA.	NA A	NA	¥		\$0.00034	\$0.00001	\$0 00001	×.	N.	<b>\$00.07</b>	\$38.07	×	\$3.50	¥.	<b>8</b> 5	<b>X</b>	\$52.58	\$137.48	\$18.00	3	N.	NA	\$38.07	\$38.07	N S	\$3.50 NA	<b>X</b>	¥	N S	\$137.48	\$18.01	\$0.0282		X X	¥	NA	\$99.10	\$99.10	\$99.10	\$99.10	\$3.50	¥	<b>⊼</b> :	<b>₹</b>	Z Z	₹	\$528.57
<b>₹</b> ₹	¥	¥	¥	¥	¥	NA	¥		\$0.0004072	\$0.000121	en 0000131	N N	N	<b>#</b> 00.00	\$39.63	NA.	\$3.50	¥.	₹ 3	<b>3</b>	\$51.37	\$136.44	\$0.0373	2	NA	¥	\$39.63	\$39.63	NA S	\$3 50 NA	<b>X</b>	NA	¥ S	\$136.44	\$21.42	\$0.0373		<b>X X</b>	¥	NA	\$54.26	\$54.26	\$54.26	\$54.26	\$3.50	N.	¥ :	<b>₹</b> 5	N X	¥	\$410.02
<b>₹</b> ₹	₹	₹	₹	A	₹	NA.	₹		\$0.0003b	\$0.00004	en 00004	₹	¥	3	¥ ₹	₹	\$3.50	¥ is	\$19.99	\$27.96	\$17.37	\$55.39	\$1833	2	₹	₹	¥	¥	¥ S	\$3 50 \$3 50	\$19.99	\$3.51	\$27.96	\$1737	\$18.33	\$0.0173		<b>₹</b> ₹	₹	₹	₹	₹	¥	<b>₹</b>	\$3.50	₹	\$19.99	\$100.59	\$100.59	\$103.36	\$311.39

								l																																							l									t				
NRC - DS3 - Incremental Charge-Manual Svc Order-Disconnect—Add1 Interoffice Transport - Dedicated - STS-1	NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect1st	NRC - DS3 - Incremental ChargeManual Svc Order - Add'l	NRC - DS3 - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Flectronic Syc Order per LSR	NRC - Manual Svc Order per LSR disconnect	NRC - Manual Syc Order per LSR	NRC - DS3 - Facility Termination - Disconnect Charge - Add'i	NDC - DC3 - Facility Termination - Disconnect Charge - 1st	NRC - DS3 - Facility Termination - Add'I	NRC - DS3 - Facility Termination -1st	DS3 -Facility Termination per month	DS3 - per mile per month	Interattice Transport - Dedicated - DS3	NRC - DS1 - Incremental ChargeManual Svc Order-DisconnectAdd'i	NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect1st	NRC -DS1 - Incremental ChargeManual Svc Order - Add'l	NRC - DS1 - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Sys Order per I SP	NRC - US1 - Facility Termination - Disconnect Charge - 1st	NRC - DS1 - Facility Termination - Add'l	NRC - DS1-Facility Termination - 1st	DS1 - Facility Termination per month	DS1 - per mile per month	Interoffice Transport - Dedicated - DS1	NRC - DS0 -incremental ChargeManual Svc Order-Disconnect—Add1	NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect1st	NRC -DS0 - Incremental ChargeManual Svc Order - Add'l	NRC - DS0 -Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Manual Ove Order, per Lox disconnect	NRC - Manual Svc Order, per LSR	NRC - DS0 - Facility Termination - Disconnect Charge - Add'l	NRC - DS0 -Facility Termination - Disconnect Charge - 1st	NRC - DS0 - Facility Termination - Add'l	NRC - DS0 - Facility Termination - 1st	DS0 - per mile per month	Interoffice Transport - Dedicated -DS0 - 64 kbps	Q	NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect—Add'i	NRC -USC - Incremental Charge - Manual Svc Order - Addil	NRC - DS0 -Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Flectronic Svc Order, per LSR	NRC - Manual 8vc Order, per Lox	NRC - USU - Facility Termination - Disconnect Charge - Addi	NRC - DS0 - Facility Termination - Disconnect Charge - 1st	NRC - DS0 - Facility Termination - Add'l	NRC - DS0 - Facility Termination - 1st	DS0 - Facility Termination per month	Interoffice Transport - Dedicated - DS0 - 56kbps	INC. +-Mile AG. Hicialiandia Chaige-hianna SAC Older-DiscollingsWari	NRC - 4-wire VG - Incremental ChargeManual Svc Order-Disconnect1st	NRC - 4-wire VG - Incremental ChargeManual Svc Order - Add'l	NRC - 4-wire VG - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	
SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN	INTE3	INTE3	INTE3	UITE3	LIATE3	11 5XX		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN	U1 F1	U1TF1	U1TF1	U1TF1	1L5XX		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOME	SOMAN	U1TD6	U1TD6	U1TD6	UITOS	1L5XX			SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	NAMON	SOMAN	U1TD5	U1TD5	U1TD5	ILDXX	1	CONST	SOMAN	SOMAN	SOMAN	SOMEC	USOC
\$19.03	\$19.03	\$38.48	\$38.48	N S	\$3.50	NA S	NA S	\$118.54	\$121.72	\$330.02	\$51177	\$898 15	\$4.98		\$12.97	\$12.97	\$27.37	\$27.37	NA	\$3.50	<b>₹</b>	\$20.4Z	\$25.44	\$148.18	\$198.15	\$79.69	\$0.6920		\$12.97	\$12.97	\$27.37	\$27.37	NA 00	\$ N	<b>X X</b>	\$5.88	\$37.16	\$48.27	\$107.11	\$0.0339		4 : 1 : 1	\$12.97	\$27.37	\$27.37	¥.	\$3.50	5	\$5.88	\$37.16	\$48.27	\$107.11	\$17.81	200	5	3 8	¥	¥	ĕi	₽
NA.	¥	NA	NA	\$0.42	\$2.77	\$3.84	\$21.56	\$108.34	\$111.56	\$305.00	\$557.69	\$1 121 93	\$4 17		¥	¥	Ą	¥	\$0.42	\$2.77	\$3.84	\$21.56	\$30.30	\$44.18	\$45.91	\$99.79	\$0.6013		¥.	¥	NΑ	¥.	\$0.42	\$3.84	\$21.56	\$12.78	\$31.01	\$54.83	\$81.31	\$0.0098			₹ 3	X X	¥	\$0.42	\$2.77	00.120	\$12.78	\$31.01	\$54.83	\$81.11	\$19.31	9	5	× ×	¥	¥.	\$0.42	2
\$18.23	\$18.23	\$37.96	\$37.96	NA S	\$3.50	NA S	NA.	\$11914	\$122.77	\$330.77	\$511 10	\$788.00	\$2.75		Ä	¥	\$18.94	\$18.94	NA	\$3.50	¥ ₹	2 5	× ×	\$111.75	\$147.07	\$78.47	\$0.4523		NA	¥	\$18.94	\$18.94	NA AN	es NA	X X	NA AN	NA	\$36.08	\$70.45	\$0.0222			K 5	\$18.94	\$18.94	W.	\$3.50		N N	<b>X</b>	\$36.08	\$79.61	\$16.0222	200	5	× ×	A	Ā	¥ :	G A
¥	¥	\$93.12	\$93.12	NA S	\$3.50	NA S	\$19 99	Z 5	NIA CO	\$516.89	\$946.23	\$1 204	\$12.62		NA	¥	NA	¥	NA	\$3.50	AN Sec. et a	\$10.00	× ×	\$231.23	\$298.18	\$55.05	\$0.4500		×	. ₩	\$37.21	\$37.21	NA 93.50	2 FO	\$19.99	AN	NA	\$56.21	\$142.33	\$0.0301			<b>₹</b>	\$37.21	\$37.21	¥.	\$3.50	\$19.99 88.81¢	\$10 00	. ₹	\$56.21	\$142.31	\$26.030	200	5	× ×	¥	NA A	¥ :	ξ
\$20.94	\$20.94	\$50.25	\$50.25	NA S	\$3.50	NA S	NA S	\$99.46	\$1001.30	\$304 90	\$611.41	\$1 101	\$14.04		\$8.06	\$8.06	\$18.14	\$18.14	NA	\$3.50	<b>₹</b> §	\$10.34	\$20.00	\$106.69	\$140.49	\$93.40	\$0.7831		\$8.06	\$8.06	\$18.14	\$18.14	\$3.50 AN	es NA	3 X	\$5.37	\$28.03	\$34.54	\$78.37	\$0.0384			\$8.06	\$18.14	\$18.14	NA	\$3.50	Z =	\$5.37	\$28.03	\$34.54	\$76.20	\$18.37		5	× ×	NA.	¥	¥;	
\$27.08	\$27.08	\$64.97	\$64.97	NA S	\$3.50	NA S	NA .	\$118.79	\$135.56 8135.56	\$477.76	\$686.74	\$744.38	\$15.02		\$11.34	\$11.31	\$25.52	\$25.52	NA	\$3.50	¥ ₹	0.12¢	\$26.56	\$147.31	\$196.28	\$74.40	\$0.6598		\$11.34	\$11.34	\$25.52	\$25.52	\$3.50 AN	es NA	X X	\$7.23	\$38.05	\$48.83	\$106.72			4	\$11.34	\$25.52	\$25.52	N.	\$3.50		\$7.23	\$38.05	\$48.83	\$106.72	\$20.0323	9	5	× ×	¥	NA A	¥ i	NIS
¥				NA S	\$3.50	NA S	NA.		+	\$57955	+		\$12.98		×	¥	\$		NA	\$3.50	<b>₹</b> §	25	<b>X</b>	\$163.75	\$217.17		\$0.0783		N.			9	NA 93.50	e NA	3 ₹	NA	NA	\$52.58	+	\$0.0282			<b>₹</b> §		\$38.07	NA.	\$3.50	2 2	N N		-		\$17.0282	H	5	<b>× ×</b>	NA.	¥	¥ ;	5
			\$	NA S	\$3.50	NA S	NA.	N S	ł	+	+	-Α	\$8 13		X	×.	\$39.63	\$39.63	NA	\$3.50	K 5	2 5	<b>X X</b>	\$1	-		\$0.7598		Ä	¥	\$39.63	\$39.63	NA SO	es an	<b>X X</b>	AN	NA	\$51.37			Н		K 5	\$39.63	\$39.63	NA.	\$3.50	2 2	Z Z	<b>X</b>			\$20.71	+	5	<b>5 5</b>	¥	NA.	¥ 8	SC
₹				N S	\$3.50	NA S	\$1999	\$100.59	_	\$31139	+	_	\$5.89		¥	₹	¥		NA	\$3.50	W 3.33	\$14.99	\$19.55	l	-		\$0.3525		¥	₹			NA SO	* N	\$19.99	\$3.51	\$27.96	\$17.37	4	\$0.1730			₹ 3		. ₹	¥	\$3.50	\$19.99 86.61¢	\$3.51	\$27.96	\$17.37	Н	\$17.74	+	3	<b>₹</b>	¥	¥	₹:	Į

								Ŧ														H							1																		F								
NRC - OC48 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st NRC - OC48 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l		1St	NRC - OCA8 - Incremental Cost - Manual Svc Order vs Electronic-Add'i	NRC - DC48 - Incremental Cost - Manual Svc Order vs Electronic-1st	lectronic Svc Order, p	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order, per LSR	NRC - OC48 - Interface OC12 on OC48 - Disconnect - Add'l	NRC - OC46 - Facility Tellitiliation - Disconnect - 1st	NRC - OC48 - Facility Termination - Disconnect Chg - Add'l	NRC - OC48 - Interface OC12 on OC48 - Add	NRC - OC48 - Interface OC12 on OC48 - 1st	NRC - OC48 - Facility Termination - Add'l	NRC - OC48 - Facility Termination - 1st	-per Inte		Interoffice Transport - Dedicated - OC48 OC48 -per mile per month	NRC - OC12 - Incremental Cost - Manual Svc Order vs. Elect-Disconnect-Add'l	NRC - OC12 - Incremental Cost - Manual Svc Order vs. Elect-Disconnect-1st	NRC - OC12 - Incremental Cost - Manual Svc Order vs. Electronic-Add'l	NRC - OC12 - Incremental Cost - Manual Svc Order vs. Electronic-1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order per LSR	NRC - OC12 - Facility Termination - Disconnect Chg - Add"	NRC - OC12- Facility Termination - Add'l		OC12 -Facility Termination	Interoffice Transport - Dedicated - OC12	AGG	1st	NRC - OC3 - Incremental Cost - Manual Svc Order vs. Electronic-Add'l	NRC - OC3 - Incremental Cost - Manual Svc Order vs. Electronic-1st	- Electronic	NRC - Manual Svc Order, per LSR disconnect		NRC - OC-3 - Facility Termination - Disconnect Charge - Add'l	NRC - OC-3 - Facility Termination - Disconnect Charge - 1st	NRC - OC-3 - Facility Termination - 1st	OC3 -Facility Termination per month	OC3 -per mile per month	Interoffice Transport - Dedicated - OC3		NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect1st	NRC - STS-1 - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NRC - STS-1 - Facility Termination - Disconnect Charge - Add'l	NRC - STS-1 - Facility Termination - Disconnect Charge - 1st	NRC - STS-1 - Facility Termination - Add'l	NRC - STS-1 - Facility Termination -1st	STS-1 - per mile per month	
SOMAN		SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN									10///	11.5XX	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	NAMOS					11 5XX	OCIVIAIN	SOMAN	SOMAN	SOMAN	SOMEC	SOME	SOMAN					1L5XX		SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	COMAN	U1TFS	U1TFS	U1TFS	1L5XX	USOC
\$19.03 \$19.03	\$38.48	\$38.48	\$38.48	\$38 48	\$3.50	NA	¥	\$118.54	\$121.72	\$11854	\$317.48	\$540.10	\$317.48	\$1,036	\$1,424	\$11.691	\$30.65	\$19.03	\$19.03	\$38.48	\$38.48	NA	\$3.50	<b>⊼</b> 5	NA OF	\$118.54	\$317.48	\$1,036	\$9,763	\$10.26	\$19.00	\$19.03	\$38.48	\$38.48	NA OC.50	\$ 50 NA	¥	\$118.54	\$121.72	\$820.85	\$2,475	\$7.35		\$19.03	\$19.03	\$38.48	NA S	\$3.50	<b>₹</b>	\$118.54	\$121.72	\$330.92	\$511.77	\$4.98	₽
¥ ¥	NA :	<b>₹</b> ₹	NA .	\$0.43	\$2.77	\$3.84	\$21.56	\$108.34	\$111.56	\$10834	\$312.05	\$543.72	\$312.05	\$1,086.66	\$1,199.42	\$12,460.76	\$34.07	NA	¥	X	¥	\$0.43	\$2.77	\$3.84	\$31.56	\$111.56	\$312.05	\$1,086.66	\$11,599.14	47.90	3	<b>₹</b>	¥	NA S	\$0.43	\$3.84	\$21.56	\$108.34	\$312.05	\$869.65	\$3,020.08	\$8.24		NA	<b>₹</b>	× ×	\$0.43	H	$\top$	T	\$111.56	П	٦.	+	7
\$18.03 \$18.03	\$37.55	\$37.55	\$37.55	\$37 55 NA	\$3.50	NA	¥	\$119.14	\$122.14	\$110.31	\$317.38	\$539.36	\$317.38	\$1,034.00	\$1,137.00	\$11.134.00	\$25.70	\$18.03	\$18.03	\$37.55	\$37.55	NA	\$3.50	X S	NA : 14	\$110.14	\$317.38	\$1,034.00	\$8,202.00	\$15.05	\$10.00	\$18.03	\$37.55	\$37.55	\$3.50	83.50 NA	₹	\$119.14	\$172.38	\$819.29	\$2,187.00	\$4.37		\$3.17	\$3.17	\$61.19	NA NA	\$3.50	₹ ₹	\$119.14	\$137.17	\$119.14	\$449.91	\$2.72	GA
X X	\$93.12	\$93.12	\$93.12	\$93 12	\$3.50	NA	\$19.99	¥ ₹	2 3	NA NA	\$516.89	\$844.21	\$516.89	\$1,713	\$1,497	\$16,017	\$138.02	¥	N <sub>A</sub>	\$93.12	\$93.12	NA	\$3.50	NA S	\$10.00	NA NA	\$516.89	\$1,713	\$12,344	\$84.88	×	<b>₹</b>	\$93.12	\$93.12	\$3.50	*3.50 NA	\$19.99	NA :	\$516.89	\$1,399	\$3,390	\$27.97		NA.	AN .12	\$93.12	NA NA	\$3.50	NA 3	P10 00	¥	\$516.89	\$946.23	\$12.62	KY
\$20.94 \$20.94	\$50.25	\$50.25	\$50.25	\$50.25 NA	\$3.50	NA	¥	\$99.46	\$102.46	\$102.16	\$304.90	\$532.13	\$304.90	\$1,147	\$1,451	\$14.950	\$128.59	\$20.94	\$20.94	\$50.25	\$50.25	NA.	\$3.50	N S	AN AN	\$102.16	\$304.90	\$1,147	\$11,517	\$74.44	\$20.54	\$20.94	\$50.25	\$50.25	WA 30.50	\$ N	¥	\$99.46	\$102.16	\$927.35	\$2,990	\$23.89		\$20.94	\$20.23	\$50.25	NA NA	\$3.50	<b>₹</b>	\$99.46	\$102.16	\$304.90	\$611.41	\$14.04	LA
\$28.59 \$28.59	\$68.62	\$68.62	\$68.62	\$68.63 NA	\$3.50	NA	¥	\$130.59	\$134.07	\$130.50	\$404.94	\$729.04	\$404.94	\$1,598.00	\$1,351.00	\$11,480	\$102.43	\$28.59	\$28.59	\$68.62	\$68.62	NA	\$3.50	X S	WA SOLUE	\$130.50	\$404.94	\$1,598	\$7,182	\$60 42	\$20.09	\$28.59	\$68.62	\$68.62	NA .50	\$3.50 NA	₹	\$130.59	\$404.94	\$1,283.00	\$1,892.00	\$18.35		\$27.08	\$27.08	\$94.50	NA	\$3.50	<b>₹</b> §	\$118.79	\$125.56	\$524.58	\$858.15	\$13.48	MS
\$29.76 \$29.76	\$69.34	\$69.34	\$69.34	\$69.34	\$3.50	NA	Z.	\$128.19	\$131.65	\$128.10	\$400.38	\$720.81	\$542.73	\$1,722	\$582.66	\$10.952	\$120.02	\$29.76	\$29.76	\$94.77	\$94.77	NA	\$3.50	NA S	NA S	\$128.19	\$509.93	\$1,381	\$7,676	\$48 55	\$23.70	\$29.76	\$94.77	\$94.77	93.50	\$3.50 NA	A	\$128.19	\$131.65	\$1,381	\$2,071	\$14.10		NA	NA O	\$55.00	NA NA	\$3.50	<b>₹</b>	\$ ₹	NA	\$436.36	\$624.86	\$6.29	NC
NA NA	\$54.26	\$54.26	\$54.26	\$54.26	\$3.50	NA	¥.	¥ §	Z Z	NA NA	\$410.02	\$635.04	\$410.02	\$1,131	\$1,561	\$967.58	\$45.92	NA	¥	\$54.26	\$54.26	NA	\$3.50	NA S	2 3	NA	\$410.02	\$1,131	\$11,132	\$30 50	5	NA	\$54.26	\$54.26	\$3.50	\$3.50 NA	A	NA	\$410.02 NA	\$915.64	\$2,802	\$9.75		NA	NA AN	\$54.26	NA NA	\$3.50	<b>₹</b>	\$ \$	NA	\$423.45	\$606.72	\$8.13	SC
₹	¥	<b>₹</b> §	₹ :	¥ ₹	\$3.50	¥	\$19.99	\$100.59	\$103.36	\$100.30	\$311.39	\$544.55	\$311.39	\$1,176	\$1,170	\$11.632	\$106.55	¥	¥	₹	¥	W	\$3.50	¥ S	\$1999	\$100.50	\$311.39	\$1,176	\$8,015	\$49.80	5	₹ ₹	¥	<b>₹</b>	₩.50	\$ ₹	\$19.99	\$100.59	\$103.36	\$950.10	\$2,124	\$13.45		¥	<b>₹</b>		₹	\$3.50	N S	\$100.59	\$103.36	\$525.25	\$858.26	\$6.88	¥

			H	F										Н			F				H			H	I		Ŧ			Ī	I				F						-	+	F				I		Ŧ	Н		Ŧ	_	
NRC -Disconnect-Addi	NRC - Per each four-fiber dark fiber arrangement - Add'l	NRC - Per each four-fiber dark fiber arrangement - 1st	month	NRC -DisconnectAdd'l	NRC -Disconnect1st	NRC - Per each four-fiber dark fiber arrangement - Add'i			NRC -DisconnectAdd"	NRC -Disconnect1st	NRC - Per each four-fiber dark fiber arrangement - Add'l	NRC - Per each four-fiber dark fiber arrangement - 1st	UNBUNDLED DARK HBER  Dark Fiber - Interoffice (four fiber strands) per route mile or fraction thereof, per month		Incremental Cost-Manual Svc. Order vs. Elect-Disconnect - Add'i	Incremental Cost Manual Svc Order vs Elect Disconnect - 1st	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	NRC - Electronic Svc Order, per LSR disconnect		NRC - Manual Svc Order, per LSR disconnect	NRC - Manual Svc Order per LSR	NRC - 1St	per VG card per month (DS0)	NRC - Add'l	NRC - 1st	per BRITE card per month	NRC - 1st	per OCU-DP(data) card per month (2.4-64kbs)	- Interface (COCI)	NRC -Add" - Disconnect	NRC - Add'l	NRC - 1st	per Channelized System (24 DS0) per month	DS1 Channelization (DS1 to DS0)	Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - Add'l	Incremental Cost-Manual Svc. Order vs. Elect-Disconnect - 1st	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -Add'l	Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	NRC - Electronic Svc Order, per LSR disconnect	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	NIC - Manual Sys Order par Sp	NRC - 1st	per Interface per month (COCI)	NRC -Add" - Disconnect	NRC -1st - Disconnect	NRC - Add'l	NRC - 1st	DS3 Channelization (DS3 to DS1)	UNBUNDLED CHANNELIZATION	Disconnect-Add	Disconnect-1st  NRC - OC48-Interface-Incremental Cost-Manual Svc. Order vs. Elec-	NRC - OC48-Interface-Incremental Cost-Manual Svc. Order vs. Elec-	
UDFL4	UDFL4	UDFL4	1 2	UDFC4	UDFC4	2 2 2 2 2 2 2	11500	5	UDF14	UDF14	UDF14	UDF14	11.5DF		SOMAN	NAMOS	SOMAN	SOMAN	SOMEC	SOMAN	SOMAN	10176	1D1VG	UC1CA	UC1CA	UC1CA	10100	1D1DD		3 3	NO.	MQ1	MQ1		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMA	UC1D1	UC1D1	MQ3	MQ3	MQ3	MO3	5		SOMAN	SOMAN	)	USOC
Z Z	\$563.09	\$1,739.00	\$70.82	A	NA	\$563.09	\$70.82		¥	A	\$563.09	\$1.739.00	\$25.80		\$0.9469	\$11.67	\$15.61	AN	\$3.50	¥ ;	98.89	\$12.05	\$0.8586	\$8.69	\$12.05	\$3.41	\$12.05	\$1.66		\$18.86	\$123.12	\$197.98	\$136.82		\$0.9469	\$11.67	\$7.39	\$15.61	NA	\$3.50	¥ ₹	WO.OS	\$12.05	\$17.22	\$52.03	\$71.76	\$188.51	\$265.87	3		\$19.03	\$19.03	2	AL
\$366.34		\$1,278.62	GEO 25	\$366.34	\$587.64	=	\$58.35	9	\$366.34	\$587.64			\$28.82		<b>₹</b>	Z X	5 ₹	\$0.43	\$2.77	\$3.84	\$21.56	\$13.16	\$1.45	\$9.43	\$13.16	\$3.83	\$13.16	\$2.20		\$18.14	\$125.18	\$182.14	\$153.60		¥	A	₹	NA	\$0.43	\$2.77	\$3.84	\$31 FG	\$13.16	\$14.40	\$58.98	\$61.64	\$188.00	\$356.40	***************************************		NA	×		FL
N N	\$562.39	\$1,737.00	677.63	NA	NA	\$562.39	\$54.63	9	NA	N.	\$562.39	\$1.737.00	\$24.96		\$0.00	\$10.70	\$14.75	NA	\$3.50	¥ :	98.00	\$12.02	\$1.17	\$8.66	\$12.02	\$3.71	\$12.02	\$1.86		\$1975	\$123.59	\$198.22	\$126.22		N	\$10.60	\$6.55	\$14.75	NA	\$3.50	₹ <del>3</del>	\$0.00	\$12.02	\$11.02	\$59.96	\$72.50	\$188.78	\$265.04	2		\$18.03	\$18.03	>	GA
X X	H	\$1,741.00	€ //O O7	NA	NA	\$563.75	\$49.07		NA	¥	\$563.75	\$1.741.00	\$31.95		X S	\$11.89	\$41.47	NA	\$3.50	N S	\$19.30	\$15.86	\$1.40	\$11.36	\$15.86	\$4.04	\$15.86	\$2.94		8 5	\$184.20	\$302.82	\$200.01		NA.	NA.	\$11.99	\$41.47	NA	\$3.50	NA S	\$10.00	\$15.86	\$8.52	¥	¥	\$303.33	\$425.32	*		N	NA.	,	КҮ
<b>₹</b>	\$5	-	C7 1/33	N	NA	+	\$64.72	9	NA	¥	+		\$32.28		NA S	\$12.43	\$19.74	NA	\$3.50	¥.	\$8.80	\$12.29	\$1.62	\$8.80	\$12.29	\$4.18	\$12.29	\$3.12		\$16.83	\$118.37	\$193.63	\$209.87		Ą	\$12.43	\$8.77	\$19.74	NA	\$3.50	<b>₹</b>	40.00	\$12.29	\$7.55	\$50.46	\$60.96	\$182.64	\$259.76	1		\$20.94	\$20.94	,	LA
<b>₹</b>	\$5	-	¢71 55	NA	NA	\$563.79	\$71.55	1	NA.	NA	\$563.79	\$1.741.00	\$33.93		X S	\$16.90	\$26.95	AN	\$3.50	<b>∑</b>	\$11.35	\$15.85	\$1.45	\$11.35	\$15.85	\$3.88	\$15.85	\$2.86		\$11 98	\$164.56	\$271.52	\$146.87		NA.	\$16.97	\$11.98	\$26.95	NA	\$3.50	<b>₹</b> 5	W	\$15.85	\$5.58	\$65.20	\$79.94	\$247.40	\$356.80	***************************************		\$28.59	\$28.59	7	MS
Z Z	_	\$1,738.00	¢55.47	N <sub>A</sub>	NA	+		1	N <sub>N</sub>	¥	\$562.82	\$1.738.00	\$29.86		\$1.48	\$18.33	\$28.13	NA	\$3.50	¥ :	\$11.28	\$15.76	\$1.64	\$11.28	\$15.76	\$3.76	\$15.76	\$2.88		\$21.14	\$161.43	\$267.19	\$177.72		\$1.48	\$18.26	\$13.33	\$28.13	NA	\$3.50	¥ §	VIA	\$15.76	\$4.61	\$63.32	\$77.90	\$243.76	\$351.95	2		\$29.76	\$29.76	)	NC
X X			_	NA		\$565.53	_	2	NA.	NA	\$565.53	_			<b>₹</b>	A6.92	\$25.59	NA	\$3.50	₹:	να. οα. οα	\$12.05	\$1.47	\$8.68	\$12.05	\$4.21	\$12.05	\$2.34	Н	+	+	-	\$147.51		NA.	NA.	\$8.92	\$25.59	NA	\$3.50	<b>₹</b> 5	40.00		+		-	-	\$321.54	+		N	N	,	SC
<b>₹</b>	\$564.08	\$1,742.00	\$60.08	₩		\$564.08			¥			\$1.742.00			\$1.46	\$14.21	\$25.66	¥	\$3.50	₹ .	\$19.99	\$12.61	\$1.25	\$9.03	\$12.61	\$3.33	\$12.61	\$2.46		\$15.81	\$119.99	\$197.21	\$165.21		\$1.46	\$14.21	\$10.46	\$21.71	NA	\$3.50	NA S	\$10.00	\$12.61	\$3.91	\$50.31	\$61.09	\$185.94	\$265.90	2		₹	₹	,	TN

Where the state Commiss herein, it is the intent of the same consistent with	Interim rates subject to true-up	NOTES:	
Where the state Commission has adopted rates for the rate elements contained herein, it is the intent of the parties to reflect such rates in this exhibit and to apply the same consistent with applicable FCC and Commission rules and orders.	re-up.		
			USOC
			P
			2
			GA
			?
			LA
			NS
			ઠ
			SC
			Į

SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
BUNDLED LOOP COMBINATIONS										
oundled Loop/Port Combinations (Note 4)										
RKET RATES (INCLUDING ALL VERTICAL FEATURES) (Note 1)										
THE TRATES (INCLUDING ALL VENTICAL FEATURES) (Note 1)										
								Greensboro-		
								Winston Salem-		
			0-1 1 51					Highpoint/		
			Orlando, Ft. Lauderdale,					Charlotte- Gastonia-Rock		
ity Zone 1 / Top 8 MSAs in BellSouth Region			Miami	Atlanta		New Orleans		Hill		Nashville
stomers with 4 or more DS0 Equivalent			mann	71001110		TOW OTTOWN				Tudo!!Viiid
rrently Combined (Note2)										
2-Wire Voice Grade Loop with 2-Wire Line Port (Res. and Bus.)										
2-Wire Voice Grade Line Port (Res.), per month										
2- wire voice unbundled port - residence	UEPRL	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled port with caller ID - residence	UEPRC	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled port outgoing only - residence	UEPRO	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAR	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPRM	NA	NA	NA	NA	NA	NA	NA	NA	NA
Outline training and a subtract of the delegation of the state of the	LIEDAG	B1A	N14	N.0	N14	044.00	N14	N/A	N/A	A14
2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAS	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAT	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled South Carolina extended local dialing parity port with Carolina extended local dialing parity port with	OLITAI	INA	INA	INA	INA	INA	INA	INA	11/0	INA
caller ID	UEPAU	NA	NA	NA	NA	NA	NA	NA	NA	NA
odio 15	021710			101	10.			10.		
2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	UEPAQ	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled Florida area calling with caller ID - residence	UEPAF	NA	\$14.00	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	UEPAG	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	UEPAH	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence										
(LW8)	UEPAJ	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	UEPAK	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
0 - 1 1 1 1 1	LIEDAI									04400
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	UEPAL	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2ininininin (TACCR)	UEPAM	NIA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	UEPAIVI	NA	INA	NA NA	INA	INA	INA	INA	INA	\$14.00
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	UEPAN	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2 wire voice disbanded Termessee Area caming port with camer 15 Testacrice (TWI 2A)	OLI 744	100	101	101	100	101	101	101	100	ψ14.00
2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)	UEPAO	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-wire voice unbundled res, low usage line port with Caller ID (LUM)	UEPAP	NA	\$14.00	\$14.00	NA	\$14.00	NA NA	\$14.00	NA	\$14.00
2-Wire Voice Grade Line Port (Bus.), per month										
2-wire voice unbundled port without Caller ID	UEPBL	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled port with unbundled port with Caller+E484 ID	UEPBC	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled outgoing only port	UEPBO	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
								1 7		
2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	UEPAW	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	UEPBM	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	UEPAX	NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice grade unburided Louisiana extended local dialing parity port with caller 1D	UEPAX	INA	INA	INA	INA	\$14.00	INA	INA	INA	INA
2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	UEPAY	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice grade unbundled South Carolina extended local dialing parity port with Carolina extended local dialing parity port with	OLIAI	INC	INC	INO	INA	INC	INA	INC	130	INA
caller ID	UEPAZ	NA	NA	NA	NA	NA	NA	NA	NA	NA
CONTO TO	OLI /VL	1973	1903	10/3	13/3	193	11/1	101	103	14/1
2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	UEPAV	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
	UEPB1	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-wire voice unbundled incoming only port with Caller ID		NA	NA	NA	NA	\$14.00	NA	NA	NA	NA
2-wire voice unbundled incoming only port with Caller ID 2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	UEPAA	INA								
2-wire voice unbundled incoming only port with Caller ID 2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC) 2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	UEPAA UEPAB	NA NA	NA NA	NA	NA	NA	NA	NA	NA	NA
2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)					NA	NA	NA	NA	NA NA	NA
2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)					NA NA	NA NA	NA NA	NA NA	NA NA	NA \$14.00

RIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-wire voice unbundled TN Bus 2-WAY Collierville and Memphis Local Calling Port (B2F)	UEPAE	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-Wire Voice Grade Loop (SL1) (Res. and Bus.)										
RC - 2-Wire Voice Grade Loop - Statewide	UEPLX	NA NA	NA 011.00	NA 010.00	NA	NA 011.05	NA NA	\$14.18	NA NA	NA 015.0
RC - 2-Wire Voice Grade Loop Zone 1	UEPLX	NA	\$14.90	\$10.80	NA	\$14.05	NA NA	NA	NA	\$15.9
RC - 2-Wire Voice Grade Loop Zone 2	UEPLX	NA	\$18.51	\$12.47	NA	\$24.14	NA NA	NA	NA	\$20.7
RC - 2-Wire Voice Grade Loop Zone 3	UEPLX	NA	\$24.25	\$19.83	NA	\$49.30	NA	NA	NA	\$27.1
Combination Rates								****		
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	Note 8	NA	NA .	NA	NA	NA	NA	\$28.18	NA	NA
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	Note 8	NA	\$28.90	\$24.80	NA	\$28.05	NA	NA	NA NA	\$29.92
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	Note 8	NA	\$32.51	\$26.47	NA	\$38.14	NA	NA	NA NA	\$34.79
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	Note 8	NA	\$38.25	\$33.83	NA	\$63.30	NA	NA	NA	\$41.1
Nonrecurring Charges										
2-Wire Voice Grade Line Port (Res. And Bus.)										
NRC - 2- wire voice grade unbundled port/loop combination - 1st, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.5
NRC - 2- wire voice grade unbundled port/loop combination - Add'l, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA NA	\$41.50	NA	\$41.5
NRC - 2- wire voice grade unbundled port/loop combination - 1st, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.5
NRC - 2- wire voice grade unbundled port/loop combination - Add'l, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.5
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent		NA	\$10.00	\$10.00	NA	\$10.00	NA	\$10.00	NA	\$10.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces	SOMEC	NA	\$2.75	\$3.50	NA	\$3.50	NA	\$3.50	NA	\$3.50
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st	SOMAN	NA	\$21.56	\$33.76	NA	\$31.92	NA	\$40.18	NA	\$30.8
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l	SOMAN	NA	\$21.56	\$7.86	NA	\$7.32	NA	\$9.45	NA	\$7.03
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Electronic		NA	TBD	TBD	NA	\$2.11	NA	\$1.42	NA	TBD
NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Manual Service Order		NA	TBD	TBD	NA	\$5.12	NA	\$10.27	NA	TBD
NRC - Electronic Service Order Disconnect		NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
NRC - Incremental Manual Service Order Disconnect		NA	\$3.84	\$20.00	NA	\$20.00	NA	\$20.00	NA	\$20.00
		NA	\$3.84	\$20.00	NA NA	\$20.00	NA NA	\$20.00	NA	\$20.00
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month		NA NA	\$3.84	\$20.00	NA NA	\$20.00	NA NA	\$20.00	NA NA	\$20.00
Wire Voice Grade Loop with 2-Wire Line Port PBX	UEPRD	NA NA	\$3.84	\$20.00	NA NA	\$20.00	NA NA	\$20.00	NA NA	
Wire Voice Grade Loop with 2-Wire Line Port PBX 2-Wire Analog Line Port (PBX), per month	UEPRD UEPPC									\$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence		NA	\$14.00	\$14.00	NA	\$14.00	NA .	\$14.00	NA NA	\$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS	UEPPC	NA NA	\$14.00 \$14.00	\$14.00 \$14.00	NA NA	\$14.00 \$14.00	NA NA	\$14.00 \$14.00	NA NA	\$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPPC UEPPO UEPP1	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00	\$14.00 \$14.00 \$14.00 \$14.00	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00	NA NA NA NA	\$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT	UEPPC UEPPO	NA NA NA	\$14.00 \$14.00 \$14.00	\$14.00 \$14.00 \$14.00	NA NA NA	\$14.00 \$14.00 \$14.00	NA NA NA	\$14.00 \$14.00 \$14.00	NA NA NA	\$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING	UEPPC UEPPO UEPP1 UEPA2	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA	\$14.0 \$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	UEPPC UEPPO UEPP1 UEPA2 UEPL2	NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00	NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA NA	\$14.0 \$14.0 \$14.0 NA
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	UEPPC UEPPO UEPP1 UEPA2	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA	\$14.0 \$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT	UEPPC UEPPO UEPP1 UEPA2 UEPL2	NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00	NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA	NA NA NA NA NA	\$14.0 \$14.0 \$14.0 \$14.0 NA NA NA
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT	UEPPC UEPPO UEPP1  UEPA2  UEPL2 UEPLD  UEPT2  UEPTO	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA NA NA	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA NA	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA	NA NA NA NA NA NA NA NA NA	\$14.0 \$14.0 \$14.0 NA NA \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT	UEPPC UEPPO UEPP1 UEPA2 UEPL2 UEPLD UEPT2 UEPT0 UEPTO UEPXA	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA NA NA	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA NA NA \$14.00	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA NA NA	NA	\$14.0 \$14.0 \$14.0 \$14.0 NA NA \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT	UEPPC UEPPO UEPP1  UEPA2  UEPL2 UEPLD  UEPT2  UEPTO	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA NA NA	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA NA	NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA	NA NA NA NA NA NA NA NA NA	\$14.0 \$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT	UEPPC UEPPO UEPP1 UEPA2 UEPL2 UEPLD UEPT2 UEPT0 UEPTO UEPXA	NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA	NA	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA NA NA \$14.00	NA NA NA NA NA NA NA	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA	NA	\$14.0 \$14.0 \$14.0 \$14.0 NA NA \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT	UEPPC UEPPO UEPP1 UEPA2 UEPL2 UEPLD UEPT2 UEPTO UEPXA UEPXB	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA S14.00 \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA S14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA NA \$14.00 S14.00 S14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA S14.00 \$14.00	NA	\$14.0 \$14.0 \$14.0 \$14.0 NA NA \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED 1NCOMING PBX TRUNK - BUSINESS  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY COMBINATION PBX USAGE PORT 2-WIRE VOICE UNBUNDLED PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS 2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS	UEPPC UEPPO UEPP1  UEPA2  UEPL2  UEPLD  UEPT2  UEPTO UEPXA  UEPXB  UEPXB	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA S14.00 \$14.00 \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA S14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA NA NA \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA S14.00 \$14.00	NA N	\$14.0 \$14.0 \$14.0 \$14.0 NA NA \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS 2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS 2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT 2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS WITCHBOARD PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD DDC CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD DDC CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD DDC CAPABLE PORT 2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING PORT	UEPPC UEPPO UEPP1  UEPA2  UEPL2 UEPLD  UEPT2  UEPTO UEPXA UEPXB UEPXC UEPXC UEPXD  UEPXE	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY COMBINATION PBX USAGE PORT 2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS 2-WIRE VOICE UNBUNDLED PBX TOLD TERMINALS PORT 2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT 2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT	UEPPC UEPPO UEPP1 UEPA2 UEPL2 UEPLD UEPT2 UEPT0 UEPXA UEPXA UEPXB UEPXC UEPXD UEPXB UEPXC UEPXD	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00	NA N	\$14.0 \$14.0 \$14.0 NA NA \$14.0 \$14.0 \$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS  2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED PBX LD TOMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD DD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED PBX LD DD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED PBX LD DD TERMINAL PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL HOTEL PORTS 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX KENTUCKY ROOM AREA CALLING PORT WITHOUT LUD 2-WIRE VOICE UNBUNDLED PBX KENTUCKY LUD AREA CALLING PORT	UEPPC UEPPO UEPP1  UEPA2  UEPL2  UEPLD  UEPT2  UEPTO UEPXA  UEPXB  UEPXB  UEPXB  UEPXC  UEPXC  UEPXC  UEPXC  UEPXC	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wire Analog Line Port (PBX), per month  2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - Residence LINE SIDE UNBUNDLED COMBINATION 2-WAY PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED OUTWARD PBX TRUNK - BUSINESS LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS 2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT 2-WIRE VOICE UNBUNDLED 1-WAY COMBINATION PBX USAGE PORT 2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOTEL PORTS 2-WIRE VOICE UNBUNDLED PBX TOLD TERMINALS PORT 2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT 2-WIRE VOICE UNBUNDLED PBX LD DDD TERMINALS PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT 2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IDD CAPABLE PORT	UEPPC UEPPO UEPP1 UEPA2 UEPL2 UEPLD UEPT2 UEPT0 UEPXA UEPXA UEPXB UEPXC UEPXD UEPXB UEPXC UEPXD	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA \$14.00 NA \$14.00 \$14.00 \$14.00 \$14.00 \$14.00	NA N	\$14.00 \$14.00 \$14.00 \$14.00 NA NA \$14.00 NA \$14.00 \$14.00 \$14.00	NA N	\$14.0 \$14.0 \$14.0 NA NA \$14.0 \$14.0 \$14.0 \$14.0 \$14.0 \$14.0

CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT	UEPXL	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM CALLING PORT	UEPXM	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT	UEPXO	NA	\$14.00	\$14.00	NA	\$14.00	NA	\$14.00	NA	\$14.00
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT	UEPXP	NA NA	NA	NA	NA NA	\$14.00	NA NA	NA	NA NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT	UEPXQ	NA	NA	NA NA	NA	NA	NA	NA	NA.	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL CALLING PORT	UEPXR	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT	UEPXS	NA NA	\$14.00	\$14.00	NA NA	\$14.00	NA NA	\$14.00	NA NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS CALLING PORT	UEPXT	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT	UEPXU	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING PORT	UEPXV	NA	NA	NA	NA	NA	NA	NA	NA	\$14.00
LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	LNPCP									
2-Wire Voice Grade Loop (SL1)										
RC - 2- Wire Voice Grade Loop (SL1)	UEPLX	NA	NA	NA	NA	NA	NA	\$14.18	NA	NA
RC - 2- Wire Voice Grade Loop - Zone 1	UEPLX	NA NA	\$14.90	\$10.80	NA	\$14.05	NA NA	NA NA	NA	\$15.92
RC - 2- Wire Voice Grade Loop - Zone 2	UEPLX	NA	\$18.51	\$12.47	NA	\$24.14	NA	NA	NA	\$20.79
RC - 2- Wire Voice Grade Loop - Zone 3	UEPLX	NA	\$24.25	\$19.83	NA	\$49.30	NA	NA	NA	\$27.18
RC - 2- Wire Voice Grade Loop - Zone 4	UEPLX	NA	NA	NA	NA	NA	NA	NA	NA	NA
Combination Rates								****		
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	Note 8	NA	NA Too oo	NA COA CO	NA NA	NA Too.or	NA NA	\$28.18	NA NA	NA \$29.92
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6) RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)	Note 8 Note 8	NA NA	\$28.90 \$32.51	\$24.80 \$26.47	NA NA	\$28.05 \$38.14	NA NA	NA NA	NA NA	\$29.92
RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 2 (Note 6)  RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note 6)	Note 8	NA NA	\$38.25	\$33.83	NA NA	\$63.30	NA NA	NA NA	NA NA	\$34.79 \$41.18
Nonrecurring Charges	Note o	INA	ψ30.23	ψ33.03	INA	ψ03.30	14/4	19/5	INA	φ41.10
NRC - 2- wire voice grade unbundled port/loop combination - 1st, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
NRC - 2- wire voice grade unbundled port/loop combination - Add'l, with change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
NRC - 2- wire voice grade unbundled port/loop combination - 1st, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
NRC - 2- wire voice grade unbundled port/loop combination - Add'l, no change		NA	\$41.50	\$41.50	NA	\$41.50	NA	\$41.50	NA	\$41.50
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent		NA	\$10.00	\$10.00	\$10.00	\$10.00	NA	\$10.00	NA	\$10.00
NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,	SOMEC	NA	\$2.75	<b>6</b> 2.50	NA	£2.50	NA	<b>6</b> 2.50	NA.	<b>6</b> 2.50
per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st	SOMAN	NA NA	\$2.75	\$3.50 \$33.67	NA NA	\$3.50 \$31.92	NA NA	\$3.50 \$40.18	NA NA	\$3.50 \$30.89
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual										
Svc.Order vs. Electronic - Add'l  NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update -	SOMAN	NA					NA		NA	\$7.03
Electronic  NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update -		NA	TBD	TBD	NA	\$2.11	NA	\$1.42	NA	TBD
Manual Service Order		NA	TBD	TBD	NA	\$5.12	NA	\$10.27	NA	TBD
NRU - Electronic Service Order Disconnect		NA	\$0.42	NA	NA	NA NA	NA	NA	NA	NA
NRC - Incremental Manual Service Order Disconnect		NA	\$3.84	\$20.00	NA	\$20.00	NA	\$20.00	NA	\$20.00
Electronic NRC - 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Manual Service Order NRC - Electronic Service Order Disconnect	SOMAN	NA NA	TBD \$0.42	TBD NA	NA NA	\$5.12 NA	NA NA		\$10.27 NA	\$1.42 NA \$10.27 NA NA NA
BASED RATES (Notes 2 & 3)										
I BASED RATES (Notes 2 & 3)				i .	1	I	I		1	
rently Combined										
rently CombinedWire Voice Grade Loop with 2-Wire Line Port										
rently Combined	UEPRL	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54

NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 4 (Note 6)  Nonrecurring Charges	- 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3	(Note	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port. Zone 1 (Note 6)	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port. Statewide	Combination Brico	RC - 2- Wire Voice Grade Loop - Zone 3	RC - 2- Wire Voice Grade Loop - Zone 2	- 2- Wire Voice Grade Loop	- 2- Wile voice Grade Loop		(B2F)	2-wire voice unbundled TN Bus 2-Way Area Calling Port Standard Option (TACC2)	2 wife voice unbundled TN Bus 2 Way Area Colling Bart Crandard Option (TACCS)	2-wire voice unbundled TN Bus 2-Way Area Calling Port Economy Option (TACC1)	2-wire voice unbundled SC Bus Area Calling Port with Caller ID (LMB)	2-wire voice unbundled LA Bus Area Calling Port with Caller ID (BUC)	2-wire voice unbundled incoming only port with Caller ID	2-wire voice grade unbundled Tennessee extended local dialing parity port with caller ID	2-wire voice grade unbundled South Carolina extended local dialing panty port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	2-wire voice unbundled outgoing only port	2-wire voice unbundled port with unbundled port with Caller+E484 ID	2-wire voice unbundled port without Caller ID	2-Wire Voice Grade Line Port (Bus.), per month	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (2MR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (1MF2X)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACSR)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (TACER)	2-wire voice unbundled Tennessee Area Calling port with Caller ID - residence (F2R)	2-wire voice unbundled South Carolina Area Calling port with Caller ID - residence (LW8)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (AC7)	2-wire voice unbundled Louisiana Area Plus with caller ID - residence (RUL)	2-wire voice unbundled Florida area calling with caller ID - residence	Swire voice grade unbundled Tennessee extended local diating parity port with caller ID	2-wire voice grade unbundled South Carolina extended local dialing parity port with caller ID	2-wire voice grade unbundled Mississippi extended local dialing parity port with caller ID	2-wire voice grade unbundled Louisiana extended local dialing parity port with caller ID	2-wire voice grade unbundled Kentucky extended local dialing parity port with caller ID	2-wire voice grade unbundled Alabama extended local dialing parity port with caller ID	2-wire volice unbundled port outgoing only - residence
USAC2	Note 8	Note 8	Note 8	Note 8	Note 8	OEPLX	OFFPLX	OFFPLX	S F X	, , ,	<u> </u>	UEPAE	CETAD	<b>n</b> 0 2	UEPAC	UEPAB	UEPAA	UEPB1	UEPAV	UEPAZ	UEPAY	UEPAX	UEPBM	UEPAW	OEPBO	UEPBC	UEPBL	CEPAP	UEPAO	UEPAN	UEPAM	UEPAL	UEPAK	UEPAJ	UEPAH	UEPAG	UEPAF	I IF PAO	UEPAU	UEPAT	UEPAS	UEPRM	UEPAR	USOC UEPRO
\$2.80 \$0.41	¥	\$44.44	\$25.51	\$16.55	¥	Ā	\$42.24	\$23.31	\$14.35	3	N	NA	NA	S	NA	Ā	X.	\$2.20	NA	NA	NA	NA	N <sub>A</sub>	\$2.20	\$2.20	\$2.20	\$2.20	\$2.20	N N	NA	N <sub>A</sub>	NA	NA	NA	NA	NA	¥ ₹	NA	N	NA	NA	NA	\$2.20	<b>AL</b> \$2.20
\$0.1964 \$0.1964	NA	\$25.60	\$19.86	\$16.25	NA	NA.	\$24.25	\$18.51	\$14.90	5	NIA	NA	N	Z	NA	X	Ā	\$1.35	NA	NA	NA	NA	NA	NA	\$1.35	\$1.35	\$1.35	\$1.35	NA	NA	NA	NA	NA	NA	N <sub>A</sub>	NA	\$1.35	NA	NA	NA	NA	NA	NA	FL \$1.35
\$2.01 \$0.3108	NA	\$21.62	\$14.26	\$12.59	NA.	¥	\$19.83	\$12.47	\$10.80	5	<u> </u>	NA	N	2	NA	¥	¥	\$1.79	NA	NA	NA.	NA	NA	N <sub>A</sub>	\$1.79	\$1.79	\$1.79	\$1.79	NA NA	NA	N <sub>A</sub>	NA	NA	NA	NA	NA	X S	NA	N <sub>A</sub>	NA	NA	NA	NA	<b>GA</b> \$1.79
\$10.00 \$10.00	NA	\$50.39	\$30.29	\$17.40	NA.	×	\$47.78	\$27.68	\$14.79	3	NIA	NA	¥	N	NA	Ā	¥	\$2.61	NA	NA	NA	NA	\$2.61	NA	\$2.61	\$2.61	\$2.61	\$2.61	NA	NA	NA	NA	NA	NA	NA	NA	¥ ₹	NA	NA	NA	¥	\$2.61	NA	\$2.61
\$3.80 \$0.29	¥	\$51.85	\$26.69	\$16.60	¥	¥	\$49.30	\$24.14	\$14.05	3	25	N <sub>A</sub>	NA	5	N <sub>A</sub>	Ā	\$2.55	\$2.55	NA	¥	NA	\$2.55	×	NA	\$2.55	\$2.55	\$2.55	\$2.55	NA	NA.	NA	NA	NA	¥	\$2.55	\$2.55	¥ ₹	N P	NA	NA	\$2.55	NA	×	<b>LA</b> \$2.55
\$5.20 \$0.41	\$38.59	\$29.75	\$21.45	\$16.71	N N	\$36.47	\$27.63	\$19.33	\$14.59	2	20	Ą	Ā	5	NA.	Ā	NA.	\$2.12	Ą	¥	\$2.12	N <sub>A</sub>	N <sub>A</sub>	¥	\$2.12	\$2.12	\$2.12	\$2.12	Ž X	N <sub>A</sub>	¥	¥	¥	NA	NA	NA	¥ ₹	Z Þ	¥	\$2.12	¥	N <sub>A</sub>	N <sub>A</sub>	<b>MS</b> \$2.12
\$2.77 \$0.40	NA	NA	<b>⊼</b>	NA S	\$16.46	X	X X	X X	\$ \$	÷ + - 0	61440	NA.	NA	2	¥	×	×	\$2.28	NA.	NA	¥	NA	NA	NA	\$2.28	\$2.28	\$2.28	\$2.28	NA	NA	NA	NA	NA	NA	NA	NA	N S	Z	NA	NA	NA	NA	NA	\$2.28
\$1.59 \$0.40	¥	\$37.68	\$29.35	\$20.71	NA.	¥	\$33.99	\$25.66	\$17.02	2425	NIA	NA	NA	5	NA	\$3.69	NA NA	\$3.69	A	\$3.69	N <sub>A</sub>	NA	NA	NA	\$3.69	\$3.69	\$3.69	\$3.69	NA	NA	N <sub>A</sub>	NA.	NA	\$3.69	¥	NA	N S	N D	\$3.69	NA	NA NA	NA	NA	\$3.69
\$1.03 \$0.2886	N <sub>A</sub>	\$26.31	\$18.96	\$17.02	X.	¥	\$21.77	\$14.42	\$12.48	3		\$4.54	\$4.54	e 2	\$4.54	S	¥	\$4.54	\$4.54	NA	¥	NA	N <sub>A</sub>	NA	\$4.54	\$4.54	\$4.54	\$4.54	\$4.54	\$4.54	\$4.54	\$4.54	\$4.54	NA	NA	NA	NA ST	\$4 54	N	NA	NA	NA	NA	<b>TN</b> \$4.54

DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change	USACC	\$2.80	\$0.1964	\$2.01	\$10.00	\$3.80	\$5.20	\$2.77	\$1.59	\$1.03
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change	USACC	\$0.41	\$0.1964	\$0.3108	\$10.00	\$0.29	\$0.41	\$0.40	\$0.40	\$0.2886
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USAS2	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
	per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual										
	Svc.Order vs. Electronic - 1st	SOMAN	\$40.71	\$21.56	\$33.67	\$19.99	\$31.92	\$43.52	\$40.18	\$43.19	\$30.89
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l	SOMAN	\$9.58	\$21.56	\$7.88	\$19.99	\$7.32	\$9.99	\$9.45	\$9.91	\$7.03
$\vdash$	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update -	SOIVIAIN	φ9.56	\$21.50	\$1.00	\$19.99	\$1.32	\$9.99	\$9.40	φ9.91	\$1.03
	Electronic		\$1.44	TBD	TBD	TBD	\$2.11	\$2.87	\$1.42	\$0.71	\$0.76
	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update -		·				·		·	1	
	Manual Service Order		\$8.25	TBD	TBD	TBD	\$5.12	\$6.88	\$10.27	\$8.91	\$7.97
	NRC - Electronic Service Order Disconnect		NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
	NRC - Incremental Manual Service Order Disconnect		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
	NRCs for New (not Currently Combined) as ordered in Georgia:										
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPRL	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPRL	NA NA	NA NA	\$15.25	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPRC	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPRC UEPRO	NA NA	NA NA	\$15.25 \$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
$\vdash$	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st  NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPRO	NA NA	NA NA	\$22.14 \$15.25	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
++	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add 1	UEPAP	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPAP	NA NA	NA NA	\$15.25	NA NA	NA NA	NA NA	NA NA	NA NA	NA.
HT	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPBL	NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPBL	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
$\vdash$	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add I	UEPBC	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Fort - New - 1st	UEPBC	NA NA	NA NA	\$15.25	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPBO	NA NA	NA NA	\$22.14	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
H	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPBO	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	UEPB1	NA	NA	\$22.14	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	UEPB1	NA	NA	\$15.25	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	USAS2	NA	NA	\$10.00	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - 1st		NA	NA	\$8.45	NA	NA	NA	NA	NA	NA
$\vdash$	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - Add'l		NA	NA	\$3.91	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic,										
	per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	NA	NA	\$3.50	NA	NA	NA	NA	NA	NA
H	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.				40.00					1	
	Electronic - New - 1st		NA	NA	\$37.06	NA	NA	NA	NA	NA	NA
	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.										
ш	Electronic - New - Add'l		NA	NA	\$8.19	NA	NA	NA	NA	NA	NA
	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update -							l			
++	Electronic		NA	NA	TBD	NA	NA	NA	NA	NA	NA
	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Manual Service Order		NA	NA	TBD	NA	NA	NA	NA	NA	NA
$\vdash$	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.		INA	INA	IBD	INA	INA	INA	INA	INA	INA
	Electronic - New - Disconnect		NA	NA	\$11.17	NA	NA	NA	NA	NA	NA
						1					-
	2- Wire Voice Grade Loop - Bus Only with 2 -Wire DID Trunk Port										
	, , , , , , , , , , , , , , , , , , , ,										
	2 - Wire Line Port - DID Trunk Port por month	UEPD1	TBD	\$9.36	\$11.35	\$10.84	\$13.12	\$14.63	\$12.12	TBD	\$8.78
++	2 - Wire Line Port - DID Trunk Port, per month	UEPUI	עמו	φυ.30	φ11.35	φ10.84	φ13.1Z	φ14.03	φ12.12	100	φο./δ
Ш	2-Wire Voice Grade Loop (SL2)										
$\vdash \vdash$	RC - 2- Wire Voice Grade Loop - Statewide	UECD1	NA	NA .	NA	NA	NA	NA	\$11.76	NA	NA
++	RC - 2- Wire Voice Grade Loop - Zone 1	UECD1	\$17.95	\$18.48	\$16.84	\$17.78	\$17.65	\$18.35	NA NA	\$21.57	\$9.60
++	RC - 2- Wire Voice Grade Loop - Zone 2	UECD1	\$29.16	\$22.43	\$19.45	\$23.96	\$30.32	\$24.33	NA NA	\$32.53	\$11.09
++	RC - 2- Wire Voice Grade Loop - Zone 3 RC - 2- Wire Voice Grade Loop - Zone 4	UECD1 UECD1	\$52.84 NA	\$27.87 NA	\$30.92 NA	\$34.96 NA	\$61.93 NA	\$34.77 \$45.88	NA NA	\$43.08 NA	\$16.74 NA
++	RC - 2- Wire Voice Grade Loop - Zone 4  Combination Rates	UECDI	NΑ	INA	NA	INA	NA	\$45.88	INA	INA	NA
++	RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$23.79	NA	NA
ш	110 2 1110 10100 Glade Loop With 2-14116 DID Fort, Statewide	INOIGO	I INC	IN.	13/7	INC	INA	I N/A	Ψ23.13	11/7	14/5

Version 3Q00:11/07/00
LOOP-PORT COMBOS

RIPTION		USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone		Note 8	28.72	\$27.84	\$28.19	28.72	\$30.77	28.72	NA	28.72	\$18.38
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone		Note 8	34.91	\$31.79	\$30.80	34.91	\$43.44	34.91	NA	34.91	\$19.87
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone	3 (Note 6)	Note 8	45.9	\$37.23	\$42.27	45.9	\$75.05	45.9	NA	45.9	\$25.52
RC - 2-Wire Voice Grade Loop with 2-Wire DID Port, Zone4	4 (Note 6)	Note 8	NA	NA	NA	NA	NA	TBD	NA	NA	NA
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Co	onversion - Switch As Is - 1st										
port		USAC1	\$14.62	\$14.62	\$166.08	TBD	\$14.60	\$14.60	\$13.26	\$14.62	\$8.76
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Co	onversion - Switch As Is										
Each Addl Port		USAC1	\$3.73	\$3.73	\$140.01	TBD	\$3.72	\$3.72	\$8.39	\$3.73	\$5.75
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Co	onversion with changes - 1st										
port		USA1C	\$14.62	\$14.62	\$166.08	TBD	\$14.60	\$14.60	\$13.26	\$14.62	\$8.76
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Co	onversion with changes -										
Each Addl port		USA1C	\$3.73	\$3.73	\$140.01	TBD	\$3.72	\$3.72	\$8.39	\$3.73	\$5.75
NRC - 2-Wire DID Subsequent Activity - Per Svc Order - Ad	dd Trunks Per Trunk	USAS1	\$53.57	\$53.57	NA NA	NA.	\$53.50	\$53.50	NA NA	\$53.57	NA NA
			400.01	400.0.			400.00	***************************************		400.0.	
NRC - 2-Wire Voice Grade Loop/Line Port Combination - O	SS LSR Charge Electronic										
per LSR received from the CLEC by one of the OSS interact		SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Inc		OOMEO	ψ0.00	Ψ2.70	ψ0.00	ψ0.00	ψ0.00	ψ0.00	ψ0.00	ψ0.00	ψ0.00
Service Order - 1st	remental Cost- Manual	SOMAN	\$19.99	\$21.56	\$37.88	\$19.99	\$19.99	\$19.99	\$53.89	\$19.99	\$41.43
		SOIVIAIN	\$19.99	\$21.00	φ31.00	\$19.99	\$19.99	\$19.99	φυσ.09	\$19.99	Ş41.4·
NRC- 2- Wire Voice Grade Loop with 2- Wire DID Port - Inc	remental Cost- Manual	SOMAN	\$19.99	\$21.56	\$16.84	\$19.99	\$19.99	\$19.99	\$11.34	\$19.99	\$9.80
Service Order - Addl		SUMAN									
NRC - Electronic Service Order Disconnect	<del></del>		\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42
NRC - Incremental Manual Service Order Disconnect			\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
Telephone Number/Trunk Group Establishment										1	
DID Trunk Termination ( one required per port)		NDT	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
DID Numbers, Establish Trunk Group and Provide First Grou	up of 20 DID Numbers (FL,									1	
GA, NC, & SC only)		NDZ	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
DID Numbers, Establish Trunk Group and Provide First Group	up of 20 DID Numbers (AL,										
KY, LA, MS, & TN). In addition, Provides Additional DID Nur	mbers for each Group of 20										
DID Numbers (Valid in All States)		ND4	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
DID Numbers, non-consective		ND5	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
2-Wire ISDN Digital Grade Loop with 2-wire ISD	N Digital Port										
2-wire ISDN Digital Port per month	N Digital 1 of	UEPPB	\$16.42	\$8.51	\$13.47	\$12.99	\$11.42	\$51.91	\$24.37	\$33.74	\$18.21
2-Wire ISDN Digital Grade Loop		OLITE	ψ10.42	ψ0.01	ψ10.47	Ψ12.55	Ψ11.42	ψ01.01	Ψ24.01	ψ00.7 Ψ	Ψ10.2
RC - 2-Wire ISDN Digital Grade Loop - Statewide		USL2X	NA	NA	NA	NA	NA	NA	\$19.08	NA	NA
RC - 2-Wire ISDN Digital Grade Loop - Statewide		USL2X	\$23.23	\$22.48	\$21.89	\$22.41	\$28.87	\$21.86	NA	\$26.68	\$16.2
RC - 2-Wire ISDN Digital Grade Loop - Zone 1	+	USL2X	\$37.74	\$27.90	\$25.27	\$31.10	\$37.63	\$28.97	NA NA	\$40.24	\$18.7
		USL2X	\$68.38	\$30.78	\$40.17	\$42.36	\$48.42	\$41.40	NA NA	\$53.29	\$28.2
RC - 2-Wire ISDN Digital Grade Loop - Zone 3											
RC - 2-Wire ISDN Digital Grade Loop - Zone 4		USL2X	NA	NA	NA	NA	NA	\$54.64	NA	NA	NA
Combination Rates											
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital	ital Port - Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$43.45	NA	NA
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital						\$34.40	\$34.84				
	tal Port - Zone 1	Note 8	\$39.65	\$30.99				S73 77	I NA	\$60.42	
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital		Note 8	\$39.65 \$54.16	\$30.99 \$36.41	\$35.36 \$38.74			\$73.77 \$80.78	NA NA	\$60.42 \$73.98	\$34.4
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Grade	ital Port - Zone 2	Note 8	\$54.16	\$36.41	\$38.74	\$44.10	\$43.20	\$80.78	NA	\$73.98	\$34.4 \$36.9
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital	ital Port - Zone 2 ital Port - Zone 3	Note 8 Note 8	\$54.16 \$84.80	\$36.41 \$39.30	\$38.74 \$53.64	\$44.10 \$55.35	\$43.20 \$59.69	\$80.78 \$93.31	NA NA	\$73.98 \$87.03	\$34.4 \$36.9 \$46.4
	ital Port - Zone 2 ital Port - Zone 3	Note 8	\$54.16	\$36.41	\$38.74	\$44.10	\$43.20	\$80.78	NA	\$73.98	\$34.4 \$36.9
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital	ital Port - Zone 2 ital Port - Zone 3	Note 8 Note 8	\$54.16 \$84.80	\$36.41 \$39.30	\$38.74 \$53.64	\$44.10 \$55.35	\$43.20 \$59.69	\$80.78 \$93.31	NA NA	\$73.98 \$87.03	\$34.4 \$36.9 \$46.4
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital	ital Port - Zone 2 ital Port - Zone 3 ital Port - Zone 4	Note 8 Note 8	\$54.16 \$84.80	\$36.41 \$39.30	\$38.74 \$53.64	\$44.10 \$55.35	\$43.20 \$59.69	\$80.78 \$93.31	NA NA	\$73.98 \$87.03	\$34.4 \$36.9 \$46.4 NA
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digi RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digi	ital Port - Zone 2 ital Port - Zone 3 ital Port - Zone 4	Note 8 Note 8 Note 8	\$54.16 \$84.80 NA	\$36.41 \$39.30 NA	\$38.74 \$53.64 NA	\$44.10 \$55.35 NA	\$43.20 \$59.69 NA	\$80.78 \$93.31 \$106.55	NA NA NA	\$73.98 \$87.03 NA	\$34.4 \$36.9 \$46.4 NA
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Grade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital	ital Port - Zone 2 ital Port - Zone 3 ital Port - Zone 4  Port - 1st conversion	Note 8 Note 8 Note 8	\$54.16 \$84.80 NA \$79.12	\$36.41 \$39.30 NA \$86.79	\$38.74 \$53.64 NA \$239.95	\$44.10 \$55.35 NA \$79.12	\$43.20 \$59.69 NA \$79.01	\$80.78 \$93.31 \$106.55 \$79.12	NA NA NA \$174.35	\$73.98 \$87.03 NA \$79.12	\$34.4 \$36.9 \$46.4 NA \$117.2
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Grade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital	ital Port - Zone 2 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Add'l conversion	Note 8 Note 8 Note 8	\$54.16 \$84.80 NA	\$36.41 \$39.30 NA	\$38.74 \$53.64 NA	\$44.10 \$55.35 NA	\$43.20 \$59.69 NA	\$80.78 \$93.31 \$106.55	NA NA NA	\$73.98 \$87.03 NA	\$34.4 \$36.9 \$46.4 NA \$117.2
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digit RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digit NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital	ital Port - Zone 2 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Add'l conversion	Note 8 Note 8 Note 8 USACB	\$54.16 \$84.80 NA \$79.12	\$36.41 \$39.30 NA \$86.79	\$38.74 \$53.64 NA \$239.95 \$156.92	\$44.10 \$55.35 NA \$79.12 \$54.04	\$43.20 \$59.69 NA \$79.01 \$53.97	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04	NA NA NA \$174.35	\$73.98 \$87.03 NA \$79.12	\$34.4 \$36.9 \$46.4 NA \$117.2
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Grade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital	ital Port - Zone 2 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Add'l conversion	Note 8 Note 8 Note 8	\$54.16 \$84.80 NA \$79.12	\$36.41 \$39.30 NA \$86.79	\$38.74 \$53.64 NA \$239.95	\$44.10 \$55.35 NA \$79.12	\$43.20 \$59.69 NA \$79.01	\$80.78 \$93.31 \$106.55 \$79.12	NA NA NA \$174.35	\$73.98 \$87.03 NA \$79.12	\$34.4 \$36.9 \$46.4 NA \$117.2
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Add1 conversion  Port - Non Feature	Note 8 Note 8 Note 8 USACB	\$54.16 \$84.80 NA \$79.12	\$36.41 \$39.30 NA \$86.79	\$38.74 \$53.64 NA \$239.95 \$156.92	\$44.10 \$55.35 NA \$79.12 \$54.04	\$43.20 \$59.69 NA \$79.01 \$53.97	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04	NA NA NA \$174.35	\$73.98 \$87.03 NA \$79.12	\$34.4 \$36.9 \$46.4 NA \$117.2
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - O:	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Add'l conversion  Port - Non Feature  SS LSR Charge, Electronic,	Note 8 Note 8 Note 8 Note 8 USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50	\$43.20 \$59.69 NA \$79.01 \$53.97	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50	NA NA NA \$174.35 \$174.35	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50	\$34.4 \$36.9 \$46.4 NA \$117.2 \$117.2
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digit RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Other LSR received from the CLEC by one of the OSS interactions.	ital Port - Zone 2 tal Port - Zone 3 tal Port - Zone 4 Port - 1st conversion Port - Addl conversion Port - Non Feature SS LSR Charge, Electronic, titve interfaces (Note 7)	Note 8 Note 8 Note 8 USACB	\$54.16 \$84.80 NA \$79.12	\$36.41 \$39.30 NA \$86.79	\$38.74 \$53.64 NA \$239.95 \$156.92	\$44.10 \$55.35 NA \$79.12 \$54.04	\$43.20 \$59.69 NA \$79.01 \$53.97	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04	NA NA NA \$174.35	\$73.98 \$87.03 NA \$79.12	\$34.4 \$36.9: \$46.4! NA \$117.2 \$117.2
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital C - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Over 150 per LSR received from the CLEC by one of the OSS interaction NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Gr	ital Port - Zone 2 tal Port - Zone 3 tal Port - Zone 4 Port - 1st conversion Port - Addl conversion Port - Non Feature SS LSR Charge, Electronic, titve interfaces (Note 7)	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50	NA NA NA \$174.35 \$174.35 \$53.50	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50	\$34.4 \$36.9: \$46.4! NA \$117.2 \$117.2 \$212.8
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Oper LSR received from the CLEC by one of the OSS interactive Carbon State Combination - Other LSR received from the CLEC by one of the OSS interactive Carbon State St	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Addi conversion  Port - Non Feature  SS LSR Charge, Electronic, tive interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50	\$43.20 \$59.69 NA \$79.01 \$53.97	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50	NA NA NA \$174.35 \$174.35	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50	\$34.4 \$36.9: \$46.4! NA \$117.2 \$117.2 \$212.8
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Of per LSR received from the CLEC by one of the OSS interactive ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digita	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Addi conversion  Port - Non Feature  SS LSR Charge, Electronic, tive interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$3.50	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$3.50	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$19.99	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$3.50	NA NA NA S174.35 \$174.35 \$53.50 \$3.50	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$3.50	\$34.4 \$36.9: \$46.4! NA \$117.2 \$212.8 \$3.50
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital C - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Orger LSR received from the CLEC by one of the OSS interactive NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 1st  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 4ddl	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Addi conversion  Port - Non Feature  SS LSR Charge, Electronic, tive interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$3.50 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$3.50 \$19.99	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99	NA NA NA NA S174.35 \$174.35 \$53.50 \$3.50 \$19.99	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$3.50 \$19.99	\$34.4 \$36.9 \$46.4 NA \$117.2 \$212.8 \$3.50 \$19.90
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Oper LSR received from the CLEC by one of the OSS interact NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - Electronic Service Order - Addl  NRC - Electronic Service Order Disconnect	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Addi conversion  Port - Non Feature  SS LSR Charge, Electronic, tive interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56 \$0.42	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$3.50 \$19.99 \$0.42	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	NA NA NA NA S174.35 \$1	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	\$34.4 \$36.9 \$46.4 NA \$117.2 \$117.2 \$212.8 \$3.50 \$19.99 \$0.42
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital C - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Orger LSR received from the CLEC by one of the OSS interactive NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 1st  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 4ddl	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Addi conversion  Port - Non Feature  SS LSR Charge, Electronic, tive interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$3.50 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$3.50 \$19.99	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99	NA NA NA NA S174.35 \$174.35 \$53.50 \$3.50 \$19.99	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$3.50 \$19.99	\$34.4 \$36.9 \$46.4 NA \$117.2 \$117.2 \$212.8 \$3.50 \$19.99 \$0.42
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Oper LSR received from the CLEC by one of the OSS interact NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - Electronic Service Order - Addl  NRC - Electronic Service Order Disconnect	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Addi conversion  Port - Non Feature  SS LSR Charge, Electronic, tive interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56 \$0.42	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$3.50 \$19.99 \$0.42	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	NA NA NA NA S174.35 \$1	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	\$34.4 \$36.9 \$46.4 NA \$117.2 \$117.2 \$212.8 \$3.50 \$19.99 \$0.42
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Oper LSR received from the CLEC by one of the OSS interact NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - Electronic Service Order - Addl  NRC - Electronic Service Order Disconnect	ital Port - Zone 2 tal Port - Zone 3 tial Port - Zone 3 tial Port - Zone 4  Port - 1st conversion  Port - Addi conversion  Port - Non Feature  SS LSR Charge, Electronic, tive interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56 \$0.42	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$3.50 \$19.99 \$0.42	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	NA NA NA NA S174.35 \$1	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	\$34.4 \$36.9: \$46.4! NA \$117.2 \$117.2 \$212.8 \$3.50 \$19.99 \$0.42
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digit RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Grade Loop/2-wire ISDN Digital INRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital INRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital INRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Orger LSR received from the CLEC by one of the OSS interact NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 1st  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - Addl  NRC - Electronic Service Order Disconnect  NRC - Incremental Manual Service Order Disconnect	ital Port - Zone 2 tal Port - Zone 3 tal Port - Zone 3 tal Port - Zone 4  Port - 1st conversion  Port - Add'l conversion  Port - Non Feature  SS LSR Charge, Electronic, titve interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56 \$0.42	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$3.50 \$19.99 \$0.42	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	NA NA NA NA S174.35 \$1	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	\$34.4 \$36.9: \$46.4! NA \$117.2 \$117.2 \$212.8 \$3.50 \$19.99 \$0.42
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Grade Loop/2-wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 4st  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - Add  NRC - Incremental Manual Service Order Disconnect  4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS	ital Port - Zone 2 tal Port - Zone 3 tal Port - Zone 3 tal Port - Zone 4  Port - 1st conversion  Port - Add'l conversion  Port - Non Feature  SS LSR Charge, Electronic, titve interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USACB SOMEC SOMAN	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56 \$21.56 \$0.42 \$3.84	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	NA NA NA NA S174.35 \$174.35 \$53.50 \$3.50 \$19.99 \$0.42 \$20.00	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$34.4 \$36.9: \$46.44 NA \$117.2 \$117.2 \$212.8 \$3.50 \$19.99 \$0.42 \$20.00
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital C - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital RC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Over LSR received from the CLEC by one of the OSS interaction NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 1st NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - Electronic Service Order Disconnect NRC - Incremental Manual Service Order Disconnect  4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS A - Wire ISDN DS1 Digital Trunk Port	ital Port - Zone 2 tal Port - Zone 3 tal Port - Zone 3 tal Port - Zone 4  Port - 1st conversion  Port - Add'l conversion  Port - Non Feature  SS LSR Charge, Electronic, titve interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USASB	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56 \$0.42	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99 \$19.99	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$3.50 \$19.99 \$0.42	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	NA NA NA NA S174.35 \$1	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99	\$34.4 \$36.9: \$46.44 NA \$117.2 \$117.2 \$212.8 \$3.50 \$19.99 \$0.42 \$20.00
RC - 2-Wire ISDN Digital Grade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital Crade Loop with 2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Grade Loop/2-wire ISDN Digital Grade Loop/2-wire ISDN Digital Subsequent Activity  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - 4st  NRC - 2-Wire ISDN Digital Grade Loop/2-wire ISDN Digital Manual Service Order - Add  NRC - Incremental Manual Service Order Disconnect  4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS	ital Port - Zone 2 tal Port - Zone 3 tal Port - Zone 3 tal Port - Zone 4  Port - 1st conversion  Port - Add'l conversion  Port - Non Feature  SS LSR Charge, Electronic, titve interfaces (Note 7)  Port - Incremental Cost-	Note 8 Note 8 Note 8 Note 8 USACB USACB USACB USACB USACB SOMEC SOMAN	\$54.16 \$84.80 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$36.41 \$39.30 NA \$86.79 \$54.04 \$53.50 \$2.75 \$21.56 \$21.56 \$0.42 \$3.84	\$38.74 \$53.64 NA \$239.95 \$156.92 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$44.10 \$55.35 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$43.20 \$59.69 NA \$79.01 \$53.97 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$80.78 \$93.31 \$106.55 \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	NA NA NA NA S174.35 \$174.35 \$53.50 \$3.50 \$19.99 \$0.42 \$20.00	\$73.98 \$87.03 NA \$79.12 \$54.04 \$53.50 \$19.99 \$19.99 \$0.42 \$20.00	\$34.4° \$36.92 \$46.46

Version 3Q00:11/07/00

DES	CRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
	RC - 4- Wire DS1 Digital Loop- Zone 2	USL4P	\$84.05	\$119.68	\$3.21	\$135.15	\$129.12	\$67.58	NA	\$89.90	\$75.40
	RC - 4- Wire DS1 Digital Loop- Zone 3	USL4P	\$152.29	\$194.70	\$101.93	\$186.26	\$344.16	\$96.58	NA	\$119.06	\$98.59
	RC - 4-Wire DS1 Digital Loop - Zone 4	USL4P	NA	NA	NA	NA	NA	\$127.47	NA	NA	NA
	Combination Rates										
	RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$241.72	NA	NA
	RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 1	Note 8	\$237.76	\$187.87	\$218.69	\$219.25	\$208.25	\$264.20	NA	\$274.40	\$136.13
	RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 2	Note 8	\$270.07	\$215.07	\$227.29	\$248.36	\$236.67	\$280.79	NA	\$304.69	\$153.80
	RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 3	Note 8	\$338.31	\$290.08	\$265.09	\$299.47	\$451.57	\$309.79	NA	\$333.85	\$176.99
	RC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port - Zone 4	Note 8	NA	NA	NA	NA	NA	\$340.68	NA	NA	NA
	Local Number Portability										
	Local Number Portability (1 per port)	LNPCN	\$1.75	\$1.75	\$1.75	\$1.75	\$1.75	\$1.75	\$1.75	\$1.75	\$1.75
	Interface (Provsioning Only)										
	Voice/Data	PR71V	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Digital Data	PR71D	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Inward Data	PR71E	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	Non-Recurring Charges										
	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination - 1st conversion	USACP	\$240.30	\$247.97	\$269.96	\$240.30	\$239.95	\$240.30	\$481.51	\$240.30	\$328.53
	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination - Add'l										•
	conversion  NRC -4 - Wire DS1 Digital Loop with 4 - Wire ISDN DS1 Digital Trunk Port -	USACP	\$157.17	\$157.17	\$269.96	\$157.17	\$156.94	\$157.17	\$481.51	\$157.17	\$328.53
	Subsequent Channel Activation - Per Channel	USASP	\$29.06	\$29.06	\$28.71	\$29.06	\$29.01	\$29.06	\$36.92	\$29.06	\$28.39
ı	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination - Subsequent Inward/2-way Telephone Numbers	PR7TG	\$0.98	\$0.9804	\$0.9686	\$0.98	\$0.98	\$0.98	\$1.17	\$0.98	\$0.9353
П	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination -										
	Subsequent Outward Telephone numbers	PR7TP	\$23.02	\$23.02	\$22.75	\$23.02	\$22.99	\$23.02	\$28.17	\$23.02	\$22.36
	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination - Subsequent Inward Telephone Numbers	PR7ZT	\$46.05	\$46.05	\$45.49	\$46.05	\$45.98	\$46.05	\$56.33	\$46.05	\$44.71
	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN DS1 Digital Port Combination - Subsequent Service Order Per Order	USASP	\$147.47	\$147.47	\$147.47	\$147.47	\$147.47	\$147.47	\$255.25	\$147.47	\$189.76
						· ·				·	
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
	NRC - 4-Wire DS1 Digital Loop with 4-wire ISDN Digital Port - Incremental Cost-	JOIVILO	ψ3.30	Ψ2.73	ψ3.30	ψ3.30	ψ3.30	ψ3.50	ψ3.30	ψ3.30	ψ3.30
	Manual Service Order - 1st	SOMAN	\$19.99	\$21.56	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
	NRC - 4-Wire ISDN Digital Loop with 4-wire ISDN Digital Port - Incremental Cost-	SOMAN	Ψ15.55	ΨZ 1.30	ψ13.33	ψ13.33	ψ19.99	ψ13.33	Ψ13.33	ψ13.33	ψ13.33
	Manual Service Order - Addl	SOMAN	\$19.99	\$21.56	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99	\$19.99
	NRC - Electronic Service Order Disconnect	00.12.11	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42	\$0.42
	NRC - Incremental Manual Service Order Disconnect		\$20.00	\$3.84	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
			4-0.00	*****	4-0.00	4-0.00	4	4=0.00	<del>+</del>	4=0.00	<del></del>
	4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port										
	4 - Wire DDITS Digital Trunk Port (Formerly DID Trunk Port)	UDD1T	\$130.23	\$63.31	\$120.80	\$83.28	\$149.27	\$146.46	\$123.65	TBD	\$120.00
	4 - Wire DS1 Digital Loop	USLDC	ψ130.23	ψ03.31	\$120.00	ψ03.20	⊕143.21	\$140.40	\$123.03	100	\$120.00
	4 - Wire DS1 Digital Loop - Statewide	USLDC	NA	NA		NA	NA	NA	\$62.71	NA	NA
	4 - Wire DS1 Digital Loop - Statewide	USLDC	\$51.74	\$64.69	\$55.53	\$106.04	\$56.32	\$50.99	NA	\$59.61	\$57.73
	4 - Wire DS1 Digital Loop - Zone 2	USLDC	\$84.05	\$94.71	\$64.13	\$135.15	\$96.73	\$67.58	NA NA	\$89.90	\$75.40
	4 - Wire DS1 Digital Loop - Zone 3	USLDC	\$152.29	\$208.93	\$101.93	\$186.26	\$197.57	\$96.58	NA NA	\$119.06	\$98.59
	4 - Wire DS1 Digital Loop - Zone 3	USLDC	NA	NA	NA	NA	NA	\$127.47	NA NA	NA	\$96.59 NA
H	Combination Rates	UULDU	14/5	14/1	14/5	14/5	14/7	Ψ121.41	14/1	14/7	INA
H	4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Statewide	Note 8	NA	NA	NA	NA	NA	NA	\$186.36	NA	NA
$\vdash$	4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Statewide  4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 1	Note 8	\$181.97	\$128.00	\$176.33	\$189.32	\$205.59	\$197.45	NA	TBD	\$93.28
+	4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 1	Note 8	\$214.28	\$158.02	\$184.93	\$218.43	\$246.00	\$214.04	NA NA	TBD	\$110.95
+	4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 3	Note 8	\$282.52	\$272.24	\$222.73	\$269.54	\$346.84	\$243.04	NA NA	TBD	\$134.14
+	4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Zone 4	Note 8	NA	NA	NA	NA	NA	\$273.93	NA NA	NA NA	NA
+	22. Signal 200p mail 1 1110 25110 Hullet 1011 2010 4			. 4/1				Ψ2.0.00		. 473	103
$\vdash$	Local number Portability per DSO Activated	LNPCP	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15	\$3.15
+	Central Office Terminating Point	CTG	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	*	0.0	\$0.00	ψ0.00	Ψ0.00	\$0.00	ψ0.00	Ψ0.00	ψ0.00	ψ0.00	ψ0.00
4	Telephone Number / Trunk Group establishment	LIDTOY	<b>#0.00</b>	<b>\$0.00</b>	<b>60.00</b>	60.00	<b>60.00</b>	<b>60.00</b>	<b>60.00</b>	<b>60.00</b>	<b>60.00</b>
-	Telephone Number for 2-Way Trunk Group	UDTGX	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
+	Telephone Number for 1-Way Outward Trunk Group	UDTGZ	\$0.00 \$0.00								
	Telephone Number for 1-Way Inward Trunk Group Without DID DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers (FL,										
$\vdash$	GA, NC, & SC only)  DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers (AL,	NDZ	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
	KY, LA, MS, & TN). In addition, Provides Additional DID Numbers for each Group of 20										
	DID Numbers (Valid in All States)	ND4	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
П	DID Numbers, Non- consecutive DID Numbers , Per Number	ND5	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00

																						ł					H					H	DESC	ם ם
Loop with 4 - Wire DDITS Trunk PortFixed cost 0-8 miles (Facilities Termination) - Disconnect Additional - New Only	NRC - Interoffice Chamnel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk PortFixed cost 0.8 miles (Facilities Termination) - Disconnect - 1st - New Only	NRC - Interoffice Channel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk PontFixed cost 0-8 miles (Facilities Termination) - Additional - New Only	NRC - Interoffice Channel Mileage - (Dedicated DS1) FX/FCO for 4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk PortFixed cost 0-8 miles (Facilities Termination) - 1st - New Only	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Digital Trunk Port - Subsequent Telephone Numbers	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Digital Trunk Port - Subsequent Signaling Changes	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel Activation - Per Channel - 2-Way DID with User Transfer	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel Activation - Per Channel - 1-Way Inward Trunk With DID	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel Activation - Per Channel - 1-Way Inward Trunk Without DID	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel Activation - Per Channel - 1-Way Outward Trunk	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Subsequent Channel Activation - Per Channel - 2-Way Trunk	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - Additional Disconnect	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - 1st - Disconnect	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - New - Additional	lō.	Wire DS	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with Change - Trunks - Additional	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with Change - Trunks - 1st	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with DS1 Changes - Additional	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion with DS1 changes - 1st	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port - Conversion - Switch as is - Additional	NRC -4 - Wire DS1 Digital Loop with 4 - Wire DDITS Trunk Port -Conversion - Switch as is - 1st	NRC - Incremental Manual Service Order Disconnect	NRC- 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port - Incremental Cost-Manual Service Order - Add!	NRC- 4-Wire DS1 Digital Loop with 4-Wire DDITS Trunk Port - Incremental Cost- Manual Service Order - 1st	NRC - AVIrie DSI Digital Loop with 4-Wire DDITS Trunk Port Combination - QSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	Non-recurring Charges	Enhanced Performance Charges - as negotiated in contract	Enhanced Performance Charges	Fixed cost 25 + miles (Facilities Termination) Additional costs 25 + miles	additional costs per mile 9-25 miles	Additional costs per mile 0-8 miles  Fixed cost 9-25 miles (Facilities Termination)	Loop with 4 - wire DDI 3 Trunk Port Fixed cost 0-8 miles (Facilities Termination)	Interpretable A Wife DRIFE Trans Days	to interior:
1LNO4	1LNO3	1LNO2	1LNO1			UDTTE	UDTTD	UDTTC	UDTTB	UDTTA	UDDIT	UDDIT	UDDIT	UDDIT	USAS4	USAWB	USAWB	USAWA	USAWA	USAC4	USAC4		SOMAN	SOMAN	SOMEC		UDTPC		1LNO3	1LNOB	1LNOA	1LNO1	0800	5
\$20.42	\$25.44	\$148.18	\$198.15	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	N	¥	NA	TBD	TBD	TBD	TBD	TBD	TBD	ТВО	\$20.00	\$19.99	\$19.99	\$3.50		TBN		\$0.00 \$0.6920	\$0.6920	\$0.6920	\$79.69	Ä	-
NA	NA	NA	۸N	TBD	TBD	\$28.96	\$28.96	\$28.96	\$28.96	\$28.96	NA	NA	¥	NA	\$54.00	\$134.07	\$268.82	\$134.07	\$268.82	\$134.07	\$268.82	\$3.84	\$21.56	\$21.56	\$2.75		TBN		\$0.00 \$0.2000	\$0.2000	\$0.00	\$92.62	7	4
NA	NA	\$111.75	\$147.07	TBD	TBD	\$28.71	\$28.71	\$28.71	\$28.71	\$28.71	TBD	TBD	\$514.02	\$858.30	\$147.47	\$269.96	\$269.96	\$269.96	\$269.96	\$269.96	\$269.96	\$20.00	\$16.84	\$37.88	\$3.50		TBN		\$0.00 \$0.3068	\$0.3068	\$0.3068	\$63.39	GA	)
NA	N	NA	NA	TBD	TBD	\$28.96	\$28.96	\$28.96	\$28.96	\$28.96	TBD	¥	¥.	NA	TBD	\$134.08	\$261.15	\$134.08	\$261.15	\$134.08	\$261.15	\$20.00	\$19.99	\$19.99	\$3.50		TBN		\$0.00	\$0.4500	\$0.4500	\$55.05	2	5
NA	N	N <sub>A</sub>	NA	TBD	TBD	\$26.60	\$26.60	\$26.60	\$26.60	\$26.60	N <sub>A</sub>	N <sub>A</sub>	¥	NA.	\$68.57	\$123.16	\$266.76	\$123.16	\$266.76	\$123.16	\$266.76	\$20.00	\$19.99	\$19.99	\$3.50		TBN		\$0.00 \$0.7831	\$0.7831	\$0.00	\$93.40	5	
NA	N	N <sub>A</sub>	NA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	NA	₹	NA.	TBD	TBD	TBD	TBD	TBD	TBD	TBD	\$20.00	\$19.99	\$19.99	\$3.50		TBN		\$0.00	\$0.6598	\$0.6598	\$74.40	S.	
NA	NA	N <sub>A</sub>	NA	\$120.96	\$29.65	\$146.91	\$146.91	\$146.91	\$146.91	\$146.91	\$11.98	NA	¥	NA	\$127.63	\$490.38	\$490.38	\$490.38	\$490.38	\$490.38	\$490.38	\$20.00	\$19.99	\$19.99	\$3.50		TBN		\$0.00 \$0.0783	\$0.0783	\$0.0783	\$71.29	200	ś
NA	NA	NA	NA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	N <sub>A</sub>	¥	NA	TBD	TBD	TBD	TBD	TBD	TBD	TBD	\$20.00	\$19.99	\$19.99	\$3.50		TBN		\$0.00 \$0.7598	\$0.7598	\$0.7598	\$94.98	y.	3
NA	N <sub>A</sub>	NA	NA	\$88.68	\$22.92	\$108.67	\$108.67	\$108.67	\$108.67	\$108.67	\$15.29	N <sub>A</sub>	N	NA	\$94.88	\$312.91	\$312.91	\$312.91	\$312.91	\$312.91	\$312.91	\$20.00	\$19.99	\$19.99	\$3.50		TBN		\$0.00 \$0.3525	\$0.3525	\$0.3525	\$75.83	Z	1 2

BIRDLAND AND SECURITARY FORMS - CONTROL POWER FORMS - CONTROL PO	SCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	SC	TN
NBC - Separations Forms - Conversion at new results 11											
RNC - Sperificial Formal - Compression at ease in read Indisordal   RNC - Sperificial Formal - Compression of State Sperificial Formal - Compression of St		CCOSF	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
No. 5 - Entended Superfumer Forms - Charge or Subsequent Active y - 1st   CCCCSF   50.00   5											\$0.00
MOC. Extended Sequentime Forms: Convey or Sussessment Analysis - Asstronal Society   Security - Asstronal Society - Asstrona											\$0.00
REC Extended Superframe Format - Annother Conversion of New Install and Additional COCRET St. 20.00 \$0.00			40.00	******	40.00	40.00	40.00	40.00	70.00	40.00	******
REC Extended Squarthare Formari - Conversion or New Install stational COCEF   \$0.000   \$	NRC - Extended Superframe Format - Change or Subsequent Activity - Additional	CCOSE	\$600.00	\$655.00	\$600.00	\$730.00	\$605.00	\$600.00	\$615.00	\$605.00	\$590.0
RIC: Exercised Sepentrane Formari - Contension of New Internal - Additional Notice - A											\$0.00
NBC Extended Superframe Formari - Change or Subsequent Actively - Additional COOPEF   50:000   50											\$0.00
NGC - Extended Signeritame Format - Change of Sidnespeert Activity - Additional CCOEF 5600.00 \$650.00 \$500.00		CCOEE									\$0.00
Abernate Mark Inversion (AM)  No.C. Stenderformer Format. 1st  MCOSP  \$1.00  \$0	1910 - Extended Supermanie i Simat - Change of Subsequent Activity - 1st	CCOLI	\$0.00	\$0.00	\$0.00	ψ0.00	\$0.00	ψ0.00	ψ0.00	\$0.00	\$0.00
NRC - Spentrame Formst - 14st   MCOSF   \$0.00	NRC - Extended Superframe Format - Change or Subsequent Activity - Additional	CCOEF	\$600.00	\$655.00	\$600.00	\$730.00	\$605.00	\$600.00	\$615.00	\$605.00	\$590.0
NRC - Sugeritame Formal - 1451 NCOSEF 50.00 \$0.0	Alternate Mark Inversion (AMI)										
NRC - Spender Spender no Formal - Additional MCOPD 50.00 \$0.		MCOSE	\$0.00	en on	¢0.00	00.00	en nn	00.00	00.00	\$0.00	\$0.00
NRC - Estended Superframe Format - 1st											
NCCPC    \$0.00   \$0.											\$0.00
Wire Voice Grade Loop with 2-Wire Line Port PBX  2-Wise Analog Line Port (PBX), per month  2-Wise Voice Line Line Line Port PBX Voice Line Line Line Line Line Line Line Lin											\$0.00
2-Wire Analog Line Port (PBX), per month   2-Wire VOICE LINBURDLED COMBINATION 2-WAY PBX TRUM: - Business   UEPPC   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.89   \$1.01   \$1.02   \$1.00	INCC - Exterioed Supermanie Pormat - Additional	MCOFO	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00	\$0.00
2-Wire Analog Line Port (PBX), per month   2-Wire VOICE LINBURDLED COMBINATION 2-WAY PBX TRUM: - Business   UEPPC   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.89   \$1.01   \$1.02   \$1.00	W. V. C. I.I. W. C. I. C.										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PEX TRUMS. Residence   UEPPD   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$1.00   \$1.0	Wire Voice Grade Loop with 2-Wire Line Port PBX										
2 WIRE VOICE UNBUNDLED COMBINATION 2-WAY PEX TRUMS*. Residence   UEPRO   \$2.20   \$1.35   \$1.79   \$2.61   \$2.56   \$2.12   \$2.28   \$3.69   \$1.00   \$1.	2-Wire Analog Line Port (PBX), per month										
LINE SIDE LINBUNDLED COMMINATION 2-WAY PEX TRUM.* BUSINESS   UEPPC   \$2.20   \$1.35   \$1.79   \$2.61   \$2.56   \$2.12   \$2.28   \$3.69   \$1.00		UEPRD	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
LINE SIDE LINBUNDLE DICTUMARD PEX TRINK - BUSINESS   UEPPO   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$1.00											\$4.54
LINE SIDE LINBUNDLED INCOMINO PEX TRIMN- BUSINESS   UEPP1   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$3.24   \$3.25   \$3.24   \$3.25   \$3.24   \$3.25											
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX ALABAMA CALLING PORT UEPA2 \$2.20 NA											\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING   UEPL2	LINE SIDE UNBUNDLED INCOMING PBX TRUNK - BUSINESS	UEPP1	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX LOUISIANA CALLING   UEPL2											
UEPL2		UEPA2	\$2.20	NA NA	NA	NA	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED PEX FAMILY PORTS											
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING PORT   UEPTO NA	PORT	UEPL2	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
PORT	2-WIRE VOICE UNBUNDLED PBX LD TERMINAL PORTS	UEPLD	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
PORT	2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX TENNESSEE CALLING										
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT   UEPXA   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.24   \$2.26   \$3.69   \$3.24   \$2.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.25   \$3.21   \$3.25   \$3	PORT	UEPT2	NA	\$4.54							
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT   UEPXA   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.24   \$2.26   \$3.69   \$3.24   \$2.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.25   \$3.21   \$3.25   \$3											
2-WIRE VOICE UNBUNDLED 2-WAY COMBINATION PBX USAGE PORT   UEPXA   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.24   \$2.26   \$3.69   \$3.24   \$2.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$2.12   \$2.28   \$3.69   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.25   \$3.21   \$3.25   \$3.25   \$3.25   \$3.21   \$3.25   \$3	2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX TENNESSEE CALLING PORT	UEPTO	NA	\$4.54							
2-WIRE VOICE UNBUNDLED PBX TOLL TERMINAL HOPEL PORTS UEPXB \$2.20 \$1.35 \$1.79 \$2.61 \$2.55 \$2.12 \$2.28 \$3.69 \$2.24 WIRE VOICE UNBUNDLED PBX LD DDT PERMINAL SWITCHBOARD PORT UEPXD \$2.20 \$1.35 \$1.79 \$2.61 \$2.55 \$2.12 \$2.28 \$3.69 \$2.24 WIRE VOICE UNBUNDLED PBX LD DTERMINAL SWITCHBOARD PORT UEPXD \$2.20 \$1.35 \$1.79 \$2.61 \$2.55 \$2.12 \$2.28 \$3.69 \$2.24 WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD IND CAPABLE PORT UEPXD \$2.20 \$1.35 \$1.79 \$2.61 \$2.55 \$2.12 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2.24 \$2.24 \$2.28 \$3.69 \$3.24 \$2		UEPXA	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED PBX LD TERMINALS PORT   UEPXC   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.20   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.20											\$4.54
2-WIRE VOICE UNBUNDLED PBX LD TERMINAL SWITCHBOARD PORT   UEPX   S2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$5.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$5.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$5.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$3.											\$4.54
Depart   D											\$4.54
PORT		UEFAD	\$2.20	\$1.33	\$1.79	\$2.01	\$2.55	φ2.12	\$2.20	\$3.09	<b>\$4.54</b>
2-WIRE VOICE UNBUNDLED 2-WAY PBX KENTUCKY ROOM AREA CALLING PORT   UEPXF NA NA NA \$2.61 NA		HEDVE	00.00	04.05	04.70	00.04	00.55	00.40	60.00	00.00	0454
WITHOUT LUD		UEPXE	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED PBX KENTUCKY LID AREA CALLING PORT UEPXG NA NA NA \$2.61 NA											
2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT UEPXH NA NA NA \$2.61 NA											NA
2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT WITHOUT LUD NA											NA
LUD 2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING PORT UEPXK NA	2-WIRE VOICE UNBUNDLED PBX KENTUCKY PREMIUM CALLING PORT	UEPXH	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
LUD UEPXJ NA NA NA \$2.61 NA	2-WIRE VOICE UNBUNDLED 2-WAY KENTUCKY AREA CALLING PORT WITHOUT										
2-WIRE VOICE UNBUNDLED 2-WAY PBX LOUISIANA LOCAL OPTIONAL CALLING PORT UEPXK NA		UEPXJ	NA	NA	NA	\$2.61	NA	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ADMINISTRATIVE CALLING PORT UEPXL \$2.20 \$1.35 \$1.79 \$2.61 \$2.55 \$2.12 \$2.28 \$3.69 \$5 2-WIRE VOICE UNBUNDLED 2-WAY PBX HOTEL/HOSPITAL ECONOMY ROOM CALLING PORT UEPXM \$2.20 \$1.35 \$1.79 \$2.61 \$2.55 \$2.12 \$2.28 \$3.69 \$5 2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORT UEPXN NA			NA	NIA	NA		\$2.55	NA	NA	NA	NA
ADMINISTRATIVE CALLING PORT   UEPXL   \$2.20   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$2.90   \$1.35   \$1.79   \$2.61   \$2.55   \$2.12   \$2.28   \$3.69   \$3.69   \$3.60		OLITAN	INA	140	INA	11/0	ΨΖ.ΟΟ	1307	IN	INC	INA
CALLING PORT	ADMINISTRATIVE CALLING PORT	UEPXL	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL ECONOMY ADMINIATRATIVE CALLING PORT UEPXN NA		LIEBYAY	00.00	04.5-	04 ==	00	00	00 :-	00.00	00.00	
ADMINATRATIVE CALLING PORTTENNESSEE CALLING PORT  2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX HOTEL/HOSPITAL DIACOUNT ROOM CALLING PORT  2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT  2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBX LOUISIANA LOCAL DISCOUNT CALLING PORT  2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT  2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL CONOMY CALLING PORT  1		UEPXM	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
ROOM CALLING PORT	ADMINIATRATIVE CALLING PORTTENNESSEE CALLING PORT	UEPXN	NA	\$4.54							
CALLING PORT	ROOM CALLING PORT	UEPXO	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL ECONOMY CALLING PORT UEPXQ NA		UEPXP	NA	NA	NA	NA	\$2.55	NA	NA	NA	NA
2-WIRE VOICE UNBUNDLED 2-WAY PBX MISSISSIPPI LOCAL OPTIONAL CALLING PORT UEPXR NA											
PORT		UEPXQ	NA	NA	NA	NA	NA	\$2.12	NA NA	NA	NA
2-WIRE VOICE UNBUNDLED 1-WAY OUTGOING PBXMEASURED PORT UEPXS \$2.20 \$1.35 \$1.79 \$2.61 \$2.55 \$2.12 \$2.28 \$3.69 \$ 2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS CALLING						1				1	
2-WIRE VOICE UNBUNDLED 2-WAY PBX SOUTH CAROLINA AREA PLUS CALLING											NA
		UEPXS	\$2.20	\$1.35	\$1.79	\$2.61	\$2.55	\$2.12	\$2.28	\$3.69	\$4.54
		UEPXT	NA	\$3.69	NA						
2-WIRE VOICE UNBUNDLED PBX COLLIERVILLE & MEMPHIS CALLING PORT UEPXU NA											\$4.54

Version 3Q00:11/07/00
LOOP-PORT COMBOS

2-Wire Voice Grade Loop with 2-Wire Line Port - New -		NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Addi	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add"	Ņ	2-Wire Voice Grade Loop with 2-Wire Line Port	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	- 2-Wire Voice Grade Loop with 2-Wire Line Port - New -	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	New -	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	2-Wire Voice Grade Loop with 2-Wire Line Port - New -	2-Wire Voice Grade Loop with 2-Wire Line Port - New -	2-Wire Voice Grade Loop with 2-Wire Line Port - New -	NRCs for New (not Currently Combined) as ordered in Georgia:	NRC - Incremental Manual Service Order Disconnect	NRC - Electronic Service Order Disconnect	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Manual Service Order	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Electronic	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - Add'l	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Incremental Cost - Manual Svc.Order vs. Electronic - 1st	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch with change	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch with change	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Add'l, Switch as is	NRC - 2-Wire Voice Grade Loop/Line Port Combination - 1st, Switch as is	e Loop with 2-Wire Line Port, Zone 4 (Note	- 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 3 (Note	(Note	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Zone 1 (Note 6)	RC - 2-Wire Voice Grade Loop with 2-Wire Line Port, Statewide	RC - 2- Wire voice Grade Loop - Zone 4  Combination Rates	Zone	RC - 2- Wire Voice Grade Loop - Zone 2	RC - 2- Wire Voice Grade Loop - Zone 1	2-Wire Voice Grade Loop (SL1)  RC - 2- Wire Voice Grade Loop - Statewide	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	TOX.	ESCRIPTION  2-WIRE VOICE UNBUNDLED 2-WAY PBX TENNESSEE REGIONSERV CALLING
UEPXM	UEPXL	UEPXL		CEPXD	UEPXD	UEPXC	UEPXC	UEPXB	UEPXB	UEPXA	UEPXA	UEPLD	UEPLD	UEPP1	UEPPO	UEPPO	UEPPC	UEPPC	UEPRD	IEBBD					SOMAN	SOMAN	SOMEC	USAS2	USACC	USACC	USAC2	USAC2	Note 8	Note 8	Note 8	Note 8	Note 8	OEPLX	UEPLX	UEPLX	UEPLX	UEPLX	LNPCP	OEPXV	USOC
NA	¥.	¥ §	2 3	<b>X X</b>	¥	NA NA	NA	₹	NA	¥	NA	¥ :	Z 3	NA NA	¥	NA	NA	NA	<b>∑</b> 5	NA	\$20.00	TBD	\$8.25	\$1.44	\$9.58	\$40.71	\$3.50	\$10.00	\$0.41	\$2.80	\$0.41	\$2.80	NA.	\$44.44	\$25.51	\$16.55	NA	¥	\$42.24	\$23.31	\$14.35	NA		¥	£ }
NA A	NA A	X 3	Z Z	Z Z	¥	NA	NA	₹	¥	A	NA	¥.	N S	N N	×	NA A	NA	NA	¥.	NA	\$3.84	\$0.42	TBD	TBD	\$21.56	\$21.56	\$2.75	\$10.00	\$3.80	\$15.82	\$3.80	\$15.82	N <sub>A</sub>	\$25.60	\$19.86	\$16.25	¥	NA	\$24.25	\$18.51	\$14.90	N.		Ā	; P
\$22.14	\$15.25	\$22.14	\$15.05	\$15.25	\$22.14	\$15.25	\$22.14	\$15.25	\$22.14	\$15.25	\$22.14	\$15.25	\$22.14	\$22.14	\$15.25	\$22.14	\$15.25	\$22.14	\$15.25	\$22.14	\$20.00	TBD	TBD	TBD	\$7.88	\$33.67	\$3.50	\$10.00	\$0.3108	\$2.01	\$0.3108	\$2.01	NA	\$21.62	\$14.26	\$12.59	NA	N.	\$19.83	\$12.47	\$10.80	NA		¥	GA GA
NA	NA.	¥ §	2 3	× ×	. ₹	NA NA	NA	N N	¥	¥	NA	¥ :	N 5	N K	. ₹	N <sub>A</sub>	NA	NA	¥ ;	NA	\$20.00	TBD	TBD	TBD	\$19.99	\$19.99	\$3.50	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	NA.	#VALUE!	#VALUE!	#VALUE!	NA	¥	\$47.78	\$27.68	\$14.79	NA		¥	হ হ
NA	NA	₹ 3	Z §	<b>8</b>	X.	NA	NA	₹	¥	NA	NA	N :	Z 3	NA NA	NA.	NA	NA	NA	X :	NA	\$20.00	TBD	\$5.12	\$2.11	\$7.32	\$31.92	\$3.50	\$10.00	\$0.29	\$3.80	\$0.29	\$3.80	NA	\$51.85	\$26.69	\$16.60	NA	¥	\$49.30	\$24.14	\$14.05	¥		Ä	<b>.</b> .
NA	N	₹ 3	Z 3	<b>3 5</b>	X.	NA	NA	¥	NA	A	NA	¥ :	8 3	NA NA	NA.	NA	NA	NA	¥ ;	NA	\$20.00	TBD	\$6.88	\$2.87	\$0.99	\$43.52	\$3.50	\$10.00	\$0.41	\$5.20	\$0.41	\$5.20	\$38.59	\$29.75	\$21.45	\$16.71	¥	\$36.47	\$27.63	\$19.33	\$14.59	¥		Ā	NS NS
NA NA	Ν	<b>₩</b>	N N	N X	¥	ΝA	NA	¥	NA	NA.	NA (	×.	N S	NA KA	X.	Ą	NA	NA	¥.	NA	\$20.00	TBD	\$10.27	\$1.42	\$9.45	\$40.18	\$3.50	\$10.00	\$0.40	\$2.77	\$0.40	\$2.77	NA	NA	NA	Z <sub></sub>	\$16.46	N	¥ ×	A	N	\$14.18		VA.	S NO
NA A	NA	₹ ₹	Z 3	3 5	X.	NA	NA	¥	NA	NA	NA (	N :	Z 3	NA NA	NA NA	NA	NA	NA	<b>X</b> :	NA	\$20.00	TBD	\$8.91	\$0.71	\$9.91	\$43.19	\$3.50	\$10.00	\$0.40	\$1.59	\$0.40	\$1.59	NA	\$37.68	\$29.35	\$20.71	NA	X	\$33.99	\$25.66	\$17.02	¥		¥	SC
NA	NA	8	N S	3 8	X.	NA	NA	Š	N	¥	NA I	X.	X 3	NA NA	NA.	NA	NA	NA	X :	NA	\$20.00	TBD	\$7.97	\$0.76	\$7.03	\$30.89	\$3.50	\$10.00	\$0.2886	\$1.03	\$0.2886	\$1.03	NA A	\$26.31	\$18.96	\$17.02	NA	NA	\$21.77	\$14.42	\$12.48	NA		\$4.54	T N

						l	5_															DE
Where the state Commission has adopted rates for the rate elements contained herein, it is the intent of the parties to reflect such rates in this exhibit and to apply the same consistent with applicable FCC and Commission rules and orders.	7 In the absence of ordered OSS rates by a state commission, BellSouth will offer regionwide rates 8 There is not a unique combination USOC. CLEC should submit the loop and port USOCs. 9 Rates in TN and FL are interim and shall be trued-up when final rates are ordered.	6 Geographically Deaveraged UNE Zones and applicable rates have been established for certaint services, as shown in this Agreement. Where Geographically Deaveraged UNE Zones and applicable rates are established. Statewide rates are obsolete. Further, BelfSouth is in the process of enhancing its billing systems in order to accomodate this Geographically Deaveraged UNE Zone Rate Structure. Until these enhancements are accomplished, estimated to be mid 2001, the UNE Zone 1 rate will be billed for all services residing in Zones 1, 2, 3 or 4, i.e., Rates for services residing in UNE Zones 2, 3 and UNE Zone 4, where applicable, will mot be billed. Once billing enhancements are complete, all applicable UNE Zone rates reflected in this Agreement will be billed. Reference internet Website thrus/www.interconnection.besouth.com/become_clec/docs/interconnection/deavurzns.pdf to view Geographically Deaveraged UNE Zone Designations by Central Office.	4 End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section of this rate exhibit shall apply to all combinations of loop/port network elements.  5 Detected	3 In the absence of ordered rates by a State Commission, the recurring rates for Currently Combined combinations of loop/port network elements will be the sum of the recurring rates for the UNEs which make up the combinations, and the nonrecurring rates shall be as set forth in this section.	In Georgia, rates will apply for Currently Combined as well as not Currently Combined loop/port combinations unless otherwise identified.	Inherim rates subject to true-up.  1 Market Rates will apply in those areas where BellSouth is not required to provide circuit switching pursuant to FCC rules.	TEO.	LOCAL NUMBER PORTABILITY (REQUIRES ONE PER PORT)	All Other Loop/Port Combinations	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.  Electronic - New - Disconnect	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Manual Service Order	NRC- 2 Wire Voice Grade Loop/Line Port Combination - Subsequent Database Update - Electronic	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs. Electronic - New - Add'l	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - Incremental Cost Manual vs.  Electronic - New - 1st	NRC - 2-Wire Voice Grade Loop/Line Port Combination - OSS LSR Charge, Electronic, per LSR received from the CLEC by one of the OSS interactive interfaces (Note 7)	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - Add'l	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Disconnect - 1st	NIDO - 2-Mire Visios Grade I not/I ina Bort Combination - Subsequent	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st  NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add!	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Add'l	NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - Addi NRC - 2-Wire Voice Grade Loop with 2-Wire Line Port - New - 1st	DESCRIPTION
	ide rates							LNPCX							SOMEC		CONOR	IIGAGS	UEPXS	UEPXO	UEPXO	USOC
									TBD	N <sub>A</sub>	×.	NA	N <sub>A</sub>	NA	N N	₹	<b>₹</b>	No.	<b>X X</b>	AN	¥ ¥	AL AL
									TBD	NA	NA	NA	NA	NA.	<b>⊼</b>	NA	N S	NIA :	<b>X</b>	NA	<b>₹</b>	<u> </u>
									Note 2	\$11.17	TBD	TBD	\$8.19	\$37.06	\$3.50	\$3.91	\$8.45	\$1000	\$15.25	\$15.25	\$15.25	Ĝ, GA
									TBD	NA	NA	NA NA	NA.	N <sub>A</sub>	<b>⊼</b>	N <sub>A</sub>	X 5	NIA :	<b>X</b> X	NA.	<b>₹</b>	হ হ
									TBD	NA	NA	NA	NA	NA	¥	NA	₩ 5	NA :	8	NA.	¥ ¥	5 5
									TBD	NA	NA	NA	N <sub>A</sub>	NA	Z S	NA	X 5	No. 1	<b>₹</b>	NA	¥ ¥	, MS
									TBD	NA	NA	NA	NA	NA.	<b>⊼</b>	NA	N S	NIA :	<b>₩</b>	NA	<b>₹</b>	S C
									TBD	NA	¥	NA	¥	NA	¥	NA	¥ §	No.	8	NA	¥ ¥	SC
									TBD	N	N <sub>A</sub>	NA	N <sub>A</sub>	NA	Z S	NA	X 5	No.	¥ ¥	NA	¥	Į.

		1								Н	ł	F	F		H	+	ł						ł		Н	1		Н	1				ł					1		П	I			1		F					П	1	
NRC- DS1 interoffice Facility Termination - 1st  NRC-DS1 interoffice Facility Termination - Add1	Non-Recurring Charges - New EEL (Note 2)(Note 3)	2-wire ISDN(BRITE COCI) per month	DS1 Channelized System per month	Interoffice Channel - Dedicated - DS1 - Per mile per month	2-wire ISDN Loop per month, Zone 4 (Note 1)	2-wire ISDN Loop per month, Zone 3 (Note 1)	2-wire ISDN Loop per month, Zone 2 (Note 1)	2-wire ISDN Loop per month  2-wire ISDN Loop per month  3-wire ISDN Loop per month	Recurring Charges	DS1 Interoffice Channel and 2-wire ISDN Local Loop:	NRC-DS I Interornice channel and 4-wire vs Local Loop Combination - Ivianual Svc Or	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Manual Svc Or	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Manual Svc Ord	Manual Svc	NRC-DS1 interoffice channel and 4-wire VG Local Loop Combination - Manual Svc Or	Electronic	VG(COCI	NRC-DS1 Channelization System - Add'l	NRC-DS1 Channelization System -1st	NRC-4-wire VG Local Loop - 1st	NRC-DS1 interoffice Facility Termination - Add'l	NRC- DS1 interoffice Facility Termination - 1st	Non-Recurring Charges - New FET (Note 2) (Note 3)	DS1 Channelized System per month	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	Interoffice Channel - Dedicated - DS1 - per mile per month	4-wire VG Loop per month, Zone 3 (Note 1)	4-wire VG Loop per month, Zone 2 (Note 1)	4-wire VG Loop per month, Zone 1 (Note 1)	4-wire VG Loop per month	DS1 Interoffice Channel and 4-wire VG Local Loop EEL:	INTO-2-WIE VG LOCAL COOP AND CHARMENZED DST INTERIORICE COMBINATION - MANAGED	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual St	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual Sv	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual St	NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Electronic NRC- 2-wire VG Local Loop and Channelized DS1 Interoffice Combination - Manual St	NRC-VG(COCI)interface card - Add'	NRC-VG(COCI)interface card -1st	NRC-DS1 Channelization System -1st	NRC-2-wire VG Local Loop - Add'l	NRC-DS1 interoffice Facility Termination - Add"	NRC- DS1 interoffice Facility Termination - 1st	Non-Recurring Charges - New EEL (Note 2)(Note 3)	VG (COCI) interface card per month	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	Interoffice Channel - Dedicated - DS1 - per mile per month	Zone 4	2-wire VG Loop per month. Zone 2. (Note 1)		2-wire VG Loop per month, statewide	DS1 Interoffice Channel and 2-wire VG Local Loop EEL:	transport and roop [channelization in applicable].	ENHANCED EXTENDED LINKS (EELS)  New EEL rates are the sum of the individual UNE network elements (interoffice transport and two fichages).
U1TF1		UC1CA	M -	1L5XX	NA	U1L2X	U1L2X	U1L2X			OUMAN	- 0	•	α.	SOMAN	)	1D1VG	MQ1	MQ1	UEAL4	U1TF1	U1TF1	פארער	MQ1	U1TF1	1L5XX	UEAL4	UEAL4	UEAL4	UFAI 4			SOMAN			SOMEC	1D1VG	1D1VG	MQ	UEAL2	U1TF1	U1TF1		1D1VG	UITE	1L5XX	NA	UEAL2	UEAL2	UEAL2			USOC
N N		<b>₹</b> 5	N S	N N	AN	NA	¥ 5	N A			Š	X X	ΝA	NA.	NA S	Z S	¥	AN	<b>₹</b> §	× ×	AN	NA	¥	×.	NA.	¥ ₹	X X	NA	¥ 5	N		NA	¥ ¥	AN	NA S	X X	NA	X	X X	NA S	X X	NA.		<b>₹</b>	× ×	¥	¥.	X X	¥	¥			Ą
\$171.01 \$101.84		\$3.83	\$153.60	\$0.6013	NA	\$104.47	\$47.35	\$32 3A			Ä	X X	NA	NA.	\$25.40	\$3.17	\$12.16	\$137.77	\$275.18	\$329.76	\$101.84	\$171.01	\$1.45	\$153.60	\$99.79	\$0.6013	\$78.35	\$35.51	\$24.26	N		N.	X X	NA	NA NA	\$3.17	\$8.77	\$12.16	\$275.18	\$35.15	\$101.84	\$171.01		\$1.45	\$99.79	\$0.6013	W	TB E	\$20.52	¥			72
\$142.64		\$3.71	\$18.23	\$0.4523	NA	\$40.17	\$25.27	\$21 80			\$11.85	\$19.88	\$27.49	\$33.63	NA NA	\$3.50	\$12.02	\$137.06	\$206.09	\$228.99	\$94.87	\$142.64	\$2.67	\$18.23	\$78.47	\$0.4523	\$41.99	\$26.42	\$22.88	NA		\$11.00	\$19.88	\$27.49	\$33.63	\$3.50	\$8.66	\$12.02	\$206.09	\$51.57	\$94.87	\$142.64		\$2.20	\$78.47	\$0.4523	NA	\$28.26	\$15.40	¥			GA
X X		₹ 5	N S	N N	AN	NA	¥ 5	N X			¥.	₹ ₹	NA.	NA	NA S	Z F	X X	NA	<b>₹</b> §	× ×	NA	NA	¥	¥	N <sub>A</sub>	<b>₹</b>	X X	NA	¥ 5	NA.		3	<b>X</b>	NA	N S	X X	NA	<b>₹</b>	¥ ¥	N S	X X	NA.		<b>₹</b>	X X	¥	W.	N K	¥	¥			হ
\$160.49 \$123.03		\$4.18	\$209.40	\$0.7831	NA	\$74.19	\$36.22	\$21 15			\$8.06	\$45.91	\$153.37	\$242.20	NA NA	\$3.50	\$12.29	\$135.20	\$220.07	\$128.42	\$123.03	\$160.49	\$1.62	\$209.87	\$93.40	\$0.7831	\$86.47	\$41.85	\$24.36	NA.		\$0.00	\$45.91	\$153.37	\$242.20	\$3.50	\$8.80	\$12.29	\$220.07	\$93.60	\$123.03	\$160.49		\$1.62	\$93.40	\$0.7831	NA.	\$61.32	\$17.65	¥			Σ
N N		<b>₹</b> 5	N S	X X	NA	NA	¥ 5	N X			Š	<b>X X</b>	NA	NA.	NA S	Z Z	<b>X X</b>	NA	<b>₹</b> ₹	<b>X X</b>	NA	NA	¥	<b>.</b> ¥	NA A	<b>₹</b>	<b>X X</b>	NA	¥ ₹	NA		NA	<b>X X</b>	NA	NA S	X X	NA	¥	₹ ₩	N S	8	NA		¥ ₹	<b>X X</b>	A	NA :	Z Z	×	¥			MS
\$217.17 \$163.75		\$3.76	\$177.72	\$0.5753	NA	TBD	TBD G	\$24.98			X	₹ ₹	\$51.40	\$66.20	NA NA	\$3.50	\$15.76	\$182.57	\$301.74	\$288.47	\$462.69	\$534.48	\$1.64	\$177.72	\$71.29	\$0.0783	₹ ₹	NA	NA.	\$27.49		NA.	<b>X X</b>	\$51.40	\$66.20	\$3.50	\$11.28	\$15.76	\$193.63	\$106.56	\$163.75	\$217.17		\$1.64	\$71.29	\$0.5753	NA	# E	TBD	\$19.50			ĸ
N N		X 5	Z S	N N	NA	NA	X S	N A			¥.	X X	NA.	¥	NA S	Z S	X X	NA	<b>₹</b> §	× ×	NA	NA	¥	¥	NA	¥ ₹	X X	NA	¥ 5	NA		N.	X X	NA	NA S	X X	NA	X X	¥	NA S	X X	NA		¥ §	× ×	N.	NA :	N K	¥	¥			SC
\$165.53 \$124.84		\$3.33	\$165.21	\$0.3525	NA	\$28.02	\$19.55	\$15.54			NA	¥	NA.	NA.	\$19.99	\$3.50	\$12.61	\$135.80	\$222.87	\$103.76	\$124.84	\$165.53	\$1.25	\$165.21	\$75.83	\$0.3525	\$27.18	\$20.79	\$15.92	NA		3	X X	NA	NA NA	\$3.50	\$9.03	\$12.61	\$222.87	\$65.84	\$124.84	\$165.53		\$1.25	\$75.83	\$0.3525	NA	\$79.55	\$15.54	¥			뒫

## NETWORK ELEMENTS AND OTHER SERVICES

		Н																																														Ŧ	Ŧ			1						
DS1 Loop per month, Zone 1 (Note 1)	DS1 Loop per month	DS1 Interoffice Channel and DS1 Interoffice Local Loop:	NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual St	NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual Svo	NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual Sv	NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual Sv	NRC-DS1 interoffice channel and 4-wire 64kbps Local Loop Combination - Manual Svo	NRC-4-wire 64kbps(COCI) interface card -Add1	NRC-4-wire 64kbps(COCI)interface card -1st	NRC-DS1 Channelization System - Add'l	NRC-DS1 Channelization System -1st	NRC-4-wire 64kbps Local Loop - Add'l	NRC-4-wire 64kbps Local Loop - 1st	NRC- DS1 interoffice - Add'll	NRC- DS1 interoffice - 1st	4-wire 64kops card COCI per month	DS1 Channelized System per month	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	Interoffice Channel - Dedicated - DS1 - per mile per month	4-wire 64kbps Loop per month, Zone 4 (Note 1)	4-wire 64kbps Loop per month, Zone 3 (Note 1)	Zone 2	Zone 1	Recurring Charges	DS1 Interoffice Channel and 4-wire 64 kbps Local Loop:	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Svo	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Svo	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Svo	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Manual Svo	NRC-DS1 interoffice channel and 4-wire 56kbps Local Loop Combination - Electronic	NRC-4-wire 56kbps(COCI)interface card -Add"	NRC-4-wire 56kbps(COCI)interface card -1st	NRC-DS1 Channelization System - Add"	NRC-17S1 Channelization System -1st	NRC-4-wire 56kbps Local Loop - 1st	NRC-DS1 interoffice Facility Termination - Add'l	NRC- DS1 interoffice Facility Termination - 1st	Non-Recurring Charges - New EEL (Note 2) (Note 3)	DS1 Channelized System per month		Channel - Dedicated - DS1 - per mile per month		Loop per month, Zone 3	4-wire 56kbps Loop per month, Zone 1 (Note 1)		Recurring Charges	DS1 Interoffice Channel and 4-wire 56 kbps Local Loop:	NRC-UST Interoffice channel and 2-wire ISDN Local Loop Combination - Manual Svc		Manual Svc	Manual Svc		NRC-DS1 interoffice channel and 2-wire ISDN Local Loop Combination - Electronic Sv	NRC-2-wire BRITE/COCI)litterface card -Add"	NRC-DS1 Channelization System - Addii	NRC-DS1 Channelization System -1st	ISDN Local L	NRC: 2-wire ISDN Local Loop - 1st
USLXX	USLXX		'd SOMAN					SOMEC	1D1DD	MQ1	MQ1	UDL64	UDL64	UITF1	INTE	10100	MQ1	U1TF1	1L5XX	N.	UDL64	UDI 64	00004	2				SOMAN	Т	S		1D1DD	MO	UDL56	UDL56	U1TF1	U1TF1		MQ	U1TF1	1L5XX	NA	UDL56	UDL56	UDL56			SOMAN	0	0	0	)	SOMEC	IC1CA	MQ1	MQ1	U1L2X	USOC
NA	NA A		X	NA.	NA	<b>₹</b>	₹ 5	N N	¥	X.	¥	NA.	NA	¥ ₹	NA	¥	<b>X</b>	¥	NA	NA	X.	Z 3	N 5	N .		AN	¥.	¥ 3	N N	¥	AN	NA	₹ ;	N N	¥	ΝA	₹	3	× ×	<b>X</b>	N	NA	N S	NA NA	¥			NA	; <sub>₹</sub>	Ą	NA	N S	₹ ₹	N S	N N	¥	¥	N A
\$64.69	N N		X	¥	NA	¥	\$25.40	\$8.77	\$12.16	\$137.77	\$275.18	\$148.55	\$329.76	\$101.84	\$171.01	\$1.00	\$153.60	\$99.79	\$0.6013	N.	\$126.22	\$57.21	\$30.08	N		NA	¥ :	X 5	\$25.40	\$3.17	\$8.77	\$12.16	\$137.77	\$148.55	\$329.76	\$101.84	\$171.01	02.2¢	\$153.60	\$99.79	\$0.6013	NA	\$126.22	\$39.08	×			NA	× ×	¥	N.	\$25.40	\$3.17	\$8 77	\$137.77	\$275.18	\$148.55	\$329.76
\$52.40	NA A		\$11.85	\$19.88	\$27.49	\$33.63	NA S	\$3.50	\$12.02	\$137.06	\$206.09	\$206.98	\$395.14	\$94.87	\$142 64	\$1.06	\$18.23	\$78.47	\$0.4523	NA	\$46.53	\$30.53	\$28.4	20		\$11.85	\$19.88	\$27.49	NA NA	\$3.50	\$8.66	\$12.02	\$137.06	\$206.98	\$395.14	\$94.87	\$142.64	\$1.00	\$18.23	\$78.47	\$0.4523	NA.	\$46.53	\$26.44	NA			\$11.00	\$19.88	\$27.49	\$33.63	NA NA	\$3.50	88.88	\$137.06	\$206.09	\$51.57	<b>GA</b> \$77.54
NA	×.		X	¥ ¥	NA.	Z.	₹ 5	N N	NA.	NA NA	¥	NA	NA.	<b>₹</b>	N	NA	<b>X</b>	<b>X</b>	¥	NA.	¥ ;	Z 3	N 5	200		NA	¥.	NA 5	<b>S S</b>	×.	NA	NA	X.	N X	N.	¥	¥.	5	₹ ₹	× ×	¥	NA.	¥ 5	S S	NA.			- NA	· ×	¥	NA.	N S	K 5	N S	8 8	. ₹	¥	⊼ ኛ
\$56.32	NA.		\$8.06	\$45.91	\$153.37	\$242.20	NA S	\$3.50	\$12.29	\$135.20	\$220.07	\$230.50	\$333.28	\$123.03	\$160.49	\$3.12	\$209.87	\$93.40	\$0.7831	NA	\$96.48	\$47.24	\$37.50	NIA		\$8.06	\$45.91	\$153.37	\$242.20 A	\$3.50	\$8.80	\$12.29	\$135.20	\$230.50	\$333.28	\$123.03	\$160.49	\$3.12	\$209.87	\$93.40	\$0.7831	NA	\$96.48	\$27.50	ž ×			\$8.06	\$16.12	\$36.31	\$57.58	NA NA	\$3.50	\$8.2.29 62.21¢	\$135.20	\$220.07	\$172.63	\$223.27
NA	N.		X	X X	NA.	N.	<b>₹</b> 5	N N	<b>X</b>	¥	¥	¥	NA	¥ §	NA	NA.	K K	. ₹	A	NA.	₩ :	N 5	N 5	N		NA	¥.	N 5	N K	¥	NA	NA	¥ ;	NA NA	¥	NA.	¥	5	× ×	<b>X</b>	¥	NA.	¥ 5	N N	¥			NA.	: X	¥	NA.	N.	K 5	N S	N K	. ₹	¥	N S
TBD	\$62.78		X	¥	\$50.83	\$65.01	NA S	\$11.28	\$15.76	\$288.33	\$301.74	\$337.51	\$489.04	\$163.75	\$217.17	\$2.58	\$177.72	\$71.29	\$0.5753	¥	da l	3 5	TBD (	422.67		NA	¥	\$50.83	PER DA	\$3.50	\$11.28	\$15.76	\$182.57	\$337.51	\$489.04	\$163.75	\$217.17	90.20	\$177.72	\$71.29	\$0.5753	NA	TB O	<b>B</b> B	\$32.67			NA	× ×	\$38.07	\$38.07	NA NA	\$3.50	\$11.28	\$182.57	\$301.74	\$251.31	NC \$325.91
NA	NA NA		×	X.	NA	Z.	N S	8 8	¥	X.	×	NA.	NA	¥ ₹	N.	N	<b>X</b>	¥	NA	N <sub>A</sub>	¥ §	Z 3	2 3	25		NA	Z.	N S	Z Z	X X	NA	NA	<b>X</b> :	N X	¥	NA	¥	3	3 8	<b>X</b>	NA	NA	¥ 5	8 8	¥			NA	· ×	NA.	NA	NA S	8	2 3	3 8	¥	ΝĀ	SC SC
NA	A		X	<b>X</b>	NA	Z.	\$19.99	\$9.03	\$12.61	\$135.80	\$222.87	\$65.84	\$103.76	\$124.84	\$165.53	\$2.46	\$165.21	\$75.83	\$0.3525	×	\$65.75	\$45.87	426 42	ZI		NA	¥ :	X 5	\$19.99	\$3.50	\$9.03	\$12.61	\$135.80	\$421.26	\$643.00	\$124.84	\$165.53	\$2.40	\$165.21	\$75.83	\$0.3525	NA	\$65.75	\$36.45	NA.			NA.	¥ ¥	¥	N	\$19.99	\$3.50	\$903	\$135.80	\$222.87	\$31.00	<b>TN</b>

	ENHANCED EXTENDED LINKS (EELs)	USOC	AL	P	GA	ΚY	LA	MS	NC	SC	ī
	DS1 Loop per month, Zone 2 (Note 1)	USLXX	¥	\$94.71	\$60.51	¥	\$96.73	N	ТВD	¥	NA.
	DS1 Loop per month, Zone 3 (Note 1)	USLXX	¥	\$208.93	\$96.18	¥	\$197.57	¥	TBD	¥	NA
	DS1 Loop per month, Zone 4 (Note 1)	NA	NA	NA	NA	NA	NA	NA	NA	NA	AN
	Interoffice Channel - Dedicated - DS1 - per mile per month	1L5XX	¥	\$0.6013	\$0.4523	¥	\$0.7831	N	\$0.5753	¥	¥
	Interoffice Channel - Dedicated - DS1 - Facility Termination per month	U1TF1	Ϋ́	\$99.79	\$78.47	NA	\$93.40	N	\$71.29	NA	¥
	Non-Recurring Charges - New EEL (Note 2) (Note 3)										
	NRC- DS1 interoffice - 1st	U1TF1	NA	\$171.01	\$142.64	NA	\$160.49	NA	\$217.17	NA	NA
	NRC- DS1 interoffice - Add'l	U1TF1	NA	\$101.84	\$94.87	NA	\$123.03	NA	\$163.75	NA	NA
	NRC-DS1 Local Loop - 1st	USLXX	¥	\$627.78	\$627.44	¥	\$502.73	N	\$714.84	¥	N.
	NRC-DS1 Local Loop - Add"	USLXX	Ą	\$377.43	\$231.49	Ą	\$293.92	Ā	\$421.47	Ā	M
	NRC-DS1 interoffice channel and DS1 Local Loop Combination - Electronic Svc Order	SOMEC	¥	\$3.17	\$3.50	NA	\$3.50	Ā	\$3.50	¥	\$3.50
	NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order, pe	SOMAN	¥	\$25.40	¥	NA	X	Ā	X	¥	\$19.9
Π	NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - 1	SOMAN	NA	ΝA	\$33.63	NA	\$242.20	NA A	\$65.01	A	NA.
	NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - A	SOMAN	¥	¥	\$27.49	Ā	\$153.37	Ā	\$50.83	¥	¥
Π	NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - D	SOMAN	A	¥	\$19.88	NA A	\$45.91	Ā	X	X	NA
	NRC-DS1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - D	SOMAN	NA	NA	\$11.85	NA	\$8.06	NA A	NA	Ą	¥
П											
	DS3 Interoffice Channel and DS3 Local Loop:										
	Recurring Charges										
	DS3 Loop per Facility Termination per month	UE3PX	¥	\$404.58	\$390.34	₹	\$669.01	Ā	\$387.01	¥	\$607.2
	DS3 Loop per mile	1L5ND	A	\$11.77	\$8.90	A	\$30.34	Ā	\$32.53	X	\$23.7
	Interoffice Channel - Dedicated - DS3 - FacilityTermination per month	U1TF3	NA	\$1,121.93	\$717.60	NA	\$1,101	NA A	\$720.38	X	\$760.2
	Interoffice Channel - Dedicated - DS3 - per mile per month	1L5XX	NA	\$4.17	\$6.46	NA	\$14.04	NA A	\$12.98	NA NA	\$5.89
	Non-Recurring Charges - New EEL (Note 2)(Note 3)										
	NRC- DS3 interoffice - 1st	U1TF3	¥	\$154.30	\$633.41	¥	\$713.57	¥	\$794.94	¥	\$729.2
	NRC- DS3 interoffice - Add'l	U1TF3	¥	\$77.50	\$449.91	¥	\$404.36	Ā	\$579.55	¥	\$411.9
Г	NRC-DS3 Local Loop - 1st	UE3PX	NA	\$1,020.45	\$761.81	NA	\$811.30	NA A	\$964.04	X	\$829.5

## NETWORK ELEMENTS AND OTHER SERVICES

NRC-STS-1 interoffice - Add1 NRC-STS-1 Local Loop - 1st NRC-STS-1 Local Loop - Add1 NRC-STS-1 interoffice channe	NRC-STS-1 int NRC-STS-1 Loo NRC-STS-1 Loo NRC-STS-1 inte	NRC-STS-1 intel NRC-STS-1 Loo NRC-STS-1 Loo NRC-STS-1 intel	NRC-STS-1 int NRC-STS-1 Loo NRC-STS-1 inte NRC-STS-1 inte NRC-STS-1 inte NRC-STS-1 inte NRC-STS-1 inte	NRC-STS-1 into  NRC-STS-1 Loc  NRC-STS-1 into  NRC-STS-1 into  NRC-STS-1 into  NRC-STS-1 into  NRC-STS-1 into	NRC-STS-1 into  NRC-STS-1 Loc  NRC-STS-1 into  NRC-STS-1 into  NRC-STS-1 into  NRC-STS-1 into	NRC- STS-1 into NRC-STS-1 Loc NRC-STS-1 Loc NRC-STS-1 into NRC-STS-1 into	NRC-STS-1 into NRC-STS-1 Log NRC-STS-1 Log NRC-STS-1 into	NRC-STS-1 into NRC-STS-1 Log NRC-STS-1 Log	NRC-STS-1 into	NRC- STS-1 into		NRC- STS-1 interoffice - 1st	Non-Recurring	Interoffice Chan	Interoffice Chan	STS-1 Loop per mile	STS-1 Loop per	Recurring Charges	STS-1 Interoffic	NRC-DS3 interc	NRC-DS3 Local Loop - Add	ENHANCED EXTE						
NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Electronic Svc Ol NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice Channel and STS-1 Local Loop Combination - Manual Svc Orde NRC-STS-1 interoffice Channel and DS1 Local Loop:	al Loub Asu I of Local Loop Combination - Electronic Svc O Inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde Inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde Inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde Inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde Inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde Inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde Inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde Inoffice channel Inoffice Cha	Jai Loup Avai  Jofflee channel and STS-1 Local Loop Combination - Electronic Svc O' Jrofflee channel and STS-1 Local Loop Combination - Manual Svc Orde Jrofflee channel and STS-1 Local Loop Combination - Manual Svc Orde Jrofflee channel and STS-1 Local Loop Combination - Manual Svc Orde Jrofflee channel and STS-1 Local Loop Combination - Manual Svc Orde Jrofflee channel and STS-1 Local Loop Combination - Manual Svc Orde Jrofflee channel and STS-1 Local Loop Combination - Manual Svc Orde	ar Loup: ANJ 1 rofflee channel and STS-1 Local Loop Combination - Electronic Svc O  rofflee channel and STS-1 Local Loop Combination - Manual Svc Orde  rofflee channel and STS-1 Local Loop Combination - Manual Svc Orde  rofflee channel and STS-1 Local Loop Combination - Manual Svc Orde  rofflee channel and STS-1 Local Loop Combination - Manual Svc Orde  rofflee channel and STS-1 Local Loop Combination - Manual Svc Orde	art Loup - Asu I indiffer and STS-1 Local Loop Combination - Electronic Svc O inoffice channel and STS-1 Local Loop Combination - Manual Svc Orde groffice channel and STS-1 Local Loop Combination - Manual Svc Orde anoffice channel and STS-1 Local Loop Combination - Manual Svc Orde anoffice channel and STS-1 Local Loop Combination - Manual Svc Orde	al Loop - NALI  yoffice channel and STS-1 Local Loop Combination - Electroric Svc Oracle channel and STS-1 Local Loop Combination - Manual Svc Orde  roffice channel and STS-1 Local Loop Combination - Manual Svc Orde	and Loup - Auguration - Auguration - Electronic Svc Ora jorffice channel and STS-1 Local Loop Combination - Electronic Svc Orace aroffice channel and STS-1 Local Loop Combination - Manual Svc Orace	eroffice channel and STS-1 Local Loop Combination - Electronic Svc O	Jai Luup - Aug i	vall oop Addi	pal Loop - 1st	eroffice - Add'l	eroffice - 1st	Non-Recurring Charges - New EEL (Note 2)(Note 3)	nteroffice Channel - Dedicated - STS-1 - per mile per month	Interoffice Channel - Dedicated - STS-1 - FacilityTermination per month	mile	STS-1 Loop per Facility Termination per month	rges	STS-1 Interoffice Channel and STS-1 Local Loop:	NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order - 🕩	NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order - 🛡	NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order - A	NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order - 1	NRC-DS3 interoffice channel and DS3 Local Loop Combination - Manual Svc Order, pe	NRC-DS3 interoffice channel and DS3 Local Loop Combination - Electronic Svc Order	Loop - Add'l	ENHANCED EXTENDED LINKS (EELs)	
UDLS1 UDLS1 SOMEC SOMAN SOMAN SOMAN SOMAN							7	UDLS1 UDLS1	UDLS1		U1TFS	U1TFS		1L5XX	U1TFS	1L5ND	UDLS1				SOMAN	4 SOMAN			SOMEC	UE3PX	JOSU	
NA NA	X X X 3	X	¥ ¥ §	N S	5	<u></u>	NA	NA	A	NA	A	NA.		W	A	NA	NA			۸N	NA	NA	A	AN	¥	AN	ΑL	
			₹	¥	NA	N	\$25.40	\$3.17	\$513.74	\$1,020.45	\$77.50	\$154.30		\$4.17	\$1,105.98	\$11.77	\$446.09			NA	NA	NA	N	\$25.40	\$3.17	\$513.74	P	
			\$18.23	\$18.23	\$37.96	\$37.96	NA	\$3.50	\$545.54	\$761.81	\$449.91	\$633.41		\$2.72	\$788.00	\$8.90	\$421.59			\$18.03	\$18.03	\$37.55	\$37.55	NA	\$3.50	\$545.54	GA	
			NA	¥	Ä	A	NA	NA	NA	NA	N	NA		NA	N	NA	¥			AN	NA	NA.	A	NA	¥	NA	ĸγ	
			\$41.88	\$41.88	\$100.50	\$100.50	NA	\$3.50	\$502.09	\$811.30	\$404.36	\$713.57		\$14.04	\$1,101	\$38.98	\$497.08			\$41.88	\$41.88	\$100.50	\$100.50	NA	\$3.50	\$502.09	5	
			NA	Ā	NA	NA	NA	NA	NA	NA	NA A	NA		NA	NA A	NA	NA			AN	NA	NA	NA	NA	Ä	NA	SM	
	_		N	¥	\$84.76	\$123.62	NA	\$3.50	\$542.73	\$964.04	\$436.36	\$624.86		\$6.29	\$800.94	\$32.53	\$387.01			AN	NA	\$69.01	\$83.19	NA	\$3.50	\$542.73	NC	
			NA.	Ā	NA	ΝĀ	NA	NA	ΝĀ	NA	Ϋ́	NA		NA	Ϋ́	NA	NA			AN	NA	ΝĀ	NA	NA	Š	NA	SC	
			¥.	¥	NA	¥	\$19.99	\$3.50	\$512.23	\$829.52	\$625.84	\$961.62		\$6.88	\$838.65	\$30.53	\$400.21			AN	NA	X.	¥	\$19.99	\$3.50	\$512.23	N	

							I																									F														1				I	Ξ	Ι	$\exists$
NRC.2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual S) NRC.2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual S) NRC.2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual S)	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual St NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Manual St	NRC-2-wire VG interoffice channel and 2-wire VG Local Loop Combination - Electronic	NRC-2-wire VG Local Loop - 1st	NRC- 2-wire VG interoffice - Add'l	NRC- 2-wire VG interoffice - 1st	Interoffice Channel - Dedicated - 2-wire VG - per mile per month  Non-Recurring Charges - New EEL (Note 2)(Note 3)	Interoffice Channel - Dedicated - 2-wire VG - FacilityTermination per month	2-wire VG Loop per month, Zone 4 (Note 1)	2-wire VG Loop per month, Zone 3 (Note 1)	2-wire VG Loop per month, Zone 2 (Note 1)	2-wire VG Loop per month, statewide	Recurring Charges	2-wire VG Interoffice Channel and 2-wire VG Local Loop:	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Manual Svc Order,	NRC-STS-1 interoffice channel and DS1 Local Loop Combination - Electronic Svc Orde	NRC-DS1(COCI)interface card -Add'l	NRC-DS1(COCI) interface card -1st	NRC-D83 Channelization System - 1st	NRC- STS-1 interoffice - Add'l	NRC- STS-1 interoffice - 1st	NRC-DS1 Local Loop - Add'l	Non-Recurring Charges - New EEL (Note 2)(Note 3)  NRC-DS1 Local Loop - 1st	DS3 Interface per month (DS1 COCI)	DS3 Channelized System per month	Interoffice Channel - Dedicated - STS-1 - Facility Lemmation per month	Escilit-Termination per	Loop per month, Zone 3 (Note	DS1 Loop per month, Zone 2 (Note 1)	Zono 4 (Noto	STS-1 Interoffice Channel and DS1 Local Loop:	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - E	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - I	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - A	NRC-DS3 interoffice channel and DS1 Local Loop Combination - Manual Svc Order - 1	nic S	NRC-DS1(COCI)interface card -Add'l	NRC-DS3 Criatineitzation System - Addi	NRC-D3 Channelization System - Ist	NRC-DS1 Local Loop - Add"	NRC-DS1 Local Loop - 1st	NRC- DS3 interoffice - Add'l	NRC- DS3 interoffice - 1st	Non-Recurring Charges - New EEL (Note 2)(Note 3)	DS3 Channelized System per month	Interoffice Channel - Dedicated - DS3 - per mile per month	Interoffice Channel - Dedicated - DS3 - FacilityTermination per month	DS1 Loop per month, Zone 3 (Note 1)	IDS1 Loop per month, Zone Z (Note 1)	DS1 Loop per month, Zone 1 (Note 1)	ENHANCED EXTENDED LINKS (EELS)
SOMAN	1		UEAL2	U1TV2	U1TV2	1L5XX	U1TV2	AN	UEAL2	UEAL2	UEAL2			SOMAN	SOMAN			SOMEC		UC1D1	MOG	U1TFS	U1TFS	USLXX	USLXX	UC1D1	MQ3	11.577	NA	USLXX	USLXX	5		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	UC1D1	UC1D1	MOG	USLXX	USLXX	U1TF3	U1TF3	UCTUT	MQ3	1L5XX	U1TF3	NA NA	I SE XX	USLXX	USOC
NA NA	NA NA	NA.	NA NA	AN	NA	NA	NA.	AN	NA	N K	¥			NA	AN	¥ §	N N	AN AN	NA	¥ ₹	NA NA	¥	AN	N.	NA	AN	NA S	NA NA	NA A	NA :	Z Z	<u> </u>		NA	¥	NA :	¥ ₹	N X	NA	AN	NA NA	¥ ¥	AN	AN	¥	NA	N N	AN	NA :	X §	N A	N A	AL
X	\$25.40 NA	\$3.17	\$38.02	\$67.61	\$112.10	\$0.01	\$26.52	AN	TBD	TBD	\$20.52			NA	NA	X 3	\$25.40	\$3.17	\$8.77	\$12.16	\$404.85	\$77.50	\$154.30	\$377.43	\$627.78	\$14.40	\$220.97	\$1,105.98	NA 105 00	\$208.93	\$94.71	200		Ä	¥	NA :	NA O+.	\$3.17	\$8.77	\$12.16	\$168.26	\$124.84	\$338.52	\$377.43	\$627.78	\$14.40	\$220.97	\$4.17	\$1,121.93	NA 9200.93	\$208 93	\$64.69	7
\$27.36 NA NA	\$37.88	\$3.50	\$104.17	\$36.08	\$79.61	\$0.0222	\$17.07	NA	\$28.26	\$17.78	\$16.51			\$11.85	\$19.88	\$27.49	\$33 63	\$3.50	\$8.66	\$12.02	\$316.28	\$94.87	\$142.64	\$282.38	\$520.09	\$0.67	\$202.91	\$6.46	NA 717.60	\$96.18	\$60.51	950		\$11.85	\$19.88	\$27.49	NA O	\$3.50	\$8.66	\$12.02	\$171.73	\$94.87	\$142.64	\$282.38	\$520.09	\$0.67	\$202.91	\$6.46	\$717.60	NA . 10	\$96.18	\$52.40	GA
N N N	NA NA	N.	X X	¥	NA	X	X.	AN	NA	¥ ₹	₹			NA	NA	¥ ₹	2 3	₹ ₹	NA	<b>₹</b> §	N N	₹	NA	¥.	¥	AN	NA S	NA A	N N	¥.	¥ ¥	N		X	₹	NA :	₹ ₹	¥	NA	N S	NA KA	¥	¥	AN	¥	¥	¥	¥	NA :	¥ ₹	2 2	N A	3
\$26.20 \$19.47 \$8.06	\$36.28	\$3.50	\$128.42	\$39.91	\$104.23	\$0.0384	\$19.10	NA	\$61.93	\$30.32	\$19.35			\$29.00	\$50.49	\$58.31	\$68.30	\$3.50	\$8.80	\$12.29	\$320.72	\$404.36	\$713.57	\$293.92	\$502.73	\$7.55	\$245.84	\$1,101	e NA	\$197.57	\$96.73	9000		\$8.06	\$19.47	\$26.20	\$36.28	\$3.50	\$8.80	\$12.29	\$320.72	\$293.92	\$502.73	\$404.36	\$713.57	\$7.55	\$245.84	\$14.04	\$1,101	NA.	\$107.57	\$56.32	5
X X X	NA NA	N.	N N	NA NA	NA	X	¥	NA	NA	X S	<b>X</b>			NA	NA	¥ §	N N	X X	NA	<b>₹</b>	NA NA	<b>X</b>	NA	¥	¥	NA	NA S	NA A	N A	W.	× ×	NA		NA	¥	NA :	<b>X</b> 5	X X	NA	N S	NA A	X X	NA NA	NA	¥	¥	¥	Ą	NA :	<b>₹</b>	N N	N X	MS
XXX	X X	X S	Z Z	NA	NA	×	X.	NA	NA	<b>₹</b>	<b>.</b> ×			NA	NA	\$67.76	\$81 QA	\$3.50	\$11.28	\$15.76	\$351.95	\$436.36	\$624.86	\$421.47	\$714.84	\$4.61	\$226.81	\$307.01	NA NA	TBD	TB G	=		×	N <sub>A</sub>	\$104.02	\$118.20	\$3.50	\$11.28	\$15.76	\$373.76	\$421.47	\$714.84	\$579.55	\$794.94	\$4.61	\$226.81	\$12.98	\$720.38	¥ ē	룅룡	曹	No
N N	X X	¥.	X X	NA.	NA	×	NA.	NA	NA	<b>₹</b>	¥			NA.	NA	¥ §	N N	<b>S S</b>	NA	<b>₹</b> 5	N N	¥	NA	N.	N.	NA	N S	NA NA	N N	NA.	₹ ₹			×	NA.	NA :	<b>₹</b>	× ×	NA	N S	NA NA	<b>.</b>	NA.	NA	¥	¥	<b>X</b>	NA.	NA :	¥ ₹	N E	N A	SC
X X X	\$19.99 NA	\$3.50	\$192.97 \$140.72	\$20.88	\$83.35	\$0.0173	\$18.33	NA	\$28.02	\$19.55	\$18.00			NA	NA	<b>₹</b>	Z Z	¥	NA.	<b>₹</b> 5	NA KA	<b>X</b>	NA	¥	¥	Ą	N S	NA A	N A	X.	<b>₹</b>	N		×	<b>.</b>	NA :	<b>₹</b>	X X	NA	X §	Z A	<b>X X</b>	Ą	NA	¥	¥	<b>X X</b>	NA	NA :	<b>₹</b>	N Z	N N	ī

## NETWORK ELEMENTS ND OTHER SERVICES

E	H	+	F			1		7			H			-										H					1	F			1											F	F						H	1	Ŧ	т
NRC - 1st	NRC - Ordinarily Combined in GA (Note 5)	Zone 3	Zone 2	Zone 1	2-Wire Analog Voice Grade Loop -per month	2-Wire Anabo Voice Grade Loop - per mile per month	2.Wire Analog Voice Grade I con - Service I evel 2	Network Elements used in Existing Combinations at UNE Rates (Note4)	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - N	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - N	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - N	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - M	NRC-4-wire 64kbps interoffice channel and 4-wire 64kbps Local Loop Combination - Iv	NRC-4-wire 64kbps Local Loop - Add'l	NRC-4-wire 64kbps Local Loop - 1st	NRC- 4-wire 64kbps interoffice - Add'l	NRC- 4-wire 64kbps interoffice - 1st	Interoffice Channel - Dedicated - 4-wire 64kbps - per mile per month	Interoffice Channel - Dedicated - 4-wire 64kbps - FacilityTermination per month	4-wire 64kbps Loop per month, Zone 4 (Note 1)	4-wire 64kbps Loop per month, Zone 2 (Note 1)	per month	4-wire 64 kbps Interoffice Channel and 4-wire 64 kbps Local Loop:  Recurring Charges		NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - IV	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - N	NRC-4-wire 56kbps interoffice channel and 4-wire 56kbps Local Loop Combination - E	NRC-4-wire 56kbps Local Loop - 1st	NRC- 4-wire 56kbps interoffice - Add'l	NRC- 4-wire 56kbps interoffice - 1st	Interoffice Channel - Dedicated - 4-wire 56kbps - per mile per month	- Dedicated - 4-wire 56kbps		4-wire 56kbps Loop per month, Zone 3 (Note 1)	per month, Zone 1	4-wire 56 kbps Interoffice Channel and 4-wire 56kbps Local Loop:  Recurring Charges	NKC-4-WIRE ARTHURIONICE CHAINELAND 4-WIRE ART FOOD COMMUNICINE. MANING CHAINE ARTHURIST CONTRIBUTION - MANING CHAINE CHAINE ARTHURIST CONTRIBUTION - MANING CHAINE CHAINE - MANING	VG interoffice channel and 4-wire VG Local Loop Combination -	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Manual S	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Manual Si	NRC-4-wire VG interoffice channel and 4-wire VG Local Loop Combination - Electronic	NRC-4-wire VG Local Loop - Add'l	NRC-4-wire VG Local Loop - 1st	NRC- 4-wire VG interoffice - Add'l	Non-Recurring Charges - New EEL (Note 2)(Note 3)	er month	Interoffice Channel - Dedicated - 4-wire VG - FacilityTermination per month	4-wire VG Loop per month, Zone 3 (Note 1) 4-wire VG Loop per month. Zone 4 (Note 1)	month, Zone 2	4-wire VG Loop per month, Zone 1 (Note 1)	4-wire VG Interoffice Channel and 4-wire VG Local Loop:	ENHANCED EXTENDED LINKS (EELS)
UEAL2	C I	UEAL2	UEAL2	UEAL2	UEAL2	UNCVX		USOC	SOMAN	SOMAN			SOMEC		UDL64	U1TD6	U1TD6	1L5XX	U1TD6	NA	UDL64	UDL64			SOMAN				SOMEC	UDL56	U1TD5	U1TD5	1L5XX	U1TD5	NA	UDL56	UDL56		OCWA	SOMAN			SOMAN		UEAL4	U1TV4	INTW	1L5XX	U1TV4	NA L4	UEAL4	UEAL4		USOC
NA		\$52.84 NA	\$29.16	\$17.95	NA	1L5ND		AL	NA	NA NA	NA	¥ 3	Z Z	NA.	NA	NA	NA M	X	NA	NA S	3 8	NA.			NA S	N N	X.	NA	¥ ₹	× ×	NA	¥	NA	NA.	NA.	X 3	<b>.</b>		5	<b>₹</b>	NA	Z :	NA NA	: <sub>K</sub>	NA	NA :	NA	NA.	NA	X X	X X	¥		A
NA		TBD	TBD	\$20.52	N.	\$0.00		Ð	×	Ä	AN	NA STOCK	\$3.17	\$148.55	\$329.76	\$73.44	\$160.33	\$0.0098	\$19.31	NA NA	\$57.21	\$39.08			NA S	Z Z	<b>X</b>	\$25.40	\$3.17	\$329.76	\$73.44	\$160.33	\$0.0098	\$23.64		\$126.22	\$39.08		3	<b>₹</b>	NA A	NA S	\$3.17	\$148.55	\$329.76	\$73.44	\$160 33	\$0.0098	\$23.64	\$/8.35	\$35.51	\$24.26		2
\$104.17		\$28.26	\$17.78	\$15.40	NA.	\$0.00		GA	\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	\$241.20	\$348.55	\$36.08	\$79.61	\$0.0222	\$16.45	NA S	\$30.53	\$26.44			NA S	\$27.3b	\$37.88	NA	\$3.50	\$348.55	\$36.08	\$79.61	\$0.0222	\$16.45	NA.	\$46.53	\$26.44		N.	<b>X</b>	NA	¥ ;	Z Z	: ≰	NA.	X.	NA	×	NA	<b>X X</b>	X X	¥		GA
NA		\$55.78	\$32.32	\$17.27	Ą	\$0.00		ζ.	Ą	¥	AN	¥ 5	NA NA	NA.	¥	NA :	NA A	¥	NA.	NA S	S 8	NA.			NA S	N K	¥	NA	X §	S A	Ā	¥	NA	NA.	NA.	8	<b>X X</b>		3	¥ ¥	NA	N.	N X	: ₹	NA.	X.	N	NA	NA	8	×	¥		ΚY
NA		\$61.93	\$30.32	\$17.65	Ą	\$0.00		LA	\$41.88	\$41.88	\$100.50	\$100.50	\$3.50	\$502.09	\$811.30	\$404.36	\$713.57	\$0.0384	\$18.37	NA NA	\$47.24	\$27.50			NA .	\$25.20	\$36.28	NA	\$3.50	\$421.27	\$39.91	\$104.23	\$0.0384	\$18.37	NA	\$86.47	\$24.36		100	<b>¥ ¥</b>	NA	X.	N N	: ₹	¥	X.	NA	ΝA	NA	X X	X X	¥		F
NA	# 10.00	\$34.77	\$24.33	\$18.35	¥	\$0.00		MS	Ą	×	AN	¥ §	NA NA	×	NA.	NA.	NA A	¥	NA.	NA S	N K	NA.			NA S	N K	¥	NA	¥ ₹	N N	NA AN	¥	NA	NA.	NA.	¥ ₹	¥ ¥		5	¥ ¥	NA	¥.	NA KA	: ⊀	NA.	¥.	NA	ΝA	NA	<b>X X</b>	X X	¥		NS
NA		NA	TBD	TBD	\$19.50	\$0.00		NC	NA	¥	\$50.83	\$65.01	\$3.50	\$337.51	\$489.04	\$52.58	\$137.48	\$0.0282	\$17.40	NA C	3 8	TBD			NA S	\$50.83	\$65.01	NA	\$3.50	\$489.04	\$52.58	\$137.48	\$0.0282	\$17.40	NA	TBD E	曹		3	<b>₹</b>	NA	¥.	N X	: 🛚	A	NA :	N	¥	NA	8	X X	NA		NC
NA		\$31.87	\$27.59	\$16.58	NA	\$0.00		SC	NA	¥	NA	¥ 5	N X	¥	NA.	NA :	NA NA	¥	NA	NA S	X X	¥			NA S	N K	¥.	NA	<b>₹</b>	X X	NA.	₹	NA	¥	NA	<b>₹</b>	S &		3	₹ ₹	NA	¥ ;	N X	: ¥	¥	N :	NA	¥	NA	<b>X X</b>	X X	¥		SC
NA		\$28.02	\$19.55	\$15.54	¥	\$0.00		Ź	¥	<b>X</b>	NA.	N S	\$19.50	\$512.23	\$829.52	\$411.98	\$729.27	\$0.1730	\$17.74	NA S	\$45.87	\$36.45			N S	8	¥	\$19.99	\$3.50	\$643.00	\$20.88	\$83.35	\$0.1730	\$17.74	NA.	\$27.18	\$15.92		5	X X	NA	¥ ;	N N	: ≰	NA	¥.	NA	¥	NA	<b>X X</b>	<b>X</b>	¥		T

E			-	-												1								I							I																			Ŧ	F		Ŧ			H	Ŧ	Ŧ	F	<u> </u>
NRC - Incremental Charge - Manual Service Order - Add'l		NRC - Electronic Svc Order, per LSR	+   -	7	NRC - 1st	NRC - Ordinarily Combined in GA (Note 5)	Zone 4	Zone 3	Zone 1	2-Wire ISDN Digital Grade Loop per month	2-Wire ISDN Loop per mile per month	2-Wire ISDN Digital Grade Loop	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - Incremental Charge - Manual Service Order - Addi	NRC - Incremental Charge - Manual Service Order - 1st	<ul> <li>Electronic Svc Order, per LSR</li> </ul>	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	Zone 4	Zone 3	Zone 2	Zone 1	4-Wire Analog Voice Grade Loop per mile per month	4-Wire Analog Voice Grade Loop	NKC-24-WIRE COMBINATION - SWIGH AS IS CONVEISION CHARGE - DISCONNECT - Addi	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Electronic Syc Order per I SB		NRC - Add'l	NRC - 1st	NRC - Ordinarily Combined in GA (Note 5)	Zone 4	Zone 2	Zone 1	2-Wire Analog Voice Grade Loop - Rev Bat - per month	2-Wire Analog Voice Grade Loop - Loopf Start - per mile per month	2-Wire Analog Voice Grade Loop - Service Level 2 - (reverse battery)	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC - 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"I	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)  NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC - Incremental Charge - Manual Service Order - Disconnect		NRC - Incremental Charge - Manual Service Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - Disconnect Charge - 1st	NRC - Add'l	ENHANCED EXTENDED LINKS (EELs)
SOMAN	SOMAN	SOMEC	01127	0122	U1L2X		U1L2X	111 2X	U1L2X	U1L2X	1L5ND		UNCCC	UNCCC	UNCCC		OCIVIAIN	SOMAN	SOMAN	SOMEC	UEAL4	UEAL4	UEAL4	I JEAI 4	UEAL4	UEAL4	UEAL4	UEAL4	1L5ND			UNCCC	UNCCC	UNCCC		SOMAN	SOMAN	SOME	OFAR2	UEAR2	UEAR2	UEAR2	C C	UTIAR2	UE ARZ	UEAR2	UEAR2	1L5ND		UNCCC	UNCCC	UNCCC	LINCCC	SOMAN	SOMAN	SOMAN	SOMEC	UEAL2	UEAL2	USOC
NA	NA	<b>₹</b>	NA NA	5	Z X		NA CO	\$68.38	\$23.23	NA NA	\$0.00		\$0.00	\$0.00	\$32.11	\$54.03	Š	\$	NA NA	NA NA	N.	NA	¥.	NA	Ā	\$70.67	\$39.00	\$24.01	\$0.00		\$0.00	\$0.00	\$32.11	\$54.03		AN	X S	N N	Z Z	X X	NA.	NA	100	40.2C¢	\$29.16	\$17.95	NA	\$0.00		\$0.00	\$0.00	\$32.11	\$54.03	X	NA	NA S	N N	; <sub>K</sub>	: NA	AL
A	NA.	<b>₹</b>	3	5	N N		¥.	\$104.35	\$32.34	N N	\$0.00		\$16.77	\$16.77	\$32.75	\$32 75	5	<b>X</b>	¥	¥	¥	NA.	¥ ;	NA	¥	\$78.35	\$35.51	\$24.26	\$0.00		\$10.77	\$16.77	\$32.75	\$32.75		¥	X S	Z S	N N	<b>X</b>	¥	NA.		AN AN	\$22.43	\$23.23	\$20.52	\$0.00		\$16.77	\$16.77	\$32.75	\$32.75	¥	¥	NA S	N S	: X	: <sub>K</sub>	2
\$8.42	\$18.94	\$3.50	2 2	\$100.33	\$233.38		NA.	\$40.17	\$21.89	NA NA	\$0.00		\$12.61	\$12.61	\$26.99	\$58.43	Š	\$8.42	NA	\$3.50	N.	NA	\$170.57	\$206.95	×	\$41.99	\$26.42	\$22.88	\$0.00		\$12.01	\$12.61	\$26.99	\$58.43		NA.	\$8.42	\$18.94	63 50 NA	X X	\$78.10	\$104.17		AN 26.05	\$19.45	\$16.84	¥.	\$0.00		\$12.61	\$12.61	\$26.99	\$58.43	¥	\$8.42	\$18.94	\$3.50	<b>X X</b>	\$78.10	GA
NA	NA	<b>₹</b> §	Z Z	5	S A		N C	\$76.42	\$23.66	NA NA	\$0.00		\$0.00	\$0.00	\$32.16	\$54.09	3	3	<b>X</b>	¥	¥	NA	<b>₹</b>	NA	X	\$67.56	\$39.14	\$20.92	\$0.00		φ0.00	\$0.00	\$32.16	\$54.09		¥	¥ ;	2 3	N N	<b>.</b> ×	N <sub>A</sub>	NA	100	NA O	\$32.32	\$17.27	NA.	\$0.00		\$0.00	\$0.00	\$32.16	\$54.09	X	NA	NA S	Z Z	: K	: N	ΚY
NA	NA	<b>₹</b>	N N	5	<b>X X</b>		N :	\$74.19	\$21.15	Ž NA	\$0.00		\$0.00	\$0.00	\$32.24	\$54 23	3	<b>X</b>	X.	N.	N <sub>A</sub>	NA	<b>X</b> :	NA	×	\$85.47	\$41.85	\$24.36	\$0.00		<b>\$0.00</b>	\$0.00	\$32.24	\$54.23		NA NA	X.	2 3	Z Z	X X	NA.	NA		AN AN	\$30.32	\$17.65	NA	\$0.00		\$0.00	\$0.00	\$32.24	\$54.23	X	NA	NA S	N N	; <sub>K</sub>	: N	F
NA	NA	<b>₹</b>	N N	5	<b>X X</b>		\$54.64	\$41.40	\$21.86	NA NA	\$0.00		\$0.00	\$0.00	\$32.16	\$54.09	3	<b>X</b>	X.	X.	N <sub>A</sub>	NA	<b>X</b> :	NA	\$55.96	\$42.40	\$29.67	\$22.38	\$0.00		<b>\$0.00</b>	\$0.00	\$32.16	\$54.09		NA NA	X.	2 3	Z Z	X X	NA.	NA	610.00	\$45.88	\$24.33	\$18.35	NA	\$0.00		\$0.00	\$0.00	\$32.16	\$54.09	X	NA	NA S	N N	; <sub>K</sub>	: N	MS
N <sub>N</sub>	NA	<b>₹</b>	Z Z	5	<b>X X</b>		NA C	<del>-</del>		\$24.98	\$0.00		\$0.00	\$0.00	\$32.10	\$54.00	3	\$	NA NA	¥	¥.	NA	¥ ;	NA	NA.	\$44.38	\$38.53	\$23.51	\$0.00		\$0.00	\$0.00	\$32.10	\$54.00		NA A	X S	Z 3	Z Z	X X	NA	NA		N 0		ТВD	\$19.50	\$0.00		\$0.00	\$0.00	\$32.10	\$54.00	X	¥	N S	N S	; <sub>K</sub>	: K	NC
N.	NA	<b>₹</b>	Z Z	5	<b>X X</b>		NA.	\$40.17	\$21.40	NA	\$0.00		\$0.00	\$0.00	\$32.25	\$54.26	3	\$	NA NA	¥	¥.	NA	¥ ;	NA	NA.	\$58.85	\$44.44	\$49.47	\$0.00		\$0.00	\$0.00	\$32.25	\$54.26		NA NA	X S	2 3	Z Z	Z A	NA	NA		\$43.00	\$32.53	\$21.57	NA.	\$0.00		\$0.00	\$0.00	\$32.25	\$54.26	X	¥	N S	N S	; <sub>K</sub>	: K	SC
ΝA	NA	Z 5	3	5	N N		NA SEC.	\$28.02	\$15.54	NA NA	\$0.00		\$0.00	\$0.00	\$32.17	\$54 13	NA	<b>X</b>	NA	¥	¥	¥	¥ ;	NA	×	\$28.02	\$19.55	\$15.54	\$0.00		\$0.00	\$0.00	\$32.17	\$54.13		¥	¥ ;	Z S	8	×	¥	NA		\$20.02	\$19.55	\$15.54	NA.	\$0.00		\$0.00	\$0.00	\$32.17	\$54.13	×	NA.	NA S	N S	š Ş	: <sub>K</sub>	TN

## NETWORK ELEMENTS AND OTHER SERVICES

															H																																Ŧ	Ŧ	Ħ	Ŧ			F			m
NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Electronic Ovo Order, per Lox	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'I	NRC - 1st	NRC - Ordinarily Combined in GA (Note 5)	Zone /	Zone 2	Zone 1	4-Wire DS1 Digital Loop per month	4-Wire DS1 Digital Loop per mile per month	A.Wire DS4 Digital Dop	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'I	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - Disconnect Charge - Add'l		NRC - Add"	NRC - Ordinarily Combined in GA (Note 5)	Zone 4	Zone 3	Zone 1	4-Wire 64 kbps Digital Grade Loop per month	4-Wire 64 kbps Digital Grade Loop per mile per month	4-Wire 64 kbps Digital Grade Loop	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Addi	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC - Incremental Charge - Manual Service Order - Disconnect  NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - Incremental Charge - Manual Service Order - Add'i	NRC - Incremental Charge - Manual Service Order - 1st	<ul> <li>Electronic Svc Order</li> </ul>	NRC - Disconnect Charge - Add'l	NRC - Add'l	NRC - 1st	NRC - Ordinarily Combined in GA (Note 5)	Zone 4	Zone 3	Zone 1	4-Wire 56 kbps Digital Grade Loop per month	4-Wire 56 kbps Digital Grade Loop	a o convolución cial de maconinon	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC - Incremental Charge - Manual Service Order - Disconnect	ENHANCED EXTENDED LINKS (EELS)
UNCCC	UNCCC		SOMAN	SOMAN	COMPC	USLXX	USLXX	USLXX	USLXX	0000		USLXX	USLXX	USLXX	1L5ND		UNCCC		UNCCC		SOMAN	SOMAN	SOMEC	UDL64	UDL64	UDL64	2	UDL64	UDL64	UDL64	UDL64	1L5ND		UNCCC	UNCCC	UNCCC	UNCCC	SOMAN	SOMAN	SOMAN	SOMEC	UDL56	UDL56	UDL56		UDL56	UDL56	UDL56	UDL56	בֿאַת	0.000	UNCCC	UNCCC	UNCCC	SOMAN	USOC
\$0.00	\$54.03 \$32.11		X.	N S	N A	X X	¥	AN	Ą	3	VIN 67.701 &	\$84.05	\$51.74	AN	\$0.00		\$0.00	\$0.00	\$54.03		¥ 5	AN AN	N N	NA	¥.	AN AN		NA	\$80.45	\$27.33	AN AN	\$0.00		\$0.00	\$0.00	\$32.11	\$54.03	AN	AN	¥	AN	¥.	AN AN	AN		AN G-300	\$80.45	\$27.33	NA NA	\$0.00	<b>\$0.00</b>	00.08	\$32.11	\$54.03	NA	ΑL
\$16.77	\$32.75 \$32.75		N :	N S	NA	X X	¥	A	¥	5	\$200.50	\$94.71	\$64.69	NA.	\$0.00		\$16.77	\$16.77	\$32.75		<b>₹</b>	S &	¥	NA	¥.	N N		NA	\$50.85	\$33.90	NA	\$0.00		\$16.77	\$16.77	\$32.75	\$32.75	Ä	3	¥	A	₹ 5	8 8	¥		NA S	\$50.85	\$33.90	NA S	3	•	\$16.77	\$32.75	\$32.75	A	<b>P</b>
\$12.61	\$58.43 \$26.99		N.	\$8.42	\$3.50	Ž X	₹	\$268.18	\$429.98	Š	ψ90.10	\$60.51	\$52.40	N.	\$0.00		\$12.61	\$12.61	\$58.43		NA.	\$18.94	\$3.50	NA		\$241.20	9	NA	\$46.53	\$26.44	- N	\$0.00		\$12.61	\$12.61	\$26.99	\$58.43	K	\$8.42	\$18.94	\$3.50	¥ ₹	\$241.20	\$348.55		NA S	\$46.53	\$26.44	NA O	2000		\$12.61	\$26.99	\$58.43	X	GA
\$0.00	\$54.09 \$32.16		¥ ;	N S	NA NA	<b>X</b>	×.	NA	¥	5	\$10Z.34	\$162.34	\$50.28	NA	\$0.00		\$0.00	\$0.00	\$54.09		<b>₹</b>	N N	×.	NA	¥ :	N N		NA	<b>₹</b> §	× ×	×	\$0.00		\$0.00	\$0.00	\$32.16	\$54.09	X	Z X	×.	NA	X S	N N	NA		X S	NA NA	S A	NA O	e 00	0.00	\$0.00	\$32.16	\$54.09	NA	<b>Σ</b>
\$0.00	\$54.23 \$32.24		X :	N S	- A	X X	. ₹	NA.	¥	5	ψI Θ Ι . Ο Ι	\$96.73	\$56.32	N.	\$0.00		\$0.00	\$0.00	\$54.23		<b>₹</b> 5	N N	× ×	NA.	<b>∑</b>	N X	:	NA	\$96.48	\$27.50	NA NA	\$0.00		\$0.00	\$0.00	\$32.24	\$54.23	VA.	<b>S S</b>	. ₹	NA.	¥ 5	X X	NA.		NA S	\$96.48	\$27.50	NA O	5 3	0.00	\$0.00	\$32.24	\$54.23	×	<b>F</b>
\$0.00	\$54.09 \$32.16		¥ ;	N S	NA NA	<b>X</b>	×.	NA	¥	\$127.47	\$107.47	307.58	\$50.99	NA	\$0.00		\$0.00	\$0.00	\$54.09		<b>₹</b>	N N	×.	NA	¥ :	N N		\$64.02	\$48.51	\$25.61	NA NA	\$0.00		\$0.00	\$0.00	\$32.16	\$54.09	X	3 8	×.	NA	X S	N N	NA		\$64.02	\$48.51	\$25.61	NA S	50	#0.00	\$0.00	\$32.16	\$54.09	NA	NS
\$0.00	\$54.00 \$32.10		<b>X</b> :	Z §	NA NA	<b>X</b>	N <sub>A</sub>	NA	¥	\$	× 5		TBD	\$62.78	\$0.00		\$0.00	\$0.00	\$54.00		X 5	N X	N N	NA	¥ :	N S		NA	TBO GO		\$32.67	\$0.00		\$0.00	\$0.00	\$32.10	\$54.00	NA	Z N	N <sub>A</sub>	NA	X S	S S	NA		¥ 5	큠팅	# IBD	\$32.67	5 00	60.00	\$0.00	\$32.10	\$54.00	A	N <sub>C</sub>
\$0.00	\$54.26 \$32.25		<b>X</b> :	Z §	N N	<b>X</b>	N <sub>A</sub>	NA.	¥	3	\$104.30	\$89.93	\$53.00	NA	\$0.00		\$0.00	\$0.00	\$54.26		X 5	N X	N N	NA	¥ :	N S		NA	\$52.76	\$27.94	NA	\$0.00		\$0.00	\$0.00	\$32.25	\$54.26	NA	Z N	N <sub>A</sub>	NA.	X S	S S	NA		NA S	\$45.81	\$27.94	NA O	5 00	60.00	\$0.00	\$32.25	\$54.26	A	SC
\$0.00	\$54.13 \$32.17		<b>₹</b>	<b>₹</b>	Z A	<b>X</b>	¥	NA	₹	3	3	N A	X X	¥	\$0.00		\$0.00	\$0.00	\$54.13		¥ ₹	N N	N.	NA	<b>⊼</b> :	N S		NA	\$65.75	\$36.45	NA NA	\$0.00		\$0.00	\$0.00	\$32.17	\$54.13	Ä	\$ \$	¥	NA	<b>⊼</b> §	<b>X X</b>	NA		NA S	\$45.87	\$36.45	NA \$0.00	5000	0.00	\$0.00	\$32.17	\$54.13	¥	ī

H					Ŧ		F			1		1				F	F							F		H		1			F		F		H		1		F						1		F					H	H	7	Ŧ	Ŧ					7
NRC - OC12 - Facility Termination - Add'l	NRC - Ordinarily Combined in GA (Note 5)	facility termination per month	OC-12 Local Loop	INVO. OCO COMBINALI DIR. ORIGINALI SI SCHIBBI CHARGE DISCUII BOL. MALI	NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'i	NRC - OC3 - Incremental Cost - Manual Svc Order vs Elect-Disconnect-1st	NRC - OC3 - Incremental ChargeManual Svc Order - Addil	NRC - DC3 - Incremental Charge-Manual Syc Order - 1st	Electronic Syc Order per I s	NRC - OC3 - Facility Termination - Disconnect - 1st	OC3 - Facility Termination -	- OC3 - Facility Termination -	narily Combined in GA (Note	facility termiantion per month	per mile per month	OC-3 Local Loop	NRC - SI S - I COMBINATION - SWITCH AS IS CONVERSION CHARGE - DISCONNECT - ADDI	NRC - ST S-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-STS-1 COMBINATION - *Switch As Is* Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add	NRC - STS-1 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	NRC - STS-1 - Incremental ChargeManual Svc Order - Add"	NRC - STS-1 - Incremental Charge-Manual Syc Order - 1st	NRC - Flectronic Syc Order, per LSR	Manual Sv		NRC - STS-1 - Facility Termination - Disconnect - Add'l	NRC - STS-1 - Facility Termination - Disconnect - 1st	STS-1 - Facility Termination - /	NRC - STS-1 - Facility Termination	NRC - Ordinarily Combined in GA (Note 5)	per mile per month	STS-1 Local Loop	070 4 1000 1000	NRC- DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"l	NRC- DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - 1st		NRC - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Addil	Incremental Char		NRC - Electronic Svc Order, per LSR disconnect	NRC - Electronic Svc Order, per LSR	NRC - Manual Svc Order, per LSR disconnect	141	- Facility Termination - Disconnect -	- Facility Termination -	NRC - Facility Termination - Add'l	Termination - '	facitility termination per month	3	DS3 Local Loon	NRC- DS1 COMBNATION - Switch As Is* Conversion Charge - Disconnect - Addl	דייייי דיאדייטבט ו אוייס (כבו "י)
		ILSND	200		UNCCC	UNCCC	UNCCC	OCIONA	SOMAN	SOMAN	SOMOS	SOMAN	SOMEC						1L5ND				UNCCC	UNCCC			SOMAN	SOMAN	SOME	SOMEC	SOMAN	SOMAN	UDLS1	UDLS1	UDLS1	UDLS1	0	1L5ND	5		UNCCC	UNCCC	UNCCC	UNCCC	COMPAN	SOMAN	SOMAN	SOMAN	SOMEC	SOMEC	SOMAN	SOMAN	UE3PX	UE3PX	UE3PX	Ydsall	UE3PX	1L5ND		UNCCC	5
NA S	NIA	\$5,630	¢10 13	<b>\$0.00</b>	\$0.00	\$32.11	\$54.03	5	NA S	NA S	2 3	N S	N S	NA	NA NA	NA.		\$1,123	\$7.09		\$0.00	\$0.00	\$32.11	\$54.03		NA	NA :	NA S	2 3	NA NA	Ä	NA.	NA	NA	NA	NA	<b>*</b> 110.00	\$10.85			\$0.00	\$0.00	\$32.11	\$54.03		NA NA	3	S A	NA.	NA	Ā	NA	NA	X.	NA S	NIA	\$419.65	\$10.85		\$0.00	^
NA S	VIN	\$2,068	611 18	<b>\$10.77</b>	\$16.77	\$32.75	\$32.75	1467	NA S	NA S	25	A S	Z S	AN AN	NA AN	NA NA		\$648.60	\$8.93		\$10.77	\$16.77	\$32.75	\$32.75		ΝA	NA.	NA S	25	NA NA	NA.	N.	NA	AN	NA	N.	0.00	\$11.77			\$16.77	\$16.77	\$32.75	\$32.75	10.	NA NA	NA NA	Š	NA.	NA	NA	AN	NA	AN.	8	NA	\$404.58	\$11.77		\$16.77	1
\$413.00	00 031 13	\$2,109.00	60 21	÷	\$12.61	\$26.99	\$58.43	0.00	\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	\$122.31	\$473.00	\$947.69		\$630.21	\$6.75		\$12.01	\$12.61	\$26.99	\$58.43		\$18.03	\$18.03	\$37.55	427.75	\$3.50	S X	¥	\$119.14	\$122.31	\$426.40	\$639.50		\$8.90			\$12.61	\$12.61	\$26.99	\$58.43	0	\$18.03	\$37.55	\$37.55	NA.	\$3.50	Ą	N.	\$119.14	\$122.31	\$426.40	\$630 50	\$390.34	\$8.90		\$12.61	>
NA S	NIA	\$2,457	08 073	<b>#</b> 0.00	\$0.00	\$32.16	\$54.09	5	N S	N S	Z 3	N 5	Z 3	× ×	S N	¥		\$436.95	\$33.15		\$0.00	\$0.00	\$32.16	\$54.09		NA	N.	X S	Z 3	NA NA	<b>X</b>	¥	¥	AN	NA	N N	4.00.00	\$43.69			\$0.00	\$0.00	\$32.16	\$54.09	4.	NA NA	¥ ×	<b>X</b>	<b>X</b>	¥	¥	NA	¥	₩ :	¥ ₹	NΙΔ	\$436.95	\$43.69		\$0.00	5
NA S	NIA	\$2,571	626 40	<b>#</b> 0.00	\$0.00	\$32.24	\$54.23	5	N S	N S	Z 3	N 5	N S	8	¥ ×	¥		\$753.65	\$29.58		\$0.00	\$0.00	\$32.24	\$54.23		NA	N.	X S	Z 3	NA NA	<b>X</b>	¥	NA	AN	NA	N N	<b>*</b> 101.00	\$38.98	9		\$0.00	\$0.00	\$32.24	\$54.23	4.	NA NA	¥ ×	<b>X</b>	<b>X</b>	¥	¥	NA.	W.	₩ :	¥ ₹	NΙΔ	\$497.08	\$38.98		\$0.00	>
N S	NIA	\$2,371	650 70	<b>#</b> 0.00	\$0.00	\$32.16	\$54.09	5	N S	N 5	8 3	N 5	N S	× ×	3	¥		\$689.68	\$41.27		\$0.00	\$0.00	\$32.16	\$54.09		NA	N :	X S	8 3	Z A	<b>X</b>	¥	NA	AN	NA	N N	1	\$427.81	9		\$0.00	\$0.00	\$32.16	\$54.09		N X	\$ \$	<b>X</b>	NA.	¥	¥	NA	NA :	X S	NA 5	NΙΔ	\$427.81	\$54.39		\$0.00	Mo
X.	N	\$2,122	62028	60.00	\$0.00	\$32.10	\$54.00	5	N S	N S	5	200	2 3	S 8	\$	¥		\$611.36	\$24.69		\$0.00	\$0.00	\$32.10	\$54.00		Ν	×.	X S	5	N N	<b>X</b>	×	NA	NA	NA	¥	000	\$32.53	9		\$0.00	\$0.00	\$32.10	\$54.00		NA A	₹ ₹	. ₹	×.	NA	¥	¥	NA :	¥ ;	N 5	NA	\$387.01	\$32.53		\$0.00	5
X.	NIA	\$2,663	614	<b>#</b> 0.00	\$0.00	\$32.25	\$54.26	5	N S	N S	8 3	N 5	2 3	<b>S</b>	\$	¥		\$701.71	\$11.78		\$0.00	\$0.00	\$32.25	\$54.26		NA.	×.	X S	8 3	N N	<b>X</b>	×	NA	NA	NA	×	0	\$15.53	1		\$0.00	\$0.00	\$32.25	\$54.26	5	NA A	₹ ₹	. ₹	×.	NA	¥	¥	NA :	¥ ;	N 5	NA	\$421.60	\$15.53		\$0.00	3
N.	NIA	\$2,079	600 51	<b>#</b> 0.00	\$0.00	\$32.17	\$54.13	5	N S	N S	5	8	2 3	Š	\$ \$	¥		\$620.20	\$23.16		\$0.00	\$0.00	\$32.17	\$54.13		ΝĀ	¥.	X S	5	N N	¥	Z.	NA	NA	NA	¥	1	\$400.53			\$0.00	\$0.00	\$32.17	\$54.13	5	N A	\$ \$	\$	Z.	NA A	Ā	NA	NA :	<b>₹</b>	N S	NA	\$400.21	\$30.53		\$0.00	‡

Local Channel - Dedicated - 2-Wire VG - Rev Bat.  Monthly Recurring per month  Monthly Recurring per mile per month  MRC - Ordinarily Combined in GA (Note 5)  NRC - 2-wire VG Local Channel - 1st  NRC - 2-wire VG Local Channel - Add    NRC - 2-Wire VG Local Channel - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - 1st  NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - 1st  NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - 1st  NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add    NRC - 2-Wire VG - Incremental Charge - Manual Svc Order - Disconnect - Add	NRC - 24-Wire Vo Order, per LSR  NRC - 24-Wire VG - Incremental Charge-Manual Svc Order - 1st  NRC - 24-Wire VG - Incremental Charge-Manual Svc Order - Disconnect - 1st  NRC - 24-Wire VG - Incremental Charge-Manual Svc Order - Disconnect - 1st  NRC - 24-Wire VG - Incremental Charge-Manual Svc Order - Disconnect - Add1  NRC - 24-Wire VG - Incremental Charge-Manual Svc Order - Disconnect - Add1  NRC-24-Wire COMBINATION - Switch As Is 'Conversion Charge - 1st  NRC-24-WIRE COMBINATION - Switch As Is 'Conversion Charge - Disconnect - 1st  NRC-24-WIRE COMBINATION - Switch As Is 'Conversion Charge - Disconnect - Add1  NRC-24-WIRE COMBINATION - Switch As Is 'Conversion Charge - Disconnect - Add1  NRC-24-WIRE COMBINATION - Switch As Is 'Conversion Charge - Disconnect - Add1	Local Channels:  Local Channel - Dedicated - 2-Wire VG Local Channel - Dedicated - 2-Wire VG Monthly Recurring per month Monthly Recurring per mile per month MRC - Ordinarily Combined in GA (Note 5)  NRC - 2-wire VG Local Channel - 1st NRC - 2-wire VG Local Channel - Add!	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)  NRC-C-49 COMBINATION - "Switch As Is "Conversion Charge - Ist  NRC-CC-49 COMBINATION - "Switch As Is" Conversion Charge - Nadi  NRC-CC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st  NRC-CC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add1	NRC - Electronic Svc Order, per LSR  NRC - OCA8 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-1  NRC - OCA8 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-1  NRC - OCA8 - Interface - Manual Svc Order vs Electronic-Disconnect-1  NRC - OCA8 - Interface - Manual Svc Order vs Electronic-Disconnect-AddT  NRC - OCA8 - Interface - Manual Svc Order vs Electronic-Disconnect-AddT  NRC - OCA8 - Incremental ChargeManual Svc Order-1  NRC - OCA8 - Interface OCT2 on OCA8 - Incremental ChargeManual Svc Orde  NRC - OCA8 - Interface OCT2 on OCA8 - Incremental ChargeManual Svc Orde	DC-48 Local Loop per mile per month facility termination per month facility termination per month DC-12 interface on OC-48 Loop per month URC - Ordinarily Combined in GA (Note 5) NRC - OCA8 - Facility Termination - 1st NRC - OCA8 - Facility Termination - Add1 NRC - OCA8 - Interface OC12 on OCA8 - 1st 1 NRC - OCA8 - Facility Termination - Disconnect - 1st NRC - OCA8 - Facility Termination - Disconnect - 1st NRC - OCA8 - Facility Termination - Disconnect - 1st NRC - OCA8 - Interface OC12 on OCA8 - Disconnect - 1st NRC - OCA8 - Interface OC12 on OCA8 - Disconnect - 1st NRC - OCA8 - Interface OC12 on OCA8 - Disconnect - 1st NRC - OCA8 - Interface OC12 on OCA8 - Disconnect - 1st NRC - OCA8 - Interface OC12 on OCA8 - Disconnect - Add1	ENHANCED EXTENDED LINKS (GELs)  NRC - CC12 - Facility Termination - Disconnect - 1st NRC - CC12 - Facility Termination - Disconnect - Add1 NRC - Electronic Svc Order - per LSR NRC - CC12 - Incernential Charge - Nanual Svc Order - 1st NRC - CC12 - Incernential Charge - Nanual Svc Order - Add1 NRC - CC12 - Incernential Charge - Nanual Svc Order - Add1 NRC - CC12 - Incernential Cost-Manual Svc Order vs. Elect-Disconnect-1st NRC - CC12 - Incernential Cost-Manual Svc Order vs. Elect-Disconnect-1st NRC - CC12 - Incernential Cost-Manual Svc Order vs. Elect-Disconnect-Add1 NRC - CC12 - Incernential Cost-Manual Svc Order vs. Elect-Disconnect-Add1 NRC - CC12 - Combination 'Switch As is Conversion Charge - Add1 NRC - CC12 COMBINATION - Switch As is Conversion Charge - Disconnect - 1st NRC - CC12 COMBINATION - Switch As is Conversion Charge - Disconnect - Add1 NRC - CC12 COMBINATION - Switch As is Conversion Charge - Disconnect - Add1 NRC - CC12 COMBINATION - Switch As is Conversion Charge - Disconnect - Add1 NRC - CC12 COMBINATION - Switch As is Conversion Charge - Disconnect - Add1
ULDR2 11.5NC ULDR2 ULDR2 ULDR2 SOMEC SOMAN SOMAN SOMAN	SOMEC SOMAN SOMAN SOMAN SOMAN UNCCC UNCCC UNCCC UNCCC	ULDV2 1L5NC ULDV2 ULDV2		SOMEC F-1 SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN SOMAN	1L5ND	SOMAN
\$14.61 \$0.00 NA NA NA NA NA	NA NA NA NA NA NA S54.03 \$52.11 \$0.00	\$14.61 \$0.00 NA	\$54.03 \$32.11 \$0.00	X	\$33.22 \$1,947 \$699.62 NA NA NA NA NA NA	AL NA
\$26.31 \$0.00 NA NA NA NA NA	NA NA NA NA NA NA NA NA NA NA NA NA NA N	\$29.33 \$0.00 NA	\$32.75 \$32.75 \$16.77 \$16.77	X	\$36.67 \$1,699 \$592.09 \$8592.09 NA NA NA NA NA NA	FL NA
\$13.91 \$0.00 \$245.43 \$33.90 \$3.50 \$33.63 \$27.49 \$8.08	\$3.50 \$27.49 \$22.24 \$8.08 \$58.43 \$58.43 \$12.61 \$12.61	\$13.91 \$0.00 \$245.43 \$33.90	\$58.43 \$26.99 \$12.61 \$12.61	\$3.50 \$37.55 \$37.55 \$37.55 \$37.55 \$37.55 \$18.03 \$18.03 \$18.03	\$27.25 \$1.598.00 \$594.80 \$1.162.00 \$413.00 \$413.00 \$539.36 \$539.36 \$172.31 \$119.14 \$122.31	GA \$122.31 \$119.14 \$3.50 \$37.55 \$37.55 \$37.55 \$18.03 \$18.03 \$18.03 \$18.03 \$18.03
\$22.26 \$0.00 NA NA NA NA NA	NA NA NA NA NA NA NA NA NA NA NA NA NA N	\$22.26 \$0.00 NA	\$54.09 \$32.16 \$0.00	NA N	\$166.59 \$2,129 \$725.77 \$725.77	KY NA
\$14.94 \$0.00 NA NA NA NA NA	NA NA NA NA NA NA NA NA NA NA NA NA NA N	\$14.94 \$0.00 NA	\$54.23 \$32.24 \$0.00 \$0.00	NA N	\$119.40 \$2,268 \$723.29 NA NA NA NA NA NA	LA NA NA NA NA NA NA NA NA NA S54.23 \$32.24 \$0.00
\$17.83 \$0.00 NA NA NA NA NA	NA NA NA NA NA NA NA S54.09 \$32.16 \$0.00	\$17.83 \$0.00 NA	\$54.09 \$32.16 \$0.00 \$0.00	A A A A A A A A A A A A A A A A A A A	\$166.59 \$1.753 \$667.00 NA NA NA NA NA NA	NA N
\$14.82 \$0.00 NA NA NA NA NA	NA NA NA NA NA NA S54.00 \$52.10 \$0.00	\$14.83 \$0.00 NA	\$54.00 \$32.10 \$0.00 \$0.00	NA N	\$120.02 \$1.677 \$582.66 NA NA NA NA NA NA	NG NA NA NA NA NA NA NA NA NA NA NA NA NA
\$16.83 \$0.00 NA NA NA NA	NA NA NA NA NA NA S54.26 \$32.25 \$0.00 \$0.00	\$16.83 \$0.00 NA	\$54.26 \$32.25 \$0.00 \$0.00		\$47.57 \$1,733 \$773.40 NA NA NA NA NA NA	SC NA NA NA NA NA NA NA NA NA NA NA NA NA
\$19.02 \$0.00 NA NA NA NA	NA NA NA NA NA NA NA NA NA NA NA NA NA N	\$19.02 \$0.00 NA	\$54.13 \$32.17 \$0.00 \$0.00	X X X X X X X X X	\$93.50 \$1.832 \$570.54 NA NA NA NA NA NA	TN NA

																1																												-						Ŧ								H	Ħ	п	П
	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - STS-1 Local Channel - Incremental ChargeManual Svc Order - Disconnec - A	NRC - STS-1 Local Channel - Incremental ChargeManual Svc Order - Disconnect - 1	NRC - STS-1 Local Channel - Incremental ChargeManual Svc Order - Add'i	NRC - Electronic Svc Order, per LSX	- STS-1 Loc	NRC - STS-1 Local Channel Facility Termination - 1st	NRC - Ordinarily Combined in GA (Note 5)	STS-1 Local Channel - Facility Termination per month	STS-1 Local Channel - per mile per month	Local Channel - Dedicated - STS-1		NRC- DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Evisting Combination "Switch As Is" Conversion Charge (Note 6)	NDC - DC3   Cc3   Channel - Incremental Charge - Manual Svc Order - Disconnect - Addi	NRC - Dos Local Channel - Incremental Charge - Manual Syc Order - Additional Charge - Manual Charge - M	NRC - US3 Local Channel - Incremental Charge - Manual Svc Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - US3 Local Channel - Facility Termination - Add I	NRC - DS3 Local Channel Facility Termination - 1st	NRC - Ordinarily Combined in GA (Note 5)	DS3 Local Channel - Facility Termination per month	DS3 Local Channel - per mile per month	Local Channel - Dedicated - DS3	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'l	NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-DIS1 COMBINATION - "Switch As Is" Conversion Charge : 1st	NRC - DST Local Challed - Inciented Charge - Manual Svc Order - Discolline - Add	NRC - DS1 Local Channel - Incremental ChargeManual Svc Order - Disconnect - 1st	NRC - DST Local Channel - Incremental ChargeManual SVC Order - Addi	NRC - DS1 Local Channel - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - DS1 Local Channel - Add'l	NRC - DS1 Local Channel - 1st	NRC - Ordinarily Combined in GA (Note 5)	Monthly Recurring per month	20	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'i	NRC-2/4-WIRE COMBINATION - Which As is Conversion Charge - Add I	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	ge (Note 6)	NRC - 4-Wire VG Local Channel - Incremental Charge Manual Svc Order - Disconne	- 4-Wire VG Local Channel - Incremental ChargeManual Svc Order -	4-Wire VG Local Channel - Incremental ChargeManual Svc Order -	5	NRC - Flectronic Syc Order per LSR	Local Channel -	NRC - Ordinarily Combined in GA (Note 5)	Monthly Recurring per mile per month	Monthly Recurring per month	Local Channel - Dedicated - 4-Wire VG	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - AddI	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	III I LIOPA PYTELINEN I MIVO JEEI A
UNCCC	UNCCC		SOMAN	s SOMAN	SOMAN	SOMAN	ULDFS	ULDFS		ULDFS	1L5NC			UNCCC	UNCCC	UNCCC	UNCCC		NAMOS		SOMAN	SOMEC	OLDF3	ULDF3	1	ULDF3	1L5NC	USOC	UNCCC	UNCCC	UNCCC	I NOOO	OCIVINA	SOMAN	SOMAN	SOMAN	SOMEC	ULDF1	ULDF1		1 500	7	UNCCC	UNCCC.	UNCCC		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC.	ULDV4		1L5NC	ULDV4		UNCCC	UNCCC	UNCCC	UNCCC	200
\$0.00	\$32.11		¥	NA	<b>₹</b> ₹	8 8	¥	NA.		\$525.40	\$8.44			\$0.00	\$0.00	\$32 11	\$54.03	5	N S	N 2	3	<u> </u>	, N	×		\$535.92	\$8.44	AL	\$0.00	\$0.00	\$32.11	\$54.03	5	8 8	X X	X X	N.	NA	¥	0.00	\$0.00	900	\$0.00	\$0.00	\$54.03		N.	N.	<b>₹</b> 5	¥ ₹	Z S	<b>S S</b>		\$0.00	\$15.77		\$0.00	\$0.00	\$32.11	\$54.03	2
\$16.77	\$32.75		¥	NA	<b>₹</b> §	N N	¥	NA.		\$569.67	\$9.32			\$16.77	\$16.77	\$32 75	\$32.75	5	N S	3	<b>₹</b>	<b>X</b>	<u> </u>	<b>X</b>		\$556.27	\$9.32	P	\$16.77	\$16.77	\$32.75	\$32.75	5	N N	: ₹	<b>X</b>	NA.	NA	¥	<b>\$0.00</b>	\$0.00	200	\$16.77	\$16.77	\$32.75	1	N.	¥	<b>₹</b> 5	<b>₹</b> §	N S	X X		\$0.00	\$30.50		\$16.77	\$16.77	\$32.75	\$32.75	0
\$12.61	\$26.99		\$18.03	\$18.03	\$37.55	\$3.50	\$545.54	\$761.81		\$517.56	\$6.92		*****	\$0.00	\$0.00	\$39 60	\$71.04	6.00	\$18.03	\$49.03	\$37.55	\$3.50	\$426.40	\$639.50		\$515.91	\$6.92	GA	\$12.61	\$12.61	\$26.99	\$58.43	41	\$35.76	\$33.53	\$65.60	\$3.50	\$84.14	\$166.88	0.00	\$0.00	300000000000000000000000000000000000000	\$12.61	\$10.99	\$58.43		\$8.08	\$22.24	\$27.49	\$33.63	\$3.50	\$245.43		\$0.00	\$14.99		\$12.61	\$12.61	\$26.99	\$58.43	5
\$0.00	\$32.16		¥	NA	<b>₹</b> ₹	8 8	NA.	NA		\$635.09	\$34.00			\$0.00	\$0.00	\$32 16	\$54.09	5	N S	× ×	3 3	¥	X X	×		\$635.09	\$34.00	2	\$0.00	\$0.00	\$32.16	\$54.09	3	8 8	¥	X	NA	NA	¥.	0.00	\$0.00	9	\$0.00	\$32.10	\$54.09		NA.	NA.	<b>₹</b>	K 3	N S	Z Z		\$0.00	\$23.38		\$0.00	\$0.00	\$32.16	\$54.09	5
\$0.00	\$32.24		¥	NA	<b>₹</b> 5	N N	¥	NA.		\$558.00	\$8.77			\$0.00	\$0.00	\$32.24	\$54.23	5	NA S		5 ₹	: ₹	. X	¥		\$669.01	\$30.34	LA	\$0.00	\$0.00	\$32.24	\$54.23	5	N N	X X	X	N.	NA	¥	0.00	\$0.00	9	\$0.00	\$0.00 #2.26	\$54.23		NA.	¥	₹ 5	¥ ₹	N S	S 8		\$0.00	\$16.21		\$0.00	\$0.00	\$32.24	\$54.23	- >
\$0.00	\$32.16		¥	NA	<b>₹</b> 5	N N	¥	NA.		\$531.39	\$38.98			\$0.00	\$0.00	\$32.16	\$54.09	5	NA S		5 ₹	: ₹	. X	¥		\$533.33	NA	MS	\$0.00	\$0.00	\$32.16	\$54.09	5	N N	X X	×	N.	NA	¥	0.00	\$0.00	200	\$0.00	\$0.00	\$54.09		NA.	¥	₹ 5	¥ ₹	N S	S 8		\$0.00	\$19.03		\$0.00	\$0.00	\$32.16	\$54.09	Mo
\$0.00	\$32.10		¥	NA	<b>₹</b> §	N X	¥	NA		\$512.00	¥			\$0.00	\$0.00	\$32.10	\$54.00	5	N 5	× ×	X X	<b>X</b>	X X	× ×		\$498.87	A	NO	\$0.00	\$0.00	\$32.10	\$54.00	5	N X	<b>*</b>	<b>X</b>	NA	NA	¥	0.00	\$0.00	900	\$0.00	\$0.00	\$54.00		NA	¥	¥ 5	X 5	N N	NA NA		\$0.00	\$15.87		\$0.00	\$0.00	\$32.10	\$54.00	5
\$0.00	\$32.25		¥	NA	<b>₹</b> §	N X	¥	NA		\$481.14	\$12.08			\$0.00	\$0.00	\$32.25	\$54.26	5	N 5	× ×	X X	<b>X</b>	X X	× ×		\$493.31	\$12.08	SC	\$0.00	\$0.00	\$32.25	\$54.28	5	N X	<b>*</b>	<b>X</b>	NA	NA	¥	0.00	\$0.00	200	\$0.00	\$0.00	\$54.26		NA	¥	¥ 5	X 5	N N	NA NA		\$0.00	\$18.05		\$0.00	\$0.00	\$32.25	\$54.26	ŝ
\$0.00	\$32.17		₹	NA	<b>₹</b> §	N A	¥	¥		\$615.65	\$25.11			\$0.00	\$0.00	\$32.17	\$54 13	5	N 5	<b>S</b>		. ₹	<u> </u>	¥		\$607.28	\$23.76	₹	\$0.00	\$0.00	\$32.17	\$54.13	5	NA NA	<b>X</b>	¥	NA.	NA	¥	0.00	\$0.00	94007	\$0.00	\$0.00	\$54.13		¥	¥	¥ §	¥ §	N S	X X		\$0.00	\$20.14		\$0.00	\$0.00	\$32.17	\$54.13	T <sub>N</sub>

H	F	H		1	F	F							1	Ŧ	H					1	1	F		H		H	l		F	H				H	Ŧ		F					F		-	H	F		H		Ŧ	F	P			F	H			Н
Interoffice Channel - Dedicated - 2-wire VG - per mile per month		NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge (Note 6)	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-Add			NRC - OC-48 - Incremental ChargeManual Svc Order-1st	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-Add'l	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-1st	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-Add'i	• •	NRC - CL48 - Interface OC12 on OC48 - Disconnect - Add I				NRC - OC48 - Interface OC12 on OC48 - Add'l	NRC - OC48 - Interface OC12 on OC48 - 1st	NRC - OC48 - Facility Termination - Add'l	NRC - Ordinarily Combined in GA (Note 5)	Local Channel - OC12 interface on OC48 Facility	Local Channel - OC48 - per Facility Termination	Local Channel - OC48 - per Mile	Local Channel - OC48	NRC- OC-12 COMBINATION - "Switch As is" Conversion Charge - Disconnect - Addi	NRC- OC-12 COMBINATION - "Switch As is" Conversion Charge - Disconnect - 1st	NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - Add'l	"Switch As Is" Conversion Charge -	Conversion	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-Add'l	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-1st	NRC - OC12 - Incremental Charge - Manual Svc Order - Add'l	NRC - OC12 - Incremental Charge - Manual Svc Order - 1st	NRC - OC12 - Facility Termination - Disconnect - Additional Section Section 1 Section	NRC - OC12 - Facility Termination - Disconnect - 1st	NRC - OC12 - Facility Termination - Add'l	NRC - OC12 - Facility Termination - 1st	NRC - Ordinarily Combined in GA (Note 5)	Local Channel - OC12 - per Facility Termination	Local Channel OC12 For Mile	Topol Change Codds	NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	"Switch As Is" Conversion Charge -	on	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Add'l	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	NRC - OC3 - Incremental ChargeManual Svc Order - Add'l	NRC - DC3 - Incremental ChargeManual Svc Order - 1st	NRC - OC3 - Facility Termination - Disconnect - Add'l	onnect -	NRC - OC3 - Facility Termination - Add'l	NRC - OC3 - Facility Termination - 1st	Local Channel - OC3 - per Facility Termination	Local Channel - OC3 - per Mile	Local Channel - OC3	NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	ENHANCED EXTENDED LINKS (EELS)
1L5XX		UNCCC	UNCCC	UNCCC		SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	TBA	TBA	TBA	TBA	TBA	TBA	TBA	TBA	TBA	1L5NC	USOC	UNCCC	UNCCC	UNCCC	UNCCC		SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	TBA	TBA	TBA		TBA	1 500	500	UNCCC	UNCCC	UNCCC		SOMAN	SOMAN	SOMAN	SOMEC	TBA	TBA	TBA	TBA	TBA	1L5NC		UNCCC	USOC
\$0.0339		\$0.00	\$0.00	\$32.11	\$54.03	NA	¥	NA	NA	NA	¥.	N :	X S	NA A	S A	NA.	NA	NA	NA A	X S	NIA	\$736.71	\$1,713	\$33.22	AL	\$0.00	\$0.00	\$32.11	\$54.03		NA	¥.	N :	N S	3	¥	NA NA	NA		\$2.557	610 13	≥	\$0.00	\$0.00	\$54.03		NA	NA	<b>₹</b>	Z Z	X X	NA.	¥	¥	\$691.33	\$8.23		\$0.00	A-
\$0.0100		\$16.77	\$16.77	\$32.75	\$32 75	NA	NA	AN	AN	NA	NA.	NA.	NA S	NA AN	NA NA	NA	NA	NA	AN	NA S	N	\$586.28	\$1,944	\$36.67	FL	\$16.77	\$16.77	\$32.75	\$32.75		NA	N.	NA.	N S	AN AN	×	NA	AN		\$2.753	£11 10	1	\$16.77	\$16.77	\$32.75		AN	NA	¥ ₹	NA NA	NA	AN	ΑN	¥	\$940.35	\$7.83		\$16.77	- F
\$0.0222		\$12.61	\$12.61	\$26.99	\$58.43	\$37.55	\$37.55	\$37.55	\$37.55	\$18.03	\$18.03	\$18.03	\$18.03	\$119.14	\$122.31	\$119.14	\$122.31	\$317.38	\$539.36	\$413.00	\$1 162 00	\$594.80	\$1,598.00	\$27.25	GA	\$12.61	\$12.61	\$26.99	\$58.43		\$18.03	\$18.03	\$37.55	\$37.55	\$119.14	\$122.31	\$413.00	\$1,162.00		\$2.109.00	60 SA	)	\$12.61	\$12.61	\$58.43		\$18.03	\$18.03	\$37.55	\$37.50	\$119.14	\$122.31	\$413.00	\$947.69	\$630.21	\$6.75		\$12.61	GA
\$0.0301		\$0.00	\$0.00	\$32.16	\$54.09	NA	¥	AN	NA	NA	¥.	NA.	N S	NA	S N	₹	¥	NA	NA A	N S	NIA	\$725.77	\$2,129	\$133.84	ΚY	\$0.00	\$0.00	\$32.16	\$54.09		NA	¥.	NA :	N S	3 ₹	₹	NA	AN		\$2.457	640.00	2	\$0.00	\$0.00	\$54.09		AN	NA	<b>₹</b>	Z Z	<b>X</b>	NA.	Ā	¥	\$713.29	\$33.15		\$0.00	ΚY
\$0.0384		\$0.00	\$0.00	\$32.24	\$54.23	NA	NA	N	NA.	NA	N.	NA.	N S	NA NA	₹ ₹	NA.	NA	NA	¥	N S	NIA	\$723.29	\$2,268	\$119.40	LA	\$0.00	\$0.00	\$32.24	\$54.23		NA	¥.	NA :	N S	3	¥	NA	AN		\$2.571	e36 40	>	\$0.00	\$0.00	\$54.23		AN	NA	<b>₹</b>	Z Z	NA NA	AN	Ā	¥	\$753.65	\$29.58		\$0.00	F
\$0.0323		\$0.00	\$0.00	\$32.16	\$54.09	NA	NA A	AN	AN	NA	¥	N :	X S	N N	<b>X</b>	A	NA	NA	Ā	X S	NIA	\$667.00	\$1,753	\$166.59	MS	\$0.00	\$0.00	\$32.16	\$54.09		NA	¥	NA :	N S	8	¥	NA.	AN		\$2.371	02.033	Š	\$0.00	\$0.00	\$54.09		NA	NA	<b>₹</b>	N N	X X	AN	¥	¥	\$689.68	\$41.27		\$0.00	SW
\$0.0282		\$0.00	\$0.00	\$32.10	\$54.00	X	NA.	NA	NA	NA	¥	¥.	X S		<b>S</b>	¥	NA A	NA	¥	X S	NIA	\$582.66	\$1,677	\$120.02	NC	\$0.00	\$0.00	\$32.10	\$54.00		NA	¥	N :	X S	3	<b>X</b>	¥	NA		\$2.122	\$20.20 0.00	5	\$0.00	\$0.00	\$54.00		NA	NA	<b>₹</b> §	N N	×	NA	¥	¥	\$611.36	\$24.69		\$0.00	NC
\$0.0373		\$0.00	\$0.00	\$32.25	\$54.26	¥	NA.	NA	NA	NA	¥	N.	¥ 5	NA KA	X X	NA.	NA A	NA	¥	¥ 5	No	\$773.40	\$1,733	\$47.57	SC	\$0.00	\$0.00	\$32.25	\$54.26		NA	¥	NA :	N 5	N X	<b>X</b>	NA	NA		\$2.663	61450	3	\$0.00	\$0.00	\$54.26		NA	NA	<b>₹</b> 5	NA NA	×	NA	¥	¥	\$701.71	\$11.78		\$0.00	SC
\$0.0173		\$0.00	\$0.00	\$32.17	\$54.13	NA	NA	NA	NA	NA	¥	¥.	¥ ;	N A	¥ X	¥	NA A	NA	¥	¥ ;	NA	\$570.54	\$1,832	\$93.50	₹	\$0.00	\$0.00	\$32.17	\$54.13		NA	¥	₩.	₹ 5	\$ \$	¥	N <sub>P</sub>	NA		\$2.079	630 51	į	\$0.00	\$0.00	\$54.13		NA	NA	<b>₹</b> §	NA X	¥	NA	¥	N	\$620.20	\$23.16		\$0.00	T

NRC - DS1 Interoffice Channel - Facility Termination - 1st	Interoffice Channel - Dedicated - DS1 - Facility Termination per month  NRC - Ordinarily Combined in GA (Note 5)	Interoffice Channel - Dedicated - DS1	NRC - 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add'i	NRC- 2/4-WIRE COMBINATION - *Switch As Is* Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - 4-wire 64 kbps interoffice Channel - Incremental Charge - Manual Svc Order - D	NRC - 4-wire 64 kbps interoffice Channel - incremental Charge - Manual Svc Order - Ad	NRC - 4-wire 64 kbps Interoffice Channel - Incremental ChargeManual Svc Order - 1s	NRC - Electronic Svc Order, per LSR	NRC - 4-wire 64 kbps Interoffice Channel - Facility Termination - Add'l	NRC - 4-wire 64kbps Interoffice Channel - Facility Termination - 1st	NRC - Ordinarily Combined in GA (Note 5)	Interoffice Channel - Dedicated - DS0 - 64 kbps - Facility Termination per month	Interoffice Channel - Dedicated - DSO - 64kbps	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Addli	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	- 4-wire 56 kbps Interoffice Channel - Incremental ChargeManual Svc Order -	NRC - 4-wire 56 kbps Interoffice Channel - Incremental ChargeManual Svc Order - Di	- 4-wire 56 kbps Interoffice Channel - Incremental Charge - Manual Svc Order -	NRC - Electronic Svc Order, per LSR	NRC - 4-wire 56 kbps Interoffice Channel - Facility Termination - Add'l	NRC - 4-wire 56kbps Interoffice Channel - Facility Termination - 1st	Interoffice Channel - Dedicated - DS0 - 56 kbps - Facility Termination per month	Interoffice Channel - Dedicated - DS0 - 56kbps - per mile per month	Interoffice Channel - Dedicated - DS0 - 56kbps	NRC- 24-WIRE COMBINATION - Switch As is Conversion Charge - Disconnect - Addi	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	- 4-wire vG Interoffice Channel - Incremental Charge wanual Svc Order -	- 4-wire VG Interoffice Channel -	4-wire VG Interoffice Channel - Incremental ChargeManual Svc	- Electronic Svc Order, per LSR	NRC - 4-wire VG Interoffice Channel - Facility Termination - 1st	rily Combined in GA (Note 5)	Interoffice Channel - Dedicated 4-wire VG - Facility Termination per month	Interoffice Channel - Dedicated 4-wire VG - per mile per month	Interoffice Channel - Dedicated - 4-wire VG	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Addi	NRC- 2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-2/4-WIRE COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-2/4-WIRE COMBINATION - 'Switch As is' Conversion Charge - 1st	NRC - 2-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - Discor	Incremental ChargeManual Svc	2-wire VG Interoffice Channel - Incremental ChargeManual Svc Order -	NRC - 2-wire VG Interoffice Channel - Incremental ChargeManual Svc Order - 1st	- 2-wire VG Inter	NRC - 2-wire VG Interoffice Channel - Facility Termination - 1st	Interoffice Channel - Dedicated 2-wire VG - Facility Termination per month  NRC - Ordinarily Combined in GA (Note 5)	ENHANCED EXTENDED LINKS (EELS)
U1TF1	U1TF1	11 5XX	UNCCC	UNCCC	UNCCC	UNCCC	OCIVIAIN	SOMAN	SOMAN	- 1		U1TD6	U1TD6		U1TD6	1 5 4	UNCCC	UNCCC	UNCCC	SOMAN		SOMAN	SOMEC	U1TD5	U1TD5	U11D5	1L5XX		ONCOC	UNCCC	UNCCC	UNCCC	SOIMAN	SOMAN	SOMAN	SOMAN	SOMEC	U1TV4		U1TV4	1L5XX		UNCCC	UNCCC	UNCCC	CINCOC	SOMAN	SOMAN	SOMAN	SOMEC	U1TV2	U1TV2	U1TV2	USOC
NA	\$79.69	0009 03	\$0.00	\$0.00	\$32.11	\$54.03	A	8 8	5 €	S X	Ä	Ä	¥		\$17.81	00000	\$0.00	\$32.11	\$54.03	¥ ?	N N	₹ ₹	¥	NA	¥	\$17.81	\$0.0339		\$0.00	\$0.00	\$32.11	\$54.03	N	5 €	<b>X</b>	¥	NA	X X		AN	NA		\$0.00	\$0.00	\$32.11	\$54.03	NA.	NA	N S	X X	<b>X X</b>	Ą	\$18.49	AL
NA	\$99.79	\$0.6013	\$16.77	\$16.77	\$32.75	\$32.75	¥.	N X	¥ ¥	₹ ₹	. ₹	¥	¥		\$19.46	0000	\$16.77	\$32.75	\$32.75	¥ ₹	2 3	<u> </u>	¥	N <sub>A</sub>	₹	\$19.46	\$0.0100		\$10.77	\$16.77	\$32.75	\$32.75	3	¥ ¥	<b>X</b>	¥	NA	N X		\$23.82	\$0.0100		\$16.77	\$16.77	\$32.75	\$32.75	NA	NA	N S	N N	X X	NA	\$26.72	7
\$169.57	\$78.47	\$0.4523	\$12.61	\$12.61	\$26.99	\$58.43	S	× ×	\$18.94	\$18.94	\$3.50	\$36.08	\$79.61		\$16.45	***************************************	\$12.61	\$26.99	\$58.43	X :	\$10.94	\$18.94	\$3.50	\$36.08	\$79.61	\$16.45	\$0.0222		\$12.01	\$12.61	\$26.99	\$58.43	5	5 €	¥	¥	NA	8		NA	NA		\$12.61	\$12.61	\$26.99	\$58.43	NA	NA	\$18.94	\$18.94	\$36.08	\$79.61	\$17.07	GA
NA	\$55.05	\$0.4500	\$0.00	\$0.00	\$32.16	\$54.09	Š	N N	<b>X X</b>	<b>₹</b>	<b>X</b>	<b>X</b>	¥		\$26.95	0000	\$0.00	\$32.16	\$54.09	¥ ₹	Z Z	3	X.	¥	₹	\$26.95	\$0.0301		\$0.00	\$0.00	\$32.16	\$54.09	5	<b>X X</b>	<b>X</b>	¥	NA	X X		NA	NA		\$0.00	\$0.00	\$32.16	\$54.09	NA	NA	N S	N N	S S	NA	\$27.66	7
NA	\$93.40	\$0 7831	\$0.00	\$0.00	\$32.24	\$54.23	Š	3 8	3	<b>X</b>	NA.	X.	¥		\$18.37	000	\$0.00	\$32.24	\$54.23	¥ §	Z Z	<b>3 3</b>	×.	NA	N	\$18.37	\$0.0384		\$0.00	\$0.00	\$32.24	\$54.23	×	3	<b>X X</b>	¥	NA	Z Z	5	NA	NA		\$0.00	\$0.00	\$32.24	\$54.23	NA A	NA	N S	N N	X X	NA.	\$19.10	5
NA	\$74.40	\$0.6508	\$0.00	\$0.00	\$32.16	\$54.09	S	× ×	5 €	<b>X</b>	¥	¥	¥		\$20.64	2000	\$0.00	\$32.16	\$54.09	X :	Z	<b>S S</b>	×	NA.	¥	\$20.64	\$0.0323		\$0.00	\$0.00	\$32.16	\$54.09	5	5 €	¥	¥	NA	8		NA	NA		\$0.00	\$0.00	\$32.16	\$54.09	NA	NA	NA 5	NA NA	N N	NA	\$21.33	MS
NA	\$71.29	\$0.0783	\$0.00	\$0.00	\$32.10	\$54.00	Ä	3 8	₹ ₹	<b>X</b>	<b>X</b>	<b>X</b>	¥		\$17.40	90000	\$0.00	\$32.10	\$54.00	X S	Z	<b>3 3</b>	¥	NA	N	\$17.40	\$0.0282		\$0.00	\$0.00	\$32.10	\$54.00	5	₹	<b>X X</b>	¥	NA	8 8	2	AN	NA		\$0.00	\$0.00	\$32.10	\$54.00	NA	NA	NA 5	N N	N N	NA	\$18.01	NC
NA	\$94.98	\$0.7598	\$0.00	\$0.00	\$32.25	\$54.26	NA.	N A	¥ ¥	<u> </u>	¥	¥	¥		\$20.71	60 0373	\$0.00	\$32.25	\$54.26	¥ 5	N N	X X	NA.	NA	Z.	\$20.71	\$0.0373		\$0.00	\$0.00	\$32.25	\$54.26	3	¥ ¥	X X	¥	NA	X X	25	NA	NA		\$0.00	\$0.00	\$32.25	\$54.26	NA.	NA	N S	N N	X X	NA	\$21.42	SC
NA	\$75.83	\$0.35.05	\$0.00	\$0.00	\$32.17	\$54.13	NA	S &	3 8	₹ ₹	X.	X.	¥		\$17.74	200 4 700	\$0.00	\$32.17	\$54.13	¥ §	2 3	₹ ₹	×	NA	N.	\$17.74	\$0.1730		\$0.00	\$0.00	\$32.17	\$54.13	N	₹ ₹	<b>₹</b>	N.	NA :	N N	5	NA	NA		\$0.00	\$0.00	\$32.17	\$54.13	NA	NA	¥ ;	NA NA	<b>5</b>	NA	\$18.33	Z

## NETWORK ELEMENTS AND OTHER SERVICES

H																		H																																			F	Н	=		]
NRC - OC12 - Facility Termination - Add'l	NRC - Ordinarily Combined in GA (Note 5)	Interoffice Channel - OC12 - per Facility Termination	Interoffice Channel - OC12 Interoffice Channel - OC12 - per Mile	NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st NRC- OC-3 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-OC-3 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC - UC3 - Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-Addi	NRC - OC3 -Incremental Cost - Manual Svc. Order vs. Elect-Disconnect-1st	NRC - OC3 - Incremental ChargeManual Svc Order - Add'l	NRC - OC3 - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - OC3 - Facility Termination - Disconnect - Add'l	NRC - OC3 - Facility Termination - Disconnect - 1st	NRC - OC3 - Facility Termination - Add'l	NRC - OC3 - Facility Termination - 1st	NRC - Ordinarily Combined in GA (Note 5)	per Mile	Interoffice Channel - OC3	NRC-STS-1 CUMBINATION - "Switch As is" Conversion Charge - Disconnect - Addi	NRC- STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add"I	NRC-STS-1 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC - STS-1 Interoffice Channel - Incremental Charge - Manual Svc Order - Disconne	NRC - STS-1 Interoffice Channel - Incremental ChargeManual Svc Order - Disconner	NRC - STS-1 Interoffice Channel - Incremental ChargeManual Svc Order - Add'l	NRC - STS-1 Interoffice Channel - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - STS-1 Interoffice Channel - Facility Termination - Add'l	NRC - Ordinarily Combined in GA (Note 5)	Interoffice Channel - Dedicated - STS-1 - Facility Termination per month	Interoffice Channel - Dedicated - STS-1 - per mile per month	Interoffice Channel - Dedicated - STS-1	Conversion		NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-DS3 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC - DS3 Interoffice Channel - Incremental ChargeManual Svc Order - Disconnect		- DS3 Interoffice Channel - Incremental ChargeManual Svc Order -	office Channel - I	NRC - Electronic Svc Order, per LSR	NRC - DS3 Interoffice Channel - Facility Termination - 1st	NRC - Ordinarily Combined in GA (Note 5)	Interoffice Channel - Dedicated - DS3 - Facility Termination per month	Interoffice Channel - Dedicated - DS3 - per mile per month	nteroffice Channel - Dedicated - Dear mile ner month	1.	"Switch As Is" Conversion Charge - Disconnect -	"Switch As Is" Conversion Ch	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - 1st	- DST Interchange Combination "Switch As Is" Conversion Cha	Channel - Incremental ChargeManual Svc Order -		mental ChargeManual Svc Order -	NRC - Electronic Svc Order, per LSR	ENHANCED EXTENDED LINKS (EELS)  NRC - DS1 Interoffice Channel - Facility Termination - Add'l	The state of the s
TBA	TDA	TBA	1L5XX	UNCCC	UNCCC	UNCCC	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	TBA	TBA	TBA	TBA	9	1L5XX		UNCCC	UNCCC	UNCCC	UNCCC	SOMAN	10		SOMAN	SOMEC	UITES	INTEC	U1TFS	1L5XX		UNCCC	UNCCC	UNCCC	INCCC	SOMAN	SOMAN	SOMAN	SOMAN	SOMEC	U1TF3		U1TF3	1L5XX		UNCCC	UNCCC	UNCCC	UNCCC	O NAME OF THE O	SOMAN	SOMAN	SOMAN	SOMEC	USOC U1TF1	
¥ 5	NIA	\$9,763	\$19.26	\$0.00	\$32.11	\$54.03	¥	₹ ₹	. ₹	NA NA	NA	NA NA	NA	NA	¥	4,14	\$7.35		\$0.00	\$0.00	\$32.11	\$54.03	X	×	NA	NA	X S	N S	N	\$898.15	\$4.98		\$0.00	\$0.00	\$32.11	\$54.03	NA	¥	AN	Z.	N S	<b>×</b> ×		\$898.15	\$4.98		\$0.00	\$0.00	\$32.11	\$54.03	5	N X	X X	X	¥ :	<b>∦</b> }	:
X S	NIA	\$11,685	\$26.91	\$16.77 \$16.77	\$32.75	\$32.75	Ä	<b>3 3</b>	. ₹	NA	NA	NA	NA	NA	K	\$3,043	\$8.38		\$16.77	\$16.77	\$32.75	\$32.75	Ä	NA.	NA	NA	NA S	N S	2	\$569.67	\$9.32		\$16.77	\$16.77	\$32.75	\$32.75	Ä	NA	NA	Z.	K 5	₹ ₹		\$1,130	\$4.25		\$16.77	\$16.77	\$32.75	\$32.75	5	8 8	<b>X X</b>	X	¥ :	<b>₹</b> ₽	!
\$317.38	\$4 004 00	\$8,202.00	\$15.05	\$12.61 \$12.61	\$26.99	\$58.43	\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	\$119.14	\$122.31	\$413.00	\$947.69	92,107.00	\$4.37		\$12.61	\$12.61	\$26.99	\$58.43	\$3.17	\$3.17	\$61.19	\$61.19	\$3.50	\$238.28	¢507 00	\$788.00	\$2.72		\$12.61	\$12.61	\$26.99	\$58.43	\$18.76	\$30.42	\$38.87	\$51.27	\$3.50	\$578.97		\$717.60	\$6.46		\$12.61	\$12.61	\$26.99	\$58.43	30.70	\$15.13	\$17.77	\$23.98	\$3.50	<b>GA</b> \$112.77	:
¥ 5	+	\$12,344		\$0.00	\$32.16	\$54.09	×	<b>S S</b>	. ₩	NA NA	NA	¥	W	NA	¥	$^{\dagger}$	\$27.97		\$0.00	\$0.00	\$32.16	\$54.09	X	×	NA	NA	X S	N S	NIA	\$1,204	\$12.62		\$0.00	\$0.00	\$32.16	\$54.09	NA	X	NA	Z.	N 5	× ×		\$1,112.02	\$12.06		\$0.00	\$0.00	\$32.16	\$54.09	5	N K	X X	¥	¥.	₹ ₹	
X S	20	\$11,517	\$74.44	\$0.00	\$32.24	\$54.23	NA	¥	¥	ΝA	NA	NA NA	NA	NA	¥	92,330	\$23.89		\$0.00	\$0.00	\$32.24	\$54.23	NA	¥	NA	NA	X S	N S	N	\$1,101	\$14.04		\$0.00	\$0.00	\$32.24	\$54.23	NA.	NA.	NA	N.	N S	<b>¥ ¥</b>		\$1,131.09	\$16.15		\$0.00	\$0.00	\$32.24	\$54.23	5	N K	: <sub>K</sub>	X	¥	₹ <b>5</b>	
¥ §	N	\$7,182.00	\$60.42	\$0.00	\$32.16	\$54.09	X	3	×.	NA.	¥	NA.	NA	NA	¥.	φ1,00.200	\$18.35		\$0.00	\$0.00	\$32.16	\$54.09	NA	¥	NA	NA	N S	Z S	2	\$744.38	\$15.02		\$0.00		\$32.16	\$54.09	NA	NA.	NA	¥.	Z 3	<b>₹</b>		\$686.84	\$13.48		\$0.00	\$0.00	\$32.16	\$54.09	5	X X	: <sub>K</sub>	NA	<b>⊼</b>	NS ⊗	;
X S		\$7,676.00	€	\$0.00	\$32.10	\$54.00	NA	3	¥	NA.	NA.	NA.	NA	NA	¥		\$14.10		\$0.00	\$0.00	\$32.10	\$54.00	NA	¥	NA	NA	N S	N S	25	\$720.38	\$12.98		\$0.00	\$0.00	\$32.10	\$54.00	NA A	NA.	NA	¥.	N S	<b>₹</b>		\$720.38	\$12.98		\$0.00	\$0.00	\$32.10	\$54.00	5	N A	: N	N.	<b>⊼</b>	× δ	;
X S		\$11,132		\$0.00	\$32.25	\$54.26	NA	3	X.	NA	NA	NA	NA	NA	¥.	\$2,002	\$9.75		\$0.00	\$0.00	\$32.25	\$54.26	NA A	NA.	NA	NA	NA S	N S	2	\$967.70	\$8.13		\$0.00	\$0.00	\$32.25	\$54.28	¥	NA	NA	Z.	Z 3	<b>X X</b>		\$904.49	\$19.14		\$0.00	\$0.00	\$32.25	\$54.26	3	8 8	: <sub>K</sub>	A	¥	SC	-
¥ §	N	\$8,015	\$49.80	\$0.00	\$32.17	\$54.13	X	\$ \$	×.	NA	NA	NA	NA	NA	¥.	\$2,124	\$13.45		\$0.00	\$0.00	\$32.17	\$54.13	X	NA.	NA	NA	NA S	Z S	<u> </u>	\$760.20	\$5.89		\$0.00	\$0.00	\$32.17	\$54.13	NA A	NA	NA	Z.	8	Z Z		\$840.61	\$6.88		\$0.00	\$0.00	\$32.17	\$54.13	3	Z Z	: <sub>K</sub>	NA	¥	<b>₹</b>	!

## NETWORK ELEMENTS AND OTHER SERVICES

NRC - Channel Activation VG - 1st NRC - Channel Activation VG - Add'l	NRC - DS1 Channelization - Add'l	NRC - DS1 Channelization - 1st	2-wire ISDN(BRITE card) per month	VG interface card per month	OCU-DP(data) interface card per month (2.4-64kbs)	DS1 Channelization	THING DOWNTO I COMBINATION OFFICE TO COMPOSITION OF DISCOMPLANT	NRC- DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - A	NRC-DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge - Add'l	NRC-DS3/STS-1 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	Channelization - Incremental Charge Manual Svc Order - Disconnect -	- DS3 Channelization - Incremental ChargeManual Svc Order - Add'l	NRC - DS3 Channelization - Incremental ChargeManual Svc Order - 1st	NRC - Electronic Svc Order, per LSR	NRC - Channel Activation - Addl		NRC - DS3 Channelization - 1st	NRC - Ordinarily Combined in GA (Note 5)	DS3 Interface per month (DS1 COCI)	DS3 Channelized System per month	Channelization:	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC- OC-48 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-OC-48 COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-Add	NRC - OC48 - Interface OC12 on OC48 - Incremental ChargeManual Svc Order-1st	NRC - OC-48 - Incremental ChargeManual Svc Order-Add'l	NRC - OC-48 - Incremental Charge-Manual Svc Order-1et	NRC - OC48 - Interface - Manual Svc Order vs Electronic-Disconnect-1st	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-Add'l	NRC - OC48 - Facility Termination-Manual Svc Order vs Electronic-Disconnect-1st	NRC - Electronic Svc Order, per LSR	NRC - OC48 - Interface OC12 On OC48 - Disconnect - Add'l	NRC - OC48 - Facility Termination - Disconnect - Add'l	NRC - OC48 - Facility Termination - Disconnect - 1st	NRC - OC48 - Interface OC12 on OC48 - Add'l	NRC - OC48 - Interface OC12 on OC48 - 1st	NRC - OCA8 - Facility Termination - Addil		Interoffice Channel - OC12 interface on OC48 Facility	Interoffice Channel - OC48 - per Facility Termination	Interoffice Channel - OC48 - per Mile	Interoffice Channel - OC48	NRC- OC-12 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC- OC-12 COMBINATION - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - Add'i	NRC-OC-12 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-Addi  NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - OC12 - Incremental Cost-Manual Svc. Order vs. Elect-Disconnect-1st		NRC -0C12 - Incremental Charge - Manual Svc Order - 1st	NRC - Electronic Syc Order, per LSR	NRC - OC12 - Facility Termination - Disconnect - 1st	ENHANCED EXTENDED LINKS (EELs)
1D1VG 1D1VG	MQ1	MQ1	UC1CA	1D1VG	1D1DD		0.000	UNCCC	UNCCC	UNCCC	NAMIOS	SOMAN	SOMAN	SOMAN	SOMEC	101DU	EDW EDW	MQ3		UC1D1	MO3		CCC	UNCCC	UNCCC	0000	SOMAN	NAMOS	NAMOS	NAMOS	NAMOS	SOMAN	NAMOS	SOMEC	TRA	TBA	TBA	TBA	TBA	TRA	100	TBA	TBA	1L5XX		UNCCC	UNCCC	UNCCC	UNCCC	SOMAN	SOMAN	NAMOS	SOMAN	SOMEC	TBA	USOC
N N	NA :	¥	\$3.41	\$0.8586	\$1.66		<b>\$0.00</b>	\$0.00	\$32.11	\$54.03	Š	<b>X X</b>	¥	NA.	Ν	¥ 5	N N	<b>X</b>			\$225.36		\$0.00	\$0.00	\$32.11	en 03	NA.	ΝA	¥ ;	Z 3	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	₩	NA	¥ ₹	N S	× ×	M	NA	N S	Z Z	2	\$1,424	\$11,691	\$30.65		\$0.00	\$0.00	\$32.11	\$54.03	NA	¥	ΝA	NA :	X 5	NA NA	2
X X	NA.	¥	\$3.86	\$1.46	\$154./4	2	6	\$16.77	\$32.75	\$32.75	5	<b>X X</b>	NA.	NA	ΝĀ	¥ 5	Z Z	¥		\$14.51	\$222.61		\$16.77	\$16.77	\$32.75	37 003	¥	NA	¥ ;	2 3	3	. ₹	NA	<b>₹</b>	Z 5	<b>X X</b>	¥	NA	N S	N N	5	\$1,208	\$12,554	\$34.66		\$16.77	\$16.77	\$32.75	\$32.75	Š	¥	¥	NA :	X S	N N	<b>P</b>
\$12.02 \$8.66	\$137.06	\$206.09	\$3.71	\$1.17	\$1.86	2	-	\$12.61	\$26.99	\$58.43	\$10.00	\$30.09	\$38.45	\$50.72	\$3.50	\$8.66	\$1/1./2	\$316.28		\$11.02	\$182.04		\$12.61	\$12.61	\$26.99	660 45	\$18.03	\$18.03	\$18.03	\$18.03	\$37.55	\$37.55	\$37.55	\$3.50	\$37.55	\$119.14	\$122.31	\$317.38	\$539.36	\$1,034.00	2	\$1,137.00	\$11,134.00	\$25.70		\$12.61	\$12.61	\$26.99	\$58.43	\$18.03	\$18.03	\$37.55	\$37.55	\$3.50	\$122.31	GA
N N	NA :	₹	\$4.04	\$1.40	\$2.94		90.00	\$0.00	\$32.16	\$54.09	3	<b>Z</b>	NA.	NA.	N <sub>A</sub>	¥ ₹	N N	<b>X</b>		\$8.52	\$236.32		\$0.00	\$0.00	\$32.16	en 200	¥	NA.	¥ ₹	Z 5	N N	\	NA	¥ ₹	N S	₹ ₹	₹	NA :	N S	Z Z	2	T	t	-		\$0.00	\$0.00	\$32.16	\$54.09	X	¥	¥	NA :	X 5	N N	<b>Ξ</b>
N N	NA	¥	\$4.18	\$1.62	\$3.12	200	90.00	\$0.00	\$32.24	\$54.23	N	Z X	N	NA	N <sub>A</sub>	X S	Z Z	X.		\$7.55	\$245.84		\$0.00	\$0.00	\$32.24	9 6 1 3 3	N	NA.	₹ 5	K 5	N N	×.	NA	X S	¥ 3	××	NA	¥.	<b>X</b> 3	N N	5	\$1,451	\$14,950	\$128.59		\$0.00	\$0.00	\$32.24	\$54.23	NA	X	NA	NA :	X S	N A	<b>5</b>
NA NA	NA.	₹	\$3.88	\$1.45	\$2.86		60.00	\$0.00	\$32.16	\$54.09	3	₹ ₹	NA NA	NA.	NA.	¥ 5	N X	¥		\$5.58	\$229.30	İ	\$0.00	\$0.00	\$32.16	en 4 00	N	N <sub>A</sub>	¥ s	2 5	NA NA	NA.	NA	¥ ₹	N S	¥ ¥	¥	NA :	NA 5	2 2	-	\$1,351.00	\$11,480	\$102.43		\$0.00	\$0.00	\$32.16	\$54.09	NA	¥	¥	NA S	X 5	N N	SW
X X	NA	¥	\$3.76	\$1.64	\$2.88	9	#0.00	\$0.00	\$32.10	\$54.00	3	<b>3</b> ¥	NA	NA	NA	¥ 5	N X	¥		\$4.61	\$226.81	İ	\$0.00	\$0.00	\$32.10	en 4 00	N	NA.	¥ ;	N 5	NA NA	NA.	NA	¥ §	N S	× ×	¥	NA :	NA 5	N N	2	+	6	\$120.02		\$0.00	\$0.00	\$32.10	\$54.00	NA	¥	NA.	NA :	X 5	N N	N <sub>C</sub>
¥ ¥	NA	¥	\$4.21	\$1.47	\$2.34	9	00.00	\$0.00	\$32.25	\$54.26	NA.	X X	¥	NA A	NA.	X S	3 8	<b>X</b>		\$11.99	\$200.01		\$0.00	\$0.00	\$32.25	96 78	NA	NA.	¥ ;	2 3	S &	N.	NA	¥ ₹	N S	× ×	¥	NA .	N S	Z Z	2	\$1,561	\$967.58	\$45.92		\$0.00	\$0.00	\$32.25	\$54.26	NA	¥	NA	NA :	X 5	NA NA	SC
N N	NA	¥	\$3.33	\$1.25	\$2.46		90.00	\$0.00	\$32.17	\$54.13	×	<b>X X</b>	A	NA	¥	¥ 5	Z Z	¥		\$3.91	\$222.98		\$0.00	\$0.00	\$32.17	en 43	¥	ΝĀ	¥ 5	8 3	<b>S S</b>	¥	NA	¥ 5	N S	<b>X X</b>	NA.	NA	N S	N N	5	\$1,170	\$11,632	\$106.55		\$0.00	\$0.00	\$32.17	\$54.13	×	¥.	NA	X.	X S	Z Z	Z

H										Н							1										-				-										I		Ŧ				Ŧ	Ŧ	E	Н			
NRC - Clear Channel Capability (B8ZS/SF) Option - Subsequent - Manual Service Orde	NRC - Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel - NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel -	NRC - Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel -	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - Manual Service Or	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - Manual Service O	NRC - Clear Channel Capability (B82S/ESF) Option - Subsequent - per DS1 Channel NRC - Clear Channel Capability (B82S/ESF) Option - Subsequent - per DS1 Channel	1	- Clear Channel Capability (B8ZS/ESF) Option -	Optional Features & Functions:	NRC- Node - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC- Node - "Switch As Is" Conversion Charge - Disconnect - 1st	NRC-Node - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - Node - Incremental Charge - Manual Service Order - Disconnect - Add'l	NRC - Node - Incremental Charge - Manual Service Order - Disconnect -1st	NRC - Node - Incremental Charge - Manual Service Order - Add'l	NRC - Node - Incremental Charge - Manual Service Order - 1st	NR.C Node - Disconnect - Addil	NRC - Node - Add"	NRC - Node - 1st	NRC - Ordinarily Combined in GA:	Node (Synchronet)  Node per month	2	NRC- DCS COMBINATION - "Switch As Is" Conversion Charge - Disconnect - Add"	NRC-DCS COMBINATION - "Switch As Is" Conversion Charge - Add"	NRC-DCS COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-D3 D3C Terrification "Switch As Is" Conversion Charge (Note 6)	NRC- DS3 DSC Termination with DS1 Switching - Disconnect - 1st	NRC- DS3 DSC Termination with DS1 Switching - Add'l	Termination with DS1 Switching - 1st	NRC- DS1 DSC Termination with NRC- DS1 Switching - Disconnect - Add"	Termination with NRC- DS1 Switching - Add'l	hing - 1st	Disconnect -	NRC- DS1 DSC Termination with DS0 Switching - Disconnect - 1st	Termination with DS0 Switching -	- Customer Configuration E	NRC - Ordinarily Combined in GA:	DS3 DSC Termination with DS1 Switching	DS1 DSC Termination with DS1 Switching	Access to DCS - Customer Reconfiguration (FlexServ)		NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 COMBINATION - "Switch As Is" Conversion Charge - Add NRC- DS1 CONVERSION - Add NRC- DS1 CONVERSION - "Switch As Is" Conversion Charge - Add NRC- DS1 CONVERSION - "Switch As Is" Conversion Charge - Add NRC- DS1 CONVERSION - "Switch As Is" Conversion Charge - Add NRC- DS1 CONVERSION - "Switch As Is" Conversion - "Switch As Is" Conversion - "Switch As Is" - "Switch As Is" Conversion - "Switch As Is" - "	NRC-DS1 COMBINATION - "Switch As Is" Conversion Charge - 1st	NRC-All Existing Combination "Switch As Is" Conversion Charge (Note 6)	NRC - DS1 Channelization - Incremental ChargeManual Svc Order - Disconnect - Add	- DS1 Channelization - Incremental ChargeManual Svc Order DS1 Channelization - Incremental ChargeManual Svc Order -	NRC - DS1 Channelization - Incremental ChargeManual Svc Order - 1st	- Electronic Svc Order, per LSR			NRC - Channel Activation OCI-DP- Add"	ENHANCED EXTENDED LINKS (EELS)
	CCOSF		CCOSF	SOMAN			CCOEF	CCOEF		UNCCC	UNCCC	UNCCC	500	SOMAN	SOMAN	SOMAN	SOMO	LINCAL	UNCNI	UNCNT		UNCNT		UNCCC	UNCCC	UNCCC																UNCCC	UNCCC	UNCCC	5	_	SOMAN	SOMAN	SOMEC	UCICA	UCICA	10100	USOC
\$29.23	\$1.99	\$23.81	\$184.85	\$3.93	\$29.23	\$0.77	\$23.81	\$184.85		\$0.00	\$0.00	\$54.03		NA	NA	X.	NA S	N S	Z X	NA		\$15.77		\$0.00	\$32.11	\$54.03	100	T BD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	\$0.00	\$0.00	\$54.03		NA A	<b>⊼</b> ₹	: X	NA.	NA	¥ ₹	N A	AL.
\$21.73	\$2.07	\$23.82	\$184.92	\$3.87	\$21.73	\$0.80	\$23.82	\$184.92		\$16.77	\$16.77	\$32.75	200	N	¥	₹ 5	NA S	N S	\$ ₹	¥		\$16.35		\$16.77	\$32.75	\$32.75	\$24.30	\$31.06	\$39.64	\$51.50	\$16.73	\$25.36	\$37.23	\$24.98	\$39.64	\$51.50	\$3.44	\$2 97	\$154.31	\$12.23	£20 72	\$16.77	\$16.77	\$32.75	3	¥	<b>₹</b>	<b>.</b> ×	: N	NA :	<b>₹</b>	N K	Z [2
\$29.33	\$2.03	\$23.78	\$184.62	\$3.93	\$29.33	\$0.79	\$23.78	\$184.62		\$12.61	\$12.61	\$58.43		NA	\$3.87	A E	\$21.73	\$3 05	\$4.47	\$47.19		\$13.98		\$12.61	\$26.99	\$58.43	\$ZU.10	\$20.16	\$31.49	\$32.07	\$12.10	\$17.49	\$18.07	\$20.16	\$20.16	\$32.07	\$3.36	\$3 91	\$151.85	\$8.64	20 00	\$12.61	\$10.99	\$58.43		\$0.00	\$10.70	\$14.75	\$3.50	\$8.66	\$12.02	\$8.66	GA C
\$29.24	\$1.99	\$23.82	\$184.91	\$3.94	\$29.24	\$1.99	\$23.82	\$184.91		\$0.00	\$0.00	\$32.16	2	¥	¥	¥ ;	N S	N S	\$ ₹	¥		\$18.11		\$0.00	\$32.16	\$54.09	- 00	T BB	TBD	TBD	TBD I	TBD	TBD	TBD 6	TBO	TBD	TBD	Ŧ.	TBD	TBD	Ħ.	\$0.00	\$0.00	\$54.09		¥	<b>₹</b>	× ×	: N	NA :	<b>₹</b>	N K	X X
\$29.20	\$1.97	\$23.70	\$184.65	\$3.92	\$29.20	\$1.97	\$23.70	\$184.65		\$0.00	\$0.00	\$54.23	200	NA	NA	¥ :	NA S	N N	<b>X X</b>	NA.		\$15.43		\$0.00	\$32.24	\$54.23	100	T BB	TBD	TBD	TBD I	TBD	TBD	TBD 6	TBD TBD	TBD	TBD	Ā	TBD	TBD	Ŧ,	\$0.00	\$0.00	\$54.23		NA A	X X	- X	: N	NA	<b>⊼</b> 3	N N	<b>5 5</b>
\$29.33	\$1.96	\$23.78	\$184.60	\$3.93	\$29.33	\$1.96	\$23.78	\$184.60		\$0.00	\$0.00	\$32.16	9	¥	¥	¥ ;	N S	N S	\$ ₹	¥	-	\$16.15		\$0.00	\$32.16	\$54.09	00	# #	TBD	TBD	TBD I	TBD	TBD	큠	<b>B B</b>	d de	TBD is	평	TBD	de la	ġ	\$0.00	\$0.00	\$54.09		¥	<b>₹</b>	× ×	: N	NA :	<b>₹</b>	N K	NS NS
\$29.33	\$1.99	\$23.60	\$184.76	\$3.93	\$29.33	\$1.99	\$23.60	\$184.76		\$0.00	\$0.00	\$54.00	9	NA.	¥.	¥ ;	NA S	2 3	3 3	¥		\$16.00		\$0.00	\$32.10	\$54.00	-	曹	TBD	da Ga	B 0	間	TBD	큠		TBD	TB IS	큠	TBD	B B	4	\$0.00	\$0.00	\$54.00		N.	¥ ¥	₹ ₹	: ¥	NA	X S	8	N C
\$29.33	\$1.99	\$23.86	\$185.26	\$3.93	\$29.33	\$1.99	\$23.86	\$185.26		\$0.00	\$0.00	\$32.25	9	NA.	¥	¥ ;	N S	2 3	3 3	¥		\$14.55		\$0.00	\$32.25	\$54.26	-	T BB	ΠBD	da Ga	B 0	間	TBD	TBD G	TBO	TBD	TBD	TBN	TBD	TBD	4	\$0.00	\$0.00	\$54.26		¥.	¥ ¥	₹ ₹	: ¥	NA :	X S	2 8	SC
\$29.33	\$2.03	\$23.85	\$185.16	\$3.93	\$29.33	\$2.03	\$23.85	\$185.16		\$0.00	\$0.00	\$54.13		¥	N <sub>A</sub>	¥ ?	NA 5	Z S	3	¥		\$17.11		\$0.00	\$32.17	\$54.13	000	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	T T T T T T T T T T T T T T T T T T T	\$0.00	\$0.00	\$54.13		N <sub>A</sub>	<b>₩</b>	<u> </u>	: <u>N</u>	NA :	<b>₹</b>	N N	, T

	Н															
Where the state Commission has adopted rates for the rate elements contained herein, it is the intent of the parties to reflect such rates in this exhibit and to apply the same consistent with applicable FCC and Commission rules and orders.	6 The "Switch As Is" NRC is a conversion charge. One SAI charge is applicable per circuit.	5 The Ordinarily Combined in GA NRC applies to new combinations within the State of Georgia.	- DS3 Local Channel + DS3 Interoffice Channel + DS3 Channelization System + DS1 Interface Card	- DS1 Loop + DS1 Interface Card + DS3 Channelization System + DS3 Interoffice Channel	- 2-wire VG Loop + Voice Grade Interface Card + DS1 Channelization System + DS1 Interoffice Channel	* Examples:	4 Add together the recurring rates of all the applicable network elements in order to obtain total monthly recurring rate	3 Unapproved rates are subject to true up.	Tennessee - Nashville	N. Carolina - Greensboro, Charlotte	Louisiana - New Orleans	Florida - Miami, Orlando, Ft. Lauderdale	2 New EELs will only be available in the State of Georgia and in density Zone 1 of the following MSAs in the BellSouth Region:	Interim rates subject to true-up.  Interim rates subject to true-up.  Indexim rates subject to true-up.  I Geographically Deaveraged UNE Zones and applicable rates have been established for certain services, as shown in this Agreement. Where Geographically Deaveraged UNE Zones and applicable rates are established. Statewide rates are obsolete. Further, BellSouth is in the process of enhancing its billing systems in order to accomplate this Geographically Deaveraged UNE Zone Rate Structure. Until these enhancements are accomplished, estimated to be mid 2001, the UNE Zone 1 rate will be billed for all services residing in Zones 1, 2, 3 or 4, i.e., Rates for services residing in UNE Zone 4, where applicable, will not be billed. Once billing enhancements are complete, all applicable UNE Zone rates reflected in this Agreement will be billed. Reterence Internet Vestate http://www.interconnection.bellsouth.com/become_clec/ doss/interconnection/deavurzis.pdf to view Geographically Deaveraged UNE Zone Designations by Central Office.	NRC - Clear Channel Capability (B8ZS/ESF) Option - Subsequent - Manual Service Or	ENHANCED EXTENDED LINKS (EELs)
U	ouit.	Georgia.	1 Interface Car	Channel	31 Interoffice Ch		ain total monthly						ollowing MSAs	() — U.	SOMAN	USOC
			ď.		nannel		recurring rate						in the BellSout		\$3.93	AL
							•						h Region:		\$3.87	2
															\$3.93	GA
															\$3.94	Ŕ
															\$3.92	LA
															\$3.93	SW
															\$3.93	NC
															\$3.93	SC
															\$3.93	ī

NRC - Aggr	NRC - 1st	eatures	Incremental Charge - Manual Service		NRC - 1St	120	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Addi'i	NRC - 1st	Multiple Inter LATA Carrier Routing per Carrier Requested per 8XX #	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Addill	NRC - 1st	- Incremental Charge - Manual Service Order -	NRC - Incremental Charge - Manual Service Order - Add'l	- Manual Service Order -	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Addl'I	NRC - 1st	0	NRC - Incremental Charge - Manual Service Order - Disconnect	NBC Incremental Charge Manual Service Order Addit	NRC - Disconnect Charge - Addi	NRC - Disconnect Charge - 1st	NRC - Addit	NRC - 1St	Per 8XX # Established w/o POTS (w/8XX No.) Translations	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Addl'I	NRC - 1st	Reservation Charge per 8XX number reserved	with Optional Complex Features, per message	GR OCIGEILING OVC. WIT OF GROOT	8XX Access Ten Digit Screening Svc. W/POTS No. Delivery	···/Ontional Commission England	8XX Access Ten Digit Screening Svc. W/800 No. Delivery	with Optional Complex Features, per query	per query per query	8XX Access Ten Digit Screening Svc. W/POTS No. Delivery	for 8XX Numbers, with Optional Complex Features, per query		8XX Access Ten Digit Screening Svc. W/8XX No. Delivery	8XX Access Ten Digit Screening Service (Note 1)	message	provisioned	Enhanced Optional Daily Usage File: Message Processing , Per Message	Enhanced Optional Daily Usage File (EODUF)	RC - ADUF, Data Transmision (CONNECT:DIRECT), per message	RC - ADUF, Message Distribution, per Magnetice Tape provisioned	RC - ADUF Message Processing per message	Access Daily Heave Eile (ADITE)	Provisioned	RC- OSS OLEC Daily Usage File: Message Processing, Per Message	age	Recovery of incremental OSS costs, per CLP, per month	Operational Support Systems	DESCRIPTION
NBFUX	N8FDX		SOMAN	NATAN	N8TAX	1	SOMAN	SOMAN	N8FMX	N8FMX		SOMAN	SOMAN	N8FCX	N8FCX	SOMAN	SOMAN	SOMAN	N8FTX	N8FTX	N8FTX	N8FTX	000	NAMOS	SOMAN	NAMOS	N/A	NA A	NA A		SOMAN	SOMAN	N8R1X	N8R1X		NA.	N/A	WA	N/A	N/>	N/A	N/A		N/A	N/A	147	N/A														USOC
×	\$5.69		NA.	\$07.27 16.0¢	\$8.10	,	NA.	NA	\$3.81	\$6.66		NA	NA	\$2.85	\$5.69	\$17.75	N.	\$27.37	\$0.97	\$10.04	\$1.97	\$15.88		\$17.75	\$27.37	\$0.97	\$10.04	\$1.97	\$15.88	9	NA.	\$27.37	\$0.97	\$7.13		Z :	NA	NA	S S	2	NA	NA		NA	¥	0000	\$0.0005	¥	NA	\$0.004		\$0.001	NA S	\$0.004	\$0.00004	\$55.19	\$0.0033	\$0.0002	NA		AL
NA.	NA	1	¥ ₹	NA	₹ ₹		Ā	NA	NA	X		NA	NA	NA	NA	NA	N	NA	Ā	NA	NA	NA A		N 3	5	NA	NA	<u> </u>	\$		NA NA	NA	NA	NA		X S	NA	NA	3	NA	NA	NA		NA	¥	\$0.00000	\$0.0006531	ΝĀ	NA	\$0.004		\$0.001	NA S	\$0.004	\$0.001	\$54.95	\$0.004	\$0.008	NA		F
\$4.46	\$4.72		NA S	\$0.76	\$7.33		¥	¥	\$2.99	\$5.22		XA	NA A	\$2.23	\$4.46	NA.	¥.	\$18.94	¥	¥	\$1.45	\$12.81	5	N S	ψ10.94	£1004	N N	\$1.45	\$12.81		¥	\$18.94	\$0.76	\$6.57		¥ ;	NA	¥	<u> </u>	NA	NA A	NA A		NA	¥	#0.000 0.000	\$0 0004868	¥	NA A	\$0.0034555		\$0.0000434	NA SOL	\$0.0136327	\$0.0000434	\$28.85	\$0.0082548	\$0.0001275	NA A		GA
\$6.97	\$6.97		Z ₹	\$1.19	\$11.24		NA	NA.	\$4.67	\$8.16		NA	NA	\$3.49	\$6.97	NA	N	NA A	Ā	NA	\$3.22	\$30.59	157	N 3	× §	N A	NA NA	\$3.22	\$30.59		N	NA	\$1.19	\$10.05		N S	NA	NA	NA NA	2	\$0.0011	\$0.0010		\$0.0011	\$0.0010	167	NA	NA.	NA	\$0.004		\$0.001	NA.	\$0.004	\$0.000365	\$55.68	\$0.0032357	\$0.0008611	NA		₹
\$4.27	\$4.27		N G	\$0.73	\$7.01	)	¥	NA	\$2.86	\$5.00		NA	NA	\$2.14	\$4.27	\$11.40	N.	\$18.14	\$0.73	\$8.30	\$1.39	\$12.27	-	\$11.40	\$10.14	\$0.73	\$8.30	\$1.39	\$1227	9	¥	\$18.14	\$0.73	\$6.29		X S	NA	¥	¥ ¥	N	NA	Ą		NA	¥	0.00000	\$0 0005305	¥	NA	\$0.004		\$0.0000305	N S	\$0.004	\$0.0000300			-			₽
\$5.63	\$5.63		NA S	\$0.96	\$9.42	;	NA.	¥	\$3.77	\$6.59		NA	NA A	\$2.81	\$5.63	\$16.05	Z,	\$25.52	\$0.96	\$11.32	\$1.93	\$17.04	0.00	\$16.05	\$20.52	\$U.96	\$11.32	\$1.93	\$17.04		¥	\$25.52	\$0.96	\$8.46		¥ ;	NA	¥	š	25	NA.	NA		NA	₹	0.0000	\$0 0005321	¥	NA A	\$0.004		\$0.001	NA S	\$0.004	\$0.0000354	\$54.62	\$0.0032089	\$0.0001179	NA		MS
₹	\$5.63		¥20.54	\$0.96	\$8.01		₹	¥	\$3.77	\$6.59		¥	¥	\$2.82	\$5.63	¥	₹	\$41.35	₹	¥	\$2.73	\$23.82	3	<b>E 3</b>	WA 1.33	\$41.2F	\$	\$2.73	\$23.82		₹	\$26.94	\$0.96	\$7.05		<b>₹</b> ;	£	£	≨ ≨	N .	\$0.00431	\$0.00383		\$0.00431	\$0.00365	0.0000	_	₹	¥	\$0.004		\$0.001	₩.	\$0 004	\$0.0004	\$54.61	\$0.0032	\$0.0003	\$305.00		NC
\$5.64	\$5.64		NA 420	\$0.9583	\$7.34		¥	¥	\$3.78	\$6.60		NA.	NA	\$2.82	\$5.64	NA	¥	¥	₹	\$42.95	\$2.73	\$22.63	5	N 5	5	N A	\$42.95	\$2.73	\$22.63	9	¥	\$27.84	\$0.9583	\$6.38		¥ ;	N	¥	\$ ₹	N/	NA	¥		NA	¥	\$0.0000EF	\$0 0005227	¥	NA	\$0.004		\$0.001	N S	\$0.004	\$0.0000357	\$54.72	\$0.0032344	\$0.0002862	NA A		SC

\$48.50 \$0.50 NA

\$3.50 \$2.00 NA

\$3.00 \$1.50 NA

\$3.00

NA \$0.008 \$0.004 \$54.95 \$0.001

į

\$30.00 \$0.50 NA

 $\exists \ \exists$ 

 $\mathbb{R}$ 

 $\exists \exists$ 

\$0.000519

₹ ₹

\$0.004 NA

\$0.004 NA \$0.001

\$1.50 NA NA NA NA

\$67.50 \$1.50 NA NA NA NA

NRC - Incremental Charge-Manual Svc Order - NRC- addi NRC - Incremental Charge-Manual Svc Order - NRC-Disconnect	NRC - Incremental Charge-Manual Svc Order - NRC - 1st	NRC - Disconnect Charge - Add'l	NRC - Disconnect Charge - 1st	NRC - Add'l	NRC - 1st	Directory Transport - Local Channel DS1, per month	Directory Transport	Recording Charge per Branded Announcement - Disconnect - Subsequent	Recording Charge per Branded Announcement - Disconnect - Initial	EBAS or 0- automation loading, per NAV shelf	AABS or back-end loading, per IVS	DRAM or front-end loading, per TOPS switch	Professional recording of name (DA and OCP alone)	Professional recording of name (DA alone)	Directory Assistance Access Service Calls, per call	Number Services intercept per intercept Query Update	number services intercept per query	Call Completion Access Term charge per completed call	Directory Assist Call Completion Access Svc (DACC), per call attempt	DIRECTORY ASSISTANCE SERVICES	verification and Emergency Interrupt, per call	Verification, per call	verification and Emergency Interrupt, per minute	vernication, per minute	INWARD OPERATOR SERVICES	Recording Charge per Branded Announcement – Disconnect – Subsequent	Disconnect -		AABS or back-end loading, per IVS	DRAM or front-end loading, per TOPS switch	Professional recording of name (DA and OCP alone)	Professional recording of name (OCP alone)	Fully Automated Call Handling per call - Using Foreign LIDB	Fully Automated Call Handling per call - Using BST LIDB	Operator Provided Call Handling, per call	Call Completion Access Termination Charge per call attempt	Operator Provided Call Handling per min - Using Foreign LIDB	Call Completion Access Termination Charge per call attempt	Operator Provided Call Handling per min - Using BST LIDB	OPERATOR CALL PROCESSING		NRC	CCS7 Signaling Usage Surrogate, per link per LATA per mo (9)	(applicable when measurement and billing capability exists.)	CCS7 Signaling Usage, per TCAP message	(applicable when measurement and billing capability exists.)	CCS7 Signaling Usage, per ISUP message	CCS7 Signaling Termination, per STP port per month	NRC - Incremental Charge - Manual Service Order - Disconnect	NRC - Incremental Charge - Manual Service Order	NRC - Disconnect	NRC	CCS7 Signaling Connection, per link (B link) (also known as D link) per month	NDC - Incremental Charge - Manual Service Order - Disconnect	NEC - Discollect	NRC Disconnect	NDC	CCG/ Gignaling Connection not link / Link) not month	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Incremental Charge - Electronic Service Order	LIDB Originating Point Code Establishment or Change - NRC	LIDB Validation per query	LIDB Common Transport per query	DESCRIPTION
SOMAN	SOMAN	WA	NA A	N/A	N/A	N/A		N/A	N/A	N/A	NA	N/A	N/A	WA	N/A	WA	N/A	NA NA	NA NA		WA	\[\frac{1}{2}\]	Z A	WA		WA	NA	USOD2	USOD2	USOD2	USOD1	USOD1	N/A	NA A	N/A	WA	N/A	NA	NA										SOMAN	SOMAN			CONSTR	NAMOS	NAMOS				SOMAN	SOMAN		N/A	OQU	OQT	OSOC
\$29.27	\$61.99	\$32.18	\$46.28	\$442.84	\$503.57	\$35.52		\$9.61	\$9.61	\$270.00	\$225.00	\$250.00	\$7,000.00	\$3,000.00	\$0.275	NA.	\$0.0235		\$0.10		NA	<b>X</b>	\$1.16	\$1.16	)	\$9.61	\$9.61	\$270.00	\$225.00	\$250.00	\$7,000.00	\$7,000.00	\$0.13	\$0.11	8	\$0.08	\$1.25	\$0.08	\$1.21		000	\$62.00	\$3/6.12	201	\$0.0001		\$0.00004	\$148.72	\$16.31	\$25.93	\$135.70	\$171.98	\$18.79	\$16.31	# 100.70	\$171.90	\$171.09	61070	X	\$25.93	NA	\$64.36	\$0.041003	\$0.00004	AL
NA NA	NA.	¥	¥	\$226.44	\$242.45	\$43.64		N S	NA O	\$270.00	\$225.00	\$250.00	\$7,000.00	\$3,000.00	\$0.275	NA.	\$0.01	Ž X	\$0.10		\$1.00	\$0.80	Ž.	×		¥	¥.	\$270.00	\$225.00	\$250.00	\$7,000.00	\$7,000.00	\$0.10	\$0.10	¥	¥	\$1.00	¥	\$1.00		000	\$62.00	\$64.00		\$0.00004		\$0.00001	\$113.00	¥ :	NA S	NA S	\$400.00	\$5.00	2 2	5	\$400.00	\$400.00	96	NA	×.	NA	NA.	\$0.041003	\$0.0003	2
N X	\$44.22	NA NA	¥	\$312.89	\$356.15	\$38.36		X 5	NA O	\$270.00	\$225.00	\$250.00	\$7,000.00	\$3,000.00	\$0.275	× ×	\$0.009/49/		\$0.10		NA	. ₹	\$0.921083			¥	NA	\$270.00	\$225.00	\$250.00	\$7,000.00	\$7,000.00	\$0.0976984	\$0.0776409	¥	X	\$1.02	¥	\$0.9680296		000	\$62.00	\$340.67	9	\$0.0000870		\$0.0000354	\$133.99	N S	\$18.94	NA S	\$131.96	\$17.05	\$10.94	91001	\$101.90	\$131.06	977	×	\$18.94	NA	\$50.30	\$0.0105974	\$0.0000338	GA
NA NA	NA.	NA NA	¥	\$546.94	\$637.46	\$36.32		N 5	NA O	\$270.00	\$225.00	\$250.00	\$7,000.00	\$3,000.00	\$0.275	\$0.0055	\$0.0086	NA NA	\$0.10		\$1.111	\$1.00	S	<b>X</b>		X	NA NA	\$270.00	\$225.00	\$250.00	\$7,000.00	\$7,000.00	\$0.1071	\$0.0856	K	A	\$1.6249	¥	\$1.6016		00.00	\$62.00	\$329.98		\$0.000102042	-	\$0.	\$174.08	X.	NA S	NA S	\$354.95	\$16.31		2 5	\$10 AIV	\$10.01	94004	X	. ₹	NA	\$107.60	\$0.00938	\$0.00006	হ
\$19.48	\$42.34	\$23.32	\$33.02	\$298.29	\$339.69	\$43.83		N S	NA O	\$270.00	\$225.00	\$250.00	\$7,000.00	\$3,000.00	\$0.275	×	\$0.02	Ž X	\$0.10	•	¥.	¥ ¥	\$0.86	\$0.86		¥	NA.	\$270.00				\$7,000.00		\$0.10	8	¥	\$0.96	¥	\$0.91		000	\$62.00	\$406.71		\$0.0001052	_	\$(	\$161.99	\$11.40	\$18.14	\$101.10	\$126.34	\$19.48	\$10.14	6101.10	\$120.34	\$1363A	940	NA	\$18.14				\$0.0	5
\$27.41	\$59.58	\$33.02	\$46.85	\$435.28	\$494.83	\$38.91		N S	NA CO	\$270.00	\$225.00	_		\$3,000.00	_	×	\$0.018826	P NA	\$0.10	3	×	<b>X</b>	\$1.14	\$1.14	·	¥	Ą					\$7,000.00		\$0.107288	8	¥	\$1.24	¥	\$1.19		000	\$62.00	\$406.53		\$0.00011		\$(	\$161.12	\$16.05	\$25.52	\$134.08	\$169.72	\$21.58	946.05	\$1.04.00	\$109.72	\$160.73	637	NA	\$25.52	NA		\$0.01421;	\$0.00004	NS
\$1.//	\$86.15	¥	₹	\$462.69	\$534.48	\$35.68		₹ 5	AL COO	\$270.00	\$225.00			-	_				\$0.062		\$0.65	\$0.54	\$1.15	\$1.15	·	₹	¥	\$270.00	\$225.00	\$250.00	\$7,000.00	\$7,000.00	6 \$0.12	4 \$0.11	₹	¥	\$1.24	¥	\$1.20		901.00	\$62.00	\$338.98	9	15 \$0.00009	1		\$132.88	₹ :	₹ 5	¥ ::	\$510.00	\$155.00	2	5	\$310.00	\$510.00	946600	¥	\$26.94	\$62.26	\$91.00	32 \$0.013400	6 \$0.0003	NC
\$3.11	\$87.99	¥	¥	\$462.81	\$534.81	\$37.20		X 5	NA COO	\$270.00	\$225.00	_		\$3,000.00	+	t	\$0.0124036	\$0.08	\$0.10		×	<b>X</b>	\$1.15	\$1.15	·	¥						\$7,000.00				\$0.08	\$1.25	\$0.08	\$1.21		\$0E.00	\$62.00	\$396.55	9	\$0.0001108		\$0.0000452	\$156.33	¥ :	NA S	\$42.95	\$277.07	\$21.79	2 5	νιν νιν	\$42.0E	\$27.70	20, 70	\$27.84	\$27.84	NA	\$61.62	\$0.0141003	\$0.0000442	SC
X §	N <sub>A</sub>	Ā	¥	\$486.83	\$868.97	\$133.81		8 5	NA 0.00	\$270.00	\$225.00	\$250.00	\$7,000.00	\$3,000.00	\$0.275	× ×			\$0.10		\$1.95	\$0.90	Š	<b>X</b>		¥	NA	\$270.00	\$225.00	\$250.00	\$7,000.00	\$7,000.00	\$0.15	\$0.15	\$0.30	¥	¥	¥	¥		#0F:00	\$62.00	\$395.00		\$0.00005	-	40	\$355.00	<b>∀</b> :	N S	NA S	\$510.00	pendina	2 5	5	\$310.00	\$510.00	945	×	\$91.00	×	¥	\$0.041003	\$0.0003	¥

													AIN, S	<u> </u>													AIN - E	AIN, per me		Direct	Director							Direct	DA Int	Acces	Switch	Ħ	I		I	Ŧ		Direct	DESC
Trigger Access Charge, per trigger, per DN, CDP		NRC - Disconnect Trioger Access Charge per trioger per DN 10-Digit PODP	NRC	Access Charge per trigger per DN Off-Hook	9	NRC - Disconnect  Trigger Access Charge, per trigger per DN, Off-Hook Delay	NRC	Trigger Access Charge, per trigger, per DN, Term. Attempt	NRC - Disconnect	12.1	NRC - Disconnect	Service Establishment Charge, per state, initial set-up NRC	AIN, Service Creation Tools	CO. Performed Session, per minute  AIN - ReliSouth AIN Toolkit Service	Session per minute	NRC - Disconnect	NRC	NRC - Disconnect	D COURT   DOI 10001 10	NRC - Disconnect		NRC - Disconnect  Port Connection - ISDN Access	NRC	Port Connection - Dial/Shared Access	NRC - Disconnect	Service Establishment Charge, per state, initial set-up	AIN - BellSouth AIN SMS Access Service	AIN, per message	(2.4)	Directory Assistance Database Service, per month	rectory Assistance Database Service (DADS) Directory Assistance Database Service charge per listing	NRC - Manual Service Order - Add'l	NRC - Manual Service Order - 1st	NRC - Incremental Charge - Manual Service Order - Add'l	NRC - Disconnect Charge - Add'l  NRC - Incremental Charge - Manual Service Order - 1st	NRC - Disconnect Charge - 1st	NRC - ISC	Directory Transport-Installation NRC, per trunk or signaling connection	DA Interconnection, per DA Access Service Call	Access Tandem Switching per DA Access Service per call per mile	Fransport per DA Access Service per call	NRC - Incremental Charge - Manual Service Order - Disconnect - Add"l	NRC - Incremental Charge - Manual Service Order - Add'l NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	NRC - Incremental Charge - Manual Service Order - 1st	NRC - Disconnect Charge - Add'l	NRC - Add'l	NRC - 1st	Directory Transport - Dedicated DS1 Level Interoffice per mile per mo  Directory Transport - Dedicated DS1 Level Interoffice per facility termination per mo	DESCRIPTION
2	BAPTO	BAPTM	BAPTM	BAPTD	BAPTD	BAPTT	BAPTT		BAPVX		BAPSC	BAPSC	CAMBP		WA S	CAMRC	CAMRC	CAMAU	CAMAU	CAM1P	CAM1P	CAMUT	CAMDP	0.000	CAMSE	OVIVOE	CAM	CAM		DBSOF	N/A			SOMAN	SOMAN	N/A	NA A	N/A	WA	Z Z	N/A	SOMAN	SOMAN	SOMAN	Z Z	W W	WA	N/A	USOC
407.00	\$117.98	\$27.04	\$49.64	\$27.04	\$49.64	\$27.04	\$49.64	1	\$8,363.00		\$114.22	\$192.69	NA.		\$0.0892	\$35.26	\$142.13	\$70.05	\$141.84	\$27.04	\$64.05	\$27.04	\$64.05		\$114.22	\$107.40		¥		\$128.55	\$0.0446	¥	₹ ₩	NA	\$5.95	\$173.46	\$5.95	\$360.60	\$0.00269	\$0.00003	\$0.0003	\$12.97	\$27.37	\$27.37	\$20.42	\$148.18 \$25.44	\$198.15	\$0.6923	AL.
5	X X	A	Ą	¥	NA	NA	X.	1	8 8		Z.	¥	TBD		N S	N A	¥	¥	NA	¥	NA	¥	¥		N A	N .		(interim)		\$100.00	\$0.001	¥	₹ ₩	NA	X X	AN	\$4.71	2000	NA	\$0.00055	\$0.0003	NA :	<b>X X</b>	NA.	¥ ₹	\$44.18 NA	\$45.91	\$0.6013	P
5	\$70.06	NA	\$19.13	¥	\$114.80	NA	\$19.13	5	\$8,348.00		X.	\$86.74	NA		\$0.0795604	NA NA	\$35.44	¥	\$84.43	¥	\$29.66	¥	\$29.66		4V. C	\$00.0E		¥		\$95.50	\$0.0445	X	<b>₹</b>	NA	\$44.22	AN	\$4.42	200	\$0.00269	\$0.0000100	\$0.0002906	NA :	X X	\$18.94	¥ ₹	\$111.75 NA	\$147.07	\$0.4523 \$78.47	GA
5	X X	NA	NA	¥	NA	NA	N.	5	N N		N.	N <sub>A</sub>	NA	¥	X.	N A	NA.	¥	NA	×	NA	¥	¥		N N	NA		¥		\$120.76	\$0.0193	¥	X X	NA	<b>X X</b>	AN	\$13.32	900	NA	\$0.000783	\$0.000175	NA :	X X	NA.	<b>₹</b> 5	\$231.18 NA	\$298.18	\$0.45 \$55.05	ই
91.01	\$92.99	\$18.60	\$41.08	\$18.60	\$41.08	\$18.60	\$41.08	3	\$8,315.00		\$78.05	\$153.25	NA	\$1.97	\$0.10	\$24.40	\$125.33	\$48.95	\$104.95	\$18.61	\$50.07	\$18.61	\$50.07		\$78.06	645004		¥		\$90.54	\$0.0443	¥	<b>X</b>	\$4.23	\$130.05	NA	\$4.23	9	N	\$0.0000175		\$8.06	\$18.14	\$18.14	\$16.34	\$106.69	\$140.49	\$0.78	, L
<b>910.11</b>	\$106.90	\$37.70	\$39.30	\$37.70	\$39.30	\$37.70	\$39.30	5	\$8,379.00		\$135.96	\$169.31	Ą	\$2.09	\$0.0975650	\$45.77	\$131.54	\$79.91	\$129.83	\$37.70	\$53.47	\$37.70	\$53.47	4.00.00	\$135.06	617403		¥		\$126.17	\$0.0447	¥	×	\$5.85	\$171.49	AN	\$5.85	62 67 73	NA	\$0.0000202	\$0.0002997	\$11.34	\$25.52	\$25.52	\$21.61	\$147.31	\$196.28	\$0.6598	NS.
3	\$149.95	¥	\$72.76	₹	\$72.76	¥	\$72.76	3	\$8,363.00		₹	\$290.05	⋠		\$0.0791		\$172.05	¥	\$200.83	₹	\$86.94	₹	\$86.94		\$294.//	27 VOC\$		₹		\$126.26	\$0.04460	\$10.98	\$407.53	W	<b>₹</b>	NA.	8	<u>.</u>		\$0.0003		П	\$38.07	\$38.07	<b>₹</b>	\$163.75	\$217.17	\$0.5753 \$71.29	K
157	\$150.25	NA	\$73.02	¥	\$73.02	NA	\$73.02	3	\$8,333.00		Z.	\$291.41	NA	\$2.07	\$0.0942966	\$0,0028 NA	\$172.26	¥	\$202.08	X	\$87.29	N	\$87.29		\$296.16	\$206.46	NA	¥		\$127.23	\$0.0444	¥	X X	NA	<b>X X</b>	NA	\$11.00	¢ 407 04	\$0.000269	\$0.0000303	\$0.000327	NA :	\$39.63 NA	\$39.63	<b>₹</b> §	\$162.70	\$216.27	\$0.7598 \$94.98	SC
5	₹ ≱	¥	¥	¥	¥	X	<b>X</b>	5	Z Z		Z.	¥	¥			X X		¥	¥	Ş	NA	¥	¥		Z	NA	NA.	₹ 0	IBD I	¥	¥	¥	<b>X</b>	NA.	<b>₹</b>	NA	<b>₹</b>	N	NA :	N S	₹	¥	Z Z	¥.	<b>₹</b> §	\$100.49	\$100.49	\$23.00	Į

Awire Cross-Connect RC IRC - 1st NRC - 1st NRC - Add! - Manual Serv NRC - Disconnect - Add! NRC - 1st NRC - 1st NRC - 1st NRC - 1st NRC - Add! - Manual Serv NRC - Disconnect - 1st NRC - Serv NRC - 1st NRC - Serv NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - 1st NRC - 1st NRC - 1st NRC - 1st NRC - 1st NRC - Add! - Manual Serv NRC - Add! - Nanual Serv NRC - 1st NRC - 1st NRC - 1st NRC - 1st NRC - 1st NRC - 1st NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - Disconnect - 1st NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - Disconnect - Add! NRC - Disconnect - Add!	Z-Wife Cross-Con RC - 1st NRC - 2st NRC - 1st	A white Cross-Con	Awire Cross-Con RC - 1st	A-wire Cross-Con	Z-Wire Cross-Con   RC   1st     NRC - 1st     NRC - 1st     NRC - 1st     NRC - 1st     NRC - 2st     NRC - 2st     NRC - 2st     NRC - 1st     NRC - 1st     NRC - 1st     NRC - 2st     NRC - 2st	A-wire Cross-Con	Zwine Cross-Con RC IRC IRC IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 1st IRC - 2dd IRC - 1st IRC - 3dd IRC - 3dd IRC - 3dd IRC - 1st IRC - 3dd	Zwine Cross-Con RC RC RC RC RC RC RC RC RC RC RC RC RC	Zwine Cross-Con RC RC RC RC RC RC NRC - 1st NRC - 1st NRC - Add' NRC - Disc NRC - Disc NRC - Disc NRC - Disc NRC - Disc NRC - Disc NRC - St NRC - St NRC - 1st	RC	2-wire Cross-Con  RC  NRC - 1st  NRC - 2dd'   NRC - 1st-  NRC - Dist   Z-WIFE Cross-Con RC NRC - 1st NRC - 1st -	RC RC - 1st NRC - 1st NRC - 1st NRC - 4dd NRC	RC NRC - 1st NRC - 1st -	RC NRC - 1st	2-wire Cross-Con		VIRTUAL COLLOCATION	NRC - Incre	Customized rout NRC	NRC NRC	SELECTIVE ROUTING (Note 5)	* Volume and term	method to t	CALLING NAME (CI	NRC - Disconnect	NRC	Call Event special Stud	NRC	NRC - Disconnect	NRC	Special Study - per AIN	NRC Disc	Monthly Report -	2	Type 1 Nod	NRC - Disconnect	NRC NRC	NRC - Disc	NRC		
INRO - 1st INRO - 1st Manual Service Order INRO - Add'I INRO - Add'I - Manual Service Order INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - Add'I INRO - Disconnect - Add'I INRO - Disconnect - Add'I INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - 1st INRO - Disconnect - Add'I INRO - Disconnect - Add'I INRO - Disconnect - Add'I INRO - Disconnect - Add'I INRO - Disconnect - Add'I	Manual Service Order  Manual Service Order  mect - 1st  mect  mect - Add'l  mect - Add'l  mect - Add'l  mect - Add'l	Manual Service Order  n-Manual	Manual Service Order	Manual Service Order  Alanual Sarvice Order  Alanual Sarvice Order  Mect - St  Mect - Add1	Manual Service Order - Manual Service Order nnect - 1st nnect - Add¹	Manual Service Order - Manual Service Order nnect - 1st nnect - Add1	Manual Service Order  - Manual Service Order  nnect - 1st	Manual Service Order Manual Service Order	Monte Contine Order			nect	nnect - 1st	NRC - Add'l - Manual Service Order	Add'I 1st - Manual Service Order		nect	TION	NRC - Incremental Charge - Manual Service Order	Customized routing per unique line class code, per request, per switch  NRC	TUITN, EACTI	IG (Note 5)	Volume and term arrangements are also available.	(Non-Database Owner), Per Query * method to transmit the names to the BellSouth CNAM database	ALLING NAME (CNAM) QUERY SERVICE CNAM (Database Owner), Per Query	onnect		NRC - Disconnect  Call Event special Study - per AIN Toolkit Service Subscription	NRC	nnect		Special Study - per AIN Toolkit Service Subscription	opposit in the same of the sam	Monthly Report - per AIN Toolkit Service Subscription		Type 1 Node Charge, per AIN Toolkit Subscription, per node, per query	onnect	Ingger Access Charge, per trigger, per DN, Feature Code NRC	onnect		
UEAC4 UEAC4 UEAC4 CNC2F CNC2F CNC2F CNC2F CNC4F	UEAC4 UEAC4 UEAC4 CNC2F CNC2F CNC2F CNC2F CNC2F CNC2F CNC4F CNC4F CNC4F	UEAC4 UEAC4 UEAC4  CNC2F CNC2F CNC2F CNC2F CNC2F CNC2F CNC2F	UEAC4 UEAC4 CNG2F CNG2F CNG2F CNG2F CNG2F	UEAC4 UEAC4 CNC2F CNC2F CNC2F	UEAC4 UEAC4 CNC2F	UEAC4	UEAC4	i .		UEAC4	UEAC4 UEAC4	OLINOR	UEAC2	i	UEAC2	UEAC2	UEAC2			USRCR				N'A	N/A	BAPES	BAPES	BAPDS	BAPDS	BAPLS	BAPLS	BAPLS	BAPMS	BAPMS			BAPTF	BAPTF	BAPTC	USOC BAPTC	_
\$55.46 \$39.18 \$16.83 \$13.27 \$21.75 \$66.71 \$50.43 \$13.31	\$55.46 \$39.18 \$16.83 \$13.27 \$21.75 \$66.71	\$55.46 \$39.18 \$16.83 \$13.27	\$55.46 \$39.18 \$16.83	\$55.46		\$12 10	\$11.39	\$12.82	₹ ₹	\$50.43	\$0.56 \$66.71		\$12.75	N.	\$29.40 NA	\$30.76	\$0.28		\$25.93	\$230.60	¥ ₹	2		\$595.00	\$0.016	\$15.90	\$47.74	\$31.84	\$44.56	\$15.90	\$47.74	\$0.10	\$44.56	\$16.00	2	\$0.024	\$37.90	\$117.98	\$37.90	<b>AL</b> \$117.98	
NA NA NA	NA NA		¥	N.	₹ ₹	<b>X X</b>	¥	NA.	<b>⊼</b> ₹	\$11.57	\$0.524 \$11.57	-	N N	X.	\$11.57	\$11.57	\$0.524		AN	\$229.65	8	2		\$595.00	\$0.016	¥	NA	¥ ¥	¥ 3	S S	NA	<b>₹</b>	X X	X X		Z Z	X X	N <sub>A</sub>	NA	₹₽	
N N	\$38.78	\$28.11 \$50.53	×	W.	\$41.56	\$15.64	¥	NA.	<b>₹</b>	\$12.60	\$0.50 \$12.60	5	N N	¥.	\$12.60 NA	\$12.60	\$0.30		\$18.94	\$180.62	X §	N		\$595.00	\$0.016	NA.	\$22.64	\$0.0028704	\$22.64	NA NA	\$22.64	\$0.0861109	\$22.64	\$1.46 \$15.96	2	\$0.0209223	NA	\$70.06	NA	<b>GA</b> \$70.06	
	\$38.78 NA NA	\$28.11 \$50.53	NA A	¥.	\$41.56	\$15.64	Ä	AN	<b>₹</b>	\$50.96	\$0.62 \$54.23	5	N N	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	\$51.07	\$54.21	\$0.31		NA	\$229.65	NA NA	Data		\$595.00	\$0.016	NA	AN	X X	N S	N A	AN	<b>₹</b>	N A	N K		¥ ¥		¥	AN	¥ <b>ર</b>	
5	\$38.37 \$16.75 \$14.20	\$34.38 \$49.81	\$10.29	\$12.84	\$29.63	\$19.13	\$8.55	\$9.53	X §	\$22.24	\$0.52 \$23.23	6	\$9.48	<b>X</b>	\$22.11	\$23.04	\$0.26		Ą	\$229.65	<b>₩</b>	N		\$595.00	\$0.016	\$37.77	\$37.77	\$21.97	\$34.61	e NA	\$37.77	\$0.08	\$34.61	\$15.89	2	\$0.0065	\$26.73	\$92.99	\$26.73	\$92.99	
¥	\$38.78 \$16.97 \$14.35	\$28.11 \$50.53	\$10.34	\$12.96	\$41.56	\$15.64	\$11.43	\$12.83	<b>₹</b>	\$29.77	\$0.7992	-	\$12.76	NA.	\$29.59 NA	\$30.93	\$0.3996		\$25.52	NA \$227.99	N S	NIA		\$595.00	\$0.016	NA	\$47.21	\$0.0027018	\$44.02	NA NA	\$47.21	\$0.0810536	\$44.02	\$16.01	2	\$0.0256138	\$48.44	\$106.90	\$48.44	<b>MS</b>	
\$0.97	\$63.56 NA	\$28.74 \$82.35	₹	¥	\$48.55	\$15.99	₹	¥.	\$4.73	\$39.25	\$0.18 \$41.91	3	₹ ₹	\$4.75	\$39.23	\$41.78	\$0.09		+	\$229.65	H	25		\$595.00	\$0.016	₹	\$47.20		\$71.80	e 15 00	\$47.20		\$71.80	\$15.98		\$0.005		\$149.95	¥	NC \$149.95	
NA	\$63.68 NA	\$27.08 \$84.07	NA NA	¥	\$48.89	\$15.06	¥	NA.	<b>₹</b>	\$38.90	\$0.7297 \$41.56		N N	×.	\$38.94	\$41.50	\$0.3648		\$27.84		N S			\$595.00	\$0.016	NA NA	\$47.35	\$0.0029092	\$72.15	e15 oA	\$47.35	\$0.0872769	\$72.15	\$15.93	2	\$0.0250662	NA A	\$150.25	NA.	<b>SC</b> \$150.25	
<b>X X</b>	\$38.7 NA	\$28.1 \$50.5	¥	¥	\$29.82	\$15.6	¥	NA NA	<b>₹</b>	\$19.2	\$0.50		N N	×.	\$19.2 NA	\$19.20	\$0.30		NA	\$229.6	TBD	T D		\$595.0	\$0.01	¥			¥ §				N N	N X		X X		¥	NA	<b>₹</b>	

DESCR	DESCRIPTION	USOC	₽	7	GA	ঽ	5	SW	S	SC	¥
7	RC		AN	NA A	NA A	AN	AN	NA A	\$12.33	AN	AN
_	NRC - 1st		NA	NA	NA	NA	NA	NA	\$69.84	NA	1
_	NRC - Add'I		¥	¥	¥	¥	₹	¥	\$49.43	¥	
_	NRC - Manual Service Order - 1st		¥	¥	¥	¥	¥	¥	\$4.70	¥	
_	NRC - Manual Service Order - Add'l		NA	NA	NA	NA	NA	NA	\$4.70	NA	
If no rat	If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BelSouth tariff or as negotiated by the parties upon request by either party.	et forth in applicable	BellSouth tari	ff or as negoti	ated by the pa	rties upon reques	t by either part	y.			
	Interim rates subject to true-up.										
1	BellSouth and CLEC shall negotiate rates for this offering. If agreement is not										
_	reached within sixty (60) days of the Effective Date, either party may petition the										
71	Florida PSC to settle the disputed charge or charges. (FL)										
2 1	This rate element is for those states w/o separate rates for 800 calls with 800 No.										
_	Delivery vs. POTS No. Delivery and calls with Optional Complex Features vs. w/o										
C	Optional Complex Features.										
3	This charge is only applicable where signaling usage measurement or billing										
С	capability does not exist.										
4	Prices for AIN to be determined upon development of mediation device. (TN)										
51 F	Price for Line Class Codes for Selective Routing shall be determined by the TRA.										
L											

Where the state Commission has adopted rates for the rate elements contained herein, it is the intent of the parties to reflect such rates in this exhibit and to apply the same consistent with applicable FCC and Commission rules and orders.

### **Attachment 3**

**Network Interconnection** 

#### TABLE OF CONTENTS

1. Network Interconnection	3
2. Interconnection Trunk Group Architectures	6
3. Network Design And Management For Interconne	ection12
4. Local Dialing Parity	15
5. Interconnection Compensation	15
3. Interconnection Compensation	13
6. Frame Relay Service Interconnection	22
7. Operational Support Systems (OSS)	24
Rates	Exhibit A
Basic Architecture	Exhibit B
One Way Architecture	
Two Way Architecture	
Supergroup Architecture	Exhibit E

The Parties shall provide interconnection with each other's networks for the transmission and routing of Local Traffic, Transit Traffic, Internet Service Provider-Bound ("ISP-Bound") Traffic, and exchange access (intraLATA toll and switched access) on the following terms:

#### 1. **Network Interconnection**

All negotiated rates, terms and conditions set forth in this Attachment pertain only to the provision of network interconnection where NewSouth owns and provides its switch(es).

- 1.1 Network Interconnection for Call Transport and Termination may be provided by the Parties at any technically feasible point. Requests to BellSouth for interconnection at points other than as set forth in this Attachment may be made through the Bona Fide Request/New Business Request process set out in General Terms and Conditions.
- 1.1.1 An Interconnection Point (IP) is the physical telecommunications equipment interface that performs the interconnection function for BellSouth and NewSouth. Each Party is responsible for providing the network on its side of the IP. Furthermore, the IP must be located within the LATA in which Local Traffic is originated.
- 1.1.2 Pursuant to the provisions of this Attachment, the location of the initial IP in a given LATA shall be established by mutual agreement of the Parties. Subject to the requirements for installing additional IPs, as set forth below, existing IPs will be accepted as initial IPs and will not require re-grooming. When the Parties mutually agree to utilize two-way interconnection trunk groups for the exchange of Local Traffic, ISP-bound Traffic and IntraLATA Toll Traffic between each other, the Parties shall mutually agree to the location of IP(s).
- 1.1.2.1 If the Parties are unable to agree to a mutual initial IP, each Party, as originating Party, may establish a single IP in the LATA for the delivery of its originated Local Traffic, ISP-bound Traffic, and IntraLATA Toll Traffic to the other Party for call transport and termination by the terminating Party.
- 1.1.2.2 Additional IPs in a particular LATA may be established by mutual agreement of the Parties. Absent mutual agreement, in order to establish additional IPs in a LATA, the proposed additional IP must meet the following criteria: (1) the traffic between NewSouth and BellSouth at the proposed additional IP must exceed 8.9 million minutes of Local Traffic and ISP-bound Traffic per month for three consecutive months during the busy hour; and (2) any end office to be designated as an IP must be more than 20 miles from an existing IP. BellSouth will not designate an IP at an end office where physical or virtual collocation space or BellSouth fiber connectivity is not available.

1.1.2.3 Upon written notification from the Party requesting the establishment of an additional IP, the receiving Party has 20 business days to analyze, respond to, and negotiate in good faith the establishment of and location of such IP. Should the Parties disagree that the traffic volumes justify an additional IP or disagree as to the location of an additional IP, the Parties shall follow the Dispute Resolution process contained in the General Terms and Conditions of this Agreement to determine whether and how the additional IP should be established.

#### 1.2 Interconnection via Dedicated Transport Facilities

- 1.2.1 The Parties shall institute a "bill and keep" compensation plan under which neither Party will charge the other Party recurring and nonrecurring charges for trunks (one-way or two-way), trunk ports and associated dedicated facilities for the exchange of Local Traffic (non-transit), ISP-bound Traffic, and IntraLATA Toll Traffic. Each Party has the obligation to install the appropriate trunks, trunk ports and associated facilities on its respective side of the IP and is responsible for bearing its own costs on its side of the IP. Both Parties, as appropriate, shall be compensated for the ordering of trunks, trunk ports and facilities used exclusively for transit traffic and for ancillary traffic types including, but not limited to, 911 and OS/DA. The Parties agree that charges for such trunks and facilities are as set forth in Exhibit A to this Attachment. Either Party may, at its option, choose to purchase such trunks and facilities from the other Party's tariff.
- 1.2.2 Pursuant to 1.2.1 above, as part of Local Interconnection Call Transport and Termination Service, the originating Party may obtain **Local Channel** facilities (i.e., entrance facilities) from the terminating Party from the originating Party's specified Interconnection Point to its Serving Wire Center. Such facilities may be purchased out of the terminating party's Commission approved access services tariff or as unbundled network elements at the rates set forth in Exhibit A to this Attachment. If tariffed access services are purchased, the portion of Local Channel facilities utilized for Local Traffic shall be determined based upon the application of the Percent Local Facility (PLF) Factor as defined in this Attachment. Additionally, the charges applied to the portion of the tariffed Local Channel used for Local Traffic as determined by the PLF are as set forth in Exhibit A to this Attachment. This factor shall be reported in addition to the switched dedicated transport jurisdictional factors specified in the BellSouth intrastate and interstate switched access tariffs.
- 1.2.3 Pursuant to 1.2.1 above, either Party may obtain **Dedicated Interoffice Transport** facilities from its designated Serving Wire Center to the other Party's first point of switching. Such facilities may be purchased out of the terminating party's access services tariff or as unbundled network elements at the rates set forth in Exhibit A to this Attachment. If tariffed access services are purchased, the portion of Dedicated Interoffice Transport facilities utilized for Local Traffic shall be determined based upon the application of the Percent Local Facility (PLF) Factor as defined in this Attachment. Additionally, the charges applied to the portion of the tariffed Dedicated

Interoffice Transport used for Local Traffic as determined by the PLF are as set forth in Exhibit A to this Attachment. This factor shall be reported in addition to the switched dedicated transport jurisdictional factors specified in the BellSouth intrastate and interstate switched access tariffs.

- 1.2.4 For the purposes of this Attachment, **Local Channel** (i.e., entrance facility) is defined as a flat-non-distance-sensitive rated switch transport facility between a Party's Interconnection Point and its Serving Wire Center.
- 1.2.5 For the purposes of this Attachment, **Serving Wire Center** is defined as the wire center owned by one Party from which the other Party would normally obtain dial tone for its Interconnection Point.
- 1.2.6 For the purposes of this Attachment, **Dedicated Interoffice Transport** is defined as a distance-sensitive rated switch transport facility between a Party's Serving Wire Center and the first point of switching on the other Party's common (shared) network.

#### 1.3 **Fiber Meet**

- 1.3.1 **Fiber Meet** is an interconnection arrangement whereby the Parties physically interconnect their networks via an optical fiber interface (as opposed to an electrical interface) at which one Party's facilities, provisioning, and maintenance responsibility begins and the other Party's responsibility ends (i.e., Interconnection Point).
- 1.3.2 If NewSouth elects to interconnect with BellSouth pursuant to a Fiber Meet, NewSouth and BellSouth shall jointly engineer and operate a Synchronous Optical Network ("SONET") transmission system by which they shall interconnect their transmission and routing of Local Traffic via a Local Channel facility at either the DS0, DS1, or DS3 level. The Parties shall work jointly to determine the specific transmission system. However, NewSouth's SONET transmission must be compatible with BellSouth's equipment in the BellSouth Interconnection Wire Center, and the Data Communications Channel (DCC) must be turned off, unless otherwise mutually agreed to by the Parties.
- 1.3.3 BellSouth shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the BellSouth Interconnection Wire Center ("BIWC").
- 1.3.4 NewSouth shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the NewSouth Interconnection Wire Center ("NewSouth Wire Center").
- 1.3.5 BellSouth shall designate a Interconnection Point outside the BIWC as a Fiber Meet point, and shall make all necessary preparations to receive, and to allow and enable NewSouth to deliver, fiber optic facilities into the Interconnection Point with sufficient spare length to reach the fusion splice point at the Interconnection Point. BellSouth shall, wholly at its own expense, procure, install, and maintain the fusion splicing point in the Interconnection Point. A Common Language Location Identification ("CLLI")

- code will be established for each Interconnection Point. The code established must be a building type code. All orders shall originate from the Interconnection Point (i.e., Interconnection Point to NewSouth, Interconnection Point to BellSouth).
- 1.3.6 NewSouth shall deliver and maintain such strands wholly at its own expense. Upon verbal request by NewSouth, BellSouth shall allow NewSouth access to the Fiber Meet entry point for maintenance purposes as promptly as possible.
- 1.3.7 The Parties shall jointly coordinate and undertake maintenance of the SONET transmission system. Each Party shall be responsible for maintaining the components of their own SONET transmission system.
- 1.3.8 Each Party will be responsible for (i) providing its own transport facilities to the Fiber Meet, and (ii) the cost to build-out its facilities to such Fiber Meet.
- 1.3.9 Neither Party shall charge the other for its portion of the Fiber Meet facility used exclusively for non-transit Local Traffic (i.e., the Local Channel). Charges for unbundled network element facilities shall be billed at the rates set forth in Attachment 3, Exhibit A. Charges for Switched and Special Access Services shall be billed in accordance with the applicable Access Service tariff (i.e. the BellSouth Interstate or Intrastate Access Services Tariff).

#### 2. Interconnection Trunk Group Architectures

- 2.1 BellSouth and NewSouth shall establish interconnecting trunk groups and trunk group configurations between networks including the establishment of one-way or two-way trunks in accordance with the following provisions set forth in this Agreement. For trunking purposes, traffic will be routed based on the digits dialed by the originating end user and in accordance with the Local Exchange Routing Guide (LERG).
- 2.2 NewSouth shall establish an interconnection trunk group(s) to at least one BellSouth access tandem within the LATA for the delivery of NewSouth's originated local and intraLATA toll traffic and for the receipt and delivery of Transit Traffic. To the extent NewSouth desires to terminate local and intraLATA toll traffic to BellSouth and Transit Traffic to third parties subtending other BellSouth access tandems within the LATA, other than the one NewSouth has established interconnection trunk groups to, NewSouth shall establish trunk groups to such other BellSouth access tandems.
- 2.2.1 Notwithstanding the forgoing, NewSouth shall establish an interconnection trunk group(s) to all BellSouth access and local tandems in the LATA where NewSouth has homed (i.e., assigned) its NPA/NXXs. NewSouth shall home its NPA/NXXs on the BellSouth tandems that serve the Exchange Rate Center Areas to which the NPA/NXXs are assigned. The specified association between BellSouth tandems and Exchange Rate Centers is defined in the national Local Exchange Routing Guide (LERG). NewSouth shall enter its NPA/NXX access and/or local tandem homing arrangement into the LERG.

- 2.3 Switched Access traffic will be delivered to and by Interexchange Carriers (IXCs) based on NewSouth's NXX Access Tandem homing arrangement as specified by NewSouth in the Local Exchange Routing Guide (LERG).
- Any NewSouth interconnection request that deviates from the interconnection trunk group architectures as described in this Agreement that affects traffic delivered to NewSouth from a BellSouth switch that requires special BellSouth switch translations and other network modifications will require NewSouth to submit a Bona Fide Request/New Business Request via the Bona Fide Request/New Business Request Process set forth in Attachment 12 of this Agreement.
- 2.5 Subject to 1.2.1, charges, both non-recurring and recurring, associated with interconnecting trunk groups between BellSouth and NewSouth are set forth in Exhibit A. To the extent a rate associated with the interconnecting trunk group is not set forth in Exhibit A, the interim rate shall be as set forth in the appropriate Party's Commission filed and effective tariff for Switched Access services. Once a cost based rate is established by BellSouth, the interim tariff rate shall be trued up and the cost based rate will be applied retroactively to the effective date of this agreement.
- Unless the Parties mutually agree otherwise, NewSouth shall be responsible for ordering and paying for any two way trunks carrying Transit Traffic. At such time as NewSouth is providing a transit function on 20% of all Transit Traffic, BellSouth and NewSouth will negotiate alternative compensation for two way trunks carrying Transit Traffic. At such time as NewSouth is providing the transit function on 30% of all Transit Traffic, the Parties shall execute an amendment implementing such alternative compensation for two way trunks carrying Transit Traffic.
- 2.7 All trunk groups will be provisioned as Signaling System 7 (SS7) capable where technically feasible. If SS7 is not technically feasible multi-frequency (MF) protocol signaling shall be used.
- In cases where NewSouth is also an IXC, the IXC's Feature Group D (FG D) trunk group(s) must remain separate from the local interconnection trunk group(s).
- Unless in response to a blocking situation or for a project, when either Party orders interconnection trunk group augmentations, a Firm Order Confirmation (FOC) shall be returned to the ordering Party within four (4) business days from receipt of a valid error free ASR. A project is defined as a new trunk group or the request of 96 or more trunks on a single or multiple trunk group(s) in a given local calling area. Blocking situations and projects shall be managed through the BellSouth Interconnection Trunking Project Management group and NewSouth's equivalent trunking group.
- 2.10 Interconnection Trunk Groups for Exchange of Local, ISP-Bound, IntraLATA Toll and Transit Traffic
- 2.10.1 If the Parties' originated local, ISP-Bound and/or intraLATA toll traffic is exchanged utilizing the same two-way trunk group, the Parties shall mutually agree to use this

type of two-way interconnection trunk group with the quantity of trunks being mutually determined and the provisioning being jointly coordinated. Furthermore, the Interconnection Point(s) for two-way interconnection trunk groups transporting both Parties local, ISP-Bound and/or intraLATA toll shall be mutually agreed upon. NewSouth shall order such two-way trunks via the Access Service Request (ASR) process in place for Local Interconnection upon determination by the Parties, in a joint planning meeting, that such trunk groups shall be utilized. BellSouth will use the Trunk Group Service Request (TGSR) to request changes in trunking. Both Parties reserve the right to issue ASRs if so required, in the normal course of business. Furthermore, the Parties shall jointly review such trunk performance and forecasts on a periodic basis. The Parties use of two-way interconnection trunk groups for the transport of local, ISP-Bound and/or intraLATA toll traffic between the Parties does not preclude either Party from establishing additional one-way interconnection trunks for the delivery of its originated local, ISP-Bound and/or intraLATA toll traffic to the other Party.

#### 2.10.2 BellSouth Access Tandem Interconnection Architectures

BellSouth Access Tandem Interconnection provides intratandem access to subtending end offices.

#### 2.10.2.1 **Basic Architecture**

2.10.2.1.1 In this architecture, NewSouth's originating Local traffic, ISP-Bound traffic, and IntraLATA Toll and originating and terminating Transit Traffic is transported on a single two-way trunk group between NewSouth and BellSouth access tandem(s) within a LATA. This group carries intratandem Transit Traffic between NewSouth and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires interconnection and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. BellSouth originated Local, ISP-bound and IntraLATA Toll traffic is transported on a single one-way trunk group terminating to NewSouth. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The Basic Architecture is illustrated in Exhibit B.

#### 2.10.2.2 **One-Way Trunk Group Architecture**

2.10.2.2.1 In this architecture, the Parties interconnect using two one-way trunk groups. One one-way trunk group carries NewSouth-originated local, ISP-Bound, and intraLATA toll traffic destined for BellSouth end-users. The other one-way trunk group carries BellSouth-originated local, ISP-Bound, and intraLATA toll traffic destined for NewSouth end-users. A third two-way trunk group is established for NewSouth's originating and terminating Transit Traffic. This group carries intratandem Transit Traffic between NewSouth and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires interconnection

and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The One-Way Trunk Group Architecture is illustrated in Exhibit C.

#### 2.10.2.3 Two-Way Trunk Group Architecture

2.10.2.3.1 The Two-Way Trunk Group Architecture establishes one two-way trunk group to carry local, ISP-Bound, and intraLATA toll traffic between NewSouth and BellSouth. In addition, a two-way transit trunk group must be established for NewSouth's originating and terminating Transit Traffic. This group carries intratandem Transit Traffic between NewSouth and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires interconnection and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. The Two-Way Trunk Group Architecture is illustrated in Exhibit D.

#### 2.10.2.4 **Supergroup Architecture**

2.10.2.4.1 In the Supergroup Architecture, the Parties' Local, ISP-Bound, and IntraLATA Toll and Transit Traffic are exchanged on a single two-way trunk group between NewSouth and BellSouth. This group carries intratandem Transit Traffic between the Parties and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which NewSouth desires interconnection and has the proper contractual arrangements. This group also carries NewSouth originated intertandem traffic transiting a single BellSouth access tandem destined to third party tandems such as an Independent Company tandem or other CLEC tandem. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required. The LERG should be referenced for current routing and tandem serving arrangements. In the event either Party desires to use Supergroup, the Parties will mutually agree on the compensation applicable for the portion of the trunks and facilities used for transit traffic. The Supergroup Architecture is illustrated in Exhibit E.

#### 2.10.3 Local Tandem Interconnection

2.10.3.1 Local Interconnection trunk group(s) may be established at BellSouth local tandems for: (1) the delivery of NewSouth-originated Local Traffic transported and terminated by BellSouth to BellSouth end offices within the local calling area as defined in BellSouth's General Subscriber Services Tariff (GSST), section A3 served by those BellSouth local tandems, and (2) for local Transit Traffic transported by BellSouth for

- third party network providers who have also established an interconnection trunk group(s) at those BellSouth local tandems.
- 2.10.3.2 When a specified local calling area is served by more than one BellSouth local tandem, NewSouth must designate a "home" local tandem for each of its assigned NPA/NXXs and establish trunk connections to such local tandems. Where requested facilities are not available, NewSouth and BellSouth will route NewSouth's traffic, if possible, to a point where facilities are available. This alternative routing will be an interim solution until requested facilities are available. Additionally, NewSouth may choose to establish an interconnection trunk group(s) at the BellSouth local tandems where it has no codes homing but is not required to do so. NewSouth may deliver Local Traffic to a "home" BellSouth local tandem that is destined for other BellSouth or third party network provider end offices subtending other BellSouth local tandems in the same local calling area where NewSouth does not choose to establish an interconnection trunk group(s). It is NewSouth's responsibility to enter its own NPA/NXX local tandem homing arrangements into the Local Exchange Routing Guide (LERG) either directly or via a vendor in order for other third party network providers to determine appropriate traffic routing to NewSouth's codes. Likewise, NewSouth shall obtain its routing information from the LERG.
- 2.10.3.3. Notwithstanding establishing an interconnection trunk group(s) to BellSouth's local tandems, NewSouth must also establish an interconnection trunk group(s) to BellSouth access tandems within the LATA on which NewSouth has NPA/NXXs homed for the delivery of Interexchange Carrier Switched Access (SWA) and toll traffic, and traffic to Type 2A CMRS connections located at the access tandems. BellSouth shall not switch SWA traffic through more than one BellSouth access tandem. SWA, Type 2A CMRS or toll traffic routed to the local tandem in error will not be backhauled to the BellSouth access tandem for completion. (Type 2A CMRS interconnection is defined in BellSouth's A35 General Subscriber Services Tariff).
- 2.10.3.4 BellSouth's provisioning of local tandem interconnection assumes that NewSouth has executed the necessary local interconnection agreements with the other third party network providers subtending those local tandems in a manner consistent with the requirements of the Act.

#### 2.10.4 **Direct End Office-to-End Office Interconnection**

- 2.10.4.1 Direct End Office-to-End Office one-way or two-way interconnection trunk groups allow for the delivery of a Party's originating local or intraLATA toll traffic to the terminating Party on a direct end office-to-end office basis.
- 2.10.4.2 The Parties shall utilize direct end office-to-end office trunk groups under the following conditions:
- 2.10.4.2.1 Tandem Exhaust If a tandem through which the Parties are interconnected is unable to, or is forecasted to be unable to support additional traffic loads for any period of time, the Parties will mutually agree on an alternative tandem trunking plan or end

- office trunking plan that will alleviate the tandem capacity shortage and ensure completion of traffic between NewSouth and BellSouth's subscribers.
- 2.10.4.2.2 Traffic Volume –To the extent either Party has the capability to measure the amount of traffic between a NewSouth switching center and a BellSouth end office, either Party shall install and retain direct end office trunking sufficient to handle actual or reasonably forecasted traffic volumes, whichever is greater, between a NewSouth switching center and a BellSouth end office where the traffic exceeds or is forecasted to exceed two DS1s of traffic per month. Either Party will install additional capacity between such points when overflow traffic between NewSouth's switching center and BellSouth's end office exceeds or is forecasted to exceed two DS1s of traffic per month. In the case of one way trunking, additional trunking shall only be required by the Party whose trunking has achieved the preceding usage threshold.
- 2.10.4.2.3 Mutual Agreement The Parties may install direct end office trunking upon mutual agreement in the absence of conditions (1) or (2) above and agreement will not unreasonably be withheld.

#### 2.10.5 Transit Traffic Trunk Group

2.10.5.1 Transit Traffic trunks can either be two-way trunks or two one-way trunks ordered by NewSouth to deliver and receive local and intraLATA toll Transit Traffic from third parties, such as Independent Companies and other CLECs, via BellSouth access tandems (or BellSouth local tandems for Local Traffic), and Switched Access traffic to and from Interexchange Carriers via BellSouth access tandems pursuant to the Transit Traffic section of this Attachment. Establishing Transit Traffic trunks at BellSouth access and local tandems provides intratandem access to the third parties also interconnected at those tandems.

#### 2.10.6 Toll Free Traffic

- 2.10.6.1 If NewSouth chooses BellSouth to handle Toll Free database queries from its switches, all NewSouth originating Toll Free traffic will be routed over the Transit Traffic Trunk Group.
- 2.10.6.2 All originating Toll Free Service (Toll Free) calls for which NewSouth requests that BellSouth perform the Service Switching Point ("SSP") function (i.e., perform the database query) shall be delivered using GR-394 format over the Transit Traffic Trunk Group. Carrier Code "0110" and Circuit Code (to be determined for each LATA) shall be used for all such calls.
- 2.10.6.3 NewSouth may handle its own Toll Free database queries from its switch. If so, NewSouth will determine the nature (local/intraLATA/interLATA) of the Toll Free call based on the response from the database. If the query determines that the call is a BellSouth local or intraLATA Toll Free number, NewSouth will route the post-query local or IntraLATA converted ten-digit local number to BellSouth over the local or intraLATA trunk group. If the query determines that the call is a third party (ICO or other CLEC) local or intraLATA Toll Free number, NewSouth will route the post-

query local or intraLATA converted ten-digit local number to BellSouth over the Transit Traffic Trunk Group. In such case, NewSouth is to provide a Toll Free billing record when appropriate. If the query reveals the call is an interLATA Toll Free number, NewSouth will route the post-query interLATA call (Toll Free number) directly from its switch for carriers interconnected with its network or over the Transit Traffic Trunk Group to the BellSouth Access Tandem for carriers not directly connected to its network. Calls will be routed to BellSouth over the local/intraLATA and Transit Traffic Trunk Groups within the LATA in which the calls originate.

2.10.6.4 All post-query Toll Free Service (Toll Free) calls for which NewSouth performs the SSP function, if delivered to BellSouth, shall be delivered using GR-394 format for calls destined to IXCs, and GR-317 format for calls destined to end offices that directly subtend the BellSouth access tandem.

#### 3. Network Design And Management For Interconnection

- 3.1 Network Management and Changes. Both Parties will work cooperatively with each other to install and maintain the most effective and reliable interconnected telecommunications networks, including but not limited to, the exchange of toll-free maintenance contact numbers and escalation procedures. Both Parties agree to provide public notice of changes in the information necessary for the transmission and routing of services using its local exchange facilities or networks, as well as of any other changes that would affect the interoperability of those facilities and networks.
- 3.2 Interconnection Technical Standards. The interconnection of all networks will be based upon accepted industry/national guidelines for transmission standards and traffic blocking criteria. Interconnecting facilities shall conform, at a minimum, to the telecommunications industry standard of DS-1 pursuant to Bellcore Standard No. TR-NWT-00499. Signal transfer point, Signaling System 7 ("SS7") connectivity is required at each interconnection point. BellSouth will provide out-of-band signaling using Common Channel Signaling Access Capability where technically and economically feasible, in accordance with the technical specifications set forth in the BellSouth Guidelines to Technical Publication, TR-TSV-000905. Facilities of each Party shall provide the necessary on-hook, off-hook answer and disconnect supervision and shall hand off calling number ID (Calling Party Number) when technically feasible.
- 3.3 <u>Quality of Interconnection</u>. The local interconnection for the transmission and routing of telephone exchange service and exchange access that each Party provides to each other will be at least equal in quality to what it provides to itself and any subsidiary or affiliate, where technically feasible, or to any other Party to which each Party provides local interconnection.
- 3.4 <u>Network Management Controls.</u> Both Parties will work cooperatively with each other to apply sound network management principles by invoking appropriate network management controls (e.g., call gapping) to alleviate or prevent network congestion.

- Common Channel Signaling. Both Parties will provide LEC-to-LEC Common Channel Signaling ("CCS") to each other, where available, in conjunction with all traffic in order to enable full interoperability of CLASS features and functions except for call return. All CCS signaling parameters will be provided, including automatic number identification ("ANI"), originating line information ("OLI") calling company category, charge number, etc. All privacy indicators will be honored, and each Party will cooperate with each other on the exchange of Transactional Capabilities Application Part ("TCAP") messages to facilitate full interoperability of CCS-based features between the respective networks. Neither Party shall alter the CCS parameters, or be a party to altering such parameters, or knowingly pass CCS parameters that have been altered in order to circumvent appropriate interconnection charges.
- 3.6 <u>Signaling Call Information</u>. BellSouth and NewSouth will send and receive 10 digits for Local Traffic. Additionally, BellSouth and NewSouth will exchange the proper call information, i.e. originated call company number and destination call company number, CIC, and OZZ, including all proper translations for routing between networks and any information necessary for billing.

#### 3.7 Forecasting for Trunk Provisioning

- 3.7.1 Within six (6) months after execution of this agreement, NewSouth shall provide an initial interconnection trunk group forecast for each LATA that it shall provide service within BellSouth's region. Upon receipt of NewSouth's forecast, the Parties shall schedule and participate in a joint planning meeting to develop a joint interconnection trunk group forecast. Each forecast provided under this Section shall be deemed "Confidential Information" under the General Terms and Conditions Part A of this Agreement.
- 3.7.1.1 At a minimum, the forecast shall include the projected quantity of Transit Trunks, NewSouth-to-BellSouth one-way trunks ("NewSouth Trunks"), BellSouth-to-NewSouth one-way trunks ("Reciprocal Trunks") and/or two-way interconnection trunks, if the Parties have agreed to interconnect using two-way trunking to transport the Parties' local and intraLATA toll. The quantities shall be projected for a minimum of six months in advance and shall include the current year plus next two years total forecasted quantities. Considering NewSouth's provided forecast, the Parties shall mutually develop Reciprocal Trunk and/or two-way interconnection trunk forecast quantities for the time periods listed and to be included within the initial forecast.
- 3.7.1.2 Additionally, all forecasts shall include, at a minimum, trunk group type (local/intraLATA toll, Transit, Operator Services, 911, etc.), A location/Z location (CLLI codes for NewSouth location and BellSouth location where the trunks shall terminate), interface type (e.g., DS1), Direction of Signaling, Trunk Group Access Code, if known, (commonly referred to as the 2-6 code) and forecasted trunks in service each year (cumulative).

- 3.7.2 Each Party shall exercise its best efforts to provide the quantity of interconnection trunks mutually forecasted. However, the provision of the forecasted quantity of interconnection trunks is subject to trunk terminations and facility capacity existing at the time the trunk order is submitted. Furthermore, the receipt and development of trunk forecasts does not imply any liability for failure to perform if capacity (trunk terminations or facilities) is not available for use at the forecasted time.
- 3.7.3 The submitting and development of interconnection trunk forecasts shall not replace the ordering process in place for local interconnection trunks.
- 3.7.4 Once initial interconnection trunk forecasts have been developed, NewSouth shall continue to provide interconnection trunk forecasts on a semiannual basis or at otherwise mutually agreeable intervals. NewSouth shall use its best efforts to make the forecasts as accurate as possible based on reasonable engineering criteria. Interconnection trunk forecasts shall be updated and provided to BellSouth on an as needed basis, but no less frequently than semiannually and no more frequently than monthly. Upon receipt of NewSouth's forecast, including forecast updates, the Parties shall confer to mutually develop BellSouth Reciprocal Trunk and/or two-way interconnection trunk forecasted quantities for the listed time periods within such subsequent forecasts.

#### 3.8 Trunk Utilization

- 3.8.1 BellSouth and NewSouth shall monitor traffic on each interconnection trunk group that is installed pursuant to the initial interconnection trunk requirements and subsequent forecasts six months after the initial installation of the trunks and any time after the end of a calendar quarter thereafter. Based on a review of the capacity utilization during such quarter for installed Reciprocal Trunk groups and/or two-way interconnection trunk groups, subject to the provisions of the section following, BellSouth may disconnect any Non-utilized Reciprocal Trunk(s) or Under-utilized Reciprocal Trunk(s), as defined in Section 3.8.1.1 below, and NewSouth shall refund to BellSouth associated trunk and facility charges paid by BellSouth retroactive to the date on which such trunks failed to maintain compliance with the threshold described in Section 3.8.1.1 below. In addition, BellSouth may request NewSouth to disconnect any Non-utilized or Under-utilized two-way interconnection trunk(s)if BellSouth has determined that the trunk group is not being utilized as described in Section 3.8.1.1 below, provided that the Parties have not otherwise agreed. NewSouth shall comply with such request, subject to Section 3.8.1.1 below.
- 3.8.1.1 The Parties agree that within 180 days of the installation of a trunk or trunks, the trunks will be utilized at 60 percent (60%) of the time consistent busy hour utilization level. The Parties agree that within 365 days of the installation of a trunk or trunks, the trunks will be utilized at eighty percent (80%) of the time consistent busy hour utilization level. Any trunk or trunks not meeting the minimum thresholds set forth in this Section are defined as "Under-utilized" trunks.

- 3.8.1.2 BellSouth's Local Interconnection Switching Center ("LISC") will notify NewSouth of any under-utilized reciprocal trunk groups and the number of trunks that BellSouth wishes to disconnect. BellSouth will provide supporting information either by email or facsimile to the designated NewSouth interface. NewSouth will provide concurrence with the disconnection in seven (7) business days of its receipt of such notification or will provide specific information supporting why the trunks should not be disconnected. Such supporting information should include expected Local Number Ported (LNP) and traffic volumes and the timeframes within which NewSouth expects to need such trunks. BellSouth's LISC project manager and Circuit Capacity Manager will discuss the information with NewSouth to determine if agreement can be reached on the number of trunks to be removed. If no agreement can be reached, BellSouth will issue disconnect orders to NewSouth. The due date of these orders will be four weeks from the date on which NewSouth received notice, in writing, of BellSouth's request to disconnect the underutilized trunk groups.
- 3.8.1.3 To the extent NewSouth requests BellSouth and BellSouth agrees to install additional Reciprocal and/or two-way interconnection trunks in any forecast period following the initial forecasting period that are not included in the forecast for that period (as such forecast may be revised from time to time), such trunks may be provisioned by BellSouth subject to the conditions set forth in the preceding sections.
- 3.8.2 To the extent that any Final interconnection trunk group is utilized at, or, based on trend (incorporating linear regression analysis using Erlang B theory with weekly tracked historical traffic data per trunk group engineered at a P.01 grade of service) will reach within six weeks, a time-consistent busy hour utilization level of eighty percent (80%) or greater, the Parties shall negotiate in good faith for the installation of augmented facilities.

#### 4. **Local Dialing Parity**

4.1 BellSouth and NewSouth shall provide local and toll dialing parity as described in the Act and required by FCC rules, regulations and policies. Dialing parity shall be provided for all originating telecommunications services that require dialing to route a call. BellSouth and NewSouth shall permit similarly situated telephone exchange service end users to dial the same number of digits to make a local telephone call notwithstanding the identity of the end user's or the called party's telecommunications service provider. In addition, NewSouth end users shall experience at least the same service quality level as BellSouth end users in terms of post-dial delay, call completion rate and transmission quality.

#### 5. **Interconnection Compensation**

#### 5.1 Compensation for Call Transportation and Termination for Local Traffic

- 5.1.1 For reciprocal compensation between the Parties pursuant to this Attachment, Local Traffic is defined as any telephone call that is originated by an end user of one Party and terminated to an end user of the other Party within a given LATA on that other Party's network, except for those calls that are originated or terminated through switched access arrangements.
- 5.1.1.1 Additionally, Local Traffic includes any cross boundary, voice-to-voice intrastate, interLATA or interstate, interLATA calls between specific wire centers established as a local call by the ruling regulatory body.
- 5.1.1.2 For purposes of this Attachment, ISP-bound Traffic is defined as any telephone call to an ESP/ISP that is dialed using a local dialing pattern (7 or 10 digits) by the calling party to an ESP/ISP server physically located within a given LATA ("ISP-bound traffic").
- 5.1.2 For purposes of this Agreement and for traffic between the Parties originating from and directed to the exchanges subject to this Agreement, the Parties agree to a bill-and-keep arrangement for usage on Local Traffic and ISP-bound traffic. Such bill-and-keep arrangement includes any per minute of use rate elements associated with the transport and termination of Local Traffic and ISP-bound Traffic (including, but not limited to end office switching, tandem switching, and common transport).
- 5.1.2.1 For the purposes of this Attachment, **Common (Shared) Transport** is defined as the transport of the originating Party's traffic by the terminating Party over the terminating Party's common (shared) facilities between the terminating Party's tandem switch and end office switch and/or between the terminating Party's tandem switches.
- 5.1.2.2 For the purposes of this Attachment, **Tandem Switching** is defined as the function that establishes a communications path between two switching offices through a third switching office (the Tandem switch).
- 5.1.2.3 For the purposes of this Attachment, **End Office Switching** is defined as the function that establishes a communications path between the trunk side and line side of the End Office switch.
- 5.1.3 Neither Party shall represent Switched Access Traffic as Local Traffic for purposes of payment of reciprocal compensation.
- Pursuant to the definition of Local Traffic in this Attachment, and for the purpose of delivery of one Party's originating traffic to the other, Local Traffic and ISP-Bound Traffic delivered to a terminating Party's end users physically located within the LATA in which the call originated and within which the Party's end user's NPA/NXX is assigned shall be subject to bill-and-keep. If either Party assigns NPA/NXXs to specific BellSouth rate centers within the LATA and assigns numbers from those NPA/NXXs to its end users physically located outside of that LATA, the originating Party's traffic originating from within the LATA where the NPA/NXXs are assigned and delivered to the terminating Party's customer physically located outside of such LATA, shall not be deemed Local Traffic, and such traffic will not be subject to bill-

and-keep. Further, the Parties agree to identify such interLATA traffic to each other and to compensate each other for originating and transporting such interLATA traffic to each other at the originating Party's Commission approved tariffed switched access rates.

- 5.1.5. **Percent Local Use.** Each Party shall report to the other a Percent Local Usage ("PLU"). The application of the PLU will determine the amount of Local Traffic and ISP-Bound Traffic, per minute of use, subject to bill and keep under this agreement. For purposes of developing the PLU, each Party shall consider every local call, every call to an enhanced/information service provider (including Internet service providers), and every long distance call, excluding Transit Traffic. Each Party shall update its PLU on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than 30 calendar days after the first of each such month based on local usage for the past three months ending the last day of December, March, June and September, respectively. Requirements associated with PLU calculation and reporting shall be as set forth in BellSouth's Percent Local Use Reporting Guidebook, as it is amended from time to time. Notwithstanding the foregoing, where the terminating Party has message recording technology that identifies the jurisdiction of traffic terminated as defined in this Agreement, such information, in lieu of the PLU factor, shall at the terminating Party's option be utilized to determine the appropriate local usage compensation to be paid.
- Percent Local Facility. Each Party shall report to the other a Percent Local Facility ("PLF"). The application of the PLF will determine the portion of switched dedicated transport ordered via BellSouth's E6 tariff to be billed per the local jurisdiction rates. The PLF shall be applied to multiplexing, local channel and interoffice channel switched dedicated transport ordered from BellSouth's E6 tariff and utilized in the provision of local interconnection trunks. Each Party shall update its PLF on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than 30 calendar days after the first of each such month to be effective the first bill period the following month, respectively. Requirements associated with PLU and PLF calculation and reporting shall be as set forth in BellSouth's Percent Local Use/Percent Local Facility Reporting Guidebook, as it is amended from time to time.
- Interstate Usage ("PIU"). All jurisdictional report to the other the projected Percent Interstate Usage ("PIU"). All jurisdictional report requirements, rules and regulations for Interexchange Carriers specified in BellSouth's Intrastate Access Services Tariff will apply to NewSouth. After interstate and intrastate traffic percentages have been determined by use of PIU procedures, the PLU and PLF factors will be used, to the extent applicable, for application and billing of local interconnection. Each Party shall update its PIUs on the first of January, April, July and October of the year and shall send it to the other Party to be received no later than 30 calendar days after the first of each such month, for all services showing the percentages of use (PIUs, PLU, and PLF) for the past three months ending the last day of December, March, June and September. Notwithstanding the foregoing, where the terminating Party has message recording technology that identifies the jurisdiction of traffic terminated as defined in

this Agreement, such information, in lieu of the PIU and PLU factors, shall at the terminating Party's option be utilized to determine the appropriate local usage compensation to be paid.

Audits. On thirty (30) days written notice, each Party must provide the other the ability and opportunity to conduct an annual audit to ensure the proper billing of traffic. BellSouth and NewSouth shall retain records of call detail for a minimum of nine months from which a PLF, PLU and/or PIU can be ascertained. The audit shall be accomplished during normal business hours at an office designated by the Party being audited. Audit requests shall not be submitted more frequently than one (1) time per calendar year. Audits shall be performed by a mutually acceptable independent auditory paid for by the Party requesting the audit. The PLF, PLU and/or PIU shall be adjusted based upon the audit results and shall apply to the usage for the quarter the audit was completed, to the usage for the quarter prior to the completion of the audit, and to the usage for the two quarters following the completion of the audit. If, as a result of an audit, either Party is found to have overstated the PLF, PLU and/or PIU by twenty percentage points (20%) or more, that Party shall reimburse the auditing Party for the cost of the audit.

#### 5.5 Rate True-up

This section applies only to Tennessee and other rates that are interim or expressly subject to true-up under this attachment.

- 5.5.1 The interim prices for Unbundled Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:
- 5.5.2 The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by a final order (including any appeals) of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions of Section 12 of the General Terms and Conditions and Attachment 1 of the Agreement.
- 5.5.3 The Parties may continue to negotiate toward final prices, but in the event that no such agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in the General Terms and Conditions of the Agreement, so long as they file the resulting Agreement with the Commission as a "negotiated agreement" under Section 252(e) of the Act.

- 5.5.4 A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
  - (a) BellSouth and CLEC is entitled to be a full Party to the proceeding;
  - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
  - (c) It shall include as an issue the geographic deaveraging of unbundled element prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

#### 5.6 **Compensation for 8XX Traffic**

- 5.6.1 <u>Compensation for 8XX Traffic.</u> Each Party shall compensate the other pursuant to the appropriate switched access charges, including the database query charge as set forth in the Party's Commission or FCC filed and approved intrastate or interstate switched access tariffs.
- 5.6.2 <u>Records for 8XX Billing</u>. Each Party will provide to the other the appropriate records necessary for billing intraLATA 8XX customers. The records provided will be in a standard EMI format.
- 5.6.3 8XX Access Screening. BellSouth's provision of 8XX Toll Free Database (TFD) to NewSouth requires interconnection from NewSouth to BellSouth 8XX SCP. Such interconnections shall be established pursuant to BellSouth's Common Channel Signaling Interconnection Guidelines and Bellcore's CCS Network Interface Specification document, TR-TSV-000905. NewSouth shall establish CCS7 interconnection at the BellSouth Local Signal Transfer Points serving the BellSouth 8XX SCPs that NewSouth desires to query. The terms and conditions for 8XX TFD are set out in BellSouth's Intrastate Access Services Tariff as amended.

#### 5.7 Mutual Provision of Switched Access Service

5.8.1 Switched Access Traffic. Switched Access Traffic is defined as telephone calls requiring local transmission or switching services for the purpose of the origination or termination of Telephone Toll Service. Switched Access Traffic includes the following types of traffic: Feature Group A, Feature Group B, Feature Group C, Feature Group D, toll free access (e.g., 800/877/888), 900 access, and their successors or similar Switched Exchange Access Services. The Parties have been unable to agree as to whether "Voice-Over-Internet Protocol" transmissions ("VOIP") which cross LATA boundaries constitute Switched Access Traffic. Notwithstanding the foregoing, and without waiving any rights with respect to either Party's position as to the jurisdictional nature of VOIP, the Parties agree to abide by any effective and applicable

FCC rules and orders regarding the nature of such traffic and the compensation payable by the Parties for such traffic, if any.

- 5.8.2 When one Party's end office switch, subtending the other Party's Access Tandem switch for receipt or delivery of switched access traffic, provides an access service connection to or from an interexchange carrier ("IXC") by either a direct trunk group to the IXC utilizing the other Party's facilities, or via the other Party's tandem switch, each Party will provide its own access services to the IXC and bill on a multi-bill, multi-tariff meet-point basis. Each Party will bill its own access services rates to the IXC with the exception of the interconnection charge. The interconnection charge will be billed by the Party providing the end office function. Each party will use the Multiple Exchange Carrier Access Billing (MECAB) guidelines to establish meet point billing for all applicable traffic. Thirty (30)-day billing periods will be employed for these arrangements. For tandem routed traffic, the tandem company agrees to provide to the Initial Billing Company as defined in MECAB, at no charge, all the switched access detail usage data, recorded at the access tandem, within no more than sixty (60) days after the recording date. The Initial Billing Company will provide the switched access summary usage data, for all originating and terminating traffic, to all Subsequent Billing Companies as defined in MECAB within 10 days of rendering the initial bill to the IXC. Each Party will notify the other when it is not feasible to meet these requirements so that the customers may be notified for any necessary revenue accrual associated with the significantly delayed recording or billing. As business requirements change, data reporting requirements may be modified as necessary.
- 5.8.3 In the event that either Party fails to provide switched access detailed usage data to the other Party within 90 days after the recording date and the receiving Party is unable to bill and/or collect access revenues due to the sending Party's failure to provide such data within said time period, then the Party failing to send the data as specified herein shall be liable to the other Party in an amount equal to the unbillable or uncollectible revenues. Each company will provide complete documentation to the other to substantiate any claim of unbillable access revenues and a negotiated settlement will be agreed upon between the Parties.
- 5.8.4 Each Party will retain for a minimum period of sixty (60) days, access message detail sufficient to recreate any data which is lost or damaged by their company or any third party involved in processing or transporting data.
- Each Party agrees to recreate the lost or damaged data within forty-eight (48) hours of notification by the other or by an authorized third party handling the data.
- Each Party also agrees to process the recreated data within forty-eight (48) hours of receipt at its data processing center.
- 5.8.7 All claims should be filed with the other Party within 120 days of the receipt of the date of the unbillable usage.

- 5.8.8 The Initial Billing Company shall keep records of its billing activities relating to jointly-provided Intrastate and Interstate access services in sufficient detail to permit the Subsequent Billing Party to, by formal or informal review or audit, to verify the accuracy and reasonableness of the jointly-provided access billing data provided by the Initial billing Party. Each Party agrees to cooperate in such formal or informal reviews or audits and further agrees to jointly review the findings of such reviews or audits in order to resolve any differences concerning the findings thereof.
- 5.8.9 NewSouth agrees not to deliver switched access traffic to BellSouth for termination except over NewSouth ordered switched access trunks and facilities.

#### **5.9** Transit Traffic Service

- 5.9.1 Each Party shall provide tandem switching and transport services for the other Party's Transit Traffic. Transit Traffic is traffic originating on one Party's network that is switched and/or transported by the other Party and delivered to a third party's network, or traffic originating on a third Party's network that is switched and/or transported by one Party and delivered to the other Party's network. Rates for local Transit Traffic shall be the applicable Call Transport and Termination charges as set forth in Exhibit A to this Attachment. Rates for intraLATA toll and Switched Access Transit Traffic shall be the applicable charges as set forth in the applicable Party's Commission approved Interstate or Intrastate Switched Access tariffs. Switched Access Transit Traffic presumes that one Party's switch is subtending the other Party's Access Tandem for switched access traffic to and from one Party's end users utilizing the other Party's facilities, either by direct trunks with the IXC, or via the BellSouth Access Tandem. Billing associated with all Transit Traffic shall be pursuant to Multiple Exchange Carrier Access Billing (MECAB) guidelines. Pursuant to these guidelines, the Initial Billing Company shall provide summary usage data, for all originating and terminating Transit Traffic, to all Subsequent Billing Companies. Traffic between NewSouth and Wireless Type 1 third parties shall not be treated as Transit Traffic from a routing or billing perspective. Traffic between NewSouth and Wireless Type 2A or third parties utilizing UNE-P shall not be treated as Transit Traffic from a routing or billing perspective until BellSouth and the Wireless carrier or third party utilizing UNE-P have the capability to properly meet-point-bill in accordance with MECAB guidelines.
- 5.9.2. The delivery of traffic which transits the BellSouth network and is transported to another carrier's network is excluded from any BellSouth billing guarantees and will be delivered at the rates stipulated in this Agreement to a terminating carrier. BellSouth agrees to deliver this traffic to the terminating carrier; provided, however, that NewSouth is solely responsible for negotiating and executing any appropriate contractual agreements with the terminating carrier for the receipt of this traffic through the BellSouth network. BellSouth will not be liable for any compensation to the terminating carrier or to NewSouth. Both Parties' agree to compensate the other for any charges or costs for the delivery of Local Transit Traffic to a connecting carrier on behalf of the other Party. Additionally, the Parties agree that any billing to a third party or other telecommunications carrier under this section shall be pursuant to MECAB procedures.

#### 6. Frame Relay Service Interconnection

- 6.1 In addition to the Local Interconnection services set forth above, BellSouth will offer a network to network Interconnection arrangement between BellSouth's and NewSouth's frame relay switches as set forth below. The following provisions will apply only to Frame Relay Service and Exchange Access Frame Relay Service in those states in which NewSouth is certified and providing Frame Relay Service as a Local Exchange Carrier and where traffic is being exchanged between NewSouth and BellSouth Frame Relay Switches in the same LATA.
- The Parties agree to establish two-way Frame Relay facilities between their respective Frame Relay Switches to the mutually agreed upon Frame Relay Service point(s) of interconnection ("POI(s)") within the LATA. All POIs shall be within the same Frame Relay Network Serving Areas as defined in Section A40 of BellSouth's General Subscriber Service Tariff except as set forth in this Attachment.
- Upon the request of either Party, such interconnection will be established where BellSouth and NewSouth have Frame Relay Switches in the same LATA. Where there are multiple Frame Relay switches in one central office, an interconnection with any one of the switches will be considered an interconnection with all of the switches at that central office for purposes of routing packet traffic.
- 6.4 The Parties agree to provision local and IntraLATA Frame Relay Service and Exchange Access Frame Relay Service (both intrastate and interstate) over Frame Relay interconnection facilities between the respective Frame Relay switches and the POIs.
- 6.5 The Parties agree to assess each other reciprocal charges for the facilities that each provides to the other according to the Percent Local Circuit Use Factor (PLCU), determined as follows:
- 6.5.1 If the data packets originate and terminate in locations in the same LATA, and consistent with the local definitions of the Agreement, the traffic is considered local. Frame Relay framed packet data is transported within Virtual Circuits (VC). For the purposes of this Agreement, if all the data packets transported within a VC remain within the LATA, then consistent with the local definitions in this Agreement, the traffic on that VC is local ("Local VC").
- 6.5.2 If the originating and terminating locations of the two way packet data traffic are not in the same LATA, the traffic on that VC is interLATA ("InterLATA VC").
- 6.5.3 The PLCU is determined by dividing the total number of Local VCs, by the total number of VCs on each Frame Relay facility. To facilitate implementation, NewSouth may determine its PLCU in aggregate, by dividing the total number of Local VCs in a given LATA by the total number VCs in that LATA. The Parties agree to renegotiate

the method for determining PLCU, at BellSouth's request, and within 90 days, if BellSouth notifies NewSouth that it has found that this method does not adequately represent the PLCU.

- 6.5.4 If there are no VCs on a facility when it is billed, the PLCU will be zero.
- 6.5.5 BellSouth will provide the circuit between the Parties' respective Frame Relay Switches. The Parties will be compensated as follows: BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring charges for the circuit based upon the rates set forth in BellSouth's Interstate Access Tariff, FCC No. 1. NewSouth will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed charges for the circuit by one-half of NewSouth's PLCU.
- The Parties agree to compensate each other for Frame Relay network-to-network interface (NNI) ports based upon the NNI rates set forth in BellSouth's Interstate Access Tariff, FCC No. 1. Compensation for each pair of NNI ports will be calculated as follows: BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring charges for the NNI port. NewSouth will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed non-recurring and recurring charges for the NNI port by NewSouth's PLCU.
- 6.7 Each Party agrees that there will be no charges to the other Party for its own subscriber's Permanent Virtual Circuit (PVC) rate elements for the local PVC segment from its Frame Relay switch to its own subscriber's premises. PVC rate elements include the Data Link Connection Identifier (DLCI) and Committed Information Rate (CIR).
- 6.8 For the PVC segment between the NewSouth and BellSouth Frame Relay switches, compensation for the PVC charges is based upon the rates in BellSouth's Interstate Access Tariff, FCC No. 1.
- 6.9 Compensation for PVC rate elements will be calculated as follows:
- 6.9.1 If NewSouth orders a VC connection between a BellSouth subscriber's PVC segment and a PVC segment from the BellSouth Frame Relay switch to the NewSouth Frame Relay switch, BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring PVC charges for the PVC segment between the BellSouth and NewSouth Frame Relay switches. If the VC is a Local VC, NewSouth will then invoice and BellSouth will pay, the total nonrecurring and recurring PVC charges billed for that segment. If the VC is not local, no compensation will be paid to NewSouth for the PVC segment.
- 6.9.2 If BellSouth orders a Local VC connection between a NewSouth subscriber's PVC segment and a PVC segment from the NewSouth Frame Relay switch to the BellSouth Frame Relay switch, BellSouth will invoice, and NewSouth will pay, the total non-recurring and recurring PVC and CIR charges for the PVC segment between the

BellSouth and NewSouth Frame Relay switches. If the VC is a Local VC, NewSouth will then invoice and BellSouth will pay the total non-recurring and recurring PVC and CIR charges billed for that segment. If the VC is not local, no compensation will be paid to NewSouth for the PVC segment.

- 6.9.3 The Parties agree to compensate each other for requests to change a PVC segment or PVC service order record, according to the Feature Change charge as set forth in the BellSouth access tariff BellSouth Tariff FCC No. 1.
- 6.9.4 If NewSouth requests a change, BellSouth will invoice and NewSouth will pay a Feature Change charge for each affected PVC segment.
- 6.9.4.1 If BellSouth requests a change to a Local VC, NewSouth will invoice and BellSouth will pay a Feature Change charge for each affected PVC segment.
- 6.9.5 The Parties agree to limit the sum of the CIR for the VCs on a DS1 NNI port to not more than three times the port speed, or not more than six times the port speed on a DS3 NNI port.
- 6.9.6 Except as expressly provided herein, this Agreement does not address or alter in any way either Party's provision of Exchange Access Frame Relay Service or interLATA Frame Relay Service. All charges by each Party to the other for carriage of Exchange Access Frame Relay Service or interLATA Frame Relay Service are included in the BellSouth access tariff BellSouth Tariff FCC No. 1.
- NewSouth will identify and report quarterly to BellSouth the PLCU of the Frame Relay facilities it uses, per section 6.5.3 above.
- 6.11 Either Party may request a review or audit of the various service components, consistent with the provisions of section E2 of the BellSouth State Access Services tariffs or Section 2 of the BellSouth FCC No.1 Tariff.
- 6.12 If during the term of this Agreement, BellSouth obtains authority to provide interLATA Frame Relay in any State, the Parties agree to renegotiate this arrangement for the exchange of Frame Relay Service Traffic within one hundred eighty (180) days of the date BellSouth receives interLATA authority. In the event the Parties fail to renegotiate this Section 6 within the one hundred eighty-day period, they will submit this matter to the appropriate State commission(s) for resolution.

#### 7. Operational Support Systems (OSS)

The terms, conditions and rates for OSS are as set forth in General Terms of this Agreement.

					R	ATES BY STA	TE			
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
LOCAL INTERCONNECTION (CALL TRANSPORT AND TERMINATION) - FOR TRANSIT	TRAFFIC									
End Office Switching, per mou	N/A	\$0.0018	NA	\$0.0016333	\$0.002562	\$0.00210	\$0.0023771	\$0.0015	\$0.0019295	\$0.0019
Direct Local Interconnection, per mou (same as End Office Switching in FL)		NA	\$0.002	NA	NA	NA	NA	NA	NA	NA
Tandem Switching, per mou	N/A	\$0.00063	\$0.00029	\$0.0006757	\$0.001096	\$0.0008	\$0.0007834	\$0.0006	\$0.0006843	\$0.000676
Tandem Local Interconnection, per mou (includes end office switching element)	N/A	NA	\$0.00325	NA	NA	NA	NA	NA	NA	NA
Multiple Tandem Switching, per mou (applies to initial tandem only), effective 10/99	N/A	NA	\$0.00125	NA	NA	NA	NA	NA	NA	NA
Local Intermediary, per mou (applies to transit traffic only)	N/A	NA	\$0.00125	NA	NA	NA	NA	NA	NA	NA
Tandem Intermediary Charge, per mou*	N/A	\$0.0015	NA	NA	\$0.001096	NA	NA	NA	NA	NA
*(This charge is applicable only to transit traffic and is applied in addition to applicable switching and/or interconnection charges.)										
TRUNK CHARGE - For trunks not subject to bill and keep										
Interim charges, both non-recurring and recurring, associated with interconnecting										1
trunk groups between BellSouth and CLEC-1 shall be as set forth in this Exhibit.										1
At such time as BellSouth develops a final cost based rate for such interconnecting										ł
trunk groups, the Parties shall amend this agreement to include such final cost										ł
based rates and shall true up such charges in accordance with this Attachment.										
Installation Trunk Side Service - per DS0										
NRC - 1st	TPP++	\$333.69	\$336.43	\$333.28	\$334.09	\$334.94	\$334.11	\$333.54	\$335.14	\$334.29
NRC - Add'l	TPP++	\$56.91	\$57.38	\$56.84	\$57.12	\$56.98	\$56.98	\$56.88	\$57.16	\$57.01
INTERCEPORE TRANSPORT F. A.										<b> </b>
INTEROFFICE TRANSPORT - For transport not subject to bill and keep										<b> </b>
Common (Shared) Transport	N1/A	<b>#</b> 0.00004	00.000010	<b>A</b> 0.00000	<b>\$0.0000010</b>	<b>A</b> 0.000000	<b>*</b> ***********************************	<b>#</b> 0.0004	<b>***</b>	00.00004
Common (Shared) Transport per mile per mou	N/A	\$0.00001	\$0.000012	*		\$0.0000083	\$0.0000091	\$0.00001	\$0.0000121	\$0.00004
Common (Shared) Transport Facilities Termination per mou	N/A	\$0.00045	\$0.0005	\$0.0004152	\$0.000426	\$0.00047	\$0.0004281	\$0.00034	\$0.0004672	\$0.00036
Interoffice Channel Transport - Dedicated - VG										l
Interoffice Transport - Dedicated - 2-wire VG	41.57/5	<b>*</b> 0.0000	<b>#</b> 0.0000	<b>#</b> 0.0000		<b>#</b> 0.0004	<b>.</b>	00.000	40.0070	00.0470
2-Wire VG - per mile per month	1L5XF	\$0.0339	\$0.0098	\$0.0222	NA	\$0.0384	NA	\$0.0282	\$0.0373	\$0.0173
2-Wire VG - Facility Termination per month	1L5XF 1L5XF	\$18.49	\$26.52	\$17.07	NA NA	\$19.10 \$76.20	NA NA	\$18.01 \$137.48	\$21.42	\$18.33 \$55.39
NRC - 2-wire VG - Facility Termination -1st		\$107.11	\$81.09	\$79.61					\$136.44	
NRC - 2-wire VG - Facility Termination - Add'l	1L5XF	\$48.27	\$54.83	\$36.08	NA	\$34.54	NA NA	\$52.58	\$51.37	\$17.37
NRC - 2-wire VG - Facility Termination - Disconnect Charge -1st	1L5XF	\$37.16	\$31.01	NA	NA	\$28.03		NA	NA NA	\$27.96
NRC - 2-wire VG - Facility Termination - Disconnect Charge -Add'l	1L5XF SOMAN	\$5.88	\$12.78 \$21.56	NA NA	NA NA	\$5.37 NA	NA NA	NA NA	NA NA	\$3.51 \$19.99
NRC - Manual Svc Order, per LSR		NA NA								
NRC - Manual Svc Order, per LSR disconnect NRC - Electronic Svc Order, per LSR	SOMAN SOMEC	NA \$3.50	\$3.84 \$2.75	NA \$3.50	NA NA	NA \$3.50	NA NA	NA \$3.50	NA \$3.50	NA \$3.50
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50 NA	\$0.42	\$3.50 NA	NA NA	\$3.50 NA	NA NA	\$3.50 NA	\$3.50 NA	\$3.50 NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	\$0.42 NA	\$18.94	NA NA	\$18.14	NA NA	\$38.07	\$39.63	NA NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.57	NA NA	\$18.94	NA NA	\$18.14	NA NA	\$38.07	\$39.63	NA NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order - Add 1	SOMAN	\$12.97	NA NA	\$18.94 NA	NA NA	\$8.06	NA NA	\$38.07 NA	\$39.63 NA	NA NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order-DisconnectAdd'l	SOMAN	\$12.97	NA NA	NA NA	NA NA	\$8.06	NA NA	NA NA	NA NA	NA NA
	SOIVIAIN	φ12.31	INA	INA	INA	φο.υυ	INA	INA	INA	INA
Interoffice Transport - Dedicated - 2 Wire VG - Kentucky & Mississippi				<u> </u>	1			<del> </del>		<del>                                     </del>
2-Wire VG - per mile per month	1L5NF	NA	NA	NA	\$0.0301	NA	\$0.0323	NA	NA	NA
2-Wire VG - Facility Termination per month	1L5NF	NA NA	NA NA	NA NA	\$27.66	NA NA	\$21.33	NA NA	NA NA	NA NA
NRC - 2-wire VG - Facility Termination -1st	1L5NF	NA NA	NA NA	NA NA	\$142.31	NA NA	\$106.72	NA NA	NA NA	NA NA
NRC - 2-wire VG - Facility Termination - 1st	1L5NF	NA NA	NA NA	NA NA	\$56.21	NA NA	\$48.83	NA NA	NA NA	NA NA
NRC - 2-wire VG - Facility Termination - Add1	1L5NF	NA NA	NA NA	NA NA	NA	NA NA	\$38.05	NA NA	NA NA	NA NA
NRC - 2-wire VG - Facility Termination - Disconnect Charge - Add'l	1L5NF	NA NA	NA NA	NA NA	NA NA	NA NA	\$7.23	NA NA	NA NA	NA NA
NRC - Manual Svc Order, per LSR	SOMAN	NA NA	NA NA	NA NA	\$19.99	NA NA	97.23 NA	NA NA	NA NA	NA NA
Version 3Q00:09/29/00	COMAN	INA	I IN/	11/7	ψ13.33	14/7	INA	11/7	IAU	11/7

			_		R	ATES BY STA	TE			
DESCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	NA	\$3.50	NA	\$3.50	NA	NA	NA
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	NA	NA	NA	\$25.52	NA	NA	NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	NA	NA	NA	\$25.52	NA	NA	NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	NA	NA	NA	NA	NA	\$11.34	NA	NA	NA
NRC - 2-wire VG - Incremental ChargeManual Svc Order-DisconnectAdd'l	SOMAN	NA	NA	NA	NA	NA	\$11.34	NA	NA	NA
Interoffice Transport - Dedicated - DS0 - 56/64 KBPS										
DS0 - per mile per month	1L5XK	\$0.0339	\$0.0098	\$0.0222	NA	\$0.0384	NA	\$0.0282	\$0.0373	\$0.1730
DS0 - Facility Termination per month	1L5XK	\$17.81	\$19.31	\$16.45	NA	\$18.37	NA	\$17.40	\$20.71	\$17.74
NRC - DS0 - Facility Termination - 1st	1L5XK	\$107.11	\$81.11	\$79.61	NA.	\$76.20	NA	\$137.48	\$136.44	\$55.39
NRC - DS0 - Facility Termination - Add'l	1L5XK	\$48.27	\$54.83	\$36.08	NA	\$34.54	NA	\$52.58	\$51.37	\$17.37
NRC - DS0 -Facility Termination - Disconnect Charge - 1st	1L5XK	\$37.16	\$31.01	NA	NA	\$28.03	NA	NA	NA	\$27.96
NRC - DS0 - Facility Termination - Disconnect Charge - Add'l	1L5XK	\$5.88	\$12.78	NA NA	NA NA	\$5.37	NA.	NA.	NA NA	\$3.51
NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	NA	NA	NA	NA	NA	\$19.99
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA NA	\$3.84	NA NA	NA NA	NA NA	NA.	NA.	NA NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	NA	\$3.50	NA	\$3.50	\$3.50	\$3.50
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
NRC - DS0 -Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	NA	\$38.07	\$39.63	NA
NRC -DS0 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	NA	\$38.07	\$39.63	NA
NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	NA	NA	NA	NA
NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	NA	NA	NA	NA
Interoffice Transport - Dedicated - DS0 - 56/64 KBPS - Kentucky & Mississippi  DS0 - per mile per month	1L5NK	NA NA	NA	NA	\$0.0301	NA	\$0.0323	NA	NA	NA
DS0 - Facility Termination per month	1L5NK	NA	NA	NA	\$26.95	NA	\$20.64	NA	NA	NA
NRC - DS0 - Facility Termination - 1st	1L5NK	NA	NA	NA	\$142.31	NA	\$106.72	NA	NA	NA
NRC - DS0 - Facility Termination - Add'I	1L5NK	NA	NA	NA	\$56.21	NA	\$48.83	NA	NA	NA
NRC - DS0 -Facility Termination - Disconnect Charge - 1st	1L5NK	NA	NA	NA	NA	NA	\$38.05	NA	NA	NA
NRC - DS0 - Facility Termination - Disconnect Charge - Add'l	1L5NK	NA	NA	NA	NA	NA	\$7.23	NA	NA	NA
NRC - Manual Svc Order, per LSR	SOMAN	NA	NA	NA	\$19.99	NA	NA	NA	NA	NA
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	NA	\$3.50	NA	\$3.50	NA	NA	NA
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	NA	NA	NA	NA	NA	NA	NA	NA
NRC - DS0 -Incremental ChargeManual Svc Order - 1st	SOMAN	NA	NA	NA	\$37.21	NA	\$25.52	NA	NA	NA
NRC -DS0 - Incremental ChargeManual Svc Order - Add'l	SOMAN	NA	NA	NA	\$37.21	NA	\$25.52	NA	NA	NA
NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	NA	NA	NA	NA	NA	\$11.31	NA	NA	NA
NRC - DS0 -Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	NA	NA	NA	NA	NA	\$11.34	NA	NA	NA
1 1										
Interoffice Transport - Dedicated - DS1										
Interoffice Transport - Dedicated - DS1  DS1 - per mile per month	1L5XL	\$0.6920	\$0.6013	\$0.4523	NA	\$0.7831	NA	\$0.5753	\$0.7598	\$0.3525
DS1 - per mile per month	1L5XL 1L5XL	\$0.6920 \$79.69	\$0.6013 \$99.79	\$0.4523 \$78.47	NA NA	\$0.7831 \$93.40	NA NA	\$0.5753 \$71.29	\$0.7598 \$94.98	\$0.3525 \$75.83
DS1 - per mile per month DS1 - Facility Termination per month	1L5XL	\$79.69	\$99.79	\$78.47	NA	\$93.40	NA	\$71.29	\$94.98	\$75.83
DS1 - per mile per month DS1 - Facility Termination per month NRC - DS1-Facility Termination - 1st	1L5XL 1L5XL	\$79.69 \$198.15	\$99.79 \$45.91	\$78.47 \$147.07	NA NA	\$93.40 \$140.49	NA NA	\$71.29 \$217.17	\$94.98 \$216.27	\$75.83 \$145.98
DS1 - per mile per month DS1 - Facility Termination per month NRC - DS1-Facility Termination - 1st NRC - DS1 - Facility Termination - Add'l	1L5XL	\$79.69 \$198.15 \$148.18	\$99.79 \$45.91 \$44.18	\$78.47	NA	\$93.40 \$140.49 \$106.69	NA	\$71.29	\$94.98	\$145.98 \$109.85
DS1 - per mile per month DS1 - Facility Termination per month NRC - DS1-Facility Termination - 1st	1L5XL 1L5XL 1L5XL	\$79.69 \$198.15	\$99.79 \$45.91	\$78.47 \$147.07 \$111.75	NA NA NA	\$93.40 \$140.49	NA NA NA	\$71.29 \$217.17 \$163.75	\$94.98 \$216.27 \$162.70	\$75.83 \$145.98

					R	ATES BY STA	TE	_		
DESCRIPTION	usoc	AL	FL	GA	KY	LA	MS	NC	sc	TN
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	NA	\$3.50	NA	\$3.50	\$3.50	\$3.50
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
NRC - DS1 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	NA	\$38.07	\$39.63	NA
NRC -DS1 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	NA	\$38.07	\$39.63	NA
NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$12.97	NA	NA	NA	\$8.06	NA	NA	NA	NA
NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	\$12.97	NA	NA	NA	\$8.06	NA	NA	NA	NA
Interoffice Transport - Dedicated - DS1 - Kentucky & Mississippi										
DS1 - per mile per month	1L5NL	NA	NA	NA	\$0.4500	NA	\$0.6598	NA	NA	NA
DS1 -Facility Termination per month	1L5NL	NA	NA	NA	\$55.05	NA	\$74.40	NA	NA	NA
NRC - DS1-Facility Termination - 1st	1L5NL	NA.	NA.	NA NA	\$298.18	NA	\$196.28	NA.	NA NA	NA.
NRC - DS1 - Facility Termination - Add'l	1L5NL	NA	NA	NA	\$231.23	NA	\$147.31	NA	NA	NA
NRC - DS1 - Facility Termination - Disconnect Charge - 1st	1L5NL	NA	NA NA	NA	NA NA	NA	\$26.56	NA.	NA NA	NA
NRC - DS1 - Facility Termination - Disconnect Charge - Add'l	1L5NL	NA NA	NA NA	NA NA	NA NA	NA NA	\$21.61	NA NA	NA NA	NA NA
NRC - Manual Svc Order, per LSR	SOMAN	NA NA	NA NA	NA NA	\$19.99	NA NA	NA NA	NA NA	NA NA	NA.
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA
NRC - Electronic Svc Order, per LSR	SOMEC	NA NA	NA NA	NA NA	\$3.50	NA NA	\$3.50	NA NA	NA NA	NA NA
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA NA	NA NA	NA NA	NA	NA NA	NA	NA NA	NA NA	NA NA
NRC - DS1 - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	\$25.52	NA NA	NA NA	NA NA
NRC -DS1 - Incremental ChargeManual Svc Order - 1st	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	\$25.52	NA NA	NA NA	NA NA
NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	\$11.31	NA NA	NA NA	NA NA
NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	NA NA	NA NA	NA NA	NA NA	NA NA	\$11.34	NA NA	NA NA	NA NA
T T							* -			
Interoffice Transport - Dedicated - DS3						4				
DS3 - per mile per month	1L5XM	\$4.98	\$4.17	\$2.72	NA	\$14.04	NA	\$12.98	\$8.13	\$5.89
DS3 -Facility Termination per month	1L5XM	\$898.15	\$1,121.93	\$788.00	NA	\$1,101	NA	\$720.38	\$967.70	\$760.20
NRC - DS3 - Facility Termination -1st	1L5XM	\$511.77	\$557.69	\$511.10	NA	\$611.41	NA	\$794.94	\$606.72	\$625.91
NRC - DS3 - Facility Termination - Add'l	1L5XM	\$330.92	\$325.61	\$330.77	NA	\$304.90	NA	\$579.55	\$423.45	\$311.39
NRC - DS3 - Facility Termination - Disconnect Charge - 1st	1L5XM	\$121.72	\$111.56	\$122.31	NA	\$102.16	NA	NA	NA	\$103.36
NRC - DS3 - Facility Termination - Disconnect Charge - Add'l	1L5XM	\$118.54	\$108.34	\$119.14	NA	\$99.46	NA	NA	NA	\$100.59
NRC - Manual Svc Order, per LSR	SOMAN	NA	\$21.56	NA	NA	NA	NA	NA	NA	\$19.99
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	NA	\$3.50	NA	\$3.50	\$3.50	\$3.50
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA	NA	NA	NA	NA	NA	NA
NRC - DS3 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA	\$37.55	NA	\$50.25	NA	\$91.26	\$54.26	NA
NRC - DS3 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$38.48	NA	\$37.55	NA	\$50.25	NA	\$91.26	\$54.26	NA
NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect1st	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	NA	NA	NA	NA
NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect—Add'l	SOMAN	\$19.03	NA	\$18.03	NA	\$20.94	NA	NA	NA	NA
		ļ								
Interoffice Transport - Dedicated - DS3 - Kentucky & Mississippi							2	ļ		
DS3 - per mile per month	1L5NM	NA	NA	NA	\$12.62	NA	\$15.02	NA	NA	NA
DS3 -Facility Termination per month	1L5NM	NA	NA	NA	\$1,204	NA	\$744.38	NA	NA	NA
NRC - DS3 - Facility Termination -1st	1L5NM	NA	NA	NA	\$946.23	NA	\$686.74	NA	NA	NA
NRC - DS3 - Facility Termination - Add'l	1L5NM	NA	NA	NA	\$516.89	NA	\$477.76	NA	NA	NA
NRC - DS3 - Facility Termination - Disconnect Charge - 1st	1L5NM	NA	NA	NA	NA	NA	\$125.56	NA	NA	NA
NRC - DS3 - Facility Termination - Disconnect Charge - Add'l	1L5NM	NA	NA	NA	NA	NA	\$118.79	NA	NA	NA
NRC - Manual Svc Order, per LSR	SOMAN	NA	NA	NA	\$19.99	NA	NA	NA	NA	NA
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NA

DC EC EC AN AN AN AN AN AN AN AN AN AN AN AN AN	\$14.61 \$14.61 \$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37 \$17.75	FL NA NA NA NA NA NA NA NA NA NA NA NA NA	\$13.91 \$382.95 \$62.40 NA NA NA \$15.50 NA \$18.94 \$8.42 NA	\$3.50 NA \$93.12 \$93.12 NA NA NA NA \$22.26 \$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	LA NA	\$3.50 NA \$64.97 \$27.08 \$27.08 \$27.08 \$17.83 \$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA	NC NA NA NA NA NA NA NA NA NA NA NA NA \$14.83 \$553.80 \$89.69 NA NA NA NA NA NA S3.50 NA NA S42.17 \$12.76	\$C NA NA NA NA NA NA NA \$16.83 \$554.00 \$88.58 NA NA NA NA NA	TN NA NA NA NA NA NA NA NA NA NA NA NA NA
V2 V2 V2 V2 V2 V2 V2 V2 V2 V2 V2 V4 V2 V4 V4 V4	NA NA NA NA NA NA NA \$14.61 \$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37 \$17.75	NA NA NA NA NA NA NA NA NA NA NA NA NA N	NA NA NA NA NA NA NA NA NA NA NA NA NA N	NA \$93.12 \$93.12 NA NA NA \$22.26 \$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	NA NA NA NA NA S14.94 \$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	NA \$64.97 \$64.97 \$27.08 \$27.08 \$17.83 \$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	NA NA NA NA NA NA NA NA NA NA NA NA NA N	NA NA NA NA NA NA \$16.83 \$554.00 \$88.58 NA NA NA NA	NA NA NA NA NA NA NA S19.02 \$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
AN IAN IAN IAN IAN IAN IAN IAN IAN IAN I	NA NA NA NA NA \$14.61 \$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37 \$17.75	NA NA NA NA NA \$29.33 \$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	NA NA NA NA NA \$13.91 \$382.95 \$62.40 NA NA NA NA NA NA NA S3.50 NA \$18.94 \$8.42	\$93.12 \$93.12 NA NA NA \$22.26 \$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	NA NA NA NA S14.94 \$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$64.97 \$64.97 \$27.08 \$27.08 \$17.83 \$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	NA NA NA NA NA \$14.83 \$553.80 \$89.69 NA NA NA NA NA NA NA S3.50 NA \$42.17	NA NA NA NA \$16.83 \$554.00 \$88.58 NA NA NA NA	NA NA NA NA NA \$19.02 \$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
IAN IAN IAN IV2 V2 V2 V2 V2 IAN IAN IEC IEC IAN IAN IAN IAN IAN IAN IAN IAN IAN IAN	NA NA NA NA S14.61 \$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	NA NA NA NA \$29.33 \$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	NA NA NA NA \$13.91 \$382.95 \$62.40 NA NA NA NA NA S3.50 NA \$18.94 \$8.42	\$93.12 NA NA \$22.26 \$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	NA NA NA NA \$14.94 \$347.49 \$53.76 \$53.76 \$6.60 NA NA \$3.50 NA \$18.14	\$64.97 \$27.08 \$27.08 \$17.83 \$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	NA NA NA \$14.83 \$553.80 \$89.69 NA NA NA NA NA \$3.50 NA	NA NA NA \$16.83 \$554.00 \$88.58 NA NA NA NA	NA NA NA NA \$19.02 \$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 V2 V2 V2 V2 IAN IAN IEC IEC IAN IAN	\$14.61 \$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	NA NA \$29.33 \$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	NA NA \$13.91 \$382.95 \$62.40 NA NA NA NA S3.50 NA \$18.94 \$8.42	\$22.26 \$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA	NA NA \$14.94 \$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$27.08 \$27.08 \$17.83 \$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	NA NA \$14.83 \$553.80 \$89.69 NA NA NA NA NA NA S3.50 NA \$42.17	NA NA \$16.83 \$554.00 \$88.58 NA NA NA NA NA	NA NA \$19.02 \$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 V2 V2 V2 V2 VA IAN IEC IEC IAN IAN	\$14.61 \$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	\$29.33 \$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	\$13.91 \$382.95 \$62.40 NA NA NA S3.50 NA \$18.94 \$8.42	\$22.26 \$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	\$14.94 \$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$27.08 \$17.83 \$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	\$14.83 \$553.80 \$89.69 NA NA NA NA NA S3.50 NA \$42.17	\$16.83 \$554.00 \$88.58 NA NA NA NA NA	NA \$19.02 \$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 V2 V2 V2 AN IAN IEC IEC IAN IAN IAN IAN IAN IAN IAN IAN IAN IAN	\$14.61 \$494.65 \$84.44 \$77.81 \$7.63 NA \$3.50 NA \$27.37 \$18.37	\$29.33 \$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	\$13.91 \$382.95 \$62.40 NA NA NA NA NA \$3.50 NA \$18.94	\$22.26 \$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	\$14.94 \$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$17.83 \$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	\$14.83 \$553.80 \$89.69 NA NA NA NA NA NA \$3.50 NA \$42.17	\$16.83 \$554.00 \$88.58 NA NA NA NA NA NA	\$19.02 \$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 V2 AN AN EC EC AN AN AN	\$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	\$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	\$382.95 \$62.40 NA NA NA NA S3.50 NA \$18.94 \$8.42	\$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	\$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	\$553.80 \$89.69 NA NA NA NA S3.50 NA \$42.17	\$554.00 \$88.58 NA NA NA NA NA NA NA	\$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 V2 AN AN EC EC AN AN AN	\$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	\$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	\$382.95 \$62.40 NA NA NA NA S3.50 NA \$18.94 \$8.42	\$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	\$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	\$553.80 \$89.69 NA NA NA NA S3.50 NA \$42.17	\$554.00 \$88.58 NA NA NA NA NA NA NA	\$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 V2 AN AN EC EC AN AN AN	\$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	\$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	\$382.95 \$62.40 NA NA NA NA S3.50 NA \$18.94 \$8.42	\$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	\$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	\$553.80 \$89.69 NA NA NA NA S3.50 NA \$42.17	\$554.00 \$88.58 NA NA NA NA NA NA NA	\$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 V2 AN AN EC EC AN AN AN	\$494.65 \$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	\$386.34 \$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	\$382.95 \$62.40 NA NA NA NA S3.50 NA \$18.94 \$8.42	\$585.15 \$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	\$347.49 \$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$487.62 \$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	\$553.80 \$89.69 NA NA NA NA S3.50 NA \$42.17	\$554.00 \$88.58 NA NA NA NA NA NA NA	\$199.33 \$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 V2 IAN IAN IEC IEC IAN IAN IAN IAN IAN IAN IAN IAN IAN IAN	\$84.44 \$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	\$66.36 \$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	\$62.40 NA NA NA NA \$3.50 NA \$18.94 \$8.42	\$98.53 \$11.99 NA \$19.99 NA \$3.50 NA NA	\$59.75 \$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$84.35 \$77.69 \$8.95 NA NA \$3.50 NA \$25.52	\$89.69 NA NA NA NA S3.50 NA \$42.17	\$88.58 NA NA NA NA S3.50 NA	\$24.16 \$54.81 \$4.80 \$19.99 NA \$3.50
V2 V2 IAN IAN IEC IEC IAN IAN	\$77.81 \$7.63 NA NA \$3.50 NA \$27.37 \$18.37	\$67.91 \$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	NA NA NA NA \$3.50 NA \$18.94 \$8.42	\$11.99 NA \$19.99 NA \$3.50 NA NA	\$53.68 \$6.60 NA NA \$3.50 NA \$18.14	\$77.69 \$8.95 NA NA \$3.50 NA \$25.52	NA NA NA NA \$3.50 NA \$42.17	NA NA NA NA \$3.50 NA	\$54.81 \$4.80 \$19.99 NA \$3.50
V2 IAN IAN IEC IEC IAN IAN	\$7.63 NA NA \$3.50 NA \$27.37 \$18.37 \$17.75	\$5.92 \$21.56 \$3.84 \$2.75 \$0.42 NA	NA NA NA \$3.50 NA \$18.94 \$8.42	NA \$19.99 NA \$3.50 NA NA	\$6.60 NA NA \$3.50 NA \$18.14	\$8.95 NA NA \$3.50 NA \$25.52	NA NA NA \$3.50 NA \$42.17	NA NA NA \$3.50 NA	\$4.80 \$19.99 NA \$3.50
IAN IEC IEC IAN IAN	NA NA \$3.50 NA \$27.37 \$18.37 \$17.75	\$21.56 \$3.84 \$2.75 \$0.42 NA NA	NA NA \$3.50 NA \$18.94 \$8.42	\$19.99 NA \$3.50 NA NA	NA NA \$3.50 NA \$18.14	NA NA \$3.50 NA \$25.52	NA NA \$3.50 NA \$42.17	NA NA \$3.50 NA	\$19.99 NA \$3.50
IAN IEC IEC IAN IAN IAN	NA \$3.50 NA \$27.37 \$18.37 \$17.75	\$3.84 \$2.75 \$0.42 NA NA	NA \$3.50 NA \$18.94 \$8.42	NA \$3.50 NA NA NA	NA \$3.50 NA \$18.14	NA \$3.50 NA \$25.52	NA \$3.50 NA \$42.17	NA \$3.50 NA	NA \$3.50
EC EC IAN IAN IAN	\$3.50 NA \$27.37 \$18.37 \$17.75	\$2.75 \$0.42 NA NA	\$3.50 NA \$18.94 \$8.42	\$3.50 NA NA NA	\$3.50 NA \$18.14	\$3.50 NA \$25.52	\$3.50 NA \$42.17	\$3.50 NA	\$3.50
IEC IAN IAN IAN	NA \$27.37 \$18.37 \$17.75	\$0.42 NA NA	NA \$18.94 \$8.42	NA NA NA	NA \$18.14	NA \$25.52	NA \$42.17	NA	
IAN IAN IAN V4	\$27.37 \$18.37 \$17.75	NA NA	\$18.94 \$8.42	NA NA	\$18.14	\$25.52	\$42.17		INA
IAN IAN V4	\$18.37 \$17.75	NA	\$8.42	NA					NA
V4	\$17.75				\$8.Ub			\$13.55	NA NA
V4	-	NA	NA		C44 40	\$11.34			NA NA
	045.77			INA	\$11.40	\$16.05	NA	NA	INA
	0.45.77								
	\$15.77	\$30.50	\$14.99	\$23.38	\$16.21	\$19.03	\$15.87	\$18.05	\$20.14
V4	\$502.43	\$387.21	\$368.44	\$585.15	\$352.75	\$495.25	\$562.23	\$562.46	\$201.53
V4	\$86.68	\$67.22	\$64.05	\$98.53	\$61.33	\$86.56	\$92.67	\$91.57	\$24.83
V4	\$78.71	\$68.78	NA	NA	\$54.36	\$78.58	NA	NA	\$55.52
V4	\$8.53	\$6.79	NA	NA	\$7.28	\$9.84	NA	NA	\$5.51
IAN	NA	\$21.56	NA	\$19.99	NA	NA	NA	NA	\$19.99
IAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	NA
IEC	NA	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50
IEC	\$3.50	\$0.42	NA	NA	NA	NA	NA	NA	NA
IAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$42.17	\$43.64	NA
AN	\$18.73	NA	\$8.42	NA	\$8.06	\$11.34	\$12.76	\$13.55	NA
110	<b>605.50</b>	£42.52	<b>#20.20</b>	£42.00	£42.00	<b>#20.04</b>	<b></b>	<b>627.20</b>	\$40.27
									\$277.35
									\$233.26
	· ·								\$33.18
									\$22.30
									\$19.99
									NA no so
									\$3.50
									NA
									NA
		I NA	NA	NA	NA	NA	\$1.77	\$3.11	NA
	\$0.00						1		1
FI FI FI M	FHG FHG FHG FHG MAN MAN MEC MEC	FHG \$35.52 FHG \$503.57 FHG \$44.284 FHG \$46.28 FHG \$32.18 MAN NA MAN NA MEC \$3.50 MEC NA	FHG \$35.52 \$43.53 FHG \$503.57 \$242.45 FHG \$442.84 \$226.44 FHG \$46.28 \$41.13 FHG \$32.18 \$28.28 MAN NA \$21.56 MAN NA \$3.84 MEC \$3.50 \$2.75 MEC NA \$0.42	FHG \$35.52 \$43.53 \$38.36 FHG \$503.57 \$242.45 \$356.15 FHG \$442.84 \$226.44 \$312.89 FHG \$46.28 \$41.13 NA FHG \$32.18 \$28.28 NA MAN NA \$21.56 NA MAN NA \$3.84 NA MEC \$3.50 \$2.75 \$3.50 MEC NA \$0.42 NA MAN \$61.95 NA \$44.22	FHG \$35.52 \$43.53 \$38.36 \$43.80 FHG \$503.57 \$242.45 \$356.15 \$538.95 FHG \$442.84 \$226.44 \$312.89 \$464.94 FHG \$46.28 \$41.13 NA NA FHG \$32.18 \$28.28 NA NA MAN NA \$21.56 NA \$19.99 MAN NA \$3.84 NA NA MEC \$3.50 \$2.75 \$3.50 \$3.50 MEC NA \$0.42 NA NA MAN \$61.95 NA \$44.22 \$87.71	FHG \$35.52 \$43.53 \$38.36 \$43.80 \$43.80 FHG \$503.57 \$242.45 \$356.15 \$538.95 \$348.56 FHG \$442.84 \$226.44 \$312.89 \$464.94 \$300.30 FHG \$46.28 \$41.13 NA NA \$24.15 FHG \$32.18 \$28.28 NA NA \$21.31 MAN NA \$21.56 NA \$19.99 NA MAN NA \$3.84 NA NA NA NA MAN NA \$3.84 NA NA NA NA MAN NA \$3.50 \$2.75 \$3.50 \$3.50 MEC NA \$0.42 NA NA NA MAN \$61.95 NA \$44.22 \$87.71 \$42.34	FHG \$35.52 \$43.53 \$38.36 \$43.80 \$43.80 \$38.91 FHG \$503.57 \$242.45 \$356.15 \$538.95 \$348.56 \$494.83 FHG \$442.84 \$226.44 \$312.89 \$464.94 \$300.30 \$435.28 FHG \$46.28 \$41.13 NA NA \$24.15 \$46.85 FHG \$32.18 \$28.28 NA NA \$21.31 \$33.02 MAN NA \$21.56 NA \$19.99 NA NA MAN NA \$3.84 NA NA NA NA NA MAN NA \$3.84 NA NA NA NA NA MEC \$3.50 \$2.75 \$3.50 \$3.50 \$3.50 MEC NA \$0.42 NA NA NA NA MAN \$61.95 NA \$44.22 \$87.71 \$42.34 \$59.58	FHG \$35.52 \$43.53 \$38.36 \$43.80 \$43.80 \$38.91 \$35.68 FHG \$503.57 \$242.45 \$356.15 \$538.95 \$348.56 \$494.83 \$534.48 FHG \$442.84 \$226.44 \$312.89 \$464.94 \$300.30 \$435.28 \$462.69 FHG \$46.28 \$41.13 NA NA \$24.15 \$46.85 NA FHG \$32.18 \$28.28 NA NA \$21.31 \$33.02 NA MAN NA \$21.56 NA \$19.99 NA NA NA NA NA NA NA NA NA NA NA NA NA	FHG \$35.52 \$43.53 \$38.36 \$43.80 \$43.80 \$38.91 \$35.68 \$37.20 FHG \$503.57 \$242.45 \$356.15 \$538.95 \$348.56 \$494.83 \$534.48 \$534.81 FHG \$442.84 \$226.44 \$312.89 \$464.94 \$300.30 \$435.28 \$462.69 \$462.81 FHG \$46.28 \$41.13 NA NA \$24.15 \$46.85 NA NA FHG \$32.18 \$28.28 NA NA \$21.31 \$33.02 NA NA NA NA NA NA NA NA NA NA NA NA NA

					R	ATES BY STA	TE			
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
Local Channel - Dedicated - DS3				_			-	_		
DS3 - Facility Termination per month	TEFHJ	\$541.78	\$570.06	\$550.01	\$697.89	\$696.07	\$533.33	\$498.87	\$498.58	\$633.15
NRC - DS3 - Facility Termination - 1st	TEFHJ	\$640.54	\$903.37	\$639.50	\$1,091.00	\$594.71	\$526.67	\$562.25	\$735.42	\$726.1
NRC - DS3 - Facility Termination - Add'l	TEFHJ	\$426.28	\$528.05	\$426.40	\$661.23	\$396.54	\$493.71	\$527.88	\$519.31	\$411.6
NRC - DS3 - Facility Termination - Disconnect - 1st	TEFHJ	\$121.72	\$221.46	\$122.31	NA	\$113.75	\$42.41	NA	NA	\$103.3
NRC - DS3 - Facility Termination - Disconnect - Add'l	TEFHJ	\$118.54	\$154.90	\$119.14	NA	\$110.80	\$40.87	NA	NA	\$100.5
NRC - Manual Svc Order, per LSR	SOMAN	NA NA	\$21.56	NA NA	\$19.99	NA NA	NA NA	NA NA	NA NA	\$19.9
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.75	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.50	\$3.5
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.42	NA NA	NA	NA	NA NA	NA NA	NA	NA
NRC - DS3 -Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA	\$37.55	\$93.12	\$50.25	\$31.49	\$56.25	\$54.26	NA
NRC - DS3 -Incremental ChargeManual Svc Order - 1st	SOMAN	\$38.48	NA NA	\$37.55	\$93.12	\$50.25	\$31.49	\$56.25	\$54.26	NA NA
NRC - DS3 - Incremental ChargeManual Svc Order - Add r	SOMAN	\$19.03	NA NA	\$18.03	NA	\$20.94	\$25.35	\$30.23 NA	934.20 NA	NA NA
NRC - DS3 - Incremental ChargeManual Svc Order-Disconnect-Add'l	SOMAN	\$19.03	NA NA	\$18.03	NA NA	\$20.94	\$25.35	NA NA	NA NA	NA NA
NRC - DS3 - Incremental Chargeivianual Svc Order-Disconnect-Add i	SOMAN	\$19.03	INA	\$18.03	INA	\$20.94	\$25.35	INA	INA	INA
HANNELIZATION				_			_		_	
DS3 Channelization (DS3 to DS1)										1
per Channelized System (28 DS1) per month	SATCS	\$188.51	\$220.97	\$188.78	NA	\$182.64	NA	\$243.76	\$234.30	\$185.
NRC - 1st	SATCS	\$71.76	\$356.40	\$72.50	NA	\$60.96	NA	\$77.90	NA	\$61.
NRC - Add'l	SATCS	\$52.03	\$188.00	\$59.96	NA	\$50.46	NA	\$63.32	NA	\$50.
NRC -1st - Disconnect	SATCS	\$17.22	\$61.64	\$11.02	NA	\$7.55	NA	\$4.61	\$11.99	\$3.9
NRC -Add'l - Disconnect	SATCS	\$12.05	\$58.98	\$12.02	NA	\$12.29	NA	\$15.76	\$12.05	\$12.0
per Interface per month (COCI)	SATCO	\$8.69	\$14.40	\$8.66	NA	\$8.80	NA	\$11.28	\$8.68	\$9.0
NRC - 1st	SATCO	NA	\$13.16	NA	NA	NA	NA	NA	NA	\$19.
NRC - Add'l	SATCO	NA	\$9.43	NA	NA	NA	NA	NA	NA	NA
NRC - Manual Svc Order, per LSR	SOMEC	\$3.50	\$21.56	\$3.50	NA	\$3.50	NA	\$3.50	\$3.50	\$3.5
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	N/
NRC - Electronic Svc Order, per LSR	SOMAN	\$15.61	NA	\$14.91	NA	\$19.74	NA	\$28.13	\$25.59	\$21.
NRC - Electronic Svc Order, per LSR disconnect	SOMAN	\$7.39	NA	\$6.63	NA	\$8.77	NA	\$13.33	\$8.92	\$10.
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	SOMAN	\$11.67	NA	\$10.82	NA	\$12.43	NA	\$18.26	NA	\$14.
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -Add'l	SOMAN	\$0.9469	NA	NA	NA	NA	NA	\$1.48	NA	\$1.4
DS3 Channelization (DS3 to DS1) - Kentucky & Mississippi										+
per Channelized System (28 DS1) per month	SATNS	NA	NA	NA	\$236.32	NA	\$247.40	NA	NA	N/
NRC - 1st	SATNS	NA	NA	NA	\$425.41	NA	\$79.94	NA	NA	N/
NRC - Add'l	SATNS	NA	NA	NA	\$303.33	NA	\$65.20	NA	NA	N/
NRC -1st - Disconnect	SATNS	NA	NA	NA	NA	NA	\$5.58	NA	NA	N/
NRC -Add'l - Disconnect	SATNS	NA	NA	NA	NA	NA	\$15.85	NA	NA	N/
per Interface per month (COCI)	SATCO	NA	NA	NA	\$8.52	NA	\$11.35	NA	NA	N/
NRC - 1st	SATCO	NA	NA	NA	\$19.99	NA	NA	NA	NA	N/
NRC - Add'l	SATCO	NA	NA	NA	NA	NA	NA	NA	NA	N/
NRC - Manual Svc Order, per LSR	SOMAN	NA	NA	NA	\$3.50	NA	\$3.50	NA	NA	N/
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	NΑ
NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	NA	NA	NA	\$26.95	NA	NA	N/
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	NA	NA	NA	NA	\$11.98	NA	NA	N/
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	SOMAN	NA	NA	NA	\$41.47	NA	\$16.97	NA	NA	N/
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -Add'l	SOMAN	NA	NA	NA	\$11.99	NA	NA	NA	NA	N/
DS1 Channelization (DS1 to DS0)										

					R	ATES BY STA	TE			
ESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
per Channelized System (24 DS0) per month	SATC1	\$136.82	\$153.60	\$126.22	NA	\$209.87	NA	\$177.72	\$147.51	\$165.
NRC - 1st	SATC1	\$197.98	\$182.14	\$198.22	NA	\$193.63	NA	\$267.19	\$220.89	\$197.
NRC - Add'l	SATC1	\$123.12	\$125.18	\$123.59	NA	\$118.37	NA	\$161.43	\$137.15	\$119.
NRC -1sr - Disconnect	SATC1	\$30.18	\$19.52	\$31.03	NA	\$26.44	NA	\$34.55	NA	\$25.6
NRC -Add'l - Disconnect	SATC1	\$18.86	\$18.14	\$19.75	NA	\$16.83	NA	\$21.14	NA	\$15.8
- Interface (COCI)								,		
per OCU-DP(data) card per month (2.4-64kbs)	SATSA	\$1.66	\$2.20	\$1.86	NA	\$3.12	NA	\$2.88	\$2.34	\$2.4
NRC - 1st	SATSA	\$12.05	\$13.16	\$12.02	NA	\$12.29	NA	\$15.76	\$12.05	\$12.
NRC - Add'l	SATSA	\$8.69	\$9.43	\$8.66	NA	\$8.80	NA	\$11.28	\$8.68	\$9.0
per BRITE card per month	SATSA	\$3.41	\$3.83	\$3.71	NA	\$4.18	NA	\$3.76	\$4.21	\$3.
NRC - 1st	SATSA	\$12.05	\$13.16	\$12.02	NA	\$12.29	NA	\$15.76	\$12.05	\$12
NRC - Add'l	SATSA	\$8.69	\$9.43	\$8.66	NA	\$8.80	NA	\$11.28	\$8.68	\$9.
per VG card per month (DS0)	SATSA	\$0.86	\$1.45	\$1.17	NA	\$1.62	NA	\$1.64	\$1.47	\$1.
NRC - 1st	SATSA	\$12.05	\$13.16	\$12.02	NA	\$12.29	NA	\$15.76	\$12.05	\$12
NRC - Add'l	SATSA	\$8.69	\$9.43	\$8.66	NA.	\$8.80	NA NA	\$11.28	\$8.68	\$9
NRC - Manual Svc Order, per LSR	SOMAN	NA NA	\$21.56	NA NA	NA.	NA	NA NA	NA NA	NA	\$19
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	\$3.84	NA	NA	NA	NA	NA	NA	I N
NRC - Electronic Svc Order, per LSR	SOMEC	\$3.50	\$2.77	\$3.50	NA	\$3.50	NA	\$3.50	\$3.50	\$3
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	\$0.43	NA	NA	NA	NA	NA	NA	N
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	SOMAN	\$15.61	NA	\$14.75	NA	\$19.74	NA	\$28.13	\$25.59	\$25
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -Add'l	SOMAN	\$7.39	NA	\$6.55	NA NA	\$8.77	NA	\$13.33	\$8.92	\$15
Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - 1st	SOMAN	\$11.67	NA	\$10.70	NA	\$12.43	NA	\$18.26	NA	\$14
Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - Add'l	SOMAN	\$0.9469	NA	\$0.00	NA NA	NA NA	NA	\$1.48	NA	\$1
OS1 Channelization (DS1 to DS0) - Kentucky & Mississippi	0.0.07314				<b>*</b>		<b>#</b> 140.07		N.1.0	<b>_</b>
per Channelized System (24 DS0) per month	SASTN1	NA	NA	NA	\$200.01	NA	\$146.87	NA	NA	N
NRC - 1st	SASTN1	NA NA	NA	NA NA	\$302.82	NA NA	\$271.52	NA	NA	N N
NRC - Add'l	SASTN1	NA NA	NA NA	NA	\$184.20	NA NA	\$164.56	NA NA	NA NA	I N
NRC -1sr - Disconnect	SASTN1	NA		NA	NA		\$36.38			I N
NRC -Add'l - Disconnect	SASTN1	NA	NA	NA	NA	NA	\$11.98	NA	NA	I I
- Interface (COCI)	0.4.TO.4	NIA.	NA	NA	<b>CO.O.4</b>	NIA	<b>60.00</b>	NA	NA	N
per OCU-DP(data) card per month (2.4-64kbs)	SATSA	NA NA			\$2.94 \$15.86	NA NA	\$2.86		NA NA	I N
NRC - 1st	SATSA	NA	NA	NA	7.5.55	NA NA	\$15.85	NA		
NRC - Add'l	SATSA	NA	NA	NA	\$11.36	NA	\$11.35	NA	NA	N
per BRITE card per month	SATSA	NA NA	NA	NA	\$4.04	NA NA	\$3.88	NA NA	NA NA	l N
NRC - 1st	SATSA	NA NA	NA	NA	\$15.86	NA NA	\$15.85	NA	NA	l N
NRC - Add'l	SATSA	NA NA	NA	NA	\$11.36	NA NA	\$11.35	NA	NA	l N
per VG card per month (DS0)	SATSA	NA	NA	NA	\$1.40	NA NA	\$1.45	NA	NA	l N
NRC - 1st	SATSA	NA NA	NA	NA NA	\$15.86	NA NA	\$15.85	NA	NA	l N
NRC - Add'l	SATSA	NA	NA	NA	\$11.36	NA	\$11.35	NA	NA	l N
NRC - Manual Svc Order, per LSR	SOMAN	NA	NA	NA	\$19.99	NA	NA	NA	NA	l N
NRC - Manual Svc Order, per LSR disconnect	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	N
NRC - Electronic Svc Order, per LSR	SOMEC	NA	NA	NA	\$3.50	NA	\$3.50	NA	NA	N
NRC - Electronic Svc Order, per LSR disconnect	SOMEC	NA	NA	NA	NA	NA	NA	NA	NA	N
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -1st	SOMAN	NA	NA	NA	\$41.47	NA	\$19.74	NA	NA	N
Channel System - Incremental Cost - Manual Svc. Order vs. Electronic -Add'l	SOMAN	NA	NA	NA	\$11.99	NA	\$8.77	NA	NA	N
Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - 1st	SOMAN	NA	NA	NA	NA	NA	\$12.43	NA	NA	N
Incremental Cost-Manual Svc. Order vs. Elect -Disconnect - Add'l	SOMAN	NA	NA	NA	NA	NA	NA	NA	NA	N
al Interconnection Mid-Span Meet					1					₩
al Interconnection Mid-Span Meet Local Channel - Dedicated - DS1			1		<del> </del>					₩
Version 3Q00:09/29/00	1	1		l	l		l	1		ь

					R	ATES BY STA	TE			
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
DS1 Monthly Recurring per month	TEFHG	\$14.84	\$17.27	\$15.95	\$17.19	\$19.93	\$16.48	\$16.58	\$15.87	\$17.90
NRC - DS1 - 1st	TEFHG	\$546.69	\$553.19	\$545.80	\$547.16	\$548.70	\$546.77	\$546.22	\$548.72	\$547.48
NRC - DS1 - Add'l	TEFHG	\$471.61	\$477.22	\$470.84	\$472.01	\$473.34	\$471.68	\$471.20	\$473.36	\$472.29
NRC - DS1 - Disconnect Chg - 1st	TEFHG	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC - DS1 - Disconnect Chg - Add'l	TEFHG	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC - DS1 - Incremental ChargeManual Svc Order - 1st	SOMAN	\$89.52	90.0705	\$88.87	\$89.09	\$89.34	\$89.03	\$88.94	\$89.34	\$89.14
NRC - DS1 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$89.52	90.0705	88.8666	89.0879	89.3388	89.0255	\$88.94	\$89.34	\$89.14
NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect	SOMAN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
Local Channel - Dedicated - DS3										<del>                                     </del>
DS3 - Facility Termination per month	TEFHJ	\$214.60	\$245.65	\$231.61	\$275.82	\$287.11	\$218.84	\$232.80	\$222.56	\$285.50
NRC - DS3 - Facility Termination - 1st	TEFHJ	\$1.066.46	\$1,079.15	\$1.064.72	\$1.067.37	\$1.070.38	\$1.066.63	\$1.065.55	\$1.070.42	\$1.068.00
NRC - DS3 - Facility Termination - Add'I	TEFHJ	\$670.88	678.85.98	\$669.79	\$671.45	\$673.34	\$670.98	\$670.31	\$673.37	\$671.85
NRC - DS1 - Disconnect Chg - 1st	TEFHJ	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC - DS1 - Disconnect Chg - Add'l	TEFHJ	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NRC - DS3 -Incremental ChargeManual Svc Order - 1st	SOMAN	\$94.49	\$95.61	\$94.34	\$94.57	\$94.84	\$94.50	\$94.41	\$94.94	\$94.63
NRC - DS3 - Incremental ChargeManual Svc Order - Add'l	SOMAN	\$94.49	\$95.61	\$94.34	\$94.57	\$94.84	\$94.50	\$94.41	\$94.94	\$94.63
NRC - DS1 - Incremental ChargeManual Svc Order-Disconnect	SOMAN	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD	TBD
NOTES:										<del>                                     </del>
If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.										
Where the state Commission has adopted rates for the rate elements contained herein, it		+								$\vdash$
is the intent of the parties to reflect such rates in this exhibit and to apply the same consistent with applicable FCC and Commission rules and orders.										

# **Attachment 4**

**Physical Collocation** 

# BELLSOUTH PHYSICAL COLLOCATION

#### 1. Scope of Attachment

1.1 <u>Scope of Attachment.</u> The rates, terms, and conditions contained within this Attachment shall only apply when NewSouth is occupying the Collocation Space as a sole occupant or as a Host within a Premises location pursuant to Section 4.

All the negotiated rates, terms and conditions set forth in this Attachment pertain to collocation and the provisioning of Collocation Space.

- 1.2 Right to occupy. BellSouth shall offer to NewSouth collocation on rates, terms, and conditions that are just, reasonable, non-discriminatory and consistent with the rules of the Federal Communications Commission ("FCC"). Subject to Section 4 of this Attachment, BellSouth hereby grants to NewSouth a right to occupy that certain area designated by BellSouth within a BellSouth Premises, of a size which is specified by NewSouth and agreed to by BellSouth (hereinafter "Collocation Space"). BellSouth Premises include BellSouth Central Offices and Serving Wire Centers, as well as all buildings or similar structures owned or leased by BellSouth that house BellSouth Network Facilities and all structures that house facilities on public rights-of-way, including but not limited to, vaults containing loop concentrators and other similar structures. To the extent this Attachment does not include all the necessary rates, terms and conditions for BellSouth Premises other than BellSouth Central Offices, the Parties will negotiate said rates, terms, and conditions at the request for collocation at BellSouth Premises other than a Central Office. Notwithstanding the foregoing, BellSouth shall consider in its designation for cageless collocation any unused space within the BellSouth Premises. The size specified by NewSouth may contemplate a request for space sufficient to accommodate NewSouth's growth within a two year period.
- 1.2.1 <u>Space Reclamation</u>. In the event of space exhaust within a Central Office Premises, BellSouth may include in its documentation for the Petition for Waiver filing any unused space in the Central Office Premises. NewSouth will be responsible for any justification of unused space within its space, if such justification is required by the appropriate state commission.
- 1.3 <u>Use of Space</u>. NewSouth shall use the Collocation Space for the purposes of installing, maintaining and operating NewSouth's equipment (to include testing and monitoring equipment) necessary to interconnect with BellSouth services and facilities, including access to unbundled network elements, for the provision of telecommunications services. Pursuant to Section 5 following, NewSouth may at its option, place NewSouth-owned fiber entrance facilities to the Collocation Space. In addition to, and not in lieu of, interconnection to BellSouth services and facilities, NewSouth may connect to other interconnectors within the designated BellSouth Premises by purchasing BellSouth's Special Access Services from BellSouth's FCC or

State Access Services Tariffs. The Collocation Space may be used for no other purposes except as specifically described herein or authorized in writing by BellSouth.

- 1.4 <u>Rates and charges</u>. NewSouth agrees to pay the rates and charges identified in Exhibit A attached hereto.
- 1.5 <u>Service Coordination</u>. The Parties shall coordinate, where necessary, to ensure that the Collocation Space is provisioned in accordance with the specifications submitted by NewSouth in its Application, as affirmed by the Bona Fide Firm Order or as jointly amended thereafter. BellSouth will continue to provide the necessary infrastructure to support NewSouth's request(s) during NewSouth's occupancy of the Collocation Space.

#### 2. Space Notification

- 2.1 <u>Availability of Space</u>. Upon submission of an application pursuant to Section 6, BellSouth will permit NewSouth to physically collocate, pursuant to the terms of this Attachment, at any BellSouth Premises, unless BellSouth has determined that there is no space available due to space limitations or that physical collocation is not practical for technical reasons.
- 2.1.1 Unless otherwise specified, BellSouth will respond to an application within ten (10) calendar days as to whether space is available or not available within a BellSouth Premises. This interval excludes National Holidays. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.
- 2.1.2 BellSouth will respond to a Florida Application within fifteen (15) calendar days as to whether space is available or not available within a BellSouth Premises. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.
- 2.1.3 BellSouth will respond to a Louisiana Application within ten (10) calendar days for space availability for one (1) to ten (10) Applications; fifteen (15) calendar days for eleven (11) to twenty (20) Applications; and for more than twenty (20) Applications, it is increased by five (5) calendar days for every five additional Applications received within five (5) business days. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.
- 2.1.4 BellSouth will respond to a Mississippi Application within ten (10) business days as to whether space is available or not available within a BellSouth Premises. If the amount of space requested is not available, BellSouth will notify NewSouth of the amount of space that is available.
- 2.2 <u>Reporting</u>. Upon request from NewSouth, BellSouth will provide a written report ("Space Availability Report") specifying the amount of Collocation Space available at

the Premises requested, the number of collocators present at the Premises, any modifications in the use of the space since the last report on the Premises requested and the measures BellSouth is taking to make additional space available for collocation arrangements.

- 2.2.1 The request from NewSouth for a Space Availability Report must be written and must include the Premises and Common Language Location Identification ("CLLI") code of the Premises. Such information regarding Premises and CLLI code is located in the National Exchange Carriers Association (NECA) Tariff FCC No. 4.
- 2.2.2 BellSouth will respond to a request for a Space Availability Report for a particular Premises within ten (10) calendar days (in Mississippi, 10 business days) of receipt of such request. BellSouth will make best efforts to respond in ten (10) calendar (in Mississippi, 10 business days) days to such a request when the request includes from two (2) to five (5) Premises within the same state. The response time for requests of more than five (5) Premises shall be negotiated between the Parties. If BellSouth cannot meet the ten calendar day (in Mississippi, 10 business days) response time, BellSouth shall notify in writing NewSouth and inform NewSouth of the time frame under which it can respond.
- 2.3 <u>Denial of Application</u>. After notifying NewSouth that BellSouth has no available space in the requested Premises ("Denial of Application"), BellSouth will allow NewSouth, upon request, to tour the entire Premises within ten (10) calendar days (in Mississippi, 10 business days) of such Denial of Application. In order to schedule said tour within ten (10) calendar days (in Mississippi, 10 business days), the request for a tour of the Premises must be received by BellSouth within five (5) calendar days of the Denial of Application.
- 2.4 <u>Filing of Petition for Waiver</u>. Upon Denial of Application BellSouth will timely file a petition with the Commission pursuant to 47 U.S.C. § 251(c)(6).
- 2.5 Waiting List. On a first-come, first-served basis governed by the date of receipt of an application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. BellSouth will notify the telecommunications carriers on the waiting list when space becomes available according to how much space becomes available and the position of telecommunications carrier on said waiting list. NewSouth must submit an updated, complete, and correct application to BellSouth within 30 business days or notify BellSouth in writing that NewSouth wants to maintain its place on the waiting list either without accepting such space or accepting an amount of space less than its original request. If NewSouth does not submit such an application or notify BellSouth in writing as described above, BellSouth will offer such space to the next CLEC on the waiting list and remove NewSouth from the waiting list. Upon request, BellSouth will advise NewSouth as to its position on the list.

- 2.5.1 In Florida, on a first-come, first-served basis governed by the date of receipt of an Application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. Sixty (60) calendar days prior to space becoming available, if known, BellSouth will notify the Florida PSC and the telecommunications carriers on the waiting list by mail when space becomes available according to the position of telecommunications carrier on said waiting list. If not known sixty (60) calendar days in advance, BellSouth shall notify the Florida PSC and the telecommunications carriers on the waiting list within two days of the determination that space is available.
- 2.6 <u>Public Notification</u>. BellSouth will maintain on its Interconnection Services website a notification document that will indicate all Central Offices that are without available space. BellSouth shall update such document within ten (10) calendar days (in Mississippi, 10 business days) of the Denial of Application date. BellSouth will also post a document on its Interconnection Services website that contains a general notice where space has become available in a Central Office previously on the space exhaust list. BellSouth shall allocate said available space pursuant to the waiting list referenced in Section 2.5.
- 2.7 <u>State Agency Procedures</u>. Notwithstanding the foregoing, should any state or federal regulatory agency impose procedures or intervals different than procedures or intervals set forth in this section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for all applications submitted for the first time after the effective date thereof.

#### 3. Collocation Options

3.1 Cageless. In accordance and compliance with local building code, BellSouth shall allow NewSouth to collocate NewSouth's equipment and facilities without requiring the construction of a cage or similar structure and without requiring the creation of a separate entrance to the Collocation Space. BellSouth shall allow NewSouth to have direct access to its equipment and facilities 24 hours/day, 7days/week, but may require NewSouth to use a central entrance to the BellSouth Premises. BellSouth shall make cageless collocation available in single bay increments pursuant to Section 7. Except where NewSouth's equipment requires special technical considerations (e.g., special cable racking, isolated ground plane), BellSouth shall assign cageless Collocation Space in conventional equipment rack lineups where feasible. For equipment requiring special technical considerations, NewSouth must provide the equipment layout, including spatial dimensions for such equipment pursuant to generic requirements contained in BellCore (Telcordia) GR-63-Core and shall be responsible for constructing all special technical requirements associated with such equipment pursuant to Section 6.5 following.

- Cages and Adjacent Arrangement Enclosures. At NewSouth's option and upon request, BellSouth shall construct enclosures in compliance with NewSouth's collocation request and in accordance and compliance with local building code. At NewSouth's request, BellSouth shall permit NewSouth to subcontract the construction of physical collocation arrangements with a contractor certified by BellSouth ("BellSouth Certified Contractor"), provided however, that BellSouth shall not unreasonably withhold approval of contractors. When NewSouth elects to have BellSouth construct an enclosure, space enclosure charges will apply as set forth in Exhibit A to this Attachment. The space enclosure charges set forth in Exhibit A to this Attachment will not apply in cases where NewSouth elects to use a BellSouth Certified Contractor to construct an enclosure.
- 3.2.1 When NewSouth subcontracts the construction, NewSouth must arrange with a BellSouth Certified Contractor to construct a collocation arrangement enclosure in accordance with BellSouth's reasonable and nondiscriminatory guidelines and specifications and at NewSouth's sole expense. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard enclosure specification, NewSouth and NewSouth's BellSouth Certified Contractor must comply with local building code requirements. NewSouth's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary permits and/or licenses for such construction. BellSouth shall cooperate with NewSouth and provide, at NewSouth's expense, the documentation, including architectural drawings, necessary for NewSouth to obtain the zoning, permits and/or other licenses. BellSouth shall pass on to NewSouth its reasonable costs of providing the documentation. The BellSouth Certified Contractor shall bill NewSouth directly for all work performed for NewSouth pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Contractor. NewSouth must provide the local BellSouth building contact with two Access Keys used to enter the locked enclosure. Except in case of emergency, BellSouth will not access NewSouth's locked enclosure prior to notifying NewSouth.
- 3.2.2 BellSouth has the right to review NewSouth's plans and specifications prior to allowing construction to start. BellSouth shall complete its review within 15 calendar days. BellSouth has the right to inspect the enclosure after construction to make sure it is designed and constructed according to BellSouth's reasonable and nondiscriminatory guidelines and specifications and to require NewSouth to remove or correct at NewSouth's cost any structure that does not meet these standards.
- 3.2 <u>Shared (Subleased) Caged Collocation</u>. NewSouth may allow other telecommunications carriers to share NewSouth's caged collocation arrangement pursuant to terms and conditions agreed to by NewSouth ("Host") and other telecommunications carriers ("Guests") and pursuant to this section in accordance and compliance with local building code, except where the BellSouth Premises is located within a leased space and BellSouth is prohibited by said lease from offering such an option. NewSouth shall notify BellSouth in writing upon execution of any agreement

between the Host and its Guest within ten (10) calendar days of its execution. Further, such notice shall include the name of the Guest(s) and the term of the agreement, and shall contain a certification by NewSouth that said agreement imposes upon the Guest(s) the same terms and conditions for Collocation Space as set forth in this Attachment between BellSouth and NewSouth.

- 3.2.1 NewSouth shall be the sole interface and responsible Party to BellSouth for the purpose of submitting applications for initial and additional equipment placements of Guest; for assessment of rates and charges contained within this Attachment; and for the purposes of ensuring that the safety and security requirements of this Attachment are fully complied with by the Guest, its employees and agents. In the event the Host and Guest jointly submit an initial Application, only one Application Fee will be assessed. A separate initial Guest application shall require the assessment of a Subsequent Application Fee, as set forth in Exhibit A, if this application is not the initial application made for the arrangement. Notwithstanding the foregoing, Guest may arrange directly with BellSouth for the provision of the interconnecting facilities between BellSouth and Guest and for the provision of the services and access to unbundled network elements and BellSouth will bill the Guest directly for such services.
- 3.2.2 NewSouth shall indemnify and hold harmless BellSouth from any and all claims, actions, causes of action, of whatever kind or nature arising out of the presence of NewSouth's Guests in the Collocation Space except to the extent caused by BellSouth's sole negligence, gross negligence, or willful misconduct.
- 3.2.3 In making shared caged arrangements available, whether or not NewSouth serves as Host, BellSouth may not increase the cost of site preparation or nonrecurring charges above the cost of provisioning such a shared arrangement of similar dimensions and material to a single collocating party.
- 3.2.4 BellSouth will not place unreasonable restrictions on NewSouth's use of a cage, and as such will allow NewSouth to contract with other CLECs to share the cage in a sublease-type arrangement. If two (2) or more CLECs who have interconnection agreements with BellSouth utilize a shared collocation cage, BellSouth will permit each CLEC to order UNEs to and provision service from that shared collocation space, regardless of which CLEC was the original Collocator.
- 3.3 Adjacent Collocation. BellSouth will provide adjacent collocation arrangements ("Adjacent Arrangement") where space within the Premises is legitimately exhausted, subject to technical feasibility, where the Adjacent Arrangement does not interfere with access to existing or planned structures or facilities on the Premises property and where permitted by zoning and other applicable state and local regulations. The Adjacent Arrangement shall be constructed or procured by NewSouth and in conformance with BellSouth's reasonable and nondiscriminatory design and construction specifications. Further, NewSouth shall construct, procure, maintain and operate said Adjacent Arrangement(s) pursuant to all of the terms and conditions set

forth in this Attachment. Reasonable and nondiscriminatory rates shall be negotiated at the time of the request for the Adjacent Arrangement.

- 3.4.1 Should NewSouth elect such option, NewSouth must arrange with a BellSouth Certified Contractor to construct an Adjacent Arrangement structure in accordance with BellSouth's reasonable and nondiscriminatory guidelines and specifications. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard specification, NewSouth and NewSouth's BellSouth Certified Contractor must comply with local building code requirements. NewSouth's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such construction. NewSouth's BellSouth Certified Contractor shall bill NewSouth directly for all work performed for NewSouth pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Contractor. NewSouth must provide the local BellSouth building contact with two cards, keys or other access device used to enter the locked enclosure. Except in cases of emergency, BellSouth shall not access NewSouth's locked enclosure prior to notifying NewSouth. BellSouth will notify NewSouth by telephone of such access within a reasonable time.
- 3.4.2 BellSouth maintains the right to review NewSouth's plans and specifications prior to construction of an Adjacent Arrangement(s). BellSouth shall complete such review within 15 business days. BellSouth may inspect the Adjacent Arrangement(s) following construction and prior to the Commencement Date, as defined in Section 4.1 following, to ensure the design and construction comply with BellSouth's reasonable and nondiscriminatory guidelines and specifications. BellSouth may require NewSouth, at NewSouth's sole cost, to correct any deviations from BellSouth's reasonable and nondiscriminatory guidelines and specifications found during such inspection(s), up to and including removal of the Adjacent Arrangement, within five (5) business days of BellSouth's inspection for deviations that may cause harm to personnel or property and within a reasonable period of time for other deviations, unless the Parties mutually agree to an alternative time frame.
- 3.4.3 NewSouth shall provide a concrete pad, the structure housing the arrangement, heating/ventilation/air conditioning ("HVAC"), lighting, and all facilities that connect the structure (i.e. racking, conduits, etc.) to the BellSouth point of interconnection. At NewSouth's option, and where the local authority having jurisdiction permits, BellSouth shall provide an AC power source and access to physical collocation services and facilities subject to the same nondiscriminatory requirements as applicable to any other physical collocation arrangement. NewSouth's BellSouth Certified Contractor shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such arrangement.
- 3.4.4 BellSouth shall allow Shared (Subleased) Caged Collocation within an Adjacent Arrangement pursuant to the terms and conditions set forth in Section 3.3 preceding.

### 4. Occupancy

- 4.1 <u>Commencement Date</u>. The "Commencement Date" shall be the day NewSouth's equipment becomes operational as described in Article 4.2, following.
- 4.2 Occupancy. BellSouth will notify NewSouth in writing that the Collocation Space is ready for occupancy. NewSouth must notify BellSouth in writing that collocation equipment installation is complete and is operational with BellSouth's network. BellSouth may, for good cause and on a reasonable and nondiscriminatory basis, not accept orders for interconnected service until receipt of such notice. For purposes of this paragraph, NewSouth's telecommunications equipment will be deemed operational when cross-connected to BellSouth's network for the purpose of service provision.
- 4.3 Termination. Except where otherwise agreed to by the Parties, NewSouth may terminate occupancy in a particular Collocation Space upon thirty (30) calendar days prior written notice to BellSouth. Upon termination of such occupancy, NewSouth at its expense shall remove its equipment and other property from the Collocation Space. NewSouth shall have thirty (30) calendar days from the termination date to complete such removal, including the removal of all equipment and facilities of NewSouth's Guests; provided, however, that NewSouth shall continue payment of monthly fees to BellSouth until such date as NewSouth has fully vacated the Collocation Space. Should NewSouth or NewSouth's Guest fail to vacate the Collocation Space within thirty (30) calendar days from the termination date, BellSouth shall have the right to remove the equipment and other property of NewSouth or NewSouth's Guest at NewSouth's expense and with no liability for damage or injury to NewSouth or NewSouth's Guest's property unless caused by the gross negligence or intentional misconduct of BellSouth. Upon termination of occupancy, NewSouth shall surrender such Collocation Space to BellSouth in the same condition as when first occupied by the NewSouth except for ordinary wear and tear unless otherwise agreed to by the Parties. NewSouth shall be responsible for the cost of removing any enclosure, together with all support structures (e.g., racking, conduits), of an Adjacent Collocation arrangement at the termination of occupancy and restoring the grounds to their original condition.

### 5. Use of Collocation Space

5.1 Equipment Type. In accordance with applicable FCC and State Commission rules and orders, BellSouth permits the collocation of any type of equipment necessary for interconnection to BellSouth's network or for access to unbundled network elements in the provision of telecommunications services. Such equipment used or useful for interconnection and access to unbundled network elements includes, but is not limited to transmission equipment including, but not limited to, optical terminating equipment and multiplexers, and digital subscriber line access multiplexers, routers, asynchronous transfer mode multiplexers, and remote switching modules. Nothing in this section

requires BellSouth to permit collocation of equipment used solely to provide enhanced services.

- 5.1.1 Such equipment must at a minimum meet the following BellCore (Telcordia) Network Equipment Building Systems (NEBS) General Equipment Requirements: Criteria Level 1 requirements as outlined in the BellCore (Telcordia) Special Report SR-3580, Issue 1; equipment design spatial requirements per GR-63-CORE, Section 2; thermal heat dissipation per GR-063-CORE, Section 4, Criteria 77-79; acoustic noise per GR-063-CORE, Section 4, Criterion 128, and National Electric Code standards. BellSouth may not impose safety requirements on NewSouth's equipment that are more stringent than the safety requirements it imposes on its own equipment. BellSouth may not object to the collocation of NewSouth's equipment on the ground that the equipment fails to comply with NEBS performance standards. If BellSouth denies collocation of a competitor's equipment, citing safety standards, BellSouth must provide to NewSouth within five (5) business days of the denial a list of all equipment that BellSouth locates with the premises in question, together with an affidavit attesting that all of that equipment meets or exceeds the safety standard that BellSouth contends NewSouth's equipment fails to meet. In the event that BellSouth believes that the collocated equipment will not be or is not being used for interconnection or access to unbundled network elements or determines that NewSouth's equipment does not meet NEBS Level 1 safety requirements, NewSouth will be given ten (10) calendar days to comply with the requirements or remove the equipment from the collocation space. If the parties do not resolve the dispute, BellSouth or NewSouth may file a complaint at the Commission seeking a formal resolution of the dispute.
- 5.1.2 NewSouth shall not use the Collocation Space for marketing purposes nor shall it place any identifying signs or markings in the area surrounding the Collocation Space or on the grounds of the Premises.
- 5.1.3 NewSouth shall place a plaque or other identification affixed to NewSouth's equipment necessary to identify NewSouth's equipment, including a list of emergency contacts with telephone numbers.
- Upon request, NewSouth will certify in writing to BellSouth that the equipment is necessary for interconnection or access to unbundled network elements. In the event that the Parties have a dispute about the type of equipment to be collocated, BellSouth may, within thirty (30) calendar days from the receipt of the written certification, file a complaint with the Commission seeking a formal determination that the equipment cannot be collocated in a BellSouth Premises. While the dispute is pending, BellSouth will not prevent or unreasonably delay installation of the disputed equipment in the Collocation space; however, NewSouth will not activate the equipment during the pendency of the dispute. NewSouth will be responsible for all costs incurred as a result of the installation should removal or modification of the equipment be required by the Commission's ruling.

- 5.2 Entrance Facilities. NewSouth may elect to place NewSouth-owned or NewSouthleased fiber entrance facilities into the Collocation Space. BellSouth will provide an interconnection point or points, physically accessible by both BellSouth and NewSouth, at which the fiber optic cable carrying NewSouth's circuits can enter BellSouth's premises, provided that BellSouth shall designate interconnection points as close as possible to its premises. NewSouth will provide and place fiber cable at the point of entrance of sufficient length to be pulled through conduit and into the splice location. NewSouth will provide and install a sufficient length of fire retardant riser cable, to which the entrance cable will be spliced, which will extend from the splice location to NewSouth's equipment in the Collocation Space. In the event NewSouth utilizes a non-metallic, riser-type entrance facility, a splice will not be required. NewSouth must contact BellSouth for instructions prior to placing the entrance facility cable in the manhole. NewSouth is responsible for maintenance of the entrance facilities. At NewSouth's option, BellSouth will accommodate where technically feasible a microwave entrance facility pursuant to separately negotiated terms and conditions. BellSouth will permit interconnection of copper or coaxial cable if such interconnection is first approved by the Commission.
- Dual Entrance. BellSouth will provide at least two interconnection points at each Premises where there are at least two such interconnection points available and where capacity exists. Upon receipt of a request for physical collocation under this Attachment, BellSouth shall provide NewSouth with information regarding BellSouth's capacity to accommodate dual entrance facilities. If conduit in the serving manhole(s) is available and is not reserved for another purpose for utilization within 12 months of the receipt of an application for collocation, BellSouth will make the requested conduit space available for installing a second entrance facility to NewSouth's arrangement. The location of the serving manhole(s) will be determined at the reasonable and nondiscriminatory discretion of BellSouth. Where dual entrance is not available due to lack of capacity, BellSouth will so state in the Application Response. BellSouth shall not deny a collocation application solely for the reason that dual entrance facilities are not available.
- 5.2.2 <u>Shared Use</u>. NewSouth may utilize spare capacity on an existing interconnector entrance facility for the purpose of providing an entrance facility to another NewSouth collocation arrangement within the same BellSouth Premises. NewSouth must arrange with BellSouth for BellSouth to splice the utilized entrance facility capacity to NewSouth-provided riser cable.
- 5.3 Splicing in the Entrance Manhole. Although not generally permitted, should NewSouth request a splice to occur in the entrance manhole(s), BellSouth, at its reasonable and nondiscriminatory discretion, may grant such a request. When the request for a splice is granted to NewSouth by BellSouth, NewSouth shall ensure its employees or agents entering and/or performing work in the entrance manhole(s) are trained and comply with BellSouth procedures and OSHA requirements regarding access to manholes and that BellSouth personnel are notified and present for all entrances and work performed in the entrance manhole(s). Manhole covers shall be

properly closed and secured at the conclusion of entry and/or work. Advance notification to BellSouth shall occur at a minimum of 48 hours prior to desired entry for normal work activities and at a minimum of 2 hours prior to desired entry in an out of service condition.

- 5.4 Demarcation Point. BellSouth, in a reasonable and nondiscriminatory manner, will designate the point(s) of demarcation between NewSouth's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For 2-wire and 4-wire connections to BellSouth's network, the demarcation point shall be a common block on the BellSouth designated conventional distributing frame. NewSouth shall be responsible for providing, and a supplier certified by BellSouth ("NewSouth's BellSouth Certified Supplier") shall be responsible for installing and properly labeling/stenciling, the common block, and necessary cabling pursuant to Section 6.4. For DS1 and DS3 connections, the demarcation point shall be a BellSouth provided DSX panel. For fiber connections, the demarcation point shall be a BellSouth provided LGX panel. NewSouth or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Section 5.5, following, and may self-provision cross-connects that may be required within the Collocation Space to activate service requests. At NewSouth's option and expense, a Point of Termination ("POT") bay or frame may be placed in the Collocation Space, but will not serve as the demarcation point. NewSouth must make arrangements with a BellSouth Certified Supplier for such placement.
- 5.4.1 In the event NewSouth's collocation space is at such a distance from the demarcation point that an intermediary transmission device is needed to prevent signal degradation and ensure compliance with industry standards, BellSouth, at its own expense, shall install such transmission device. In such cases, BellSouth will notify NewSouth and all relevant vendors that the distance poses a risk of degradation and that larger gauge cable should be used.
- NewSouth's Equipment and Facilities. NewSouth, or if required by this Attachment, NewSouth's BellSouth Certified Supplier, is solely responsible for the design, engineering, installation, testing, provisioning, performance, monitoring, maintenance and repair of the equipment and facilities used by NewSouth. Such equipment and facilities may include but are not limited to cable(s); equipment; and point of termination connections.
- BellSouth's Access to Collocation Space. From time to time BellSouth may require access to the Collocation Space. BellSouth retains the right to access such space for the purpose of making BellSouth equipment and building modifications (e.g., running, altering or removing racking, ducts, electrical wiring, HVAC, and cables). BellSouth will give three (3) business days' notice to NewSouth when access to the Collocation Space is required. NewSouth may elect to be present whenever BellSouth performs work in the Collocation Space. The Parties agree that NewSouth will not bear any of the expense associated with this work.

- 5.7 Access. Pursuant to Section 11, NewSouth shall have access to the Collocation Space twenty-four (24) hours a day, seven (7) days a week. NewSouth agrees to provide the name and social security number or date of birth or driver's license number of each employee, contractor, or agents of NewSouth provided with access keys or cards ("Access Keys") prior to the issuance of said Access Keys. BellSouth shall issue Access Keys within thirty (30) calendar days of such request for such Access Keys. Access Keys shall not be duplicated under any circumstances. NewSouth agrees to be responsible for all Access Keys and for the return of all said Access Keys in the possession of NewSouth employees, contractors, Guests, or agents after termination of the employment relationship, contractual obligation with NewSouth or upon the termination of this Attachment or the termination of occupancy of an individual collocation arrangement.
- 5.7.1 BellSouth will not repeatedly delay NewSouth's entry into a Premises or access to its collocated equipment. BellSouth will provide NewSouth with reasonable access to restroom facilities and parking.
- 5.7.2 <u>Lost or Stolen Access Keys</u>. NewSouth shall notify BellSouth in writing immediately in the case of lost or stolen Access Keys. Should it become necessary for BellSouth to re-key buildings or deactivate a card as a result of a lost Access Key(s) or for failure to return an Access Key(s), NewSouth shall pay for all reasonable costs associated with the re-keying or deactivating the card. Likewise, if NewSouth must re-key the lock on its collocation cage as a result of BellSouth losing the NewSouth provided key, BellSouth shall pay the reasonable costs for re-keying the cage and replacing keys.
- 5.8 Interference or Impairment. Notwithstanding any other provisions of this Attachment, equipment and facilities placed in the Collocation Space shall not interfere with or impair service provided by BellSouth or by any other interconnector located in the Premises; shall not endanger or damage the facilities of BellSouth or of any other interconnector, the Collocation Space, or the Premises; shall not compromise the privacy of any communications carried in, from, or through the Premises; and shall not create an unreasonable risk of injury or death to any individual or to the public. If BellSouth, on a reasonable and nondiscriminatory basis, determines that any equipment or facilities of NewSouth violates the provisions of this paragraph, BellSouth shall give written notice to NewSouth, which notice shall direct NewSouth to cure the violation within forty-eight (48) hours of NewSouth's actual receipt of written notice or, at a minimum, to commence curative measures within 24 hours and to exercise reasonable diligence to complete such measures as soon as possible thereafter. After receipt of the notice, the Parties agree to consult immediately and, if necessary, to inspect the arrangement. If NewSouth fails to take curative action within 48 hours or if the violation is of a character which poses an immediate and substantial threat of damage to property, injury or death to any person, or interference/impairment of the services provided by BellSouth or any other interconnector, then and only in that event, BellSouth may take such action as it deems appropriate to correct the violation, including without limitation the interruption of electrical power to NewSouth's equipment. BellSouth will endeavor, but is not required, to provide

notice to NewSouth prior to taking such action and shall have no liability to NewSouth for any damages arising from such action, except to the extent that such action by BellSouth constitutes willful misconduct.

- Personalty and its Removal. Subject to requirements of this Attachment, NewSouth may place or install in or on the Collocation Space such facilities and equipment, including storage for spare equipment, as it deems desirable for the conduct of business, provided that such equipment is telecommunications equipment, does not violate floor loading requirements, nor imposes or could impose or contains or could contain environmental conditions or hazards. Personal property, facilities and equipment placed by NewSouth in the Collocation Space shall not become a part of the Collocation Space, even if nailed, screwed or otherwise fastened to the Collocation Space, but shall retain their status as personalty and may be removed by NewSouth at any time. Any damage caused to the Collocation Space by NewSouth's employees, agents or representatives during the removal of such property shall be promptly repaired by NewSouth at its expense.
- Alterations. In no case shall NewSouth or any person acting on behalf of NewSouth make any rearrangement, modification, improvement, addition, repair, or other alteration which could affect in any way space, power, HVAC, and/or safety considerations to the Collocation Space or the BellSouth Premises without the written consent of BellSouth, which consent shall not be unreasonably withheld. The cost of any such specialized alterations shall be paid by NewSouth.
- Janitorial Service. NewSouth shall be responsible for the general upkeep and cleaning of the Caged Collocation Space and, if using a contractor, shall arrange directly with a BellSouth Certified Contractor for janitorial services. BellSouth shall provide a list of such contractors on a site-specific basis upon request.

#### 6. Ordering and Preparation of Collocation Space

- 6.1 Should any state regulatory or federal agency impose procedures or intervals different than procedures or intervals set forth in this section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for all applications submitted for the first time after the effective date thereof.
- 6.2 <u>Application for Space</u>. NewSouth shall submit an application document when NewSouth or NewSouth's Guest(s), as defined in Section 3.3, desires to request or modify the use of the Collocation Space.
- 6.2.1 <u>Initial Application</u>. For NewSouth or NewSouth's Guest(s) initial equipment placement, NewSouth shall submit to BellSouth a Physical Expanded Interconnection Application Document ("Application"), together with payment of the Application Fee as stated in Exhibit A. The Application is Bona Fide when it is complete and accurate,

meaning that all required fields on the application are completed with the appropriate type of information. The Bona Fide Application shall contain a detailed description and schematic drawing of the equipment to be placed in NewSouth's Collocation Space(s) and an estimate of the amount of square footage required.

- 6.2.2 Subsequent Application Fee. In the event NewSouth or NewSouth's Guest(s) desire to modify the use of the Collocation Space in a manner not reflected in the original application, NewSouth shall complete an Application document detailing all information regarding the modification to the Collocation Space. BellSouth shall determine what modifications, if any, to the Premises are required to accommodate the change requested by NewSouth in the Application. Such necessary modifications to the Premises may include but are not limited to, floor loading changes, changes necessary to meet HVAC requirements, changes to power plant requirements, and equipment additions. Where the subsequent Application does not require assessment for provisioning or construction work by BellSouth (e.g., NewSouth proposes to install a splitter), no Subsequent Application Fee will be required. The fee for an Application where the modification requested has limited effect (i.e., does not require assessment related to capital expenditure by BellSouth; e.g., NewSouth proposes to enhance power by adding a fuse) shall be the Subsequent Application Fee as set forth in Exhibit A. If the modification requires capital expenditure assessment (e.g., NewSouth proposes to add cable racking or increase floor space), the full Application Fee for the appropriate state shall apply.
- Application Response. In Alabama, Kentucky, North Carolina, and Tennessee, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond within ten (10) calendar days of receipt of an Application stating whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within twenty-three (23) business days, which will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7.
- In South Carolina, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond within ten (10) calendar days as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When multiple applications are submitted in a state within a fifteen (15) calendar day window, BellSouth will respond to the Bona Fide Applications as soon as possible, but no later than the following: within thirty (30) calendar days for Bona Fide Applications 6-10; within forty-two (42) calendar days for Bona Fide Applications 11-15. Response intervals for multiple

Bona Fide Applications submitted within the same timeframe for the same state in excess of 15 must be negotiated. All negotiations shall consider the total volume from all requests from telecommunications companies for collocation.

- In Florida, within fifteen (15) calendar days of receipt of a Bona Fide Application, BellSouth will respond as to whether space is available or not available within a particular Premises. Additionally, when space has been determined to be available or when a lesser amount of space than that requested is available, then with respect to the space available, BellSouth will provide a written response ("Application Response") including sufficient information to enable NewSouth to place a Firm Order. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. When NewSouth submits ten (10) or more Applications within ten (10) calendar days, the initial fifteen (15) day response period will increase by ten (10) days for every additional ten (10) Applications or fraction thereof.
- 6.3.3 In Georgia, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available for caged or cageless arrangements, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7.
- In Louisiana, in addition to the notice of space availability pursuant to Section 2.1, BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. BellSouth will respond as to whether space is available or not available within a particular Premises in accordance with Section 2. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 7. BellSouth will respond with a full Application Response within 30 calendar days for one to ten Applications; 35 calendar days for eleven to twenty Applications; and for requests of more than twenty Application it is increased by five calendar days for every five Applications received within five business days.
- 6.3.5 In Mississippi, in addition to the notice of space availability pursuant to Section 2., BellSouth will respond as to whether the Application is Bona Fide, and if it is not Bona Fide, the items necessary to cause the Application to become Bona Fide. When space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) business days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation

fees, as described in Section 7. When multiple applications are submitted in a state within a fifteen (15) business day window, BellSouth will respond to the Bona Fide Applications as soon as possible, but no later than the following: within thirty (30) business days for Bona Fide Applications 1-5; within thirty-six (36) business days for Bona Fide Applications 6-10; within forty-two (42) business days for Bona Fide Applications 11-15. Response intervals for multiple Bona Fide Applications submitted within the same timeframe for the same state in excess of 15 must be negotiated. All negotiations shall consider the total volume from all requests from telecommunications companies for collocation.

- Application Modifications. If a modification or revision is made to any information in the Bona Fide Application for Physical Collocation or the Bona Fide Application for Adjacent Collocation, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of NewSouth or necessitated by technical considerations, the application will be considered a new Application for the purposes of the response and provisioning intervals. If, at any time, BellSouth needs to reevaluate NewSouth's Bona Fide Application as a result of changes requested by NewSouth to NewSouth's original application, then BellSouth will charge NewSouth a Subsequent Application Fee. Major changes such as requesting additional space or adding additional equipment may require NewSouth to resubmit the Application with an Application Fee.
- 6.5 Bona Fide Firm Order. In Alabama, Kentucky, North Carolina, and Tennessee, NewSouth shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Bona Fide Firm Order to BellSouth. A Bona Fide Firm Order requires NewSouth to complete the Application/Inquiry process described in Section 6.2, preceding, and submit the Physical Expanded Interconnection Firm Order document (BSTEI-1P-F) indicating acceptance of the Application Response provided by BellSouth ("Bona Fide Firm Order"). The Bona Fide Firm Order must be received by BellSouth no later than five (5) business days after NewSouth's receipt of BellSouth's Application Response to NewSouth's Bona Fide Application. If the Bona Fide Firm Order is not received within this five (5) business day period, the construction and provisioning interval shall be extended day for day for each day after the 5<sup>th</sup> business day. If the Bona Fide Firm Order is not received within (30) calendar days of the Application Response, the Application will expire.
- 6.5.1 In South Carolina, NewSouth shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to BellSouth. A Firm Order shall be considered Bona Fide when NewSouth has completed the Application/Inquiry process described in Section 6.2, preceding and has submitted the Firm Order document indicating acceptance of the Application Response provided by BellSouth. The Bona Fide Firm Order must be received by BellSouth no later than thirty (30) calendar days after BellSouth's Application Response to NewSouth's Bona Fide Application or the Application will expire.

- In Mississippi, NewSouth shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to BellSouth. A Firm Order shall be considered Bona Fide when NewSouth has completed the Application/Inquiry process described in Section 6.2, preceding and has submitted the Firm Order document indicating acceptance of the Application Response provided by BellSouth. The Bona Fide Firm Order must be received by BellSouth no later than thirty (30) business days after BellSouth's Application Response to NewSouth's Bona Fide Application or the Application will expire.
- 6.5.3 BellSouth will establish a firm order date based upon the date BellSouth is in receipt of a Bona Fide Firm Order. BellSouth will acknowledge the receipt of NewSouth's Bona Fide Firm Order within seven (7) calendar days of receipt indicating that the Bona Fide Firm Order has been received. A BellSouth response to a Bona Fide Firm Order will include a Firm Order Confirmation containing the firm order date.
- 6.5.4 BellSouth will permit one accompanied site visit to NewSouth's designated collocation arrangement location after receipt of the Bona Fide Firm Order without charge to NewSouth.
- NewSouth must submit to BellSouth the completed Access Control Request Form (RF-2906-C) for all employees or agents requiring access to the BellSouth Premises a minimum of 30 calendar days prior to the date NewSouth desires access to the Collocation Space. NewSouth may submit such a request at any time subsequent to BellSouth's receipt of the Bona Fide Firm Order. In the event NewSouth desires access to the Collocation Space after submitting such a request but prior to access being approved, BellSouth shall permit NewSouth to access the Collocation Space, accompanied by a security escort at NewSouth's expense. NewSouth must request escorted access at least three (3) business days prior to the date such access is desired.
- 6.6 Construction and Provisioning
- In Alabama (Caged Only) Kentucky, Tennessee and North Carolina, BellSouth will complete construction for collocation arrangements within seventy-six (76) business days from receipt of an Application or as agreed to by the Parties. Under extraordinary conditions, the Company will complete construction for collocation arrangements within ninety-one (91) business days. Examples of extraordinary conditions include, but are not limited to, extended license or permitting intervals; major Company equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. In the event NewSouth submits a forecast as described in the following section three (3) months or more prior to the application date, the above intervals shall apply. In the event NewSouth submits such a forecast between two (2) months and three (3) months prior to the application date, the above intervals may be extended by one (1) additional month. In the event

NewSouth submits such a forecast less than two (2) months prior to the application date, the above intervals may be extended by sixty (60) calendar days. BellSouth will attempt to meet standard intervals for unforecasted requests and any interval adjustments will be discussed with NewSouth at the time the application is received. Raw space, which is space lacking the necessary infrastructure to provide collocation space (including but not limited to HVAC, Power, etc.), conversion time frames fall outside the normal intervals and are negotiated on an individual case basis. Additionally, installations to existing collocation arrangements for line sharing or line splitting, which include adding cable, adding cable and splitter, and adding a splitter, will be forty five (45) business days from receipt of an Application.

- 6.6.1.1 To be considered a timely and accurate forecast, NewSouth must submit to the Company the CLEC Forecast Form, as set forth in Exhibit C attached hereto, containing the following information: Central Office/Serving Wire Center CLLI, number of Caged square feet and/or Cageless bays, number of DS0, DS1, DS3 frame terminations, number of fused amps and planned application date.
- In Alabama, BellSouth will complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a Bona Fide Firm Order and ninety (90) calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- In Florida, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. For changes to collocation space after initial space completion, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of 45 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. If BellSouth does not believe that construction will be completed within the relevant time frame and BellSouth and NewSouth cannot agree upon a completion date, within 45 calendar days of receipt of the Bona Fide Firm Order for an initial request, and within 30 calendar days for Augmentations, BellSouth may seek an extension from the Florida PSC.
- 6.6.4 In Georgia, BellSouth will complete construction for caged collocation arrangements under ordinary conditions as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. BellSouth

will use best efforts to complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of 60 calendar days from receipt of a Bona Fide Firm Order and 90 calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.

- 6.6.5 In Louisiana, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of 90 calendar days from receipt of a Bona Fide Firm Order for an initial request, and within 60 calendar days for an Augmentation, or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within 120 calendar days of the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.
- 6.6.6 In Mississippi, excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of 120 calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within 180 calendar days of the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.

- In South Carolina, BellSouth will complete the construction and provisioning activities for cageless and caged collocation arrangements as soon as possible, but no later than 90 calendar days from receipt of a bona fide firm order. BellSouth may elect to renegotiate an alternative provisioning interval with NewSouth or seek a waiver from this interval from the Commission.
- Joint Planning Meeting. Unless otherwise agreed to by the Parties, a joint planning meeting or other method of joint planning between BellSouth and NewSouth will commence within a maximum of fifteen (15) calendar days from BellSouth's receipt of a Bona Fide Firm. At such meeting, the Parties will agree to the preliminary design of the Collocation Space and the equipment configuration requirements as reflected in the Bona Fide Application and affirmed in the Bona Fide Firm Order. BellSouth will complete all design work following the joint planning meeting. BellSouth will provide the floor plans to NewSouth within two weeks of a request for cageless collocation. For caged collocation, BellSouth will provide floor plans in the Application Response.
- 6.7.1 Unless otherwise agreed, the Collocation Space completion due date and Access Customer Termination Location (ACTL) codes will be provided to NewSouth during the joint planning meeting. BellSouth shall deliver Connecting Facilities Assignments (CFA) prior to Space Acceptance unless a POT Bay either serves as the demarcation point or has been installed by NewSouth in its collocation space. In the event a POT Bay does exist, the parties shall work cooperatively in the provisioning of CFAs in a timely manner.
- 6.8 <u>Permits</u>. Each Party or its agents will diligently pursue filing for the permits required for the scope of work to be performed by that Party or its agents within ten (10) calendar days of the completion of finalized construction designs and specifications.
- 6.9 Acceptance Walk Through. NewSouth and BellSouth will complete an acceptance walk through of each Collocation Space requested from BellSouth by NewSouth within fifteen (15) calendar days of BellSouth's notifying NewSouth that the collocation space is ready for occupancy. BellSouth will correct any deviations to NewSouth's original or jointly amended requirements within seven (7) calendar days after the walk through, unless the Parties jointly agree upon a different time frame.
- 6.10 <u>Use of BellSouth Certified Supplier</u>. NewSouth shall select a supplier which has been approved as a BellSouth Certified Supplier to perform all engineering and installation work required in the Collocation Space as set forth in TR 73503, which is consistent with industry standards. NewSouth may utilize its own employees to perform such work provided that NewSouth has been certified by BellSouth to perform such work. NewSouth must select BellSouth Certified Suppliers for transmission equipment, switching equipment and power equipment. BellSouth shall provide NewSouth with a list of BellSouth Certified Suppliers upon request. BellSouth shall not unreasonably withhold approval of any contractor proposed by NewSouth that meets the standard BellSouth criteria. The BellSouth Certified Supplier(s) shall be responsible for installing NewSouth's equipment and components, extending power cabling to the

BellSouth power distribution frame, performing operational tests after installation is complete, and notifying BellSouth's equipment engineers and NewSouth upon successful completion of installation. The BellSouth Certified Supplier shall bill NewSouth directly for all work performed for NewSouth pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Supplier. BellSouth shall consider certifying NewSouth or any supplier proposed by NewSouth.

- Alarm and Monitoring. BellSouth shall place environmental alarms in the Premises for the protection of BellSouth equipment and facilities. NewSouth shall be responsible for placement, monitoring and removal of environmental and equipment alarms used to service NewSouth's Collocation Space. Upon request, BellSouth will provide NewSouth with applicable tariffed service(s) to facilitate remote monitoring of collocated equipment by NewSouth. Both Parties shall use best efforts to notify the other of any verified environmental hazard known to that Party. The Parties agree to utilize and adhere to the Environmental Hazard Guidelines identified as Exhibit B attached hereto.
- 6.12 <u>Basic Telephone Service</u>. Upon request of NewSouth, BellSouth will provide basic telephone service to the Collocation Space under the rates, terms and conditions of the current tariff offering for the service requested.
- 6.13 Virtual Collocation Transition. BellSouth offers Virtual Collocation pursuant to the rates. terms and conditions set forth in its F.C.C. Tariff No. 1. For the interconnection to BellSouth's network and access to BellSouth unbundled network elements, NewSouth may purchase 2-wire and 4-wire cross-connects as set forth in Exhibit A, and NewSouth may place within its Virtual Collocation arrangements the telecommunications equipment set forth in Section 5.1. In the event physical Collocation Space was previously denied at a location due to technical reasons or space limitations, and that physical Collocation Space has subsequently become available, NewSouth may transition its virtual collocation arrangements to physical collocation arrangements and pay the appropriate non-recurring fees for physical collocation and for the rearrangement or reconfiguration of services terminated in the virtual collocation arrangement, as outlined in the appropriate BellSouth tariffs. In the event that BellSouth knows when additional space for physical collocation may become available at the location requested by NewSouth, such information will be provided to NewSouth in BellSouth's written denial of physical collocation. To the extent that (i) physical Collocation Space becomes available to NewSouth within 180 calendar days of BellSouth's written denial of NewSouth's request for physical collocation, and (ii) NewSouth was not informed in the written denial that physical Collocation Space would become available within such 180 calendar days, then NewSouth may transition its virtual collocation arrangement to a physical collocation arrangement and will receive a credit for any nonrecurring charges previously paid for such virtual collocation. NewSouth must arrange with a BellSouth Certified Supplier for the relocation of equipment from its virtual Collocation Space to its physical Collocation Space and will bear the cost of such relocation.

- 6.13.1 NewSouth may request the conversion of any existing virtual collocation arrangements to physical collocation arrangements. BellSouth will authorize the conversion of virtual collocation arrangements to physical collocation arrangements without requiring the relocation of the virtual arrangement where there are no extenuating circumstances or technical reasons that would cause the arrangement to become a safety hazard within the Premises or otherwise being in conformance with the terms and conditions of this Attachment and where (1) there is no change to the arrangement; and (2) the conversion of the virtual arrangement would not cause the arrangement to be located in the area of the Premises reserved for BellSouth's forecast of future growth; and (3) due to the location of the virtual collocation arrangement, the conversion of said arrangement to a physical arrangement would not impact BellSouth's ability to secure its own facilities. Notwithstanding the foregoing, if the BellSouth Premises is at or nearing space exhaust, BellSouth may authorize the conversion of the virtual arrangement to a physical arrangement even though BellSouth could no longer secure its own facilities.
- 6.14 <u>Cancellation</u>. If, at anytime prior to Space Acceptance, NewSouth cancels its order for the Collocation Space(s), BellSouth will bill NewSouth the applicable non-recurring rate for any and all work processes for which work has begun.
- 6.15 <u>Licenses.</u> NewSouth, at its own expense, will be solely responsible for obtaining from governmental authorities, and any other appropriate agency, entity, or person, all rights, privileges, and licenses necessary or required to operate as a provider of telecommunications services to the public or to occupy the Collocation Space.

#### 7. Rates and Charges

- 7.1 Recurring Charges. The recurring charges for space preparation begin on the date that NewSouth executes the written document accepting the Collocation Space pursuant to Section 6.9, or on the date NewSouth first occupies the Collocation Space, whichever is sooner. If NewSouth fails to schedule and complete a walkthrough pursuant to Section 6.9 within fifteen (15) days after BellSouth releases the space for occupancy, then BellSouth shall begin billing NewSouth for recurring charges as of the sixteenth (16th) day after BellSouth releases the Collocation Space.
- Order Processing and monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot, and Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation. NewSouth shall remit payment of the nonrecurring Firm Order Processing Fee coincident with submission of a Bona Fide Firm Order. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems.. In the event NewSouth opts for cageless

space, the space preparation fees will be assessed based on the total floor space dedicated to NewSouth as prescribed in Section 7.6.

- 7.3 Space Preparation Fee in Georgia. In Georgia, the Space Preparation Fee is a one-time fee, assessed per arrangement, per location. It recovers a portion of costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, power, building and support systems. This is a set fee of \$100 per square foot as established by the Georgia Public Service Commission Order in Docket No. 7061-U.. In the event NewSouth opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to NewSouth as prescribed in Section 7.6.
- Space Preparation Fee in North Carolina. In North Carolina, space preparation fees consist of monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot; Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation; and Power, assessed per the nominal –48V DC ampere requirements specified by NewSouth on the Bona Fide Application. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event NewSouth opts for cageless space, the space preparation fees will be assessed based on the total floor space dedicated to NewSouth as described in Section 7.6.
- 7.5 Cable Installation. Cable Installation Fee(s) are assessed per entrance fiber placed.
- 7.6 Floor Space. The floor space charge includes reasonable charges for lighting, HVAC, and other allocated expenses associated with maintenance of the Premises but does not include amperage necessary to power NewSouth's equipment. When the Collocation Space is enclosed, NewSouth shall pay floor space charges based upon the number of square feet so enclosed. When the Collocation Space is not enclosed, NewSouth shall pay floor space charges based upon the following floor space calculation: [(depth of the equipment lineup in which the rack is placed) + (0.5 x maintenance aisle depth) + (0.5 x wiring aisle depth)] X (width of rack and spacers). For purposes of this calculation, the depth of the equipment lineup shall consider the footprint of equipment racks plus any equipment overhang. BellSouth will assign unenclosed Collocation Space in conventional equipment rack lineups where feasible. In the event NewSouth's collocated equipment requires special cable racking, isolated grounding or other treatment which prevents placement within conventional equipment rack lineups, NewSouth shall be required to request an amount of floor space sufficient to accommodate the total equipment arrangement. Floor space charges are due beginning with the date on which BellSouth releases the Collocation Space for occupancy or on the date NewSouth first occupies the Collocation Space, whichever is sooner.

- 7.7 <u>Power</u>. BellSouth shall make available –48 Volt (-48V) DC power for NewSouth's Collocation Space at a BellSouth Power Board or BellSouth Battery Distribution Fuse Bay ("BDFB") at NewSouth's option within the Premises.
- 7.7.1 Recurring charges for -48V DC power will be assessed per ampere per month based upon the BellSouth Certified Supplier engineered and installed power feed fused ampere capacity. Rates include redundant feeder fuse positions (A&B) and common cable rack to NewSouth's equipment or space enclosure. NewSouth is responsible for contracting with a BellSouth Certified Supplier for power distribution feeder cable runs from a BellSouth BDFB or power board to NewSouth's equipment. When obtaining power from a BellSouth BDFB or miscellaneous fuse positions on a BellSouth power board, power cables must be engineered, furnished and installed by NewSouth using a BellSouth Certified power Supplier. Determination of the BellSouth BDFB or BellSouth power board as the power source will be made at BellSouth's sole, but reasonable, discretion. When obtaining power from a BDFB, fuses and power cables (A&B) must be engineered (sized), and installed by NewSouth's BellSouth Certified Supplier. When obtaining power from a BellSouth power board, power cables (A&B) must be engineered (sized), and installed by NewSouth's BellSouth Certified power Supplier. NewSouth's BellSouth Certified Supplier must also provide a copy of the engineering power specification prior to the Commencement Date. BellSouth will provide the power feeder cable support structure between the BellSouth BDFB or power board and NewSouth's arrangement area. NewSouth shall contract a BellSouth Certified Supplier who will be responsible for the following: power cable support structure within NewSouth's arrangement; power cable feeds; terminations of cable. Any terminations at a BellSouth power board must be performed by a BellSouth Certified power Supplier. NewSouth shall comply with all applicable National Electric Code (NEC), BellSouth TR-73503, BellCore (Telcordia) and ANSI Standards regarding power cabling.
- 7.7.2 If BellSouth has not previously invested in power plant capacity for collocation at a specific site, NewSouth has the option to add its own dedicated power plant; provided, however, that such work shall be performed by a BellSouth Certified Supplier who shall comply with BellSouth's reasonable and nondiscriminatory guidelines and specifications. Where the addition of NewSouth's dedicated power plant results in construction of a new power plant room, upon termination of this Agreement, NewSouth shall have the right to remove its equipment from the power plant room, but shall otherwise leave the room intact.
- 7.7.3 If NewSouth elects to install its own DC Power Plant, BellSouth shall provide AC power to feed NewSouth's DC Power Plant. Charges for AC power will be assessed per breaker ampere per month. Rates include the provision of commercial and standby AC power. When obtaining power from a BellSouth service panel, protection devices and power cables must be engineered (sized), and installed by NewSouth's BellSouth Certified Supplier except that BellSouth shall engineer and install protection devices and power cables for Adjacent Collocation. NewSouth's BellSouth Certified Supplier must also provide a copy of the engineering power specification prior to the

Commencement Date. Charges for AC power shall be assessed pursuant to the rates specified in Exhibit A. AC power voltage and phase ratings shall be determined on a per location basis. At NewSouth's option, NewSouth may arrange for AC power in an Adjacent Collocation arrangement from a retail provider of electrical power.

- 7.8 Cable Record charges. These charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.
- 7.9 Security Escort. A security escort will be required whenever NewSouth or its approved agent desires access to the entrance manhole or must have access to the Premises after the one accompanied site visit allowed pursuant to Section 6.4.2 prior to completing BellSouth's Security Training requirements and/or prior to Space Acceptance. Rates for a security escort are assessed in one-half (1/2) hour increments according to the schedule appended hereto as Exhibit A.
- 7.10 It is the Parties' intention that the rates contained in Exhibit A Rate "True-Up". hereto are the Commission approved rates where such rates exist. To the extent that no such Commission approved rates exist, the Parties agree that the prices reflected as interim herein shall be "trued-up" (up or down) based on final prices either determined by further agreement or by final order in a proceeding involving BellSouth before the regulatory authority for the state in which the services are being performed or any other body having jurisdiction over this Agreement (hereinafter "Commission"). Under the "true-up" process, the interim price for each service shall be multiplied by the volume of that service purchased to arrive at the total interim amount paid for that service ("Total Interim Price"). The final price for that service shall be multiplied by the volume purchased to arrive at the total final amount due ("Total Final Price"). The Total Interim Price shall be compared with the Total Final Price. If the Total Final Price is more than the Total Interim Price, NewSouth shall pay the difference to BellSouth. If the Total Final Price is less than the Total Interim Price, BellSouth shall pay the difference to NewSouth. Each Party shall keep its own records upon which a "true-up" can be based and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such "true-up," the Parties agree that the Commission shall be called upon to resolve such differences.
- 7.11 Other. If no rate is identified in the contract, the rate for the specific service or function will be negotiated by the Parties upon request by either Party. Payment of all other charges under this Attachment shall be due thirty (30) calendar days after receipt of the bill (payment due date). NewSouth will pay a late payment charge of the lessor of the legal rate or one and one-half percent (1-1/2%) assessed monthly on any balance which remains unpaid after the payment due date.

#### 8. Insurance

- 8.1 NewSouth shall, at its sole cost and expense, procure, maintain, and keep in force insurance as specified in this Section 8 and underwritten by insurance companies licensed to do business in the states applicable under this Attachment and having a Best's Insurance Rating of B++X (B++ ten).
- 8.2 NewSouth shall maintain the following specific coverage:
- 8.2.1 Commercial General Liability coverage in the amount of ten million dollars (\$10,000,000.00) or a combination of Commercial General Liability and Excess/Umbrella coverage totaling not less than ten million dollars (\$10,000,000.00). BellSouth shall be named as an Additional Insured on the Commercial General Liability policy as specified herein.
- 8.2.2 Statutory Workers Compensation coverage and Employers Liability coverage in the amount of one hundred thousand dollars (\$100,000.00) each accident, one hundred thousand dollars (\$100,000.00) each employee by disease, and five hundred thousand dollars (\$500,000.00) policy limit by disease.
- 8.2.3 All Risk property coverage on a full replacement cost basis insuring all of NewSouth's real and personal property situated on or within BellSouth's Central Office location(s).
- 8.2.4 NewSouth may elect to purchase business interruption and contingent business interruption insurance, having been advised that BellSouth assumes no liability for loss of profit or revenues should an interruption of service occur.
- 8.3 The limits set forth in Section 8.2 above may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days notice to NewSouth to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- All policies purchased by NewSouth shall be deemed to be primary and not contributing to or in excess of any similar coverage purchased by BellSouth. All insurance must be in effect on or before the date equipment is delivered to BellSouth's Premises and shall remain in effect for the term of this Attachment or until all NewSouth's property has been removed from BellSouth's Premises, whichever period is longer. If NewSouth fails to maintain required coverage, BellSouth may pay the premiums thereon and seek reimbursement of same from NewSouth.
- NewSouth shall submit certificates of insurance reflecting the coverage required pursuant to this Section a minimum of ten (10) business days prior to the commencement of any work in the Collocation Space. Failure to meet this interval may result in construction and equipment installation delays. NewSouth shall forward a certificate of insurance and notice of cancellation/non-renewal to BellSouth at the following address:

BellSouth Telecommunications, Inc. Attn.: Risk Management Coordinator 675 W. Peachtree Street Rm. 17H53 Atlanta, Georgia 30375

- 8.6 NewSouth must conform to recommendations made by BellSouth's fire insurance company to the extent BellSouth has agreed to, or shall hereafter agree to, such recommendations.
- 8.7 <u>Self-Insurance</u>. If NewSouth's net worth exceeds five hundred million dollars (\$500,000,000), NewSouth may elect to request self-insurance status in lieu of obtaining any of the insurance required in Sections 8.2.1 and 8.2.3. NewSouth shall provide audited financial statements to BellSouth thirty (30) days prior to the commencement of any work in the Collocation Space. BellSouth shall then review such audited financial statements and respond in writing to NewSouth in the event that self-insurance status is not granted to NewSouth. If BellSouth approves NewSouth for self-insurance, NewSouth shall annually furnish to BellSouth, and keep current, evidence of such net worth that is attested to by one of NewSouth's corporate officers. The ability to self-insure shall continue so long as the NewSouth meets all of the requirements of this Section. If the NewSouth subsequently no longer satisfies this Section, NewSouth is required to purchase insurance as indicated by Sections 8.2.1 and 8.2.3.
- 8.8 The net worth requirements set forth in Section 8.7 may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days' notice to NewSouth to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- 8.9 Failure to comply with the provisions of this Section will be deemed a material breach of this Attachment.

#### 9. Mechanics Liens

9.1 If any mechanics lien or other liens shall be filed against property of either Party (BellSouth or NewSouth), or any improvement thereon by reason of or arising out of any labor or materials furnished or alleged to have been furnished or to be furnished to or for the other Party or by reason of any changes, or additions to said property made at the request or under the direction of the other Party, the other Party directing or requesting those changes shall, within thirty (30) calendar days after receipt of written notice from the Party against whose property said lien has been filed, either pay such lien or cause the same to be bonded off the affected property in the manner provided by law. The Party causing said lien to be placed against the property of the other shall also defend, at its sole cost and expense, on behalf of the other, any action, suit or

proceeding which may be brought for the enforcement of such liens and shall pay any damage and discharge any judgment entered thereon.

# 10. Inspections

BellSouth shall conduct an inspection of NewSouth's equipment and facilities in the Collocation Space(s) prior to the activation of facilities between NewSouth's equipment and equipment of BellSouth. BellSouth may conduct an inspection if NewSouth adds equipment and may otherwise conduct routine inspections at reasonable intervals mutually agreed upon by the Parties. BellSouth shall provide NewSouth with a minimum of forty-eight (48) hours or two (2) business days, whichever is greater, advance notice of all such inspections. All costs of such inspection shall be borne by BellSouth.

# 11. Security and Safety Requirements

- 11.1 The security and safety requirements set forth in this section are as stringent as the security requirements BellSouth maintains at its own premises either for their own employees or for authorized contractors. Only BellSouth employees, BellSouth Certified Contractors and authorized employees, authorized Guests, pursuant to Section 3.3, preceding, or authorized agents of NewSouth will be permitted in the BellSouth Premises. NewSouth shall provide its employees and agents with picture identification which must be worn and visible at all times while in the Collocation Space or other areas in or around the Premises. The photo Identification card shall bear, at a minimum, the employee's name and photo, and the NewSouth name. BellSouth reserves the right to remove from its premises any employee of NewSouth not possessing identification issued by NewSouth or who have violated any of BellSouth's policies as outlined in the CLEC Security Training documents. NewSouth shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises. NewSouth shall be solely responsible for ensuring that any Guest of NewSouth is in compliance with all subsections of this Section 11.
- 11.1.1 NewSouth will be required, at its own expense, to conduct a statewide investigation of criminal history records for each NewSouth employee being considered for work on the BellSouth Premises, for the states/counties where the NewSouth employee has worked and lived for the past five years. Where state law does not permit statewide collection or reporting, an investigation of the applicable counties is acceptable.
- 11.1.2 NewSouth will be required to administer to their personnel assigned to the BellSouth Premises security training either provided by BellSouth, or meeting criteria defined by BellSouth.
- 11.1.3 NewSouth shall not assign to the BellSouth Premises any personnel with records of felony criminal convictions. NewSouth shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions, except for misdemeanor traffic

violations, without advising BellSouth of the nature and gravity of the offense(s). BellSouth reserves the right to refuse building access to any NewSouth personnel who have been identified to have misdemeanor criminal convictions. Notwithstanding the foregoing, in the even that NewSouth chooses not to advise BellSouth of the nature and gravity of any misdemeanor conviction, NewSouth may, in the alternative, certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions (other than misdemeanor traffic violations).

- 11.1.4 Neither Party shall not knowingly assign to the BellSouth Premises any individual who was a former employee of the other Party and whose employment with the other Party was terminated for a criminal offense whether or not the other Party sought prosecution of the individual for the criminal offense.
- 11.1.5 Neither Party shall not knowingly assign to the BellSouth Premises any individual who was a former contractor of the other Party and whose access to a BellSouth Premises was revoked due to commission of a criminal offense whether or not the other Party sought prosecution of the individual for the criminal offense.
- 11.1.6 For each NewSouth employee requiring access to a BellSouth Premises pursuant to this Attachment, NewSouth shall furnish BellSouth, prior to an employee gaining such access, a certification that the aforementioned background check and security training were completed. The certification will contain a statement that no felony convictions were found and certifying that the security training was completed by the employee. If the employee's criminal history includes misdemeanor convictions, NewSouth will disclose the nature of the convictions to BellSouth at that time. In the alternative, NewSouth may certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions other than misdemeanor traffic violations.
- 11.1.7 At BellSouth's request, NewSouth shall promptly remove from the BellSouth's Premises any employee of NewSouth BellSouth does not wish to grant access to its premises 1) pursuant to any investigation conducted by BellSouth or 2) prior to the initiation of an investigation in the event that an employee of NewSouth is found interfering with the property or personnel of BellSouth or another CLEC, provided that an investigation shall promptly be commenced by BellSouth.
- 11.2 Notification to BellSouth. BST reserves the right to interview NewSouth's employees, agents, or contractors in the event of wrongdoing in or around BellSouth's property or involving BellSouth's or another CLEC's property or personnel, provided that BellSouth shall provide reasonable notice to NewSouth's Security contact of such interview. NewSouth and its contractors shall reasonably cooperate with BellSouth's investigation into allegations of wrongdoing or criminal conduct committed by, witnessed by, or involving NewSouth's employees, agents, or contractors. Additionally, BellSouth reserves the right to bill NewSouth for all reasonable costs associated with reasonable and nondiscriminatory investigations involving its employees, agents, or contractors if it is established and mutually agreed in good faith

that NewSouth's employees, agents, or contractors are responsible for the alleged act. BellSouth shall bill NewSouth for BellSouth property which is stolen or damaged where an investigation determines the culpability of NewSouth's employees, agents, or contractors and where NewSouth agrees, in good faith, with the results of such investigation. NewSouth shall notify BellSouth in writing immediately in the event that the CLEC discovers one of its employees already working on the BellSouth premises is a possible security risk. Upon request of the other Party, the Party who is the employer shall discipline consistent with its employment practices, up to and including removal from the BellSouth Premises, any employee found to have violated the security and safety requirements of this section. NewSouth shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises.

- 11.3 <u>Use of Supplies</u>. Unauthorized use of telecommunications equipment or supplies either Party, whether or not used routinely to provide telephone service (e.g. plug-in cards,) will be strictly prohibited and handled appropriately. Costs associated with such unauthorized use may be charged to the offending Party, as may be all associated investigative costs.
- 11.4 <u>Use of Official Lines</u>. Except for non-toll calls necessary in the performance of their work, neither Party shall use the telephones of the other Party on the BellSouth Premises. Charges for unauthorized telephone calls may be charged to the offending Party, as may be all associated investigative costs.
- 11.5 <u>Accountability</u>. Full compliance with the Security requirements of this section shall in no way limit the accountability of either Party to the other for the improper actions of its employees.

### 12. Destruction of Collocation Space

12.1 In the event a Collocation Space is wholly or partially damaged by fire, windstorm, tornado, flood or by similar causes to such an extent as to be rendered wholly unsuitable for NewSouth's permitted use hereunder, then either Party may elect within ten (10) business days after such damage, to terminate this Attachment, and if either Party shall so elect, by giving the other written notice of termination, both Parties shall stand released of and from further liability under the terms hereof. If the Collocation Space shall suffer only minor damage and shall not be rendered wholly unsuitable for NewSouth's permitted use, or is damaged and the option to terminate is not exercised by either Party, BellSouth covenants and agrees to proceed promptly without expense to NewSouth, except for improvements not the property of BellSouth, to repair the damage. BellSouth shall have a reasonable time within which to rebuild or make any repairs, and such rebuilding and repairing shall be subject to delays caused by storms, shortages of labor and materials, government regulations, strikes, walkouts, and causes beyond the control of BellSouth, which causes shall not be construed as limiting factors, but as exemplary only. NewSouth may, at its own expense, accelerate the rebuild of its collocated space and equipment provided however that a BellSouth Certified Contractor is used and the necessary space preparation has been completed. Rebuild of equipment must be performed by a BellSouth Certified Vendor. If NewSouth's acceleration of the project increases the cost of the project, then those additional charges will be incurred by NewSouth. Where allowed and where practical, NewSouth may erect a temporary facility while BellSouth rebuilds or makes repairs. In all cases where the Collocation Space shall be rebuilt or repaired, NewSouth shall be entitled to an equitable abatement of rent and other charges, depending upon the unsuitability of the Collocation Space for NewSouth's permitted use, until such Collocation Space is fully repaired and restored and NewSouth's equipment installed therein (but in no event later than thirty (30) calendar days after the Collocation Space is fully repaired and restored). Where NewSouth has placed an Adjacent Arrangement pursuant to section 3.4, NewSouth shall have the sole responsibility to repair or replace said Adjacent Arrangement provided herein. Pursuant to this section, BellSouth will restore the associated services to the Adjacent Arrangement.

#### 13. Eminent Domain

13.1 If the whole of a Collocation Space or Adjacent Arrangement shall be taken by any public authority under the power of eminent domain, then this Attachment shall terminate with respect to such Collocation Space or Adjacent Arrangement as of the day possession shall be taken by such public authority and rent and other charges for the Collocation Space or Adjacent Arrangement shall be paid up to that day with proportionate refund by BellSouth of such rent and charges as may have been paid in advance for a period subsequent to the date of the taking. If any part of the Collocation Space or Adjacent Arrangement shall be taken under eminent domain, BellSouth and NewSouth shall each have the right to terminate this Attachment with respect to such Collocation Space or Adjacent Arrangement and declare the same null and void, by written notice of such intention to the other Party within ten (10) business days after such taking.

#### 14. Nonexclusivity

14.1 NewSouth understands that this Attachment is not exclusive and that BellSouth may enter into similar agreements with other Parties. Assignment of space pursuant to all such agreements shall be determined by space availability and made on a first come, first served basis.

#### 15. Notice of Non-Emergency and Emergency Work

15.1 BellSouth shall provide NewSouth with written notice five (5) business days prior to those instances where BellSouth or its subcontractors may be performing non-emergency work that has a substantial likelihood of directly affecting the Collocation

- Space occupied by NewSouth, or that is directly related to circuits that support NewSouth equipment.
- BellSouth will inform NewSouth by telephone of emergency related activity that BellSouth or its subcontractors may be performing that has a substantial likelihood of directly affecting the Collocation Space occupied by NewSouth, or is directly related to circuits that support NewSouth equipment. Notification of any emergency related activity shall be made as soon as practicable after BellSouth learns that such emergency activity is necessary so that NewSouth can take any action required to monitor or protect its service.

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – ALABAMA PHYSICAL COLLOCATION

# Rates marked with an asterisk (\*) are interim and are subject to true-up

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,760.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,134.00 Minimum
PE1SJ PE1SK PE1SL PE1SM	Space Preparation Fees (Note4) Firm Order Processing* Central Office Modifications* Common Systems Modifications – Cageless* Common Systems Modifications – Caged*	Per sq. ft. Per sq. ft. Per cage	\$2.24 \$3.01 \$102.16	\$1,211.00
PE1BW PE1C W	Space Enclosure (100 sq. ft. minimum) Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$178.65 \$17.52	NA NA
PE1PJ	Floor Space	Per sq. ft.	\$3.68	NA
PE1BD	Cable Installation	Per cable	NA	\$1,751.00
PE1PM	Cable Support Structure	Per entrance cable	\$19.67	NA
PE1FB PE1FD PE1FE PE1FG	Power -48V DC Power* 120V AC Power single phase* 240V AC Power single phase* 120V AC Power three phase* 277 AC Power three phase*	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$9.00 \$5.63 \$11.26 \$16.89 \$38.99	NA - - - -
PE1PL	Power (Note 3) -48V DC Power	Per amp	\$7.14	
PE1P2 PE1P4 PE1P1	Cross Connects 2-wire 4-wire DS-1	Per cross connect	\$.031 \$.062 \$1.28	First/Add'l \$33.68/\$31.79 \$33.63/\$31.67 \$52.93/\$39.87

PE1P3	DS-3		\$16.27	\$51.99/\$38.59
PE1F2	2-fiber		\$3.23	\$52.00/\$38.60
	AL	ABAMA (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1F4	4-fiber		\$5.73	\$64.54/\$51.14
PE1AX	Security Access System Security System*	Per central office	\$52.27	
PE1A1 PE1AA	New Access Card Activation* Administrative change, existing card*	Per card Per card	\$.059	\$55.57 \$15.58
PE1AR	Replace lost or stolen card*	Per card		\$45.56
PE1AK PE1AL	Initial Key Replace lost or stolen key	Per key Per key	NA NA	\$26.19 \$26.19
PE1SR	Space Availability Report*	Per premises requested		\$2,150.00
PE1PE PE1PF PE1PG PE1PH PE1B2 PE1B4	POT Bay Arrangements Prior to 6/1/99 2-Wire Cross-Connect 4-Wire Cross-Connect DS1 Cross-Connect DS3 Cross-Connect 2-Fiber Cross-Connect 4-Fiber Cross-Connect	Per cross connect	\$0.08 \$0.17 \$0.69 \$4.74 \$32.02 \$40.48	NA NA NA NA NA
PE1CR PE1CD PE1CO PE1C1 PE1C3 PE1CB	Cable Records VG/DS0 Cable VG/DS0 Cable DS1 DS3 Fiber Cable	Per request Per cable record Per each 100 pair Per T1TIE Per T3TIE Per cable record	NA NA NA NA NA	Note 2 Initial/Subsequent \$1708/\$1166 \$923.51/\$923.51 \$18.02/\$18.02 \$8.44/\$8.44 \$29.53/\$29.53 \$278.95/\$278.95

ALABAMA (continued)					
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Security Escort	Per half hr/add'l			
		half hr			
PE1BT	Basic Time		NA	\$33.85/\$21.45	
PE1OT	Overtime		NA	\$44.09/\$27.71	
PE1PT	Premium Time		NA	\$54.33/\$33.96	

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – FLORIDA PHYSICAL COLLOCATION

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request		\$3,791.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,160.00
PE1SJ PE1SK PE1SL	Space Preparation Fees (Note 5) Firm Order Processing Central Office Modifications Common Systems Modifications – Cageless	Per sq. ft. Per sq. ft.	\$2.58 \$2.96	\$1,211.00
PE1SM	Common Systems Modifications – Caged	Per cage	\$100.66	
PE1BW PE1C W	Space Enclosure (100 sq. ft. minimum) Wire Cage Wire Cage	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$205.93 \$20.20	NA NA
PE1PJ	Floor Space	Per sq. ft.	\$6.57	NA
PE1BD	Cable Installation	Per cable		\$1,826.00
PE1PM	Cable Support Structure		\$21.66	NA
PE1FB PE1FD PE1FE PE1FG	Power -48V DC Power 120V AC Power single phase 240V AC Power single phase 120V AC Power three phase 277 AC Power three phase	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$8.86 \$5.62 \$11.26 \$16.88 \$38.98	NA - - - -
PE1PL	Power (Note 4) -48V DC Power	Per amp	\$7.14	
	Cross Connects 2-wire 4-wire DS1 DS3 2-fiber 4-fiber	Per cross connect Per cross connect Per cross connect Per cross connect Per cross connect Per cross connect Per cross connect	\$.074 \$.148 \$1.29 \$17.48 \$2.96 \$5.66	First/Add'l \$34.53/\$32.51 \$34.54/\$32.53 \$54.15/\$40.94 \$53.28/\$39.65 \$53.28/\$39.66 \$66.08/\$52.47

	FLORIDA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
PE1AX	Security Access System Security System	Per premises	\$89.48		
PE1A1 PE1AA	New Access Card Activation Administrative change, existing card	Per card Per card	\$.06	\$56.03 \$15.71	
PE1AR	Replace lost or stolen card	Per card		\$45.93	
PE1AK PE1AL	Initial Kev Replace lost or stolen key	Per kev Per key	NA NA	\$26.41 \$26.41	
PE1SR	Space Availability Report	Per premises requested		\$2,168.00	
	POT Bay (Note 1)		NA	NA	
	Cable Records <sup>2</sup>			Note 3	
				initial/subsequent	
PE1CR PE1CD PE1CO PE1C1 PE1C3 PE1CB	Cable Records VG/DS0 Cable VG/DS0 Cable DS1 DS3 Fiber Cable	Per request Per cable record Per each 100 pair Per T1TIE Per T3TIE Per cable record	NA NA NA NA NA	\$1709/\$1166 \$923.86/\$923.86 \$18.03/\$18.03 \$8.44/\$8.44 \$29.54/\$29.54 \$279.05/\$279.05	
PE1BQ PE1OQ PE1PQ	Security Escort Basic Time Overtime Premium Time	Per ¼ hour	NA NA NA	\$10.89 \$13.64 \$16.40	

# Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note 1; POT Bays: BellSouth's Florida specific rates were established in the Florida Public Service Commission Docket No. 960833. The Commission did not set permanent rates for POT Bays, given the assumption by the Parties to the Proceeding that they will always provide their own POT Bays. It will be necessary for CLEC-1 to provide its own POT Bays per BellSouth specifications and provide the necessary information from which BellSouth can inventory.

Note 2: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

- Note 3: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.
- Note 4: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.
- Note 5: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – GEORGIA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and subject to true-up

USOC	Rate Element Description	Unit	Recurring	Non-Recurring
			Rate (RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,755.00
				<b>A</b>
PE1CA	Subsequent Application Fee	Per request	NA	\$3,130.00
				Minimum
75477				<b>*</b>
PE1BB	Space Preparation Fee	Per sq. ft.	NA	\$100.00
PE1BW PE1C W	Space Enclosure (100 sq. ft. minimum) Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$187.36 \$18.38	NA NA
55454	Floor Space		<b>.</b>	
PE1PJ	Zone A	Per sq. ft.	\$4.47	NA
PE1PK	Zone B	Per sq. ft.	\$4.47	NA
PE1BD	Cable Installation	Per cable	NA	\$1,693.00
				<b>+</b> 1,000100
PE1PM	Cable Support Structure	Per entrance	\$19.26	NA
		cable		
	Power	_		
PE1PL	-48V DC Power	Per amp	\$5.00	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.52	-
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.05	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.58	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.27	-
	Cross Connects	Per cross connect		First/Add'l
PE1P2	2-wire	. 3. 3.333 33.11.000	\$0.031	\$33.76/\$31.86
PE1P4	4-wire		\$0.061	\$33.77/\$31.80
PE1P1	DS-1		\$1.13	\$53.05/\$39.99
PE1P3	DS-3		\$14.43	\$52.14/\$38.71
PE1F2	2-fiber		\$2.86	\$52.14/\$38.72
PE1F4	4-fiber		\$5.08	\$64.74/\$51.31

	GEO	ORGIA (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1AX	Security Access System Security System*	Per premises	\$40.00	
PE1A1	New Access Card Activation*	Per card	\$.058	\$55.51
PE1AA	Administrative change, existing card*	Per card		\$15.56
PE1AR	Replace lost or stolen card*	Per card		\$45.50
PE1AK PE1AL	Initial Kev Replace lost or stolen key	Per kev Per key	NA NA	\$26.16 \$26.16
PE1SR	Space Availability Report*	Per premises requested		\$2,148.00
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.40	NA
PE1PF	4-Wire Cross-Connect		\$1.20	NA
PE1PG	DS1 Cross-Connect		\$1.20	NA
PE1PH	DS3 Cross-Connect		\$8.00	NA
PE1B2	2 Fiber Cross-Connect		\$38.79	NA
PE1B4	4 Fiber Cross-Connect		\$52.31	NA
	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
PE1CR	Cable Records	Per request	NA	\$1706/\$1164
PE1CD	VG/DS0 Cable	Per cable record	NA	\$922.38/\$922.38
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.00/\$18.00
PE1C1	DS1	Per T1TIE	NA	\$8.43/\$8.43
PE1C3	DS3	Per T3TIE	NA	\$29.49/\$29.49
PE1CB	Fiber Cable	Per cable record	NA	\$278.61/\$278.61
	Security Escort	Per half hr./Add'l		
	2000	half hr.		
PE1BT	Basic Time		NA	\$33.81/\$21.42
PE1OT	Overtime		NA	\$44.03/\$27.67
PE1PT	Premium Time		NA	\$54.26/\$33.92

## Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – KENTUCKY PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

	Rates marked with an asterisk (*) are interim and are subject to true-up.				
USOC	Rate Element Description	Unit	Recurring	Non-Recurring	
			Rate (RC)	Rate (NRC)	
PE1BA	Application Fee	Per request	NA	\$3,761.00	
PE1CA	Subsequent Application Fee	Per request	NA	\$3,135.00	
				Minimum	
	Space Preparation Fees				
	(Note 4)				
	Firm Order Processing*			\$1,202.00	
_	Central Office Modifications*	Per sq. ft.	\$2.38		
PE1SL	Common Systems	Per sq. ft.	\$3.30		
	Modifications – Cageless*				
PE1SM	Common Systems	Per cage	\$112.11		
	Modifications – Caged*				
	Space Enclosure (100 sq. ft.				
	minimum)				
	Welded Wire-mesh	Per first 100 sq. ft.	\$189.85	NA	
PE1C	Welded Wire-mesh	Per add'l 50 sq. ft.	\$18.62	NA	
W					
PE1PJ	Floor Space	Per sq. ft.	\$8.20	NA	
				<b>*</b> • •	
PE1BD	Cable Installation	Per cable	NA	\$1,755.00	
PE1PM	Cable Support Structure	Per entrance	\$20.14	NA	
FLIFIVI	Cable Support Structure	cable	Ψ20.14	INA	
+		Cable			
	Power				
	-48V DC Power*	Per amp	\$8.77	NA	
	120V AC Power single phase*	Per breaker amp	\$5.58	-	
	240V AC Power single phase*	Per breaker amp	\$11.16	-	
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.74	-	
	277 AC Power three phase*	Per breaker amp	\$38.65	_	
1 2 11 0	211 / 10 1 GWei allieg pilace	r or product dimp	φοσίου		
PE1PL	-48V DC Power (Note 3)	Per amp	\$7.68	NA	
	, ,		·		
	Cross Connects	Per cross connect		First/Add'l	
PE1P2	2-wire		\$0.037	\$33.67/\$31.78	
PE1P4	4-wire		\$0.075	\$33.66/\$31.70	
PE1P1	DS-1		\$1.51	\$52.97/\$39.90	
PE1P3	DS-3		\$19.15	\$52.04/\$38.62	
	2-fiber		\$3.80	\$52.04/\$38.63	
	4-fiber		\$6.75	\$64.59/\$51.18	

	KEN	TUCKY (continued)		
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
DE 4 4 ) (			<b>\$70.44</b>	
PE1AX	Security Access System Security System*	Per premises	\$78.11	
PE1A1	New Access Card Activation	Per card	\$.059	\$55.59
PE1AA	Administrative change, existing	Per card	ψ.000	\$15.59
DEAAD	card			<b>0.45.50</b>
PE1AR	Replace lost or stolen card	Per card		\$45.58
PE1AK	Initial Kev	Per kev	NA	\$26.20
PE1AL	Replace lost or stolen key	Per key	NA	\$26.20
PE1SR	Space Availability Report	Per premises		\$2,151
1 2 1010	opado / tranadinty / topott	requested		Ψ2,101
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect		
PE1PE	2-Wire Cross-Connect		\$0.06	NA
PE1PF	4-Wire Cross-Connect		\$0.15	NA
PE1PG	DS1 Cross-Connect		\$0.58	NA
PE1PH	DS3 Cross-Connect		\$4.51	NA
PE1B2	2 Fiber Cross-Connect		\$38.79	NA
PE1B4	4 Fiber Cross-Connect		\$52.31	NA
	Security Escort	Per half hr./Add'l		
		half hr.		
PE1BT	Basic Time		NA	\$33.86/\$21.46
PE1OT	Overtime		NA	\$44.10/\$27.72
PE1PT	Premium Time		NA	\$54.35/\$33.97
	Cable Records <sup>1</sup>			Note 2
				Initial/subsequent
				iiiiiai/subsequeni
PE1CR	Cable Records	Per request	NA	\$1709/1166
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.83/\$923.83
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.03/\$18.03
PE1C1	DS1	Per T1TIE	NA	\$8.44/\$8.44
PE1C3	DS3	Per T3TIE	NA	\$29.54/\$29.54
PE1CB	Fiber Cable	Per cable record	NA	\$279.05/\$279.05

## Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

- Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.
- Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.
- Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges..

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – LOUISIANA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring	Non-Recurring
			Rate (RC)	Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3756.00
DE4OA	Out a month of the Car	Danasassas	NIA	Ф04.04.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3131.00 Minimum
				Willimitani
	Space Preparation Fees (Note 4)			
PE1SJ	Firm Order Processing*			\$1,200.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.60	ψ.,200.00
PE1SL	Common Systems	Per sq. ft.	\$3.15	
	Modifications – Cageless*	'	·	
PE1SM	Common Systems	Per cage	\$105.87	
	Modifications – Caged*			
	Space Enclosure (100 sq. ft.			
	minimum)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$207.06	NA
PE1C	Welded Wire-mesh	Per add'l 50 sq. ft.	\$20.31	NA
W				
DEADI	Fl O	D-11-11-11	Ф <b>Г</b> О 4	NIA
PE1PJ	Floor Space	Per sq. ft.	\$5.94	NA
PE1BD	Cable Installation	Per cable	NA	\$1,753.00
DEADM	Coble Cure out Cture of use	Donontropo	<b>CO4.4C</b>	NIA
PE1PM	Cable Support Structure	Per entrance cable	\$21.16	NA
		Cabic		
	Power			
PE1PL	-48V DC Power*	Per amp	\$9.20	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.66	-
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.34	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$17.00	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$39.26	-
PE1PL	-48V DC Power (Note 3)	Per amp	\$7.15	NA
	,	'	·	
	Cross Connects	Per cross connect		First/Add'l
PE1P2	2-wire		\$0.036	\$33.61/\$31.76
PE1P4	4-wire		\$0.073	\$33.53/\$31.58
PE1P1	DS-1		\$1.20	\$52.80/\$39.76
PE1P3	DS-3		\$15.26	\$51.86/\$38.49

	LOUISIANA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Cross Connects (continued)	Per cross connect		First/Add'l	
PE1F2	2-fiber		\$3.03	\$51.86/\$38.49	
PE1F4	4-fiber		\$5.38	\$64.36/\$50.99	
PE1AX	Security Access System Security System*	Per premises	\$60.60		
PE1A1	New Access Card Activation*	Per card	\$.060	\$55.51	
PE1AA	Administrative change, existing card*	Per card		\$15.57	
PE1AR	Replace lost or stolen card	Per card		\$45.51	
PE1AK	Initial Kev	Per kev	NA	\$26.16	
PE1AL	Replace lost or stolen key	Per key	NA	\$26.16	
PE1SR	Space Availability Report*	Per premises requested		\$2,148	
	POT Bay Arrangements  Prior to 6/1/99	Per cross-connect			
PE1PE	2-Wire Cross-Connect		\$0.0776	NA	
PE1PF	4-Wire Cross-Connect		\$0.1552	NA	
PE1PG	DS1 Cross-Connect		\$0.6406	NA	
PE1PH	DS3 Cross-Connect		\$4.75	NA	
PE1B2	2 Fiber Cross-Connect		\$47.44	NA	
PE1B4	4 Fiber Cross-Connect		\$63.97	NA	
	Cable Records <sup>1</sup>			Note 2	
				Initial/subsequent	
PE1CR	Cable Records	Per request	NA	\$1706/\$1165	
PE1CD	VG/DS0 Cable	Per cable record	NA	\$922.51/\$922.51	
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.00/\$18.00	
PE1C1	DS1	Per T1TIE	NA	\$8.43/\$8.43	
PE1C3	DS3	Per T3TIE	NA	\$29.49/\$29.49	
PE1CB	Fiber Cable	Per cable record	NA	\$278.65/\$278.65	

	LOUISIANA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Security Escort	Per half hr./Add'l half hr.			
PE1BT	Basic Time		NA	\$33.97/\$21.53	
PE1OT	Overtime		NA	\$44.25/\$27.81	
PE1PT	Premium Time		NA	\$54.53/\$34.09	

## Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges..

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – MISSISSIPPI PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

Rates marked with an asterisk (*) are interim and are subject to true-up.				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,755.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,130.00
				Minimum
	Space Preparation Fees			
	(Note 4)			
PE1SJ	Firm Order Processing*			\$1,200.00
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.61	
PE1SL	Common Systems	Per sq. ft.	\$2.88	
	Modifications – Cageless*			
PE1SM	Common Systems	Per cage	\$97.85	
	Modifications – Caged*			
	Space Enclosure(100 sq. ft.			
	minimum)			
PE1BW	Welded Wire-mesh	Per first 100 sq. ft.	\$208.30	NA
PE1C	Welded Wire-mesh	Per add'l 50 sq. ft.	\$20.43	NA
W				
DEADI		D (1	<b>40.50</b>	
PE1PJ	Floor Space	Per sq. ft.	\$6.53	
PE1BD	Cable Installation	Per cable	NA	\$1,871.00
				ψ 1,01 1100
PE1PM	Cable Support Structure	Per entrance	\$19.90	NA
		cable	·	
	Power			
PE1PL	-48V DC Power*	Per amp	\$8.96	NA
PE1FB	120V AC Power single phase*	Per breaker amp	\$5.61	-
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.23	-
PE1FE	120V AC Power three phase*	Per breaker amp	\$16.84	-
PE1FG	277 AC Power three phase*	Per breaker amp	\$38.89	-
DE4DI	49\/ DC Dower (Note 2)	Doromo	<b>ቀ</b> ራ	K I A
PE1PL	-48V DC Power (Note 3)	Per amp	\$6.93	NA
	Cross Connects	Per cross connect		First/Add'l
PE1P2	2-wire	. 3. 3.333 33	\$.038	\$33.65/\$31.77
PE1P4	4-wire		\$.076	\$33.46/\$31.52
	, · · · · · · · · · ·		ψ.σ. σ	Ψοσο, Ψοο.

	MISSISSIPPI (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Cross Connects (continued)	Per cross connect	, ,	First/Add'l	
PE1P1	DS-1		\$1.30	\$52.73/\$39.70	
PE1P3	DS-3		\$16.55	\$51.78/\$38.43	
PE1F2	2-fiber		\$3.28	\$51.78/\$38.43	
PE1F4	4-fiber		\$5.83	\$64.27/\$50.91	
PE1AX	Security Access System Security System*	Per premises	\$85.54		
PE1A1	New Access Card Activation*	Per card	\$.061	\$55.50	
PE1AA	Administrative change, existing card*	Per card		\$15.56	
PE1AR	Replace lost or stolen card	Per card		\$45.50	
PE1AK	Initial Kev	Per kev	NA	\$26.16	
PE1AL	Replace lost or stolen key	Per key	NA	\$26.16	
PE1SR	Space Availability Report*	Per premises requested		\$2,147.00	
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect			
PE1PE	2-Wire Cross-Connect		\$0.1195	NA	
PE1PF	4-Wire Cross-Connect		\$0.2389	NA NA	
PE1PG	DS1 Cross-Connect		\$0.9862	NA	
PE1PH	DS3 Cross-Connect		\$5.81	NA	
PE1B2	2 Fiber Cross-Connect		\$38.79	NA	
PE1B4	4 Fiber Cross-Connect		\$52.31	NA	
	Cable Records <sup>1</sup>			Note 2	
				Initial/subsequent	
PE1CR	Cable Records	Per request	NA	\$1706/1164	
PE1CD	VG/DS0 Cable	Per cable record	NA	\$922.28/\$922.28	
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.00/\$18.00	
PE1C1	DS1	Per T1TIE	NA	\$8.42/\$8.42	
PE1C3	DS3	Per T3TIE	NA	\$29.49/\$29.49	
PE1CB	Fiber Cable	Per cable record	NA	\$278.58/\$278.58	

	MISSISSIPPI (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
	Security Escort	Per half hr./Add'l half hr.			
PE1BT	Basic Time		NA	\$33.80/\$21.42	
PE1OT	Overtime		NA	\$44.03/\$27.67	
PE1PT	Premium Time		NA	\$54.26/\$33.92	

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – NORTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

	Rates marked with an asterisk (*) are interim and are subject to true-up.				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
PE1BA	Application Fee*	Per request	NA	\$3,850.00	
PE1CA	Subsequent Application Fee	Per request	NA	\$3,119.00 Minimum	
	Space Preparation Fees (Note 4) Central Office Modification*	Per sq. ft.	\$1.57		
	Common Systems Modification  - Cageless*	Per sq. ft.	\$3.26		
	Common Systems Modification  – Caged*	Per cage	\$110.79		
	Power*	Per nominal –48v DC Amp	\$5.76		
	Space Enclosure (100 sq. ft. minimum)				
PE1BW PE1C W	Welded Wire-mesh* Welded Wire-mesh*	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$102.76 \$10.44	NA NA	
PE1PJ	Floor Space*	Per sq. ft.	\$3.45	NA	
PE1BD	Cable Installation*	Per cable	NA	\$2,305.00	
PE1PM	Cable Support Structure*	Per entrance cable	\$21.33	NA	
	Power				
PE1FB	-48V DC Power* 120V AC Power single phase*	Per amp Per breaker amp	\$6.65 \$5.50	NA -	
PE1FD	240V AC Power single phase*	Per breaker amp	\$11.01	-	
PE1FE PE1FG	120V AC Power three phase* 277 AC Power three phase*	Per breaker amp Per breaker amp	\$16.51 \$38.12	-	
PE1PL	-48V DC Power (Note 3)	Per amp	\$5.00	NA	
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2	Cross Connects (Note 1) 2-wire* 4-wire* DS-1* DS-3* 2-fiber	Per cross connect	\$0.32 \$0.64 \$2.34 \$42.84 \$2.94	First/Add'l \$41.78/\$39.23 \$41.91/\$39.25 \$71.02/\$51.08 \$69.84/\$49.43 \$51.97/\$38.59	

Attachment 4 Page 53

PE1F4 | 4-fiber | \$5.62 | \$64.53/\$51.15 |

	NORTH CAROLINA (continued)				
USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)	
PE1AX	Security Access System	Per premises	\$41.03		
	Security System*			<b>^</b> ••	
PE1A1	New Access Card Activation*	Per card	\$.062	\$55.30	
PE1AA	Administrative change, existing card*	Per card		\$15.51	
PE1AR	Replace lost or stolen card	Per card		\$45.34	
PE1AK	Initial Kev	Per kev	NA	\$26.18	
PE1AL	Replace lost or stolen key	Per key	NA	\$26.18	
PE1SR	Space Availability Report*	Per premises		\$2,140.00	
FLISIX	Space Availability Report	requested		\$2,140.00	
		requesteu			
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect			
PE1PE	2-Wire Cross-Connect		\$0.10	NA	
PE1PF	4-Wire Cross-Connect		\$0.19	NA	
PE1PG	DS1 Cross-Connect		\$0.79	NA	
PE1PH	DS3 Cross-Connect		\$4.85	NA	
PE1B2	2 Fiber Cross-Connect		\$45.30	NA	
PE1B4	4 Fiber Cross-Connect		\$61.09	NA	
	Security Escort	Per half hr./Add'l half hr.			
PE1BT	Basic Time		NA	\$42.92/\$25.56	
PE1OT	Overtime		NA	\$54.51/\$32.44	
PE1PT	Premium Time		NA	\$66.10/\$39.32	
	Cable December			Nata 0	
	Cable Records <sup>1</sup>			Note 2	
				Initial/subsequent	
PE1CR	Cable Records	Per request	NA	\$1707/\$1165	
PE1CD	VG/DS0 Cable	Per cable record	NA	\$923.08/\$923.08	
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.02/\$18.02	
PE1C1	DS1	Per T1TIE	NA	\$8.43/\$8.43	
PE1C3	DS3	Per T3TIE	NA	\$29.51/\$29.51	
PE1CB	Fiber Cable	Per cable record	NA	\$278.82/\$278.82	

## EXHIBIT A: BELLSOUTH/CLEC-1 RATES – NORTH CAROLINA PHYSICAL COLLOCATION (continued)

### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges..

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – SOUTH CAROLINA PHYSICAL COLLOCATION

Rates marked with an asterisk (\*) are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring	Non-Recurring		
			Rate (RC)	Rate (NRC)		
PE1BA Application Fee		Per request	NA	\$3768.00		
PE1CA	Subsequent Application Fee	Per request	NA	\$3,141.00		
_	γ,	1.22		Minimum		
	Space Preparation Fees (Note 4)					
PE1SJ	Firm Order Processing*			\$1,204.00		
PE1SK	Central Office Modifications*	Per sq. ft.	\$2.75	Ψ1,204.00		
PE1SL	Common Systems	Per sq. ft.	\$3.24			
1 L 10L	Modifications – Cageless*	1 01 34.11.	ψ0.24			
PE1SM	Common Systems	Per cage	\$110.17			
I L IOW	Modifications – Caged*	1 ci cage	φ110.17			
	Space Enclosure (100 sq. ft.					
PE1BW	minimum) Welded Wire-mesh	Dor first 100 sq. ft	¢240.40	NIA		
PE16W PE1C	Welded Wire-mesh	Per first 100 sq. ft.	\$219.19 \$21.50	NA		
W	Weided Wife-Hesti	Per add'l 50 sq. ft.	φ21.50	NA		
PE1PJ	Floor Space	Per sq. ft.	\$3.95	NA		
PE1BD	Cable Installation	Per cable	NA	\$1,621.00		
PE1PM	Cable Support Structure	Per entrance	\$21.33	NA		
	Cable Cappert Ctractare	cable	Ψ21.00			
	Power -48V DC Power*	Doromo	¢0.40	NA		
PE1FB	120V AC Power single phase*	Per amp Per breaker amp	\$9.19 \$5.67	INA		
PE1FD	240V AC Power single phase*	Per breaker amp	\$3.07 \$11.36	_		
PE1FE	120V AC Power three phase*	Per breaker amp	\$17.03	_		
PE1FG	277 AC Power three phase*	Per breaker amp	\$39.33	- -		
PE1PL	-48V DC Power (Note 3)	Per amp	\$7.09	NA		
	Cross Connects	Per cross connect		First/Add'l		
PE1P2	2-wire	i di cioss cominect	\$.034	\$33.75/\$31.86		
PE1P4	4-wire		\$.068	\$33.71/\$31.75		
PE1P1	DS-1		\$1.12	\$53.71/\$31.75 \$53.05/\$39.96		
PE1P3	DS-3		\$1.12 \$14.21	\$53.03/\$39.90		
PE1F3	2-fiber		\$2.82	\$52.11/\$38.69		
PE1F4	4-fiber		\$5.01	\$64.69/\$51.26		

	SOUTH CAROLINA (continued)							
USOC Rate Element Description		Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)				
PE1AX	Society Access System	Dorpromises	\$74.12					
PEIAA	Security Access System Security System*	Per premises	<b>⊅/4.1∠</b>					
PE1A1	New Access Card Activation*	Per card	\$.060	\$55.70				
PE1AA	Administrative change, existing	Per card	ψ.000	\$15.62				
. = ., .	card*	l or cara		ψ·0.02				
PE1AR	Replace lost or stolen card	Per card		\$45.66				
PE1AK	Initial Kev	Per kev	NA	\$26.25				
PE1AL	Replace lost or stolen key	Per key	NA	\$26.25				
DEAOD	0 4 11 1 111 5 4			<b>#0.455.00</b>				
PE1SR	Space Availability Report*	Per premises		\$2,155.00				
		requested						
	POT Bay Arrangements	Per cross-connect						
	Prior to 6/1/99	1 CI CIOSS COMICCI						
PE1PE	2-Wire Cross-Connect		\$0.1091	NA				
PE1PF	4-Wire Cross-Connect		\$0.2181	NA				
PE1PG	DS1 Cross-Connect		\$0.9004	NA				
PE1PH	DS3 Cross-Connect		\$5.64	NA				
PE1B2	2 Fiber Cross-Connect		\$37.36	NA				
PE1B4	4 Fiber Cross-Connect		\$50.38	NA				
	0 : 5 :	D 1 161 /A 1 111						
	Security Escort	Per half hr./Add'l half hr.						
PE1BT	Basic Time	Hall III.	NA	\$33.92/\$21.50				
PE1OT	Overtime		NA NA	\$44.19/\$27.77				
PE1PT	Premium Time		NA	\$54.45/\$34.04				
				40 11 10, 40 110 1				
	Cable Records <sup>1</sup>			Note 2				
				Initial/subsequent				
PE1CR	Cable Records	Per request	NA	\$1712/\$1168				
PE1CD	VG/DS0 Cable	Per cable record	NA	\$925.57/\$925.57				
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.06/\$18.06				
PE1C1	DS1	Per T1TIE	NA	\$8.45/\$8.45				
PE1C3	DS3	Per T3TIE	NA	\$29.59/\$29.59				
PE1CB	Fiber Cable	Per cable record	NA	\$279.57/\$279.57				

## EXHIBIT A: BELLSOUTH/CLEC-1 RATES – SOUTH CAROLINA PHYSICAL COLLOCATION (continued)

#### Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.

Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.

Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

# EXHIBIT A: BELLSOUTH/CLEC-1 RATES – TENNESSEE PHYSICAL COLLOCATION

\* Rates are interim and are subject to true-up.

USOC	Rate Element Description	Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)
PE1BA	Application Fee	Per request	NA	\$3,767.00
PE1CA	Subsequent Application Fee	Per request	NA	\$3,140.00 Minimum
PE1SJ PE1SK PE1SL PE1SM	Space Preparation Fees (Note 4) Firm Order Processing* Central Office Modifications* Common Systems Modifications – Cageless* Common Systems Modifications – Caged*	Per sq. ft. Per sq. ft. Per cage	\$2.74 \$2.95 \$100.14	\$1,204.00
PE1BW PE1C W	Space Enclosure (100 sq. ft. minimum) Welded Wire-mesh Welded Wire-mesh	Per first 100 sq. ft. Per add'l 50 sq. ft.	\$218.53 \$21.44	NA NA
PE1PJ	Floor Space	Per sq. ft.	\$6.75	NA
PE1BD	Cable Installation	Per cable	NA	\$1,757.00
PE1PM	Cable Support Structure	Per entrance cable	\$19.80	NA
PE1FB PE1FD PE1FE PE1FG	Power -48V DC Power* 120V AC Power single phase* 240V AC Power single phase* 120V AC Power three phase* 277 AC Power three phase*	Per amp Per breaker amp Per breaker amp Per breaker amp Per breaker amp	\$8.87 \$5.60 \$11.22 \$16.82 \$38.84	NA - - - -
PE1PL	-48V DC Power (Note 3)	Per amp	\$5.00	NA
PE1P2 PE1P4 PE1P1 PE1P3 PE1F2 PE1F4	Cross Connects 2-wire 4-wire DS-1 DS-3 2-fiber 4-fiber	Per cross connect	\$0.033 \$0.066 \$1.51 \$19.26 \$3.82 \$6.79	First/Add'l \$33.82/\$31.92 \$33.94/\$31.95 \$53.27/\$40.16 \$52.37/\$38.89 \$52.37/\$38.89 \$65.03/\$51.55

	TENNESSEE (continued)							
USOC Rate Element Description		Unit	Recurring Rate (RC)	Non-Recurring Rate (NRC)				
55.437			<b>^</b>					
PE1AX	Security Access System	Per premises	\$55.99					
DEAAA	Security System	Danasad	<b>ተ 050</b>	<b>ФГГ 07</b>				
PE1A1	New Access Card Activation	Per card	\$.059	\$55.67				
PE1AA	Administrative change, existing card	Per card		\$15.61				
PE1AR	Replace lost or stolen card	Per card		\$45.64				
PE1AK	Initial Kev	Per kev	NA	\$26.24				
PE1AL	Replace lost or stolen key	Per key	NA	\$26.24				
PE1SR	Space Availability Report*	Per premises		\$2,154.00				
		requested						
	POT Bay Arrangements Prior to 6/1/99	Per cross-connect						
PE1PE	2-Wire Cross-Connect		\$0.40	NA				
PE1PF	4-Wire Cross-Connect		\$1.20	NA				
PE1PG	DS1 Cross-Connect		\$1.20	NA				
PE1PH	DS3 Cross-Connect		\$8.00	NA				
PE1B2	2 Fiber Cross-Connect		\$38.79	NA				
PE1B4	4 Fiber Cross-Connect		\$52.31	NA				
	Security Escort	Per half hr./Add'l half hr.						
PE1BT	Basic Time		NA	\$33.91/\$21.49				
PE1OT	Overtime		NA	\$44.17/\$27.76				
PE1PT	Premium Time		NA	\$54.42/\$34.02				
	Cable Records <sup>1</sup>			Note 2				
				Initial/subsequent				
PE1CR	Cable Records	Per request	NA	\$1711/\$1168				
PE1CD	VG/DS0 Cable	Per cable record	NA	\$925.06/\$925.06				
PE1CO	VG/DS0 Cable	Per each 100 pair	NA	\$18.05/\$18.05				
PE1C1	DS1	Per T1TIE	NA	\$8.45/\$8.45				
PE1C3	DS3	Per T3TIE	NA	\$29.57/\$29.57				
PE1CB	Fiber Cable	Per cable record	NA	\$279.42/\$279.42				

## Note(s):

N/A refers to rate elements which do not have a negotiated rate.

Note1: Cable records charges apply for work required to build cable records in company systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.

- Note 2: The initial charge applies when the cables are first installed and inventoried. The subsequent charge applies when additional cables are installed and inventoried at the same location.
- Note 3: These Power rates will only apply for existing collocation arrangements provisioned prior to the execution of this agreement and Augments that make use of existing power.
- Note 4: Recurring charges for Space Preparation will not apply to existing collocation arrangements for which NewSouth paid non-recurring Space Preparation charges.

## **EXHIBIT B**

Page 1 of 4

## ENVIRONMENTAL AND SAFETY PRINCIPLES

The following principles provide basic guidance on environmental and safety issues when applying for and establishing Physical Collocation arrangements.

#### 1. GENERAL PRINCIPLES

- Compliance with Applicable Law. BellSouth and NewSouth agree to comply with applicable federal, state, and local environmental and safety laws and regulations including U.S. Environmental Protection Agency (USEPA) regulations issued under the Clean Air Act (CAA), Clean Water Act (CWA), Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), Superfund Amendments and Reauthorization Act (SARA), the Toxic Substances Control Act (TSCA), and OSHA regulations issued under the Occupational Safety and Health Act of 1970, as amended and NFPA and National Electrical Codes (NEC) and the NESC ("Applicable Laws"). Each Party shall notify the other if compliance inspections are conducted by regulatory agencies and/or citations are issued that relate to any aspect of this Attachment.
- Notice. BellSouth and NewSouth shall provide notice to the other, including Material Safety Data Sheets (MSDSs), of known and recognized physical hazards or Hazardous Chemicals existing on site or brought on site. Each Party is required to provide specific notice for known potential Imminent Danger conditions. NewSouth should contact 1-800-743-6737 for BellSouth MSDS sheets.
- 1.3 Practices/Procedures. BellSouth may make available additional environmental control procedures for NewSouth to follow when working at a BellSouth Premises (See Section 2, below). These practices/procedures will represent the regular work practices required to be followed by the employees and contractors of BellSouth for environmental protection. NewSouth will require its contractors, agents and others accessing the BellSouth Premises to comply with these practices. Section 2 lists the Environmental categories where BST practices should be followed by CLEC when operating in the BellSouth Premises.
- 1.4 <u>Environmental and Safety Inspections</u>. BellSouth reserves the right to inspect the NewSouth space with proper notification. BellSouth reserves the right to stop any NewSouth work operation that imposes Imminent Danger to the environment, employees or other persons in the area or Facility.
- 1.5 <u>Hazardous Materials Brought On Site</u>. Any hazardous materials brought into, used, stored or abandoned at the BellSouth Premises by NewSouth are owned by NewSouth. NewSouth will indemnify BellSouth for claims, lawsuits or damages to

persons or property caused by these materials. Without prior written BellSouth approval, no substantial new safety or environmental hazards can be created by NewSouth or different hazardous materials used by NewSouth at BellSouth Facility. NewSouth must demonstrate adequate emergency response capabilities for its materials used or remaining at the BellSouth Facility.

- 1.6 <u>Spills and Releases</u>. When contamination is discovered at a BellSouth Premises, the Party discovering the condition must notify BellSouth. All Spills or Releases of regulated materials will immediately be reported by NewSouth to BellSouth.
- 1.7 Coordinated Environmental Plans and Permits. BellSouth and NewSouth will coordinate plans, permits or information required to be submitted to government agencies, such as emergency response plans, spill prevention control and countermeasures (SPCC) plans and community reporting. If fees are associated with filing, BellSouth and NewSouth will develop a cost sharing procedure. If BellSouth's permit or EPA identification number must be used, NewSouth must comply with all of BellSouth's permit conditions and environmental processes, including environmental "best management practices (BMP)" (see Section 2, below) and/or selection of BST disposition vendors and disposal sites.
- Environmental and Safety Indemnification. BellSouth and NewSouth shall indemnify, defend and hold harmless the other Party from and against any claims (including, without limitation, third-party claims for personal injury or death or real or personal property damage), judgments, damages, (including direct and indirect damages, and punitive damages), penalties, fines, forfeitures, costs, liabilities, interest and losses arising in connection with the violation or alleged violation of any Applicable Law or contractual obligation or the presence or alleged presence of contamination arising out of the acts or omissions of the indemnifying Party, its agents, contractors, or employees concerning its operations at the Facility.

#### 2. CATEGORIES FOR CONSIDERATION OF ENVIRONMENTAL ISSUES

When performing functions that fall under the following Environmental categories on BellSouth's Premises, NewSouth agrees to comply with the applicable sections of the current issue of BellSouth's Environmental and Safety Methods and Procedures (M&Ps), incorporated herein by this reference. NewSouth further agrees to cooperate with BellSouth to ensure that NewSouth's employees, agents, and/or subcontractors are knowledgeable of and satisfy those provisions of BellSouth's Environmental M&Ps which apply to the specific Environmental function being performed by NewSouth, its employees, agents and/or subcontractors.

The most current version of reference documentation must be requested from BellSouth.

ENVIRONMENTAL CATEGORIES	ENVIRONMENTAL ISSUES	ADDRESSED BY THE FOLLOWING DOCUMENTATION
Disposal of hazardous material or other regulated material (e.g., batteries, fluorescent	Compliance with all applicable local, state, & federal laws and regulations	<ul><li>Std T&amp;C 450</li><li>Fact Sheet Series 17000</li></ul>
tubes, solvents & cleaning materials)	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)
Emergency response	Hazmat/waste release/spill firesafety emergency	<ul> <li>Fact Sheet Series 1700</li> <li>Building Emergency         Operations Plan (EOP)         (specific to and located on Premises)     </li> </ul>
Contract labor/outsourcing for services with environmental implications to be performed	Compliance with all applicable local, state, & federal laws and regulations	• Std T&C 450
on BellSouth Premises (e.g., disposition of hazardous material/waste; maintenance of storage tanks)	Performance of services in accordance with BST's environmental M&Ps	<ul> <li>Std T&amp;C 450-B</li> <li>(Contact E/S for copy of appropriate E/S M&amp;Ps.)</li> </ul>
	Insurance	• Std T&C 660
Transportation of hazardous material	Compliance with all applicable local, state, & federal laws and regulations	<ul><li>Std T&amp;C 450</li><li>Fact Sheet Series 17000</li></ul>
	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	Approved Environmental     Vendor List (Contact E/S     Management)
Maintenance/operations work which may produce a waste	Compliance with all application local, state, & federal laws and regulations	• Std T&C 450
Other maintenance work	Protection of BST employees and equipment	<ul> <li>29CFR 1910.147 (OSHA Standard)</li> <li>29CFR 1910 Subpart O (OSHA Standard)</li> </ul>
Janitorial services	All waste removal and disposal	P&SM Manager -

	must conform to all applicable federal, state and local regulations	Procurement
	All Hazardous Material and Waste	• Fact Sheet Series 17000
	Asbestos notification and protection of employees and equipment	<ul> <li>GU-BTEN-001BT, Chapter 3</li> <li>BSP 010-170-001BS (Hazcom)</li> </ul>
Manhole cleaning	Compliance with all applicable local, state, & federal laws and regulations	<ul> <li>Std T&amp;C 450</li> <li>Fact Sheet 14050</li> <li>BSP 620-145-011PR         Issue A, August 1996 </li> </ul>
	Pollution liability insurance	• Std T&C 660-3
	EVET approval of contractor	Approved Environmental     Vendor List (Contact E/S     Management)
Removing or disturbing building materials that may contain asbestos	Asbestos work practices	• GU-BTEN-001BT, Chapter 3

#### 3. **DEFINITIONS**

<u>Generator</u>. Under RCRA, the person whose act produces a Hazardous Waste, as defined in 40 CFR 261, or whose act first causes a Hazardous Waste to become subject to regulation. The Generator is legally responsible for the proper management and disposal of Hazardous Wastes in accordance with regulations.

<u>Hazardous Chemical</u>. As defined in the U.S. Occupational Safety and Health (OSHA) hazard communication standard (29 CFR 1910.1200), any chemical which is a health hazard or physical hazard.

Hazardous Waste. As defined in section 1004 of RCRA.

<u>Imminent Danger</u>. Any conditions or practices at a facility which are such that a danger exists which could reasonably be expected to cause immediate death or serious harm to people or immediate significant damage to the environment or natural resources.

Spill or Release. As defined in Section 101 of CERCLA.

#### 4. ACRONYMS

E/S – Environmental/Safety

**EVET** - Environmental Vendor Evaluation Team

 $\underline{DEC/LDEC} \text{ - Department Environmental Coordinator/Local Department Environmental Coordinator}$ 

<u>GU-BTEN-001BT</u> - BellSouth Environmental Methods and Procedures

NESC - National Electrical Safety Codes

<u>P&SM</u> - Property & Services Management

Std. T&C - Standard Terms & Conditions

## **Attachment 5**

**Access to Numbers and Number Portability** 

Version1Q00: 3/6/00

## TABLE OF CONTENTS

1.	Non-Discriminatory Access To Telephone Numbers	<u>3</u> 3
2.	Number Portability Permanent Solution	<u>3</u> 3
3.	Service Provider Number Portability	<u>4</u> 4
4.	SPNP Implementation	<u>4</u> 4
5.	Transition To Permanent Number Portability	<u>7</u> 7
6.	True-Up	_ <u>7</u> 7
	ites	_

#### ACCESS TO NUMBERS AND NUMBER PORTABILITY

## 1. Non-Discriminatory Access to Telephone Numbers

- 1.1 During the term of this Agreement, NewSouth shall contact the North American Numbering Plan Administrator, Neustar, for the assignment of numbering resources. In order to be assigned a Central Office Code, NewSouth will be required to complete the Central Office Code (NXX) Assignment Request and Confirmation Form (Code Request Form) in accordance with Industry Numbering Committee's Central Office Code (NXX) Assignment Guidelines (INC 95-0407-008).
- 1.2 For the purposes of the resale of BellSouth's telecommunications services by NewSouth, BellSouth will provide NewSouth with on line access to telephone numbers for reservation on a first come first served basis. Such reservations of telephone numbers, on a pre-ordering basis shall be for a period of forty-five (45) days. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth may request that NewSouth cancel its reservations of numbers. NewSouth shall comply with such request.
- 1.3. Further, upon NewSouth request and for the purposes of the resale of BellSouth's telecommunications services by NewSouth, BellSouth will reserve up to 100 telephone numbers per Common Language Location Identifier Code (CLLIC), for NewSouth's sole use. Such telephone number reservations shall be transmitted to NewSouth via electronic file transfer. Such reservations shall be valid for forty-five (45) days from the reservation date. NewSouth acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth shall use its best efforts to reserve for a forty-five (45) day period a sufficient quantity for NewSouth's reasonable need in that particular CLLIC.

#### 2. Number Portability Permanent Solution

- 2.1 The FCC, the Commissions, and industry forums have developed and BellSouth is implementing a permanent approach to providing service provider number portability. Both Parties will implement a permanent approach as developed and approved by the Commission, the FCC and industry forums. Consistent with the requirements to move to Permanent Number Portability (PNP) as set forth in Section 5 of this Attachment, Interim Service Provider Number Portability (SPNP) may be available only until such permanent solution is implemented in an end office.
- 2.2 <u>End User Line Charge</u>. Recovery of charges associated with implementing PNP through a monthly charge assessed to end users has been authorized by the FCC. This

Version1Q00: 3/6/00

end user line charge will be as filed in FCC No. 1 and will be billed to NewSouth where NewSouth is a subscriber to local switching or where NewSouth is a reseller of BellSouth telecommunications services. This charge will not be discounted.

## 3. Service Provider Number Portability

- 3.1 <u>Definition</u>. Until the industry-wide permanent solution is implemented in an end office, BellSouth shall provide Service Provider Number Portability ("SPNP"). SPNP is an interim service arrangement whereby an end user who switches subscription of his local exchange service from BellSouth to a CLEC, or vice versa, is permitted to retain the use of his existing assigned telephone number, provided that the end user remains at the same location for his local exchange service or changes locations and service providers but stays within the same serving wire center of his existing number.
- Methods of Providing Number Portability. SPNP is available through either remote call forwarding or direct inward dialing trunks, at the election of NewSouth. Remote call forwarding (SPNP-RCF) is an existing switch-based BellSouth service that redirects calls within the telephone network. Direct inward dialing trunks (SPNP-DID) allow calls to be routed over a dedicated facility to the NewSouth switch that serves the subscriber.
- 3.3 <u>Signaling Requirements</u>. SS7 Signaling is required for the provision of SPNP services. SPNP-DID is available from BellSouth on a per DS0, DS1, or DS3 basis. Where SPNP-DID is technically feasible and is provided on a DS1 or a DS3 basis, the applicable channelization rates are those specified in Section E6 in BellSouth's Intrastate Access Tariffs, incorporated herein by this reference. SPNP is available only for basic local exchange service.

#### 3.4 Rates

Rates for SPNP are set out in Exhibit A to this Attachment. If no rate is identified in the Attachment, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

### 4. SPNP Implementation

4.1 SPNP is available only where a CLEC or BellSouth is currently providing, or will begin providing concurrent with provision of SPNP, basic local exchange service to the affected end user. SPNP for a particular telephone number is available only from the central office originally providing local exchange service to the end user. SPNP for a particular assigned telephone number will be disconnected when any end user, Commission, BellSouth, or CLEC initiated activity (*e.g.*, a change in exchange boundaries) would normally result in a telephone number change had the end user retained his initial local exchange service.

Version1Q00: 3/6/00

- 4.2 SPNP-RCF, as contemplated by this Agreement, is a telecommunications service whereby a call dialed to an SPNP-RCF equipped telephone number is automatically forwarded to an assigned seven- or ten- digit telephone number within the local calling area as defined in BellSouth's General Subscriber Services Tariff. The forwarded-to number shall be specified by the CLEC or BellSouth, as appropriate. The forwarding Party will provide identification of the originating telephone number, via SS7 signaling, to the receiving Party. Identification of the originating telephone number to the SPNP-RCF end user cannot be guaranteed, however. SPNP-RCF provides a single call path for the forwarding of no more than one simultaneous call to the receiving Party's specified forwarded-to number.
- 4.3 SPNP-DID service, as contemplated by this Agreement, provides trunk side access to end office switches for direct inward dialing to the other Party's premises equipment from the telecommunications network to lines associated with the other Party's switching equipment and must be provided on all trunks in a group arranged for inward service. A SPNP-DID trunk termination charge, provided with SS7 Signaling only, applies for each trunk voice grade equivalent. In addition, direct facilities are required from the end office where a ported number resides to the end office serving the ported end user customer. The rates for a switched local channel and switched dedicated transport apply as set forth in Exhibit A of this Attachment. Transport mileage will be calculated as the airline distance between the end office where the number is ported and the Point of Interface ("POI") using the V&H coordinate method. SPNP-DID must be established with a minimum configuration of two channels and one unassigned telephone number per switch, per arrangement for control purposes. Transport facilities arranged for SPNP-DID may not be mixed with any other type of trunk group, with no outgoing calls placed over said facilities. SPNP-DID will be provided only where such facilities are available and where the switching equipment of the ordering Party is properly equipped. Where SPNP-DID service is required from more than one wire center or from separate trunk groups within the same wire center, such service provided from each wire center or each trunk group within the same wire center shall be considered a separate service. Only customer-dialed sent-paid calls will be completed to the first number of a SPNP-DID number group; however, there are no restrictions on calls completed to other numbers of a SPNP-DID number group. Sent-paid calls refer to those calls placed by an end user who physically deposits currency in a public telephone. Interface group arrangements provided for terminating the switched transport at the Party's terminal location are as set forth in of BellSouth's Intrastate Access Services Tariff, § E6.1.3.A as amended from time to time.
- 4.3.1 SPNP-DID Service requires ordering consecutive telephone numbers in blocks of twenty. To order non-consecutive telephone numbers or telephone numbers in less than blocks of twenty, the NBR process must be used. SS7 Signaling is required for the provision of either of these services.
- 4.4 The calling Party shall be responsible for payment of the applicable charges for sent-paid calls to the SPNP number. For collect, third-party, or other operator-assisted

Version1Q00: 3/6/00

non-sent paid calls to the ported telephone number, BellSouth or the CLEC shall be responsible for the payment of charges under the same terms and conditions for which the end user would have been liable for those charges. Either Party may request that the other block collect and third party non-sent paid calls to the SPNP-assigned telephone number. If a Party does not request blocking, the other Party will provide itemized local usage detail for the billing of non-sent paid calls on the monthly bill of usage charges provided at the individual end user account level. The detail will include itemization of all billable usage. Each Party shall have the option of receiving this usage data on a daily basis via a data file transfer arrangement. This arrangement will utilize the existing industry uniform standard, known as EMI standards, for exchange of billing data. Files of usage data will be created daily for the optional service. Usage originated and recorded in the sending BellSouth RAO will be provided in unrated or rated format, depending on processing system. CLEC usage originated elsewhere and delivered via CMDS to the sending BellSouth RAO shall be provided in rated format.

- 4.5 Each Party shall be responsible for obtaining authorization from the end user for the handling of the disconnection of the end user's service, the provision of new local service and the provision of SPNP services. Each Party shall be responsible for coordinating the provision of service with the other to assure that its switch is capable of accepting SPNP ported traffic. Each Party shall be responsible for providing equipment and facilities that are compatible with the other's service parameters, interfaces, equipment and facilities and shall be required to provide sufficient terminating facilities and services at the terminating end of an SPNP call to adequately handle all traffic to that location and shall be solely responsible to ensure that its facilities, equipment and services do not interfere with or impair any facility, equipment, or service of the other Party or any of its end users. In the event that either Party determines in its reasonable judgment that the other Party will likely impair or is impairing, or interfering with any equipment, facility or service or any of its end users, that Party may either refuse to provide SPNP service or may terminate SPNP service to the other Party after providing appropriate notice.
- Each Party shall be responsible for providing an appropriate intercept announcement service for any telephone numbers subscribed to SPNP services for which it is not presently providing local exchange service or terminating to an end user. Where either Party chooses to disconnect or terminate any SPNP service, that Party shall be responsible for designating the preferred standard type of announcement to be provided.
- 4.7 Each Party shall be the other Party's single point of contact for all repair calls on behalf of each Party's end user. Each Party reserves the right to contact the other Party's customers if deemed necessary for maintenance purposes.
- 4.8 Neither Party shall be responsible for adverse effects on any service, facility or equipment from the use of SPNP services. End-to-end transmission characteristics may vary depending on the distance and routing necessary to complete calls over

Version1Q00: 3/6/00

SPNP facilities and the fact that another carrier is involved in the provisioning of service. Therefore, end-to-end transmission characteristics cannot be specified by either Party for such calls. Neither Party shall be responsible to the other if any necessary change in protection criteria or in any of the facilities, operation, or procedures of either renders any facilities provided by the other Party obsolete or renders necessary modification of the other Party's equipment.

4.9 For terminating IXC traffic ported to either Party which requires use of either Party's tandem switching, the tandem provider will bill the IXC tandem switching, the interconnection charge, and a portion of the transport, and the other Party will bill the IXC local switching, the carrier common line and a portion of the transport. If the tandem provider is unable to provide the necessary access records to permit the other Party to bill the IXC directly for terminating access to ported numbers, then the tandem provider will bill the IXC full terminating switched access charges at the tandem provider's rate and will compensate the other Party at the tandem Party's tariff rates via a process to be mutually agreed to by the Parties to estimate the amount of ported switched access revenues due the other Party. If an intraLATA toll call is delivered, the delivering Party will pay terminating access rates to the other Party. This subsection does not apply in cases where SPNP-DID is utilized for number portability.

## 5. Transition to Permanent Number Portability

- Once a PNP is implemented in an end office both Parties must withdraw their SPNP offerings. The transition from existing SPNP arrangements to PNP shall occur within one hundred twenty (120) days from the date PNP is implemented in the end office. Neither Party shall charge the other Party for conversion from SPNP to PNP. The Parties shall comply with any SPNP/PNP transition processes established by the FCC and State commissions and appropriate industry number portability work groups.
- Notwithstanding the foregoing, the Parties acknowledge that the FCC has determined once LNP has been deployed pursuant to the FCC's orders, rules and regulations, that all local exchange carriers (LECs) have the duty to provide LNP. Therefore, either Party, at any time, may seek appropriate legal or regulatory relief concerning the transition from INP to LNP or other related issues.

## 6. True-up

This section applies only to Tennessee and other rates that are interim or expressly subject to true-up under this attachment.

6.1 The interim prices for Network Elements and Other Services and Local Interconnection shall be subject to true-up according to the following procedures:

The interim prices shall be trued-up, either up or down, based on final prices determined either by further agreement between the Parties, or by a final order

Version1Q00: 3/6/00

(including any appeals) of the Commission which final order meets the criteria of (3) below. The Parties shall implement the true-up by comparing the actual volumes and demand for each item, together with interim prices for each item, with the final prices determined for each item. Each Party shall keep its own records upon which the true-up can be based, and any final payment from one Party to the other shall be in an amount agreed upon by the Parties based on such records. In the event of any disagreement as between the records or the Parties regarding the amount of such true-up, the Parties agree that the body having jurisdiction over the matter shall be called upon to resolve such differences, or the Parties may mutually agree to submit the matter to the Dispute Resolution process in accordance with the provisions in the General Terms and Conditions and Attachment 1 of this Agreement.

- The Parties may continue to negotiate toward final prices, but in the event that no such agreement is reached within nine (9) months, either Party may petition the Commission to resolve such disputes and to determine final prices for each item. Alternatively, upon mutual agreement, the Parties may submit the matter to the Dispute Resolution Process set forth in the General Terms and Conditions and Attachment 1 of the Agreement incorporated herein by reference, so long as they file the resulting Agreement with the Commission as a "negotiated Agreement" under Section 252(e) of the Act.
- A final order of this Commission that forms the basis of a true-up shall be the final order as to prices based on appropriate cost studies, or potentially may be a final order in any other Commission proceeding which meets the following criteria:
  - (a) BellSouth and CLEC is entitled to be a full Party to the proceeding;
  - (b) It shall apply the provisions of the federal Telecommunications Act of 1996, including but not limited to Section 252(d)(1) (which contains pricing standards) and all then-effective implementing rules and regulations; and,
  - (c) It shall include as an issue the geographic deaveraging of network element prices, which deaveraged prices, if any are required by said final order, shall form the basis of any true-up.

#### BELLSOUTH/NEWSOUTH RATES SERVICE PROVIDER NUMBER PORTABILITY

		RATES BY STATE								
DESCRIPTION	usoc	AL	FL	GA	кү	LA	MS	NC	sc	TN
INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF (1) (2)								1.0		
RCF, per number ported (Business Line), 10 paths	TNPBL	NA	NA	NA	NA	NA	NA	\$2.25	NA	NA
RCF, per number ported (Residence Line), 6 paths	TNPRL	NA	NA	NA	NA	NA	NA	\$1.15	NA	NA
RCF, per number ported (Business Line)	TNPBL	\$2.13	NA	\$2.03	NA	\$2.29	\$2.34	\$1.66	\$2.17	\$1.50
NRC - Electronic	TNPBL	\$0.65	NA	\$0.51	NA	\$0.49	\$0.6441	\$0.71	\$0.7046	NA
NRC - Disconnect Charge	TNPBL	\$0.07	NA	NA	NA	\$0.05	\$0.0644	\$0.50	NA	NA
RCF, per number ported (Residence Line)	TNPRL	\$2.13	NA	\$2.03	NA	\$2.29	\$2.34	\$1.66	\$2.17	\$1.25
NRC	TNPRL	\$0.65	NA	\$0.51	NA	\$0.49	\$0.6441	\$0.71	\$0.7046	NA
NRC - Disconnect Charge	TNPRL	\$0.07	NA	NA	NA	\$0.05	\$0.0644	\$0.50	NA	NA
RCF, add'l capacity for simultaneous call forwarding, per additional path	N/A	\$0.32	NA	\$0.2836	NA	\$0.38	\$0.3838	\$0.32	\$0.3854	\$0.50
, , ,	(++) Bus = TNPBD	-								
RCF, per service order, per location	Res = TNPRD									
NRC - 1st	TNP++	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$2.73	\$1.37	\$25.00
NRC - Add'l	TNP++	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$2.73	\$1.37	\$25.00
NRC - Disconnect - 1st	TNP++	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Disconnect - Add'l	TNP++	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	NA	NA	\$18.14	\$25.52	\$45.80	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$27.37	NA	NA	NA	\$18.14	\$25.52	\$45.80	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	\$44.70	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	\$44.70	NA
INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID										
DID per number ported, Residence - NRC	TNPDR	\$1.18	NA	\$0.93	NA	\$0.89	\$1.17	\$2.25	\$2.25	NA
DID per number ported, Residence - NRC - Disconnect	TNPDR	\$1.18	NA	NA	NA	\$0.90	\$1.17	NA	NA	NA
DID per number ported, Business - NRC	TNPDB	\$1.18	NA	\$0.93	NA	\$0.89	\$1.17	\$2.25	\$2.25	NA
DID per number ported, Business - NRC - Disconnect	TNPDB	\$1.18	NA	NA	NA	\$0.90	\$1.17	NA	NA	NA
DID per service order, per location										
NRC - 1st	TNPRD	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$2.73	\$1.37	NA
NRC - Add'l	TNPRD	\$1.44	NA	\$2.10	NA	\$2.02	\$2.84	\$2.73	\$1.37	NA
NRC - Disconnect - 1st	TNPRD	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	\$44.70	NA
NRC - Disconnect - Add'I	TNPRD	\$1.44	NA	NA	NA	\$2.01	\$2.84	NA	\$44.70	NA
NRC - Incremental Charge - Manual Service Order - 1st	SOMAN	\$27.37	NA	\$18.94	NA	\$18.14	\$25.52	\$45.80	NA	NA
NRC - Incremental Charge - Manual Service Order - Add'l	SOMAN	\$27.37	NA	NA	NA	\$18.14	\$25.52	\$45.80	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - 1st	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
NRC - Incremental Charge - Manual Service Order - Disconnect - Add'l	SOMAN	\$17.77	NA	NA	NA	\$11.41	\$16.06	NA	NA	NA
DID, per trunk termination, Initial	TNPT2	\$11.84	NA	\$10.73	NA	\$12.46	\$13.78	\$11.43	\$13.16	NA
DID, per trunk termination, Initial - NRC	TNPT2	\$173.73	NA	\$135.47	NA	\$129.69	\$171.68	\$217.88	\$218.03	NA
DID, per trunk termination, Initial - Disconnect	TNPT2	\$50.43	NA	NA	NA	\$37.85	\$49.86	NA	NA	NA
DID, per trunk termination, Subsequent	TNPT2	\$11.84	NA	\$10.73	NA	\$12.46	\$13.78	\$11.43	\$13.16	NA
DID, per trunk termination, Subsequent - NRC	TNPT2	\$51.35	NA	\$39.53	NA	\$37.85	\$50.69	\$73.56	\$73.63	NA
DID, per trunk termination, Subsequent - Disconnect	TNPT2	\$25.00	NA	NA	NA	\$18.75	\$24.71	NA	NA	NA

#### NOTES:

If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

- 1 Until the FCC issues its order implementing a cost recovery mechanism for permanent number portability, the Company will track its costs of providing interim SPNP with sufficient detail to verify the costs. This will facilitate the Florida PSCs consideration of the recovery of these costs in Docket 950737-TP. (FL)
- 2 BellSouth and CLEC will each bear their own costs of providing remote call forwarding as an interim number portability option. (KY)

# **Attachment 6**

**Ordering and Provisioning** 

# TABLE OF CONTENTS

1.	Quality of Ordering And Provisioning	.3
2.	Access To Operational Support Systems	.4
3.	Miscellaneous Ordering And Provisioning Guidelines	.6

#### ORDERING AND PROVISIONING

#### 1. Quality of Ordering and Provisioning

1.1 BellSouth shall provide ordering and provisioning services to NewSouth that are equal to the ordering and provisioning services BellSouth provides to itself or any other CLEC, where technically feasible. Reasonable and nondiscriminatory guidelines for ordering and provisioning are set forth in the various pre-ordering, ordering and provisioning guides, as appropriate, and as they are amended from time to time during this Agreement. The guides may be referenced at the following site:

http://www.interconnection.bellsouth.com/guides/guides\_p.html.

BellSouth shall provide advance notification of such guides via carrier notification letters posted to BellSouth's web site.

- 1.2 BellSouth shall provide all ordering and provisioning services to NewSouth during the same business hours of operation that BellSouth provisions service to its affiliates or end users. Ordering and provisioning support required by NewSouth outside of these hours will be considered outside of normal business hours and will be subject to overtime billing.
- 1.2.1 For purposes of this Agreement, BellSouth's regular working hours are defined as follows:

Monday – Friday – 8:00 a.m. – 5:00 p.m. (Excluding Holidays)
(Resale/UNE non-coordinated, coordinated orders and order coordination-time specific)

Saturday - 8:00 a.m. –5:00 p.m. (Excluding Holidays)
(Resale/UNE non-coordinated orders)

1.2.2 It is understood and agreed that BellSouth technicians involved in provisioning service to NewSouth may work shifts outside of BellSouth's regular working hours as defined in Section 1.2 above (e.g., the employee's shift ends at 7:00 p.m. during daylight savings time). To the extent that NewSouth requests that work necessarily required in the provisioning of service to be performed outside BellSouth's regular working hours and that work is performed by a BellSouth technician during his or her scheduled shift such that BellSouth does not incur any additional costs in performing the work on behalf of NewSouth, BellSouth will not assess NewSouth additional charges beyond the rates and charges specified in this Agreement.

## 2. Access to Operations Support Systems

- 2.1 BellSouth shall provide NewSouth access to operations support systems ("OSS") functions for pre-ordering, ordering and provisioning, maintenance and repair and billing. Access to OSS is available through a variety of means, including electronic interfaces. BellSouth also provides manual options. The OSS functions available to CLECs through electronic interfaces are:
- 2.2 Pre-Ordering. BellSouth provides electronic access to the following pre-ordering functions: service address validation, telephone number selection, service and feature availability, due date information, and upon Commission approval of confidentiality protections, to customer record information. Access is provided through the Local Exchange Navigation System (LENS) interface, or the Telecommunications Access Gateway (TAG) interface. Customer Record Information includes but is not limited to, customer specific information in CRIS and RSAG. In addition, NewSouth shall provide to BellSouth access to customer record information including electronic access where available. Otherwise, NewSouth shall provide paper copies of customer record information within a reasonable period of time upon request by BellSouth. NewSouth agrees not to view, copy, or otherwise obtain access to the customer record information of any customer without that customer's permission and further agrees that it will obtain access to customer record information only in strict compliance with applicable laws, rules, or regulations of the State in which the service is provided.
- 2.2.1 <u>Interfaces.</u> BellSouth shall make available the following interfaces to NewSouth for access to pre-order functions: LENS; and TAG. Each such interface shall be available on a non-discriminatory basis in connection with pre-ordering for Resale services and UNEs that are available electronically.
- 2.2.2 The Parties <u>acknowledge</u> that ordering requirements necessitate the use of current, real time pre-order information to accurately build service orders. Each pre-order interface shall be available except for downtime attributable to maintenance and upload, twenty-four (24) hours a day, seven (7) days a week.
- 2.2.3 NewSouth shall be <u>permitted</u> to reserve a number, including, without limitation, a vanity number, for up to thirty (30) days for End Users.
- 2.2.4 All CSR data exchanged must be in English text, and not only USOC or FID format, provided that such <u>information</u> is maintained in textual format by BellSouth. All other data shall be in a mutually agreed upon nomenclature.
- 2.2.5 Upon request, BellSouth shall <u>provide</u> NewSouth with pre-order information in batch transmission to the extent available or provided to any other Telecommunications Carrier on the same terms and conditions and at the same rates.
- 2.2.6 Pre-ordering functions shall be provided at parity as measured by the Performance Measurement metrics included in Attachment 9 hereto.

- 2.3 Service Ordering and Provisioning. BellSouth provides electronic options for the exchange of ordering and provisioning information. Access is provided through an Electronic Data Interchange (EDI) interface, or the TAG ordering interface for non-complex and certain complex resale requests and certain network elements. NewSouth may integrate the EDI interface with the TAG pre-ordering interface or the TAG ordering interface. BellSouth provides integrated pre-ordering, ordering and provisioning capability through the LENS interface for non-complex and certain complex resale service requests and Unbundled Network Elements.
- 2.3.1 For generation of Resale service orders, ordering flows shall be available via such electronic interfaces for each of the following ordering functions: Conversion ("as is" or "with changes"); Change (features, listings, long distance); New Connect; Disconnect; From and To (change of premises with same service).
- 2.3.2 BellSouth shall provide to NewSouth electronic and manual interfaces for transmitting orders and receiving Firm Order Confirmation ("FOC"), completion notices, Due-Date Jeopardies, Design Layout Records, and, as available, other provisioning data and information. BellSouth shall provide NewSouth with a FOC for each Resale and UNE order. The FOC includes: purchase order number, telephone number, Local Service Request number, due date, and Service Order number.
- 2.3.3 BellSouth shall provision Resale Services and UNEs as prescribed in NewSouth service order requests. Access to status on electronically-submitted Resale services and UNEs shall be provided via the electronic interfaces. Access to status on manually-submitted service order requests shall be provided on BellSouth's Internet website.
- 2.3.4 BellSouth shall provide notice of a lack of facilities availability at parity to that BellSouth provides to itself, its Affiliates, or any other Telecommunications Carrier.
- 2.4 Service Trouble Reporting and Repair. Service trouble reporting and repair allows NewSouth to report and monitor service troubles and obtain repair services. BellSouth shall offer NewSouth service trouble reporting in a non-discriminatory manner that provides NewSouth the equivalent ability to report and monitor service troubles that BellSouth provides to itself. BellSouth also provides NewSouth an estimated time to repair, an appointment time or a commitment time, as appropriate, on trouble reports. BellSouth provides several options for electronic trouble reporting. For exchange services, BellSouth offers NewSouth non-discriminatory access to the Trouble Analysis Facilitation Interface (TAFI). In addition, BellSouth offers an industry standard, machine-to-machine Electronic Communications Trouble Administration (ECTA) Gateway interface. For designed services, BellSouth provides non-discriminatory trouble reporting through the ECTA Gateway. BellSouth also offers ECTA functionality through the human-to-machine EC-CPM/TA interface. If NewSouth requests BellSouth to repair a trouble after normal working hours,

NewSouth will be billed the appropriate overtime charges associated with this request pursuant to BellSouth's tariffs, in the event BellSouth is required to pay overtime charges to the technicians repairing or isolating the service.

- 2.5 <u>Change Management</u>. BellSouth provides a collaborative process for change management of the electronic interfaces through the Electronic Interface Change Control Process ("EICCP). Reasonable and nondiscriminatory guidelines for this process are set forth in the EICCP document, and as it is amended from time to time during this agreement.
- Migration of NewSouth to New Software Releases for National Standard Machine-to-Machine Electronic Interfaces. Pursuant to the change management process, BellSouth will issue new software releases for new industry standards for its industry standard, machine-to-machine electronic interfaces. When a new release of new industry standards is implemented, BellSouth will continue to support both the new release (N) and the prior release (N-1). When BellSouth makes the next release (N+1), BellSouth will eliminate support for the (N-1) release and support the two newest releases (N and N+1). Thus, BellSouth will always support the two most current releases. BellSouth will issue documents to NewSouth with sufficient notice to allow NewSouth to make the necessary changes to their systems and operations to migrate to the newest release in a timely fashion.
- 2.7 <u>Rates.</u> To the extent approved by the Commission or otherwise agreed to by the Parties, all costs incurred by BellSouth to develop and implement operational interfaces to the OSS shall be recovered from the carriers that use the services. Charge for use of OSS shall be as set forth in the General Terms and Conditions of of this Agreement.
- 2.8 The electronic OSS Charges rather than the manual ordering charges shall apply to Local Service Request submitted by NewSouth when BellSouth's existing electronic interfaces normally utilized by NewSouth are unavailable for reasons other than scheduled maintenance or other scheduled activities for which advance notification is required and provided by BellSouth.

# 3. Miscellaneous Ordering and Provisioning Guidelines

- Pending Orders. To ensure the most efficient use of facilities and resources, orders placed in the hold or pending status by NewSouth will be held for a maximum of thirty (30) days from the date the order is placed on hold. After such time, if NewSouth wishes to reinstate an order, NewSouth may be required to submit a new service order.
- 3.2 <u>Single Point of Contact</u>. NewSouth will be the single point of contact with BellSouth for ordering activity for network elements and other services used by NewSouth to provide services to its end users, except that BellSouth may accept an order directly

from another CLEC, or BellSouth, acting with authorization of the affected end user. NewSouth and BellSouth shall each execute a blanket letter of authorization with respect to customer orders. The Parties shall each be entitled to adopt their own internal processes for verification of customer authorization for orders, provided, however, that such processes shall comply with applicable state and federal law including, until superseded, the FCC guidelines and orders applicable to Presubscribed Interexchange Carrier (PIC) changes including Un-PIC. Pursuant to such an order, BellSouth may disconnect any network element associated with the service to be disconnected and being used by NewSouth to provide service to that end user and reuse such network elements or facilities to enable such other LEC to provide service to the end user. BellSouth will notify NewSouth that such an order has been processed, but will not be required to notify NewSouth in advance of such processing.

- 3.3 <u>Use of Facilities.</u> When a customer of NewSouth elects to discontinue service and transfer service to another local exchange carrier, including BellSouth, BellSouth shall have the right to reuse the facilities provided to NewSouth by BellSouth for retail or resale service, loop and/or port for that customer. In addition, BellSouth may disconnect and reuse facilities (i) where BellSouth is providing switching; (ii) when the facility is in a denied state (i.e., service is no longer being provided over the facility but the facility has not been disconnected); and (iii) BellSouth has received an order to establish new service or transfer of service from a customer or a customer's CLEC at the same address served by the denied facility.
- 3.3.1 Upon receipt of a service order, BellSouth will do the following:
- 3.3.1.1 Process disconnect and reconnect orders to provision the service which shall be due dated using current interval guidelines.
- 3.3.1.2 Reuse the serving facility for the retail, resale service, or network element at the same location.
- 3.3.1.3 Notify NewSouth after the disconnect order has been completed.
- 3.4 <u>Contact Numbers</u>. The Parties agree to provide one another with toll-free contact numbers for the purpose of ordering, provisioning and maintenance of services.

BellSouth shall provide single points of contact ("SPOC") for the provisioning of Resale Services (LCSC) and UNEs (UNE Center) ordered by NewSouth. Preordering and ordering electronic interfaces shall be available, subject to downtime for scheduled maintenance and other scheduled activities for which advance notification is required and provided by BellSouth, seven (7) days a week, 24 hours a day. BellSouth shall provide access to assistance for technical issues such as connectivity and passwords related to LENS, TAG and TAFI, and to the "EDI Central Group" for technical problems with EDI. Assistance will be available by telephone during normal business hours and through other contacts on nights, weekends and holidays.

- 3.5 <u>Subscription Functions</u>. In cases where BellSouth performs subscription functions for an inter-exchange carrier (i.e. PIC and LPIC changes via Customer Account Record Exchange (CARE)), BellSouth will provide the affected inter-exchange carriers with the Operating Company Number (OCN) of the local provider for the purpose of obtaining end user billing account and other end user information required under subscription requirements.
- 3.6 <u>Cancellation Charges</u>. If NewSouth cancels an order for network elements or other services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with FCC No. 1 Tariff, Section 5.4.
- 3.7 <u>Disaster Recovery Plan.</u> BellSouth's Disaster Recover Plan is as set forth in Attachment 11 of this Agreement.
- 3.8 <u>Ordering and Provisioning Information.</u> BellSouth shall provide the following to NewSouth upon request:
- 3.8.1 Design Layout Records ("DLRs") for designed unbundled Network Elements where applicable;
- 3.8.2 Advance information on the details and requirements for planning and implementation of NPA splits; and
- 3.8.3 Access to the Regional Street Address Guide ("RSAG") information via LENS, TAG or RoboTAG<sup>TM</sup> pre-ordering.
- 3.9 Each Party shall establish mutually acceptable methods and procedures for handling all misdirected calls from the other Party's End Users. Each Party, on a reciprocal basis, shall refer all misdirected calls that it receives from the other Party's End Users to a designated number of the other Party as set forth in this section. NewSouth and BellSouth each shall be responsible for providing the other party with its current toll free number. The foregoing shall apply only when the Party receiving such call knows or has reason to know that the call is misdirected from an End User of the other Party hereto.
- 3.10 BellSouth shall provide order format specifications to NewSouth for all available services, features, and functions and for ancillary data required by BellSouth to provision those services.
- 3.11 BellSouth shall provide NewSouth with standard expected provisioning intervals for all unbundled Network Elements.
- 3.12 BellSouth shall not reconfigure any NewSouth service arrangements of any NewSouth End User for Resale services, UNEs or Combinations, unless so directed by NewSouth. Any NewSouth End User that contacts BellSouth regarding a change to its NewSouth service (excluding changes in its local service provider) shall be advised to contact NewSouth. Any BellSouth End User that contacts NewSouth regarding a

- change in BellSouth service (excluding changes in its local service provider) shall be advised to contact BellSouth.
- 3.13 The Parties shall provide a generic intercept referral message that includes any new telephone number of an End User for the same period of time that BellSouth currently provides such a message for its own End Users. The intercept message shall be similar in format to the intercept referral message currently provided by BellSouth for its own End Users.
- 3.14 BellSouth shall perform all pre-testing necessary to ensure the services ordered meet the specifications outlined in the technical service description provided by BellSouth for the service being ordered.
- Any written "leave behind" materials that BellSouth technicians provide to NewSouth End Users shall be non-branded materials that do not identify the work being performed as being by BellSouth. These materials shall include, without limitation, non-branded forms for the Customer and non-branded "not at home" cards.
- 3.16 If a NewSouth End User requests a change of service at the time of installation, BellSouth technicians shall direct them to contact NewSouth directly and provide a toll-free number supplied by NewSouth. When a BellSouth employee visits the premise of an NewSouth End User, the BellSouth employee shall inform the Customer that he or she is there acting on behalf of NewSouth.
- 3.17 BellSouth shall provide telephone and/or facsimile notification to NewSouth of any NewSouth end user service requests and charges therefore not authorized on the NewSouth service request, and obtain NewSouth's approval prior to commencing work.
- 3.18 Each Party shall train and direct its employees who have contact with End Users of the other Party in the process of provisioning, maintenance or repair not to disparage the other Party or its services in any way to the other Party's End Users.
- When NewSouth places an order, NewSouth shall specify a requested Due Date, and BellSouth shall specify a Due Date based on the applicable intervals. In the event NewSouth's requested date is less than the standard interval, NewSouth shall contact BellSouth by telephone or use the expedite request field on the order as directed by BellSouth and the Parties shall negotiate an expedited Due Date as set forth in this Agreement. Expedite charges shall be as set-forth in Attachment 2 of this Agreement. BellSouth shall not complete the order prior to the Due Date unless authorized by NewSouth. If BellSouth misses the Due Date, BellSouth shall promptly notify NewSouth of the revised installation Due Date. If NewSouth requests that an order be expedited, BellSouth shall notify NewSouth of the status of the order (i) by the end of the same Business Day when such expedite requests are made prior to noon; or (ii) by noon the following Business Day otherwise.
- 3.20 NewSouth and BellSouth shall agree to escalation procedures and contacts for resolving questions and disputes related to ordering and provisioning procedures or to

- the processing of individual orders, subject ultimately to the dispute resolution provisions of this Agreement. The Parties shall use best efforts to notify each other of any modifications to these contacts within ten (10) days of any such modifications.
- 3.21 BellSouth shall transmit to NewSouth a FOC or, in the alternative, notification of the lack of available facilities within time periods specified hereafter, after BellSouth's receipt of a complete and correct order from NewSouth, provided, however, that an order for complex services requiring a service inquiry shall be deemed received for these purposes only after completion of the service inquiry. The FOC shall contain a commitment date, which shall be established on a nondiscriminatory basis with respect to installation dates for comparable orders at such time. If NewSouth uses LENS, EDI, or any other electronic interface for the submission of the order, the FOC or notification shall be posted by BellSouth in such interface within twenty-four (24) hours of receipt of the order for all fully mechanized requests. If NewSouth does not use these interfaces, or these interfaces are not available for the service or UNE being ordered, BellSouth shall transmit the FOC or notification by telecopier to a toll-free number provided by NewSouth within forty-eight (48) hours of BellSouth's receipt of the order. When NewSouth submits a complete and correct LSR for SPNP and an associated unbundled Loop simultaneously, BellSouth shall likewise issue a FOC for both the Loop and the SPNP simultaneously.
- 3.22 For Local Service Requests submitted via an electronic interface, BellSouth shall notify NewSouth via the same electronic interface, of Rejections/Errors contained in any of the data element(s) field(s) contained on any NewSouth Local Service Request. For Local Service Requests submitted manually, BellSouth shall notify NewSouth by facsimile of such Rejections and Errors. BellSouth will notify NewSouth of Rejections or Errors in 95% of mechanized orders within one (1) hour from BellSouth's receipt of the order. BellSouth will notify NewSouth of Rejections or Errors in 85% of non-mechanized and partially mechanized orders within forty-eight (48) hours from BellSouth's receipt of the order.

# **Attachment 7**

**Billing and Billing Accuracy Certification** 

# TABLE OF CONTENTS

1.	Payment and Billing Arrangements	3
	Billing Accuracy Certification	
3.	Billing Disputes	
4.	RAO Hosting	8
5.	Optional Daily Usage File	11
6.	Access Daily Usage File	14
7.	Enhanced Optional Daily Usage File	17
Ra	ntes	Exhibit A

#### BILLING AND BILLING ACCURACY CERTIFICATION

#### 1. Payment and Billing Arrangements

All negotiated rates, terms and conditions set forth in this Attachment pertain to billing and billing accuracy certifications.

- Billing. BellSouth agrees to provide billing through the Carrier Access Billing System (CABS) and through the Customer Records Information System (CRIS) depending on the particular service(s) that NewSouth requests. BellSouth will bill and record in accordance with this Agreement those charges NewSouth incurs as a result of NewSouth purchasing from BellSouth Network Elements and Other Services as set forth in this Agreement. BellSouth will format all bills in CBOS Standard or CLUB/EDI format, depending on the type of service ordered. For those services where standards have not yet been developed, BellSouth's billing format will change as necessary when standards are finalized by the industry forum.
- 1.1.1 For any service(s) BellSouth orders from NewSouth, NewSouth shall bill BellSouth in CABS format or in accordance with industry standards.
- 1.1.2 If either Party requests multiple billing media or additional copies of bills, the Billing Party will provide these at a reasonable cost.
- Master Account. After receiving certification as a local exchange company from the appropriate regulatory agency, NewSouth will provide the appropriate BellSouth account manager the necessary documentation to enable BellSouth to establish a master account for Local Interconnection, Network Elements and Other Services, and/or resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA"), Carrier Identification Code (CIC), Group Access Code (GAC), Access Customer Name and Address (ACNA) and a tax exemption certificate, if applicable.
- Payment Responsibility. Payment of all charges will be the responsibility of the billed Party. The billed Party shall make payment to the billing Party for all services billed. The billing Party is not responsible for payments not received by the billed Party from the billed Party's customer. The billing Party will not become involved in billing disputes that may arise between the billed Party and the billed Party's customer. Payments made to the billing Party as payment on account will be credited to an accounts receivable master account and not to an end user's account.
- 1.4 <u>Payment Due</u>. The payment will be due on or before the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by the billing Party.

If the payment due date falls on a Sunday or on a Holiday which is observed on a Monday, the payment due date shall be the first non-Holiday day following such Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If payment is not received by the payment due date, a late payment penalty, as set forth in Section 1.7, below, shall apply.

- 1.5 <u>Tax Exemption</u>. Upon proof of tax exempt certification from NewSouth, the total amount billed to NewSouth will not include those taxes or fees for which the CLEC is exempt. NewSouth will be solely responsible for the computation, tracking, reporting and payment of all taxes and like fees associated with the services provided to the end user of NewSouth.
- Late Payment. If any portion of the payment is received by the billing Party after the payment due date as set forth preceding, or if any portion of the payment is received by the billing Party in funds that are not immediately available to the billing Party, then a late payment penalty shall be due to the billing Party. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor and will be applied on a per bill basis. When BellSouth is the billing Party, the late factor shall be as set forth in Section A2 of the General Subscriber Services Tariff, Section B2 of the Private Line Service Tariff or Section E2 of the Intrastate Access Tariff, as appropriate. When NewSouth is the billing Party the late factor shall be one and one-half percent (1.5%) per month. The billed Party will be charged a fee for all returned checks as set forth in Section A2 of the General Subscriber Services Tariff or pursuant to the applicable state law.
- 1.7 <u>Discontinuing Service to NewSouth</u>. The procedures for discontinuing service to NewSouth are as follows:
- 1.7.1 BellSouth reserves the right to suspend or terminate service for nonpayment of services or in the event of prohibited, unlawful or improper use of BellSouth facilities or service or any other violation or noncompliance by NewSouth of the rules and regulations contained in BellSouth's tariffs.
- 1.7.2 If payment of <u>Undisputed Amounts due</u>, as described in Section 3 of this Attachment, is not received by the bill date in the month after the original bill date, BellSouth may provide written notice to NewSouth that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition, BellSouth may, at the same time, give thirty (30) days notice to NewSouth at the billing address to discontinue the provision of existing services to NewSouth at any time thereafter.
- 1.7.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.

- 1.7.4 If BellSouth does not discontinue the provision of the services involved on the date specified in the thirty days notice and NewSouth's noncompliance continues, nothing contained herein shall preclude BellSouth's right to discontinue the provision of the services to NewSouth without further notice.
- 1.7.5 If payment is not received or satisfactory arrangements made for payment by the date given in the written notification, NewSouth's services will be discontinued. Upon discontinuance of service on NewSouth's account, service to the NewSouth's end users will be denied. BellSouth will reestablish service at the request of the end user or NewSouth for BellSouth to reestablish service upon payment of the appropriate connection fee and subject to BellSouth's normal application procedures. NewSouth is solely responsible for notifying the end user of the proposed service disconnection. If within fifteen (15) days after an end user's service has been denied and no arrangements to reestablish service have been made consistent with this subsection, the end user's service will be disconnected.
- 1.8 Deposit Policy. When purchasing services from BellSouth, NewSouth will be required to complete the BellSouth Credit Profile and provide information regarding credit worthiness. Based on the results of the credit analysis, BellSouth reserves the right to secure the account with a suitable form of security deposit. Such security deposit shall take the form of cash, an Irrevocable Letter of Credit (BellSouth form), Surety Bond (BellSouth form) or, in BellSouth's reasonable and nondiscriminatory discretion, some other form of security. Any such security deposit shall in no way release NewSouth from its obligation to make complete and timely payments of its bill. Such security shall be required prior to the inauguration of service. If, in the reasonable opinion of BellSouth, circumstances so warrant and/or gross monthly billing has increased beyond the level initially used to determine the level of security, BellSouth reserves the right to request additional security. Interest on a security deposit, if provided in cash, shall accrue and be paid in accordance with the terms in the appropriate BellSouth tariff. Security deposits collected under this Section shall not exceed two months' estimated billing. In the event NewSouth fails to remit to BellSouth any deposit requested pursuant to this Section, service to NewSouth may be terminated in accordance with the terms of Section 1.7 of this Attachment, and any security deposits will be applied to NewSouth's account(s).
- 1.9 Notices. Notwithstanding anything to the contrary in this Agreement, all bills and notices regarding billing matters, including notices relating to security deposits, to rejection of additional orders from NewSouth and to disconnection of services for nonpayment of charges, shall be forwarded to the individual and/or address provided by NewSouth in establishment of its billing accounts with BellSouth, or to the individual and/or address subsequently provided by NewSouth as the contact for billing information. All monthly bills and the notices described in this Section shall be forwarded to the same individual and/or address; provided, however, upon written

notice from NewSouth to BellSouth's billing organization, a final notice of disconnection of services purchased by NewSouth under this Agreement shall be sent via certified mail to the individual(s) listed in the Notices provision of the General Terms and Conditions of this Agreement at least 30 days before BellSouth takes any action to terminate such services.

Rates. Rates for Optional Daily Usage File (ODUF), Enhanced Optional Daily Usage File (EODUF), Access Daily Usage File (ADUF), and Centralized Message Distribution Service (CMDS) are set out in Exhibit A to this Attachment. If no rate is identified in this Attachment, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the Parties upon request by either Party.

#### 2. Billing Accuracy Certification

- 2.1 Upon request, BellSouth and NewSouth will agree upon a billing quality assurance program for all billing elements covered in this Agreement that will eliminate the need for post-billing reconciliation. Appropriate terms for access to any BellSouth documents, systems, records, and procedures for the recording and billing of charges will be part of that program.
- As part of the billing quality assurance program, BellSouth and NewSouth will develop standards, measurements, and performance requirements for a local billing measurements process. On a regular basis BellSouth will provide NewSouth with mutually agreed upon performance measurement data that substantiates the accuracy, reliability, and integrity of the billing process for local billing. In return, NewSouth will pay all bills received from BellSouth in full by the payment due date.
- 2.3 Local billing discrepancies will be addressed in an orderly manner via a mutually agreed upon billing exemption process.
- 2.3.1 Each Party agrees to notify the other Party upon identifying a billing discrepancy. The Parties shall endeavor to resolve any billing discrepancy within sixty (60) calendar days of the notification date. A mutually agreed upon escalation process will be established for resolving local billing discrepancies as part of the billing quality assurance program.
- 2.3.2 Closure of a specific billing period will occur by joint agreement of the Parties whereby the Parties agree that such billing period is closed to any further analysis and financial transactions except those resulting from regulatory mandates. Closure will take place within a mutually agreed upon time interval from the bill date. The month being closed represents those charges that were billed or should have been billed by the designated bill date.

#### 3. Billing Disputes

- 3.1 Where the Parties have not agreed upon a billing quality assurance program, or where such process does not encompass all billing disputes, billing disputes shall be handled pursuant to the terms of this section.
- 3.1.1 Each Party agrees to notify the other Party in writing upon the discovery of a billing dispute. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the notification date. If the Parties are unable within the 60 day period to reach resolution, then the aggrieved Party may pursue dispute resolution in accordance with the General Terms and Conditions of this Agreement.
- 3.1.2 For purposes of this Section 3, a billing dispute means a dispute of a specific amount of money actually billed by either Party. The dispute must be clearly explained by the disputing Party and supported by written documentation, which clearly shows the basis for disputing the charges. By way of example and not by limitation, a billing dispute will not include the refusal to pay all or part of a bill or bills when no written documentation is provided to support the dispute, nor shall a billing dispute include the refusal to pay other amounts owed by the billed Party until the dispute is resolved. Claims by the billed Party for damages of any kind will not be considered a billing dispute for purposes of this Section 3. Once the billing dispute is resolved, the disputing Party will make immediate payment on any of the disputed amount owed to the billing Party or the billing Party shall have the right to pursue normal treatment procedures. Any credits due to the disputing Party, pursuant to the billing dispute, will be applied to the disputing Party's account by the billing Party immediately upon resolution of the dispute.
- 3.2 If a Party disputes a charge and does not pay such charge by the payment due date, or if a payment or any portion of a payment is received by either Party after the payment due date, or if a payment or any portion of a payment is received in funds which are not immediately available to the other Party, then a late payment penalty shall be assessed. For bills rendered by either Party for payment, the late payment charge for both Parties shall be calculated based on the portion of the payment not received by the payment due date times the late factor as set forth in the following BellSouth tariffs: for services purchased from the General Subscribers Services Tariff for purposes of resale and for ports and non-designed loops, Section A2 of the General Subscriber Services Tariff: for services purchased from the Private Line Tariff for purposes of resale, Section B2 of the Private Line Service Tariff; and for network elements and other services and local interconnection charges, Section E2 of the Access Service Tariff. In no event, however, shall interest be assessed by either Party on any previously assessed late payment charges. The Parties shall assess interest on previously assessed late payment charges only in a state where it has the authority pursuant to its tariffs. No interest or late payment penalties shall be assessed in the event that the billed Party prevails in a billing dispute.

## 4. RAO Hosting

- 4.1 RAO Hosting, Calling Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to NewSouth by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this Agreement, including such reasonable and nondiscriminatory revisions as may be made from time to time by BellSouth. BellSouth will use best efforts to provide NewSouth with 30 days' advanced notice of such revisions.
- 4.2 NewSouth shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- 4.3 Compensation amounts, if applicable, will be billed by BellSouth to NewSouth on a monthly basis in arrears. Amounts due from one Party to the other (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- NewSouth must have its own unique hosted RAO code. Requests for establishment of RAO status where BellSouth is the selected Centralized Message Distribution System (CMDS) interfacing host, require written notification from NewSouth to the BellSouth RAO Hosting coordinator at least eight (8) weeks prior to the proposed effective date. The proposed effective date will be mutually agreed upon between the Parties with consideration given to time necessary for the completion of required Telcordia (formerly BellCore) functions. BellSouth will request the assignment of an RAO code from its connecting contractor, currently Telcordia (formerly BellCore), on behalf of NewSouth and will coordinate all associated conversion activities.
- 4.5 BellSouth will receive messages from NewSouth that are to be processed by BellSouth, another LEC or CLEC in the BellSouth region or a LEC outside the BellSouth region.
- 4.6 BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from NewSouth.
- 4.7 All data received from NewSouth that is to be processed or billed by another LEC or CLEC within the BellSouth region will be distributed to that LEC or CLEC in accordance with the Agreement(s) which may be in effect between BellSouth and the involved LEC or CLEC.
- 4.8 All data received from NewSouth that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the

- agreement(s) which may be in effect between BellSouth and its connecting contractor (currently Telcordia (formerly BellCore)).
- 4.9 BellSouth will receive messages from the CMDS network that are destined to be processed by NewSouth and will forward them to NewSouth on a daily basis.
- 4.10 Transmission of message data between BellSouth and NewSouth will be via CONNECT:Direct.
- 4.11 All messages and related data exchanged between BellSouth and NewSouth will be formatted in accordance with accepted industry standards for EMI formatted records and packed between appropriate EMI header and trailer records, also in accordance with accepted industry standards.
- 4.12 NewSouth will ensure that the recorded message detail necessary to recreate files provided to BellSouth will be maintained for back-up purposes for a period of three (3) calendar months beyond the related message dates.
- 4.13 Should it become necessary for NewSouth to send data to BellSouth more than sixty (60) days past the message date(s), NewSouth will notify BellSouth in advance of the transmission of the data. If there will be impacts outside the BellSouth region, BellSouth will work with its connecting contractor and NewSouth to notify all affected Parties.
- In the event that data to be exchanged between the two Parties should become lost or destroyed, both Parties will work together to determine the source of the problem. Once the cause of the problem has been jointly determined and the responsible Party (BellSouth or NewSouth) identified and agreed to, the company responsible for creating the data (BellSouth or NewSouth) will make every effort to have the affected data restored and retransmitted. If the data cannot be retrieved, the responsible Party will be liable to the other Party for any resulting lost revenue. Lost revenue may be a combination of revenues that could not be billed to the end users and associated access revenues. Both Parties will work together to estimate the revenue amount based upon historical data through a method mutually agreed upon. The resulting estimated revenue loss will be paid by the responsible Party to the other Party within three (3) calendar months of the date of problem resolution, or as mutually agreed upon by the Parties.
- 4.15 Should an error be detected by the EMI format edits performed by BellSouth on data received from NewSouth, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify NewSouth of the error condition. NewSouth will correct the error(s) and will resend the entire pack to BellSouth for processing. In the event that an out-of-sequence condition occurs on subsequent packs, NewSouth will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth.

- 4.16 In association with message distribution service, BellSouth will provide NewSouth with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 4.17 In no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this Agreement.
- 4.18 RAO Compensation
- 4.18.1 Rates for message distribution service provided by BellSouth for NewSouth are as set forth in Exhibit A to this Attachment.
- 4.18.2 Rates for data transmission associated with message distribution service are as set forth in Exhibit A to this Attachment.
- A.18.3 Data circuits (private line or dial-up) will be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties.
- 4.18.4 All equipment, including modems and software that is required on the NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.
- 4.19 <u>Intercompany Settlements Messages</u>
- 4.19.1 This Section addresses the settlement of revenues associated with traffic originated from or billed by NewSouth as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one Bell operating territory and bills in another Bell operating territory is included. Traffic that originates and bills within the same Bell operating territory will be settled on a local basis between NewSouth and the involved company (ies), unless that company is participating in NICS.
- 4.19.2 Both traffic that originates outside the BellSouth region by NewSouth and is billed within the BellSouth region, and traffic that originates within the BellSouth region and is billed outside the BellSouth region by NewSouth, is covered by this Agreement (CATS). Also covered is traffic that either is originated by or billed by NewSouth,

- involves a company other than NewSouth, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).
- 4.19.3 Once NewSouth is operating within the BellSouth territory, revenues associated with calls originated and billed within the BellSouth region will be settled via Telcordia (formerly BellCore)'s, its successor or assign, NICS system.
- 4.19.4 BellSouth will receive the monthly NICS reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- 4.19.5 BellSouth will receive the monthly Calling Card and Third Number Settlement System (CATS) reports from Telcordia (formerly BellCore), its successor or assign, on behalf of NewSouth. BellSouth will distribute copies of these reports to NewSouth on a monthly basis.
- 4.19.6 BellSouth will collect the revenue earned by NewSouth from the Bell operating company in whose territory the messages are billed (CATS), less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth to the Bell operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf on NewSouth. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.
- 4.19.7 BellSouth will collect the revenue earned by NewSouth within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of NewSouth. BellSouth will remit the revenue billed by NewSouth within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit issued to NewSouth via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

BellSouth and NewSouth agree that monthly netted amounts of less than fifty dollars (\$50.00) will not be settled.

## 5. Optional Daily Usage File

Upon written request from NewSouth, BellSouth will provide the Optional Daily Usage File (ODUF) service to NewSouth pursuant to the terms and conditions set forth in this section

- 5.2 The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Optional Daily Usage File.
- 5.3 The Optional Daily Usage Feed will contain billable messages that were carried over the BellSouth Network and processed in the BellSouth Billing System, but billed to a NewSouth customer.
  - Charges for delivery of the Optional Daily Usage File will appear on the NewSouth's monthly bills. The charges are as set forth in Exhibit A to this Attachment.
- The Optional Daily Usage Feed will contain both rated and unrated messages. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of the NewSouth will be the responsibility of the NewSouth. If, however, the NewSouth should encounter significant volumes of errored messages that prevent processing by the NewSouth within its systems, BellSouth will work with the NewSouth to determine the source of the errors and the appropriate resolution.
- 5.6 The following specifications shall apply to the Optional Daily Usage Feed.
- 5.6.1 <u>Usage To Be Transmitted</u>
- 5.6.1.1 The following messages recorded by BellSouth will be transmitted to the NewSouth:
  - Message recording for per use/per activation type services (examples: Three Way Calling, Verify, Interrupt, Call Return, ETC.)
  - Measured billable Local
  - Directory Assistance messages
  - IntraLATA Toll
  - WATS & 800 Service
  - N11
  - Information Service Provider Messages
  - Operator Services Messages
  - Operator Services Message Attempted Calls (Network Element only)
  - Credit/Cancel Records
  - Usage for Voice Mail Message Service
- 5.6.1.2 Rated Incollects (originated in BellSouth and from other companies) can also be on Optional Daily Usage File. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.

- 5.6.1.3 BellSouth will perform duplicate record checks on records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 5.6.1.4 In the event that NewSouth detects a duplicate on Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).

#### 5.6.2 Physical File Characteristics

- 5.6.2.1 The Optional Daily Usage File will be distributed to NewSouth via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a variable block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

#### 5.6.3 <u>Packing Specifications</u>

- 5.6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 5.6.3.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

The data will be packed using ATIS EMI records.

#### 5.6.4 Pack Rejection

5.6.4.1 NewSouth will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. NewSouth will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to NewSouth by BellSouth.

#### 5.6.5 Control Data

NewSouth will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate NewSouth received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by NewSouth for reasons stated in the above section.

## 5.6.6 <u>Testing</u>

5.6.6.1 Upon request from NewSouth, BellSouth shall send test files to NewSouth for the Optional Daily Usage File. The Parties agree to review and discuss the file's content and/or format. For testing of usage results, BellSouth shall request that NewSouth set up a production (LIVE) file. The live test may consist of NewSouth's employees making test calls for the types of services NewSouth requests on the Optional Daily Usage File. These test calls are logged by NewSouth, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be completed within 30 calendar days from the date on which the initial test file was sent.

#### 6. Access Daily Usage File

- 6.1. Upon written request from NewSouth, BellSouth will provide the Access Daily Usage File (ADUF) service to NewSouth pursuant to the terms and conditions set forth in this section.
- 6.2 The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Access Daily Usage File.
- The Access Daily Usage Feed will contain access messages associated with a port that NewSouth has purchased from BellSouth

- Charges for delivery of the Access Daily Usage File will appear on the NewSouth's' monthly bills. The charges are as set forth in Exhibit A to this Attachment. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- Messages that error in the billing system of the NewSouth will be the responsibility of the NewSouth. If, however, the NewSouth should encounter significant volumes of errored messages that prevent processing by the NewSouth within its systems, BellSouth will work with the NewSouth to determine the source of the errors and the appropriate resolution.
- 6.6 <u>Usage To Be Transmitted</u>
- 6.6.1 The following messages recorded by BellSouth will be transmitted to NewSouth:

Originating and terminating interstate and intrastate access records associated with a port.

Terminating access records for undetermined jurisdiction access records associated with a port.

When NewSouth purchases Network Element ports from BellSouth and calls are made using these ports, BellSouth will handle the calls as follows:

Originating from Network Element and carried by Interexchange Carrier:

BellSouth will bill network element to CLEC and send access record to the CLEC via ADUF

Originating from network element and carried by BellSouth (NewSouth is BellSouth's toll customer):

BellSouth will bill resale toll rates to NewSouth and send toll record for the end user toll billing purposes via ODUF (Optional Daily Usage File). Access record will be sent to NewSouth via ADUF.

Terminating on network element and carried by Interexchange Carrier:

BellSouth will bill network element to NewSouth and send access record to NewSouth.

Terminating on network element and carried by BellSouth:

BellSouth will bill network element to NewSouth and send access record to NewSouth.

- 6.6.3 BellSouth will perform duplicate record checks on records processed to the Access Daily Usage File. Any duplicate messages detected will be dropped and not sent to NewSouth.
- 6.6.4 In the event that NewSouth detects a duplicate on the Access Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth.)

#### 6.6.5 <u>Physical File Characteristics</u>

- 6.6.5.1 The Access Daily Usage File will be distributed to NewSouth via an agreed medium with CONNECT:Direct being the preferred transport method. The Daily Usage Feed will be a fixed block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (210 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

#### 6.6.6 Packing Specifications

- 6.6.6.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 6.6.6.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data

exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

The data will be packed using ATIS EMI records.

#### 6.6.7 Pack Rejection

NewSouth will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. NewSouth will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to NewSouth by BellSouth.

#### 6.6.8 <u>Control Data</u>

NewSouth will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate NewSouth received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by NewSouth for reasons stated in the above section.

#### 6.6.9 <u>Testing</u>

6.6.9.1 Upon request from NewSouth, BellSouth shall send test files to NewSouth for the Access Daily Usage File. Testing shall consist of actual calls made from live accounts. A call log shall be supplied along with test request information. The Parties agree to review and discuss the file's content and/or format.

#### 7. Enhanced Optional Daily Usage File

- Upon written request from NewSouth, BellSouth will provide the Enhanced Optional Daily Usage File (EODUF) service to NewSouth pursuant to the terms and conditions set forth in this section. EODUF will only be sent to existing ODUF subscribers who request the EODUF option.
- 7.2 The NewSouth shall furnish all relevant information required by BellSouth for the provision of the Enhanced Optional Daily Usage File.
- 7.3 The Enhanced Optional Daily Usage File (EODUF) will provide usage data for local calls originating from resold Flat Rate Business and Residential Lines.

Charges for delivery of the Enhanced Optional Daily Usage File will appear on the NewSouth' monthly bills. The charges are as set forth in Exhibit A to this Attachment.

- 7.4 All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.
- 7.5 Messages that error in the billing system of the NewSouth will be the responsibility of the NewSouth. If, however, the NewSouth should encounter significant volumes of errored messages that prevent processing by the NewSouth within its systems, BellSouth will work with the NewSouth to determine the source of the errors and the appropriate resolution.
- 7.6 The following specifications shall apply to the Optional Daily Usage Feed.
- 7.6.1 <u>Usage To Be Transmitted</u>
- 7.6.1.1 The following messages recorded by BellSouth will be transmitted to the NewSouth:

Customer usage data for flat rated local call originating from CLEC end user lines (1FB or 1FR). The EODUF record for flat rate messages will include:

Date of Call From Number To Number

Connect Time

Conversation Time

Method of Recording

From RAO

Rate Class

Message Type

**Billing Indicators** 

Bill to Number

- 7.6.1.2 BellSouth will perform duplicate record checks on EODUF records processed to Optional Daily Usage File. Any duplicate messages detected will be deleted and not sent to NewSouth.
- 7.6.1.3 In the event that NewSouth detects a duplicate on Enhanced Optional Daily Usage File they receive from BellSouth, NewSouth will drop the duplicate message (NewSouth will not return the duplicate to BellSouth).

#### 7.6.2 Physical File Characteristics

- 7.6.2.1 The Enhanced Optional Daily Usage Feed will be distributed to NewSouth over their existing Optional Daily Usage File (ODUF) feed. The EODUF messages will be intermingled among NewSouth's Optional Daily Usage File (ODUF) messages. The EODUF will be a variable block format (2476) with an LRECL of 2472. The data on the EODUF will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays).
- 7.6.2.2 Data circuits (private line or dial-up) may be required between BellSouth and NewSouth for the purpose of data transmission. Where a dedicated line is required, NewSouth will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. NewSouth will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to NewSouth. Additionally, all message toll charges associated with the use of the dial circuit by NewSouth will be the responsibility of NewSouth. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the Parties. All equipment, including modems and software that is required on NewSouth end for the purpose of data transmission will be the responsibility of NewSouth.

## 7.6.3 <u>Packing Specifications</u>

- 7.6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 7.6.3.2 The Operating Company Number (OCN), From Revenue Accounting Office (RAO), and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to NewSouth which BellSouth RAO that is sending the message. BellSouth and NewSouth will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by NewSouth and resend the data as appropriate.

The data will be packed using ATIS EMI records.

# BELLSOUTH/NEWSOUTH RATES ODUF/EDOUF/ADUF/CMDS

		RATES BY STATE								
DESCRIPTION	USOC	AL	FL	GA	KY	LA	MS	NC	sc	TN
ODUF/EODUF/ADUF/CMDS										
ODUF: Recording, per message	N/A	\$0.0002	\$0.008	\$0.008	\$0.0008611	\$0.00019	\$0.0001179	\$0.008	\$0.0002862	\$0.008
ODUF: Message Processing, per message	N/A	\$0.0033	\$0.004	\$0.004	\$0.0032357	\$0.0024	\$0.0032089	\$0.004	\$0.0032344	\$0.004
EODUF: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
ADUF: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
CMDS: Message Processing, per message	N/A	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004	\$0.004
ODUF: Message Processing, per magnetic tape provisioned	N/A	\$55.19	\$54.95	\$54.95	\$55.68	\$47.30	\$54.62	\$54.95	\$54.72	\$54.95
EODUF: Message Processing, per magnetic tape provisioned	N/A	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30	\$47.30
ODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.00004	\$0.001	\$0.001	\$0.0000365	\$0.00003	\$0.0000354	\$0.001	\$0.0000357	\$0.001
EODUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364	\$0.0000364
ADUF: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001
CMDS: Data Transmission (CONNECT:DIRECT), per message	N/A	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001	\$0.001

#### NOTES:

If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the parties upon request by either party.

# **Attachment 8**

Rights-of-Way, Conduits and Pole Attachments

# Rights-of-Way, Conduits and Pole Attachments

BellSouth will provide nondiscriminatory access to any pole, duct, conduit, or right-of-way owned or controlled by BellSouth pursuant to 47 U.S.C. § 224, as amended by the Act, pursuant to terms and conditions of a license agreement subsequently negotiated with BellSouth's Competitive Structure Provisioning Center.

# **Attachment 9**

**Performance Measurements** 

# **Performance Measurements**

Upon a particular Commission's issuance of an Order pertaining to Performance Measurements and/or Remedies in a proceeding expressly applicable to all CLECs generally, BellSouth shall implement in that state such Performance Measurements as of the date specified by the Commission.

# AGREEMENT IMPLEMENTATION TEMPLATE (Residence)

#### for

## NewSouth

# **BellSouth Standard Interconnection Agreement**

Agreement Effective Date:	Agreement Expiration Date:
Account Manager:	Account Manager Tel No:

Attachment Name/Number	Section Number	Version Date	Planned Activities
Terms/Conditions PartA	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	6	2/29/00	
	7	2/29/00	
	8	2/29/00	
	9	2/29/00	
	10	2/29/00	
	11	2/29/00	
	12	2/29/00	
	13	2/29/00	
	14	2/29/00	
	15	2/29/00	
	16	2/29/00	
	17	2/29/00	
	18	2/29/00	
	19	2/29/00	
	20	2/29/00	
	21	2/29/00	
	22	2/29/00	
	23	2/29/00	
	24	2/29/00	
	25	2/29/00	
	26	2/29/00	
Terms/Conditions Part B	-	2/29/00	

Version 1Q00:3/6/00 Attachment 10-Residence

# AGREEMENT IMPLEMENTATION TEMPLATE (Residence)

#### for

# NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment Name/Number	Section Number	Version Date	Planned Activities
1-Resale	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	6	2/29/00	
	7	2/29/00	
	8	2/29/00	
	9	2/29/00	
	10	2/29/00	
	11	2/29/00	
	12	2/29/00	
	13	2/29/00	
	Exhibit A	2/29/00	
	Exhibit B	2/29/00	
	Exhibit C	2/29/00	
	Exhibit D	2/29/00	
	Exhibit E	2/29/00	
	Exhibit F	2/29/00	
	Exhibit G	2/29/00	
		2/29/00	
2-Network Elements & Other Services	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	6	2/29/00	
	7	2/29/00	
	8	2/29/00	
	9	2/29/00	

Version 1Q00:3/6/00 Attachment 10-Residence

# AGREEMENT IMPLEMENTATION TEMPLATE (Residence)

## for

# NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment	Section	Version	Planned Activities
Name/Number	Number	Date	
	10	2/29/00	
	11	2/29/00	
	12	2/29/00	
	13	2/29/00	
	14	2/29/00	
	15	2/29/00	
	16	2/29/00	
	17	2/29/00	
	Exhibit A	2/29/00	
	Exhibit B	2/29/00	
	Exhibit C	2/29/00	
3-Local Interconnection	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	6	2/29/00	
	7	2/29/00	
	8	2/29/00	
	Exhibit A	2/29/00	
	Exhibit B	2/29/00	
	Exhibit C	2/29/00	
	Exhibit D	2/29/00	
	Exhibit E	2/29/00	
4-Physical Collocation	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	6	2/29/00	

Version 1Q00:3/6/00 Attachment 10-Residence

# AGREEMENT IMPLEMENTATION TEMPLATE (Residence)

## for

## NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment	Section	Version	Planned Activities
Name/Number	Number	Date	
	_ , 0,, 0 _		
	7	2/29/00	
	8	2/29/00	
	9	2/29/00	
	10	2/29/00	
	11	2/29/00	
	12	2/29/00	
	13	2/29/00	
	14	2/29/00	
	Exhibit A	2/29/00	
	Exhibit B	2/29/00	
5-Access to Numbers &		2/29/00	
Number Portability	1		
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	6	2/29/00	
	7	2/29/00	
	8	2/29/00	
	Exhibit A	2/29/00	
6-Ordering/Provisioning	1	2/29/00	
	2	2/29/00	
	3	2/29/00	
7-Billing & Billing		2/29/00	
Accuracy Certification	1		
	2	2/29/00	
	3	2/29/00	
	4	2/29/00	
	5	2/29/00	
	6	2/29/00	
	7	2/29/00	

Version 1Q00:3/6/00 Attachment 10-Residence

# AGREEMENT IMPLEMENTATION TEMPLATE (Residence)

## for

## NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment Name/Number	Section Number	Version Date	Planned Activities
1 (41116) 1 (41118) 61	1 (4111001	2400	
	Exhibit A	2/29/00	
8-ROW/Conduits/PoleAtt	1	2/29/00	
9-Perf Measurement	Pre-Ordering	2/29/00	
	Ordering	2/29/00	
	Provisioning	2/29/00	
	Maint/Repair	2/29/00	
	Billing	2/29/00	
	Opr Svcs/DA	2/29/00	
	E911	2/29/00	
	Trunk Grp Perf	2/29/00	
	Collocation	2/29/00	
	Appendix A	2/29/00	
	Appendix B	2/29/00	
	Appendix C	2/29/00	
10-Executive Summary		2/29/00	
		2/29/00	
11-Disaster Recovery		2/29/00	
		2/29/00	

Version 1Q00:3/6/00 Attachment 10-Residence

## for

## NewSouth

# **BellSouth Standard Interconnection Agreement**

Agreement Effective Date:	Agreement Expiration Date:
Account Manager:	Account Manager Tel No:

Attachment	Section No.	Version	Planned Activities
Name		Date	
Terms/Conditions PartA	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		
	13		
	14		
	15		
	16		
	17		
	18		
	19		
	20		
	21		
	22		
	23		
	24		
	25		
	26		
Terms/Conditions Part B			

## for

# NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment	Section No.	Version	Planned Activities
Name		Date	
1-Resale	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		
	11		
	12		
	13		
	Exhibit A		
	Exhibit B		
	Exhibit C		
	Exhibit D		
	Exhibit E		
	Exhibit F		
	Exhibit G		
	Exhibit H		
2-Network Elements & Other Services	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		

Attachment 10-Business

## for

# NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment	Section No.	Version	Planned Activities
Name		Date	
	10		
	11		
	12		
	13		
	14		
	15		
	16		
	17		
	Exhibit A		
	Exhibit B		
	Exhibit C		
3-Local Interconnection	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	Exhibit A		
4-Physical Collocation	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	9		
	10		

## for

# NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment	Section No.	Version	Planned Activities
Name		Date	
	11		
	12		
	13		
	14		
	Exhibit A		
	Exhibit B		
5-Access to Numbers &			
Number Portability	1		
	2		
	3		
	4		
	5		
	6		
	7		
	8		
	Exhibit A		
6-Ordering/Provisioning	1		
	2		
	3		
7-Billing & Billing			
Accuracy Certification	1		
	2		
	3		
	4		
	5		
	6		
	7		
o powyg 1 ty p t	Exhibit A		
8-ROW/Conduits/PoleAtt	1		
9-Perf Measurement	Pre-Ordering		
	Ordering		10.7

Attachment 10-Business

## for

# NewSouth

# **BellSouth Standard Interconnection Agreement**

Attachment	Section No.	Version	Planned Activities
Name		Date	
	Provisioning		
	Maint/Repair		
	Billing		
	Opr Svcs/DA		
	E911		
	Trunk Grp Perf		
	Collocation		
	Appendix A		
	Appendix B		
	Appendix C		

# Attachment 11 BellSouth Disaster Recovery Plan

# 2000 BELLSOUTH

# DISASTER RECOVERY PLANNING

For

**CLECS** 

#### **CONTENTS PAGE** 1.0 Purpose 4 2.0 Single Point of Contact 4 3.0 Identifying the Problem 4 3.1 Site Control 5 3.2 **Environmental Concerns** 6 4.0 The Emergency Control Center (ECC) 6 5.0 Recovery Procedures 5.1 CLEC Outage 7 7 5.2 BellSouth Outage 5.2.1 Loss of Central Office 8 5.2.2 Loss of a Central Office with Serving Wire Center Functions 8 8 5.2.3 Loss of a Central Office with Tandem Functions 5.2.4 Loss of a Facility Hub 9 5.3 Combined Outage (CLEC and BellSouth Equipment 9 6.0 T1 Identification Procedures 9 10 7.0 Acronyms

### 1.0 PURPOSE

In the unlikely event of a disaster occurring that affects BellSouth's long-term ability to deliver traffic to a Competitive Local Exchange Carrier (CLEC), general procedures have been developed to hasten the recovery process. Since each location is different and could be affected by an assortment of potential problems, a detailed recovery plan is impractical. However, in the process of reviewing recovery activities for specific locations, some basic procedures emerge that appear to be common in most cases.

These general procedures should apply to any disaster that affects the delivery of traffic for an extended time period. Each CLEC will be given the same consideration during an outage and service will be restored as quickly as possible.

This document will cover the basic recovery procedures that would apply to every CLEC.

### 2.0 SINGLE POINT OF CONTACT

When a problem is experienced, regardless of the severity, the BellSouth Network Management Center (NMC) will observe traffic anomalies and begin monitoring the situation. Controls will be appropriately applied to insure the sanity of BellSouth's network; and, in the event that a switch or facility node is lost, the NMC will attempt to circumvent the failure using available reroutes.

BellSouth's NMC will remain in control of the restoration efforts until the problem has been identified as being a long-term outage. At that time, the NMC will contact BellSouth's Emergency Control Center (ECC) and relinquish control of the recovery efforts. Even though the ECC may take charge of the situation, the NMC will continue to monitor the circumstances and restore traffic as soon as damaged network elements are revitalized.

The telephone number for the BellSouth Network Management Center in Atlanta, as published in Telcordia's National Network Management Directory, is 404-321-2516.

## 3.0 IDENTIFYING THE PROBLEM

During the early stages of problem detection, the NMC will be able to tell which CLECs are affected by the catastrophe. Further analysis and/or first hand observation will determine if the disaster has affected CLEC equipment only; BellSouth equipment only or a combination. The initial restoration activity will be largely determined by the equipment that is affected.

Once the nature of the disaster is determined and after verifying the cause of the problem, the NMC will initiate reroutes and/or transfers that are jointly agreed upon by the affected CLECs' Network Management Center and the BellSouth NMC. The type and percentage of controls used will depend upon available network capacity. Controls necessary to stabilize the situation will be invoked and the NMC will attempt to re-establish as much traffic as possible.

For long term outages, recovery efforts will be coordinated by the Emergency Control Center (ECC). Traffic controls will continue to be applied by the NMC until facilities are re-established. As equipment is made available for service, the ECC will instruct the NMC to begin removing the controls and allow traffic to resume.

## 3.1 SITE CONTROL

In the total loss of building use scenario, what likely exists will be a smoking pile of rubble. This rubble will contain many components that could be dangerous. It could also contain any personnel on the premises at the time of the disaster. For these reasons, the local fire marshal with the assistance of the police will control the site until the building is no longer a threat to surrounding properties and the companies have secured the site from the general public.

During this time, the majority owner of the building should be arranging for a demolition contractor to mobilize to the site with the primary objective of reaching the cable entrance facility for a damage assessment. The results of this assessment would then dictate immediate plans for restoration, both short term and permanent.

In a less catastrophic event, i.e., the building is still standing and the cable entrance facility is usable, the situation is more complex. The site will initially be controlled by local authorities until the threat to adjacent property has diminished. Once the site is returned to the control of the companies, the following events should occur.

An initial assessment of the main building infrastructure systems (mechanical, electrical, fire & life safety, elevators, and others) will establish building needs. Once these needs are determined, the majority owner should lead the building restoration efforts. There may be situations where the site will not be totally restored within the confines of the building. The companies must individually determine their needs and jointly assess the cost of permanent restoration to determine the overall plan of action.

Multiple restoration trailers from each company will result in the need for designated space and installation order. This layout and control is required to maximize the amount of restoration equipment that can be placed at the site, and the priority of placements.

Care must be taken in this planning to insure other restoration efforts have logistical access to the building. Major components of telephone and building equipment will need to be removed and replaced. A priority for this equipment must also be jointly established to facilitate overall site restoration. (Example: If the AC switchgear has sustained damage, this would be of the highest priority in order to regain power, lighting, and HVAC throughout the building.)

If the site will not accommodate the required restoration equipment, the companies would then need to quickly arrange with local authorities for street closures, rights of way or other possible options available.

## 3.2 ENVIRONMENTAL CONCERNS

In the worse case scenario, many environmental concerns must be addressed. Along with the police and fire marshal, the state environmental protection department will be on site to monitor the situation.

Items to be concerned with in a large central office building could include:

- 1. Emergency engine fuel supply. Damage to the standby equipment and the fuel handling equipment could have created "spill" conditions that have to be handled within state and federal regulations.
- 2. Asbestos containing materials that may be spread throughout the wreckage. Asbestos could be in many components of building, electrical, mechanical, outside plant distribution, and telephone systems.
- 3. Lead and acid. These materials could be present in potentially large quantities depending upon the extent of damage to the power room.
- 4. Mercury and other regulated compounds resident in telephone equipment.
- 5. Other compounds produced by the fire or heat.

Once a total loss event occurs at a large site, local authorities will control immediate clean up (water placed on the wreckage by the fire department) and site access.

At some point, the companies will become involved with local authorities in the overall planning associated with site clean up and restoration. Depending on the clean up approach taken, delays in the restoration of several hours to several days may occur.

In a less severe disaster, items listed above are more defined and can be addressed individually depending on the damage.

In each case, the majority owner should coordinate building and environmental restoration as well as maintain proper planning and site control.

## 4.0 THE EMERGENCY CONTROL CENTER (ECC)

The ECC is located in the Colonnade Building in Birmingham, Alabama. During an emergency, the ECC staff will convene a group of pre-selected experts to inventory the damage and initiate corrective actions. These experts have regional access to BellSouth's personnel and equipment and will assume control of the restoration activity anywhere in the nine-state area.

In the past, the ECC has been involve with restoration activities resulting from hurricanes, ice storms and floods. They have demonstrated their capabilities during these calamities as well as

during outages caused by human error or equipment failures. This group has an excellent record of restoring service as quickly as possible.

During a major disaster, the ECC may move emergency equipment to the affected location, direct recovery efforts of local personnel and coordinate service restoration activities with the CLECs. The ECC will attempt to restore service as quickly as possible using whatever means is available; leaving permanent solutions, such as the replacement of damaged buildings or equipment, for local personnel to administer.

Part of the ECC's responsibility, after temporary equipment is in place, is to support the NMC efforts to return service to the CLECs. Once service has been restored, the ECC will return control of the network to normal operational organizations. Any long-term changes required after service is restored will be made in an orderly fashion and will be conducted as normal activity.

## 5.0 RECOVERY PROCEDURES

The nature and severity of any disaster will influence the recovery procedures. One crucial factor in determining how BellSouth will proceed with restoration is whether or not BellSouth's equipment is incapacitated. Regardless of who's equipment is out of service, BellSouth will move as quickly as possible to aid with service recovery; however, the approach that will be taken may differ depending upon the location of the problem.

### 5.1 CLEC OUTAGE

For a problem limited to one CLEC (or a building with multiple CLECs), BellSouth has several options available for restoring service quickly. For those CLECs that have agreements with other CLECs, BellSouth can immediately start directing traffic to a provisional CLEC for completion. This alternative is dependent upon BellSouth having concurrence from the affected CLECs.

Whether or not the affected CLECs have requested a traffic transfer to another CLEC will not impact BellSouth's resolve to re-establish traffic to the original destination as quickly as possible.

## **5.2 BELLSOUTH OUTAGE**

Because BellSouth's equipment has varying degrees of impact on the service provided to the CLECs, restoring service from damaged BellSouth equipment is different. The outage will probably impact a number of Carriers simultaneously. However, the ECC will be able to initiate immediate actions to correct the problem.

A disaster involving any of BellSouth's equipment locations could impact the CLECs, some more than others. A disaster at a Central Office (CO) would only impact the delivery of traffic to and from that one location, but the incident could affect many Carriers. If the Central Office is a Serving Wire Center (SWC), then traffic from the entire area to those Carriers served from that switch would also be impacted. If the switch functions as an Access Tandem, or there is a tandem in the building, traffic from every CO to every CLEC could be interrupted. A disaster that destroys a facility hub could disrupt various traffic flows, even though the switching equipment may be unaffected.

The NMC would be the first group to observe a problem involving BellSouth's equipment. Shortly after a disaster, the NMC will begin applying controls and finding re-routes for the

completion of as much traffic as possible. These reroutes may involve delivering traffic to alternate Carriers upon receiving approval from the CLECs involved. In some cases, changes in translations will be required. If the outage is caused by the destruction of equipment, then the ECC will assume control of the restoration.

## 5.2.1 Loss of a Central Office

When BellSouth loses a Central Office, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Begin restoring service to CLECs and other customers.

## 5.2.2 Loss of a Central Office with Serving Wire Center Functions

The loss of a Central Office that also serves as a Serving Wire Center (SWC) will be restored as described in section 5.2.1.

## 5.2.3 Loss of a Central Office with Tandem Functions

When BellSouth loses a Central Office building that serves as an Access Tandem and as a SWC, the ECC will

- a) Place specialists and emergency equipment on notice;
- b) Inventory the damage to determine what equipment and/or functions are lost;
- c) Move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) Begin reconnecting service for Hospitals, Police and other emergency agencies;
- e) Re-direct as much traffic as possible to the alternate access tandem (if available) for delivery to those CLECs utilizing a different location as a SWC;
- f) Begin aggregating traffic to a location near the damaged building. From this location, begin re-establishing trunk groups to the CLECs for the delivery of traffic normally found on the direct trunk groups. (This aggregation point may be the alternate access tandem location or another CO on a primary facility route.)
- g) Begin restoring service to CLECs and other customers.

## 5.2.4 Loss of a Facility Hub

In the event that BellSouth loses a facility hub, the recovery process is much the same as above. Once the NMC has observed the problem and administered the appropriate controls, the ECC will assume authority for the repairs. The recovery effort will include

- a) Placing specialists and emergency equipment on notice;
- b) Inventorying the damage to determine what equipment and/or functions are lost;
- c) Moving containerized emergency equipment to the stricken area, if necessary;
- d) Reconnecting service for Hospitals, Police and other emergency agencies; and
- e) Restoring service to CLECs and other customers. If necessary, BellSouth will aggregate the traffic at another location and build temporary facilities. This alternative would be viable for a location that is destroyed and building repairs are required.

## 5.3 COMBINED OUTAGE (CLEC AND BELLSOUTH EQUIPMENT)

In some instances, a disaster may impact BellSouth's equipment as well as the CLECs'. This situation will be handled in much the same way as described in section 5.2.3. Since BellSouth and the CLECs will be utilizing temporary equipment, close coordination will be required.

## 6.0 T1 IDENTIFICATION PROCEDURES

During the restoration of service after a disaster, BellSouth may be forced to aggregate traffic for delivery to a CLEC. During this process, T1 traffic may be consolidated onto DS3s and may become unidentifiable to the Carrier. Because resources will be limited, BellSouth may be forced to "package" this traffic entirely differently then normally received by the CLECs. Therefore, a method for identifying the T1 traffic on the DS3s and providing the information to the Carriers is required.

## 7.0 ACRONYMS

CO - Central Office (BellSouth)

DS3 - Facility that carries 28 T1s (672 circuits)

ECC - Emergency Control Center (BellSouth)

CLEC - Competitive Local Exchange Carrier

NMC - Network Management Center

SWC - Serving Wire Center (BellSouth switch)

T1 - Facility that carries 24 circuits

## **Hurricane Information**

During a hurricane, BellSouth will make every effort to keep CLECs updated on the status of our network. Information centers will be set up throughout BellSouth Telecommunications. These centers are not intended to be used for escalations, but rather to keep the CLEC informed of network related issues, area damages and dispatch conditions, etc.

Hurricane-related information can also be found on line at <a href="http://www.interconnection.bellsouth.com/network/disaster/dis\_resp.htm">http://www.interconnection.bellsouth.com/network/disaster/dis\_resp.htm</a>. Information concerning Mechanized Disaster Reports can also be found at this website by clicking on CURRENT MDR REPORTS or by going directly to <a href="http://www.interconnection.bellsouth.com/network/disaster/mdrs.htm">http://www.interconnection.bellsouth.com/network/disaster/mdrs.htm</a>.

## **BST Disaster Management Plan**

BellSouth maintenance centers have geographical and redundant communication capabilities. In the event of a disaster removing any maintenance center from service another geographical center would assume maintenance responsibilities. The contact numbers will not change and the transfer will be transparent to the CLEC.

# **Attachment 12**

Bona Fide Request and New Business Requests Process

## **BONA FIDE REQUEST AND NEW BUSINESS REQUESTS PROCESS**

- 1.0 The Parties agree that NewSouth is entitled to order any Network Element, Interconnection option, service option or Resale Service required to be made available by the Communications Act of 1934, as modified by the Telecommunications Act of 1996 (the "Act"), FCC requirements or State Commission requirements. NewSouth also shall be permitted to request the development of new or revised facilities or service options, which are not required by the Act. Procedures applicable to requesting the addition of such facilities or service options are specified in this Attachment 12.
- 2.0 Bona Fide Requests ("BFR") are to be used when NewSouth makes a request of BellSouth to provide a new or modified network element, interconnection option, or other service option pursuant to the Act that was not previously included in the Agreement. New Business Requests ("NBRs") are to be used when NewSouth makes a request of BellSouth to provide a new or custom capability or function to meet NewSouth's business needs that was not previously included in the Agreement. The BFR/NBR process is intended to facilitate the two-way exchange of information between NewSouth and BellSouth, necessary for accurate processing of requests in a consistent and timely fashion.
- 3.0 A BFR/NBR shall be submitted in writing by NewSouth and shall specifically identify the required service date, technical requirements, space requirements and/or such specifications that clearly define the request such that BellSouth has sufficient information to analyze and prepare a response. Such a request also shall include a NewSouth's designation of the request as being (i) pursuant to the Telecommunications Act of 1996 (i.e. a "BFR") or (ii) pursuant to the needs of the business (i.e. a "NBR"). The request shall be sent to NewSouth's Account Executive.
- 4.0 NewSouth may cancel a BFR or NBR at any time. If NewSouth cancels the request more than three (3) business days after submitting it, NewSouth shall pay BellSouth's reasonable and demonstrable costs of processing and/or implementing the BFR or NBR up to the date of cancellation. If NewSouth does not cancel a BFR or NBR, NewSouth shall pay BellSouth's reasonable and demonstrable costs of processing and implementing the request.
- 5.0 Within fifteen (15) business days of its receipt of a BFR or NBR from NewSouth, BellSouth shall respond to NewSouth by providing a preliminary analysis of such Interconnection, Network Element, or other facility or service option that is the subject of the BFR or NBR. The preliminary analysis shall confirm that BellSouth will either offer access to the Interconnection, Network Element, or other facility or service

option, or provide an explanation of why it is not technically feasible and/or why the request does not qualify as an Interconnection, Network Element, or is not otherwise required to be provided under the Act.

- 6.0 If BellSouth determines that the Interconnection, Network Element, or other facility or service option that is the subject of the BFR is technically feasible, BellSouth shall propose a firm price and a detailed implementation plan within forty (40) business days after receipt of the BFR. BellSouth may, but shall not be required, to provide a firm time and cost proposal for a NBR.
- 7.0 Within thirty (30) business days after its receipt of (i) a refusal of BellSouth to provide a BFR or NBR price quote, or (ii) the BFR or NBR price quote and implementation plan from BellSouth, NewSouth must either confirm or cancel its order for such facility or service option. If it believes such quote is not consistent with the requirements of the Act, NewSouth may at that time seek FCC or state Commission arbitration of its request, as appropriate. Any such arbitration applicable to Network Elements and/or Interconnection shall be conducted in accordance with standards prescribed in Section 252 of the Act.
- Unless NewSouth agrees otherwise, all prices shall be consistent with the pricing principles of the Act, FCC and/or the State Commission.
- 9.0 If either Party to a BFR or NBR believes that the other Party is not requesting, negotiating, or processing the Bona Fide Request in good faith, or disputes a determination, or price or cost quote, such Party may seek FCC or state Commission resolution of the dispute, as appropriate.
- 10.0 Upon agreement to the terms of a BFR or NBR, an amendment to the Agreement may be required.

# AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications Corp ("NewSouth") a Delaware corporation and shall be deemed effective as of the date of the last signature of both Parties ("Effective Date").

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and NewSouth dated May 18, 2001 (the "Interconnection Agreement");

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and NewSouth hereby convenant and agree as follows:

- 1. The Parties agree to include the following Sections in Attachment 2 of the Agreement.
  - 3.1.3.1 Unbundled Local Switching, together with Common Transport and, if necessary, Tandem Switching, provides to NewSouth local subscribers local calling and the ability to presubscribe to a primary carrier for intraLATA toll service and a primary carrier for interLATA toll service.
  - 3.1.3.2 Provided that NewSouth purchases unbundled local switching from BellSouth and uses the BellSouth CIC for its end users' LPIC or if a BellSouth local end user selects BellSouth as its LPIC, then the Parties will consider as local any calls originated by an NewSouth local end user, or originated by a BellSouth local end user and terminated to an NewSouth local end user. where such calls originate and terminate in the same LATA, except for those calls originated and terminated through switched access arrangements (i.e., calls that are transported by a party other than BellSouth). For such calls, BellSouth will charge NewSouth the UNE elements for the BellSouth facilities utilized. Neither Party shall bill the other originating or terminating switched access charges for such calls. Intercarrier compensation for local calls between BellSouth and NewSouth shall be as described in BellSouth's UNE Local Call Flows set forth on BellSouth's web site.

- 3.1.3.3 Where NewSouth purchases unbundled local switching from BellSouth but does not use the BellSouth CIC for its end users' LPIC, BellSouth will consider as local those direct dialed telephone calls that originate from an NewSouth end user and terminate within the basic local calling area or within the extended local calling areas and that are dialed using 7 or 10 digits as defined and specified in Section A3 of BellSouth's General Subscriber Services Tariffs. For such local calls, BellSouth will charge NewSouth the UNE elements for the BellSouth facilities utilized. Intercarrier compensation for local calls between BellSouth and NewSouth shall be as described in BellSouth's UNE Local Call Flows set forth on BellSouth's web site.
- 3.1.3.4 For any calls that originate and terminate through switched access arrangements (i.e., calls that are transported by a party other than BellSouth), BellSouth shall bill NewSouth the UNE elements for the BellSouth facilities utilized. Each Party may bill the toll provider originating or terminating switched access charges, as appropriate.
- 3.1.3.5 Reverse billed toll calls, such as intraLATA 800 calls, calling card calls and third party billed calls, where BellSouth is the carrier shall also be considered as local calls and NewSouth shall not bill BellSouth originating or terminating switched access for such calls.
- 3.1.3.6 BellSouth shall assess retroactive charges for UNE transport and switching associated with using the BellSouth LPIC if a CLEC has been able to previously select BellSouth as the end user LPIC prior to the option allowing the selection of a BellSouth provided LATA-wide local calling area being offered.
- 5. The Parties agree that all of the other provisions of the Interconnection Agreement, dated May 18, 2001 shall remain in full force and effect.
- 6. For electronic filing purposes in the State of Louisiana, the CLEC Louisiana Certification Number is required and must be provided by NewSouth prior to filing of the Agreement. The CLEC Louisiana Certification Number for NewSouth is TSP00231.
- 7. Either or both of the Parties is authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.

BellSouth Telecommunications, Inc.	NewSouth Communications Corp
Signature	Signature
Name	Name
Title	Title
Date	Date

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated

below.

# AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications Corp ("NewSouth") a Delaware corporation.

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and NewSouth dated May 18, 2001 (the "Interconnection Agreement") in order to incorporate rates established by the Tennessee Regulatory Authority ("TRA") in Docket Number 00-00544, on September 26, 2000 and November 7, 2000;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and NewSouth hereby convenant and agree as follows:

- 1. Those interim rates established by the TRA in Docket No. 00-00544 for certain Unbundled Network Elements in Tennessee are as set forth in Exhibit 1-TN-Int attached hereto and incorporated herein by this reference.
- 2. To the extent that any rate element set forth in Exhibit 1-TN-Int corresponds to a rate element set forth in the Interconnection Agreement, such rate element in the Interconnection Agreement is hereby deleted and replaced with the corresponding rate element in Exhibit 1-TN-Int. These rates shall be subject to retroactive true-up once permanent rates for such products and services have been established by the TRA.
- 3. To the extent that the existing Interconnection Agreement does not contain terms and conditions for such products and services, then prior to NewSouth's ordering any such elements pursuant to this Amendment, NewSouth and BellSouth shall amend the existing Interconnection Agreement to incorporate such terms and conditions.
- 4. Any rate element in the Interconnection Agreement that is not expressly replaced by the rates set forth in Exhibit 1-TN-Int as described in paragraphs 2 and 3 above shall remain in full force and effect in accordance with the terms of the Interconnection Agreement.
- 5. The Parties agree that all of the other provisions of the Interconnection Agreement, dated July 14, 2000 shall remain in full force and effect.
- 6. The Parties further agree that either or both of the Parties is authorized to submit this Amendment to the Tennessee Regulatory Authority or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject to Section 252(e) of the federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated below.

BellSouth Telecommunications, Inc.	NewSouth Communications Corp
Signature	Signature
C.W. BOCTZ Name	Jake E. Jennings Name
MANASING DIRECTOR	Vice President Regulatory Affairs <b>Title</b>
<i>6-20-01</i> Date	June 19, 2001  Date

				<del></del>											OSS RATES			
					<u> </u>		KATES				Incremental Incremental Incremental Incremental							
		. 1									Svc Order	Svc Order	Charge -	Charge -	Charge -	Charge -		
	UNBUNDLED NETWORK ELEMENT	Interim or Permanent	Zone	BCS	USOC				Nonreci	ırrina	Submitted	Submitted	Manual Svc	Manual Svc	Manual Svc	Manual S		
	ONDONDEED WORK EEEWELVE		20.10							-	Elec	Manually	Order vs.	Order vs. Electronic-	Order vs. Electronic-	Order vs. Electronic		
					L			urring	Discon		per LSR	per LSR SOMAN	Electronic-1st SOMAN	SOMAN	SOMAN	SOMAN		
						Rec	First	Add'l	First	Add'l	SOMEC	SUMAN	SOMAN	SOME	SOMA	JOINE.		
OP MOD	IFICATION										-	<del> </del>				<del></del>		
	Unbundled Loop Modification - Load Coil / Equipment Removal		l	UAL, UHL,										***				
	per 2 Wire pair - short	1		UCL	ULM2L		\$65.40				\$3.50		\$19.99	\$19.99				
	Unbundled Loop Modification - Load Coil / Equipment Removal			UAL, UHL,					ļ									
	per 2 Wire pair - long	I	Į.	UCL	ULM2G	İ	\$710.71	\$23.77			\$3.50		\$19.99	\$19.99				
	Unbundled Loop Modification - Load Coil / Equipment Removal			UAL, UHL,														
	per 4 Wire pair - short	I		UCL	ULM4L		\$65.40				\$3.50		\$19.99	\$19.99		<del></del>		
	Unbundled Loop Modification - Load Coil / Equipment Removal		1	UAL, UHL,							1							
	per 4 Wire pair - long	1	ì	UCL	ULM4G		\$710.71	\$23.77			\$3.50		\$19.99	\$19.99				
_+-	Unbundled Loop Modification - Bridged Tap Removal, per pair			UAL, UHL,										***				
	unloaded	I		UCL	ULMBT		\$65.44				\$3.50	<u> </u>	\$19.99	\$19.99		-		
_+-	umoadka														<b>_</b>	+		
B-LOOPS									-		<del> </del>	<b>—</b> —			<del>                                     </del>			
			L.=		<b>├</b>				<del>                                     </del>		+-	t	<u> </u>					
Sut	-Loop Distribution		<u> </u>	ļ							<del>                                     </del>	<del></del>				1		
	Sub-Loop - Per Building Equipment Room - CLEC Feeder	_	1	HEADI	USBSC		\$313.01					l .	\$20.35	\$10.54	\$13.32			
	Facility Set-Up	I	<b>↓</b>	UEANL	OSBSC		\$313.01					<b></b>	T					
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-				USBSD		\$108.06	\$108.06					\$20.35	\$10.54	\$13.32			
	Up	I	<b>├</b> ──	UEANL	USBR2	\$1.35	\$94.56	\$29.35				1	\$20.35	\$10.54	\$13.32			
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	1		UEANL UEANL	USBR4	\$2.26	\$116.14	\$37.10					\$20.35	\$10.54	\$13.32			
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	I		UEANL	USBR4	32.20	\$110.14	ψ37.10			-					Ι		
			<del> </del>	-	<del>                                     </del>										L			
Un	oundled Network Terminating Wire (UNTW)	<u> </u>		UENTW	UENPP	\$0.4555	\$2.48	\$2.48						I	L			
	Unbundled Network Terminating Wire (UNTW) per Pair		+	CENTIN	+ <del>  </del>	90. 1000					1							
			<del> </del>	<del>                                     </del>	<del>  </del>								L					
OP MAK	LMU - Preordering Without Reservation, per working facility		+	<del> </del>	† — †							1	<u>i</u>					
	queried (Manual).	I			UMKLW		\$100.00	\$100.00				ļ			ļ			
	LMU - Preordering Without Reservation, per spare facility queried		t	· · · · · · · · · · · · · · · · · · ·	†	-					ĺ	l			l			
	(Manual)	1		1	UMKLW		\$100.00	\$100.00							<b>_</b>	<del></del>		
	LMU - Preordering With Reservation, per spare facility queried			1	1	.,									1			
ĺ	(Manual)	I	ı	l .	UMKLP		\$100.00	\$100.00			<u> </u>				<del> </del>	+		
	LMU - Preordering Without Reservation, per working facility							I				1			1	İ		
	queried (Mechanized)	1	1		1 1		\$0.6888	\$0.6888							<b></b>	+		
	LMU - Preordering Without Reservation, per spare facility queried	1	1				1											
	(Mechanized)	1	1	1			\$0.6888	\$0.6888					<del> </del> -		<del> </del>	+		
	LMU - Preordering With Reservation, per spare facility queried										l l	1						
i i	(Mechanized)	I					\$0.6888	\$0.6888						<del>                                     </del>	<b>-</b>			
No	te: Max number of spare facilities per manual LMUSI is 3. Max			1					1		1		1		1			
INO	nber of spare facilities per mechanized LMUSI is 10.	İ		I .	1		<u></u>							<del> </del>				
NE SHAF												<del>-</del>	\$20.35	\$10.54	\$13.32	+		
TE STEEL	2-Wire Analog VG (SL1) for Line Sharing *	P	T	UEANL	UEAL2	\$13.19	\$31.99	\$20.02	<u> </u>			+	\$20.33	\$10.54	413.32	-		
	2 1,2412mg - ()							ļ			<del>                                     </del>		<del>                                      </del>	<del></del>		+		
Lin	ne Sharing Splitter							60.00	6150.00	60.00	\$3.50	+	\$19.99	\$19.99	\$19.99	\$19.9		
	Line Sharing Splitter, per System 96 Line Capacity	I	I		ULSDA	\$100.00	\$150.00	\$0.00	\$150.00 \$150.00	\$0.00 \$0.00	\$3.50	+	\$19.99	\$19.99	\$19.99	\$19.9		
	Line Sharing Splitter, per System 24 Line Capacity	I			ULSDB	\$25.00	\$150.00	\$0.00	\$130.00	30.00	\$3.50	+	\$19.99	\$19.99	\$19.99	\$19.9		
	Line Sharing Splitter - per Line Activation **	I	4		ULSDC	\$0.61	\$40.00	\$21.39	1		- 00.00	<del>                                     </del>	<b>***</b>	1	1	1		
	Line Sharing Splitter - per Subsequent Activity per Line				1,,,,,,,,	ĺ	620.00	\$15.00			\$3.50		\$19.99	\$19.99	\$19.99	\$19.9		
	Rearrangement	I			ULSDS	ļ	\$30.00	\$15.00	<del> </del>	-	35.50	<b>—</b>	1	1	1	1		
		1	——	<b>_</b>	<u> </u>	<b> </b>	1	+	<del>                                     </del>		+	1	<del>                                     </del>	1		$\top$		
No	te: Interim rates are subject to retroactive true-up.  For all loops offered by BellSouth to provide xDSL services, regardles	<u> </u>	<u> </u>	1	11	hall ample Par	upring and no	recurring rates	for this rate ele	ment are ne	manent rates a	is ordered in D	ocket 97-01262	1		$\top$		
	TI M. 11. D. IIC. who a marrida vDCI cominge regardles	s of the Relisc	outh produ	ent for that rate	.i loop rates s	шан арріу. Кес	minik and uo	" will all a successions and a succession and a successio	LUL CILL TRIC CIL					+				

#### Unbundled Network Elements TENNESSEE

				BCS				RATES						RATES		
U	NBUNDLED NETWORK ELEMENT	Interim or Permanent	Zone		USOC				Nonre	curring	Svc Order Submitted Elec	Svc Order Submitted Manually	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Charge - Manual Svo Order vs.
					,	Rec	Nonre First	curring Add'l	Disco First	nnect Add'l	per LSR SOMEC	per LSR SOMAN	Electronic-1st SOMAN	Electronic- SOMAN	Electronic- SOMAN	Electronic- SOMAN
LOOP MODIFICATION					-	Rec	11130	Add I	THSt	Auu	JOINEC	SOMAN	SOMAN	SOMAN	SOMAIN	SOMAIN
	and Madification I and Cail / Equipment Remark			UAL, UHL,			İ				1					
per 2 Wire pa		I		UCL	ULM2L		\$65.40				\$3.50		\$19.99	\$19.99		
Unbundled Lo per 2 Wire pa	oop Modification - Load Coil / Equipment Removal	I		UAL, UHL, UCL	ULM2G		\$710.71	\$23.77			\$3.50		\$19.99	\$19.99		
	oop Modification - Load Coil / Equipment Removal			UAL, UHL,												
per 4 Wire pa		I		UCL	ULM4L		\$65.40				\$3.50		\$19.99	\$19.99		
Unbundled Lo	oop Modification - Load Coil / Equipment Removal			UAL, UHL,				1								
per 4 Wire pa		I		UCL	ULM4G		\$710.71	\$23.77			\$3.50		\$19.99	\$19.99	!	
	pop Modification - Bridged Tap Removal, per pair		1	UAL, UHL,							1					
unloaded		I		UCL	ULMBT		\$65.44				\$3.50		\$19.99	\$19.99		
SUB-LOOPS																
Cub Loss Distribu				-			1				1		<del>                                     </del>		<del> </del>	-
Sub-Loop Distribu	Per Building Equipment Room - CLEC Feeder			<b>-</b>	ŧ		1				-		-			
Facility Set-U		ī		UEANL	USBSC		\$313.01						\$20.35	\$10.54	\$13.32	
	Per Building Equipment Room - Per 25 Pair Panel Set-		-	CEANL	USBSC		\$313.01						\$20.55	\$10.54	\$15.52	-
Up Up	er Building Equipment Room - Fei 23 Fan Fanei Set-	ī	1	UEANL	USBSD		\$108.06	\$108.06					\$20.35	\$10.54	\$13.32	1
	Wire Intrabuilding Network Cable (INC)	<u>†</u>	-	UEANL	USBR2	\$1.35	\$94.56	\$29.35			1		\$20.35	\$10.54	\$13.32	
	Wire Intrabulding Network Cable (INC)	<u> </u>		UEANL	USBR4	\$2.26	\$116.14	\$37.10			1		\$20.35	\$10.54	\$13.32	
Sub-E00p 4-1	wife intrabunding Network Cable (INC)			OLANE	CSDR4	\$2.20	\$110.17	\$57.10			<b> </b>		\$20.55	\$10.54	\$15.52	-
Unbundled Natura	rk Terminating Wire (UNTW)		1				1				1	1				
	etwork Terminating Wire (UNTW) per Pair	ī	<del>                                     </del>	UENTW	UENPP	\$0.4555	\$2.48	\$2,48			1					
Chounteer	etwork Terrimating whe (ONT w) per Tail			CLIVI W	CLIVII	\$0.4333	\$2.70	\$2.40			1					
LOOP MAKE-UP	the state of the s				1		1									
	dering Without Reservation, per working facility	-	<del>                                     </del>													
queried (Man	ual).	1			UMKLW		\$100.00	\$100.00								<u> </u>
LMU - Preord (Manual)	dering Without Reservation, per spare facility queried	I			UMKLW		\$100.00	\$100.00								
LMU - Preore (Manual)	dering With Reservation, per spare facility queried	ı		,	UMKLP		\$100.00	\$100.00								
	dering Without Reservation, per working facility			1			1	-								
queried (Mecl	hanized)	I					\$0.6888	\$0.6888								
(Mechanized)		I					\$0.6888	\$0.6888								
LMU - Preord (Mechanized)	dering With Reservation, per spare facility queried	I					\$0.6888	\$0.6888								
	of spare facilities per manual LMUSI is 3. Max cilities per mechanized LMUSI is 10.															
LINE SHARING	times per mechanized Envior is 10.		<del> </del>	t			<u> </u>		1		t	1			<del> </del>	
	og VG (SL1) for Line Sharing *	P		UEANL	UEAL2	\$13.19	\$31.99	\$20.02					\$20.35	\$10.54	\$13.32	
2-Wile Alialo	g vG (SE1) for Enic Starting			OLITIC	OLILL	Ψ15.17	\$31.55	<b>\$20.02</b>					<b>\$2</b> 0.55	0.0.5	V13.32	
Line Sharing Split									<u> </u>		1		1			1
	Splitter, per System 96 Line Capacity	I	<u> </u>		ULSDA	\$100.00	\$150.00	\$0.00	\$150.00	\$0.00	\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
	Splitter, per System 24 Line Capacity	I	<u> </u>		ULSDB	\$25.00	\$150.00	\$0.00	\$150.00	\$0.00	\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
	Splitter - per Line Activation **	I	<u> </u>		ULSDC	\$0.61	\$40.00	\$21.39			\$3.50	<b>_</b>	\$19.99	\$19.99	\$19.99	\$19.99
Line Sharing Rearrangemen	Splitter - per Subsequent Activity per Line ent	I			ULSDS		\$30.00	\$15.00	1		\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
	M.,				1			<u> </u>								
Note: Interim rates	are subject to retroactive true-up.				1		I									
* For all loops offer	red by BellSouth to provide xDSL services, regardless	of the Bellsou	th produc	t title, these SL	l loop rates si	hall apply. Rec	urring and non	recurring rates	for this rate el	ement are peri	nanent rates as	ordered in Do	cket 97-01262.			
** Recurring rate for	or Line Sharing Splitter per Line Activation is as a res	ult of a regiona	al settleme	nt for that rate	element.		T					T				

# AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications Corp., a Delaware corporation.

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and NewSouth dated May 18, 2001 (the "Interconnection Agreement") in order to incorporate rates established by the Tennessee Regulatory Authority ("TRA") in Docket Number 97-01262, on December 19, 2000, as amended by BellSouth's corrected submissions of January 31, 2001 and February 12, 2001;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and NewSouth hereby convenant and agree as follows:

- 1. Those permanent rates established by the TRA in Docket No. 97-01262 for certain Unbundled Network Elements and Local Interconnection in Tennessee are as set forth in Exhibit 1-TN attached hereto and incorporated herein by this reference.
- 2. To the extent that any product or service set forth in Exhibit 2-TN corresponds to a product or service set forth in the Interconnection Agreement, all rate elements and rates associated with such product or service in the Interconnection Agreement are hereby deleted and replaced with the corresponding rates and rate elements in Exhibit 1-TN.
- 3. Any rate element and rate for products or services in the Interconnection Agreement that is not expressly replaced by the rates and rate elements set forth in Exhibit 1-TN as described in paragraph 2 above shall remain in full force and effect in accordance with the terms of the Interconnection Agreement.
- 4. To the extent NewSouth and BellSouth have not previously negotiated terms and conditions corresponding to any rate element set forth in Exhibit 1-TN, then any Order for such element shall be provisioned in accordance to the terms and conditions set forth in the Competitive Local Exchange Carrier Tariff for the State of Tennessee, incorporated herein by this reference.
- 5. These rates shall be subject to retroactive true-up in accordance with the Agreement. Such true-up shall be retroactive to December 19, 2000.

- 6. The Parties agree that all of the other provisions of the Interconnection Agreement, dated July 14, 2000, shall remain in full force and effect.
- 7. The Parties further agree that either or both of the Parties is authorized to submit this Amendment to the Tennessee Regulatory Authority or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject to Section 252(e) of the federal Telecommunications Act of 1996.
- 8. This Amendment is made effective upon the date that it is signed by both Parties.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated below.

BellSouth Telecommunications, Inc.	NewSouth Communications Corp.
Signature	Joh E. J. Signature
C.W. POUTZ Name	Jake E. Jennings  Name
MANAGING DIRECTOR	Vice President Regulatory Affairs  Title
6-00-0/ Date	June 19, 2001  Date

				Τ	Г				RA1	FES					OSS	RATES		
															Incremental		Incremental	
		UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETWORK ELEMENT AS	Not in TRA Docket	Zone	BCS	USOC				Nonrecurring		Svc Order	Svc Order	Incremental	Charge - Manual Svc		Charge - Manual Svc Order vs.
		UNDUNDED HE WORK ELEMENT	STATED IN DOCKET 97-01262						Nonrecurring				Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-1st	Electronic-	Svc Order vs. Electronic-Disc 1st	
								Rec	First	Add'l	First	Disconnect First Add'l		SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				-				THE THE	, 1100				SOMEC					
ODUI	/EDOUF/G	CMDS/CNAM-Resale								-								
<u> </u>	OPTIONA	AL DAILY USAGE FILE (ODUF)															ļ	
	-		OSS OLEC Daily Usage File Recording per message				N/A	\$0.0000044										
	ODL		OSS OLEC Daily Usage File Message Distribution per message				N/A	\$0.0027366										
	ODU	UF: Message Processing, per Magnetic Tape	OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned				N/A	\$52.75										
	ODL	UF: Data Transmission (CONNECT:DIRECT),	OSS OLEC Daily Usage File Data				N/A	\$0.0000339										
				<del> </del>	<del>  -</del> -	_	<b>├</b> ──-			+		+			<del>  -</del>	1	t	

		UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262		П				RATES					oss	RATES	Incremental		
	UNBUNDLED NETWORK ELEMENT		Not in TRA Docket 97- 01262		BCS	usoc				Nonrec		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Charge - Manuel Svc of Order vs. Electronic-Disc	Incrementa Charge - Manual Svo Order vs. Electronic-Di Add'i
				ı		- 1		Nonrec	urring Add'l	Pirst	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
							Rec	First	Addi	PIFST	Addi	SOMEC	30,000				
	Geographically Deaveraged UNE Zones and											1			1	i	
ap	plicable rates have been established for certain					ļ							1				1
	services, as shown in this Agreement. Where		İ											İ			İ
1	Geographically Deaveraged UNE Zones and									i							
app	licable rates are established, Statewide rates are		1						'					1		i	
0	bsolete. Further, BellSouth is in the process of												,				İ
enh	nancing its billing systems in order to accomodate															l .	
th	nis Geographically Deaveraged UNE Zone Rate			'				i '							1		
	Structure. Until these enhancements are											İ			i	ļ	
acc	complished, the UNE Zone 1 rate will be billed for														ļ		
	services residing in Zones 1, 2, 3 or 4, i.e., once			ĺ								1			1	1	1
bi	Illing enhancements are complete, all applicable NE Zone rates reflected in this Agreement will be															Į.	İ
U	billed. Reference Internet Website													1		1	
L	pilled. Hererence internet website p://www.interconnection.bellsouth.com/become_cl												1	1		I	
nttp	ec/ docs/interconnection/deavuzns.pdf to view			1				[								I	1
God	ographically Deaveraged UNE Zone Designations				Ì					i i			1	1		1	
Get	by Central Office.												1			L	
	by Central Cinco.		<del></del>	<u> </u>			,	<del>                                     </del>	<del> </del>			-					
										L		<b>├</b>		+			
BUNE	DLED EXCHANGE ACCESS LOOP				i							ļ		<b>_</b>		·	+
DOM	DEED EXCURITE ACCESS 200.											1		<u> </u>	<u> </u>	<u> </u>	
_					<u> </u>								1				
2-V	VIRE ANALOG VOICE GRADE LOOP	and A Landa Canda Lana	<del></del>	₩-				<del>                                     </del>		-			1				
	2-Wire Analog Voice Grade Loop - Service	2-Wire Analog Voice Grade Loop	'	1	UEANL	UEAL2	\$13.19	\$31.99	\$20.02	\$10.65	\$1.41	1	1	\$20.35	\$10.54	\$13.32	
-	LCVCI I LONG I	Service Level 1 - Zone 1		+-	UEAINL	UEALZ	\$13.19	Ψ01.33	ΨE0.02	4.0.00	****	<b>-</b>					
		2-Wire Analog Voice Grade Loop	'	1 2	UEANL	UEAL2	\$17.23	\$31.99	\$20.02	\$10.65	\$1.41	l .		\$20.35	\$10.54	\$13.32	
ı		Service Level 1 - Zone 2	<del> </del>	12	UEANL	UEALZ	\$17.23	401.99	Ψ20.02	<b>\$10.00</b>	<b>V</b>		<u> </u>		-	1	1
	2-Wire Analog Voice Grade Loop - Service	2-Wire Analog Voice Grade Loop	•	١.	1	l.,_,,	*00.50	\$31.99	\$20.02	\$10.65	\$1.41	1	l .	\$20.35	\$10.54	\$13.32	
İ	Level 1- Zone 3	Service Level 1 - Zone 3	ļ	3	UEANL	UEAL2	\$22.53	\$31.99	\$20.02	\$10.00	Ψ11	<b></b>	<del>                                     </del>	V-V-			
	Manual Order Coordination for UVL-SL1s	2-WAVGL - SL1 - Manual Order		l	I			\$36.52	\$36.52	\$9.18	\$9.18			li .		i	
	(per loop)*	Coordination		<b>↓</b>	UEANL	UEAMC	<u> </u>	\$30.52	\$30.52	\$3.10	Ψ3.10	<del>                                     </del>	<del>                                     </del>	<del> </del> -			
_		2-WAVGL - SL1 - Order			ŀ	Į.	1	ļ					1			l	
i	Order Coordination for Specified Conversion	Coordination for Specified		1	l	l	l	1	204.00	1		ı				ı	
	Time for UVL-SL1 (per LSR) *	Conversion Time			UEANL	OCOSL		\$34.29	\$34.29	<u> </u>	<del> </del>	+		+		<u> </u>	
_	2-Wire Analog Voice Grade Loop - Service			1	ľ	Į.				i i	İ	l				ı	
	Level 2 w/Loop or Ground Start Signaling -	2-Wire Analog Voice Grade Loop	-			l		1	040.00	****	\$17.64	1		\$20.35	\$10.54	\$13.32	
	Zone 1	Service Level 2 - Zone 1		1	UEA	UEAL2	\$16.56	\$75.06	\$48.20	\$28.70	\$17.04		-	Ψ20.00	ψ,σ.σ.	<del>  ,,,,,,</del>	-
+	2-Wire Analog Voice Grade Loop - Service				1			1				k				li .	
	Level 2 w/Loop or Ground Start Signaling -	2-Wire Analog Voice Grade Loop	•			1		l			\$17.64			\$20.35	\$10.54	\$13.32	
1	Zone 2	Service Level 2 - Zone 2		2	UEA	UEAL2	\$21.63	\$75.06	\$48.20	\$28.70	\$17.04	<del></del>		Ψ20.00	Ψ10.0 ·	+ ****	
	2-Wire Analog Voice Grade Loop - Service				1	1	Į.	1				l	1				
ļ	Level 2 w/Loop or Ground Start Signaling -	2-Wire Analog Voice Grade Loop	-	1					1		047.04	1	Į.	\$20.35	\$10.54	\$13.32	
	Zone 3	Service Level 2 - Zone 3		3	UEA	UEAL2	\$28.28	\$75.06	\$48.20	\$28.70	\$17.64			\$20.55	\$10.04	ψ10.0 <u>2</u>	+
+		2-WAVGL - SL2 - Order				1	t	1			İ	i	1			1	
	Order Coordination for Specified Conversion	Coordination for Specified		1	Į.	1	1	I.				1	l l			1	
- Î	Time (per LSR)	Conversion Time			UEA	OCOSL		\$34.29	\$34.29	<del></del>		ļ	<b>_</b>		<del> </del>		+
	2-Wire Analog Voice Grade Loop - Service			$\top$								1		ļ		1	
	Level 2 w/Reverse Battery Signaling - Zone	2-Wire Analog Voice Grade Loop	-	1	1	l	I	1.	1			1	1	\$20,35	\$10.54	\$13.32	
	1	Service Level 2 - Zone 1		1	UEA	UEAR2	\$16.56	\$75.06	\$48.20	\$28.70	\$17.64	<b>_</b>	<del>                                      </del>	\$20.35	\$10.34	\$10.02	+
	2-Wire Analog Voice Grade Loop - Service									1		1	1	1	1	I	
İ	Level 2 w/Reverse Battery Signaling - Zone	2-Wire Analog Voice Grade Loop	-	1	1	1	ŀ		1.	1	1	1	1	\$20.35	\$10.54	\$13.32	
	2	Service Level 2 - Zone 2		2	ŲEA	UEAR2	\$21.63	\$75.06	\$48.20	\$28.70	\$17.64	+	+	\$20.35	\$10.34	\$10.02	+
-+-	2-Wire Analog Voice Grade Loop - Service	<del>                                     </del>										Ī	1	1		I	
	Level 2 w/Reverse Battery Signaling - Zone	2-Wire Analog Voice Grade Loop	-		1	1	1	1		1.		ı	1	***	640.54	\$13.32	
	cever 2 w/neverse battery signating * 2011e	Service Level 2 - Zone 3		3	UEA	UEAR2	\$28.28	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	+
	<u>                                     </u>	2-WAVGL - SL2 - Order		Ť	T	1				1		1	1	1		1	
	Order Coordination for Specified Conversion			1	1	1	1	1		1		ĺ	1	1		1	
	Time (per LSR)	Conversion Time		1	UEA	ocosi	.l	\$34.29	\$34.29	I		1					
		CONVEISION INDO															

							ATES			OSS RATES   Incremental   Incremental							
	UNBUNDLED NETWORK ELEMENT		Not in TRA Docket 97- 01262	Zone	BCS	usoc				Nonrec	•	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st		Charge - Manual Svc Il Order vs. Electronic-Disc	Charge - Manual Sv Order vs. Electronic-D
								Nonrec		Disco	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-							Rec	First	Add'l	First	Addi	SOMEC	30mAit	0011174			
$\neg$															<del></del>		<del>                                     </del>
4-W	RE ANALOG VOICE GRADE LOOP														<del> </del>		<del>                                     </del>
<del>- ''</del>	TIE AUTO-	4-Wire Analog Voice Grade Loop -				l		2400.70	605 57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
, ,		Zone 1		1	UEA	UEAL4	\$24.70	\$122.76	\$85.57	\$70.33	φυ <del>σ</del> . 10			<u> </u>	1		T
$\neg \uparrow$		4-Wire Analog Voice Grade Loop -		ا . ا			\$32.25	\$122.76	\$85.57	\$76.35	\$39.16		1	\$20.35	\$10.54	\$13.32	
		Zone 2		2	UEA	UEAL4	\$32.25	\$122.70	φου.υ/	\$70.00	Ψ00.10			<b></b>			
$\neg$		4-Wire Analog Voice Grade Loop -		3	UEA	UEAL4	\$42.17	\$122.76	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	L
	4-Wire Analog Voice Grade Loop - Zone 3	Zone 3		13	OEA	OLALT	Ψ42.17		<b>V</b>	•						l	
		4-Wire AVGL - Order Coordination for Specified Conversion Time			UEA	OCOSL		\$34.29	\$34.29								
				L		ļ		<b>-</b>						t	<u> </u>		1
2-W	RE ISDN DIGITAL GRADE LOOP			<u> </u>		L	<u></u>	<b></b>			-		<del>                                     </del>	<b>-</b> -			
		2-Wire ISDN Digital Grade Loop -		ا ا	LIDA:	1141.04	£22.00	\$142.76	\$88.88	\$76.35	\$39.16	•	1	\$20.35	\$10.54	\$13.32	l
		Zone 1		₽.	UDN	U1L2X	\$22.00	\$142.70	φου.σο	\$70.00	\$00.10	<del></del>	<b></b>	<u> </u>	1 -	1	
		2-Wire ISDN Digital Grade Loop -		2	UDN	U1L2X	\$29.02	\$142.76	\$88.88	\$76.35	\$39.16		İ	\$20.35	\$10.54	\$13.32	
		Zone 2 2-Wire ISDN Digital Grade Loop -		Ľ	UDIN	0122	Ψ23.0E	₩ 1-12.70	400.00	<b>V</b> 1 5155							ì
	·	Zone 3		3	UDN	U1L2X	\$37.95	\$142.76	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
	2-Wire ISDN Digital Grade Loop - Zone 3	2-Wire ISDN Digital Grade Loop -		<del>ا</del>	UDI.	1	40	1						1			
	Order Coordination For Specified	Order Coordination for Specified			Į.	1		Į.					1		İ		
		Conversion Time		1	UDN	OCOSL		\$34.29	\$34.29				<u> </u>	<u> </u>		<del> </del>	+
_	Conversion Time (per LSh.)	CONTENSION TIME		t									<u> </u>			<b>_</b>	∔
	IRE ASYMMETRICAL DIGITAL SUBSCRIBE	OLINE (ADSL) COMPATIBLE LOC	חר	1		T -		T				l				<b>1</b>	
<u>2-W</u>	IRE ASYMMETRICAL DIGITAL SUBSCRIBE	R LINE (ADSL) COMPATIBLE LOC	<del></del>	+	<del></del>	<del>                                     </del>		1					ŀ	1			
1	2-Wire ADSL Compatible Loop Including Man Svc Inquiry and Facility Reservation -	2-Wire ADSL Compatible Loop -			l	ł		1					1		040.54	610.00	
		Zone 1		1	UAL	UAL2X	\$13.82	\$270.01	\$234.63	\$74.54	\$39.14	<u> </u>	<b>_</b>	\$20.35	\$10.54	\$13.32	<del></del>
	2-Wire ADSL Compatible Loop Including			1						1				1		l	
	Man Svc Inquiry and Facility Reservation -	2-Wire ADSL Compatible Loop -		ı	ļ.	1		1.			****	<u> </u>	1	\$20.35	\$10.54	\$13.32	
		Zone 2		2	UAL	UAL2X	\$18.05	\$270.01	\$234.63	\$74.54	\$39.14	1	<del> </del>	φ20.00	ψ10.01	¥10.0±	+
_	2-Wire ADSL Compatible Loop Including												ı		1	l .	1
	Man Svc Inquiry and Facility Reservation -	2-Wire ADSL Compatible Loop -		١.	l	l	***	\$270.01	\$234.63	\$74.54	\$39.14	ł		\$20.35	\$10.54	\$13.32	
	Zone 3	Zone 3	<u> </u>	3	UAL	UAL2X	\$23.60	\$270.01	\$234.03	\$74.54	Ψ00.14	+	<del>                                     </del>				1 -
		2-Wire ADSL Digital Grade Loop -		1	ŀ	1				ļ		l .				1	1
	Order Coordination for Specified Conversion	Order Coordination for Specified	ì	1	UAL	OCOSL		\$34.29	\$34.29	1	i			<u> </u>		<b>_</b>	
	Time	Conversion Time		+	UAL.	10000										L	
		I WE (IDOL) COMPATIBLE LOO		+-	<b>├</b> ─	+	+	+	<del>                                      </del>								
2-14	IRE HIGH BIT RATE DIGITAL SUBSCRIBER	LINE (HDSL) COMPATIBLE LOO		+	<del>                                     </del>	+	<u> </u>	<del>                                     </del>	-	<u> </u>					1	1	
	2-Wire HDSL Compatible Loop including	2-Wire High Bit Rate DSL		ı	1	1	l .	1				1				1	
	manual service inquiry and facility	Compatible Loop - Zone 1		1 1	UHL	UHL2X	\$10.83	\$270.01	\$234.63	\$74.54	\$39.14	L		\$20.35	\$10.54	\$13.32	
<del> </del> −	reservation - Zone 1 2-Wire HDSL Compatible Loop Including	Compandic Ecop Zone :		$\top$	1							Į.	1	1		1	
	manual service inquiry and facility	2-Wire High Bit Rate DSL		1		1		1		1	1	1		\$20.35	\$10.54	\$13.32	ì
	reservation - Zone 2	Compatible Loop - Zone 2		2	UHL	UHL2X	\$14.15	\$270.01	\$234.63	\$74.54	\$39.14		<del></del>	\$20.35	\$10.54	\$13.52	+
$\vdash$	2-Wire HDSL Compatible Loop Including						1	1				1	1	1			
	manual service inquiry and facility	2-Wire High Bit Rate DSL		1		1		1 6070 04	6004.60	\$74.54	\$39.14		1	\$20.35	\$10.54	\$13.32	
	reservation - Zone 3	Compatible Loop - Zone 3	<u> </u>	3	UHL	UHL2X	\$18.50	\$270.01	\$234.63	₹7/4.54	φ39.14	+	+	1 +	7.5.5.	1	
1		2-Wire HDSL Loop - Order	!	1	1	1	1			1			1	1		1	1
1	Order Coordination for Specified Conversion	Coordination for Specified		1	UHL	ocosi		\$34.29	\$34.29	i			1				
	Time	Conversion Time		+	1 Uni	10008	╁───	¥57.25	\$37.E3		†		1				
		L		<del> </del> —	1	+-		+		<del>                                     </del>		1	<b>1</b>				
4-1	VIRE HIGH BIT RATE DIGITAL SUBSCRIBE	R LINE (HDSL) COMPATIBLE LOC	P	-	<del> </del>	1	+	+	<del> </del>	<del> </del>	·	<b>+</b>					
	4-Wire HDSL Compatible Loop Including	t .	1	1	1		ı	1	1	1				ì		l	
T	manual service inquiry and facility	4-Wire HDSL Compatible Loop -	1							L	1				\$10.54	\$13.32	

														000	RATES		
									RATES			<u> </u>		OSS	HATES	Incremental	Incremental
					ì								l			Charge - Manual Svc	Charge - Manual Svc
	AND THE STATE OF STAT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-	Zone	BCS	usoc				Maara	umina	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	incremental Charge - Manual	Order vs.	Order vs.
ĺ	UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	01262	201.0						Nonrecurring		Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Dis
				l 1	1	l.		Nonrec		Disco		per LSR	LSR	Electronic-1st SOMAN	Electronic-Add'l	1st SOMAN	Add'I SOMAN
				Ш			Rec	First	Add'i	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAIN	JOMAN
_	4-Wire HDSL Compatible Loop including				ŀ												
	manual service inquiry and facility	4-Wire HDSL Compatible Loop -		ا ۽ ا	1		040.00	\$279.60	\$244.22	\$74.54	\$39.14	l .		\$20.35	\$10.54	\$13.32	
	I ESEI VALIOI ZOIC E	Zone 2		2	UHL	UHL4X	\$18.20	\$279.60	\$244.22	\$74.54	φυσ. 14_	<b>-</b>	<b>-</b>			`	
	4-Wire HDSL Compatible Loop Including			ll	l l			ŀ	ļ						i	l.	
		4-Wire HDSL Compatible Loop -		l 3 l	UHL	UHL4X	\$23.80	\$279.60	\$244.22	\$74.54	\$39.14		1	\$20.35	\$10.54	\$13.32	
		Zone 3 4-Wire HDSL Loop Order		"	OTIL	OTILAX	ΨΕ0.00	1 12.5.55	<u> </u>							1	
	Order Coordination for Specified Conversion	Coordination for Specified				1											
	Time	Conversion Time			UHL	OCOSL		\$34.29	\$34.29					<u> </u>	ļ		<del></del>
-	Time	CONTROLOGIC TIME		1 1											Ļ	ļ	
				$\vdash$													<u> </u>
4-1	WIRE DS1 DIGITAL LOOP			+	-									1			
		A ME - DOA DI-HALL 7		ا ، ا	USL	USLXX	\$57.73	\$313.08	\$219.72	\$96.86	\$40.45	1	ļ	\$18.98	\$8.43	\$11.95	\$0.00
$\perp$	4-Wire DS1 Digital Loop - Zone 1	4-Wire DS1 Digital Loop - Zone 1		┼┼	USL	JOLAN	ψ01.70	\$5.0.00	¥=.5.72	1							
		l		ا ؍ ا	USL	USLXX	\$75.40	\$313.08	\$219.72	\$96.86	\$40.45	I		\$18.98	\$8.43	\$11.95	\$0.00
	4-Wire DS1 Digital Loop - Zone 2	4-Wire DS1 Digital Loop - Zone 2		2	USL	USLAA	\$75.40	\$313.00	φ <u>ε</u> 13.72	ψ30.00	Ψ-10. 10	-		<del>                                     </del>			
	BOA BY WALL	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	\$98.59	\$313.08	\$219.72	\$96.86	\$40.45	l		\$18.98	\$8.43	\$11.95	\$0.00
	4-Wire DS1 Digital Loop - Zone 3	4-Wire DS1 Digital Loop - Zone 3		╅			<del></del>	1									
	Order Coordination for Specified Conversion			1				l				1		1	ĺ	-	
	Time	Conversion Time			USL	OCOSL		\$34.59	\$34.59			<u> </u>	<b>↓</b>			<b>_</b>	
	Time	Sometime		T	-							<u> </u>				<b>_</b>	<del></del>
	WEDS OF ON ALKERD DIGITAL CRADE LOOK			1-1				Ī					1				
4-	WIRE 56 OR 64 KBPS DIGITAL GRADE LOOP	4-Wire 56 or 64 Kbps Digital											1		040.54	\$13.32	
	4-Wire 56 Kbps Digital Grade Loop - Zone 1			1	UDL.	UDL56	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18		<u> </u>	\$20.35	\$10.54	\$13.32	+
-	4-14 IIE 30 KDPS Digital Grade Coop 25110 .	4-Wire 56 or 64 Kbps Digital							1	1	*****			\$20.35	\$10.54	\$13.32	i
ĺ	4-Wire 56 Kbps Digital Grade Loop - Zone 2	Grade Loop - Zone 2		2	UDL	UDL56	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18	<b>∔</b> -	<b>-</b>	\$20.55	\$10.54	\$10.0 <u>2</u>	+
_		4-Wire 56 or 64 Kbps Digital			l		050.44	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
	4-Wire 56 Kbps Digital Grade Loop - Zone 3	Grade Loop - Zone 3		3	UDL	UDL56	\$53.11	\$207.01	\$141.30	\$50.70	\$44.10	<del> </del>		1		1	
		4-Wire 56/64 kbps Dig. GL - Order		1				1				1	1	l.		1	
	Order Coordination for Specified Conversion	Conversion Time		1	UDL	OCOSL	ļ.	\$34.29	\$34.29				L				
$\rightarrow$	Time	4-Wire 56 or 64 Kbps Digital		+-	- JPE	10000		1	<u>-</u> -	i -		"					
	4-Wire 64 Kbps Digital Grade Loop - Zone 1			1	UDL	UDL64	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18	<u>L.</u>		\$20.35	\$10.54	\$13.32	<b>-</b>
	4-Wire 64 Kbps Digital Grade Loop - Zone 1	4-Wire 56 or 64 Kbps Digital		十一		"						1	ı	****	\$10.54	\$13.32	
	4-Wire 64 Kbps Digital Grade Loop - Zone 2			2	UDL	UDL64	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18		<b>_</b>	\$20.35	\$10.54	\$13.32	+
-		4-Wire 56 or 64 Kbps Digital		T	T				0444.00	200 70	\$44.18	1		\$20.35	\$10.54	\$13.32	
	4-Wire 64 Kbps Digital Grade Loop - Zone 3	Grade Loop - Zone 3		3	UDL	UDL64	\$53.11	\$207.01	\$141.38	\$90.70	\$44.10	┼──	<del></del>	Ψ20.00		7.5.5	+
		4-Wire 56/64 kbps Dig. GL - Order		1						ļ		I	1	ł		1	
i	Order Coordination for Specified Conversion	Coordination for Specified		1	UDL	OCOSL		\$34.29	\$34.29	1			1	1	I		
	Time	Conversion Time		+-	1000	10000	<u> </u>	1 40	7	1							
			<del></del>	+	<u> </u>	<del> </del>		1	<del>                                     </del>	1							
SUB-L	OOPS		-	+	† — ·	t								<u> </u>		<del>                                      </del>	+
	Sub-Loop Distribution		1					L			<u> </u>	1	4	<b></b>		<del> </del>	+
- 13	Sub-Loop - Per Cross Box Location - CLEC			1						1			l l	\$20.35	\$10.54	\$13.32	
	Feeder Facility Set-Up		*		UEANL	USBSA	<u> </u>	\$517.25	\$517.25	<del> </del>	-	+	+-	\$20.33	\$10.54	ψ10.02	+
$\neg$	Sub-Loop - Per Cross Box Location - Per 25		-			Lugges	i	640.00	640.60	1				\$20.35	\$10.54	\$13.32	
	Pair Panel Set-Up		*	┷	UEANL	USBSB	<del> </del>	\$42.68	\$42.68	<del> </del>	+	+	+	<b>\$20.00</b>	7,0.01	1	$\top$
	Sub-Loop Distribution Per 2-Wire Analog				UEANL	USBN2	\$10.02	\$148.84	\$112.34	\$73.14	\$36.65	1		\$20.35	\$10.54	\$13.32	
	Voice Grade Loop - Statewide	Loop Distribution - per 2-WAVGL	+	sw	DEANL	USBINZ	\$10.02	ψ170.04	ψ. 12.0 <sup>-1</sup>	1	1 230.30	<b>1</b>	1 -				
$\neg \top$	Sub-Loop Distribution - Order Coordination,				UEANI	USBMC	:	\$36.52							1		
	per sub-loop pair	Sub-Loop Distribution - Order	<del>                                     </del>	+-	1	†	-	1	1	1						Í	
- [	Sub Lean Distribution - Order Coordination	Coordination for Specified			I	1									1	1	
	Sub-Loop Distribution - Order Coordination for Specified Time Conversion, per LSR	Conversion Time		1	UEANL	ocosi	.	\$34.29									

<del>-  </del>								RATES					OSS	RATES		Increment
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc				Nonrec		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Charge - Manual St Order va Electronic-I Add'l
						Rec	Nonrec First	urring Add'i	First	Add'i	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			⊢		_	nec	Filat	AUUT	1	7						
Sub-Loop Feeder																
DS0 Set-up per Cross Box location - CLEC				UEA,							i					ĺ
Distribution Facility set-up (for access to				UDN,UC								i	\$20.35	\$10.54	\$13.32	
Feeder)		*	<u> </u>	L,UDL	USBFW		\$517.25	\$517.25					Ψ20.00	<b>410.01</b>	<u> </u>	
			l	UEA.			ļ									
DS0 Set-up per Cross Box location - per 25			1	UDN.UC	ļ .								****	01051	***	
pair pasnel set-up (for access to Feeder)		*		L,UDL	USBFX		\$42.68	\$42.68					\$20.35	\$10.54	\$13.32	<del>                                     </del>
			Г									<u> </u>	1		Į.	
Sub-Loop Feeder- Per 2-Wire Analog Voice	: 0.44401			UEA	USBFA	\$12.05	\$122.24	\$85.05	\$76.35	\$39.16		1	\$20.35	\$10.54	\$13.32	
Grade Ground-Start Loop - Statewide	Loop Feeder per 2-WAVGL		sw	UEA	USBLA	ψ12.03	Ψ122.E-7	<b>\$00.00</b>	<b>V. U.UU</b>	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			t			
Sub-Loop Feeder- Per 2-Wire Analog Voice	Loop Feeder per 2-WAVGL		sw	UEA	USBFB	\$12.05	\$122.24	\$85.05	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
Grade Loop-Start Loop - Statewide	200p r 0000. pc. 2 11711 02	t	T	1									1			
Sub-Loop Feeder - Per 2-Wire Analog Voice			l			***		005.05	\$76.35	\$39.16		1	\$20.35	\$10.54	\$13.32	
Grade Reverse Battery Loop - Statewide	Loop Feeder per 2-WAVGL		sw	UEA	USBFC	\$12.05	\$122.24	\$85.05	\$76.35	\$39.10		<del>                                     </del>	φ20.00	ψ10.5+	<b>\$10.02</b>	†
	Sub-Loop Feeder - Order Coordination for Specified		1									Į.			İ	
	Conversion Time			UEA	OCOSL		\$34.29					<u>i</u>				
Specified Conversion Time, per LSR	Conversion time		┢													
Not and Interfere Deulee (MID)			T													
Network Interface Device (NID)	NID per 2-Wire Loop		1	UENTW	UNDC2	\$1.15	\$0.74				\$3.50		\$19.99	\$19.99	\$19.99	\$19.
THOUTH MILETIAN DE LA CONTRACTION DEL CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTION DE LA CONTRACTIO		<del>                                     </del>	<b>-</b>		UNDC4	\$1.27	\$0.74				\$3.50		\$19.99	\$19.99	\$19.99	\$19.
Network Interface Device Cross Connect	NID per 4-Wire Loop	<del> </del> -	1	02,	0.120	<u> </u>										
		-	+-	<del>                                     </del>				,					Í		<u>]</u>	
UNDLED LOOP CONCENTRATION	Loop Channelization system - DLC	<del></del>	$\vdash$	<del>                                     </del>											1	
Loop Channelization System	(Inside CO)		ı	ULC	ULCCS	\$307.07	\$307.34	\$74.37	\$4.18		\$3.50	<u> </u>	\$20.35	\$10.54	\$13.32	<b>_</b>
	CO Channel Interface - 2-Wire						1	\$9.52	\$8.66	\$8.60	\$3.50		\$20.35	\$10.54	\$13.32	
CO Charlic Interides 2 11110 10100 0	Voice Grade		1-	ULC	ULCC2	\$1.20	\$9.57	\$9.52	\$6.00	\$6.00	\$3.30	<del> </del>	\$20.00	<b>V.0.0.</b>	¥10.02	<u> </u>
Unbundled Loop Concentration - Channel	CO Channel Interfere - 3 Wire			i			l .								1	
Interface-2 Wire Voice-Loop Start or Ground	Voice Grade	ļ	1	UEA	ULCC2	\$1.20	\$9.57	\$9.52	\$8.66	\$8.60	\$3.50		\$20.35	\$10.54	\$13.32	
Start	Voice drade	-	T												ļ	1
UNDLED SUB-LOOP CONCENTRATION (OUTS	IDE CO)		†				T								<del> </del>	
UNDLED SUB-LOOP CONCENTRATION (COTS	Loop Concentration -	1	1	†	<b>1</b>							1	l l			
	Channelization System (Outside		1			****	6054.00	£200 40	\$207.92	\$50.94	\$3.50		\$20.35	\$10.54	\$13.32	
Loop Concentration - Channelization System	CO)	ļ	╄	TBD	UCT8A	\$328.28	\$651.09	\$283.42	\$207.92	\$50.94	\$5.50	+	1 \$25.55	+	1	
a channel by deep California	Loop Concentration - Remote Channel Interface - 2-WAVGL		1									1	1			
Remote Channel Interface - 2-Wire Voice	(Outside CO)		1	TBD	ULCC2	\$0.88	\$9.43	\$9.40	\$4.71	\$4.70	\$3.50		\$20.35	\$10.54	\$13.32	
Loop Concentration - Remote terminal	Loop Concentration - Remote		1		1					-			ŀ			
Cabinet (Outside CO)	Terminal Cabinet (Outside CO)				<u> </u>	ICB			<u> </u>	ļ.——	\$3.50		<del></del>		+	+
	Loop Concentration - Remote		1	1	1	1					1					1
Channel Interface - 2 Wire Voice-Loop Start	Channel Interface - 2-WAVGL			TBD	ULCC2	\$0.88	\$9.43	\$9.40	\$4.71	\$4.70	\$3.50	1	\$20.35	\$10.54	\$13.32	
or Ground Start	(Outside CO)	+	+	+ '65	102002	\$5.00	+ +	755	1			1				
	170)	+	+	+-	<b>-</b>		+	1			1					
BUNDLED LOCAL EXCHANGE SWITCHING(POP	(18)	+	+	+	+		+-	+	1	<u> </u>						
		+	+	+	+				1	<b>†</b>		T	1			
Exchange Ports		1 -	+	1	<b>+</b>	<del>                                     </del>	<del>                                     </del>	+	I							
1				1	1	1	1		1		1	1	1		1	1
				1		[	1				1	i				
Exchange Ports - 2-Wire Analog Line Port-	Exchange Ports - 2-Wire Analog	ĺ	1		1	1			1				\$20.35	\$10.54	\$13.32	\$1.
Res.	Line Port (Res./Bus.)		1	UEPSF	UEPRL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	<u></u>

		Τ' ''-						RATES					OSSI	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec	urring	Disco	curring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Increment Charge - Manual St Order va Electronic-E Add'I
		i				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Note: Although the Port rate includes all available	vortical features in TN, the desired	feature(s)	) will r	need to be	ordered.)		1					i				
Exchange Ports - 2-Wire Analog Line Port	Exchange Ports - 2-Wire Analog	, routaro(s)	<u> </u>	1	Í											
with Caller ID - Res.	Line Port (Res./Bus.)			UEPSR	UEPRC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
Exchange Ports - 2-Wire Analog Line Port	Exchange Ports - 2-Wire Analog								l	***		}	\$20.35	\$10.54	\$13.32	\$1.40
outgoing only - Res.	Line Port (Res./Bus.)			UEPSR	UEPRO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92		<b>_</b>	\$20.35	\$10.54	\$10.02	Ψ1.40
Exchange Ports - 2-Wire VG unbundled TN																
extended local dialing parity Port with Caller	Exchange Ports - 2-Wire Analog		1	HEDGE	UEPAQ	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92		l	\$20.35	\$10.54	\$13.32	\$1.40
ID - Res.	Line Port (Res./Bus.)	ļ	╁	UEFSH	OLI AG	ψ1.03	45.50	Ψ0.10	- 45.55	<b>V</b>						
Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID -	Evchange Ports - 2-Wire Analog	1														
Res (F2R)	Line Port (Res./Bus.)			UEPSR	UEPAK	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.4
Exchange Ports - 2-Wire VG unbundled		T	1									ł	l			
Tennessee Area Calling port with Caller ID -	Exchange Ports - 2-Wire Analog			l	l I			00.40		\$2.92		1	\$20.35	\$10.54	\$13.32	\$1.4
Res (TACER)	Line Port (Res./Bus.)		<u> </u>	UEPSR	UEPAL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.33	φ10.54	₩10.0E	Ψ1,3
Exchange Ports - 2-Wire VG unbundled			1		l 1		li .									
Tomoses the same and the same a	Exchange Ports - 2-Wire Analog			HEPSB	UEPAM	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.4
Res (TACSR) Exchange Ports - 2-Wire VG unbundled	Line Port (Res./Bus.)	+	<b>├</b> ─	OLFOIT	OLI AWI	Ψ1.03	40.00	+ +		·			1			
Tennessee Area Calling port with Caller ID -	Exchange Ports - 2-Wire Analog		1	l			l .	İ					1			
Res (1MF2X)	Line Port (Res./Bus.)			UEPSR	UEPAN	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.4
Exchange Ports - 2-Wire VG unbundled	Zillo V dir (Vidoli Dedi)			†												
Tennessee Area Calling port with Caller ID -	Exchange Ports - 2-Wire Analog		1	l	l l		l		***	***	ł		\$20.35	\$10.54	\$13.32	\$1.4
Res (2MR)	Line Port (Res./Bus.)			UEPSR	UEPAO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92	<u> </u>	<del>-</del>	\$20.55	\$10.54	₩ 10.0L	<del>  ••••</del>
					1		1				l	1	1		l	
	Exchange Ports - 2-Wire Analog		1		UEPAP	04.00	\$9.93	\$9.19	\$3.66	\$2.92	i		\$20.35	\$10.54	\$13.32	\$1.4
low usage line port with Caller ID (LUM)	Line Port (Res./Bus.)	+	╄	UEPSR		\$1.89		7	\$3.00	Ψ2.32	·	<del>                                     </del>	1 020,00	<del>- 4.5.5</del>	<u> </u>	1
Subsequent Activity		*	1_	UEPSR	USASC		\$10.00	\$10.00	<del> </del>		-	<del>                                     </del>	<del> </del>	<del></del>		
	[Exchange Ports includes all			UEPSR	UEPVF	\$0.00	\$0.00	\$0.00			1	1				İ
All Available Vertical Features	Applicable Features.]		+	UEFOR	DEFVE	\$0.00	\$0.00	Ψ0.00	<del> </del>		<del>                                     </del>		1			
Exchange Ports - 2-Wire Analog Line Port	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)			UEPSB	UEPBL	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92	l .	1	\$20.35	\$10.54	\$13.32	\$1.4
without Caller ID - Bus Exchange Ports - 2-Wire VG unbundled Line			+-	102102	100,00	Ψ1.55	1 . 40.00	-	,		T					
Port with unbundled port with Caller+E484 ID	Exchange Ports - 2-Wire Analog	İ					ļ		!			Į.				
- Bus.	Line Port (Res./Bus.)		1	UEPSB	UEPBC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92	L		\$20.35	\$10.54	\$13.32	\$1.4
Exchange Ports - 2-Wire Analog Line Port	Exchange Ports - 2-Wire Analog		$T^-$						l				\$20.35	\$10.54	\$13.32	\$1.4
outgoing only - Bus.	Line Port (Res./Bus.)		1	UEPSB	UEPBO	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$10.02	41.5
Exchange Ports - 2-Wire VG unbundled TN			1	1	1		1			1		į	1		Ì	
extended local dialing parity Port with Caller	Exchange Ports - 2-Wire Analog		1	UEDER	UEPAV	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.4
ID - Bus.	Line Port (Res./Bus.)	-	╫	UEPOD	DEPAV	\$1.09	ψ9.50	ψ3.13	ψυ.υυ	40.02	<del>                                     </del>		1		<u> </u>	T
Exhange Ports - 2-Wire VG unbundled	Exchange Ports - 2-Wire Analog Line Port (Res./Bus.)		1	UEPSB	UEPB1	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92		l	\$20.35	\$10.54	\$13.32	\$1.4
incoming only port with Caller ID - Bus Exchange Ports - 2-Wire VG unbundled TN	Line Folt (nes./bus.)	+	+	1	1	<b></b>	1			1	1					
Bus 2-Way Area Calling Port Economy	Exchange Ports - 2-Wire Analog	ĺ	1		1				1	1	l			040-4		\$1.4
Option - Bus (TACC1)	Line Port (Res./Bus.)			UEPSB	UEPAC	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92	<u> </u>		\$20.35	\$10.54	\$13.32	\$1.4
Exchange Ports - 2-Wire VG unbundled TN			T			I			I		Ī	1	1			1
Bus 2-Way Area Calling Port Standard	Exchange Ports - 2-Wire Analog		1	1	1		60.00	60.10	\$3.66	\$2.92	1	1	\$20.35	\$10.54	\$13.32	\$1.
Option - Bus (TACC2)	Line Port (Res./Bus.)			UEPSE	UEPAD	\$1.89	\$9.93	\$9.19	Ф3.00	<b>⊕</b> 2.92	+	<del>                                     </del>	\$20.00	+ +	1	T
Exchange Ports-2-Wire VG unbundled TN	E change Barra O Mine Applica		1	,	1		1		1		1	1	1		Į.	
Bus 2-Way Collierville & Memphis Local	Exchange Ports - 2-Wire Analog		1	UEPSE	UEPAE	\$1.89	\$9.93	\$9.19	\$3.66	\$2.92	i	1	\$20.35	\$10.54	\$13.32	\$1.4
Calling Port-Bus (B2F)	Line Port (Res./Bus.)		+	UEPSE	_	T * 7.00	\$10.00	\$10.00	T	1		1	T			
Subsequent Activity	[Exchange Ports includes all	<del>-</del>	+	UEFOE	JOSAGO	<b></b>	Ψ10.00	¥10.00	1	+		1	1		T	T
All a Theta Madical Factores	Applicable Features.		1	UEPSE	UEPVF	\$0.00	\$0.00	\$0.00		1		1				1
All Available Vertical Features  Exchange Ports - 4-Wire Analog Voice	Exchange Ports - 4-Wire Analog	<del> </del>		1 - 5	1	1	1	1						1	<b></b>	
Grade Ports - 4-wire Analog Voice	Voice Grade Port		ı	1	UEP4A	\$8.27	\$9.93	\$9.19	\$3.66	\$2.92		1	\$20.35	\$10.54	\$13.32	\$1.
Subsequent Activity	1	-	_	1	USASC		\$10.00	\$10.00	1		l .			1	1	1

								RATES					OSS	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97-	Zone	BCS	USOC				Nonrec	curring	Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Increment Charge - Manual St Order vs Electronic-E
	JIAILD III DOOLLI DA GILLE	01262					Nonrec		Disco		Elec per LSR	Manualty per LSR SOMAN	Svc Order vs. Electronic-1st SOMAN	Svc Order vs. Electronic-Add'i SOMAN	1st SOMAN	Add'I SOMAN
			╙			Rec	First	Add'1	First	Add'l	SOMEC	SOMAN	SUMAN	SOMAN	30MAN	- JOHN FAIR
	[Exchange Ports includes all Applicable Features.]				UEPVF	\$0.00	\$0.00	\$0.00								
Exchange Ports - 2-Wire DID Port	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	\$8.97	\$47.75	\$47.01	\$9.21	\$8.47			\$20.35	\$10.54	\$13.32	\$1.40
				EDEX			\$10.00	\$10.00								
Subsequent Activity	Exchange Ports includes all		<del> </del> —	UEPEX	USASC		\$10.00	\$10.00			<u> </u>			· · ·		
All Available Vertical Features	Applicable Features.]		<u> </u>	UEPEX	UEPVF	\$0.00	\$0.00	\$0.00								-
Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability	Exchange Ports - 4-Wire DID Port			UEPDD	UEPDD	\$35.74	\$75.93	\$38.15	\$8.77	\$8.04		<u> </u>	\$20.35	\$10.54	\$13.32	\$1.40
Subsequent Activity		*	1	UEPDD	USACP		\$10.00	\$10.00					ļ	ļ		
Exchange Ports - 2-Wire ISDN Port (See	Exchange Ports - 2-Wire ISDN							400.40	0440	64.10		1	\$41.43	\$42.17	\$9.80	\$9.80
	Port	ļ	1_	U1PMA		\$16.26	\$30.23	\$29.49	\$4.10	\$4.10			\$41.43	\$42.17	ψ3.00	ψο.ο
Subsequent Activity		*	Ь.	UEPDD	USASC		\$10.00	\$10.00		<u> </u>		<b></b>			<del>                                     </del>	-
	[Exchange Ports includes all		Į.	LUCDOD	UEPVF	\$0.00	\$0.00	\$0.00	1			İ	1			1
All Available Vertical Features NOTE: Transmission/usage charges associated	Applicable Features.]		<b>├</b>	DEPUD	UEPVF	\$0.00	\$0.00	\$0.00				<b>1</b>				
with POTS circuit switched usage will also apply							İ					ļ	i		1	
to circuit switched voice and/or circuit switched				l								1				
data transmission by B-Channels associated with 2-wire ISDN ports.												<u></u>				<u> </u>
NOTE: Access to B Channel or D Channel			T								1					
Packet capabilities will be available only through			1	l												
BFR/New Business Request Process. Rates for			1				i		İ		•				i	
the packet capabilities will be determined via the			1	<u> </u>	1											
Bona Fide Request/New Business Request  Exchange Ports - 2-Wire ISDN Port		<del>                                     </del>	╂								T					1
Channel Profiles		*			U1UMA	\$0.00	\$0.00	\$0.00		ļ		<u> </u>		<u> </u>		<del>                                     </del>
Chamier Fromes	Exchange Ports - 4-Wire ISDN	·	1	1					l	****			\$40.69	\$42,17	\$9.07	\$10.
	DS1 Port			UEPEX	UEPEX	\$75.04	\$148.66	\$147.18	\$38.46	\$36.98	<b>-</b>		\$40.09	\$42.17	\$9.07	<b>\$10</b> .
2 *************************************	Exchange Ports - 2-Wire Analog			UEPSE	UEPRD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92		1	\$20.35	\$10.54	\$13.32	\$1.
Res	Line Port (PBX)	<del> </del>	╄	UEPSE	DEPRO	\$1.79	\$9.55	ψ3.13	+ 0.00	Ψ.J.			1			
2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus	Line Port (PBX)			UEPSP	UEPPC	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92	ļ		\$20.35	\$10.54	\$13.32	\$1.4
E TITO TO EMILO OTED OTTO	Exchange Ports - 2-Wire Analog			UEPSP	UEPPO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92		l l	\$20.35	\$10.54	\$13.32	\$1.
PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming	Line Port (PBX) Exchange Ports - 2-Wire Analog	-	+	10E1 01	OL. I C	\$1.75	+ +0.00	+ ****	1		1 "					1
DBY Trunk - Bus	Line Port (PBX)			UEPSP	UEPP1	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92	ļ		\$20.35	\$10.54	\$13.32	\$1.
2-Wire Analog Long Distance Terminal PBX	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPLD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92		l	\$20.35	\$10.54	\$13.32	\$1.
Trunk - Bus 2-Wire Analog TN 2-Way Calling Plan PBX	Exchange Ports - 2-Wire Analog		+	102, 0.	102.12	1							T	440.54	040.00	\$1.
Trunk - Bus	Line Port (PBX)		L	UEPSP	UEPT2	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92	<b>_</b>		\$20.35	\$10.54	\$13.32	\$1.
2-Wire TN Outward Calling Plan PBX Trunk - Bus	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPTO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92		<u> </u>	\$20.35	\$10.54	\$13.32	\$1.
2-Wire Voice Unbundled PBX LD Terminal	Exchange Ports - 2-Wire Analog Line Port (PBX)	1	1	UEPSP	UEPLD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92		1_	\$20.35	\$10.54	\$13.32	\$1.
Ports 2-Wire Voice Unbundled 2-Way PBX	Exchange Ports - 2-Wire Analog	+	$\top$	1 "					\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.
Tennessee Calling Port	Line Port (PBX) Exchange Ports - 2-Wire Analog	+	+	UEPSF	UEPT2	\$1.79	\$9.93	\$9.19	\$3.00	1	<del> </del>	<b>†</b>	1 -			<u> </u>
2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee Calling Port	Line Port (PBX)		_	UEPSF	UEPTO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92	<del> </del>	<del>                                     </del>	\$20.35	\$10.54	\$13.32	\$1.4
2-Wire Vice Unbundled 2-Way PBX Usage Port	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSF	UEPXA	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92	<u> </u>		\$20.35	\$10.54	\$13.32	\$1.
2-Wire Voice Unbundled PBX Toll Terminal	Exchange Ports - 2-Wire Analog Line Port (PBX)		Τ	UEPSF	UEPXE	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.
2-Wire Voice Unbundled PBX LD DDD	Exchange Ports - 2-Wire Analog	+	-	1	1	1	-1		T					1	\$13.32	\$1.

				_					RATES					OSS	RATES		
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORIK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec		Nonre	curring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'i
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPXD	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
+	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPXE	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPXL	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
$\dagger$	2-Wire Voice Unbundled 2-Way PBX	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPXM	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Economy Admin Calling Port TN Calling Port	<u> </u>			UEPSP	UEPXN	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
T	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port	Exchange Ports - 2-Wire Analog Line Port (PBX)			UEPSP	UEPXO	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	Exchange Ports - 2-Wire Analog Line Port (PBX) Exchange Ports - 2-Wire Analog		igspace	UEPSP	UEPXS	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
1	2-Wire Voice Unbundled PBX Collierville and Memphis Calling Port 2-Wire Voice Unbundled 2-Way PBX	Line Port (PBX)  Exchange Ports - 2-Wire Analog  Exchange Ports - 2-Wire Analog			UEPSP	UEPXU	\$1.79	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
1	Tennessee RegionServ Calling Port Subsequent Activity	Line Port (PBX)	*	-	UEPSP UEPSP	UEPXV	\$1.79	\$9.93 \$10.00	\$9.19 \$10.00	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
	All Available Vertical Features	[Exchange Ports includes all Applicable Features.]			UEPSP	UEPVF	\$0.00	\$0.00	\$0.00							240.00	\$1,40
	Exchange Ports - Coin Port	Exchange Ports - Coin Port		ـــــ	<u> </u>	<del>                                     </del>	\$2.11	\$9.93	\$9.19	\$3.66	\$2.92			\$20.35	\$10.54	\$13.32	\$1.40
IBU	INDLED LOCAL SWITCHING, PORT USAGE		1-									<u> </u>	<u> </u>				
Т			<u> </u>	1						<del></del>	<del> </del>	<del> </del>					
	End Office Switching (Port Usage)			<b>⊥</b>								<del>                                     </del>	ļ	+	-	-	+
	End Office Switching Function, Per MOU	End Office Switching Function End Office Interoffice Trunk Port -		╁	-	<u> </u>	\$0.0008041		-			-	· · · ·	<del>                                     </del>			
-	End Office Trunk Port - Shared, Per MOU	Shared, per MOU		╁	-									-			
	Tandem Switching (Port Usage) (Local or Acc	cess Tandem)		T						<u> </u>	ļ			ļ		<b></b>	+
	Tandem Switching Function Per MOU	Tandem Switching Function Tandem Interoffice Trunk Port -		$\perp$			\$0.0009778					-		<del> </del>			
_	Tandem Trunk Port - Shared, Per MOU	Shared, per MOU	-	+	<u> </u>	<del>-</del>						Ī				<u> </u>	
NBL	JNDLED TRANSPORT			F	1							<u> </u>		<u> </u>			1
_		<del>                                     </del>	+	+	<del> </del>	+	<del>                                     </del>	t		<b>1</b>							
-	COMMON TRANSPORT (Shared)	Common Transport - per mile, per	-	╁		<u> </u>	\$0.0000064			1						:	
1	Common Transport - Per Mile, Per MOU  Common Transport - Facilities Termination	MOU Common Transport - Facilities Termination per MOU	+	$\top$	1		\$0.0003871										
	Per MOU			T									-	<del>                                     </del>	-	<del> </del>	-
$\Box$	INTEROFFICE CHANNEL - DEDICATED TRAN			+	+									1			
	Interoffice Channel - Dedicated Transport - Wire Voice Grade - Per Mile per month	Voice Grade - per mile per MOU			U1TVX	1L5XX	\$0.0174			<u> </u>	-	<b> </b>	ļ				-
	Interoffice Channel - Dedicated Transport- 2 Wire Voice Grade - Facility Termination per month	2- Interoffice Transport - Dedicated 2-Wire Voice Grade - Facility Termination	•		U1TV	U1TV2	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54

									RATES					oss	RATES		
ГТ									TATES							Incremental	Incremental
			Not in TRA							<del> </del>		Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svo
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	USOC				Nonre	curring	Submitted	Submitted	Charge - Manual	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order vs. Electronic-Dis
1		STATED III DOGICE: ST STEEL	01262							Diese	onnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Electronic-Add'i	1st	Add'i
					l i		Rec	Nonrec First	urring Add'l	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	D. C			_			1.00										
	nteroffice Channel - Dedicated Transport- 2-Wire Voice Grade Rev Bat Per Mile per	Interoffice Transport - Dedicated -	Ì	l									1	ļ.			
		Voice Grade - per mile per MOU		1	U1TVX	1L5XX	\$0.0174	1									
+-In	month- nteroffice Channel-Dedicated Transport- 2-	Interoffice Transport - Dedicated -										ļ		l	i '	1	
	Wire Voice Grade Rev BatFacility	2-Wire Voice Grade - Facility	ì	l								1	1	****	\$21.09	\$9.80	\$10.54
1 1	Termination per month	Termination			U1TVX	U1TR2	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51		<b></b>	\$20.35	\$21.09	\$9.00	\$10.54
+	Termination per mental							<u> </u>					<b>_</b>				-
INITE	ROFFICE CHANNEL - DEDICATED TRANS	PORT- 56/64 KBPS								L			<u> </u>				
INTE	nteroffice Channel - Dedicated Transport -	Interoffice Transport - Dedicated -							1					i	1		
	56 kbps - per mile per month	DSO - 56/64 kbps - per mile		<u> </u>	U1TDX	1L5XX	\$0.0174	L		<b>├</b>					<u> </u>		·
+-+		Interoffice Transport - Dedicated -								1		l .	i				
1 1	Interoffice Channel - Dedicated Transport -	DSO - 56/64 kbps - Facility		l					\$17.37	\$27.96	\$3.51	1	1	\$20.35	\$21.09	\$9.80	\$10.54
	56 kbps - Facility Termination per month	Termination		<b>!</b>	U1TDX	U1TD5	\$17.98	\$55.39	\$17.37	\$27.90	\$3.51	<del> </del>	<u> </u>	125.55			
	Interoffice Channel - Dedicated Transport -	Interoffice Transport - Dedicated -		l		41.500	000474					l	į.			l	
	64 kbps - per mile per month	DSO - 56/64 kbps - per mile	<u> </u>	<u> </u>	U1TDX	1L5XX	\$0.0174	ļ	<del></del>	<b>├</b> ──	<del>                                     </del>	<del>                                     </del>	1	<del>                                     </del>	· · · · · · ·		
		Interoffice Transport - Dedicated -		1				1						1	l.		
	Interoffice Channel - Dedicated Transport -	DSO - 56/64 kbps - Facility			U1TDX	U1TD6	\$17.98	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.54
(	64 kbps - Facility Termination per month	Termination	<del> </del>		UIIDA	OTTE	ψ17.50	1 400.00	<b>V</b>	+ ****							
				₩				<del>                                     </del>	-	<del>                                     </del>		<del>                                     </del>	<del>                                     </del>				
INTE	ROFFICE CHANNEL - DEDICATED TRANS	SPORT - DS1		<b>⊢</b> -		L		<b>-</b>		<del> </del>		<b></b>			<del> </del>	1	T
	Interoffice Channel - Dedicated Channel -	Interoffice Transport - Dedicated -			U1TD1	1L5XX	\$0.3562	1					1	l		1	
	DS1 - Per Mile per month	DS1 - per mile	ļ	-	וטווטו	ILSAA	\$0.3502	<del>                                     </del>	<del>                                     </del>		+		<b>†</b>				
	Interoffice Channel - Dedicated Tranport -	Interoffice Transport - Dedicated -		1	U1TD1	U1TF1	\$77.86	\$112.40	\$76.27	\$19.55	\$14.99		1	\$20.35	\$21.09	\$9.80	\$10.54
_ !	DS1 - Facility Termination per month	DS1 - Facility Termination			101101	01111	Ψ/7.00	<b>VIII.</b>	<del></del>	V				1	1		
			<del></del>	₽-	ļ		<b>_</b>	<del>                                     </del>			<del> </del>	1				1	
	AL CHANNEL - DEDICATED TRANSPORT					<u> </u>	<del>                                      </del>	<b>-</b>	<del> </del>	+					<del>                                     </del>		
	Local Channel - Dedicated - 2-Wire Voice				ULDVX	1L5NC	\$0.00			1			ı			1	
	Grade Per Mile per month	Dedicated 0	<del>-</del>	₩	OLDVX	ILSING	\$0.00		+	+-	+	<del>                                     </del>	<b>-</b>		1		
		Local Channel - Dedicated - 2-		1		ļ	1	į.	1			1					
	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here deaveraged]		lι	ULDVX	ULDV2	\$17.18	\$199.33	\$24.16	\$54.81	\$4.80	1		\$20.35	\$10.54	\$13.30	\$0.00
	Grade per month - Zone 1	Local Channel - Dedicated - 2-		╁	1 OLD VA	102512	******	1,11111	<u> </u>	1							İ
	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here		1	1		1	1				1		1			
	Grade per month - Zone 2	deaveraged		2	LULDVX	ULDV2	\$22.44	\$199.33	\$24.16	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	\$0.00
-	Grade per month - zone z	Local Channel - Dedicated - 2-	-	<b>†</b>		<b>1</b>											i
1	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here		1	1			1		1	l'	1		\$20.35	\$10.54	\$13.30	\$0.00
	Grade per month - Zone 3	deaveraged]		3	ULVDX	ULDV2	\$29.34	\$199.33	\$24.16	\$54.81	\$4.80			\$20.35	\$10.54	\$13.30	Ψ0.00
	Local Channel - Dedicated - 2-Wire Voice			Т	Γ							i				1	
	Grade Rev. Bat. Per Mile per month		*	L.	ULDVX	1L5NC	\$0.00			<del> </del>	<del></del>	<b>├</b> ──		-	<del></del>	<del>                                     </del>	+
+-		Local Channel - Dedicated - 2-		1			1			1		Į.	l l			1	
	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here		1.			\$17.18	\$199.33	\$24.16	\$54.81	\$4.80	•	1	\$20.35	\$10.54	\$13.30	\$0.00
	Grade Rev. Bat. Per month - Zone 1	deaveraged]		1	ULDVX	ULDR2	\$17.18	\$199.33	\$24.10	\$54.01	ψ4,00		1	1	<del>                                     </del>		
		Local Channel - Dedicated - 2-		i i	i		1			1		1	i				
	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here		1 2	ULDVX	ULDR2	\$22.44	\$199.33	\$24.16	\$54.81	\$4.80	1	1	\$20.35	\$10.54	\$13.30	\$0.00
	Grade Rev. Bat. Per Month - Zone 2	deaveraged]	→	+-	OLDVA	OLDRZ	Ψ22.44	ψ133.00	Ψ <u>Ε</u> -1.10	1 30	*						
		Local Channel - Dedicated - 2-		1	1	1	1	1	1			1	1	I			
1	Local Channel - Dedicated - 2-Wire Voice	Wire Voice Grade [shown here		lз	LULDVX	ULDR2	\$29.34	\$199.33	\$24.16	\$54.81	\$4.80	1	1	\$20.35	\$10.54	\$13.30	\$0.00
$\perp$	Grade Rev. Bat. Per Month - Zone 3	deaveraged]	+-	+ -	1555	†	1	1		1		T					1
	Local Channel - Dedicated - 4-Wire Voice				ULDDX	1L5NC	\$0.00					1		1		<b>_</b>	
_	Grade Per Mile per month	Local Channel - Dedicated - 4-	+	+	1	1	1								Į	I	
	Local Channel - Dedicated - 4-Wire Voice	Wire Voice Grade [shown here		1	1	I	1	1		1		1				000.00	***
-	Grade per month - Zone 1	deaveraged		1	ULDDX	ULDV4	\$18.18	\$201.53	\$24.83	\$55.52	\$5.51	<b>↓</b>		\$20.35	\$10.54	\$13.30	\$0.00
+	Grade per month - Zone i	Local Channel - Dedicated - 4-	+		1	1			1	1		1					
1	Local Channel - Dedicated - 4-Wire Voice	Wire Voice Grade [shown here		1		1		\$201.53	\$24.83	\$55.52	\$5.51		1	\$20.35	\$10.54	\$13.30	\$0.00
1						( ULDV4	\$23.74										

		Γ		1	г			F	RATES					oss	RATES		
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not In TRA Docket 97- 01262		BCS	usoc				Nonrec	urring	Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incrementa Charge - Manual Svo Order vs. Electronic-Di
				1	l			Nonrec	urring	Disco	nnect	per LSR	LSR	Electronic-1st	Electronic-Add'i	1st	Add'l
- 1							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Local Channel - Dedicated - 4-		T				· ·									
	Local Channel - Dedicated - 4-Wire Voice	Wire Voice Grade Ishown here		ŀ									l	ŀ			
	Grade per month - Zone 3	deaveraged]		3	ULDDX	ULDV4	\$31.05	\$201.53	\$24.83	\$55.52	\$5.51		L	\$20.35	\$10.54	\$13.30	\$0.00
-+	Local Channel - Dedicated - DS1 Per Mile	douverageu	-	1									1			1	
	per month		*	1	ULDD1	1L5NC	\$0.00										
_	Local Channel - Dedicated - DS1 per month -	Local Channel - Dedicated DS1		+										l .			
	Zone 1	[shown here deaveraged]		1 1	ULDD1	ULDF1	\$36.24	\$277.35	\$233.26	\$33.18	\$22.30			\$45.68	\$1.76	\$21.75	\$1.76
-	Local Channel - Dedicated - DS1 per month -																
	Zone 2	[shown here deaveraged]		2	ULDD1	ULDF1	\$47.33	\$277.35	\$233.26	\$33.18	\$22.30			\$45.68	\$1.76	\$21.75	\$1.76
	Local Channel - Dedicated - DS1 per month -	I ocal Channel - Dedicated DS1	<del> </del>	+=						,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		1		l			l
	Zone 3	[shown here deaveraged]		3	ULDD1	ULDF1	\$61.89	\$277.35	\$233.26	\$33.18	\$22.30	ì		\$45.68	\$1.76	\$21.75	\$1.76
-	Zone 3	[SHOWH Here deaveraged]		Ť	1		1									i	
				+	₩				-							Ī.,	T
<b>JULTI</b>	PLEXERS		<del> </del>	+-	<b>↓</b> —	ļ.—				-				<b>!</b>		· · · · · · · · · · · · · · · · · · ·	
T	Channelization - DS1 to DS0 Channel	Channelization - Channel System		1	LINETE:	1 401	600 77	\$141.87	\$77.11	\$14.51	\$13.46	Ī		\$20.35	\$9.80	\$11.49	\$1.18
	System	DS1 to DSO		1-	UXTD1	MQ1	\$80.77	Φ141.0/	\$77.11	φ14.01	φ10. <del>7</del> 0		<del>                                     </del>	1	1	<del></del>	1
	OCU-DP COCI (data) - DS1 to DS0 Channel	Interface Unit - Interface DS1 to		1	l	1,0400	\$1.82	\$6.07	\$4.66			I	I	\$20.35	\$9.80	\$11.49	\$1.18
	System - per month (2.4-64kbs)	DSO - OCU - DP Card		╄	UDL	1D1DD	\$1.82	\$6.07	\$4.00	<b></b>		<del></del>		<b>VEC.55</b>	40.00	V. 11.10	+
	2-wire ISDN COCI (BRITE) - DS1 to DS0	Interface Unit - Interface DS1 to		1				00.07	***			l	1	\$20.35	\$9.80	\$11.49	\$1.18
	Channel Systsem - per month	DSO - Brite Card		_	UDN	UC1CA	\$3.10	\$6.07	\$4.66	ļ — —		<b>├</b> ──	<del>                                     </del>	Ψ20.00	Ψ3.00	<b>\$11.10</b>	+ 1115
	Voice Grade COCI - DS1 to DS0 Channel	Interface Unit - Interface DS1 to			1	l						l		\$20.35	\$9.80	\$11.49	\$1.18
	System - per month	DSO - Voice Grade Card			UEA	1D1VG	\$.91	\$6.07	\$4.66			<del>-</del>		\$20.00	Ψ3.00	- Ψ11. <del>13</del>	+ + + + + + + + + + + + + + + + + + + +
-	1	Channelization - Channel System		1						l	* 40.00	j	l .	\$20.35	\$9.80	\$11.49	\$1.18
	DS3 to DS1 Channel System per month	DS3 to DS1		j	UXTD3	MQ3	\$222.98	\$308.03	\$108.47	\$44.47	\$42.62			\$20.35	\$9.60	\$11.43	\$1.10
		Channelization - Channel System												****	\$9.80	\$11.49	\$1.18
	STS1 to DS1 Channel System per month	DS3 to DS1		l	UXTS1	MQ3	\$222.98	\$308.03	\$108.47	\$44.47	\$42.62	<b>.</b>		\$20.35	\$9.00	\$11.49	\$1.10
	DS3 Interface Unit (DS1 COCI) used with	Interface Unit - Interface DS3 to		1			1						l .	****	\$9.80	\$11.49	\$1.18
	Loop per month	DS1		1	USL	UC1D1	\$17.58	\$6.07	\$4.66				ļ	\$20.35	\$9.60	\$11.49	\$1.10
	DS3 Interface Unit (DS1 COCI) used with	Interface Unit - Interface DS3 to								1				****	20.00	644.40	\$1.18
	Local Channel per month	DS1			ULDD1	UC1D1	\$17.58	\$6.07	\$4.66	ļ				\$20.35	\$9.80	\$11.49	\$1.10
-+	DS3 Interface Unit (DS1 COCI) used with	Interface Unit - Interface DS3 to							İ			l	ļ	l	20.00		£4.40
	Interoffice Channel per month	DS1	ļ	1	U1TD1	UC1D1	\$17.58	\$6.07	\$4.66			L		\$20.35	\$9.80	\$11.49	\$1.18
	interestine entantes per menu.									1			L.,	<u> </u>			
			+	1	†	†·				1						l .	1
DARK	FIBER			+-	<del>                                     </del>	<del>                                     </del>			-								T
	Dark Fiber, Four Fiber Strands, Per Route	Dark Fiber, per 4 fiber strands, per						l				i	1	1			1
	Mile or Fraction Thereof per month - Local	route mile or fraction thereof	'	ı	UDF	1L5DC	\$53.23	Į.	1	1				1			
	Channel	Dark Fiber, per 4 fiber strands, per	,	+-	100.	12000	<b>\$00.20</b>		<del> </del>	1		1					
i		route mile or fraction thereof	' [		UDF	UDFC4		\$1,219.22	\$169.75	\$453.22	\$339.34	i	1			1	
	NRC Dark Fiber - Local Channel	route mile or traction thereof	<del> </del>		1 051	100,00	<del> </del> -	V.,	V.00		-	<del>                                      </del>					
	Dark Fiber, Four Fiber Strands, Per Route	Dark Fiber, per 4 fiber strands, per	.	ı			1					1	l .	1	1	1	
- 1	Mile or Fraction Thereof per month -	route mile or fraction thereof	'	1	UDF	1L5DF	\$53.23					1	1	L		1	
	Interoffice Channel			-	1 001	12001	\$50.E0						T				
F		Dark Fiber, per 4 fiber strands, per	'	1	UDF	UDF14	1	\$1,219.22	\$169.75	\$453.22	\$339.34	1	I	1		1	
	NRC Dark Fiber - Interoffice Channel	route mile or fraction thereof	+	+	1 000	100114		¥ 1,2 10.22	\$.00.70	1	7		1	1		1	
- 1	Dark Fiber, Four Fiber Strands, Per Route	Davis Filter per 4 filter strands per		1	1	l	I			1		1	I	1	1	1	
- 1	Mile or Fraction Thereof per month - Local	Dark Fiber, per 4 fiber strands, pe	'	1	UDF	1L5DL	\$53.23								ì	I	
	Loop	route mile or fraction thereof	-	+	1 00	+ 12352	+ 400.20	t		1				1			
		Dark Fiber, per 4 fiber strands, pe	'	1	UDF	UDFL4	J	\$1 219 22	\$169.75	\$453.22	\$339.34	.1		1	1	1	
	NRC Dark Fiber - Local Loop	route mile or fraction thereof	+		1 000	100, 14	1	¥ 1,2 10.EZ	#100.70	1	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	<b>†</b>			1		
				_	-			<del></del>	+	<del> </del>	<del></del>	+	+	<del>-</del>	1	1	+
BXX A	CCESS TEN DIGIT SCREENING							ļ	<del> </del>		<del></del>	<del></del>		<del> </del>	+	<del> </del>	+
		800 Access Ten digit screening		1	1	1	1	1						1		1	
	8XX Access Ten Digit Screening, Per Call	(800 ATDS) per call					\$0.0005192	<u> </u>	-	<b>↓</b>	-	<del>-</del>	<del></del>	+		1	+
+	8XX Access Ten Digit Screening,	800 Access Ten digit screening			1	1	1	1			1	1	1	1		1	
	Reservation Charge Per 8XX Number	Reservation Charge per 800				1	.1	l		1			I .	\$20.35			
. 1	Reserved	Number Reserved				N8R1X	/ I	\$5.21	\$0.76		1			あとい.づつ	1		1

				_					RATES					oss	RATES		
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORIK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262		BCS	usoc		Nonrec			curring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic-Dis Add'i
-						1	Rec	First	Add'l	First	Add'i	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translations	800 Access Ten digit screening per 800 # established w/o POTS Translations						\$11.47	\$1.46	\$7.34	\$0.7602			\$20.35		\$13.28	
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translations	800 Access Ten digit screening per 800 # established with POTS Translations				N8FTX		\$11.47	\$1.46	\$7.34	\$0.7602			\$20.35		\$13.28	
	Customized Area of Service Per 8XX Number	800 Access Ten digit screening Customized Area of Service per 800 Number				N8FCX		\$4.47_	\$2.24								
	InterLATA CXR Routing Per CXR Requested Per 8XX No.	per 800 #				N8FMX		\$5.23	\$3.00				!				
		800 Access Ten digit screening, Change Charge per Request 800 Access Ten digit screening,		-		N8FAX		\$5.97	\$0.76					\$20.35			
	8XX Access Ten Digit Screening, Call Handling and Destination Features	Call Handling and Destination Features	-	-		N8FDX		\$4.47									
IE IN	NFORMATION DATA BASE ACCESS (LIDB)	LIDB Common Transport per				ļ											
	LIDB Common Transport Per Query	Query  LIDB Validation per Query		╄-		OQU	\$0.0000354 \$0.0117403			<u> </u>	<del>  -</del>	<del>-</del>					
+	LIDB Validation Per Query LIDB Originating Point Code Establishment or Change	LIDB Originating Point Code Establishment or Change	-			040	ψο.στιτ-100	\$49.03						\$20.35			
GNA	LING (CCS7)		-														
	CCS7 Signaling Termination, Per STP Port	CCS7 Signaling Termination per STP Port CCS7 Signaling Usage per TCAP		<u> </u>			\$138.41			1			, , , , , , , , , , , , , , , , , , , ,				<u> </u>
-	CCS7 Signaling Usage, Per TCAP Message	Message CCS7 Signaling Connection per 56	5	-		-	\$0.0000916							\$20.35			
+	CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link)	kbps faciliity  CCS7 Signaling Connection per 56  kbps faciliity	3	-			\$17.84 \$17.84	\$130.84 \$130.84						\$20.35		-	
+	(also known as D link)  CCS7 Signaling Usage, Per ISUP Message	CCS7 Signaling Usage, per call setup message					\$0.0000373									ļ	
	CCS7 Signaling Usage Surrogate, per link per LATA	CCS7 Signaling Usage Surrogate, per 56 kbps facility per LATA per month					\$352.30						<u>.</u>	\$20.35			
			+	_	╂	+-	<del> </del>	<del> </del>	<del> </del>	<del> </del>		·					
ELEC	Selective Routing Per Unique Line Class	Selective Routing (Interim Solution Line Class Codes) per Unique Line Class Code per Request per				USRCR		\$179.60						\$20.35			
+	Code Per Request Per Switch	Switch		#		1555				-							
IN - E	BELLSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service  Establishment, Per State, Initial Setup	AIN SMS Access Service - Service Establishment, per State, Initial Setup	9	-		CAMSE		\$135.56									
	AIN SMS Access Service - Port Connection	- AIN SMS Access Service - Port Connection - Dial/Shared Access				CAMDE		\$41.75									
$\dashv$	Dial/Shared Access  AIN SMS Access Service - Port Connection ISDN Access		1	1	† <del>-</del>	CAM1F		\$41.75									

							l .	F	RATES						RATES	Incremental	Incrementa
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec		Disco	curring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'l	Charge - Manual Svc Order vs. Electronic-Disc 1st SOMAN	Charge - Manual Sv Order vs. Electronic-D Add'i
1						<u> </u>	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SUMAN	SOMAN
+		AIN SMS Access Service - User		T				-					ì			i	İ
		Identification Codes - Per User ID			ł		<b>!</b>										
		Code				CAMAU		\$96.63		1							
┷		AIN SMS Access Service -		$\vdash$											1		
	AIN SMS Access Service - Security Card,	Security Card, per User ID Code, Initial or Replacement				CAMRC		\$113.67				Į.					
	Per User ID Code, Initial or Replacement	Initial of Replacement			<del></del>	07 4111 10	-	*************								j	
	AIN SMS Access Service - Storage, Per Unit	AIN SMS Access Service					\$0,0024			1	ĺ		1				
i		Storage, per unit (100 kilobytes)		₩	<b> </b>	<del></del>	\$0.0024			<del>                                     </del>		<del>                                     </del>					
		AIN SMS Access Service -	1	1		1	** *****			I		ľ		1		ļ	
		Session, per minute		<u> </u>			\$0.0820123			<b></b>	<del> </del>	+	<del>                                     </del>	<del> </del>			<b>—</b>
_		AIN SMS Access Service -		1	l					l			1				
	AIN SMS Access Service - Company	company performed session, per		ı	1				ĺ	l		l .				l .	1
	Performed Session, Per Minute	minute		1		1	\$2.27			L						<del> </del>	
+	Performed Session, Fer Minute	nindto .		_						1		1	1	1		L.,	
$\perp$				+	-	+			<u> </u>	<del></del>		1	1	1		I	
- BE	ELLSOUTH AIN TOOLKIT SERVICE											<del>                                     </del>	<del>                                     </del>			t	1 "
T	AIN Toolkit Service - Service Establishment	AIN TS - Service Establishment		1		1	1			Į.	1	į.	l	1		1	
1	Charge, Per State, Initial Setup	Charge per State, Initial Setup	İ	1		BAPSC		\$132.04		<u> </u>	ļ <u> </u>	1		<b>├</b>	+	<del></del> -	
+	AIN Toolkit Service - Training Session, Per	AIN TS - Training Session, per		$\mathbf{T}$									1				
	7 47 1 0 0 11 11 1 1 1 1 1 1 1 1 1 1 1 1 1	Customer		1	1	BAPVX		\$7,915	1							<u> </u>	<del></del>
	Customer	Customer	<del> </del>	╅										1			
1		AINLTO Tilenes Assess Charge		1				l		1			1				1
	AIN Toolkit Service - Trigger Access Charge,	AIN 15 - Irigger Access Charge,		Į.	1	BAPTT	j	\$31.21	+				1				İ
	Per Trigger, Per DN, Term. Attempt	per Trigger per DN, Term Attempt		1	<del></del>	DAPII		φ31.21			+		<del></del>	<u> </u>	1		
1		AIN TS - Trigger Access Charge,					1	1	1	1		1	1	1	1		
	AIN Toolkit Service - Trigger Access Charge,	per Trigger per DN, Off Hook	İ	1	ļ		i		1	1		1	1	1	İ	l	
	Per Trigger, Per DN, Off-Hook Delay	Delay	1	1	1	BAPTD		\$31.21			J		<b></b>				
+	Fel Higger, Fel Bit, On Floor Bolay	AIN TS - Trigger Access Charge,		1								1		1		1	
	AIN Toolkit Service - Trigger Access Charge,	per Trigger per DN, Off Hook	İ	1		1	1			1	1		ļ				
ı	AIN TOOKIT Service - Trigger Access Criarge,	Immediate		1	ì	BAPTM	ı	\$31.21		1						<u> </u>	
	Per Trigger, Per DN, Off-Hook Immediate	Immediate		+	<del>                                     </del>					1	, ,					1	
							1	i		1					1		
	AIN Toolkit Service - Trigger Access Charge,	AIN TS - Trigger Access Charge,		ı	1	BAPTO	. 1	\$85.24		1		1	1				
İ	Per Trigger, Per DN, 10-Digit PODP	per Trigger per DN, 10-Digit PODF	<u>'</u>		<u> </u>	BAPIO		\$05.24	+			<del></del>	<del>                                     </del>			T	
	AIN Toolkit Service - Trigger Access Charge	AIN TS - Trigger Access Charge,	1		1	1						1		1		1	
-	Per Trigger, Per DN, CDP	per Trigger per DN, CDP		l _	l	BAPTO		\$85.24	ļ		+			1	+	+	+
+	1 or ringgory or 211, 421			Т			1		1		i		1	1		1	
	AIN Toolkit Service - Trigger Access Charge	AIN TS - Tringer Access Charge.	į	1		1	1	1				l l		1			
	Per Trigger, Per DN, Feature Code	per Trigger per DN, Feature Code	ĺ	1	l .	BAPTE	: [	\$85.24	1	1							+
+		per migger per sint recise even		+-	t	<b>-</b>							l			1	
	AIN Toolkit Service - Query Charge, Per	AIN TS - Query Charge, Per Query				1	\$.0211882	1									
	Query	AIN TS - Query Charge, Per Query	-	+	+	+-	1 2.02	t			1					1	
	AIN Toolkit Service - Type 1 Node Charge,	1		1	1	1	1	l	1	1	1	1	1	1		1	
	Per AIN Toolkit Subscription, Per Node, Per		*		1	1	C 00E4774	1				1	1	I .		I	
- [	Query	per query	<u> </u>	4_			\$.0054774		+			+	1	+		1	1
+		AIN TS - SCP Storage charge, per	1	1	1	1				1	1	1	1	1		1	
	AIN Toolkit Service - SCP Storage Charge,	SMS access account, per 100		ı		1	1	1		1	1	1	1	1		l .	
	Per SMS Access Account, Per 100 Kilobytes	kilobytes		1	1	1	\$1.50	<u> </u>				<del></del>	<b>—</b>				-
+	AIN Tablis Consider Monthly report Dor	AIN TS - Monthly Report - per AIN	1	1	1	$\top$						1	1			1	
-	AIN Toolkit Service - Monthly report - Per		`	1	1	BAPMS	\$17.43	\$33.52		1		1	<u> </u>				1
	AIN Toolkit Service Subscription	TS Subscription	+	+	+	+-····	+	† · · · · · · · · · · ·	1			-					1
7	AIN Toolkit Service - Special Study - Per All	AIN 15 - Special Study - per AIN	1	1	1	BAPLS	\$0.1321116	\$36.23	1	1	1	1	1			1	1
	Toolkit Service Subscription	TS Subscription			+-	BAPLS	\$0.1321110	\$30.23		<del></del>	+	<del></del>					
+	AIN Toolkit Service - Call Event Report - Per	r AIN TS - Call Event Report - per			1	1		1				1		1	1	1	i
	AIN Toolkit Service Subscription	AIN TS Subscription				BAPDS	\$ \$17.35	\$33.52		+-	+		+		+	<del></del>	+
+-				T				1	1	1			1	1	1	ı	
	AIN Toolkit Service - Call Event Special	AIN TS - Call Event Special Study		ı	1	1		1		1		1	1	1	1	1	
1	Study - Per AIN Toolkit Service Subscription	ner AIN TS Subscription	1	1	1	BAPES	\$0.0511435	\$36.23	1	<u> </u>						<del></del>	
	Study - Per AIN TOOIKIT Service Subscription	per Air 10 dubocilpilori		+-	+-			1							1	l	
				-	——	+-	<del>                                     </del>	<del>                                     </del>	+	+	-						
NIE/	EDOUF/ADUF/CMDS							<b></b>					+		<del> </del>	<del></del>	+
		<del></del>			-T		1	1	1	1	1	1	1	1	1	1	
JUF/							i i										

			Ţ		Γ			RATES					oss	RATES	incremental	Incremental
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262		BCS	usoc				Nonre		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	incremental Charge - Manual Svc Order vs. Electronic-Add'i	Charge - Manual Svc Order vs. Electronic-Disc	Charge - Manual Svi Order vs. Electronic-Di
			1 1			Rec	Nonrec First	Add'I	First	nnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
100	SS OLEC Daily Usage File		Н	-		nec	r ii at	7441	1							
	ecording per message				1	\$0.0000044										
ODOF: Recording, per message	coloning per meccage		<b>i</b> —										ł .			
	SS OLEC Daily Usage File		1									1				
	essage Distribution per message		ļ.,			\$0.0027366			<u> </u>		<u> </u>					
OS	SS OLEC Daily Usage File		l		1						l .					1
ODUF: Message Processing, per Magnetic Me	essage Distribution per magnetic					\$52.75					i		l			
Tape provisioned tap	pe provisioned SS OLEC Daily Usage File Data		+			Ψ32.70			<b>-</b>							
	ansmission (Connect: Direct),				1	1										
	r message		l			\$0.0000339					<u> </u>					<del> </del>
(OCTATEOTISTICST), por massage																
UNBUNDLED PORT/LOOP COMBINATIONS - COS	ST BASED RATES															
Cost Based Rates are applied where BellSouth is																
required by FCC and/or State Commission rule to			-								i	1	1		1	
provide Unbundled Local Switching or Switch			1									ļ	i			
Ports. BellSouth is required to offer unbundled									ļ			1			ļ	
port/loop combinations that are Currently									1		ļ				1	
Combined in Tennessee unless the end user has											i .		ł		l	
four or more DS0 equivalent lines and is in Zone 1 of one of the Top 8 MSAs in BellSouth's region.					i			Ì					<u></u>			
In the absence of ordered rates by a State Commission, the recurring rates for combinations of port/loop network elements will the sum of the											į					
recurring rates for the UNEs which make up the									<b>!</b>				1		1	
combinations, and the nonrecurring rates shall			1						1		l				i	
be as set forth in this section.			4		1	ļ			ļ			<del>                                     </del>	<del> </del>	-	<del> </del>	+
Vertical features shall apply to the Unbundled																
Port/Loop Combination - Cost Based Rate											k		L		l	
section in the same manner as they are applied														İ		
to the Stand-Alone Unbundled Port section of							İ					1			l .	
this Rate Exhibit.			<b>-</b>		<u> </u>			-	<del> </del>	-	+	<del>                                      </del>				
End Office and Tandem Switching Usage and					1					Ì	1	Į.			1	
Common Transport Usage rates in the Port							İ					1				
section of this rate exhibit shall apply to all combinations of loop/port network elements							1		i	1					1	
except for UNE Coin Port/Loop Combinations			l								1	1				
which have a flat rate usage charge (USOC:		ļ		i			•		l			1				
URECU).			4	ļ	1			-	1	-	<del>                                     </del>		<del> </del>			+
For Currently Combined Combos in TN, the				1		1			1		1		1			
nonrecurring charges shall be those identified in									1			1			İ	
the Nonrecurring Charges - Currently Combined									1				t		1	
sections.								<del> </del>		<b>_</b>	<del>                                      </del>	+	<del>                                     </del>	<del> </del>	-	+
									<del></del>	<del> </del>		<del> </del>			<del> </del>	+
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE	PORT (RES)	<u> </u>	+	₩	+	<del></del>	<b>_</b>	+	1-	-	<del>                                     </del>	<del>                                     </del>	<b>†</b>		<b>†</b>	1
			+	1	+	1	1	+	<del> </del>	1	1	1			1	
UNE Port/Loop Combination Rates	-Wire Voice Grade Loop with 2-		+-	<del>                                     </del>	1		1	1	1						1	
	Vire Line Port - Zone 1		1	1_		\$14.18		1					<b>_</b>	<del></del>		+
	-Wire Voice Grade Loop with 2-	1	1	1	1		1		1		1	1				
	Vire Line Port - Zone 2		2	<u> </u>		\$18.01	L	ļ		-			4		<del> </del>	+
	-Wire Voice Grade Loop with 2-								1				1			
2-Wire VG Loop/Port Combo - Zone 3	Vire Line Port - Zone 3		3	Ļ		\$23.02	<b> </b>	1		+		<del> </del>	-	+	+	+
		ļ	$\bot$	<b>_</b>	-		<u> </u>		+	-	+	+	<del></del>	+	1	+
UNE Loop Rates						<u> </u>					_					

			Г	I	1		F	RATES					oss	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc				Nonre	curring	Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	incremental Charge - Manual Svc Order vs. Electronic-Disc	incremental Charge - Manual Svo Order vs. Electronic-Di
		0.202					Nonrec	urrino	Disco	nnect	per LSR	LSR	Electronic-1st	Electronic-Add'i	1st	Add'l
					1	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
O Wise Vision Conde Lang (CL1) Zono 1		**	1	LIEPRX	UEPLX	\$12.29										
2-Wire Voice Grade Loop (SL1) - Zone 1		**	2	UEPRX	UEPLX	\$16.12					<b>-</b>					
2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		**		UEPRX	UEPLX	\$21.13										
2-vvire voice drade Loop (3L1) - Zorie 3			Ť													
2-Wire Voice Grade Line Port Rates (Res)												ŀ				
		**		HEPRY	UEPRL	\$1.89				_						
2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID -			1	OLI IIX	102,	<del></del>										
		**	l	LIEPRX	UEPRC	\$1.89	ŀ						<u> </u>			
res 2-Wire voice unbundled port outgoing only -		<del> </del>	1	1021		<u> </u>										
		**		UEPRX	UEPRO	\$1.89										
res 2-Wire voice Grade unbundled Tennessee				† <del></del>	1	• • • • • • • • • • • • • • • • • • • •					1				Ì	
extended local dialing parity port with Caller				l .					ľ							
ID - res		**	1	UEPRX	UEPAQ	\$1.89	ļ.		l							
2-Wire voice unbundled Tennessee Area			t	1	1											
Calling port with Caller ID - res (F2R)		**	1	UEPRX	UEPAK	\$1.89		ļ	l							
2-Wire voice unbundled Tennessee Area			t —													
Calling port with Caller ID - res (TACER)		**	1	UEPRX	UEPAL	\$1.89			l		L					
2-Wire voice unbundled Tennessee Area				<u> </u>	1			1	1				1		1	
Calling port with Caller ID - res (TACSR)		**	1	UEPRX	UEPAM	\$1.89					1					
2-Wire voice unbundled Tennessee Area			1		1						1	1				
Calling port with Caller ID - res (1MF2X)		**	1	UEPRX	UEPAN	\$1.89						<b>1</b>				
2-Wire voice unbundled Tennessee Area			1										1		l	
Calling port with Caller ID - res (2MR)		**		UEPRX	UEPAO	\$1.89								ļ		
2-Wire voice unbundles res, low usage line							1	ļ			ı	l				
port with Caller ID (LUM)		**		UEPRX	UEPAP	\$1.89	<u> </u>				<u> </u>	<b></b>				
							ļ			<u> </u>			<del></del>			
FEATURES			1	1	l _		<u> </u>		<u> </u>		<u> </u>					
T LATORIES	[Exchange Ports includes all		1				i					1				
All Available Vertical Features	Applicable Features.]	**		UEPRX	UEPVF	\$0.00	\$0.00	\$0.00			1	1	ļ	-		
								<u> </u>	L							
LOCAL NUMBER PORTABILITY			1				<b>I</b>		1		<u> </u>	<u> </u>				
		**	1	UEPRX	LNPCX	\$0.35									i	
Local Number Portability (1 per port)		<del> </del>	+	10-11-11	,	T	†··	1								
			+	<b>├</b>			-				"	1				
NONRECURRING CHARGES (NRCs) - CURREN	TLY COMBINED		<b>↓</b>			<u> </u>	-	<del>                                     </del>	<del> </del>	+	<del>-</del>					
	2-Wire Voice Grade Loop/Line	1		LIEDDA	USAC2		\$1.03	\$0.29			1		\$30.89	\$7.03		
	Port Combo - Switch-as-is	—	+	UEFRA	USAUZ	<del> </del>	\$1.00	Ψ0.23	<b></b>		1					
2-Wire Voice Grade Loop / Line Port			ŀ	1	l l	Į.	I	1	ı		1		1			
Combination - Conversion - Switch with		*	l	LIEPRY	USACC	:	\$1.03	\$0.29	1		i	i	\$30.89	\$7.03		
change 2-Wire Voice Grade Loop / Line Port	2-Wire Voice Grade Loop/Line		+	102110	1 00, 100	-	4,1100		<u> </u>	T		1				
	Port Combo - Subsequent	ĺ	l.	ı	l							1	1	İ	1	
	Database Update		1	1		1	\$0.76						\$7.97			
Database Optiate		1	$\top$	$T^{-}$				1					L	1 .	L	l
ADDITIONAL NIDO		<del>                                     </del>	+	1	+	1	-		1			T		1	L	
ADDITIONAL NRCs		<del> </del>	+	1	+-										I	
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity		**		UEPR	K USAS2	:1	\$10.00	\$10.00	1		I	<u> </u>			L	
Compination - Subsequent Activity			+	1			1		Γ	1					<u> </u>	<u> </u>
<u> </u>	IE DODE (DUO)	-	+	+	+	<del> </del>		† · · · ·	<b>1</b>				1		[	
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LIN	E PORT (BUS)	1	+-	+	+		+	-	†	1	1		1		Ť ·	
			+		+-	<del></del>	+		+	<del> </del>	+	1	<del></del>			<b>†</b>
UNE Port/Loop Combination Rates			+				<del> </del>	+		+	+	1	1	1	1	†
	2-Wire Voice Grade Loop with 2-		1.		1	\$14.18			1			1	I		1	
2-Wire VG Loop/Port Combo - Zone 1	Wire Line Port - Zone 1	+	1	+	+-	<b>\$14.18</b>	<del> </del>		<del> </del>		<del></del>		1		<b>1</b>	1
	2-Wire Voice Grade Loop with 2-	1	2	1	1	\$18.01		i	1		1		1		1	
2-Wire VG Loop/Port Combo - Zone 2	Wire Line Port - Zone 2 2-Wire Voice Grade Loop with 2-	+	+-	+	+-	\$10.01	+	+	1 -	+						
			lз	1	I	\$23.02	1	1	1	1	1				1	
2-Wire VG Loop/Port Combo - Zone 3	Wire Line Port - Zone 3	L		<u> </u>		Ψ20.02				1						

<del> </del>			_				F	RATES					OSSI	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec			curring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increments Charge - Manual Sv Order vs. Electronic-D Add'i
						Rec	First	Addʻl	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			$\vdash$	-		1,50	- 11								ĺ	
			<b>—</b>			<b></b>			·				<b>1</b>			
UNE Loop Rates		**	<b>⊢</b> .		UEDLY	240.00							-			
2-Wire Voice Grade Loop (SL1) - Zone 1			1_	UEPBX	UEPLX	\$12.29					<b>-</b>		<del>                                     </del>			
2-Wire Voice Grade Loop (SL1) - Zone 2	. <u></u>	**	2	UEPBX	UEPLX	\$16.12						l				<del> </del>
2-Wire Voice Grade Loop (SL1) - Zone 3		**	3	UEPBX	UEPLX	\$21.13					1		ļ		<u> </u>	+
2 1110 10100 01000 01000																
2-Wire Voice Grade Line Port (Bus)		·		1	<u> </u>										<b></b>	
2-Wire voice unbundled port without Caller		<del></del>	$\vdash$				1					ı				1
ID - bus		**	1	UEPBX	UEPBL	\$1.89	1		l	l	<u> </u>	L				<del> </del>
2-Wire voice unbundled port with Caller +			1						I			1				
E484 ID - bus		**	1	UEPBX	UEPBC	\$1.89	İ		l		L					-
2-Wire voice unbundled port outgoing only -			<del>                                     </del>								ŀ	ŀ				
1 1		**	i i	UEPBX	UEPBO	\$1.89										↓
2-Wire voice Grade unbundled Tennessee			_	<u> </u>			1						i		1	
extended local dialing parity port with Caller					j		ŀ									
IID - bus		**		UEPBX	UEPAV	\$1.89					<u> </u>					+
2-Wire voice unbundled incoming only port			<b>†</b>								1					1
with Caller ID - Bus	1	**		UEPBX	UPEB1	\$1.89		i			L					
2-Wire voice unbundled Tennessee Bus 2-			T													
Way Area Calling Port Economy Option				1	1	1	l .		1			Į.			l	
(TACC1)		**	1	UEPBX	UEPAC	\$1.89	1		l		<u> </u>					
2-Wire voice unbundled Tennessee Bus 2-			<b>†</b> –								1	1	1			
Way Area Calling Port Standard Option			1		l		1									
(TACC2)		**	1	UEPBX	UEPAD	\$1.89	ı									
2-Wire voice unbundled Tennessee Bus 2-			†—	<u> </u>								1			Ī	
Way Collierville and Memphis Local Calling		İ	1										1		1	1
Port (B2F)	ĺ	**		UEPBX	UEPAE	\$1.89				ļ					<b>└</b>	
r or (bzr )			1												L	
			+	1			1					1	Į.		l	
LOCAL NUMBER PORTABILITY		**	╂	UEPBX	LNPCX	\$0.35				1						
Local Number Portability (1 per port)		-	╄	UCFBA	LINEUX	\$0.55	<del></del>		<del></del>			†				
		ļ	<b>-</b>		<b>├</b>			-	<del> </del>	+	+	<del>                                     </del>	<del>                                     </del>			1
FEATURES			1_		<u> </u>	<u> </u>		<u> </u>	<b>├</b>	+		<b>-</b>		-	<u> </u>	-
	[Exchange Ports includes all	**					\$0.00	\$0.00						İ	1	
All Available Vertical Features	Applicable Features.]	<u> </u>	1	DEPRY	UEPVF	\$0.00	\$0.00	\$0.00				+	·			
			1_			<u> </u>			<b>_</b>	1	+	<del></del>	<b>_</b>	<del> </del>	<del>                                     </del>	+
NONRECURRING CHARGES (NRCs) - CURREI	NTLY COMBINED		1	L	L						<b>_</b>	-	<b>—</b>		<del> </del>	+
2-Wire Voice Grade Loop / Line Port	2-Wire Voice Grade Loop/Line		Г				1 .	1.	1	1	1	1	400.00	\$7.03	1	
Combination - Conversion - Switch-as-is	Port Combo - Switch-as-is	.1	L	UEPBX	USAC2	<u></u>	\$1.03	\$0.29	<u> </u>		+		\$30.89	\$7.03	<del> </del>	+
2-Wire Voice Grade Loop / Line Port		1	Г		1	1	1		1	1	1				1	
Combination - Conversion - Switch with			1	1	1	.1	1		1		1	1	\$30.89	\$7.03	1	
change		*		UEPBX	USACC	·	\$1.03	\$0.29	<del> </del>	+-		<del>-</del>	\$30.89	\$7.03	1	+
2-Wire Voice Grade Loop / Line Port	2-Wire Voice Grade Loop/Line		1	1	1	1	1		1		1	1			1	
Combination - Conversion - Subsequent	Port Combo - Subsequent		1	1	1		1		1	1	1	1	\$7.97		1	
Database Update	Database Update	1		<u>↓</u>		L	\$0.76	<u> </u>	<b>_</b>	<del> </del>		<del> </del>	\$1.51	+	<del>                                     </del>	+
			1						<b>↓</b>	1	<b>—</b>			<del> </del>		+
ADDITIONAL NRCs			T								4			-	<del></del>	+-
2-Wire Voice Grade Loop/Line Port		T "	T				1	1.	Ī		1	1	1	1	1	[
Combination - Subsequent Activity		**		UEPBX	USAS2	<u> </u>	\$10.00	\$10.00							1	+
Cambridge Consolidation Consol			Т				1					1			$\bot$	+
- WHITE WOLDE OR ART LOOP WITH A WISSELL	NE DODT (DEC - DRY)		$\top$		1						1	1	I			
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LI	NE FUNT (NES-FBA)	+	+	+-	1	1	1	1	1	·					1	
	1	+	+		<del></del>	1	<del></del>	+	1		1	1	<u> </u>			1
UNE Port/Loop Combination Rates	O Million Creade Long with O	+	+		<del> </del>	+						- <del></del>	1			
	2-Wire Voice Grade Loop with 2-	1	1.	1	1	\$14.18	1			1	1	I			1	
2-Wire VG Loop/Port Combo - Zone 1	Wire Line Port - Zone 1	.1	1			1 914.10		1								

l l							7	RATES						RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec	urring	Nonrei		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'l
						Rec	First	Add'i	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 2		2			\$18.01										
	2-Wire Voice Grade Loop with 2-											1		ļ		
2-Wire VG Loop/Port Combo - Zone 3	Wire Line Port - Zone 3		3			\$23.02										
UNE Loop Rates																
2-Wire Voice Grade Loop (SL 1) - Zone 1		**	1	UEPRG	UEPLX	\$12.39										
		**	2	UEPRG	UEPLX	\$16.22							1			
2-Wire Voice Grade Loop (SL 1) - Zone 2		**	3	UEPRG	UEPLX	\$21.23										
2-Wire Voice Grade Loop (SL 1) - Zone 3			3	UEPRG	UEFLX	\$21.23										
2-Wire Voice Grade Line Port Rates (RES - PB)	K)											<u> </u>	<u> </u>			
2-Wire VG Unbundled Combination 2-Way		**											1			
PBX Trunk Port - Res			-	UEPRG	UEPRD	\$1.79		<del></del>	<del>                                     </del>							
LOCAL NUMBER PORTABILITY											ļ					<u> </u>
Local Number Portability (1 per port)		**		UEPRG	LNPCP	\$3.15			ł		Í					
Local Number Portability (1 per porty	-													ļ		<del></del>
FEATURES													ļ		<del> </del>	<del></del>
LATORES	[Exchange Ports includes all										1	1				
All Available Vertical Features	Applicable Features.]	**	_	UEPRG	UEPVF	\$0.00	\$0.00	\$0.00		-	<del>                                     </del>	<del>                                     </del>		1	<del>                                     </del>	+
			L.				<u> </u>		<del> </del>		1	+	<del>                                     </del>			1
NONRECURRING CHARGES (NRCs) - CURREN	TLY COMBINED		L						<b>-</b>	<del> </del>	<del>-</del>				<del>                                     </del>	1
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As- Is	-2-Wire Voice Grade Loop/Line Port Combo - Switch-as-is			UEPRG	USAC2		\$1.03	\$0.29					\$30.89	\$7.03		
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change				UEPRG	USACC		\$1.03	\$0.29	L				\$30.89	\$7.03		
2-Wire Voice Grade Loop / Line Port	2-Wire Voice Grade Loop/Line Port Combo - Subsequent						ļ				1					
Combination - Conversion - Subsequent Database Update	Database Update		L_				\$0.76			-	ļ		\$7.97		<u> </u>	<u> </u>
			<u> </u>						<u> </u>			<del> </del>	<b>-</b>			<del> </del>
ADDITIONAL NRCs		<del> </del>	<u> </u>	<u> </u>	<b>—</b>	<b> </b>	<del> </del>			<del> </del>	<del>                                     </del>	+	†			+
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity		**	<u> </u>	UEPRG	USAS2		\$10.00	\$10.00				-	\$30.89	\$7.03	<del>                                     </del>	<u> </u>
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LIF	NE PORT (BUS - PBX)	<u> </u>							<b>_</b>							
			1_	<b>Ļ</b>			<del>                                     </del>	<del>                                     </del>	<del> </del>	-	<del> </del>	+	<del> </del>	+	<del>                                     </del>	+
UNE Port/Loop Combination Rates	0.11	-	1—	<b>-</b>	<del> </del>	ļ	<del> </del>		<del> </del>	<del>                                     </del>	<del> </del>	+	<del> </del>	1	1	+
2-Wire VG Loop/Port Combo - Zone 1	2-Wire Voice Grade Loop with 2- Wire Line Port - Zone 1		1			\$14.18									<u> </u>	
	2-Wire Voice Grade Loop with 2-		T			610.01										
2-Wire VG Loop/Port Combo - Zone 2	Wire Line Port - Zone 2 2-Wire Voice Grade Loop with 2-	+	2	<u> </u>	-	\$18.01	<del>                                     </del>	<u> </u>	<del> </del>	+	1	<del>                                     </del>				
2-Wire VG Loop/Port Combo - Zone 3	Wire Line Port - Zone 3		3	1	<u> </u>	\$23.02									<del> </del>	
UNE Loop Rates				Ī					L			1	<u> </u>		1	+
2-Wire Voice Grade Loop (SL 1) - Zone 1		**	1			\$12.39	<del></del>		<del> </del>		<del>                                     </del>	+	+		<del> </del>	-
0 140 - 14-1 Orado 1 (OL 4) 2 0		**	2		UEPLX		<del></del>	<del>-</del>	<del> </del>	1	<del>                                     </del>			+	+	+
2-Wire Voice Grade Loop (SL 1) - Zone 2																
2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		**	3	UEPPX	UEPLX	\$21.23	1	<del>                                     </del>	<del> </del>	<del> </del>	-				<del>                                     </del>	

			Π		F 1		F	RATES					OSS	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262		BCS	usoc				Nonrec	•	Svc Order Submitted Elec	Svc Order Submitted Manually per	Svc Order vs.		Incremental Charge - Manual Svc Order vs. Electronic-Disc	Charge - Manual Svo Order vs. Electronic-Di
					1		Nonrec	urring Add'i	Disco: First	Add'l	per LSR SOMEC	LSR	SOMAN	Electronic-Add'i	SOMAN	SOMAN
			$\vdash$			Rec	First	A001	FIFBT	Addi	SOMEC	SOMAN	00	00		
Line Side Unbundled Combination 2-Way		**		LIEDDY	UEPPC	\$1.79	1		1							
PBX Trunk Port - Bus			$\vdash$	UEFFX	OEFFC	ψ1.7 <del>3</del>			1							
Line Side Unbundled Outward PBX Trunk		**		HEDDY	UEPPO	\$1.79			1			1				
Port - Bus			$\vdash$	OEFFX	OLITO	\$1.75										
Line Side Unbundled Incoming PBX Trunk		**	'	UEPPX	UEPP1	\$1.79	1									
Port - Bus				OLI I X	02111	Ψ1.70										
2-Wire Voice Unbundled PBX LD Terminal		**		HEPPY	UEPLD	\$1.79						l	1			
Ports				OLI IX	02. 22	<b>V</b> 1 <b>U</b>	-					1				
2-Wire Voice Unbundled 2-Way		**		UEPPX	UEPT2	\$1.79										
Combination PBX Tennessee Calling Port 2-Wire Voice Unbundled 1-Way Outgoing			<del> </del>	OZ. TA	<u> </u>											
PBX Tennessee Calling Port		**	1	UEPPX	UEPTO	\$1.79					1	l				
2-Wire Voice Unbundled 2-Way																
Combination PBX Usage Port		**		UEPPX	UEPXA	\$1.79			1							
2-Wire Voice Unbundled PBX Toll Terminal																
Hotel Ports		**		UEPPX	UEPXB	\$1.79										
2-Wire Voice Unbundled PBX LD DDD			$t^-$									•				
Terminals Port		**	1	UEPPX	UEPXC	\$1.79	l .									
2-Wire Voice Unbundled PBX LD Terminal			1	1									l .			
Switchboard Port		**		UEPPX	UEPXD	\$1.79					<u> </u>	L				
2-Wire Voice Unbundled PBX LD Terminal			T -								1	1				
Switchboard IDD Capable Port		**		UEPPX	UEPXE	\$1.79	l .				L.,.	<u> </u>				
2-Wire Voice Unbundled 2-Way PBX			t —	$\vdash$									Į.		ŀ	
Hotel/Hospital Economy Administrative			1				1				1		ł		i	
Calling Port		**	1	UEPPX	UEPXL	\$1.79					<u> </u>					<u> </u>
2-Wire Voice Unbundled 2-Way PBX		1	T								1	1				
Hotel/Hospital Economy Room Calling Port		**	ı	UEPPX	UEPXM	\$1.79			<u> </u>					-		
2-Wire Voice Unbundled 1-Way Outgoing								1	1		l.				i	
PBX Hotel/Hospital Economy Admin Calling				ļ.									1			
Port TN Calling Port		**		UEPPX	UEPXN	\$1.79			ļ		<b>-</b>	<del> </del>				
2-Wire Voice Unbundled 1-Way Outgoing			1			1			<u> </u>			1	Į.			
PBX Hotel/Hospital Discount Room Calling			1	l	l				1		1	1	į.		İ	ļ
Port		**	1	UEPPX	UEPXO	\$1.79	<u> </u>		<del></del>		<del></del>	<del>                                     </del>				<u> </u>
2-Wire Voice Unbundled 1-Way Outgoing		**	i i				i .			]	1	1				Ì
PBX Measured Port		**	1_	UEPPX	UEPXS	\$1.79							·	<del> </del>		_
2-Wire Voice Unbundled PBX Collierville		**			UEOVII	04.70		İ			i	1				
and Memphis Calling Port			<u> </u>	UEPPX	UEPXU	\$1.79	ļ	<del> </del> -	<del></del>		+	<del>                                     </del>	<del> </del>			-
2-Wire Voice Unbundled 2-Way PBX		**		LIEDDY	LUEDVA	\$1.79	i				l			1	1	
Tennessee RegionServ Callling Port				UEPPX	UEPXV	\$1.79	<b>!</b>	<del> </del>			-					
								<u> </u>			<del> </del>		<del> </del>	4	<del> </del>	<del>                                     </del>
CAL NUMBER PORTABILITY				<u> 1</u>	<u> </u>				ļ	ļ		<del>                                     </del>				
Local Number Portability (1 per port)		**		UEPPX	LNPCP	\$3.15						<u> </u>	<b></b>	-		<del> </del>
Eccar Normber 1 Citability (1 per port)			<b>†</b>								l					
1511050			+	<del>                                     </del>	<b>—</b> —								l		ļ	
ATURES	Exchange Ports includes all	+	+-	<b>1</b>	1	T										
All Available Vertical Features	Applicable Features.]	**	1	UEPPX	UEPVF	\$0.00	\$0.00	\$0.00	1		<u> </u>					-
All Avanable vertical Features	- ppoutro ( outer out		1	1	1		1	1				l	I			
	TI V COMPINED	+	╅┈╴	<del>                                     </del>	<b>1</b>										l	
ONRECURRING CHARGES (NRCs) - CURREN	ILT COMBINED	+	+	+	<del>                                     </del>		<del>                                     </del>	_		1	<u> </u>		1	1		
2-Wire Voice Grade Loop/ Line Port	2-Wire Voice Grade Loon/Line		1	1	1								1			
Combination (PBX) - Conversion - Switch-As-	Port Combo - Switch-as-is		1	LUEPPA	USAC2	1	\$1.03	\$0.29	l	1		1	\$30.89	\$7.03	1	
1s 2-Wire Voice Grade Loop/ Line Port	FOIL COUIDO - CANTON-40-10	+	1	1	1	1	1		1	1	T	<u> </u>				
Combination (PBX) - Conversion - Switch			1	1	1						1		I			
Combination (PDA) - Conversion - Switch			1	LUEDDY	USACC	.1	\$1.03	\$0.29	1		1	1	\$30.89	\$7.03	1	1

								ATES					OSS	RATES		
UNBUNDLED NETWORK ELEMENT		Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec		Nonre		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order va. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic-Di Add'i
			li			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Combination - Conversion - Subsequent	2-Wire Voice Grade Loop/Line Port Combo - Subsequent Database Update						\$0.76						\$7.97			
			-													
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity		**		UEPPX	USAS2		\$10.00	\$10.00					\$30.89	\$7.03		
2-WIRE VOICE GRADE LOOP- BUS ONLY - WIT	'H 2-WIRE DID TRUNK PORT															
2-WIRE VOICE GRADE EGG! BGG GRE!																
UNE Port/Loop Combination Rates														<b> </b>	ļ	<del> </del>
2-Wire VG Loop/2-Wire DID Trunk Port Combo - Zone 1	2-Wire Voice Grade Loop with 2- Wire DID Trunk Port - Zone 1 2-Wire Voice Grade Loop with 2-		1			\$18.38										
Combo - Zone 2	Wire DID Trunk Port - Zone 2 2-Wire Voice Grade Loop with 2-		2			\$19.87				-	ļ			ļ [		
	Wire DID Trunk Port - Zone 3		3			\$25.52				<u></u>						
			┢							<del></del>		1				
NONRECURRING CHARGES - CURRENTLY CO	2-Wire Voice Grade Loop with 2-		ł				<del></del> -					····			,	
1	Wire DID Trunk Port Combo -				l '							ŀ				1
Trunk Port Combination - Switch-as-is	Switch-as-is		<u> </u>	UEPPX	USAC1		\$8.76	\$5.75					\$41.43	\$9.80		
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination Conversion with Changes		*		UEPPX	USA1C		\$8.76	\$5.75			\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
			↓_				ļ		<del></del>			<del> </del>	1	-	<del>                                     </del>	-
Telephone Number/Trunk Group Establisment			l				ł				i	1	1			
Charges		**	┼	UEPPX	NDT	\$0.00	\$0.00	\$0.00			\$3.50		\$41.43	\$9.80	\$20.00	\$20.00
DID Trunk Termination (One Per Port)			1	DEFFX	ישוי	Ψ0.00	Ψ0.00	Ψ0.00	† — —	1	1111		1			T '
DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers		**	┡	UEPPX	NDZ	\$0.00	\$0.00	\$0.00			\$3.50	<u> </u>	\$41.43	\$9.80	\$20.00	\$20.0
Additional DID Numbers for each Group of 20 DID Numbers		**	<u> </u>	UEPPX	ND4	\$0.00	\$0.00	\$0.00	-		\$3.50		\$41.43	\$9.80	\$20.00	\$20.0
DID Numbers, Non- consecutive DID Numbers , Per Number		**		UEPPX	ND5	\$0.00	\$0.00	\$0.00	<u> </u>	-	\$3.50		\$41.43	\$9.80	\$20.00	\$20.0
LOCAL NUMBER PORTABILITY							ļ		<b>_</b>			ļ				<u> </u>
Local Number Portability (1 per port)		**	╁	UEPPX	LNPCP	\$3.15	\$0.00	\$0.00								
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-V	WIRE ISDN DIGITAL LINE SIDE POP	RT .	<u> </u>						<u> </u>	-	<del> </del>	<u> </u>		-		
		<del>                                     </del>	$\vdash$	<b>├</b>	<b>-</b>		<del>                                     </del>	<del>                                     </del>	<b> </b>	-		†	·-			1
UNE Port/Loop Combination Rates	2-Wire ISDN Digital Grade Loop		1	<b>†</b>	<b>†</b>	l						1				
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - Zone 1	with 2-Wire ISDN Digital Line Side Port - Zone 1		1		ļ	\$32.27			-	1	-	<del> </del>	\$19.99	\$19.99	\$20.00	\$20.0
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - Zone 2	2-Wire ISDN Digital Grade Loop with 2-Wire ISDN Digital Line Side Port - Zone 2		2	<u> </u>		\$34.78	<u></u>				-		\$19.99	\$19.99	\$20.00	\$20.0
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - Zone 3	2-Wire ISDN Digital Grade Loop with 2-Wire ISDN Digital Line Side Port - Zone 3		3		<u></u>	\$44.32							\$19.99	\$19.99	\$20.00	\$20.0
	AMBINED.	-	+	1	-	<u> </u>	<del>                                     </del>		-	-	1	<del> </del>	<del>-</del>			
NONRECURRING CHARGES - CURRENTLY C	OMBINED											-	-			

								RATES					OSSI	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262		BCS	usoc				7	curring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Charge - Manual Svc Order vs. Electronic-Dis Add'i
						Rec	Nonrec First	urring Add'l	First	Add'1	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire ISDN Digital Grade Loop / 2-Wire	2-Wire ISDN Digital Grade Loop 2-		┢			1100										
ISDN Line Side Port Combination - Conversion	Wire ISDN Line Side Port Combo - Switch-as-is		<u> </u>	UEPPB	USACB		\$117.23	\$117.23			\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
ADDITIONAL NRCs		-	E													
2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination-Non Feature Subs Activity	2-Wire ISDN Digital Grade Loop 2- Wire ISDN Line Side Port Combo- Non Feature Subsequent Activity		<u> </u>	UEPPB	USASB		\$212.88				\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
LOCAL NUMBER PORTABILITY		-	t	_												
Local Number Portability (1 per port)		**		UEPPR	LNPCX	\$0.35	\$0.00	\$0.00	<u> </u>							
			<b>-</b>				ļ.——·		-				1			
B-CHANNEL USER PROFILE ACCESS:		**	┼┈	HEPPR	U1UCA	\$0.00	\$0.00	\$0.00					T			
CVS/CSD (DMS/5ESS)		**	$\vdash$	L	U1UCB	\$0.00	\$0.00	\$0.00		T	1					
CVS (EWSD)		**	$\dagger$		U1UCC	\$0.00	\$0.00	\$0.00								
005				Ĺ					<u> </u>			ļ			ļ	
B-CHANNEL AREA PLUS USER PROFILE			T								1	Ì			1	
ACCESS: (AL,KY,LA,MS SC,MS, & TN)		**	1	UEPPB	U1UCD	\$0.00	\$0.00	\$0.00	<u> </u>	<u> </u>		-				
CVS/CSD (DMS/5ESS) CVS (EWSD)		**	1	UEPPB	U1UCE	\$0.00	\$0.00	\$0.00								
CSD -		**	T	UEPPB	U1UCF	\$0.00	\$0.00	\$0.00								<u> </u>
		**	1	I				20.00			<u> </u>	ļ		-		-
USER TERMINAL SERVICE PROFILE (EWSD		**	╁	UEPPB	U1UMA	\$0.00	\$0.00	\$0.00			<u> </u>					
VERTICAL FEATURES			1										1			
One per Channel B User Profile		**		UEPPB	UEPVF	\$0.00	\$0.00	\$0.00	<u> </u>	<del> </del>		<del>                                     </del>		<del> </del>		<u> </u>
							<u> </u>		<del>                                     </del>		<del>                                     </del>	<del> </del>	-			1
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE IS	ON DS1 DIGITAL TRUNK PORT		+				<del>                                     </del>	<del>                                     </del>	<b>-</b> -							
UNE Port/Loop Combination Rates			+-	<del> </del>				1								
UNE PONTLOOP Combination hates	4-Wire DS1 Digital Loop with 4-		+										1		1	
4W DS1 Digital Loop/4W ISDN DS1 Digital		-	١.			\$132.58	1		ŀ		1		\$19.99	\$19.99	\$20.00	\$20.00
Trunk Port - Zone 1	Zone 1 4-Wire DS1 Digital Loop with 4-		<b>+</b> ∸		<u> </u>	\$132.56	·				<b>-</b>	1			<b>T</b>	
4W DS1 Digital Loop/4W ISDN DS1 Digita		-		1								1	\$19.99	\$19.99	\$20.00	\$20.00
Trunk Port - Zone 2	Zone 2	-	2	<b>↓</b>		\$150.25				<del> </del>			\$19.99	\$19.99	\$20.00	\$20.00
4W DS1 Digital Loop/4W ISDN DS1 Digita	4-Wire DS1 Digital Loop with 4- Wire ISDN DS1 Digital Trunk Port	_	1	1	1							1			1.	
Trunk Port - Zone 3	Zone 3		3		ļ	\$173.44	<u> </u>			-	<del> </del>		\$19.99	\$19.99	\$20.00	\$20.00
		L	1	ļ	<u> </u>					-	<u> </u>		-		<b>+</b>	ļ
NONRECURRING CHARGES - CURRENTLY	COMBINED		+	1	1-		+	+	1	+	+	<del>                                     </del>	<del>                                     </del>	+	<del>                                     </del>	
4-Wire DS1 Digital Loop / 4-Wire ISDN DS Digital Trunk Port Combination - Conversion	on-Wire ISDN DS1 Digital Trunk Port			1			1		1			1	1			
Switch-as-is	Combo - Switch-as-is		┿.	UEPPF	USACP		\$328.53	\$328.53	-	-		<del> </del>	\$19.99	\$19.99	\$20.00	\$20.00
ADDITIONAL MOO		+	+	1-	1		<del>                                     </del>	+	-	†	<u> </u>					<u> </u>
ADDITIONAL NRCs		+	$T^-$					1								
4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Trunk Port Combination-Sub Channel Activation-Per Channel	4-Wire DS1 Digital Loop with 4- Wire ISDN DS1 Digital Trunk Port Combo - Subsequent Channel Activation - Per Channel			HEDDE	USASP		\$28.39						\$19.99	\$19.99	\$20.00	\$20.00

1				ГΠ				R	RATES					OSS	RATES		
	UNBUNDLED NETWORK ELEMENT		Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonreci	urring	Nonrec		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Charge - Manual Sv Order vs. Electronic-D Add'i
						t	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
T	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig	4-Wire DS1 Digital Loop with 4- Wire ISDN DS1 Digital Trunk Port															
L		way Telephone Numbers  4-Wire DS1 Digital Loop with 4-			UEPPP	PR7TG		\$0.94						\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Trunk Port Combination-Sub Outward Telephone Numbers	Wire ISDN DS1 Digital Trunk Port Combo - Subsequent Outward Telephone Numbers 4-Wire DS1 Digital Loop with 4-			UEPPP	PR7TP		\$22.36						\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Trunk Port Combination-Subsequent Inward Telephone Numbers	Wire ISDN DS1 Digital Trunk Port			UEPPP	PR7ZT		\$44.71						\$19.99	\$19.99	\$20.00	\$20.00
	4-Wire DS1 Dig Loop / 4-Wire ISDN DS1 Dig Trunk Port Combination-Subsequent Service	Wire ISDN DS1 Digital Trunk Port			UEPPP	USASP		\$189.76					-	\$19. <del>9</del> 9	\$19.99	\$20.00	\$20.00
LO	DCAL NUMBER PORTABILITY																
Ŧ	Local Number Portability (1 per port)		**		UEPPP	LNPCN	\$1.75										
IN'	TERFACE (Provsioning Only)				L				<u></u>				<del>                                     </del>	<del> </del>			-
$\top$	Voice/Data		**	<u> </u>	UEPPP	PR71V	\$0.00		<u> </u>				<del></del>	<del>                                     </del>			
	Digital Data		**		UEPPP	PR71D	\$0.00			ļ			<u> </u>	<del> </del>			
#	Inward Data		**	-	UEPPP	PR71E	\$0.00										
C/	ALL TYPES																
1	Inward		**		UEPPP	PR7C1	\$0.00						<b>_</b>				<u> </u>
+	Outward		**	T	UEPPP	PR7C0	\$0.00							<u> </u>			
$\pm$	Two-way		**		UEPPP	PR7CC	\$0.00										
4-1	WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDI	rs trunk port		1													-
L				↓	<b>_</b>	L						<del> </del>	<del>                                     </del>				<u> </u>
UN	NE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port	4-Wire DS1 Digital Loop With 4- Wire DID Trunk Port - Zone 1		<u> </u>			\$93.28				-		.,				
+	Zone 1  4W DS1 Digital Loop/4W DDITS Trunk Port Zone 2	4-Wire DIST Digital Loop With 4- Wire DID Trunk Port - Zone 2		1 2			\$110.95										
$^{\dagger}$	4W DS1 Digital Loop/4W DDITS Trunk Port Zone 3	4-Wire DS1 Digital Loop With 4- Wire DID Trunk Port - Zone 3		3			\$134.14										-
$\perp$			<u> </u>	+-	<del> </del>	<u> </u>					<del> </del>	+	<del>                                     </del>	1		<b></b>	
N	ONRECURRING CHARGES - CURRENTLY CO	4-Wire DS1 Digital Loop/4-Wire		1	<del>                                     </del>						-	<u> </u>					1
+	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is 4-Wire DS1 Digital Loop / 4-Wire DDITS	DID Trunk Port Combo - Switch-as- is		$\vdash$	UEPDC	USAC4		\$312.91	\$312.91			\$3.50		\$19.99	\$19.99	\$20.00	\$20.
+	Trunk Port Combination - Conversion with DS1 Changes  4-Wire DS1 Digital Loop / 4-Wire DDITS	4-Wire DS1 Digital Loop/4-Wire DID Trunk Port Combo - Switch-as		+	UEPDC	USAWA		\$312.91	\$312.91			\$3.50		\$19.99	\$19.99	\$20.00	\$20.
1	Trunk Port Combination - Conversion with Change - Trunk	is		1	UEPDC	USAWB		\$312.91	\$312.91	1		\$3.50	ļ	\$19.99	\$19.99	\$20.00	\$20.

					I			R	ATES					OSS	RATES		
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc				Nonrec		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Increment Charge Manual S Order vi Electronic- Add'i
l l								Nonrect		First	Add'i	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				$\sqcup$			Rec	First	Add'l	FH'ST	AUG I	SOMEO					
		4-Wire DS1 Digital Loop 4-Wire						1				į	1				
١.	4-Wire DS1 Dig Loop / 4-Wire DDITS Trunk	DID Trunk Port Combo -		1				1				1		1			
	Port Combination-Subsequent Service	Subsequent Service Order Per			1	1		1				l		\$19.99	\$19.99	\$20.00	\$20.00
	Activity Per Svc Order	Order		'	UEPDC	USAS4		\$94.88	\$94.88			\$3.50		\$19.99	\$19.99	\$20.00	φ20.0
	Activity 1 et ove order	4-Wire DS1 Digital Loop/4-Wire		1								i	1		l		
-	4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port	DID Trunk Port Combo -				i 1								1	1		
ľ	Combination-NRC-Sub Channel Activation-	Subsequent Channel Activation	ì		1	1		,									
	Combination Time Can Chamber	Per Channel			UEPDC	UDTTA		\$108.67				\$3.50	<u> </u>	\$19.99	\$19.99	\$20.00	\$20.0
	Per Chani-2-Way Trk	4-Wire DS1 Digital Loop/4-Wire		<del> </del> —										ŀ			
	DOLDLI	DID Truck Bort Combo -						1						Ì			
١.	4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port	DID Trunk Port Combo -		1		1							l.				
	Combo-Sub Channel Activation-Per Chanl-1-	Subsequent Channel Activation	ļ		UEPDC	LIDTER		\$108.67				\$3.50	1	\$19.99	\$19.99	\$20.00	\$20.0
- 1	Way Outward Trk	Per Channel	<u> </u>	1—	OEFDO	00110		\$100.07									
		4-Wire DS1 Digital Loop/4-Wire		1	i 1			1						1		i	
Į.	4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port	DID Trunk Port Combo -		1		1		l i					l .	1		1	
1	Combo-Sub Chan Activation-Per Chan-1-	Subsequent Channel Activation	!	1								\$3.50	1	\$19.99	\$19.99	\$20.00	\$20.0
	Way Inwrd Trk w/o DID	Per Channel		1	UEPDC	UDTTC		\$108.67				\$3.50	-	Ψ13.33	ψ10.00	<del>                                      </del>	¥_5
-		4-Wire DS1 Digital Loop/4-Wire						1							İ	1	
- 1	4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port	DID Trunk Port Combo -		1									1				1
- 1	Combo-Sub Chan Activation-Per Chan-1-	Subsequent Channel Activation		1									1	****	640.00	\$20.00	\$20.6
	Way Inwd Trk with DID	Per Channel		1	UEPDC	UDTTD		\$108.67				\$3.50		\$19.99	\$19.99	\$20.00	\$20.
-	way inwo Trk with DID	4-Wire DS1 Digital Loop/4-Wire		+-				1					l .		İ		
- 1	4-Wire DS1 Dig Loop/4-Wire DDITS Trk Port	DID Trunk Port Combo		1				1		ŀ		l	1			li .	
	4-Wire DS1 Dig Loop/4-Wire DDITS TIK FOR	Subsequent Channel Activation	1		l			i		1		i	1			•	1 .
	Combo-Sub Chan Activ'n-Per Chan-2-Way				HEDDO	UDTTE		\$108.67				\$3.50	l	\$19.99	\$19.99	\$20.00	\$20.
	DID with User Trans	Per Channel		╁	OLFDO	ODITE		+ 4,00.07				34-		T			
	4-Wire DS1 Digital Loop / 4-Wire DDITS	4-Wire DS1 Digital Loop 4-Wire		1		i I				İ		ł	1				1
	Trunk Port Combination - Subsequent	DID Trunk Port Combo -				1		\$22.92				\$3.50		\$19.99	\$19.99	\$20.00	\$20.
	Signaling Changes	Subsequent Signaling Changes			UEPDC			\$22.92	<b>_</b>			Ψ0.00	· · · · ·		+		
		4-Wire DS1 Digital Loop 4-Wire	1	1				1		i			1			1	
	4-Wire DS1 Digital Loop / 4-Wire DDITS	DID Trunk Port Combo -		1	i			1		1		1		1			
	Trunk Port Combination - Subsequent	Subsequent Service Order Per		1	1					l		\$3.50		\$19.99	\$19.99	\$20.00	\$20.
	Service Order Per Order	Order		1	UEPDC	USAS4		\$94.88	\$94.88	L	<u> </u>	\$3.50		φ13.33	Ψ19.55	- <del>VEO.00</del>	+ +
_	DETVICE CITACIT OF CITACI							1		1	1	i		1			
		4-Wire DS1 Digital Loop 4-Wire			1							1	l .	1	i	1	
	4-Wire DS1 Dig Loop/4-Wire ISDN DS1	4-Wire DST Digital Loop 4-Wire	1	1		l .										1	
	Digital Trunk Port Combination - Subsequen	DID Trunk Port Combo -	İ	1	UEPDC	1	l .	\$88.68		1		\$3.50		\$19.99	\$19.99	\$20.00	\$20.
	Telephone Numbers	Subsequent Telephone Numbers	ļ	4—	UEPUC	<del></del>		\$50.00	<del> </del>			44					
BIP	OLAR 8 ZERO SUBSTITUTION			_				ļ		<del> </del>		-			-	+	_
	Superframe Format - Conversion or New				1				****	1		l .	1			L	
	install		**		UEPDC	CC0SF	\$0.00	\$0.00	\$0.00			<b>-</b>			<del> </del>	<del> </del>	+
_	Extended Superframe Format - Conversion			T		I				Į.							İ
	or New Install		**		UEPDC	CC0EF	\$0.00	\$0.00	\$0.00	<u> 1</u>			<u> </u>			_	_
	or New Iristali			_	<b>1</b>								l .	1			***
	a company and Amitoday		**	1	UEPDC	CCOSF	\$0.00	\$590.00	\$590.00			\$3.50		\$19.99	\$19.99	\$20.00	\$20.
	Superframe Format - Subsequent Activity		+	+	102.20	10000		1	<u> </u>							l l	
	Extended Superframe Format - Subsequent		**	1	LIEBRO	CCOEF	\$0.00	\$590.00	\$590.00	1		\$3.50		\$19.99	\$19.99	\$20.00	\$20.
	Activity		+	+-	TOEFDO	1000	Ψ0.00	1 4000.00	2000.00	-		1	1	1			
				┷	<del></del>	1	<b> </b>	<b>-</b>		+	+	<del>                                     </del>	+		-	1	1
AH	ernate Mark Inversion					<u> </u>	<b></b>	<del></del>		<b>├</b> ──	<del> </del>			<del>                                     </del>	+	1	1
				1	1	1	I	1		1			1			l	1
	Superframe Format		**	1	UEPDC	MCOSF	L	\$0.00	\$0.00		J	<del></del>	<del></del>	<del> </del>	+	<del></del>	<del> </del>
	oupontaino i onnai		T	T	T			1	1	I		1	1	1	1	1	
	Extended SuperFrame Format		**	1	UEPDO	MCOPC	l	\$0.00	\$0.00		L						
_	Extended SuperFrame Format		+	+	1	1						1					
			+	4-	+-	+		+	+	<b>1</b> -		1					
Ŧ-1	lephone Number/Trunk Group Establismer	t	1	1	1	1	I	I		1	1	1	1	I	1	i	
1 61	arges				<del>_</del>	<b>_</b>	<b>ļ.</b>	+	<del> </del>	+	+	1		610.00	\$19.99	\$20.00	\$20
			**		Lucaso	LUCTON	\$0.00	1	1	1	1	\$3.50	1	\$19.99	p19.99	<b>⊅∠∪.∪∪</b>	\$2U
			**	1	UEPDO	UDTGX	\$0.00										- 1
	Telephone Number for 2-Way Trunk Group Telephone Number for 1-Way Outward		**	+	UEPDC	TODIGX	\$0.00	+	-		1	\$3.50		\$19.99	\$19.99	\$20.00	\$20

						,								220	RATES		
1									RATES					0,33	I I	Incremental	Incremental
			Not in TRA									Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
-	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	usoc				Nonre	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs. Electronic-Disc	Order vs. Electronic-Dis
	i	STATED IN DOORE. S. CIECE	01262	1				Nonrec		Dieco	nnect	Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'i	1st	Add'i
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-	Telephone Number for 1-Way Inward Trunk			-												200.00	\$20.00
-	Group Without DID		**	i	UEPDC	UDTGZ	\$0.00					\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	DID Numbers, Establish Trunk Group and		**				** **			ļ		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	Provide First Group of 20 DID Numbers		**	<b>├</b>	UEPDC	NDZ	\$0.00					Ψ0.00			*		
	Additional DID Numbers for each Group of		**	l	UEPDC	ND4	\$0.00				!	\$3.50	<u> </u>	\$19.99	\$19.99	\$20.00	\$20.00
_	20 DID Numbers DID Numbers, Non- consecutive DID			<del>                                     </del>	02.00									***	***	\$20.00	\$20.00
	Numbers , Per Number		**	乚	UEPDC	ND5	\$0.00					\$3.50	<u> </u>	\$19.99	\$19.99	\$20.00	\$20.00
												<b>├</b> ─-				<del> </del>	
Dec	dicated DS1 (Interoffice Channel Mileage) -			l	ŀ	l				1	l			l			
	/FCO for 4-Wire DS1 Digital Loop with 4-			l	1	1											<u> </u>
Wir	re DDITS Trunk Port Interoffice Channel Mileage - Fixed rate 0-8			<b>-</b>		<b></b>									010.00	****	\$20.00
İ	miles (Facilities Termination)		**	l	UEPDC	1LNO1	\$75.83			ļ		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	Interoffice Channel Mileage - Additional rate		**									\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
	per mile - 0-8 miles		**	₩	UEPDC	1LNOA	\$0.3525			<u> </u>	<del></del>	\$0.50		<del>- 0.0.00</del>		1 ,	
	Interoffice Channel Mileage - Fixed rate 9-25		**	1	UEPDO	1LNO2	\$0.0000	1								L	
+	miles (Facilities Termination) Interoffice Channel Mileage - Additional rate			_	102,00	1	*******									200.00	\$20.00
	per mile - 9-25 miles		**	١.	UEPDO	1LNOB	\$0.3525				<u> </u>	\$3.50	<b>_</b>	\$19.99	\$19.99	\$20.00	\$20.00
	Interoffice Channel Mileage - Fixed rate 25+			П	l								ł	1			
1	miles (Facilities Termination)		**	<del> </del> —	UEPDO	1LNO3	\$0.0000	$\vdash$		<del>                                     </del>	<u> </u>	<del>                                     </del>					
	Interoffice Channel Mileage - Additional rate		**	1	UEPDO	1LNOC	\$0.3525			1		\$3.50		\$19.99	\$19.99	\$20.00	\$20.00
_	per mile - 25+ miles		**	┼	UEPDO	LNPCP	\$3.15									L	
_	Local Number Portability, per DS0 Activated		**	╁	UEPDO		\$0.00	<del>                                     </del>	<del>                                     </del>							L	
	Central Office Termininating Point	· · · · · · · · · · · · · · · · · · ·	+	}	TOE! DO	1-0.0	Ψ0.00		-	$t^-$		T -					
			<del> </del>	1	<del> </del>	<del>                                     </del>		<b>-</b>		<b>†</b>							
IHAN	CED EXTENDED LINK (EELs)			+-		†											ļ
_			+	1	1	<del> </del>	<u> </u>							Į.		1	
	NOTE: New (not currently combined)				-	1	Ļ	1		1			l		l		
	loop/transport (EEL) combinations are only vailable in density Zone 1 of the Nashville MSA						1	l					1			ŀ	
av	in TN for customers w/ 4 or more lines.							1		<b>_</b>		<del> </del>	<del>                                     </del>	<b>-</b>	-	<del>                                      </del>	+
F	or Currently Combined EELs, network element						į.								1		
	ecurring and Switch As Is Charges apply. For						1	1		1		1		i .			
İ	new EELs, network element recurring and						l .										
	nonrecurring apply (no Switch As Is Charge).		+	1		1											
2-1	WIRE VOICE GRADE EXTENDED LOOP WITH	DEDICATED DS1 INTEROFFICE	TRANSP	ORT	(EEL)								<del></del>	<u> </u>	-	ļ	<del></del>
-  -	WITE TOIGE GIBBE EXTERNS ESTATEMENT	2-Wire Voice Grade Extended						1		1			İ				
		Loop With DS1 Dedicated		1	1	1						1		1		1	ĺ
	First 2-Wire Analog Voice Grade Loop - SL2/DS1 Interofficed Transport Combination	Interoffice Transport (First 2-Wire	'	1	1	ŀ	İ								İ	Į.	
	Zone 1	mileage) - Zone 1		1	UNCV	K UEAL2	\$176.10									<u> </u>	<del></del>
	Zone i	2-Wire Voice Grade Extended	T	1										1	ŀ		
		Loop With DS1 Dedicated		1	1	1				1		1		1		1	
	First 2-Wire Analog Voice Grade Loop -	Interoffice Transport (First 2-Wire	'		1					1		1	1				
	SL2/DS1 Interofficed Transport Combination	mileage) - Zone 2		2	UNCV	X UEAL2	\$181.17									<u> </u>	
	Zone 2	2-Wire Voice Grade Extended	<del>                                     </del>	+	1									1			
		Loop With DS1 Dedicated		1		1	1	Į.				1	1	1		1	
	First 2-Wire Analog Voice Grade Loop -	Interoffice Transport (First 2-Wire	9	1						1		1	}			1	
	SL2/DS1 Interofficed Transport Combination	Voice Grade with DS1 excluding		1 2	UNCV	X UEAL2	\$187.82			1		l					
	Zone 3	mileage) - Zone 3		+	15,404	T	+	+		1	-	T			-	1	
	Interoffice Transport - Dedicated - DS1	Interoffice Transport Dedicated					t .		1						1		

									RATES					OSS	RATES		
									TAILS					T	1	Incremental	Incremental Charge -
			Not in TRA									Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Manual Svc
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	USOC	ļ			Nonre	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs. Electronic-Disc
		<b>3</b> 1000000000000000000000000000000000000	01262					Nonrec		Disco		Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'i	Electronic-Disc	Add'l
				i			Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			<del> </del>	<del> </del>			nec	1.11.01	7								
		2-Wire Voice Grade Extended								l		ŀ		l		1	-
		Loop With DS1 Dedicated										1	İ		1		
	Each Additional 2-Wire Analog VG Loop-	Interoffice Transport (Additional 2-			ļ			Ļ				1					
		Wire Voice Grade Loop in same		1				l					1				
	Combination-Zone 1	DS1 excluding mileage) - Zone 1		1	UNCVX	UEAL2	\$17.47						<del> </del>	<del> </del>		1	
				l						l		l	ì	1			
		2-Wire Voice Grade Extended	ļ	i	ļ												
		Loop With DS1 Dedicated Interoffice Transport (Additional 2-		1	l		1	<u> </u>									1
	Each Additional 2-Wire Analog VG Loop-	Wire Voice Grade Loop in same					1									1	
	SL2 in the same DS1 Interoffice Transport Combination-Zone 2	DS1 excluding mileage) - Zone 2		2	UNCVX	UEAL2	\$22.54							<u> </u>		<u> </u>	
	Combination-Zone 2	D31 excluding nineage) Zone Z		<b>†</b> -=−													
		2-Wire Voice Grade Extended		1	1		1	1									
		Loop With DS1 Dedicated		1		l		1		1			Į.	1		]	
İ	Each Additional 2-Wire Analog VG Loop-	Interoffice Transport (Additional 2-		l										1	ļ	1	
	SL2 in the same DS1 Interoffice Transport	Wire Voice Grade Loop in same		ı		1				i		l .	1				
	Combination-Zone 3	DS1 excluding mileage) - Zone 3		3	UNCVX	UEAL2	\$29.19					<del> </del>		<del> </del>		<del>                                     </del>	<del></del>
		Non-Recurring Cost for Extended		ı							ļ	1	I				
	1	Loop or Local Channel and Interoffice Combination - Switch-as				l				l			1	1			
	Nonrecurring Currently Combined Network	is Interoffice Combination - Switch-as	5		LINCIX	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12		1				
	Elements Switch -As-Is Charge	IS	+	╁	ONOIX	0.1000		7	,					,,,,,,	1		
	-WIRE VOICE GRADE EXTENDED LOOP WIT	U DEDICATED DEL INTERCEICE	TDANSD	OPT (	(EEL)			<del>                                     </del>				***					
4	-WIRE VOICE GRADE EXTENDED LOOP WIT	4-Wire Voice Grade Extended	INANSF	T	T T	<u> </u>		<del>                                     </del>									
		Loop With DS1 Dedicated		1	1					1		1	1				
		Interoffice Transport (First 4-Wire								1		1	1				ĺ
İ		56 or 64 kbps Digital Grade Loop		l				1				1	1			1	
	First 4-Wire Analog Voice Grade Loop/DS1	with DS1 excluding mileage) -		1									l .				
	Interoffice Transport Combination - Zone 1	Zone 1		1	UNCVX	UEAL4	\$184.24	<u> </u>	<del></del>	<del>                                     </del>		<del>                                      </del>	<b>-</b>		<del></del>	<b> </b>	
		4-Wire Voice Grade Extended			1			į				ı	1				
		Loop With DS1 Dedicated Interoffice Transport (First 4-Wire		1	l	İ	L						1				
		56 or 64 kbps Digital Grade Loop		1	1	1	ļ.						ı	1			
1	First 4-Wire Analog Voice Grade Loop/DS1	with DS1 excluding mileage) -			1							1		1		l	i
	Interoffice Transport Combination - Zone 2			2	UNCVX	UEAL4	\$191.80					<u> </u>					ļ
	interonice transport combination 25/16 2	4-Wire Voice Grade Extended			Ť	1										1	
		Loop With DS1 Dedicated			1												
		Interoffice Transport (First 4-Wire								1		l .	1	1		i	
		56 or 64 kbps Digital Grade Loop		1	1	1		1			Ì	i					
	First 4-Wire Analog Voice Grade Loop/DS1	with DS1 excluding mileage) -		1 3	LINO	UEAL4	\$201.72					1	1				
	Interoffice Transport Combination - Zone 3	Zone 3 Interoffice Transport - Dedicated -	+	13	UNCVA	UEAL	\$201.72	<del>                                     </del>		<del>                                     </del>		1		<b>T</b>			
	Interoffice Transport - Dedicated - DS1	DS1 - Per Mile			UNC1X	1L5XX	\$0.3562			1		İ	<u>L</u>	1		<u> </u>	
-	combination - Per Mile Per Month	4-Wire Voice Grade Extended	+~	+	1	1	1			1	T*						
		Loop With DS1 Dedicated			1	I	1		1					1	1	1	
	Additional 4-Wire Analog Voice Grade Loop	Interoffice Transport (Additional 4	-	1	1	1	1	1		1				1		1	
	in same DS1 Interoffice Transport	Wire 56 or 64 kbps in same DS1		1	1					1		1	1				
	Combination - Zone 1	excluding mileage) -Zone 1		1	UNCVX	UEAL4	\$25.61	<u> </u>	-	+		+		<del>                                     </del>	+	+	+
		4-Wire Voice Grade Extended		1	1	1		1		1		1	1	1	1	1	
		Loop With DS1 Dedicated		1	1		1	1		1			1			1	
	Additional 4-Wire Analog Voice Grade Loop	Interoffice Transport (Additional 4 Wire 56 or 64 kbps in same DS1	1			1	1	1				1				I	
	in same DS1 Interoffice Transport	excluding mileage) -Zone 2		1 2	UNCV	( UEAL4	\$33.17	1		1			1	1			
i	Combination - Zone 2	excluding mileage) -Zone Z			10.1017												

_	<del></del>		T					F	RATES					oss	RATES		
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc				Nonrec	urring	Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Dia
								Nonrec		Disco		per LSR	LSR	Electronic-1st	Electronic-Add'i	1st	Add'I SOMAN
			_				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
$\overline{}$		4-Wire Voice Grade Extended											1				
		Loop With DS1 Dedicated						l									
	, additional transfer and a series	Interoffice Transport (Additional 4-						l i									
		Wire 56 or 64 kbps in same DS1		ا ا	UNCVX	LIEALA	\$43.09	1					ļ		ļ		
		excluding mileage) -Zone 3 Non-Recurring Cost for Extended		٦	UNCVA	UEAL4	\$40.05						<b>-</b>				
		Loop or Local Channel and		Ĭ '	1							1					İ
		Interoffice combination Switch-as-												1			
	Elements Switch -As-Is Charge	is			UNC1X	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12		L				
	Elements Switch -AS-15 Charge																
-	WIRE 56 KBPS EXTENDED DIGITAL LOOP W	THE DEDICATED DS1 INTEROFF	CE TRANS	SPOR	T (EEL)	i										<u> </u>	
4-	WIRE 56 KBPS EXTENDED DIGITAL LOOP W	4-Wire 56 or 64 kbps Extended		T	I (===,	<b>-</b>											
		Digital Loop With Dedicated DS1		l		ł		1									
1		Interoffice Transport (First 4-Wire			]	1		1				l		Į.			
		56 or 64 kbps Digital Grade Loop		1	1			1				1		1			
	Thou to the deliberation and the second			١.	l			1						I			
	Interoffice Transport Combination - Zone 1	Zone 1		1	UNCDX	UDL56	\$190.64						<del>                                     </del>	<del>-</del>			
		4-Wire 56 or 64 kbps Extended	İ	1									i	ļ		j	
		Digital Loop With Dedicated DS1 Interoffice Transport (First 4-Wire		1								1		1		1	}
		56 or 64 kbps Digital Grade Loop		1		ļ.	i	i									i
	First 4-Wire 56Kbps Digital Grade Loop/DS1			l l		Ī		ì					1			l	
- 1	Interoffice Transport Combination - Zone 2	Zone 2	i	1 2	UNCDX	UDL56	\$200.16		ļ			1	1				ļ
-	Interoffice Transport Combination - Zone 2	4-Wire 56 or 64 kbps Extended	1	亡	10110211		1										
		Digital Loop With Dedicated DS1		ı		Į.		1									
		Interoffice Transport (First 4-Wire	1	1	l	1				l			ł			ŀ	
		56 or 64 kbps Digital Grade Loop		1	1		1			l		1		1			
ı	First 4-Wire 56Kbps Digital Grade Loop/DS1	with DS1 excluding mileage) -			l .					l							
	Interoffice Transport Combination - Zone 3	Zone 3		3	UNCDX	UDL56	\$212.65			<u> </u>		<b>-</b>	<del></del>		<del> </del>	<del>                                     </del>	
$\top$	Interoffice Transport - Dedicated - DS1	Interoffice Transport - Dedicated -		1		1.500	00.0560	1					1				
	combination - Per Mile Per Month	DS1 - Per Mile	-	╄-	UNC1X	1L5XX	\$0.3562	. <b>.</b>				<u> </u>	<del> </del>	- <del></del>			<del>                                     </del>
		4-Wire 56 or 64 kbps Extended Digital Loop With Dedicated DS1		1		l .	İ	1		ļ							
		Interoffice Transport (Additional 4	_		ļ			1			1	i	1				
	Additional 4-Wire 56Kbps Digital Grade	Wire 56 or 64 kbps Digital Grade		1	1			1						i .			İ
	Loopin same DS1 Interoffice Transport	Loop in same DS1 excluding	İ				1			ļ							
	Combination - Zone 1	mileage) -Zone 1		1	UNCDX	UDL56	\$32.01	1	l	1				L			<del></del>
-	Combination - Zone 1	4-Wire 56 or 64 kbps Extended	1	1	1	1											
		Digital Loop With Dedicated DS1		1	1	1	1		1	1		1		1		1	
		Interoffice Transport (Additional 4		1	I	l .		1									
	Additional 4-Wire 56Kbps Digital Grade	Wire 56 or 64 kbps Digital Grade		1	1	I							1				
	Loopin same DS1 Interoffice Transport	Loop in same DS1 excluding		1.			045.50							1			
	Combination - Zone 2	mileage) -Zone 2	-	12	UNCDX	UDL56	\$41.52		<del> </del>	<del> </del>	-	1	+		+	1	
		4-Wire 56 or 64 kbps Extended	1	1	1	I	1	1				1		1		1	
		Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4	.	1	1	1	!	1		1		1					
	Additional Abbito Folkhop Digital Oroda	Wire 56 or 64 kbps Digital Grade		1	l	1	1	1		1	1		1	l		1	
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	Loop in same DS1 excluding		1	1					1		1	1	1		1	
	Combination - Zone 3	mileage) -Zone 3		3	UNCDX	UDL56	\$54.02			<u> </u>	L						
-+	Compitation - Zone 3	Nonrecurring Cost for Extended	-	Ť	1	1	1	1				T		1			1
		Loop or Local Channel and		1	1	1	1			ŀ		1	1	1		1	1
	Nonrecurring Currently Combined Network	Interoffice Combination Switch-as	;-	1			1	1			<b></b>		1		1		
	Elements Switch -As-Is Charge	is		L	UNC1X	UNCC	·	\$52.73	\$24.62	\$9.12	\$9.12	+	<del> </del>				+
$\dashv$										L		1	_				
	I-WIRE 64 KBPS EXTENDED DIGITAL LOOP	<del></del>				т —	T			1	1	1	1	1	1	1	

UNIVERSE AND PROPRIES ALE MONTH   Security					-					RATES					OSS	RATES		
April   Series   West   Series   Seri		UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	usoc			IA I LO	Nonre	curring	Submitted	Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.
Additional AlWine-Selfation Transport Combin				0.202					Nonrec	urring	Disco	nnect				Electronic-Add'l	1st	
Additional 4-Wires-RRODo Digital Crante Combination 7-Zone I  Additional 4-Wires-RRODo Digital Crante Combination 7-Zone I							l	Rec			First	Add'l_		SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Digital Loop With Dedicated USD 1 Interedifier Transport Commission - Zone 1 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 2 Interedifier Transport Commission - Zone 3 Interedifier Transport Commission	$\vdash$		4-Wire 56 or 64 kbps Extended		1		-											
First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 4-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Digital Grade Loop wire 5-0  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 64/0pp Digital Grade Loop 05  First 6-Wire 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 6-Wire 05  First 6-Wire 05  First 6-Wire 05  First 6-Wire 05  First 6-Wire 05  First 6-Wire 05	1 1						l											
First 4-Wire 64/Ops Digital Grade Loop Digital Conduction Privage Private Priv	1						ł	1						1	i			
Inter-Wire Aldrago Diginal Grade Loop DS1 (with D61 cardularing mileage) - Inter-Wire Storage Diginal Carde Loop With D61 cardularing mileage) - Inter-Wire Storage Diginal Carde Loop Diginal Carde Loop Control Control Carde Card					1	1	ĺ				i		1					
Interdice Transport Contribution - Zone 1		First 4-Wire 64Khns Digital Grade Loop/DS1				l							l					
Digital Lossy Mine Decidated State Company (1997)  First 4-Wire 64Kbps Digital Grade Lospy Dist in Decidated State Company (1997)  Interoffice Transport Combination - Zone 2  Zone 2	1		Zone 1		1	UNCDX	UDL64	\$190.64					ļ	<del></del>	<u> </u>			
First 4-Wire 64Kbps Digital Grade Loopp Signal Grade Loopp Signal Grade Loopp Signal Grade Loop Signal Grade Signal Grade Signal Grade Signal Grade Signal Grade Signal Grade Signal Grade Signa											1		ì				l	
First 4-Wire 64Kbps Digital Grade LoopDS1 interordice Transport Combination - Zone 2														Į			l .	
First 4-Wire 64Kbps Digital Grade Loop/DSI Interoffice Transport Controllation - Zone 2					1		l							Ì	Ì			
Interoffice Transport Combination - Zone 2  Interoffice Transport Combination - Zone 3  First 4-Wire 64 (Kips Digital Grade Loop)  First 4-Wire 64 (Kips Digital Grade Loop)  First 4-Wire 64 (Kips Digital Grade Loop)  First 6-Wire 64 (Kips Digital Grade Loop)  First 6-Wire 64 (Kips Digital Grade Loop)  First 6-Wire 64 (Kips Digital Grade Loop)  Interoffice Transport Combination - Zone 3  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Interoffice Transport Combin					1	1	1				ł		1					
Additional 4-Wires-46Klopa Digital Grade Loopins ame DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI thereoffice Transport (Additional 4-Wires-46Klopa Digital Grade Loopin same DSI technologin same DSI t					١,	LINCOX	HDI 64	\$200.16						ļ				L
Digital Loop With Dedicated DS1 Interoffice Transport (Fee A-Wire S or 6 k/ops Digital Grade Loop) First 4-Wire 64(Klops Digital Grade Loop) Si or 6 k/ops Digital Grade Loop Interoffice Transport Contribution - Zows 3 Zoze. Interoffice Transport Contribution - Zows 3 Zoze. Interoffice Transport Contribution - Zows 1 Zoze. Interoffice Transport Contribution - Zows 1 Zoze. Interoffice Transport Contribution - Zows 2 Zoze. Interoffice Transport Contribution - Zows 2 Zoze. Interoffice Transport Contribution - Zows 2 Zoze. Interoffice Transport Contribution - Zows 2 Zoze. Interoffice Transport Contribution - Zows 2 Zoze. Additional 4-Wires646kbps Digital Grade Looph same DS1 Interoffice Transport Contribution - Zows 2 Zoze. Additional 4-Wires646kbps Digital Grade Looph same DS1 Interoffice Transport Contribution - Zows 2 Zoze. Additional 4-Wires646kbps Digital Grade Looph same DS1 Interoffice Transport Contribution - Zows 2 Zoze. Additional 4-Wires646kbps Digital Grade Looph same DS1 Interoffice Transport Contribution - Zows 2 Zoze. Additional 4-Wires646kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4-Wire 85 or 64 kbps Digital Grade Looph same DS1 Interoffice Transport Additional 4		Interoffice Transport Combination - Zone 2	4-Wire 56 or 64 kbps Extended		<del>ا</del>	ONODA	0525.	0200:10							Ī			
Interoffice Transport Combination - Zone 3  First 4-Wire 64kbps Digital Grade Loop Store of klaps Digital Grade Loop Interoffice Transport Combination - Zone 3 2 combination - Zone 3 2 combination - Zone 3 2 combination - Zone 3 2 combination - Zone 3 2 combination - Zone 3 2 combination - Zone 3 2 combination - Zone 3 combination - Zone 3 combination - Zone 3 combination - Zone 3 combination - Zone 3 combination - Zone 3 combination - Zone 3 combination - Zone 4 combination - Zone 5 combination - Zone 5 combination - Zone 5 combination - Zone 6 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 1 combination - Zone 2 combination - Zone 1 combination - Zone 2 combination - Zone 2 combination - Zone 2 combination - Zone 3 com						1	1				1							
Se of 8 ktyps Digital Grade Loop Digital Interoffice Transport Combination - Zone 3   Zone		1			1										ŀ			
First 4-Wire 64Kbps Digital Grade Loop(DST) with DST excluding mileage) - 3 UNCDX UDL64   S212.65     Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - Dedicated - DSI   Interoffice Transport - DSI   Inte					1	i	l	1						1	1		1	
Interoffice Transport Combination - Zone 3 Interoffice Transport Combination - Per Mile Per Month  Interoffice Transport Combination - Per Mile Per Month  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 1  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 1  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 1  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 3  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 3  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 3  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 3  Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 3  Nonrecurring Currently Combined Network Itements Switch -As-is Change Interoffice Combination Switch-as-is Change Interoffice Combination Switch-as-is Change Interoffice Combination Switch-as-is Change Interoffice Transport Combination - Zone 1  Additional A-Wire546Kbps Digital Grade Loopin same DSI standard DSI Interoffice Transport Combination Switch-as-is Change Interoffice Combination Switch-as-is Change Interoffice Combination Switch-as-is Change Interoffice Combination Switch-as-is Change Interoffice Transport Combination - Zone 1  Additional A-Wire546Kbps Digital Grade Loopin Switch-as-is Change Interoffice Transport Combination Switch-as-is Change Interoffice Transport Combination Switch-as-is Change Interoffice Transport Combination Switch-as-is Change Interoffice Transport Combination Switch-as-is Change Interoffice Transport Combination Switch-as-is Change Interoffice Transport Combination Switch-as-is Change Interoffice Transport Combination Switch-as-is Change Interoffice	1 1	First 4-Wire 64Khps Digital Grade Loop/DS1				1	1		ŀ		1	Ì					1	
Interoffice Transport - Dedicated - DSI combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Month Combination - Per Mile Per Mile Per Month Combination - Per Mile Per Mi			Zone 3	l	3	UNCDX	UDL64	\$212.65					ļ	<del>                                     </del>	<u> </u>			
combination - Per Mile Per Month Chiving 55 or 54 Rips Extended Digital Loop With Declared DIS Interesting Transport Additional 4-Wire546K0pp Digital Grade Loopin same DS1 Intereffice Transport Combination - Zone 1  Additional 4-Wire546K0pp Digital Grade Loop in same DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended Digital Loop With Declared DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended Digital Loop With Declared DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended Digital Loop With Declared DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended Digital Loop With Declared DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended Digital Loop With Declared DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Digital Grade Loopin same DS1 Intereffice Transport (Additional 4-Wire56 or 64 Rips Extended DS1 Intereffice Transport (Ties 2-Wire Local Channel with DS1 Extended S2-Wire Voice Grade Declared Loopin Same DS1 Intereffice Transport (Ties 2-Wire Local Channel with DS1 Intereffice Transport (Ties 2-Wire Local Channel with DS1 Intereffice Transport (Ties 2-Wire Local Channel with DS1 Excluding Declared Loopin Revoluting DS1 Excluding Declared Loopin Revoluting DS1 Excluding Declared Loopin Revoluting DS1 Excluding Declared Loopin Revoluting DS1 Excluding Declared Loopin Revoluting DS1 Excluding DS2 Excluding DS3 Excluding DS3 Excluding DS3 Excluding DS3 Excluding DS3 Excluding DS3 Excluding DS3 Excluding DS3 Excluding DS4 Excluding DS4 Excluding DS4 Excluding DS4 Excluding D	1		Interoffice Transport - Dedicated -														l	ļ
Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 1  Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 1  Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 2  Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same			DS1 - Per Mile		<u> </u>	UNC1X	1L5XX	\$0.3562				ļ			<del>                                     </del>	<del> </del>	<del>                                     </del>	
Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport (Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport (Combination - Zone 1					1	l					1							i
Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport Combination - Zone 1  Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in same DS1 interoffice Transport (Additional 4-WireS46Kbps Digital Grade Loop in sam						1		1						I	1			
Loopin same DS1 interoffice Transport Combination - Zone 1  Loopin same DS1 interoffice Transport Additional 4-Wire546Kbps Digital Grade Loopin same DS1 interoffice Transport Combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loopin same DS1 interoffice Transport Combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loopin same DS1 interoffice Transport Combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loopin same DS1 excluding Milesge) - Zone Mi		But 10 and									Į						1	
Combination - Zone 1 mileage) - Zone 1 1 UNCDX UDL64 \$32.01  Additional 4-Wire546kbps Digital Grade Loopin same DS1 interoffice Transport (Additional 4-Wire546kbps Digital Grade Loopin same DS1 extended Digital Loop With Dedicated DS1 interoffice Transport (Additional 4-Wire546kbps Digital Grade Loopin same DS1 extended Digital Loop With Dedicated DS1 interoffice Transport (Additional 4-Wire546kbps Digital Grade Loopin same DS1 interoffice Transport (Additional 4-Wire546kbps Digital Grade Loopin same DS1 interoffice Transport (Additional 4-Wire56 or 64 kbps Digital Grade Loopin same DS1 extended Cloopin same DS1 extended Loop in same DS1 extended Loop in same DS1 extended Loop in same DS1 extended Loop in contain the Contained of the Contained Contained in the Contained Contained in the Contained Containe					1			Į.	i	İ	1		į.	l .	1			
Combination - Zone 1  Additional 4-Wire54eRkbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2  Additional 4-Wire54eRkbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2  - Wire55 or 64 Rbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2  - Wire55 or 64 Rbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3  - Nonrecurring Currently Combined Network Elements Switch - As-1s Charge - Nonrecurring Courrently Combined Network Elements Switch - As-1s Charge - Nonrecurring Courrently Combination - Zone 3  - Wire5 or Courted Courter - Switch -	1				1	UNCDX	UDL64	\$32.01		i				İ				
Additional 4-Wire546Kbps Digital Grade Loopin same DSI Interoffice Transport Additional 4-Wire 56 or 64 kbps Digital Grade Loopin same DSI Interoffice Transport Combination - Zone 2  Combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loopin same DSI provided and the combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loopin same DSI theroffice Transport Combination - Zone 3  Nonrecurring Currently Combined Network Elements Switch -As-Is Charge  2-Wire Voice Grade Dedicated DSI Interoffice Transport Dedicated Local Channel with DSI Prist 2-Wire Local Channel with DSI Interoffice Transport Combination - Zone 1  First 2-Wire Local Channel with/DSI Interoffice Transport Combination - Zone 1  First 2-Wire Local Channel with/DSI Interoffice Transport Combination - Zone 1  First 2-Wire Local Channel with/DSI Interoffice Transport Combination - Zone 1  First 2-Wire Local Channel with/DSI Interoffice Transport Combination - Zone 1  First 2-Wire Local Channel with/DSI Interoffice Transport Combination - Zone 1  First 2-Wire Local Channel with/DSI Interoffice Transport Combination - Zone 2  First 2-Wire Local Channel with/DSI Channel with DSI excluding Interoffice Transport (First 2-Wire Local Channel with DSI excluding Interoffice Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding Transport (First 2-Wire Local Channel with DSI excluding	$\vdash$	Combination - Zone 1	4-Wire 56 or 64 kbps Extended		+÷	0	1	1									ļ	
Additional 4-Wire546Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2  Combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loop in same DS1 Interoffice Transport Combination - Zone 2  Additional 4-Wire546Kbps Digital Grade Loop in same DS1 Interoffice Transport Combination - Zone 3  Additional 4-Wire546Kbps Digital Grade Loop in same DS1 Interoffice Transport Combination - Zone 3  Nonrecurring Currently Combined Network Elements Switch - As-Is Charge is  Additional 4-Wire546Kbps Digital Grade Loop in same DS1 Interoffice Combination Switch-as-Is Charge is  Additional 4-Wire546Kbps Digital Grade Loop in same DS1 Interoffice Combination - Zone 1  Nonrecurring Currently Combined Network Elements Switch - As-Is Charge is  Average Combination - Zone 1  Existing 2-Wire Local Channel with DS1 Channel with Dedicated DS1 Interoffice Transport Combination - Zone 1  Interoffice Transport Combination - Zone 1  Existing 2-Wire Local Channel with DS1 Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1	1 1				1	Į.			ì		1		<u> </u>					
Loopin same DS1 Interoffice Transport Combination - Zone 2  Loop in same DS1 excluding mileage) - Zone 2  AWire 56 or 64 ktops Extended Digital Loop With Dedicated DS1 interoffice Transport (Additional 4- Wire 56 or 64 ktops Digital Grade Loopin same DS1 Interoffice Transport (Additional 4- Wire 56 or 64 ktops Digital Grade Loopin same DS1 Interoffice Transport (Additional 4- Wire 56 or 64 ktops Digital Grade Loopin same DS1 excluding mileage) - Zone 3  Nonrecurring Cost for Extended Loop or Local Channel and Interoffice Combination Switch-as- is  UNCIX UNCCC \$52.73 \$24.62 \$9.12 \$9.12			Interoffice Transport (Additional 4-		1		l l					1		1	1			
Loop in same DS1 Interoffice Transport Combination - Zone 2    Loop in same DS1 excluding   Additional - Zone 2   Advire 56 or 84 kbps Extended		Additional 4-Wire546Kbps Digital Grade			1	1	1				l .			1			<u>l</u>	i
Combination - Zone 2    Interoffice Transport (Additional 4-Wire 56 of 84 kbps Extended Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4-Wire 56 of 64 kbps Digital Grade Loopin same DS1 Interoffice Transport (Additional 4-Wire 56 of 64 kbps Digital Grade Loopin same DS1 Interoffice Transport (Post by 100 pin same DS1 excluding mileage) - Zone 3   Nonrecurring Currently Combined Network Elements Switch - As-is Charge   Section of the Combination Switch-as-is		Loopin same DS1 Interoffice Transport			1.	l			ì		1		ļ	Į.	l .			
Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4- Wire 56 of 64 kbps Digital Grade Loop in same DS1 Interoffice Transport Combination - Zone 3 Nonrecurring Cost for Extended Loop or Local Channel and Nonrecurring Currently Combined Network Elements Switch - As-Is Charge  2-Wire VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL.)  Extended 2-Wire Voice Grade Dedicated Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated Local Channel with Dedicated Local Channel Interoffice Transport (First 2-Wire Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated			mileage) -Zone 2	<u> </u>	2	UNCD	UDL64	\$41.52		ļ	<b>├</b> ─-	-	<b>├</b> ──	<del> </del>	<del> </del>		1	
Additional 4-Wire546Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3  Nonrecurring Cost for Extended Loop or Local Channel and Interoffice Combination Switch-as- Elements Switch -As-Is Charge  2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL.)  Extended 2-Wire Voice Grade Dedicated Local Interoffice Transport (First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with/DS1 Extended 2-Wire Voice Grade Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated Local Channel with Dedicated Local Channel with Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated Local Channel with Dedicated Local Channel with Dedicated Local Channel with Dedicated Local Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated Local Channel with DS1 Channel with DS1 Extended DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 Channel with DS1 Extended DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 Channel with DS1 Extended DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 Exten					1				ļ					i			1	
Additional 4-Wire546Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3  Nonrecurring Currently Combined Network Elements Switch - As-Is Charge  2-Wire VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)  Extended 2-Wire Volce Grade Dedicated DS1 Interoffice Transport Combination - Zone 1 Interoffice Transport Combination - Zone 1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade Dedicated DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade Dedicated DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade Dedicated DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade Dedicated DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade Dedicated DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport Combination - Zone 1 Extended 2-Wire Volce Grade DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding DS1 Extended 2-Wire Local Channel With DS1 excluding DS1 Extended 2-Wire Local Channel With DS1 excluding DS1 Extended 2-Wire Local Channel With DS1 excluding DS1 Extended 2-Wire Local Channel With DS1 excluding DS1 Extended 2-Wire Local Channel With DS1 excluding DS1 Extended 2-Wire Local Channel With DS1 excluding DS1 Extend	1 1					ı		l .						Į.	l .	ļ		
Loopin same DS1 Interoffice Transport Combination - Zone 3  Loop in same DS1 (Standard Methods)  Nonrecurring Cost for Extended Loop or Local Channel and Interoffice Combination Switch - As- Is Change					1								1		1	1	Į.	
Combination - Zone 3    Monrecurring Cost for Extended Loop or Local Channel and Interoffice Combination Switch-assis   UNCIX UNCCC   \$52.73   \$24.62   \$9.12   \$9.12				İ	1			l	1	ļ	l		1				1	
Nonrecurring Cost for Extended Loop or Local Channel and Interoffice Combination Switch-asis   UNC1X UNCCC   \$52.73   \$24.62   \$9.12   \$9.12    2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)    Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with/DS1   Interoffice Transport Combination - Zone 1   1   \$176.72    First 2-Wire Local Channel with/DS1   Channel with DS1 excluding   Interoffice Transport (First 2-Wire Local Channel with DS1 excluding   Dedicated DS1 Interoffice   Transport (First 2-Wire Local Channel with DS1 excluding   Transport (First 2-Wire Local Channel with DS1 excluding   Dedicated DS1 Interoffice   Transport (First 2-Wire Local Channel with DS1 excluding   Transport (First 2-Wire Local Channel with DS1 excluding   Dedicated DS1 Interoffice   Transport (First 2-Wire Local Channel with DS1 excluding   Transport (First 2-Wire Local Channel with DS1 excluding   Dedicated DS1 Interoffice   Transport (First 2-Wire Local Channel with DS1 excluding   Dedicated DS1 Interoffice   Dedicated DS1 Interoffice   Dedicated DS2 Interoffice   Dedicated DS3 Interoffice   Dedicated DS3 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS4 Interoffice   Dedicated DS5 Interof					1 3	UNCD	UDL64	\$54.02			1		<u> </u>					
Nonrecurring Currently Combined Network Elements Switch -As-Is Charge   Loop or Local Channel and Interoffice Combination Switch-as-is   UNC1X UNCCC \$52.73 \$24.62 \$9.12 \$9.12    2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL.)    Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with/DS1 (Interoffice Transport Combination - Zone 1   Extended 2-Wire Voice Grade Dedicated Local Channel with DS1 excluding mileage) - Zone 1   S176.72	-	Combination - Zone 3	Nonrecurring Cost for Extended		+-	<u> </u>								1		İ	l .	
Elements Switch -As-Is Charge is UNC1X UNCCC \$52.73 \$24.62 \$9.12 \$9.12  2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with/DS1 excluding Interoffice Transport Combination - Zone 1 interoffice Transport Combination - Zone 1 interoffice Transport (First 2-Wire Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding Channel with DS1 excluding Channel with DS1 excluding Channel with DS1 excluding			Loop or Local Channel and		1	1		1	i	1	1	Ì		l				
Elements Switch -As-Is Charge is UNCTX UNCCC \$5.7.3 \$24.02 \$5.12 \$	1 !	Nonrecurring Currently Combined Network	Interoffice Combination Switch-as-		1			.1	1		1	00.10	1	1				
2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding mileage) - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated Local Channel with Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding	1 1		is		Ь.	UNC1)	UNCC	ــــــــــــــــــــــــــــــــــــــ	\$52.73	\$24.62	\$9.12	\$9.12	<del> </del>		<del></del>			
Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding	H				L		Щ_	1	1	<b> </b>	<u> </u>		<del> </del>	+	+	+	1	+
Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding	2.	WIRE VOICE GRADE DEDICATED EXTENDE	D LOCAL CHANNEL WITH DEDIC	CATED D	31 IN	EROFFI	CE TRAN	SPORT (EEL)	ļ	<del></del>		1	<del> </del>		<del> </del>		+	<del></del>
Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with Dedicated Local Channel with DS1 excluding  First 2-Wire Local Channel with/DS1  Channel with DS1 excluding	<del> -</del>		Extended 2-Wire Voice Grade			1		1	1		1			1			1	
First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1  Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding	1			1		1	1	1			1		1	1	1		1	
First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1  Extended 2-Wire Voice Grade Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with/DS1  First 2-Wire Local Channel with/DS1  Channel with DS1 excluding  1 \$176.72					1	1	1	1	1			1	1		1	1	ı	
Interoffice Transport Combination - Zone 1 mileage) - Zone 1 1 \$176.72  Extended 2-Wire Voice Grade Dedicated Dost Interoffice Transport (First 2-Wire Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding					1		1		1						1	1		
Interomice Transport Combination - 2016 - Interomice Transport Combination - 2016 - Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local First 2-Wire Local Channel with/DS1					1.	1		\$176.72	Į		1		1	1		<u> </u>		
Dedicated Local Channel with Dedicated DS1 Inheroffice Transport (First 2-Wire Local Channel with/DS1 Channel with DS1 excluding		Interoffice Transport Combination - Zone 1	Extended 2-Wire Voice Grade	+-	+-	+-	+	1	+	<b>—</b> —	1			T				
Dedicated DS1 Interoffice Transport (First 2-Wire Local First 2-Wire Local Channel with/DS1 Channel with DS1 excluding	1		Dedicated Local Channel with					1			1		1	1	1	1	1	1
Transport (First 2-Wire Local First 2-Wire Local Channel with/DS1 Channel with DS1 excluding					1	1	l	1	1		1		1	1	1	1		
First 2-Wire Local Channel with/DS1 Channel with DS1 excluding						1	1			1	1				1		1	
		First 2-Wire Local Channel with/DS1			1			1	1	1	1		1	1	1	1	1	
		Interoffice Transport Combination - Zone 2			2			\$181.98										1.

									ATES			<u> </u>		oss	RATES		
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec		Nonre	curring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'i
				ll			Rec	First	Add'l	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Translat A Wire Veige Grade						- 1 11 21	7.00.								
	First 2-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 3	Extended 2-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 2-Wire Local Channel with DS1 excluding mileage) - Zone 3 Interoffice Transport - Dedicated -		3			\$188.88										
	Interoffice Transport - Dedicated - DS1	DS1 - Per Mile		1	UNC1X	U1TF1	\$0.3562						<u> </u>				
	combination - Per Mile Per Month	DS1 - Per Mile		<del>  -</del> -	0.10.51	-						Ī		l .			
	Additional 2-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 1	excluding mileage) -Zone 1		1			\$18.09										
-+	2010 1	Extended 2-Wire Voice Grade		1		1	<b>,</b>				1	1	1				
	Additional 2-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 2	lexcluding mileage) -Zone 2		2			\$23.35			<u> </u>							
		Extended 2-Wire Voice Grade	1	1	l			ŀ		l		1	1	i		1	
	Additional 2-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 3	excluding mileage) -Zone 3		3			\$30.25										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	Nonrecurring Cost for Extended Loop or Local Channel and Interoffice combination Switch-as- is			UNC1X	UNCCC		\$52.73	\$24.62	\$9.12	\$9.12		<u> </u>		-		
				1		<u> </u>	L			-	<b></b>		<del> </del>		+		+
<del></del>	WIRE VOICE GRADE DEDICATED EXTENDE	DIOCAL CHANNEL WITH DEDIC	ATED DS	1 INT	EROFFIC	E TRAN	SPORT (EEL)	l									+
4-	First 4-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 1	Extended 4-Wire Volce Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 1		1			\$177.73										
	First 4-Wire Local Channel with/DS1 Interoffice Transport Combination - Zone 2	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 2		2			\$183.29					:					
	First 4-Wire Local Channel with/DS1	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (First 4-Wire Local Channel with DS1 excluding mileage) - Zone 3		3			\$190.59										
$\perp \perp$	Interoffice Transport Combination - Zone 3	Interoffice Transport - Dedicated -	+	┿		<del>                                     </del>	1		1	T			1		1	l .	
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	DS1 - Per Mile	<u> </u>	上	UNC12	K U1TF1	\$0.3562		<u> </u>							1	

								RATES					OSS	RATES		
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec	urring	Disco	curring ennect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'i	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st SOMAN	Incremental Charge - Manual Svc Order vs. Electronic-Dit Add'I SOMAN
			ll			Rec	First	Add'l	First	Addʻl	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SUMAN
Additional 4-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 1	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 4-Wire Voice Grade Channel in same DS1 excluding mileage) -Zone 1		1			\$19.09										
Additional 4-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 2	Extended 4-Wire Voice Grade Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 4-Wire Voice Grade Channel in same DS1 excluding mileage) -Zone 2 Extended 4-Wire Voice Grade		2			\$24.66										
Additional 4-Wire Voice Grade Channel in Same DSI Interoffice Transport Combination Zone 3	Dedicated Local Channel with Dedicated DS1 Interoffice Transport (Additional 4-Wire Voice		3			\$31.96										
Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	Loop or Local Channel and Interoffice Combination Switch-as- is				UNCCC		\$52.73	\$24.62	\$9.12	\$9.12						
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH	DEDICATED DS1 INTEROFFICE T	RANSPO	RT (E	EL)												<del></del>
First 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport Zone 1	With Dedicated DS1 Interoffice Transport (First 4-Wire DS1 Digital			UNC1X	USLXX	\$135.59										
First 4-Wire DS1 Digital Loop in	Extended 4-Wire DS1 Digital Loop With Dedicated DS1 Interoffice Transport (First 4-Wire DS1 Digital Loop with DS1 excluding mileage) Zone 2		2	UNC1X	USLXX	\$153.26										
First 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport Zone 3	Extended 4-Wire DS1 Digital Loop With Dedicated DS1 Interoffice Transport (First 4-Wire DS1 Digita Loop with DS1 excluding mileage) Zone 3	ı	3	UNC1X	USLXX	\$176.45										
Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mont	Interoffice Transport - Dedicated - h DS1 - Per Mile		_	UNC1X	U1TF1	\$0.3562	ļ				<u> </u>				-	
Additional 4-Wire DS1 Digital Loop in same same DS1 Interoffice Transport Combination - Zone 1			1	UNC1X	USLXX	\$57.73										
Additional 4-Wire DS1 Digital Loop in same same DS1 Interoffice Transport Combinatio - Zone 2	Extended 4-Wire DS1 Digital Loop With Dedicated DS1 Interoffice Transport (Additional 4-Wire DS1 n Digital Loop in same DS1 excluding mileage) - Zone 2		2	UNC1>	USLXX	\$75.40										

			_					RATES					OSS	RATES		
			1		ŀ			IAILS				1			Incremental	Incrementa Charge -
		Not in TRA			ļ						Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Manual Sve
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS STATED IN DOCKET 97-01262	Docket 97-	Zone	BCS	USOC				Nonrec	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
3,133,132	\$1A1ED IN DOCKET 97-01202	01262	1		. 1		1				Elec	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'i	Electronic-Disc 1st	Electronic-Di Add'l
					l		Nonrec		Disco	Add'l	per LSR SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				<u> </u>		Rec	First	Add'i	First	Addi	SOMEC	JOMAN				
							]									
	Extended 4-Wire DS1 Digital Loop										1		1		l .	
	With Dedicated DS1 Interoffice						Į.									1
Additional 4-Wire DS1 Digital Loop in same	Transport (Additional 4-Wire DS1						1					Ŀ				
same DS1 Interoffice Transport Combination	Digital Loop in same DS1		1								i .	i	i	İ	ŀ	1
- Zone 3	excluding mileage) - Zone 3		3	UNC1X	USLXX	\$98.59	Ļ				<del>                                     </del>		<del>                                     </del>			<del></del>
	Nonrecurring Cost for Extended			ļ			l									
	Loop or Local Channel and		1	i								l l				1
Nonrecurring Currently Combined Network	Interoffice Combination Switch-as-			l			\$52.73	\$24.62	\$9.12	\$9.12	İ	1	1			
Elements Switch -As-Is Charge	is		<u> </u>	UNC1X	UNCCC		\$52.73	\$24.02	φ <del>5.12</del>	φ3.12	<del>                                     </del>	<del></del>				T
			L	<u> </u>	L,		<b>_</b>					<del> </del>	<del>                                     </del>		-	<del></del>
Currently Combined Network Transport Eleme	ents (Non-Switched Combinations	Resulting	fron	n a Conve	rsion)						<b>↓</b>	-	<del>                                     </del>	<u> </u>		+
Local Channel - Dedicated - 2-Wire VG							<u> </u>				1		200.05	\$10.54	\$13.30	+
Monthly Recurring per month		**		UNCVX	ULDV2	\$19.43	\$199.33	\$24.16	\$54.81	\$4.80	<del>                                     </del>	<b>-</b>	\$20.35	\$10.54	\$13.30	+
		**	T	UNCVX		\$0.00							<b>↓</b>	<b>_</b>	<b>_</b>	+
Monthly Recurring per mile per month			$T^-$										L			
		<del> </del>	t	<del>                                     </del>												↓
Local Channel - Dedicated - 2-Wire VG - Rev		**	+	UNCVX	ULDR2	\$19.43	\$199.33	\$24.16	\$54.81	\$4.80		i	\$20.35	\$10.54	\$13.30	
Monthly Recurring per month		**	+-	UNCVX		\$0.00	<b>V</b>	7		†		1			l	
Monthly Recurring per mile per month			<b>├</b> -	UNCVA	ILSINO	φυ.σσ			<del>                                     </del>		· · · · · · · · · · · · · · · · · · ·				Ī	
			ـــــ	ļ					├		<del>                                     </del>	1				
Local Channel - Dedicated - 4-Wire VG			<u>L</u> _		L				<del> </del>	05.54	<del>                                     </del>	<del> </del>	\$20.35	\$10.54	\$13.30	
Monthly Recurring per month		**	١	UNCVX	ULDV4	\$20.56	\$201.53	\$24.83	\$55.52	\$5.51	ļ	<del></del>	\$20.33	\$10.54	\$10.00	+
		**		UNCVX	1L5NC	\$0.00	1	i								
Monthly Recurring per mile per month		<del> </del>	╁	10.1017	720110		<b>-</b>									
			╂—	<b>├</b> ┈─				<del>                                     </del>								
Local Channel - Dedicated - DS1		**	-		554	***	0004 50	\$24.83	\$55.52	\$5.51		1	\$20.35	\$10.54	\$13.30	T
DS1 Monthly Recurring per month			↓_	UNC1X	ULDF1	\$20.56	\$201.53	\$24.03	\$55.5Z	\$5.51		<del>                                     </del>	1	*******		
Monthly Recurring per mile per month		**	L	UNC1X	1L5NC	\$0.00	<b>_</b>		ļ	<del> </del>	<del></del>		<del> </del>		1	+
, , , , , , , , , , , , , , , , , , ,			1		L			ļ		ļ		<del> </del>	<del></del>	+	-	+
Interoffice Channel - Dedicated - 2-wire VG			T						l							+
Interoffice Channel - Dedicated 2-wire VG -		†	<b>†</b>	1								l		1	i	
per mile per month	İ	**		UNCVX	1L5XX	\$0.0174			L	ļ		<del></del>	<b>_</b>	<del> </del>		+-
Interoffice Channel - Dedicated 2-wire VG -				T			1			***	1	1	\$20.35	\$21.09	\$9.80	\$10.
Facility Termination per month		**		UNCVX	U1TV2	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51		<del></del>	\$20.35	\$21.09	\$9.00	+ 0.0
Tability Torrisa		T	T						1		<u> </u>				<del>                                     </del>	
Interoffice Channel - Dedicated - 2-wire VG Re	v Rattery	† · · · · ·	1										<b>_</b>		ļ	
nteroffice Channel - Dedicated 2-wire VG -	Dattery	<del>                                     </del>	t		1						Į.	l l				i
per mile per month		**	1	UNCV	1L5XX	\$0.02			1					ļ	<b>_</b>	
nteroffice Channel - Dedicated 2-wire VG -		1	1	1						1			600.05	\$21.09	\$9.80	\$10.
Facility Termination per month		**	İ	UNCV	U1TR2	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51	<b>_</b>	<del></del>	\$20.35	\$21.09	\$9.00	Ψ10
Tacilly Tollington por more			1									<u> </u>				+
Interoffice Channel - Dedicated - 56kbps			T		I				1				1		<del></del>	+-
Interoffice Channel - Dedicated - 56kbps -		1	1	<u> </u>							1					
per mile per month		**	1	UNCD	( 1L5XX	\$0.0174				1			<b>_</b>	+	+	+
Interoffice Channel - Dedicated - 56 kbps -	<del>                                     </del>	T	1	T			1	1	1	1	1	1	600.05	\$21.09	\$9.80	\$10.
Facility Termination per month		**		UNCD	U1TD5	\$17.98	\$55.39	\$17.37	\$27.96	\$3.51	<b>_</b>	+-	\$20.35	\$21.09	ψ <del>σ.ου</del>	+ 910.
r doney reminated per month			T			l			1		<b></b> _		<del> </del>	<del> </del>	<del> </del>	
the Manual Pullurand California	1	<del> </del>	1	$\top$								1			<b>-</b>	
Interoffice Channel - Dedicated - 64kbps		+-	+-	+-	1	T		1					1	1	1	
Interoffice Channel - Dedicated - 64 kbps -		**	1	UNCD	1L5XX	\$0.02	1				1			<b>_</b>		
per mile per month Interoffice Channel - Dedicated - 64 kbps -	<del>                                     </del>	+	+-		1		$\top$					1		001.00	60.00	640
		**	1	UNCD	U1TD6	\$17.98	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.
Facility Termination per month		+	+	+	1	T	1 -				_	1	I			

			_		<del></del>	-		RATES					oss	RATES		
					1		1	TATEO							Incremental Charge -	Incrementa Charge -
UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETEORK ELEMENT AS	Not in TRA Docket 97-		BCS	USOC				Nonre	curring	Svc Order Submitted	Svc Order Submitted		incremental Charge - Manual	Manual Svc Order vs.	Manual Svo Order vs.
	STATED IN DOCKET 97-01262	01262	l				1				Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs. Electronic-Add'i	Electronic-Disc	Electronic-Di Add'i
			l	ļ		Rec	Nonrec First	urring Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			-		$\vdash$	- 100	1118	744.								
Interoffice Channel - Dedicated - DS1			_				<u> </u>									
Interoffice Channel - Dedicated - DS1 - per mile per month		**		UNC1X	1L5XX	\$0.3562						<u></u>				
Interoffice Channel - Dedicated - DS1 -										01100	1		\$20.35	\$21.09	\$9.80	\$10.54
Facility Termination per month		**		UNC1X	U1TF1	\$77.86	\$112.40	\$76.27	\$19.55	\$14.99		<del>                                       </del>	\$20.00	Ψ21.03	40.00	
		<u> </u>		<b>-</b>									<u> </u>			
Local Loop - Dedicated - 2-Wire VG			╌	<u> </u>	$\vdash$	<del> </del>	<del>                                      </del>			·						
Local Loop - Dedicated - 2-Wire VG - Monthly Recurring per month - Zone 1		**	1	UNCVX	UEAL2	\$16.56	\$75.06	\$48.20	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
Local Loop - Dedicated - 2-Wire VG -			T	†									\$20.35	\$10.54	\$13.32	
Monthly Recurring per month - Zone 2		**	2	UNCVX	UEAL2	\$21.63	\$192.97	\$140.72	\$28.70	\$17.64	<del> </del>	<u> </u>	\$20.35	\$10.54	ψ10.0z	
Local Loop - Dedicated - 2-Wire VG -		**	3	UNCVX	UEAL2	\$28.28	\$192.97	\$140.72	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
Monthly Recurring per month - Zone 3			³-	ONCVA	VEALE	\$20.20	\$132.37	ψ140.7±	1 420.74	<b>V</b> 11.01	T	·				
Local Loop - Dedicated - 2-Wire VG - Monthly Recurring per mile per month		**	1	UNCVX	1L5ND	\$0.00			<u>L</u>				<u> </u>	ļ		
Monthly Recurring per time per month								L		1			<u> </u>	<u> </u>		<del> </del>
Local Loop - Dedicated - 2-Wire VG - Rev Bat.								ļ	<u> </u>	-		<del> </del>				-
Local Loop - Dedicated - 2-Wire VG - Rev		**	1.	LINGUA	UEAR2	\$16.56	\$75.06	\$48.20	\$28.70	\$17.64	ì	1	\$20.35	\$10.54	\$13.32	
Bat - Monthly Recurring per month - Zone 1			+-'	DINCVA	UEARZ	\$10.50	\$73.00	ψ-το.2.0	<del>  \$20.70</del>	<b>V</b>				1		
Local Loop - Dedicated - 2-Wire VG - Rev Bat - Monthly Recurring per month - Zone 2		**	2	UNCVX	UEAR2	\$21.63	\$192.97	\$140.72	\$28.70	\$17.64			\$20.35	\$10.54	\$13.32	
Local Loop - Dedicated - 2-Wire VG - Rev			<b>†</b>						T	047.04	i	1	\$20.35	\$10.54	\$13.32	
Bat - Monthly Recurring per month - Zone 3		**	3	UNCV	UEAR2	\$28.28	\$192.97	\$140.72	\$28.70	\$17.64	<del></del>	<del>                                     </del>	\$20.33	\$10.54	ψ10.0 <u>L</u>	<del>-</del>
		**	1	UNCVX	1L5ND	\$0.00		İ			ł	1	l .			
Monthly Recurring per mile per month			+	011077	LONE	40.00	+			T						
Dull- and A Wiles VC		1	+	+	<del>                                     </del>								<u> </u>			
Local Loop - Dedicated - 4-Wire VG		+	+-	<del>                                     </del>						T			****	\$10.54	\$13.32	
Monthly Recurring per month - Zone 1		**	1	UNCV	UEAL4	\$24.70	\$122.76	\$85.57	\$76.35	\$39.16	<del>                                      </del>	<del> </del>	\$20.35	\$10.54	\$10.02	
Local Loop - Dedicated - 4-Wire VG -		**	٦	LINION	UEAL4	\$32.25	\$122.76	\$85.57	\$76.35	\$39.16	1		\$20.35	\$10.54	\$13.32	
Monthly Recurring per month - Zone 2		<u> </u>	2	UNCV	UEAL4	\$32.25	\$122.70	Ψ03.57	1 470.00	4000						
Local Loop - Dedicated - 4-Wire VG - Monthly Recurring per month - Zone 3		**	3	UNCV	UEAL4	\$42.17	\$122.76	\$85.57	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
Monthly Recurring per month - 2016 5			1	T									İ			1
Monthly Recurring per mile per month		**	$\perp$	UNCV	( 1L5ND	\$0.00		<u> </u>	ļ. —	+	+	+-	<del> </del>	+		
			4	↓	┷				<del>  -</del>	<del></del>	+	+		<del> </del>	<del>                                     </del>	-
Local Loop - Dedicated - 2-Wire ISDN Digital			<b>→</b> —	<del> </del>	┼─	-		-	<del>                                     </del>	-	<b>-</b>	-		T		
Local Loop - Dedicated - 2-Wire ISDN -		**	11	UNCVX	U1L2X	\$22.00	\$142.76	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$13.32	
Monthly Recurring per month - Zone 1 Local Loop - Dedicated - 2-Wire ISDN -			+								1	1	\$20.35	\$10.54	\$13.32	İ
Monthly Recurring per month - Zone 2		**	2	UNCV	U1L2X	\$29.02	\$142.76	\$88.88	\$76.35	\$39.16			\$20.35	\$10.54	\$10.02	
Local Loop - Dedicated - 2-Wire ISDN -		**	3	UNCV	U1L2X	\$37.95	\$142.76	\$88.88	\$76.35	\$39.16	1		\$20.35	\$10.54	\$13.32	1
Monthly Recurring per month - Zone 3		+	٠ϰ	UNCV	101127	\$37.93	Ψ142.70	400.00	\$1,0.00	+						
Monthly Recurring per mile per month		**		UNCV	X 1L5NE	\$0.00		1			<b>_</b>	$\bot$			<del>                                     </del>	+-
- Housing per time per month			Ι						<u> </u>	4		<b>_</b>	+			+
Local Loop - Dedicated - 4-Wire 56 kbps				4			+			+		+		+	+	
Local Loop - Dedicated - 4-Wire 56 kbps			1						1		1	1				
Digital - Monthly Recurring per month - Zon	9	**	₁	UNCV	UDL5	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
Local Loop - Dedicated - 4-Wire 56 kbps			+	1	1											
Digital - Monthly Recurring per month - Zon	e		1	l			000000		,   ,,,, -,	64440	. 1	1	\$20.35	\$10.54	\$13.32	
2		**	_ 2	UNCV	C UDL5	\$40.61	\$207.01	\$141.38	\$ \$90.70	\$44.18	+		Ψ20.33	\$10.04	7,0.02	<del> </del>
Local Loop - Dedicated - 4-Wire 56 kbps	_1		1						1				1			
Digital - Monthly Recurring per month - Zon	e	**	١,	LINCV	K UDL5	5 \$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	

							F	RATES				Incremental	I Increment			
				l										1	Charge -	Charge
	UNBUNDLED NETEORK ELEMENT AS	Not in TRA		BCS	usoc						Svc Order	Svc Order Submitted	Incremental Charge - Manual	Incremental Charge - Manual	Manual Svc Order vs.	Manual :
UNBUNDLED NETWORK ELEMENT	STATED IN DOCKET 97-01262	Docket 97- 01262	Zone	BCS	USCC		Nonrecurring		Nonrecurring Disconnect		Submitted Elec	Manually per LSR SOMAN	Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Dis Add'1 SOMAN
			1		L						per L\$R		SOMAN	Electronic-Add'i	1st SOMAN	
						Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	JOHNAN	
4-Wire 56 kbps Digital Grade Loop per mile per month		**	<u> </u>	UNCVX	1L5ND	\$0.00										
			1									"				
ocal Loop - Dedicated - 4-Wire 64 kbps			<b>├</b> -	$\vdash$												
Local Loop - Dedicated - 4-Wire 64 kbps							· ·				l					
Digital - Monthly Recurring per month - Zone		**	1	UNCVX	UDL64	\$31.10	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	
Local Loop - Dedicated - 4-Wire 64 kbps			1			ļ					1		l .			
Digital - Monthly Recurring per month - Zone		**	2	UNCVX	UDL64	\$40.61	\$207.01	\$141.38	\$90.70	\$44.18		ļ	\$20.35	\$10.54	\$13.32	+
Local Loop - Dedicated - 4-Wire 64 kbps		1	1	l i					<b>,</b>		Į.		1		1	
Digital - Monthly Recurring per month - Zone		**	3_	UNCVX	UDL64	\$53.11	\$207.01	\$141.38	\$90.70	\$44.18			\$20.35	\$10.54	\$13.32	-
4-Wire 64 kbps Digital Grade Loop per mile		**	•		41.510	00.00					ł				1	<u> </u>
per month		**	$\vdash$	UNCVX	1L5ND	\$0.00										
Local Loop - Dedicated - DS1											<u> </u>	<del>                                     </del>		<del>                                     </del>		+
Local Loop - Dedicated - DS1 - Monthly		**	١,	UNC1X	USLXX	\$57.73	\$313.08	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	
Recurring per month - Zone 1		-	+÷	GNOIX	100000	ψ010	1		T -				1		*****	
Local Loop - Dedicated - DS1 - Monthly Recurring per month - Zone 2		**	2	UNC1X	USLXX	\$75.40	\$313.08	\$219.72	\$96.86	\$40.45	<b> </b>		\$18.98	\$8.43	\$11.95	-
Local Loop - Dedicated - DS1 - Monthly		**	١.	l	1	***	mana no	\$219.72	\$96.86	\$40.45			\$18.98	\$8.43	\$11.95	
Recurring per month - Zone 3		**	3	UNC1X	USLXX	\$98.59	\$313.00	\$219.72	ψ30.00	Ψ-1010_	<del>                                     </del>					T
Local Loop - Dedicated - DS1 - Monthly Recurring per mile per month		**	$\perp$	UNC1X	1L5ND	\$0.00			<u> </u>			<del>                                      </del>			<del>                                     </del>	<del> </del>
			<b>⊥</b> _				<u> </u>		<del>                                     </del>		+	<del>-</del>	<del>                                     </del>		<b>†</b>	
DS1 Channelization				L	<u> </u>		<u> </u>		4	\$13.46	<del> </del>		\$20.35	\$9.80	\$11.49	\$1
DS1 to DS0 Channel System per month		**		UNC1X	MQ1	\$80.77	\$141.87	\$77.11	\$14.51	\$13.46	<b>∔</b>	<del>-</del>	\$20.00	45.55	<b>V.III.</b>	1
OCU-DP COCI (data) - DS1 to DS0 Channel		**		UNCDX	1D1DD	\$1.82	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1
System - per month (2.4-64kbs)  2-wire ISDN COCI (BRITE) - DS1 to DS0		**	十	LINOUS S	1110101	\$3.10	\$6.07	\$4.66			1		\$20.35	\$9.80	\$11.49	\$1
Channel Systsem - per month	L		4-	UNCNX	UC1CA	\$3.10	\$0.07	φ-4.00	+-	<del>                                     </del>	+	T	1			
Voice Grade COCI - DS1 to DS0 Channel System - per month		**		UNCVX	1D1VG	\$0.91	\$6.07	\$4.66	ļ	<u> </u>		<del> </del>	\$20.35	\$9.80	\$11.49	\$1
			$\bot$		<b>.</b>		+	<del> </del> -	+-	-		+-				$\top$
MULTIPLEXERS		**	4-	LINICAY	MQ3	\$222.98	\$308.03	\$108.47	\$44.47	\$42.62	+	<del>                                     </del>	\$20.35	\$9.80	\$11.49	\$1
DS3 to DS1 Channel System per month		**	+	UNC3X		\$222.98	\$308.03		_	\$42.62			\$20.35	\$9.80	\$11.49	\$1
STS1 to DS1 Channel System per month		**	4	UNCSX	MQ3	\$222.98	\$300.03	\$100.47	+ +++++/-	+	$t^{}$				T	
DS3 Interface Unit (DS1 COCI) used with Loop per month		**		UNC1X	UC1D1	\$17.58	\$6.07	\$4.66	<b>_</b>	<u> </u>			\$20.35	\$9.80	\$11.49	\$
			$\mathbf{L}$	L	L	L									<del>                                      </del>	+
* Rates for these elements, products or services were not	and and the the TDA in Docket No. 97-015	es Howeve	r such	elements.	oroducts or s	ervices are nece	essary for the p	provision of ot	ther elements.	products or	services for w	nich permanent	rates were order	eu		+-

				г – т				DA.	TES					oss	RATES		
				ì l				- 110	1							Incremental	Incrementa
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETWORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97-	Zone	BCS	usoc				Nonrec	urring	Svc Order Sub-mitted	Svc Order Submitted	incremental Charge - Manual	Incremental Charge - Manual	Charge - Manual Svc Order vs.	Charge - Manual Sve Order vs.
		DOCKET 97-01202	01262						Ì			Elec	Manually per	Svc Order vs.	Svc Order vs. Electronic-Add'i	Electronic-Disc	Electronic-D Add'l
				i I				Nonrecu		Discor		per LSR SOMEC	LSR	Electronic-1st SOMAN	SOMAN	SOMAN	SOMAN
							Rec	First	Add'i	First	A001	SOMEC	SUMAN	SUMAIN	JOHAN	COMPLE	
Т				_										L			
CA	L INTERCONNECTION (CALL TRANSPORT AN	ID TERMINATION)															
T.			ļ					-									
E	ND OFFICE SWITCHING			$\vdash$			\$0.0008041										
T	End Office Switching Function, Per MOU	End office switcing function	ļ	$\vdash$			\$0.0000041										
				$\vdash$				- "									
T	ANDEM SWITCHING	Tandem switching function		$\vdash$			\$0.0009778										L
+-	Tandem Switching Function Per MOU	Tarident Switching Idriction		<b>├</b> -													
1	L INTERCONNECTION (TRANSPORT)			1 1													<u> </u>
TA	LINTERCONNECTION (TRANSPORT)			1											ļ		<u> </u>
-	OMMON TRANSPORT (Shared)			1											ļ		
-	Common Transport - Per Mile, Per MOU	Common transport - per mile, per MOU					\$0.0000064										-
+-	Common Transport - Facilities Termination Per	Common Transport - Facilities Termination												1			
	MOU	per MOU		<u> </u>			\$0.0003871					<u> </u>	<u> </u>				<del></del>
+-			<u> </u>		L							-	<del></del>		<del> </del>	-	
IN	TEROFFICE CHANNEL - DEDICATED TRANSF	PORT - VOICE GRADE		<b>!</b>	L					-		-					
Т		Interoffice Transport - Dedicated - Voice		1	11170	1L5XF	\$0.0174					l	Į.			l	
	Wire Voice Grade - Per Mile per month	Grade		-	UIIVA	ILDAF	\$0.0174										
	Interoffice Channel - Dedicated Transport- 2-	Interesting Transport Dedicated - 2-Mira				l	1									1	İ
	Wire Voice Grade - Facility Termination per	Interoffice Transport - Dedicated - 2-Wire Voice Grade - Facility Termination		1	UITVX	1L5XF	\$18.58	\$55.39	\$17.37	\$27.96	\$3.51			\$20.35	\$21.09	\$9.80	\$10.5
$\perp$	month	Voice Grade - Pacinty Termination	-	1-													
1.	ITEROFFICE CHANNEL - DEDICATED TRANSF	PORT - 56/64 KBPS		<b>1</b>								L		L		<u> </u>	ļ
1117	TENOFFICE CHANNEL - DEBICATED TRANS	Interoffice Transport - Dedicated DSO -								1				l		1	
	Interoffice Channel - Dedicated Transport - 56	56/64 kbps - Interoffice Transport -				l		İ				l .	į .	l		1	
	kbps - per mile per month	Dedicated - DSO - per mile			U1TDX	1L5XK	\$0.0174					<u> </u>	ļ	<del>                                       </del>		<del> </del>	<del></del>
+	nope per rime per riena.	Interoffice Transport - Dedicated DSO -					1		1					1	1		
	Interoffice Channel - Dedicated Transport - 56	56/64 kbps - Interoffice Transport -		1			047.00	655.00	\$17.37	¢27.06	¢2.51	Į	1	\$20.35	\$21.09	\$9.80	\$10.5
ì	kbps - Facility Termination per month	Dedicated - DSO - Facility Termination		<b>-</b>	U1TDX	1L5XK	\$17.98	\$55.39	\$17.37	φε/1.50	\$3.31	<del></del>		\$20.00	<b>V2.1.00</b>	1	1
T		Interoffice Transport - Dedicated DSO -	İ	į .			1										
	Interoffice Channel - Dedicated Transport - 64	56/64 kbps - Interoffice Transport -		1	U1TDX	1L5XK	\$0.0174					Į.	1				
_	kbps - per mile per month	Dedicated - DSO - per mile Interoffice Transport - Dedicated DSO -	-	+-	OTIDA	120/41	40.0111	<b>-</b>				<b>1</b>					
	Libert Wine Observal Dedicated Transport 64	56/64 kpps - Interoffice Transport -	Ì	1			i			1	ļ	1		1		1	
1	Interoffice Channel - Dedicated Transport - 64	Dedicated - DSO - Facility Termination		1	U1TDX	1L5XK	\$17.98	\$55.39	\$17.37	\$27.96	\$3.51	L	<u> </u>	\$20.35	\$21.09	\$9.80	\$10.5
+	kbps - Facility Termination per month	Dedicated Boo Facility Formation															
-	TEROFFICE CHANNEL - DEDICATED TRANSI	PORT - DS1		T	L					<u> </u>		<u> </u>					<del></del>
-  "	Interoffice Channel - Dedicated Channel - DS1 -	Interoffice Transport - Dedicated - DS1 -				į .			ĺ			l	i	1			
Ì	Per Mile per month	per mile		<u> </u>	U1TD1	1L5XiL	\$0.3562			<u> </u>	<u> </u>	<b></b>	<b>├</b>	<b>-</b>		1	+
T	Interoffice Channel - Dedicated Tranport - DS1	- Interoffice Transport - Dedicated - DS1 -		1	LIATE	41 5/11	\$77.86	\$112.40	\$76.27	\$19.55	\$14.99		l	\$20.35	\$21.09	\$9.80	\$10.5
	Facility Termination per month	Facility Termination			U1TD1	1L5XIL	\$77.00	\$112.40	ψ/ U.Z.	Ψ10.50	Ψ14.00	t	1	1			
$\perp$			+	+	<del> </del>	<del>                                     </del>		<u> </u>		t	1	1					
L	OCAL CHANNEL - DEDICATED TRANSPORT	Local Channel - Dedicated - 2-Wire Voice	+	1-	+	t	1			1		Ī	1				1
	Local Channel - Deldicated - 2-Wire Voice	Grade [shown here deaveraged]		1	ULDVX	TEFV3	\$17.18	\$199.33	\$24.16	\$54.81	\$4.80	<u> </u>		\$20.35	\$10.54	\$13.30	\$0.00
+	Grade per month - Zone 1  Local Channel - Deldicated - 2-Wire Voice	Local Channel - Dedicated - 2-Wire Voice	<b>-</b>	+-										1			
-	Grade per month - Zone 2	Grade [shown here deaveraged]		2	ULDVX	TEFV3	\$22.44	\$199.33	\$24.16	\$54.81	\$4.80	<b>-</b>	1	\$20.35	\$10.54	\$13.30	\$0.0
+	Local Cahnnel - Dedicated 2-Wire Voice Grade		1	1				I	1					600.05	640.54	612.20	\$0.0
	per month - Zone 3	Grade [shown here deaveraged]		3	ULDVX	TEFV3	\$29.34	\$199.33	\$24.16	\$54.81	\$4.80	<u> </u>	<b>_</b>	\$20.35	\$10.54	\$13.30	\$0.0
+	Local Channel - Dedicated - 4-Wire Voice	Local Channel - Dedicated - 4-Wire Voice			1	1	1		00:00	A== ==	05.51	1		\$20.35	\$10.54	\$13.30	\$0.0
	Grade per month - Zone 1	Grade [shown here deaveraged]	1	11	ULDDX	TEFV4	\$18.18	\$201.53	\$24.83	\$55.52	\$5.51	<del>-</del>	+	\$20.35	\$10.54	ψ10.00	+ 40.0
+	Local Channel - Dedicated - 4-Wire Voice	Local Channel - Dedicated - 4-Wire Voice	1	۔ ا	1		f00.74	#201 F2	\$24.92	\$55.52	\$5.51	1	1	\$20.35	\$10.54	\$13.30	\$0.0
_	Grade per month - Zone 2	Grade [shown here deaveraged]		2	ULDDX	TEFV4	\$23.74	\$201.53	<b>⊅∠4.03</b>	\$55.52	ψυ.σι	+	+	<del>*************************************</del>	1	1 ,,,,,,,,,	+
$\top$	Local Channel - Dedicated - 4-Wire Voice	Local Channel - Dedicated - 4-Wire Voice		3	ULDDX	TEFV4	\$31.05	\$201.53	\$24.83	\$55.52	\$5.51			\$20.35	\$10.54	\$13.30	\$0.0
- 1	Grade per month - Zone 3	Grade [shown here deaveraged]		13	OLDDX	1EFV4	φυ1.00	9201.33	Ψ2-7.00	+ 400.02	40.01	1	1	1	1		T
_	Local Channel - Dedicated - DS1 per month -	Local Channel - Dedicated - Dedicated	1	1												\$21.75	\$1.7

## LOCAL INTERCONNECTION Tennessee

				_		— т		P/	TES					oss	RATES		
															incremental	Incremental Charge - Manual Svc	Incremental Charge - Manual Svo
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETWORK ELEMENT AS STATED IN DOCKET 97-01262	Not in TRA Docket 97- 01262	Zone	BCS	usoc		Nonrec		Nonre	curring	Svc Order Sub-mitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Charge - Manual	Order vs. Electronic-Disc	Order vs.
						1 1	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				1													<del></del>
+	Local Chamber Doubles	Local Channel - Dedicated - Dedicated DS1 [shown here deaveraged]		2	ULDD1	TEFHG	\$47.33	\$277.35	\$233.26	\$33.18	\$22.30			\$45.68	\$1.76	\$21.75	\$1.76
	Local Channel - Dedicated - DS1 per month -	Local Channel - Dedicated - Dedicated DS1 [shown here deaveraged]		3	ULDD1	TEFHG	\$61.89	\$277.35	\$233.26	\$33.18	\$22.30			\$45.68	\$1.76	\$21.75	\$1.76
	Zone 3	DOT (SHOWN HOLD GOLD LAGGE)											<u> </u>		<del> </del>		
M	ULTIPLEXERS	Channelization - Channel System DS1 to	-	-	LIVEDI	SATC1	\$80.77	\$141.87	\$77.11	\$14.51	\$13.46			\$20.35	\$9.80	\$11.49	\$1.18
-+	OCU-DP COCI (data) - DS1 to DS0 Channel	Interface Unit - Interface DS1 to DSO - OCU - DP Card	<del>                                     </del>		UDL	SATSA	\$1.82	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
-+	System - per month (2.4-64kbs)  2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month	Interface Unit - Interface DS1 to DSO - Brite Card			UDN	SATSA	\$3.10	\$6.07	\$4.66				<u> </u>	\$20.35	\$9.80	\$11.49	\$1.18
1	Voice Grade COCI - DS1 to DS0 Channel	Interface Unit - Interface DS1 to DSO - Voice Grade Card			UEA	SATSA	\$.91	\$6.07	\$4.66					\$20.35	\$9.80	\$11.49	\$1.18
$\vdash$	System - per month DS3 to DS1 Channel System per month	Channelization - Channel System DS3 to DS1			UXTD3	SATCS			\$108.47	\$44.47	\$42.62			\$20.35 \$20.35	\$9.80 \$9.80	\$11.49 \$11.49	\$1.18 \$1.18
$\vdash +$	DS3 to DS1 Channel System per month	Interface Unit - Interface DS3 to DS1		<u> </u>	USL	SATCO	\$17.58	\$6.07	\$4.66		-		<del> </del> -	φ <u>2</u> 0.35	ψ3.60	ψ11.43	
$\vdash$				<b>↓</b> —	<del> </del>	1		+-	1	$\vdash$	<del>                                     </del>						

_				т-				RA1	FS			-		OSSI	RATES		
	UNBUNDLED NETWORK ELEMENT	UNBUNDLED NETWORK ELEMENT AS STATED IN DO	Not in TRA Docket 97- 01262		BCS	usoc				Nonrecurring		Svc Order Submitted Elec	Svc Order Submitted Manually per	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add	Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'l
					l			Nonre First	Add'l		nnect	per LSR SOMEC	LSR	SOMAN	SOMAN	SOMAN	SOMAN
				₩-	——	<b> </b>	Rec	FIRST	Addi	FIFSE	Audi	JOHLO	00				
DUF/EC	DOUF/ADUF/CMDS			==													
T -				-	<b>├</b>					t	-	I					
OP.	TIONAL DAILY USAGE FILE (ODUF)				<b>⊢</b> —	<del>                                     </del>		1-	-	1							
	ODUF: Recording, per message	OSS OLEC Daily Usage File Recording per message		<u> </u>		N/A	\$0.0000044			L.							<u> </u>
	ODUF: Message Processing, per message	OSS OLEC Daily Usage File Message Distribution per message		l	ļ	N/A_	\$0.0027366			<u> </u>							-
	ODUF: Message Processing, per Magnetic Tape provisioned	OSS OLEC Daily Usage File Message Distribution per magnetic tape provisioned				N/A	\$52.75					<u></u>					
	ODUF: Data Transmission (CONNECT:DIRECT), per message	OSS OLEC Daily Usage File Data Transmission (Connect: Direct), per message				N/A	\$0.0000339			<u> </u>		!					

ODUF/ADUF/CMDS

Tennessee

- 2-WIRE ANALOG VOICE GRADE LOOP
- 4-WIRE ANALOG VOICE GRADE LOOP
- 2-WIRE ISDN DIGITAL GRADE LOOP
- 2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP
- 2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP
- 4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP
- 4-WIRE DS1 DIGITAL LOOP
- 4-WIRE 56 OR 64 KBPS DIGITAL GRADE LOOP
- **SUB-LOOP DISTRIBUTION**
- **SUB-LOOP FEEDER**
- **NETWORK INTERFACE DEVICE (NID)**
- UNBUNDLED LOOP CONCENTRATION
- UNBUNDLED SUB-LOOP CONCENTRATION (OUTSIDE CO)
- UNBUNDLED LOCAL EXCHANGE SWITCHING(PORTS)
- UNBUNDLED LOCAL SWITCHING, PORT USAGE
- **COMMON TRANSPORT (SHARED)**
- INTEROFFICE CHANNEL DEDICATED TRANSPORT VOICE GRADE
- INTEROFFICE CHANNEL DEDICATED TRANSPORT- 56/64 KBPS
- INTEROFFICE CHANNEL DEDICATED TRANSPORT DS1
- **LOCAL CHANNEL DEDICATED TRANSPORT**
- MULTIPLEXERS
- DARK FIBER
- **8XX ACCESS TEN DIGIT SCREENING**
- LINE INFORMATION DATA BASE ACCESS (LIDB)
- SIGNALING (CCS7)
- **SELECTIVE ROUTING**
- AIN BELLSOUTH AIN SMS ACCESS SERVICE
- AIN BELLSOUTH AIN TOOLKIT SERVICE
- **OPTIONAL DAILY USAGE FILE (ODUF)**
- UNBUNDLED PORT/LOOP COMBINATIONS COST BASED RATES
- 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)
- 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)
- 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES PBX)
- 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS PBX)
- 2-WIRE VOICE GRADE LOOP- BUS ONLY WITH 2-WIRE DID TRUNK PORT
- 2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT
- 4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT
- 4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT
- 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)
- 4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)
- 4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)
- 4-WIRE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)
- 2-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)
- 4-WIRE VOICE GRADE DEDICATED EXTENDED LOCAL CHANNEL WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)
- 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)

## AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications Corp ("NewSouth") a Delaware corporation and shall be deemed effective as of the date of the last signature of both Parties ("Effective Date").

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and NewSouth dated May 18, 2001 (the "Interconnection Agreement");

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and NewSouth hereby convenant and agree as follows:

- 1. The Parties agree to delete in its entirety Exhibit C to Attachment 2 of the Agreement and replace it with the new Exhibit C, which is attached hereto and incorporated herein by this reference.
- 2. The Parties agree to delete Sections 4 and 5 of Attachment 2 of the Agreement and replace them with the new Sections 4 and 5 contained in Exhibit A to this Amendment, which is attached hereto and incorporated herein by this reference.
- 3. The Parties agree that all of the other provisions of the Interconnection Agreement, dated May 18, 2001 shall remain in full force and effect.
- 4. Either or both of the Parties is authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated below.

BellSouth Telecommunications, Inc.	NewSouth Communications Corp
Signature	Signature
Name	Name
Title	Title
Date	 Date

## 4. Enhanced Extended Link (EEL)

- 4.1 For purposes of this Section, references to "Already Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.
- Where necessary to comply with an effective FCC and/or State Commission order, or as otherwise mutually agreed by the Parties, BellSouth shall offer access to loop and transport combinations, also known as the Enhanœd Extended Link ("EEL") as defined in Section 4.3 below.
- 4.2.2 Subject to Section 4.2.3 below, BellSouth will provide access to the EEL in the combinations set forth in 4.3 following. This offering is intended to provide connectivity from an end user's location through that end user's SWC to NewSouth's POP serving wire center. The circuit must be used for the purpose of provisioning telecommunications services, including telephone exchange service, to NewSouth's enduser customers. Except as provided for in paragraph 22 of the FCC's Supplemental Order Clarification, released June 2, 2000, in CC Docket No. 9698 ("June 2, 2000 Order"), the EEL will be connected to NewSouth's facilities in NewSouth's collocation space at the POP SWC. NewSouth may purchase BellSouth's access facilities between NewSouth's POP and NewSouth's collocation space at the POP SWC.
- 4.2.3 BellSouth shall provide EEL combinations to NewSouth in the state of Georgia and Tennessee regardless of whether or not such EELs are Already Combined. In all other states, BellSouth shall make available to NewSouth those EEL combinations described in Section 4.3 below only to the extent such combinations are Already Combined.
- 4.2.4 BellSouth will make available EEL combinations to NewSouth in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs, regardless of whether or not such EELs are Already Combined.
- 4.2.5 Additionally, BellSouth shall make available to NewSouth a combination of an unbundled loop and tariffed special access interoffice facilities. To the extent NewSouth will require multiplexing functionality in connection with such combination, BellSouth will provide access to multiplexing within the central office pursuant to the terms, conditions and rates set forth in its Access Services Tariffs. The combination of an unbundled loop and tariffed special access interoffice facilities and any associated tariffed services, including but not limited to multiplexing, shall not be eligible for conversion to UNEs as described in Section 4.5 below. Where multiplexing functionality is required in connection with loop and transport combinations, such multiplexing will be provided at the rates and on the terms set forth in this Agreement.

4.3	EEL Combinations
4.3.1	DS1 Interoffice Channel + DS1 Channelization + 2wire VG Local Loop
4.3.2	DS1 Interoffice Channel + DS1 Channelization + 4wire VG Local Loop
4.3.3	DS1 Interoffice Channel + DS1 Channelization + 2-wire ISDN Local Loop
4.3.4	DS1 Interoffice Channel + DS1 Channelization + 4wire 56 kbps Local Loop
4.3.5	DS1 Interoffice Channel + DS1 Channelization + 4wire 64 kbps Local Loop
4.3.6	DS1 Interoffice Channel + DS1 Local Loop
4.3.7	DS3 Interoffice Channel + DS3 Local Loop
4.3.8	STS-1 Interoffice Channel + STS-1 Local Loop
4.3.9	DS3 Interoffice Channel + DS3 Channelization + DS1 Local Loop
4.3.10	STS-1 Interoffice Channel + DS3 Channelization + DS1 Local Loop
4.3.11	2-wire VG Interoffice Channel + 2-wire VG Local Loop
4.3.12	4-wire VG Interoffice Channel + 4-wire VG Local Loop
4.3.13	4-wire 56 kbps Interoffice Channel + 4wire 56 kbps Local Loop
4.3.14	4-wire 64 kbps Interoffice Channel + 4wire 64 kbps Local Loop
4.4	Other Network Element Combinations
	In the state of Georgia and Tennessee, BellSouth shall make available to NewSouth, in accordance with Section 4.6 below: (1) combinations of network elements other than EELs that are Already Combined; and (2) combinations of network elements other than EELs that are not Already Combined but that BellSouth ordinarily combines in its network. In all other states, BellSouth shall make available to NewSouth, in accordance with Section 4.5 below, combinations of network elements other than EELs only to the extent such combinations are Already Combined.
4.5	Special Access Service Conversions
4.5.1	NewSouth may not convert special access services to combinations of loop and transport network elements, whether or not NewSouth selements provides its entrance facilities (or obtains entrance facilities from a third

party), unless NewSouth uses the combination to provide a "significant amount of local exchange service" (as described in Section 4.5.2 below), in addition to exchange access ærvice, to a particular customer. Such conversions of existing special access services pursuant to this section may include facilities within a single density zone (as described in 47 C. F. R. 69.123) or across Density Zones.

- 4.5.1.2 For the purpose of special access conversions under Section 4.5.1, a "significant amount of local exchange service" is as defined in the FCC's June 2, 2000 Order. The Parties agree to incorporate by reference paragraph 22 of the June 2, 2000 Order. When NewSouth requests conversion of special access circuits, NewSouth will selfcertify to BellSouth in the manner specified in paragraph 29 of the June 2, 2000 Order that the circuits to be converted qualify for conversion. In addition there may be extraordinary circumstances where NewSouth is providing a significant amount of local exchange service, but does not qualify under any of the three options set forth in paragraph 22 of June 2, 2000 Order, or under a fourth option set forth below in Section 4.5.2. In such case, NewSouth may petition the FCC for a waiver of the local usage options set forth in the June 2, 2000 Order. If a waiver is granted, then upon NewSouth's request the Parties shall amend this Agreement to the extent necessary to incorporate the terms of such waiver for such extraordinary circumstance.
- 4.5.1.3 The recurring charges for such combinations shall be the sum of the recurring charge for the applicable UNE loop and transport segments (including multiplexing, if applicable), as set forth in Exhibit C to this Attachment. The nonrecurring charges for such combinations shall be an amount equal to all applicable conversion charges set forth in Exhibit C to this Attachment for conversion of special access circuits to EELs, plus all applicable nonrecurring cross connect charges (set forth in Attachment 4 to this Agreement) required to connect the facility to NewSouth's collocation arrangement. EELs that terminate in NewSouth collocation arrangements may be connected by NewSouth via crossconnects to BellSouth services used by NewSouth to transport traffic between NewSouth's collocation space and NewSouth's POP.
- 4.5.1.4 Upon request for conversions of up to 15 circuits from special access to EELs, BellSouth shall perform such conversions within seven (7) days from BellSouth's receipt of a valid, error free service order from NewSouth. Requests for conversions of fifteen (15) or more circuits from special access to EELs will be provisioned on a project basis. Except as set forth in Section 4.5.3 below, conversions should not require the special access circuit to be disconnected and reconnected because only the billing information or other administrative information associated with the circuit will change when NewSouth requests a conversion. Submission of a spreadsheet identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR), only until such time as the LSR process is modified to accommodate such requests.

- 4.5.1.5 BellSouth may, at its sole expense, and upon thirty (30) days notice to NewSouth, audit NewSouth's records not more than once in any twelve month period, unless an audit finds noncompliance with the local usage options referenced in the June 2, 2000 Order, in order to verify the type of traffic being transmitted over combinations of loop and transport network elements. If, based on its audits, BellSouth concludes that NewSouth is not providing a significant amount of local exchange traffic over the combinations of loop and transport network elements,BellSouth may file a complaint with the appropriate Commission, pursuant to the dispute resolution process set forth in this Agreement. In the event that BellSouth prevails, BellSouth may convert such combinations of loop and transport network elements to special access services and may seek appropriate retroactive reimbursement from NewSouth.
- 4.5.2 In addition to the circumstances under which NewSouth may identify special access circuits that qualify for conversions to EELs (referenced in Section 4.5.1.2 above), NewSouth also shall be entitled to convert special access circuits to unbundled network elements pursuant to the terms of this section 4.5.2 et seq.
- 4.5.2.1 Upon request by NewSouth, BellSouth will convert special access circuits to combinations of an unbundled loop connected to special access transport provided that: (1) the combination terminates to a NewSouth collocation arrangement; and (2) NewSouth certifies, in the manner set forth in Section 4.5.2 above, that at least 75% of the unbundled network element(s) component of the facility is used to provide originating and terminating local voice traffic. The recurring charges for such combinations shall be the sum of the recurring charge for the applicable UNE loop, as set forth in Exhibit C to this Attachment, and all applicable recurring charges for the special access transport facility, as set forth in the BellSouth tariff under which such facilities were ordered. The nonrecurring charges for such combinations shall be an amount equal to all applicable conversion charges set forth in Exhibit C to this Attachment for conversion of special access circuits to EELs, plus the applicable nonrecurring cross connect charges (set forth in Attachment 4 to this Agreement) required to connect the facility to NewSouth's collocation arrangement. Such combinations that terminate in NewSouth collocation arrangements may be connected by NewSouth via crossconnects to BellSouth services used by NewSouth to transport traffic between NewSouth's collocation space and NewSouth'sPOP.
- 4.5.2.2 Upon request from NewSouth to convert special access circuits pursuant to Section 4.5.2, BellSouth shall have the right, upon 10 business days notice, to conduct an audit prior to any such conversion to determine whether the subject facilities meet local usage requirements set forth in Section 4.5.2. An audit conducted pursuant to this Section shall take into account a usage period of the past three (3) consecutive months, and shall be subject to the requirements for audits as set forth in the June 2, 2000 Order, except as expressly modified herein.

- 4.5.3 In consideration of Section 4.5.2.1 above, and subject to Section 4.5.7 below, for those special access circuits identified by NewSouth in writing as of January 19, 2001 as being eligible for conversion pursuant to the terms of this Agreement, BellSouth will provide to NewSouth a credit in an amount equal to three times the difference between the monthly special access rates for such circuits and the monthly rates for the combinations to which those circuits are converted.
- 4.5.3.1 For circuits converted pursuant to one of the three options made available to NewSouth in Section 4.5.1, the credit will be in an amount equal to three times the difference between the monthly special access rate for such circuits and the monthly UNE recurring charges for the loop, transport and multiplexing (if applicable), as set forth in Exhibit C to this Attachment, that, in combination, form an EEL.
- 4.5.3.2 For circuits converted pursuant to the fourth option made available to NewSouth in Section 4.5.2, the credit will be in an amount equal to three times the difference between the monthly special access rates for such circuits and the sum of the monthly UNE recurring charges for the loop, as set forth in Exhibit C to this Attachment, and the monthly recurring charge for the special access transport facility, as set forth in the BellSouth tariff under which such facility was ordered.
- 4.5.3.3 Such credits will be applied to NewSouth's bill within sixty (60)days following execution of this Agreement.
- 4.5.3.4 Within ten (10) days following execution of this Agreement, NewSouth shall certify to BellSouth in writing that the circuits designated as of January 19, 2001 meet significant local use requirements ofone of the four conversion options set forth above. Such certification shall include a designation by NewSouth of which of the particular four conversion options specified herein is applicable to each of the individual circuits designated as of January 19 2001.
- 4.5.3.5 BellSouth shall assign a project management team and designate a project manager to facilitate the timely conversion of special access circuits. BellSouth and NewSouth will participate in a joint implementation meeting within fifteen (15) days following execution of this Agreement, or within 15 days of any subsequent request for conversion, to establish a schedule for conversion of the identified special access circuits. BellSouth shall complete conversions of all circuits identified by NewSouth as of January 19, 2001 within 3 months of the joint implementation meeting, unless an alternative completion date is agreed to by the Parties. For purposes of conversion of the circuits identified by NewSouth as of January 19, 2001, NewSouth's speadsheet identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR). For subsequent conversion requests pursuant to Sections 4.5.1 and 4.5.2 above, submission of a spreadsheet identifying the circuits to be converted shall serve as a substitute for submission of a local service request (LSR), only until such time as the LSR process is modified to accommodate such requests.

- 4.5.4 For all special access circuits converted under this Agreement, NewSouth shall pay BellSouth any termination charges applicable to the special access circuits converted, as specified in BellSouth's tariffs.
- 4.5.5 The Parties acknowledge that the conversion option described in Section 4.5.2 and the credits offered NewSouthin Section 4.5.3 constitute a reasonable negotiated alternative to those developed by the FCC in the June 2, 2000 Order. However, BellSouth has agreed to the terms of Sections 4.5.2 and 4.5.3 based upon the assumption that the FCC's current rules regarding special access conversions will remain in effect throughout the 2001 calendar year. In the event that the FCC modifies its rules regarding conversion of special access circuits in a manner that is inconsistent with BellSouth's stated position on the issue, then BellSouth cannot realize the value of the alternative option made available to NewSouth hereunder. In the event that the FCC rules regarding special access conversions are modified in the manner described herein with an effective date prior to January 1, 2002, NewSouth will reimburse BellSouth one-seventh of the credits extended to NewSouth under Section 4.5.3 above for each month or portion thereof prior to January 1, 2002, that such modified FCC rules are in effect.
- 4.6 Rates
- 4.6.1 Georgia
- 4.6.1.1 The non-recurring and recurring rates for the EEL Combinations of network elements set forth in 4.3, whether Already Combined or new, are as set forth in this Attachment.
- 4.6.1.2 On an interim basis, for combinations of loop and transport network elements not set forth in Section 4.3, where the elements are not Already Combined but are ordinarily combined in BellSouth's network, the non recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.
- 4.6.1.3 To the extent that NewSouth seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, NewSouth, at its option, can request that such rates be determined pursuant to the Bona Fide Request/New Busines Request (NBR) process set forth in this Agreement.
- 4.6.2 Tennessee
- 4.6.2.1 The non-recurring and recurring rates for the EEL Combinations of network elements set forth in 5.3.4 whether Currently Combined or new, are as set forth in Exhibit C of this Attachment.

- 4.6.2.2 Where the elements are not Currently Combined but are ordinarily combined in BellSouth's network, the nonrecurring and recurring charges for such UNE combinations shall be the sum of the standalone nonrecurring and recurring charges of the network elementswhich make up the combination.
- 4.6.2.3 To the extent that NewSouth seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, NewSouth, at its option, can request that such rates be determined pursuant to the Bona Fide Request/New Business Request (NBR) process set forth in this Agreement.

### 4.6.3 All Other States

- 4.6.3.1 Subject to Section 4.2.3 and 4.4 preceding, all other states, the rates for (1) Already Combined EEL combinations set forth in Section 4.3, and (2) other combinations of network elements that are Already Combined in the network will be the sum of the recurring rates for the individual network elements plus a nonrecurring charge as specified in Exhibit C of this Attachment.
- 4.6.3.2 Rates for new EEL combinations in Density Zone 1 in the Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs shall be as set forth in Exhibit C hereto; provided, however, that to the extent a rate is not established in Exhibit C, the rate shall be the sum of the recurring and nonrecurring charges for the individual network elements as set forth in Exhibit C to this Attachment, unless otherwise established by the Commission.

## 5. Port/Loop Combinations

- For purposes of this Section, references to "Already Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location. For purposes of this Section, "soft dial tone" (i.e., where network elements are connected through from the end user premises to the BellSouth end office and no dispatch is required to initiate service) shall be considered "Already Combined".
- At NewSouth's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in Section 5.5 below, that are Already Combined in BellSouth's network except as specified in Sections 5.2.1 and 5.2.2 below, consistent with the requirements of 47 C.F.R. 315(b) and all applicable FCC and Commission rules and policies.
- 5.2.1 BellSouth shall not provide access to combinations of unbundled port and loop network elements in locations whee, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.

- 5.2.2 In accordance with effective and applicable FCC rules, BellSouth shall not provide unbundled circuit switching in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, of the Atlanta, Miami, Orlando, Fort Lauderdale, Charlotte, New Orleans, Greensboro and Nashville MSAs to NewSouth if NewSouth's customer has 4 or more DS0 equivalent lines.
- 5.3 Combinations of port and loopnetwork elements provide local exchange service for the origination or termination of calls. BellSouth shall make available the following loop and port combinations at the terms and at the rates set forth below:
- 5.3.2.1 In Georgia and Tennessee, BellSouth shallprovide to NewSouth combinations of port and loop network elements to NewSouth on an unbundled basis regardless of whether or not such combinations are Currently Combined except in those locations where BellSouth is not required to provide circuit switching, as set forth in Section 5.2.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit C of this Attachment.
- In all other states, BellSouth shall provide to NewSouth combinations of port and loop network elements on anunbundled basis if such combinations are Currently Combined, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit C of this Attachment.
- 5.3.2.3 In all states other than Georgia and Tennessee, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2, BellSouthshall provide to NewSouth combinations of port and loop network elements that are not Currently Combined. The rates for such combinations shall be negotiated by the Parties.
- In those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2, BellSouth shall provide to NewSouth combinations of port and loop network elements whether or not such combinations are Currently Combined. The rates for Currently Combined combinations are the market bæed rates as set forth in Exhibit C. The rates for not Currently Combined combinations shall be negotiated by the Parties.
- When NewSouth orders loop/port combinations, and identifies to BellSouth the type of telecommunications service it intends to deliver to its end user customer through that combination (e.g., POTS, ISDN), BellSouth will provide the requested elements with all the functionality, and with at least the same quality of performance and operations systems support (ordering, provisioning, maintenance, billing and recording), that BellSouth provides through its own network to its local exchange service customers receiving equivalent service, unless NewSouth requests a lesser or greater quality of performance through the Bona Fide Request process. BellSouth will provide ordering, provisioning and maintenance

services, including intervals, at parity with the same services BellSouth provides to its own end users or resold services as measured in Attachment 9 Performance Measures. The intervalsthat BellSouth provides for its products and services are as set forth in the Products and Services Interval Guide which can be found on the BellSouth Interconnection website at www.interconnection.BellSouth.com. The Products and Services Interval Guide may be amended from time to time. Any intervals contained in The Products and Service Interval Guide will not be increased unless ordered to do otherwise by the appropriate regulatory or judicial body. BellSouth's intervals begin with the receipt of an error free local service request (LSR). At the time of this interconnection agreement, not all combinations can be ordered electronically. All residence, business, and PBX port loop services can be electronically ordered. BellSouth will provide manual orderingprocesses for loop port combinations which cannot be electronically processed. BellSouth will provide notice of additional electronic ordering functionality via the Change Control Process.

- 5.4 Rates for Combinations of Loop and Port Network Elements
- 5.4.1 Rates for combinations of loop and port network elements, as set forth in Section 5.4, are provided in Exhibit A of this Attachment.
- 5.4.2 Rates for Circuit Switching
- 5.4.2.1 Rates for circuit switching, where BellSouth is not required, pursuant to Sections 5.1.1 and 5.1.2, to provide circuit switching are as set forth in Exhibit A of this Attachment.
- 5.5 Port/Loop Combination Offerings
- 5.5.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.2 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mie per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.3 2-wire CENTREX port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.4. 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

- 5.5.5 2-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem swtching, and tandem trunk port.
- 5.5.6 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 5.5.7 4-wire DS1Loop with normal serving wire center channelization interface, 2 wire voice grade ports (PBX), 2-wire DID ports, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

ORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
							Nonre	curring			Svc Order Submitted Elec per LSR		Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic-Di Add'I
						Rec	First	Add'I	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
						1100	11100	Aug I	1 11 54	Auu	COMEG	COMPAN	COMPA	COMPA	COMPA	COMPAR
		hown in the sections for stand-alone loops or loops as part of a combination refers to Greenection.bellsouth.com/become_a_clec/html/interconnection.htm	eographically	Deaveraged UNE Zor	nes. To view Geo	ographically Dea	everaged UNE Z	one Designation	ns by Central (	Office, refer to	Internet Wel	osite:				
NDLED I	EXCHANGE	ACCESS LOOP														
2-	-WIRE ANA	LOG VOICE GRADE LOOP														
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	1	UEANL	UEAL2	15.24	59.03	43.14	15.21	3.22			27.37	12.97	17.77	
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	2	UEANL	UEAL2	24.75	59.03	43.14	15.21	3.22			27.37	12.97	17.77	
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	3	UEANL	UEAL2	44.85	59.03	43.14	15.21	3.22			23.97	12.97	17.77	
_		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone	2	UEPSR, UEPSB UEPSR, UEPSB	UEALS UEALS	15.24 24.75	59.03 59.03	43.14 43.14	15.21 15.21	3.22			27.37 27.37	12.97 12.97	17.77 17.77	
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone	3	UEPSR, UEPSB	UEALS	44.85	59.03	43.14		3.22			23.97	12.97	17.77	
-+		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone Engineering Information Document (E	3	UEANL	UEALO	44.00	28.75	28.75	15.21	3.22			23.81	12.31	17.77	17
— †		Manual Order Coordination for UVL-SL1s (per loop		UEANL	UEAMC	1	51.29	51.29	1	<b> </b>						<b>—</b>
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR		UEANL	OCOSL	1	45.99	45.99	1							<b>†</b>
		the second secon		22/3/2	22002		.5.00	.5.00						<u> </u>		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1	1	UEA	UEAL2	17.95	145.46	108.4	40.31	26.01			27.37	12.97	17.77	17
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 2	2	UEA	UEAL2	29.16	145.46	108.4	40.31	26.01			27.37	12.97	17.77	17
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -	3													
		Zone 3 Order Coordination for Specified Conversion Time (per LS	3	UEA UEA	UEAL2 OCOSL	52.84	145.46 45.99	108.4	40.31	26.01			27.37	12.97	17.77	17
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zon	1	UEA	UEAR2	17.95	145.46	108.4	40.31	26.01			27.37	12.97	17.77	17
			2													
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zon	2	UEA	UEAR2	29.16	145.46	108.4	40.31	26.01			27.37	12.97	17.77	17
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zon Order Coordination for Specified Conversion Time (per LS	3	UEA UEA	UEAR2 OCOSL	52.84	145.46 45.99	108.4	40.31	26.01			27.37	12.97	17.77	17
4-		LOG VOICE GRADE LOOP		OLA	OCOSE		45.55									•
		4-Wire Analog Voice Grade Loop - Zone	1	UEA	UEAL4	24.01	293.7	241.76	108.96	57.01			27.37	12.97	17.77	1
		4-Wire Analog Voice Grade Loop - Zone	2	UEA	UEAL4	39	293.7	241.76	108.96	57.01			27.37	12.97	17.77	17
		4-Wire Analog Voice Grade Loop - Zone	3	UEA	UEAL4	70.67	293.7	241.76	108.96	57.01			27.37	12.97	17.77	1
		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		45.99									
2-		DIGITAL GRADE LOOP														
		2-Wire ISDN Digital Grade Loop - Zone	1	UDN	U1L2X	23.23	331.85	255.87	108.95	57.01			27.37	12.97	17.77	
		2-Wire ISDN Digital Grade Loop - Zone	2	UDN	U1L2X	37.74	331.85	255.87	108.95	57.01			27.37	12.97	17.77	
		2-Wire ISDN Digital Grade Loop - Zone Order Coordination For Specified Conversion Time (per LS)	3	UDN UDN	U1L2X OCOSL	68.38	331.85	255.87	108.95	57.01			27.37	12.97	17.77	1
		Order Coordination For Specified Conversion Time (per LS		UDIN	UCUSL		45.99									
2.	-WIRF Univ	ersal Digital Channel (UDC) COMPATIBLE LOOP														
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	I 1	UDC	UDC2X	16.84	104.17	78.1	108.95	57.01			18.94	8.42	17.77	1
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	I 2	UDC	UDC2X	19.45	104.17	78.1	108.95	57.01			18.94	8.42	17.77	
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	1 3	UDC	UDC2X	30.92	104.17	78.1	108.95	57.01			18.94	8.42	17.77	17
2-	-WIRE ASYI	MMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP														
		2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP  2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -														
		Zone 1  2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -	1	UAL	UAL2X	12.09	514.21	464.58	106.65	56.98			27.37	12.97	17.77	1
		Zone 2	2	UAL	UAL2X	19.64	514.21	464.58	106.65	56.98			27.37	12.97	17.77	1
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 3	3	UAL	UAL2X	35.59	514.21	464.58	106.65	56.98			27.37	12.97	17.77	1
		Order Coordination for Specified Conversion Time (per LS		UAL	OCOSL	1	45.99			-						
			1	1	UAL2W	12.09	204.88	129.08	100.52	15.82			27.37	12.97	17.77	17
		2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zon	1	UAL	UNLEVV						1					
		Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zon     Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zon	2	UAL	UAL2W	19.64	204.88	129.08	100.52	15.82			27.37	12.97	17.77	17
		2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zon	2				204.88							12.97		
			2	UAL	UAL2W	19.64 35.59		129.08	100.52	15.82			27.37		17.77 17.77	

2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP													
2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -													
Zone 1  2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -	₩	1 UHL	UHL2X	9.41	514.21	464.58	106.65	56.98		27.37	12.97	17.77	
Zone 2		2 UHL	UHL2X	15.29	514.21	464.58	106.65	56.98		27.37	12.97	17.77	
2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -	1												
Zone 3	<u> </u>	3 UHL	UHL2X	27.7	514.21	464.58	106.65	56.98		27.37	12.97	17.77	
Order Coordination for Specified Conversion Time (per LS  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -	┿	UHL	OCOSL		45.99								
Zone 1		1 UHL	UHL2W	9.41	222.2	146.4	100.52	15.82		27.37	12.97	17.77	
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -	1												
Zone 2	<u> </u>	2 UHL	UHL2W	15.29	222.2	146.4	100.52	15.82		27.37	12.97	17.77	
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		3 UHL	UHL2W	27.7	222.2	146.4	100.52	15.82		27.37	12.97	17.77	
Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL	21.1	45.99	146.4	100.52	15.82		21.31	12.97	17.77	
4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP													
4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation	1												
Zone 1		1 UHL	UHL4X	11.52	541.13	491.5	106.65	56.98		27.37	12.97	17.77	
4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation	1	2 UHI	111111 437	40.74	544.40	404.5	400.05	50.00		07.07	40.07	4	
Zone 2  4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation		2 UHL	UHL4X	18.71	541.13	491.5	106.65	56.98		27.37	12.97	17.77	
Zone 3		3 UHL	UHL4X	33.9	541.13	491.5	106.65	56.98		27.37	12.97	17.77	
Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		45.99								
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -				44.50	070.00	000 50	400.00	00.7		07.07	40.07	4	
Zone 1  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -	+	1 UHL	UHL4W	11.52	279.39	203.59	109.99	20.7		27.37	12.97	17.77	
Zone 2		2 UHL	UHL4W	18.71	279.39	203.59	109.99	20.7		27.37	12.97	17.77	
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -													
Zone 3		3 UHL	UHL4W	33.9	279.39	203.59	109.99	20.7		27.37	12.97	17.77	
Order Coordination for Specified Conversion Time (per LS	+	UHL	OCOSL		45.99								
4-WIRE DS1 DIGITAL LOOP													
4-Wire DS1 Digital Loop - Zone		1 USL	USLXX	51.74	610.13	380.26	134.77	55.97		27.37	12.97	17.77	
4-Wire DS1 Digital Loop - Zone :	<b>├</b>	2 USL	USLXX	84.05 152.29	610.13	380.26	134.77	55.97		27.37	12.97	17.77	
4-Wire DS1 Digital Loop - Zone :  Order Coordination for Specified Conversion Time (per LS	+	3 USL USL	OCOSL	152.29	610.13 49.18	380.26	134.77	55.97		27.37	12.97	17.77	
		002	00002		10.10								
4-WIRE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP													
4 Wire Unbundled Digital 19.2 Kbps	<del> </del>	1 UDL	UDL19	27.33	498.05	343.7	129.62	64.25		27.37	12.97	17.77	
4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital 19.2 Kbps	<del></del>	2 UDL 3 UDL	UDL19 UDL19	44.4 80.45	498.05 498.05	343.7 343.7	129.62 129.62	64.25 64.25		27.37 27.37	12.97 12.97	17.77 17.77	
4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital Loop 56 Kbps - Zone	+	1 UDL	UDL19	27.33	498.05	343.7	129.62	64.25		27.37	12.97	17.77	
4 Wire Unbundled Digital Loop 56 Kbps - Zone	t	2 UDL	UDL56	44.4	498.05	343.7	129.62	64.25		27.37	12.97	17.77	
4 Wire Unbundled Digital Loop 56 Kbps - Zone		3 UDL	UDL56	80.45	498.05	343.7	129.62	64.25		27.37	12.97	17.77	
Order Coordination for Specified Conversion Time (per LS	<del> </del>	UDL	OCOSL		45.99								
4 Wire Unbundled Digital Loop 64 Kbps - Zone 4 Wire Unbundled Digital Loop 64 Kbps - Zone	+	1 UDL 2 UDL	UDL64 UDL64	27.33 44.4	498.05 498.05	343.7 343.7	129.62 129.62	64.25 64.25		27.37 27.37	12.97 12.97	17.77 17.77	
4 Wire Unbundled Digital Loop 64 Kbps - Zone	+	3 UDL	UDL64	80.45	498.05	343.7	129.62	64.25		27.37	12.97	17.77	
Order Coordination for Specified Conversion Time (per LS		UDL	OCOSL		45.99								
2-WIRE Unbundled COPPER LOOP  2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility													
reservation - Zone 1		1 UCL	UCLPB	11.9	283.37	163.68	120.15	22.37		18.94	8.42		
2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility													
reservation - Zone 2	<del> </del>	2 UCL	UCLPB	13.74	283.37	163.68	120.15	22.37		18.94	8.42		
2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3		3 UCL	UCLPB	21.83	283.37	163.68	120.15	22.37		18.94	8.42		
Order Coordination for Unbundled Copper Loops (per loc	<del>                                     </del>	UCL	UCLMC	21.03	51.29	51.29	120.13	22.31		10.54	0.42		
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility	1												
reservation - Zone 1		1 UCL	UCLPW	11.9	104.17	78.1				18.94	8.42		
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2		2 UCL	UCLPW	13.74	104.17	78.1				18.94	8.42		
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility	<del>  '-</del>	2 0CL	OCLI W	13.74	104.17	70.1				10.34	0.42		
reservation - Zone (	1	3 UCL	UCLPW	21.83	104.17	78.1				18.94	8.42		
Order Coordination for Unbundled Copper Loops (per loc	$\perp$	UCL	UCLMC		51.29	51.29							
2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility		1 UCL	UCL2L	35.43	270.20	150.59	120.15	22.37		40.04	9.40		
reservation - Zone 1  2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	+	I UCL	UCLZL	35.43	270.28	150.59	120.15	22.31		18.94	8.42		
		2 UCL	UCL2L	40.91	270.28	150.59	120.15	22.37		18.94	8.42		
reservation - Zone 2													
reservation - Zone 2 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility							120.15	22.37	1 1	18.94	8.42		
reservation - Zone :  2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ;		3 UCL	UCL2L	65.02	270.28	150.59	120.15	22.01		10.01	0.12		
reservation - Zone ?  2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ?  Order Coordination for Unbundled Copper Loops (per loc		3 UCL UCL	UCL2L UCLMC	65.02	270.28 51.29	150.59 51.29	120.15	22.01		10.01	0.12		
reservation - Zone :  2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility	1	UCL	UCLMC		51.29	51.29	120.15	22.01					
reservation - Zone ?  2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ?  Order Coordination for Unbundled Copper Loops (per loc	1	UCL		35.43			120.15	22.01		18.94	8.42		

2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility			1101	1101 0141	05.00	101.17	70.4			40.04	0.40	
reservation - Zone ( Order Coordination for Unbundled Copper Loops (per loc	- !	3	UCL UCL	UCL2W UCLMC	65.02	104.17 51.29	78.1 51.29			18.94	8.42	
2-Wire Unbundled Copper Loop - Non-Designed Zone	1	1	UEQ	UEQ2X	11.01	44.69	22.4	25.65	7.06	27.37	12.97	
2 Wire Unbundled Copper Loop - Non-Designed - Zone	i	2	UEQ	UEQ2X	12.67	44.69	22.4	25.65	7.06	27.37	12.97	
2 Wire Unbundled Copper Loop - Non-Designed - Zone	- 1	3	UEQ	UEQ2X	20.22	44.69	22.4	25.65	7.06	27.37	12.97	
Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per lo			UEQ	USBMC		51.29	51.29					
Engineering Information Documer			UEQ UEQ	URET1		28.75 78.92	28.75 78.92					
Loop Testing - Basic 1st Half Hou Loop Testing - Basic Additional Half Hou			UEQ	URETA		23.33	23.33					
Loop rooming Saddo Maditional Hair Hox			oz q	OKETA		20.00	20.00					
4-WIRE COPPER LOOP												
4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - 2	d											
		1	UCL	UCL4S	16.65	331.78	212.09	130.69	27.6	27.37	8.42	
4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - 2	d											
4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - 2		2	UCL	UCL4S	19.22	331.78	212.09	130.69	27.6	18.94	8.42	
3	1	3	UCL	UCL4S	30.55	331.78	212.09	130.69	27.6	18.94	8.42	
Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		36.46	36.46					
A Miles Connection (Chart with out annual annual facility annual facility annual facility annual facility annual facility annual facility annual facility annual facility annual facility annual facility annual facility and facility annual facility and facility annual facility and facility annual facility and facility			HO		40.05	404.47	70.4			40.04	0.40	
4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zo	ı I	1	UCL	UCL4W	16.65	104.17	78.1			18.94	8.42	
4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zo	1	2	UCL	UCL4W	19.22	104.17	78.1			18.94	8.42	
AW(5-0 C   Ch		_	1101		20.55	404.47	70.4			40.04	0.40	
4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zo Order Coordination for Unbundled Copper Loops (per loc	l I	3	UCL UCL	UCL4W UCLMC	30.55	104.17 36.46	78.1 36.46			18.94	8.42	
4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			OOL	OOLIVIO		30.40	50.40					
reservation - Zone 1		1	UCL	UCL4L	47.56	318.7	199	130.69	27.6	18.94	8.42	
4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		_	UCL	UCL4L	54.00	040.7	400	400.00	07.0	40.04	0.40	
reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		2	UCL	UCL4L	54.92	318.7	199	130.69	27.6	18.94	8.42	
reservation - Zone (		3	UCL	UCL4L	87.3	318.7	199	130.69	27.6	18.94	8.42	
Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		36.46	36.46					
4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reserva	i .		1101	1101.40	47.50	404.47	70.4			40.04	0.40	
Zone 1  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reserva	i !	1	UCL	UCL40	47.56	104.17	78.1			18.94	8.42	
Zone 2	1	2	UCL	UCL4O	54.92	104.17	78.1			18.94	8.42	
4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reserva	i											
Zone 3 Order Coordination for Unbundled Copper Loops (per loc	- 1	3	UCL UCL	UCL40 UCLMC	87.3	104.17 36.46	78.1			18.94	8.42	
Order Coordination for Onbundled Copper Loops (per loc			UCL	UCLIVIC		30.40	36.46					
LOOP MODIFICATION												
Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal t	٠.		UAL, UHL, UCL, UEQ, ULS	ULM2L		67.39	67.39					
Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18	<u> </u>		UCL, ULS	ULM2G		337.5	337.5					
Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18	I		UHL, UCL	ULM4L		67.39	67.39					
Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18			UCL	ULM4G		337.5	337.5					
Onburidica Ecop Modification Nethoval of Ecod Colls - 4 Ville pail gledter trial 10			UAL, UHL, UCL,	OLIVI40		331.3	337.3			1		
Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled le	- 1		UEQ, UEF, ULS	ULMBT		78.1	78.1					
SUB-LOOPS	-											
000 2001 0										+		
Sub-Loop Distribution		L										
Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L			UEANL	USBSA		421.08	421.08			18.94	8.42	
Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L	!		UEANL	USBSB		67.1	67.1			18.94	8.42	
Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I			UEANL UEANL	USBSC		394.74 154.57	394.74 154.57		+	18.94 18.94	8.42 8.42	+
Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statew	<u> </u>	sw	UEANL	USBN2	9.12	207.01	171.32			18.94	8.42	
Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		45.99	45.99					
Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Statewi	-	SW	UEANL	USBN4	8.32	219.35	72.99	123.72	28.77	18.94	8.42	
Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (IN(		-	UEANL UEANL	USBMC USBR2	1.61	45.99 137.03	45.99 41.59	115.85	19.17	18.94	8.42	
Order Coordination for Unbundled Sub-Loops, per sub-loop pair	<u> </u>		UEANL	USBMC	1.01	45.99	45.99	110.00	13.17	10.34	0.42	
Sub-Loop 4-Wire Intrabuilding Network Cable (INC	1		UEANL	USBR4	2.96	176.46	55.11	122.17	19.57	18.94	8.42	
Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		45.99	45.99					
2 Wire Copper Unbundled Sub-Loop Distribution - Statewi		SW	UEF	UCS2X	5.54	175.16	55.5	108.86	24.53	18.94	8.42	
Order Coordination for Unbundled Sub-Loops, per sub-loop pair 4 Wire Copper Unbundled Sub-Loop Distribution - Statewi		SW	UEF UEF	USBMC UCS4X	6.89	45.99 219.35	45.99 72.99	123.72	28.77	18.94	8.42	+
Order Coordination for Unbundled Sub-Loops, per sub-loop pair		311	UEF	USBMC	0.00	45.99	45.99	.20.12	25.11	10.54	0.72	
Sub-Loop Feeder	L	<u> </u>										

Page 3 of 20 Version 2Q01: 08/30/01

		UEA,											
	1101 F - 1 - D00 0 4 0 - D - 1 C 01 F 0 D - 4 C	UDN,UCL,UDL,UD	HODEW		404.00								
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-	UEA,	USBFW		421.08								
		UDN.UCL.UDL.UD											
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u	C	USBFX		67.1	67.1							
	USL Feeder DS1 Set-up at DSX location, per DS1 terminatio	USL	USBFZ		519.95	11.32							
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade- Statew	sw UEA	USBFA	8.58	206.44	170.05	119.95	27.04		18.94	8.42		
	Order Coordination for Specified Conversion Time, per LSR	UEA	OCOSL		45.99								
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Statew	sw UEA	USBFB	8.58	206.44	170.05	119.95	27.04		18.94	8.42		
	Order Coordination for Specified Time Conversion, per LSR	UEA	OCOSL		45.99								
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade Loop - Statewide	sw UEA	HODEO	8.58	206.44	170.05	440.05	07.04		40.04	8.42		
	Order Coordination For Specified Conversion Time, per LS	sw UEA UEA	USBFC OCOSL	8.58	45.99	170.05	119.95	27.04		18.94	8.42		
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Statew	sw UEA	USBFD	19.91	243.41	81.32	134.77	33.93		18.94	8.42		
	Order Coordination For Specified Conversion Time, Per LS	UEA	OCOSL	13.51	45.99	01.02	104.77	33.33		10.54	0.42		
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Statew	sw UEA	USBFE	19.91	243.41	81.32	134.77	33.93		18.94	8.42		
	Order Coordination For Specified Conversion Time, Per LS	UEA	OCOSL		45.99								
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Statewii	sw UDN	USBFF	17.73	208.5	62.31	119.68	29.58		19.99	19.99	19.99	19.9
	Order Coordination For Specified Conversion Time, Per LS	UDN	OCOSL		45.99								
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	sw UDC	USBFS	17.73	208.5	62.31	119.68	29.58		19.99	19.99	19.99	19.9
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewic	sw USL	USBFG	79.3	203.69	128.76	124.09	34.8		19.99	19.99	19.99	19.9
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi	sw UCL	OCOSL USBFH	7.22	45.99 195.38	63.15	119.68	29.58		18.94	8.42		
	Order Coordination For Specified Conversion Time, per LS	UCL	OCOSL	1.22	45.99	03.13	119.00	29.56		10.94	0.42		
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewin	sw UCL	USBFJ	13.72	243.41	81.32	134.77	33.93		18.94	8.42		
	Order Coordination For Specified Conversion Time, per LS	UCL	OCOSL	10.72	45.99	01.02		55.55		.0.04	0.12		
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Lo	sw UDL	USBFN	24.5	243.41	81.32	134.77	33.93		19.99	19.99	19.99	19.9
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewi	sw UDL	USBFO	24.5	243.41	81.32	134.77	33.93		19.99	19.99	19.99	19.9
	Order Coordination For Specified Time Conversion, per LS	UDL	OCOSL		45.99								
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewi	sw UDL	USBFP	24.5	243.41	81.32	134.77	33.93		19.99	19.99	19.99	19.9
	Order Coordination For Specified Conversion Time, per LS	UDL	OCOSL		45.99								
Unb	pundled Sub-Loop Modification												
Olib	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W												
	PR	UEF	ULM2X		355.71	12.26				18.94	8.42		
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W	02.	CLINEX		000.71	12.20					02		
	PR	UEF	ULM4X		355.71	12.26				18.94	8.42		
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR												
	unloaded	UEF	ULM4T		560.55	14.3				18.94	8.42		
Unb	oundled Network Terminating Wire (UNTW)												
	Unbundled Network Terminating Wire (UNTW) per Pa	UENTW	UENPP	1.37	2.48	2.48	1.74	1.74		18.94	8.42		
Note	work Interface Device (NID)												
INCL	Network Interface Device (NID) - 1-2 line	UENTW	UND12		86.46	56.75				18.94	8.42		
	Network Interface Device (NID) - 1-6 line	UENTW	UND16		127.93	98.21				18.94	8,42		
	Network Interface Device Cross Connect - 2 V	UENTW	UNDC2		11.73	11.73				18.94	8.42		
	Network Interface Device Cross Connect - 4V	UENTW	UNDC4		11.73	11.73				18.94	8.42		
UNBUNDLED LO	OOP CONCENTRATION												
	Unbundled Loop Concentration - System A (TR008	ULC	UCT8A	441.42	650.81	650.81				19.99	19.99	19.99	19.9
	Unbundled Loop Concentration - System B (TR008	ULC	UCT8B	52.97	271.17	271.17				19.99	19.99	19.99	19.9
	Unbundled Loop Concentration - System A (TR30)	ULC	UCT3A UCT3B	478.93	650.81 271.17	650.81 271.17				19.99	19.99	19.99	19.9
	Unbundled Loop Concentration - System B (TR30: Unbundled Loop Concentration - DS1 Loop Interface Ca	ULC	UCTCO	89.26 5.04	126.57	92.14	33.57	9.4		19.99	19.99	19.99	19.9
	Unbundled Loop Concentration - ISDN Loop Interface Ca	UDN	ULCC1	8	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.9
	Unbundled Loop Concentration - UDC Loop Interface (Brite Car	UDC	ULCCU	8	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.9
	Unbundled Loop Concentration 2 Wire Voice-Loop Start or Ground Start Loop Interface												
	(POTS Card)	UEA	ULCC2	2	21.07	20.96	10.78	10.71		18.94	8.42		
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface (SPOT\$												
	Card)	UEA	ULCCR	11.89	21.07	20.96	10.78	10.71		18.94	8.42		
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca	UEA	ULCC4	7.09	21.07	20.96	10.78	10.71		18.94	8.42		
	Unbundled Loop Concentration - TEST CIRCUIT Car	ULC	UCTTC	34.67	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.9
		UDL	ULCC7 ULCC5	10.51 10.51	21.07 21.07	20.96 20.96	10.78 10.78	10.71 10.71		19.99 19.99	19.99 19.99	19.99 19.99	19.9 19.9
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa	LIDI		10.51				10.71	1				19.9
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa	UDL		10.51	21.07	20.96	10.78			19 90	19 99		
		UDL	ULCC6	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa			10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
UNBUNDLED SU	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa			10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
UNBUNDLED SU	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa			10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  Unbundled Loop Concentration (Outside Co)			10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  Unbundled Loop Concentration (OUTSIDE CO)  (OVISIONING ONLY - NO RATE	UDL	ULCC6	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  UB-LOOP CONCENTRATION (OUTSIDE CO)  OUTSIONING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation	UDL	ULCC6 UNDBX	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  Unbundled Loop Concentration (OUTSIDE CO)  (OVISIONING ONLY - NO RATE	UENTW UENTW	ULCC6	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  UB-LOOP CONCENTRATION (OUTSIDE CO)  OUTSIONING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation	UDL	ULCC6 UNDBX	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	

Page 4 of 20

		UAL,UCL,UDC,UDL											
	Unbundled Contact Name Provisioning Only no rate	,UDN,UEA,UHL,UL C	UNECN	0	0								
	Unbundled Contact Name, Provisioning Only - no rate	C	UNECN	U	U								
		UEA,UDN,UCL,UD											
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra	C	USBFQ	0	0								
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra	UEA,USL,UCL,UDL	USBFR	0	0								<b></b>
	Unbundled DS1 Loop - Superframe Format Option - no ra	USL	CCOSF	0	0								
	Unbundled DS1 Loop - Expanded Superframe Format option - no ra	USL	CCOEF	0	0								
HIGH CAPACITY UNBU	INDLED LOCAL LOOP												
	month minimum billing period												
	High Capacity Unbundled Local Loop - DS3 - Per Mile per mon	UE3	1L5ND	10.16									
	High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor	UE3	UE3PX	374.52	903.03	527.87	238.97	167.16		31.31	31.31	3.93	3.93
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor	UDLSX	1L5ND	10.16									<b></b>
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor	UDLSX	UDLS1	387.67	903.03	527.87	238.97	167.16		31.31	31.31	3.93	3.93
LOOP MAKE-UP													1
LOUP MAKE-UP	Loop Makeup - Preordering Without Reservation, per working or spare facility queried												
	(Manual).	UMK	UMKLW		131.22	131.22							1
	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).	UMK	UMKLP		136.93	136.93							
	Loop MakeupWith or Without Reservation, per working or spare facility queried												
	(Mechanized)	UMK	PSUMK		0.9809855	0.9809855							1
LINE SHARING													<b></b>
	Line Obesies Califfred and Contant OC Line Contain	111.0	LII CDA	450.7	224.00	-	054.70		-	1	1		<del></del>
<del></del>	Line Sharing Splitter, per System 96 Line Capaci I Line Sharing Splitter, per System 24 Line Capaci I	ULS ULS	ULSDA ULSDB	152.7 38.18	221.09 221.09	0	254.79 254.79	0	0	+	-		
	Line Sharing Splitter, per System 24 Line Capaci	ULS	ULSD8	12.73	221.09	0	254.79	0	0	+	1		
	Line Sharing - per Line Activatio	ULS	ULSDC	0.61	39.09	20.94	22.15	9.46		27.37	12.97	17.77	17.77
	Line Sharing - per Subsequent Activity per Line Rearrangeme I	ULS	ULSDS		34.9	16.18				27.37	12.97		
													<b></b>
	Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lines		ULSDG				44.00						[
	(16 pair) I	ULS	ULSDG		57.7		11.39						
UNBUNDLED TRANSP	ORT												
ONDONDEED INVINO													
COMMON	TRANSPORT (Shared)												
	Common Transport - Per Mile, Per MOI			0.00001									
	Common Transport - Facilities Termination Per MO			0.00045									<b></b>
NOTE IN	TERRETOR CHANNEL PERIOATER TRANSPORT	4 200 1 1 (											+
NOTE: IN	TEROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS3 = one mor	nth, DS3 and above four m	ontns										
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE												
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per mo	U1TVX	1L5XX	0.0101									
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per												
	month	U1TVX	U1TV2	24.15	81.07	54.82	33.47	13.79		31.31	31.31	3.93	3.93
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per												[
	month	U1TVX	1L5XX	0.0101									+
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination per month	U1TVX	U1TR2	24.15	81.07	54.82	33.47	13.79		31.31	31.31	3.93	3.93
	inonui	UIIVA	UTINZ	24.13	01.07	34.02	33.41	13.13		31.31	31.31	5.95	3.93
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month	U1TVX	1L5XX	0.0101						1			I
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination per												
	month	U1TVX	U1TV4	21.41	81.07	54.82	33.47	13.79		31.31	31.31	3.93	3.93
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per moi	U1TDX	1L5XX	0.0101	04.07	54.82	20.47	40.70		24.24	24.24	0.00	0.00
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mo	U1TDX U1TDX	U1TD5 1L5XX	17.28 0.0101	81.07	54.82	33.47	13.79		31.31	31.31	3.93	3.93
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per moi Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mo	U1TDX	U1TD6	17.28	81.07	54.82	33.47	13.79		31.31	31.31	3.93	3.93
		31107	0.100	.7.20	31.01	3 1.02	55.77			31.01	31.01	5.33	5.55
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT - DS1												
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mon	U1TD1	1L5XX	0.2067									
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor	U1TD1	U1TF1	68.75	178.53	163.61	32.7	28.88		31.31	31.31	3.93	3.93
	FIGE CHANNEL DEDICATED TRANSPORT DC2					1				1			l
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT - DS3  Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor	U1TD3	1L5XX	4.67						1	+		
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor	U1TD3	U1TF3	804.02	557.49	325.51	120.39	116.91		31.31	31.31	3.93	3.93
	Interesting Statistics Sedicated Harisport See Facility Fernillation per III0	01100	01110	004.02	337.43	020.01	120.03	. 10.01		01.01	31.01	5.55	0.50
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT- STS-1												
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mor	U1TS1	1L5XX	4.67									
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mo	U1TS1	U1TFS	801.57	557.49	325.51	120.39	116.91		31.31	31.31	3.93	3.93
				1						4			+
						1				1			<b>—</b>
				-					_	+	-		
1 1	I			-	1	+	1	<del>                                     </del>		1		1	
ו טכאו כו	HANNEL - DEDICATED TRANSPORT												

	NOTE: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month,	193 and above—four months							T			
	Local Channel - Dedicated - 2-Wire Voice Grade Per Month	ULCVX	ULDV2	15.96	386.19	66.33	73.28	6.39	31.31	31.31	3.93	3.9
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor	ULCVX	ULDR2	15.96	386.19	66.33	73.28	6.39	31.31	31.31		
1											3.93	3.9
	Local Channel - Dedicated - 4-Wire Voice Grade per mon	UNCVX	ULDV4	17.06	387.19	67.2	74.22	7.33	31.31	31.31	3.93	3.9
	Local Channel - Dedicated - DS1 per month - Zone	1 ULDD1	ULDF1	41.52	354.94	307.43	44.38	30.52	31.31	31.31	3.93	3.9
	Local Channel - Dedicated - DS1 per month - Zone	2 ULDD1	ULDF1	61.05	354.94	307.43	44.38	30.52	31.31	31.31	3.93	3.9
	Local Channel - Dedicated - DS1 per month - Zone	3 ULDD1	ULDF1	47.29	354.94	307.43	44.38	30.52	31.31	31.31	3.93	3.9
	Local Channel - Dedicated - DS3 - Per Mile per mon	ULDD3	1L5NC	7.91								
	Local Channel - Dedicated - DS3 - Facility Termination per mon	ULDD3	ULDF3	476.04	903.03	527.87	238.87	167.16	31.31	31.31	3.93	3.9
	Local Channel - Dedicated - STS-1- Per Mile per mon	ULDS1	1L5NC	7.91	000.00	027.07	200.07	107.10	01.01	01.01	0.00	0.0
			ULDFS		000.00	507.07	000.07	407.40	24.24	24.24	2.02	2.0
	Local Channel - Dedicated - STS-1 - Facility Termination per mor	ULDS1	ULDFS	466.84	903.03	527.87	238.87	167.16	31.31	31.31	3.93	3.9
MULTIPLE												
	Channelization - DS1 to DS0 Channel Syster	UXTD1	MQ1	122.5	182.08	125.14	21.07	19.58	31.31	31.31	3.93	3.9
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb	UDL	1D1DD	1.36	13.15	9.43						
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per mon!	UDN	UC1CA	2.92	13.15	9.43						
	Voice Grade COCI - DS1 to DS0 Channel System - per mon	UEA	1D1VG	0.64	13.15	9.43						
	DS3 to DS1 Channel System per monti	UXTD3	MQ3	201.37	356.28	187.94	66.51	63.65	31.31	31.31	3.93	3.
			MOS		330.20	107.94	00.51	03.03				
	STS1 to DS1 Channel System per montl	UXTS1	MQ3	201.37				-	31.31	31.31	3.93	3.
	DS3 Interface Unit (DS1 COCI) used with Loop per mont	USL	UC1D1	15.39	13.15	9.43						
DARK FIBE	ER										1	
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local											
	Channel	UDF	1L5DC	68.84							1	
<del></del>		UDF	UDFC4	00.04	1278.17	275.73	634.11	395.32	31.31	31.31	3.93	3.
<b></b>	NRC Dark Fiber - Local Channe	UDF	UDFC4	-	12/8.1/	215.13	034.11	393.32	31.31	31.31	3.93	3.
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Interoffice										1	
	Channel	UDF	1L5DF	25.53					1			
	NRC Dark Fiber - Interoffice Channe	UDF	UDF14		1278.17	275.73	634.11	395.32	31.31	31.31	3.93	3.
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local			1		1			1			
1	Loop	UDF	1L5DL	68.84							ı	
	NRC Dark Fiber - Local Loop	UDF	UDFL4	00.0.	1278.17	275.73	634.11	395.32	31.31	31.31	3.93	3.
TRANSPOR	TOTALED	ODI	JUI L4	+	1210.11	210.10	004.11	000.02	01.01	01.01	3.33	٥.
TRANSPUR	RIOINER											
	Optional Features & Functions:										1	
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chanr	UNC1X	CCOEF		184.85	23.81	1.99	0.77	29.23	3.93		
<b>—</b>	Clear Channel Capability (88ZS/SF) Option - Subsequent - per DS1 Chanr	UNC1X	CCOSF		184.85	23.81	1.99	0.77	29.23	3.93	-	
0VV 4005	SS TEN DIGIT SCREENING	UNCIX	UUUSF	1	104.00	۵.0۱	1.33	0.77	43.43	3.83		
OAA ACCE		0110	+	0.0005		<del>                                     </del>	1		+	+		
<u></u>	8XX Access Ten Digit Screening, Per Ca	OHD		0.0005							<del></del>	
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserva	OHD	N8R1X		7.13	0.97	<u> </u>		27.37	27.37	17.75	17.7
1	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation	OHD		1	15.88	1.97	10.04	0.97	27.37	27.37	17.75	17.
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translatio	OHD	N8FTX		15.88	1.97	10.04	0.97	27.37	27.37	17.75	17.
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb	OHD	N8FCX		5.69	2.85	10.01	0.07	27.37	27.37	17.75	17.
		OHD	INDICA		3.03	2.00			21.31	21.31	17.73	17.
	8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested										1	
	Per 8XX No.	OHD	N8FMX		6.66	3.81			27.37	27.37	17.75	17.
	8XX Access Ten Digit Screening, Change Charge Per Reque	OHD	N8FAX		8.1	0.97			27.37	27.37	17.75	17.
	8XX Access Ten Digit Screening, Call Handling and Destination Feature	OHD	N8FDX		5.69				27.37	27.37	17.75	17.
LINE INFO	RMATION DATA BASE ACCESS (LIDB)								1	1		
LINE IN O	LIDB Common Transport Per Quer	OQT		0.00004						+	+	
									+			
	LIDB Validation Per Quen	OQU								+		
	LIDB Originating Point Code Establishment or Chang			0.0142								
		OQT, OQU	NRPBX	0.0142	64.36				27.37	27.37	17.75	17.
		OQT, OQU	NRPBX	0.0142	64.36				27.37	27.37	17.75	17.
SIGNALING		OQT, OQU			64.36						17.75	
SIGNALING	G (CCS7)				64.36							
SIGNALING	3 (CCS7) CCS7 Signaling Termination, Per STP Por	1DB	NRPBX PT8SX	148.72	64.36				27.37	27.37	17.75	
SIGNALING	G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messagi	1DB 1DB	PT8SX	148.72 0.0001		171.09	125.7	125.7	25.93	25.93	16.31	16.
SIGNALING	G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A link	1DB 1DB 1DB	PT8SX TPP++	148.72 0.0001 18.79	171.98	171.98	135.7	135.7	25.93 25.93	25.93 25.93	16.31	16.
SIGNALING	3 (CCS7) CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin	1DB 1DB 1DB 1DB	PT8SX	148.72 0.0001 18.79 18.79		171.98 171.98	135.7 135.7	135.7 135.7	25.93	25.93	16.31	16.
SIGNALING	G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag	1DB 1DB 1DB 1DB 1DB	PT8SX TPP++ TPP++	148.72 0.0001 18.79 18.79 0.00004	171.98				25.93 25.93 25.93	25.93 25.93 25.93	16.31 16.31 16.31	16. 16.
SIGNALING	G (CCS7)  G (CCS7)  G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT	1DB 1DB 1DB 1DB	PT8SX TPP++	148.72 0.0001 18.79 18.79	171.98				25.93 25.93	25.93 25.93	16.31	16. 16.
SIGNALING	G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag	1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98	171.98			25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93	16.31 16.31 16.31	16. 16.
SIGNALING	G (CCS7)  G (CCS7)  G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT	1DB 1DB 1DB 1DB 1DB	PT8SX TPP++ TPP++	148.72 0.0001 18.79 18.79 0.00004	171.98				25.93 25.93 25.93	25.93 25.93 25.93	16.31 16.31 16.31	16. 16. 16.
SIGNALING	G (CS7)  G (CS7)  CCS7 Signaling Termination, Per STP Por  CCS7 Signaling Usage, Per TCAP Messag  CCS7 Signaling Connection, Per link (A link)  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Usage, Per ISUP Messag  CCS7 Signaling Usage, Per ISUP Messag  CCS7 Signaling Usage Surrogate, per link per LAT  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected	1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98	171.98			25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93	16.31 16.31 16.31	16. 16. 16.
SIGNALING	GCCS7)  GCCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per	1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
SIGNALING	G (CS7)  G (CS7)  CCS7 Signaling Termination, Per STP Por  CCS7 Signaling Usage, Per TCAP Messag  CCS7 Signaling Connection, Per link (A link)  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Usage, Per ISUP Messag  CCS7 Signaling Usage, Per ISUP Messag  CCS7 Signaling Usage Surrogate, per link per LAT  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected	1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98	171.98			25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93	16.31 16.31 16.31	17 16 16 16 16
	G (CCS7)  G (CCS7)  G (CCS7)  G (CCS7)  CCS7 Signaling Termination, Per STP Por  CCS7 Signaling Usage, Per TCAP Messag  CCS7 Signaling Connection, Per link (A linh  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Usage, Per ISUP Messag  CCS7 Signaling Usage Surrogate, per link per LAT  CCS7 Signaling Deint Code, per Originating Point Code Establishment or Change, per STP affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected	1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
	G (CCS7)  G (CCS7)  G (CCS7)  G (CCS7)  CCS7 Signaling Termination, Per STP Por  CCS7 Signaling Usage, Per TCAP Messag  CCS7 Signaling Connection, Per link (A linh  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Usage, Per ISUP Messag  CCS7 Signaling Usage Surrogate, per link per LAT  CCS7 Signaling Deint Code, per Originating Point Code Establishment or Change, per STP affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected	1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
SIGNALING	G (CCS7)  G (CCS7)  G (CCS7)  G (CCS7)  CCS7 Signaling Termination, Per STP Por  CCS7 Signaling Usage, Per TCAP Messag  CCS7 Signaling Connection, Per link (A linh  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Connection, Per link (B link) (also known as D lin  CCS7 Signaling Usage, Per ISUP Messag  CCS7 Signaling Usage Surrogate, per link per LAT  CCS7 Signaling Deint Code, per Originating Point Code Establishment or Change, per STP affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected	1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
E911 SERV	G (CCS7) G (CCS7) G (CCS7) G (CCS7) G (CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected	1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
E911 SERV	G (CCS7) G (CCS7) G (CCS7) G (CCS7) G (CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected	1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
E911 SERV	G CCS7)  G CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) CCS7 Signaling Connection, Per link (B link) CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16 16 16 16
E911 SERV	G (CCS7) G (CCS7) G (CCS7) G (CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  INCE  WAME (CNAM) SERVICE CNAM for DB Owners, Per Quen	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16 16 16 16
E911 SER\	G CCS7)  G CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) CCS7 Signaling Connection, Per link (B link) CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16 16 16
E911 SERV	G (CCS7) G (CCS7) G (CCS7) G (CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  INCE  WAME (CNAM) SERVICE CNAM for DB Owners, Per Quen	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16 16 16
E911 SERV	G (CCS7)  G (CCS7)  G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE  CNAM for DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM for Non DB Owners, Per Quen	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16 16 16
E911 SERV	G (CCS7) G (CCS7) G (CCS7) G (CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  INCE  WAME (CNAM) SERVICE CNAM for DB Owners, Per Quen	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
E911 SERV	G CCS7)  G CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE  CNAM for DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX TPP++ TPP++ STU56 CCAPO CCAPD	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40 8	40			25.93 25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31 16.31	16. 16. 16. 16.
E911 SERV	G (CCS7)  G (CCS7)  G (CCS7)  CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE  CNAM for DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM for Non DB Owners, Per Quen	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX  TPP++ TPP++ STU56  CCAPO	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40	171.98 40			25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31	16. 16. 16.
E911 SERV	G CCS7)  G CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE  CNAM for DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX TPP++ TPP++ STU56 CCAPO CCAPD	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40 8	40			25.93 25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31 16.31	16. 16. 16. 16.
E911 SERV	G CCS7)  G CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE  CNAM for DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX TPP++ TPP++ STU56 CCAPO CCAPD	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40 8	40			25.93 25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31 16.31	16. 16. 16. 16.
E911 SERV	G CCS7)  G CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Affected  WICE  VICE  CNAM for DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM for Non DB Owners, Per Quen CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	PT8SX TPP++ TPP++ STU56 CCAPO CCAPD	148.72 0.0001 18.79 18.79 0.00004 376.12	171.98 171.98 40 8	40			25.93 25.93 25.93 25.93 25.93 25.93	25.93 25.93 25.93 25.93 25.93 25.93 25.93	16.31 16.31 16.31 16.31 16.31	16. 16. 16. 16. 16.

												Т	
	OPERATOR SERVICES AND DIRECTORY ASSISTANCE												
OPERATOR	OR CALL PROCESSING				1.0								
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIC Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC	$\rightarrow$			1.2 1.24						+		
	Oper. Call Processing - Oper. Provided, Per Mill Osing Poreign Lt.  Oper. Call Processing - Fully Automated, per Call - Using BST LIC	-			0.2						+		
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE				0.2								
INWARD OF	PPERATOR SERVICES				4.45								
	Inward Operator Services - Verification, Per Minu Inward Operator Services - Verification and Emergency Interrupt - Per Mini	$\rightarrow$			1.15 1.15						+		
	inward operator ocrytecs - verification and Emergency interrupt - i er with	-			1.10						+		
BRANDING	G - OPERATOR CALL PROCESSING												
	Recording of Custom Branded OA Announcement			CBAOS		7000	7000			19.99	19.99	19.99	19.99
	Loading of Custom Branded OA Announcement per shelf/NAV			CBAOL		500	500			19.99	19.99		
DIRECTOR	RY ASSISTANCE SERVICES										<del></del>	<del> </del>	
DIRECTOR	DIRECTORY ASSISTANCE ACCESS SERVICE	-++									+	<del></del>	
	Directory Assistance Access Service Calls, Charge Per Ca	_			0.275						+		
	DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)												
	Directory Assistance Call Completion Access Service (DACC), Per Call Attern				0.1								
	UNBRANDING	_			+			1			+	++	
	DIRECTORY TRANSPORT	-									+	++	
	Directory Transport - Local Channel DS	+			35.52	503.57	442.84	46.28	32.18	61.99	61.99	29.27	29.27
	Directory Transport - DS1 Level Interoffice Per Mil				0.6923								
	Directory Transport - DS1 Level Interoffice Per Facility Termination				79.69	198.15	148.18	25.44	20.42	27.37	27.37	12.97	12.97
	Switched Common Transport Per DA Access Service Per Ca				0.0003								
	Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Ca				0.00003 0.0023								
	Directory Transport - DA Interconnection Per DA Service Ca				0.0023						+	+	
	Directory Transport - Installation NRC, Per Trunk or Signaling Connection				0.000	260.69	5.95	173.46	5.95	61.99	61.99	29.27	29.27
	DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)												
	Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per moni	$\rightarrow$		DBSOF	0.04								
BEANDING	G - DIRECTORY ASSISTANCE	$\rightarrow$		DBSOF	150						+		
BRANDING	Custom Branding Announcement, per Recording to be used with the provision of Di	A	AMT	CBADA		3000	3000				+	+	
	Loading of Custom Branded Announcement per DRAM Card/Switch		AMT	CBADC		690	690				<u> </u>		
SELECTIVE	E ROUTING												
	Selective Routing Per Unique Line Class Code Per Request Per Swit			USRCR		230.6	230.6			40.71	9.58		
VIDTUAL CO	COLLOCATION	$\longrightarrow$											
VIKTOAL	DELECCATION	-++	ueanl,uea,udn,udc,u								+	<del></del>	
	Virtual Collocation - 2-wire Cross Connects (loop		al,uhl,ucl,uec	UEAC2	0.28	30.76	29.4	12.75	11.38	19.99	19.99	19.99	19.99
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splittin	1	UEPSR, UEPSB	VE1LS	0.28	30.76	29.4	12.75	11.38	19.99	19.99	19.99	19.99
	Virtual Collocation - 2-wire Cross Connects (por			VE1R2	0.28	30.76	29.4	12.75	11.38	19.99	19.99	19.99	19.99
	Virtual Collocation - 4-wire Cross Connects (loop		uea,uhl,ucl,ud	UEAC4	0.56	66.71	50.43	12.82	11.39	19.99	19.99	19.99	19.99
	Virtual Collocation - 4-wire Cross Connects (por Virtual Collocation - 2-Fiber Cross Connect		CLO	VE1R4 CNC2F	0.56 12.1	66.71 55.46	50.43 39.18	12.82 16.83	11.39 13.27	19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Virtual Collocation - 2-1 iber Cross Connect:	-+-+	CLO	CNC4F	21.75	66.71	50.43	21.86	18.31	19.99	19.99	19.99	19.99
	Virtual Collocatin - DS1 Cross Connect:		USL,ULC,CLO	CNC1X	7.5	155	14				10.00		
AIN SELEC	CTIVE CARRIER ROUTING		000								<del></del>		
			SRC	SRCEC		202197.82		17181.39 3.39	3.39	27.37 27.37	27.37 27.37	27.37 27.37	27.37
<b>—</b>	Regional Service Establishment			00000		000.75					27.37	27.37	27.3
	End Office Establishment		SRC	SRCEO	0.0021412	339.75	339.75	3.39	0.00	21.01			
		1		SRCEO	0.0031412	339.75	339.75	3.39	0.00	27.07	+	-	
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE	1	SRC		0.0031412								
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE AIN SMS Access Service - Service Establishment, Per State, Initial Setup	1	SRC	CAMSE	0.0031412	197.49	197.49	114.22	114.22	27.37	27.37	17.75	17.75
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS ACCESS SErvice - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access	1	SRC	CAMSE CAMDP	0.0031412	197.49 64.05	197.49 64.05	114.22 27.04	114.22 27.04	27.37 27.37	27.37	17.75	17.75
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE AIN SMS Access Service - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access	1	SRC	CAMSE CAMDP CAM1P	0.0031412	197.49 64.05 64.05	197.49 64.05 64.05	114.22 27.04 27.04	114.22 27.04 27.04	27.37 27.37 27.37	27.37 27.37	17.75 17.75	17.75 17.75
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code	1	SRC	CAMSE CAMDP CAM1P CAMAU	0.0031412	197.49 64.05 64.05 141.84	197.49 64.05 64.05 141.84	114.22 27.04 27.04 70.05	114.22 27.04 27.04 70.05	27.37 27.37 27.37 27.37	27.37 27.37 27.37	17.75 17.75 17.75	17.75 17.75 17.75
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - Port Connection - Gent Per User ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement	1	SRC	CAMSE CAMDP CAM1P		197.49 64.05 64.05	197.49 64.05 64.05	114.22 27.04 27.04	114.22 27.04 27.04	27.37 27.37 27.37	27.37 27.37	17.75 17.75	17.75 17.75
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service Establishment, Per State, Initial Setup  AIN SMS Access Service - Port Connection - Dial/Shared Access  AIN SMS Access Service - Port Connection - ISDN Access  AIN SMS Access Service - Port Identification Codes - Per User ID Code  AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement  AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)		SRC	CAMSE CAMDP CAM1P CAMAU	0.0026	197.49 64.05 64.05 141.84	197.49 64.05 64.05 141.84	114.22 27.04 27.04 70.05	114.22 27.04 27.04 70.05	27.37 27.37 27.37 27.37	27.37 27.37 27.37	17.75 17.75 17.75	17.75 17.75 17.75
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - Very Identification Codes - Per User ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Service - For Unit (100 Kilobytes) AIN SMS Access Service - Session, Per Minute		SRC	CAMSE CAMDP CAM1P CAMAU	0.0026 0.0892	197.49 64.05 64.05 141.84	197.49 64.05 64.05 141.84	114.22 27.04 27.04 70.05	114.22 27.04 27.04 70.05	27.37 27.37 27.37 27.37	27.37 27.37 27.37	17.75 17.75 17.75	17.75 17.75 17.75
AIN - BELLS	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service Establishment, Per State, Initial Setup  AIN SMS Access Service - Port Connection - Dial/Shared Access  AIN SMS Access Service - Port Connection - ISDN Access  AIN SMS Access Service - Port Identification Codes - Per User ID Code  AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement  AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)		SRC	CAMSE CAMDP CAM1P CAMAU	0.0026	197.49 64.05 64.05 141.84	197.49 64.05 64.05 141.84	114.22 27.04 27.04 70.05	114.22 27.04 27.04 70.05	27.37 27.37 27.37 27.37	27.37 27.37 27.37	17.75 17.75 17.75	17.75 17.75 17.75
	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS ACCESS SErvice - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code AIN SMS Access Service - User Identification Codes - Per User ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Storage, Per Unit (100 Kilobytes) AIN SMS Access Service - Session, Per Minute  AIN SMS Access Service - Company Performed Session, Per Minute  LSOUTH AIN TOOLKIT SERVICE		SRC	CAMSE CAMDP CAM1P CAMAU CAMRC	0.0026 0.0892	197.49 64.05 64.05 141.84 142.13	197.49 64.05 64.05 141.84 142.13	114.22 27.04 27.04 70.05 35.26	114.22 27.04 27.04 70.05 35.26	27.37 27.37 27.37 27.37 27.37 27.37	27.37 27.37 27.37 27.37 27.37	17.75 17.75 17.75 17.75 17.75	17.75 17.75 17.75 17.75
	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Session, Per Minute AIN SMS Access Service - Company Performed Session, Per Minute LSOUTH AIN TOOLKIT SERVICE AIN TOOLKIT SERVICE  AIN TOOLKIT SERVICE		SRC	CAMSE CAMDP CAM1P CAMAU CAMRC	0.0026 0.0892	197.49 64.05 64.05 141.84 142.13	197.49 64.05 64.05 141.84 142.13	114.22 27.04 27.04 70.05	114.22 27.04 27.04 70.05	27.37 27.37 27.37 27.37 27.37 27.37	27.37 27.37 27.37 27.37 27.37	17.75 17.75 17.75 17.75 17.75	17.75 17.75 17.75 17.75 17.75
	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS ACCESS SERVICE - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code AIN SMS Access Service - User Identification Codes - Per User ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Service, Per Unit (100 Kilobytes) AIN SMS Access Service - Company Performed Session, Per Minute AIN SMS Access Service - Company Performed Session, Per Minute  LSOUTH AIN TOOLKIT SERVICE  AIN TOOIKIT SERVICE AIN TOOIKIT SERVICE - Training Session, Per Customer		SRC	CAMSE CAMDP CAM1P CAMAU CAMRC  BAPSC BAPVX	0.0026 0.0892	197.49 64.05 64.05 141.84 142.13	197.49 64.05 64.05 141.84 142.13	114.22 27.04 27.04 70.05 35.26	114.22 27.04 27.04 70.05 35.26	27.37 27.37 27.37 27.37 27.37 27.37 27.37	27.37 27.37 27.37 27.37 27.37	17.75 17.75 17.75 17.75 17.75	17.75 17.75 17.75 17.75 17.75
	End Office Establishment Query NRC, per query  LSOUTH AIN SMS ACCESS SERVICE  AIN SMS Access Service - Service Establishment, Per State, Initial Setup AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement AIN SMS Access Service - Session, Per Minute AIN SMS Access Service - Company Performed Session, Per Minute LSOUTH AIN TOOLKIT SERVICE AIN TOOLKIT SERVICE  AIN TOOLKIT SERVICE	1	SRC	CAMSE CAMDP CAM1P CAMAU CAMRC	0.0026 0.0892	197.49 64.05 64.05 141.84 142.13	197.49 64.05 64.05 141.84 142.13	114.22 27.04 27.04 70.05 35.26	114.22 27.04 27.04 70.05 35.26	27.37 27.37 27.37 27.37 27.37 27.37	27.37 27.37 27.37 27.37 27.37	17.75 17.75 17.75 17.75 17.75	17.75 17.75 17.75 17.75 17.75

Page 7 of 20

	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate			BAPTM		49.64	49.64	27.04	27.04		27.37	27.37	17.75	17.75
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP			BAPTO		117.98	117.98	37.9	37.9		27.37	27.37	17.75	17.75
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP			BAPTC		117.98	117.98	37.9	37.9		27.37	27.37	17.75	17.75
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code			BAPTF		117.98	117.98	37.9	37.9		27.37	27.37	17.75	17.75
	AIN Toolkit Service - Query Charge, Per Query				0.024									
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per													
	Query				0.006									
	ANIT HE OF CORD OF THE OWN A THE CORD OF THE OWN A				4.00									
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes			BAPMS	1.63									
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			BAPIS	16	44.56 47.74	44.56 47.74	31.84	31.84		27.37	27.37	17.75	17.75
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription  AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			BAPLS	0.1 15.9	47.74	47.74	15.9 31.84	15.9 31.84		27.37 27.37	27.37 27.37	17.75 17.75	17.75 17.75
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription  AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			BAPES	0.003	47.74	44.56	31.84	31.84		27.37	27.37	17.75	17.75
	Aliv Toolkit Service - Call Event Special Study - Fel Aliv Toolkit Service Subscription			DAFES	0.003	47.74	47.74				21.31	21.31	17.75	17.75
ODLIE/EDOL	JF/ADUF/CMDS													
ODOI7EDOC	T/ADDI/ONIDO													
	ACCESS DAILY USAGE FILE (ADUF)													
	ADUF: Message Processing, per messag				0.004									
	ADUF: Data Transmission (CONNECT:DIRECT), per messag				0.001									
	ENHANCED OPTIONAL DAILY USAGE FILE (EODUF)													
	EODUF: Message Processing, per messag				0.004									
	OPTIONAL DAILY USAGE FILE (ODUF)													
	ODUF: Recording, per message	-			0.0002		1	1	1			1	-	
	ODUF: Recording, per messagi ODUF: Message Processing, per messag			1	0.0002		<b>-</b>		1					
	ODUF: Message Processing, per Magnetic Tape provisions				55.19									
	ODUF: Data Transmission (CONNECT:DIRECT), per messag				0.00004									
ENHANCED	EXTENDED LINK (EELs)													
	NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mian			e, TN; New Orlean	s, LA;									
	NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be													
	NOTE: In all states, EEL network elements shown below also apply to currently combined facilities w				rge applies to cu	rrently combine	d facilities conv	erted to UNEs	.(Non-recurri	ng rates do not apply.)				
	NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the	GA PS	C order.(No Switch As Is C	harge.)										
		ļ.,												
	2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE	L)												
-	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone		1 UNCVX	UEAL2	17.95									
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zon		2 UNCVX	UEAL2	29.16									
	Filst 2-wire vo Grade Loop(SL2) in a DST interofficed Transport Combination - 2011		2 UNCVA	UEALZ	29.10									
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zon		3 UNCVX	UEAL2	52.84									
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor		UNC1X	1L5XX	0.2067									
							1							
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo		UNC1X	U1TF1	68.75									
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo DS1 Channelization System Per Mont		UNC1X UNC1X	MQ1	122.5									
	DS1 Channelization System Per Monti Voice Grade COCI - DS1 To Ds0 Interface - Per Monti													
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat		UNC1X UNCVX	MQ1 1D1VG	122.5 0.64									
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat  - Zone 1		UNC1X	MQ1	122.5									
	DS1 Channelization System Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Mont! Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat		UNC1X UNCVX	MQ1 1D1VG UEAL2	122.5 0.64 17.95									
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat  - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat  - Zone 2		UNC1X UNCVX	MQ1 1D1VG	122.5 0.64									
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat  - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat  - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination		UNC1X UNCVX  1 UNCVX  2 UNCVX	MQ1 1D1VG UEAL2 UEAL2	122.5 0.64 17.95 29.16									
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati Zone 3		1 UNCVX 2 UNCVX 3 UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2	122.5 0.64 17.95 29.16 52.84									
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor		1 UNCVX 2 UNCVX 3 UNCVX UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2 1D1VG	122.5 0.64 17.95 29.16	11 18	11 18	13.96	13.96		21 31	21 21	3 03	34
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati Zone 3		1 UNCVX 2 UNCVX 3 UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2	122.5 0.64 17.95 29.16 52.84	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-is Chan		1 UNCVX 2 UNCVX 3 UNCVX UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2 1D1VG	122.5 0.64 17.95 29.16 52.84	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.6
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor	L)	1 UNCVX 2 UNCVX 3 UNCVX UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2 1D1VG	122.5 0.64 17.95 29.16 52.84	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE	L)	1 UNCVX 2 UNCVX 3 UNCVX UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2 1D1VG	122.5 0.64 17.95 29.16 52.84	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE	L)	UNC1X UNCVX  1 UNCVX  2 UNCVX  3 UNCVX UNCVX UNC1X  1 UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 1D1VG UNCCC	122.5 0.64 17.95 29.16 52.84 0.64	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE 15TS1 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zou 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zou 2	L) he	UNC1X UNCVX 1 UNCVX 2 UNCVX 3 UNCVX UNCVX UNC1X	MQ1 1D1VG  UEAL2  UEAL2  UEAL2  1D1VG  UNCCC	122.5 0.64 17.95 29.16 52.84 0.64	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1	L) he	1 UNCYX 2 UNCVX 3 UNCVX UNCYX 1 UNCVX 1 UNCYX 1 UNCYX UNCYX UNCYX 2 UNCVX	MQ1 1D1VG UEAL2 UEAL2 UEAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL4 UEAL4	122.5 0.64 17.95 29.16 52.84 0.64	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinat - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoi 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoi 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoi 3	L) he	1 UNCVX 2 UNCVX 3 UNCVX UNCVX 1 UNCVX 1 UNCVX 2 UNCVX 2 UNCVX 2 UNCVX 3 UNCVX	MQ1 1D1VG UEAL2 UEAL2 1D1VG UNCCC UEAL4 UEAL4 UEAL4 UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COC1 - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zor 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zor 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	L) he	1 UNCVX 2 UNCVX 3 UNCVX UNCVX 1 UNCVX 1 UNCVX 2 UNCVX UNCYX UNCYX UNCYX 1 UNCVX 2 UNCVX 3 UNCVX UNCYX	MQ1 1D1VG  UEAL2  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  UEAL4  1L5XX	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zor 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zor 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor	L) he	1 UNCYX 2 UNCVX 3 UNCVX UNCVX 4 UNCVX 1 UNCVX 2 UNCVX UNCYX UNCYX 2 UNCVX 3 UNCVX UNCYX UNCYX UNCYX UNCYX UNCYX	MQ1 1D1VG UEAL2 UEAL2 1D1VG UNCCC UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.87 0.2067 68.75	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.:
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinationel 3-Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoiden 1	L) he	1 UNCVX 2 UNCVX 3 UNCVX UNCVX 1 UNCVX 4 UNCVX 2 UNCVX 2 UNCVX 4 UNCVX 2 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 4 UNCVX 4 UNCVX 5 UNCVX 6 UNCVX 6 UNCVX	MQ1 1D1VG  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 68.75 122.5	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COC1 - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational Zone 3  Voice Grade COC1 - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid Interoffice Transport Combination - Zoid Interoffice Transport - Dedicated - DS1 - Combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor Channelization - Channel System Combination - Per Mor Voice Grade COC1 - DS1 to DS0 Channel System combination - per mor	L) he	1 UNCYX 2 UNCVX 3 UNCVX UNCVX 4 UNCVX 1 UNCVX 2 UNCVX UNCYX UNCYX 2 UNCVX 3 UNCVX UNCYX UNCYX UNCYX UNCYX UNCYX	MQ1 1D1VG UEAL2 UEAL2 1D1VG UNCCC UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.87 0.2067 68.75	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 1  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination - Per Mor Noice Grade COCI - DS1 to DS0 Channel System combination - Per mor Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	L) he	1 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 5 UNCVX 2 UNCVX 2 UNCVX 4 UNCVX 4 UNCVX 5 UNCVX 6 UNCVX 7 UNCVX 7 UNCVX 7 UNCVX 7 UNCVX 8 UNCVX 8 UNCVX	MQ1 1D1VG  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  1L5XX  U1TF1  MQ1 1D1VG	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 68.75 122.5 0.64	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.3
	DS1 Channelization System Per Mont  Voice Grade COC1 - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COC1 - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Channel System - Additional - Zone	L) he	1 UNCVX 2 UNCVX 3 UNCVX UNCVX 1 UNCVX 4 UNCVX 2 UNCVX 2 UNCVX 4 UNCVX 2 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 4 UNCVX 4 UNCVX 5 UNCVX 6 UNCVX 6 UNCVX	MQ1 1D1VG  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 68.75 122.5	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational - Zone 1  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational - Zone 2  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zon 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zon 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination - Per Mor Noice Grade COCI - DS1 to DS0 Channel System combination - Per mor Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	L) he	1 UNCYX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 2 UNCVX 3 UNCVX 4 UNCYX 2 UNCYX 2 UNCYX 3 UNCYX 4 UNCYX 4 UNCYX 4 UNCYX 5 UNCYX 6 UNCYX 7 UNCYX 7 UNCYX 8 UNCYX 8 UNCYX 9 UNCYX 9 UNCYX 1 UNCYX	MQ1 1D1VG  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  1L5XX  U1TF1  MQ1 1D1VG	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 68.75 122.5 0.64	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinat - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zon 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zon 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in Same DS1 Interoffice Transport Combination - Zone Additional 4-Wire Analog Voice Grade Loop in	L) he	1 UNCYX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 2 UNCVX 3 UNCVX 4 UNCYX 2 UNCYX 2 UNCYX 3 UNCYX 4 UNCYX 4 UNCYX 4 UNCYX 5 UNCYX 6 UNCYX 7 UNCYX 7 UNCYX 8 UNCYX 8 UNCYX 9 UNCYX 9 UNCYX 1 UNCYX	MQ1 1D1VG  UEAL2  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 68.75 122.5 0.64	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 1  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Chare  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 1  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination - Per Mor Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone	L) he	1 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 2 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 1 UNCVX 2 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 1 UNCVX 2 UNCVX 3 UNCVX	MQ1 1D1VG  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  1L5XX  U1TF1  MQ1 1D1VG  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 0.2067 122.5 0.64 24.01 39 70.67	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COC1 - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 3  Voice Grade COC1 - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Charanous Combinational Combinational Per More Combinational Per M	L) he	UNC1X	MQ1 1D1VG  UEAL2  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  1L5XX  U1TF1  MQ1 1D1VG  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 68.75 122.5 0.64 24.01									
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 1  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch -As-Is Chare  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zoid 1  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination - Per Mor Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone / Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone	L) he	1 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 2 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 1 UNCVX 2 UNCVX 2 UNCVX 3 UNCVX 4 UNCVX 1 UNCVX 1 UNCVX 2 UNCVX 3 UNCVX	MQ1 1D1VG  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  UEAL4  1L5XX  U1TF1  MQ1 1D1VG  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 0.2067 122.5 0.64 24.01 39 70.67	11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Channelization System Per Mont  Voice Grade COC1 - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COC1 - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Chan  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zon 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zon 3  Interoffice Transport - Dedicated - DS1 - Sacility Termination Per Mor Interoffice Transport Dedicated - DS1 - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination - Per Mon Voice Grade COC1 - DS1 to DS0 Channel System combination - Per mor Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone :  Voice Grade COC1 - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Chan	L) he he	UNC1X	MQ1 1D1VG  UEAL2  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  1L5XX  U1TF1  MQ1 1D1VG  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 0.2067 122.5 0.64 24.01 39 70.67									
	DS1 Channelization System Per Mont  Voice Grade COC1 - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combinational 2-Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 2-Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinational 3  Voice Grade COC1 - DS1 to DS0 Channel System combination - per mor Nonrecurring Currently Combined Network Elements Switch - As-Is Charanous Combinational Combinational Per More Combinational Per M	L) he he he	UNC1X	MQ1 1D1VG  UEAL2  UEAL2  UEAL2  1D1VG  UNCCC  UEAL4  UEAL4  1L5XX  U1TF1  MQ1 1D1VG  UEAL4	122.5 0.64 17.95 29.16 52.84 0.64 24.01 39 70.67 0.2067 0.2067 122.5 0.64 24.01 39 70.67									

First 4-wire 56Kbps Digital Grade Loop in a DS	1 Intereffice Transport Combination - Zone	1 1					I					1	
2	Timeronice Transport Combination - Zone	2	UNCDX	UDL56	44.4								
First 4-Wire 56Kbps Digital Grade Loop in a DS	61 Interoffice Transport Combination - Zone												
3 Interoffice Transport - Dedicated - DS1 combin	ation Doublin Doublin	3	UNCDX	UDL56	80.45								
interoffice Transport - Dedicated - DST combin	ation - Per Mile Per Mor		UNC1X	1L5XX	0.2067								
Interoffice Transport - Dedicated - DS1 - comb	ination Facility Termination Per Mo		UNC1X	U1TF1	68.75					31.31	31.31	3.93	3.
Channelization - Channel System DS1 to DS0	combination Per Mon		UNC1X	MQ1	122.5								
OCU-DP COCI (data) - DS1 to DS0 Channel S			UNCDX	1D1DD	1.36								
Additional 4-Wire 56Kbps Digital Grade Loopin Combination - Zone (	same DS1 Interoffice Transport	1	UNCDX	UDL56	27.33					31.31	31.31	3.93	3.
Additional 4-Wire 56Kbps Digital Grade Loopin	same DS1 Interoffice Transport	- '	UNCDX	UDLS6	21.33					31.31	31.31	3.93	3.
Combination - Zone 2	Same DOT Interentee Transport	2	UNCDX	UDL56	44.4					31.31	31.31	3.93	3.
Additional 4-Wire 56Kbps Digital Grade Loopin	same DS1 Interoffice Transport												
Combination - Zone (		3	UNCDX	UDL56	80.45					31.31	31.31	3.93	3.
OCU-DP COCI (data) - DS1 to DS0 Channel S			UNCDX	1D1DD	1.36								
Nonrecurring Currently Combined Network Ele			UNC1X	UNCCC	1.36	11.18	11.18	13.96	13.96	31.31	31.31	3.93	3.
Nomecuning currently combined Network Lie	ments Switch -As-is Chair		ONCIX	UNCCC		11.10	11.10	13.90	13.90	31.31	31.31	3.93	٥.
4-WIRE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDIC	ATED DS1 INTEROFFICE TRANSPORT (EEL)												
First 4-Wire 64Kbps Digital Grade Loop in a DS	61 Interoffice Transport Combination - Zone												
1		1	UNCDX	UDL64	27.33								
First 4-Wire 64Kbps Digital Grade Loop in a DS	51 Interoffice Transport Combination - Zone		LINORY	LIDLOA	44.4								
First 4-Wire 64Kbps Digital Grade Loop in a DS	S1 Interoffice Transport Combination - Zono	2	UNCDX	UDL64	44.4								
3		3	UNCDX	UDL64	80.45								
Interoffice Transport - Dedicated - DS1 combin	ation - Per Mile Per Mor		UNC1X	1L5XX	0.2067								
		1 7											
Interoffice Transport - Dedicated - DS1 combin	nation - Facility Termination Per Mo		UNC1X	U1TF1	68.75								
Channelization - Channel System DS1 to DS0	combination Per Mon		UNC1X	MQ1	122.5								
OCU-DP COCI (data) - DS1 to DS0 Channel S	vstem combination - per month (2.4-64kt		UNCDX	1D1DD	1.36	0	0						
Additional 4-Wire 64Kbps Digital Grade Loopin			CHODA	15155	1.00	Ü	Ů						
Combination - Zone 1	•	1	UNCDX	UDL64	27.33								
Additional 4-Wire 64Kbps Digital Grade Loopin	same DS1 Interoffice Transport	2											
Combination - Zone   Additional 4-Wire 64Kbps Digital Grade Loopin	name DC1 Intereffice Transport	2	UNCDX	UDL64	44.4								
Combination - Zone (	Same D31 interoffice Transport	3	UNCDX	UDL64	80.45								
Official and Edited			ONODA	ODLOT	00.40								
OCU-DP COCI (data) - DS1 to DS0 Channel S	ystem combination - per month (2.4-64kt		UNCDX	1D1DD	1.36								
Nonrecurring Currently Combined Network Ele	ments Switch -As-Is Char		UNC1X	UNCCC		11.18	11.18	13.96	13.96	31.31	31.31	3.93	3.
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATE	D DC4 INTEROFFICE TRANSPORT (EEL)												
4-Wire DS1 Digital Loop in Combination with D		1	UNC1X	USLXX	51.74								
4-Wire DS1 Digital Loop in Combination with D		2	UNC1X	USLXX	84.05								
4-Wire DS1 Digital Loop in Combination with D	S1 Interoffice Transport - Zone	3	UNC1X	USLXX	152.29								
Interoffice Transport - Dedicated - DS1 combin	ation - Per Mile Per Mor		UNC1X	1L5XX	0.2067								
Interesting Transport Destinated DC4 combin			LINGAY	HATEA	CO 75								
Interoffice Transport - Dedicated - DS1 combin Nonrecurring Currently Combined Network Ele			UNC1X UNC1X	U1TF1 UNCCC	68.75	11.18	11.18	13.96	13.96	31.31	31.31	3.93	3.
Nonrecaring currently combined Network Ele	ments owiten 7/3 is onar		ONOTA	011000		11.10	11.10	10.50	10.50	01.01	01.01	0.00	0.
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATE	D DS3 INTEROFFICE TRANSPORT (EEL)												
First DS1Loop in DS3 Interoffice Transport Cor		1	UNC1X	USLXX	51.74								
First DS1Loop in DS3 Interoffice Transport Cor		2	UNC1X	USLXX	84.05								
First DS1Loop in DS3 Interoffice Transport Cor Interoffice Transport - Dedicated - DS3 combin		3	UNC1X UNC3X	USLXX 1L5XX	152.29 4.67				-				
Interoffice Transport - Dedicated - DS3 combin		+ +	UNC3X	U1TF3	804.02						1		
DS3 to DS1 Channel System combination per	mont		UNC3X	MQ3	201.37								
DS3 Interface Unit (DS1 COCI) combination pe	r montl		UNC1X	UC1D1	15.39								
Additional DS1Loop in DS3 Interoffice Transpo		1	UNC1X	USLXX	51.74								
Additional DS1Loop in DS3 Interoffice Transpo Additional DS1Loop in DS3 Interoffice Transpo		3	UNC1X UNC1X	USLXX	84.05 152.29						1		
DS3 Interface Unit (DS1 COCI) combination pe		3	UNC1X	UC1D1	15.39						1		
Nonrecurring Currently Combined Network Ele			UNC3X	UNCCC	. 5.00	11.18	11.18	13.96	13.96	31.31	31.31	3.93	3.
, ,													
2-WIRE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE		$\perp$											
2-WireVG Loop used with 2-wire VG Interoffic 2-WireVG Loop used with 2-wire VG Interoffic		1 2	UNCVX	UEAL2 UEAL2	17.95 29.16					_			
2-WireVG Loop used with 2-wire VG Interoffice		3	UNCVX	UEAL2	29.16 52.84								
Interoffice Transport - Dedicated - 2-wire VG co		- 3	UNCVX	1L5XX	0.0101						1		
Interoffice Transport - Dedicated - 2- Wire Voice													
per month	•		UNCVX	U1TV2	24.15					31.31	31.31	3.93	3
Nonrecurring Currently Combined Network Ele	ments Switch -As-Is Char	$\perp$	UNCVX	UNCCC		11.18	11.18	13.96	13.96	31.31	31.31	3.93	3
4-WIRE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE	CDADE INTEROFFICE TRANSPORT (FFL)	+								_			
4-Wire VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE 4-WireVG Loop used with 4-wire VG Interoffice		1	UNCVX	UEAL4	24.01								
4-WireVG Loop used with 4-wire VG Interoffice	Transport Combination - Zon	2	UNCVX	UEAL4	39								
4-WireVG Loop used with 4-wire VG Interoffice	Transport Combination - Zon	3	UNCVX	UEAL4	70.67								
Interoffice Transport - Dedicated - 4-wire VG co	ombination - Per Mile Per Moi		UNCVX	1L5XX	0.0101								

	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination		UNCVX	U1TV4	21.41									
ĺ	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCVX	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	
DS3 DIGITAL	EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)													
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per moi		UNC3X	1L5ND	10.16							1		
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per mo Interoffice Transport - Dedicated - DS3 - Per Mile per mon		UNC3X UNC3X	UE3PX 1L5XX	374.52 4.67							+		
<u> </u>	interoffice Transport - Dedicated - D33 - Fer Wille per Hioff		UNCSX	ILJAX	4.07									
ı	Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mo		UNC3X	U1TF3	804.02									
1	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC3X	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	
STS1 DIGITA	L EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)											+		
	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per moi		UNCSX	1L5ND	10.16									
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per mo Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor		UNCSX	UDLS1 1L5XX	387.67 4.67						+			
ľ	interomee transport bedieated of or combination it et while per mor		ONOOX	TEOXIX	4.07									
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mo		UNCSX	U1TFS	801.57									
1	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCSX	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	
2-WIRE ISDN	EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)											+		
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	1	UNCNX	U1L2X	23.23									
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	2	UNCNX	U1L2X	37.74									
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	3	UNCNX	U1L2X	68.38						+			
	Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo		UNC1X UNC1X	1L5XX U1TF1	0.2067 68.75					_	+	+		
	Channelization - Channel System DS1 to DS0 combination - per mor		UNC1X	MQ1	122.5							+		
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon		UNCNX	UC1CA	2.92									
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	1	UNCNX	U1L2X	23.23							+		
ļ	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	2	UNCNX	U1L2X	37.74									
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	3	UNCNX	U1L2X	68.38									
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per mon	3	UNCNX	UC1CA	2.92									
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC1X	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	
	DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	51.74							+		
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	84.05							+		
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	152.29									
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor		UNCSX	1L5XX	4.67									
	Interoffice Transport - Dedicated - STS1 combination - Facility Terminati		UNCSX	U1TFS	801.57									
1	STS1 to DS1 Channel System conbination per mont DS3 Interface Unit (DS1 COCI) combination per montl		UNCSX UNC1X	MQ3 UC1D1	201.37 15.39									
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	51.74							+		
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	84.05									
Į.	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	152.29									
	DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	15.39									
l l	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCSX	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	
4-WIRE 56 KE	BPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)													
4	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zon	1	UNCDX	UDL56	27.33									
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zon	2	UNCDX	UDL56	44.4									
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zon	3	UNCDX	UDL56	80.45						+	+		
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per N  Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat		UNCDX	1L5XX U1TD5	0.0101 17.28						+	+		
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCDX	UNCCC	17.20	11.18	11.18	13.96	13.96		31.31	31.31	3.93	
	BPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)		I II I I I I I I	LIPLO	07.00						1	<del>                                     </del>		
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon	1 2	UNCDX	UDL64 UDL64	27.33 44.4						+	+		
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon	3	UNCDX	UDL64	80.45						+	+		
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per N	J	UNCDX	1L5XX	0.0101							1		
I	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminat		UNCDX	U1TD6	17.28									
1	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCDX	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	
L NETWORK E	ELEMENTS													
When used a	is a part of a currently combined facility, the non-recurring charges do not apply, but a	Switch As I	s charge does apply	,								1		
	is a part of a currently combined facility, the non-recurring charges do not apply, but a is ordinarilty combined network elements in Georgia, the non-recurring charges apply a													
	l l													
Node (Synchi	roNet) Node per month		UNCDX	UNCNT	15.77									

	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion													
	Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge		UNCVX	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charq		UNC1X	UNCCC		11.18	11.18	13.96	13.96		31.31	31.31	3.93	3.9
	DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charger STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion		UNC3X	UNCCC		11.18	11.18	13.96	13.96	;	31.31	31.31	3.93	3.9
	Charge		UNCSX	UNCCC		11.18	11.18	13.96	13.96	:	31.31	31.31	3.93	3.9
	NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS	S3 and al	bove=four months											
SED ATION	AL SUPPORT SYSTEMS													
	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st	ate speci	ific electronic service order	ring charges as or	rdered by the Sta	ate Commissions								
	NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit	is the Be	IlSouth regional electronic	service ordering	charge									
	NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LS	electron	ic service ordering charge:	s, or CLEC-1 may	y elect the region	al electronic ser	vice ordering ch	arge.						
	NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LS	Dasis												
	Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces													
	(Regional)			SOMEC	+	3.5								
	The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to G	eograph	ically Deaveraged UNE Zo	nes. To view Ge	ographically Dea	veraged UNE Z	one Designation	s by Central	Office, refer to	Internet Website:		1	1	
	http://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm													
JNDLED	LOCAL EXCHANGE SWITCHING(PORTS)													
	Exchange Ports													
	Exchange Ports NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will	need to	be ordered using retail U	SOCs										
	2-WIRE VOICE GRADE LINE PORT RATES (RES)													
	Exchange Ports - 2-Wire Analog Line Port- Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re		UEPSR UEPSR	UEPRL UEPRC	2.07	21.93 21.93	21.93 21.93	6.21 6.21	6.21 6.21		27.37 27.37	12.97 12.97	17.77 17.77	1
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re		UEPSR	UEPRO	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	Exchange Ports - 2-Wire VG unbundled AL extended local dialing parity Port with Callet	r ID	OLFOR	OLFRO	2.07	21.93	21.93	0.21	0.21		21.31	12.57	17.77	
	- Res.		UEPSR	UEPAR	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LL Subsequent Activity		UEPSR UEPSR	UEPAP	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	EATURES		OLFOR	USASC	U	0	U							
	All Available Vertical Feature		UEPSR	UEPVF	5.55	0	0				27.37	12.97	17.77	1.
	2-WIRE VOICE GRADE LINE PORT RATES (BUS)													
	Exchange Ports - 2-Wire Analog Line Port without Caller ID - B		UEPSB	UEPBL	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484	ŀ	UEPSB	UEPBC	2.07	21.93	24.02	6.21	6.21		27.37	12.97	17 77	
	- Bus.  Exchange Ports - 2-Wire Analog Line Port outgoing only - Bu		UEPSB	UEPBO	2.07	21.93	21.93 21.93	6.21	6.21		27.37 27.37	12.97	17.77	1
	Exchange Ports - 2-Wire VG unbundled AL extended local dialing parity Port with Called	r ID												
	- Bus.  Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B		UEPSB UEPSB	UEPAW UEPB1	2.07	21.93 21.93	21.93 21.93	6.21 6.21	6.21 6.21		27.37 27.37	12.97 12.97	17.77 17.77	1
	Subsequent Activity		UEPSB	USASC	2.07	21.95	21.93	0.21	0.21		21.01	12.57	17.77	
	EATURES													
	All Available Vertical Feature		UEPSB	UEPVF	5.55	(	0				27.37	12.97	17.77	1
-	EXCHANGE PORT RATES (DID & PBX)  Exchange Ports - 2-Wire DID Port		UEPEX	UEPP2	9.2	238.61	37.48	119.79			19.99	19.99	19.99	19
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabili		UEPDD	UEPDD	68.67	404.04	191.38	145.18	4.92		19.99	19.99	19.99	19
	Exchange Ports - 2-Wire ISDN Port (See Notes below		UEPTX UEPSX	U1PMA	11.19	145.54	105.97	95.57	21.47		19.99	19.99	19.99	19
	All Features Offered		UEPTX UEPSX	UEPVF	5.55	0	0	CDN						
	NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/Ne	w Rusine	ess Request Process Rate	es for the nacket	canabilities will b	e determined via	the Bona Fide	Request/New	Rusiness Re	quest Process				
	Exchange Ports - 2-Wire ISDN Port Channel Profiles	- Dusine	UEPTX UEPSX	U1UMA	0	0	0	requestries	Business ite	quest i rocess.				
	Exchange Ports - 4-Wire ISDN DS1 Por		UEPEX	UEPEX	96.37	407.62	203.11	158.35	40.11		54.75	54.75	11.53	11
	2-Wire VG Unbundled 2-Way PBX Trunk - Re:		UEPSE	UEPRD	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu	-	UEPSP UEPSP	UEPPC UEPPO	2.07	21.93 21.93	21.93 21.93	6.21 6.21	6.21 6.21		27.37 27.37	12.97 12.97	17.77 17.77	1
	2-Wire VG Line Side Unbundled Onward PBX Trunk - Bu	l	UEPSP	UEPP1	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	2-Wire Analog Long Distance Terminal PBX Trunk - Bu		UEPSP	UEPLD	2.07	21.93	21.93	6.21	6.21	:	27.37	12.97	17.77	1
I	2-Wire Voice Unbundled 2-Way PBX Alabama Calling Pc		UEPSP	UEPA2	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Vice Unbundled 2-Way PBX Usage Po	-	UEPSP UEPSP	UEPLD UEPXA	2.07	21.93 21.93	21.93 21.93	6.21 6.21	6.21 6.21		27.37 27.37	12.97 12.97	17.77 17.77	1
	2-Wire Vice Unbundled 2-Way PBX Usage P0  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por	1	UEPSP	UEPXA	2.07	21.93	21.93	6.21	6.21		27.37 27.37	12.97	17.77	1
	2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPSP	UEPXC	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPSP	UEPXD	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	
-+	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt	-	UEPSP	UEPXE	2.07	21.93	21.93	6.21	6.21	<del>                                     </del>	27.37	12.97	17.77	1
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P		UEPSP	UEPXL	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	
$\neg$	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPSP	UEPXM	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling P		UEPSP	UEPXO	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1
	, , , , , , , , , , , , , , , , , , , ,				-									

	OW: V: III II II II II II II II II II II II I	LIEDOD	LIEDVO	0.07	04.00	04.00	0.04	0.04			10.07	47.77	
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc Subsequent Activity	UEPSP UEPSP	UEPXS USASC	2.07	21.93	21.93	6.21	6.21		27.37	12.97	17.77	1.44
FEATURES	oubboddon rounty	02, 0	00,100	- U						-	+		
	All Available Vertical Feature	UEPSP UEPS	E UEPVF	5.55	0	0				27.37	12.97	17.77	1.44
EXCHANGE	PORT RATES (COIN)												<b></b>
	Exchange Ports - Coin Por			2.34	21.93	21.93	5.21	5.21		25.93	12.97	16.33	0.48
NOTE: Tran	smission/usage charges associated with POTS circuit switched usage will also apply to circuit s	witched voice and/or circ	uit switched data tra	nemission by R-	Channels associa	ated with 2-wire	ISDN ports			+	+	<del>                                     </del>	
NOTE: Acce	ess to B Channel or D Channel Packet capabilities will be available only through BFR/New Busin	ness Request Process. I	Rates for the packet	capabilities will I	be determined via	a the Bona Fide	Request/Nev	Business Request Proce	SS.	+			
							1			-	+		
UNBUNDLED LOCAL SW	ITCHING, PORT USAGE												
													<b></b>
	Switching (Port Usage) End Office Switching Function, Per MOI			0.0018									<b>——</b>
	End Office Trunk Port - Shared, Per MOL			0.0018						+	+	+	
										-	+		
Tandem Swi	tching (Port Usage) (Local or Access Tandem)												
	Tandem Switching Function Per MOI			0.00063									
Common Tra	anenort									+		-	
Common Tra	Common Transport - Per Mile, Per MOI			0.00001						+	+		
	Common Transport - Facilities Termination Per MO			0.00045						-	+		
UNBUNDLED PORT/LOO	P COMBINATIONS - COST BASED RATES		-							+	+		1
Cost Recod I	Rates are applied where BellSouth is required by FCC and/or State Commission rule to provide	Unbundled Local Switch	ing or Switch Ports							+	+	-	
Features sha	all apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same manner.	er as they are applied to	the Stand-Alone Un	bundled Port se	ction of this Rate	Exhibit.				+			
End Office a	nd Tandem Switching Usage and Common Transport Usage rates in the Port section of this rat	e exhibit shall apply to a	II combinations of lo	op/port network	elements except	for UNE Coin	Port/Loop Cor	mbinations.					
For Georgia	and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Combined a	and Not Currently Comb	ned Combos and the	e first and additi	ional Port nonrec	urring charges	apply to Not C	Currently Combined Comb	os. For Cur	rently			1
Combined C	ombos in GA, TN and all other states, the nonrecurring charges shall be those identified in the I	Nonrecurring - Currently	Combined sections.										
0 WID = 1101	DE ORANG LOOP WITH A WIRE LINE PORT (DEC)												<del></del>
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)									+			
UNE Port/Lo	op Combination Rates										+		
	2-Wire VG Loop/Port Combo - Zone	1		16.55						+	+		
	2-Wire VG Loop/Port Combo - Zone	2		25.51									
	2-Wire VG Loop/Port Combo - Zone	3		44.44									<b>—</b>
UNE Loop R	of on									+			
ONE LOOP R	2-Wire Voice Grade Loop (SL1) - Zone	1 UEPRX	UEPLX	14.35							+		
	2-Wire Voice Grade Loop (SL1) - Zone	2 UEPRX	UEPLX	23.31						1	1		
	2-Wire Voice Grade Loop (SL1) - Zone	3 UEPRX	UEPLX	42.24									
0.145 14.1	On the Property (Pro)												+
	e Grade Line Port Rates (Res) 2-Wire voice unbundled port - residenc	UEPRX	UEPRL	2.2						40.71	9.58		
	2-Wire voice unbundled port - resident. 2-Wire voice unbundled port with Caller ID - re	UEPRX	UEPRC	2.2						40.71	9.58		
	2-Wire voice unbundled port outgoing only - re	UEPRX	UEPRO	2.2						40.71	9.58		
	2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID -												
	res	UEPRX UEPRX	UEPAR UEPAP	2.2						40.71 40.71	9.58		t
	2-Wire voice unbundles res, low usage line port with Caller ID (LU	UEFKA	UEFAF	2.2						40.71	9.58		
										+	+		
FEATURES													
	All Features Offered	UEPRX	UEPVF	5.55	0	0				40.71	9.58		<b> </b>
LOCAL MUM	BER PORTABILITY	+ +								+	+		
LUCAL NUM	Local Number Portability (1 per por	UEPRX	LNPCX	0.35						+	+		ſ
							L			1			
	RING CHARGES (NRCs) - CURRENTLY COMBINED									1			
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	UEPRX	USAC2	-	2.8	0.41				40.71	9.58		1
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char	UEPRX	USACC		2.8	0.41				40.71	9.58		1
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with Chair 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database	OLFIX	JOACC		2.0	0.41				70.71	9.30		
	Update				1.44					8.25			
													+
ADDITIONAL		HERRY	HCACC	-		0				+	+		<b>—</b>
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ	UEPRX	USAS2	0	0	0				+	+	$\vdash$	
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)									+	1		ĺ
	17												
UNE Port/Lo	op Combination Rates									1			
	2-Wire VG Loop/Port Combo - Zone	1		16.55						4			<b>—</b>
	2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	3		25.51 44.44	1	1				+	+	$\vdash$	
	Z WITHO VIO ECOP/T OT CONTIDO - ZONE			77.77						+			
							1				+	1	
UNE Loop R	ates												
	ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 UEPBX 2 UEPBX	UEPLX UEPLX	14.35 23.31									

2-Wire Voice Grade Loop (SL1) - Zone		3 UEPBX	UEPLX	42.24							_
2-Wire Voice Grade Line Port (Bus)	-+										—
2-Wire voice unbundled port without Caller ID - bi	-+	UEPBX	UEPBL	2.2				40.71	9.58		—
2-Wire voice unbundled port with Caller + E484 ID - bi	_	UEPBX	UEPBC	2.2				40.71	9.58		_
2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	2.2				40.71	9.58		
2-Wire voice Grade unbundled Alabama extended local dialing parity port with Caller ID -											
bus		UEPBX	UEPAW	2.2				40.71	9.58		
2-Wire voice unbundled incoming only port with Caller ID - B		UEPBX	UPEB1	2.2				40.71	9.58		
LOCAL NUMBER PORTABILITY											
Local Number Portability (1 per por		UEPBX	LNPCX	0.35							
FEATURES											
		LIEDDY	LIEDVE				+ + + + + + + + + + + + + + + + + + + +	40.74	0.50		
All Features Offered		UEPBX	UEPVF	5.55	0	0		40.71	9.58		_
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED							+ + + + + + + + + + + + + + + + + + + +				_
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	-	UEPBX	USAC2		2.8	0.41		40.71	9.58		-
2-wire voice Grade Loop / Line Fort Combination - Conversion - Switch-as	-+	OLFBA	USACZ		2.0	0.41		40.71	9.30		_
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char		UEPBX	USACC		2.8	0.41					
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database		OL: DA	00/100		2.0	5.11					-
Update					1.44			8.25			
											_
ADDITIONAL NRCs											Т
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX	USAS2					40.71	9.58		Т
											Ξ
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)											
UNE Port/Loop Combination Rates											_
2-Wire VG Loop/Port Combo - Zone		1		16.55							Ξ
2-Wire VG Loop/Port Combo - Zone		2		25.51							
2-Wire VG Loop/Port Combo - Zone		3		44.44							
											_
UNE Loop Rates											
2-Wire Voice Grade Loop (SL 1) - Zone		1 UEPRG	UEPLX	14.35							_
2-Wire Voice Grade Loop (SL 1) - Zone		2 UEPRG	UEPLX	23.31							_
2-Wire Voice Grade Loop (SL 1) - Zone		3 UEPRG	UEPLX	42.24							_
2-Wire Voice Grade Line Port Rates (RES - PBX)							+ + + + + + + + + + + + + + + + + + + +		-		_
2-Wire Voice Grade Line Port Rates (RES - PBX)  2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	2.2			+ + + + + + + + + + + + + + + + + + + +	40.71	0.50		_
2-wire vG Onbundled Combination 2-way PBA Trunk Port - Re	-	UEPRG	UEPRU	2.2				40.71	9.58		_
LOCAL NUMBER PORTABILITY									+		_
Local Number Portability (1 per por		UEPRG	LNPCP	3.5			+ + + + + + + + + + + + + + + + + + + +				_
Ecolar Number 1 Gradinty (1 per por	-+	OLITIO	LINI OI	0.0					<del>                                     </del>		-
FEATURES											-
All Features Offered		UEPRG	UEPVF	5.55	0	0		40.71	9.58		-
					Ţ.				0.00		_
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED											Τ
											Τ
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPRG	USAC2		2.8	0.41		40.71	9.58		
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with											
Change		UEPRG	USACC		2.8	0.41		40.71	9.58		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database											
Update					1.44			8.25			
									$\vdash$		_
ADDITIONAL NRCs									$\vdash$		_
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ	$\rightarrow$	UEPRG	USAS2	0	0	0					_
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou					14.64	14.64		19.99	19.99	19.99	_
	$-\!\!\!+$		1	1					$\vdash$		_
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	-+		-								_
2-TAILE AOIDE GLADE FOOL AALL 5-MILE FINE LOVE (DOS - LDV)	-+										-
UNE Port/Loop Combination Rates	-+		<del>                                     </del>	1			+ + +			+	-
2-Wire VG Loop/Port Combo - Zone	-+	1	<b>I</b>	16.55			+ + +		<del>                                     </del>		-
2-Wire VG Loop/Port Combo - Zone		2		25.51			+ +		<del> </del>		-
2-Wire VG Loop/Port Combo - Zone		3		44.44					$\vdash$	-	-
	-+	-	1								-
UNE Loop Rates	-										-
2-Wire Voice Grade Loop (SL 1) - Zone		1 UEPPX	UEPLX	14.35							-
2-Wire Voice Grade Loop (SL 1) - Zone		2 UEPPX	UEPLX	23.31							_
		3 UEPPX	UEPLX	42.24							Ī
2-Wire Voice Grade Loop (SL 1) - Zone											
2-Wire Voice Grade Loop (SL 1) - Zone											
2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Line Port Rates (BUS - PBX)	=										
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt		UEPPX	UEPPC	2.2				40.71	9.58	·	_
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt  Line Side Unbundled Outward PBX Trunk Port - Bt		UEPPX	UEPPO	2.2				40.71	9.58		_
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt  Line Side Unbundled Outward PBX Trunk Port - Bt  Line Side Unbundled Incoming PBX Trunk Port - Bt		UEPPX UEPPX	UEPPO UEPP1	2.2				40.71 40.71	9.58 9.58		-
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt Line Side Unbundled Outward PBX Trunk Port - Bt Line Side Unbundled Incoming PBX Trunk Port - Bt 2-Wire Voice Unbundled 2-Way Combination PBX Alabama Calling Pt		UEPPX UEPPX UEPPX	UEPPO UEPP1 UEPA2	2.2 2.2 2.2				40.71 40.71 40.71	9.58 9.58 9.58		_
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt  Line Side Unbundled Outward PBX Trunk Port - Bt  Line Side Unbundled Incoming PBX Trunk Port - Bt		UEPPX UEPPX	UEPPO UEPP1	2.2				40.71 40.71	9.58 9.58		_

2-Wire Voice Unbundled PBX Toll Terminal Hotel Por	UEPPX	UEPXB	2.2	40.71 9.58
2-Wire Voice Unbundled PBX LD DDD Terminals Po	UEPPX	UEPXC	2.2	40.71 9.58
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	UEPPX	UEPXD	2.2	40.71 9.58
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pr	UEPPX	UEPXE	2.2	40.71 9.58
CMC V C II I I I I I I I I I I I I I I I I	HEDDY	LIEDVI		40.74
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P	UEPPX	UEPXL	2.2	40.71 9.58
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Pi	UEPPX	UEPXM	2.2	40.71 9.58
SW: V: III II II II II II II II II II II II I	HEDDY	LIEBYO		40.74
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling P	UEPPX	UEPXO	2.2	40.71 9.58
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	UEPPX	UEPXS	2.2	40.71 9.58
	<del>                                     </del>			
LOCAL NUMBER PORTABILITY				
Local Number Portability (1 per por	UEPPX	LNPCP	3.15	
	<del>                                     </del>			
FEATURES				
All Features Offerec	UEPPX	UEPVF	5.55 0 0	40.71 9.58
	<del>                                     </del>			
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED				
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As	UEPPX	USAC2	2.8 0.41	40.71 9.58
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with				
Change	UEPPX	USACC	2.8 0.41	40.71 9.58
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database				
Update			1.44	8.25
ADDITIONAL NRCs				
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ	UEPPX	USAS2	0 0 0	
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grot			14.64 14.64	19.99 19.99 19.99
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT				
UNE Port/Loop Combination Rates				
2-Wire VG Coin Port/Loop Combo – Zone			16.88	
2-Wire VG Coin Port/Loop Combo – Zone			25.84	
2-Wire VG Coin Port/Loop Combo – Zone			44.77	
UNE Loop Rates				
2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	14.35	
2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	23.31	
2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	42.24	
2-Wire Voice Grade Line Ports (COIN)				
2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)				
	UEPCO	UEPRF	2.53	40.71 9.58
2-Wire Coin 2-Way with Operator Screening (AL, K'	UEPCO	UEPRE	2.53	40.71 9.58
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K				
LA, MS)	UEPCO	UEPRA	2.53	40.71 9.58
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, M	UEPCO	UEPRB	2.53	40.71 9.58
2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local				
(AL, KY, LA, MS)	UEPCO	UEPCD	2.53	40.71 9.58
2-Wire Coin Outward with Operator Screening and 011 Blocking (AL, F	UEPCO	UEPRK	2.53	40.71 9.58
2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL,	1 22.33	1		
KY, LA, MS)	UEPCO	UEPRH	2.53	40.71 9.58
2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, and Local				1.5 1 0.00
(AL, KY, LA, MS)	UEPCO	UEPCN	2.53	40.71 9.58
2-Wire 2-Way Smartline with 900/976 (all states except Li	UEPCO	UEPCK	2.53	40.71 9.58
2-Wire Coin Outward Smartline with 900/976 (all states except L	UEPCO	UEPCR	2.53	40.71 9.58
ADDITIONAL UNE COIN PORT/LOOP (RC)	02.00	02.0.0		10
UNE Coin Port/Loop Combo Usage (Flat Rate	UEPCO	URECU	1.56 0 0	<del>-                                     </del>
STE CONT FOR ECOP CONTROL COAge Trial Mark	OLF CO	UNECO		<del>-                                     </del>
LOCAL NUMBER PORTABILITY		1		
Local Number Portability (1 per por	UEPCO	LNPCX	0.35	<del>-                                     </del>
Essai Number Fortability (1 per per	OLF CO	LINEON	5.55	<del>-                                     </del>
FEATURES	<del></del>	+		<del>-                                     </del>
All Features Offered	UEPCO	UEPVF	5.55 0 0	27.37 12.97 17.77
Saturdo Orioto.	02100		<u> </u>	27.07 12.07 17.77
NONRECURRING CHARGES - CURRENTLY COMBINED		1		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	UEPCO	USAC2	2.8 0.41	40.71 9.58
2 10 Total Ordad 2009 / Emo Total Combination Conversion Contact at	02100	33/102	2.0 0.41	.5
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char	UEPCO	USACC	2.8 0.41	40.71 9.58
2 WHO VOICE State Edgy / Line Fort Combination - Conversion - Switch with Char	OLF CO	USACC	2.0 0.41	40.71 3.30
ADDITIONAL NRCs	<del></del>	+		
	UEPCO	LICACO	0 0	40.71 9.58
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ	UEPCO	USAS2	0 0	40.71 9.58
2 WIDE VOICE CRADE LOOP, BUS ONLY, WITH 2 WIDE DID TRUNK DODT	+	+		
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT	<del></del>	+		
INIT D. All O. All C D. A.	+			
UNE Port/Loop Combination Rates				
		1	29.59	
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	1 2	_	36.58	

2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	3		45.06								
UNE Loop Rates									-	-	
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	1 UEPPX	UECD1	20.42					19.99	19.99	19.99	
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	2 UEPPX	UECD1	27.41					19.99	19.99	19.99	
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	3 UEPPX	UECD1	35.89					19.99	19.99	19.99	
UNE Port Rate	LIEDDY	LIEDDA	0.47					40.00	40.00	40.00	
Exchange Ports - 2-Wire DID Por	UEPPX	UEPD1	9.17					19.99	19.99	19.99	
NONRECURRING CHARGES - CURRENTLY COMBINED									+	+ + + + + + + + + + + + + + + + + + + +	
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as	UEPPX	USAC1		14.61	3.73			19.99	19.99	19.99	
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable											
Changes	UEPPX	USA1C		14.61	3.73			19.99	19.99	19.99	
ADDITIONAL NRCs	LIEDDY	110404		50.50	50.50			40.00		40.00	
2-Wire DID Subsequent Activity - Add Trunks, Per Trun	UEPPX	USAS1		53.56	53.56			19.99	19.99	19.99	
Telephone Number/Trunk Group Establisment Charges									+	+ + + + + + + + + + + + + + + + + + + +	
DID Trunk Termination (One Per Port	UEPPX	NDT	0	0	0			19.99	19.99	19.99	
Additional DID Numbers for each Group of 20 DID Numbe	UEPPX	ND4	0	0	0			19.99	19.99	19.99	
DID Numbers, Non- consecutive DID Numbers , Per Number	UEPPX	ND5	0	0	0		19.99				
Reserve Non-Consecutive DID number	UEPPX	ND6	0	0	0		19.99				
Reserve DID Numbers	UEPPX	NDV	0	0	0		19.99				
LOCAL NUMBER PORTABILITY									+	+	
Local Number Portability (1 per por	UEPPX	LNPCP	3.15						+	+	
Local Namber Fortability (1 per por	OLITA	LIVIOI	0.10						+		
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT									1		
UNE Port/Loop Combination Rates										-	
	4 115000 115000		00.00								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	1 UEPPB UEPPR		36.62		+				+	+	
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	2 UEPPB UEPPR		44.49						+		
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	3 UEPPB UEPPR		55.39								
UNE Loop Rates									+	+	
	1 UEPPB UEPPR	USL2X	27.2					10.00	10.00	40.00	
2-Wire ISDN Digital Grade Loop - UNE Zone			27.2					19.99	19.99	19.99	
2-Wire ISDN Digital Grade Loop - UNE Zone	2 UEPPB UEPPR	USL2X	35.07					19.99	19.99	19.99	
2-Wire ISDN Digital Grade Loop - UNE Zone	3 UEPPB UEPPR	USL2X	45.97					19.99	19.99	19.99	
UNE Port Rate											
Exchange Port - 2-Wire ISDN Line Side Po	UEPPB UEPPR	UEPPB	9.42					19.99	19.99	19.99	
NONRECURRING CHARGES - CURRENTLY COMBINED									+	+	
NOTICE OF THE COMMENTED									+	<del>                                     </del>	
2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Convers	UEPPB UEPPR	USACB	0	77.01	54.04			19.99	19.99	19.99	
ADDITIONAL NRCs											
LOCAL NUMBER PORTABILITY									+	+	
Local Number Portability (1 per por	UEPPB UEPPR	LNPCX	0.35	0	0						
	OLFFB OLFFR	LINFOX	0.55	U	0						
B-CHANNEL USER PROFILE ACCESS:											
CVS/CSD (DMS/5ESS)	UEPPB UEPPR	U1UCA	0	0	0						
CVS (EWSD)	UEPPB UEPPR	U1UCB	0	0	0				1	1	
CSD	UEPPB UEPPR	U1UCC	0	0	0						
			1 -								
B-CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)			-	_				1		1	
CVS/CSD (DMS/5ESS)	UEPPB UEPPR	U1UCD	0	0	0				+		
CVS (EWSD)	HEDDO HEDDO	HALICE	0	0							
CVS (EWSD)	UEPPB UEPPR	U1UCE	U	U	0				+	++	
CSD	UEPPB UEPPR	U1UCF	0	0	0						
USER TERMINAL PROFILE			1					+	+	+	
									1		
User Terminal Profile (EWSD only)	UEPPB UEPPR	U1UMA	0	0	0			1	<del></del>	1	
VERTICAL FEATURES									<u> </u>		
		=									
All Vertical Features - One per Channel B User Profile	UEPPB UEPPR	UEPVF	5.55	0	0	I		1			

Interoffice Channel mileage each, including first mile and facilities termination	UEPPB UEPPR	M1GNC	17.81	107.11	48.27		19.99	19.99	19.99	
Interoffice Channel mileage each, additional mile	UEPPB UEPPR	M1GNM	0.0339	0	0		0	19.99	19.99	,
A WIDE DOLD DISTALL COD WITH A WIDE INDIVIDUAL PROPERTY TRUNK DODT										
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT								+	+	
UNE Port/Loop Combination Rates									+	T
4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	1 UEPPP		198.29							
4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	2 UEPPP		274							-
4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	3 UEPPP		425.41							-
UNE Loop Rates								-	+	t
4-Wire DS1 Digital Loop - UNE Zone	1 UEPPP	USL4P	101.92				19.99	19.99	19.99	9
4-Wire DS1 Digital Loop - UNE Zone :	2 UEPPP	USL4P	177.63				19.99	19.99	19.99	
4-Wire DS1 Digital Loop - UNE Zone :	3 UEPPP	USL4P	329.04				19.99	19.99	19.99	9
UNE Port Rate									+	1
Exchange Ports - 4-Wire ISDN DS1 Por	UEPPP	UEPPP	96.37				19.99	19.99	19.99	
NOVIDEO URBINO OLIABORO OLIBRENTI VIOLABINIER										
NONRECURRING CHARGES - CURRENTLY COMBINED  4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion -										-
Switch-as-is	UEPPP	USACP	0	238.13	157.11		19.99	19.99	19.99	
ADDITIONAL NRCs								-		
4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within										T
Std Allowance	UEPPP	PR7TF		0.9801			19.99	19.99	19.99	L
4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All State except NC)	UEPPP	PR7TO		23.02	23.02		19.99	19.99	19.99	
4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above										
Std Allowance	UEPPP	PR7ZT		46.05	46.05		19.99	19.99	19.99	-
LOCAL MUNICIPAL PROPERTY										
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per por	UEPPP	LNPCN	1.75							-
	92.11	E. III O. I	1.70							
INTERFACE (Provsioning Only)										
Voice/Data	UEPPP UEPPP	PR71V	0	0	0					-
Digital Data Inward Data	UEPPP	PR71D PR71E	0	0	0			-	+	
New or Additional "B" Channel	LIEDDO	DD7D1/	_							
New or Additional - Voice/Data B Channel  New or Additional - Digital Data B Channel	UEPPP UEPPP	PR7BV PR7BF	0	29.05			19.99	19.99	19.99	9
New or Additional Inward Data B Channel	UEPPP	PR7BD	0	29.05 29.05			19.99 19.99	19.99 19.99	19.99	
New or Additional Useage Sensitive Voice Data B Channel	UEPPP	PR7BS	0	29.05			19.99	19.99	19.99 19.99	
New or Additional Useage Sensitive Digital Data B Channel	UEPPP	PR7BU	0	29.05			19.99	19.99	19.99	
CALL TYPES			_		_					
Inward	UEPPP	PR7C1	0	0	0					-
Outword	LIEDDD									
Outward Two-way	UEPPP	PR7C0	0						+	1
Outward Two-way	UEPPP UEPPP	PR7C0 PR7CC	0	0	0				+	
Two-way Interoffice Channel Mileage	UEPPP	PR7CC	0	0	0					
Two-way  Interoffice Channel Mileage  Fixed Each Including First Mile	UEPPP	PR7CC 1LN1A	0 80.382				19.99	19.99	19.99	9
Two-way Interoffice Channel Mileage	UEPPP	PR7CC	0	0	0		19.99	19.99	19.99	)
Two-way  Interoffice Channel Mileage  Fixed Each Including First Mile	UEPPP	PR7CC 1LN1A	0 80.382	0	0		19.99	19.99	19.99	9
Two-way  Interoffice Channel Mileage  Fixed Each Including First Mile  Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates	UEPPP UEPPP UEPPP	PR7CC 1LN1A	0 80.382 0.692	0	0					
Interoffice Channel Mileage Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4/W DS1 Digital Loop/4/W DDITS Trunk Port - UNE Zone	UEPPP UEPPP UEPPP 1 UEPDC	PR7CC 1LN1A	0 80.382 0.692	0	0		19.99	19.99	19.99	
Two-way  Interoffice Channel Mileage Fixed Each Including First Mili Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	UEPPP UEPPP UEPPP 1 UEPDC 2 UEPDC	PR7CC 1LN1A	0 80.382 0.692 170.59 246.3	0	0		19.99 19.99	19.99 19.99	19.99	9
Interoffice Channel Mileage Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4/W DS1 Digital Loop/4/W DDITS Trunk Port - UNE Zone	UEPPP UEPPP UEPPP 1 UEPDC	PR7CC 1LN1A	0 80.382 0.692	0	0		19.99	19.99	19.99	9
Two-way  Interoffice Channel Mileage Fixed Each Including First Mili Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	UEPPP UEPPP UEPPP  1 UEPDC 2 UEPDC 3 UEPDC	PR7CC  1LN1A  1LN1B	0 80.382 0.692 170.59 246.3 397.71	0	0		19.99 19.99 19.99	19.99 19.99 19.99	19.99	9
Two-way  Interoffice Channel Mileage Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  WDS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  WDS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	UEPPP	PR7CC  1LN1A 1LN1B  USLDC	0 80.382 0.692 170.59 246.3	0	0		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	
Two-way  Interoffice Channel Mileage Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone:  UNE Loop Rates  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:	UEPPP	PR7CC  1LN1A 1LN1B  USLDC USLDC	0 80.382 0.692 170.59 246.3 397.71	0	0	23.41	19.99 19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99 19.99	
Two-way  Interoffice Channel Mileage Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  WDS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  WDS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	UEPPP	PR7CC  1LN1A 1LN1B  USLDC	0 80.382 0.692 170.59 246.3 397.71	0	0	23.41	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	9
Interoffice Channel Mileage Fixed Each Including First Mili Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone:	1 UEPPC 2 UEPPC 3 UEPDC 1 UEPDC 2 UEPDC 3 UEPDC 3 UEPDC	PR7CC  1LN1A 1LN1B  USLDC USLDC USLDC	0 80.382 0.692 170.59 246.3 397.71 101.92	0	0	23.41	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	
Two-way  Interoffice Channel Mileage Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE POrt/Loop Combination Rates 4-W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone:	UEPPP	PR7CC  1LN1A 1LN1B  USLDC USLDC	0 80.382 0.692 170.59 246.3 397.71	0	0	23.41	19.99 19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99 19.99	9
Interoffice Channel Mileage  Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone:  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone:  UNE Loop Rates  4-Wire DS1 Digital Loop - UNE Zone 4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Trunk Por	1 UEPPC 2 UEPDC 3 UEPDC 1 UEPDC 2 UEPDC 3 UEPDC UEPDC UEPDC UEPDC UEPDC	1LN1A 1LN1B  USLDC USLDC USLDC USLDC USLDC	0 80.382 0.692 170.59 246.3 397.71 101.92	198.15	148.18 25.44	23.41	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	
Interoffice Channel Mileage Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4-W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone: 4W DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Loop - UNE Zone: 4-Wire DS1 Digital Trunk Por  NONRECURRING CHARGES - CURRENTLY COMBINED 4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as	1 UEPPC 2 UEPPC 3 UEPDC 1 UEPDC 2 UEPDC 3 UEPDC 3 UEPDC	PR7CC  1LN1A 1LN1B  USLDC USLDC USLDC	0 80.382 0.692 170.59 246.3 397.71 101.92	0	0	23.41	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	
Interoffice Channel Mileage  Fixed Each Including First Mile Each Airline-Fractional Additional Mil  4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone:  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone:  UNE Loop Rates  4-Wire DS1 Digital Loop - UNE Zone 4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Loop - UNE Zone:  4-Wire DS1 Digital Trunk Por	1 UEPPC 2 UEPDC 3 UEPDC 1 UEPDC 2 UEPDC 3 UEPDC UEPDC UEPDC UEPDC UEPDC	1LN1A 1LN1B  USLDC USLDC USLDC USLDC USLDC	0 80.382 0.692 170.59 246.3 397.71 101.92	198.15	148.18 25.44	23.41	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	

Attachment 2 Exhibit C

ADDITIONAL NRCs												1	
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel													_
Activation/Chan - 2-Way Trunl		UEPDC	UDTTA		28.85	28.95				19.99	19.99	19.9	99
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-					1								Ī
Way Outward Trunk		UEPDC	UDTTB		28.85	28.85				19.99	19.99	19.9	99
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan Inward													
Trunk w/out DID		UEPDC	UDTTC		28.85	28.85				19.99	19.99	19.9	99
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan - Inwa	rd												
Trunk with DID		UEPDC	UDTTD		28.85	28.85				19.99	19.99	19.9	99
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way													
DID w User Trans		UEPDC	UDTTE		28.85	28.85				19.99	19.99	19.9	99
BIPOLAR 8 ZERO SUBSTITUTION													_
B8ZS -Superframe Format		UEPDC	CCOSF		0	600				19.99	19.99	19.9	
B8ZS - Extended Superframe Forma		UEPDC	CCOEF		0	600				19.99	19.99	19.9	96
													_
Alternate Mark Inversion						_							_
AMI -Superframe Format		UEPDC	MCOSF MCOPO		0	0							_
AMI - Extended SuperFrame Forma		UEPDC	MCOPO		0	0							_
													-
Talanhana Nimba-/Tamih Casar Fatablianant Channa													_
Telephone Number/Trunk Group Establisment Charges  Telephone Number for 2-Way Trunk Grou	ļ	UEPDC	UDTGX	0	-		+	+ +			1		_
Telephone Number for 1-Way Outward Trunk Grou		UEPDC	UDTGY	0	+				19.99	1		1	-
Telephone Number for 1-Way Jouward Trunk Group Without Di	-	UEPDC	UDTGZ	0	+		1	+ +	19.99	1		1	-
DID Numbers for each Group of 20 DID Number	-	UEPDC	ND4	0	0		1	+ +	19.99	1		1	-
DID Numbers for each Group of 20 DID Number  DID Numbers, Non- consecutive DID Numbers , Per Number		UEPDC	ND4 ND5	0	U				19.99	1		1	-
Reserve Non-Consecutive DID Numbers , Per Numbers Reserve Non-Consecutive DID Nos	-	UEPDC	ND5 ND6	0	0	0			19.99	-		+	_
	-			_	_	0	_			1		1	_
Reserve DID Numbers	ļ	UEPDC	NDV	0	0	U	+	+ +	19.99		1		_
Dedicated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DD	TS True	nk Port	+	1	-		-	+ +		1		+	-
Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination	irul	UEPDC	1LNO1	79.69	198.15	148.18	25.44	20.42		19.99	19.99	19.99	-
Interoffice Channel Mileage - Additional rate per mile - 0-8 mil	-	UEPDC	1LNOA	0.692	198.15	148.18	20.44	20.42		19.99	19.99	19.99	-
Interoffice Channel Mileage - Additional rate per fille - 0-8 mil	-	UEPDC	1LNO2	0.692	0	0	1	+ +		1		1	-
Interoffice Channel Mileage - Additional rate per mile - 9-25 mil		UEPDC	1LNOB	0.692	0	0							-
Interoffice Channel Mileage - Additional rate per fille - 9-25 mile  Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination		UEPDC	1LNO3			0	0						-
Interoffice Channel Mileage - Additional rate per mile - 25+ mil		UEPDC	1LNOC	0.692	0	0	U						-
Interonice Channel Mileage - Additional rate per mile - 25+ mil							0						_
Local Number Portability, per DS0 Activate		UEPDC UEPDC	LNPCP	3.15	0	0	U						_
Central Office Termininating Poir		DEFDC	CIG	- 0									-
													-
4-WIRE DS1 LOOP WITH CHANNELIZATION WITH PORT													-
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations													-
Each System can have up to 24 combinations of rates depending on type and number of ports used													-
Each System start day to 2 r combinations of factor deponding on type and number of porter dece													-
UNE DS1 Loop													_
4-Wire DS1 Loop - UNE Zone 1		1 UEPMG	USLDC	101.92	0	0							_
4-Wire DS1 Loop - UNE Zone 2		2 UEPMG	USLDC	177.63	0	0							_
4-Wire DS1 Loop - UNE Zone (		3 UEPMG	USLDC	329.04	0	0							_
													Τ
UNE DSO Channelization Capacities (D4 Channel Bank Configurations)													Τ
24 DSO Channel Capacity - 1 per DS		UEPMG	VUM24	115.89	0	0							Т
48 DSO Channel Capacity - 1 per 2 DS1		UEPMG	VUM48	231.78	0	0							Τ
96 DSO Channel Capacity -1per 4 DS1:		UEPMG	VUM96	463.56	0	0				1			-
144 DS0 Channel Capacity - 1 per 6 DS1:		UEPMG	VUM14	695.34	Ó	0				1		İ	_
192 DS0 Channel Capacity -1 per 8 DS1		UEPMG	VUM19	980	0	0				1		İ	_
240 DS0 Channel Capacity - 1 per 10 DS1		UEPMG	VUM20	1158.9	0	0				1			-
288 DS0 Channel Capacity - 1 per 12 DS1		UEPMG	VUM28	1390.68	0	0				1			-
384 DS0 Channel Capacity - 1 per 16 DS1		UEPMG	VUM38	1854.24	0	0				1			-
480 DS0 Channel Capacity - 1 per 20 DS1		UEPMG	VUM40	2317.8	0	0							Τ
576 DS0 Channel Capacity -1 per 24 DS1:		UEPMG	VUM57	2781.36	0	0							-
672 DS0 Channel Capacity - 1 per 28 DS1		UEPMG	VUM67	3244.92	0	0	1			1			-
5.2.2.5 Statistic depactly in por 20 00 in		OE: MO	. 5300	221102	7	1							-
Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - Co	nversio	n Charge Based on a Svs	tem										-
A Minimum System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DSO Ports										1			-
Multiples of this configuration functioning as one are considered Add'l after the minimum system con										1			-
	inguratio		USAC4	0	300.95	16.72				19.99	19.99	19.99	-
System Additions at End User Locations Where 4-Wire DS1 Loop with Channelization with Port Co			2007	ľ	000.00		1			.0.00		70.00	-
New (Not Currently Combined) In Georgia & Tennessee Only	oati	on carrointy Exists and	+		+					1		1	-
NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature Activation -	-			+						1	-	1	-
New GA & TN Only	1	UEPMG	VUMD4	0	716.11	468.04	148.75	17.65		19.99	19.99	19.99	
Bipolar 8 Zero Substitution	-	UEPIVIG	V UIVID4	U	/10.11	408.04	148./5	17.00		19.99	19.99	19.99	_
		LIEDMO	00005	-	0	000	-			40.00	40.00	10.00	_
Clear Channel Capability Format, superframe - Subsequent Activity Or		UEPMG	CCOSF	U	U	600				19.99	19.99	19.99	_
Clear Channel Capability Format - Extended Superframe - Subsequent Activity O		UEPMG	CCOEF	U	0	600				19.99	19.99	19.99	_
Alternate Mark Inversion (AMI)			1	1						1			_
			MCOSF	In	0	0	1	1		1		1	
Superframe Forma		UEPMG		U		-							_
		UEPMG	MCOPO	0	0	0							I

Exchange Ports													
Line Side Combination Channelized PBX Trunk Port - Busine:	UEPPX	UEPCX	1.58	0	0	0	0		19.99				
Line Side Outward Channelized PBX Trunk Port - Busines	UEPPX	UEPOX	1.58	0	0	0	0		19.99				
Line Side Inward Only Channelized PBX Trunk Port without DI	UEPPX	UEP1X	1.58	0	0	0	0		19.99				
2-Wire Trunk Side Unbundled Channelized DID Trunk Po	UEPPX	UEPDM	9.2	0	0	0	0		19.99				
2-Wire Channelized PBX Area Calling Service Combination Port (AL On	UEPPX	UEPA4	1.58	0	0				19.99				
2 Wire Channelized PBX Area Calling Service Outgoing Only Port (AL On	UEPPX	UEPA3	1.58	0	0				19.99				
Feature Activations - Unbundled Loop Concentration	OE. TX	OL: 710	1.00						10.00				
Feature (Service) Activation for each Line Side Port Terminated in D4 Ba	UEPPX	1PQWM	0.64	25.39	13.41	4.19	4.16			19.99	19.99	19.99	19.99
Feature (Service) Activation for each Trunk Side Port Terminated in D4 Ba	UEPPX	1PQWU	0.64	78.13	18.42	59.24	11.58			19.99	19.99	19.99	19.99
	UEPPX	IPQWU	0.64	78.13	18.42	59.24	11.58			19.99	19.99	19.99	19.99
Telephone Number/ Group Establishment Charges for DID Service													
DID Trunk Termination (1 per Port	UEPPX	NDT	0										
DID Numbers - groups of 20 - Valid all State	UEPPX	ND4	0	0	0				19.99				
Non-Consecutive DID Numbers - per numbe	UEPPX	ND5	0	0	0				19.99				
Reserve Non-Consecutive DID Number:	UEPPX	ND6	0	0	0								
Reserve DID Numbers	UEPPX	NDV	0	0	0								
Local Number Portability													
Local Number Portability - 1 per po	UEPPX	LNPCP	3.15	0	0								
FEATURES - Vertical and Optional	OLFFX	LINEOF	3.13	0	U								
Local Switching Features Offered with Line Side Ports Only													
All Features Available	UEPPX	UEPVF	5.55	0	0				19.99				
										1			
ED PORT LOOP COMBINATIONS - MARKET RATES													1
										1			1
Market Rates shall apply where BellSouth is not required to provide unbundled local switching or switch po	orts per ECC and/or State	Commission rules				+	+		<b> </b>	1	1	_	1
These scenarios include:	poi Co dilujoi otate		_	1	+	+	+		<b>!</b>	+	+		t
Unbundled port/loop combinations that are Not Currently Combined in all of the BellSouth states except	nt on noted for Consists	d Tannasaa	_	+	+	+	+		-	+	+		<b>+</b>
1. Unbundled porvious combinations that are not currently combined in an or the BellSouth states except	pt as noted for Georgia ar	u rennessee.	-1	P00	in a lane til	+	+		<del>                                     </del>	1	+		<del> </del>
2. Unbundled port/loop combinations that are Currently Combined or Not Currently Combined in Zone 1 o							+		1	1	1		1
The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale, Miami); CA (New Orlando, Ft. Lauderdale	reans); NC (Greensboro-\	vinston Salem-High	point/Charlotte-	astonia-Rock F	ıııı); IN (Nashvil	ie).	.1		1	1			ļ
BellSouth currently is developing the billing capability to mechanically bill the recurring and non-recurring N	Market Rates in this section	n. In the interim Be	IlSouth shall bill	the rates in the (	Cost-Based sect	ion preceding	in lieu of the M	arket Rates	and reserve	es the right to	true-up the	illing differenc	€.
End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section of this re	ate exhibit shall apply to a	all combinations of I	oop/port networ	celements excep	ot for UNE Coin	Port/Loop Co	mbinations whi	ch have a fla	at rate usag	e charge (US	OC: URECL	).	
For Not Currently Combined scenarios where Market Rates apply, the Nonrecurring charges are listed in the	he First and Additional NE	C columns for each	h Port USOC. F	or Currently Com	bined scenarios	the Nonrecu	rring charges a	re listed in th	he NRC - Ci	urrently Comb	ined section	. Additional NE	Cs may apr
are categorized accordingly.						,							,,
are categorized accordingly.							1			1	1		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)													
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates													
UNE Port/Loop Combination Rates	1		28.35										
UNE Port/Loop Combination Rates  [2-Wire VG Loop/Port Combo - Zone			28.35 37.31										
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2		37.31										
UNE Port/Loop Combination Rates  [2-Wire VG Loop/Port Combo - Zone													
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	2		37.31										
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone UNE Loop Rates	2 3	UEDLY	37.31 56.24										
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone UNE Loop Rates  2-Wire Voice Grade Loop (\$L1) - Zone	2 3 3 1 UEPRX	UEPLX	37.31 56.24 14.35										
UNE Loop Carbination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 UEPRX 2 UEPRX	UEPLX	37.31 56.24 14.35 23.31										
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone UNE Loop Rates  2-Wire Voice Grade Loop (\$L1) - Zone	2 3 3 1 UEPRX		37.31 56.24 14.35										
UNE Loop Carbination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 UEPRX 2 UEPRX	UEPLX	37.31 56.24 14.35 23.31										
UNE Loop Carbination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 UEPRX 2 UEPRX	UEPLX	37.31 56.24 14.35 23.31										
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Line Port (Res)	1 UEPRX 2 UEPRX	UEPLX	37.31 56.24 14.35 23.31	90	90					40.71	9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Une Dundled Loop (SL1) - Zone 2-Wire Voice Unbundled port - residenc	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX	UEPLX UEPLX UEPRL	37.31 56.24 14.35 23.31 42.24							40.71			
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	37.31 56.24 14.35 23.31 42.24	90	90					40.71	9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re	2 3 3 4 1 UEPRX 2 UEPRX 3 UEPRX UEPR	UEPLX UEPLX UEPRL UEPRC UEPRO	37.31 56.24 14.35 23.31 42.24 14 14	90 90	90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	37.31 56.24 14.35 23.31 42.24	90	90					40.71	9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - Residenc 2-Wire voice unbundled port with Caller ID - Re 2-Wire voice unbundled port outgoing only - Re 2-Wire voice unbundled sers, low usage line port with Caller ID (LU	2 3 3 4 1 UEPRX 2 UEPRX 3 UEPRX UEPR	UEPLX UEPLX UEPRL UEPRC UEPRO	37.31 56.24 14.35 23.31 42.24 14 14	90 90	90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled sers, low usage line port with Caller ID (LU  LOCAL NUMBER PORTABILITY	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14	90 90	90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - Residenc 2-Wire voice unbundled port with Caller ID - Re 2-Wire voice unbundled port outgoing only - Re 2-Wire voice unbundled sers, low usage line port with Caller ID (LU	2 3 3 4 1 UEPRX 2 UEPRX 3 UEPRX UEPR	UEPLX UEPLX UEPRL UEPRC UEPRO	37.31 56.24 14.35 23.31 42.24 14 14	90 90	90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port of Unique only - re 2-Wire voice unbundled port of utigging only - re 2-Wire voice unbundled port of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14	90 90	90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port with Caller ID - te 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port of Unique only - re 2-Wire voice unbundled port of utigging only - re 2-Wire voice unbundled port of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re 2-Wire voice unbundled sort of utigging only - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14	90 90	90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port with Caller ID - te 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port with Caller ID - te 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 5-Wire voice unbundled port outgoing only - re 5-Wire voice unbundled port outgoing only - re 6-Wire voice unbundled port outgoing only - re 6-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled po	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outgoing only - re 5-Wire voice unbundled port outgoing only - re 5-Wire voice unbundled port outgoing only - re 6-Wire voice unbundled port outgoing only - re 6-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port outgoing only - re 7-Wire voice unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port unbundled port	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port ortioging only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port out	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled po	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port ortioging only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port out	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port with Caller ID - re 4-Wire voice unbundled port with Caller ID - re 4-Wire voice unbundled port with Caller ID - re 4-Wire voice unbundled port with Caller ID - re 4-Wire voice unbundled port with Caller ID - re 4-Wire voice unbundled port with Caller ID - re 4-Wire voice unbundled port with Caller ID - re 4-Wire voice unbundled port with Call	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundleds pres, low usage line port with Caller ID (LU  LOCAL NUMBER PORTABILITY Local Number Portability (1 per por  FEATURES  All Features Offerec  NONRECURRING CHARGES - CURRENTLY COMBINED  ADDITIONAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  UNE Port/Loop Combination Rates	2 3 UEPRX 2 UEPRX 3 UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 10.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port outgoing only - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loo	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 10.35 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 4-Wire voice unbundled port outg	2 3 UEPRX 2 UEPRX UEPXX UEPRX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 0.35 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port outgoing only - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loop Combination Rates 3-Wire voice Grade Loop/Loo	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 10.35 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2 -Wire VG Loop/Port Combo - Zone 2 -Wire VG Loop/Port Combo - Zone 2 -Wire VG Loop/Port Combo - Zone 2 -Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2 -Wire Voice Grade Loop (SL1) - Zone 2 -Wire Voice Grade Loop (SL1) - Zone 2 -Wire Voice Grade Loop (SL1) - Zone 2 -Wire Voice Grade Loop (SL1) - Zone  2 -Wire Voice Grade Line Port (Res)  2 -Wire voice unbundled port - residenc 2 -Wire voice unbundled port outgoing only - re 2 -Wire voice unbundled port outgoing only - re 2 -Wire voice unbundled port outgoing only - re 2 -Wire voice unbundled port outgoing only - re 2 -Wire voice unbundled port outgoing only - re 2 -Wire voice unbundled port outgoing only - re 3 - Wire voice unbundled port outgoing only - re 4 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 6 - Wire voice unbundled port outgoing only - re 7 - Wire voice unbundled port outgoing only - re 8 - Wire voice unbundled port versidence 9 - Wire voice unbundled port unbund	2 3 UEPRX 2 UEPRX UEPXX UEPRX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX UEPXX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 0.35 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice unbundled port - residenc 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 3 - Wire voice unbundled port outgoing only - re 4 - Wire voice unbundled port outgoing only - re 4 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 4 - Wire voice unbundled port outgoing only - re 5 - Wire voice Unbundled port outgoing only - re 5 - Wire Voice Grability (1 per por  FEATURES  All Features Offerec  NONRECURRING CHARGES - CURRENTLY COMBINED  ADDITIONAL NRCS  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone UNE Loop Rates	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP  LNPCX  UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 10.35 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice Grade Loop (SL1) - Zone 2 - Wire Voice unbundled port - residenc 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 2 - Wire voice unbundled port outgoing only - re 3 - Wire voice unbundled port outgoing only - re 4 - Wire voice unbundled port outgoing only - re 4 - Wire voice unbundled port outgoing only - re 5 - Wire voice unbundled port outgoing only - re 4 - Wire voice unbundled port outgoing only - re 5 - Wire voice Unbundled port outgoing only - re 5 - Wire Voice Grability (1 per por  FEATURES  All Features Offerec  NONRECURRING CHARGES - CURRENTLY COMBINED  ADDITIONAL NRCS  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone 2 - Wire VG Loop/Port Combo - Zone UNE Loop Rates	2 3 1 UEPRX 2 UEPRX 3 UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP  LNPCX  UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 10.35 0.35	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outging only - re 2-Wire voice unbundled port outging only - re 2-Wire voice unbundled port outging only - re 2-Wire voice unbundles res, low usage line port with Caller ID (LU  LOCAL NUMBER PORTABILITY Local Number Portability (1 per por  FEATURES All Features Offerec  NONRECURRING CHARGES - CURRENTLY COMBINED  ADDITIONAL NRCs NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  2-Wire VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  UNE Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2 3 3 UEPRX 2 UEPRX 3 UEPRX UE	UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP  LNPCX  UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 10 0.35 0 0 28.35 37.31 56.24	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 3-Wire voice unbundled port outgoing only - re 3-Wire voice Unbundled port outgoing only - re 3-Wire voice Unbundled port outgoing only - re 3-Wire Voice Grade Loop/Line Port With Caller ID (LU  LOCAL NUMBER PORTABILITY Local Number Portability (1 per por  FEATURES All Features Offerec  NONRECURRING CHARGES - CURRENTLY COMBINED  ADDITIONAL NRCs NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  UNE Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2 3 3 UEPRX 2 UEPRX UEPX	UEPLX UEPRL UEPRC UEPRO UEPRO UEPAP  LNPCX  UEPVF  USAS2	37.31 56.24 14.35 23.31 42.24 14 14 14 15 0.35 0 0	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outging only - re 2-Wire voice unbundled port outging only - re 2-Wire voice unbundled port outging only - re 2-Wire voice unbundles res, low usage line port with Caller ID (LU  LOCAL NUMBER PORTABILITY Local Number Portability (1 per por  FEATURES All Features Offerec  NONRECURRING CHARGES - CURRENTLY COMBINED  ADDITIONAL NRCs NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  2-Wire VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  UNE Port/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2 3 3 UEPRX 2 UEPRX 3 UEPRX UE	UEPLX  UEPRL UEPRC UEPRO UEPRO UEPAP  LNPCX  UEPVF	37.31 56.24 14.35 23.31 42.24 14 14 14 10 0.35 0 0 28.35 37.31 56.24	90 90 90	90 90 90 90					40.71 40.71	9.58 9.58		

	2-Wire voice unbundled port without Caller ID - bt		UEPBX	UEPBL	14	90	90			40.71	9.58		
:	2-Wire voice unbundled port with Caller + E484 ID - bi		UEPBX	UEPBC	14	90	90			40.71	9.58		
	2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	14	90	90			40.71			
	DED BODTADUUTV												
	BER PORTABILITY Local Number Portability (1 per por		UEPBX	LNPCX	0.35								
	Local Number Portability (1 per por		UEPBX	LINPCX	0.35								
FEATURES													
1 2711 01120													
NONRECURE	RING CHARGES - CURRENTLY COMBINED												
ADDITIONAL	NRCs												
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequε		UEPBX	USAS2		0	0						
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)												
	op Combination Rates												
	2-Wire VG Loop/Port Combo - Zone	1			28.35								
	2-Wire VG Loop/Port Combo - Zone	2			37.31								
, <i> </i>	2-Wire VG Loop/Port Combo - Zone	3			56.24								
<u> </u>													
UNE Loop Ra													
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRG	UEPLX	14.35								
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRG	UEPLX	23.31								
,	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRG	UEPLX	42.24								
2 Miro Vair-	Grade Line Port Rates (RES - PBX)			-									-
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	14	90	90			40.71	9.58		-
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - RE		UEPRG	UEPRD	14	90	90			40.71	9.58		
LOCAL NILIM	BER PORTABILITY												
	Local Number Portability (1 per por		UEPRG	LNPCP	3.15								
·	Local Number Fortability (1 per por		OLFING	LINFOF	3.13								
FEATURES													
TEATORES													
NONRECLIRE	RING CHARGES - CURRENTLY COMBINED												
HONKEOOKI	KING OFFICED - CONNENTED COMBINED												
ADDITIONAL	NRCs												
ADDITIONAL	MICOS												
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurri					0	0						
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou					14.64	14.64			19.99	19.99	19.99	
	F DX Subsequent Activity - Change/Realrange Multiline Hunt Glot					14.04	14.04			13.33	13.33	19.99	
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RUS - PRY)												
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)												
UNE Port/Loc	op Combination Rates	1			28.35								
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone				28.35 37.31								
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2			37.31								
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone												
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2			37.31								
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates	2	UEPPX	UEPLX	37.31								
UNE Port/Loo	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2 3	UEPPX UEPPX	UEPLX UEPLX	37.31 56.24								
UNE Port/Loo	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volice Grade Loop (SL1) - Zone	3			37.31 56.24 14.35								
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX	UEPLX	37.31 56.24 14.35 23.31								
UNE Port/Loo UNE Loop Ra	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone (Grade Line Port Rates (BUS - PBX)	1 2	UEPPX UEPPX	UEPLX UEPLX	37.31 56.24 14.35 23.31 42.24								
UNE Port/Loc  UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 1-Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt	1 2	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC	37.31 56.24 14.35 23.31 42.24	90	90			40.71	9.58		
UNE Loop Ra	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  - Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt Line Side Unbundled Outward PBX Trunk Port - Bt	1 2	UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO	37.31 56.24 14.35 23.31 42.24	90	90			40.71	9.58		
UNE Loop Ra	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 1-Corade Line Port Rates (BUS - PBX) 1-De Side Unbundled Combination 2-Way PBX Trunk Port - Bu 1-De Side Unbundled Combination PBX Trunk Port - Bu 1-De Side Unbundled Combination PBX Trunk Port - Bu 1-De Side Unbundled Incoming PBX Trunk Port - Bu 1-De Side Unbundled Incoming PBX Trunk Port - Bu	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX  UEPPC UEPPO UEPP1	37.31 56.24 14.35 23.31 42.24 14 14	90 90	90 90			40.71 40.71	9.58 9.58		
UNE Loop Ra  UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX  UEPPC UEPPO UEPP1 UEPA2	37.31 56.24 14.35 23.31 42.24 14 14 14	90 90 90	90 90 90			40.71 40.71 40.71	9.58 9.58 9.58		
UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPA2 UEPLD	37.31 56.24 14.35 23.31 42.24 14 14 14 14	90 90 90 90	90 90 90 90			40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58		
UNE Loop Re	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt Line Side Unbundled Coutward PBX Trunk Port - Bt Line Side Unbundled Incoming PBX Trunk Port - Bt Line Side Unbundled Incoming PBX Trunk Port - Bt 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PSX LD Terminal Port 2-Wire Voice Unbundled PSX LD Terminal Port	1 2	UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPA2 UEPLD UEPXA	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14	90 90 90 90 90	90 90 90 90 90			40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58		
UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  - Grade Line Port Rates (BUS - PBX) - Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu - Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu - Line Side Unbundled Owward PBX Trunk Port - Bu - Line Side Unbundled C-Way Combination PBX Alabama Calling Pt - 2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc - 2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc - 2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc - 2-Wire Voice Unbundled PBX Toll Terminal Hotel Por	1 2	UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPA2 UEPLD UEPXA UEPXB	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14	90 90 90 90 90 90	90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58		
UNE Loop Re	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  - Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt Line Side Unbundled Dutward PBX Trunk Port - Bt Line Side Unbundled Incoming PBX Trunk Port - Bt Line Side Unbundled Incoming PBX Trunk Port - Bt Line Side Unbundled PMX Trunk Port - Bt Line Side Unbundled PMX Trunk Port - Bt 2-Wire Voice Unbundled PMX LD Terminal Port 2-Wire Voice Unbundled PSX Toll Terminal Port 2-Wire Voice Unbundled PSX Toll Terminal Hotel Por 2-Wire Voice Unbundled PSX Toll Terminals Por	1 2	UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPO UEPPO UEPP1 UEPA2 UEPLD UEPXA UEPXB UEPXC	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14	90 90 90 90 90 90 90	90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPA2 UEPLD UEPXA UEPXA UEPXB UEPXC UEPXD	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90	90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  - Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt Line Side Unbundled Dutward PBX Trunk Port - Bt Line Side Unbundled Incoming PBX Trunk Port - Bt Line Side Unbundled Incoming PBX Trunk Port - Bt Line Side Unbundled PMX Trunk Port - Bt Line Side Unbundled PMX Trunk Port - Bt 2-Wire Voice Unbundled PMX LD Terminal Port 2-Wire Voice Unbundled PSX Toll Terminal Port 2-Wire Voice Unbundled PSX Toll Terminal Hotel Por 2-Wire Voice Unbundled PSX Toll Terminals Por	1 2	UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPO UEPPO UEPP1 UEPA2 UEPLD UEPXA UEPXB UEPXC	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14	90 90 90 90 90 90 90	90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc  UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 1-Grade Line Port Rates (BUS - PBX) 1-Grade Line Port Rates (BUS - PBX) 1-Grade Line Port Rates (BUS - PBX) 1-Grade Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu 1-Ine Side Unbundled Combination 2-Way PBX Trunk Port - Bu 1-Ine Side Unbundled Combination PBX Trunk Port - Bu 1-Ine Side Unbundled Combination PBX Port - Bu 1-Ine Side Unbundled PBX LOT Trunk Port - Bu 2-Wire Voice Unbundled PBX LO DD Terminal Port 2-Wire Voice Unbundled PBX LD DD Terminal For 2-Wire Voice Unbundled PBX LD DD Terminal Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPP1 UEPA2 UEPLD UEPXA UEPXA UEPXA UEPXC UEPXC UEPXD	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPO UEPPO UEPPO UEPPA2 UEPLO UEPXA UEPXB UEPXC UEPXB UEPXC UEPXD UEPXE	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 1-Grade Line Port Rates (BUS - PBX) 1-Grade Line Port Rates (BUS - PBX) 1-Grade Line Port Rates (BUS - PBX) 1-Grade Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu 1-Ine Side Unbundled Combination 2-Way PBX Trunk Port - Bu 1-Ine Side Unbundled Combination PBX Trunk Port - Bu 1-Ine Side Unbundled Combination PBX Port - Bu 1-Ine Side Unbundled PBX LOT Trunk Port - Bu 2-Wire Voice Unbundled PBX LO DD Terminal Port 2-Wire Voice Unbundled PBX LD DD Terminal For 2-Wire Voice Unbundled PBX LD DD Terminal Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPP1 UEPA2 UEPLD UEPXA UEPXA UEPXA UEPXC UEPXC UEPXD	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPC UEPPO UEPP1 UEPA2 UEPLD UEPXA UEPXB UEPXC UEPXC UEPXC UEPXC UEPXC UEPXC	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPO UEPPO UEPPO UEPA2 UEPA2 UEPA3 UEPX6 UEPXC UEPXB UEPXB UEPXC UEPXE UEPXL UEPXL UEPXL UEPXL UEPXL UEPXL	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPC UEPPO UEPP1 UEPA2 UEPLD UEPXA UEPXB UEPXC UEPXC UEPXC UEPXC UEPXC UEPXC	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPO UEPPO UEPPO UEPA2 UEPA2 UEPA3 UEPX6 UEPXC UEPXB UEPXB UEPXC UEPXE UEPXL UEPXL UEPXL UEPXL UEPXL UEPXL	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc  UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPC UEPPO UEPPO UEPPI UEPAZ UEPLD UEPXA UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc  UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPO UEPPO UEPPO UEPA2 UEPA2 UEPA3 UEPX6 UEPXC UEPXB UEPXB UEPXC UEPXE UEPXL UEPXL UEPXL UEPXL UEPXL UEPXL	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc  UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPC UEPPO UEPPO UEPPI UEPAZ UEPLD UEPXA UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc  UNE Loop Ra  2-Wire Voice	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPC UEPPO UEPPO UEPPI UEPAZ UEPLD UEPXA UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc  UNE Loop Re  2-Wire Voice  LOCAL NUM  FEATURES	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone ates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPC UEPPO UEPPO UEPPI UEPAZ UEPLD UEPXA UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		
UNE Port/Loc  UNE Loop Re  2-Wire Voice  LOCAL NUM  FEATURES	op Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPPC UEPPC UEPPC UEPPO UEPPO UEPPI UEPAZ UEPLD UEPXA UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB UEPXC UEPXB	37.31 56.24 14.35 23.31 42.24 14 14 14 14 14 14 14 14 14 14 14 14 14	90 90 90 90 90 90 90 90 90 90 90	90 90 90 90 90 90 90 90 90 90 90			40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71 40.71	9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58 9.58		

2-Wire Voice Grade Loop/ Line Port Combination - Subsequε	UEPPX	USAS2		0	0				
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurri				0	0				
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou				14.64	14.64	19.99	19.99	19.99	
The state of the s					11121				
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT									
UNE Port/Loop Combination Rates									
2-Wire VG Coin Port/Loop Combo – Zone			28.35						
2-Wire VG Coin Port/Loop Combo – Zone			37.31						
2-Wire VG Coin Port/Loop Combo – Zone			56.24						
UNE Loop Rates									
	LIEDGO	LIEDLY	44.05						
2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	UEPCO UEPCO	UEPLX	14.35 23.31		<del>                                     </del>				
2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	42.24						
2-Wile Voice Grade Loop (SLT) - Zorie	UEFCO	UEFLA	42.24						
2-Wire Voice Grade Line Port Rates (Coin)									
2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)	UEPCO	UEPRF			90	40.71	9.58		
2-Wire Coin 2-Way with Operator Screening (AL, K'	UEPCO	UEPRE	14	90	90	40.71	9.58		
2-Wire Coin 2-Way with Operator Screening (AL, K 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K	UEFCO	UEFRE	14	90	90	40.71	9.56		
LA, MS, SC)	UEPCO	UEPRA	14	90	90	40.71	9.58		
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, M	UEPCO	UEPRB	14	90	90	40.71	9.58		
2-Wire Coin 2-Way with Operator Screening and 611 Blocking (AL, EA, WI	OLFCO	OLFIND	14	30	90	40.71	3.30		
(AL. KY. LA. MS)	UEPCO	UEPCD	14	90	90	40.71	9.58		
2-Wire Coin Outward with Operator Screening and 011 Blocking (AL, F	UEPCO	UEPRK	14	90	90	40.71	9.58		
2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL,	3L1 00	JEI KK				40.71	0.00		
KY, LA, MS)	UEPCO	UEPRH	14	90	90	40.71	9.58		
2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local									
(AL, KY, LA, MS)	UEPCO	UEPCN	14	90	90	40.71	9.58		
LOCAL NUMBER PORTABILITY									
Local Number Portability (1 per por	UEPCO	LNPCX	0.35						
NONRECURRING CHARGES - CURRENTLY COMBINED									
ADDITIONAL NRCs					1				
2-Wire Voice Grade Loop/ Line Port Combination - Subseque	UEPCO	USAS2		0	0				

Page 20 of 20 Version 2Q01: 08/30/01

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
								Nonrec	curring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual St Order vs Electronic Disc Add
													Dis	sconnect			
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		shown in the sections for stand-alone loops or loops as part of a combination refers to nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm	Geograp	ohical	y Deaveraged UNE	Zones.	To view Geogr	aphically Deaver	aged UNE Zone	Designation	ns by Central	Office, refer	to Internet	Website:		T	
BUNDLE	D EXCHANG	E ACCESS LOOP															
	2-WIRE AN	ALOG VOICE GRADE LOOP															
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone		1	UEANL	UEAL2	11.74	44.68	20.57	23.1	5.92		10.73			1.65	
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone		2	UEANL	UEAL2	16.26	44.68	20.57	23.1	5.92		10.73			1.65	
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone		3	UEANL	UEAL2	30.75	44.68	20.57	23.1	5.92		10.73			1.65	
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone		1	UEPSR, UEPSB		11.74	44.68	20.57	23.1	5.92		10.73			1.65	
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zoni		2	UEPSR, UEPSB		16.26	44.68	20.57	23.1	5.92		10.73			1.00	
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zoni		3		UEALS	30.75	44.68	20.57	23.1	5.92	+	10.73	1	1	1.65	
		Engineering Information Document (E	<del>                                     </del>	J	UEANL	ULALO	50.75	28.77	28.77	20.1	0.32	1	10.10	1	<b> </b>	1.00	l
		Engineering memberior bootinent (E			OLANIE			20.11	20.77	1	1	1		1	1	<b> </b>	
		Manual Order Coordination for UVL-SL1s (per loop			UEANL	UEAMC		8.12	8.12								
	<del> </del>	manual order coordination for ove-seria (per loop	<b></b>		OLAINL	JEANIC		0.12	0.12	+	<b>†</b>	+		+	1		<b> </b>
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR			UEANL	OCOSL		20.75	20.75								
		Order Coordination for Specified Conversion Time for UVL-SLT (per LSR			UEANL	UCUSL		20.75	20.75								
		0.000															
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling	-														
		Zone 1		1	UEA	UEAL2	13.43	122.38	74.35	57.28	10.83		10.73			1.65	
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling	-														
		Zone 2		2	UEA	UEAL2	18.6	122.38	74.35	57.28	10.83		10.73			1.65	
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling	-														
		Zone 3		3	UEA	UEAL2	35.18	122.38	74.35	57.28	10.83		10.73			1.65	
		Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL		20.75									
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo	ne														
		1		1	UEA	UEAR2	13.43	122.38	74.35	57.28	10.83		10.73			1.65	
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo	ne			0 = 1 10 1 =			1								
		2		2	UEA	UEAR2	18.6	122.38	74.35	57.28	10.83		10.73			1.65	
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo	00	-	OLA	OLMINE	10.0	122.00	14.00	01.20	10.00		10.70			1.00	
		2-Wile Allalog Voice Grade Loop - Service Level 2 Wileverse Ballery Signaling - 20	iie	3	UEA	UEAR2	35.18	122.38	74.35	57.28	10.83		10.73			1.65	
				3	ULA	ULAINZ	33.10	122.30	74.33	37.20	10.03		10.73			1.03	
		Order Consideration for Consideral Conversion Time (con I C			UEA	OCOSL		20.75									
	4 14/105 441	Order Coordination for Specified Conversion Time (per LS			UEA	UCUSL		20.75									
	4-WIRE AN	ALOG VOICE GRADE LOOP															
		4-Wire Analog Voice Grade Loop - Zone		1	UEA	UEAL4	21.23	151.34	103.82	60.47	14.02		10.73			1.65	
		4-Wire Analog Voice Grade Loop - Zone		2	UEA	UEAL4	29.41	151.34	103.82	60.47	14.02		10.73			1.65	
		4-Wire Analog Voice Grade Loop - Zone		3	UEA	UEAL4	55.63	151.34	103.82	60.47	14.02		10.73			1.65	
		Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL		20.75									
	2-WIRE ISD	N DIGITAL GRADE LOOP															
		2-Wire ISDN Digital Grade Loop - Zone		1	UDN	U1L2X	20.44	133.15	85.12	56.1	9.65		10.73			1.65	
		2-Wire ISDN Digital Grade Loop - Zone		2	UDN	U1L2X	28.31	133.15	85.12	56.1	9.65	1	10.73	1	1	1.65	
		2-Wire ISDN Digital Grade Loop - Zone		3	UDN	U1L2X	53.56	133.15	85.12	56.1	9.65		10.73			1.65	
		Order Coordination For Specified Conversion Time (per LS			UDN	OCOSL		20.75									
					<u> </u>												
	2-WIRE Uni	versal Digital Channel (UDC) COMPATIBLE LOOP								İ	İ	1			İ		
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		1	UDC	UDC2X	20.44	133.15	85.12	56.1	9.65	1	10.73	1	1	1.65	l
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		2	UDC	UDC2X	28.31	133.15	85.12	56.1	9.65	1	10.73	1	1	1.65	l
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		3	UDC	UDC2X	53.56	133.15	85.12	56.1	9.65	1	10.73	1	1	1.65	l
	<del> </del>	2 This Shire sair Digital Channel (ODO) Compatible Loop - Zone	<b></b>	J	000	JUJEN	55.50	100.10	00.12	JU. 1	5.05	+	10.10	+	1	1.00	<b> </b>
	2 WIDE AC	YMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP					<b> </b>			+	1	+	<b> </b>	+	1		-
	Z-WIKE AS	I MINIE I MOAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOUP				_				+	<b>+</b>	+	-	1	1		-
		A MIDE ADVANCEDION DIGITAL OUDOODIDED LINE (ADOL) CONTRIBUTE STORE												1			1
	1	2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOO								-	1	1	1	1	1	-	<b> </b>
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -								1	1	1		1			l
		Zone 1		1	UAL	UAL2X	11.52	134.8	93.62	67.66	14.09	1	10.73	1		1.65	ļ
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -					1			1	1	1		1			l
		Zone 2		2	UAL	UAL2X	15.96	134.8	93.62	67.66	14.09	1	10.73			1.65	
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -	l T	Ţ		1 7	1			1	1	1			1		1
		Zone 3		3	UAL	UAL2X	30.19	134.8	93.62	67.66	14.09	1	10.73	1	<u> </u>	1.65	
	1											1	1				
		Order Coordination for Specified Conversion Time (per LS			UAL	OCOSL	1	20.75	1	1	1	1	0	1	1	i .	i

				,				1				
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1	1	UAL	UAL2W	11.52	112.55	64.12	54.67	8.22		10.73	1.65
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2	2	UAL	UAL2W	15.96	112.55	64.12	54.67	8.22		10.73	1.65
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3	3	UAL	UAL2W		112.55	64.12	54.67	8.22		10.73	1.65
		3			30.19		04.12	34.07	0.22		10.73	1.03
	Order Coordination for Specified Conversion Time (per LS		UAL	OCOSL		20.75						
2-WIRE HIG	SH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP											
i	2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOO											
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 1	1	UHL	UHL2X	9.12	143.43	102.25	67.66	14.09		10.73	1.65
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 2	2	UHL	UHL2X		143.43	102.25	67.66	14.09		10.73	
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -											1.65
	Zone 3	3	UHL	UHL2X	23.9	143.43	102.25	67.66	14.09		10.73	1.65
I	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		20.75						
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation											
	Zone 1 2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation	1	UHL	UHL2W		121.17	72.75	54.67	8.22		10.73	1.65
	Zone 2  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation	2	UHL	UHL2W	12.63	121.17	72.75	54.67	8.22		10.73	1.65
	Zone 3	3	UHL	UHL2W	23.9	121.17	72.75	54.67	8.22		10.73	1.65
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		20.75						
1-WIDE UIO	SH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	+		1							-	
4-WINE HIG	4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -	+		1								
	Zone 1	1	UHL	UHL4X	14.24	174.28	125.3	69.56	11.37		10.73	1.65
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2	2	UHL	UHL4X	19.72	174.28	125.3	69.56	11.37		10.73	1.65
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 3	3	UHL	UHL4X	37.31	174.28	125.3	69.56	11.37		10.73	1.65
					07.01		120.0	03.00	11.07		10.70	1.00
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		20.75						
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1	1	UHL	UHL4W	14.24	152.02	104.11	56.57	10.12		10.73	1.65
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2	2	UHL	UHL4W	19.72	152.02	104.11	56.57	10.12		10.73	1.65
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3	3	UHL	UHL4W		152.02	104.11	56.57	10.12		10.73	1.65
		J			07.01		104.11	30.57	10.12		10.75	1.00
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		20.75						
	1 DIGITAL LOOP											
	4-Wire DS1 Digital Loop - Zone	1	USL	USLXX		282.15	163.51	47.4	10.22		10.73	1.65
	4-Wire DS1 Digital Loop - Zone :	2	USL	USLXX		282.15	163.51	47.4	10.22		10.73	1.65
	4-Wire DS1 Digital Loop - Zone :	3	USL	USLXX	181.38	282.15	163.51	47.4	10.22		10.73	1.65
	Order Coordination for Specified Conversion Time (per LS		USL	OCOSL		20.75						
4-WIRE 19.1	2, 56 OR 64 KBPS DIGITAL GRADE LOOP	+		1								
	4 Wire Unbundled Digital 19.2 Kbps	1	UDL	UDL19		145.66	98.14	60.47	14.02		10.73	1.65
	4 Wire Unbundled Digital 19.2 Kbps	2	UDL	UDL19		145.66	98.14	60.47	14.02		10.73	1.65
	4 Wire Unbundled Digital 19.2 Kbps	3	UDL	UDL19		145.66	98.14	60.47	14.02		10.73	1.65
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	1	UDL	UDL56		145.66	98.14	60.47	14.02		10.73	1.65
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	2	UDL	UDL56		145.66	98.14	60.47	14.02		10.73	1.65
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	3	UDL	UDL56	64.14	145.66	98.14	60.47	14.02		10.73	1.65
			LIE:	0000		00						
	Order Coordination for Specified Conversion Time (per LS	1	UDL	OCOSL	04.10	20.75	00.11	00 :-	44.00		10.73	
	4 Wire Unbundled Digital Loop 64 Kbps - Zone		UDL	UDL64		145.66	98.14	60.47	14.02			1.65
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	2	UDL	UDL64		145.66	98.14	60.47	14.02		10.73	1.65
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	3	UDL	UDL64	64.14	145.66	98.14	60.47	14.02		10.73	1.65
	Order Coordination for Specified Conversion Time (per LS		UDL	OCOSL		20.75						
L		1 -										
	hundled CORPER LOOP	+ + +						1	i e	1		1
2-WIRE Unb	bundled COPPER LOOP   2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility									İ		
2-WIRE Unb	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1	1	UCL	UCLPB	11.52	133.88	92.7	67.66	14.09		10.73	1.65
2-WIRE Unb	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility	1 2	UCL	UCLPB		133.88	92.7 92.7	67.66 67.66	14.09		10.73	1.65

			1	I		I	T				
	Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC		8.12	8.12				
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone '	1	UCL	UCLPW	11.52	111.62	63.19	54.67	8.22	10.73	1.65
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :	2	UCL	UCLPW		111.62	63.19	54.67	8.22	10.73	1.65
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone (	3	UCL	UCLPW		111.62	63.19	54.67	8.22	10.73	1.65
	Order Coordination for Unbundled Copper Loops (per loop)		UCL	UCLMC		8.12	8.12				
	2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone '	1	UCL	UCL2L	33.57	133.88	92.7	67.66	14.09	10.73	1.65
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2	2	UCL	UCL2L	46.5	133.88	92.7	67.66	14.09	10.73	1.65
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone (	3	UCL	UCL2L	87.96	133.88	92.7	67.66	14.09	10.73	1.65
	Order Coordination for Unbundled Copper Loops (per loop)		UCL	UCLMC		8.12	8.12				
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone '	1	UCL	UCL2W	33.57	111.62	63.19	54.67	8.22	10.73	1.65
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone (	2	UCL	UCL2W	46.5	111.62	63.19	54.67	8.22	10.73	1.65
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc	3	UCL	UCL2W UCLMC		111.62 8.12	63.19 8.12	54.67	8.22	10.73	1.65
	Order Coordination for Oribunided Copper Loops (per loc		UCL	JULIVIU		0.12	0.12				
	2-Wire Unbundled Copper Loop - Non-Designed Zone	1	UEQ	UEQ2X	11.01	44.69	22.4	25.65	7.06	10.73	1.65
	2 Wire Unbundled Copper Loop - Non-Designed - Zone	2	UEQ	UEQ2X		44.69	22.4	25.65	7.06	10.73	1.65
	2 Wire Unbundled Copper Loop - Non-Designed - Zone  Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loc	3	UEQ UEQ	UEQ2X USBMC	20.22	44.69 8.12	22.4 8.12	25.65	7.06	10.73	1.65
	Engineering Information Documer		UEQ	CODIVIO		28.77	28.77				
	Loop Testing - Basic 1st Half Hou		UEQ	URET1		78.92	78.92				
	Loop Testing - Basic Additional Half Hoι		UEQ	URETA		23.33	23.33				
4-WIRE COI	PPER LOOP										
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1	1	UCL	UCL4S	16.18	160.36	119.69	69.56	15.99	10.73	1.65
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 2	2	UCL	UCL4S	22.41	160.36	119.69	69.56	15.99	10.73	1.65
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 3	3	UCL	UCL4S	42.39	160.36	119.69	69.56	15.99	10.73	1.65
	Order Coordination for Unbundled Copper Loops (per loc  4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -		UCL	UCLMC		8.12	8.12				
	Zone 1  4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -	1	UCL	UCL4W	16.18	138.1	90.19	56.57	10.12	10.73	1.65
	Zone 2  4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -	2	UCL	UCL4W	22.41	138.1	90.19	56.57	10.12	10.73	1.65
	Zone 3 Order Coordination for Unbundled Copper Loops (per loc	3	UCL	UCL4W UCLMC	42.39	138.1 8.12	90.19 8.12	56.57	10.12	10.73	1.65
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone '	1	UCL	UCL4L	57.88	160.36	119.69	69.56	15.99	10.73	1.65
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	2	UCL	UCL4L	80.18	160.36	119.69	69.56	15.99	10.73	1.65
	4-wire Unburided Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc	3	UCL UCL	UCL4L UCLMC	151.67	160.36 8.12	119.69 8.12	69.56	15.99	10.73	1.65
	A-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '	1	UCL	UCL40		138.1	90.19	56.57	10.12	10.73	4.05
	reservation - Zone	2	UCL	UCL40		138.1	90.19	56.57	10.12	10.73	1.65
	#-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone \$	3	UCL	UCL40		138.1	90.19	56.57	10.12	10.73	1.65
	Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC		8.12	8.12	33.37	10.12		1.00
MODIFICATION											
NODIFICATION	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to		UAL, UHL, UCL, UEQ, ULS	ULM2L	_0	0	0	0	0		
	18k ft			1		309.32	309.32				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18		UCL, ULS	ULM2G		309.32	309.32				
			UCL, ULS UHL, UCL	ULM2G		0	0				
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18		UCL, ULS								

	<del> </del>	_						+ +		-		+ + + + + + + + + + + + + + + + + + + +
Sub-Loop Distribution												
	Per Cross Box Location - CLEC Feeder Facility Set-L		LIFANI	LICDCA		407.00	467.08				40.70	4.05
	Per Cross Box Location - CLEC Feeder Facility Set-L I Per Cross Box Location - Per 25 Pair Panel Set-L I			USBSA USBSB		467.08 11.27	11.27				10.73 10.73	1.65 1.65
Sub-Loop - I	Per Closs Box Location - Per 25 Pair Patier Set-C		UEAINL	USBSB		11.21	11.21				10.73	1.65
Sub-Loop - I	Per Building Equipment Room - CLEC Feeder Facility Set-I		UEANL	USBSC		152.58	152.58				10.73	1.65
Sub-Loop - I	rei Bullullig Equipment Room - CLEC reedel Facility Set-t		UEAINL	USBSC		132.36	132.30				10.73	1.65
Sub-Loop -	Per Building Equipment Room - Per 25 Pair Panel Set-l		UEANL	USBSD		43.54	43.54				10.73	1.65
	Distribution Per 2-Wire Analog Voice Grade Loop - Zone	1		USBN2	6.9	54.26	19.64	37.03	4.1		10.73	1.65
	Distribution Per 2-Wire Analog Voice Grade Loop - Zone	2		USBN2	9.56	54.26	19.64	37.03	4.1		10.73	1.65
	Distribution Per 2-Wire Analog Voice Grade Loop - Zone	3		USBN2	18.08	54.26	19.64	37.03	4.1		10.73	1.65
	dination for Unbundled Sub-Loops, per sub-loop pair	_		USBMC	10.00	8.12	8.12	07.00			10.70	1.00
	Distribution Per 4-Wire Analog Voice Grade Loop - Zon€	1		USBN4	7.35	62.05	27.42	37.98	5.05		10.73	1.65
	Distribution Per 4-Wire Analog Voice Grade Loop - Zone	2		USBN4	10.18	62.05	27.42	37.98	5.05		10.73	1.65
	Distribution Per 4-Wire Analog Voice Grade Loop - Zone	3		USBN4	19.25	62.05	27.42	37.98	5.05		10.73	1.65
	dination for Unbundled Sub-Loops, per sub-loop pair	Ť		USBMC		8.12	8.12	00				
	-Wire Intrabuilding Network Cable (INC			USBR2	3.33	46.74	12.11	37.03	4.1		10.73	1.65
Order Coord	dination for Unbundled Sub-Loops, per sub-loop pair			USBMC	0.00	8.12	8.12	07.00			10.70	1.00
	-Wire Intrabuilding Network Cable (INC			USBR4	6.32	50.41	15.78	37.98	5.05		10.73	1.65
	dination for Unbundled Sub-Loops, per sub-loop pair			USBMC		8.12	8.12	00				
	per Unbundled Sub-Loop Distribution - Zone	1		UCS2X	5.66	54.26	19.64	37.03	4.1		10.73	1.65
	per Unbundled Sub-Loop Distribution - Zone			UCS2X	7.83	54.26	19.64	37.03	4.1		10.73	1.65
	per Unbundled Sub-Loop Distribution - Zone			UCS2X	14.82	54.26	19.64	37.03	4.1		10.73	1.65
	dination for Unbundled Sub-Loops, per sub-loop pair	T -		USBMC		23.24	23.24					
	per Unbundled Sub-Loop Distribution - Zone	1		UCS4X	4.72	62.05	27.42	37.98	5.05		10.73	1.65
	per Unbundled Sub-Loop Distribution - Zone			UCS4X	6.53	62.05	27.42	37.98	5.05		10.73	1.65
	per Unbundled Sub-Loop Distribution - Zone			UCS4X	12.36	62.05	27.42	37.98	5.05		10.73	1.65
Order Coord	dination for Unbundled Sub-Loops, per sub-loop pair			USBMC		8.12	8.12					
Sub-Loop Feeder												
			UEA,									
			UDN,UCL,UDL,UD									
USL-Feeder	r, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-		С	USBFW		467.08						
	, ,		UEA,									
			UDN,UCL,UDL,UD									
USL Feeder	r - DS0 Set-up per Cross Box location - per 25 pair set-u		С	USBFX		11.27	11.27					
USL Feeder	r DS1 Set-up at DSX location, per DS1 terminatio		USL	USBFZ		522.41	11.32					
	Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zoni	1		USBFA	7.6	83.62	46.2	45.57	10.19		10.73	1.65
	Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zoni	2		USBFA	10.53	83.62	46.2	45.57	10.19		10.73	1.65
	Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zon	3		USBFA	19.92	83.62	46.2	45.57	10.19		10.73	1.65
Order Coord	dination for Specified Conversion Time, per LSR		UEA	OCOSL		20.75						
Unbundide S	Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni	1	UEA	USBFB	7.6	83.62	46.2	45.57	10.19		10.73	1.65
Unbundled S	Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni	2	UEA	USBFB	10.53	83.62	46.2	45.57	10.19		10.73	1.65
Unbundled \$	Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zoni	3	UEA	USBFB	19.92	83.62	46.2	45.57	10.19		10.73	1.65
Order Coord	dination for Specified Time Conversion, per LSR		UEA	OCOSL		20.75						
Unbundled S	Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon	1	UEA	USBFC	7.6	83.62	46.2	45.57	10.19		10.73	1.65
	Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon	2		USBFC	10.53	83.62	46.2	45.57	10.19		10.73	1.65
Unbundled S	Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zone											
3		3	UEA	USBFC	19.92	83.62	46.2	45.57	10.19		10.73	1.65
	dination For Specified Conversion Time, per LS		UEA	OCOSL		20.75						
	Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zoni	1		USBFD	16.05	96.4	58.12	48.55	11.33		10.73	1.65
	Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone	2		USBFD	22.23	96.4	58.12	48.55	11.33		10.73	1.65
Unbundled S	Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zoni	3	UEA	USBFD	42.06	96.4	58.12	48.55	11.33		10.73	1.65
	dination For Specified Conversion Time, Per LS			OCOSL		20.75						
	Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	1		USBFE	16.05	96.4	58.12	48.55	11.33		10.73	1.65
	Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	2		USBFE	22.23	96.4	58.12	48.55	11.33		10.73	1.65
Unbundled S	Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	3	UEA	USBFE	42.06	96.4	58.12	48.55	11.33		10.73	1.65
		1					1					
Order Coord	dination For Specified Conversion Time, Per LS			OCOSL		20.75						
Unbundled S	Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone	1		USBFF	16.18	98.91	60.12	46.95	9.74		10.73	1.65
Unbundled S	Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	2		USBFF	22.41	98.91	60.12	46.95	9.74	The state of the s	10.73	1.65
Unbundled S	Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	3	UDN	USBFF	42.39	98.91	60.12	46.95	9.74		10.73	1.65
	dination For Specified Conversion Time, Per LS		UDN	OCOSL		20.75	1					
	Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	1		USBFS	16.18	98.91	60.12	46.95	9.74		10.73	1.65
	Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	2		USBFS	22.41	98.91	60.12	46.95	9.74	The state of the s	10.73	1.65
	Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	3		USBFS	42.39	98.91	60.12	46.95	9.74		10.73	1.65
	Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	1		USBFG	43.64	120.61	70.34	65.07	16.2		10.73	1.65
	Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	2	USL	USBFG	60.45	120.61	70.34	65.07	16.2		10.73	1.65
Unbundled S	Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	3	USL	USBFG	114.36	120.61	70.34	65.07	16.2		10.73	1.65
	dination For Specified Conversion Time, Per LS		USL	OCOSL		20.75						
	Sub-Loop Feeder, 2-Wire Copper Loop - Zone	1		USBFH	6.65	76.87	38.08	45.64	8.43		10.73	1.65
	Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	2		USBFH	9.22	76.87	38.08	45.64	8.43		10.73	1.65
	Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	3	UCL	USBFH	17.44	76.87	38.08	45.64	8.43		10.73	1.65

		i.	,				,					
Order Coordination For Specified Conversion Time, per LS		UCL	OCOSL		20.75							
Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	1	UCL	USBFJ	12.76	89.85	51.57	46.59	9.38	10.73		1.65	
Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	2	UCL	USBFJ	17.67	89.85	51.57	46.59	9.38	10.73		1.65	
Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	3	UCL	USBFJ	33.43	89.85	51.57	46.59	9.38	10.73		1.65	
Order Coordination For Specified Conversion Time, per LS		UCL	OCOSL		20.75							
Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	1	UDL	USBFN	17.52	90.72	52.43	48.55	11.33	10.73		1.65	
Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	3	UDL UDL	USBFN	24.28 45.92	90.72 90.72	52.43 52.43	48.55 48.55	11.33 11.33	10.73 10.73		1.65 1.65	
Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loor Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	1	UDL	USBFO	17.52	90.72	52.43	48.55	11.33	10.73		1.65	
Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	2	UDL	USBFO	24.28	90.72	52.43	48.55	11.33	10.73		1.65	
Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	3	UDL	USBFO	45.92	90.72	52.43	48.55	11.33	10.73		1.65	
Order Coordination For Specified Time Conversion, per LS		UDL	OCOSL		20.75							
Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1	UDL	USBFP	17.52	90.72	52.43	48.55	11.33	10.73		1.65	
Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	2	UDL	USBFP	24.28	90.72	52.43	48.55	11.33	10.73		1.65	
Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	3	UDL	USBFP	45.92	90.72	52.43	48.55	11.33	10.73		1.65	
Order Coordination For Specified Conversion Time, per LS		UDL	OCOSL		20.75							
Order Coordination For Specified Conversion Time, per LS		UDL	UCUSL		20.75							
Unbundled Sub-Loop Modification												
Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W												
PR		UEF	ULM2X		9.11	9.11			10.73		1.65	
Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W					-							
PR		UEF	ULM4X		9.11	9.11			10.73		1.65	
Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR												
unloaded		UEF	ULM4T		14.05	14.05			10.73		1.65	
Unbundled Network Terminating Wire (UNTW)												
Unbundled Network Terminating Wire (UNTW) per Pa		UENTW	UENPP	0.3682	21.85	21.85			10.73		1.65	
Network Interface Device (NID)												
Network Interface Device (NID) - 1-2 line		UENTW	UND12		63.72	40.94			10.73		1.65	
Network Interface Device (NID) - 1-2 line  Network Interface Device (NID) - 1-6 line		UENTW	UND12		105.96	83.17			10.73		1.65	
Network Interface Device Cross Connect - 2 V		UENTW	UNDC2		7.12	7.12			10.73		1.65	
Network Interface Device Cross Connect - 4V		UENTW	UNDC4		7.12	7.12			10.73		1.65	
UNBUNDLED LOOP CONCENTRATION												
Unbundled Loop Concentration - System A (TR00)		ULC	UCT8A	461.86	324.01	324.01			10.73		1.65	
Unbundled Loop Concentration - System B (TR00)		ULC	UCT8B	54.91	135	135			10.73		1.65	
Unbundled Loop Concentration - System A (TR30:		ULC	UCT3A	500.74	324.01	324.01			10.73		1.65	
Unbundled Loop Concentration - System B (TR30)		ULC	UCT3B	92.53	135	135			10.73		1.65	
Unbundled Loop Concentration - DS1 Loop Interface Ca		ULC	истсо	5.18	64.65	46.45	16.67	4.35	10.73		1.65	
Unbundled Loop Concentration - DST Loop Interface Ca		UDN	ULCC1	8.22	14.96	14.88	6.11	6.07	10.73		1.65	
Unbundled Loop Concentration - ISBN Loop Interface (Brite Cal		UDC	ULCCU	8.22	14.96	14.88	6.11	6.07	10.73		1.65	
Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop		ODO	OLOGO	0.22	14.50	14.00	0.11	0.07	10.70		1.00	
Interface (POTS Card		UEA	ULCC2	2.06	14.96	14.88	6.11	6.07	10.73		1.65	
Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface												
(SPOTS Card)		UEA	ULCCR	12.22	14.96	14.88	6.11	6.07	10.73		19.99	
Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca		UEA	ULCC4	7.29	14.96	14.89	6.11	6.07	10.73		1.65	
Unbundled Loop Concentration - TEST CIRCUIT Cai		ULC	UCTTC	35.63	14.96	14.88	6.11	6.07	10.73		1.65	
Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa		UDL	ULCC7	10.8	14.96	14.88	6.11	6.07	10.73		1.65	
Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa	+	UDL UDL	ULCC5 ULCC6	10.8 10.8	14.96 14.96	14.88 14.88	6.11	6.07 6.07	10.73 10.73		1.65 1.65	
onbundied Loop Concentration - Digital 64 Kbps Data Loop Interia		UDL	ULUUD	10.8	14.90	14.00	0.11	0.07	10.73	+	1.00	
						+				+	+	
UNBUNDLED SUB-LOOP CONCENTRATION (OUTSIDE CO)												
UNE OTHER, PROVISIONING ONLY - NO RATE						1						
NID - Dispatch and Service Order for NID installation		UENTW	UNDBX			1						
		UENTW	UENCE									
UNTW Circuit Id Establishment, Provisioning Only - No Rate			1									
		UEANL,UEF,UEQ,	LINIEGO					1				
UNTW Circuit Id Establishment, Provisioning Only - No Rate Unbundled Contract Name, Provisioning Only - No Rate		UENTW	UNECN									
		UENTW UAL,UCL,UDC,UDL										
Unbundled Contract Name, Provisioning Only - No Rate		UENTW UAL,UCL,UDC,UDL ,UDN,UEA,UHL,UL										
		UENTW UAL,UCL,UDC,UDL		0	0							
Unbundled Contract Name, Provisioning Only - No Rate		UENTW UAL,UCL,UDC,UDL ,UDN,UEA,UHL,UL C	UNECN	0	0							
Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate		UENTW UAL,UCL,UDC,UDL ,UDN,UEA,UHL,UL	UNECN	0	0							
Unbundled Contract Name, Provisioning Only - No Rate		UENTW UAL,UCL,UDC,UDL ,UDN,UEA,UHL,UL C UEA,UDN,UCL,UD	UNECN									
Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra		UENTW UAL,UCL,UDC,UDL ,UDN,UEA,UHL,UL C UEA,UDN,UCL,UD	UNECN									
Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate		UENTW UAL,UCL,UDC,UDL ,UDN,UEA,UHL,UL C UEA,UDN,UCL,UD C	UNECN	0	0							

		Unbundled DS1 Loop - Expanded Superframe Format option - no ra		USL	CCOEF	0	0							
IGH CAPA	CITY UNBUN	DLED LOCAL LOOP												
		onth minimum billing period												
		High Capacity Unbundled Local Loop - DS3 - Per Mile per mon		UE3	1L5ND	10.06								
		High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor			UE3PX	387.1	501.59	309.24	125.43	87.3		10.73		1.65
		High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon		UDLSX	1L5ND	10.06								1.00
		High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor		UDLSX	UDLS1	426.68	501.59	309.24	125.43	87.3		10.73		1.65
OOP MAK	F-I IP		-											
OUT WAR	L-01	Loop Makeup - Preordering Without Reservation, per working or spare facility gueried	-											
		(Manual).		UMK	UMKLW		43.1	43.1						
		Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).			UMKLP		45.72	45.72						
		Loop MakeupWith or Without Reservation, per working or spare facility queried												
		(Mechanized)		UMK	PSUMK		0.6757	0.6757						
NE SHARI	ING													
		Line Sharing Splitter, per System 96 Line Capaci			ULSDA	100	150	0	150	0		0		
		Line Sharing Splitter, per System 24 Line Capaci	[		ULSDB		150	0	150	0		0		
		Line Sharing Splitte, Per System, 8 Line Capaci			ULSD8		150	0	150	0		0		
	1	Line Sharing - per Line Activatio			ULSDC	0.61	40	22				10.73		1.65
	+	Line Sharing - per Subsequent Activity per Line Rearrangeme I		ULS	ULSDS		30	15				10.73		
	+													
	+		-+											
IBUNDI F	D TRANSPOR	RT												
	11													
	COMMON TO	RANSPORT (Shared)												
		Common Transport - Per Mile, Per MOl				0.0000039								
		Common Transport - Facilities Termination Per MO				0.0004579								
	NOTE: INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS3 = one m	onth	, DS3 and above for	ur month	\$								
		CE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE												
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per moi		U1TVX	1L5XX	0.0084								
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month		LIATIO	U1TV2	00.00	42.69	00.00	40.54	0.04		40.70		4.05
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per	-	U1TVX	U11V2	26.02	42.69	28.66	16.51	6.34		10.73		1.65
		month		U1TVX	1L5XX	0.0084								
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination		OTTVX	ILOXXX	0.0004								
		per month		U1TVX	U1TR2	26.02	42.69	28.66	16.51	6.34		10.73		1.65
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month	- 1	U1TVX	1L5XX	0.0084								
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination												
		per month	- 1	U1TVX	U1TV4	23.2	42.69	28.66	16.51	6.34		10.73		1.65
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per mon		U1TDX	1L5XX	0.0084								
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mor		U1TDX	U1TD5	18.95	42.69	28.66	16.51	6.34		10.73		1.65
	+	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per mon Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mor	_	U1TDX	1L5XX U1TD6	0.0084 18.95	42.69	20.66	16.51	6.34		10.73		1.65
	+	interoffice Chairner - Dedicated Transport - 64 KDps - Facility Termination per moi		U1TDX	01106	16.95	42.69	28.66	16.01	0.34		10.73		1.65
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS1										1		
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mor	-+	U1TD1	1L5XX	0.171								
		Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor		U1TD1	U1TF1	90.87	95.16	88.78	16.74	14.85		10.73		1.65
			†											
		CE CHANNEL - DEDICATED TRANSPORT- DS3												
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor		U1TD3	1L5XX	3.57	-							
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mor	]	U1TD3	U1TF3	1101	302.43	197.7	64.94	63.61		10.73		1.65
												1		
		CE CHANNEL - DEDICATED TRANSPORT- STS-1		IUTO:	41.500	0								
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon		U1TS1	1L5XX	3.57	200.40	407.7	04.04	00.01		40.70		1.05
	+	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor	-+	U1TS1	U1TFS	1085	302.43	197.7	64.94	63.61		10.73		1.65
	1											1		
	1		-+											
	LOCAL CHA	NNEL - DEDICATED TRANSPORT												
	NOTE: LOCA	AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month, DS3	3 and	d above=four month	S									
		Local Channel - Dedicated - 2-Wire Voice Grade per month - Zone	1	ULCVX	ULDV2	21.04	239.67	42.34	33.93	3.61		10.73		1.65
		Local Channel - Dedicated - 2-Wire Voice Grade per month - Zone	2	ULCVX	ULDV2	29.15	239.67	42.34	33.93	3.61		10.73	-	1.65
			3		ULDV2	55.14	239.67	42.34	33.93	3.61		10.73		1.65
		2000 Chamic Boalcated 2 Who Voice Grade Nov. Bat. 1 of month 2011	1		ULDR2	21.04	239.67	42.34	33.93	3.61		10.73		1.65
			2		ULDR2	29.15	239.67	42.34	33.93	3.61		10.73		1.65
			3		ULDR2	55.14 21.91	239.67	42.34 42.97	33.93 34.47	3.61 4.15	-	10.73 10.73		1.65
														1.65
			2		ULDV4		240.3 240.3	42.97	34.47	4.15		10.73		1.65

		Local Channel - Dedicated - 4-Wire Voice Grade per month - Zone	3	UNCVX	ULDV4	57.4	240.3	42.97	34.47	4.15		10.73	1.65
		Local Channel - Dedicated - DS1 per month - Zone	1	ULDD1	ULDF1	34.49	195.33	165.48	21.9	15.28		10.73	1.65
		Local Channel - Dedicated - DS1 per month - Zone	2	ULDD1	ULDF1	47.78	195.33	165.48	21.9	15.28		10.73	1.65
		Local Channel - Dedicated - DS1 per month - Zone	3	ULDD1	ULDF1	90.38	195.33	165.48	21.9	15.28		10.73	1.65
		Local Channel - Dedicated - DS3 - Per Mile per mon		ULDD3	1L5NC	7.83							
		Local Channel - Dedicated - DS3 - Facility Termination per mon		ULDD3	ULDF3	554.83	501.59	309.24	125.43	87.3		10.73	1.65
		Local Channel - Dedicated - STS-1- Per Mile per mon		ULDS1	1L5NC	7.83							
		Local Channel - Dedicated - STS-1 - Facility Termination per mon		ULDS1	ULDFS	563.73	501.59	309.24	125.43	87.3		10.73	1.65
		Econ chamber Boaloated Ore 1 1 downly reminiation per mon		OLDO!	020.0	000.70	001.00	000.21	120.10	07.0			1.00
MULTIPLEXE	RS												
MOLTIF LLXL	N.S	Channelization - DS1 to DS0 Channel Syster		UXTD1	MQ1	151.74	91.44	64.57	10	9.46		10.73	1.65
		OCULDE COCI (data) DC4 to DC0 Channel System	_		1D1DD	2.16			10	9.46		10.73	1.00
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb		UDL			9.08	6.38					
		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per mont		UDN	UC1CA	3.76	9.08	6.38					
		Voice Grade COCI - DS1 to DS0 Channel System - per mon		UEA	1D1VG	1.42	9.08	6.38					
		DS3 to DS1 Channel System per mont		UXTD3	MQ3	218.7	179.66	106.96	36.37	35.22		10.73	1.65
		STS1 to DS1 Channel System per mont		UXTS1	MQ3	218.7	179.66	106.96	36.37	35.22		10.73	1.65
		DS3 Interface Unit (DS1 COCI) used with Loop per mont		USL	UC1D1	14.24	9.08	6.38					
DARK FIBER													
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local											
		Channe		UDF	1L5DC	54.11							
		NRC Dark Fiber - Local Channe		UDF	UDFC4	04.11	677.34	174.79	277.72	179.41		10.73	1.65
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -		UDF	UDFC4		077.34	174.79	211.12	179.41		10.73	1.00
				LIDE	41.505	05.14							
		Interoffice Channe	+	UDF	1L5DF	25.14	077.04	474.70	077.76	470.44		10.70	105
		NRC Dark Fiber - Interoffice Channe	+	UDF	UDF14		677.34	174.79	277.72	179.41		10.73	1.65
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local			1								
		Loop	$\perp$	UDF	1L5DL	54.11							
		NRC Dark Fiber - Local Loop		UDF	UDFL4		677.34	174.79	277.72	179.41		10.73	1.65
TRANSPORT	OTHER		Т					L					
			┸										
C	Optional Fe	eatures & Functions:											
		Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chanı		UNC1X	CCOEF		184.92	23.82	2.07	0.8		10.73	1.65
		Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chan		UNC1X	CCOSF		184.92	23.82	2.07	0.8		10.73	1.65
WW 400500	TEN DIGIT	T SCREENING		UNCIX	CCUSF		184.92	23.82	2.07	0.8		10.73	1.00
BXX ACCESS	TEN DIGI												
		8XX Access Ten Digit Screening, Per Ca		OHD		0.0006165							
		8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv		OHD	N8R1X		3.74	0.64				10.73	1.65
		8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation		OHD			7.92	1.06	5.2	0.64		10.73	1.65
		8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation		OHD	N8FTX		7.92	1.06	5.2	0.64		10.73	1.65
		8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb		OHD	N8FCX		3.74	1.87				10.73	1.65
		8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested											
		Per 8XX No.		OHD	N8FMX		4.37	2.5				10.73	1.65
		8XX Access Ten Digit Screening, Change Charge Per Reque		OHD	N8FAX		4.37	0.64				10.73	1.65
		8XX Access Ten Digit Screening, Call Handling and Destination Featur		OHD	N8FDX		3.74	0.01				10.73	1.65
		8XX Access Ten Digit Screening, w/ 8XX No. Delivery, per que		OHD	1401 DX	0.0006165	0.14					10.70	1.00
		8XX Access Ten Digit Screening, w/ box No. Delivery, per que		OHD		0.0006165							
		one Access Terribigit Screening, W/ FOTS No. Delivery, per que		UND	-	0.0000100							
INE INCORE	ATION DA	TA BASE ACCESS (LIDB)			-								
LINE INFORM	IATION DA												
		LIDB Common Transport Per Quer	$\perp$	OQT	1	0.0000195							
		LIDB Validation Per Quer	$\perp$	OQU	1	0.0132254							
		LIDB Originating Point Code Establishment or Chang		OQT, OQU	NRPBX		49.71	49.71	49.71	49.71		10.73	1.65
SIGNALING (C	CCS7)		⊥_ ⊺										
		CCS7 Signaling Termination, Per STP Por		1DB	PT8SX	129.77						10.73	1.65
		CCS7 Signaling Usage, Per TCAP Messag		1DB		0.0000592							
		CCS7 Signaling Connection, Per link (A link		1DB	TPP++	18.39	39.28	39.28	16.51	16.51		10.73	1.65
		CCS7 Signaling Connection, Per link (B link) (also known as D lin		1DB	TPP++	18.39	39.28	39.28	16.51	16.51		10.73	1.65
		CCS7 Signaling Usage, Per ISUP Messag	1 1	1DB	1	0.0000148			. 5.0				
		CCS7 Signaling Usage Surrogate, per link per LAT	1 1	1DB	STU56							10.73	1.65
		CCS7 Signaling Osage Surrogate, per link per EXT  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per	+ +	.50	0.000	0.0.00					-		1.00
		STP affected		1DB	CCAPO		41.5	41.5				10.73	1.65
		CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per	+	וטט	COAFU		41.0	41.0				10.10	1.00
		Stp Affected		1DB	CCAPD		8	8				10.73	1.65
		SIP ATTECLEU	+	IDP	CCAPD		ō	0				10.13	1.00
044 OFB\"	_		+		+			1					
911 SERVIC	Ł		+		1								
								1					
		Local Channel - Dedicated - 2-wr Voice Grade - Zone				21.04	239.67	42.34	33.93	3.61		10.73	1.65
		Local Channel - Dedicated - 2-wr Voice Grade - Zone				29.15	239.67	42.34	33.93	3.61		10.73	1.65
		Local Channel - Dedicated - 2-wr Voice Grade - Zone				55.14	239.67	42.34	33.93	3.61		10.73	1.65
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mi				0.0084							
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Terminati				26.02	42.69	28.66	16.51	6.34		10.73	1.65
		Local Channel - Dedicated - DS1 - Zone				34.49	195.33	165.48	21.9	15.28		10.73	1.65
		Local Channel - Dedicated - DS1 - Zone	+		1	47.78	195.33	165.48	21.9	15.28		10.73	1.65
		Local Channel - Dedicated - DS1 - Zone  Local Channel - Dedicated - DS1 - Zone	+ +		+	90.38	195.33	165.48	21.9	15.28		10.73	1.65
-						30.30	133.33	100.40	41.9	10.20		10.70	1.00
						0.171						1	
		Interoffice Transport - Dedicated - DS1 Per Mil Interoffice Transport - Dedicated - DS1 Per Facility Termination				0.171 90.87	95.16	88.78	16.74	14.85		10.73	1.65

LLING NA	ME (CNAM) SERVICE						1				
	CNAM for DB Owners, Per Quen		OQV		0.0010161						
	CNAM for Non DB Owners, Per Query		OQV		0.0010161						
	CNAM For DB Owners - Service Establishmen		OQV	+	0.0010101	22.85	22.85	17.14	17.14	10.73	1.65
	ON WIT OF DE OWNERS OF VICE Establishmen					22.00	22.00	17.14	17.14	10.75	1.00
	CNAM For Non DB Owners - Service Establishmer		OQV			22.85	22.85	17.14	17.14	10.73	1.65
	CNAM For DB Owners - Service Provisioning With Point Code Establishme		OQV			1435	1061	317.7	233.6	10.73	1.65
	ON WIT OF BE OWNER OF TOTAL CHANGE THE COME COME COME.					1 100	1001	0	200.0	10.10	
	CNAM For Non Db Owners - Service Provisioning With Point Code Establishme		OQV			492.73	355.07	322.83	233.6	10.73	1.65
	CNAM (Non-Databs Owner), NRC, applies when using the Character Based User										1177
	Interface (CHUI)		OQV	CDDCH	ı	595	595			10.73	1.65
			7.7.							1311	- 1100
QUERY	SERVICE										
	LNP Charge Per query				0.000842						
	LNP Service Establishment Manua					12.46	12.46	9.35	9.35	10.73	1.65
	LNP Service Provisioning with Point Code Establishme					591.01	301.93	218.42	160.6	10.73	1.65
(	OPERATOR SERVICES AND DIRECTORY ASSISTANCE										
						-					
RATOR	CALL PROCESSING										
	Oper. Call Processing - Oper. Provided, Per Min Using BST LID				1.2						
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC				1.24	-					
	Oper. Call Processing - Fully Automated, per Call - Using BST LID				0.2						
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE				0.2						
						-					
ARD OPE	ERATOR SERVICES										
	Inward Operator Services - Verification, Per Ca				1						
	Inward Operator Services - Verification and Emergency Interrupt - Per C				1.95						
Anding -	OPERATOR CALL PROCESSING										
	Recording of Custom Branded OA Announcement			CBAOS	;	7000	7000			10.73	1.65
	Loading of Custom Branded OA Announcement per shelf/NAV			CBAOL		500	500			10.73	
ECTORY	ASSISTANCE SERVICES										
	DIRECTORY ASSISTANCE ACCESS SERVICE										
	Directory Assistance Access Service Calls, Charge Per Ca				0.275						
l l	DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)										
	Directory Assistance Call Completion Access Service (DACC), Per Call Attem				0.1						
	UNBRANDING										
	DIRECTORY TRANSPORT										
	Directory Transport - Local Channel DS				43.64	242.45	226.44			10.73	1.65
	Directory Transport - DS1 Level Interoffice Per Mi				0.6013						
	Directory Transport - DS1 Level Interoffice Per Facility Terminati				99.79	45.91	44.18			10.73	1.65
	Switched Common Transport Per DA Access Service Per Ca				0.0003						
	Switched Common Transport Per DA Access Service Per Call Per Mi				0.00001						
	Access Tandem Switching Per DA Access Service Per Ca				0.00055						
	Directory Transport - Installation NRC, Per Trunk or Signaling Connection					206.06	4.71			10.73	1.65
	DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)										
	Directory Assistance Data Base Service Charge Per Listir				0.04						
	Directory Assistance Data Base Service, per mont			DBSOF	150						
ANDING -	DIRECTORY ASSISTANCE										
	Custom Branding Announcement, per Recording to be used with the provision of DA		AMT	CBADA		3000	3000				
	Loading of Custom Branded Announcement per DRAM Card/Switch		AMT	CBADC	;	690	690				
						-					
ECTIVE I	ROUTING										
T											
	Selective Routing Per Unique Line Class Code Per Request Per Swit			USRCR		84.33	84.33	11.46	11.46	10.73	1.65
							1				
TUAL CO	LLOCATION										
			ueanl,uea,udn,udc								
	Virtual Collocation - 2-wire Cross Connects (loop		al,uhl,ucl,uec			33.86	31.95			10.73	1.65
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splittin	1	UEPSR, UEPSB			33.86	31.95			10.73	1.65
	Virtual Collocation - 2-wire Cross Connects (por			VE1R2		11.57	11.57			10.73	1.65
	Virtual Collocation - 4-wire Cross Connects (looş		uea,uhl,ucl,ud	UEAC4		33.99	32			10.73	1.65
	Virtual Collocation - 4-wire Cross Connects (por			VE1R4		11.57	11.57			10.73	1.65
	Virtual Collocatin - DS1 Cross Connect		USL,ULC,CLO	CNC1X	1.37	53.3	40.2				
CELECTI	VE CARRIER ROUTING										
SELECTI			CDC	LCDCEC	-1	191575	1	6974	1	10.73	1.65
SELECTI	Regional Service Establishment		SRC	SRCEC							

	Io NPO		000	1			Т			Т	Т	T	
	Query NRC, per query		SRC		0.0030998								
AIN - BELLSOUTH AIN S	SMS ACCESS SERVICE												
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			CAMSE		39.27	39.27	33.04	33.04		10.73		1.65
	AIN SMS Access Service - Port Connection - Dial/Shared Access			CAMDP	•	7.79	7.79	7.38	7.38		10.73		1.65
	AIN SMS Access Service - Port Connection - ISDN Access			CAM1P		7.79	7.79	7.38	7.38		10.73		1.65
	AIN SMS Access Service - User Identification Codes - Per User ID Code			CAMAU		34.85	34.85	21.97	21.97		10.73		1.65
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			CAMRC	0.000	73.76	73.76	9.51	9.51		10.73		1.65
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)  AIN SMS Access Service - Session, Per Minute				0.0029 0.7985								
	AIN SMS Access Service - Session, Per Minute  AIN SMS Access Service - Company Performed Session, Per Minute				0.7985								
	7 and Gwid Access delivine Company i entitlined dession, i el williade				0.4133								
AIN - BELLSOUTH AIN	TOOLKIT SERVICE												
	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			BAPSC		39.27	39.27	33.04	33.04		10.73		1.65
	AIN Toolkit Service - Training Session, Per Customer			BAPVX		8406	8406				10.73		1.65
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt			BAPTD		7.79	7.79	7.38	7.38		10.73		1.65
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay			DAPID		7.79	7.79	7.38	7.38		10.73		1.65
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate			BAPTM		7.79	7.79	7.38	7.38		10.73		1.65
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP			BAPTO		34.32	34.32	11.66	11.66		10.73		1.65
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP			BAPTC		34.32	34.32	11.66	11.66		10.73		1.65
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code			BAPTF		34.32	34.32	11.66	11.66		10.73		1.65
	AIN Toolkit Service - Query Charge, Per Query				0.0509436								
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query				0.0062787								
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes				0.06								
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription			BAPMS		7.79	7.79	4.47	4.47		10.73		1.65
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			BAPLS	3.85	8.62	8.62				10.73		1.65
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			BAPDS	4.28	7.79	7.79	4.47	4.47		10.73		1.65
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			BAPES	0.13	8.62	8.62				10.73		1.65
ODUF/EDOUF/ADUF/CN	Inc												
ODOI7EDOOI7ADOI7CII													
ACCESS D	AILY USAGE FILE (ADUF)												
	ADUF: Message Processing, per messag				0.013928								
	ADUF: Data Transmission (CONNECT:DIRECT), per messag				0.00012927								
ENHANCEL	O OPTIONAL DAILY USAGE FILE (EODUF)				0.222451								
	EODUF: Message Processing, per messagi				0.222451								
OPTIONAL	DAILY USAGE FILE (ODUF)												
	ODUF: Recording, per message				8800000.0								
	ODUF: Message Processing, per message				0.006614								
	ODUF: Message Processing, per Magnetic Tape provisions				48.77		1						
	ODUF: Data Transmission (CONNECT:DIRECT), per messag				0.00010772								
ENHANCED EXTENDED	) LINK (EELs)												
Ziiii ii ii ii ii zii zii zii													
	v EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, I				TN; New Orlea	ns, LA;							
	rlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below												
	Il states, EEL network elements shown below also apply to currently combined facilities whi					harge applies to	currently combi	ined facilities	converted	to UNEs.(No	n-recurring	rates do not a	apply.)
NOTE: In G	eorgia, the EEL network elements apply to ordinarily combined network elements per the G	A PSC	order.(No Switch A	As Is Cha	irge.)								
2-WIRE VO	ICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)												
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	1	UNCVX	UEAL2	13.43	115.02	54.58	43.28	5.68		10.73		1.65
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	2	UNCVX	UEAL2	18.6	115.02	54.58	43.28	5.68		10.73		1.65
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	3	UNCVX	UEAL2	35.18	115.02	54.58	43.28	5.68		10.73		1.65
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor		UNC1X	1L5XX	0.171								
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor		UNC1X	U1TF1		157.3	110.42	41.12	16.18		10.73		1.65
	DS1 Channelization System Per Mont		UNC1X	MQ1	151.74	51.63	13.29	1.35	1.21				
	Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport		UNCVX	1D1VG		6.05	4.36	40.00					
	Combination - Zone ' Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport	1	UNCVX	UEAL2		115.02	54.58	43.28	5.68		10.73		1.65
	Combination - Zone :	2	UNCVX	UEAL2	18.6	115.02	54.58	43.28	5.68		10.73		1.65

# Unbundled Network Elements FLORIDA

	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport	ГΤ	-									
	Combination - Zone :	3	UNCVX	UEAL2	35.18	115.02	54.58	43.28	5.68	10.73		1.65
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mon		UNCVX	1D1VG	1.42	6.05	4.36					
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		8.1	8.1	8.1	8.1	10.73		1.65
4-WIRE VOI	DICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)											
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -		1110101		04.00	445.00	54.50	40.00	F 00	40.70		4.00
	Zone 1 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -	1	UNCVX	UEAL4	21.23	115.02	54.58	43.28	5.68	10.73		1.65
ı	Zone 2	2	UNCVX	UEAL4	29.41	115.02	54.58	43.28	5.68	10.73		1.65
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -		UNCVA	UEAL4	29.41	113.02	34.36	43.20	3.00	10.73		1.00
	Zone 3	3	UNCVX	UEAL4	55.63	115.02	54.58	43.28	5.68	10.73		1.65
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.171	110.02	01.00	10.20	0.00	10.10		1.00
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor		UNC1X	U1TF1	90.87	157.3	110.42	41.12	16.18	10.73		1.65
	Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	151.74	51.63	13.29	1.35	1.21			
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mon		UNCVX	1D1VG	1.42	6.05	4.36					
i e	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	ı										
	Combination - Zone	1	UNCVX	UEAL4	21.23	115.02	54.58	43.28	5.68	10.73		1.65
ı	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	ı . l										
<u> </u>	Combination - Zone :	2	UNCVX	UEAL4	29.41	115.02	54.58	43.28	5.68	10.73		1.65
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport		1110101		55.00	445.00	54.50	40.00	5.00	40.70		4.00
	Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon	3	UNCVX	1D1VG	55.63	115.02	54.58	43.28	5.68	10.73		1.65
	voice Grade GOGI - DOT to DOU Grianner System combination - per mon	-+	UNCVX	10176	1.42	6.05	4.36				+ +	-
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		8.1	8.1	8.1	8.1	10.73		1.65
	Transparing Surrounty Combined Network Eldinetits Switch -As-is Origin	-	OI4OIA	011000		0.1	0.1	0.1	0.1	10.73		1.00
4-WIRE 56	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	$\neg$										
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	$\vdash$										
	Zone 1	1	UNCDX	UDL56	24.48	115.02	54.58	43.28	5.68	10.73		1.65
1	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	, 🗆										
ļ	Zone 2	2	UNCDX	UDL56	33.91	115.02	54.58	43.28	5.68	10.73		1.65
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -											
	Zone 3	3	UNCDX	UDL56	64.14	115.02	54.58	43.28	5.68	10.73		1.65
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	-+	UNC1X	1L5XX	0.171							
ĺ	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor	ı	UNC1X	U1TF1	90.87	157.3	110.42	41.12	16.18	10.73		1.65
<del>                                     </del>	Channelization - Channel System DS1 to DS0 combination Per Mor	-+	UNC1X	MQ1	151.74	51.63	13.29	1.35	1.21	10.73		1.00
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)	-	UNCDX	1D1DD	2.16	6.05	4.36	1.55	1.21			
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	-	0.105/1	18.88	2.10	0.00	1.00					
	Combination - Zone	1 1	UNCDX	UDL56	24.48	115.02	54.58	43.28	5.68	10.73		1.65
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport											
	Combination - Zone :	2	UNCDX	UDL56	33.91	115.02	54.58	43.28	5.68	10.73		1.65
ı	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport											
<b></b>	Combination - Zone (	3	UNCDX	UDL56	64.14	115.02	54.58	43.28	5.68	10.73		1.65
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-		LINODY	40400	0.40	0.00	0.00					
	64kbs)	-+	UNCDX	1D1DD	2.16	9.08	6.38					
ı	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X				0.4	8.1	8.1	10.73		
	Nonrecurring Currently Combined Network Elements Switch -As-is Char	-		LINICCC				0.1	0.1			1.66
4-WIRF 64	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	-		UNCCC		8.1	8.1			10.70		1.65
			-	UNCCC		8.1	8.1			10.75		1.65
04	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	$\dashv$		UNCCC		8.1	8.1			16.73		1.65
		1	UNCDX	UNCCC UDL64	24.48	115.02	54.58	43.28	5.68	10.73		1.65
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	1	UNCDX	UDL64		115.02	54.58			10.73		1.65
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	1 2			24.48			43.28 43.28	5.68 5.68			
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -		UNCDX	UDL64	33.91	115.02 115.02	54.58 54.58	43.28	5.68	10.73		1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3	1 2 3	UNCDX UNCDX UNCDX	UDL64 UDL64	33.91 64.14	115.02	54.58			10.73		1.65
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -		UNCDX	UDL64	33.91	115.02 115.02	54.58 54.58	43.28	5.68	10.73		1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNCDX UNCDX UNCDX UNCDX UNC1X	UDL64 UDL64 UDL64 1L5XX	33.91 64.14 0.171	115.02 115.02 115.02	54.58 54.58 54.58	43.28 43.28	5.68 5.68	10.73 10.73 10.73		1.68 1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mot		UNCDX UNCDX UNCDX UNC1X UNC1X	UDL64  UDL64  UDL64  1L5XX  U1TF1	33.91 64.14 0.171 90.87	115.02 115.02 115.02	54.58 54.58 54.58	43.28 43.28 41.12	5.68 5.68	10.73		1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor		UNCDX UNCDX UNCDX UNCDX UNC1X	UDL64 UDL64 UDL64 1L5XX	33.91 64.14 0.171	115.02 115.02 115.02	54.58 54.58 54.58	43.28 43.28	5.68 5.68	10.73 10.73 10.73		1.68 1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		UNCDX UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X	UDL64 UDL64 UDL64 1L5XX U1TF1 MQ1	33.91 64.14 0.171 90.87 151.74	115.02 115.02 115.02 157.3 51.63	54.58 54.58 54.58 110.42 13.29	43.28 43.28 41.12	5.68 5.68	10.73 10.73 10.73		1.68 1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)		UNCDX UNCDX UNCDX UNC1X UNC1X	UDL64  UDL64  UDL64  1L5XX  U1TF1	33.91 64.14 0.171 90.87	115.02 115.02 115.02	54.58 54.58 54.58	43.28 43.28 41.12	5.68 5.68	10.73 10.73 10.73		1.68 1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		UNCDX UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X	UDL64 UDL64 UDL64 1L5XX U1TF1 MQ1	33.91 64.14 0.171 90.87 151.74	115.02 115.02 115.02 157.3 51.63	54.58 54.58 54.58 110.42 13.29	43.28 43.28 41.12	5.68 5.68	10.73 10.73 10.73		1.68 1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4- 64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNC1X	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD	33.91 64.14 0.171 90.87 151.74 2.16	115.02 115.02 115.02 115.02 157.3 51.63 6.05	54.58 54.58 54.58 110.42 13.29 4.36	43.28 43.28 41.12 1.35	5.68 5.68 16.18 1.21	10.73 10.73 10.73 10.73		1.66 1.66 1.66
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNC1X	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD	33.91 64.14 0.171 90.87 151.74 2.16	115.02 115.02 115.02 115.02 157.3 51.63 6.05	54.58 54.58 54.58 110.42 13.29 4.36	43.28 43.28 41.12 1.35	5.68 5.68 16.18 1.21	10.73 10.73 10.73 10.73		1.66 1.66 1.66
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs) Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91	115.02 115.02 115.02 157.3 51.63 6.05 115.02	54.58 54.58 54.58 110.42 13.29 4.36 54.58	43.28 43.28 41.12 1.35 43.28 43.28	5.68 5.68 16.18 1.21 5.68	10.73 10.73 10.73 10.73 10.73		1.66 1.65 1.66 1.66 1.66
TV	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDX UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64	33.91 64.14 0.171 90.87 151.74 2.16 24.48	115.02 115.02 115.02 157.3 51.63 6.05 115.02	54.58 54.58 54.58 110.42 13.29 4.36 54.58	43.28 43.28 41.12 1.35	5.68 5.68 16.18 1.21 5.68	10.73 10.73 10.73 10.73		1.68 1.68 1.68 1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Fer Acility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64  UDL64	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91 64.14	115.02 115.02 115.02 157.3 51.63 6.05 115.02 115.02	54.58 54.58 54.58 110.42 13.29 4.36 54.58 54.58	43.28 43.28 41.12 1.35 43.28 43.28	5.68 5.68 16.18 1.21 5.68	10.73 10.73 10.73 10.73 10.73		1.66 1.65 1.66 1.66 1.66
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91	115.02 115.02 115.02 157.3 51.63 6.05 115.02	54.58 54.58 54.58 110.42 13.29 4.36 54.58	43.28 43.28 41.12 1.35 43.28 43.28	5.68 5.68 16.18 1.21 5.68	10.73 10.73 10.73 10.73 10.73		1.66 1.65 1.66 1.66 1.66
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  CU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Combination - Zone :  Combination - Zone :		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64  UDL64  1D1DD	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91 64.14	115.02 115.02 115.02 115.02 157.3 51.63 6.05 115.02 115.02 115.02 6.05	54.58 54.58 54.58 110.42 13.29 4.36 54.58 54.58 4.36	43.28 43.28 41.12 1.35 43.28 43.28	5.68 5.68 16.18 1.21 5.68 5.68	10.73 10.73 10.73 10.73 10.73 10.73		1.66 1.66 1.66 1.66 1.66
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Fer Acility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64  UDL64	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91 64.14	115.02 115.02 115.02 157.3 51.63 6.05 115.02 115.02	54.58 54.58 54.58 110.42 13.29 4.36 54.58 54.58	43.28 43.28 41.12 1.35 43.28 43.28	5.68 5.68 16.18 1.21 5.68	10.73 10.73 10.73 10.73 10.73		1.66 1.65 1.66 1.66 1.66
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  OCU-DP COCI (data) - DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Chars		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64  UDL64  1D1DD	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91 64.14	115.02 115.02 115.02 115.02 157.3 51.63 6.05 115.02 115.02 115.02 6.05	54.58 54.58 54.58 110.42 13.29 4.36 54.58 54.58 4.36	43.28 43.28 41.12 1.35 43.28 43.28	5.68 5.68 16.18 1.21 5.68 5.68	10.73 10.73 10.73 10.73 10.73 10.73		1.66 1.66 1.66 1.66 1.66
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  CCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Chars	1 2 3	UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91 64.14 2.16	115.02 115.02 115.02 115.02 157.3 51.63 6.05 115.02 115.02 115.02 6.05 8.1	54.58 54.58 54.58 110.42 13.29 4.36 54.58 54.58 4.36 8.1	43.28 43.28 41.12 1.35 43.28 43.28 43.28	5.68 5.68 16.18 1.21 5.68 5.68 5.68	10.73 10.73 10.73 10.73 10.73 10.73 10.73		1.68 1.68 1.68 1.68 1.68 1.68
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor  OCU-DP COCI (data) - DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Chars		UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64  UDL64  UDL64  1L5XX  U1TF1  MQ1  1D1DD  UDL64  UDL64  UDL64  1D1DD	33.91 64.14 0.171 90.87 151.74 2.16 24.48 33.91 64.14	115.02 115.02 115.02 115.02 157.3 51.63 6.05 115.02 115.02 115.02 6.05	54.58 54.58 54.58 110.42 13.29 4.36 54.58 54.58 4.36	43.28 43.28 41.12 1.35 43.28 43.28	5.68 5.68 16.18 1.21 5.68 5.68	10.73 10.73 10.73 10.73 10.73 10.73		1.66 1.66 1.66 1.66 1.66

Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.171							
Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Moi		UNC1X	U1TF1	90.87	157.3	110.42	41.12	16.18	10	73	1.65
Nonrecurring Currently Combined Network Elements Switch -As-Is Charc		UNC1X	UNCCC		8.1	8.1	8.1	8.1	10	73	1.65
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)											100
First DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX		196.32	109.65	46.38	13.03	10		1.65
First DS1Loop in DS3 Interoffice Transport Combination - Zone First DS1Loop in DS3 Interoffice Transport Combination - Zone	3	UNC1X UNC1X	USLXX		196.32 196.32	109.65 109.65	46.38 46.38	13.03	10		1.65 1.65
Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Mor	3	UNC3X	1L5XX	3.57	190.32	109.00	40.30	13.03	10	.13	1.65
Interoffice Transport - Dedicated - DS3 - Combination - Per Mile Per Mor		UNC3X	U1TF3		288.5	124.61	34.8	16.96	10	73	1.65
DS3 to DS1 Channel System combination per mon		UNC3X	MQ3	218.7	104.13	50.98	10.96	3.84	10	70	1.00
DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1		6.05	4.36	10.00	0.01			
Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX		196.32	109.65	46.38	13.03	10	73	1.65
Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X	USLXX		196.32	109.65	46.38	13.03	10		1.65
Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	3	UNC1X	USLXX		196.32	109.65	46.38	13.03	10	73	1.65
DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	14.24	6.05	4.36					
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC3X	UNCCC		8.1	8.1	8.1	8.1	10	73	1.65
2-WIRE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)											
2-Wire VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)  2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	1	UNCVX	UEAL2	13.43	115.02	54.58	43.28	5.68	10	73	1.65
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	2	UNCVX	UEAL2		115.02	54.58	43.28	5.68	10		1.65
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	3	UNCVX	UEAL2		115.02	54.58	43.28	5.68	10		1.65
Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor	- 5	UNCVX	1L5XX	0.0084		300	.0.20	5.50	10.		1.00
Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility		0.10171	120,00	0.0007							+ + + + + + + + + + + + + + + + + + + +
Termination per month		UNCVX	U1TV2	26.02	85.38	47.42	40.82	16.25	10	73	1.65
Nonrecurring Currently Combined Network Elements Switch -As-Is Charç	+	UNCVX	UNCCC		8.1	8.1	8.1	8.1	10	73	1.65
4-WIRE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)											+ +
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zon€	1	UNCVX	UEAL4		115.02	54.58	43.28	5.68	10		1.65
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	2	UNCVX	UEAL4		115.02	54.58	43.28	5.68	10		1.65
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	3	UNCVX	UEAL4		115.02	54.58	43.28	5.68	10	73	1.65
Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor		UNCVX	1L5XX	0.0084							
Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility		UNCVX	U1TV4	22.2	85.38	47.42	40.82	40.05	10	70	1.65
Termination per month		UNCVA		23.2	85.38	41.42	40.82	16.25	10.	./3	1.00
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq		UNCVX	UNCCC		8.1	8.1	8.1	8.1	10	73	1.65
DS3 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)											
High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor		UNC3X	1L5ND	10.06							
High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per											
month		UNC3X	UE3PX	387.1	220.36	139.5	60.49	23.69			
Interoffice Transport - Dedicated - DS3 - Per Mile per mon		UNC3X	1L5XX	3.57							
Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mo		UNC3X	U1TF3	1101	288.5	124.61	34.8	16.96	10	73	1.65
Nonrecurring Currently Combined Network Elements Switch -As-Is Charç		UNC3X	UNCCC		8.1	8.1	8.1	8.1	10	73	1.65
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)											
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor	+	UNCSX	1L5ND	10.06							+ + +
High Capacity Unbundled Local Loop - STS1 combination - Fer Mile per High High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per		5.100A	.20140	. 5.00							+ + +
month		UNCSX	UDLS1	426.68	220.36	139.5	60.49	23.69			
Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor		UNCSX	1L5XX	3.57							
Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mo		UNCSX	U1TFS	1085	288.5	124.61	34.8	16.96	10	73	1.65
Nonrecurring Currently Combined Network Elements Switch -As-Is Charc		UNCSX	UNCCC		8.1	8.1	8.1	8.1	10	73	1.65
		UNCOX	UNCCC		0.1	0.1	0.1	0.1	10.		1.03
2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)											
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	1	UNCNX	U1L2X		115.02	54.58	43.28	5.68	10		1.65
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	2	UNCNX	U1L2X U1L2X		115.02 115.02	54.58 54.58	43.28 43.28	5.68 5.68	10		1.65 1.65
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mi	3	UNCNX UNC1X	1L5XX		115.02	54.58	43.28	5.68	10.	13	1.65
Interoffice Transport - Dedicated - DS1 combination - Per Mi  Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor	+	UNC1X	U1TF1	90.87	157.3	110.42	41.12	16.18	10	73	1.65
Channelization - Channel System DS1 to DS0 combination - per mor	+	UNC1X	MQ1	151.74	51.63	13.29	1.35	1.21	10.		1.00
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon		UNCNX	UC1CA		6.05	4.36	1.00	1.41			+ + +
Additional 2 wire IDSN Loop in some DS4 Intereffice Transport Combination 7-	1	UNCNX	U1L2X	20.44	115.02	54.58	43.28	5.68	10	72	1.65
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon											1.65
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	2	UNCNX	U1L2X	28.31	115.02	54.58	43.28	5.68	10	73	1.65
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	3	UNCNX	U1L2X	53.56	115.02	54.58	43.28	5.68	10.	73	1.65
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per mon	- 5	UNCNX	UC1CA	3.76	6.05	4.36	70.20	0.00	10.		1.03
, , , , , , , , , , , , , , , , , , , ,			1 - 2 - 27,								

	1											
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL)												
First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	69.22	196.32	109.65	46.38	13.03		10.73		1.65
First DS1 Loop in STS1 Interoffice Transport Combination - Zone	2	UNC1X	USLXX		196.32	109.65	46.38	13.03		10.73		1.65
First DS1 Loop in STS1 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	181.38	196.32	109.65	46.38	13.03		10.73		1.65
Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor		UNCSX	1L5XX	3.57								
Interoffice Transport - Dedicated - STS1 combination - Facility Terminati		UNCSX	U1TFS	1085	288.5	124.61	34.8	16.96		10.73		1.65
STS1 to DS1 Channel System conbination per mon		UNCSX	MQ3	218.7								
DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	14.24	6.05	4.36						
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	69.22	196.32	109.65	46.38	13.03		10.73		1.65
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	95.89	196.32	109.65	46.38	13.03		10.73		1.65
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	181.38	196.32	109.65	46.38	13.03		10.73		1.65
DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	14.24	6.05	4.36						
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq		UNCSX	UNCCC		8.1	8.1	8.1	8.1		10.73		1.65
4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)												
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone	1	UNCDX	UDL56		115.02	54.58	43.28	5.68		10.73		1.65
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone	2	UNCDX	UDL56		115.02	54.58	43.28	5.68		10.73		1.65
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone	3	UNCDX	UDL56	64.14	115.02	54.58	43.28	5.68		10.73		1.65
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per M		UNCDX	1L5XX									
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminati	1	UNCDX	U1TD5	19.31	85.38	47.42	40.82	16.25		10.73		1.65
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCDX	UNCCC		8.1	8.1	8.1	8.1		10.73		1.65
4-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)												
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone	1	UNCDX	UDL64	24.48	115.02	54.58	43.28	5.68		10.73		1.65
	2	UNCDX	UDL64		115.02	54.58	43.28	5.68		10.73	+	1.65
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone									<b> </b>			
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone	3	UNCDX	UDL64	64.14	115.02	54.58	43.28	5.68		10.73		1.65
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M	+	UNCDX	1L5XX	0.0098	110.50		74.05	04.04		10.70		
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminati		UNCDX	U1TD6	19.31	149.56	86	71.35	31.91		10.73		1.65
Nonrecurring Currently Combined Network Elements Switch -As-Is Charç		UNCDX	UNCCC		8.1	8.1	8.1	8.1		10.73		1.65
IAL NETWORK ELEMENTS												
When used as ordinarilty combined network elements in Georgia, the non-recurring charges apply and	a tne S	witch As is Charge	does not									
Node (SynchroNet)		LINCDY	LINCAIT	16.25								
Node (SynchroNet) Node per month		UNCDX	UNCNT	16.35								
Node per month	binatio		UNCNT	16.35								
Node per month	binatio	n)		16.35	8.1	8.1	8.1	8.1		10.73		1.65
Node per month  Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each combination of the Charge (One applies to each combination	binatio		UNCCC	16.35	8.1	8.1	8.1	8.1		10.73		1.65
Node per month	binatio	n)		16.35	8.1	8.1	8.1	8.1 8.1		10.73		1.65
Node per month  Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each coml 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion	binatio	n) UNCVX	UNCCC	16.35								
Node per mont  Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each coml 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	binatio	UNCVX UNCDX	UNCCC	16.35	8.1	8.1	8.1	8.1		10.73		1.65
Nonecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each combined Network Elements "Switch As Is" Charge (One applies to each combined Network Elements "Switch As Is" Conversion Charge	binatio	UNCVX UNCDX UNC1X UNC3X	UNCCC UNCCC UNCCC	16.35	8.1 8.1 8.1	8.1 8.1 8.1	8.1 8.1 8.1	8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each coml 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge		UNCVX UNCDX UNC1X UNC3X UNC3X	UNCCC	16.35	8.1 8.1	8.1 8.1	8.1	8.1 8.1		10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each coml 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 an		UNCVX UNCDX UNC1X UNC3X UNC3X	UNCCC UNCCC UNCCC	16.35	8.1 8.1 8.1	8.1 8.1 8.1	8.1 8.1 8.1	8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each combined Network Elements "Switch As Is" Charge (One applies to each combined Network Elements "Switch As Is" Conversion Charge	nd abov	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC		8.1 8.1 8.1 8.1	8.1 8.1 8.1 8.1	8.1 8.1 8.1	8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comi 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and ONAL SUPPORT SYSTEMS	ad abov	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC	arges as order	8.1 8.1 8.1 8.1 ed by the State C	8.1 8.1 8.1 8.1	8.1 8.1 8.1	8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each combined Network Elements "Switch As Is" Charge (One applies to each combined Network Elements "Switch As Is" Conversion Charge	ad abov	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC	arges as order	8.1 8.1 8.1 8.1 ed by the State C	8.1 8.1 8.1 8.1	8.1 8.1 8.1	8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comi 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and ONAL SUPPORT SYSTEMS	pecific e BellSe ronic se	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC UNCCC	arges as order	8.1 8.1 8.1 8.1 ed by the State C	8.1 8.1 8.1 8.1 commissions	8.1 8.1 8.1 8.1	8.1 8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge Conversion Charg	pecific e BellSe ronic se	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC UNCCC dering chinic services, or Cl	arges as order	8.1 8.1 8.1 8.1 8.1 cd by the State C trige ct the regional elicenters.	8.1 8.1 8.1 8.1 commissions	8.1 8.1 8.1 8.1	8.1 8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the state sp NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the electronic (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the electronic (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR bar	pecific e BellSe ronic se	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC UNCCC	arges as order	8.1 8.1 8.1 8.1 ed by the State C	8.1 8.1 8.1 8.1 commissions	8.1 8.1 8.1 8.1	8.1 8.1 8.1 8.1		10.73 10.73 10.73		1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge Conversion Charg	oecific e BellSironic si	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SOMEC	arges as order be ordering chi LEC-1 may ele	8.1 8.1 8.1 8.1 ced by the State C tripe ct the regional elicities and the state C and the sta	8.1 8.1 8.1 8.1 commissions ectronic service	8.1 8.1 8.1 8.1 ordering cha	8.1 8.1 8.1 8.1	Office, refer	10.73 10.73 10.73 10.73	ebsite:	1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge Conversion Charge Conversion Charge Channel used in a COMBINATION - "Switch As Is" Conversion Charge Charge Channel used in a COMBINATION - "Switch As Is" Conversion Charge Charge Charge or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the state sp NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the electr NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR base (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Geographtp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	oecific e BellSironic si	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SOMEC	arges as order be ordering chi LEC-1 may ele	8.1 8.1 8.1 8.1 ced by the State C tripe ct the regional elicities and the state C and the sta	8.1 8.1 8.1 8.1 commissions ectronic service	8.1 8.1 8.1 8.1 ordering cha	8.1 8.1 8.1 8.1	Office, refer	10.73 10.73 10.73 10.73	ebsite:	1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and NOTE: (1) Concluded: CLEC-1 should contact its contract negotiator if it prefers the state sp NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the electr NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR bai (Regional)	oecific e BellSironic si	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SOMEC	arges as order be ordering chi LEC-1 may ele	8.1 8.1 8.1 8.1 ced by the State C tripe ct the regional elicities and the state C and the sta	8.1 8.1 8.1 8.1 commissions ectronic service	8.1 8.1 8.1 8.1 ordering cha	8.1 8.1 8.1 8.1	Office, refer	10.73 10.73 10.73 10.73	ebsite:	1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge Conversion Charge Conversion Charge Channel used in a COMBINATION - "Switch As Is" Conversion Charge Charge Channel used in a COMBINATION - "Switch As Is" Conversion Charge Charge Charge or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the state sp NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the electr NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR base (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Geographtp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	oecific e BellSironic si	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SOMEC	arges as order be ordering chi LEC-1 may ele	8.1 8.1 8.1 8.1 ced by the State C tripe ct the regional elicities and the state C and the sta	8.1 8.1 8.1 8.1 commissions ectronic service	8.1 8.1 8.1 8.1 ordering cha	8.1 8.1 8.1 8.1	Office, refer	10.73 10.73 10.73 10.73	ebsite:	1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the state specific Commission ordered rates the slate specific Commission ordered rates the electronic NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR bare (Regional)   Electronic CSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)   The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Geograph (Ports)   CLCAL EXCHANGE SWITCHING(PORTS)   CLCAL EXCHANGE SWITCHING(PORTS)   CLCAL EXCHANGE SWITCHING(PORTS)   CLCAL EXCHANGE SWITCHING(PORTS)   COMBINATION - "Switch As Is" Conversion Charge conversion Combinat	d abovecific to BellScronic si	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months electronic service or outh regional electronic service or outh regional electronic service or outh regional electronic service ordering char	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SOMEC	arges as order be ordering chi LEC-1 may ele	8.1 8.1 8.1 8.1 ced by the State C tripe ct the regional elicities and the state C and the sta	8.1 8.1 8.1 8.1 commissions ectronic service	8.1 8.1 8.1 8.1 ordering cha	8.1 8.1 8.1 8.1	Office, refer	10.73 10.73 10.73 10.73	ebsite:	1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and NOTE: (1) Concluded: CLEC-1 should contact its contract negotiator if it prefers the state sp NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the electr NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR baid Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)    The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Geographitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm    ED LOCAL EXCHANGE SWITCHING(PORTS)    Exchange Ports   NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will need.	d abovecific to BellScronic si	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months electronic service or outh regional electronic service or outh regional electronic service or outh regional electronic service ordering char	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SOMEC	arges as order be ordering chi LEC-1 may ele	8.1 8.1 8.1 8.1 ced by the State C tripe ct the regional elicities and the state C and the sta	8.1 8.1 8.1 8.1 commissions ectronic service	8.1 8.1 8.1 8.1 ordering cha	8.1 8.1 8.1 8.1	Office, refer	10.73 10.73 10.73 10.73	ebsite:	1.65 1.65
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each comic 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3 and NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the state sp NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the electron NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR base (Regional)   Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)   The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Geographtp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm   Exchange Ports	d abovecific to BellScronic si	UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX re=four months electronic service or outh regional electronic service or outh regional electronic service or outh regional electronic service ordering char	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SOMEC	arges as order be ordering chi LEC-1 may ele	8.1 8.1 8.1 8.1 ced by the State C tripe ct the regional elicities and the state C and the sta	8.1 8.1 8.1 8.1 commissions ectronic service	8.1 8.1 8.1 8.1 ordering cha	8.1 8.1 8.1 8.1	Office, refer	10.73 10.73 10.73 10.73	ebsite:	1.65 1.65

		1	1 1			1				
-	Tuebone Deste C. Mire Analog Line Dest with Celler ID. De	UEPSR	UEPRC	4.24	3.37	3.27	1.69	4.00	10.73	4.05
E)	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re	UEPSR	UEPRC	1.34	3.37	3.27	1.69	1.62	10.73	1.65
-	Total Control	LIEDOD	LIEDDO	4.04	0.07	0.07	4.00	4.00	40.70	4.05
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Re	UEPSR	UEPRO	1.34	3.37	3.27	1.69	1.62	10.73	1.65
	Exchange Ports - 2-Wire VG unbundled Florida area calling with Caller ID - Re	UEPSR	UEPAF	1.34	3.37	3.27	1.69	1.62	10.73	1.65
E	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU	UEPSR	UEPAP	1.34	3.37	3.27	1.69	1.62	10.73	1.65
	Subsequent Activity	UEPSR	USASC	0	0	0				
FEATURES										
A.	All Available Vertical Feature	UEPSR	UEPVF	2.17	0	0			10.73	1.65
			1			-			17117	
2-WIRE VOICE	E GRADE LINE PORT RATES (BUS)		1							
	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bi	UEPSB	UEPBL	1.34	3.37	3.27	1.69	1.62	10.73	1.65
		UEFOD	UEFBL	1.34	3.31	3.21	1.09	1.02	10.73	1.03
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484		1							
ID	D - Bus.	UEPSB	UEPBC	1.34	3.37	3.27	1.69	1.62	10.73	1.65
E.	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bu	UEPSB	UEPBO	1.34	3.37	3.27	1.69	1.62	10.73	1.65
E.	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B	UEPSB	UEPB1	1.34	3.37	3.27	1.69	1.62	10.73	1.65
S	Subsequent Activity	UEPSB	USASC	0	0	0				
FEATURES	Subsequent returns	02.05	00,100							
	All Available Vertical Feature	UEPSB	UEPVF	2.17	0	0			10.73	1.65
		UEFSB	UEFVF	2.17	- 0	U			10.73	1.03
	PORT RATES (DID & PBX)		1							
E.	Exchange Ports - 2-Wire DID Port	UEPEX	UEPP2	8.81	70.69	14.26	37.81	3.84	10.73	1.65
1			1 T	Т						
E	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabilit	UEPDD	UEPDD	52.73	136.24	70.1	44	2.8	10.73	1.65
F	Exchange Ports - 2-Wire ISDN Port (See Notes below	UEPTX UEPSX		8.46	42.22	45.69	24.91	10.75	10.73	1.65
	All Features Offered	UEPTX UEPSX		2.17	0	0			100	1.00
	smission/usage charges associated with POTS circuit switched usage will also apply to circuit switch						with 2-wire IS	DN porte		
									Dusiness Demost Description	
NOIE: Acces	ss to B Channel or D Channel Packet capabilities will be available only through BFR/New Business	Request Process. I	rates for th				bona Fide F	request/New	business Request Process.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles	UEPTX UEPSX		0	0	0				
E.	Exchange Ports - 4-Wire ISDN DS1 Por	UEPEX	UEPEX	79.35	157.42	85.8	44.89	16.43	10.73	1.65
2.	2-Wire VG Unbundled 2-Way PBX Trunk - Re:	UEPSE	<b>UEPRD</b>	1.34	35.22	16.39	11.14	0.648	10.73	1.65
									17117	
2	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu	UEPSP	UEPPC	1.34	35.22	16.39	11.14	0.648	10.73	1.65
2-	2-Wile VG Line Side Offburided 2-Way PBA Truffk - Bu	UEFSF	UEFFC	1.34	33.22	10.39	11.14	0.046	10.73	1.03
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu	UEPSP	UEPPO	1.34	35.22	16.39	11.14	0.648	10.73	1.65
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu	UEPSP	UEPP1	1.34	35.22	16.39	11.14	0.648	10.73	1.65
	2-Wire Analog Long Distance Terminal PBX Trunk - Bu	UEPSP	UEPLD	1.34	35.22	16.39	11.14	0.648	10.73	1.65
2-	2-Wire Voice Unbundled PBX LD Terminal Port	UEPSP	UEPLD	1.34	35.22	16.39	11.14	0.648	10.73	1.65
2.	2-Wire Vice Unbundled 2-Way PBX Usage Po	UEPSP	UEPXA	1.34	35.22	16.39	11.14	0.648	10.73	1.65
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Por	UEPSP	UEPXB	1.34	35.22	16.39	11.14	0.648	10.73	1.65
2	2-Wire Voice Unbundled PBX LD DDD Terminals Po	UEPSP	UEPXC	1.34	35.22	16.39	11.14	0.648	10.73	1.65
2-	e-wife voice officialities For	OLFOF	OLIAC	1.54	33.22	10.55	11.14	0.040	10.73	1.03
	With Miles Hall BRY I B Touris I G 1914 and B	LIEDOD	LIEDVO	4.04	05.00	40.00	44.44	0.040	40.70	4.05
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	UEPSP	UEPXD	1.34	35.22	16.39	11.14	0.648	10.73	1.65
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc	UEPSP	UEPXE	1.34	35.22	16.39	11.14	0.648	10.73	1.65
2-	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling									
P <sup>,</sup>	Port	UEPSP	UEPXL	1.34	35.22	16.39	11.14	0.648	10.73	1.65
2.	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P	UEPSP	UEPXM	1.34	35.22	16.39	11.14	0.648	10.73	1.65
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling							2.3.0	10.10	
	2-wire voice Unbundled 1-way Outgoing PBX Hotel/Hospital Discount Room Calling	UEPSP	UEPXO	1.34	35.22	16.39	11.14	0.648	10.73	1.65
	OIL .	UEPSP	UEPAU	1.04					10.73	
	W. W. W. Hall Bridge Control BRV March 18	LIEDOD	LIEDYC	101						1.65
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	UEPSP	UEPXS	1.34	35.22	16.39	11.14	0.648	10.73	
2-							11.14	0.648	10.73	
2- Si	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc Subsequent Activity	UEPSP UEPSP	UEPXS	0	35.22 0	16.39	11.14	0.648	10.73	
2-							11.14	0.648	10.73	
SI FEATURES							11.14	0.648	10.73	1.65
SI FEATURES	Subsequent Activity  All Available Vertical Feature	UEPSP	USASC	0	0	0	11.14	0.648		1.65
SI FEATURES AI EXCHANGE PO	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)	UEPSP	USASC	2.17	0	0			10.73	
SI FEATURES AI EXCHANGE PO	Subsequent Activity  All Available Vertical Feature	UEPSP	USASC	0	0	0	1.69	1.62		1.65
SI FEATURES AI EXCHANGE PE	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por	UEPSP UEPSP UEPSE	USASC	0 2.17 1.34	0 0 3.37	0 0 3.27	1.69	1.62	10.73	
SI FEATURES AI EXCHANGE PI EXC	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switched	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF uit switched	0 2.17 1.34	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI FEATURES AI EXCHANGE PI EXC	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF uit switched	0 2.17 1.34	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI FEATURES AI EXCHANGE PI EXCHANGE Transr NOTE: Access	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switc	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF uit switched	0 2.17 1.34	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI FEATURES AI EXCHANGE PI EXCHANGE Transr NOTE: Access	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switched	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF uit switched	0 2.17 1.34	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI FEATURES AI EXCHANGE PI EXCHANGE Transr NOTE: Access	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switc	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF uit switched	0 2.17 1.34	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI FEATURES AI EXCHANGE PI E NOTE: Transr NOTE: Access D LOCAL SWIT	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switces to B Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF uit switched	0 2.17 1.34	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI FEATURES AI EXCHANGE PI NOTE: Transr NOTE: Access D LOCAL SWIT	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switc	UEPSP UEPSP UEPSE hed voice and/or circ	USASC  UEPVF  uit switched Rates for th	0 2.17 1.34 I data transmise e packet capa	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SE SE SE SE SE SE SE SE SE SE SE SE SE	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switces to B Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE  witching (Port Usage)  End Office Switching Function, Per MOL	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF  uit switched Rates for th	0 2.17 1.34 I data transmis e packet capa 0.0007341	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SE SE SE SE SE SE SE SE SE SE SE SE SE	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switc	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF  uit switched Rates for th	0 2.17 1.34 I data transmise e packet capa	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SEATURES AND AND AND AND AND AND AND AND AND AND	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switc	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF  uit switched Rates for th	0 2.17 1.34 I data transmis e packet capa 0.0007341	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SEATURES AND AND AND AND AND AND AND AND AND AND	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switces to B Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE  witching (Port Usage)  End Office Switching Function, Per MOL	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF  uit switched Rates for th	0 2.17 1.34 I data transmis e packet capa 0.0007341	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SECTION STATE OF THE SECTIO	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  Smission/usage charges associated with POTS circuit switched usage will also apply to circuit switces to B Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE  witching (Port Usage)  End Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL  ching (Port Usage) (Local or Access Tandem)	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF  uit switched Rates for th	0 2.17 1.34 I data transmis e packet capa 0.0007341	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SEATURES ALL SECHANGE PLANTE: Transmote: Access D LOCAL SWIT End Office Sw Error Transmoters Switch Swit	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  smission/usage charges associated with POTS circuit switched usage will also apply to circuit switched to Both Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE  witching (Port Usage)  and Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL  ching (Port Usage) (Local or Access Tandem)  Fandem Switching Function Per MOL	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF UEPVF uuit switched Rates for th	0 2.17 1.34 I data transmis e packet capa 0.0007341 0.0001571	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SEATURES ALL SECHANGE PLANTE: Transmote: Access D LOCAL SWIT End Office Sw Error Transmoters Switch Swit	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  Smission/usage charges associated with POTS circuit switched usage will also apply to circuit switces to B Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE  witching (Port Usage)  End Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL  ching (Port Usage) (Local or Access Tandem)	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF UEPVF uuit switched Rates for th	0 2.17 1.34 I data transmise e packet capa 0.0007341 0.0001571	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SE SE SE SE SE SE SE SE SE SE SE SE SE	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  Imission/usage charges associated with POTS circuit switched usage will also apply to circuit switches to B Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE  witching (Port Usage)  and Office Switching Function, Per MOU  End Office Trunk Port - Shared, Per MOU  Ching (Port Usage) (Local or Access Tandem)  fandem Switching Function Per MOU  fandem Switching Function Per MOU  fandem Trunk Port - Shared, Per MOU  fandem Trunk Port - Shared, Per MOU	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF UEPVF uuit switched Rates for th	0 2.17 1.34 I data transmis e packet capa 0.0007341 0.0001571	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	
SI SEATURES AND ALL SEA	Subsequent Activity  All Available Vertical Feature  PORT RATES (COIN)  Exchange Ports - Coin Por  Imission/usage charges associated with POTS circuit switched usage will also apply to circuit switches to B Channel or D Channel Packet capabilities will be available only through BFR/New Business  TCHING, PORT USAGE  witching (Port Usage)  and Office Switching Function, Per MOU  End Office Trunk Port - Shared, Per MOU  Ching (Port Usage) (Local or Access Tandem)  fandem Switching Function Per MOU  fandem Switching Function Per MOU  fandem Trunk Port - Shared, Per MOU  fandem Trunk Port - Shared, Per MOU	UEPSP UEPSP UEPSE hed voice and/or circ	USASC UEPVF UEPVF uuit switched Rates for th	0 2.17 1.34 I data transmis e packet capa 0.0007341 0.0001571	0 0 3.37 sion by B-Chanr	0 0 3.27	1.69	1.62 DN ports.	10.73	

ED POR I/LOC	OP COMBINATIONS - COST BASED RATES													
0	Date of the last Dallow History in the FOO of the Control of the last					tal Barta								
Features sh	Rates are applied where BellSouth is required by FCC and/or State Commission rule to all apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same	provide ne mani	ner as	they are applied	to the Stand	-Alone Unbur	dled Port section	of this Rate Ex	hibit.			<del></del>		_
	and Tandem Switching Usage and Common Transport Usage rates in the Port section o									Combinations.				
	and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cor			ot Currently Com	bined Comb	os and the fire	st and additional	Port nonrecurrin	g charges apply to No	ot Currently Co	mbined Combo	s. For Currently Combined C	ombos in GA, T	'N a
states, the n	nonrecurring charges shall be those identified in the Nonrecurring - Currently Combined s	sections	S.					1						
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)													_
UNE Port/Lo	oop Combination Rates													
	2-Wire VG Loop/Port Combo - Zone		2			13.01 17.15								
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :		3			30.45								_
						00.10								_
UNE Loop F														
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		2	UEPRX UEPRX	UEPLX	11.89 16.03								
	2-Wire Voice Grade Loop (SL1) - Zone		3	UEPRX	UEPLX	29.33								
2-Wire Voic	e Grade Line Port Rates (Res)													
	2-Wire voice unbundled port - residenc			UEPRX	UEPRL	1.12		+			10.73		1.65	
	2-Wire voice unbundled port with Caller ID - re			UEPRX	UEPRC	1.12					10.73		1.65	
	·													
	2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r			UEPRX UEPRX	UEPRO UEPAF	1.12 1.12		1			10.73 10.73		1.65 1.65	
	2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundles res, low usage line port with Caller ID (LU!			UEPRX	UEPAF	1.12		+			10.73		1.65	—
	port man dunor to (Ed)													
FEATURES	All Features Offered			UEPRX	UEPVF	2.17	0	0			10.73		1.65	
	All Features Offered			UEPRA	UEPVF	2.17	U	U			10.73		1.05	
LOCAL NUM	MBER PORTABILITY													
	Local Number Portability (1 per port			UEPRX	LNPCX	0.35								
NONDECLID	RRING CHARGES (NRCs) - CURRENTLY COMBINED													
NONKECOK	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as			UEPRX	USAC2		0.092	0.092			10.73	<del></del>		_
	THIS FORE STAGE ESSEY EITHER SIX COMMUNICION CONTOURS CON													
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan			UEPRX	USACC		0.092	0.092			10.73			
ADDITIONA	I NRCs													
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ			UEPRX	USAS2	0	0	0						
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)													
UNE Port/Lo	oop Combination Rates											<del></del>		_
	2-Wire VG Loop/Port Combo - Zone		1			13.01								
	2-Wire VG Loop/Port Combo - Zone		2			17.15								
	2-Wire VG Loop/Port Combo - Zone		3			30.45								
UNE Loop F	Rates											<del></del>		
	2-Wire Voice Grade Loop (SL1) - Zone		1	UEPBX	UEPLX	11.89								
	2-Wire Voice Grade Loop (SL1) - Zone		2	UEPBX	UEPLX	16.03		-						
	2-Wire Voice Grade Loop (SL1) - Zone		3	UEPBX	UEPLX	29.33		+		_		<del>- + +</del>	<del>  </del> -	
2-Wire Voic	e Grade Line Port (Bus)													
	2-Wire voice unbundled port without Caller ID - bu			UEPBX	UEPBL	1.12					10.73		1.65	
	2 Wire voice unbundled port with Coller + E494 ID by			UEPBX	UEPBC	1.12					10.73		1.65	
	2-Wire voice unbundled port with Caller + E484 ID - bu			UEPBA	UEPBC	1.12		+			10.73		1.00	
	2-Wire voice unbundled port outgoing only - bu			UEPBX	UEPBO	1.12		<u> </u>			10.73		1.65	
	2-Wire voice unbundled incoming only port with Caller ID - Bı			UEPBX	UPEB1	1.12					10.73		1.65	
LOCAL NUM	MBER PORTABILITY							+					<del>  </del> -	
LOCAL NON	Local Number Portability (1 per port			UEPBX	LNPCX	0.35		+				<del></del>		
					270									
FEATURES				LIEBBY .	LIES -	0.1-					40 ===		1.05	
	All Features Offered			UEPBX	UEPVF	2.17	0	0			10.73		1.65	
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED							<del> </del>		+		<del></del>		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as			UEPBX	USAC2		0.092	0.092			10.73		1.65	_
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan													_
				UEPBX	USACC		0.092	0.092	1 1	1	1		1	

Attachment 2 Exhibit C

	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX	USAS2				10.73	
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)								
UNE Port/Lo	oop Combination Rates								
	2-Wire VG Loop/Port Combo - Zone	1			13.01				
	2-Wire VG Loop/Port Combo - Zone	2			17.15				
	2-Wire VG Loop/Port Combo - Zone	3			30.45				
UNE Loop R	Pates								
ONE LOOP I	2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPRG	UEPLX	11.89				
	2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPRG	UEPLX	16.03				
	2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPRG	UEPLX	29.33				
2-Wire Voice	e Grade Line Port Rates (RES - PBX)								
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	4.40			10.73	4.05
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - RE		UEPRG	UEPRD	1.12			10.73	1.65
LOCAL NUM	MBER PORTABILITY								
	Local Number Portability (1 per port		UEPRG	LNPCP	3.5				
FEATURES									
	All Features Offered		UEPRG	UEPVF	2.17	0	0	10.73	1.65
NONDECLID	RRING CHARGES (NRCs) - CURRENTLY COMBINED								+ + + + + + + + + + + + + + + + + + + +
NUNKECUK	ANING CHARGES (NACS) - CORRENTLY COMBINED								+ + + + + + + + + + + + + + + + + + + +
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPRG	USAC2		7.62	1.72	10.73	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with		323	33,.32		7.02	2	10.75	
	Change		UEPRG	USACC		7.62	1.72	10.73	
ADDITIONAL									
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USAS2	0	0	0		
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					7.09	7.09	10.73	1.6
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)								
2	er old bereit in the enter old (boo 15%)								
UNE Port/Lo	oop Combination Rates								
	2-Wire VG Loop/Port Combo - Zone	1			13.01				
	2-Wire VG Loop/Port Combo - Zone :	2			17.15				
	2-Wire VG Loop/Port Combo - Zone	3			30.45				
UNIE L B									
UNE Loop R	2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPPX	UEPLX	11.89				
	2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPPX	UEPLX	16.03				
	2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPPX	UEPLX	29.33				
2-Wire Voice	e Grade Line Port Rates (BUS - PBX)								
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı		UEPPX	UEPPC	1.12			10.73	1.65
	Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	1.12			10.73	1.65
	Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPP0	1.12			10.73	
	2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	1.12			10.73	
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	1.12			10.73	
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	1.12			10.73	
1	2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	1.12			10.73	1.65
1	O Mary Visited History III A DDV I D Transferd C 15 15 15 15		LIESS.		4.45				
+	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	-	UEPPX	UEPXD	1.12			10.73	1.65
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		UEPPX	UEPXE	1.12	-	+	10.73	1.65
1	Port Port PBX Hotel/Hospital Economy Administrative Calling		UEPPX	UEPXL	1.12			10.73	1.65
			OLITA	OLI AL	1.14			10.73	1.03
1	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX	UEPXM	1.12			10.73	1.65
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling								
1	Port		UEPPX	UEPXO	1.12			10.73	1.65
1	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX	UEPXS	1.12			10.73	1.65
LOCAL MITT	MPER ROPTARII ITV	-				1	-		+ + + + + + + + + + + + + + + + + + + +
LOCAL NUM	MBER PORTABILITY		HEDDY	LNDCS	2.45				+ + + + + + + + + + + + + + + + + + + +
	Local Number Portability (1 per port		UEPPX	LNPCP	3.15	-	+		+ + + + + + + + + + + + + + + + + + + +
FEATURES							+		<del>                                     </del>
. LAI UILO	All Features Offered		UEPPX	UEPVF	2.17	0	0	10.73	1.65
1	, iii i dataroo onorot		OLFFA	OEL AL.	4.11	U		10.73	1.65
							1		
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED			1 1					
NONRECUR	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPPX	USAC2					1.65

2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with										
Change		UEPPX	USACC		7.62	1.72		10.73	3 1.65	_
ADDITIONAL NRCs										-
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPPX	USAS2	0	0	0				$\rightarrow$
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi		OLITA	COMOZ		7.09	7.09		10.73	3 1.65	1.65
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT										_
UNE Port/Loop Combination Rates										-
2-Wire VG Coin Port/Loop Combo – Zone 1				13.01					<del>                                     </del>	+
2-Wire VG Coin Port/Loop Combo – Zone 2				17.15						$\dashv$
2-Wire VG Coin Port/Loop Combo – Zone 3				30.45						
UNE Loop Rates										
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	11.89					+ + + + + + + + + + + + + + + + + + + +	-
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	16.03						_
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	29.33						
O.W. Velia Ocala Usa Basta (OOM)										_
2-Wire Voice Grade Line Ports (COIN)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (FL)									+ + + + + + + + + + + + + + + + + + + +	-
2 Wile Court 2 Way with Operator Corectning and Blocking, 611, 500/510, 11888 (12)		UEPCO	UEP2F	1.12				10.73	3 1.65	1.65
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (FL)		UEPCO	UEPFA	1.12				10.73		
2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, an Local (FL)	d	LIEBOO	LIEBOO	4.40					,	
Local (FL)  2-Wire Coin Outward with Operator Screening and 011 Blocking (AL, FL)		UEPCO UEPCO	UEPCG UEPRK	1.12 1.12			+ + + -	10.73 10.73		
2-Wire Coin Outward with Operator Screening and 817 blocking: 900/976, 1+DDD, 011+			JEI KK	2						
(FL)		UEPCO	UEPOF	1.12				10.73	3 1.65	.65
2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)		UEPCO	UEPCQ	1.12				10.73	3 1.65	
2-Wire 2-Way Smartline with 900/976 (all states except LA)		UEPCO	UEPCK	1.12				10.73		
2-Wire Coin Outward Smartline with 900/976 (all states except LA)		UEPCO	UEPCR	1.12				10.73		
ADDITIONAL UNE COIN PORT/LOOP (RC)										_
UNE Coin Port/Loop Combo Usage (Flat Rate		UEPCO	URECU	1.86	0	0				
		021 00	UKLUU	1.00	·					-
LOCAL NUMBER PORTABILITY										
Local Number Portability (1 per port		UEPCO	LNPCX	0.35						_
FEATURES										-
All Features Offered		UEPCO	UEPVF	2.17	(	0	0	10	0.73 1.65	.65
										_
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPCO	USAC2		0.092	0.092		10.73	3 1.68	1.05
2-Wife Voice Grade Loop / Line Fort Combination - Conversion - Switch-as		UEFCO	USACZ		0.092	0.092		10.73	1.00	.00
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		UEPCO	USACC		0.092	0.092		10.73	3 1.65	1.65
ADDITIONAL NRCs										+
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPCO	USAS2		0	0		10.73		+
		02.00	CONCE			Ŭ		70.70		
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT										I
UNE Port/Loop Combination Rates										-
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	1		+ +	22.22					+ + + + + + + + + + + + + + + + + + + +	+
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	2			27.39						土
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	3			43.79						_
UNE Loop Rates							+ + +		+ + + + + + + + + + + + + + + + + + + +	+
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	1	UEPPX	UECD1	13.43			1	10.73	3 1.65	+
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	2	UEPPX	UECD1	18.6				10.73	3 1.65	
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	3	UEPPX	UECD1	35.18				10.73	3 1.65	+
UNE Port Rate									+ + + + + + + + + + + + + + + + + + + +	+
Exchange Ports - 2-Wire DID Por		UEPPX	UEPD1	8.79				10.73	3 1.65	<u>コ</u>
NONDEGUIDDING GUADGEG GUIDDENTI V COMPINED										#
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-		UEPPX	USAC1		7.08	1.69	<del>                                     </del>	10.73	3 1.65	+
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switchas:  2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowab	le	OLFFX	USACI		7.00	1.03		10.73	1.03	+
Changes		UEPPX	USA1C		7.08	1.69		10.73	3 1.65	_
ADDITIONAL NIPCO										$\dashv$
ADDITIONAL NRCs  2-Wire DID Subsequent Activity - Add Trunks, Per Trun		UEPPX	USAS1		29.08	29.08	+ + +	10.73	3 1.65	+
		OLITA	00/101		20.00	25.00		10.73	1.03	_
Telephone Number/Trunk Group Establisment Charges										1
DID Trunk Termination (One Per Port		UEPPX UEPPX	NDT NDZ	0	0	0		10.73		
DID Numbers Establish Trunk Group and Provide First Croup of 20 DID Number				W		U	1 1	10.73	1.65	
DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbe Additional DID Numbers for each Group of 20 DID Numbe		UEPPX	ND4	0	0	0		10.73	3 1.65	

Reserve Non-Consecutive DID number Reserve DID Numbers	UEPPX UEPPX	ND6 NDV	0	0	0			10.73 10.73	1.65 1.65	
LOCAL NUMBER PORTABILITY					1	1				
Local Number Portability (1 per port	UEPPX	LNPCP	3.15							
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT										
UNE Port/Loop Combination Rates										
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zonε	1 UEPPB 1 UEPPR		30.29							
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	2 UEPPB UEPF	PR	36.51							
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zonε	3 UEPPB UEPF	PR	56.45							
UNE Loop Rates										
2-Wire ISDN Digital Grade Loop - UNE Zone	1 UEPPB UEPP	PR USL2X	13.43					10.73	1.65	5
2-Wire ISDN Digital Grade Loop - UNE Zone	2 UEPPB UEPF	PR USL2X	29.44					10.73	1.65	5
2-Wire ISDN Digital Grade Loop - UNE Zone	3 UEPPB UEPP	PR USL2X	49.38					10.73	1.65	5
UNE Port Rate Exchange Port - 2-Wire ISDN Line Side Po	UEPPB UEPP	PR UEPPB	7.07					10.73	1.65	5
NONRECURRING CHARGES - CURRENTLY COMBINED					+	1				
2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion	UEPPB UEPP	R USACB	0	27.61	15.33			10.73	1.65	5
ADDITIONAL NRCs										
LOCAL NUMBER PORTABILITY										
Local Number Portability  Local Number Portability (1 per port	UEPPB UEPP	PR LNPCX	0.35	0	0					
B-CHANNEL USER PROFILE ACCESS:	02113 0211	IX EIII OX	0.00							
CVS/CSD (DMS/5ESS)	UEPPB UEPP	PR U1UCA	0	0	0					
CVS (EWSD)	UEPPB UEPP	R U1UCB	0	0	0					
CSD	UEPPB UEPP	R U1UCC	0	0	0					
B-CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)										
USER TERMINAL PROFILE										
User Terminal Profile (EWSD only)	UEPPB UEPP	PR U1UMA	0	0	0					
VERTICAL FEATURES										
All Vertical Features - One per Channel B User Profile	UEPPB UEPP	PR UEPVF	2.17	0	0					
INTEROFFICE CHANNEL MILEAGE										
Interoffice Channel mileage each, including first mile and facilities termination	UEPPB UEPP	R M1GNC	19.79	42.69	28.66	16.51	6.34	10.73	1	1.65
Interoffice Channel mileage each, additional mile	UEPPB UEPP	R M1GNM	0.0084	0	0			10.73	1	1.65
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT										
UNE Port/Loop Combination Rates										
4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	1 UEPPP 2 UEPPP 3 UEPPP		148.57 175.24 260.73							
UNE Loop Rates										Ŧ
4-Wire DS1 Digital Loop - UNE Zone	1 UEPPP	USL4P	69.22					10.73	1.65	
4-Wire DS1 Digital Loop - UNE Zone 4-Wire DS1 Digital Loop - UNE Zone	2 UEPPP 3 UEPPP	USL4P USL4P	95.89 181.38					10.73 10.73	1.65	
UNE Port Rate										
Exchange Ports - 4-Wire ISDN DS1 Por  NONRECURRING CHARGES - CURRENTLY COMBINED	UEPPP	UEPPP	79.35					10.73	1.65	5
4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion -Switch-as-is	UEPPP	USACP	0	61.25	55.34	1		10.73	1.65	

Attachment 2 Exhibit C

ADDITIONAL NR	RCe .										
	Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within		LIEDDO	DD-77-		0.4070	1			40.70	4.05
	d Allowance		UEPPP	PR7TF		0.4879				10.73	1.65
	Nire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All										
Stat	ates except NC		UEPPP	PR7TO		11.46	11.46			10.73	1.65
4-W	Vire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos										
	ove Std Allowance		UEPPP	PR7ZT		22.92	22.92			10.73	1.65
LOCAL NUMBER	R PORTABILITY										
	cal Number Portability (1 per port		UEPPP	LNPCN	1.75						
LUC	tal Number Fortability (1 per por		UEFFF	LINFCIN	1.75						
INTERESTOR (D.				-							
	rovsioning Only)										
	ice/Data		UEPPP	PR71V	0	0	0				
Digi	gital Data		UEPPP	PR71D	0	0	0				
Inwa	vard Data		UEPPP	PR71E	0	0	0				
New or Addition	nal "B" Channel										-
	w or Additional - Voice/Data B Channel		UEPPP	PR7BV	0	13.96				10.73	1.65
					-						
	w or Additional - Digital Data B Channel	1	UEPPP	PR7BF	0	13.96				10.73	1.68
	w or Additional Inward Data B Channel		UEPPP	PR7BD	0	13.96				10.73	1.65
	w or Additional Useage Sensitive Voice Data B Channel		UEPPP	PR7BS	0	13.96				10.73	19.99
Nev	w or Additional Useage Sensitive Digital Data B Channel		UEPPP	PR7BU	0	13.96				10.73	1.65
	· · · · · · · · · · · · · · · · · · ·				-						
CALL TYPES		1		1			1				
Inw	ward	+	UEPPP	PR7C1	0	0	0		+		
		1									
	tward	$\perp$	UEPPP	PR7C0	0	0	0				
Two	o-way		UEPPP	PR7CC	0	0	0				
Interoffice Chan											
Fixe	ed Each Including First Mili		UEPPP	1LN1A	91.04	95.15	88.78	16.74 14.85		10.73	1.65
Eac	ch Airline-Fractional Additional Mil		UEPPP	1LN1B	0.171						
4-WIRE DS1 DIG	GITAL LOOP WITH 4-WIRE DDITS TRUNK PORT										
LINE Port/Loon	Combination Rates			1							
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	1	UEPDC	-	121.95					10.73	1.65
				-							
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	2	UEPDC		148.62					10.73	1.65
4W	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	3	UEPDC		234.11					10.73	1.65
<b>UNE Loop Rates</b>											
4-V	Wire DS1 Digital Loop - UNE Zone	1	UEPDC	USLDC	69.22					10.73	1.65
4-V	Wire DS1 Digital Loop - UNE Zone	2	UEPDC	USLDC				10.22		10.73	1.65
	Wire DS1 Digital Loop - UNE Zone	3	UEPDC	USLDC	181.38					10.73	1.65
UNE Port Rate											
4 M	Vire DDITS Digital Trunk Por		UEPDC	UDD1T	52.73					10.73	1.65
4-11	Wife DDTS Digital Trunk For		OLI DO	ODDII	32.13					10.73	1.00
NONDECLIDEIN	NG CHARGES - CURRENTLY COMBINED	+		1			1				
		-	LIEDDO	LICACI		74.00	40.44			40.70	4.05
	Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-	1	UEPDC	USAC4		71.29	42.11			10.73	1.65
	Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with			l							
	1 Changes		UEPDC	USAWA		71.29	42.11			10.73	1.65
	Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with						1				
Cha	ange - Trunk		UEPDC	USAWB		71.29	42.11			10.73	1.65
ADDITIONAL NR									$\neg$		
4-V	Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel										
	tivation/Chan - 2-Way Truni		UEPDC	UDTTA		14.14	14.14			10.73	1.65
	Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-								+		
	ay Outward Trunk		UEPDC	UDTTB		14.14	14.14			10.73	1.65
	Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan	1	OLI DO	30110		17.17	17.17		+	70.70	1.00
	vard Trunk w/out DIE		UEPDC	UDTTC		14.14	14.14			10.73	1.65
		+	OLFDO	00110		19.14	14.14		+	10.73	1.03
	Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -		HEDDO	LIDTES		4444	4444			40.70	4.05
	vard Trunk with DIC	$\perp$	UEPDC	UDTTD		14.14	14.14			10.73	1.65
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way										
	D w User Trans		UEPDC	UDTTE		14.14	14.14			10.73	1.65
BIPOLAR 8 ZER	RO SUBSTITUTION										
							1				
B82	ZS -Superframe Format		UEPDC	CCOSF		0	655			10.73	 1.65
B82	ZS - Extended Superframe Forma		UEPDC	CCOEF		0	655			10.73	1.65
				1		-					
		-									
Alternate Mark I	Inversion										
Alternate Mark I	Inversion										

AMI - Extended SuperFrame Forma			UEPDC	MCOPC	)	0	0				
Telephone Number/Trunk Group Establisment Charges											
Telephone Number for 2-Way Trunk Grou			UEPDC	UDTGX	0					10.73	
Telephone Number for 1-Way Outward Trunk Grou			UEPDC	UDTGX					-	10.73	
Telephone Number for 1-Way Inward Trunk Group Withou	DI		UEPDC	UDTGZ						10.73	
DID Numbers, Establish Trunk Group and Provide First Gr			UEPDC	NDZ	0	0	0			10.73	
DID Numbers for each Group of 20 DID Number	up of 20 BIB Harribe		UEPDC	ND4	0	Ů				10.73	
DID Numbers, Non- consecutive DID Numbers , Per Numb			UEPDC	ND5	0					10.73	
Reserve Non-Consecutive DID Nos			UEPDC	ND6	0	0	0			10.73	
Reserve DID Numbers			UEPDC	NDV	0	0	0			10.73	
Dedicated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS		runk P									
Interoffice Channel Mileage - Fixed rate 0-8 miles (Faciliti			UEPDC	1LNO1		95.16	88.78	16.74 14.85		10.73	1.65
Interoffice Channel Mileage - Additional rate per mile - 0-	mil		UEPDC	1LNOA		0	0				
Interoffice Channel Mileage - Fixed rate 9-25 miles (Facili	es Terminatio		UEPDC	1LNO2		0	0				
Interoffice Channel Mileage - Additional rate per mile - 9-2	5 mil		UEPDC	1LNOB		0	0				
Interoffice Channel Mileage - Fixed rate 25+ miles (Facilit			UEPDC	1LNO3		0	0	0			
Interoffice Channel Mileage - Additional rate per mile - 25	· mil		UEPDC	1LNOC		0	0				
Local Number Portability, per DS0 Activate Central Office Termininating Poir			UEPDC UEPDC	LNPCP	3.15 0	0	0	0			
Central Office Termininating Poir			UEPDC	CIG	U						
				ļ	+	+			+	1	
4-WIRE DS1 LOOP WITH CHANNELIZATION WITH PORT				1	1	1			+		
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Act	vations			1	1	1			+		
Each System can have up to 24 combinations of rates depending on											
Lacin Oysicin can have up to 24 combinations of rates depending on	Type and number of ports used			ļ	+	+			+	1	
UNE DS1 Loop				<del>                                     </del>	1					1	
4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	60.22	0	0				
4-Wire DS1 Loop - UNE Zone 1			UEPMG	USLDC		0	0				
4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3			UEPMG	USLDC		0					
4-Wire DST Loop - UNE Zone 3		3	UEPING	USLDC	181.38	U	0				
UNE DSO Channelization Capacities (D4 Channel Bank Configuration	->										
	S)		LIEBMO	1// 18404							
24 DSO Channel Capacity - 1 per DS1			UEPMG UEPMG	VUM24	121.31	0	0				
48 DSO Channel Capacity - 1 per 2 DS1s					242.62	0	0				
96 DSO Channel Capacity -1per 4 DS1s			UEPMG		485.24	0	0				
144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	727.86	0	0				
192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG		970.48	0	0				
240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20		0	0				
288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG		1455.72	0	0				
384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG		1940.96	0	0				
480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG		2426.2	0	0				
576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG		2911.44	0	0				
672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3396.68	0	0				
Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with				ystem							
A Minimum System configuration is One (1) DS1, One (1) D4 Channe											
Multiples of this configuration functioning as one are considered Ad											
NRC - Conversion (Currently Combined) with or without Be			UEPMG	USAC4	0	72.61	3.82			10.73	1.65
System Additions at End User Locations Where 4-Wire DS1 Loop wit	Channelization with Port Combin	nation	Currently Exists ar	d							
New (Not Currently Combined) In Georgia & Tennessee Only						1					
NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port a	d Assoc Feature Activation -			L				l l			
New GA & TN Only			UEPMG	VUMD4	0	726.11	468.21	145.32 17.24		10.73	1.65
Bipolar 8 Zero Substitution									1		
				l							
Clear Channel Capability Format, superframe - Subseque	t Activity Only		UEPMG	CCOSF	0	0	655			10.73	1.65
				l							
Clear Channel Capability Format - Extended Superframe	Subsequent Activity Only		UEPMG	CCOEF	0	0	655			10.73	1.65
Alternate Mark Inversion (AMI)									1		
					L	L			1		
Superframe Format			UEPMG	MCOSF		0	0				
Extended Superframe Format			UEPMG	MCOPC	0	0	0		1		
						1					
Exchange Ports Associated with 4-Wire DS1 Loop with Channelization	n with Port										
Exchange Ports											
				l	1	1			1		
Line Side Combination Channelized PBX Trunk Port - Bus	ness		UEPPX	UEPCX	1.34	0	0	0 0		10.73	1.65
Line Side Outward Channelized PBX Trunk Port - Busines			UEPPX	UEPOX		0	0	0 0		10.73	1.65
Line Side Inward Only Channelized PBX Trunk Port without	DID		UEPPX	UEP1X	1.34	0	0	0 0		10.73	1.65
			UEPPX	UEPDM	0 01	lo.	0	10	1	10.73	1.65
2-Wire Trunk Side Unbundled Channelized DID Trunk Por			UEPPX	OEFDIVI	0.01	U	U	lo lo		10.73	1.00

	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX	1PQWM 0	.66 25	5.4	13.41	3.96	3.93	10.73	1.65	
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		UEPPX	1PQWU 0	.66 78	3.16	18.42	56.03	10.95	10.73	1.65	
Telepho	one Number/ Group Establishment Charges for DID Service											
	DID Trunk Termination (1 per Port)		UEPPX	NDT 0	1					10.73		
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)		UEPPX	NDZ 0	0		0			10.73		
	DID Numbers - groups of 20 - Valid all States		UEPPX	ND4 0	0		0			10.73		
	Non-Consecutive DID Numbers - per number		UEPPX	ND5 0	0		0			10.73		
	Reserve Non-Consecutive DID Numbers		UEPPX	ND6 0	0		0			10.73		
	Reserve DID Numbers		UEPPX	NDV 0	0		0			10.73		
Local No	umber Portability		LIEBBY .				_					
	Local Number Portability - 1 per port		UEPPX	LNPCP 3	.15 0		0					
	RES - Vertical and Optional											
Local St	witching Features Offered with Line Side Ports Only		LIEDOV									
	All Features Available		UEPPX	UEPVF 2	.17 0		0			10.73	1.65	
											+	
INDI ED DODE	LOOP COMBINATIONS - MARKET RATES						-					
UNDLED PORT	LOOP COMBINATIONS - MARKET RATES										+	
Market F	Rates shall apply where BellSouth is not required to provide unbundled local switching of	r ouitab parta pa	r ECC and/or Stat	to Commission	ruloo							
	rates shall apply where belisouth is not required to provide unbuildied local switching to scenarios include:	or switch ports pe	FCC and/or Stat	le Commission	Tules.							
1 Unbu	undled port/loop combinations that are Not Currently Combined in all of the BellSouth s	tates except as n	oted for Georgia	and Tennesse	ι Δ							
	undled port/loop combinations that are Currently Combined or Not Currently Combined					s with 4 or mo	re DS0 equivale	nt lines				
The Ton	o 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); L	A (New Orleans)	· NC (Greenshore	-Winston Sale	em-Highnoint/C	harlotte-Gasto	nia-Rock Hill):	TN (Nashvill	e)			
1110 100	o o mono in Bollodano rogion are. Le Conando, La Eddaordalo, illianinj, ori (illiania), E	ar (iron orioano)	, 110 (01001100010	, motor out	Jiii Tiigiipoiite C	nanotto Gaott	71110 1 (OOK 1 1111);	(1440)	ОД.			
BallSout	th currently is developing the billing capability to mechanically bill the recurring and nor	-recurring Marko	t Pates in this sec	tion In the in	torim BallSout	h chall hill the	rates in the Co	t-Basad sa	ction proceding in li	au of the Market Pates and reser	nues the right to true-up the	hilling di
	rket Rate for unbundled ports includes all available features in all states.	-recurring warke	i itales ili lilis sec	Zuon. In the In	iteriiri, Delioout	II SHAII DIII IHE	Tates III the Co.	SI-Daseu se	ction preceding in in	ed of the warket reales and reser	ives the right to true-up the	billing u
End Offi	ice and Tandem Switching Usage and Common Transport Usage rates in the Port secti	on of this rate ev	hihit shall annly to	all combination	one of loon/nor	t network elen	nents except fo	r LINE Coin	Port/Loon Combina	tions which have a flat rate usan	e charge (LISOC: LIRECLI)	
End Offi	Currently Combined scenarios where Market Rates apply, the Nonrecurring charges are	listed in the Fire	t and Additional N	IBC solumns f	or oach Dort II	POC For Cur	rently Combine	d cooperion	the Monropurring of	pergos ere lieted in the NBC. Cu	reputh Combined section	Λ ddition
	ply also and are categorized accordingly.	e listeu ili tile Fils	i and Additional N	NAC COIGITIES I	or each Foll of	SOC. FOI Cui	rently Combine	a scenanos,	the Nonreculting Ci	larges are listed in the NKC - Ct	Trentily Combined Section. A	Addition
illay app	pry also and are categorized accordingly.											
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)											
	TOTAL DIGITAL TOTAL (N.E.D.)											
UNE Po	rt/Loop Combination Rates											
OIL I O	2-Wire VG Loop/Port Combo - Zone	1			25.89							
	2-Wire VG Loop/Port Combo - Zone :	2			30.03							
	2-Wire VG Loop/Port Combo - Zone	3			43.33							
UNE Loc	on Pates											
		1	UEPRX	UEPLX	11.89							
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPRX UEPRX	UEPLX UEPLX	11.89 16.03							
	2-Wire Voice Grade Loop (SL1) - Zone											
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRX	UEPLX	16.03							
2-Wire V	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone Voice Grade Line Port (Res)	2	UEPRX UEPRX	UEPLX UEPLX	16.03 29.33							
2-Wire V	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRX	UEPLX	16.03	90	90			10.73	1.65	
2-Wire V	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc	2	UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL	16.03 29.33							
2-Wire V	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone Voice Grade Line Port (Res)	2	UEPRX UEPRX	UEPLX UEPLX	16.03 29.33	90	90			10.73	1.65	
2-Wire \	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re	2	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	16.03 29.33 14	90	90			10.73	1.65	
2-Wire \	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC	16.03 29.33 14 14	90 90	90			10.73	1.65	
2-Wire \	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	16.03 29.33 14 14 14 14	90 90 90	90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
2-Wire V	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC	16.03 29.33 14 14	90 90	90			10.73	1.65	
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LU1)	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	16.03 29.33 14 14 14 14	90 90 90	90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LUI)	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAF UEPAP	14 14 14 14 14 14 14 14	90 90 90	90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundles res, low usage line port with Caller ID (LU1)	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	16.03 29.33 14 14 14 14	90 90 90	90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LU)  NUMBER PORTABILITY Local Number Portability (1 per port	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAF UEPAP	14 14 14 14 14 14 14 14	90 90 90	90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LUI)  NUMBER PORTABILITY Local Number Portability (1 per port	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAF UEPAF UEPAF	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90	90 90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LU)  NUMBER PORTABILITY Local Number Portability (1 per port	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAF UEPAP	14 14 14 14 14 14 14 14	90 90 90	90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LUI  NUMBER PORTABILITY Local Number Portability (1 per porl	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPAF UEPAF UEPAF UEPAP	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90	90 90 90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  Voice Grade Line Port (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LUI)  NUMBER PORTABILITY Local Number Portability (1 per port	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAF UEPAF UEPAF	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90	90 90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per port  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAF UEPAF UEPVF	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90	90 90 90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LUI)  NUMBER PORTABILITY  Local Number Portability (1 per port  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPAF UEPAF UEPAF UEPAP	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90	90 90 90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per port  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAF UEPAF UEPVF	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90	90 90 90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundles res, low usage line port with Caller ID (LUI)  NUMBER PORTABILITY  Local Number Portability (1 per port  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAF UEPAF UEPVF	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90	90 90 90 90 90			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID (LUI)  NUMBER PORTABILITY Local Number Portability (1 per port  RES All Features Offerec 2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chan	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID (LUI)  NUMBER PORTABILITY Local Number Portability (1 per port  RES All Features Offerec 2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chan	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY Local Number Portability (1 per port  RES All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chan	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per porl  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chan-  DNAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID - r 2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY Local Number Portability (1 per port  RES All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chan  DNAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 0.35	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per porl  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chan  DNAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  1	2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 14 0.35	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per porl  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chani  DNAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  Int/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 0.35 0	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per porl  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chan  DNAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  1	2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 10.35	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC 2-WIRE UNE Poi	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per porl  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chani  DNAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  Int/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 0.35 0	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	
LOCAL FEATUR ADDITIC 2-WIRE UNE Poi	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID - r  2-Wire voice unbundled Florida Area Calling with Caller ID (LUI  NUMBER PORTABILITY  Local Number Portability (1 per porl  RES  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination - Switch-as- 2-Wire Voice Grade Loop / Line Port Combination - Switch with chani  DNAL NRCs  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)  rt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRC UEPAF UEPAF UEPAF UEPAF UEPAF UEPAC UEPAF UEPAC	16.03 29.33 14 14 14 14 14 14 0.35 0	90 90 90 90 90 0 41.5 41.5	90 90 90 90 90 0 41.5			10.73 10.73 10.73	1.65 1.65 1.65	

2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	29.33						
2-Wire Voice Grade Line Port (Bus)										
2-Wire voice Grade Line Port (Bus)  2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	14	90	90			10.73	1.65
2-Wile Voice unbuilded port without Callet 10 - bt		OLI BX	OLFBL	14	90	30			10.73	1.03
2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBX	UEPBC	14	90	90			10.73	1.65
2 Wire union unb und la dinasti autonion anh.		UEPBX	UEPBO	14	90	90			10.73	1.65
2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	14	90	90			10.73	1.65
OCAL NUMBER PORTABILITY										
Local Number Portability (1 per port		UEPBX	LNPCX	0.35						
EATURES										
-EATONES										
NONRECURRING CHARGES - CURRENTLY COMBINED										
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-		UEPBX	USAC2		41.5	41.5				
2 Wire Vales Conda Lang (Line Boot Combination Could with above		UEPBX	USACC		41.5	41.5				
2-Wire Voice Grade Loop / Line Port Combination - Switch with change		UEPBA	USACC		41.5	41.5				
ADDITIONAL NRCs										
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPBX	USAS2		0	0				
NUMBER VOICE OF ARE LOOP WITH A WIRE LINE PORT (REG. PRV)										
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)										
JNE Port/Loop Combination Rates										
2-Wire VG Loop/Port Combo - Zone	1			25.89						
2-Wire VG Loop/Port Combo - Zone	2			30.03	-					
2-Wire VG Loop/Port Combo - Zone	3			43.33				1		
JNE Loop Rates										
2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRG	UEPLX							
2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRG	UEPLX	16.03						
2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRG	UEPLX	29.33						
2-Wire Voice Grade Line Port Rates (RES - PBX)										
THE TOISE STAND LINE FOR THAISE (N.25 T. 27)										
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Rε		UEPRG	UEPRD	14	90	90			10.73	1.65
Local Number Portability Local Number Portability (1 per port		UEPRG	LNPCP	3.15						
Local Number Fortability (1 per por		OLFING	LINI CI	3.13						
EATURES										
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-		UEPRG	USAC2		41.5	41.5				
2-Wile Voice Grade Loop/ Line Fort Combination - Switch-As-		OLFING	USAUZ		41.5	41.5				
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change		UEPRG	USACC		41.5	41.5				
ADDITIONAL NRCs										
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-										
Nonrecurring					0	0				
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					7.09	7.09			10.73	1.
WIDE VOICE OR ADE LOOP WITH A WIDE LINE DORT (DUC. DDV)										
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)										
JNE Port/Loop Combination Rates										
2-Wire VG Loop/Port Combo - Zone	1			25.89						
2-Wire VG Loop/Port Combo - Zone :	2			30.03						
2-Wire VG Loop/Port Combo - Zone	3			43.33		1				
JNE Loop Rates										
2-Wire Voice Grade Loop (SL1) - Zone	1	UEPPX	UEPLX	11.89						
2-Wire Voice Grade Loop (SL1) - Zone	2	UEPPX UEPPX	UEPLX	16.03 29.33				1		
2-Wire Voice Grade Loop (SL1) - Zone	3	UEPPX	UEPLX	29.33		1				
P-Wire Voice Grade Line Port Rates (BUS - PBX)										
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı		UEPPX	UEPPC	14	90	90			10.73	1.65
Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	14	90	90			10.73	1.65
Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX	UEPP0	14	90	90			10.73	1.65
2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	14	90	90			10.73	1.65
2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	14	90	90		$\perp$	10.73	1.65
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	14	90	90			10.73	1.65
2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	14	90	90			10.73	
						- 50				
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	14	90	90	1 1	1	10.73	1.65

UEPXL UEPXS	14 14 14 14 3.15 3.15 25.89 30.03 43.33	90 90 90 90 90 41.5 41.5 0 0 7.09	90 90 90 90 90 41.5 41.5 0 0 7.09		10.73 10.73 10.73 10.73 10.73	1.65 1.65 1.65 1.65
UEPXN UEPXS	14 14 14 3.15 25.89 30.03	90 90 90 90 41.5 41.5	90 90 90 41.5 41.5		10.73 10.73 10.73	1.65 1.65 1.65
UEPXC UEPXS USAC2 USAC2 USAS2	14 14 3.15 25.89 30.03	90 90 90 41.5 41.5 0	90 90 41.5 41.5 0		10.73	1.65
UEPXC UEPXS USAC2 USAC2 USAS2	14 14 3.15 25.89 30.03	90 90 90 41.5 41.5 0	90 90 41.5 41.5 0		10.73	1.65
LNPCF  USAC2  USAC2	14 3.15 25.89 30.03	41.5 41.5 0	41.5 41.5 0		10.73	1.65
LNPCF  USAC2  USAC2	14 3.15 25.89 30.03	41.5 41.5 0	41.5 41.5 0		10.73	1.65
USACCUSACCUSACCUSACCUSACCUSACCUSACCUSAC	3.15 3.15 25.89 30.03	41.5 41.5 0	41.5 41.5 0			
USAC2 USAC2 USAS2	25.89	0 0	0 0		10.73	1.6
USAC2 USAC2 USAS2	25.89	0 0	0 0		10.73	1.6
USACC USAS2	30.03	0 0	0 0		10.73	1.6
USACC USAS2	30.03	0 0	0 0		10.73	1.6
USACC USAS2	30.03	0 0	0 0		10.73	1.6
USACC USAS2	30.03	0 0	0 0		10.73	1.6
USACC USAS2	30.03	0 0	0 0		10.73	1.6
USAS2	30.03	0	0		10.73	1.6
	30.03	0	0		10.73	1.4
	30.03	0	0		10.73	1.6
LIEDLY	30.03				10.73	1.6
LIEDLY	30.03				10.73	1.6
LIEDLY	30.03	7.09	7.09		10.73	1.6
LIEDLY	30.03					
LIEDLY	30.03					
LIEPLY	30.03					
LIEDI	30.03					
HERIX	30.03					
HEDLY						
LIEDLY			1			1
LIEDLY						
LIED! V						
IUEPLX	11.89					
UEPLX	16.03					
) UEPLX	29.33					
-+-						
-+-						+
UEP2F	14	90	90		10.73	1.6
						1.6
					1300	
UEPCC	14	90	90		10.73	1.6
UEPR	14	90	90		10.73	1.6
UEPOF	14	90	90		10.73	1.6
UEPCC	14	90	90		10.73	1.65
LNDO	0.25					
LNPCX	0.35					
USAC2		41.5	41.5	+		<del>                                     </del>
USACC		41.5	41.5			
USAS2		0	0			
		1	<u> </u>			
)		UEPCG 14 D UEPK 14 D UEPOF 14 D UEPCQ 14 D UEPCQ 14 D UEPCQ 14	0 UEPCG 14 90 14 90 15 16 16 17 17 18 18 18 18 18 18 18 18 18 18 18 18 18	0 UEPCG 14 90 90 90 00 UEPRK 14 90 90 90 00 00 UEPOF 14 90 90 90 00 00 UEPCQ 14 90 90 90 00 00 UEPCQ 14 90 90 90 00 UEPCQ 14 90 90 90 00 00 USACZ 41.5 41.5 00 USACZ 41.5 41.5	0 UEPCG 14 90 90 90	0 UEPCG 14 90 90 10.73 0 UEPK 14 90 90 10.73 0 UEPOF 14 90 90 10.73 0 UEPCQ 14 90 90 10.73 0 UEPCQ 14 90 90 10.73 0 UEPCQ 14 90 90 10.73 0 USAC2 41.5 41.5

RY	NOTES	UNBUNDLED NETWORK ELEMENT Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
							Nonra	curring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR			Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increr Cha Manu Orde Elect Disc
-							Nonre	curring					connect			
-						Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOI
-						Rec	FIISL	Auu i	FIISL	Add I	SOWIEC	SUMAN	SOWAN	SUMAN	SOWAN	30
-+																1
		shown in the sections for stand-alone loops or loops as part of a combination refers to Geograp terconnection.bellsouth.com/become_a_clec/html/interconnection.htm	hically	Deaveraged UNE 2	Zones. To	view Geograp	phically Deavera	ged UNE Zone	Designations	s by Central O	ffice, refer	to Internet V	/ebsite:	T		
LED	EXCHANGE	E ACCESS LOOP														
Ŧ																
2	2-WIRE ANA	LOG VOICE GRADE LOOP														
- 17		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	1	UEANL	UEAL2	14.21	42.54	31.33					18.94	8.42		
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	2	UEANL	UEAL2	16.41	42.54	31.33					18.94	8.42		
-+		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	3	UEANL	UEAL2	26.08	42.54	31.33					18.94	8.42		
-+		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone	1	UEPSR, UEPSB	UEALS	14.21	42.54	31.33					18.94	8.42		
-+		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zoni 2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zoni	2	UEPSR, UEPSB		16.41	42.54	31.33	+				18.94	8.42	1.65	1
+		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zoni 2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zoni		UEPSR, UEPSB		26.08	42.54		1				18.94	8.42	1.05	+
+		Engineering Information Document (E	3	UEANL	UEALS	20.00	28.72	31.33 28.72	1				10.94	0.42		+
+		Engineering information Document (E		UEANL	1		20.12	20.12	+					1		1
		Manual Order Coordination for UVL-SL1s (per loop		UEANL	UEAMC		36.46	36.46	1							
+		ivianuai Oruei Guoruination foi UVL-SETS (per 100p		UEANL	UEAIVIC		30.40	30.40	+	1						+
		Order Coordination for Pagaified Conversion Time for LIV/L SL1 (par LSD		UEANL	OCOSL		34.22	34.22	1							1
+		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR		UEANL	UCUSL		34.22	34.22	+	1						+
+		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -			1				+							$\vdash$
				UEA	UEAL2	40.04	104.17	70.4					40.04	0.40		
+		Zone 1	1	UEA	UEALZ	16.84	104.17	78.1					18.94	8.42		-
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 2	2	UEA	UEAL2	19.45	104.17	78.1					18.94	8.42		
+			2	UEA	UEAL2	19.45	104.17	78.1					18.94	8.42		-
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -	_													
_		Zone 3	3	UEA	UEAL2	30.92	104.17	78.1					18.94	8.42		_
														ļ		
4		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		35.74									
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone														
_		1	1	UEA	UEAR2	16.84	104.17	78.1					18.94	8.42		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone														
_		2	2	UEA	UEAR2	19.45	104.17	78.1					18.94	8.42		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone														
4		3	3	UEA	UEAR2	30.92	104.17	78.1					18.94	8.42		
														ļ		
		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		35.74									
4		LOG VOICE GRADE LOOP														
		4-Wire Analog Voice Grade Loop - Zone	1	UEA	UEAL4	22.26	206.95	170.57					18.94	8.42		
		4-Wire Analog Voice Grade Loop - Zone	2	UEA	UEAL4	25.7	206.95	170.57					18.94	8.42		
		4-Wire Analog Voice Grade Loop - Zone	3	UEA	UEAL4	40.86	206.95	170.57					18.94	8.42		
				-	1				1							1
Ш		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		35.74	<u> </u>	1	<u> </u>				<u> </u>		L
2		N DIGITAL GRADE LOOP														╚
I		2-Wire ISDN Digital Grade Loop - Zone	1	UDN	U1L2X	21.89	233.38	180.35					18.94	8.42		
⅃		2-Wire ISDN Digital Grade Loop - Zone	2	UDN	U1L2X	25.27	233.38	180.35					18.94	8.42		Ľ
		2-Wire ISDN Digital Grade Loop - Zone	3	UDN	U1L2X	40.17	233.38	180.35					18.94	8.42		
Т					1		1		1							
		Order Coordination For Specified Conversion Time (per LS	L I	UDN	OCOSL		35.74		1					<u>                                     </u>		L
I																
2	2-WIRE Univ	versal Digital Channel (UDC) COMPATIBLE LOOP		-					1							
T		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1	1	UDC	UDC2X	21.89	44.69	31.55	25.65	7.06			18.94	8.42		
Т		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2	2	UDC	UDC2X	25.27	44.69	31.55	25.65	7.06			18.94	8.42		
1		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 3	3	UDC	UDC2X	40.17	44.69	31.55	25.65	7.06			18.94	8.42		
$\top$		•	-		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,											
2	2-WIRF ASV	MMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP			1				1					$\vdash$		t
ď	AO I	MINISTER DISTRICT SOCIONALINE LINE (ADDE) COM ATIBLE LOOF			1				+							1
		2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOO			1		1		1							
+					1		+		+							+
		Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 1	1	UAL	UAL2X	11.23	44.69	31.55	25.65	7.06			18.94	8.42		
+			- 1	UAL	UMLZÁ	11.23	44.09	31.55	∠0.00	7.00			10.94	0.42		+
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -	2	1161	HALOY	10.07	44.00	24 55	25.05	7.00			10.04	0.40		
+		Zone 2	2	UAL	UAL2X	12.97	44.69	31.55	25.65	7.06		-	18.94	8.42		₩
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -			1141 011	00.00	44.00	04.55	05.05	7.00			40.04	0.40		
		Zone 3	3	UAL	UAL2X	20.62	44.69	31.55	25.65	7.06			18.94	8.42		₩.
+																

2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1	1	1	UAL	UAL2W	11.23	44.69	31.55	25.65	7.06	18.94	8.42	
2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2		2	UAL	UAL2W	12.97	44.69	31.55	25.65	7.06	18.94	8.42	
2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -												
Zone 3		3	UAL	UAL2W	20.62	44.69	31.55	25.65	7.06	18.94	8.42	
Order Coordination for Specified Conversion Time (per LS			UAL	OCOSL		35.74						
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP												
THE HOLD IN THE BOOK OF THE CONTROL	-											
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOO     2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -												
Zone 1  2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -		1	UHL	UHL2X	7.88	44.69	31.55	25.65	7.06	18.94	8.42	
Zone 2		2	UHL	UHL2X	9.09	44.69	31.55	25.65	7.06	18.94	8.42	
2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 3		3	UHL	UHL2X	14.46	44.69	31.55	25.65	7.06	18.94	8.42	
2010 0	-	J	OHE	OTILEX	14.40	44.00	01.00	20.00	7.00	10.54	0.42	
Order Coordination for Specified Conversion Time (per LS)			UHL	OCOSL		35.74						
Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1	1	1	UHL	UHL2W	7.88	44.69	31.55	25.65	7.06	18.94	8.42	
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -		2		UHL2W	9.09	44.69	31.55		7.06	18.94	8.42	
Zone 2  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation	_'_		UHL	UHLZVV	9.09	44.09	31.55	25.65	7.06	18.94	8.42	
Zone 3	_1	3	UHL	UHL2W	14.46	44.69	31.55	25.65	7.06	18.94	8.42	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		35.74						
4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP												
4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 1	1	1	UHL	UHL4X	10.39	44.69	31.55	25.65	7.06	18.94	8.42	
4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2		2	UHL	UHL4X	12	44.69	31.55	25.65	7.06	18.94	8.42	
4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -	-											
Zone 3	1	3	UHL	UHL4X	19.07	44.69	31.55	25.65	7.06	18.94	8.42	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		35.74						
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -												
Zone 1  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation		1	UHL	UHL4W	10.39	44.69	31.55	25.65	7.06	18.94	8.42	
Zone 2	_1	2	UHL	UHL4W	12	44.69	31.55	25.65	7.06	18.94	8.42	
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		3	UHL	UHL4W	19.07	44.69	31.55	25.65	7.06	18.94	8.42	
					10.01		01.00	20.00	7.00	10.01	0.12	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		35.74						
4-WIRE DS1 DIGITAL LOOP												
4-Wire DS1 Digital Loop - Zone		1	USL	USLXX	55.53	429.98	268.18			18.94	8.42	
4-Wire DS1 Digital Loop - Zone :		2	USL	USLXX	64.13	429.98	268.18			18.94	8.42	
4-Wire DS1 Digital Loop - Zone :		3	USL	USLXX	101.93	429.98	268.18			18.94	8.42	
Order Coordination for Specified Conversion Time (per LS			USL	OCOSL		35.74						
4-WIRE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP												
4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	25.75	348.55	241.2			18.94	8.42	
4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	29.74	348.55	241.2			18.94	8.42	
4 Wire Unbundled Digital 19.2 Kbps		3	UDL	UDL19	47.27	348.55	241.2			18.94	8.42	
4 Wire Unbundled Digital Loop 56 Kbps - Zone		1	UDL	UDL56	25.75	348.55	241.2			18.94	8.42	
4 Wire Unbundled Digital Loop 56 Kbps - Zone	$\rightarrow$	2	UDL	UDL56	29.74	348.55	241.2			18.94	8.42	
4 Wire Unbundled Digital Loop 56 Kbps - Zone	$\rightarrow$	3	UDL	UDL56	47.27	348.55	241.2			18.94	8.42	
. Aftic orioditated bigital Loop of Raps - Zotte		J	JDL	CDLOG	71.41	5-5.55	471.4			10.54	0.42	
Order Coordination for Specified Conversion Time (per LS			UDL	OCOSL		35.74						
4 Wire Unbundled Digital Loop 64 Kbps - Zone		1	UDL	UDL64	25.75	348.55	241.2			18.94	8.42	
4 Wire Unbundled Digital Loop 64 Kbps - Zone	T	2	UDL	UDL64	29.74	348.55	241.2			18.94	8.42	
4 Wire Unbundled Digital Loop 64 Kbps - Zone		3	UDL	UDL64	47.27	348.55	241.2			18.94	8.42	
			UDL	OCOSL		35.74						
Order Coordination for Specified Conversion Time (per LS								<u> </u>				
Order Coordination for Specified Conversion Time (per LS	=											
2-WIRE Unbundled COPPER LOOP												
2-WIRE Unbundled COPPER LOOP  2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility		1	LICI	LICI PP	12.02	44.60	21 55	25.65	7.06	10.04	9.42	
2-WIRE Unbundled COPPER LOOP  2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1  2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility		1	UCL	UCLPB	12.02	44.69	31.55	25.65	7.06	18.94	8.42	
2-WIRE Unbundled COPPER LOOP  2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1		1 2 3	UCL UCL UCL	UCLPB UCLPB	12.02 13.88 22.07	44.69 44.69 44.69	31.55 31.55 31.55	25.65 25.65 25.65	7.06 7.06 7.06	18.94 18.94	8.42 8.42 8.42	

		1				ı	1				1	1			
Order Coordination for Unbundled Copper Loops (per les			UCL	UCLMC		16.11	16 11								
Order Coordination for Unbundled Copper Loops (per loc 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility			UCL	UCLIVIC		10.11	16.11								
reservation - Zone '	1	1	UCL	UCLPW	12.02	44.69	31.55	25.65	7.06			18.94	8.42		
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility		-	002	OOLI W	12.02	44.00	01.00	20.00	7.00			10.54	0.42		
reservation - Zone 2	- 1	2	UCL	UCLPW	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility															
reservation - Zone (	- 1	3	UCL	UCLPW	22.07	44.69	31.55	25.65	7.06			18.94	8.42		
Order Coordination for Unbundled Copper Loops (per loop)															
			UCL	UCLMC		16.11	16.11								
2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility															
reservation - Zone '		1	UCL	UCL2L	35.56	44.69	31.55	25.65	7.06			18.94	8.42		
2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2		2	UCL	UCL2L	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			UCL	UCLZL	41.07	44.09	31.00	25.65	7.00			10.94	0.42		
reservation - Zone (		3	UCL	UCL2L	65.28	44.69	31.55	25.65	7.06			18.94	8.42		
							000								
Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		16.11	16.11								
2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility															
reservation - Zone '	- 1	1	UCL	UCL2W	35.56	44.69	31.55	25.65	7.06			18.94	8.42		
2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility															
reservation - Zone 2	ı	2	UCL	UCL2W	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		_	1101	1101.014	05.00	44.00	04.55	05.05	7.00			40.04	0.40		
reservation - Zone ( Order Coordination for Unbundled Copper Loops (per loc	ı	3	UCL UCL	UCL2W UCLMC	65.28	44.69 16.11	31.55 16.11	25.65	7.06			18.94	8.42		
Order Coordination for Oribundled Copper Loops (per loc			UCL	UCLIVIC		10.11	10.11								
<del>-                                      </del>	<u> </u>	1												+	
2-Wire Unbundled Copper Loop Non-Designed - S\	- 1	sw	UEQ	UEQ2X	12.8	44.69	22.4	25.65	7.06			18.94	8.42	-	
Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loc			UEQ	USBMC	-	16.11	16.11								
Engineering Information Documer			UEQ			28.72	28.72								
Loop Testing - Basic 1st Half Hou			UEQ	URET1		78.92	78.92								
Loop Testing - Basic Additional Half Hoι			UEQ	URETA		23.33	23.33								
4-WIRE COPPER LOOP															
4-Wire Copper Loop/Short - including manual service inquiry and facility reservation -															
Zone 1		1	UCL	UCL4S	12.02	44.69	31.55	25.65	7.06			18.94	8.42		
4-Wire Copper Loop/Short - including manual service inquiry and facility reservation -		- '	UCL	UCL45	12.02	44.09	31.00	25.65	7.00			10.94	0.42		
Zone 2		2	UCL	UCL4S	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
4-Wire Copper Loop/Short - including manual service inquiry and facility reservation -		_	002	00210	10.00	11.00	01.00	20.00	7.00			10.01	0.12		
Zone 3		3	UCL	UCL4S	22.07	44.69	31.55	25.65	7.06			18.94	8.42		
Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		16.11	16.11								
4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -															
Zone 1		1	UCL	UCL4W	12.02	44.69	31.55	25.65	7.06			18.94	8.42		
4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -															
Zone 2	ı	2	UCL	UCL4W	13.88	44.69	31.55	25.65	7.06			18.94	8.42		
4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -		3	LICI		20.07	44.69	24.55	25.05	7.00			40.04	0.40		
Zone 3 Order Coordination for Unbundled Copper Loops (per loc	- '	3	UCL	UCL4W UCLMC	22.07	16.11	31.55 16.11	25.65	7.06			18.94	8.42		
4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			UCL	UCLIVIC		10.11	10.11								
reservation - Zone '		1	UCL	UCL4L	35.56	44.69	31.55	25.65	7.06			18.94	8.42		
4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			002	00212	00.00	11.00	01.00	20.00	7.00			10.01	0.12		
reservation - Zone 2		2	UCL	UCL4L	41.07	44.69	31.55	25.65	7.06			18.94	8.42		
4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
reservation - Zone (		3	UCL	UCL4L	65.28	44.69	31.55	25.65	7.06			18.94	8.42		
Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		16.11	16.11								
		1	1	1											
4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility								25.65	7.06			18.94	8.42		
reservation - Zone '	1	1	UCL	UCL40	35.56	44.69	31.55	23.03		1	Į.				
reservation - Zone 4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility	1	1							7.06			10.04	Ω ΔΩ	1	
reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '	I	2	UCL	UCL40	35.56 41.07	44.69 44.69	31.55 31.55	25.65	7.06			18.94	8.42	$\longrightarrow$	
reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility	1	2	UCL	UCL4O	41.07	44.69	31.55	25.65						+	
reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '	1								7.06 7.06			18.94 18.94	8.42 8.42		
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (	1		UCL	UCL40	41.07	44.69 44.69	31.55 31.55	25.65							
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc	1		UCL	UCL40	41.07	44.69 44.69	31.55 31.55	25.65							
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  Order Coordination for Unbundled Copper Loops (per loc	I I		UCL UCL UCL	UCL40	41.07	44.69 44.69	31.55 31.55	25.65							
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal	I I		UCL UCL UCL UCL UCL	UCL40 UCL40 UCLMC	41.07 65.28	44.69 44.69 16.11	31.55 31.55	25.65 25.65	7.06						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ;  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ;  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft	I I to		UCL UCL UCL UCL UCL UCL UAL, UHL, UCL,	UCL4O UCL4O UCLMC	41.07 65.28	44.69 44.69 16.11	31.55 31.55 16.11	25.65 25.65	7.06						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18			UCL UCL UCL UCL UCL	UCL40 UCL40 UCLMC	41.07 65.28	44.69 44.69 16.11	31.55 31.55	25.65 25.65	7.06						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ;  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ;  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft			UCL UCL UCL UCL UAL, UHL, UCL, UEQ, ULS UCL, ULS	UCL40 UCL40 UCLMC ULM2L ULM2L	41.07 65.28 0 0	44.69 44.69 16.11 0 0	31.55 31.55 16.11 0 0	25.65 25.65 0 0	7.06 0 0						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18			UCL UCL UCL UCL UCL UCL UAL, UHL, UCL,	UCL4O UCL4O UCLMC	41.07 65.28	44.69 44.69 16.11	31.55 31.55 16.11	25.65 25.65	7.06						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18 ft			UCL UCL UCL UAL, UHL, UCL, UEQ, ULS UCL, ULS UHL, UCL	UCL4O UCLMC UCLMC ULM2L ULM2G ULM4L	41.07 65.28 0 0	44.69 44.69 16.11 0 0	31.55 31.55 16.11 0 0	25.65 25.65 0 0	7.06 0 0						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18			UCL UCL UCL UCL UAL, UHL, UCL, UEQ, ULS UCL, ULS UHL, UCL	UCL40 UCL40 UCLMC ULM2L ULM2L	41.07 65.28 0 0	44.69 44.69 16.11 0 0	31.55 31.55 16.11 0 0	25.65 25.65 0 0	7.06 0 0						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18 ft			UCL UCL UCL UAL, UHL, UCL, UEQ, ULS UCL, ULS UHL, UCL	UCL4O UCLMC UCLMC ULM2L ULM2G ULM4L	41.07 65.28 0 0	44.69 44.69 16.11 0 0	31.55 31.55 16.11 0 0	25.65 25.65 0 0	7.06 0 0						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18 ft  Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18			UCL UCL UCL UAL, UHL, UCL, UEQ, ULS UCL, ULS UHL, UCL ULS UCL ULS UCL	UCL40 UCL40 UCLMC ULM2L ULM2G ULM4L	0 0 0 0	44.69 44.69 16.11 0 0 0	31.55 31.55 16.11 0 0 0	25.65 25.65 0 0 0	7.06 0 0 0						
reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  Order Coordination for Unbundled Copper Loops (per loc  OP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18 ft  Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18			UCL UCL UCL UAL, UHL, UCL, UEQ, ULS UCL, ULS UHL, UCL ULS UCL ULS UCL	UCL40 UCL40 UCLMC ULM2L ULM2G ULM4L	0 0 0 0	44.69 44.69 16.11 0 0 0	31.55 31.55 16.11 0 0 0	25.65 25.65 0 0 0	7.06 0 0 0						

Sub-Lo	op Distribution														T
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L			UEANL	USBSA		421.08	421.08				18.94	8.42		
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L	i i	+	UEANL	USBSB		67.1	67.1				18.94	8.42		
			+												
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I	1	_	UEANL	USBSC		394.74	394.74				18.94	8.42		
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I			UEANL	USBSD		154.57	154.57				18.94	8.42		
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi	+	+		USBN2	9.12	207.01	171.32				18.94	8.42		+
		+	SW			9.12						18.94	8.42		-
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		34.22	34.22							
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Statewi		SW	UEANL	USBN4	8.32	219.35	72.99	123.72	28.77		18.94	8.42		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		T	UEANL	USBMC		34.22	34.22							
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC		1		USBR2	1.61	137.03	41.59	115.85	19.17		18.94	8.42		1
-	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	<del>- '</del> -	+-	UEANL	USBMC	1.01	34.22	34.22	115.65	13.17		10.54	0.42		+
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC	1		UEANL	USBR4	2.96	176.46	55.11	122.17	19.57		18.94	8.42		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		T	UEANL	USBMC		34.22	34.22							
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone		1	UEF	UCS2X	5.54	175.16	55.5	108.86	24.53		18.84	8.42		1
		<del></del>	2	UEF	UCS2X	5.54	175.16	55.5	108.86	24.53		18.94	8.42		+
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone														_
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone		3	UEF	UCS2X	5.54	175.16	55.5	108.86	24.53		18.94	8.42		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.22	34.22							
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone		1	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77		18.94	8.42		
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone	- i	2	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77		18.94	8.42		+
+		<del></del>	3												+
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone		3	UEF	UCS4X	6.89	219.35	72.99	123.72	28.77		18.94	8.42		+-
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		34.22	34.22							
Sub-Lo	op Feeder	<del></del>	4												+
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-			UEA, UDN,UCL,UDL,UDC	USBFW		\$421.08								
		<b>†</b>	1	UEA,											1
L	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u	<del></del>	4	UDN,UCL,UDL,UDC	USBFX		67.1	67.1							4
	USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		521.57	11.3							
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade- Statew		SW	UEA	USBFA	8.58	206.44	170.05				18.94	8.42		
	Order Coordination for Specified Conversion Time, per LSR		1	UEA	OCOSL		34.22								
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Statew	+	SW	UEA	USBFB	8.58	206,44	170.05				18.94	8.42		+
		+	SW			0.00	34.22	170.00				10.94	0.42		+-
	Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		34.22								
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade Loop -														
	Statewide		SW	UEA	USBFC	8.58	206.44	170.05				18.94	8.42		
	Order Coordination For Specified Conversion Time, per LS			UEA	OCOSL		34.22								
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Statew	+	SW	UEA	USBFD	\$19.91	\$243,41	\$81.32	\$134.77	\$33.93		18.94	8.42		
						*				400.00					
	Order Coordination For Specified Conversion Time, Per LS			UEA	OCOSL		34.22								
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Statew		SW	UEA	USBFE	19.91	\$243.41	\$81.32	\$134.77	\$33.93		18.94	8.42		
		T													
	Order Coordination For Specified Conversion Time, Per LS			UEA	OCOSL		34.22								
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Statewin	+	SW	UDN	USBFF	\$17.73	\$208.50	\$62.31	\$119.68	\$29.58		18.94	8.42		+
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - StateWil		SW	UDIN	USBFF	\$17.73	\$208.50	\$62.31	\$119.00	\$29.58		18.94	8.42		+-
	Order Coordination For Specified Conversion Time, Per L\$			UDN	OCOSL		34.22								
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		SW	UDC	USBFS	17.73	208.5	62.31	119.68	29.58		19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Statewin	+	sw	USL	USBFG	79.3		128.76	124.09	34.8				19.99	
								120.70	124.03	34.0		10 00	10.00		_
	Oribundied Sub-Loop Feeder Loop, 4-Wire DST - Statewin	+	_	002	CODIC		203.69					19.99	19.99	10.00	
												19.99	19.99	10.00	
	Order Coordination For Specified Conversion Time, Per L\$			USL	OCOSL		34.22							70.00	
			SW			7.22		63.15	119.68	29.58		19.99	19.99 8.42	10.00	
	Order Coordination For Specified Conversion Time, Per L\$			USL	OCOSL		34.22	63.15	119.68	29.58				10.00	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi			USL UCL	OCOSL USBFH		34.22 195.38	63.15	119.68	29.58				10.00	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi Order Coordination For Specified Conversion Time, per LS		SW	USL UCL UCL	OCOSL USBFH OCOSL	7.22	34.22 195.38 34.22					18.94	8.42	10.00	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi			USL UCL	OCOSL USBFH		34.22 195.38	63.15 81.32	119.68	29.58				10.00	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik		SW	USL UCL UCL UCL	OCOSL USBFH OCOSL USBFJ	7.22	34.22 195.38 34.22 243.41					18.94	8.42	10.00	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS		SW	USL UCL UCL UCL	OCOSL USBFH OCOSL USBFJ	7.22	34.22 195.38 34.22 243.41 34.22	81.32	134.77	33.93		18.94	8.42		
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS		SW	USL UCL UCL UCL	OCOSL USBFH OCOSL USBFJ	7.22	34.22 195.38 34.22 243.41 34.22	81.32		33.93		18.94	8.42		
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Lox		SW SW	USL UCL UCL UCL UCL UCL UCL	OCOSL USBFJ OCOSL USBFJ OCOSL USBFN	7.22 13.72 24.5	34.22 195.38 34.22 243.41 34.22 243.41	81.32 81.32	134.77	33.93		18.94	8.42 8.42	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS		SW	USL UCL UCL UCL	OCOSL USBFH OCOSL USBFJ	7.22	34.22 195.38 34.22 243.41 34.22	81.32	134.77	33.93		18.94	8.42		
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic		SW SW	USL UCL UCL UCL UCL UCL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO	7.22 13.72 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41	81.32 81.32	134.77	33.93		18.94	8.42 8.42	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewik Order Coordination For Specified Time Conversion, per LS		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL	7.22 13.72 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22	81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic		SW SW	USL UCL UCL UCL UCL UCL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO	7.22 13.72 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41	81.32 81.32	134.77	33.93		18.94	8.42 8.42	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewik Order Coordination For Specified Time Conversion, per LS		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL	7.22 13.72 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22	81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewik Order Coordination For Specified Time Conversion, per LS		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL	7.22 13.72 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22	81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewik		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP	7.22 13.72 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41	81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
Unbuno	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewi  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewi  Order Coordination For Specified Conversion Time, per LS		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP	7.22 13.72 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41	81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
Unbunc	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 243.41 34.22 243.41 34.22	81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewich Corder Coordination For Specified Conversion Time, per LS  Illed Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pa		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP	7.22 13.72 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41	81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewi  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewi  Order Coordination For Specified Conversion Time, per LS  Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewi  Order Coordination For Specified Conversion Time, per LS  Ilded Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pa		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL	OCOSL USBFJ OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP OCOSL	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22	81.32 81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99	8.42 8.42 19.99 19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewich Corder Coordination For Specified Conversion Time, per LS  Illed Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pa		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL	OCOSL USBFH OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 243.41 34.22 243.41 34.22	81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99	8.42 8.42 19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewi  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewi  Order Coordination For Specified Conversion Time, per LS  dled Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pa  k Interface Device (NID) Network Interface Device (NID) - 1-2 line		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL UD	OCOSL USBFH OCOSL USBFN USBFN USBFO OCOSL USBFP OCOSL UENPP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22	81.32 81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99	8.42 8.42 19.99 19.99 19.99 8.42	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 10-10-10-10-10-10-10-10-10-10-10-10-10-1		SW SW SW	USL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL UD	OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP OCOSL UENPP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22	81.32 81.32 81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99 18.94 18.94	8.42 19.99 19.99 19.99 8.42 8.42	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Conversion Time, per LS  Ilded Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pa  k Interface Device (NID)  Network Interface Device (NID) - 1-2 line Network Interface Device (NID) - 1-6 line Network Interface Device (Oros Connect - 2 V		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL UD	OCOSL USBFH OCOSL USBFN USBFO OCOSL USBFP OCOSL USBFP USBFP USBFP UENPP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22 82.48	81.32 81.32 81.32 81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99	8.42 8.42 19.99 19.99 19.99 8.42	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewic  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewic  Order Coordination For Specified Conversion Time, per LS  sub-Loop Feeder - Per 4-Wire 10-10-10-10-10-10-10-10-10-10-10-10-10-1	1	SW SW SW	USL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL UD	OCOSL USBFJ OCOSL USBFN USBFO OCOSL USBFP OCOSL UENPP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22	81.32 81.32 81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99 18.94 18.94	8.42 19.99 19.99 19.99 8.42 8.42	19.99	
Network	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Conversion Time, per LS  dled Network Terminating Wire (UNTW)  Unbundled Network Terminating Wire (UNTW) per Pa  k Interface Device (NID) Network Interface Device (NID) - 1-2 line Network Interface Device (NID) - 1-6 line Network Interface Device Cross Connect - 2 V Network Interface Device Cross Connect - 4V		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL UD	OCOSL USBFH OCOSL USBFN USBFO OCOSL USBFP OCOSL USBFP USBFP USBFP UENPP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22 82.48	81.32 81.32 81.32 81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99 18.94 18.94	8.42 19.99 19.99 19.99 8.42 8.42	19.99	
Network	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Ktps Digital Grade Lov Sub-Loop Feeder - Per 4-Wire 5k Ktps Digital Grade Loop - Statewi  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 5k Ktps Digital Grade Loop - Statewi  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Ktps Digital Grade Loop - Statewi  Order Coordination For Specified Conversion Time, per LS  Idled Network Terminating Wire (UNTW) Unbundled Network Terminating Wire (UNTW) per Pa  k Interface Device (NID) Network Interface Device (NID) - 1-2 line Network Interface Device (NID) - 1-6 line Network Interface Device Cross Connect - 2V Network Interface Device Cross Connect - 4V  CONCENTRATION		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL UD	OCOSL USBFH OCOSL USBFN USBFN USBFP OCOSL USBFP OCOSL USBFP UND12 UND12 UND12 UND12 UND12	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22 243.41 34.22 6.15	\$1.32 \$1.32 \$1.32 \$1.32 \$1.32 \$2.48 \$2.48 \$6.69 98.21 6.15 6.15	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99 18.94 18.94 18.94	8.42 19.99 19.99 19.99 8.42 8.42 8.42	19.99 19.99	
Network	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Statewi  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Statewik  Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loo Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Time Conversion, per LS Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Statewik  Order Coordination For Specified Conversion Time, per LS  dled Network Terminating Wire (UNTW)  Unbundled Network Terminating Wire (UNTW) per Pa  k Interface Device (NID) Network Interface Device (NID) - 1-2 line Network Interface Device (NID) - 1-6 line Network Interface Device Cross Connect - 2 V Network Interface Device Cross Connect - 4V		SW SW SW	USL UCL UCL UCL UCL UDL UDL UDL UDL UDL UDL UDL UDL UDL UD	OCOSL USBFH OCOSL USBFN USBFO OCOSL USBFP OCOSL USBFP USBFP USBFP UENPP	7.22 13.72 24.5 24.5 24.5	34.22 195.38 34.22 243.41 34.22 243.41 243.41 34.22 243.41 34.22 243.41 34.22 82.48	81.32 81.32 81.32 81.32 81.32 81.32 81.32	134.77 134.77 134.77	33.93 33.93 33.93 33.93		18.94 18.94 19.99 19.99 19.99 18.94 18.94	8.42 19.99 19.99 19.99 8.42 8.42	19.99	

	Unbundled Loop Concentration - System A (TR30:		ULC	UCT3A		650.81	650.81				19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System B (TR30)		ULC	UCT3B	89.26	271.17	271.17				19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - DS1 Loop Interface Ca	-	ULC	UCTCO	5.04	126.57	92.14	33.57	9.4		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Car	-	UDN	ULCC1	8	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - UDC Loop Interface (Brite Car	-	UDC	ULCCU	8	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop				_									
	Interface (POTS Card)	-	UEA	ULCC2	2	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface													
	(SPOTS Card)		UEA	ULCCR	11.89	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca		UEA	ULCC4		21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - TEST CIRCUIT Cai		ULC	UCTTC		21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa		UDL	ULCC7	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa		UDL	ULCC5	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa		UDL	ULCC6	10.51	21.07	20.96	10.78	10.71		19.99	19.99	19.99	19.99
UNDUNDUE	D SUB-LOOP CONCENTRATION (OUTSIDE CO)													
UNBUNDLEL	D SUB-LOOP CONCENTRATION (OUTSIDE CO)													
UNE OTHER	R, PROVISIONING ONLY - NO RATE													
OILE OTHER	, THE THE STATE OF THE THE TENTE													
	NID - Dispatch and Service Order for NID installation		UENTW	UNDBX										
	UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW	UENCE										
			UEANL,UEF,UEQ											
	Unbundled Contract Name, Provisioning Only - No Rate		UENTW	UNECN										
	onbuilded contract Name, Fronsioning Only No Nate		UAL,UCL,UDC,UD											
			,UDN,UEA,UHL,U											
	Unbundled Contact Name, Provisioning Only - no rate		C	UNECN	0	0								
	Official Name, Provisioning Only - no face		C	UNECIN	U	U								
		_	UEA,UDN,UCL,UI	,										
	Habitandlad Cub Lana Fandar O Wisa Casas Batt Institute and		C C	USBFQ	0	0								
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra	_	C	USBFQ	U	U								
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra		UEA,USL,UCL,UD	USBFR	0	0								
	Onburidled Sub-Loop Feeder-4 Wife Cross Box Jumper - no ra	_	UEA,USL,UCL,UD	USBFR	U	U								
	Habitan Had DCA Lana Considerate Farmet Ontine and the		LICI	CCOCE	0	0								
	Unbundled DS1 Loop - Superframe Format Option - no ra		USL	CCOSF	U	0								
	Unbundled DS1 Loop - Expanded Superframe Format option - no re		USL	CCOEF	0	0								
+	Chanaca Bot Edop Expanaca dapernanie i diniai opiidii ilio ii		JOL	OOOLI		U								
HIGH CAPAC	CITY UNBUNDLED LOCAL LOOP													
	NOTE: 4 month minimum billing period													
	High Capacity Unbundled Local Loop - DS3 - Per Mile per mon		UE3	1L5ND	8.9									
<b>-</b>	High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor		UE3	UE3PX		639.5	426.4	122.31	119.14		37.55	37.55	18.03	18.03
-	High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon		UDLSX	1L5ND	8.9	033.3	420.4	122.51	113.14		31.33	37.33	10.00	10.03
<b>-</b>	High Capacity Unbundled Local Loop - STS-1 - Fel Mile per Mon	_		ILJIAD				122.31	119.14					
	Inigh Capacity Unbundled Local Loop - 515-1 - Facility Termination per mor		LIDLCV	LIDI C4		C20 F					27.55	27.55	40.00	40.00
1			UDLSX	UDLS1		639.5	426.4	122.31	119.14		37.55	37.55	18.03	18.03
I OOD MAKE	=1ID		UDLSX	UDLS1		639.5	426.4	122.31	119.14		37.55	37.55	18.03	18.03
LOOP MAKE			UDLSX	UDLS1		639.5	426.4	122.31	119.14		37.55	37.55	18.03	18.03
LOOP MAKE	Loop Makeup - Preordering Without Reservation, per working or spare facility queried							122.31	119.14		37.55	37.55	18.03	18.03
LOOP MAKE	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).		UMK	UMKLW		35	35	122.31	119.14		37.55	37.55	18.03	18.03
LOOP MAKE	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).							122.31	119.14		37.55	37.55	18.03	18.03
LOOP MAKE	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried		UMK UMK	UMKLW		35 45	35 45	122.31	119.14		37.55	37.55	18.03	18.03
LOOP MAKE	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).		UMK	UMKLW		35	35	122.31	119.14		37.55	37.55	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)		UMK UMK	UMKLW		35 45	35 45	122.31	119.14		37.55	37.55	18.03	18.03
LOOP MAKE	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)		UMK UMK	UMKLW		35 45	35 45	122.31	119.14		37.55	37.55	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  MG		UMK UMK UMK	UMKLW UMKLP PSUMK	421.59	35 45 0.075	35 45 0.075				37.55	37.55	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci		UMK UMK UMK	UMKLW UMKLP PSUMK	421.59	35 45 0.075	35 45 0.075	0	0	0	37.55	37.55	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci  Line Sharing Splitter, per System 24 Line Capaci		UMK UMK UMK	UMKLW UMKLP PSUMK ULSDA	421.59 421.59	35 45 0.075	35 45 0.075	0 0	0 0	0	37.55	37.55	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci	1	UMK UMK  UMK  ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDA ULSDB ULSD8	421.59 131 32 11	35 45 0.075	35 45 0.075	0 0 0	0 0 0				18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio		UMK UMK  UMK  ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC	421.59 131 32 11	35 45 0.075 0 0 0 0 10.51	35 45 0.075 0 0 0 0 7.7	0 0	0 0	0	18.94	8.42	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci		UMK UMK  UMK  ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDA ULSDB ULSD8	421.59 131 32 11	35 45 0.075	35 45 0.075	0 0 0	0 0 0	0			18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio		UMK UMK  UMK  ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC	421.59 131 32 11	35 45 0.075 0 0 0 0 10.51	35 45 0.075 0 0 0 0 7.7	0 0 0	0 0 0	0	18.94	8.42	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci  Line Sharing Splitter, per System 24 Line Capaci  Line Sharing Splitter, per System 8 Line Capaci  Line Sharing - per Line Activatio  Line Sharing - per Line Activatio  Line Sharing - per Subsequent Activity per Line Rearrangeme	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC	421.59 131 32 11	35 45 0.075 0 0 0 0 10.51	35 45 0.075 0 0 0 0 7.7	0 0 0	0 0 0	0	18.94	8.42	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDS ULSDS	421.59 131 32 11	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci  Line Sharing Splitter, per System 24 Line Capaci  Line Sharing Splitter, per System 8 Line Capaci  Line Sharing - per Line Activatio  Line Sharing - per Line Activatio  Line Sharing - per Subsequent Activity per Line Rearrangeme	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC	421.59 131 32 11	35 45 0.075 0 0 0 0 10.51	35 45 0.075 0 0 0 0 7.7	0 0 0	0 0 0	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDS ULSDS	421.59 131 32 11	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop Makeup - With or Without Reservation, per working or spare facility queried (Mechanized)  MG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 92 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing - CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDS ULSDS	421.59 131 32 11	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)  D TRANSPORT  COMMON TRANSPORT (Shared)	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDS ULSDS	421.59 131 32 11	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)  D TRANSPORT  COMMON TRANSPORT (Shared)	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDS ULSDS	421.59 131 32 11	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop Makeup - With or Without Reservation, per working or spare facility queried (Mechanized)  MG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 92 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing - CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)	1 1 1 1 1	UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDS ULSDS	131 32 11 0.61	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)  D TRANSPORT  COMMON TRANSPORT (Shared) Common Transport - Per Mile, Per MOL Common Transport - Per Mile, Per MOL Common Transport - Facilities Termination Per MO		UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC ULSDC	131 32 11 0.61	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)  D TRANSPORT  COMMON TRANSPORT (Shared) Common Transport - Per Mile, Per MOL Common Transport - Per Mile, Per MOL Common Transport - Facilities Termination Per MO		UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC ULSDC	131 32 11 0.61	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)  D TRANSPORT  COMMON TRANSPORT (Shared)  Common Transport - Per Mile, Per MOL Common Transport - Facilities Termination Per MO  NOTE: INTEROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS		UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC ULSDC	131 32 11 0.61	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03
LINE SHARIN	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)  NG  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 98 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lin (16 pair)  D TRANSPORT  COMMON TRANSPORT (Shared) Common Transport - Per Mile, Per MOL Common Transport - Per Mile, Per MOL Common Transport - Facilities Termination Per MO		UMK UMK  UMK  ULS ULS ULS ULS ULS ULS ULS ULS	UMKLW UMKLP PSUMK  ULSDA ULSDB ULSDB ULSDB ULSDC ULSDC	131 32 11 0.61	35 45 0.075 0 0 0 10.51 36.23	35 45 0.075 0 0 0 0 7.7 13.23	0 0 0 0 7	0 0 0 0 4.2	0	18.94	8.42	18.03	18.03

	Intereffice Changel Dedicated Transport 2 Wire Value Code Traility Transporting		1 1				1			1	1	
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination	11477.07	1147710	47.07	70.04	00.00			40.04	40.04		
	per month	U1TVX	U1TV2	17.07	79.61	36.08			18.94	18.94		
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per											
	month	U1TVX	1L5XX	0.0222								
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination											
	per month	U1TVX	U1TR2	17.07	79.61	36.08	0	0	18.94	18.94		
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per mon	U1TDX	1L5XX	0.0222								
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mor	U1TDX	U1TD5	16.45	79.61	36.08			18.94	18.94		
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per mon	U1TDX	1L5XX	0.0222								
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mor	U1TDX	U1TD6	16.45	79.61	36.08	0	0	18.94	18.94		
									.,,,,			
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT - DS1		_									
INTEROIT	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mor	U1TD1	1L5XX	0.4523								
	Interoffice Channel - Dedicated Channel - DS1 - Fer Mile per mor	U1TD1	U1TF1	78.47	147.07	111.75			18.94	18.94		
	Interoffice Channel - Dedicated Transport - DS1 - Facility Termination per mor	UIIDI	UIIFI	78.47	147.07	111./5			18.94	18.94		
WITEDOE	FIGE CHANNEL DEPLOATED TRANSPORT DOS											
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT- DS3											
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor	U1TD3	1L5XX	2.72								
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mor	U1TD3	U1TF3	788	511.1	330.77	122.31	119.14	37.55	37.55	18.03	18.03
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT- STS-1											
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon	U1TS1	1L5XX	2.72	1			L	 	<u> </u>		
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor	U1TS1	U1TFS	783.63	511.1	449.91	122.31	119.14	61.19	61.19	3.17	3.17
										l		1
LOCAL C	HANNEL - DEDICATED TRANSPORT						1		<del>-  </del>	1		
	DCAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month, DS3	S and ahove-four month	ne			l	1			1	<del>                                     </del>	<del>                                     </del>
INOTE: LO	Local Channel - Dedicated - 2-Wire Voice Grade Per Month	ULCVX	ULDV2	13.91	382.95	62.4	1		18.94	8.42	-	<b> </b>
											-	<del>                                     </del>
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor	ULCVX	ULDR2	13.91	382.95	62.4			18.94	18.94		
	Local Channel - Dedicated - 4-Wire Voice Grade per mon	UNCVX	ULDV4	14.99	368.44	64.05			18.94	8.42		
	Local Channel - Dedicated - DS1 per mont	ULDD1	ULDF1	38.36	356.15	312.89	122.31	119.14	44.22	44.22	18.03	18.03
	Local Channel - Dedicated - DS3 - Per Mile per mon	ULDD3	1L5NC	6.92								
	Local Channel - Dedicated - DS3 - Facility Termination per mon	ULDD3	ULDF3	515.91	639.5	426.31	122.31	119.14	37.55	37.55	18.03	18.03
	Local Channel - Dedicated - STS-1- Per Mile per mon	ULDS1	1L5NC	6.92								
	Local Channel - Dedicated - STS-1 - Facility Termination per mon	ULDS1	ULDFS	517.56	639.5	426.31	122.31	119.14	18.94	18.94		
MULTIPLEXERS												
	Channelization - DS1 to DS0 Channel System	UXTD1	MQ1	126.22	198.22	123.59	31.03	19.75	14.75	6.55	10.7	
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)	UDL	1D1DD	1.86	12.02	8.66	31.03	13.73	14.73	0.55	10.7	
+		UDN		3.37								
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per mont		UC1CA		12.02	8.66						
	Voice Grade COCI - DS1 to DS0 Channel System - per mon	UEA	1D1VG	1.17	12.02	8.66	70.5	50.00	44.75	0.55	40.0	
	DS3 to DS1 Channel System per mont	UXTD3	MQ3	182.04	265.91	188.78	72.5	59.96	14.75	6.55	10.6	
	STS1 to DS1 Channel System per mont	UXTS1	MQ3	182.04	265.91	188.78	72.5	59.96	18.94	18.94		
	DS3 Interface Unit (DS1 COCI) used with Loop per mont	USL	UC1D1	11.02	12.02	8.66						
DARK FIBER												
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local											
	Channe	UDF	1L5DC	44.22								
	NRC Dark Fiber - Local Channe	UDF	UDFC4		1355.29	273.69						
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -	05.	05.0.		1000.20	270.00						
	Interoffice Channe	UDF	1L5DF	44.22								1
	NRC Dark Fiber - Interoffice Channe	UDF	UDF14	77.22	1355.29	273.69	0	0	18.94	18.94	<del>                                     </del>	<del>                                     </del>
		UDF	UDF 14		1335.28	213.09	U	U	16.94	10.94	-	<del>                                     </del>
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local	LIDE	41.50	44.00								
	Loop	UDF	1L5DL	44.22	4055.00	070.00	-		40.01	40.04	-	ļ
	NRC Dark Fiber - Local Loop	UDF	UDFL4		1355.29	273.69	0	0	18.94	18.94		
RANSPORT OTHER							1					
							1			<u> </u>		
Optional F	Features & Functions:		T		1			L	 	<u> </u>		
							1			1		1
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chanı	UNC1X	CCOEF		184.62	23.78	2.03	0.79	29.33	3.93		
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chann	UNC1X	CCOSF		184.62	23.78	2.03	0.79	29.33	3.93	1	
XX ACCESS TEN DIG	SIT SCREENING											
	8XX Access Ten Digit Screening, Per Ca	OHD		0.0004868		l	1			1	<del>                                     </del>	<del>                                     </del>
	8YY Access Ten Digit Screening, Personation Charac Box SVV Number Boson	OHD	N8R1X	0.0004008	6.57	0.76	1		18.94	18.94	1	<del>                                     </del>
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv		NOR I A								-	<del>                                     </del>
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation	OHD	No		12.81	1.45	1		18.94	18.94	-	ļ
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation	OHD	N8FTX		12.81	1.45			18.94	18.94	1	-
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb	OHD	N8FCX		4.46	2.23			18.94	18.94		ļ
	8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested											
	Per 8XX No.	OHD	N8FMX		5.22	2.99			18.94	18.94		
			110=110		7.00	0.70	1		18.94	18.94		1
	8XX Access Ten Digit Screening, Change Charge Per Reque	OHD	N8FAX		7.33	0.76				10.54		
	8XX Access Ten Digit Screening, Change Charge Per Reque 8XX Access Ten Digit Screening, Call Handling and Destination Featur	OHD OHD	N8FAX N8FDX		4.72	4.46			18.94	18.94		
INE INFORMATION D.												

LIDB Validation Per Quer	OQU		0.0105974		T								i
LIDB Originating Point Code Establishment or Chanç	OQT, OC	QU NRPBX	(	50.3						18.94	18.94		L
													L
SIGNALING (CCS7)													L
CCS7 Signaling Termination, Per STP Por	1DB	PT8SX	133.99							18.94	18.94		
CCS7 Signaling Usage, Per TCAP Messag	1DB		0.000087										<b>.</b>
CCS7 Signaling Connection, Per link (A link	1DB	TPP++		131.96	131.96					18.94	18.94		
CCS7 Signaling Connection, Per link (B link) (also known as D lin	1DB	TPP++		131.96	131.96					18.94	18.94		
CCS7 Signaling Usage, Per ISUP Messag	1DB		0.0000354										L
CCS7 Signaling Usage Surrogate, per link per LAT	1DB	STU56	340.67							18.94	18.94		L
CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per										'			I
STP affectec	1DB	CCAPC	)	40	40					18.94	18.94		L
CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per										1			ı
Stp Affected	1DB	CCAPE	)	8	8					18.94	18.94		L
													L
E911 SERVICE													l
													L
													L
CALLING NAME (CNAM) SERVICE										1			ı
CNAM for DB Owners, Per Query	OQV		0.016										ı
CNAM for Non DB Owners, Per Quen	OQV		0.01										ı
													ı
													ı
CNAM (Non-Databs Owner), NRC, applies when using the Character Based User								1	1				
Interface (CHUI)	OQV	CDDCH	4	595	595					18.94	18.94		I
			T		T						ĺ		i
			1	1	1						ĺ		İ
													l
LNP QUERY SERVICE			T		T						ĺ		i
			1	1	1						ĺ		İ
													ı
OPERATOR SERVICES AND DIRECTORY ASSISTANCE													İ
													i
OPERATOR CALL PROCESSING													i
Oper. Call Processing - Oper. Provided, Per Min Using BST LID			1.2		+	1	-						ĺ
Oper. Call Processing - Oper. Provided, Per Min Using Foreign LID			1.24		+	1	-						ĺ
Oper. Call Processing - Fully Automated, per Call - Using BST LID			0.2		+								i
Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE		-+	0.2	+	+	+	-						·
open dan recodering rany reasonated, per dan dening recognized		-+	- 0.2	+	+	+	-						·
INWARD OPERATOR SERVICES			+	+	+	1							i
Inward Operator Svcs - Verification, Per Minute			1.15		+								i
Inward Operator Services - Verification and Emergency Interrupt - Per Minu		-+	1.15	+	+	+	-						·
inward Operator Services Vermeation and Emergency interrupt 1 or Mine		-+	1.10	+	+	+	-						·
BRANDING - OPERATOR CALL PROCESSING			+	+	+	1							i
Recording of Custom Branded OA Announcement	<del></del>	CBAOS	\$	7000	7000	+				19.99	19.99	19.99	19.99
Loading of Custom Branded OA Announcement per shelf/NAV		CBAOL		500	500	+	-			19.99	19.99	10.00	10.00
Edding of Gustom Branded Ox Announcement per sneimer		CDAOL	+	300	300			-		15.55	13.33	<del>                                     </del>	
DIRECTORY ASSISTANCE SERVICES	<del>-      </del>	-+-	+		+	+				<del>                                     </del>			
DIRECTORY ASSISTANCE ACCESS SERVICE			+	+	+	+				t'		+	
			0.075							<u> </u>	<b>——</b>		
Directory Assistance Access Service Calls, Charge Per Ca			0.275	+	+	+					<b></b>		
DIPERTORY ASSISTANCE AND COMPLETION ASSESSMENT (PAGE)										<u> </u>	<b>——</b>		
DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)											+	ļI	
Directory Assistance Call Completion Access Service (DACC), Per Call Attem			0.1							<u> </u>	+	<b>_</b>	<del></del>
UNBRANDING			+	+	+	+		1	<del>                                     </del>	<b></b> '		<b> </b>	
		+-	+	+	+	++				<b></b> '			
DIRECTORY TRANSPORT			+									$oxed{oxed}$	<b></b>
Directory Transport - Local Channel DS			38.36	356.15	312.89	4			ļ	44.22	44.22		<b></b>
Directory Transport - DS1 Level Interoffice Per Mi			0.4523	<del></del>	<del> </del>							$oxed{oxed}$	<b></b>
Directory Transport - DS1 Level Interoffice Per Facility Termination			78.47	147.07	111.75					18.94	18.94	$oxed{oxed}$	<b></b>
Switched Common Transport Per DA Access Service Per Ca			0.0002906							<b></b> '	<del></del>	$oxed{oxed}$	<b></b>
Switched Common Transport Per DA Access Service Per Call Per Mi			0.0000186		<b></b>					<b> </b>	<b></b>		<b></b>
Access Tandem Switching Per DA Access Service Per Ca			0.0019152							<b></b> '	<del></del>	$oxed{oxed}$	<b></b>
Directory Transport - DA Interconnection Per DA Service C€			0.00269		<del></del>							$oxed{oxed}$	<b></b>
Directory Transport - Installation NRC, Per Trunk or Signaling Connecti				204.23	4.42					44.22	44.22		<b></b>
										<b></b> '	<del></del>	$oxed{oxed}$	<b></b>
DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)						4			ļ	<b></b> '	<b></b>		<b></b>
Directory Assistance Data Base Service Charge Per Listir			0.04		<b></b>	4				ļ	<b></b>		<b></b>
Directory Assistance Data Base Service, per mont		DBSOF	F 150		<b></b>	4				ļ	<b></b>		<b></b>
BRANDING - DIRECTORY ASSISTANCE					4	1				ļ	<b></b>		<b></b>
Custom Branding Announcement, per Recording to be used with the provision of DA	AMT	CBADA		3000	3000					<u> </u>			L
Loading of Custom Branded Announcement per DRAM Card/Switch	AMT	CBADO		690	690	T							
SELECTIVE ROUTING									1				I
													l
Selective Routing Per Unique Line Class Code Per Request Per Swit		USRCF	₹	180.62	180.62					33.67	7.88		I
Selective Routing Fer Unique Line Class Code Fer Request Fer Swit					7			1	1				
Selective Routing Fer Orlique Line Class Code Fer Request Fer Swit										l .	Į	1	
VIRTUAL COLLOCATION					+	++							

			ueanl,uea,udn,udc,u										
	Virtual Collocation - 2-wire Cross Connects (loop		al,uhl,ucl,uec l	UEAC2	0.0283	24.56	23.56	9.2	8.3	19.99	19.99	19.99	19.99
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splittin	- 1		VE1LS	0.0283	24.56	23.56	9.2	8.3	19.99	19.99	19.99	19.99
	Virtual Collocation - 2-wire Cross Connects (por			VE1R2	0.0283	24.56	23.56	9.2	8.3	19.99	19.99	19.99	19.99
	Virtual Collocation - 4-wire Cross Connects (looş Virtual Collocation - 4-wire Cross Connects (por			UEAC4 VE1R4	0.0566 0.0566	24.75 24.75	23.7	9.03 9.03	8.1 8.1	19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Virtual Collocation - 4-wire Cross Connects (por Virtual Collocation - 2-Fiber Cross Connect			CNC2F	2.88	41.72	30.36	10.43	8.36	19.99	2.2	19.99	19.99
	Virtual Collocation - 2-Fiber Cross Connects  Virtual Collocation - 4-Fiber Cross Connects			CNC4F	5.76	51.03	39.67	13.71	11.65	2.2	2.2		
	Virtual Collocation - The Cross Connect			CNC1X	7.5	155	14	10.71	11.00	2.2	2.2		
AIN SELEC	TIVE CARRIER ROUTING												
	Regional Service Establishment			SRCEC		391788				19.99	19.99	19.99	19.99
	End Office Establishment			SRCEO		320.53	320.53			19.99	19.99	19.99	19.99
	Line/Port NRC, per end user  Query NRC, per query		SRC	SRCLP	0.000448	2.06	2.06			19.99	19.99	19.99	19.99
	Query NRC, per query		SRC		0.000448								
AIN - BELLS	SOUTH AIN SMS ACCESS SERVICE												
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup		C	CAMSE		90.25	90.25			18.94	18.94		
	AIN SMS Access Service - Port Connection - Dial/Shared Access		C	CAMDP		29.66	29.66			18.94	18.94		
	AIN SMS Access Service - Port Connection - ISDN Access			CAM1P		00.00	00.00			40.04	40.04		
	AIN SWS Access Service - Port Connection - ISDN Access			CAWITP		29.66	29.66			18.94	18.94		
	AIN SMS Access Service - User Identification Codes - Per User ID Code			CAMAU		84.43	84.43			18.94	18.94		
	Ant civic Access cervice Coer Identification Codes 1 of Coer ID Code			0/11/1/10		04.43	04.43			10.54	10.54		
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			CAMRC		35.44	35.44			18.94	18.94		
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)				0.0023								
	AIN SMS Access Service - Session, Per Minute				0.0795604								
	AIN SMS Access Service - Company Performed Session, Per Minute				2.08								
AIN - BELLS	SOUTH AIN TOOLKIT SERVICE												
	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			BAPSC		86.74	86.74			18.94	18.94		
	AIN Toolkit Service - Training Session, Per Customer			BAPVX BAPTT		8348 19.13	8348 19.13			18.94 18.94	18.94 18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay			BAPTD		19.13	19.13			18.94	18.94		
	Ally Toolkit Service - Higger Access Charge, Fer Higger, Fer Div, Olf-Hook Delay			DAFID		114.6	114.8			16.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate		F	ВАРТМ		19.13	19.13			18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP			BAPTO		70.06	70.06			18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP			BAPTC		70.06	70.06			18.94	18.94		
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code			BAPTF		70.06	70.06			18.94	18.94		
	AIN Toolkit Service - Query Charge, Per Query				0.0209223								
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Pe	r											
	Query				0.0053137								
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobyte	S			1.46								
	AIN Tabilit Carries Manthly second Bas AIN Tabilit Carries Cubacristics			BAPMS	45.00	00.04	00.04			40.04	40.04		
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			BAPLS	15.96 0.0861109	22.64 22.64	22.64 22.64			18.94 18.94	18.94 18.94		
	And Toolkit Service - Special Study - Fer And Toolkit Service Subscription			DAI LO	0.0661109	22.04	22.04			10.94	10.94		
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription		l e	BAPDS	15.87	22.64	22.64			18.94	18.94		
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			BAPES	0.0028704	22.64	22.64			18.94	18.94		
ODUF/EDO	UF/ADUF/CMDS												
	ACCESS DAILY USAGE FILE (ADUF)												
	ADUF: Message Processing, per messag				0.0136327		<b></b>				1		
-	ADUF: Data Transmission (CONNECT:DIRECT), per messag				0.0000434		-				1		
-	ENHANCED OPTIONAL DAILY USAGE FILE (EODUF)		+				<del> </del>						-
-	EODUF: Message Processing, per message				0.0034555		<b> </b>						
					0.0004000								
	OPTIONAL DAILY USAGE FILE (ODUF)												
	ODUF: Recording, per message				0.0001275						1		
	ODUF: Message Processing, per message				0.0082548								
	ODUF: Message Processing, per Magnetic Tape provisions		+		28.85		-						
	ODUF: Data Transmission (CONNECT:DIRECT), per messag		+		0.0000434		<del>                                     </del>						
ENHANCED	EXTENDED LINK (EELs)		+										
LITTIANCEL							<b> </b>						
	NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; N	liami. FL: F	. Lauderdale. FLI: Nash	ville. TN	I: New Orleans	. LA:							
	NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rate	s below exc	ept Switch As Is Charge	e.									
	NOTE: In all states, EEL network elements shown below also apply to currently combined facilit	es which ar	e converted to UNE rate	es. A Sw	itch As Is Cha	rge applies to c	urrently combi	ned facilities	converted to UNEs.(Non-	recurring rates do not	apply.)		
										-			
	NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements pe	r the GA PS	C order.(No Switch As Is	s Charge	e.)					<u> </u>		<u> </u>	

	ICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (	EEL)													
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone		1	UNCVX	UEAL2	16.84	104.14	78.1							
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone		2	UNCVX	UEAL2	19.45	104.14	78.1	0	0					
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone		3	UNCVX	UEAL2	30.92	104.14	78.1	0	0					
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor			UNC1X	1L5XX	0.4523									
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor			UNC1X	U1TF1	78.47	194.63	141.51	132.25	46.16		33.63	27.49	19.88	11
	DS1 Channelization System Per Mont			UNC1X	MQ1 1D1VG	1.17	12.02	8.66							
	Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport			UNCVX	IDIVG	1.17	12.02	8.00							
	Combination - Zone		1	UNCVX	UEAL2	16.84	104.14	78.1	0	0		18.94	8.42		
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport														
+	Combination - Zone :  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport		2	UNCVX	UEAL2	19.45	104.14	78.1	0	0		18.94	8.42		
	Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon		3	UNCVX	UEAL2 1D1VG	30.92 1.17	104.14 12.02	78.1 8.66	0	0		18.94	8.42		
_						1.17			40.04	40.04		45.40	45.70		
_	Nonrecurring Currently Combined Network Elements Switch -As-Is Char			UNC1X	UNCCC		12.97	11.27	12.61	12.61		45.46	15.72		
4-WIRE VO	DICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (	EEL)													
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	22.26	206.95	170.57							
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -														
	Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -		2	UNCVX	UEAL4	25.7	206.95	170.57	0	0					
	Zone 3		3	UNCVX	UEAL4	40.86	206.95	170.57	0	0					
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor			UNC1X	1L5XX	0.4523									
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor			UNC1X	U1TF1	78.47	194.63	141.51	132.25	46.16		33.63	27.49	19.88	1
	Channelization - Channel System DS1 to DS0 combination Per Mor			UNC1X UNCVX	MQ1 1D1VG	1.17	0 12.02	0 8.66	0	0					
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mon Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport			UNCVX	1D1VG	1.17	12.02	8.66							
	Combination - Zone		1	UNCVX	UEAL4	22.26	206.95	170.57	0	0		18.94	8.42		
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone :		2	UNCVX	UEAL4	25.7	206.95	170.57	0	0		18.94	8.42		
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport														
	Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon		3	UNCVX	UEAL4 1D1VG	40.86 1.17	206.95 12.02	170.57 8.66	0	0		18.94	8.42		
+	Voice Grade COCI - DST to DS0 Chariner System Combination - per mon			UNCVX		1.17	12.02	0.00							
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			UNC1X	UNCCC		12.97	11.27	12.61	12.61		45.46	15.72		
1															
4-WIRE 56	 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR	T (EEL)													
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	T (EEL)													
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1	T (EEL)	1	UNCDX	UDL56	25.75	384.56	241.2							
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	T (EEL)	1						0	0					
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	T (EEL)	1 2	UNCDX	UDL56	25.75 29.74	384.56 384.56	241.2 241.2	0	0					
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	T (EEL)	1 2 3						0	0					
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	T (EEL)		UNCDX	UDL56	29.74	384.56	241.2	,	-					
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	T (EEL)		UNCDX UNCDX UNC1X	UDL56 UDL56 1L5XX	29.74 47.27 0.4523	384.56 384.56	241.2 241.2	0	0					
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor	T (EEL)		UNCDX UNCDX UNC1X UNC1X	UDL56 UDL56 1L5XX U1TF1	29.74 47.27 0.4523 78.47	384.56 384.56 194.63	241.2 241.2 141.51	0 132.25	0 46.16		33.63	27.49	19.88	1
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor	T (EEL)		UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X	UDL56 UDL56 1L5XX U1TF1 MQ1	29.74 47.27 0.4523 78.47 0	384.56 384.56 194.63 0	241.2 241.2 141.51 0	0	0		33.63	27.49	19.88	1
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor	T (EEL)		UNCDX UNCDX UNC1X UNC1X	UDL56 UDL56 1L5XX U1TF1	29.74 47.27 0.4523 78.47	384.56 384.56 194.63	241.2 241.2 141.51	0 132.25	0 46.16		33.63	27.49	19.88	1
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone -	T (EEL)		UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X	UDL56 UDL56 1L5XX U1TF1 MQ1	29.74 47.27 0.4523 78.47 0	384.56 384.56 194.63 0	241.2 241.2 141.51 0	0 132.25	0 46.16		33.63	27.49	19.88	1
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor CCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone: Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	T (EEL)	3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56	29.74 47.27 0.4523 78.47 0 1.86 25.75	384.56 384.56 194.63 0 12.02 384.56	241.2 241.2 141.51 0 8.66 241.2	0 132.25 0	0 46.16 0		18.94	8.42	19.88	1
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :	T (EEL)	3	UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX	UDL56  UDL56  1L5XX  U1TF1  MQ1  1D1DD  UDL56  UDL56	29.74 47.27 0.4523 78.47 0 1.86 25.75	384.56 384.56 194.63 0 12.02 384.56	241.2 241.2 141.51 0 8.66 241.2	0 132.25 0	0 46.16 0		18.94 18.94	8.42 8.42	19.88	
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor CCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb) Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :	T (EEL)	3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56	29.74 47.27 0.4523 78.47 0 1.86 25.75	384.56 384.56 194.63 0 12.02 384.56	241.2 241.2 141.51 0 8.66 241.2	0 132.25 0	0 46.16 0		18.94	8.42	19.88	1
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 - combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :	T (EEL)	3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56  1L5XX  U1TF1  MQ1  1D1DD  UDL56  UDL56	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27	384.56 384.56 194.63 0 12.02 384.56 384.56	241.2 241.2 141.51 0 8.66 241.2 241.2	0 132.25 0	0 46.16 0		18.94 18.94	8.42 8.42	19.88	1
4-WIRE 56	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  CCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  COU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)	T (EEL)	3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56  UDL56  UDL56  1D1DD	29.74 47.27 0.4523 78.47 0 1.86 25.75	384.56 384.56 194.63 0 12.02 384.56 384.56 384.56	241.2 241.2 141.51 0 8.66 241.2 241.2 241.2 8.66	0 132.25 0 0	0 46.16 0		18.94 18.94 18.94	8.42 8.42 8.42	19.88	1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor Channelization - Channel System DS1 to DS0 combination Per Mor CCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-Is Charq		3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56  1L5XX  U1TF1  MQ1  1D1DD  UDL56  UDL56  UDL56	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27	384.56 384.56 194.63 0 12.02 384.56 384.56	241.2 241.2 141.51 0 8.66 241.2 241.2	0 132.25 0	0 46.16 0		18.94 18.94	8.42 8.42	19.88	1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs) Nonrecurring Currently Combined Network Elements Switch -As-Is Charş  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR		3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56  UDL56  UDL56  1D1DD	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27	384.56 384.56 194.63 0 12.02 384.56 384.56 384.56	241.2 241.2 141.51 0 8.66 241.2 241.2 241.2 8.66	0 132.25 0 0	0 46.16 0		18.94 18.94 18.94	8.42 8.42 8.42	19.88	1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kb)  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -		1 2 3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56  UDL56  UDL56  1D1DD  UNCCC	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27 1.86	384.56 384.56 194.63 0 12.02 384.56 384.56 12.02 12.97	241.2  241.2  141.51 0 8.66  241.2  241.2  241.2  8.66  11.27	0 132.25 0 0	0 46.16 0		18.94 18.94 18.94	8.42 8.42 8.42	19.88	1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  CCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  COU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Chark  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone :		3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56  UDL56  UDL56  1D1DD	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27	384.56 384.56 194.63 0 12.02 384.56 384.56 384.56	241.2 241.2 141.51 0 8.66 241.2 241.2 241.2 8.66	0 132.25 0 0	0 46.16 0		18.94 18.94 18.94	8.42 8.42 8.42	19.88	1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kb)  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -		1 2 3	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56  UDL56  UDL56  1D1DD  UNCCC	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27 1.86	384.56 384.56 194.63 0 12.02 384.56 384.56 12.02 12.97	241.2  241.2  141.51 0 8.66  241.2  241.2  241.2  8.66  11.27	0 132.25 0 0	0 46.16 0		18.94 18.94 18.94	8.42 8.42 8.42	19.88	1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 - combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		3 1 2 3 1 1 2	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  1L5XX  U1TF1  MQ1  1D1DD  UDL56  UDL56  UDL56  UDL56  UDL56  UDL56  UDL56  UDL64	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27 1.86	384.56 384.56 194.63 0 12.02 384.56 384.56 12.02 12.97 348.55	241.2 241.2 141.51 0 8.66 241.2 241.2 241.2 8.66 11.27 241.2 241.2	0 132.25 0 0 0 0 12.61	0 46.16 0 0 0 0		18.94 18.94 18.94	8.42 8.42 8.42	19.88	
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 - combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kb)  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		1 2 3	UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56  UDL56  UDL56  UDL56  UDL56  UDL56  UDL64  UDL64  UDL64	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27 1.86	384.56 384.56 194.63 0 12.02 384.56 384.56 12.02 12.97	241.2  241.2  141.51 0  8.66  241.2  241.2  8.66  11.27	0 132.25 0 0 0 0	0 46.16 0 0 0 0		18.94 18.94 18.94	8.42 8.42 8.42	19.88	
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		3 1 2 3 1 1 2	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  1L5XX  U1TF1  MQ1  1D1DD  UDL56  UDL56  UDL56  UDL56  UDL56  UDL56  UDL56  UDL64	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27 1.86	384.56 384.56 194.63 0 12.02 384.56 384.56 12.02 12.97 348.55	241.2 241.2 141.51 0 8.66 241.2 241.2 241.2 8.66 11.27 241.2 241.2	0 132.25 0 0 0 0 12.61	0 46.16 0 0 0 0 12.61		18.94 18.94 18.94	8.42 8.42 8.42	19.88	1
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Pacility Termination Per Mot Channelization - Channel System DS1 to DS0 combination Per Mor OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb) Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor		3 1 2 3 1 1 2	UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNC1X UNCDX UNC1X	UDL56  UDL56  1L5XX  U1TF1  MQ1  1D1DD  UDL56  UDL56  1D1DD  UNCCC  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27 1.86  25.75 29.74 47.27 0.4523 78.47	384.56 384.56 194.63 0 12.02 384.56 384.56 12.02 12.97 348.55 348.55 348.55	241.2 241.2 141.51 0 8.66 241.2 241.2 241.2 241.2 241.2 241.2 241.2 241.2 141.51	0 132.25 0 0 0 0 12.61	0 46.16 0 0 0 0 12.61		18.94 18.94 18.94	8.42 8.42 8.42	19.88	
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 - combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport  Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR  First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		3 1 2 3 1 1 2	UNCDX  UNCDX  UNC1X  UNC1X  UNC1X  UNC1X  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL56  UDL56 1L5XX  U1TF1 MQ1 1D1DD  UDL56	29.74 47.27 0.4523 78.47 0 1.86 25.75 29.74 47.27 1.86 25.75 29.74 47.27 0.4523	384.56 384.56 194.63 0 12.02 384.56 384.56 12.02 12.97 348.55 348.55	241.2 241.2 141.51 0 8.66 241.2 241.2 241.2 8.66 11.27 241.2 241.2 241.2 241.2	0 132.25 0 0 0 0 12.61	0 46.16 0 0 0 0 12.61		18.94 18.94 18.94 45.46	8.42 8.42 8.42		1

COMMISSION - 2005	1	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	т	T											T		
Additional service Geological Operation Control and Collision Transport (Control Collision Colli				4	LINCDY	LIDL64	25.75	349.55	2/1 2	0	0			18 04	8 42		
Descriptions 2006.1   Descriptions 2006.1				+-'-+	UNCDA	ODL04	23.73	340.33	241.2	0	- 0			10.54	0.42		
National Act Note of England Content Logon Sample Content Co					LINCDY	LIDI C4	20.74	240.55	244.2	0	0			40.04	0.40		
Contemporary - Zones   Contemporary - Zones   Contemporary   Contemporary - Zones   Conte			+		UNCDX	UDL64	29.74	348.55	241.2	U	U			18.94	8.42		
COLLIGE FOOD:   Long										_	_						
MACROX   1900   1.86   12.27   1.26				3	UNCDX	UDL64	47.27	348.55	241.2	0	0			18.94	8.42		
Awy Do Dark A Common Name Research Switch Ask County   1987   1987   1988   1																	
### 1997   1997	6	64kbs)			UNCDX	1D1DD	1.86	12.02	8.66								
MOTILAL PATTINETION DOOR WITH DESCRICATION OF MATERIAL PROPERTY AND ADDRESS OF MATERIAL PROPERTY																	
HWW CSD   Digital cap in Combination with Distriction Transport 2-200   1 UNCX   USEXX   SSS   4452   198.66   0   0   0   0   0   0   0   0   0	N.	Nonrecurring Currently Combined Network Elements Switch -As-Is Charc			UNC1X	UNCCC		12.97	11.27	12.61	12.61			45.46	15.72		
CVYVID CEST   Digal Loss on Combination with Distriction Français - Zons   1   UNCIX   USBAN   44:22   198.88   0   0			-														
CVYVID CEST   Digal Loss on Combination with Distriction Français - Zons   1   UNCIX   USBAN   44:22   198.88   0   0	4-WIRE DS1	DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (F	EI)	+													
With CSD Digital Loss Conferences and Dil Interfice Transport 2				1	LINC1V	Hel VV	EE EO	442.2	120.60						+		
Liver DRI Digital Log in Companion with Tiber Service   3   Liver X   107.30   445.2   198.68   0   0   0	4	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	+							_	^						
Interdiffic Transport Celebrate, DSI constraints - Pot Mile Per Mar.   UNIX. V. UTFT - 794			+														
Memoritine Transport - Deficated - BS1 continuation - Facility Termination Per Mo.   UNCIX   U1TT1   78.47   134.65   141.51   132.25   48.16   33.63   27.49   15.72   15.7				3				443.2	138.69	0	0						
Nonecouring Currently Cerebined Nesson Elements Switch. As is Charl   AdMitted Digital Activated Digital Transport Cerebinator - Zero   UNCT   UNCOL   USBOX   25.53   443.3   138.66   0   0   0   0   0   0   0   0   0	Ir	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor			UNC1X	1L5XX	0.4523										
Advise D GGFTAL EXTENDED LOPF WITH DEDICATED OSS INTEROPPEE TRANSPORT (EEL)   UNCTA																	
### SPAN DISCONDING STREET CONTRIBUTION OF MITTER DESIRES PROFILE TRANSPORT (EEL)  First DST Loop in DSD Intended Transport Combination - Zone  JUNCIN USUXX 64.13 443.2 138.69 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1 1	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor			UNC1X	U1TF1	78.47	194.63	141.51	132.25	46.16			33.63	27.49	19.88	11
### DSY DIATE STENDED LOOP WITH SECURITOR DISTANCE TENSPORT (EEL)  First DST Loop in DSS Interding Transport Combination - Zons		· · · · · · · · · · · · · · · · · · ·															
### DSY DIATE STENDED LOOP WITH SECURITOR DISTANCE TENSPORT (EEL)  First DST Loop in DSS Interding Transport Combination - Zons	l N	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNC1X	UNCCC		12 97	11 27	12 61	12 61			15.72	15.72		
First OSE Loop in DSS Interest Transport Combination - Zerot   1 UND X USBAX   55.52   441.2   136.69   0   0   0   0   0   0   0   0   0		Nonrecurring Currently Combined Network Elements Cwitch 7/3 is Chart	+	+-+	ONOIX	011000		12.01	11.27	12.01	12.01			10.72	10.72		
First DSI Loop in DSS Intentifier Transport Combination - Zone	4 WIDE DO4	DIGITAL EXTENDED LOOP WITH DEDICATED DES INTEROFFICE TRANSPORT (F		+		_											
First DSE Loop in DSE Interfice Transport Combination - Zone			EL)	+													
First DS LLock in DS Intendince Transport Combination - Zone   3   UNCIX   USUXX   1918.0   443.2   138.68   0   0   0																	ļ
Interesting Tangonic Delication - 1981 Interfines ( - 1981				2	UNC1X		64.13			0	0						
Interestina Transport Dedicated: DSS combination. Per Mile Per Mor   UNCXX   1,10XX   2,72   19.85   19.85   15.15   5.	l F	First DS1Loop in DS3 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	101.93	443.2	138.69	0	0						
Interesting Transport - Deskoards - D83 - Seality Termination part more   UNCXX UTFT   788   198.46   133.15   9.4.4   35.99   33.63   27.49   19.88   10.50	lr.	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Mor			UNC3X	1L5XX	2.72										
DSS to DST Channel System combinations per mont   DRCSX   MGG   MGG   MGC   MGG			1					198.45	153.15	95.4	35.99		1	33.63	27.49	19.88	
DSS Interface Link (D81 COCI) combination per mont			+	+									1	00.00		10.00	
Additional DSI Loop in DS3 Interriffical Transport Combination - Zoni	r	DOS to be a criatine system combination per mon	+	+-+						- 0	10.12						
Additional DSI Loop in DSI Interdiffice Transport Combination - Zono   2 UNCTX USEXX 1013 4432 138.68 0 0 0 18.94 8.42			+	+						•				40.04	0.40		
Additional DST Loop in DSS Interestine Transport Corribonation - Zeon   3 UNCIX USLX   101.93   138.89   0 0   19.94   8.42										-	-						
DS3 Interface Line (DS1 COCI) combination per mont    UNCYX   UNCOX   12.07   12.01   12.01   45.46   15.72										-							
Nonrecurring Currently Combined Network Elements Switch -As-is Chary   UNCX   UNCCC   12.97   11.27   12.61   12.61   45.46   15.72	ρ	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	101.93	443.2	138.69	0	0			18.94	8.42		
2-Wire Voice RRADE EXTENSED LOOP? 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL.)   1 UNCVX   U.E.L.2   16.84   104.14   78.1   0   0   18.94   8.42   1.2.25   1.2		DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1	11.02	12.02	8.66								
### 24/18/19 Computed with 2-view VP Olice GRADE INTEROFFICE TRANSPORT (EEL.)    2-VirieVPC Loop used with 2-view VP Olimeroffice Transport Combination - Zorn		, ,	1														
### VINCE GRADE EXTENDED LOOP? 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL.)    2-WINEVO Loop used with 2-wire V O Interoffice Transport Combination - Zorn	l N	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNC3X	UNCCC		12 97	11 27	12 61	12 61			45 46	15.72		
2-Wire/VG Loop used with 2-wire VG Interoffice Transport Combination - Zone   1 UNCVX UEAL2   16.64   104.14   78.1   0 0   18.94   8.42	† * * * * * * * * * * * * * * * * * * *	Nonicounting Currently Combined Notwork Elements Current 716 to Orlan	+	+	0.100/1	0.1000		12.01	111.27	12.01	12.01			10.10	10.72		
2-Wire/VG Loop used with 2-wire VG Interoffice Transport Combination - Zone   1 UNCVX UEAL2   16.64   104.14   78.1   0 0   18.94   8.42	2 WIDE VOIC	CE COADE EVIENDED I CODI A MIDE VOICE COADE INTEROFFICE TRANSPORT	(EEL)	+		_											-
2   Wiew'Q Logo used with 2-wire VG Interoffice Transport Combination - Zone   2   Wick'y   UEAL2   19.45   104.14   78.1   0   0   18.94   8.42			(EEL)	+													
2-Wire/Q Logo used with Z-wire V S Interoffice Transport Combination - Zont   18,44   8,42   Interoffice Transport - Dedicated - 2-w File Per Mor   UNCVX																	
Interoffice Transport - Dedicated - 2wire Viola of Combination - Facility   UNCVX   U17V2   17.07   79.61   36.08   18.94											0						
Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility   UNCVX U1TV2   17.07   79.61   36.08   18.94	2	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone		3	UNCVX	UEAL2	30.92	104.14	78.1	0	0			18.94	8.42		
Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility   UNCVX U1TV2   17.07   79.61   36.08   18.94	lr.	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor			UNCVX	1L5XX	0.0222										
Termination per mont	l)	Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility	1														
Nonrecurring Currently Combined Network Elements Switch-As-Is Char					LINCVX	U1TV2	17.07	79.61	36.08					18 94	18 94		
### VIEWOR GRADE EXTENDED LOOP! A WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)  ### A WireYG Loop used with 4-wire VG interoffice Transport Combination - Zont 1 UNCVX UEAL4 22.26 20.69.5 170.57 0 0 18.94 8.42 4.20 4.40 4.40 4.60 20.90 used with 4-wire VG interoffice Transport Combination - Zont 3 UNCVX UEAL4 40.86 20.69.5 170.57 0 0 18.94 8.42 4.40 4.40 4.60 20.90 used with 4-wire VG interoffice Transport Combination - Per Mile Per Mor UNCVX USAL4 40.86 20.69.5 170.57 0 0 18.94 8.42 4.40 4.40 4.60 20.90 used with 4-wire VG interoffice Transport - Dedicated -4-Wire VG combination - Per Mile Per Mor UNCVX UNCVX USAL4 40.86 20.69.5 170.57 0 0 0 18.94 8.42 4.40 4.40 4.40 4.40 4.40 8.40 20.90 used with 4-wire VG interoffice Transport - Dedicated -4-Wire VG combination - Per Mile Per Mor UNCVX UNCVX USAL4 40.86 20.69.5 170.57 0 0 0 18.94 8.42 4.40 4.40 8.40 20.90 used with 4-wire VG interoffice Transport - Dedicated -4-Wire Voice Grade combination - Fer Mile Per Mor UNCVX UNC			-						0.0.00						1		
### WIRE VOICE GRADE EXTENDED LOOP! 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)  ### Wire VG Loop used with 4-wire VG interoffice Transport Combination - Zont		Nonrocurring Currently Combined Naturals Flomenta Switch As Is Char			LINCVY	LINICCC		12.07	11 27	12.61	12.61			4E 4G	15 70		
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zont   1 UNCVX UEAL4   25.7   206.95   170.57   0 0   18.94   8.42   1	15	Nonrecurring Currently Combined Network Elements Switch -As-is Chart	+	+-+	UNCVA	UNCCC		12.31	11.27	12.01	12.01			43.40	13.72		-
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zond   1 UNCVX URAL4   25.7   206.95   170.57   0 0   18.94   8.42				+		_											-
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zont   2 UNCVX UEAL4   42.67   266.95   170.57   0   0   18.94   8.42			(EEL)	$\perp$													
4-Wire/G Loop used with 4-wire VG Interoffice Transport Combination - 2 on								206.95		0	0				8.42		
Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility   UNCVX   U1TV4   17.07   79.61   36.08   18.94   18.				2													
Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility   UNCVX U1TV4   17.07   79.61   36.08   18.94	4	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone		3	UNCVX	UEAL4	40.86	206.95	170.57	0	0			18.94	8.42		
Interoffice Transport - Dedicated - 4-Wire Voice Grade combination - Facility   UNCVX U1TV4   17.07   79.61   36.08   18.94	lr.	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor			UNCVX	1L5XX	0.0222										
Nonrecurring Currently Combined Network Elements Switch -As-Is Char;   UNCVX   UNCCC   12.97   11.27   12.61   12.61   45.46   15.72			1														
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			1		LINCVY	11111/4	17.07	70.61	36.09			1	1	18 04	18 04		ĺ
SS DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)   High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor   UNC3X   1L5ND   8.9   UNC3X   1L5ND   119.14   UNC3X   119.14   UNC3X   119.14   UNC3X   119.14   UNC3X   119.14   UNC3X		remination per mont	+	+	UNCVA	01114	17.07	10.61	30.08			<b> </b>	1	10.94	10.94		<del> </del>
SS DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)			1									1	1				
High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor   UNC3X   1L5ND   8.9   High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month   UNC3X   UE3PX   390.34   639.5   426.4   122.31   119.14	l N	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		$\perp$	UNCVX	UNCCC		12.97	11.27	12.61	12.61		<b> </b>	45.46	15.72		1
High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor   UNC3X   1L5ND   8.9   High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month   UNC3X   UE3PX   390.34   639.5   426.4   122.31   119.14																	<u></u>
High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month   UNC3X   UE3PX   390.34   639.5   426.4   122.31   119.14	DS3 DIGITAL	EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)															
High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month   UNC3X   UE3PX   390.34   639.5   426.4   122.31   119.14		High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor	1		UNC3X	1L5ND	8.9										
Month   UNC3X   UE3PX   390.34   639.5   426.4   122.31   119.14	I F		-														
Interoffice Transport - Dedicated - DS3 - Per Mile per mon						LIESDY	200.24	620 E	426.4	122 21	110 14						
Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mo	Н	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per							420.4	122.31	119.14						-
Nonrecurring Currently Combined Network Elements Switch -As-Is Char;   UNC3X   UNCCC   12.97   11.27   12.61   12.61   45.46   15.72	H	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month		+				000.0									
Nonrecurring Currently Combined Network Elements Switch -As-Is Char;   UNC3X   UNCCC   12.97   11.27   12.61   12.61   45.46   15.72	H	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month		$\pm \pm$				000.0									_
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor  High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  UNCSX UDLS1 421.59 639.5 426.4 122.31 119.14  UNCSX 1L5XX 2.72  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  UNCSX UTFS 783.63 198.45 449.91 95.4 35.99 37.55 37.55 18.03  Nonrecurring Currently Combined Network Elements Switch -As-Is Char;  UNCSX UNCCC 12.97 11.27 12.61 12.61 45.46 15.72	H m Ir	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon		+	UNC3X	1L5XX	2.72										
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor  High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  UNCSX UDLS1 421.59 639.5 426.4 122.31 119.14  UNCSX 2.72  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  UNCSX UTFS 783.63 198.45 449.91 95.4 35.99 37.55 37.55 18.03  Nonrecurring Currently Combined Network Elements Switch -As-Is Char;  UNCSX UNCCC 12.97 11.27 12.61 12.61 45.46 15.72	H m Ir	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon	П	$\frac{1}{1}$	UNC3X	1L5XX	2.72		153.15	95.4	35.99			37.55	37.55	18.03	
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor  High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  UNCSX UDLS1 421.59 639.5 426.4 122.31 119.14  UNCSX 2.72  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mor  UNCSX UTFS 783.63 198.45 449.91 95.4 35.99 37.55 37.55 18.03  Nonrecurring Currently Combined Network Elements Switch -As-Is Char;  UNCSX UNCCC 12.97 11.27 12.61 12.61 45.46 15.72	H m lr	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mo	Н		UNC3X	1L5XX U1TF3	2.72	198.45								18.03	
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor   UNCSX   1L5ND   8.9	H m lr	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mo	))		UNC3X	1L5XX U1TF3	2.72	198.45								18.03	
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor   UNCSX   1L5ND   8.9	H n lr	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Charq	Ы		UNC3X	1L5XX U1TF3	2.72	198.45								18.03	
High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month   UNCSX   UDLS1   421.59   639.5   426.4   122.31   119.14     UNCSX   UDLS1   119.14     UNCSX   UDLS1   119.14   UNCSX   UDLS1   119.14   UNCSX   UDLS1   119.14   UNCSX   UDLS1   12.61	H m lr	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Charq	ni		UNC3X	1L5XX U1TF3	2.72	198.45								18.03	
Month   UNCSX   UDLS1   421.59   639.5   426.4   122.31   119.14	Ir N	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon  Nonrecurring Currently Combined Network Elements Switch - As-Is Charg  AL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)	))		UNC3X UNC3X UNC3X	1L5XX U1TF3 UNCCC	788	198.45								18.03	
Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor	H m Ir	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon  Nonrecurring Currently Combined Network Elements Switch - As-Is Charg  NEXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mon	))		UNC3X UNC3X UNC3X	1L5XX U1TF3 UNCCC	788	198.45								18.03	
Interoffice Transport - Dedicated - STS1 combination - Facility Termination per moi   UNCSX   U1TFS   783.63   198.45   449.91   95.4   35.99   37.55   37.55   18.03	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month  Nonrecurring Currently Combined Network Elements Switch - As-Is Charg  AL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Pacility Termination per Mile per month  High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per   - Transport - DS3 - DS4 - DS5 - DS5 - DS6 - DS	)		UNC3X UNC3X UNC3X UNC3X	1L5XX U1TF3 UNCCC	2.72 788 8.9	198.45	11.27	12.61	12.61					18.03	
Nonrecurring Currently Combined Network Elements Switch -As-Is Char; UNCSX UNCCC 12.97 11.27 12.61 12.61 45.46 15.72	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per monuncer in Company Comp	))		UNC3X  UNC3X  UNC3X  UNC3X  UNCSX  UNCSX	1L5XX U1TF3 UNCCC 1L5ND UDLS1	2.72 788 8.9 421.59	198.45	11.27	12.61	12.61					18.03	
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg UNCSX UNCCC 12.97 11.27 12.61 12.61 45.46 15.72	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per monuncer in Company Comp	31		UNC3X  UNC3X  UNC3X  UNC3X  UNCSX  UNCSX	1L5XX U1TF3 UNCCC 1L5ND UDLS1	2.72 788 8.9 421.59	198.45	11.27	12.61	12.61					18.03	
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg UNCSX UNCCC 12.97 11.27 12.61 12.61 45.46 15.72	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon  Nonrecurring Currently Combined Network Elements Switch - As-Is Charg  IL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mon  High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per  month  Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor	)		UNC3X  UNC3X  UNC3X  UNC3X  UNCSX  UNCSX  UNCSX	1L5XX U1TF3 UNCCC 1L5ND UDLS1 1L5XX	2.72 788 8.9 421.59 2.72	198.45 12.97 639.5	11.27	12.61	12.61			45.46	15.72		
	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon  Nonrecurring Currently Combined Network Elements Switch - As-Is Charg  IL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mon  High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per  month  Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor	21		UNC3X  UNC3X  UNC3X  UNC3X  UNCSX  UNCSX  UNCSX	1L5XX U1TF3 UNCCC 1L5ND UDLS1 1L5XX	2.72 788 8.9 421.59 2.72	198.45 12.97 639.5	11.27	12.61	12.61			45.46	15.72		
	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon  Nonrecurring Currently Combined Network Elements Switch - As-Is Charg  IL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mon  High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per  month  Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor	31		UNC3X  UNC3X  UNC3X  UNC3X  UNCSX  UNCSX  UNCSX	1L5XX U1TF3 UNCCC 1L5ND UDLS1 1L5XX	2.72 788 8.9 421.59 2.72	198.45 12.97 639.5	11.27	12.61	12.61			45.46	15.72		
2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month	31		UNC3X UNC3X UNC3X  UNC3X  UNCSX UNCSX UNCSX UNCSX UNCSX	UNCCC  1L5ND  UDLS1 1L5XX  U1TFS	2.72 788 8.9 421.59 2.72	198.45 12.97 639.5	11.27 426.4 449.91	12.61 122.31 95.4	119.14			45.46 37.55	15.72 37.55		
2-WIKE ISUN EXTENDED LOUP WITH UST INTEKUFFICE TRANSPORT (EEL)	STS1 DIGITAL	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Per Mile per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month	11		UNC3X UNC3X UNC3X  UNC3X  UNCSX UNCSX UNCSX UNCSX UNCSX	UNCCC  1L5ND  UDLS1 1L5XX  U1TFS	2.72 788 8.9 421.59 2.72	198.45 12.97 639.5	11.27 426.4 449.91	12.61	119.14			45.46 37.55	15.72 37.55		
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 UNCNX U1L2X 21.89 233.38 180.38	STS1 DIGITAL H	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per month  Interoffice Transport - Dedicated - DS3 - Per Mile per mon  Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month  Nonrecurring Currently Combined Network Elements Switch -As-Is Charg  AL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month  Interoffice Transport - Dedicated - STS1 combination - Per Mile per more  Interoffice Transport - Dedicated - STS1 combination - Per Mile per more  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per month  Nonrecurring Currently Combined Network Elements Switch -As-Is Charg	31		UNC3X UNC3X UNC3X  UNC3X  UNCSX UNCSX UNCSX UNCSX UNCSX	UNCCC  1L5ND  UDLS1 1L5XX  U1TFS	2.72 788 8.9 421.59 2.72	198.45 12.97 639.5	11.27 426.4 449.91	12.61	119.14			45.46 37.55	15.72 37.55		

	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		2	UNCNX	U1L2X	25.27	233.38	180.38							
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		3	UNCNX	U1L2X	40.17	233.38	180.38							
	Interoffice Transport - Dedicated - DS1 combination - Per Mi			UNC1X	1L5XX	0.4523									
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per mor			UNC1X	U1TF1	78.47	194.63	141.51	132.25	46.16		33.63	27.49	19.88	11.85
	Channelization - Channel System DS1 to DS0 combination - per mor			UNC1X	MQ1	0	0	0	0	0					
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon		-	UNCNX	UC1CA	3.37	12.02	8.66							
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon		1	UNCNX	U1L2X	21.89	233.38	180.38				18.94	8.42		
	Additional 2-wire IDSN 200p in same DS finteronice fransport Combination - 2011		+ ' +	UNCINA	UILZX	21.03	233.30	100.30				10.34	0.42		
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon		2	UNCNX	U1L2X	25.27	233.38	180.38				18.94	8.42		
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon		3	UNCNX	U1L2X	40.17	233.38	180.38				18.94	8.42		
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per mon			UNCNX	UC1CA	3.37	12.02	8.66							
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			UNC1X	UNCCC		12.97	11.27	12.61	12.61		45.46	15.72	0	0
4 WIDE DO	1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT	(551)	+-+												
4-WIKE DS	First DS1 Loop in STS1 Interoffice Transport Combination - Zone	(EEL)	1	UNC1X	USLXX	55.53	443.2	138.69	0	0					
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	64.13	443.2	138.69	0	0					
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	101.93	443.2	138.69	0	0					
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor		<b>-</b>	UNCSX	1L5XX	2.72	110.2	100.00	- ŭ						
	Interoffice Transport - Dedicated - STS1 combination - Facility Terminati			UNCSX	U1TFS	783.63	198.45	449.91	95.4	35.99		33.63	27.49	19.88	11.85
	STS1 to DS1 Channel System conbination per mon			UNCSX	MQ3	182.04	103.24	87.41	0	18.12					
	DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1	11.02	12.02	8.66							
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	55.53	443.2	138.69	0	0		18.94	8.42		
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	64.13	443.2	138.69	0	0		18.94	8.42		
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	101.93	443.2	138.69	0	0		18.94	8.42		
	DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1	11.02	12.02	8.66							
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charc			UNCSX	UNCCC		12.97	11.27	12.61	12.61		45.46	15.72	0	0
	Nonrecurring Currently Combined Network Elements Switch -As-is Char			UNCOA	UNCCC		12.97	11.21	12.01	12.01		43.40	13.72	U	U
4-WIRE 56	KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)														
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		1	UNCDX	UDL56	25.75	384.56	241.2							
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		2	UNCDX	UDL56	29.74	384.56	241.2							
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		3	UNCDX	UDL56	47.27	384.56	241.2							
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per M			UNCDX	1L5XX	0.0222									
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminati			UNCDX	U1TD5	16.45	147.07	111.75				33.63	27.49	19.88	11.85
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNCDX	UNCCC		12.97	11.27	12.61	12.61		45.46	15.72		
4 WIDE 64	│ KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)														
4-WINE 04	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone	1	1	UNCDX	UDL64	25.75	348.55	241.2							
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		2	UNCDX	UDL64	29.74	348.55	241.2							
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		3	UNCDX	UDL64	47.27	348.55	241.2							
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M			UNCDX	1L5XX	0.0222									
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminati			UNCDX	U1TD6	16.45	147.07	111.75				33.63	27.49	19.88	11.85
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNCDX	UNCCC		12.97	11.27	11.27	12.61		45.46	15.72		
ADDITIONAL NETWORK	( EL EMENTO														
ADDITIONAL NETWORK	K ELEMENIS		+												
When used	as a part of a currently combined facility, the non-recurring charges do not apply,	hut a Swi	itch As	le charne does ar	nlv										
	as ordinarilty combined network elements in Georgia, the non-recurring charges a														
1111011 4004	as oralizating combined notifier combined in seeingle, the new recurring oral geo a	pp.y aa .	1	ton 7 to 10 ontaingo											
Node (Sync	chroNet)														
	Node per month			UNCDX	UNCNT	13.98									
Nonrecurri		ah aambi	nation		1										
Nomecum	ng Currently Combined Network Elements "Switch As Is" Charge (One applies to ea		nauon)												
Nomecum	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion		nation)												
Nomecum	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	on	nation)	UNCVX	UNCCC		12.97	11.27	12.61	12.61		18.94	18.94		
Nomecum	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 156/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion	on	nation)												
Nonecum	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	on	nation)	UNCVX	UNCCC		12.97 12.97	11.27	12.61	12.61		18.94 18.94	18.94 18.94		
Nonecum	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	on	nation)	UNCDX	UNCCC		12.97	11.27	12.61	12.61		18.94	18.94		
Noneculli	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 156/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion	on	nation)												
Nomecum	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char	oh I	nation)	UNCDX UNC1X	UNCCC		12.97 12.97	11.27 11.27	12.61	12.61 12.61		18.94	18.94 18.94		
Noneturn	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charden Channel used in a COMBINATION - "Switch As Is" Conversion Charden Channel used in a COMBINATION - "Switch As Is" Conversion Charden Channel used in a COMBINATION - "Switch As Is" Conversion Charden Channel used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a COMBINATION - "Switch As Is" Conversion Charden Channel Used in a Combination Charden Channel Used in a Combination Charden Channel Used Chann	oh I		UNCDX	UNCCC		12.97	11.27	12.61	12.61		18.94	18.94		
Noneturn	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char	oh I		UNCDX UNC1X	UNCCC		12.97 12.97	11.27 11.27	12.61	12.61 12.61		18.94	18.94 18.94		
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge	oh i		UNCDX UNC1X UNC3X UNCSX	UNCCC UNCCC		12.97 12.97 12.97	11.27 11.27 11.27	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94	18.94 18.94		
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  at Channel - Dedicated Transport - minimum billing period - Below DS3=one month,	oh i		UNCDX UNC1X UNC3X UNCSX UNCSX	UNCCC UNCCC UNCCC		12.97 12.97 12.97 12.97	11.27 11.27 11.27 11.27	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94 18.94	18.94 18.94 18.94 18.94		
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Topic Charge  DS3 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  All Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon	oh i		UNCDX  UNC1X  UNC3X  UNCSX  sfour months  UNCXV	UNCCC UNCCC UNCCC UNCCC	13.91	12.97 12.97 12.97 12.97 272.07	11.27 11.27 11.27 11.27	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94 18.94	18.94 18.94 18.94 18.94		
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  al Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 4-Wire Voice Grade per mon	oh i		UNCDX  UNC1X  UNC3X  UNCSX  Efour months  UNCXV  UNCXV	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC	14.99	12.97 12.97 12.97 12.97 12.97 272.07	11.27 11.27 11.27 11.27 60.43 60.43	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94 18.94	18.94 18.94 18.94 18.94		
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Topic Charge  DS3 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  All Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon	oh i		UNCDX  UNC1X  UNC3X  UNCSX  sfour months  UNCXV	UNCCC UNCCC UNCCC UNCCC		12.97 12.97 12.97 12.97 272.07	11.27 11.27 11.27 11.27	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94 18.94	18.94 18.94 18.94 18.94		
NOTE: Loca	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversior Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversior Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  al Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - DS1 Per Mont	oh i		UNCDX  UNC1X  UNC3X  UNCSX  Efour months  UNCXV  UNCXV	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC	14.99	12.97 12.97 12.97 12.97 12.97 272.07	11.27 11.27 11.27 11.27 60.43 60.43	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94 18.94	18.94 18.94 18.94 18.94		
NOTE: Loc:	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char ST31 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  al Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon Local Channel - Dedicated - 1-Wire Voice Grade per mon	DS3 and	above=	UNCDX  UNC1X  UNC3X  UNCSX  sfour months  UNCXV  UNCXV  UNCXV  UNCXV  UNC1X	UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV2 ULDV4 ULDF1	14.99 38.36	12.97 12.97 12.97 12.97 272.07 272.07 164.99	11.27 11.27 11.27 11.27 11.27 60.43 60.43 113.76	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94 18.94	18.94 18.94 18.94 18.94		
NOTE: Loci	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversior Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversior Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  al Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - DS1 Per Mont	DS3 and	above=	UNCDX UNC1X UNC3X UNCSX UNCSX IOUNCSY UNCXY UNCXY UNCXY UNC1X UNC1X UNC1X	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV4 ULDV4 ULDF1	14.99 38.36 ges as ordered	12.97 12.97 12.97 12.97 272.07 272.07 164.99	11.27 11.27 11.27 11.27 11.27 60.43 60.43 113.76	12.61 12.61 12.61	12.61 12.61 12.61		18.94 18.94 18.94 18.94	18.94 18.94 18.94 18.94		

NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LS	R basis												_
Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces													_
(Regional)			SOMEC		3.5								
The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to G	eographically	y Deaveraged UNE	Zones. To	view Geograph	ically Deaverag	ged UNE Zone I	Designations by	Central Office,	refer to Internet We	ebsite:			
http://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm													
			_					,					
D LOCAL EXCHANGE SWITCHING(PORTS)													_
Exchange Ports													Τ
NOTE: Although the Port Rate includes all available features in GA & TN, the desired features wil	need to be o	ordered using reta	il USOCs										Τ
													Ξ
2-WIRE VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port- Re:		UEPSR	UEPRL	1.85	17.16	17.16				18.94	8.42		_
Exchange Forts - 2-Wile Analog Line Fort- Re:		UEFSK	UEFKL	1.00	17.16	17.10				10.94	0.42		-
Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re		UEPSR	UEPRC	1.85	17.16	17.16				18.94	8.42		
Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU		UEPSR UEPSR	UEPRO UEPAP	1.85 1.85	17.16	17.16 17.16				18.94 18.94	8.42 8.42		_
Exchange Forts - 2-vviie vo unbunded les, low usage line port with caller to (Lo		OLF SIX	OLIA	1.03		17.10				10.54	0.42		-
Subsequent Activity	$-\!\!\!\!+\!\!\!\!\!-$	UEPSR	USASC	0	0	0							_
FEATURES All Available Vertical Feature		UEPSR	UEPVF	0	0	0				18.94	8.42	<del> </del>	_
		OLI SIX	OLI VI	0	U	0				10.34	0.42		-
2-WIRE VOICE GRADE LINE PORT RATES (BUS)													Ξ
Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bi  Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484		UEPSB	UEPBL	1.85	17.16	17.16				18.94	8.42		_
ID - Bus.		UEPSB	UEPBC	1.85	17.16	17.16				18.94	8.42		
Exchange Ports - 2-Wire Analog Line Port outgoing only - Bu		UEPSB	UEPBO	1.85	17.16	17.16				18.94	8.42		_
Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B		UEPSB	UEPB1	1.85	17.16	17.16				18.94	8.42		_
Subsequent Activity		UEPSB	USASC	0	0	0							
FEATURES													_
All Available Vertical Feature  EXCHANGE PORT RATES (DID & PBX)		UEPSB	UEPVF	0	0	0				18.94	8.42		_
Exchange Ports - 2-Wire DID Port		UEPEX	UEPP2	11.35	61.91	61.91				19.99	19.99	19.99	-
Exchange Folio E Mile Bib Foli			OZ. I Z	11.00		01.01				10.00	10.00	10.00	_
Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabilit		UEPDD	UEPDD	120.8	108.38	60.88				19.99	19.99	19.99	_
Exchange Ports - 2-Wire ISDN Port (See Notes below All Features Offerec		UEPTX UEPSX UEPTX UEPSX		13.47	47.37 0	47.37 0				39.98	39.98		_
NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to o	circuit switche	d voice and/or circu	uit switched	data transmissio	on by B-Channe	els associated v	rith 2-wire ISDN	ports.					_
NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/Ne	w Business R	Request Process. F	Rates for the	e packet capabili	ities will be det	ermined via the	Bona Fide Requ	est/New Busine	ess Request Proces	SS.			_
Exchange Ports - 2-Wire ISDN Port Channel Profiles		UEPTX UEPSX	U1UMA	0	0	0							Ξ
Exchange Ports - 4-Wire ISDN DS1 Por		UEPEX	UEPEX	163.16	186.8	186.8				37.88	37.88		_
2-Wire VG Unbundled 2-Way PBX Trunk - Re:		UEPSE	UEPRD	1.85	17.16	17.16				18.94	8.42		
2 17110 10 Officialistic 2-174 ay 1 DA Truth - 1/C:	-	OLT SE	OLFRD	1.03	17.10	17.10				10.34	0.42		_
2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu		UEPSP	UEPPC	1.85	17.16	17.16				18.94	8.42		_
O Wise VC Line Cide Helphandled Out and DDV To all D		LIEDOD	LIEDDO	4.05	47.40	47.40				40.04	0.40		
2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu	-+	UEPSP UEPSP	UEPPO UEPP1	1.85 1.85	17.16 17.16	17.16 17.16				18.94 18.94	8.42 8.42	<del>                                     </del>	_
2-Wire Vo Line Side Orbotholed Incoming PBX Trunk - Bt  2-Wire Analog Long Distance Terminal PBX Trunk - Bu	-+	UEPSP	UEPLD	1.85	17.16	17.16				18.94	8.42		-
2-Wire Voice Unbundled PBX LD Terminal Port		UEPSP	UEPLD	1.85	17.16	17.16				18.94	8.42		Ξ
2-Wire Vice Unbundled 2-Way PBX Usage Po 2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPSP UEPSP	UEPXA UEPXB	1.85 1.85	17.16 17.16	17.16 17.16				18.94 18.94	8.42 8.42		_
z-wire voice undunded PBA Toll Terminal Hotel Por	-+	UEPSP	UEPXB	1.65	17.76	17.76				16.94	6.42		-
2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPSP	UEPXC	1.85	17.16	17.16				18.94	8.42		
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc		UEPSP UEPSP	UEPXD	1.85 1.85	17.16 17.16	17.16 17.16				18.94 18.94	8.42 8.42		_
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pf  2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		UEPSP	UEPAE	1.85	17.16	17.10				10.94	6.42		-
Port		UEPSP	UEPXL	1.85	17.16	17.16				18.94	8.42		
					.=								
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling		UEPSP	UEPXM	1.85	17.16	17.16				18.94	8.42		_
12-VVIIIE VOICE URDURIUSEU 1-VVAY OULGOING PBA HOLEI/HOSPITAL DISCOURT KOOM CAILING		UEPSP	UEPXO	1.85	17.16	17.16				18.94	8.42		
Port			UEPXS	1.85	17.16	17.16				18.94	8.42		
		UEPSP	OLI AU										
Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc													
Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc Subsequent Activity		UEPSP	USASC	0	0	0							_
Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc			USASC		0	0				18.94	8.42		_

NOTE: Transmission/usage charges associated with POTS circuit sw											Duele see D	Dece				
NOTE: Access to B Channel or D Channel Packet capabilities will be	available only through BFR/New	Busines	ss Re	quest Process. Ra	tes for the	e packet capab	ilities will be dete	ermined via the	Bona Fide I	Request/New	Business Re	equest Proce	ISS.			
DLED LOCAL SWITCHING, PORT USAGE		+	-													
DLED LOCAL SWITCHING, FORT USAGE		+														
End Office Switching (Port Usage)		-+														
End Office Switching Function, Per MOI	-	-+				0.0016333										+
End Office Trunk Port - Shared, Per MOL		-				0.0001564										
End office frankt of office of the		-				0.0001001										
Tandem Switching (Port Usage) (Local or Access Tandem)																
Tandem Switching Function Per MOI						0.0006757										
Tandem Trunk Port - Shared, Per MOI						0.0002126										
Common Transport																
Common Transport - Per Mile, Per MOL						800000.0										
Common Transport - Facilities Termination Per MO						0.0004152										
DLED PORT/LOOP COMBINATIONS - COST BASED RATES																
Cost Based Rates are applied where BellSouth is required by FCC ar	id/or State Commission rule to pro	ovide U	Inbun	dled Local Switchin	g or Swite	h Ports.										
Features shall apply to the Unbundled Port/Loop Combination - Cost	Based Rate section in the same r	manner	as th	ey are applied to th	e Stand-	Alone Unbundle	ed Port section of	of this Rate Exh	ibit.	L.,						
End Office and Tandem Switching Usage and Common Transport Us																
For Georgia and Tennessee, the recurring UNE Port and Loop charge	as listed apply to Currently Combi	ined and	d Not	Currently Combine	d Combo	s and the first a	ind additional Po	ort nonrecurring	charges ap	ply to Not Cur	rently Comb	oined Combo	s. For			
Currently Combined Combos in GA, TN and all other states, the nonre	ecurring charges shall be those id	Jentified	l in the	e Nonrecurring - Cι	rrently Co	ombined sectio	ns.									
																1
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)																1
																1
UNE Port/Loop Combination Rates																
2-Wire VG Loop/Port Combo - Zone			1			12.59										
2-Wire VG Loop/Port Combo - Zone			2			14.26										
2-Wire VG Loop/Port Combo - Zone			3			21.62										
UNE Loop Rates																
2-Wire Voice Grade Loop (SL1) - Zone		-	1	UEPRX	UEPLX	10.8										
2-Wire Voice Grade Loop (SL1) - Zone			3	UEPRX	UEPLX	12.47 19.83										-
2-Wire Voice Grade Loop (SL1) - Zone		-+	3	UEPRX	UEPLX	19.83										-
2-Wire Voice Grade Line Port Rates (Res)		-+	-													
2-Wire voice unbundled port - residenc		-		UEPRX	UEPRL	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
2-Wife voice unbullated port - residenc		-+	-	UEFKA	UEFKL	1.79	22.14	15.25	0.40	3.91			33.07	1.00		
2-Wire voice unbundled port with Caller ID - re				UEPRX	UEPRC	1.79	22.14	15.25	8.45	3.91			37.06	7.88		
2-Wile Voice dribundled port with Galler 15 - 16	-	-+		OLFIX	OLFIC	1.75	22.14	13.23	0.43	3.51			37.00	7.00		
2-Wire voice unbundled port outgoing only - re				UEPRX	UEPRO	1.79	22.14	15.25	8.45	3.91			33.67	7.88		
2-Wire voice unbundles res, low usage line port with Call	ler ID (LUI	-		UEPRX	UEPAP		22.14	15.25	8.45	3.91			33.67	7.88		1
E TYTE TOIGE UNDANGING TOO; TOW GOOGO THIS PORT WAT OUT	o. 15 (201			OL: TO	021711	1.70		10.20	0.10	0.01			00.01	7.00		
FEATURES																
All Features Offered				UEPRX	UEPVF	0	0	0					33.67	7.88		
LOCAL NUMBER PORTABILITY																
Local Number Portability (1 per port				UEPRX	LNPCX	0.35										
				·												
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED																
2-Wire Voice Grade Loop / Line Port Combination - Conv	/ersion - Switch-as		Ţ	UEPRX	USAC2		2.01	0.3108					33.67	7.88		
					l											
2-Wire Voice Grade Loop / Line Port Combination - Con	version - Switch with chan	$\longrightarrow$		UEPRX	USACC		2.01	0.3108	1	<b>_</b>			33.67	7.88		1
ADDITIONAL NIDO		-+	-+												-	+
ADDITIONAL NRCs		$-\!\!\!\!-\!\!\!\!\!+$		I IEEE N					1	<b>_</b>						+
2-Wire Voice Grade Loop/Line Port Combination - Subse	equent Activ	-+		UEPRX	USAS2	0	0	0	1	1						+
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)		-+														+
2-WINE VOICE GRADE LOUP WITH 2-WINE LINE PORT (BUS)		-+	-+		-				1		-				<b> </b>	+
UNE Part/Loop Combination Deta-		-+														+-
UNE Port/Loop Combination Rates		-+	1			12 59										+
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone		-+	2			12.59										+
2-Wire VG Loop/Port Combo - Zone			3		-	21.62			1						l	+
z-vviie vo Loop/Fort Combo - Zone	<del></del>	-+	3			21.02			1	1				1		+
UNE Loop Rates	<del></del>	-+	-						1	1				1		+
2-Wire Voice Grade Loop (SL1) - Zone	<del></del>	-+	1	UEPBX	UEPLX	10.8			<b>+</b>							+
2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	<del></del>		2	UEPBX	UEPLX	12.47			1							+
2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	<del></del>		3	UEPBX	UEPLX	19.83			1							+
2 1110 10100 01000 2000 (02.1) 2010	-			02. 5/.	JE. 2X	10.00										1
	-	-														1
2-Wire Voice Grade Line Port (Bus)							1		4	4				-		1-
2-Wire Voice Grade Line Port (Bus)  2-Wire voice unbundled port without Caller ID - bu		-	Į	UEPBX	UEPBI	1.79	22.14	15.25	8.45	3,91			33.67	7.88		
2-Wire Voice Grade Line Port (Bus)  2-Wire voice unbundled port without Caller ID - bu		_		UEPBX	UEPBL	1.79	22.14	15.25	8.45	3.91			33.67	7.88		

		HEDDY	LIEDDO	4.70	00.44	45.05	0.45	0.04		0.07	7.00		
2-Wire voice unbundled port outgoing only - bu 2-Wire voice unbundled incoming only port with Caller ID - Bu		UEPBX UEPBX	UEPBO UPEB1	1.79 1.79	22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91		3.67 3.67	7.88 7.88		+
		OL. DA	0. 25.	10	22.11	10.20	0.10	0.01	Ŭ,	0.07	7.00		1
LOCAL NUMBER PORTABILITY													
Local Number Portability (1 per port		UEPBX	LNPCX	0.35									+
FEATURES													+
All Features Offered		UEPBX	UEPVF	0	0	0			3:	3.67	7.88		T
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED		LIEDDY	110400		0.04	0.0100				0.07	7.00		4
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	-	UEPBX	USAC2		2.01	0.3108			3.	3.67	7.88		+
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		UEPBX	USACC		2.01	0.3108							
ADDITIONAL NRCs		LIEBBY											_
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX	USAS2						33	3.67	7.88		+
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)													+
													T
UNE Port/Loop Combination Rates													
2-Wire VG Loop/Port Combo - Zone	1			12.59									_
2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :	3			14.26 21.62									+-
2-Wile VG Loop/Fort Combo - Zone	- 3			21.02									+
UNE Loop Rates													I
2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPRG	UEPLX	10.8									
2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPRG	UEPLX	12.47									+
2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPRG	UEPLX	19.83									+
2-Wire Voice Grade Line Port Rates (RES - PBX)													+
													Т
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - R€		UEPRG	UEPRD	1.79	22.14	15.25	8.45	3.91	3:	3.67	7.88		_
LOCAL NUMBER PORTABILITY	$\longrightarrow$												+
Local Number Portability (1 per port		UEPRG	LNPCP	3.5									+
Essai Hamber I Stability (1 per per		021110	2.1. 0.	0.0									T
FEATURES													
All Features Offered		UEPRG	UEPVF	0	0	0			3:	3.67	7.88		┷
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED													+
NONRECORRING CHARGES (INCS) - CORRENTET COMBINED													+
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPRG	USAC2		2.01	0.3108			3:	3.67	7.88		
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with													
Change		UEPRG	USACC		2.01	0.3108			3:	3.67	7.88		+-
ADDITIONAL NRCs													+
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USAS2	0	0	0							T
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64			19	9.99	19.99	19.99	
													#
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)													+
UNE Port/Loop Combination Rates													_
2-Wire VG Loop/Port Combo - Zone	1 2			12.59									+
2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :	3			14.26 21.62									+
2 Wile vo Edoph of Combo Zone				21.02									T
UNE Loop Rates													
2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPPX	UEPLX	10.8									_
2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPPX UEPPX	UEPLX	12.47 19.83									+
2 Wile Voice Grade 2009 (GE 1) Zone		OLITA	OLI LX	13.00									+
2-Wire Voice Grade Line Port Rates (BUS - PBX)													
													1
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı	-+	UEPPX	UEPPC	1.79	22.14	15.25	8.45	3.91	3:	3.67	7.88		+
Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	1.79	22.14	15.25	8.45	3.91	3.	3.67	7.88		
Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX	UEPP1	1.79	22.14	15.25	8.45	3.91		3.67	7.88		T
2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	1.79	22.14	15.25	8.45	3.91	3:	3.67	7.88		Т
2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc	$\Box$	UEPPX	UEPXA	1.79	22.14	15.25	8.45	3.91		7.06	7.88	-	╨
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por	-+	UEPPX	UEPXB	1.79	22.14	15.25	8.45	3.91	3:	3.67	7.88		+
2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	1.79	22.14	15.25	8.45	3.91	3:	3.67	7.88		
1000 010001000 1 07 E5 555 101111100110	-	JEI I A	521 70		22.17	.5.25	0.40	5.51	3.				T
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	1.79	22.14	15.25	8.45	3.91	3:	3.67	7.88		
2-Wire Voice Unburidled PBX LD Terminal Switchboard IDD Capable Pc		UEPPX	UEPXE		22.14	15.25	8.45	3.91		3.67	7.88		

2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		UEPPX	UEPXL	1.79	22.14	15.25	8.45	3.91	33.67	7.88		
			OLI AL	1.79		13.23	0.40	5.51	33.07	7.00		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX	UEPXM	1.79	22.14	15.25	8.45	3.91	33.67	7.88		
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port		UEPPX	UEPXO	1.79	22.14	15.25	8.45	3.91	33.67	7.88		
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX	UEPXS	1.79	22.14	15.25	8.45	3.91	33.67			
LOCAL NUMBER PORTABILITY		UEPPX	LNDCD	2.45								
Local Number Portability (1 per port		UEPPX	LNPCP	3.15								
FEATURES												
All Features Offered		UEPPX	UEPVF	0	0	0			33.67	7.88		
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED												-
The state of the s												
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPPX	USAC2		2.01	0.3108			33.67	7.88		
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change		UEPPX	USACC		2.01	0.3108			33.67	7.88		
Onlange		OLITA	USACC		2.01	0.5100			33.07	7.00		
ADDITIONAL NRCs												
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ  PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi		UEPPX	USAS2	0	0 14.64	0 14.64			19.99	19.99	19.99	-
PBX Subsequent Activity - Change/Rearrange Multilline Hunt Grot					14.04	14.04			19.99	19.99	19.99	
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT												
NE Book and Continue Date												
JNE Port/Loop Combination Rates    2-Wire VG Coin Port/Loop Combo – Zone 1	-			12.69								-
2-Wire VG Coin Port/Loop Combo – Zone 2				14.36								+
2-Wire VG Coin Port/Loop Combo – Zone 3				21.72								
JNE Loop Rates												
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	10.8								
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	12.47								
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	19.83								_
-Wire Voice Grade Line Ports (COIN)												
2-Wire Coin 2-Way with Operator Screening (GA)		UEPCO	UEPGC	1.89	22.14	15.25	8.45	3.91	33.67	7.88		
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA)		LIEBOO	LIEDOO	4.00	00.44	45.05	0.45	0.04	00.07	7.00		
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA)		UEPCO UEPCO	UEP2G UEPGA	1.89 1.89	22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91	33.67 33.67			-
2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA)		UEPCO	UEPGB	1.89	22.14	15.25	8.45	3.91	33.67			
2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and		UEPCO	UEPCH	1.89	22.14	45.05	8.45	3.91	00.07	7.88		
Local (GA)  2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, M		UEPCO	UEPRJ	1.89	22.14	15.25 15.25	8.45	3.91	33.67 33.67			
2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+,												
and Local (FL, GA)  2-Wire 2-Way Smartline with 900/976 (all states except L/		UEPCO	UEPCQ	1.89	22.14	15.25	8.45 8.45	3.91	33.67			
2-Wire 2-way Smartline with 900/976 (all states except L)  2-Wire Coin Outward Smartline with 900/976 (all states except L)		UEPCO UEPCO	UEPCK	1.89 1.89		15.25 15.25	8.45	3.91 3.91	33.67 33.67			
ADDITIONAL UNE COIN PORT/LOOP (RC)												
					_							
UNE Coin Port/Loop Combo Usage (Flat Rate		UEPCO	URECU	3.59	0	0				-		+
OCAL NUMBER PORTABILITY												
Local Number Portability (1 per port		UEPCO	LNPCX	0.35								
FEATURES	+							-				-
All Features Offered		UEPCO	UEPVF	0	0	(			18.94	8.42		1
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPCO	USAC2		2.01	0.3108			33.67	7.88		+
2-vviile voice Grade Loop / Line Fort Combination - Conversion - Switch-as		UEFCU	USAUZ		2.01	0.3108			33.67	1.00		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		UEPCO	USACC		2.01	0.31			33.67	7.88		
ADDITIONAL NRCs										_	1	+
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPCO	USAS2		0	0	1		33.67	7.88	1	
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT											1	-
UNE Port/Loop Combination Rates										-		+
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	1			28.19								+
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	2			30.8								
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	3			42.27			-	-				-
			1	1						_	-	+
JNE Loop Rates  2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone  2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	1 2	UEPPX UEPPX	UECD1	16.84 19.45	104.78 104.78	78.1 78.1			19.99 19.99		19.99 19.99	1

UNE Port Rate													_
Exchange Ports - 2-Wire DID Por		U	EPPX	UEPD1	11.35	61.91	61.91			19.99	19.99	19.99	Ξ
NONRECURRING CHARGES - CURRENTLY COMBINED													_
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-			EPPX	USAC1		93.38	93.38			19.99	19.99	19.99	_
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable			LIIX	CONOT		30.30	33.30			13.33	13.55	13.33	-
Changes		U	EPPX	USA1C		93.38	93.38			19.99	19.99	19.99	
													Ξ
ADDITIONAL NRCs													_
Telephone Number/Trunk Group Establisment Charges													-
DID Trunk Termination (One Per Port		U	EPPX	NDT	0	0	0			19.99	19.99	19.99	-
DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbe		Ü	EPPX	NDZ	0	0	0			19.99	19.99	19.99	Τ
Additional DID Numbers for each Group of 20 DID Numbe		U	EPPX	ND4	0	0	0			19.99	19.99	19.99	Ξ
DID Numbers, Non- consecutive DID Numbers , Per Numbe			EPPX	ND5	0	0	0		19.99				_
Reserve Non-Consecutive DID number			EPPX	ND6	0	0	0		19.99				_
Reserve DID Numbers		U	EPPX	NDV	0	0	0		19.99				_
LOCAL NUMBER PORTABILITY													-
Local Number Portability (1 per port		U	EPPX	LNPCP	3.15								
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT													_
2-WIRE ISON DIGITAL GRADE LOOP WITH 2-WIRE ISON DIGITAL LINE SIDE PORT													-
UNE Port/Loop Combination Rates													
OW ICON Digital Conda Lagra (OW ICON Digital Line Cide Dest. LINE 7			EPPB EPPR		25.20								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	1	U	EPPK		35.36								-
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	2	UEPPB	UEPPF	2	38.74								
OW IODN District Control on Ion Ion Ion Ion Ion Ion Ion Ion Ion		UEPPB	UEPPF		50.04								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	3	UEPPB	UEPPI	C	53.64								-
UNE Loop Rates													
2-Wire ISDN Digital Grade Loop - UNE Zone	1	UEPPE	UEPPF	USL2X	21.89	252.32	188.77			19.99	19.99	19.99	-
2-Wire ISDN Digital Grade Loop - UNE Zone	2	UEPPB	UEPPF	USL2X	25.27	252.32	188.77			19.99	19.99	19.99	
O Maria IODNI District Constitution of INVESTIGATION		LIEDDE	LIEDDE	LIOLOV	40.47	050.00	100 77			40.00	40.00		
2-Wire ISDN Digital Grade Loop - UNE Zone	3	UEPPE	UEPPF	USL2X	40.17	252.32	188.77			19.99	19.99	19.99	_
UNE Port Rate													-
Exchange Port - 2-Wire ISDN Line Side Po		UEPPE	3 UEPPR	UEPPB	13.47	47.37				19.99	19.99	19.99	Ξ
NONRECURRING CHARGES - CURRENTLY COMBINED													_
2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination -													-
Conversion		UEPPE	3 UEPPR	USACB	0	93.38	93.38			19.99	19.99	19.99	
													Ξ
ADDITIONAL NRCs													_
2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy - Non Feature/Add													
Trunk		UEPPE	UEPPF	USASB		165.95				19.99	19.99	19.99	-
LOCAL NUMBER PORTABILITY													-
													_
Local Number Portability (1 per port		UEPPE	UEPPF	LNPCX	0.35	0	0						_
B-CHANNEL USER PROFILE ACCESS:													_
													-
CVS/CSD (DMS/5ESS)		UEPPE			0	0	0						_
CVS (EWSD)			UEPPR			0	0						_
CSD		UEPPE	3 UEPPR	U1UCC	0	0	0						_
B-CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)													-
USER TERMINAL PROFILE													_
User Terminal Profile (EWSD only)		UEPPE	UEPPR	U1UMA	0	0	0						
VERTICAL FEATURES													_
All Vertical Features - One per Channel B User Profile		UEPPE	UEPPR	UEPVF	0	0	0						
INTEROFFICE CHANNEL MILEAGE													-
-	_	1								İ			_
Interoffice Channel mileage each, including first mile and facilities termination		UEPPI	3 UEPPR	M1GNC	16.47	79.61	36.08			19.99	19.99	19.99	_

UNE Port/Loop Combination Rates						<del> </del>			+		1		$\dashv$
4W DS1 Digital Loop/4W	ISDN DS1 Digital Trunk Port - UNE Zone	1	UEPPP		218.69								1
4W DS1 Digital Loop/4W	ISDN DS1 Digital Trunk Port - UNE Zone	2	UEPPP		227.29								
4W DS1 Digital Loop/4W	ISDN DS1 Digital Trunk Port - UNE Zone	3	UEPPP		265.09								
UNE Loop Rates													
4-Wire DS1 Digital Loop	- UNE Zone	1	UEPPP	USL4P	55.53	448.92	276.6			19.99	19.99	19.99	
4-Wire DS1 Digital Loop		2	UEPPP	USL4P	64.13	448.92	276.6			19.99	19.99	19.99	
4-Wire DS1 Digital Loop	- UNE Zone	3	UEPPP	USL4P	101.93	448.92	276.6			19.99	19.99	19.99	
UNE Port Rate													_
Exchange Ports - 4-Wire	ISDN DS1 Por		UEPPP	UEPPP	163.16	186.8	186.8			19.99	19.99	19.99	
NONDECLIDENIA CUADOSO, OUDS	ENTLY COMPINED												4
NONRECURRING CHARGES - CURR	/ 4-Wire ISDN DS1 Digital Trunk Port Combination -												_
Conversion -Switch-as-is	7 4-Wile IODIN DOT Digital Trunk Fort Combination -		UEPPP	USACP	0	269.96	269.96			19.99	19.99	19.99	
ADDITIONAL NRCs	DN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within												4
Std Allowance	DN Digti Trk Port - Subsqt Activy- inward/two way tei nos within		UEPPP	PR7TF		0.9686				19.99	19.99	19.99	
	e ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All					0.0000				10.00	10.00	10.00	
States except NC			UEPPP	PR7TO		22.75	22.75			19.99	19.99	19.99	4
4-Wire DS1 Loop / 4-Wir Above Std Allowance	e ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos		UEPPP	PR7ZT		45.49	45.49			19.99	19.99	19.99	
Above dia Allowallet			OLI II	110721		70.70	40.40			10.99	10.00	10.00	
COAL NUMBER ROPTARY													
LOCAL NUMBER PORTABILITY  Local Number Portability	(1 per port		UEPPP	LNPCN	1.75	<del>                                     </del>			+				-
Local Number Portability	(1 per port		UEFFF	LINECIN	1.70	<del> </del>			+				+
INTERFACE (Provsioning Only)													
Voice/Data			UEPPP	PR71V	0	0	0						
Digital Data Inward Data			UEPPP UEPPP	PR71D PR71E	0	0	0						_
inward Data			UEPPP	PR/IE	U	U	0		+ +				_
New or Additional "B" Channel													
New or Additional - Voice			UEPPP	PR7BV	0	28.71				19.99	19.99	19.99	
New or Additional - Digital			UEPPP	PR7BF	0	28.71				19.99	19.99	19.99	
New or Additional Inward			UEPPP	PR7BD	0	28.71				19.99	19.99	19.99	
	e Sensitive Voice Data B Channel		UEPPP	PR7BS	0	28.71				19.99	19.99	19.99	
New or Additional Useag	e Sensitive Digital Data B Channel		UEPPP	PR7BU	0	28.71				19.99	19.99	19.99	_
CALL TYPES													1
Inward			UEPPP	PR7C1	0	0	0						
Outward			UEPPP	PR7C0	0	0	0						
Two-way			UEPPP	PR7CC	0	0	0						
Interoffice Channel Mileage													-
Fixed Each Including Fire	st Mili		UEPPP	1LN1A	78.9223	147.07	111.75	0		19.99	19.99	19.99	1
Each Airline-Fractional A	dditional Mil		UEPPP	1LN1B	0.4523								
4-WIRE DS1 DIGITAL LOOP WITH 4-	WIRE DOITS TRUNK PORT												_
	The second secon												
UNE Port/Loop Combination Rates			-										
4W DS1 Digital Loop/4W	DDITS Trunk Port - UNE Zone	1 2	UEPDC		176.33	<del>                                     </del>				19.99	19.99	19.99	4
	DDITS Trunk Port - UNE Zone DDITS Trunk Port - UNE Zone	3	UEPDC UEPDC		184.93 222.73	<del>                                     </del>				19.99 19.99	19.99 19.99	19.99 19.99	4
551 Digital 2500/4444			021 00							10.59	.5.55	.3.33	
UNE Loop Rates			·								<u> </u>		
4-Wire DS1 Digital Loop		1	UEPDC	USLDC	55.53	448.92	276			19.99	19.99	19.99	_
4-Wire DS1 Digital Loop 4-Wire DS1 Digital Loop	- UNE ZONE :	3	UEPDC UEPDC	USLDC	64.13 101.93	448.92 448.92	276.6 276.6		+ +	19.99 19.99	19.99 19.99	19.99 19.99	$\dashv$
			021 00	33200	.07.00		2.0.0			10.59	.5.55	.3.33	
UNE Port Rate						L					L		]
4-Wire DDITS Digital Tru	nk Por		UEPDC	UDD1T	120.8	89.44	52.46			19.99	19.99	19.99	-
NONRECURRING CHARGES - CURR	ENTLY COMBINED												+
4-Wire DS1 Digital Loop	/ 4-Wire DDITS Trunk Port Combination - Switch-as-		UEPDC	USAC4		269.96	269.96			19.99	19.99	19.99	
4-Wire DS1 Digital Loop	/ 4-Wire DDITS Trunk Port Combination - Conversion with												7
DS1 Changes	/ A Wise DDITC Trust Dark Combination Commission 19		UEPDC	USAWA		269.96	269.96			19.99	19.99	19.99	4
4-Wire DS1 Digital Loop Change - Trunk	/ 4-Wire DDITS Trunk Port Combination - Conversion with		UEPDC	USAWB		269.96	269.96			19.99	19.99	19.99	
			021 00	COMVD		200.00	200.00			15.55	10.00	10.00	7
ADDITIONAL NRCs													
	DDITS Trunk Port - Subsequent Service Activity Per Service												

													_
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel Activation/Chan - 2-Way Trunl		UEPDC	UDTTA		28.71	28.71				19.99	19.99	19.99	
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1 Way Outward Trunk	-	UEPDC	UDTTB		28.71	28.71				19.99	19.99	19.99	
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan Inward Trunk w/out DIC		UEPDC	UDTTC		28.71	28.71				19.99	19.99	19.99	
4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -			UDTTD							19.99			
Inward Trunk with DIC 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way		UEPDC			28.71	28.71					19.99	19.99	
DID w User Trans BIPOLAR 8 ZERO SUBSTITUTION		UEPDC	UDTTE		28.71	28.71				19.99	19.99	19.99	
B8ZS -Superframe Format	-	UEPDC	CCOSF		0	600				19.99	19.99	19.99	
B8ZS - Extended Superframe Forma		UEPDC	CCOEF		0	600				19.99	19.99	19.99	
Alternate Mark Inversion													
AMI -Superframe Format		UEPDC	MCOSF		0	0							
AMI - Extended SuperFrame Forms		UEPDC	MCOPC	)	0	0							
Telephone Number/Trunk Group Establisment Charges													
Telephone Number for 2-Way Trunk Grou		UEPDC	UDTGX	0									T
Telephone Number for 1-Way Outward Trunk Grou		UEPDC	UDTGY	0					19.99				
Telephone Number for 1-Way Inward Trunk Group Without DI		UEPDC	UDTGZ						19.99				-
DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbe DID Numbers for each Group of 20 DID Number		UEPDC UEPDC	NDZ ND4	0	0	0			19.99 19.99				+
DID Numbers, Non- consecutive DID Numbers , Per Numbe		UEPDC	ND5	0					19.99				+
Reserve Non-Consecutive DID Nos		UEPDC	ND6	0	0	0			19.99				T
Reserve DID Numbers		UEPDC	NDV	0	0	0			19.99				
D. C. ( ) DOA ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) (	DITO T												-
Dedicated DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DI Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination	JIIS Irunk i	UEPDC	1LNO1	78.47	147.07	111.75	0	0		19.99	19.99	19.99	+
Interoffice Channel Mileage - Additional rate per mile - 0-8 mil		UEPDC	1LNOA		0	0				13.33	15.55	15.55	╁
Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination		UEPDC	1LNO2	0	0	0							
Interoffice Channel Mileage - Additional rate per mile - 9-25 mil		UEPDC	1LNOB		0	0							
Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination		UEPDC	1LNO3		0	0	0						-
Interoffice Channel Mileage - Additional rate per mile - 25+ mil Local Number Portability, per DS0 Activate		UEPDC UEPDC	1LNOC LNPCP		0	0	0						+
Central Office Termininating Poir		UEPDC	CTG	0		-	0						
													+
4-WIRE DS1 LOOP WITH CHANNELIZATION WITH PORT System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations Each System can have up to 24 combinations of rates depending on type and number of ports w	sed												
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u	sed												
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  [4-Wire DS1 Loop - UNE Zone 1	sed 1	UEPMG	USLDC	55.53	0	0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2	1 2	UEPMG	USLDC	64.13	0	0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  [4-Wire DS1 Loop - UNE Zone 1	1		USLDC USLDC USLDC	64.13	-	0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3	1 2	UEPMG	USLDC	64.13	0								
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  [24 DS0 Channel Capacity - 1 per DS1	1 2	UEPMG UEPMG UEPMG	USLDC USLDC VUM24	64.13 101.93 102.64	0								
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DSO Channel Capacity - 1 per DS1  48 DSO Channel Capacity - 1 per 2 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG	USLDC USLDC VUM24 VUM48	64.13 101.93 102.64 205.28	0	0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  ach System can have up to 24 combinations of rates depending on type and number of ports u  JNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  JNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 4 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96	64.13 101.93 102.64 205.28 410.56	0 0 0 0 0 0	0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports usually support of the combination of	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14	64.13 101.93 102.64 205.28 410.56 615.84	0 0 0 0 0 0 0 0 0 0	0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 6 DS1s  144 DS0 Channel Capacity - 1 per 6 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14 VUM19	64.13 101.93 102.64 205.28 410.56 615.84 821.12	0 0 0 0 0 0 0	0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DSO Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 4 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  142 DS0 Channel Capacity - 1 per 8 DS1s  142 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 8 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14 VUM19 VUM20	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4	0 0 0 0 0 0 0	0 0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  JNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  JNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 8 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  192 DS0 Channel Capacity - 1 per 8 DS1s  248 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14 VUM19 VUM20 VUM28	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68	0 0 0 0 0 0 0 0	0 0 0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DSO Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 4 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  142 DS0 Channel Capacity - 1 per 8 DS1s  142 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 8 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14 VUM19 VUM20 VUM28	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24	0 0 0 0 0 0 0	0 0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  JNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  JNE DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 6 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 16 DS1s  384 DS0 Channel Capacity - 1 per 16 DS1s  384 DS0 Channel Capacity - 1 per 16 DS1s  480 DS0 Channel Capacity - 1 per 16 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14 VUM19 VUM20 VUM28 VUM38 VUM40 VUM57	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8 2463.36	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 4 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 16 DS1s  480 DS0 Channel Capacity - 1 per 16 DS1s	1 2	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMMG UEPMMG UEPMMG UEPMMG UEPMMG UEPMMG	VUM24 VUM48 VUM96 VUM14 VUM19 VUM20 VUM28 VUM38 VUM40 VUM57	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8	0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 6 DS1s  194 DS0 Channel Capacity - 1 per 6 DS1s  192 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 12 DS1s  384 DS0 Channel Capacity - 1 per 12 DS1s  675 DS0 Channel Capacity - 1 per 20 DS1s  576 DS0 Channel Capacity - 1 per 28 DS1s  672 DS0 Channel Capacity - 1 per 28 DS1s	1 2 3 3	UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14 VUM19 VUM20 VUM28 VUM38 VUM40 VUM57 VUM67	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8 2463.36	0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 4 DS1s  144 DS0 Channel Capacity - 1 per 6 DS1s  144 DS0 Channel Capacity - 1 per 6 DS1s  192 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 10 DS1s  480 DS0 Channel Capacity - 1 per 12 DS1s  384 DS0 Channel Capacity - 1 per 12 DS1s  480 DS0 Channel Capacity - 1 per 20 DS1s  576 DS0 Channel Capacity - 1 per 28 DS1s  672 DS0 Channel Capacity - 1 per 28 DS1s  Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - CA Minimum System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS0 OF TO SET T	1 2 3 3 Sonversion Conversion Control with Fe	UEPMG UEPMG	VUM24 VUM48 VUM96 VUM14 VUM19 VUM20 VUM28 VUM38 VUM40 VUM57 VUM67	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8 2463.36	0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0							
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channel S1 Loop - UNE Zone 3  UNE DS0 Channel Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 8 DS1s  192 DS0 Channel Capacity - 1 per 8 DS1s  192 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 16 DS1s  384 DS0 Channel Capacity - 1 per 16 DS1s  480 DS0 Channel Capacity - 1 per 16 DS1s  480 DS0 Channel Capacity - 1 per 16 DS1s  576 DS0 Channel Capacity - 1 per 20 DS1s  576 DS0 Channel Capacity - 1 per 24 DS1s  672 DS0 Channel Capacity - 1 per 28 DS1s  Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - CA  Molinimum System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS0 PM  Multiples of this configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS0 PM  Multiples of this configuration is one (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS0 PM  Multiples of this configuration is one (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS0 PM  Multiples of this configuration is one (1) DS1, One (1) D4 Channel Add Tafter the minimum system	1 2 3 3 Sonversion Conversion Control with Fe	UEPMG UEPMG	VUM24 VUM48 VUM49 VUM14 VUM19 VUM20 VUM20 VUM38 VUM40 VUM57 VUM67	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8 2463.36 2873.92	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0				19.99	19.99	19.99	19
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channelization Capacities (D4 Channel Bank Configurations)  24 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 6 DS1s  194 DS0 Channel Capacity - 1 per 8 DS1s  195 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 16 DS1s  384 DS0 Channel Capacity - 1 per 16 DS1s  480 DS0 Channel Capacity - 1 per 16 DS1s  480 DS0 Channel Capacity - 1 per 12 DS1s  675 DS0 Channel Capacity - 1 per 28 DS1s  672 DS0 Channel Capacity - 1 per 28 DS1s  Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - C	1 2 3 3 Sonversion Conversion Conversion Control with Fen configuration	UEPMG UEPMG	VUM24 VUM48 VUM46 VUM14 VUM19 VUM20 VUM20 VUM38 VUM40 VUM57 VUM67	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8 2463.36 2873.92	0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0				19.99	19.99	19.99	19
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations  Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channel Capacity - USE Zone 3  UNE DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per DS1s  96 DS0 Channel Capacity - 1 per B DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  144 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 10 DS1s  248 DS0 Channel Capacity - 1 per 10 DS1s  288 DS0 Channel Capacity - 1 per 16 DS1s  288 DS0 Channel Capacity - 1 per 16 DS1s  384 DS0 Channel Capacity - 1 per 16 DS1s  480 DS0 Channel Capacity - 1 per 18 DS1s  576 DS0 Channel Capacity - 1 per 2 DS1s  576 DS0 Channel Capacity - 1 per 2 DS1s  672 DS0 Channel Capacity - 1 per 2 DS1s  Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - CA Minimum System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS0 PNRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes  System Additions at End User Locations Where 4-Wire DS1 Loop with Channelization with Port CNew (Not Currently Combined) in Georgia & Tennessee Only	1 2 3 3 Sonversion Conversion Control with Fen configuration	UEPMG UEPMG	VUM24 VUM48 VUM46 VUM14 VUM19 VUM20 VUM20 VUM38 VUM40 VUM57 VUM67	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8 2463.36 2873.92	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0				19.99	19.99	19.99	19
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations Each System can have up to 24 combinations of rates depending on type and number of ports u  UNE DS1 Loop  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  UNE DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per DS1  48 DS0 Channel Capacity - 1 per 2 DS1s  96 DS0 Channel Capacity - 1 per 6 DS1s  192 DS0 Channel Capacity - 1 per 8 DS1s  194 DS0 Channel Capacity - 1 per 8 DS1s  240 DS0 Channel Capacity - 1 per 10 DS1s  280 DS0 Channel Capacity - 1 per 10 DS1s  280 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 10 DS1s  384 DS0 Channel Capacity - 1 per 10 DS1s  480 DS0 Channel Capacity - 1 per 10 DS1s  480 DS0 Channel Capacity - 1 per 12 DS1s  576 DS0 Channel Capacity - 1 per 20 DS1s  576 DS0 Channel Capacity - 1 per 28 DS1s  672 DS0 Channel Capacity - 1 per 28 DS1s  Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with ChannelEation with Port - C  A Minimum System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS0 PM  Multiples of this configuration functioning as one are considered Add" after the minimum system   NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes	1 2 3 3 Sonversion Conversion Control with Fen configuration	UEPMG UEPMG	VUM24 VUM48 VUM46 VUM14 VUM19 VUM20 VUM20 VUM38 VUM40 VUM57 VUM67	64.13 101.93 102.64 205.28 410.56 615.84 821.12 1026.4 1231.68 1642.24 2052.8 2463.36 2873.92	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0	144.05	17.09		19.99	19.99	19.99	19

	Class Channel Carability Farmet avantages Cuban wat Anti-ity Only		l .	JEPMG	CCOSF	0	0	600					19.99	19.99	19.99	19.99
	Clear Channel Capability Format, superframe - Subsequent Activity Only			JEPMG JEPMG	CCOEF		0	600					19.99	19.99	19.99	19.99
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only ark Inversion (AMI)			JEPMG	CCOEF	U	U	600					19.99	19.99	19.99	19.99
							_									
	Superframe Format Extended Superframe Format			JEPMG JEPMG	MCOSF		0	0								
	Extended Superframe Format			JEPMG	MCOPO	0	0	0							_	
Exchange P	Ports Associated with 4-Wire DS1 Loop with Channelization with Port															
Exchange P																
	Line Side Combination Channelized PBX Trunk Port - Business		L	JEPPX	UEPCX	1.79	0	0	0	0		19.99				
	Line Side Outward Channelized PBX Trunk Port - Business			JEPPX	UEPOX		0	0	0	0		19.99				
	Line Side Inward Only Channelized PBX Trunk Port without DID			JEPPX	UEP1X	1.79	0	0	0	0		19.99			_	
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port		ι	JEPPX	UEPDM	11.35	0	0	0	0		19.99				
	ivations - Unbundled Loop Concentration			, L	02. 0	11.00	Ů	-	0			10.00				
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		L	JEPPX	1PQWM	0.62	25.09	13.25	3.99	3.97			19.99	19.99	19.99	19.99
			[ [	-						1						1
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		L	JEPPX	1PQWU	0.62	77.21	18.2	56.49	11.04	-		19.99	19.99	19.99	19.99
	Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port)		<del>   </del>	JEPPX	NDT	0		_	_	_						1
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			JEPPX	NDZ	0	0	0	-	_	1	19.99				1
	DID Numbers - groups of 20 - Valid all States			JEPPX	ND4	0	0	0		+	1	19.99				1
	Non-Consecutive DID Numbers - per number			JEPPX	ND5	0	0	0			1	19.99		1		1
	Reserve Non-Consecutive DID Numbers		l	JEPPX	ND6	0	0	ō				T				
	Reserve DID Numbers		l	JEPPX	NDV	0	0	0								
	er Portability															
	Local Number Portability - 1 per port		L	JEPPX	LNPCP	3.15	0	0								1
	- Vertical and Optional		+					_		_						1
Local Switch	hing Features Offered with Line Side Ports Only  All Features Available		<del>                                     </del>	JEPPX	UEPVF	0	0	0				19.99			_	
	All Fediules Available			JEFFA	UEFVF	U	U	U				19.99				
															-	
D PORT LOO	DP COMBINATIONS - MARKET RATES															
D PORT LOO	DP COMBINATIONS - MARKET RATES															
Market Rates	s shall apply where BellSouth is not required to provide unbundled local switching or sv	witch port	s per F0	CC and/or State C	ommission	rules.										
Market Rates These scena	s shall apply where BellSouth is not required to provide unbundled local switching or sv arios include:															
Market Rates These scena 1. Unbundle	s shall apply where BellSouth is not required to provide unbundled local switching or sv arios include: ed port/loop combinations that are Not Currently Combined in all of the BellSouth state	s except	as note	d for Georgia and	Tennesse	э.	re with 4 or m	DS0 equiva	lent lines							
Market Rates These scena 1. Unbundle 2. Unbundle	s shall apply where BellSouth is not required to provide unbundled local switching or sv arios include:	s except	as noted the Top	d for Georgia and 8 MSAS in BellSo	Tennesse	e. n for end use				le).						
Market Rates These scena 1. Unbundle 2. Unbundle The Top 8 M	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I	s except one 1 of the New Orles	as noted the Top ans); NO	d for Georgia and 8 MSAS in BellSo C (Greensboro-W	Tennesse outh's regio inston Sale	e. n for end use m-Highpoint/0	Charlotte-Gas	tonia-Rock Hill)	TN (Nashvil							
Market Rates These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: add port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (t	s except one 1 of the New Orles	as noted the Top ans); NO	d for Georgia and 8 MSAS in BellSo C (Greensboro-W	Tennesse outh's regio inston Sale	e. n for end use m-Highpoint/0	Charlotte-Gas	tonia-Rock Hill)	TN (Nashvil		g in lieu of th	e Market Ra	ttes and rese	rves the right	to true-up the	billing di
Market Rates These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include:  ad port/loop combinations that are Not Currently Combined in all of the BellSouth state do port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (turrently is developing the billing capability to mechanically bill the recurring and non-received the properties of	s except one 1 of the New Orles	as noted the Top ans); No arket Ra	d for Georgia and 8 MSAS in BellSc C (Greensboro-W ates in this section	Tennesses outh's regio inston Sale	e. n for end use m-Highpoint/0 erim, BellSou	th shall bill th	e rates in the Co	TN (Nashvil	ection precedin						
Market Rates These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Istrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of the common state of the common states.	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibi	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	e. n for end use m-Highpoint/( erim, BellSou ons of loop/po	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rates These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Curr	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. Interpret of the port is compared to the port section of th	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibi	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	e. n for end use m-Highpoint/( erim, BellSou ons of loop/po	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rates These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Curr apply also ar	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are lis and are categorized accordingly.	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibi	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	e. n for end use m-Highpoint/( erim, BellSou ons of loop/po	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rates These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Curr apply also ar	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. Interpret of the port is compared to the port section of th	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibi	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	e. n for end use m-Highpoint/( erim, BellSou ons of loop/po	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Curr apply also ar 2-WIRE VOIC	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Isurrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of the combined scenarios where Market Rates apply, the Nonrecurring charges are listed are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibi	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	e. n for end use m-Highpoint/( erim, BellSou ons of loop/po	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market f End Office a For Not Curr apply also ar 2-WIRE VOIC UNE Port/Lo	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (furrently is developing the billing capability to mechanically bill the recurring and non-received for unbundled ports includes all available features in all states.  International Combined Scenarios where Market Rates apply, the Nonrecurring charges are listed acategorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibi	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market I End Office a For Not Curr apply also ar  2-WIRE VOIC	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include:  d port/loop combinations that are Not Currently Combined in all of the BellSouth state at port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Isarrently is developing the billing capability to mechanically bill the recurring and non-received the combined ports includes all available features in all states.  Interpret of the port is compared to the port of the port section of the port of the por	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Curr apply also ar 2-WIRE VOIC UNE Port/Lo	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are lis nd are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CD Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	e. n for end use m-Highpoint/C erim, BellSou ens of loop/po or each Port U	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scena 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Curr apply also ar 2-WIRE VOIC UNE Port/Lo	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include:  d port/loop combinations that are Not Currently Combined in all of the BellSouth state at port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Isarrently is developing the billing capability to mechanically bill the recurring and non-receivate for unbundled ports includes all available features in all states.  Interpretate the North Combined Scenarios where Market Rates apply, the Nonrecurring charges are listed are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates  2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarion 1. Unbundled 2. Unbundled The Top 8 M BellSouth cu The Market is End Office a For Not Curr apply also ar 2-WIRE VOIC UNE Port/Lo	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (furrently is developing the billing capability to mechanically bill the recurring and non-recently combined specifications all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSo C (Greensboro-W ates in this section t shall apply to all	Tennesses outh's regio inston Sale i. In the int	e. n for end use m-Highpoint/C erim, BellSou ens of loop/po or each Port U	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarion 1. Unbundle 2. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market f End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (furrently is developing the billing capability to mechanically bill the recurring and non-recently combined specifications all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt C (Greensboro-W ates in this section t shall apply to all ind Additional NRC UEPRX	Tennessei uuth's regioinston Sale in In the int combinatic columns fo	e. n for end use m-Highpoint/C erim, BellSou ens of loop/po or each Port U	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarion These scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. Interest of the combined scenarios where Market Rates apply, the Nonrecurring charges are lis and are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE Unive VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibite First are 1 2 3	d for Georgia and 8 MSAS in BellSt C (Greensboro-W stee in this section t shall apply to all ad Additional NRC UEPRX UEPRX	Tennesse; suth's regionston Sale in the int combination columns for	a. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U  24.8 26.47 33.83	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarion These scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo	s shall apply where BellSouth is not required to provide unbundled local switching or sy arios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (furrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of the company of the Combined scenarios where Market Rates apply, the Nonrecurring charges are listed are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); Noted arket Rase e exhibit of First and 1 2 3	d for Georgia and 8 MSAS in BellSt C (Greensboro-W ates in this section t shall apply to all ind Additional NRC UEPRX	Tennessei uuth's regioinston Sale in In the int combinatic columns fo	a. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U  24.8 26.47 33.83	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarion The Se scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WIRE VOIC UNE POrt/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z (SAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt C (Greensboro-W stee in this section t shall apply to all ad Additional NRC UEPRX UEPRX	Tennesse; suth's regionston Sale in the int combination columns for	a. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U  24.8 26.47 33.83	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarior These scenarior 1. Unbundle 2. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market f End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (If urrently is developing the billing capability to mechanically bill the recurring and non-recently for unbundled ports includes all available features in all states.  Market Port (Sage and Common Transport Usage rates in the Port section crently Combined scenarios where Market Rates apply, the Nonrecurring charges are listed are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE CHOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt. Greensboro-W ates in this section t shall apply to all ad Additional NRC UEPRX UEPRX UEPRX UEPRX	Tennesse; uuth's regionston Sale i. In the int combination combination columns for  UEPLX UEPLX UEPLX	n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U 24.8 26.47 33.83	Charlotte-Gas th shall bill th	e rates in the Comments except furrently Combination	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarior These scenarior 1. Unbundle 2. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market f End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z (SAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt C (Greensboro-W stee in this section t shall apply to all ad Additional NRC UEPRX UEPRX	Tennesse; suth's regionston Sale in the int combination columns for	a. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U  24.8 26.47 33.83	charlotte-Gas th shall bill th tr network ele	e rates in the Co	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarion These scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WiRE VOIC UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or surios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Isarrently is developing the billing capability to mechanically bill the recurring and non-recently combined scenarios that all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of the composition of the provided scenarios where Market Rates apply, the Nonrecurring charges are listed are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSc (Greensboro-W ates in this section t shall apply to all hid Additional NRC UEPRX UEPRX UEPRX	Tennesse; uth's regionston Sale in the int combinatic columns fo  UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX	n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U  24.8 26.47 33.83  10.8 12.47 19.83	Charlotte-Gas th shall bill th rt network elel ISOC. For Ci	e rates in the Cr ements except f urrently Combination	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usage the NRC - Cu	pe charge (US)	OC: URECU)	
Market Rate: These scenarion These scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WiRE VOIC UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (If urrently is developing the billing capability to mechanically bill the recurring and non-recently for unbundled ports includes all available features in all states.  Market Port (Sage and Common Transport Usage rates in the Port section crently Combined scenarios where Market Rates apply, the Nonrecurring charges are listed are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE CHOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt. Greensboro-W ates in this section t shall apply to all ad Additional NRC UEPRX UEPRX UEPRX UEPRX	Tennesse; uuth's regionston Sale i. In the int combination combination columns for  UEPLX UEPLX UEPLX	n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U 24.8 26.47 33.83	Charlotte-Gas th shall bill th	e rates in the Comments except furrently Combination	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag	ge charge (US	OC: URECU)	
Market Rate: These scenarion These scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or surios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. Into Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are lis nd are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt. Greensboro-W stees in this section t shall apply to all and Additional NRC UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	Tennesse; suth's regionston Sale . In the int combinatic columns fo  UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC	2. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U 24.8 26.47 33.83 10.8 12.47 19.83 14 14 14	charlotte-Gas th shall bill th rt network ele SOC. For Cri  90  90	e rates in the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furness and the Crements except furness except fur	TN (Nashvil	ection precedin	mbinations w	hich have a	33.67 33.67	rently Comb  7.88  7.88	OC: URECU)	
Market Rate: These scenarion These scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z (SAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I urrently is developing the billing capability to mechanically bill the recurring and non-recently combined scenarios includes all available features in all states.  Ind Tandem Switching Usage and Common Transport Usage rates in the Port section of are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port - residenc	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSc C (Greensboro-W stes in this section t shall apply to all hid Additional NRC UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	Tennesse; uth's regionston Sale . In the int combinatic columns for UEPLX UEPLX UEPLX UEPLX UEPRC	2. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U 24.8 26.47 33.83 10.8 12.47 19.83 14	charlotte-Gas th shall bill th th network ele SOC. For Cu	e rates in the Comments except if the comment	TN (Nashvil	ection precedin	mbinations w	hich have a	flat rate usag the NRC - Cu	rently Comb  7.88	OC: URECU)	
Market Rate: These scenarion The Se scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-WIRE VOIC UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or surios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I surrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are lis and are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port soutgoing only - re 2-Wire voice unbundled port outgoing only - re	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt. Greensboro-W stees in this section t shall apply to all and Additional NRC UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	Tennesse; suth's regionston Sale . In the int combinatic columns fo  UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC	2. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U 24.8 26.47 33.83 10.8 12.47 19.83 14 14 14	charlotte-Gas th shall bill th rt network ele SOC. For Cri  90  90	e rates in the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furness and the Crements except furness except fur	TN (Nashvil	ection precedin	mbinations w	hich have a	33.67 33.67	rently Comb  7.88  7.88	OC: URECU)	
Market Rate: These scenarion The Se scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-Wire Voice UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or svarios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (If urrently is developing the billing capability to mechanically bill the recurring and non-received for unbundled ports includes all available features in all states.  Interest in Switching Usage and Common Transport Usage rates in the Port section crently Combined scenarios where Market Rates apply, the Nonrecurring charges are listed as categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt. Greensboro-W stees in this section to shall apply to all and Additional NRC UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	Tennesse; uth's regionston Sale i. In the int combination combination columns for UEPLX UEPLX UEPLX UEPLX UEPLX UEPRO UEPRO UEPRO UEPRO UEPRO	24.8 26.47 33.83 10.8 11.8 11.8 11.8 11.8 11.8 11.8 11.8	charlotte-Gas th shall bill th rt network ele SOC. For Cri  90  90	e rates in the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furness and the Crements except furness except fur	TN (Nashvil	ection precedin	mbinations w	hich have a	33.67 33.67	rently Comb  7.88  7.88	OC: URECU)	
Market Rate: These scenarion The Se scenarion 1. Unbundle 2. Unbundle The Top 8 M BellSouth cu The Market F End Office a For Not Currapply also ar 2-Wire Voice UNE Port/Lo  UNE Loop R	s shall apply where BellSouth is not required to provide unbundled local switching or surios include: ad port/loop combinations that are Not Currently Combined in all of the BellSouth state ad port/loop combinations that are Currently Combined or Not Currently Combined in Z ISAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (I surrently is developing the billing capability to mechanically bill the recurring and non-rec Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are lis and are categorized accordingly.  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port soutgoing only - re 2-Wire voice unbundled port outgoing only - re	s except one 1 of New Orles	as noted the Top ans); No arket Ra e exhibit First ar	d for Georgia and 8 MSAS in BellSt. Greensboro-W stees in this section t shall apply to all and Additional NRC UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	Tennesse; suth's regionston Sale . In the int combinatic columns fo  UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC	2. n for end use m-Highpoint/C erim, BellSou ons of loop/po or each Port U 24.8 26.47 33.83 10.8 12.47 19.83 14 14 14	charlotte-Gas th shall bill th rt network ele SOC. For Cri  90  90	e rates in the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furrently Combination of the Crements except furness and the Crements except furness except fur	TN (Nashvil	ection precedin	mbinations w	hich have a	33.67 33.67	rently Comb  7.88  7.88	OC: URECU)	

All Features Offered		UEPRX	UEPVF	0	0	0							
											L	L	J
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	+-+	UEPRX	USAC2		41.5	41.5				<del>                                     </del>			$\dashv$
2-Wire Voice Grade Loop / Line Port Combination - Switch with change		UEPRX	USACC		41.5	41.5							
ADDITIONAL NRCs													$\dashv$
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque	+	UEPRX	USAS2		0	0							+
		02.100	COMOL		Ü	Ů							$\exists$
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)													+
UNE Port/Loop Combination Rates													$\forall$
2-Wire VG Loop/Port Combo - Zone	1			24.8									I
2-Wire VG Loop/Port Combo - Zone	2			26.47									
2-Wire VG Loop/Port Combo - Zone	3			33.83						<u> </u>	<del></del>	<b>_</b>	_
UNE Loop Rates											+		+
2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	10.8								<del>                                     </del>	+
2-Wire Voice Grade Loop (SL1) - Zone	2	UEPBX	UEPLX	12.47						1			+
2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	19.83									T
2-Wire Voice Grade Line Port (Bus)													4
2-Wire voice unbundled port without Caller ID - bu	-	UEPBX	UEPBL	14	90	90				33.67	7.88	<u> </u>	+
2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBX	UEPBC	14	90	90				33.67	7.88		
2-Wire voice unbundled port outgoing only - bu	+	UEPBX	UEPBO	14	90	90				33.67	7.88		+
LOCAL NUMBER PORTABILITY													1
Local Number Portability (1 per port	<del>                                     </del>	UEPBX	LNPCX	0.35							<del>                                     </del>		+
FEATURES													I
NONRECURRING CHARGES - CURRENTLY COMBINED													+
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	<del>                                     </del>	UEPBX	USAC2		41.5	41.5							+
2 Wile Voice Grade 200p / Enter of Combination Owner as		OLI DX	00/102		41.5	41.0				1			+
2-Wire Voice Grade Loop / Line Port Combination - Switch with change		UEPBX	USACC		41.5	41.5							_
ADDITIONAL NRCs													+
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPBX	USAS2		0	0							4
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)													
UNE Port/Loop Combination Rates	<del>                                     </del>											<u> </u>	4
2-Wire VG Loop/Port Combo - Zone	1			24.8							<del>                                     </del>	-	+
2-Wire VG Loop/Port Combo - Zone	2			26.47						1			+
2-Wire VG Loop/Port Combo - Zone	3			33.83									T
													1
UNE Loop Rates	<b>_</b>									<u> </u>	<del></del>	<b>_</b>	4
2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	1 2	UEPRG UEPRG	UEPLX	10.8 12.47									+
2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRG	UEPLX	19.83								<del> </del>	+
2 Wile Voice Grade Loop (GLT) Zone		OLITIO	OLI LX	10.00									$^{\dagger}$
2-Wire Voice Grade Line Port Rates (RES - PBX)													1
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	14	90	90				33.67	7.88		
													4
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port	<del>                                     </del>	UEPRG	LNPCP	3.15							<del>                                     </del>		+
Local Number Portability (1 per port		OLFING	LINEOF	3.13									+
FEATURES													
												<u> </u>	Ţ
NONRECURRING CHARGES - CURRENTLY COMBINED		LIEBBO	11010		41.5	4:-				-			4
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-	+	UEPRG	USAC2		41.5	41.5				-	<del></del>		+
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change		UEPRG	USACC		41.5	41.5							
ADDITIONAL NRCs	++						-	-		<del> </del>	<del></del>	-	+
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-	+				<b> </b>						<b>—</b>	<b>-</b>	+
Nonrecurring					0	0							
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Gro	$+ \mp$		1		14.64	14.64	<b>—</b>	<b>-</b>	<b>-</b>	19.99	19.99	19.99	4
	+	1	-		<u> </u>								$\forall$
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)													
													+
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone	1			24.8									#
	1 2			24.8 26.47									#

UNE Loop Rates													
2-Wire Voice Grade Loop (SL1) - Zone	1	UEPPX	UEPLX	10.8									
2-Wire Voice Grade Loop (SL1) - Zone	2	UEPPX	UEPLX	12.47									
2-Wire Voice Grade Loop (SL1) - Zone	3		UEPLX	19.83									
			, _, _, _,										
2-Wire Voice Grade Line Port Rates (BUS - PBX)	-												
2 Time Void Citate Line ( Citate ( Citate ( Citate ) Citate ( Cita													
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bi		UEPPX	UEPPC	14	90	90				33.67	7.88		
Line Side Oribunded Combination 2-way PBA Trunk Port - Bi		UEFFA	UEFFC	14	90	90				33.07	7.00		+
Live City Hallow Hall Convert BRY To all Box 1		UEPPX	UEPPO		90	90				00.07	7.88		
Line Side Unbundled Outward PBX Trunk Port - Bu				14						33.67			4
Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX	UEPP1	14	90	90				33.67	7.88		
2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	14	90	90				33.67	7.88		
2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	14	90	90				33.67	7.88		
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	14	90	90				33.67	7.88		
2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	14	90	90				33.67	7.88		
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	14	90	90				33.67	7.88		
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc		UEPPX	UEPXE	14	90	90				33.67	7.88		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		OLITA	OLIAL	17	30	30				00.07	7.00		+
Port		UEPPX	UEPXL	14	90	90				33.67	7.88		
Polt		UEFFA	UEFAL	14	90	90				33.07	7.00		-
O Miss Vision Habitania do Mart DDV 1111 1111 1111 1111 1111 1111 1111	1	UEDE:	HEDVA	4.4	00	60				22.27	7.00		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX	UEPXM	14	90	90	1		1	33.67	7.88		+-
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling			1										
Port		UEPPX	UEPXO	14	90	90	1		ļ	33.67	7.88		1
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX	UEPXS	14	90	90				33.67	7.88		1
							1		1				1
LOCAL NUMBER PORTABILITY			T			L	L <sup></sup>	 <u> </u>	L <sup></sup>			L <sup></sup>	Ш¯
Local Number Portability (1 per port		UEPPX	LNPCP	3.15								l	
FEATURES													
NONRECURRING CHARGES - CURRENTLY COMBINED													
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-		UEPPX	USAC2		41.5	41.5							
z-wire voice Grade Loop/ Line Fort Combination - Switch-As-		UEFFA	USACZ		41.5	41.5							+
O Wire Voice Conde Loop / Line Port Combination Coulteb with Char		HEDDY	LICACO		44.5	44.5							
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Chan		UEPPX	USACC		41.5	41.5							_
ADDITIONAL NRCs													_
2-Wire Voice Grade Loop/ Line Port Combination - Subseque		UEPPX	USAS2		0	0							
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-													
Nonrecurring					0	0							
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64				19.99	19.99	19.99	1
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT													
UNE Port/Loop Combination Rates													
UNE Port/Loop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone 1				24.8									
2-Wire VG Coin Port/Loop Combo – Zone 1				24.8 26.47									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2				26.47									
2-Wire VG Coin Port/Loop Combo – Zone 1													
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3				26.47									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates		LIEDCO	I I E DI Y	26.47 33.83									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	26.47 33.83									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	26.47 33.83 10.8 12.47									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone			UEPLX	26.47 33.83									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	26.47 33.83 10.8 12.47									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO UEPCO	UEPLX UEPLX	26.47 33.83 10.8 12.47 19.83									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX UEPLX	26.47 33.83 10.8 12.47	90	90				33.67	7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO UEPCO	UEPLX UEPLX UEPGC	26.47 33.83 10.8 12.47 19.83									
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Line Port Rates (Coin)  2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA)		UEPCO UEPCO UEPCO	UEPLX UEPLX UEPGC UEPGC	26.47 33.83 10.8 12.47 19.83	90	90				33.67	7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA)		UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPGC UEPGC UEP2G UEPGA	26.47 33.83 10.8 12.47 19.83 14	90 90	90 90				33.67 33.67	7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Line Port Rates (Coin) 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA)		UEPCO UEPCO UEPCO	UEPLX UEPLX UEPGC UEPGC UEP2G UEPGA	26.47 33.83 10.8 12.47 19.83	90	90				33.67	7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Line Port Rates (Coin)  2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA)		UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPGC UEPGC UEP2G UEPGA	26.47 33.83 10.8 12.47 19.83 14	90 90	90 90				33.67 33.67	7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 901/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+,and Local (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+,and Local (GA)		UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPGC UEPGC UEP2G UEPGA	26.47 33.83 10.8 12.47 19.83 14	90 90	90 90				33.67 33.67	7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 901/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+,and Local (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+,and Local (GA)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPGC UEP2G UEPGA UEPGB UEPCH	26.47 33.83 10.8 12.47 19.83 14 14 14 14	90 90 90	90 90 90				33.67 33.67 33.67	7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+,and Local (GA) 2-Wire Coin Outward with Operator Screening and 011Blocking (GA, KY, MS)		UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPGC UEP2G UEPGA UEPGB UEPCH	26.47 33.83 10.8 12.47 19.83 14 14 14 14	90 90 90	90 90 90				33.67 33.67 33.67	7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and 011Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (CA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (CA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (CA)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPGC UEPGG UEPGA UEPGB UEPGB UEPCH UEPRJ	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14	90 90 90 90	90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+,and Local (GA) 2-Wire Coin Outward with Operator Screening and 011Blocking (GA, KY, MS)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPGC UEPGG UEPGA UEPGB UEPGB UEPCH UEPRJ	26.47 33.83 10.8 12.47 19.83 14 14 14 14	90 90 90	90 90 90				33.67 33.67 33.67	7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and 011Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (CA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (CA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (CA)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPGC UEPGG UEPGA UEPGB UEPGB UEPCH UEPRJ	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14	90 90 90 90	90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and 911Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPGC UEPGG UEPGA UEPGB UEPGB UEPCH UEPRJ	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14	90 90 90 90	90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  LOCAL NUMBER PORTABILITY		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPGC UEPGG UEPGA UEPGB UEPGB UEPCH UEPCH UEPCJ	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14 14 14	90 90 90 90	90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and 911Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPGC UEPGG UEPGA UEPGB UEPGB UEPCH UEPCH UEPCJ	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14	90 90 90 90	90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  LOCAL NUMBER PORTABILITY Local Number Portability (1 per port		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPGC UEPGG UEPGA UEPGB UEPGB UEPCH UEPCH UEPCJ	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14 14 14	90 90 90 90	90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 011Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  LOCAL NUMBER PORTABILITY Local Number Portability (1 per port		UEPCO X UEPLX UEPCC UEPCG UEPCG UEPCG UEPCG UEPCG UEPCG UEPCH UEPCD UEPCC	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14 14 14	90 90 90 90 90 90	90 90 90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88			
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  LOCAL NUMBER PORTABILITY Local Number Portability (1 per port		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPCC UEPCG UEPCG UEPCG UEPCG UEPCG UEPCG UEPCH UEPCD UEPCC	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14 14 14	90 90 90 90	90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88		
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  LOCAL NUMBER PORTABILITY Local Number Portability (1 per port		UEPCO X UEPGC UEPGG UEPCG UEPCA UEPCB UEPCH UEPCH UEPCL UEPCA UEPCA UEPCA UEPCA UEPCA UEPCA UEPCA UEPCA	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14 14 14	90 90 90 90 90 90 90	90 90 90 90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88			
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Orade Line Port Rates (Coin) 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 900/976 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and 901Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  LOCAL NUMBER PORTABILITY Local Number Portability (1 per porl  NONRECURRING CHARGES - CURRENTLY COMBINED 2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-I		UEPCO X UEPGC UEPGG UEPCG UEPCA UEPCB UEPCH UEPCH UEPCL UEPCA UEPCA UEPCA UEPCA UEPCA UEPCA UEPCA UEPCA	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14 14 14	90 90 90 90 90 90	90 90 90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88			
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way with Operator Screening (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (GA) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (GA) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (GA) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)  LOCAL NUMBER PORTABILITY Local Number Portability (1 per port		UEPCO X UEPCS UEPGG UEPGA UEPGA UEPGB UEPCH UEPRJ UEPCQ UEPCQ UEPCQ UEPCQ UEPCQ UEPCQ UEPCQ UEPCQ UEPCQ UEPCQ UEPCQ	26.47 33.83 10.8 12.47 19.83 14 14 14 14 14 14 14	90 90 90 90 90 90 90	90 90 90 90 90 90 90				33.67 33.67 33.67 33.67 33.67	7.88 7.88 7.88 7.88 7.88			

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zone	BCS	USOC		R/	ATES (\$)					OSS R	ATES (\$)		
							Nonrec	urring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic-Di Add'I
							Homeo	L					sconnect			
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		shown in the sections for stand-alone loops or loops as part of a combination refers to nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm	Geographically	Deaveraged UNE Z	ones. To	view Geographio	cally Deaverage	d UNE Zone	Designation	s by Central (	Office, refer	to Internet \	Website:			
MRIINDI EI	D EXCHANG	E ACCESS LOOP														
NDUNDLE	DEXCHANG	E ACCESS LOUP														
	2-WIRE ANA	ALOG VOICE GRADE LOOP														
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	1	UEANL	UEAL2	13.54	70.44	44.05	46.93	10.4		19.99				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	2	UEANL	UEAL2	19.73	70.44	44.05	46.93	10.4		19.99				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	3	UEANL	UEAL2	28.27	70.44	44.05	46.93	10.4		19.99				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone	1	UEPSR, UEPSB	UEALS	13.54	70.44	44.05	46.93	10.4		19.99				
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zoni	2	UEPSR, UEPSB	UEALS	19.73	70.44	44.05	46.93	10.4		19.99				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zoni	3	UEPSR, UEPSB	UEALS	28.27	70.44	44.05	46.93	10.4		19.99				
		Engineering Information Document (E		UEANL			28.76	28.76								
		Manual Order Coordination for UVL-SL1s (per loop		UEANL	UEAMC		16.31	16.31								
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR		UEANL	OCOSL		36.18	36.18								
		Order Coordination for Specified Conversion Time for Ove-SET (per ESIX		OLANE	OCOSL		30.10	30.10								
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling Zone 1	1	UEA	UEAL2	17.27	236.75	177.1				19.99				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling Zone 2	2	UEA	UEAL2	32.32	236.75	177.1				19.99				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling Zone 3	3	UEA	UEAL2	55.78	236.75	177.1				19.99				
		Order Coordination for Specified Conversion Time (per LS)		UEA	OCOSL		36.18									
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo	ne													
		Service Level 2 w/Reverse Battery Signaling - Zo	ne 1	UEA	UEAR2	17.27	236.75	177.1				19.99				
		2	2	UEA	UEAR2	32.32	236.75	177.1				19.99				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo 3	ne 3	UEA	UEAR2	55.78	236.75	177.1				19.99				
	4-WIRE ANA	Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		36.18									
		4-Wire Analog Voice Grade Loop - Zone	1	UEA	UEAL4	20.92	457.14	348.83				19.99				
		4-Wire Analog Voice Grade Loop - Zone	2	UEA	UEAL4	39.14	457.14	348.83				19.99				
		4-Wire Analog Voice Grade Loop - Zone	3	UEA	UEAL4	67.57	457.14	348.83				19.99				
		•														
		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		36.18									
	2-WIRE ISD	N DIGITAL GRADE LOOP														
		2-Wire ISDN Digital Grade Loop - Zone	1	UDN	U1L2X	23.66	541.28	431.61				19.99				
		2-Wire ISDN Digital Grade Loop - Zone	2	UDN	U1L2X	44.28	541.28	431.61				19.99				
		2-Wire ISDN Digital Grade Loop - Zone	3	UDN	U1L2X	76.42	541.28	431.61				19.99				
		Order Coordination For Specified Conversion Time (per LS		UDN	OCOSL		36.18									
	0 MIDE !! :															
	∠-WIRE Uni	versal Digital Channel (UDC) COMPATIBLE LOOP		LIDO	HDOOM	05.70	222 47	450.51	405.40	20.40		40.00	1			<u> </u>
	1	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	1	UDC	UDC2X	25.73	233.47	158.51	105.49	20.48		19.99	1			<del>                                     </del>
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	3	UDC UDC	UDC2X UDC2X	34.83 45.56	233.47 233.47	158.51 158.51	105.49 105.49	20.48 20.48		19.99 19.99				<del>                                     </del>
		2-Wile Oniversal Digital Chariner (ODC) Compatible Loop - Zone	3	ODC	UDUZA	45.50	255.47	130.31	103.43	20.40		13.33				
	2-WIRE AS	YMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP														
		2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOO 2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation														
		Zone 1	1	UAL	UAL2X	8.79	713.5	609.44				19.99				
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 2	2	UAL	UAL2X	16.46	713.5	609.44				19.99				
		Zone 2  Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 3	3	UAL	UAL2X	28.4	713.5	609.44				19.99				
		Order Coordination for Specified Conversion Time (per LS		UAL	OCOSL		36.18									

	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1	1	UAL	UAL2W	8.79	205.25	129.42	100.89	15.88	19.99			
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -	2	UAL	UAL2W	16.46	205.25	129.42	100.89	15.88	19.99			
	Zone 2  2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -	2	UAL	UALZW	10.40	205.25	129.42	100.89	15.88	19.99		-+-	-
	Zone 3	3	UAL	UAL2W	28.4	205.25	129.42	100.89	15.88	19.99			
	Order Coordination for Specified Conversion Time (per LS		UAL	OCOSL		36.18							
2-WIRE HIG	GH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP											-+-	-
	2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOO												
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 1	1	UHL	UHL2X	6.29	713.5	609.44			19.99			
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -		OTIL	UTILZX	0.23		003.44			13.33			_
	Zone 2  2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation	2	UHL	UHL2X	11.78	713.5	609.44			19.99			_
	Zone 3	3	UHL	UHL2X	20.33	713.5	609.44			19.99			
	O de O de l'artire ( a O de l'artire d'artire ( a o O de l'artire d'artire			00001		00.40							
+	Order Coordination for Specified Conversion Time (per LSI  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation		UHL	OCOSL		36.18							-
	Zone 1	1	UHL	UHL2W	6.29	222.58	146.75	100.89	15.88	19.99			
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2	2	UHL	UHL2W	11.78	222.58	146.75	100.89	15.88	19.99			
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation												
	Zone 3	3	UHL	UHL2W	20.33	222.58	146.75	100.89	15.88	19.99			+
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		36.18							
4-WIRE HIC	GH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP												_
- WIIVE THE	4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -												-
	Zone 1	1	UHL	UHL4X	7.68	748.93	646.17			19.99		$\longrightarrow$	_
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 2	2	UHL	UHL4X	14.38	748.93	646.17			19.99			
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -												
+	Zone 3	3	UHL	UHL4X	24.82	748.93	646.17			19.99			-
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		36.18							
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1	1	UHL	UHL4W	7.68	279.79	203.96	109.64	20.64	19.99			
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation												
+	Zone 2  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation	2	UHL	UHL4W	14.38	279.79	203.96	109.64	20.64	19.99			-
	Zone 3	3	UHL	UHL4W	24.82	279.79	203.96	109.64	20.64	19.99			
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		36.18							
			0112	00002		00.10							
4-WIRE DS	81 DIGITAL LOOP  4-Wire DS1 Digital Loop - Zone	1	USL	USLXX	50.26	849.8	523.27			19.99		-	_
-	4-Wire DS1 Digital Loop - Zone	2	USL	USLXX	94.06	849.8	523.27			19.99			_
	4-Wire DS1 Digital Loop - Zone :	3	USL	USLXX	162.34	849.8	523.27			19.99			
	Order Coordination for Specified Conversion Time (per LS		USL	OCOSL		36.18							
													$\Box$
4-WIRE 19.	1.2, 56 OR 64 KBPS DIGITAL GRADE LOOP  4 Wire Unbundled Digital 19.2 Kbps	1	UDL	UDL19	35.92	250.99	176.03	116.85	27.85	19.99		-+-	-
+	4 Wire Unbundled Digital 19.2 Kbps	2	UDL	UDL19	40.32	250.99	176.03	116.85	27.85	19.99			-
-	4 Wire Unbundled Digital 19.2 Kbps	3	UDL	UDL19	37.9	250.99	176.03	116.85	27.85	19.99			
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	1	UDL	UDL56	35.92	250.99	176.03	116.85	27.85	19.99			
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	2	UDL	UDL56	40.32	250.99	176.03	116.85	27.85	19.99			
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	3	UDL	UDL56	37.9	250.99	176.03	116.85	27.85	19.99			
+	Order Coordination for Specified Conversion Time (per LS	1	UDL	OCOSL	25.22	36.18	470.00	440.05	07.05	10.00			-+
+	4 Wire Unbundled Digital Loop 64 Kbps - Zone	2	UDL	UDL64	35.92	250.99	176.03 176.03	116.85 116.85	27.85 27.85	19.99	+-	-+-	+
+	4 Wire Unbundled Digital Loop 64 Kbps - Zone 4 Wire Unbundled Digital Loop 64 Kbps - Zone	3	UDL UDL	UDL64 UDL64	40.32 37.9	250.99 250.99	176.03	116.85 116.85	27.85 27.85	19.99 19.99		-	
+	4 Wife Oribunuled Digital Loop 64 Kbps - Zone	3	UDL	UDL64	31.9	250.99	170.03	110.00	21.00	19.99		-	-+
1	Order Coordination for Specified Conversion Time (per LS		UDL	OCOSL		36.18							
													-
2-WIRE Uni	houndled COPPER LOOP												
2-WIRE Uni	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility	1	IICI	LICI DB	14 04	283 77	164.04	120 6	22.45	19 99			
2-WIRE Uni	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility	1	UCL	UCLPB	14.94	283.77	164.04	120.6	22.45	19.99	_		-
2-WIRE Uni	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1	1 2	UCL	UCLPB UCLPB	14.94 15.15	283.77 283.77	164.04 164.04	120.6 120.6	22.45	19.99			#

	1	,	1									
Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		16.31	16.31					
2-Wire Unbundled Copper Loop/Short without manual serv	rice inquiry and facility											
reservation - Zone ' 2-Wire Unbundled Copper Loop/Short without manual serv	des lessies and facility	1	UCL	UCLPW	14.94	203.39	127.56	100.89	15.88	19.99		
reservation - Zone 2	rice inquiry and facility	2	UCL	UCLPW	15.15	203.39	127.56	100.89	15.88	19.99		
2-Wire Unbundled Copper Loop/Short without manual serv	rice inquiry and facility	3	1101	LIOL DW	45.70	200.00	407.50	400.00	45.00	40.00		
reservation - Zone (		3	UCL	UCLPW	15.73	203.39	127.56	100.89	15.88	19.99		
Order Coordination for Unbundled Copper Loops (per loop)	,		UCL	UCLMC		16.31	16.31					
2-Wire Unbundled Copper Loop/Long - includes manual sn reservation - Zone '	vc. inquiry and facility	1	UCL	UCL2L	36.19	270.38	150.65	120.6	22.45	19.99		
2-Wire Unbundled Copper Loop/Long - includes manual sv	c. inquiry and facility											
reservation - Zone 2 2-Wire Unbundled Copper Loop/Long - includes manual sv	o inquiry and facility	2	UCL	UCL2L	49.31	270.38	150.65	120.6	22.45	19.99		
reservation - Zone (	rc. Inquiry and racility	3	UCL	UCL2L	80.78	270.38	150.65	120.6	22.45	19.99		
Order Coordination for Unbundled Copper Loops (per loop)	)		UCL	UCLMC		16.31	16.31					
2-Wire Unbundled Copper Loop/Long - without manual ser	vice inquiry and facility		UCL	UCLIVIC		10.31	10.31					
reservation - Zone '		1	UCL	UCL2W	36.19	190	114.17	100.89	15.88	19.99		
2-Wire Unbundled Copper Loop/Long - without manual ser reservation - Zone 2	vice inquiry and facility	2	UCL	UCL2W	49.31	190	114.17	100.89	15.88	19.99		
2-Wire Unbundled Copper Loop/Long - without manual ser	vice inquiry and facility											
reservation - Zone ( Order Coordination for Unbundled Copper Loops (per loc		3	UCL UCL	UCL2W UCLMC	80.78	190 16.31	114.17 16.31	100.89	15.88	19.99		
Crasi coordination for characted copper coope (per for			002	CCLING		10.01	10.01					
2-Wire Unbundled Copper Loop - Non-Designed Zone		I 1	UEQ	UEQ2X	11.01	44.69	22.4	25.65	7.06	19.99		
2 Wire Unbundled Copper Loop - Non-Designed - Zone		1 2	UEQ	UEQ2X	12.67	44.69	22.4	25.65	7.06	19.99		
2 Wire Unbundled Copper Loop - Non-Designed - Zone		I 3	UEQ	UEQ2X	20.22	44.69	22.4	25.65	7.06	19.99		
Order Coordination 2 Wire Unbundled Copper Loop - Non-	Designed (per loc		UEQ UEQ	USBMC		16.31 28.76	16.31 28.76					
Engineering Information Documer  Loop Testing - Basic 1st Half Hou			UEQ	URET1		78.92	78.92					
Loop Testing Basic Additional Half Hou			UEQ	URETA		23.33	23.33					
4-WIRE COPPER LOOP												
4-Wire Copper Loop/Short - including manual service inqui	iry and facility reservation -											
Zone 1		1	UCL	UCL4S	25.26	332.2	212.46	130.27	27.51	19.99		
4-Wire Copper Loop/Short - including manual service inqui Zone 2	iry and facility reservation -	2	UCL	UCL4S	23	332.2	212.46	130.27	27.51	19.99		
4-Wire Copper Loop/Short - including manual service inqui	iry and facility reservation -											
Zone 3		3	UCL	UCL4S	19.08	332.2	212.46	130.27	27.51	19.99		
Order Coordination for Unbundled Copper Loops (per loc 4-Wire Copper Loop/Short - without manual service inquiry	, and facility reconstation		UCL	UCLMC		16.31	16.31					
Zone 1	and facility reservation -	1	UCL	UCL4W	25.26	251.82	175.99	109.64	20.64	19.99		
4-Wire Copper Loop/Short - without manual service inquiry Zone 2	and facility reservation -	2	UCL	UCL4W	23	251.82	175.99	109.64	20.64	19.99		
4-Wire Copper Loop/Short - without manual service inquiry	and facility reservation -	2	UCL	UCL4VV	23	251.82	175.99	109.64	20.04	19.99		
Zone 3	,	3	UCL	UCL4W	19.08	251.82	175.99	109.64	20.64	19.99		
Order Coordination for Unbundled Copper Loops (per loc	1 1 14 111		UCL	UCLMC		16.31	16.31					
4-Wire Unbundled Copper Loop/Long - includes manual sv reservation - Zone '	c. inquiry and facility	1	UCL	UCL4L	61.02	318.81	199.07	130.27	27.51	19.99		
4-Wire Unbundled Copper Loop/Long - includes manual sv	c. inquiry and facility	_										
reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - includes manual sv	c inquiry and facility	2	UCL	UCL4L	55.74	318.81	199.07	130.27	27.51	19.99		
reservation - Zone (	o. Inquiry and rability	3	UCL	UCL4L	88.97	318.81	199.07	130.27	27.51	19.99		
Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		16.31	16.31					
4-Wire Unbundled Copper Loop/Long - without manual svc	c. inquiry and facility	1	LICI	1101.40	C4 00	238.42	400.0	400.04	20.64	40.00		
reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - without manual svc	. inquiry and facility		UCL	UCL4O	61.02	238.42	162.6	109.64	20.64	19.99		
reservation - Zone 2		2	UCL	UCL4O	55.74	238.42	162.6	109.64	20.64	19.99		
4-Wire Unbundled Copper Loop/Long - without manual svc reservation - Zone (	c. inquiry and facility	3	UCL	UCL4O	88.97	238.42	162.6	109.64	20.64	19.99		
Order Coordination for Unbundled Copper Loops (per loc		3	UCL	UCLMC	00.97	16.31	16.31	109.64	20.04	19.99		
OOP MODIFICATION				1								
Unbundled Loop Modification, Removal of Load Coils - 2 W	Vire pair less than or equal	to	UAL, UHL, UCL,		-							
18k ft Unbundled Loop Modification, Removal of Load Coils - 2 w	viro grooter than 19		UEQ, ULS UCL, ULS	ULM2L ULM2G		65.2 341.64	65.2				_	
Unbundled Loop Modification, Removal of Load Coils - 2 W Unbundled Loop Modification Removal of Load Coils - 4 W		ĸ	UCL, ULS	ULM2G		341.64	341.64					
Unbundied Loop Modification Removal of Load Colls - 4 W	riie iess tilali vi equal to 18	1.	UHL, UCL	ULM4L		65.2	65.2					
ft												
ft									1			
ft Unbundled Loop Modification Removal of Load Coils - 4 W	Vire pair greater than 18		UCL UAL UHL UCL	ULM4G		341.64	341.64					
ft Unbundled Loop Modification Removal of Load Coils - 4 W Unbundled Loop Modification Removal of Bridged Tap Rei			UCL UAL, UHL, UCL, UEQ, UEF, ULS	ULM4G ULMBT		341.64 65.24						

3															
Sub-Loop Distrib															
	-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L	ı		UEANL	USBSA		600.03	600.03				19.99			
Sub-l	-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L			UEANL	USBSB		45.28	45.28				19.99			
0.1.1	1				LIODOO		070.00	070.00				40.00			
Sub-I	-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I			UEANL	USBSC		379.89	379.89				19.99			
Cb. I	Lana Bar Building Equipment Barra Barra Sa Bair Barral Cat I			UEANL	LICDOD		111.55	111.55				19.99			
	-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I		1	UEANL	USBSD	9.03	131.64	61.93	90.83	13.44		19.99			-
	-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone -Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone			UEANL	USBN2 USBN2	12.25	131.64	61.93		13.44		19.99			_
	-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone -Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		3	UEANL	USBN2	16.71	131.64	61.93	90.83	13.44		19.99			
	er Coordination for Unbundled Sub-Loops, per sub-loop pair	- '	3	UEANL	USBMC	10.71	36.18	36.18	90.83	13.44		19.99			
	-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		1	UEANL	USBN4	10.18	158.12	88.41	99.1	18.08		19.99			-
	-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		2	UEANL	USBN4	9.44	158.12	88.41	99.1	18.08		19.99			_
Sub-l	-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		3	UEANL	USBN4	13.38	158.12	88.41	99.1	18.08		19.99			
Orde	er Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC	10.00	36.18	36.18	33.1	10.00		10.00			
	-Loop 2-Wire Intrabuilding Network Cable (INC	-		UEANL	USBR2	3.23	106.06	36.35	90.83	13.44		19.99			
	er Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC	0.20	36.18	36.18	30.03	10.44		10.00			
	-Loop 4-Wire Intrabuilding Network Cable (INC	1		UEANL	USBR4	6.29	118.54	48.84	99.1	18.08		19.99			
	er Coordination for Unbundled Sub-Loops, per sub-loop pair	·		UEANL	USBMC	0.20	36.18	36.18	00.1	10.00		10.00			
	ire Copper Unbundled Sub-Loop Distribution - Zone		1	UEF	UCS2X	8.01	131.64	61.93	90.83	13.44		19.99			7
	ire Copper Unbundled Sub-Loop Distribution - Zone	i		UEF	UCS2X	9.18	131.64	61.93	90.83	13.44		19.99			7
	ire Copper Unbundled Sub-Loop Distribution - Zone	i		UEF	UCS2X	11.02	131.64	61.93	90.83	13.44		19.99			7
	er Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC	<del></del>	36.18	36.18							7
	ire Copper Unbundled Sub-Loop Distribution - Zone	1		UEF	UCS4X	10.65	158.12	88.41	99.1	18.08		19.99			7
	ire Copper Unbundled Sub-Loop Distribution - Zone	i		UEF	UCS4X	9.71	158.12	88.41	99.1	18.08		19.99			7
	ire Copper Unbundled Sub-Loop Distribution - Zone	i		UEF	UCS4X	8.45	158.12	88.41	99.1	18.08		19.99			
Orde	er Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		36.18	36.18							
															T
Sub-Loop Feeder	r														T
				UEA.											
				UDN,UCL,UDL,U	D										
USL-	-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-			С	USBFW		600.03								
				UEA,											
				UDN,UCL,UDL,U	D										
USL	Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u			С	USBFX		45.28	45.28							
	Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		527.98	11.32							
	undled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zon		1	UEA	USBFA	10.36	184.97	111.91	108.76	26.76		19.99			
	undled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone		2	UEA	USBFA	13.62	184.97	111.91	108.76	26.76		19.99			
Unbu	undled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zon		3	UEA	USBFA	19.69	184.97	111.91	108.76	26.76		19.99			
	er Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		36.18								
	undlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni		1	UEA	USBFB	10.36	184.97	111.91	108.76	26.76		19.99			
	undled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni		2	UEA	USBFB	13.62	184.97	111.91	108.76	26.76		19.99			
Unbu	undled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zoni		3	UEA	USBFB	19.69	184.97	111.91	108.76	26.76		19.99			
	er Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		36.18								
Unbu	undled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon		1	UEA	USBFC	10.36	184.97	111.91	108.76	26.76		19.99			_
	undled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon		2	UEA	USBFC	13.62	184.97	111.91	108.76	26.76		19.99			
Unbu	undled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zo	ne	_												
3			3	UEA	USBFC	19.69	184.97	111.91	108.76	26.76		19.99			
					0000		00.40								
	er Coordination For Specified Conversion Time, per LS			UEA	OCOSL	20.00	36.18	100.0	400.04	22.24		10.00			4
	undled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone		1	UEA	USBFD	30.69	213.56	138.6	122.64	33.64		19.99		1	4
Unbu	undled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zong		2	UEA	USBFD	36.12	213.56	138.6	122.64	33.64		19.99		1	4
Unbu	undled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zoni	-	3	UEA	USBFD	22.9	213.56	138.6	122.64	33.64		19.99			+
0	or Coordination For Specified Conversion Time - Beat 6			1154	OCOSL		26.40								
	er Coordination For Specified Conversion Time, Per LS undled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	-	1	UEA UEA	USBFE	30.69	36.18 213.56	138.6	122.64	33.64		19.99			+
	undled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni undled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni		2	UEA	USBFE	36.12	213.56	138.6	122.64	33.64		19.99		1	$\dashv$
	undled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni undled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni		3	UEA	USBFE	36.12 22.9	213.56	138.6	122.64	33.64		19.99		1	$\dashv$
Unbu	undied Sub-Loop i seder Loop, 4 Wile Loop-Start, Voice Gidde - Zoni		٥	UEA	USBFE	22.3	213.30	130.0	122.04	33.04		10.00		1	$\dashv$
Ordo	er Coordination For Specified Conversion Time, Per LS			UEA	OCOSL		36.18							1	
	undled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone		1	UDN	USBFF	17.75	211.3	136.34	111.02	26.01	1	19.99		1	$\dashv$
	undled Sub-Loop Feeder Loop, 2 Wife ISDN BRI - Zone		2	UDN	USBFF	23.67	211.3	136.34	111.02	26.01		19.99			$\dashv$
	undled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone		3	UDN	USBFF	29.9	211.3	136.34	111.02	26.01		19.99		1	+
01100	andida 545 250p ( 5646) 200p, 2 11/10 105/1 5/11 2016			0014	OOD!!	20.0	211.0	100.04	111.02	20.01					7
Orde	er Coordination For Specified Conversion Time, Per LS			UDN	OCOSL		36.18							1	
	undled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		1	UDC	USBFS	17.75	211.3	136.34	111.02	26.01	1 1	19.99	1		$\dashv$
	undled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible		2	UDC	USBFS	23.67	211.3	136.34	111.02	26.01		19.99			7
	undled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		3	UDC	USBFS	29.9	211.3	136.34	111.02	26.01		19.99	1		$\dashv$
	undled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		1	USL	USBFG	75.1	202.14	127.18	122.64	33.64		19.99			7
	undled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		2	USL	USBFG	104.53	202.14	127.18	122.64	33.64		19.99			7
	undled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		3	USL	USBFG	152.36	202.14	127.18	122.64	33.64		19.99			
31.00															T
Orde	er Coordination For Specified Conversion Time, Per LS			USL	OCOSL		36.18								
	undled Sub-Loop Feeder, 2-Wire Copper Loop - Zone		1	UCL	USBFH	8.29	167.62	92.66	106.42	21.41		19.99			T
	undled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		2	UCL	USBFH	7.3	167.62	92.66	106.42	21.41		19.99			T
l loubu					USBFH	6.03						19.99			

Attachment 2 Exhibit C

		1 1		1 1		1		1							
	Order Coordination For Specified Conversion Time par LS		UCL	OCOSL		36.18							ļ	l 1	, !
	Order Coordination For Specified Conversion Time, per LS	1		USBFJ	16.55	202.05	127.09	115.43	26.43		19.99			$\vdash$	
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	2	UCL	USBFJ	15.35	202.05	127.09	115.43	26.43		19.99				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	3	UCL	USBFJ	12.52	202.05	127.09	115.43	26.43		19.99				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	3	UCL	USBFJ	12.02	202.00	127.09	115.45	20.43		19.99				
	Order Consideration For Consideral Consumption Times and I S		UCL	OCOSL		36.18							ļ	,	, '
	Order Coordination For Specified Conversion Time, per LS				07.00		407.40	100.01	00.04		10.00			$\vdash$	لــــــا
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	1	UDL	USBFN	27.38	202.14	127.18	122.64	33.64		19.99			$\vdash$	لــــــا
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	2	UDL	USBFN	33.41	202.14	127.18 127.18	122.64	33.64 33.64		19.99 19.99			$\vdash$	لــــــا
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	3	UDL	USBFN	24.47	202.14		122.64							
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	1	UDL	USBFO	27.38	202.14	127.18	122.64	33.64		19.99				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	2	UDL	USBFO	33.41	202.14	127.18	122.64	33.64		19.99				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	3	UDL	USBFO	24.47	202.14	127.18	122.64	33.64		19.99			<b>├</b>	
													ļ	,	, '
	Order Coordination For Specified Time Conversion, per LS		UDL	OCOSL		36.18								<b>├</b>	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1	UDL	USBFP	27.38	202.14	127.18	122.64	33.64		19.99				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	2	UDL	USBFP	33.41	202.14	127.18	122.64	33.64		19.99			<b>├</b>	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zon€	3	UDL	USBFP	24.47	202.14	127.18	122.64	33.64		19.99				
													ļ	,	i
	Order Coordination For Specified Conversion Time, per LS		UDL	OCOSL		36.18								<b></b> '	ļ
															<b>.</b>
Unbundled :	Sub-Loop Modification														<b></b>
	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W	T										T		ı 7	i –
	PR		UEF	ULM2X		355.83	12.27	1			19.99		ļ	ı '	 
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W				-								-		
	PR		UEF	ULM4X		355.83	12.27				19.99		l	1	į I
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR														
	unloaded		UEF	ULM4T		560.74	14.3				19.99		l	1	
		+ +						<b>†</b>				1			
Unhundled	Network Terminating Wire (UNTW)					1									
O I DU I I I I I	Unbundled Network Terminating Wire (UNTW) per Pa	+	UENTW	UENPP	0.64	62.83	62.83				19.99				
	one and a section of the following wife (of the party) per tra	+	OLIVIA	OLIVET	0.04	02.00	02.00								
Notwork Inte	erface Device (NID)														
			UENTW	UND12		89.66	57.24				19.99				
	Network Interface Device (NID) - 1-2 line		UENTW												
	Network Interface Device (NID) - 1-6 line			UND16		129.24	99.52 11.78				19.99 19.99				
	Network Interface Device Cross Connect - 2 V		UENTW	UNDC2		11.78									
	Network Interface Device Cross Connect - 4V		UENTW	UNDC4		11.78	11.78				19.99			$\vdash$	لــــــا
INDUNDUED LOOP COL	NOTATO ATION													$\vdash$	لـــــــا
JNBUNDLED LOOP COM															
	Unbundled Loop Concentration - System A (TR00)		ULC	UCT8A	522.17	651.04	651.04				19.99			<b>├</b>	
	Unbundled Loop Concentration - System B (TR00)		ULC	UCT8B	63.59	271.27	271.27				19.99			<b>├</b>	
	Unbundled Loop Concentration - System A (TR30:		ULC	UCT3A	567.21	651.04	651.04				19.99			<b>├</b>	
	Unbundled Loop Concentration - System B (TR30:		ULC	UCT3B	107.16	271.27	271.27				19.99				<u> </u>
													ļ	,	, '
	Unbundled Loop Concentration - DS1 Loop Interface Ca		ULC	UCTCO	6.04	126.61	92.17	33.46	9.37		19.99				<u> </u>
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Car		UDN	ULCC1	9.59	21.08	20.96	10.75	10.68		19.99				<u> </u>
	Unbundled Loop Concentration - UDC Loop Interface (Brite Car		UDC	ULCCU	9.59	21.08	20.96	10.75	10.68		19.99				<u> </u>
	Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop												ļ	,	,
	Interface (POTS Card)		UEA	ULCC2	2.4	21.08	20.96	10.75	10.68		19.99			<b></b> '	ļ
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface												ļ	,	,
	(SPOTS Card)		UEA	ULCCR	14.26	21.08	20.96	10.75	10.68		19.99				
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca		UEA	ULCC4	8.51	21.08	20.96	10.75	10.68		19.99		l	ļ'	1
	Unbundled Loop Concentration - TEST CIRCUIT Car		ULC	UCTTC	41.58	21.08	20.96	10.75	10.68		19.99			<u> </u>	
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa		UDL	ULCC7	12.6	21.08	20.96	10.75	10.68		19.99			ļ	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa		UDL	ULCC5	12.6	21.08	20.96	10.75	10.68		19.99				
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa		UDL	ULCC6	12.6	21.08	20.96	10.75	10.68		19.99			<u> </u>	<b></b>
						1							l	ļ'	
						1							l	ļ'	1
JNBUNDLED SUB-LOOF	CONCENTRATION (OUTSIDE CO)							<u> </u>						'	
JNE OTHER, PROVISION	NING ONLY - NO RATE														
														· ·	
	NID - Dispatch and Service Order for NID installation		UENTW	UNDBX									l	1	
	·													[	
	UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW	UENCE									l		
			UEANL,UEF,UEC			1									
	Unbundled Contract Name, Provisioning Only - No Rate		UENTW	UNECN				1					ļ	ı '	
	The state of the s	- 1	UAL,UCL,UDC,UI			1	1								
			,UDN,UEA,UHL,U	ii l									l	1	
	Unbundled Contact Name, Provisioning Only and rate		C	UNECN	0	0							l	1	
	Unbundled Contact Name, Provisioning Only - no rate	+	U	UNECIN	U	U		-						$\vdash$	لــــــا
		+	HEATIDM HIGH T					-						$\vdash$	
	Habitadlad Cub Laca Fandas 2 Wita Casas Barris 1		UEA,UDN,UCL,U		0	_							l		I
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra	+	С	USBFQ	0	0				<b></b>				, ——'	
						_							l		I
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra		UEA,USL,UCL,U[	DL USBFR	0	0								<b></b> '	ļ!
							1	1		1	1				, ,
													1	l i	
	Unbundled DS1 Loop - Superframe Format Option - no ra		USL	CCOSF	0	0								<sup> </sup>	

								1		1	1		1		
		Unbundled DS1 Loop - Expanded Superframe Format option - no ra		USL	CCOEF	0	0								
HIGH CAP		NDLED LOCAL LOOP													
	NOTE: 4 m	onth minimum billing period													
		High Capacity Unbundled Local Loop - DS3 - Per Mile per mon		UE3	1L5ND	11.53									
		High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor		UE3	UE3PX	379.72	903.34	528.05	238.2	166.62		19.99			
		High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon		UDLSX	1L5ND	11.53									
		High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor		UDLSX	UDLS1	394.76	903.34	528.05	238.2	166.62		19.99			
LOOP MAI	KE-UP														
		Loop Makeup - Preordering Without Reservation, per working or spare facility queried													
		(Manual).		UMK	UMKLW		47.98	47.98							
		Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).		UMK	UMKLP		50.88	50.88							
		Loop MakeupWith or Without Reservation, per working or spare facility queried													
		(Mechanized)		UMK	PSUMK		0.6746	0.6746							
LINE SHAF	RING														
		Line Sharing Splitter, per System 96 Line Capaci		ULS	ULSDA	203.33	377.71	0	357.29	0		0			
		Line Sharing Splitter, per System 24 Line Capaci		ULS	ULSDB	50.83	377.71	0	357.29	0		0			
		Line Sharing Splitte, Per System, 8 Line Capaci		ULS	ULSD8	16.94	377.71	0	357.29	0		0			
		Line Sharing - per Line Activatio		ULS	ULSDC	0.61	37.02	21.2	20.1	9.87		19.99			
		Line Sharing - per Subsequent Activity per Line Rearrangeme		ULS	ULSDS		32.78	16.38				19.99			
	-		_	-	+		1	+		1			1	1	
		Live Olaria al FO/DLFO O and Oliveria along			+-+		1	1					-	1	
		Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lines			LII CDC		57.70	1	44.40						
	-	(16 pair)		ULS	ULSDG		57.72	1	11.43				+	+	
UNDUNC	.ED TRANSPO	DDT.		-	+		+	1					+	+	
OMBONDE	ED IKANSPO	JKI			+		1	1			-		-	+	-
	COMMON	TRANSPORT (Shared)			+		1	1			-		-	+	-
	COMMON	Common Transport - Per Mile, Per MOI			+	0.0000049	1	1	-				+	+	-
						0.0000048									
		Common Transport - Facilities Termination Per MO				0.000426									
	NOTE: INT		no mont	th DC2 and above for	ur montho										
	NOTE. INTE	EKOFFICE CHANNEL - DEDICATED TRANSFORT - HIIIIIIIIIIIII biiiiiig period. below b33 = 0	ne mon	iii, Doo aliu above it	our monus										
	INTEROFFI	ICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE													
	E.to	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per moi		U1TVX	1L5XX	0.0118									
		Interoffice Channel - Dedicated Transport 2- Wire Voice Grade - Facility Termination		OTTVX	TEOXX	0.0110									
		per month		U1TVX	U1TV2	29.51	81.07	54.84	33.36	13.75		19.99			
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per		011177	02	20.01	01.07	0	00.00	10.70		10.00			
		month		U1TVX	1L5XX	0.0118									
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination		OTTVX	TEOXX	0.0110									
		per month		U1TVX	U1TR2	29.51	81.07	54.84	33.36	13.75		19.99			
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month		U1TVX	1L5XX	0.0118									
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination													
		per month		U1TVX	U1TV4	26.22	81.1	54.84	33.36	13.75		19.99			
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per mon		U1TDX	1L5XX	0.0118									
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mor		U1TDX	U1TD5	21.26	81.11	54.84	33.36	13.75		19.99			
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per mon		U1TDX	1L5XX	0.0118									
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mor		U1TDX	U1TD6	21.26	81.11	54.84	33.36	13.75		19.99			
						-									
	INTEROFFI	ICE CHANNEL - DEDICATED TRANSPORT - DS1													
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mor		U1TD1	1L5XX	0.2407									
		Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor		U1TD1	U1TF1	97.38	178.59	163.67	32.59	28.79		19.99			
	,														
	INTEROFFI	ICE CHANNEL - DEDICATED TRANSPORT- DS3				·									
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor		U1TD3	1L5XX	5.1									
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mor		U1TD3	U1TF3	1191.53	557.69	325.62	120	116.54		19.99			
		ICE CHANNEL - DEDICATED TRANSPORT- STS-1													
	INTEROFFI					5.1		1							
	INTEROFFI	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon		U1TS1	1L5XX				120		i e	19.99			1
	INTEROFFI			U1TS1 U1TS1	U1TFS	1165.53	557.69	325.62	120	116.54		13.33			
	INTEROFFI	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon					557.69	325.62	120	116.54		19.99			
	INTEROFFI	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon					557.69	325.62	120	116.54		13.33			
	INTEROFFI	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon					557.69	325.62	120	116.54		13.33			
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor					557.69	325.62	120	116.54		15.55			
	LOCAL CH	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor  ANNEL - DEDICATED TRANSPORT		U1TS1	U1TFS		557.69	325.62	120	116.54		19.99			
	LOCAL CH	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor  IANNEL - DEDICATED TRANSPORT CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month	, DS3 ar	U1TS1	U1TFS	1165.53									
	LOCAL CH	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor  INTERPORT - Pacific Transport - STS-1 - Facility Termination per mor	, DS3 ar	U1TS1  nd above=four montl ULCVX	U1TFS  Ins  ULDV2	1165.53	386.33	66.35	73.04	6.37		19.99			
	LOCAL CH	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor  IANNEL - DEDICATED TRANSPORT CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor	, DS3 ar	nd above=four montl ULCVX ULCVX	U1TFS  uns  ULDV2 ULDR2	1165.53 18.81 18.81	386.33 386.33	66.35 66.35	73.04 73.04	6.37 6.37		19.99			
	LOCAL CH	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor  IANNEL - DEDICATED TRANSPORT  CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 4-Wire Voice Grade Rev Bat per mor Local Channel - Dedicated - 4-Wire Voice Grade Per mon		ultrs1  d above=four montl ULCVX ULCVX UNCVX	U1TFS  DIS  ULDV2  ULDR2  ULDV4	18.81 18.81 20.12	386.33 386.33 387.2	66.35 66.35 67.22	73.04 73.04 73.98	6.37 6.37 7.31		19.99 19.99 19.99			
	LOCAL CH	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor  IANNEL - DEDICATED TRANSPORT CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 4-Wire Voice Grade Rev Bat per mor Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - 5-Wire Voice Grade per mon Local Channel - Dedicated - DS1 per month - Zone	1	U1TS1  above=four montl ULCVX ULCVX ULCVX UNCVX UNCVX ULDD1	ULDV2 ULDV2 ULDV4 ULDF1	18.81 18.81 20.12 44.63	386.33 386.33 387.2 355.06	66.35 66.35 67.22 307.53	73.04 73.04 73.98 44.24	6.37 6.37 7.31 30.42		19.99 19.99 19.99 19.99			
	LOCAL CH	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor  IANNEL - DEDICATED TRANSPORT  CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one month Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 4-Wire Voice Grade Rev Bat per mor Local Channel - Dedicated - 4-Wire Voice Grade Per mon	1 2	ultrs1  d above=four montl ULCVX ULCVX UNCVX	U1TFS  DIS  ULDV2  ULDR2  ULDV4	18.81 18.81 20.12	386.33 386.33 387.2	66.35 66.35 67.22	73.04 73.04 73.98	6.37 6.37 7.31		19.99 19.99 19.99			

Page 6 of 22

Local County   Solidates   150   Part   Pa												
Color   Decision   Secti		Local Channel - Dedicated - DS3 - Per Mile per mon	ULDD3	1L5NC	8.98							
MINTERERS		Local Channel - Dedicated - DS3 - Facility Termination per mon				903.34	528.05	238.2	166.62	19.99		
MATHEMATICAL   Mathematical   Math		Local Channel - Dedicated - STS-1- Per Mile per mon										
Discreption   Discreption		Local Channel - Dedicated - STS-1 - Facility Termination per mon	ULDS1	ULDFS	550.34	903.34	528.05	238.2	166.62	19.99		
Discreption   Discreption												
COLUMN COLUMN	MULTIPLEXERS											
Part   EDIFFO CO   69/11   - 93 1 to 50 Covers   69/11   - 93 1		Channelization - DS1 to DS0 Channel Syste						21	19.52	19.99		
Decoration Court   Special C		OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)	UDL			13.16						
Color   Colo												
STR 10.07 Clavore System pare more   UPTS   MO3   19.8     SS     SS     SS     SS     SS     SS     SS     SS     SS     SS   SS     SS		Voice Grade COCI - DS1 to DS0 Channel System - per mon	UEA									
District   District						356.4	188	66.3	63.44			
DANK PREED.  Day Filer. Four Filer Stands Are Route file or Fincision Trend per memb- Lucal Counts.  Day Filer. Four Filer Stands Are Route file or Fincision Trend per memb- Lucal Counts.  Day Filer. Four Filer Stands Are Route file or Fincision Trend per memb- Lucal Lucy Counts file or Fincision Trend per memb- Lucal Lucy Counts file. Four Filer Stands Are Route file or Fincision Trend per memb- Lucal Lucy Counts file. Four Filer Stands Are Route file or Fincision Trend per memb- Lucal Lucy Counts file. Four File Stands Are Route file or Fincision Trend per memb- Lucal Lucy Counts file. Four File File Stands Are Route file or Fincision Trend per memb- Lucal Lucy Counts file. Four File File Stands Are Route file or Fincision Trend per memb- Lucal Lucy Counts file. Four File File Stands Are Route file or Fincision Trend per memb- Lucal Lucy Counts file. Four File File Stands Are Route file										19.99		
Day Filter, For Fiber Stronds, Per Robe Nike of Fraction Theory or month - Local   1927   11,000   28   1278.61   1278.62		DS3 Interface Unit (DS1 COCI) used with Loop per monti	USL	UC1D1	14.53	13.16	9.43					
Day Filter, For Fiber Stronds, Per Robe Nike of Fraction Theory or month - Local   1927   11,000   28   1278.61   1278.62												
Chance   C	DARK FIBER											
MRC Dath Filter Local Channel   USP   USPC   1278-01   275-62   532-07   394-05   19-95												
Dark Flant Four Place Stands, Per Roule Miles of Frestion Thereof part rough   11 pp			UDF	1L5DC	48							
Interesting Chance			UDF	UDFC4		1278.61	275.82	632.07	394.05	19.99		
MSC Dan Flate - Intercenter, Change   Lipid   1778-51   2778-50   6227   394-00   1939		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -										
Data Figure Four Four Four Four Mile or Fraction Theoret per month - Local USF   USP   U		Interoffice Channe	UDF	1L5DF	31.51							
Data Figure Four Four Four Four Mile or Fraction Theoret per month - Local USF   USP   U		NRC Dark Fiber - Interoffice Channe	UDF	UDF14		1278.61	275.82	632.07	394.05	19.99		
Logs												
NRC Des Fiber Local Log			UDF	1L5DL	48							
TRANSPORT OTHER					-	1278.61	275.82	632.07	394.05	19.99		
Optional Features & Functions:  Clear Channel Capability (BRZSES) Option - Subsequent - per DS1 Chain  WACCES TEN DIOTI SCREENING  BXX ACCESS TEN DIOTI SCREEN	TRANSPORT OTHER											
Clear Coursed Capability (B8ZSESF) Option - Subsequent - per DS1 Chara   UNC1X   CCOEF   184.91   23.82   1.99   0.78   19.99												
Clear Coursed Capability (B8ZSESF) Option - Subsequent - per DS1 Chara   UNC1X   CCOEF   184.91   23.82   1.99   0.78   19.99												
Clear Coursed Capability (B8ZSESF) Option - Subsequent - per DS1 Chara   UNC1X   CCOEF   184.91   23.82   1.99   0.78   19.99												
Clear Coursed Capability (B8ZSESF) Option - Subsequent - per DS1 Chara   UNC1X   CCOEF   184.91   23.82   1.99   0.78   19.99	Ontional F	eatures & Functions:										
Clear Channel Capability (823-98) Cytion - Subsequent - per DS1 Chan	Optional I			+								$\overline{}$
Clear Channel Capability (823-98) Cytion - Subsequent - per DS1 Chan		Clear Channel Canability (B8ZS/ESE) Ontion - Subsequent - per DS1 Chang	LINC1Y	CCOFF		184 91	23.82	1 00	0.78	19 90		
SXX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv		Clear Channel Canability (B87S/SE) Ontion - Subsequent - per DS1 Chang					23.02					
BXX Access Ten Digit Screening, Pack XN, Especial W/D POTS Translation   Child Pack XN, Access Ten Digit Screening, Pack XN, Established W/D POTS Translation   Child Pack XN, Access Ten Digit Screening, Pack XN, Established W/D POTS Translation   Child Pack XN, Access Ten Digit Screening, Pack XN, Established W/D POTS Translation   Child Pack XN, Access Ten Digit Screening, Multiple InterLATA CXR Routing Pack XN, Pack XN, Access Ten Digit Screening, Multiple InterLATA CXR Routing Pack XN, Pack XN, Access Ten Digit Screening, Multiple InterLATA CXR Routing Pack XN, Pack XN, Access Ten Digit Screening, Multiple InterLATA CXR Routing Pack XN, Pack XN, Access Ten Digit Screening, Multiple InterLATA CXR Routing Pack XN, Access Ten Digit Screening, Change Cha	SAA VCCESS LEN DIGI	IT SCREENING	UNCIX	CCUSF		104.91	23.82	1.99	0.78	19.99	1	
BXX Access Ten Digit Screening, Per XXX No. Established WWD POTS Translation	BAA ACCESS TEN DIGI		OUD	NODAY		40.05	4.40			10.00		
BIXX Access Ten Digit Screening, Det SXX Access Ten Digit Screening, Usufurnized Area of Service Per 8XX Numb   BIXX Access Ten Digit Screening, Untiline Interface Access Ten Digit Screening, Untiline Interface Access Ten Digit Screening, Untiline Interface Access Ten Digit Screening, Untiline Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening, William Interface Access Ten Digit Screening William Interface Access Ten Digit		6XX Access 1en Digit Screening, Reservation Charge Per 6XX Number Reserv		Norta								
BXX Access Ten Digit Screening, William Enter LATS CAR Requised Park AX No.   19.99		8XX Access 1en Digit Screening, Per 8XX No. Established W/O POTS Translation		NOTTY								
SXA Access Ten Digit Screening, Multiple InterCATA CXR Routing Per CXR Requested Per EXX No. 2014   19.90		8XX Access 1en Digit Screening, Per 8XX No. Established with POTS Translatio										
Per 8X No.   OHD   N8FMX   8.16   4.67   19.99			OHD	N8FCX		6.97	3.49			19.99		
BXX Access Ten Digit Screening, Call Handing and Bestination Feature   OHD NBFDX   11,24   11,19   119,99   19,99		8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested	0110									
BXX Access Ten Digit Screening, Walt XN. Delivery, per que												
BXX Access Ten Digit Screening, wil RXX No. Delivery, per que   OHD   0.001		8XX Access Ten Digit Screening, Change Charge Per Reque					1.19					
BXX Access Ten Digit Screening w/BXX No. Delivery, per que   OHD   0.0011				N8FDX		6.97				19.99		
Complex Features, per quer   OHD		8XX Access Ten Digit Screening, w/ 8XX No. Delivery, per que	OHD		0.001							
SIXX Access Ten Digit Screening w/ POTS No. Delivery, per que   OHD   0.001												
BXX Access Ten Digit Screening w/ POTS No. Delivery, with Optional Complex Features, per quen   OHD   0.0011												
Features, per quen		8XX Access Ten Digit Screening, w/ POTS No. Delivery, per que	OHD		0.001							
LIDE INFORMATION DATA BASE ACCESS (LIDB)												
LIDB Common Transport Per Quer   OQT   0.00006		Features, per query	OHD		0.0011							
LIDB Common Transport Per Quer   OQT   0.00006												
LIDB Validation Per Quer   LIDB Originating Point Code Establishment or Chang   OQT, OQU   NPBX   107.6   19.99	LINE INFORMATION DA	ATA BASE ACCESS (LIDB)										
LIDB Validation Per Quer   LIDB Originating Point Code Establishment or Chang   OQT, OQU   NPBX   107.6   19.99		LIDB Common Transport Per Quer	OQT		0.00006							
LIDB Originating Point Code Establishment or Chang												
SignaLing (CCS7)				NRPBX		107.6				19.99		
CCS7 Signaling Termination, Per STP Por   1DB   PT8SX   174.08   19.99			1 1			-						
CCS7 Signaling Termination, Per STP Por   1DB   PT8SX   174.08   19.99	SIGNALING (CCS7)											
CCS7 Signaling Usage, Per TCAP Messag	13337	CCS7 Signaling Termination, Per STP Por	1DB	PT8SX	174.08		İ			19.99		
CCS7 Signaling Connection, Per link (A linh   1DB TPP++ 16.31 354.95 354.95 174.08 19.99				1.1227								
CCS7 Signaling Connection. Per link (B link) (also known as D lin   1DB   TPP++ 16.31   354.95   354.95   174.08   19.99				TPP++		354.95	354.95	174.08	174.08	19 99		
CCS7 Signaling Usage, Per ISUP Messag			1DB		16.31							
CCST Signaling Usage Surrogate, per link per LAT  CCST Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec  STP affectec  CCST Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCST Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  1DB CCAPD  8 8 8  19.99  E911 SERVICE  CALLING NAME (CNAM) SERVICE  CNAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User		CCS7 Signaling Usage, Per ISUP Messag			0.000037893	1100				10.00		
CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected 1DB CCAPD 8 8 8 19,99  E911 SERVICE 1DB CCAPD 8 8 8 19,99  CALLING NAME (CNAM) SERVICE		CCS7 Signaling Usage Surrogate, per link per LAT		STU56						19 99	1	
STP_affectec			100	0.000	020.30					10.00	1	$\overline{}$
CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  Stp Affected  1DB CCAPD  8 8  19.99  E911 SERVICE  CALLING NAME (CNAM) SERVICE  CNAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User			1DP	CCAPO		40	40			19.00		
Stp Affected   1DB   CCAPD   8   8   19.99		CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per	100	COALO		40	40			15.55		
E911 SERVICE  CALLING NAME (CNAM) SERVICE  CONAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User			1DP	CCAPD		8	8			19.00		
CALLING NAME (CNAM) SERVICE  CNAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User		0.07 / 11/00/00	100	COALD		U	U			10.00		
CALLING NAME (CNAM) SERVICE  CNAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	E911 SERVICE			+								
CNAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	LU. I OLIVIOL			$\rightarrow$			1				1	
CNAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	<del>                                     </del>		<del>                                     </del>				1					
CNAM for DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM for Non DB Owners, Per Quen  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	CALLING NAME (CHAM	N SEDVICE	<del>                                     </del>				1					
CNAM for Non DB Owners, Per Quen  OQV  0.01  CNAM (Non-Databs Owner), NRC, applies when using the Character Based User	CALLING NAME (CNAM		001/	+	0.040		-					
CNAM (Non-Databs Owner), NRC, applies when using the Character Based User				+			-					
CNAM (Non-Databs Owner), NRC, applies when using the Character Based User Interface (CHUI)  OQV CDDCH 595 595 19.99		CNAM for Non DB Owners, Per Query	UQV		0.01		1	-				
CNAM (Non-Databs Owner), NRC, applies when using the Character Based User Interface (CHUI)  OQV CDDCH 595 595 19.99			<del>                                     </del>	$\rightarrow$			-					
CNAM (Non-Databs Owner), NRC, applies when using the Character Based User   OQV CDDCH   595   595   19.99			1 1				-					
Interface (CHUI)				1								
		Interface (CHUI)	OQV	CDDCH		595	595			19.99		
							1				 <u> </u>	

LNP QUERY SERVICE													
OPERATO	OR SERVICES AND DIRECTORY ASSISTANCE												
OPERATOR CALL PRO	OCESSING												
	Oper. Call Processing - Oper. Provided, Per Min Using BST LID				1.2								
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LID				1.24								
	Oper. Call Processing - Fully Automated, per Call - Using BST LID				0.2								
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE				0.2								
	Oper. Call Flocessing - Fully Automated, per Call - Osing Foreign Etc				0.2								
INWARD OPERATOR	CED//ICEC			-									
INWARD OPERATOR :													
	Inward Operator Services - Verification, Per Ca				1								
	Inward Operator Services - Verification and Emergency Interrupt - Per C				1.95								
BRANDING - OPERAT	FOR CALL PROCESSING												
	Recording of Custom Branded OA Announcement			CBAOS		7000	7000		19.99				
	Loading of Custom Branded OA Announcement per shelf/NAV			CBAOL		500	500		19.99	19.99	19.99		
DIRECTORY ASSISTA	ANCE SERVICES												
	DRY ASSISTANCE ACCESS SERVICE												
DIRECTO		<b></b>		+	0.075		-		+				
$\vdash$	Directory Assistance Access Service Calls, Charge Per Ca				0.275								
				1				<del>                                     </del>					
DIRECTO	DRY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)			1									
	Directory Assistance Call Completion Access Service (DACC), Per Call Attem			1	0.1								
UNBRANI													
	DRY TRANSPORT												
20.0	Directory Transport - Local Channel DS			1	36.32	637.46	546.94		19.99				
	Directory Transport - DS1 Level Interoffice Per Mi		+	+	0.45	007.70	040.04		10.00				
	Directory Transport - DS1 Level Interoffice Per Facility Termination			+	55.05	298.18	231.18	<del>                                     </del>	19.99				
						298.18	231.16		19.99				
	Switched Common Transport Per DA Access Service Per Ca				0.000175								
	Switched Common Transport Per DA Access Service Per Call Per Mi				0.000004								
	Access Tandem Switching Per DA Access Service Per Ca				0.000783								
	Directory Transport - Installation NRC, Per Trunk or Signaling Connection					501.98	13.32		19.99				
DIRECTO	DRY ASSISTANCE DATA BASE SERVICE (DADS)												
	Directory Assistance Data Base Service Charge Per Listir				0.04								
	Directory Assistance Data Base Service, per mont			DBSOF	150								
BRANDING - DIRECTO	ORY ASSISTANCE												
DIVARDING DIRECTO	Custom Branding Announcement, per Recording to be used with the provision of DA		AMT	CBADA		3000	3000						
			AMT	CBADA		690	690						
	Loading of Custom Branded Announcement per DRAM Card/Switch		AIVII	CBADC		690	690						
SELECTIVE ROUTING	3												
	Selective Routing Per Unique Line Class Code Per Request Per Swit			USRCR		229.65	229.65			19.99	19.99		
VIRTUAL COLLOCATI	TON												
			ueanl,uea,udn,udc										
	Virtual Collocation - 2-wire Cross Connects (loop		al,uhl,ucl,uec	UEAC2	0.31	54.21	51.07		19.99				
	Virtual Collocation 2 Wire Cross Connects (Loop) for Line Splittin	1	UEPSR, UEPSB	VE1LS	0.31	54.21	51.07		19.99				
<del></del>	Virtual Collocation - 2-wire Cross Connects (Loop) for Line Splittin  Virtual Collocation - 2-wire Cross Connects (por	-	OLF SIN, DEFSB	VE1R2	0.31	54.21	51.07		19.99				
			and the section										
<del></del>	Virtual Collocation - 4-wire Cross Connects (loop		uea,uhl,ucl,ud	UEAC4	0.62	54.23	50.96	<del>                                     </del>	19.99				
	Virtual Collocation - 4-wire Cross Connects (por		01.0	VE1R4	0.62	54.23	50.96		19.99				
	Virtual Collocation - 2-Fiber Cross Connect		CLO	CNC2F	15.64	41.56	29.82			19.99	19.99	19.99	19.99
	Virtual Collocation - 4-Fiber Cross Connects		CLO	CNC4F	28.11	50.53	38.78			19.99	19.99	19.99	19.99
	Virtual Collocatin - DS1 Cross Connect		USL,ULC,CLO	CNC1X	1.5	44.07	31.86	12.76 11.53					
				1			1						
AIN SELECTIVE CARR	RIER ROUTING			1									
	Regional Service Establishment		SRC	SRCEC		391788			19.99				
	End Office Establishment		SRC	SRCEO		320.53	320.53		19.99				
	Line/Port NRC, per end user		SRC	SRCLP		2.06	2.06		19.99				
1 1			SRC	ONOLI	0.000448	2.00	2.00	<del>                                     </del>	10.00				
	Query NRC per query			1	0.000440		1		1				
	Query NRC, per query												
AIN PELLOCUTY													
AIN - BELLSOUTH AIN	N SMS ACCESS SERVICE												
	N SMS ACCESS SERVICE												
AIN - BELLSOUTH AIN	N SMS ACCESS SERVICE												
AIN - BELLSOUTH AIN	N SMS ACCESS SERVICE												
AIN - BELLSOUTH AIN	N SMS ACCESS SERVICE												
	N SMS ACCESS SERVICE												
AIN - BELLSOUTH AIN ODUF/EDOUF/ADUF/C	N SMS ACCESS SERVICE N TOOLKIT SERVICE CMDS												
AIN - BELLSOUTH AIN ODUF/EDOUF/ADUF/C	N SMS ACCESS SERVICE N TOOLKIT SERVICE CMDS DALLY USAGE FILE (ADUF)				0.001								
AIN - BELLSOUTH AIN ODUF/EDOUF/ADUF/C	N SMS ACCESS SERVICE  N TOOLKIT SERVICE  CMDS  DAILY USAGE FILE (ADUF)    ADUF: Message Processing, per messag				0.004								
AIN - BELLSOUTH AIN ODUF/EDOUF/ADUF/C	N SMS ACCESS SERVICE N TOOLKIT SERVICE CMDS DALLY USAGE FILE (ADUF)				0.004 0.001								
AIN - BELLSOUTH AIN  ODUF/EDOUF/ADUF/C  ACCESS	N SMS ACCESS SERVICE  N TOOLKIT SERVICE  CMDS  DAILY USAGE FILE (ADUF)  ADUF: Message Processing, per messag  ADUF: Data Transmission (CONNECT:DIRECT), per messag												
AIN - BELLSOUTH AIN ODUF/EDOUF/ADUF/C ACCESS	N SMS ACCESS SERVICE  N TOOLKIT SERVICE  CMDS  DAILY USAGE FILE (ADUF)  ADUF: Message Processing, per messag  ADUF: Data Transmission (CONNECT:DIRECT), per messag  ED OPTIONAL DAILY USAGE FILE (EDDUF)				0.001								
AIN - BELLSOUTH AIN ODUF/EDOUF/ADUF/C ACCESS	N SMS ACCESS SERVICE  N TOOLKIT SERVICE  CMDS  DAILY USAGE FILE (ADUF)  ADUF: Message Processing, per messag  ADUF: Data Transmission (CONNECT:DIRECT), per messag												

OPTIONAL DAILY USAGE FILE (ODUF)												
ODUF: Recording, per message				0.0008611								+
ODUF: Message Processing, per message				0.0032357								
ODUF: Message Processing, per Magnetic Tape provisions				55.68								
ODUF: Data Transmission (CONNECT:DIRECT), per messag				0.0000365								+
NHANCED EXTENDED LINK (EELs)												+
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, Fl				ew Orleans, LA;								
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below												
NOTE: In all states, EEL network elements shown below also apply to currently combined facilities which a				s Is Charge appli	es to currently	combined faci	lities conver	ted to UNEs.(Non-	recurring rates do r	not apply.)		+
NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the GA f	order.(	(No Switch As I	s Charge.)									+
2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)												+
First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	1	UNCVX	UEAL2	17.27								+
First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zoni	2	UNCVX	UEAL2	32.32								+
First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zono	3	UNCVX	UEAL2	55.78								
Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor		UNC1X	1L5XX	0.2407								+
Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor		UNC1X	U1TF1	97.38								
DS1 Channelization System Per Mont		UNC1X	MQ1	139.65								
Voice Grade COCI - DS1 To Ds0 Interface - Per Montt  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport	_	UNCVX	1D1VG	0.7676							<del>                                     </del>	+
Combination - Zone	1	UNCVX	UEAL2	17.27								
Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport												
Combination - Zone :	2	UNCVX	UEAL2	32.32								4
Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone (	3	UNCVX	UEAL2	55.78								
Voice Grade COCI - DS1 to DS0 Channel System combination - per mon	3	UNCVX	1D1VG	0.7676								+
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		11.19	11.19	13.91	13.91	19.99			
4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -												+
Zone 1	1	UNCVX	UEAL4	20.92								
First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -			OL/IL!	20.02								+
Zone 2	2	UNCVX	UEAL4	39.14								
First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3	3	UNCVX	UEAL4	67.57								
Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	3	UNC1X	1L5XX	0.2407								+
Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor		UNC1X	U1TF1	97.38								+
Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	139.65								
Voice Grade COCI - DS1 to DS0 Channel System combination - per mon		UNCVX	1D1VG	0.7676								
Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone	1	UNCVX	UEAL4	20.92								
Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport		OHOVA	OEALT	20.02								+
Combination - Zone :	2	UNCVX	UEAL4	39.14								
Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport												
Combination - Zone (	3	UNCVX	UEAL4	67.57								+
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		11.19	11.19	13.91	13.91	19.99			
4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL	)											
First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1	1	UNCDX	UDL56	25.02								
First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	1	UNCDX	UDLOB	35.92							<del>                                     </del>	+
Zone 2	2	UNCDX	UDL56	40.32								
First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -												
Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	3	UNCDX UNC1X	UDL56 1L5XX	37.9 0.2407								-
interornee Transport - Dedicated - DST combination - Per Mile Per Mor	+++	UNCIA	ILOXX	0.2407								+
Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor		UNC1X	U1TF1	97.38					19.99			
Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	139.65	-							
OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb:	+	UNCDX	1D1DD	1.63								+
Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone	1	UNCDX	UDL56	35.92					19.99			
Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		5ODX	33200	55.52					13.33			1
Combination - Zone 2	2	UNCDX	UDL56	40.32					19.99			
Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	_	LINODY	LID! 50	27.0					10.00			
Combination - Zone ( OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-	3	UNCDX	UDL56	37.9					19.99		<del>                                     </del>	+
64kbs)		UNCDX	1D1DD	1.63								
				-								T
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		11.19	11.19	13.91	13.91	19.99			
4-WIRE 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	)											+
Zone 1	1	UNCDX	UDL64	35.92								

1	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	2	UNCDX	UDL64	40.00							
+	Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	2	UNCDX	UDL64	40.32							
	Zone 3	3	UNCDX	UDL64	37.9							
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.2407							
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor		UNC1X	U1TF1	97.38							
+	Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	139.65							_
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		UNCDX	4D4DD	4.00	0	0					
+	64kbs) Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	+ +	UNCDX	1D1DD	1.63	0	0					_
	Combination - Zone	1	UNCDX	UDL64	35.92							
_	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		ONODA	ODLOT	00.02							
	Combination - Zone :	2	UNCDX	UDL64	40.32							
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport											
	Combination - Zone :	3	UNCDX	UDL64	37.9							
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-											
	64kbs)		UNCDX	1D1DD	1.63							
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		11.19	11.19	13.91	13.91	19.99		
	Nonlecuring Currently Combined Network Elements Switch -As-is Chart		UNCIA	UNCCC		11.19	11.19	13.91	13.91	19.99		
4-WIRE DS1	1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)											
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	1	UNC1X	USLXX	50.26							
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	2	UNC1X	USLXX	94.06							
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	3	UNC1X	USLXX	162.34							
+	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.2407							_
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor	1 1	UNC1X	U1TF1	97.38							
+	interonice transport - Dedicated - D5 i combination - Facility Termination Per Mol		UNCIX	UIIFI	97.38						1	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg	1 1	UNC1X	UNCCC		11.19	11.19	13.91	13.91	19.99		
4-WIRE DS1	1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)											
	First DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	50.26							
	First DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	94.06							
	First DS1Loop in DS3 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	162.34							
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS3 - Facility Termination per mor		UNC3X UNC3X	1L5XX U1TF3	5.1 1191.53							+
	DS3 to DS1 Channel System combination per mon		UNC3X	MQ3	194.82							_
_	DS3 Interface Unit (DS1 COCI) combination per mont		UNC1X	UC1D1	14.53							1
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	50.26							
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	94.06							
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	162.34							
	DS3 Interface Unit (DS1 COCI) combination per montl	-	UNC1X	UC1D1	14.53							
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charc		UNC3X	UNCCC		11.19	11.19	13.91	13.91	19.99		
_	Nonrecurring Currently Combined Network Elements Switch -As-is Chart		ONCOX	UNCCC		11.13	11.13	13.31	13.91	13.33		1
2-WIRE VOI	ICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)											
	2 WiseVC Lass used with 2 wise VC Istartfiles Transport Combination 7	1	11110101	UEAL2	17.27							
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone		UNCVX									
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	2	UNCVX	UEAL2	32.32							
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone		UNCVX	UEAL2 UEAL2	32.32 55.78							
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor	2	UNCVX	UEAL2	32.32							
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility	2	UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX	32.32 55.78 0.0118					19 99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor	2	UNCVX	UEAL2 UEAL2	32.32 55.78					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility	2	UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX	32.32 55.78 0.0118	11.19	11.19	13.91	13.91	19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Char	2	UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX U1TV2	32.32 55.78 0.0118	11.19	11.19	13.91	13.91			
4-WIRE VOI	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Char ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)	3	UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC	32.32 55.78 0.0118 29.51	11.19	11.19	13.91	13.91			
4-WIRE VOI	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Char  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)  4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	3	UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC	32.32 55.78 0.0118 29.51	11.19	11.19	13.91	13.91			
4-WIRE VOI	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Char  CE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	1 2	UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC UEAL4 UEAL4	32.32 55.78 0.0118 29.51 20.92 39.14	11.19	11.19	13.91	13.91			
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	3	UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4	32.32 55.78 0.0118 29.51 20.92 39.14 67.57	11.19	11.19	13.91	13.91			
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Charç  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Pomeline Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Pomeline Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Pomeline Pomer	1 2	UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC UEAL4 UEAL4	32.32 55.78 0.0118 29.51 20.92 39.14	11.19	11.19	13.91	13.91			
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	1 2	UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4	32.32 55.78 0.0118 29.51 20.92 39.14 67.57	11.19	11.19	13.91	13.91			
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 4- Wire VGe Grade combination - Per Mile Per Mor	1 2	UNCVX	UEAL2 UEAL2 UEAL2 IL5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 UEAL4 IL5XX U1TV4	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118				13.91			
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 4- Wire VGe Grade combination - Per Mile Per Mor	1 2	UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 1L5XX	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118	11.19	11.19	13.91	13.91			
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charç  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG Combination - Pack Mile Per Mor Interoffice Transport - Dedicated - 4- Wire VG Corable Combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq	1 2	UNCVX	UEAL2 UEAL2 UEAL2 IL5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 UEAL4 IL5XX U1TV4	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2-Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)	1 2	UNCVX UNCVX	UEAL2 UEAL2 1L5XX  U1TV2 UNCCC  UEAL4 UEAL4 UEAL4 1L5XX  U1TV4  UNCCC	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118 26.22					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 4-wire VG combination - Facility Termination per mont!  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)	1 2	UNCVX	UEAL2 UEAL2 UEAL2 IL5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 UEAL4 IL5XX U1TV4	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG Combination - Pacility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL) High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon	1 2	UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 IL5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 UEAL4 UL5XX U1TV4 UNCCC	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118 26.22					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 4-Wire VG combination - Facility Termination per montf  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per montf	1 2	UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 1L5XX U1TV4 UNCCC	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118 26.22					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG Combination - Pacility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL) High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon	1 2	UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 IL5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 UEAL4 UL5XX U1TV4 UNCCC	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118 26.22					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 4-Wire VG combination - Facility Termination per montf  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)  High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per montf	1 2	UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX U1TV2 UNCCC UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 1L5XX U1TV4 UNCCC	32.32 55.78 0.0118 29.51 20.92 39.14 67.57 0.0118 26.22					19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charç  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-Wire VG Grade combination - Zone Interoffice Transport - Dedicated - 4-Wire VG Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charç  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL) High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon	1 2	UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 IL5XX  U1TV2 UNCCC  UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 IL5XX  U1TV4  UNCCC  1L5ND UE3PX 1L5XX U1TF3	32.32 55.78 0.0118 29.51 29.51 20.92 39.14 67.57 0.0118 26.22 11.53 379.72 5.1		11.19	13.91	13.91	19.99		
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2-Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-wire VG Grade combination - Zone Interoffice Transport - Dedicated - 4-wire VG ordination - Facility Termination per montf  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL) High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon Interoffice Transport - Dedicated - Josa Combination - Per Mile Per mon Interoffice Transport - Dedicated - PDS3 - Per Mile Per mon	1 2	UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 1L5XX  U1TV2 UNCCC  UEAL4 UEAL4 UEAL4 1L5XX  U1TV4  UNCCC  1L5ND  UE3PX 1L5XX	32.32 55.78 0.0118 29.51 29.51 20.92 39.14 67.57 0.0118 26.22 11.53 379.72 5.1					19.99		
DS3 DIGITA	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charç  ICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - 4-Wire VG Grade combination - Zone Interoffice Transport - Dedicated - 4-Wire VG Grade combination - Facility Termination per montt  Nonrecurring Currently Combined Network Elements Switch -As-Is Charç  LEXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL) High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mon Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 - Per Mile per mon Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mon	1 2	UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	UEAL2 UEAL2 IL5XX  U1TV2 UNCCC  UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 IL5XX  U1TV4  UNCCC  1L5ND UE3PX 1L5XX U1TF3	32.32 55.78 0.0118 29.51 29.51 20.92 39.14 67.57 0.0118 26.22 11.53 379.72 5.1	11.19	11.19	13.91	13.91	19.99		

	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per												i		
	month		UNCSX	UDLS1	394.76										
	Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor		UNCSX	1L5XX	5.1									<u> </u>	<b></b>
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mo		UNCSX	U1TFS	1165.53									<b></b>	<b></b>
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCSX	UNCCC		11.19	11.19	13.91	13.91		19.99			<b></b>	<b></b>
0.14/105	ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)														
2-WIRE				1111.01										<b></b>	<b></b>
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	1	UNCNX	U1L2X	23.66									<b></b>	<b></b>
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	2	UNCNX	U1L2X	44.28									<b></b>	<del> </del>
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	3	UNCNX	U1L2X	76.42									<b></b>	<b></b>
	Interoffice Transport - Dedicated - DS1 combination - Per Mi		UNC1X	1L5XX	0.2407									<b></b>	<del> </del>
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per mor		UNC1X	U1TF1	97.38									<b></b>	<b></b>
	Channelization - Channel System DS1 to DS0 combination - per mor		UNC1X	MQ1	139.65									<b>_</b>	<b>_</b>
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon		UNCNX	UC1CA	3.5									<b></b>	<b></b>
	A Life and a sign POM and its property of the Taylor of Carlot and Taylor	1	LINIONIN	U1L2X	00.00										
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	1	UNCNX	U1L2X	23.66										
	A LEGIC AND TO THE PORT OF THE	2	LINIONIY	1141.07	44.00										
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	2	UNCNX	U1L2X	44.28										
	Additional Quality IDCN Language DCAInter Will Towns to Continuous T		LINIONIN	1141.07	70.40	1							i		
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	3	UNCNX	U1L2X	76.42									<b></b>	<b></b>
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per mon	$\rightarrow$	UNCNX	UC1CA	3.5	-					-			-	<del></del>
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg	$\rightarrow$	UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99			-	<del></del>
4 14/105	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL)														
4-WIKE				1101101										<b></b>	<b></b>
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	50.26									<b></b>	<b></b>
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	94.06									<b></b>	<b></b>
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	162.34									<b>_</b>	<b>_</b>
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor		UNCSX	1L5XX	5.1									<b>_</b>	<b>_</b>
	Interoffice Transport - Dedicated - STS1 combination - Facility Terminati		UNCSX	U1TFS	1165.53									<b>_</b>	<b>_</b>
	STS1 to DS1 Channel System conbination per mon		UNCSX	MQ3	194.82										
	DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	14.53									<u> </u>	<b>_</b>
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	50.26									<u> </u>	<b>_</b>
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	94.06									<u> </u>	
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	162.34									<u> </u>	
	DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	14.53										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCSX	UNCCC		11.19	11.19	13.91	13.91		19.99				
4-WIRE	56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)														
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone	1	UNCDX	UDL56	35.92										
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone	2	UNCDX	UDL56	40.32										
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone	3	UNCDX	UDL56	37.9										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per M		UNCDX	1L5XX	0.0118										
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminati		UNCDX	U1TD5	21.26										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCDX	UNCCC		11.19	11.19	13.91	13.91		19.99				
4-WIRE	64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)														
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone	1	UNCDX	UDL64	35.92										
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone	2	UNCDX	UDL64	40.32										
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone	3	UNCDX	UDL64	37.9										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M		UNCDX	1L5XX	0.0118										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminati	$\bot$	UNCDX	U1TD6	21.26						T				
				$\top$							$\neg$				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charc		UNCDX	UNCCC		11.19	11.19	13.91	13.91		19.99				
		$\Box$												<u> </u>	
ADDITIONAL NETWO	ORK ELEMENTS														
When us	sed as a part of a currently combined facility, the non-recurrng charges do not apply, but a S	Switch As I	ls charge does a	ipply.											
When us	sed as ordinarilty combined network elements in Georgia, the non-recurring charges apply ar	nd the Swit	tch As Is Charge	does not.											
Nonrecu	urring Currently Combined Network Elements "Switch As Is" Charge (One applies to each con	nbination)													
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion		-						l			-			
	Charge		UNCVX	UNCCC		11.19	11.19	13.91	13.91		19.99				
	56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion														
	Charge		UNCDX	UNCCC		11.19	11.19	13.91	13.91		19.99		i		
	DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99		i		
						1.2									
	DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		UNC3X	UNCCC		11.19	11.19	13.91	13.91		19.99		i		
	STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion														
	Charge		UNCSX	UNCCC		11.19	11.19	13.91	13.91		19.99		i		
				,										+	+
	-									1	II.				

	NOTE: Loca	Il Channel - Dedicated Transport - minimum billing period - Below DS3=one month,	DS3 and above:	four months								
SOTE: 1) Flexione, Entro Child. EREC I should contact a scantial appearance in a control of page and p												
MOTE   CI   Controllar, The extension curves contempt groups according regular to the control of the control					<u> </u>							
NOTE: (1) Exercised. CLEE* I may deep deep the Note does deposit Commission or part of control of flows. In he life the name of Ministry in he life to appeal of sections control or programs. According to the programs of Ministry in he life to appeal of sections of the programs. According to the programs of Ministry in he life to appeal of the programs. According to the programs of Ministry in he life to appeal of the programs of Ministry in he life to appeal of the programs. According to the programs of t	NOTE: (1) E	lectronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the	state specific ele	ectronic service orde	ering char	ges as ordered by	the State Comr	nissions				
NOTE: 157 Names Service Deer change: described, the search of Totals to be Silved on part LRR search	NOTE: (1) C	ontinued: The electronic service ordering charge currently contained in this rate exhibit	oit is the BellSou	th regional electroni	ic service	ordering charge						
Description Colic Charge, por LSR, Examined via BSTs OSS notations in elemants   Source   S				vice ordering charge	es, or CLE	C-1 may elect the	regional electro	onic service ordering change.				
The Total State of the State alone tops or loop as part of a combination rivers to Goographically Deservaged Life. Zone. To year disagraphically Deservaged Life. Zone Designations by Control Offices, refer to Internet Wedness: 100,000,000,000,000,000,000,000,000,000	NOTE: (2) N	wanuar Service Order charge: disconnect, in the state of Florida, to be bliled on a per	LSK basis									
The Total State of the State alone tops or loop as part of a combination rivers to Goographically Deservaged Life. Zone. To year disagraphically Deservaged Life. Zone Designations by Control Offices, refer to Internet Wedness: 100,000,000,000,000,000,000,000,000,000		Electronia OCC Charge, par LCP, authmitted via BCT's OCC interactive interference							-			
Name					COMEC		2.5					
Medium   Discrete		(Regional)			SOME		3.0					-
INSERTING FOR SET   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES STATE PARTS   PERS   INSERT CONTINUES	The "Zone" s	shown in the sections for stand-alone loops or loops as part of a combination refers to	Geographically	Deaveraged LINE 7	ones To	view Geographic	ally Deaverage	LINE Zone Designations by Central Office, refe	r to Internet \	Nehsite:	+	<u> </u>
September   Port   Exchange Ports   Port   Rate Scholade all evolution features will resel to be ordered using read USCOs			o o o o grap moany	Douvoiagou oive E	.01100. 10	Tion Googlapino	any Douvorago	a crez zono zooignationo zy contrai cinico, rore		robono.		
Continger Ports   Continger												
Contemporate   Cont												
Contemp Potes   Contemp	UNBUNDLED LOCAL EX	CHANGE SWITCHING(PORTS)										
NOTE: Although the Port Rate includes all available features will Ad 11 A 11 A 12 A 11 A 12 A 11 A 12 A 12		,										
2-WIRE VOICE GRADE LIVE PORT TATES (RES)	Exchange P	orts										
2-WIRE VOICE GALDE LIMIT PORT FAITE STATES   UEPSR   UEPSR   UEPSR   UEPSR   UEPSR   24.98   24.98   19.99	NOTE: Altho	ough the Port Rate includes all available features in GA & TN, the desired features v	will need to be o	rdered using retail l	USOCs							
Exchange Ports - 2-Wire Analog Line Port Mit Caller ID - Re		·										
Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re	2-WIRE VOI	CE GRADE LINE PORT RATES (RES)										
Echange Parts - 2-Wire Analog Line Port outgoing only - Re		Exchange Ports - 2-Wire Analog Line Port- Re:		UEPSR	UEPRL	2.61	24.98	24.98	19.99			
Exchange Ports - 2-Wire Analog Line Port edgeing only - Re									1			
Exchange Pots - 2-Wire VG unbunded KY extended local daing party Pot with Caller ID (LU UEPSR UEPRP 261 34.98 24.98 19.99		Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re		UEPSR	UEPRC	2.61	24.98	24.98	19.99			<u> </u>
Exchange Pots - 2-Wire VG unbunded KY extended local daing party Pot with Celer   UEPSR   UEPSR   UEPSR   2.51   2.4.98   2.4.98   19.99					1				1			
D. Res.   UEPRS   UEPRS   UEPRS   24.98   24.98   19.99				UEPSR	UEPRO	2.61	24.98	24.98	19.99			
Exchange Ports - 2-Wire VO unbunded res, low usage line port with Caller ID (LU   UEPSR   UE			ller	LIEBOD	UEDD:	0.51	04.55		40.00			
Subsequent Activity   UEPSR   USASC   UEPVF   3.39   UEPSR												ļ
PEATURES   All Available Vertical Feature		Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU	<del>                                     </del>	UEPSR	UEPAP	2.61	24.98	24.98	19.99	-		<u> </u>
PEATURES   All Available Vertical Feature		Subacquent Activity		LIEDED	116460		0		1			
All Available Vertical Feature  All Available Vertical Feature  All Paralters (BUS)  Exchange Ports - 2-Wire Analog Line Port without Caller ID- 8  Exchange Ports - 2-Wire Analog Line Port without Caller ID- 8  Exchange Ports - 2-Wire Volume of the Windows of the Port with unbunded port with Callers E464  UEPS8 UEPS8 UEPS0 261 37.55 37.55 119.99  Exchange Ports - 2-Wire Volume of the Wire V	EEATURES	Subsequent Activity		UEPSK	USASC	U	U	0				<del> </del>
2-WIRE VOICE GRADE LINE PORT RATES (BUS)	FEATURES	All Available Vertical Feature	+ + + + + + + + + + + + + + + + + + + +	LIEPSR	LIED\/E	3 30	n	0	10 00			
Exchange Ports - 2-Wire Vandroded Line Port without Caller (I) - Bit   Exchange Ports - 2-Wire Vandroded Line Port with unbundled port with Caller (E44   UEPSB   UEPBC   2.61   37.55   37.55   19.99		All Available vertical realure		OLF SIX	OLI VI	3.33	0	0	19.99			
Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller (E-484   UEPSB   UEPBC   2.61   37.55   37.55   19.99	2-WIRE VOI	CE GRADE LINE PORT RATES (BUS)										
Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller-E484   UEPBB   UEPBC   2.61   37.55   37.55   119.99	2 111112 1011			UEPSB	UEPBL	2.61	37.55	37.55	19.99			
B			84						10.00			
Exchange Ports - 2-Wire VG unbundled KY extended local dialing parity Port with Caler   UEPSB   UEPS			1	UEPSB	UEPBC	2.61	37.55	37.55	19.99			
Exchange Ports - 2-Wire VG unbundled KY extended local dialing parity Port with Caler   UEPSB   UEPS												
Exchange Ports - 2-Wire VG unbundled KY extended local dialing parity Port with Caller ID - Bu		Exchange Ports - 2-Wire Analog Line Port outgoing only - Bu		UEPSB	UEPBO	2.61	37.55	37.55	19.99			
D - Bus   Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B		Exchange Ports - 2-Wire VG unbundled KY extended local dialing parity Port with Ca	ller									
Subsequent Activity		ID - Bus.										
FEATURES		Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B		UEPSB	UEPB1	2.61	37.55	37.55	19.99			
FEATURES												
All Available Vertical Feature   UEPSB   UEPVF   3.39   0   0   119.99		Subsequent Activity		UEPSB	USASC	0	0	0				
EXCHANGE PORT RATES (DID & PBX)	FEATURES			LIEDOD	LIED) (E	0.00			40.00			-
Exchange Ports - 2-Wire DID Port	EVCHANCE			UEPSB	UEPVF	3.39	U	0	19.99		+	-
Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabilit   UEPDD   B3.28   404.18   191.44   144.71   4.9   19.99				HEDEY	HEDD2	10.07	238 60	37.40 110.4 7.5	10.00			-
Exchange Ports - 2-Wire ISDN Port (See Notes below   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPXE   UEPX UEPSX   UEPXE   UEPX		Exchange Forts - 2-Wile DID Fort		OLITEX	OLITZ	10.51	230.03	37.49 119.4 7.5	13.33			-
Exchange Ports - 2-Wire ISDN Port (See Notes below   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPYE   3.39   0 0 0   UEPTX UEPSX   UEPXE   UEPX UEPSX   UEPXE   UEPX		Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID canabilit		UEPDD	UEPDD	83 28	404 18	191.44 144.71 4.9	19 99			
All Features Offerex										1		
NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by B-Channels associated with 2-wire ISDN ports.									. 0.00	İ		
Exchange Ports - 2-Wire ISDN Port Channel Profile   UEPTX UEPSX   U1UMA   0   0   0   0   0   0   0   0   0	NOTE: Tran	ismission/usage charges associated with POTS circuit switched usage will also apply to		voice and/or circuit	switched	data transmission	by B-Channels	associated with 2-wire ISDN ports.				
Exchange Ports - 4-Wire ISDN DS1 Por   UEPEX   UEPEX   UEPEX   113.21   407.77   203.18   157.84   39.98   19.99	NOTE: Acce		New Business Re				es will be deterr		Request Proc	ess.		
2-Wire VG Unbundled 2-Way PBX Trunk - Re:  UEPSE UEPRD 2.61 36.47 36.47 19.99  2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu UEPSP UEPPC 2.61 36.47 36.47 19.99  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu UEPSP UEPPD 2.61 36.47 36.47 19.99  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu UEPSP UEPPD 2.61 36.47 36.47 19.99  2-Wire Valou Unbundled Incoming PBX Trunk - Bu UEPSP UEPPD 2.61 36.47 36.47 19.99  2-Wire Valou Unbundled PBX LD Terminal Port UEPSP UEPSP UEPLD 2.61 36.47 36.47 19.99  2-Wire Valou Unbundled PBX LD Terminal Hotel Por UEPSP U												
2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  19.99  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  19.99  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  19.99  2-Wire Voice Unbundled DBX Trunk - Bu  19.99  2-Wire Voice Unbundled PBX LD Terminal Port  19.99  2-Wire Voice Unbundled PBX LD Terminal Port  19.99  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  19.99  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  19.99  2-Wire Voice Unbundled PBX LD DDD Terminals Po  10.00  10.0		Exchange Ports - 4-Wire ISDN DS1 Por		UEPEX	UEPEX	113.21	407.77	203.18 157.84 39.98	19.99			
2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  19.99  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  19.99  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu  19.99  2-Wire Voice Unbundled DBX Trunk - Bu  19.99  2-Wire Voice Unbundled PBX LD Terminal Port  19.99  2-Wire Voice Unbundled PBX LD Terminal Port  19.99  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  19.99  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  19.99  2-Wire Voice Unbundled PBX LD DDD Terminals Po  10.00  10.0					1				1			
2-Wire VG Line Side Unbundled Dutward PBX Trunk - Bu		2-Wire VG Unbundled 2-Way PBX Trunk - Re:		UEPSE	UEPRD	2.61	36.47	36.47	19.99			
2-Wire VG Line Side Unbundled Dutward PBX Trunk - Bu												
2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bt   UEPSP   UEPP1   2.61   36.47   36.47   19.99		2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu	$\vdash$	UEPSP	UEPPC	2.61	36.47	36.47	19.99			-
2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bt   UEPSP   UEPP1   2.61   36.47   36.47   19.99		O Wise VO Line Cide Helandled Outstand POV To 14 D		LIEDOD	LIEDDO	0.04	20. 17	20.47	40.00			
2-Wire Analog Long Distance Terminal PBX Trunk - Bu			+							-		-
2-Wire Voice Unbundled PBX LD Terminal Port   UEPSP   UEPLD   2.61   36.47   36.47   19.99											1 1	<del>                                     </del>
2-Wire Vice Unbundled 2-Way PBX Usage Po   UEPSP   UEPXA   2.61   36.47   36.47   19.99		2-Wire Voice Unbundled PRX I D Terminal Port	<del>                                     </del>									
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por   UEPSP   UEPXB   2.61   36.47   36.47   19.99									10.00			
2-Wire Voice Unbundled PBX LD DDD Terminals Po  UEPSP UEPXC 2.61 36.47 36.47 19.99  2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt  UEPSP UEPXD 2.61 36.47 36.47 19.99  UEPSP UEPXE 2.61 36.47 36.47 19.99										İ		
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc UEPSP UEPXD 2.61 36.47 36.47 19.99 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc UEPSP UEPXE 2.61 36.47 36.47 19.99												
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc UEPSP UEPXD 2.61 36.47 36.47 19.99 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt UEPSP UEPXE 2.61 36.47 36.47 19.99		2-Wire Voice Unbundled PBX LD DDD Terminals Po	<u>1                                      </u>	UEPSP	<b>UEPXC</b>	2.61	36.47	36.47	19.99			
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt UEPSP UEPXE 2.61 36.47 36.47 19.99												-
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc										
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Po		UEPSP	UEPXE	2.61	36.47	36.47	19.99			
					1				1			
2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port Without LL UEPSP UEPXF 2.61 36.47 36.47 19.99		2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port Without LL	+	UEPSP	UEPXF	2.61	36.47	36.47	19.99			
O Miles Visite Helbrindled DDV (Keetrolled HD Area Celling De		O Wise Vision Habrard and DDV Kontrolay LUD Asset Only on De		LIEDOD	LIEBYS	0.04	20. 17	20.47	40.00			
2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Pc UEPSP UEPSG 2.61 36.47 36.47 19.99		z-wire voice unbundled PBX Kentucky LUD Area Calling Pc	<del>                                     </del>	UEPSP	UEPXG	2.61	36.47	36.47	19.99		1 1	<b>└</b>

	1 1								
2-Wire Voice Unbundled PBX Kentucky Premium Callling Pc		UEPSP	UEPXH	2.61	36.47	36.47	19.99		
2-Wire Voice Unbundled 2-Way PBX Kentucky Area Callling Port Without LL 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		UEPSP	UEPAJ	2.61	36.47	36.47	19.99	<del>                                     </del>	
Port		UEPSP	UEPXL	2.61	36.47	36.47	19.99		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPSP	UEPXM	2.61	36.47	36.47	19.99		
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling									
Port		UEPSP	UEPXO	2.61	36.47	36.47	19.99		
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPSP	UEPXS	2.61	36.47	36.47	19.99		
Subsequent Activity		UEPSP	USASC	0	0	0			
FEATURES Subsequent Activity		UEPSP	USASC	U	U	0			
All Available Vertical Feature		UEPSP UEPSE	UEPVF	3.39	0	0	19.99		
EXCHANGE PORT RATES (COIN)		OLI OI OLI OL	OLI VI	0.00			10.33		
Exchange Ports - Coin Por				3.04	40.71	40.71	19.99		
Local Switching Features offered with Port									
NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to	circuit switched	voice and/or circuit	t switched	data transmission	by B-Channels	s associated w	vith 2-wire ISDN ports.		
NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/h	New Business Re	equest Process. Ra				mined via the	Bona Fide Request/New Business Request Process.		
Exchange port - 4-wire ISDN trunk port -all available features includ			UEPEX	275.48	181.27	116.42	19.99		
DLED LOCAL SWITCHING, PORT USAGE									
Find Office Switching (Post House)			1			1		<del>                                     </del>	
End Office Switching (Port Usage)  End Office Switching Function, Per MOI			1	0.002562		1		<del>                                     </del>	
End Office Switching Function, Pel MOL			1	0.002562		+		+	
Tandem Switching (Port Usage) (Local or Access Tandem)			+			1		+	
Tandem Switching Function Per MOI			1	0.001096		1		+	
Tandon Omicining Candidate of MOC				3.551000		1			
Common Transport									
Common Transport - Per Mile, Per MOL				0.0000049					
Common Transport - Facilities Termination Per MO				0.000426					
DLED PORT/LOOP COMBINATIONS - COST BASED RATES									
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission rule t Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cr.	me manner as the of this rate exhiboration and Not ombined and Not ombined and Not on the manner of the one of the manner of the one of the manner as the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of of one of	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled I ons of loop/port ne	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section	me manner as the of this rate exhiboration and Not ombined and Not ombined and Not on the manner of the one of the manner of the one of the manner as the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of of one of	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled I ons of loop/port ne	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Co  in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	me manner as the of this rate exhiboration and Not ombined and Not ombined and Not on the manner of the one of the manner of the one of the manner as the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of of one of	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled I ons of loop/port ne	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Coin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates	me manner as the of this rate exhibition and Normal - Currently Co	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled I ons of loop/port ne s and the first and	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  [2-Wire VG Loop/Port Combo - Zone	me manner as the of this rate exhiboration and Not ombined and Not ombined and Not on the manner of the one of the manner of the one of the manner as the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of the one of of one of	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled I ons of loop/port ne s and the first and	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cr in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	me manner as the of this rate exhibiting and Noting - Currently Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Currently - Company - Cu	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled in one of loop/port ne s and the first and the first and the first 20.34	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  [2-Wire VG Loop/Port Combo - Zone	me manner as the of this rate exhibition and Normal - Currently Co	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled I ons of loop/port ne s and the first and	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	me manner as the of this rate exhibiting and Noting - Currently Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Currently - Company - Cu	ney are applied to the oit shall apply to all to Currently Combine	ne Stand-A combination	Alone Unbundled in one of loop/port ne s and the first and the first and the first 20.34	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling Comborting Cayling Comborting Cayling Comborti	me manner as the of this rate exhibiting and Noting - Currently Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Company - Currently - Currently - Company - Cu	ney are applied to it it shall apply to all it. Currently Combine mbined sections.	ne Stand-Acombination	Alone Unbundled i ons of loop/port ne is and the first and 16.15 22.34 30.88	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cr  in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	me manner as the of this rate exhibition of this rate exhibition of the combined and Noting - Currently Combined and Noting - Currently - Combined and Noting - Currently - Combined and Noting - Currently - Combined and Noting - Currently - Combined and Noting - Currently - Combined and Noting - Currently - Combined and Noting - Currently - Combined and Noting - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Combined - Currently - Currently - Combined - Currently - Currently - Currently - Currently - Currently - Currently - Currently - Currently - Currently	ney are applied to this shall apply to all currently Combine mbined sections.	ne Stand-Acombination ed Combos  UEPLX	Alone Unbundled i ons of loop/port ne s and the first and 16.15 22.34 30.88	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling Comborting Cayling Comborting Cayling Comborti	me manner as the of this rate exhibition of this rate exhibition of this rate exhibition of the rate o	ney are applied to it it shall apply to all it. Currently Combine mbined sections.	ne Stand-Acombination	Alone Unbundled i ons of loop/port ne is and the first and 16.15 22.34 30.88	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayw	me manner as the of this rate exhibit sombined and Nothing - Currently Colonia - Currently	ney are applied to it it shall apply to all it. Currently Combine mbined sections.	ne Stand-Accombination and Combos  UEPLX UEPLX UEPLX	None Unbundled I ons of loop/port ne s and the first and  16.15 22.34 30.88  13.54 19.73	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling Components of the Nonrecurring Cayling Volce Grade LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Line Port Rates (Res)	me manner as the of this rate exhibit sombined and Nothing - Currently Colonia - Currently	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX	uePLX UEPLX UEPLX UEPLX UEPLX	None Unbundled is ons of loop/port ne s and the first and 16.15 22.34 30.88 13.54 19.73 28.27	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring 2-Wire VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  1-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	me manner as the of this rate exhibit sombined and Nothing - Currently Colonia - Currently	ney are applied to it it shall apply to all it. Currently Combine mbined sections.	ne Stand-Accombination and Combos  UEPLX UEPLX UEPLX	None Unbundled I ons of loop/port ne s and the first and  16.15 22.34 30.88  13.54 19.73	etwork element	s except for l	UNE Coin Port/Loop Combinations.	ntly Combined	
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayward Cayw	me manner as the of this rate exhibit sombined and Nothing - Currently Colonia - Currently	uey are applied to this shall apply to all it currently Combine in the sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-Accombination and Combos UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX	16.15 22.34 30.88  13.54 19.73 28.27	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling Components of the Nonrecurring Cayling Volce Grade LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Loop (SL1) - Zone 2-Wire Volce Grade Line Port Rates (Res)	me manner as the of this rate exhibit sombined and Nothing - Currently Colonia - Currently	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX	uePLX UEPLX UEPLX UEPLX UEPLX	None Unbundled is ons of loop/port ne s and the first and 16.15 22.34 30.88 13.54 19.73 28.27	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section. For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cr in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Caywing Cay	me manner as the of this rate exhibit sombined and Nothing - Currently Colonia - Currently	uey are applied to this shall apply to all it currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A- combination and Combos  UEPLX UEPLX UEPLX UEPLX UEPRL	16.15 22.34 30.88  13.54 19.73 28.27	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayl	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it currently Combine in the sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-Accombination and Combos UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX	16.15 22.34 30.88  13.54 19.73 28.27	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section. For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cr in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Caywing Cay	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A- combination and Combos  UEPLX UEPLX UEPLX UEPLX UEPRL	16.15 22.34 30.88  13.54 19.73 28.27	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cr in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayl	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A- combination and Combos  UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section.  For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Ct in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling Ca	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A-combination of the co	\text{None Unbundled I on son of loop/port ne s and the first and the first and the first and 16.15 \\ 22.34 \\ 30.88 \\ 13.54 \\ 19.73 \\ 28.27 \\ \text{2.61} \\ 2.61 \\ 2.61 \\ \text{2.61} \\	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cri in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Kentucky extended local dialing parity port with Caller - residenc 2-Wire voice unbundled sers, low usage line port with Caller ID (LUI	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A-combination of the co	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayli	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A-combination of the co	16.15 16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61	additional Por	s except for Innonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cri in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring CA, TN and all other states (RES)  2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Kentucky extended local dialing parity port with Caller - residenc 2-Wire voice unbundled sers, low usage line port with Caller ID (LUI	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A-combination of the co	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61	etwork element	s except for l	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayli	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A-combination of the co	16.15 16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61	additional Por	s except for Innonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Cr in GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling and Tennessee, the recurring Charges shall be those identified in the Nonrecurring Cayling C	me manner as the of this rate value. The state of the state value of the state value of the stat	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX	ue Stand-A-combination of the co	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 3.39	additional Por	s except for Innonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayli	me manner as the of this rate value of the state value of the state of	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ue Stand-A-combination of the co	16.15 16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61	additional Por	s except for Innonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling and Tennessee, the recurring Charges shall be those identified in the Nonrecurring Cayling Ca	me manner as the of this rate value of the state value of the state of	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX	ue Stand-A-combination of the co	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 3.39	additional Por	s except for Innonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling and Tennessee, the recurring Charges shall be those identified in the Nonrecurring Cayling Ca	me manner as the of this rate value of the state value of the state of	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO UEPRAP	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 3.39	additional Por	s except for In nonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling and Tennessee, the recurring Charges shall be those identified in the Nonrecurring Cayling Ca	me manner as the of this rate value of the state value of the state of	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX	ue Stand-A-combination of the co	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 3.39	additional Por	s except for Innonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling and Tennessee, the recurring Charges shall be those identified in the Nonrecurring Cayling Ca	me manner as the of this rate value of the state value of the state of	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX	UEPLX UEPRO UEPRO UEPAP	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 3.39	o 0	s except for In nonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling and Tennessee, the recurring Charges shall be those identified in the Nonrecurring Cayling Ca	me manner as the of this rate value of the state value of the state of	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO UEPRAP	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 3.39	additional Por	s except for In nonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99  19.99		
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Crin GA, TN and all other states, the nonrecurring charges shall be those identified in the Nonrecurring Cayling and Tennessee, the recurring Charges shall be those identified in the Nonrecurring Cayling Ca	me manner as the of this rate value of the state value of the state of	uey are applied to this shall apply to all it shall apply to all it. Currently Combine mbined sections.  UEPRX	UEPLX UEPRO UEPRO UEPAP	16.15 22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 3.39	o 0	s except for In nonrecurring	UNE Coin Port/Loop Combinations.  charges apply to Not Currently Combined Combos. For Curre  19.99  19.99  19.99  19.99  19.99  19.99  19.99  19.99		

2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	$\perp$											
LINE Descrip	Out Fred to But a	1											
	oop Combination Rates	1			10.15								
	2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2			16.15 22.34								
	2-Wire VG Loop/Port Combo - Zone :	3		_	30.88								
	2-Wile VG Looph Oil Combo - Zone	3			30.00								
UNE Loop R	Rates												
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	13.54								
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPBX	UEPLX	19.73								
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	28.27								
2-Wire Voice	e Grade Line Port (Bus)												
	2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	2.61					19.99			
	O.W		HEDDY	UEPBC	0.04					40.00			
	2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBX	UEPBC	2.61					19.99			
	2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	2.61					19.99			
	2-Wire voice Grade unbundled Kentucky extended local dialing parity port with Caller ID		UEFBA	UEFBU	2.01					19.99			
	- bus		UEPBX	UEPBM	2.61					19.99			
	2-Wire voice unbundled incoming only port with Caller ID - Bu		UEPBX	UPEB1	2.61					19.99			
	2 / 1												
LOCAL NUN	MBER PORTABILITY												
	Local Number Portability (1 per port		UEPBX	LNPCX	0.35								
FEATURES													
	All Features Offered	-	UEPBX	UEPVF	3.39	0	0			19.99			
NONDECUE	DRING CHARGES (AIRCS), CHRRENTI V COMPINIED						-						
NUNKECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	+ +	UEPBX	USAC2		10	10			19.99			
	2-vvire voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPBA	USACZ		10	10			19.99			
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		UEPBX	USACC		10	10						
	2 WHIC VOICE CHARLE LOOP / EINE FOR COMBINATION CONVENSION CONTENT CHARLE		OLI DX	ОСЛОС		10	10						
ADDITIONAL	L NRCs												
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX	USAS2						19.99			
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)												
	oop Combination Rates												
	2-Wire VG Loop/Port Combo - Zone	1			16.15								
	2-Wire VG Loop/Port Combo - Zone :	2			22.34								
	2-Wire VG Loop/Port Combo - Zone	3			30.88								
UNE Loop R	Potos												
ONE LOOP N	2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPRG	UEPLX	13.54								
	2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPRG	UEPLX	19.73								
	2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPRG	UEPLX	28.27								
	E TINO TOIGG OTAGG EGOP (GE T) EGNO		020	OL: EX	EU.E.								
2-Wire Voice	e Grade Line Port Rates (RES - PBX)												
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - R€		UEPRG	UEPRD	2.61						19.99	19.99	
LOCAL NUN	MBER PORTABILITY												
	Local Number Portability (1 per port	-	UEPRG	LNPCP	3.5	1							
FEATURES		+ +		+		-							
FEATURES		+ +	UEPRG	UEPVF	3.39	0	0			19.99			
			UEPKG	UEPVF	3.39	U	U			19.99			
	All Features Offered			- 1				1					
NONRECLIP													
NONRECUR	All Features Offeret  RING CHARGES (NRCs) - CURRENTLY COMBINED												
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED		UEPRG	USAC2		10	10			19.99			
NONRECUR	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPRG	USAC2		10	10			19.99			
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED		UEPRG UEPRG	USAC2		10	10			19.99			
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change												
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change  L NRCs		UEPRG	USACC		10	10						
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change LNRCs LNRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ				0	10	10			19.99			
	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change  L NRCs		UEPRG	USACC	0	10	10						
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change LNRCs LNRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USACC	0	10	10			19.99			
ADDITIONA	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change  L NRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi		UEPRG	USACC	0	10	10			19.99			
ADDITIONA	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change LNRCs LNRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USACC	0	10	10			19.99			
ADDITIONAI	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change LNRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)		UEPRG	USACC	0	10	10			19.99			
ADDITIONAL 2-WIRE VOI	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change  L NRCs  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ  PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  oop Combination Rates		UEPRG	USACC		10	10			19.99			
ADDITIONAL 2-WIRE VOI	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change  L NRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ  PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  oop Combination Rates  2-Wire VG Loop/Port Combo - Zone	1	UEPRG	USACC	16.15	10	10			19.99			
ADDITIONAL  2-WIRE VOI	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change  L NRCs  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ  PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  oop Combination Rates	1 2 3	UEPRG	USACC		10	10			19.99			

Attachment 2 Exhibit C

JNE Loop F	2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPPX	UEPLX	13.54					-		
	2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPPX	UEPLX	19.73							
	2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPPX	UEPLX	28.27							
2-Wire Voic	e Grade Line Port Rates (BUS - PBX)											
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu		UEPPX	UEPPC	2.61				19.99			
	, and the second											
	Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	2.61				19.99			
	Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX	UEPP1	2.61				19.99			
	2-Wire Voice Unbundled PBX LD Terminal Port	_	UEPPX	UEPLD	2.61				19.99			
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc 2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX UEPPX	UEPXA	2.61 2.61				19.99 19.99			
	2-Wire Voice Unburidled PBX Toli Terminal Hotel Por		UEPPX	UEPAB	2.01				19.99			
	2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	2.61				19.99			
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	2.61				19.99			
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc		UEPPX	UEPXE	2.61				19.99			
	2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without Ll		UEPPX	UEPXF	2.61					19.99	19.99	
	2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Pc		UEPPX	UEPXG	2.61		1			19.99	19.99	
	2-Wire Voice Unbundled PBX Kentucky Premium Calling Pc		UEPPX	UEPXH	2.61		1			19.99	19.99	
	2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LU		UEPPX	UEPXJ	2.61		+			19.99	19.99	
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port		UEPPX	UEPXL	2.61				19.99			
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX	UEPXM	2.61				19.99			
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling											
	Port		UEPPX	UEPXO	2.61				19.99			
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX	UEPXS	2.61					19.99	19.99	
OCAL NU	MBER PORTABILITY								_			
	Local Number Portability (1 per port		UEPPX	LNPCP	3.15							
EATURES												
EATURES	All Features Offered		UEPPX	UEPVF	3.39	0	0		19.99			
	THIT CALAICS ONCICE		OLITA	OEI VI	0.00		, i		13.55			
NONRECUE	RRING CHARGES (NRCs) - CURRENTLY COMBINED											
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPPX	USAC2		10	10		19.99			
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change		UEPPX	USACC		10	10		19.99			
	Change		UEPPX	USACC		10	10		19.99			
ADDITIONA	AL NRCs											
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPPX	USAS2	0	0	0					
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64		19.99			
-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT											
JNE Port/Lo	oop Combination Rates											
	2-Wire VG Coin Port/Loop Combo – Zone 1				16.15							
	2-Wire VG Coin Port/Loop Combo – Zone 2				22.64		1					
	2-Wire VG Coin Port/Loop Combo – Zone 3				31.09		1					
							1					
JNE Loop F	Rates											
JNE Loop F	Rates		LIEBOO	LIEDLY	13.54							
JNE Loop F	Rates 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	13.54							
JNE Loop F	Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	19.73							
JNE Loop F	Rates 2-Wire Voice Grade Loop (SL1) - Zone											
	Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	19.73							
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO UEPCO	UEPLX UEPLX	19.73 28.27							
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)		UEPCO UEPCO	UEPLX UEPLX UEPRF	19.73 28.27				19.99	19.99		
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		UEPCO UEPCO	UEPLX UEPLX	19.73 28.27				19.99 19.99	19.99		
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)		UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPRF	19.73 28.27 2.91 2.91					19.99		
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening (AL, KY) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)		UEPCO UEPCO	UEPLX UEPLX UEPRF UEPRF	19.73 28.27				19.99	19.99		
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)  2-Wire Coin 2-Way with Operator Screening (AL, KY)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking: 900/976, 1+DDD, 011+, &		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPRF UEPRE UEPRA UEPKA	2.91 2.91 2.91 2.91				19.99 19.99 19.99	19.99		
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening (AL, KY) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPLX UEPRF UEPRA UEPKA UEPCD	2.91 2.91 2.91 2.91 2.91 2.91				19.99 19.99 19.99	19.99		
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)  2-Wire Coin 2-Way with Operator Screening (AL, KY) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS) 2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)		UEPCO F UEPRF UEPRE UEPRA UEPKA UEPCD UEPRN	2.91 2.91 2.91 2.91 2.91 2.91 2.91 2.91				19.99 19.99 19.99 19.99 19.99	19.99			
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening (AL, KY) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS) 2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPLX UEPRF UEPRA UEPKA UEPCD	2.91 2.91 2.91 2.91 2.91 2.91				19.99 19.99 19.99	19.99		
Every Force	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)  2-Wire Coin 2-Way with Operator Screening (AL, KY) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS) 2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)		UEPCO A UEPRA UEPKA UEPCD UEPRN UEPRN UEPRN	2.91 2.91 2.91 2.91 2.91 2.91 2.91 2.91				19.99 19.99 19.99 19.99 19.99	19.99			
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening (AL, KY) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin Outward with Operator Screening and Without Operator Screening (KY, LA, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)		UEPCO A UEPRA UEPRA UEPCD UEPRN UEPRN UEPRN UEPRN	2.91 2.91 2.91 2.91 2.91 2.91 2.91 2.91				19.99 19.99 19.99 19.99 19.99 19.99	19.99			
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin Outward with Operator Screening and 011 Blocking (KY, LA, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)		UEPCO X UEPLX UEPLX UEPLX UEPRF UEPRE UEPRA UEPKA UEPCD UEPRN UEPRJ UEPRH UEPCN	2.91 2.91 2.91 2.91 2.91 2.91 2.91 2.91				19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99			
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening (AL, KY) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY) 2-Wire Coin Outward with Operator Screening and Without Operator Screening (KY, LA, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)		UEPCO A UEPRA UEPRA UEPCD UEPRN UEPRN UEPRN UEPRN	2.91 2.91 2.91 2.91 2.91 2.91 2.91 2.91				19.99 19.99 19.99 19.99 19.99 19.99	19.99			

UN	NE Coin Port/Loop Combo Usage (Flat Ratı		ι	JEPCO	URECU	2.57	0	0				<u></u>
LOCAL NUMBE	ER PORTABILITY									<del></del>		-
	cal Number Portability (1 per port		ι	JEPCO	LNPCX	0.35						
FEATURES	Features Offered			JEPCO	UEPVF	3.39		0 0	19.99			┢
All	T eatures Offerer			DEFCO	OLI VI	3.39			19.99	-		t
	NG CHARGES - CURRENTLY COMBINED											
2-\	Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		L	JEPCO	USAC2		10	10	19.99			-
2-\	Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		ι	JEPCO	USACC		10	10	19.99			
ADDITIONAL N	IRCs Wire Voice Grade Loop/Line Port Combination - Subsequent Activ			JEPCO	USAS2		0	0	19.99			
	GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT			DE1 00	OOMOZ		Ů		10.00			
	Combination Rates											
	Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	1				28.72						-
2-\	Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	2 3				34.9 45.9						
UNE Loop Rate												H
	Wire Analog Voice Grade Loop - (SL2) - UNE Zone Wire Analog Voice Grade Loop - (SL2) - UNE Zone	1 2		JEPPX JEPPX	UECD1	17.78 23.96			19.99 19.99			-
	Wire Analog Voice Grade Loop - (SL2) - UNE Zone Wire Analog Voice Grade Loop - (SL2) - UNE Zone	3	1	JEPPX JEPPX	UECD1	34.96			19.99	+		$\vdash$
UNE Port Rate	change Ports - 2-Wire DID Por			JEPPX	UEPD1	10.94			19.99			
	NG CHARGES - CURRENTLY COMBINED			JEFFX	UEFUI	10.94			19.99			
	Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowab	ile										<del>                                     </del>
Ch	nanges		Į	JEPPX	USA1C		14.62	3.73	19.99			
ADDITIONAL N												
2-\	Wire DID Subsequent Activity - Add Trunks, Per Trun		·	JEPPX	USAS1		53.58	53.58	19.99			
Telephone Nur	mber/Trunk Group Establisment Charges											
	D Trunk Termination (One Per Port		U	JEPPX	NDT	0	0	0	19.99			
	Iditional DID Numbers for each Group of 20 DID Numbe D Numbers, Non- consecutive DID Numbers , Per Numbe			JEPPX JEPPX	ND4 ND5	0	0	0	19.99 19.99			+
	eserve Non-Consecutive DID number			JEPPX	ND6	0	0	0	19.99			<del>                                     </del>
	eserve DID Numbers			JEPPX	NDV	0	0	0	19.99			
LOCAL NUMBE	ER PORTABILITY											-
	cal Number Portability (1 per port		ι	JEPPX	LNPCP	3.15						<del>                                     </del>
	DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT			J	2.11 01	0.10						
	D Combination Rates  VISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	1		JEPPB JEPPR		35.4						
	V ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone			3 UEPPR	1	44.09						
2V	V ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	3	UEPPE	3 UEPPR	2	55.35						
UNE Loop Rate								1				$\vdash$
•	Wire ISDN Digital Grade Loop - UNE Zone	1	UEPPI	B UEPPR	USL2X	22.41			19.99			
2-\	Wire ISDN Digital Grade Loop - UNE Zone	2	UEPPE	B UEPPR	USL2X	31.1			19.99			
2-1	Wire ISDN Digital Grade Loop - UNE Zone	3	UEPPI	B UEPPR	USL2X	42.36			19.99			
UNE Port Rate					1							$\vdash$
	xchange Port - 2-Wire ISDN Line Side Po		UEPP	B UEPPR	UEPPB	12.99			19.99		=	
	NG CHARGES - CURRENTLY COMBINED											L
					1							
2-	Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination -		UEPP	B UEPPR	USACB	0	77.04	54.04	19.99			
2-	-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - nversion		UEPP	B UEPPR	USACB	0	77.04	54.04	19.99			

Attachment 2 Exhibit C

	No. of the Control of		HEDDO	LIEDDD	LNDOV	0.05								T	Т
Local	Number Portability (1 per port		UEPPB	UEPPR	LNPCX	0.35	0	0					-	+	$\exists$
B-CHANNEL USER	PROFILE ACCESS:														I
CVS/C	CSD (DMS/5ESS)		UEPPB	UEPPR	HILLOA	0	0	0							
	EWSD)			UEPPR		0	0	0					+	+	$\dashv$
CSD				UEPPR		0	0	0						+	$\forall$
D OLIANDEL ADEA	PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)														4
	CSD (DMS/5ESS)		LIEPPR	UEPPR	U1UCD	0	0	0					+	+	+
															T
	EWSD)		UEPPB	UEPPR		0	0	0							+
CSD			UEPPB	UEPPR	U1UCF	0	0	0							4
USER TERMINAL P	PROFILE														$\exists$
Hear T	Farminal Brafile (FIMCD only)		HEDDD	UEPPR	114111140	•		0							T
User I	Ferminal Profile (EWSD only)		UEPPB	UEPPR	UTUNA	0	0	0					+	+	+
VERTICAL FEATUR	RES														1
All Ver	rtical Features - One per Channel B User Profile		UEPPB	UEPPR	UEPVF	3.39	0	0							
INTEROFFICE CHA	ANNEL MILEAGE														+
		1												†	$\forall$
Interof	ffice Channel mileage each, including first mile and facilities termination		UEPPB	UEPPR	M1GNC	26.98	142.31	56.21			19.99			+	+
Interof	ffice Channel mileage each, additional mile		UEPPB	UEPPR	M1GNM	0.0301	0	0			0				4
4-WIRE DS1 DIGITA	AL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT														1
UNE Port/Loop Cor	mhination Pates														+
	S1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	1	UEF	PPP		219.25							+	+	$\dashv$
	S1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	2	UEF	PPP		248.36							<b>+</b>	+	T
	S1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	3	UEF	PPP		299.47									4
UNE Loop Rates													+	+	+
	e DS1 Digital Loop - UNE Zone	1	UEF	PPP	USL4P	106.04					19.99		+	+	+
4-Wire	e DS1 Digital Loop - UNE Zone	2	UEF	PPP	USL4P	135.15					19.99				
4-Wire	e DS1 Digital Loop - UNE Zone	3	UEF	PPP	USL4P	186.15					19.99				4
UNE Port Rate													+	+	+
	nge Ports - 4-Wire ISDN DS1 Por		UEF	PPP	UEPPP	113.21					19.99				Ī
NONDECTIONING O	CHARGES - CURRENTLY COMBINED														+
	e DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -			-									+	+	+
	rsion -Switch-as-is		UEF	PPP	USACP	0	238.22	157.17			19.99			-	4
ADDITIONAL NRCs															T
4-Wire	e DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within lowance		LIE	PPP	PR7TF	-	0.9804		· ·		19.99	· ·			T
4-Wire	DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All													+	+
	except NC]  DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos		UEF	PPP	PR7TO		23.02	23.02			19.99			+	+
	Std Allowance		UEF	PPP	PR7ZT		46.05	46.05			19.99				4
															#
LOCAL NUMBER P	PORTABILITY Number Portability (1 per port		LIE	PPP	LNPCN	1.75								-	+
			OL.		Litti Oit	0									Ī
INTERFACE (Provs		$\perp$		-	DDT			<u> </u>							4
Voice/ Digital				PPP PPP	PR71V PR71D	0	0	0					+		+
Inward				PPP	PR71E	0	0	0							7
New or Additional	"P" Channel												<u> </u>	1	4
	"B" Channel r Additional - Voice/Data B Channel	1	LIFE	PPP	PR7BV	0	29.06				19.99		+	+	+
	r Additional - Voice/Data B Channel	1		PPP	PR7BF	0	29.06		1		19.99		<b>†</b>	+	+
	r Additional Inward Data B Channel			PPP	PR7BD	0	29.06				19.99		<b>†</b>	†	+
New o	r Additional Useage Sensitive Voice Data B Channel		UEF	PPP	PR7BS	0	29.06				19.99			1	$\top$
	r Additional Useage Sensitive Digital Data B Channel		UEF	PPP	PR7BU	0	29.06				19.99				I
CALL TYPES															4
II ALL IVDES			1					1			1		1	<b></b>	4
Inward	·		UEF	ממכ	PR7C1	0	0	0							

	Two-way	+	UEPPP	PR7CC	0	0	U						
ntoroffice '	Channel Mileage	-					1						
nteronice (	Fixed Each Including First Mili		UEPPP	1LN1A	FF F	298.18	231.23	0		40	.99		
	Each Airline-Fractional Additional Mil		UEPPP	1LN1A 1LN1B	55.5 0.45	298.18	231.23	U		19	.99		
	Each Ainme-Fractional Additional Will		UEFFF	ILNID	0.45								
1-WIDE DS	S1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT												
F-WINE DO	ST DIGITAL LOOF WITH 4-WIKE DDITS TRONK FORT												
INE Port/I	Loop Combination Rates												
DIAL FOLUL	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	1	UEPDC		189.32					10	.99		
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	2	UEPDC		218.43						.99		
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	3	UEPDC		269.54						.99		
	4W DST Digital Loop/4W DDITS Truffk Fort - ONE Zoffe	3	UEFDC		209.54					18	.99		
JNE Loop	Pates												
DIAL LOOP	4-Wire DS1 Digital Loop - UNE Zone	1	UEPDC	USLDC	106.04					10	.99		
	4-Wire DS1 Digital Loop - ONE Zone	2	UEPDC	USLDC	100.04				23.33		.99		
	4-Wire DS1 Digital Loop - UNE Zone	3	UEPDC	USLDC	186.15				25.55		.99		
	4-Wire DOT Digital Loop - ONE Zone	3	OLFDC	USEDO	100.13					10	.55		
JNE Port R	Pate												
JIL I OILI	4-Wire DDITS Digital Trunk Por		UEPDC	UDD1T	83.28					10	.99		
	4-Wile DDITS Digital Trunk For		UEFDC	UDDII	03.20					18	.99		
NONRECHI	RRING CHARGES - CURRENTLY COMBINED												
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-	_	UEPDC	USAC4		261.15	134.08			10	.99		
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-	+	OLFDO	COACH		201.10	104.00	1		18			
	DS1 Changes		UEPDC	USAWA		261.15	134.08			10	.99		
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with		OLI DO	OUNTA		201.10	104.00			13			
	Change - Trunk		UEPDC	USAWB		261.15	134.08			10	.99		
	onango mani		OLI DO	COMVE		201.10	104.00			10			
ADDITIONA	AL NRCs						1						
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel						1						
	Activation/Chan - 2-Way Trunk		UEPDC	UDTTA		28.96	28.96			10	.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-		521 00	32117		25.50	20.00			18			
	Way Outward Trunk		UEPDC	UDTTB		28.96	28.96		l l	10	.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan		02.00	05.15		20.00	20.00			- 10	.00		
	Inward Trunk w/out DIC		UEPDC	UDTTC		28.96	28.96			10	.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -		02.00	00110		20.00	20.00			- 10	.00		
	Inward Trunk with DIC		UEPDC	UDTTD		28.96	28.96			10	.99		
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way		02.00	00110		20.00	20.00			- 10	.00		
	DID w User Trans		UEPDC	UDTTE		28.96	28.96			19	.99		
BIPOLAR 8	8 ZERO SUBSTITUTION												
	B8ZS -Superframe Format		UEPDC	CCOSF		0	730			19	.99		
	· ·												
	B8ZS - Extended Superframe Forma		UEPDC	CCOEF		0	730			19	.99		
Alternate N	Mark Inversion												
	AMI -Superframe Format		UEPDC	MCOSF		0	0						
	AMI - Extended SuperFrame Forma		UEPDC	MCOPO		0	0						
Telephone	Number/Trunk Group Establisment Charges												
	Telephone Number for 2-Way Trunk Grou		UEPDC	UDTGX	0								
	Telephone Number for 1-Way Outward Trunk Grou		UEPDC	UDTGY	0						.99		
	Telephone Number for 1-Way Inward Trunk Group Without DI		UEPDC	UDTGZ	0	1	1				.99		
	relephone Number for 1-way inward Trunk Group Without Di							1			.99		
	DID Numbers for each Group of 20 DID Number		UEPDC	ND4	0						.99	1	1
	DID Numbers for each Group of 20 DID Number DID Numbers, Non- consecutive DID Numbers , Per Numbe		UEPDC	ND5	0								
	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos		UEPDC UEPDC	ND5 ND6	0	0	0			19	.99		
	DID Numbers for each Group of 20 DID Number DID Numbers, Non- consecutive DID Numbers , Per Numbe		UEPDC	ND5	0	0	0			19			
	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers		UEPDC UEPDC UEPDC	ND5 ND6	0		-			19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS	Frunk Port	UEPDC UEPDC UEPDC	ND5 ND6 NDV	0 0 0	0	0			19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non- consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminatic	Frunk Port	UEPDC UEPDC UEPDC	ND5 ND6 NDV	0 0 0 55.05	298.18	231.23	0	0	19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminative Interoffice Channel Mileage) - Additional rate per mile - 0-8 mile	Frunk Port	UEPDC UEPDC UEPDC UEPDC UEPDC	ND5 ND6 NDV	0 0 0 55.05 0.45	0 298.18 0	231.23	0	0	19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers, Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS ' Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ	Frunk Port	UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2	0 0 0 55.05 0.45	298.18 0	0 231.23 0 0	0	0	19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Non Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS 'Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminatik Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminatik Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminatik Interoffice Channel Mileage - Additional rate per mile - 9-25 mil	Frunk Port	UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNO4 1LNO2 1LNOB	0 0 0 55.05 0.45 0	298.18 0 0 0	0 231.23 0 0		0	19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminative Channel Mileage - Additional rate per mile 0-8 mile (Facilities Terminative Channel Mileage - Additional rate per mile 0-8 miles (Facilities Terminative Channel Mileage - Additional rate per mile 0-8 miles (Facilities Terminative Channel Mileage - Fixed rate 9-25 miles (Facilities Terminative Channel Mileage - Fixed rate 25+ miles (Facilities T	Frunk Port	UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3	0 0 0 55.05 0.45 0 0.45	298.18 0 0 0 0	0 231.23 0 0 0	0	0	19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers, Per Numbe Reserve Non-Consecutive DID Nos Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS ' Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 0-25 mil Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 9-25 mile 9-25 mil Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per miles - 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per miles - 25+ mil	Frunk Port	UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC	0 0 0 55.05 0.45 0 0.45 0	0 0 298.18 0 0 0 0	0 231.23 0 0 0 0	0	0	19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminatik Interoffice Channel Mileage) - Rixed rate 9-25 miles (Facilities Terminatik Interoffice Channel Mileage) - Rixed rate 9-25 miles (Facilities Terminatik Interoffice Channel Mileage) - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 miles (Interoffice Channel Mileage) - Rixed rate 25+ miles (Facilities Terminatik Interoffice Channel Mileage) - Additional rate per mile - 25+ mile Local Number Portability, per DSO Activate	Frunk Port	UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC LNPCP	0 0 0 55.05 0.45 0 0.45 0 0.45 3.15	298.18 0 0 0 0	0 231.23 0 0 0		0	19 19	.99		
Dedicated I	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers, Per Numbe Reserve Non-Consecutive DID Nos Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS ' Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 0-25 mil Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 9-25 mile 9-25 mil Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per miles - 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per miles - 25+ mil	Frunk Port	UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC	0 0 0 55.05 0.45 0 0.45 0	0 0 298.18 0 0 0 0	0 231.23 0 0 0 0	0	0	19 19	.99		
Dedicated	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminatik Interoffice Channel Mileage) - Rixed rate 9-25 miles (Facilities Terminatik Interoffice Channel Mileage) - Rixed rate 9-25 miles (Facilities Terminatik Interoffice Channel Mileage) - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 miles (Interoffice Channel Mileage) - Rixed rate 25+ miles (Facilities Terminatik Interoffice Channel Mileage) - Additional rate per mile - 25+ mile Local Number Portability, per DSO Activate	Frunk Port	UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC LNPCP	0 0 0 55.05 0.45 0 0.45 0 0.45 3.15	0 0 298.18 0 0 0 0	0 231.23 0 0 0 0	0	0	19 19	.99		
	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS* Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 0-8 mil Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 3-25 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 25+ mil Local Number Portability, per DSO Activate Central Office Termininating Poir	Frunk Port	UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC LNPCP	0 0 0 55.05 0.45 0 0.45 0 0.45 3.15	0 0 298.18 0 0 0 0	0 231.23 0 0 0 0	0	0	19 19	.99		
4-WIRE DS	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS 'Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 0-8 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Terminativ Interoffice Terminativ Interoffice Terminativ	Frunk Port	UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC LNPCP	0 0 0 55.05 0.45 0 0.45 0 0.45 3.15	0 0 298.18 0 0 0 0	0 231.23 0 0 0 0	0	0	19 19	.99		
4-WIRE DS System is 1	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS* Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminatik Interoffice Channel Mileage - Additional rate per mile - 0-8 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 2-5+ mil Local Number Portability, per DS0 Activate Central Office Termininating Poir  1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations	Frunk Port	UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC LNPCP	0 0 0 55.05 0.45 0 0.45 0 0.45 3.15	0 0 298.18 0 0 0 0	0 231.23 0 0 0 0	0	0	19 19	.99		
4-WIRE DS System is 1	DID Numbers for each Group of 20 DID Number DID Numbers, Non-consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID Nos Reserve DID Numbers  DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS 'Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Terminativ Interoffice Channel Mileage - Additional rate per mile - 0-8 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Terminativ Interoffice Terminativ Interoffice Terminativ Interoffice Terminativ	Frunk Port	UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC UEPDC	ND5 ND6 NDV 1LNO1 1LNOA 1LNO2 1LNOB 1LNO3 1LNOC LNPCP	0 0 0 55.05 0.45 0 0.45 0 0.45 3.15	0 0 298.18 0 0 0 0	0 231.23 0 0 0 0	0	0	19 19	.99		

T	DOLL INST		luoi no luo	-	-		1					
4-Wire	DS1 Loop - UNE Zone 1 DS1 Loop - UNE Zone 2	1 UEPMG 2 UEPMG	USLDC 106.04	0	0							$\vdash$
			USLDC 135.15	0	0					+		
4-Wire	DS1 Loop - UNE Zone 3	3 UEPMG	USLDC 186.15	0	0							
												<b></b>
	zation Capacities (D4 Channel Bank Configurations)				_							
	O Channel Capacity - 1 per DS1	UEPMG	VUM24 136.99	0	0							
	O Channel Capacity - 1 per 2 DS1s	UEPMG	VUM48 273.98	0	0							
	O Channel Capacity -1per 4 DS1s	UEPMG	VUM96 547.96	0	0							
	60 Channel Capacity - 1 per 6 DS1s	UEPMG	VUM14 821.94	0	0							
	60 Channel Capacity -1 per 8 DS1s	UEPMG	VUM19 1095.92	0	0							
	60 Channel Capacity - 1 per 10 DS1s	UEPMG	VUM20 1369.9	0	0							
	60 Channel Capacity - 1 per 12 DS1s	UEPMG	VUM28 1643.88	0	0							
	60 Channel Capacity - 1 per 16 DS1s	UEPMG	VUM38 2191.84	0	0							
480 DS	60 Channel Capacity - 1 per 20 DS1s	UEPMG	VUM40 2739.8	0	0							
	60 Channel Capacity -1 per 24 DS1s	UEPMG	VUM57 3287.76	0	0							
672 DS	60 Channel Capacity - 1 per 28 DS1s	UEPMG	VUM67 3835.72	0	0							
Non-Recurring Cha	rges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - Con	version Charge Baser	d on a System									
	configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DSO Port									+	+	
Multiples of this cor	nfiguration functioning as one are considered Add'l after the minimum system c	onfiguration is countr	ed.							+	+	
	Conversion (Currently Combined) with or without BellSouth Allowed Changes	UEPMG	USAC4 0	301.05	16.72				19.99			
	t End User Locations Where 4-Wire DS1 Loop with Channelization with Port Cor			001.00	10.72				10.00	+	+	
	Combined) In Georgia & Tennessee Only	IIDINIALION CUNCINITY E.	Aloto unu							+	+	
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature Activation -									+	+	
	A & TN Only	UEPMG	VUMD4 0	716.36	468.2	149.3	17.71		19.99			
Bipolar 8 Zero Subs		OLI MO	VOIND4 0	7 10.30	400.2	143.3	17.71		13.33	+	+	
Dipolal o Zelo Subs	Sutution									+		<b>├</b>
Clear	Channel Capability Format, superframe - Subsequent Activity Only	UEPMG	CCOSF 0	0	730			] [	19.99		1	1
Ciedi C	Statistics Capability Format, Supername Gubsequent Activity Only	OLF MG	00001 0	U	130		1	+ +	13.33	+	+	
Clear	Channel Capability Format - Extended Superframe - Subsequent Activity Only	UEPMG	CCOEF 0	0	730			] [	19.99		1	1
Alternate Mark Inve		OLI WIG	CCOLI	U	130	+		1	15.55	+	+	$\vdash$
And mare Wark mive	i sivii (niiii)	-+-	<del>-                                     </del>				1	+ +		+	+	
Suporfr	rame Format	UEPMG	MCOSF 0	0	0							
	ed Superframe Format	UEPMG	MCOPO 0	0	0					+		<b> </b>
Exterior	ed Superirame Format	UEPING	MCOPOU	U	U							
Fort and Books As	and the first AME and the second of the seco											
	sociated with 4-Wire DS1 Loop with Channelization with Port											
Exchange Ports		-								+		
	I. C. II. C. II. DOV. T. I. DOV. T. I. D. I. D. I.	HEDDY	UEDOV 4 00						40.00			
Line Sid	de Combination Channelized PBX Trunk Port - Business	UEPPX	UEPCX 1.66	0	0	0	0		19.99			
	I. O. I I Ol I DDV T I D D	LIEPPX	UEDOV 1 as		_	_	_					
	de Outward Channelized PBX Trunk Port - Business	02	UEPOX 1.66	0	0	0	0		19.99			
Line Sid	de Inward Only Channelized PBX Trunk Port without DID	UEPPX	UEP1X 1.66	0	0	0	0		19.99			
						_	_					
	Trunk Side Unbundled Channelized DID Trunk Port	UEPPX	UEPDM 10.97	0	0	0	0		19.99			
Feature Activations	s - Unbundled Loop Concentration											
Feature	e (Service) Activation for each Line Side Port Terminated in D4 Bank	UEPPX	1PQWM 0.77	25.4	13.41	4.17	4.15		19.99			
	e (Service) Activation for each Trunk Side Port Terminated in D4 Bank	UEPPX	1PQWU 0.77	78.15	19.68	59.05	11.54		19.99			
	/ Group Establishment Charges for DID Service	$\longrightarrow$					1					-
	unk Termination (1 per Port)	UEPPX	NDT 0									
	imbers - groups of 20 - Valid all States	UEPPX	ND4 0	0	0				19.99		<u> </u>	<sup> </sup>
	onsecutive DID Numbers - per number	UEPPX	ND5 0	0	0				19.99			
	e Non-Consecutive DID Numbers	UEPPX	ND6 0	0	0							
	e DID Numbers	UEPPX	NDV 0	0	0		[					
Local Number Porta												
Local N	lumber Portability - 1 per port	UEPPX	LNPCP 3.15	0	0							
FEATURES - Vertica											1	
Local Switching Fea	atures Offered with Line Side Ports Only									1		
	tures Available	UEPPX	UEPVF 3.39	0	0		1	1	19.99		1	
7			12		- 1		1	1			+	
					1	1		1			+	
	BINATIONS - MARKET RATES	-					1	+ -		+	+	$\overline{}$
D PORT LOOP COM	PROMISE INDICES			+		+	1	+		+	+	$\overline{}$
D PORT LOOP COM								+ -		+	+	$\overline{}$
D PORT LOOP COM			0.00					+ +		+	+	$\overline{}$
	apply where Rell South is not required to provide unbundled local switching or switch	h norts per ECC and/or				-	1	+		+	+	$\overline{}$
Market Rates shall a	apply where BellSouth is not required to provide unbundled local switching or switch	h ports per FCC and/or	r State Commission rules.					1		1		
Market Rates shall a	lude:											1
Market Rates shall a These scenarios inc 1. Unbundled port/li	lude: oop combinations that are Not Currently Combined in all of the BellSouth states ex	xcept as noted for Geo	orgia and Tennessee.	are with 4 or more	DS0 equivale	ant lines						
Market Rates shall a These scenarios inc 1. Unbundled port/li 2. Unbundled port/li	clude: oop combinations that are Not Currently Combined in all of the BellSouth states ex oop combinations that are Currently Combined or Not Currently Combined in Zone	xcept as noted for Geo	orgia and Tennessee. in BellSouth's region for end use	ers with 4 or more	DS0 equivale	ent lines.	)					
Market Rates shall a These scenarios inc 1. Unbundled port/li 2. Unbundled port/li	lude: oop combinations that are Not Currently Combined in all of the BellSouth states ex	xcept as noted for Geo	orgia and Tennessee. in BellSouth's region for end use	ers with 4 or more Charlotte-Gastoni	DS0 equivale a-Rock Hill);	ent lines. TN (Nashville	).					
Market Rates shall a These scenarios inc 1. Unbundled port/k 2. Unbundled port/l The Top 8 MSAs in	dude: oop combinations that are Not Currently Combined in all of the BellSouth states eo oop combinations that are Currently Combined or Not Currently Combined in Zone BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New	xcept as noted for Geo e 1 of the Top 8 MSAS w Orleans); NC (Greens	orgia and Tennesseb. in BellSouth's region for end use sboro-Winston Salem-Highpoint	Charlotte-Gastoni	a-Rock Hill);	TN (Nashville						
Market Rates shall a These scenarios inc 1. Unbundled port/l 2. Unbundled port/l The Top 8 MSAs in l BellSouth currently i	slude:  oop combinations that are Not Currently Combined in all of the BellSouth states exporper on the state of the states exporper of the states exporper of the states exporper of the states of the states exporper of the states	xcept as noted for Geo e 1 of the Top 8 MSAS w Orleans); NC (Greens	orgia and Tennesseb. in BellSouth's region for end use sboro-Winston Salem-Highpoint	Charlotte-Gastoni	a-Rock Hill);	TN (Nashville		in lieu of the	Market Ra	tes and reserv	res the right to	true-up the b
Market Rates shall a These scenarios inc 1. Unbundled port/li 2. Unbundled port/l The Top 8 MSAs in BellSouth currently i The Market Rate for	dude: oop combinations that are Not Currently Combined in all of the BellSouth states eo oop combinations that are Currently Combined or Not Currently Combined in Zone BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New	xcept as noted for Geo e 1 of the Top 8 MSAS w Orleans); NC (Greens ring Market Rates in this	rgia and Tennesseb. in BellSouth's region for end ussboro-Winston Salem-Highpoint	Charlotte-Gastoni	a-Rock Hill); tes in the Co	TN (Nashville	ion preceding					

Page 19 of 22

2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)								
	pop Combination Rates				07.54				
	2-Wire VG Loop/Port Combo - Zone	1 2			27.54 33.73				
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :	3			42.27				+
UNE Loop F	Paten								
UNE LOOP F	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRX	UEPLX	13.54				+
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRX	UEPLX	19.73				
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRX	UEPLX	28.27				
2-Wire Voic	e Grade Line Port (Res)								
	2-Wire voice unbundled port - residenc		UEPRX	UEPRL	14	90	90	19.99	+
	2-Wire voice unbundled port with Caller ID - re		UEPRX	UEPRC	14	90	90	19.99	
<u></u>	2-Wire voice unbundled port outgoing only - re		UEPRX	UEPRO	14	90	90	19.99	
	2-Wire voice unbundles res, low usage line port with Caller ID (LU!		UEPRX	UEPAP	14	90	90	19.99	
LOCAL NUM	MBER PORTABILITY								
	Local Number Portability (1 per port		UEPRX	LNPCX	0.35				
FEATURES									
	All Features Offered	+	UEPRX	UEPVF	0	0	0		+
ADDITIONA	NRCs NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPRX	USAS2		0	0		+
			<u> </u>						
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)								+
	pop Combination Rates								
	2-Wire VG Loop/Port Combo - Zone	1			27.54				
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :	3			33.73 42.27				
UNE Loop F	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	13.54				
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPBX	UEPLX	19.73				
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	28.27				
	e Grade Line Port (Bus)								
	2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	14	90	90	19.99	+
	2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBX	UEPBC	14	90	90	19.99	
	2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	14	90	90	19.99	
LOCAL MUB	/BER PORTABILITY								
LUCAL NUM	Local Number Portability (1 per port		UEPBX	LNPCX	0.35				
FEATURES									
NONRECUR	RING CHARGES - CURRENTLY COMBINED								
ADDITIONA									
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPBX	USAS2		0	0		+
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)								
UNE Port/Lo	pop Combination Rates								
	2-Wire VG Loop/Port Combo - Zone	1			27.54				1
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :	2 3			33.73 42.27				
UNE Loop F									+
SITE LOOP P	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRG	UEPLX	13.54	<del> </del>		<del>                                     </del>	+ +
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRG	UEPLX	19.73				
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRG	UEPLX	28.27				
	I control of the cont					ĺ	1 1 1		1 1

LOCAL NUMBER F								 					_
Local	Number Portability (1 per port		UEPRG	LNPCP	3.15								
													Т
FEATURES													
NONRECURRING (	CHARGES - CURRENTLY COMBINED												
													_
ADDITIONAL NRC													_
2 Wire	e Loop/Line Side Port Combination - Non feature - Subsequent Activity-					_							
	ecurring					0	0						_
PBX S	Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64		19.99				_
2-WIDE VOICE GD	RADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)												_
2-WINE VOICE GIV	RADE EOOF WITH 2-WIRE EINE FORT (BOS - FBX)												-
UNE Port/Loop Co	ombination Rates												_
	e VG Loop/Port Combo - Zone	1			27.54								_
2-Wire	e VG Loop/Port Combo - Zone	2			33.73								_
	e VG Loop/Port Combo - Zone	3			42.27								Т
													Т
UNE Loop Rates													
	e Voice Grade Loop (SL1) - Zone	1	UEPPX	UEPLX	13.54								Ξ
2-Wire	e Voice Grade Loop (SL1) - Zone	2	UEPPX	UEPLX	19.73								
2-Wire	e Voice Grade Loop (SL1) - Zone	3	UEPPX	UEPLX	28.27								_
													_
2-Wire Voice Grad	de Line Port Rates (BUS - PBX)	$\perp$					1				<b></b>	1	_
	Did Halo Halo Control Control		HESSY		4.0	-			40.00				
Line S	Side Unbundled Combination 2-Way PBX Trunk Port - Bı		UEPPX	UEPPC	14	90	90		19.99			1	_
	Side Unbounded Outstand DDV Terrals Dark Dr.		LIEDDY	LIEDDO	44	00	00		40.00				
Line S	Side Unbundled Outward PBX Trunk Port - Bu Side Unbundled Incoming PBX Trunk Port - Bu	-	UEPPX UEPPX	UEPPO UEPP1	14 14	90 90	90 90		19.99 19.99				_
Line S	e Voice Unbundled PBX LD Terminal Port		UEPPX	UEPP1 UEPLD	14 14	90	90		19.99			-	_
	e Voice Unbundled PBX LD Terminal Port e Voice Unbundled 2-Way Combination PBX Usage Pc	-+	UEPPX	UEPXA	14	90	90		19.99		1	1	-
	e Voice Unbundled 2-way Combination PBX Usage PC e Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	14	90	90		19.99				-
2 ***	2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 -		U=11X	02170		30	30		.0.00		1		-
2-Wire	e Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	14	90	90		19.99				
													_
2-Wire	e Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	14	90	90		19.99				
2-Wire	e Voice Unbundled PBX LD Terminal Switchboard IDD Capable Po		UEPPX	UEPXE	14	90	90		19.99				Ξ
2-Wire	e Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without Ll		UEPPX	UEPXF	14	90	90		19.99				Ξ
					-								
	e Voice Unbundled PBX Kentucky LUD Area Calling Pc		UEPPX	UEPXG	14	90	90		19.99				
2-Wire	e Voice Unbundled PBX Kentucky Premium Calling Pc		UEPPX	UEPXH	14	90	90		19.99				_
2-Wire	e Voice Unbundled 2-Way Kentucky Area Calling Port without LL		UEPPX	UEPXJ	14	90	90		19.99				_
	e Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling												
Port			UEPPX	UEPXL	14	90	90		19.99				_
0.140	a Voice Unbundled 2 Way BRY Hotel/Hoorital Faceaux Bases Calling D		HEDDY	UEPXM	14	90	90		10.00				
2-Wire	e Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P	+	UEPPX	UEPXM	14	90	90		19.99		1	1	_
2-Wire Port	e Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling		UEPPX	UEPXO	14	90	90		19.99				
7.Mire	e Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX	UEPXS	14	90	90		19.99				_
Z-VVITE	C VOICE CITEMINED 1-VVay Outgoing F DA Wedsuled Ft	-+	ULFFA	ULFAS	14	30	30		10.00		1	1	_
LOCAL NUMBER F	PORTABILITY						1					<b>†</b>	-
	Number Portability (1 per port		UEPPX	LNPCP	3.15							1	-
Local			JE/ I A	2 01	50							1	-
FEATURES							İ						-
													_
NONRECURRING (	CHARGES - CURRENTLY COMBINED												_
													Ξ
ADDITIONAL NRC													_
	re Voice Grade Loop/ Line Port Combination - Subseque		UEPPX	USAS2		0	0						Ξ
	e Loop/Line Side Port Combination - Non feature - Subsequent Activity-										· -		
	ecurring					0	0						_
PBX S	Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64		19.99				_
	AADE LOOP WITH A WIDE ANALOO LINE CONTROL			$\perp$									_
z-WIRE VOICE GR	RADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT	$\perp$					1				<b></b>	1	_
		$\perp$					1				<b></b>	1	_
UNE Port/Loop Co		$\perp$					1				<b></b>	1	_
	e VG Coin Port/Loop Combo – Zone 1				27.54							1	_
2-Wire	e VG Coin Port/Loop Combo – Zone 2			$\perp$	33.73								_
2-Wire	e VG Coin Port/Loop Combo – Zone 3				42.27		-					1	_
INE Loop Date:							-					1	_
UNE Loop Rates	- Vaine Conda Lana (CLA) - Zana		LIEBOO	HERLY	10.51		-					1	_
2-Wire	e Voice Grade Loop (SL1) - Zone		UEPCO UEPCO	UEPLX	13.54 19.73								_
	e Voice Grade Loop (SL1) - Zone e Voice Grade Loop (SL1) - Zone	-+-+	UEPCO	UEPLX	19.73								_
2-Wire	e voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	28.27		1		-	-		-	_

	2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)								
		UEPCO	UEPRF	14	90	90	19.99		
	2-Wire Coin 2-Way with Operator Screening (AL, KY)	UEPCO	UEPRE	14	90	90	19.99		
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL,								
	KY, LA, MS, SC)	UEPCO	UEPRA	14	90	90	19.99		
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY)	UEPCO	UEPKA	14	90	90	33.67	7.88	
	2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, &								
	Local (AL, KY, LA, MS)	UEPCO	UEPCD	14	90	90	19.99		
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)	UEPCO	UEPRN	14	90	90	19.99		
	2-Wire Coin Outward with Operator Screening and 011Blocking (GA, KY, MS)	UEPCO	UEPRJ	14	90	90	19.99		
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL,								
	KY, LA, MS)	UEPCO	UEPRH	14	90	90	19.99		
	2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local								
	(AL, KY, LA, MS)	UEPCO	UEPCN	14	90	90	19.99		$\dashv$
LOCAL NU	JMBER PORTABILITY								=
	Local Number Portability (1 per port	UEPCO	LNPCX	0.35					_
	JRRING CHARGES - CURRENTLY COMBINED								
ADDITION	AL NRCs								
	2-Wire Voice Grade Loop/ Line Port Combination - Subseque	UEPCO	USAS2		0	0			
									-
									$\neg$

Page 22 of 22 Version 2Q01: 08/30/01

GORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
								Nonre	curring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge Manual S Order v Electron Disc Ad
														sconnect			
							Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
		shown in the sections for stand-alone loops or loops as part of a combination refers to terconnection.bellsouth.com/become_a_clec/html/interconnection.htm	Geogra	phical	ly Deaveraged UNE	Zones.	To view Geogr	aphically Deave	raged UNE Zor	e Designati	ons by Centra	Il Office, ref	er to Interne	et Website:	Ī		ı
NDI ED	EVCHANC	E ACCESS LOOP															
NDLED	EXCHAING	E ACCESS LOOP															
2	-WIRE AND	ALOG VOICE GRADE LOOP															
	- WIIVE AND	2-Wire Analog Voice Grade Loop - Service Level 1- Zone		1	UEANL	UEAL2	14.96	40.69	29.96	11.48	3.36			18.14	8.06	11.41	11.4
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone		2	UEANL	UEAL2	25.69	40.69	29.96	11.48	3.36			18.14	8.06	11.41	11.4
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone		3	UEANL	UEAL2	52.47	40.69	29.96	11.48	3.36			18.14	8.06	11.41	11.4
<del></del>		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zon		2	UEPSR, UEPSB UEPSR, UEPSB		14.96 25.69	40.69 40.69	29.96 29.96	11.48 11.48	3.36		<b>†</b>	18.14 18.14	8.06	11.41 11.41	11.4
-		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zoni					52.47			11.48	3.36		1	18.14	8.06	11.41	11.4 11.4
-		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zoni		3	UEPSR, UEPSB UEANL	UEALS	52.47	40.69	29.96	11.46	3.36		1	10.14	8.06	11.41	11.4
		Engineering Information Document (E			UEANL	1		28.72	28.72			1	1				l
		Manual Order Coordination for LIVI. St. to (			I I E A A II	LIEANA		24.0	24.0								
		Manual Order Coordination for UVL-SL1s (per loop			UEANL	UEAMC		34.9	34.9		1			1	<b> </b>		-
		0-10			115.440	00001		00.77	00.77								
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR			UEANL	OCOSL		32.77	32.77								-
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling															
			-	1	LIEA	LIEALO	47.05	00.00	74.70	20.72	40.07			40.44	0.00	44.44	44
		Zone 1		-	UEA	UEAL2	17.65	99.69	74.73	28.73	18.87			18.14	8.06	11.41	11.
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling	-	2			30.32	00.00	74.70	00.70	40.07			40.44	0.00	44.44	
		Zone 2		2	UEA	UEAL2	30.32	99.69	74.73	28.73	18.87			18.14	8.06	11.41	11.
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling	-	_													
		Zone 3		3	UEA	UEAL2	61.93	99.69	74.73	28.73	18.87			18.14	8.06	11.41	11.
		Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL		32.77									
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo	ne														
		1		1	UEA	UEAR2	17.65	99.69	74.73	28.73	18.87			18.14	8.06	11.41	11.4
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo	ne														
		2		2	UEA	UEAR2	30.32	99.69	74.73	28.73	18.87			18.14	8.06	11.41	11.4
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo	ne														
		3		3	UEA	UEAR2	61.93	99.69	74.73	28.73	18.87			18.14	8.06	11.41	11.4
		Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL		32.77									
4	-WIRE ANA	ALOG VOICE GRADE LOOP															
		4-Wire Analog Voice Grade Loop - Zone		1	UEA	UEAL4	24.36	198.1	163.26	74.27	39.44			18.14	8.06	11.41	11.
		4-Wire Analog Voice Grade Loop - Zone		2	UEA	UEAL4	41.85	198.1	163.26	74.27	39.44			18.14	8.06	11.41	11.
		4-Wire Analog Voice Grade Loop - Zone		3	UEA	UEAL4	85.47	198.1	163.26	74.27	39.44			18.14	8.06	11.41	11.
		Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL		32.77									
2	-WIRE ISD	N DIGITAL GRADE LOOP															
		2-Wire ISDN Digital Grade Loop - Zone		1	UDN	U1L2X	21.15	223.27	172.63	74.27	39.44		İ	18.14	8.06	11.41	11.
		2-Wire ISDN Digital Grade Loop - Zone		2	UDN	U1L2X	36.32	223.27	172.63	74.27	39.44		İ	18.14	8.06	11.41	11.
-+		2-Wire ISDN Digital Grade Loop - Zone		3	UDN	U1L2X	74.19	223.27	172.63	74.27	39.44			18.14	8.06	11.41	11.
		2 Will look bigital olddo 2005 2010			ODIN	UTLLA	7 11.10	ELU.E.	112.00	7 1.121	00.11			10.11	0.00		
		Order Coordination For Specified Conversion Time (per LS			UDN	OCOSL		32.77									
-		2.22. 22. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2.			3314	JUGGE		U	1		1			1	1		
2	-WIRF Univ	versal Digital Channel (UDC) COMPATIBLE LOOP							1		1			1	1		
-		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		1	UDC	UDC2X	28.55	233.18	158.33	105.01	20.41		1	18.14	8.06	11.41	11.
-+		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		2	UDC	UDC2X	37.12	233.18	158.33	105.01	20.41		1	18.14	8.06	11.41	11.
-+		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		3	UDC	UDC2X	51.88	233.18	158.33	105.01	20.41		<b>†</b>	18.14	8.06	11.41	11.
-+		2 17110 Chinasan Digital Charino (ODO) Compatible Loop - Zone		J	550	ODOZA	01.00	200.10	100.00	100.01	20.41			10.14	0.00	11.991	- 11
_	WIDE AC	/MMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP				1						1	1				<b>-</b>
2	-vvike ASY	MINIETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP				-											-
		A WIDE ADVANCEDION DIGITAL OUROODIDED LINE (ADOL) CONSTRUCTOR															1
		2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOO				1	-				-	<b> </b>	1	1			<del>                                     </del>
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -															
_		Zone 1		1	UAL	UAL2X	11.9	343.13	310.03	72.54	39.42			18.14	8.06	11.41	11.
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -															
		Zone 2		2	UAL	UAL2X	20.43	343.13	310.03	72.54	39.42			18.14	8.06	11.41	11.
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -															1
		Zone 3		3	UAL	UAL2X	41.73	343.13	310.03	72.54	39.42			18.14	8.06	11.41	11.4
			_	1 7		1							1		1		
		Order Coordination for Specified Conversion Time (per LS		1	UAL	OCOSL	1	32.77	1	1	1	1	1	1	1		1

2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1		1	UAL	UAL2W	11.9	204.74	129.02	100.41	15.81	18.14	8.06	11.41	1
2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -		2	UAL	UAL2W	20.43	204.74	129.02	100.41	15.81	18.14	8.06	11.41	1
Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservator -								100.41	10.01	10.14	6.00		
Zone 3		3	UAL	UAL2W	41.73	204.74	129.02	100.41	15.81	18.14	8.06	11.41	-
Order Coordination for Specified Conversion Time (per LS)			UAL	OCOSL		32.77							
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	++	+											
	+ +	_		-									
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOO 2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation		_		-									
Zone 1		1	UHL	UHL2X	8.97	343.13	310.03	72.54	39.42	18.14	8.06	11.41	
2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation Zone 2		2	UHL	UHL2X	15.41	343.13	310.03	72.54	39.42	18.14	8.06	11.41	
2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation	-												
Zone 3	+	3	UHL	UHL2X	31.48	343.13	310.03	72.54	39.42	18.14	8.06	11.41	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		32.77							
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 1	†	1	UHL	UHL2W	8.97	222.04	146.33	100.41	15.81	18.14	8.06	11.41	
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation		2											
Zone 2 2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation		2	UHL	UHL2W	15.41	222.04	146.33	100.41	15.81	18.14	8.06	11.41	+
Zone 3		3	UHL	UHL2W	31.48	222.04	146.33	100.41	15.81	18.14	8.06	11.41	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		32.77							
4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	++	+											
4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation	on -		-		-								
Zone 1  4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation	00 -	1	UHL	UHL4X	12.97	361.45	328.35	72.54	39.42	18.14	8.06	11.41	
Zone 2		2	UHL	UHL4X	21.76	361.45	328.35	72.54	39.42	18.14	8.06	11.41	
4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation Zone 3		3	UHL	UHL4X	44.44	361.45	328.35	72.54	39.42	18.14	8.06	11.41	
							020.00	72.01	00.12	10.11	0.00		
Order Coordination for Specified Conversion Time (per LSI  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation		_	UHL	OCOSL		32.77							
Zone 1		1	UHL	UHL4W	12.97	279.17	203.45	111.45	20.98	18.14	8.06	11.41	
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 2		2	UHL	UHL4W	21.76	279.17	203.45	111.45	20.98	18.14	8.06	11.41	
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation													
Zone 3	+ + :	3	UHL	UHL4W	44.44	279.17	203.45	111.45	20.98	18.14	8.06	11.41	
Order Coordination for Specified Conversion Time (per LS)		_	UHL	OCOSL		32.77							
4-WIRE DS1 DIGITAL LOOP	+ +			_									
4-Wire DS1 Digital Loop - Zone		1	USL	USLXX	56.32	410.38	255.48	92.35	38.44	18.14	8.06	11.41	
4-Wire DS1 Digital Loop - Zone : 4-Wire DS1 Digital Loop - Zone :		3	USL	USLXX	96.73 197.57	410.38 410.38	255.48 255.48	92.35 92.35	38.44 38.44	18.14 18.14	8.06 8.06	11.41 11.41	
	1 1	-			107.07		200.40	32.00	50.44	10.14	0.00	11.41	
Order Coordination for Specified Conversion Time (per LS		-	USL	OCOSL		33.05							
4-WIRE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP					-								
4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	27.5 47.24	333.28	230.5 230.5	87.99	44.24	18.14	8.06	11.41	
4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital 19.2 Kbps		3	UDL UDL	UDL19 UDL19	96.48	333.28 333.28	230.5	87.99 87.99	44.24 44.24	18.14 18.14	8.04 8.06	11.41 11.41	
4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital Loop 56 Kbps - Zone		1	UDL	UDL56	27.5	333.28	230.5	87.99	44.24	18.14	8.06	11.41	
4 Wire Unbundled Digital Loop 56 Kbps - Zone		2	UDL	UDL56	47.24	333.28	230.5	87.99	44.24	18.14	8.06	11.41	+
4 Wire Unbundled Digital Loop 56 Kbps - Zone		3	UDL	UDL56	96.48	333.28	230.5	87.99	44.24	18.14	8.06	11.41	
4 Wile Official Coop of Napa Zone			ODL	ODESO	30.40	000.20	200.0	07.55	77.27	10.14	0.00	11.41	
Order Coordination for Specified Conversion Time (per LS			UDL	OCOSL		32.77							
4 Wire Unbundled Digital Loop 64 Kbps - Zone		1	UDL	UDL64	27.5	333.28	230.5	87.99	44.24	18.14	8.06	11.41	
4 Wire Unbundled Digital Loop 64 Kbps - Zone		2	UDL	UDL64	47.24	333.28	230.5	87.99	44.24	18.14	8.06	11.41	
4 Wire Unbundled Digital Loop 64 Kbps - Zone		3	UDL	UDL64	96.48	333.28	230.5	87.99	44.24	18.14	8.06	11.41	4
Order Coordination for Specified Conversion Time (per LS			UDL	OCOSL		32.77							
		$\pm$											H
2-WIRE Unbundled COPPER LOOP  2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility	+			$\bot$									H
		1	UCL	UCLPB	15.97	283.15	163.59	120.02	22.34	19.99	19.99	19.99	
reservation - Zone 1													
2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2		2	UCL	UCLPB	17.99	283.15	163.59	120.02	22,34	19.99	19.99	19.99	
2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility		3	UCL	UCLPB	17.99 18.18	283.15 283.15	163.59 163.59	120.02	22.34	19.99	19.99	19.99	

											,				
	Order Coordination for Unbundled Copper Loops (per loc	1		UCL	UCLMC	ļ	34.9	34.9							
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility														
	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility	$\vdash$	1	UCL	UCLPW	15.97	202.88	127.16	100.41	15.81		19.99	19.99	19.99	19.99
	reservation - Zone 2		2	UCL	UCLPW	17.47	202.88	127.16	100.41	15.81		19.99	19.99	19.99	19.99
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone (	ı	3	UCL	UCLPW	18.18	202.88	127.16	100.41	15.81		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loop)		3			10.10	202.00	127.10	100.41	13.01		15.55	13.33	15.55	15.55
				UCL	UCLMC		34.9	34.9							
	2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone '	ı	1	UCL	UCL2L	42.85	269.99	150.43	120.02	22.34		19.99	19.99	19.99	19.99
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	i						.=							
	reservation - Zone 2 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		2	UCL	UCL2L	57.79	269.99	150.43	120.02	22.34		19.99	19.99	19.99	19.99
	reservation - Zone :		3	UCL	UCL2L	100.8	269.99	150.43	120.02	22.34		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loop)	ı		UCL	UCLMC	ļ	34.9	34.9							
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility														
	reservation - Zone ' 2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility	<del>                                     </del>	1	UCL	UCL2W	42.85	189.73	114.01	100.41	15.81		19.99	19.99	19.99	19.99
	reservation - Zone 2		2	UCL	UCL2W	57.79	189.73	114.01	100.41	15.81		19.99	19.99	19.99	19.99
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone (	1	3	UCL	UCL2W	100.8	189.73	114.01	100.41	15.81		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc		3	UCL	UCLMC	100.0	34.9	34.9	100.41	10.01		19.99	19.99	19.99	19.99
		$\vdash$										<del>                                     </del>	$\vdash$	1	
	2-Wire Unbundled Copper Loop - Non-Designed Zone	ı	1	UEQ	UEQ2X	11.01	44.69	22.4	25.65	7.06		18.14	8.06	11.41	11.41
	2 Wire Unbundled Copper Loop - Non-Designed - Zone		2	UEQ	UEQ2X	12.67	44.69	22.4	25.65	7.06		18.14	8.06	11.41	11.41
	2 Wire Unbundled Copper Loop - Non-Designed - Zone Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per lo	$\vdash$	3	UEQ UEQ	UEQ2X USBMC	20.22	44.69 34.9	22.4 34.9	25.65	7.06		18.14	8.06	11.41	11.41
	Engineering Information Documer	$\vdash$		UEQ	OSBIVIC		28.72	28.72					<del>                                     </del>		
	Loop Testing - Basic 1st Half Hou			UEQ	URET1		78.92	78.92							
	Loop Testing - Basic Additional Half Hou			UEQ	URETA		23.33	23.33				+			
		-													
4-WIRE CO	PPER LOOP											<del> </del>	<del></del>		
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation -														
	Zone 1		1	UCL	UCL4S	28.68	331.51	211.94	132.43	27.97		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 2	1	2	UCL	UCL4S	26.01	331.51	211.94	132.43	27.97		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation -		2	UCL	UCL45	20.01	331.31	211.94		21.91		19.99	19.99	19.99	19.99
	Zone 3	$\longrightarrow$	3	UCL	UCL4S	27.9	331.51	211.94	132.43	27.97		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc	$\longrightarrow$		UCL	UCLMC		34.9	34.9					<b></b>		
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 1	1	1	UCL	UCL4W	28.68	251.24	175.52	111.45	20.98		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -	i I													
	Zone 2  4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -	1	2	UCL	UCL4W	26.01	251.24	175.52	111.45	20.98		19.99	19.99	19.99	19.99
	Zone 3		3	UCL					111.45	20.98			1	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc			UCL	UCL4W	27.9	251.24	175.52		20.90		19.99	19.99		
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			UCL	UCL4W UCLMC	27.9	251.24 34.9	175.52 34.9	111.43	20.96		19.99	19.99	19.99	
			1	UCL	UCLMC		34.9	34.9							10.00
	reservation - Zone 1		1			76.23			132.43	27.97		19.99	19.99	19.99	19.99
	reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ;		1 2	UCL	UCLMC		34.9	34.9							
	reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		2	UCL UCL UCL	UCL4L UCL4L	76.23 112.23	34.9 318.35 318.35	34.9 198.79 198.79	132.43	27.97 27.97		19.99	19.99 19.99	19.99	19.99
	reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone (			UCL UCL UCL	UCL4L UCL4L UCL4L	76.23	34.9 318.35 318.35 318.35	34.9 198.79 198.79 198.79	132.43	27.97		19.99	19.99	19.99	19.99
	reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ' 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		2	UCL UCL UCL	UCL4L UCL4L UCL4L UCL4L UCLMC	76.23 112.23	34.9 318.35 318.35	34.9 198.79 198.79	132.43	27.97 27.97		19.99	19.99 19.99	19.99	19.99
	reservation - Zone '		2	UCL UCL UCL	UCL4L UCL4L UCL4L	76.23 112.23	34.9 318.35 318.35 318.35	34.9 198.79 198.79 198.79	132.43	27.97 27.97		19.99	19.99 19.99	19.99	19.99
	reservation - Zone '  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone '  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone '  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility		3	UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4C	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09	34.9 198.79 198.79 198.79 34.9 162.37	132.43 132.43 132.43	27.97 27.97 27.97 20.98		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99
	reservation - Zone '		2 3 1 2	UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4L UCLMC	76.23 112.23 138.01	34.9 318.35 318.35 318.35 34.9	34.9 198.79 198.79 198.79 34.9	132.43 132.43 132.43	27.97 27.97 27.97		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99
	reservation - Zone '  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone (  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone (  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (		3	UCL UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09	34.9 198.79 198.79 198.79 34.9 162.37 162.37	132.43 132.43 132.43	27.97 27.97 27.97 20.98		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99
	reservation - Zone '		2 3 1 2	UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09	34.9 198.79 198.79 198.79 34.9 162.37	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
	reservation - Zone '  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone (  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone (  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (		2 3 1 2	UCL UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09	34.9 198.79 198.79 198.79 34.9 162.37 162.37	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ;  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ;  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ;  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ;  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone ;  Order Coordination for Unbundled Copper Loops (per loc		3 1 2 3	UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09	34.9 198.79 198.79 198.79 34.9 162.37 162.37	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal	to	3 1 2 3	UCL UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O UCL4O UCL4O UCL4O	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09 34.9	34.9 198.79 198.79 198.79 34.9 162.37 162.37 34.9	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft	to	3 1 2 3	UCL UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCLMC UCL4L UCL4L UCL4C UCL4O UCL4O UCL4O UCL4O UCL4O UCL4O UCLMC	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09 34.9	34.9 198.79 198.79 198.79 34.9 162.37 162.37 34.9	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal		3 1 2 3	UCL UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O UCL4O UCL4O UCL4O UCLMC	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09 34.9	34.9 198.79 198.79 198.79 34.9 162.37 162.37 34.9	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18		3 1 2 3	UCL UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCLMC UCL4L UCL4L UCL4C UCL4O UCL4O UCL4O UCL4O UCL4O UCL4O UCLMC	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09 34.9	34.9 198.79 198.79 198.79 34.9 162.37 162.37 34.9	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone '		3 1 2 3	UCL UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O UCL4O UCL4O UCLMC	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 34.9 65.11 341.16 65.11	34.9 198.79 198.79 198.79 34.9 162.37 162.37 34.9 65.11 341.16	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal 18k ft  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18		3 1 2 3	UCL  UCL  UCL  UCL  UCL  UCL  UCL  UCL	UCLMC UCL4L UCL4L UCL4C UCL4O UCL4O UCL4O UCL4O UCL4O UCL4O UCLMC	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 238.09 34.9	34.9 198.79 198.79 198.79 34.9 162.37 162.37 34.9	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99
P MODIFICATION	reservation - Zone '		3 1 2 3	UCL UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCL4L UCL4L UCL4L UCL4C UCL4O UCL4O UCL4O UCL4O UCLMC	76.23 112.23 138.01 76.23	34.9 318.35 318.35 318.35 34.9 238.09 238.09 34.9 65.11 341.16 65.11	34.9 198.79 198.79 198.79 34.9 162.37 162.37 34.9 65.11 341.16	132.43 132.43 132.43 111.45	27.97 27.97 27.97 20.98		19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.9 19.9 19.9

PS																
Sub-Loop I	Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L			UEANL	USBSA		612.58	612.58					18.14	8.06	11.41	1
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L			UEANL	USBSB		45.22	45.22					18.14	8.06	11.41	1
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I			UEANL	USBSC		379.35	379.35					18.14	8.06	11.41	1
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I	1		UEANL	USBSD		110.17	110.17					18.14	8.06	11.41	1
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zon€		1	UEANL	USBN2	10.23	131.45	61.84	90.39	13.37			18.14	8.06	11.41	1
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		2	UEANL	USBN2	13.92	131.45	61.84	90.39	13.37			18.14	8.06	11.41	1
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		3	UEANL	USBN2	20.22	131.45	61.84	90.39	13.37			18.14	8.06	11.41	
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		32.77	32.77								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		1	UEANL	USBN4	15.14	157.89	88.29	100.74	18.38			18.14	8.06	11.41	
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		2	UEANL	USBN4	19.64	157.89	88.29	100.74	18.38			18.14	8.06	11.41	٠.
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		3	UEANL	USBN4	24	157.89	88.29	100.74	18.38			18.14	8.06	11.41	١.
	Order Coordination for Unburgled Cub Large and all large and		3	UEANL	USBMC	24	32.77	32.77	100.74	10.30			10.14	0.00	11.41	
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	-	_													
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC			UEANL	USBR2	3.18	105.91	36.3	90.39	13.37			18.14	8.06	11.41	
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		32.77	32.77								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC	1		UEANL	USBR4	7.2	118.39	48.77	100.74	18.38			18.14	8.06	11.41	1
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		32.77	32.77								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone	- 1	1	UEF	UCS2X	8.41	131.45	61.84	90.39	13.37			18.14	8.06	11.41	
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone	i			UCS2X	11	131.45	61.84	90.39	13.37			18.14	8.06	11.41	
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone	T i	3		UCS2X	13.5	131.45	61.84	90.39	13.37			18.14	8.06	11.41	٠.
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC	. 5.0	32.77	32.77	23.00		+	<del> </del>		00		1
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone			UEF	UCS4X	10.13	157.89	88.29	100.74	18.38	<b>+</b>	-	18.14	8.06	11.41	
-		++									-					
-	4 Wire Copper Unbundled Sub-Loop Distribution - Zone	+ !		UEF	UCS4X	13.06	157.89	88.29	100.74	18.38	-		18.14	8.06	11.41	
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone			UEF	UCS4X	17.02	157.89	88.29	100.74	18.38			18.14	8.06	11.41	
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		32.77	32.77								1
Sub-Loop I	Feeder															
				UEA,												
				UDN,UCL,UDL,UD												
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-			С	USBFW		612.58									
				UEA,												
				UDN,UCL,UDL,UD												
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u			C	USBFX		45.22	45.22								
	USL Feeder DS1 Set-up per Cross Box location - per 25 pair set-u		+	USL	USBFZ		568.98	11.3								
			1	UEA		10.98	184.76		108.27	20.00			19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zon		2	UEA	USBFA	13.31	184.76	111.8	108.27	26.66			19.99	19.99	19.99	-
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zong				USBFA			111.0		26.66						
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zon		3	UEA	USBFA	21.51	184.76	111.8	108.27	26.66			19.99	19.99	19.99	
	Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		32.77									
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni		1	UEA	USBFB	10.98	184.76	111.8	108.27	26.66			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni		2	UEA	USBFB	13.31	184.76	111.8	108.27	26.66			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zon		3	UEA	USBFB	21.51	184.76	111.8	108.27	26.66			19.99	19.99	19.99	
	Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		32.77									
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon		1	UEA	USBFC	10.98	184.76	111.8	108.27	26.66			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon		2	UEA	USBFC	13.31	184.76	111.8	108.27	26.66			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Z	one		OLA	005.0	10.01	1010	11110	100.27	20.00			10.00	10.00	10.00	
	onbundled Sub-Loop reeder Loop, 2 write Analog Neverse Battery, voice Grade - 2	une	3	UEA	USBFC	21.51	184.76	111.8	108.27	26.66			19.99	19.99	19.99	
	3		3	UEA	USBFC	21.01	104.70	111.0	100.27	20.00			19.99	19.99	19.99	1
	Order Consideration For Considerat Consumation Time and C	1		1154	ocosl		20.77									
	Order Coordination For Specified Conversion Time, per LS	1	+	UEA		05.00	32.77	100 17	10176	04.05			10.00	40.00	40.00	1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zoni	1	1	UEA	USBFD	25.83	213.33	138.47	124.72	34.25			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone		2	UEA	USBFD	26.93	213.33	138.47	124.72	34.25			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zoni		3	UEA	USBFD	25.51	213.33	138.47	124.72	34.25			19.99	19.99	19.99	
		1 -	1 7		1 7			l			Ι Τ		٦		l	1 -
	Control Contro	1		UEA				I							1	
	Order Coordination For Specified Conversion Time, Per LS				OCOSL		32.77			34.25			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni		1	UEA	USBFE	25.83	213.33	138.47	124.72					19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni			UEA	USBFE		213.33		124.72 124.72				19.99			
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon		2	UEA UEA	USBFE USBFE	36.93	213.33 213.33	138.47	124.72	34.25					19 99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni			UEA	USBFE		213.33		12 11.7 2				19.99 19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni		2	UEA UEA UEA	USBFE USBFE USBFE	36.93	213.33 213.33 213.33	138.47	124.72	34.25					19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Order Coordination For Specified Conversion Time, Per L\$		3	UEA UEA UEA UEA	USBFE USBFE USBFE OCOSL	36.93 25.51	213.33 213.33 213.33 32.77	138.47 138.47	124.72 124.72	34.25 34.25			19.99	19.99		
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone		3	UEA UEA UEA UEA UDN	USBFE USBFE USBFE OCOSL USBFF	36.93 25.51 19.39	213.33 213.33 213.33 213.33 32.77 211.05	138.47 138.47 136.19	124.72 124.72 110.52	34.25 34.25 25.91			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni Order Coordination For Specified Conversion Time, Per L\$ Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone		2 3 1 2	UEA UEA UEA UEA UDN UDN	USBFE USBFE USBFE OCOSL USBFF USBFF	36.93 25.51 19.39 24.12	213.33 213.33 213.33 213.33 32.77 211.05 211.05	138.47 138.47 136.19 136.19	124.72 124.72 124.72 110.52	34.25 34.25 25.91 25.91			19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone		3	UEA UEA UEA UEA UDN	USBFE USBFE USBFE OCOSL USBFF	36.93 25.51 19.39	213.33 213.33 213.33 213.33 32.77 211.05	138.47 138.47 136.19	124.72 124.72 110.52	34.25 34.25 25.91			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone		2 3 1 2	UEA UEA UEA UEA UDN UDN UDN	USBFE USBFE USBFE OCOSL USBFF USBFF USBFF	36.93 25.51 19.39 24.12	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05	138.47 138.47 136.19 136.19	124.72 124.72 124.72 110.52	34.25 34.25 25.91 25.91			19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Order Coordination For Specified Conversion Time, Per L\$ Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per L\$		1 2 3	UEA UEA UEA UDN UDN UDN UDN	USBFE USBFE USBFE OCOSL USBFF USBFF USBFF	36.93 25.51 19.39 24.12 32.5	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05 32.77	138.47 138.47 136.19 136.19 136.19	124.72 124.72 124.72 110.52 110.52 110.52	34.25 34.25 25.91 25.91 25.91			19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone		2 3 1 2	UEA UEA UEA UDN UDN UDN UDN	USBFE USBFE USBFE OCOSL USBFF USBFF USBFF	36.93 25.51 19.39 24.12	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05	138.47 138.47 136.19 136.19	124.72 124.72 124.72 110.52	34.25 34.25 25.91 25.91			19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		2 3 1 2 3	UEA UEA UEA UDN UDN UDN UDN	USBFE USBFE USBFE USBFF USBFF USBFF USBFF USBFF	36.93 25.51 19.39 24.12 32.5	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05 32.77	138.47 138.47 136.19 136.19 136.19	124.72 124.72 124.72 110.52 110.52 110.52	34.25 34.25 25.91 25.91 25.91 25.91			19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatib) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatib)		1 2 3 1 2 3	UEA UEA UEA UDA UDN UDN UDN UDN UDN UDN UDN UDN UDC UDC	USBFE USBFE USBFE OCOSL USBFF USBFF USBFF OCOSL USBFS USBFS	36.93 25.51 19.39 24.12 32.5 19.39 24.12	213.33 213.33 213.33 213.33 32.77 211.05 211.05 32.77 211.05 211.05	138.47 138.47 136.19 136.19 136.19 136.19	124.72 124.72 124.72 110.52 110.52 110.52	34.25 34.25 25.91 25.91 25.91 25.91 25.91			19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		1 2 3 1 2 3 1 2 3	UEA UEA UEA UDA UDN UDN UDN UDN UDN UDN UDN UDD UDC UDC	USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFS USBFS	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5	213.33 213.33 213.33 213.33 32.77 211.05 211.05 32.77 211.05 211.05 211.05 211.05	138.47 138.47 136.19 136.19 136.19 136.19 136.19 136.19	124.72 124.72 110.52 110.52 110.52 110.52 110.52 110.52	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 25.91			19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		2 3 1 2 3 1 2 3 1	UEA UEA UEA UDN UDN UDN UDN UDN UDN UDD UDC UDC UDC UDC UDC USL	USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5 64.62	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05	138.47 138.47 138.47 136.19 136.19 136.19 136.19 136.19 136.19	124.72 124.72 124.72 110.52 110.52 110.52 110.52 110.52 124.72	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 34.25			19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		2 3 3 1 2 3 1 2 3 1 2	UEA UEA UEA UDA UDN UDN UDN UDN UDC UDC UDC USL USL	USBFE USBFE USBFE OCOSL USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFG USBFG	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5 64.62 92.63	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05 211.05 211.05 211.05 211.05 201.93	138.47 138.47 136.19 136.19 136.19 136.19 136.19 136.19 127.07	124.72 124.72 110.52 110.52 110.52 110.52 110.52 110.52 124.72	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 34.25 34.25			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		2 3 1 2 3 1 2 3 1	UEA UEA UEA UDN UDN UDN UDN UDN UDN UDD UDC UDC UDC UDC UDC USL	USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5 64.62	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05	138.47 138.47 138.47 136.19 136.19 136.19 136.19 136.19 136.19	124.72 124.72 124.72 110.52 110.52 110.52 110.52 110.52 124.72	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 34.25			19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		2 3 3 1 2 3 1 2 3 1 2	UEA UEA UEA UDN UDN UDN UDN UDN UDN UDC UDC UDC UDC UDC USL USL	USBFE USBFE USBFE OCOSL USBFF USBFF USBFS USBFS USBFS USBFS USBFG USBFG	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5 64.62 92.63	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 201.93 201.93	138.47 138.47 136.19 136.19 136.19 136.19 136.19 136.19 127.07	124.72 124.72 110.52 110.52 110.52 110.52 110.52 110.52 124.72	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 34.25 34.25			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		1 2 3 1 2 3 1 2 3 1 2 3	UEA UEA UEA UDA UDN UDN UDN UDN UDN UDC UDC UDC UDC UDC USL USL USL	USBFE USBFE USBFF OCOSL USBFF USBFF OCOSL USBFS USBFS USBFS USBFG USBFG USBFG USBFG	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5 64.62 92.63 309.79	213.33 213.33 213.33 32.77 211.05 211.05 211.05 211.05 211.05 211.05 211.05 201.93 201.93 201.93	138.47 138.47 138.47 136.19 136.19 136.19 136.19 136.19 127.07 127.07	124.72 124.72 110.52 110.52 110.52 110.52 110.52 110.52 124.72 124.72	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 25.91 34.25 34.25 34.25			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 5-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 5-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 5-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 5-Wire Copper Loop - Zone		2 3 1 2 3 1 2 3 1 2 3	UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL USL UCL	USBFE USBFE USBFF OCOSL USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG USBFG USBFG	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5 64.62 92.63 309.79	213.33 213.33 213.33 213.33 32.77 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05 211.05	138.47 138.47 138.47 136.19 136.19 136.19 136.19 136.19 127.07 127.07	124.72 124.72 110.52 110.52 110.52 110.52 110.52 110.52 124.72 124.72 124.72	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 34.25 34.25 34.25			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		1 2 3 1 2 3 1 2 3 1 2 3	UEA UEA UEA UDA UDN UDN UDN UDN UDN UDC UDC UDC UDC UDC USL USL USL	USBFE USBFE USBFF OCOSL USBFF USBFF OCOSL USBFS USBFS USBFS USBFG USBFG USBFG USBFG	36.93 25.51 19.39 24.12 32.5 19.39 24.12 32.5 64.62 92.63 309.79	213.33 213.33 213.33 32.77 211.05 211.05 211.05 211.05 211.05 211.05 211.05 201.93 201.93 201.93	138.47 138.47 138.47 136.19 136.19 136.19 136.19 136.19 127.07 127.07	124.72 124.72 110.52 110.52 110.52 110.52 110.52 110.52 124.72 124.72	34.25 34.25 25.91 25.91 25.91 25.91 25.91 25.91 25.91 34.25 34.25 34.25			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	

District Confession for Experience Connection from part of 1   1.03   0.0095   1.000								1			, ,				
Charle on Francisco   Charles of Service   Charles   C		Order Coordination For Specified Conversion Time, per LS		LICI	00081		32.77								
Part   Part			1			20.46		126.91	117.34	26.87		19.99	19.99	19.99	19.99
Dest Coveration for Spetting General Transport Conversion Transport Co		Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	2	UCL			201.76								
But Lapp Force: The 4 Word 25 State Digital Golds List		Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	3	UCL	USBFJ	12.89	201.76	126.91	117.34	26.87		19.99	19.99	19.99	19.99
But Lapp Force: The 4 Word 25 State Digital Golds List		0.1.0.0		1101	00001		00.77								
Sub-Loc Protect   Per 4 Wiley   10 Floor Debt Gross   1			4			07.44		407.07	404.70	24.05		10.00	40.00	40.00	40.00
Specings   Specing   Spe					USBEN										
Start Logic Fault   Fig. 4-100; 60 (SEE) (DIR) Carda Locar, 20x   1   LGC   MSPT   2711   2712   2712   2712   2712   2712   1			3		USBFN										
Sub-to-controlled Park Park Park SC Ages (public Gross Loope 2 and 1979															
Outstand Section   Per Score (Consistency Per Score)   Per Score															
Sub-loop Feature - New Yorks & Filting Supplied Control Coop - York   1 UND   10590F   7711   2019 2   1477   3425   1500   15		Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	3	UDL	USBFO	24.95	201.93	127.07	124.72	34.25		19.99	19.99	19.99	19.99
Sub-loop Feature - New Yorks & Filting Supplied Control Coop - York   1 UND   10590F   7711   2019 2   1477   3425   1500   15		0.1.0.0		LIDI	00001		00.77								
Sel-book Finetric Put-Africe St (Sept. Put-Africa)   Sept.		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zons	1			27 11		127.07	124 72	34.25		10.00	10.00	10.00	10.00
Octavior Concentration For Specified Conversion Time, per LG   USA   OCSA   S2.77															
Octavior Concentration For Specified Conversion Time, per LG   USA   OCSA   S2.77	-		3												
Companies   Comp															
Instrumed Sisk-Loop Modification - AVE Copper Dist Lood Cool Plays Removal pay 4 W   UEF   ULMSX   355.32   12.25     16.14   8.08   11.41   11.41		Order Coordination For Specified Conversion Time, per LS		UDL	OCOSL		32.77								
Instrumed Sisk-Loop Modification - AVE Copper Dist Lood Cool Plays Removal pay 4 W   UEF   ULMSX   355.32   12.25     16.14   8.08   11.41   11.41															
PR	Unbundled														
Discunded Sul-Note Note (Sul-Note Note)   February		Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W		HEE	LILMOY		255.22	40.05				40.44	0.00	44.44	44.46
PR		Linkundled Sub-less Medification 4 M Copper Diet Load Cail/Equip Berevul 4 M		UEF	ULM2X		355.32	12.25				18.14	გ.06	11.41	11.41
Unbounded Sub-toop Modification 2-wide w Copper Dist Bridged Tap Removal, per PR UEF ULMAT 559.94 14.28				UEF	I II MAV		355 32	12.25				18 14	8 06	11 /1	11 /1
UBDATE   U		P.12	+++	OLI	JLIVI4X		300.32	12.20				10.14	0.00	11.41	11.41
Unbundled Nework Terminating Wire (URTW)   URDPY   U				UEF	ULM4T		559.94	14.28				18.14	8.06	11.41	11.41
Usbanded Network Fernmatring Wire (UKTW) par Pa   Network Interface Device (MD)   1-8 ine   USKTW   USFP   0.36   0.20					1										
Network Interface Device (ND) - 1-2 line	Unbundled	Network Terminating Wire (UNTW)													
Network inferface Device (NID) - 1-2 line		Unbundled Network Terminating Wire (UNTW) per Pa		UENTW	UENPP	0.35	62.26	62.26				18.14	8.06	11.41	1.41
Network inferface Device (NID) - 1-2 line															
Network inferface Device (NR) -1-6 line   UENTW UNDG   127.79   89.11   15.14   2.06   11.41	Network Int														
Network literateae Device Cross Connect: 4"   UENTW   UNICC    11/2   11/2   11/2   18.14   8.06   11.41   11.41     Network interface Device Cross Connect: 4"   UENTW   UNICC    11/2   11/2   18.14   8.06   11.41   11.41     Network interface Device Cross Connect: 4"   UENTW   UNICC    11/2   11/2   18.14   8.06   11.41   11.41     Network interface Device Cross Connect: 4"   URL   UCTR   43.25   4.05   11.22   11.7															
Network literface Device Cross Connect - 4V															
NBUNDLED LOOP CONCENTRATION    International Components of Concentration   System & 17800;   U.C.				LIENTW	LINDC4		11.72	11.72							
Unbundled Logo Concentration - System & (TRO0)		Notwork interface Device cross connect 44		OLIVIV	011004		11.72	11.72				10.14	0.00	11.41	11541
Unbundled Loop Concentration - System & (1780)   19.99   19.	NBUNDLED LOOP CO	DICENTRATION													
Unbunded Logo Concentration - System A (TR30): U.C. UCT38 10.39		Unbundled Loop Concentration - System A (TR00)			UCT8A									19.99	
Unbundled Loop Concentration - System B (TR30):   ULC   UCT30   103.99   270.88   270.88   19.99   1															
Unbundled Loop Concentration - DS1 Loop Interface Car															
Inhundled Loop Concentration - ISDN Loop Interface (Brite Can   UDN ULCCI   9.39   21.05   20.93   10.92   10.86   19.99   1		Unbundled Loop Concentration - System B (TR30:		ULC	UCT3B	103.99	270.88	270.88				19.99	19.99	19.99	19.99
Inhundled Loop Concentration - ISDN Loop Interface (Brite Can   UDN ULCCI   9.39   21.05   20.93   10.92   10.86   19.99   1		Unbundled Lean Concentration DC1 Lean Interface Co.		III.C	LICTCO	E 01	126 12	02.04	24.02	0.52		10.00	10.00	10.00	10.00
Inhundred Loop Concentration - 2Wire Voice - Voice voice op Start or Ground Start Loop   Inhundred Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface   ULCC2		Unbundled Loop Concentration - DST Loop Interface Ca													
Unbundled Loop Concentration - 2 Wire Voice-Loop Start or Ground Start Loop Interface (POTS Card Unbundled Loop Concentration - 2 Wire Voice Loop Interface (Specials Ca UEA ULCC 13.95 21.05 20.93 10.92 10.86 19.99															
Interface (POTS Card   UDA unded Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface (SPECIAIS )   UDA unded Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca   UEA   ULCC   1.3.95   21.0.5   20.9.3   10.92   10.86   19.99															
CSPOTS Card    UEA   ULCCR   13.95   21.05   20.93   10.92   10.86   19.99   19.99   19.99   19.99   19.99   19.99   19.99   Unbundled Loop Concentration - TEST CIRCUIT Car   ULC   UCTC   40.67   21.05   20.93   10.92   10.86   19.99   19.99   19.99   19.99   19.99   Unbundled Loop Concentration - Digital 19.2 Khps Data Loop Interfa   ULC   UCTC   40.67   21.05   20.93   10.92   10.86   19.99   19.99   19.99   19.99   Unbundled Loop Concentration - Digital 19.2 Khps Data Loop Interfa   UDL   ULCCS   12.33   21.05   20.93   10.92   10.86   19.99   19.99   19.99   19.99   Unbundled Loop Concentration - Digital 56 Khps Data Loop Interfa   UDL   ULCCS   12.33   21.05   20.93   10.92   10.86   19.99   19				UEA	ULCC2	2.35	21.05	20.93	10.92	10.86		19.99	19.99	19.99	19.99
Unbundled Loop Concentration - A Wire Voice Loop Interface (Specials Ca   UEA   ULCC4   8.32   21.05   20.93   10.92   10.86   119.99															
Unbundled Loop Concentration - TEST CIRCUIT Cas   UCTT						13.95									
Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa   UDL   ULCG7   12.33   21.05   20.93   10.92   10.86   19.99   1	$\longrightarrow$														
Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa															
Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa															
INBUNDLED SUB-LOOP CONCENTRATION (OUTSIDE CO)  INBUNDLED SUB-LOOP CONCENTRATION (OUTSIDE CO)  INE OTHER, PROVISIONING ONLY - NO RATE  INIT - Dispatch and Service Order for NID installation  UENTW UNDBX  UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW UNECK  UINTW UNECK  U		Unbundled Loop Concentration - Digital 50 Kbps Data Loop Interfa							10.92						
INE OTHER, PROVISIONING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UENTW  UNDBX  UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW  UENCE  UEANL, UEF, UEQ, UENTW  UNECN  ULCL, UDC, UDL UDL UDL, UDL, UDL UDL UDL UDL UDL UDL UDL UDL UDL UDL							0								. 3.00
INE OTHER, PROVISIONING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UENTW  UNDBX  UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW  UENCE  UEANL, UEF, UEQ, UENTW  UNECN  ULCL, UDC, UDL UDL UDL, UDL, UDL UDL UDL UDL UDL UDL UDL UDL UDL UDL															
NID - Dispatch and Service Order for NID installation  UENTW UNDBX  UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW UNDBX  UENCE  UENCE  UENCE  UENCE  UENCE  UAL, UCL, UDC, UDL  UDR, UBCN  UNECN	NBUNDLED SUB-LOO	DP CONCENTRATION (OUTSIDE CO)													
NID - Dispatch and Service Order for NID installation  UENTW UNDBX  UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW UNDBX  UENCE  UENCE  UENCE  UENCE  UENCE  UAL, UCL, UDC, UDL  UDR, UBCN  UNECN	55	CONCENTRATION (COTOLDE CO)													
NID - Dispatch and Service Order for NID installation  UENTW UNDBX  UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW UENCE  UENLUEF, UEQ, UENLUEF, UEQ, UENTW UNECN  UAL, UCL, UDC, UDL, UDN, UEA, UHL, UL C UHECN UNDEC		on concentration (corolle co)													
UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW UENCE  Unbundled Contract Name, Provisioning Only - No Rate  UENTW UENCB  ULENTW UENCB  ULENTW UENCB  ULENTW UNECN  UN															
UNTW Circuit Id Establishment, Provisioning Only - No Rate  UENTW UENCE  Unbundled Contract Name, Provisioning Only - No Rate  UENTW UENCB  ULENTW UENCB  ULENTW UENCB  ULENTW UENCB  ULENTW ULECN UNECN  UNDE															
Unbundled Contract Name, Provisioning Only - No Rate  ULANL, UEF, UEQ, ULNTW UNECN  UAL, UCL, UDC, UDL , UDN, UEA, UHL, UL C UNECN U		NING ONLY - NO RATE		HENTW	LINDDY										
Unbundled Contract Name, Provisioning Only - No Rate  ULANL, UEF, UEQ, ULNTW UNECN  UAL, UCL, UDC, UDL , UDN, UEA, UHL, UL C UNECN U		NING ONLY - NO RATE		UENTW	UNDBX										
Unbundled Contract Name, Provisioning Only - No Rate		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation													
Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  UEA,UDX,UCL,UDL USBFQ 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation		UENTW											
Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  UEA,UDN,UCL,UDL USBFQ 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW UEANL,UEF,UEQ,	UENCE										
Unbundled Contact Name, Provisioning Only - no rate  C UNECN 0 0  UEA,UDN,UCL,UD  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  UEA,UDN,UCL,UD USBFQ 0 0  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  UEA,USL,UCL,UDL USBFR 0 0		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW UEANL,UEF,UEQ, UENTW	UENCE										
Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  UEA,UDN,UCL,UD C USBFQ 0 0 Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  UEA,USL,UCL,UDL USBFR 0 0		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate	l	UENTW UEANL,UEF,UEQ, UENTW JAL,UCL,UDC,UDL	UENCE										
Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  C USBFQ 0 0  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  UEA,USL,UCL,UDL USBFR 0 0		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate	l	UENTW UEANL,UEF,UEQ, UENTW JAL,UCL,UDC,UDL, UDN,UEA,UHL,UL	UENCE	0	0								
Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  UEA,USL,UCL,UDL USBFR 0 0		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate	Į,	UENTW UEANL,UEF,UEQ, UENTW JAL,UCL,UDC,UDL UDN,UEA,UHL,UL C	UENCE	0	0								
		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate	Į,	UENTW UEANL,UEF,UEQ, UENTW JAL,UCL,UDC,UDL UDN,UEA,UHL,UL C	UENCE UNECN										
		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate	Į,	UENTW UEANL,UEF,UEQ, UENTW JAL,UCL,UDC,UDL UDN,UEA,UHL,UL C	UENCE UNECN										
Unbundled DS4 Loop. Superframe Formet Option, no re		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra		UENTW UEANL,UEF,UEQ, UENTW JAL,UCL,UDC,UDL UDN,UEA,UHL,UL C UEA,UDN,UCL,UD C	UENCE UNECN UNECN USBFQ	0	0								
		NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra		UENTW UEANL,UEF,UEQ, UENTW JAL,UCL,UDC,UDL UDN,UEA,UHL,UL C UEA,UDN,UCL,UD C	UENCE UNECN UNECN USBFQ	0	0								

												1		
	Unbundled DS1 Loop - Expanded Superframe Format option - no ra		USL	CCOEF	0	0								
HIGH CAPACITY UNBU														
NOTE: 4 m	nonth minimum billing period													
	High Capacity Unbundled Local Loop - DS3 - Per Mile per mon		UE3	1L5ND	12.75									ļ
	High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor		UE3	UE3PX	419.1	902.06	527.3	242.15	169.38		31.27	31.27	3.92	3.92
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon		UDLSX	1L5ND	12.75									ļ
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor		UDLSX	UDLS1	433.21	902.06	527.3	242.15	169.38		31.27	31.27	3.92	3.92
														<b>,</b>
LOOP MAKE-UP	Land Hall Brown Br													<b></b> '
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried		UMK	UMKLW		47.04	47.04							1 '
	(Manual).					47.91	47.91							<b></b>
	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).		UMK	UMKLP		50.8	50.8							<b></b>
	Loop MakeupWith or Without Reservation, per working or spare facility queried													
	(Mechanized)		UMK	PSUMK		0.6852	0.6852							<b></b>
														<b></b>
LINE SHARING														<b></b>
														<b></b>
	Line Sharing Splitter, per System 96 Line Capaci		ULS	ULSDA	215.88	377.17	0	355.58	0	0				+
	Line Sharing Splitter, per System 24 Line Capaci		ULS	ULSDB	53.97	377.17	0	355.58	0	0				+
	Line Sharing Splitte, Per System, 8 Line Capaci	!	ULS	ULSD8	17.99	377.17	0	355.58	0	0				<b></b>
	Line Sharing - per Line Activatio		ULS	ULSDC	0.61	36.97	21.17	20	9.82		18.14	8.06	11.41	11.41
	Line Sharing - per Subsequent Activity per Line Rearrangeme		ULS	ULSDS	-	32.74	16.35	1			18.14	8.06		<del>                                     </del>
					ļ		ļ			1	1	<b></b>		
	Use Objects OFEO/DEEO Oscillostrusti CO	_								1	1			
	Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 line	es .		111.000		57.04		44.07						ĺ
	(16 pair)	<u> </u>	ULS	ULSDG		57.64		11.37						<b>!</b>
														<b></b>
UNBUNDLED TRANSPO	ORT													<b></b>
														<b></b>
COMMON	TRANSPORT (Shared)													<b></b>
	Common Transport - Per Mile, Per MOL				0.0000083									<b></b>
	Common Transport - Facilities Termination Per MO				0.00047									<b></b>
	TERRORE OLIVANIEL DEPLOATED TRANSPORT													+
NOTE: INT	FEROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS	3 = one	month, DS3 and above to	our month	\$									+
INTEROFE	FICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE													<u> </u>
INTEROFF			LIATIO	41.53/3/	0.0405									
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per moi		U1TVX	1L5XX	0.0165									<b></b>
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination		U1TVX	U1TV2	00.40	00.00	5470	00.04	13.97		04.07	31.27	0.00	0.00
	per month		UTIVX	U11V2	26.12	80.98	54.76	33.91	13.97		31.27	31.27	3.92	3.92
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile p	er	LIATIO	41.53/3/	0.0405									
	month		U1TVX	1L5XX	0.0165									<b>—</b>
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination		LIATIO	LIATEDO	00.40	00.00	5470	00.04	40.07		04.07	04.07	0.00	0.00
	per month		U1TVX	U1TR2	26.12	80.98	54.76	33.91	13.97		31.27	31.27	3.92	3.92
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month		U1TVX	1L5XX	0.0165									
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Fer Mile per month	,	UTIVA	ILSAA	0.0165									
	per month		U1TVX	U1TV4	22.89	80.98	54.76	33.91	13.97		31.27	31.27	3.92	3.92
	per month		UTIVA	01174	22.89	80.98	54.76	33.91	13.97		31.27	31.27	3.92	3.92
	Liver (for Observat Bullion of FOUR		LUTDY	41.57/7	0.0405									+
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per mon		U1TDX	1L5XX	0.0165	00.00	5470	00.04	40.07		04.07	04.07	0.00	0.00
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mor		U1TDX	U1TD5	18.04	80.99	54.76	33.91	13.97		31.27	31.27	3.92	3.92
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per mon		U1TDX	1L5XX	0.0165	00.00	5470	00.04	40.07		04.07	04.07	0.00	0.00
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mor		U1TDX	U1TD6	18.04	80.99	54.76	33.91	13.97	1	31.27	31.27	3.92	3.92
INTEROFF	FICE CHANNEL - DEDICATED TRANSPORT - DS1			-				1		1	1			
INTEROFF			U1TD1	1L5XX	0.3367			1		1	1	1		
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mor					470.04	400.44	22.42	20.27		24.27	24.07	2.00	20.00
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor		U1TD1	U1TF1	81.45	178.34	163.44	33.13	29.27	+	31.27	31.27	3.92	39.92
INTEROFE	FICE CHANNEL - DEDICATED TRANSPORT- DS3			+										
INTEROFF			U1TD3	41.577	7.07									<b>—</b>
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mor		U1TD3	1L5XX U1TF3	7.67 982.87	556.98	325.16	121.99	118.47	+	31.27	31.27	3.92	3.92
	interoffice Channer - Dedicated Transport - D53 - Facility Termination per mor		01103	UIIF3	902.07	556.96	323.10	121.99	110.47	1	31.27	31.21	3.92	3.92
INTERACE	FICE CHANNEL - DEDICATED TRANSPORT- STS-1			1				1		1	1	1		
INTEROFF	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon		U1TS1	1L5XX	7.67		<b> </b>	+		1	1	<b> </b>		
			U1TS1	U1TFS	959.46	556.89	325.16	121.00	110 47	1	31.27	31.27	3.92	2.02
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor		U1151	UITES	959.46	98.000	325.76	121.99	118.47	1	31.27	31.27	3.92	3.92
			<del>                                     </del>	-	<b> </b>		<b> </b>	+		1	1	<b> </b>		
				1				1		1	1			
				1						+	+			
LOCAL CH	HANNEL - DEDICATED TRANSPORT			1						+	+			
	CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one m	onth C	IS3 and above-four month	ns						+	+			
INOTE. EU	Local Channel - Dedicated - 2-Wire Voice Grade Per Month	Julii, L	ULCVX	ULDV2	21.44	385.78	66.26	74.25	6.48	<b>†</b>	31.27	31.27	3.92	3.92
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor		ULCVX	ULDR2	21.44	385.78	66.26	74.25	6.48	+	31.27	31.27	3.92	3.92
										1				
	Local Channel - Dedicated - 4-Wire Voice Grade per mon		1 UNCVX	ULDV4	22.7	385.78	66.26	74.25 44.97	6.48	1	31.26	31.26	3.92	3.92
	Local Channel - Dedicated - DS1 per month - Zone		. 02001	ULDF1	45.1	354.56	307.1		30.93	+	31.27	31.27	3.92	3.92
	Local Channel - Dedicated - DS1 per month - Zone		2 ULDD1	ULDF1	85.62	354.56	307.1	44.97	30.93	-	31.27	31.27	3.92	3.92
	Local Channel - Dedicated - DS1 per month - Zone		3 ULDD1	ULDF1	144.54	354.56	307.1	44.97	30.93	1	31.27	31.27	3.92	3.92

	Local Channel - Dedicated - DS3 - Per Mile per mon	ULDD3	1L5NC	9.92	902.06	527.3	242.15	100.00	31.27	31.27	3.92	2.00
	Local Channel - Dedicated - DS3 - Facility Termination per mon  Local Channel - Dedicated - STS-1- Per Mile per mon	ULDD3 ULDS1	ULDF3 1L5NC	543.82 9.92	902.06	527.3	242.15	169.38	31.27	31.27	3.92	3.92
	Local Channel - Dedicated - STS-1 - Facility Termination per mon	ULDS1	ULDFS	529.62	902.06	527.3	242.15	169.38	31.27	31.27	3.92	3.92
	Local Channel - Dedicated - OC12 - Facility Termination per mon	02501	025.0	3372.85	1181.77	408.27	121.99	118.47	50.25	50.25	20.94	20.94
=												
ULTIPLEX		UXTD1	MQ1	121.5	101.00	125.01	04.05	10.01	31.27	04.07	0.00	0.00
	Channelization - DS1 to DS0 Channel Syster  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)	UDL	1D1DD	1.6	181.88 13.14	9.42	21.35	19.84	31.21	31.27	3.92	3.92
			UC1CA			9.42						
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per mont Voice Grade COCI - DS1 to DS0 Channel System - per mon	UDN UEA	1D1VG	3.43 0.7509	13.14 13.14	9.42						
	DS3 to DS1 Channel System per mont	UXTD3	MQ3	233.25	355.89	187.73	67.4	64.49	31.27	31.27	3.92	3.92
	STS1 to DS1 Channel System per mont	UXTS1	MQ3	233.25	355.89	187.73	67.4	64.49	31.27	31.27	3.92	3.92
	DS3 Interface Unit (DS1 COCI) used with Loop per mont	USL	UC1D1	13.64	13.14	9.42	07.4	04.45	01.27	31.27	0.02	0.02
	Boo michado om (Bor coo) accamar 200 por mona	332	- 00151	10.01		0.12						
RK FIBER	R											
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local											
	Channe	UDF	1L5DC	62.92								
	NRC Dark Fiber - Local Channe	UDF	UDFC4		1276.79	275.43	642.55	400.58	31.27	31.27	3.92	3.9
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -											
	Interoffice Channe	UDF	1L5DF	31.78								
	NRC Dark Fiber - Interoffice Channe	UDF	UDF14		1276.79	275.43	642.55	400.58	31.27	31.27	3.92	3.9
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local					1						
	Loop	UDF	1L5DL	62.92	4070	077 10	040	100		04.00	0.00	
ANODO	NRC Dark Fiber - Local Loop	UDF	UDFL4		1276.79	275.43	642.55	400.58	31.27	31.27	3.92	3.9
ANSPOR	TOTHER		+			-	1			1	1	<u> </u>
		<del>                                      </del>	+			1						
		++	+			+			<del>-  </del>		1	
	Optional Features & Functions:	<del>     </del>	+			<b> </b>						
	optional i catalos a i anotions.	<del>     </del>	+			<b> </b>						
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chanı	UNC1X	CCOEF		184.65	23.7	1.97	0.77	29.2	3.92		
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chanr	UNC1X	CCOSF		184.65	23.7	1.97	0.77	29.2	3.92		
X ACCES	SS TEN DIGIT SCREENING	Siteix	- 0000		101.00	20.1	1.01	0.77	20.2	0.02		
	8XX Access Ten Digit Screening, Per Ca	OHD	+	0.0005305								
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv	OHD	N8R1X	0.0000000	6.29	0.73	18.14		18.14	18.14	11.4	11.
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation	OHD			12.27	1.39	8.3	0.73	18.14	18.14	11.4	11.
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation	OHD	N8FTX		12.27	1.39	8.3	0.73	18.14	18.14	11.4	11.
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb	OHD	N8FCX		4.27	2.14			18.14	18.14	11.4	11.
	8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested											
	Per 8XX No.	OHD	N8FMX		5	2.86			18.14	18.14	11.4	11.4
	8XX Access Ten Digit Screening, Change Charge Per Reque	OHD	N8FAX		7.01	0.73			18.14	18.15	11.4	11.
	8XX Access Ten Digit Screening, Call Handling and Destination Featur	OHD	N8FDX		4.27				18.14	18.14	11.4	11.
NE INFOR	MATION DATA BASE ACCESS (LIDB)		_									
	LIDB Common Transport Per Quer	OQT		0.0000418								
	LIDB Validation Per Quer	OQU	NDDDV	0.0103774	40.47				10.11	40.44	44.4	- 44
	LIDB Originating Point Code Establishment or Chanç	OQT, OQU	NRPBX		48.17				18.14	18.14	11.4	11.4
SNALING	(CCS7)											
GNALING												
		1DB	DTOCY	161.00					10.14	10.14	11.4	11
	CCS7 Signaling Termination, Per STP Por	1DB	PT8SX	161.99					18.14	18.14	11.4	11.
	CCS7 Signaling Usage, Per TCAP Messag	1DB		0.0001052	126 34	126 34	101.1	101.1				
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A link	1DB 1DB	TPP++	0.0001052 19.48	126.34 126.34	126.34 126.34	101.1	101.1 101.1	18.14	18.14	11.4	11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A lini CCS7 Signaling Connection, Per link (B link) (also known as D lin	1DB		0.0001052	126.34 126.34	126.34 126.34	101.1 101.1	101.1 101.1				11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linł CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag	1DB 1DB 1DB 1DB	TPP++ TPP++	0.0001052 19.48 19.48					18.14	18.14 18.14	11.4	11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT	1DB 1DB 1DB	TPP++	0.0001052 19.48 19.48 0.000043					18.14 18.14	18.14	11.4 11.4	11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A lin) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectee	1DB 1DB 1DB 1DB	TPP++ TPP++	0.0001052 19.48 19.48 0.000043					18.14 18.14	18.14 18.14	11.4 11.4	11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A lin) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per	1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56	0.0001052 19.48 19.48 0.000043	126.34	126.34			18.14 18.14 18.14	18.14 18.14 18.14	11.4 11.4 11.4	11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A lin) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectee	1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56	0.0001052 19.48 19.48 0.000043	126.34	126.34			18.14 18.14 18.14	18.14 18.14 18.14	11.4 11.4 11.4	11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected	1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
1 SERVI	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected	1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11. 11.
11 SERVI	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected	1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS3 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected	1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A lin) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Diange Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCE  AME (CNAM) SERVICE	1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043 406.71	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec. CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCE  AME (CNAM) SERVICE  CNAM for DB Owners, Per Quer)	1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043 406.71	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A lin) CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Diange Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCE  AME (CNAM) SERVICE	1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043 406.71	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec. CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCE  AME (CNAM) SERVICE  CNAM for DB Owners, Per Quer)	1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043 406.71	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
011 SERVI	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CE  AME (CNAM) SERVICE  CNAM for DB Owners, Per Quen CNAM for Non DB Owners, Per Quen	1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043 406.71	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec. CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCE  AME (CNAM) SERVICE  CNAM for DB Owners, Per Quer)	1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO	0.0001052 19.48 19.48 0.000043 406.71	126.34	126.34			18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4	11. 11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linb CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Origination Point Code Establishment or Change, Per Origination Point Code Establishment or Change, Per Origina	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO CCAPD	0.0001052 19.48 19.48 0.000043 406.71	40	40			18.14 18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4 11.4	11. 11. 11. 11.
	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linb CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Origination Point Code Establishment or Change, Per Origination Point Code Establishment or Change, Per Origina	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO CCAPD	0.0001052 19.48 19.48 0.000043 406.71	40	40			18.14 18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4 11.4	11. 11. 11. 11.
LLING NA	CCS7 Signaling Usage, Per TCAP Messag CCS3 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment o	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO CCAPD	0.0001052 19.48 19.48 0.000043 406.71	40	40			18.14 18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4 11.4	11. 11. 11. 11.
LLING NA	CCS7 Signaling Usage, Per TCAP Messag CCS7 Signaling Connection, Per link (A linb CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage, Per ISUP Messag CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, Per Origination Point Code Establishment or Change, Per Origination Point Code Establishment or Change, Per Origina	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO CCAPD	0.0001052 19.48 19.48 0.000043 406.71	40	40			18.14 18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4 11.4	11. 11. 11.
LLING NA	CCS7 Signaling Usage, Per TCAP Messag CCS3 Signaling Connection, Per link (A linh CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Connection, Per link (B link) (also known as D lin CCS7 Signaling Usage Surrogate, per link per LAT CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affectec CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected  CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment or Change, per Stp Affected  CCS7 Signaling Point Code, per Origination Point Code Establishment o	1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB 1DB	TPP++ TPP++ STU56 CCAPO CCAPD	0.0001052 19.48 19.48 0.000043 406.71	40	40			18.14 18.14 18.14 18.14 18.14	18.14 18.14 18.14 18.14 18.14	11.4 11.4 11.4 11.4 11.4	11. 11. 11. 11.

BRANDING - OP	ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  RATOR SERVICES  Inward Operator Services - Verification, Per Minul				1.2 1.24									-
BRANDING - OP	Oper. Call Processing - Oper. Provided, Per Min Using BST LID Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC Oper. Call Processing - Fully Automated, per Call - Using BST LID Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  RATOR SERVICES				1.2									+
BRANDING - OP  DIRECTORY AS	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC Oper. Call Processing - Fully Automated, per Call - Using BST LID Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC RATOR SERVICES				1.2									
BRANDING - OP  DIRECTORY AS	Oper. Call Processing - Fully Automated, per Call - Using BST LID Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC RATOR SERVICES													
BRANDING - OP  DIRECTORY AS	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE  RATOR SERVICES										<b></b> '			
BRANDING - OP  DIRECTORY AS	RATOR SERVICES				0.2							⊢		<del> </del>
BRANDING - OP  DIRECTORY AS					0.2						<u> </u>	⊢		<del> </del>
BRANDING - OP  DIRECTORY AS											<u> </u>	⊢		<del> </del>
DIRECTORY AS DIR					4.45						<u> </u>	⊢		<del> </del>
DIRECTORY AS DIR					1.15									<b>↓</b>
DIRECTORY AS DIR	Inward Operator Services - Verification and Emergency Interrupt - Per Minu				1.15						<u> </u>	⊢		
DIRECTORY AS DIR	PERATOR CALL PROCESSING										<del></del>	$\vdash$	$\vdash$	+
DIR				CBAOS		7000	7000				19.99	40.00	19.99	19.99
DIR	Recording of Custom Branded OA Announcement											19.99	19.99	19.99
DIR	Loading of Custom Branded OA Announcement per shelf/NAV			CBAOL		500	500				19.99	19.99		
DIR											ļ	<b></b>		<b>_</b>
											ļ	<b></b>		<b>↓</b>
	IRECTORY ASSISTANCE ACCESS SERVICE										ļ	<b></b>		
	Directory Assistance Access Service Calls, Charge Per Ca				0.275						<b></b>			<b>_</b>
DIR	IRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)													
	Directory Assistance Call Completion Access Service (DACC), Per Call Attem				0.1									
	NBRANDING											<u> </u>		
DIR	IRECTORY TRANSPORT													
	Directory Transport - Local Channel DS				43.83	339.69	298.29	33.02	23.32		42.34	42.34	19.48	19.48
	Directory Transport - DS1 Level Interoffice Per Mi				0.78									
	Directory Transport - DS1 Level Interoffice Per Facility Termination				93.4	140.49	106.69	20	16.34		18.14	18.14	8.06	8.06
	Switched Common Transport Per DA Access Service Per Ca				0.0003274									
	Switched Common Transport Per DA Access Service Per Call Per Mi				0.0000175								,	
	Access Tandem Switching Per DA Access Service Per Ca				0.0025257									
	Directory Transport - Installation NRC, Per Trunk or Signaling Connection					195.54	4.23				130.05	4.23	,	
											,			
DIR	IRECTORY ASSISTANCE DATA BASE SERVICE (DADS)													1
	Directory Assistance Data Base Service Charge Per Listir				0.04							[		1
	Directory Assistance Data Base Service, per mont			DBSOF	150							[		1
BRANDING - DIF	DIRECTORY ASSISTANCE													1
	Custom Branding Announcement, per Recording to be used with the provision of DA		AMT	CBADA		3000	3000							<b>†</b>
	Loading of Custom Branded Announcement per DRAM Card/Switch		AMT	CBADC		690	690							+
	20daling of Oddom Brandod ramounionness por Brown Oddaromion		7	ODADO		030	030					<b>——</b>		+
SELECTIVE RO	OLITING										<b></b>		-	+
DEELECTIVE NO.	551116										<b></b>		-	+
	Selective Routing Per Unique Line Class Code Per Request Per Swit			USRCR		229.65	229.65				31.92	7.32	1	
	Selective Routing Fer Ornque Line Class Code Fer Request Fer Swit			USKCK		229.00	229.00				31.92	1.32		+
VIRTUAL COLLO	LOCATION													+
VIKTOAL COLL	ECCATION										-		$\vdash$	
	Virtual Callegation 2 with Const Constant (Inc.		ueanl,uea,udn,udc,u	UEAC2	0.26	23.04	20.44	9.48	0.54		19.99	19.99	19.99	40.00
	Virtual Collocation - 2-wire Cross Connects (loop	-	al,uhl,ucl,uec UEPSR, UEPSB		0.26	23.04	22.11 22.11		8.54 8.54					19.99
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splittin	'	UEPSR, UEPSB					9.48			19.99	19.99	19.99	19.99
	Virtual Collocation - 2-wire Cross Connects (por			VE1R2	0.26	23.04	22.11	9.48	8.54		19.99	19.99	19.99	19.99
	Virtual Collocation - 4-wire Cross Connects (loop		uea,uhl,ucl,ud	UEAC4	0.52	23.23	22.24	9.53	8.55		19.99	19.99	19.99	19.99
	Virtual Collocation - 4-wire Cross Connects (por		01.0	VE1R4	0.52	23.23	22.24	9.53	8.55		19.99	19.99	19.99	19.99
	Virtual Collocation - 2-Fiber Cross Connect		CLO	CNC2F	19.13	41.07	29.63	12.84	10.29		19.99	19.99	19.99	19.99
	Virtual Collocation - 4-Fiber Cross Connects		CLO	CNC4F	34.38	49.81	38.37	16.75	14.2		19.99	19.99	19.99	19.99
	Virtual Collocatin - DS1 Cross Connect		USL,ULC,CLO	CNC1X	7.5	43.61	30.6	9.56	8.63					+
AIN OF FOTO	IC CARRIED POLITING			-				-			<b></b> '	<del></del> '		+
AIN SELECTIVE	E CARRIER ROUTING		200	000==		004=00					40.00	40.55	40.00	
	Regional Service Establishment		SRC	SRCEC		391788					19.99	19.99	19.99	19.99
	End Office Establishment		SRC	SRCEO		320.53	320.53				19.99	19.99	19.99	19.99
	Line/Port NRC, per end user		SRC	SRCLP		2.06	2.06				19.99	19.99	19.99	19.99
	Query NRC, per query	Γ	SRC		0.000448							l	1 7	
AIN - BELLSOU	UTH AIN SMS ACCESS SERVICE													
												·	[	
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			CAMSE		153.31	153.31	78.06	78.06		18.14	18.14	11.4	11.4
														1
	AIN SMS Access Service - Port Connection - Dial/Shared Access			CAMDP		50.07	50.07	18.61	18.61		18.14	18.14	11.4	11.4
		-					22.0.							
	AIN SMS Access Service - Port Connection - ISDN Access			CAM1P		50.07	50.07	18.61	18.61		18.14	18.14	11.4	11.4
	- III CING / ISSUES COLVICE   OIL COMMODICAL INDIVIOUSS	-		J/ UVI II		50.07	50.07	10.01	10.01	l	10.14	10.14	11.4	+
	AIN SMS Access Service - User Identification Codes - Per User ID Code			CAMAU		104.95	104.95	48.95	48.95	1	18.14	18.14	11.4	11.4
	Any Swis Access Service - Oser Identification Codes - Per Oser ID Code			CAIVIAU		104.95	104.95	40.90	40.95		10.14	10.14	11.4	11.4
	AIN CMC Access Coming Committee Cond Partition ID Code In 1997 to Development			CAMRC		405.00	405.00	24.4	24.4	1	40.44	40.44	44.4	44.4
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			CAMRC		125.33	125.33	24.4	24.4	<b></b>	18.14	18.14	11.4	11.4
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)				0.0029						<b></b> '	ļ'	ļ'	4
	AIN SMS Access Service - Session, Per Minute				0.1									1
	AIN SMS Access Service - Company Performed Session, Per Minute			<u> </u>	1.97					<u> </u>	<u> </u>	'	'	
												1		
	UTH AIN TOOLKIT SERVICE													

							_							
AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			BAPSC		153.25	153.25	78.05	78.05			18.14	18.14	11.4	11.4
AIN Toolkit Service - Service Establishment Charge, Fer State, Initial Setup  AIN Toolkit Service - Training Session, Per Customer	_		BAPVX		8315	8315	76.00	76.03			18.14	18.14	11.4	11.4
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt	-		BAPTT		41.08	41.08	18.6	18.6			18.14	18.14	11.4	11.4
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay			BAPTD		41.08	41.08	18.6	18.6			18.14	18.14	11.4	11.4
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate			BAPTM		41.08	41.08	18.6	18.6			18.14	18.14	11.4	11.4
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP			BAPTO		92.99	92.99	26.73	26.73			18.14	18.14	11.4	11.4
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code	-+		BAPTE		92.99 92.99	92.99 92.99	26.73 26.73	26.73 26.73			18.14 18.14	18.14 18.14	11.4 11.4	11.4
AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Peature Code  AIN Toolkit Service - Query Charge, Per Query			DAFIF	0.03	92.99	92.99	20.73	20.73			18.14	18.14	11.4	11.4
AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per	-			0.00										
Query				0.0065										
AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes				1.79										
AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription			BAPMS	15.89	34.61	34.61	21.97	24.07			18.14	18.14	11.4	11.4
AIN Toolkit Service - Worlding report - Per AIN Toolkit Service Subscription  AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription	-		BAPLS	0.08	37.77	37.77	21.97	21.97			18.14	18.14	11.4	11.4
And toolkin dervice apecial olday if eliving rooming dervice dabsorption			D/ (I LO	0.00	57.77	07.77					10.14	10.14	11.4	111.4
AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			BAPDS	15.81	34.61	34.61	21.97	21.97			18.14	18.14	11.4	11.4
AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			BAPES	0.0026	37.77	37.77					18.14	18.14	11.4	11.4
JF/EDOUF/ADUF/CMDS														
ACCECC DAILY LICACE FILE (ADLIE)	-	1	1			1								
ACCESS DAILY USAGE FILE (ADUF)  ADUF: Message Processing, per messag	-+	+	1	0.004										
ADUF: Message Flocessing, per messag  ADUF: Data Transmission (CONNECT:DIRECT), per messag				0.0000305										
neon sala namono (contreone neon), per mocag				0.0000000										
ENHANCED OPTIONAL DAILY USAGE FILE (EODUF)														
EODUF: Message Processing, per message				0.004										
OPTIONAL DAILY HOLDE FILE (OPHE)														
OPTIONAL DAILY USAGE FILE (ODUF)  ODUF: Recording, per message	-			0.00019										
ODUF: Necording, per message ODUF: Message Processing, per message	-			0.0024										
ODUF: Message Processing, per Magnetic Tape provisions				47.3										
ODUF: Data Transmission (CONNECT:DIRECT), per messag				0.00003										
	,													
	-													
ANCED EXTENDED LINK (EELs)														
	ni El · l	Et Laudordalo El I: I	Jachvillo 1	TN: Now Orload	ne I A:									
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia				TN; New Orlear	ns, LA;									
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b	elow ex	cept Switch As Is Ch	arge.			o currently com	nbined facilit	ies converte	I to UNEs.(N	lon-recurrin	g rates do no	ot apply.)		
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia	elow ex	ccept Switch As Is Ch are converted to UNE	arge. rates. A S	Switch As Is Ch		currently com	nbined facilit	ies converte	I to UNEs.(N	on-recurrin	g rates do no	ot apply.)		
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the	which a	ccept Switch As Is Ch are converted to UNE	arge. rates. A S	Switch As Is Ch		currently com	nbined facilit	ies converte	I to UNEs.(N	lon-recurrin	g rates do no	ot apply.)		
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL	elow ex which a he GA P	ccept Switch As Is Chare converted to UNE	arge. rates. A S As Is Cha	Switch As Is Ch rge.)	arge applies to				I to UNEs.(N	lon-recurrin				
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the	which a	ccept Switch As Is Chare converted to UNE	arge. rates. A S	Switch As Is Ch		o currently com	nbined facilit	ies converted	I to UNEs.(N	lon-recurrin	g rates do no	at apply.)	3.92	3.92
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	which a he GA P	scept Switch As Is Chare converted to UNE SC order.(No Switch UNCVX	rates. A S As Is Char	Switch As Is Ch rge.) 17.65	arge applies to	92.77	82.08	12.22	I to UNEs.(N	lon-recurrin	31.27	31.27		3.92
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL	elow ex which a he GA P	scept Switch As Is Chare converted to UNE SC order.(No Switch UNCVX	arge. rates. A S As Is Cha	Switch As Is Ch rge.)	arge applies to				I to UNEs.(N	lon-recurrin			3.92	3.9.
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combined network elements apply to ordinarily combine	which a he GA P	ccept Switch As Is Chare converted to UNE ISC order.(No Switch UNCVX	rates. A S As Is Char	Switch As Is Ch rge.) 17.65	arge applies to	92.77	82.08	12.22	I to UNEs.(N	lon-recurrin	31.27	31.27		3.9
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhili, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the State of the NOTE: In State of the NOTE	below ex which a he GA P	ccept Switch As Is Chare converted to UNE ISC order.(No Switch UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	AS IS Character UEAL2  UEAL2  UEAL2  UEAL2  1L5XX	17.65 30.32 61.93 0.3367	193.82 193.82	92.77 92.77 92.77	82.08 82.08 82.08	12.22 12.22 12.22	l to UNEs.(N	lon-recurrin	31.27 31.27 31.27	31.27 31.27 31.27	3.92	3.9
NOTE: New ELS available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements per the NOTE: In Georgia, the EEL network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor	below ex which a he GA P	ccept Switch As Is Chare converted to UNE SC Order.(No Switch UNCVX UNCVX UNCVX UNCVX UNCVX UNCYX UNCYX UNC1X	uEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	17.65 30.32 61.93 0.3367 81.45	193.82 193.82 193.82 295.39	92.77 92.77 92.77 213.86	82.08 82.08 82.08 78.86	12.22 12.22 12.22 34.76	I to UNEs.(N	lon-recurrin	31.27	31.27	3.92	
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Service of the Service of the Service of the Service of the Service of the Service of the Service of the Service of the Service of the Service of the Service of the Service of the Service of Serv	below ex which a he GA P	are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X	UEAL2	witch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5	193.82 193.82 193.82 193.82 295.39 123.37	92.77 92.77 92.77 213.86 26.67	82.08 82.08 82.08	12.22 12.22 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27	31.27 31.27 31.27	3.92	3.9
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Pirst 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!	below ex which a he GA P	ccept Switch As Is Chare converted to UNE SC Order.(No Switch UNCVX UNCVX UNCVX UNCVX UNCVX UNCYX UNCYX UNC1X	uEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	17.65 30.32 61.93 0.3367 81.45	193.82 193.82 193.82 295.39	92.77 92.77 92.77 213.86	82.08 82.08 82.08 78.86	12.22 12.22 12.22 34.76	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27	31.27 31.27 31.27	3.92	3.9
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Service of the Service of the Service of the Service of Servic	below ex which a he GA P	ccept Switch As Is Chare converted to UNESC Order. (No Switch UNCVX UNCVX UNCVX UNCVX UNCVX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509	193.82 193.82 193.82 193.82 295.39 123.37 12.15	92.77 92.77 92.77 92.77 213.86 26.67 8.76	82.08 82.08 82.08 78.86 3.42	12.22 12.22 12.22 34.76 3.08	I to UNEs.(N	ion-recurrin	31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27	3.92 3.92 3.92	3.93
NOTE: New ELS available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhili, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Pirst 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone	pelow ex which a he GA P	ccept Switch As Is Chare converted to UNESC Order. (No Switch UNCVX UNCVX UNCVX UNCVX UNCVX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX	UEAL2	witch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5	193.82 193.82 193.82 193.82 295.39 123.37	92.77 92.77 92.77 213.86 26.67	82.08 82.08 82.08 78.86	12.22 12.22 12.22 34.76	i to UNEs.(N	lon-recurrin	31.27 31.27 31.27	31.27 31.27 31.27	3.92	3.9
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Service of the Service of the Service of the Service of Servic	pelow ex which a he GA P	ccept Switch As Is Chare converted to UNE SC Order.(No Switch UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509	193.82 193.82 193.82 193.82 295.39 123.37 12.15	92.77 92.77 92.77 92.77 213.86 26.67 8.76	82.08 82.08 82.08 78.86 3.42	12.22 12.22 12.22 34.76 3.08	I to UNEs.(N	ion-recurrin	31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27	3.92 3.92 3.92	3.9 3.9 3.9
NOTE: New ELS available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the NOTE: In STANSPORT (EEL First 2-Wire VG Loop(SL2) in a DS1 Interoffice Transport Combination - Zone  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont Voice Grade COC1 - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice	pelow ex which a he GA P: ) 1 2 3	coept Switch As Is CT are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX	uEAL2	30.32 61.93 0.3367 81.45 121.5 0.7509 17.65	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82	92.77 92.77 92.77 213.86 26.67 8.76 92.77	82.08 82.08 82.08 78.86 3.42 82.08	12.22 12.22 12.22 34.76 3.08 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9
NOTE: New ELLs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements ply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements ply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements ply to ordinarily combined network elements per the Prist 2-Wire VG Loop (SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Grade Loop (SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor DS1 Channelization System Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop (SL2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop (SL2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop (SL2) in the same DS1 Interoffice Transport Combination - Zone :	pelow ex which a he GA P	are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX	ueage.  Trates. A S As Is Chair  UEAL2  UEAL2  UEAL2  1L5XX  U1TF1  MQ1  1D1VG  UEAL2  UEAL2  UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82	92.77 92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77	82.08 82.08 82.08 78.86 3.42	12.22 12.22 12.22 12.22 34.76 3.08	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92	3.9 3.9 3.9
NOTE: New ELLs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Prist 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Montl Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the sa	pelow ex which a he GA P: ) 1 2 3	coept Switch As Is CT are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX	uEAL2	30.32 61.93 0.3367 81.45 121.5 0.7509 17.65	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82	92.77 92.77 92.77 213.86 26.67 8.76 92.77	82.08 82.08 82.08 78.86 3.42 82.08	12.22 12.22 12.22 34.76 3.08 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9
NOTE: New ELLs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Prist 2-Wire VG Edepth State S	pelow ex which a he GA P: ) 1 2 3	coept Switch As Is C Var Converted to UNE SC Order. (No Switch UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	uerge.  Frates. A S As Is Chair  UEAL2  UEAL2  UEAL2  UEAL2  ULEAL2  UEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 193.82	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77	82.08 82.08 82.08 78.86 3.42 82.08 82.08	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Prist 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mille per mor Interoffice Transport - Dedicated - DS1 combination - Per Mille per mor DS1 Channelization System Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Mont Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone :	pelow ex which a he GA P: ) 1 2 3	are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX	ueage.  Trates. A S As Is Chair  UEAL2  UEAL2  UEAL2  1L5XX  U1TF1  MQ1  1D1VG  UEAL2  UEAL2  UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82	92.77 92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77	82.08 82.08 82.08 78.86 3.42 82.08	12.22 12.22 12.22 34.76 3.08 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92	3.9
NOTE: New ELLs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Prist 2-Wire VG Edepth State S	pelow ex which a he GA Pr ) 1 2 3	coept Switch As Is C Var Converted to UNE SC Order. (No Switch UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	uerge.  Frates. A S As Is Chair  UEAL2  UEAL2  UEAL2  UEAL2  ULEAL2  UEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 193.82	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77	82.08 82.08 82.08 78.86 3.42 82.08 82.08	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22	I to UNEs.(N	ion-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the State NOTE: In Georgia, the NOTE: In Georgia, the State NOTE: In Georgia, the NOTE: In Georgia, the State NOTE: In Georgia, the NOTE: In Georgia, the State NoTE: In Georgia, the NOTE: In Georgia, the State NoTE: In Georgia, the NOTE: In Georgia, the State NoTE: In Georgia, the NOTE: In Georgia, the State NoTE: In Georgia, the NOTE: In Georgia, the State NoTE: In Georgia, the NOTE: In Georgia, the NoTE: In Georgia, the	pelow ex which a he GA Pr ) 1 2 3	coept Switch As Is CV SC order.(No Switch UNCVX UNCVX UNCVX UNCVX UNCTX UNCTX UNCTX UNCTX UNCTX UNCTX UNCYX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 193.82 11.17	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77 8.76	82.08 82.08 82.08 78.86 3.42 82.08 82.08 14.14	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22 12.22 14.14	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Prist 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont - Voice Grade COCI - DS1 To DS0 Interface - Per Montl - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport - Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport - Combination - Zone - Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport - Combination - Zone - S	pelow ex which a he GA Pr ) 1 2 3	coept Switch As Is C Var Converted to UNE SC Order. (No Switch UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	uerge.  Frates. A S As Is Chair  UEAL2  UEAL2  UEAL2  UEAL2  ULEAL2  UEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2  ULEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 193.82	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77	82.08 82.08 82.08 78.86 3.42 82.08 82.08	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, and the NOTE: In Georgia (EEL) and State of Transport Combination - Zone Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - DS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor Interoffice Grane COCI - DS1 To Ds0 Interface - Per Montl  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Interoffice Transport Combination - Interoffice Transport Combination - Interoffice Transport Combination - Interoffice Transport Combination - Interoffice Transport C	which a same of the control of the c	are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509	193.82 193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77 8.76 11.7	82.08 82.08 82.08 78.86 3.42 82.08 82.08 14.14	12.22 12.22 12.22 12.22 12.22 12.22 12.22 14.14	I to UNEs.(N	ion-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9 3.9
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Post of the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Post of the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Post of the NOTE: In Georgia, and the NOTE: In Georgia and Interoffice Transport (Post of Transport Combination - Zone Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont Voice Grade COCI - DS1 To Ds0 Interface - Per Mont Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	pelow ex which a he GA Pr ) 1 2 3	are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 193.82 11.17	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77 8.76	82.08 82.08 82.08 78.86 3.42 82.08 82.08 14.14	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22 12.22 14.14	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, and the NOTE: In Georgia Combination - Zone Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 tombination - Per Mile per mor Interoffice Transport - Dedicated - DS1 tombination - Per Mile per mor Interoffice Transport - DS1 To DS0 Interface - Per Mont!  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -	which a house GA P	are converted to UNE SC order (No Switch UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX	uEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509  24.36  41.85	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 8.76 11.7 92.77	82.08 82.08 82.08 76.86 3.42 82.08 82.08 14.14	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22 14.14	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9 3.9 3.9
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Polician of the Note: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Polician of the Note: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Polician of the Note: Interoffice Transport Combination - Zone:    First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	which a same of the control of the c	coept Switch As Is CV SC Order, (No Switch UNCVX UNCVX UNCVX UNCVX UNCTX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509  24.36  41.85	193.82 193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77 8.76 11.7	82.08 82.08 82.08 78.86 3.42 82.08 82.08 14.14	12.22 12.22 12.22 12.22 12.22 12.22 12.22 14.14	I to UNEs.(N	ion-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9 3.9
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, and the NOTE: In Georgia Combination - Zone Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 tombination - Per Mile per mor Interoffice Transport - DS1 Tombination - Zone Substitute Voice Grade CoCl - DS1 To DS0 Interface - Per Montl - Substitute Transport - Combination - Zone :  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport - Combination - Zone :  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport - Combination - Zone :  Voice Grade COCl - DS1 to DS0 Channel System combination - per mon  Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2  First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Co	which a house GA P	are converted to UNE SC order (No Switch UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX	uEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509  24.36  41.85	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 8.76 11.7 92.77	82.08 82.08 82.08 76.86 3.42 82.08 82.08 14.14	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22 14.14	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9 3.9 3.9 3.9
NOTE: New ELS available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined not apply to ordinarily combined not apply to ordinarily combined not apply to ordinarily combination - Zone Interoffice Transport Combination - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mont	which a house GA P	coept Switch As Is Cf are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNCTX UNCTX UNCTX UNCTX UNCTX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4 UEAL4 UEAL4	30.32 61.93 0.7509 17.65 30.32 61.93 0.7509 24.36 41.85	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 8.76 11.7	82.08 82.08 82.08 76.86 3.42 82.08 82.08 82.08 82.08 82.08	12.22 12.22 12.22 34.76 3.08 12.22 12.22 14.14	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9 3.9 3.9
NOTE: New ELS available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combination - Zone Interoffice Transport Combination - Zone Interoffice Transport Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mont	which a house GA P	Comparison   Com	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509  24.36  41.85  85.47  0.3367  81.45	193.82 193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77	82.08 82.08 82.08 76.86 3.42 82.08 82.08 14.14 82.08 82.08	12.22 12.22 12.22 12.22 12.22 12.22 12.22 14.14 12.22 12.22 12.22 12.22 12.22	I to UNEs.(N	ion-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9
NOTE: New ELS available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, and the NOTE: In Section 1 (Internation - Zone) and Interoffice Transport (Interoffice Transport ordinarion - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor DS1 Channelization system Per Mont - Voice Grade CCC1 - DS1 To DS0 Interface - Per Montl - Voice Grade CCC1 - DS1 To DS0 Interface - Per Montl - Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone :  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone :  Voice Grade CCC1 - DS1 to DS0 Channel System combination - per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	which a house GA P	are converted to UNE SC order (No Switch UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX	uEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	Switch As Is Ch rge.)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509  24.36  41.85	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 8.76 11.7 92.77	82.08 82.08 82.08 76.86 3.42 82.08 82.08 14.14	12.22 12.22 12.22 12.22 34.76 3.08 12.22 12.22 14.14	I to UNEs.(N	ion-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9
NOTE: New ELS available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mont	which a house GA P	coept Switch As Is Cf are converted to UNE SC order.(No Switch UNCVX UNCVX UNCVX UNCTX UNCTX UNCTX UNCTX UNCTX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4 UEAL4 UEAL4	30.32 61.93 0.7509 17.65 30.32 61.93 0.7509 24.36 41.85	193.82 193.82 193.82 193.82 295.39 123.37 12.15 193.82 193.82 12.15 11.17	92.77 92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 8.76 11.7	82.08 82.08 82.08 76.86 3.42 82.08 82.08 82.08 82.08 82.08	12.22 12.22 12.22 34.76 3.08 12.22 12.22 14.14	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9 3.9 3.9 3.9
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates be NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Prist 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont Voice Grade COCI - DS1 To DS0 Interface - Per Montl Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone : Voice Grade COCI - DS1 to DS0 Channel System combination - per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Com	which a house GA P	coept Switch As Is CF SC order, (No Switch UNCVX UNCVX UNCVX UNCVX UNCTX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4 .)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509  24.36  41.85  85.47  0.3367  81.45  121.5	193.82 193.82 193.82 193.82 193.82 193.82 193.82 193.82 193.82 11.17	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77	82.08 82.08 82.08 76.86 3.42 82.08 82.08 14.14 82.08 82.08	12.22 12.22 12.22 12.22 12.22 12.22 12.22 14.14 12.22 12.22 12.22 12.22 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.9 3.9 3.9 3.9 3.9 3.9 3.9	
NOTE: New ELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Mia NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates b NOTE: In all states, EEL network elements shown below also apply to currently combined facilities NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the NOTE: In Georgia, the EEL network elements apply to ordinarily combined network elements per the Per State St	which a house GA P	coept Switch As Is CF SC order, (No Switch UNCVX UNCVX UNCVX UNCVX UNCTX	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4 .)  17.65  30.32  61.93  0.3367  81.45  121.5  0.7509  17.65  30.32  61.93  0.7509  24.36  41.85  85.47  0.3367  81.45  121.5	193.82 193.82 193.82 193.82 193.82 193.82 193.82 193.82 193.82 11.17	92.77 92.77 92.77 213.86 26.67 8.76 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77 92.77	82.08 82.08 82.08 76.86 3.42 82.08 82.08 14.14 82.08 82.08	12.22 12.22 12.22 12.22 12.22 12.22 12.22 14.14 12.22 12.22 12.22 12.22 12.22	I to UNEs.(N	lon-recurrin	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27 31.27	3.92 3.92 3.92 3.92 3.92 3.92 3.92 3.92	3.3 3.3 3.3 3.3 3.3 3.3 3.3	

Page 9 of 23

Add	ditional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport												Т
Com	mbination - Zone :	2	UNCVX	UEAL4	41.85	193.82	92.77	82.08		31.27	31.27	3.92	
	ditional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	3	UNCVX	UEAL4	85.47	193.82	92.77	82.08	12.22	31.27	31.27	3.92	
	ce Grade COCI - DS1 to DS0 Channel System combination - per mon		UNCVX	1D1VG	0.7509	12.15	8.76					0.00	
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charc		UNC1X	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	
			ONCIA	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.32	
	S EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT	(EEL)											
Zone	at 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - lee 1	1	UNCDX	UDL56	27.5	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
	tt 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -												
Zone	ne 2 st 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	2	UNCDX	UDL56	47.24	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
Zone	ne 3	3	UNCDX	UDL56	96.48	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
Inter	eroffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.3367								4
Inte	eroffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor		UNC1X	U1TF1	81.45	295.39	213.71	78.86	34.78	31.27	31.27	3.92	
Cha	annelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	121.5	123.37	26.67	3.42	3.08				
	U-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb: ditional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDX	1D1DD	1.6	12.15	8.76						-
Com	mbination - Zone	1	UNCDX	UDL56	27.5	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
	ditional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	UNCDX	UDL56	47.24	193.82	92.77	82.08	12.22	31.27	31.27	3.92	
Add	ditional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport												1
	mbination - Zone (	3	UNCDX	UDL56	96.48	193.82	92.77	82.08	12.22	31.27	31.27	3.92	_
64kb	U-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-bs)		UNCDX	1D1DD	1.6	13.14	9.42						
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	+
4-WIRE 64 KBPS	S EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT	(EEL)											
	st 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -												
Zone	te 1  tt 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	1	UNCDX	UDL64	27.5	193.82	92.77	92.08	12.22	31.27	31.27	3.92	-
Zone	ne 2	2	UNCDX	UDL64	47.24	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
First	st 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	3	UNCDX	UDL64	96.48	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
	eroffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	3	UNC1X	1L5XX	0.3367	193.62	92.11	92.00	12.22	31.27	31.21	3.92	+
Inte	eroffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor annelization - Channel System DS1 to DS0 combination Per Mor		UNC1X UNC1X	U1TF1 MQ1	81.45 121.5	295.39 123.37	213.71 26.67	78.86 3.42	34.78 3.08	31.27	31.27	3.92	+
OCL	U-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-												T
64kb	bs) ditional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDX	1D1DD	1.6	12.15	8.76						+
Com	mbination - Zone	1	UNCDX	UDL64	27.5	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
	ditional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	LINORY	LIDIOA	47.04	400.00	00.77	00.00	10.00	04.07	04.07	0.00	
	mbination - Zone : ditional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	UNCDX	UDL64	47.24	193.82	92.77	92.08	12.22	31.27	31.27	3.92	+
Com	mbination - Zone (	3	UNCDX	UDL64	96.48	193.82	92.77	92.08	12.22	31.27	31.27	3.92	
OCU 64kb	U-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		UNCDX	1D1DD	1.6	12.15	8.76						
					1.0								
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	4
4-WIRE DS1 DIG	GITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EE	L)											1
	/ire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	1	UNC1X	USLXX	56.32	348.15	207.55	89.88	28	31.27	31.27	3.92	
	/ire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	3	UNC1X	USLXX	96.73 197.57	348.15 348.15	207.55	89.88 89.88	28 28	31.27 31.27	31.27 31.27	3.92 3.92	+
	/ire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone eroffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	3	UNC1X UNC1X	USLXX 1L5XX	0.3367	346.15	207.55	69.66	28	31.27	31.27	3.92	+
	eroffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo		UNC1X	U1TF1	81.45	295.39	213.86	78.86	34.76	31.27	31.27	3.92	
					01.40								T
Non	nrecurring Currently Combined Network Elements Switch -As-Is Chare		UNC1X	UNCCC		11.17	11.7	14.14	13.91	31.27	31.27	3.92	+
	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EE												
	t DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	56.32	348.15	92.77	89.88	12.22	31.27	31.27	3.92	_
	t DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X UNC1X	USLXX	96.73 197.57	348.15 348.15	92.77 92.77	89.88 89.88	12.22 12.22	31.27 31.27	31.27 31.27	3.92 3.92	4
	at DS1Loop in DS3 Interoffice Transport Combination - Zone Proffice Transport - Dedicated - DS3 combination - Per Mile Per Mor	3	UNC1X UNC3X	1L5XX	7.67	340.15	92.11	09.60	12.22	31.21	31.21	3.92	+
	eroffice Transport - Dedicated - DS3 - Facility Termination per mor		UNC3X	U1TF3	982.87	507.48	221.03	54.07	28.12	31.27	31.27	3.92	+
	3 to DS1 Channel System combination per mon		UNC3X	MQ3	233.25	323.12	127.13	37.72	19.37	01.27	01.27	0.02	7
DS3	3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	13.64	12.16	8.77						1
Add	ditional DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	56.32	348.15	92.77	89.88	12.22	31.27	31.27	3.92	
Add	ditional DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	96.73	348.15	92.77	89.88	12.22	31.27	31.27	3.92	J
rida		3	LINICAV	LICLVV	197.57	348.15	92.77	89.88	12.22	31.27	31.27	3.92	Т
Add	ditional DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interface Unit (DS1 COCI) combination per montl	3	UNC1X UNC1X	USLXX UC1D1	197.37	346.13	32.11	09.00	12.22	31.27	31.21	3.92	_1

Nonrecurring Currently Combined Network Elements Switch -As-Is Charç			UNC3X	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	1
2-WIRE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT	(EEL)												+
2-Wire VG Loop used with 2-wire VG Interoffice Transport Combination - Zone	,,	1	UNCVX	UEAL2	17.65	193.82	92.77	82.08	12.22	31.27	31.27	3.92	+
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	1	2	UNCVX	UEAL2	30.32	193.82	92.77	82.08	12.22	31.27	31.27	3.92	t
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone		3	UNCVX	UEAL2	61.93	193.82	92.77	82.08	12.22	31.27	31.27	3.92	1
Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor			UNCVX	1L5XX	0.0165								T
Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility													T
Termination per month			UNCVX	U1TV2	26.12	149.36	85.88	78.02	34.89	31.27	31.27	3.92	
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			UNCVX	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	_
4-WIRE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT	(FFL)												+
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	(LLL)	1	UNCVX	UEAL4	24.36	193.82	92.77	82.08	12.22	31.27	31.27	3.92	+
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone		2	UNCVX	UEAL4	41.85	193.82	92.77	82.08	12.22	31.27	31.27	3.92	+
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone		3	UNCVX	UEAL4	85.47	193.82	92.77	82.08	12.22	31.27	31.27	3.92	+
Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor			UNCVX	1L5XX	0.0165		· ·						Ť
Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility													Ť
Termination per month			UNCVX	U1TV4	22.89	149.36	85.88	78.02	34.89	31.27	31.27	3.92	
Nonrecurring Currently Combined Network Elements Switch -As-Is Charç			UNCVX	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	+
DS3 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)													$\dagger$
High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor			UNC3X	1L5ND	12.75								T
High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per													T
month			UNC3X	UE3PX	419.1	387.7	258.22						1
Interoffice Transport - Dedicated - DS3 - Per Mile per mon		-	UNC3X	1L5XX	7.67								4
Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per m	O		UNC3X	U1TF3	982.87	610.38	249.26	72.2	39.88	31.27	31.27	3.92	
	-		S.100A	51113	002.07	0.0.00	2.5.20		55.56	31.27	U1.21	0.02	$^{+}$
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			UNC3X	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)						1							+
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor		-	UNCSX	1L5ND	12.75								+
High Capacity Unbundled Local Loop - STS1 combination - Fer Mile per mor			UNCOX	TESIND	12.75								+
month			UNCSX	UDLS1	433.21	387.7	258.22						
Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor			UNCSX	1L5XX	7.67	307.7	230.22						+
Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mo			UNCSX	U1TFS	959.46	610.38	249.26	72.2	39.88	31.27	31.27	3.92	+
Nonrecurring Currently Combined Network Elements Switch -As-Is Charc			UNCSX	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	
											J.,		İ
2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)													
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		1	UNCNX	U1L2X	21.15	193.82	92.77	82.08	12.22	31.26	31.26	3.91	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		2	UNCNX	U1L2X	36.32	193.82	92.77	82.08	12.22	31.27	31.27	3.92	_
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		3	UNCNX	U1L2X	74.19	193.82	92.77	82.08	12.22	31.27	31.27	3.92	_
Interoffice Transport - Dedicated - DS1 combination - Per Mi			UNC1X	1L5XX	0.3367								4
Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per mor			UNC1X	U1TF1	81.45	295.39	213.86	78.86	34.76	31.27	31.27	3.92	_
Channelization - Channel System DS1 to DS0 combination - per mor			UNC1X UNCNX	MQ1 UC1CA	121.5	123.37	26.67	3.42	3.08				+
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon			UNCNX	UCTCA	3.43	12.15	8.76		+				+
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon		1	UNCNX	U1L2X	21.15	193.82	92.77	82.08	12.22	31.27	31.27	3.92	
													T
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon		2	UNCNX	U1L2X	36.32	193.82	92.77	82.08	12.22	31.27	31.27	3.92	+
Additional 2-wire IDSN Loop in same DS1Intereffice Transport Combination 7-		3	UNCNX	U1L2X	74 19	193.82	92 77	82.08	12.22	31.27	31.27	3.92	
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per mon	+	3	UNCNX	UC1CA	3.43	193.82	8.76	82.08	12.22	31.2/	31.21	3.92	+
and the second s					10								$^{\dagger}$
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			UNC1X	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	1
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT	r (EEL)	+ +				<del> </del>							+
First DS1 Loop in STS1 Interoffice Transport Combination - Zone	. ()	1	UNC1X	USLXX	56.32	348.15	207.55	89.88	28	31.27	31.27	3.92	+
First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1	2	UNC1X	USLXX	96.73	348.15	207.55	89.88	28	31.27	31.27	3.92	+
First DS1 Loop in STS1 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	197.57	348.15	207.55	89.88	28	31.27	31.27	3.92	+
Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor			UNCSX	1L5XX	7.67	2.3.10				31.21	27.2		$^{+}$
Interoffice Transport - Dedicated - STS1 combination - Facility Terminati			UNCSX	U1TFS	959.46	610.38	249.52	72.52	39.88	31.27	31.27	3.92	T
STS1 to DS1 Channel System conbination per mon			UNCSX	MQ3	233.25	220.23	98.9	19.27	7.61				╛
DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1	13.64	12.16	8.77						T
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	56.32	348.15	207.55	89.88	28	31.27	31.27	3.92	╛
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	96.73	348.15	207.55	89.88	28	31.27	31.27	3.92	╛
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	197.57	348.15	207.55	89.88	28	31.27	31.27	3.92	
DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1	13.64	12.16	8.77						
Nonrocurring Currently Combined Naturals Florence Cuites As In City			LINCOV	LINICOC		44.47	44.7	11.11	1444	24.07	24 07	2.00	
Nonrecurring Currently Combined Network Elements Switch -As-Is Charc	-		UNCSX	UNCCC		11.17	11.7	14.14	14.14	31.27	31.27	3.92	+
						1						+	+
4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL	.)						l l						

1 1	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		2 UNCDX	UDL56	47.24	193.82	92.77	82.08	12.22		31.27	31.27	3.92	3.92
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		3 UNCDX	UDL56	96,48	193.82	92.77	82.08	12.22		31.27	31.27	3.92	3.92
						193.82	92.77	82.08	12.22		31.27	31.27	3.92	3.92
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per M		UNCDX	1L5XX	0.0165									
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminati		UNCDX	U1TD5	18.04	149.36	86	78.02	34.89		31.27	31.27	3.92	3.92
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charc		UNCDX	UNCCC		11.17	11.7	14.14	14.14		31.27	31.27	3.92	2.02
	Nonrecurring Currently Combined Network Elements Switch -As-is Chart		UNCDX	UNCCC		11.17	11.7	14.14	14.14		31.27	31.27	3.92	3.92
4-WIRE 6	64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)													
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		1 UNCDX	UDL64	27.5	193.82	92.77	82.08	12.22		31.27	31.27	3.92	3.92
					47.24									
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone			UDL64		193.82	92.77	82.08	12.22		31.27	31.27	3.92	3.92
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		3 UNCDX	UDL64	96.48	193.82	92.77	82.08	12.22		31.27	31.27	3.92	3.92
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M		UNCDX	1L5XX	0.0165									
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminati		UNCDX	U1TD6	18.04	149.36	86	78.02	34.89		31.27	31.27	3.92	3.92
							7.		0.1.00		U.I.		0.00	0.00
						_								
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCDX	UNCCC		0	11.7	14.14	14.14		31.27	31.27	3.92	3.92
ADDITIONAL NETWO	ORK ELEMENTS													
		L												
	sed as a part of a currently combined facility, the non-recurrng charges do not apply, b													
When us	sed as ordinarilty combined network elements in Georgia, the non-recurring charges ag	ply and th	e Switch As Is Charge	does not.										
	The page of the pa			+										
Access t	to DCS - Customer Reconfiguration (FlexServ)													
								1	1					
Node (St	ynchroNet)													
	Node per month		UNCDX	UNCNT	15.43	<del> </del>	<b> </b>							
<b>+</b>	Node per monu		UNCDX	UNCIVI	15.43			<b> </b>						
Nonrecu	arring Currently Combined Network Elements "Switch As Is" Charge (One applies to each	ch combin	ation)											
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion		T											-
			UNCVX	UNCCC		11.17	11.7	14.14	14.14		31.27	31.27	3.92	3.92
<b>+</b>	Charge		UNCVX	UNCCC		11.17	11.7	14.14	14.14		31.21	31.27	3.92	3.92
	56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion													
	Charge		UNCDX	UNCCC		11.17	11.7	14.14	14.14		31.27	31.27	3.92	3.92
	~													
	DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		UNC1X	UNCCC		11.17	44.7	4444	4444		31.27	31.27	2.02	2.02
	DST Interoffice Channel used in a COMBINATION - Switch as is Conversion Char		UNCIX	UNCCC		11.17	11.7	14.14	14.14		31.27	31.27	3.92	3.92
	DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		UNC3X	UNCCC		11.17	11.7	14.14	14.14		31.27	31.27	3.92	3.92
	STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion													
			LINIOOV	1111000		44.47	44.7				04.07	04.07	0.00	0.00
	Charge		UNCSX	UNCCC		11.17	11.7	14.14	14.14		31.27	31.27	3.92	3.92
NOTE: L	ocal Channel - Dedicated Transport - minimum billing period - Below DS3=one month,	DS3 and a	bove=four months											
	DODT SYSTEMS													
LODED ATIONIAL CLIDE														
OPERATIONAL SUPP														
	Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the:	state speci	ific electronic service o	rdering char	ges as order	ed by the State (	Commissions							
NOTE: (1	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the						Commissions							
NOTE: (1 NOTE: (1	Blectronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the DO Continued: The electronic service ordering charge currently contained in this rate exhibition.	it is the Be	ellSouth regional electr	onic service	ordering cha	arge		ordering of	arge					
NOTE: (1 NOTE: (1 NOTE: (1	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the . 1) Continued: The electronic service ordering charge currenly contained in this rate exhib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the	it is the Be e electron	ellSouth regional electr ic service ordering cha	onic service	ordering cha	arge		e ordering ch	arge.					
NOTE: (1 NOTE: (1 NOTE: (1	Blectronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the DO Continued: The electronic service ordering charge currently contained in this rate exhibition.	it is the Be e electron	ellSouth regional electr ic service ordering cha	onic service	ordering cha	arge		e ordering cl	arge.					
NOTE: (1 NOTE: (1 NOTE: (1	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the . 1) Continued: The electronic service ordering charge currenly contained in this rate exhib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the	it is the Be e electron	ellSouth regional electr ic service ordering cha	onic service	ordering cha	arge		e ordering cl	arge.					
NOTE: (1 NOTE: (1 NOTE: (1	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the off-should contact its contract negotiator if it prefers the off-should contained in this rate exhibits 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the should be contained in the state of Florida, to be billed on a per lease of the contained in the state of Florida, to be billed on a per lease of the contained in the state of Florida.	it is the Be e electron	ellSouth regional electr ic service ordering cha	onic service	ordering cha	arge		e ordering cl	arge.					
NOTE: (1 NOTE: (1 NOTE: (1	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shib. I Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per lacet of the contract of the	it is the Be e electron	ellSouth regional electr ic service ordering cha	onic service rges, or CLE	ordering cha	rge ct the regional e		e ordering cl	arge.					
NOTE: (1 NOTE: (1 NOTE: (1	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the off-should contact its contract negotiator if it prefers the off-should contained in this rate exhibits 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the should be contained in the state of Florida, to be billed on a per lease of the contained in the state of Florida, to be billed on a per lease of the contained in the state of Florida.	it is the Be e electron	ellSouth regional electr ic service ordering cha	onic service	ordering cha	arge		e ordering cl	arge.					
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service							
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shib. I Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per lacet of the contract of the	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			I Office, refer to Internet	Website:			
NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate shib. Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left the contraction of the contractive interfaces (Regional).  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional).	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			I Office, refer to Internet	t Website:			
NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	t Website:			
NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate shib. Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left the contraction of the contractive interfaces (Regional).  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional).	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			l Office, refer to Internet	Website:			
NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate shib. Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left the contraction of the contractive interfaces (Regional).  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional).	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2 The "Zon http://ww	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left the contraction of	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2 The "Zon http://ww	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate exhibit Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per lacet in the state of Florida, the	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			I Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shib. I Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a perill Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  LEXCHANGE SWITCHING(PORTS)	e electron LSR basis	ellSouth regional electric service ordering cha	onic service rges, or CLE SOMEC	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2 NOTE: (2  The "Zon http://ww  UNBUNDLED LOCAL	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate shib. I Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left the contraction of the contractive interfaces (Regional)  The shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  The contractive interfaces (Regional)  The shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  The contractive interfaces (Regional)  The shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec  SOMEC  SOMEC  Zones. To	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2 NOTE: (2  The "Zon http://ww  UNBUNDLED LOCAL	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate shib. I Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left the contraction of the contractive interfaces (Regional)  The shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  The contractive interfaces (Regional)  The shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  The contractive interfaces (Regional)  The contractive interfaces (Regional)  The contractive interfaces (Regional)  The contractive interfaces (Regional)	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec  SOMEC  SOMEC  Zones. To	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOTE: (2 NOTE: (2  The "Zon http://ww  UNBUNDLED LOCAL	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shib. I Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a perill Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  LEXCHANGE SWITCHING(PORTS)	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec  SOMEC  SOMEC  Zones. To	ordering cha C-1 may ele	arge ct the regional e	lectronic service			I Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shrib. Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a period of the control of the con	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec  SOMEC  SOMEC  Zones. To	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	: Website:			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate exhibit Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  net shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features woolce GRADE LINE PORT RATES (RES)	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec SOMEC SOMEC E Zones. To	ordering cha C-1 may ele	atge ct the regional e	lectronic service	e Designatio	ons by Centra	I Office, refer to Internet				
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shrib. Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a period of the control of the con	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec  SOMEC  SOMEC  Zones. To	ordering cha C-1 may ele	arge ct the regional e	lectronic service			Il Office, refer to Internet	: Website:	8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate exhibit Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  net shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features woolce GRADE LINE PORT RATES (RES)	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec SOMEC SOMEC E Zones. To	ordering cha C-1 may ele	atge ct the regional e	lectronic service	e Designatio	ons by Centra	I Office, refer to Internet		8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left the control of the co	it is the Be e electron LSR basis	ellSouth regional electric service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering ret	somec ome Some Some Some Some Some Some	ordering character	arge ct the regional e	lectronic service	e Designatio	ons by Centra	Il Office, refer to Internet	18.14			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the 1) Continued: The electronic service ordering charge currently contained in this rate exhibit Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per left (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  net shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features woolce GRADE LINE PORT RATES (RES)	it is the Be e electron LSR basis	ellSouth regional electric service ordering cha	somec SOMEC SOMEC E Zones. To	ordering cha C-1 may ele	atge ct the regional e	lectronic service	e Designatio	ons by Centra	I Office, refer to Internet		8.06 8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the of 1) Continued: The electronic service ordering charge currently contained in this rate shib. It concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a perillectronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)    Regional	it is the Be e electron LSR basis	ellSouth regional electric service ordering challing chal	onic service gges, or CLE SOMEC SOMEC E Zones. To	ordering chackers and selected may be over the control of the cont	3.5 aphically Deave	raged UNE Zon	e Designation	4.38	Il Office, refer to Internet	18.14	8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate exhib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  nee* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering ret	somec ome Some Some Some Some Some Some	ordering character	arge ct the regional e	lectronic service	e Designatio	ons by Centra	I Office, refer to Internet	18.14			
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate exhib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  nee* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challing chal	onic service gges, or CLE SOMEC SOMEC E Zones. To	ordering chackers and selected may be over the control of the cont	3.5 aphically Deave	raged UNE Zon	e Designation	4.38	Il Office, refer to Internet	18.14	8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port -Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Caller Line Ports outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Caller Line Ports - VIII outgoing only - Re	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering retrieval of the ordered using retrieval or the ordered using retrieval or the ordered using retrieval ordered using retrieval or the ordered using retrie	onic service rges, or CLE SOMEC SOMEC E Zones. To	ordering chackers and selected may be over the control of the cont	3.5 aphically Deave	raged UNE Zon  16.43  16.43	e Designation 4.38 4.38 4.38	4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14	8.06 8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  neo* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port-Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challing chal	onic service gges, or CLE SOMEC SOMEC E Zones. To	ordering chackers and selected may be over the control of the cont	3.5 aphically Deave	raged UNE Zon	e Designation	4.38	Il Office, refer to Internet	18.14	8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate exhib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne" shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Call ID - Res.	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordered using retained to the ordered using retained its service ordered using retained its service or ordere	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRL UEPRC UEPRO UEPAS	ordering chains of the control of th	3.5 aphically Deave	16.43 16.43 16.43	4.38 4.38 4.38	4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14	8.06 8.06	10.39 10.39	10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port -Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Caller Line Ports outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Caller Line Ports - VIII outgoing only - Re	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challing chal	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRL UEPRC UEPRO UEPAS UEPAG	ordering checkers and selected may be over the control of the cont	3.5 aphically Deave  16.43 16.43 16.43 16.43	raged UNE Zon  16.43  16.43	4.38 4.38 4.38	4.38 4.38 4.38 4.38	Il Office, refer to Internet	18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06	10.39	10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  LEXCHANGE SWITCHING(PORTS)  apperors  Although the Port Rate includes all available features in GA & TN, the desired features who in the section of the state of Florida, and the state of Florida, to be billed on a per l  WOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port- Re:  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL)  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL)	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challing chal	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRL UEPRC UEPRO UEPAS UEPAG	ordering checkers and selected may be over the control of the cont	3.5 aphically Deave  16.43 16.43 16.43 16.43	16.43 16.43 16.43	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06	10.39 10.39 10.39	10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate exhib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne" shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Call ID - Res.	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordered using retained to the ordered using retained its service ordered using retained its service or ordere	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRL UEPRC UEPRO UEPAS	ordering chains of the control of th	3.5 aphically Deave	16.43 16.43 16.43	4.38 4.38 4.38	4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14	8.06 8.06	10.39 10.39	10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  LEXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features where the ports -2-Wire Analog Line Port-Re:  Exchange Ports -2-Wire Analog Line Port with Caller ID - Re  Exchange Ports -2-Wire Analog Line Port outgoing only - Re  Exchange Ports -2-Wire VG unbundled LA extended local dialing parity Port with Call ID - Res.  Exchange Ports -2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL Exchange Ports -2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL Exchange Ports -2-Wire VG unbundled res, low usage line port with Caller ID (LU Exchange Ports -2-Wire VG unbundled res, low usage line port with Caller ID (LU	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering retrieval to the ordered using	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRC UEPRC UEPRO UEPAS UEPAG UEPAG	crdering chackers are considered as a considered considered as a considered considered as a considered conside	3.5 aphically Deave  16.43 16.43 16.43 16.43 16.43	16.43 16.43 16.43 16.43 16.43	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06	10.39 10.39 10.39	10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port-Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Wire Unbundled La extended local dialing parity Port with Caller ID - Res.  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID (LU  Subsequent Activity	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challing chal	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRL UEPRC UEPRO UEPAS UEPAG	ordering checkers and selected may be over the control of the cont	3.5 aphically Deave  16.43 16.43 16.43 16.43	16.43 16.43 16.43	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06	10.39 10.39 10.39	10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne" shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  EXCHANGE SWITCHING(PORTS)  EXCHANGE SWITCHING(PORTS)  EXCHANGE SWITCHING(PORTS)  Exchange Ports - 2-Wire Analog Line Port Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Call ID - Res.  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU Subsequent Activity	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed in service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordered using retained in the service of the service ordered using retained its service ordered using retained in the service or	somec service rges, or CLE SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOME SOMEC SOME SOME SOME SOME SOME SOME SOME SOME	cordering chains of the control of t	3.5 aphically Deave  16.43 16.43 16.43 16.43 16.43	16.43 16.43 16.43 16.43 16.43	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06 8.06 8.06	10.39 10.39 10.39 10.39 10.39	10.39 10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne" shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  EXCHANGE SWITCHING(PORTS)  EXCHANGE SWITCHING(PORTS)  EXCHANGE SWITCHING(PORTS)  Exchange Ports - 2-Wire Analog Line Port Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Call ID - Res.  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU Subsequent Activity	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering retrieval to the ordered using	somec service rges, or CLE SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOME SOMEC SOME SOME SOME SOME SOME SOME SOME SOME	crdering chackers are considered as a considered considered as a considered considered as a considered conside	3.5 aphically Deave	16.43 16.43 16.43 16.43 16.43	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06 8.06 8.06	10.39 10.39 10.39 10.39 10.39	10.39 10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib. 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  EXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port-Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire Wire Unbundled La extended local dialing parity Port with Caller ID - Res.  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID (LU  Subsequent Activity	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed in service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordered using retained in the service of the service ordered using retained its service ordered using retained in the service or	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRC UEPRC UEPRO UEPAS UEPAG UEPAG	cordering chains of the control of t	16.43 16.43 16.43 16.43 16.43	16.43 16.43 16.43 16.43 16.43 0	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06	10.39 10.39 10.39	10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate exhibit Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed in service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordered using retained in the service of the service ordered using retained its service ordered using retained in the service or	somec service rges, or CLE SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOME SOMEC SOME SOME SOME SOME SOME SOME SOME SOME	cordering chains of the control of t	16.43 16.43 16.43 16.43 16.43	16.43 16.43 16.43 16.43 16.43 0	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06 8.06 8.06	10.39 10.39 10.39 10.39 10.39	10.39 10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  LEXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port-Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Call ID - Res.  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU)  Subsequent Activity  EXES  All Available Vertical Feature  VOICE GRADE LINE PORT RATES (BUS)	dit is the Be e electron. SR basis  Geograph	UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRL UEPRC UEPRO UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAF	cordering checkers and selected may be selecte	3.5 aphically Deave  16.43 16.43 16.43 16.43 16.43 0	16.43 16.43 16.43 16.43 0	4.38 4.38 4.38 4.38 4.38	4.38 4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06 8.06 8.06 8.06	10.39 10.39 10.39 10.39 10.39	10.39 10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate exhibit Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the state of Florida, to be billed on a per laborate of the contract of the	dit is the Be e electron. SR basis  Geograph	ellSouth regional electric service ordering challed in service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordering challed its service ordered using retained in the service of the service ordered using retained its service ordered using retained in the service or	somec service rges, or CLE SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOMEC SOME SOMEC SOME SOME SOME SOME SOME SOME SOME SOME	cordering chains of the control of t	16.43 16.43 16.43 16.43 16.43	16.43 16.43 16.43 16.43 16.43 0	4.38 4.38 4.38	4.38 4.38 4.38 4.38	I Office, refer to Internet	18.14 18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06 8.06 8.06	10.39 10.39 10.39 10.39 10.39	10.39 10.39 10.39 10.39 10.39
NOTE: (1 NOTE: (1 NOTE: (1 NOTE: (2 NOT	1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the: 1) Continued: The electronic service ordering charge currently contained in this rate shib: 1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the 2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per l  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  ne* shown in the sections for stand-alone loops or loops as part of a combination refers to ww.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  LEXCHANGE SWITCHING(PORTS)  ge Ports  Although the Port Rate includes all available features in GA & TN, the desired features w  VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port-Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Call ID - Res.  Exchange Ports - 2-Wire VG unbundled Louisiana Area Plus with Caller ID - Res (RL  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU)  Subsequent Activity  EXES  All Available Vertical Feature  VOICE GRADE LINE PORT RATES (BUS)	dit is the Be e electron. SR basis  Geograph	UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR	onic service rges, or CLE SOMEC SOMEC E Zones. To UEPRL UEPRC UEPRO UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS UEPAF	cordering checkers and selected may be selecte	3.5 aphically Deave  16.43 16.43 16.43 16.43 16.43 0	16.43 16.43 16.43 16.43 0	4.38 4.38 4.38 4.38 4.38	4.38 4.38 4.38 4.38 4.38	Il Office, refer to Internet	18.14 18.14 18.14 18.14 18.14 18.14	8.06 8.06 8.06 8.06 8.06 8.06	10.39 10.39 10.39 10.39 10.39	10.39 10.39 10.39 10.39 10.39

Attachment 2 Exhibit C

# Unbundled Network Elements LOUISIANA

1	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484													
	ID - Bus.	UEPSB	UEPBC	2.2	16.43	16.43	4.38	4.38			18.14	8.06	10.39	10.
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bu	UEPSB	UEPBO	2.2	16.43	16.43	4.38	4.38			18.14	8.06	10.39	10.
	Exchange Ports - 2-Wire Arialog Line Port outgoing only - Bu  Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Caller	UEFSB	UEFBU	2.2	10.43	10.43	4.30	4.36			10.14	0.00	10.39	10.
	ID - Bus.	UEPSB	UEPAX	2.2	16.43	16.43	4.38	4.38			18.14	8.06	10.39	10.
	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B	UEPSB	UEPB1	2.2	16.43	16.43	4.38	4.38			18.14	8.06	10.39	10.
	Exchange Ports - 2-Wire VG unbundled Louisiana Bus Area Calling Port with Caller ID -													
	Bus (BUC)	UEPSB	UEPAA	2.2	16.43	16.43	4.38	4.38			18.14	8.06	10.39	10.
	Subsequent Activity	UEPSB	USASC	0	0	0								
FEATURES	oubsequent Neuvici	OLI OD	00/100											
	All Available Vertical Feature	UEPSB	UEPVF	8.28	0	0					18.14	8.06	10.39	10.
	PORT RATES (DID & PBX)													
	Exchange Ports - 2-Wire DID Port	UEPEX	UEPP2	9.52	238.35	37.44	121.38	7.63			18.14	8.06	10.39	10
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabilii	UEPDD	UEPDD	77.66	403.61	191.17	147.11	4.98			19.99	19.99	19.99	19
	Exchange Ports - 2-Wire ISDN Port (See Notes below	UEPTX UEPSX		11.42	145.38	105.86	95.47	21.45			38.29	38.29	6.65	6
	All Features Offered	UEPTX UEPSX	UEPVF	8.28	0	0								
	nsmission/usage charges associated with POTS circuit switched usage will also apply to circuit s													
NOTE: Acc	sess to B Channel or D Channel Packet capabilities will be available only through BFR/New Busi						he Bona Fid	e Request/Ne	w Business	Request Pr	ocess.			
<b>_</b>	Exchange Ports - 2-Wire ISDN Port Channel Profiles  Exchange Ports - 4-Wire ISDN DS1 Por	UEPTX UEPSX		0	0	0	400.40	40.65			00.40	00.40	7.70	-
	Exchange Ports - 4-Wire ISDN DS1 Por	UEPEX	UEPEX	107.55	407.19	202.89	160.46	40.65			33.18	33.18	7.73	7.
	2-Wire VG Unbundled 2-Way PBX Trunk - Re:	UEPSE	UEPRD	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8.
	2 Wile ve enbanded 2 Way 1 BX Walle No.	02.02	OZ. N.D		10.10	10.10	0.11	0			10.11	0.00	0.01	
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu	UEPSP	UEPPC	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu	UEPSP	UEPPO	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bt 2-Wire Analog Long Distance Terminal PBX Trunk - Bu	UEPSP UEPSP	UEPP1 UEPLD	2.2	16.43 16.43	16.43 16.43	3.77 3.77	3.77 3.77			18.14 18.14	8.06 8.06	8.94 8.94	8
	2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Pc	UEPSP	UEPL2	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled PBX LD Terminal Port	UEPSP	UEPLD	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	1
	2-Wire Vice Unbundled 2-Way PBX Usage Po	UEPSP	UEPXA	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Por	UEPSP	UEPXB	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled PBX LD DDD Terminals Po	UEPSP	UEPXC	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	UEPSP	UEPXD	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc	UEPSP	UEPXE	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Callling P	UEPSP	UEPXK	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling													
	Port	UEPSP	UEPXL	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P	UEPSP	UEPXM	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling	OLF SF	OLI XIVI	2.2	10.43	10.45	5.11	5.77			10.14	0.00	0.34	
	Port Port	UEPSP	UEPXO	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling P	UEPSP	UEPXP	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	UEPSP	UEPXS	2.2	16.43	16.43	3.77	3.77			18.14	8.06	8.94	8
	Subsequent Activity	UEPSP	USASC	0	0	0								
FEATURES	Subsequent Activity	UEFSF	USASC	0	U	U								
	All Available Vertical Feature	UEPSP UEPSE	UEPVF	8.28	0	0					18.14	8.06	8.94	
	PORT RATES (COIN)													
	Exchange Ports - Coin Por			2.5	16.43	16.43	4.15	4.15			18.14	8.06	9.86	9
	nsmission/usage charges associated with POTS circuit switched usage will also apply to circuit s sess to B Channel or D Channel Packet capabilities will be available only through BFR/New Busi								D	Decised De	l			
	tess to b Channel or D Channel Packet capabilities will be available only through BFR/New Busi	less Request Process. 1	Rates for	пе раскет сар	adilities will be d	etermined via t	ne Bona Fio	e Requestine	w business	Request Pr	ocess.			
NOIE: Acc														
	WITCHING, PORT USAGE													
	WITCHING, PORT USAGE													
D LOCAL SV	WITCHING, PORT USAGE  Switching (Port Usage)													
D LOCAL SW	Switching (Port Usage)  End Office Switching Function, Per MOI			0.0021										
D LOCAL SW	Switching (Port Usage)			0.0021 0.0002										
D LOCAL SV	Switching (Port Usage)  End Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL													
D LOCAL SV	Switching (Port Usage)  End Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL  witching (Port Usage) (Local or Access Tandem)			0.0002										
D LOCAL SV	Switching (Port Usage) End Office Switching Function, Per MOL End Office Trunk Port - Shared, Per MOL witching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOL			0.0002										
D LOCAL SV	Switching (Port Usage)  End Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL  witching (Port Usage) (Local or Access Tandem)			0.0002										
End Office S Tandem Sw Common Tr	Switching (Port Usage)  End Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL  vitching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOL  Tandem Trunk Port - Shared, Per MOL  ransport			0.0002 0.0008 0.0003										
End Office S  Tandem Sw  Common Tr	Switching (Port Usage) End Office Switching Function, Per MOL End Office Trunk Port - Shared, Per MOL witching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOL Tandem Trunk Port - Shared, Per MOL Tansport Common Transport - Per Mile, Per MOL			0.0002 0.0008 0.0003 0.000083										
End Office S Tandem Sw Common Tr	Switching (Port Usage)  End Office Switching Function, Per MOL  End Office Trunk Port - Shared, Per MOL  vitching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOL  Tandem Trunk Port - Shared, Per MOL  ransport			0.0002 0.0008 0.0003										
End Office S  Tandem Sw  Common Tr	Switching (Port Usage)  End Office Switching Function, Per MOL End Office Trunk Port - Shared, Per MOL  intching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOL Tandem Trunk Port - Shared, Per MOL  ransport  Common Transport - Per Mile, Per MOL  Common Transport - Facilities Termination Per MO			0.0002 0.0008 0.0003 0.000083										
End Office S Tandem Sw Common Tr	Switching (Port Usage) End Office Switching Function, Per MOL End Office Trunk Port - Shared, Per MOL witching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOL Tandem Trunk Port - Shared, Per MOL Tansport Common Transport - Per Mile, Per MOL			0.0002 0.0008 0.0003 0.000083										

	onrecurring charges shall be those identified in the Nonrecurring - Currently Combined sect									
	, ,									
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)									
	pop Combination Rates									
	2-Wire VG Loop/Port Combo - Zone	1			16.6		<del>                                     </del>			
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :	3			26.69 51.85					
	2-Wile VG Loop/Port Combo - Zone	3			51.85					+
UNE Loop R	ates									-
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRX	UEPLX	14.05					-
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRX	UEPLX	24.14					
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRX	UEPLX	49.3					
	e Grade Line Port Rates (Res)							-	_	
	2-Wire voice unbundled port - residenc		UEPRX	UEPRL	2.55			31.9	2 7.32	+
	2-Wire voice unbundled port with Caller ID - re		UEPRX	UEPRC	2.55			31.9	2 7.32	
	2 Trine voice and and a port with Callet 1D - 16		ULFIX	OLFRO	2.00			31.:	1.32	+
	2-Wire voice unbundled port outgoing only - re		UEPRX	UEPRO	2.55			31.9	2 7.32	
	2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Calle									
	- res		UEPRX	UEPAS	2.55			31.9	2 7.32	
										1
	2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RU		UEPRX	UEPAG	2.55			31.9		
	2-Wire voice unbundles res, low usage line port with Caller ID (LUI		UEPRX	UEPAP	2.55			31.9	2 7.32	+
										+
FEATURES										+
	All Features Offered		UEPRX	UEPVF	8.28	0	0	 31.9	2 7.32	+
					5.20	Ŭ		01	7.02	+
LOCAL NUN	IBER PORTABILITY									
	Local Number Portability (1 per port		UEPRX	LNPCX	0.35					
	RING CHARGES (NRCs) - CURRENTLY COMBINED								_	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPRX	USAC2		3.8	0.29	31.9	2 7.32	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		UEPRX	USACC		3.8	0.29	31.9	2 7.32	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with Chair 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa		UEFKA	USACC		3.0	0.29	31.:	2 1.32	+
	Update					2.11		5.1	2	
ADDITIONAL										
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPRX	USAS2	0	0	0			
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)						<del>                                     </del>			
LINE Bort/Lo	pop Combination Rates									+
	2-Wire VG Loop/Port Combo - Zone	1			16.6					+
	2-Wire VG Loop/Port Combo - Zone	2			26.69					20
	2-Wire VG Loop/Port Combo - Zone	3			51.85					
UNE Loop R			-							
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	14.05					
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPBX	UEPLX	24.14					
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	49.3					+
2-Wire Voice	e Grade Line Port (Bus)									+
	2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	2.55			31.	2 7.32	+
	2 Trine voice and and and port without called 1D - Dt		ULF DA	OLFBL	2.00			31.:	1.32	+
	2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBX	UEPBC	2.55			31.9	2 7.32	1
										Т
	2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	2.55			31.9	2 7.32	
	2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Calle		==							1
	- bus		UEPBX	UEPAX	2.55			31.9	2 7.32	
	2-Wire voice unbundled incoming only port with Caller ID - Bu 2-Wire voice unbundled Louisiana Bus Area Calling Port with Caller ID (BU		UEPBX UEPBX	UPEB1 UEPAA	2.55 2.55			31. 31.	2 7.32 2 7.32	+
	2-YVITE VOICE UNDUTUIEU LOUISIANA DUS AIEA CAIIING FOR WITH CAIIEN ID (BU		UEFBA	UEFAA	2.00			31.3	£ 1.32	+
LOCAL NUM	IBER PORTABILITY									+
	Local Number Portability (1 per port		UEPBX	LNPCX	0.35					+
										+
FEATURES										
	All Features Offered		UEPBX	UEPVF	8.28	0	0	31.9	2 7.32	
	RING CHARGES (NRCs) - CURRENTLY COMBINED									

# Unbundled Network Elements LOUISIANA

2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa			UEPBX	USACC		3.8	0.29				
Update						2.11			5.12		
ADDITIONAL NRCs	i										
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ			UEPBX	USAS2					31.92	7.32	
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)											
UNE Port/Loop Combination Rates											
2-Wire VG Loop/Port Combo - Zone		1			16.6						
2-Wire VG Loop/Port Combo - Zone		2			26.69						
2-Wire VG Loop/Port Combo - Zone		3			51.85						
UNE Loop Rates  2-Wire Voice Grade Loop (SL 1) - Zone		1	UEPRG	UEPLX	14.05						
2-Wire Voice Grade Loop (SL 1) - Zone		2	UEPRG	UEPLX	24.14						
2-Wire Voice Grade Loop (SL 1) - Zone		3	UEPRG	UEPLX	49.3						
2-Wire Voice Grade Line Port Rates (RES - PBX)	i										
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re			UEPRG	UEPRD	2.55				31.92	7.32	
LOCAL NUMBER PORTABILITY											
Local Number Portability (1 per port			UEPRG	LNPCP	3.5						
FEATURES											
All Features Offered			UEPRG	UEPVF	8.28	0	0		31.92	7.32	
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED											
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As			UEPRG	USAC2		3.8	0.29		31.92	7.32	
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPRG	USACC		3.8	0.29		31.92	7.32	
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa			OLITIO	CONCO		0.0	0.25		01.02	7.52	
Update	<b>—</b>					2.11			5.12		
ADDITIONAL NRCs											
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ			UEPRG	USAS2	0	0	0				
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					•	14.64	14.64		19.99	19.99	19.99
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)											
UNE Port/Loop Combination Rates											
2-Wire VG Loop/Port Combo - Zone		1			16.6						
2-Wire VG Loop/Port Combo - Zone	1	2			26.69						
2-Wire VG Loop/Port Combo - Zone		3			51.85						
UNE Loop Rates											
2-Wire Voice Grade Loop (SL 1) - Zone		1	UEPPX	UEPLX							
				UEPLX	14.05						
2-Wire Voice Grade Loop (SL 1) - Zone		2	UEPPX	UEPLX	24.14						
2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone		3									
2-Wire Voice Grade Loop (SL 1) - Zone			UEPPX	UEPLX	24.14						
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)			UEPPX UEPPX	UEPLX UEPLX	24.14 49.3				04.00		
2-Wire Voice Grade Loop (SL 1) - Zone			UEPPX	UEPLX	24.14				31.92	7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt  Line Side Unbundled Outward PBX Trunk Port - Bu			UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO	24.14 49.3 2.55 2.55				31.92	7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPC UEPPO UEPP1	24.14 49.3 2.55 2.55 2.55				31.92 31.92	7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bt  Line Side Unbundled Outward PBX Trunk Port - Bt  Line Side Unbundled Incoming PBX Trunk Port - Bt  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPL2	24.14 49.3 2.55 2.55 2.55 2.55				31.92 31.92 31.92	7.32 7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu  Line Side Unbundled Outward PBX Trunk Port - Bu  Line Side Unbundled Incoming PBX Trunk Port - Bu  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port			UEPPX	UEPLX UEPLX UEPLX UEPPC UEPPO UEPP1 UEPL2 UEPLD	24.14 49.3 2.55 2.55 2.55 2.55 2.55 2.55				31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu  Line Side Unbundled Outward PBX Trunk Port - Bu  Line Side Unbundled Incoming PBX Trunk Port - Bu  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Treminal Port  2-Wire Voice Unbundled PBX LD Treminal Port			UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPL2	24.14 49.3 2.55 2.55 2.55 2.55 2.55 2.55 2.55				31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port  2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por			UEPPX	UEPLX UEPC UEPPO UEPP1 UEPL2 UEPLD UEPXA UEPXB	24.14 49.3 2.55 2.55 2.55 2.55 2.55 2.55 2.55 2.				31.92 31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu  Line Side Unbundled Outward PBX Trunk Port - Bu  Line Side Unbundled Incoming PBX Trunk Port - Bu  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Treminal Port  2-Wire Voice Unbundled PBX LD Treminal Port			UEPPX	UEPLX UEPC UEPPO UEPP1 UEPL2 UEPLD UEPXA	24.14 49.3 2.55 2.55 2.55 2.55 2.55 2.55 2.55				31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  2-Wire Voice Unbundled PBX LD DD Terminal Port  2-Wire Voice Unbundled PBX LD DD Terminal Por			UEPPX UEPPX	UEPLX UEPPC UEPPO UEPP1 UEPL2 UEPLD UEPXB UEPXB UEPXC	24.14 49.3 2.55 2.55 2.55 2.55 2.55 2.55 2.55 2.				31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  2-Wire Voice Unbundled PBX Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port  2-Wire Voice Unbundled PBX LD Terminal Hotel Por  2-Wire Voice Unbundled PBX LD DD Terminals Po  2-Wire Voice Unbundled PBX LD DD Terminals Po  2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc			UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPC UEPPO UEPP1 UEPL2 UEPLD UEPXB UEPXB UEPXC UEPXD UEPXE	2.55 2.55 2.55 2.55 2.55 2.55 2.55 2.55				31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu Line Side Unbundled Outward PBX Trunk Port - Bu Line Side Unbundled Datward PBX Trunk Port - Bu Line Side Unbundled PBX LO Terminal Port 2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  2-Wire Voice Unbundled PBX LD DD Terminals Po  2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc			UEPPX UEPPX	UEPLX UEPPC UEPPO UEPP1 UEPL2 UEPLD UEPXB UEPXB UEPXC	24.14 49.3 2.55 2.55 2.55 2.55 2.55 2.55 2.55 2.				31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  2-Wire Voice Unbundled PBX LD DD Terminals Po  2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt  2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling P  2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPC UEPPO UEPPI UEPLD UEPLD UEPXB UEPXB UEPXB UEPXC UEPXB	2.55 2.55 2.55 2.55 2.55 2.55 2.55 2.55				31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port  2-Wire Voice Unbundled PBX Toll Terminal Port  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  2-Wire Voice Unbundled PBX LD DD Terminals Po  2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc  2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling P  2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX UEPPX	UEPLX UEPC UEPPC UEPPC UEPP1 UEPL2 UEPL2 UEPXA UEPXB UEPXC UEPXC UEPXC UEPXC UEPXC UEPXC UEPXC UEPXC	2.55 2.55 2.55 2.55 2.55 2.55 2.55 2.55				31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32 7.32 7.32	
2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı  Line Side Unbundled Outward PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  Line Side Unbundled Incoming PBX Trunk Port - Bı  2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port  2-Wire Voice Unbundled PBX Toll Terminal Hotel Por  2-Wire Voice Unbundled PBX LD DD Terminals Po  2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt  2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt  2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling P  2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPC UEPPO UEPPI UEPLD UEPLD UEPXB UEPXB UEPXB UEPXC UEPXB	2.55 2.55 2.55 2.55 2.55 2.55 2.55 2.55				31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92 31.92	7.32 7.32 7.32 7.32 7.32 7.32 7.32 7.32	

# Unbundled Network Elements LOUISIANA

							i						
2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling P		U	EPPX	UEPXP	2.55		ı			31.92	7.32		
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc			EPPX	UEPXS	2.55		I			31.92	7.32		
							<b>.</b>						
LOCAL NUMBER PORTABILITY							<b></b>						
Local Number Portability (1 per port		UF	EPPX	LNPCP	3.15			++-	-++-	-++		ļ	
FEATURES												-	
All Features Offered			IEPPX	UEPVF	8.28	0	0			31.92	7.32	-	
All Features Offeret		- 01	EPPX	UEPVF	0.20	0				31.92	7.32	<del>                                     </del>	
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED				_			i	+					+
							i	_					
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		U	EPPX	USAC2		3.8	0.29			31.92	7.32		
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with							i						
Change		UF	EPPX	USACC		3.8	0.29			31.92	7.32		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa	4						ı				l.		
Update						2.11				5.12		<b>_</b>	
ADDITIONAL NRCs												-	
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ	++		EPPX	USAS2	0	0	0					<b>_</b>	
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi	+		EFFA	USASZ		14.64	14.64			19.99	19.99	19.99	٠
1 57 Cabboquotit / tottitity   Orlango/ total ango matalinio main oron				_						10.00	10.00	10.00	
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT													
							i						
UNE Port/Loop Combination Rates													
2-Wire VG Coin Port/Loop Combo – Zone 1	$\perp \perp \perp$			$\bot$	16.79	lacksquare		$\bot$					
2-Wire VG Coin Port/Loop Combo – Zone 2	$\perp \perp \downarrow$				26.88		<b></b>					<u> </u>	1
2-Wire VG Coin Port/Loop Combo – Zone 3					52.04							<b>_</b>	
UNE Loop Rates	++			+				+					+
2-Wire Voice Grade Loop (SL1) - Zone	+-+	11/	EPCO	UEPLX	14.05	$\vdash$		+	-	-+-+		<del> </del>	+
2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone			EPCO	UEPLX	24.14							<del>                                     </del>	
2-Wire Voice Grade Loop (SL1) - Zone	_		EPCO	UEPLX	49.3		i	-					
E Will Voice Glade Eddy (GET) Edile				- 02.27				-					
2-Wire Voice Grade Line Ports (COIN)							I						
2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS	,)						i						
		UF	EPCO	UEPRF	2.74		<b>.</b>			31.92	7.32		
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL	,						ı						
KY, LA, MS)			EPCO	UEPRA	2.74					31.92	7.32	<b>_</b>	
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, MS) 2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, &		Ut	EPCO	UEPRB	2.74					31.92	7.32	-	
Local (AL, KY, LA, MS)		1.07	EPCO	UEPCD	2.74		ı			31.92	7.32		
2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)	_	Ur	EPCO	UEPRN	2.74		i	-		31.92	7.32		
2-Wire Coin Outward with Operator Screening and 011 Blocking (LA)			EPCO	UEPLA	2.74					31.92	7.32		
2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (A	L,						i						
KY, LA, MS)		UF	EPCO	UEPRH	2.74		<b>.</b>			31.92	7.32		
2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, and Lo	4						ı				l.		
(AL, KY, LA, MS)			EPCO	UEPCN	2.74		<b></b>			31.92	7.32		
2-Wire Coin 2-Way Smartline with 900/976 (Louisiana only)		Ut	EPCO EPCO	UEPNA	2.74 2.74					31.92 31.92	7.32 7.32	-	
2-Wire Coin Outward Smartline with 900/976 (Louisiana only)  ADDITIONAL UNE COIN PORT/LOOP (RC)	++	UE	<u>=PCO</u>	DEPCB	2.74	<del>                                     </del>				31.92	7.32	<b>_</b>	+
ADDITIONAL ONE COMPTONIZEOUT (NC)												<del>                                     </del>	
UNE Coin Port/Loop Combo Usage (Flat Rate		Ur	EPCO	URECU	1.81	0	0				l.		
one com one cody ( narran				- UNLOC	1.01		. <del></del>						1
LOCAL NUMBER PORTABILITY													1
Local Number Portability (1 per port		UI	EPCO	LNPCX	0.35								L
FEATURES	$oxed{\Box}$			$\bot$		oxdot							
All Features Offered		UF	EPCO	UEPVF	8.28	0	<b></b>	0		18.14	8.06	8.94	
NONDECTIDDING CHARGES, CURRENTLY COMPINED	+-+	-		$-\!\!\!-\!\!\!\!-$		$\vdash$		+	-	$\longrightarrow$		<del>                                     </del>	1
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	++		EPCO	USAC2		3.8	0.29	+		31.92	7.32		+
z-write voice Grade Loop / Line Port Combination - Conversion - Switch-as	+-+	UE	<u>=</u> PUU	USACZ		3.8	0.29	+	<del></del>	31.92	1.52	<del>                                     </del>	+-
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		1.07	EPCO	USACC		3.8	0.29			31.92	7.32		
2 THE VOICE GRADE LOOP / LINE FOR COMBINATION - CONVERSION - CWITCH WITH CHAIN	-	- 00	00	COACC		5.0	0.23	+		31.32	1.02		+
ADDITIONAL NRCs													1
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UI	EPCO	USAS2		0	0			31.92	7.32		L
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT													
						oxed	<b></b>					<u> </u>	1
UNE Port/Loop Combination Rates	4—4						<b></b>					<u> </u>	1
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		1			28.99				$\longrightarrow$			<u> </u>	1
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		2		+	35.35 49.88			+					+
z-wire vg loop/z-wire DID Trunk Port Combo - UNE Zone	+-+	3		+-+	49.88	$\vdash$		++-	-	-+-+		<del></del>	+
	+-+	-		+				+	<del></del>	<del>-   -  </del>		<del>                                     </del>	+
				1		1		1 1					+
UNE Loop Rates  2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	+	1 UE	EPPX	UECD1	19.5		-			19.99	19.99	19.99	1

	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	3	UE	PPX	UECD1	40.39					19.99	19.99	19.99	
UNE Port Ra	ate													
	Exchange Ports - 2-Wire DID Por		UE	PPX	UEPD1	9.49					19.99	19.99	19.99	
NONDEGUE	DENIS CHARGES CHEREFULLY COMPINED													
NONRECUR	RRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as		ш	PPX	USAC1		14.6	3.72			19.99	19.99	19.99	
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-		UE	PPA	USACI		14.6	3.12			19.99	19.99	19.99	$\vdash$
	Changes		UE	PPX	USA1C		14.6	3.72			19.99	19.99	19.99	
ADDITIONAL	L NRCs   2-Wire DID Subsequent Activity - Add Trunks, Per Trun			PPX	USAS1		53.5	53.5			19.99	19.99	19.99	-
	2-Wife DID Subsequent Activity - Add Trunks, Per Trun		UE	PPX	USAST		53.5	53.5			19.99	19.99	19.99	
Telephone N	Number/Trunk Group Establisment Charges													
	DID Trunk Termination (One Per Port			PPX	NDT	0	0	0			19.99	19.99	19.99	1
	Additional DID Numbers for each Group of 20 DID Numbe			PPX	ND4	0	0	0			19.99	19.99	19.99	
	DID Numbers, Non- consecutive DID Numbers , Per Numbe Reserve Non-Consecutive DID number			PPX	ND5 ND6	0	0	0		19.99 19.99				
	Reserve DID Numbers			PPX	NDV	0	0	0		19.99				
			· ·											
LOCAL NUM	MBER PORTABILITY													
	Local Number Portability (1 per port		UE	PPX	LNPCP	3.15								
2-WIRE ISDI	N DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT								+					+
2E 10DI	THE STATE OF THE S													1
UNE Port/Lo	pop Combination Rates													
				PPB										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	1	UE	PPR		34.43								+
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	2	UEPPB	UEPPR		43.62								
	ETT TOSTE SIGNAL GIAGO ESOSPETI TOSTE SIGNAL ENTO GIAGO FOR GIACO ESTA		02118	02.11.		10.02								
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	3	UEPPB	UEPPR		58.76								
UNE Loop R	Rates													
ONE LOOP N	Adles													
	2-Wire ISDN Digital Grade Loop - UNE Zone	1	UEPPB	UEPPR	USL2X	24.92					19.99	19.99	19.99	
	2-Wire ISDN Digital Grade Loop - UNE Zone	2	UEPPB	UEPPR	USL2X	33.11					19.99	19.99	19.99	ļ .
	2-Wire ISDN Digital Grade Loop - UNE Zone	3	UEPPB	UEPPR	USL2X	49.25					19.99	19.99	19.99	1
UNE Port Ra											19.99			ļ.,
	Exchange Port - 2-Wire ISDN Line Side Po		UEPPB	UEPPR	UEPPB	9.51					19.99	19.99	19.99	1
NONRECUR	RRING CHARGES - CURRENTLY COMBINED													
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination -													
	Conversion		UEPPB	UEPPR	USACB	0	76.93	53.97			19.99	19.99	19.99	
ADDITIONAL	I NRCs													
7.55.110.0.0														
LOCAL NUM	MBER PORTABILITY													
	Land Novel of Dord 179 of America		HEDDD	HEDDO	LNDOV	0.05								
	Local Number Portability (1 per port		UEPPB	UEPPR	LNPCX	0.35	0	0						
B-CHANNEL	USER PROFILE ACCESS:													
	CVS/CSD (DMS/5ESS)		UEPPB	UEPPR		0	0	0						
	CVS (EWSD) CSD			UEPPR UEPPR		0	0	0						
	000		OLFFD	OLFFIX	01000	0	U	U						
B-CHANNEL	AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)													
	CVS/CSD (DMS/5ESS)		UEPPB	UEPPR	U1UCD	0	0	0						
	CVC (FWCD)		LIEDDD	LIEDDD	HALICE									
	CVS (EWSD)		UEPPB	UEPPR	U1UCE	0	0	0	1					+
	CSD		UEPPB	UEPPR	U1UCF	0	0	0						
USER TERM	MINAL PROFILE	$\perp$												
	User Terminal Profile (EWSD only)		HEPPR	UEPPR	LITLIMA	0	0	0						
	OSEL TETHINALFTONIC (EWSD ONly)		OLFFB	OLFFR	JIOWA	U	U	U	<del>                                     </del>					+
VERTICAL F	FEATURES													t
	l l													
	All Vertical Features - One per Channel B User Profile		UEPPB	UEPPR	UEPVF	8.28	0	0						

Inte	roffice Channel mileage each, including first mile and facilities termination	ι	JEPPB UEPPR	M1GNC	18.4	76.2	34.54				19.99	19.99	19.99	1
Inte	eroffice Channel mileage each, additional mile	ι	JEPPB UEPPR	M1GNM	0.0384	0	0			0				
4-WIRE DS1 DIG	SITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT													
LINE Port/Loop (	Combination Rates				ļ									
	DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	1	UEPPP		207.97									
4W	DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	2	UEPPP		236.26									
	DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	3	UEPPP		450.74									
<b>UNE Loop Rates</b>														
	Vire DS1 Digital Loop - UNE Zone	1	UEPPP	USL4P	100.42						19.99	19.99	19.99	
	Vire DS1 Digital Loop - UNE Zone	2	UEPPP	USL4P	128.71						19.99	19.99	19.99	
4-V	Vire DS1 Digital Loop - UNE Zone	3	UEPPP	USL4P	343.19						19.99	19.99	19.99	
UNE Port Rate														+
	hange Ports - 4-Wire ISDN DS1 Por		UEPPP	UEPPP	107.55						19.99	19.99	19.99	
LXU	mange Forts - 4-Wile ISDN DS FFO		OLITI	OLITI	107.55						13.33	19.99	13.33	
NONRECURRING	G CHARGES - CURRENTLY COMBINED													
	Vire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -													
	oversion -Switch-as-is		UEPPP	USACP	0	237.88	156.94				19.99	19.99	19.99	
ADDITIONAL NR														1
	/ire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within		LIEDES	DD		0.575					46.55		40.00	
	Allowance	-+	UEPPP	PR7TF		0.979			_		19.99	19.99	19.99	
	/ire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All		UEPPP	PR7TO		22.99	22.99				19.99	19.99	19.99	
2-14/	tes except NC] //ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos		UEFFF	1ºK/10		22.33	22.33		_		19.99	19.99	19.99	+
	ove Std Allowance		UEPPP	PR7ZT		45.98	45.98				19.99	19.99	19.99	
,,,,,,														
LOCAL NUMBER														
Loca	al Number Portability (1 per port		UEPPP	LNPCN	1.75									
INTERFACE (Pro	euriceire Oalul				ļ									+
	ovsioning Only) ce/Data		UEPPP	PR71V	0	0	0							+
	ital Data		UEPPP	PR71D	0	0	0							
	ard Data		UEPPP	PR71E	0	0	0							
			02	TIOTIE		·	Ů							
New or Addition	al "B" Channel													
	v or Additional - Voice/Data B Channel		UEPPP	PR7BV	0	29.01					19.99	19.99	19.99	
New	v or Additional - Digital Data B Channel		UEPPP	PR7BF	0	29.01					19.99	19.99	19.99	
New	v or Additional Inward Data B Channel		UEPPP	PR7BD	0	29.01					19.99	19.99	19.99	
New	v or Additional Useage Sensitive Voice Data B Channel		UEPPP	PR7BS	0	29.01					19.99	19.99	19.99	
New	v or Additional Useage Sensitive Digital Data B Channel		UEPPP	PR7BU	0	29.01					19.99	19.99	19.99	
CALL TYPES														
Inwa	ard ward		UEPPP UEPPP	PR7C1 PR7C0	0	0	0							-
	D-Way		UEPPP	PR7CC	0	0	0							+
I WO	, way		OLFFF	1 10700	U	U	U		-					+
Interoffice Chani	nel Mileage							+	_		+			+
	ed Each Including First Mile		UEPPP	1LN1A	94.1831	140.49	106.69	20			19.99	19.99	19.99	
Eac	h Airline-Fractional Additional Mil		UEPPP	1LN1B	0.7831									
1-WIDE DS1 DIG	GITAL LOOP WITH 4-WIRE DDITS TRUNK PORT													
4-WIKE DOT DIG	STAL LOOF WITH 4-WIKE DOITS TROUK FORT													1
UNE Port/Loop (	Combination Rates													
4W	DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	1	UEPDC		178.08						19.99	19.99	19.99	
4W	DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	2	UEPDC		206.37						19.99	19.99	19.99	
4W	DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	3	UEPDC		420.85						19.99	19.99	19.99	
IINE Loca Deco											1			+
UNE Loop Rates		1	UEPDC	USLDC	100.42				_		19.99	10.00	19.99	
4-11	Vire DS1 Digital Loop - UNE Zone Vire DS1 Digital Loop - UNE Zone	2	UEPDC	USLDC	100.42			23.72		-	19.99	19.99 19.99	19.99	
4-1/	Vire DS1 Digital Loop - UNE Zone	3	UEPDC	USLDC	343.19			23.12	_		19.99	19.99	19.99	
		Ĭ			2.2.10						. 5.00			T
UNE Port Rate														Ĺ
	/ire DDITS Digital Trunk Por		UEPDC	UDD1T	77.66						19.99	19.99	19.99	
			-											LΞ
	G CHARGES - CURRENTLY COMBINED													1
						266.76	123.16	1		1	19.99	19.99	19.99	
4-W	Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as- Vire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with		UEPDC	USAC4	l	200.70	123.10				19.99	19.99	10.00	+-

# Unbundled Network Elements LOUISIANA

DDITIONAL N 4- Ac 4-	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with change - Trunk		UEPDC	USAWB		266.76	123.16				19.99	19.99	19.99	
DDITIONAL N 4- Ac 4-			OLI DC	USAWB		200.70	123.10				13.33	19.99	13.33	
4- Ac 4-	NRCs													
4- Ac 4-												+	+	+
Ac 4-	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel											+	+	+
4-	A-Wile DST Loop / 4-Wile DDTTS Trulk Fort - NRC - Subsequent Channel		UEPDC	UDTTA		26.6	26.6				19.99	19.99	19.99	
			UEPDC	UDITA		20.0	20.0				19.99	19.99	19.99	
	-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1	1-	LIEBBO	LIDITED		00.0	00.0				40.00		40.00	
	Vay Outward Trunk		UEPDC	UDTTB		26.6	26.6				19.99	19.99	19.99	
	-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan													
	nward Trunk w/out DIC		UEPDC	UDTTC		26.6	26.6				19.99	19.99	19.99	
	-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -													
	nward Trunk with DIC		UEPDC	UDTTD		26.6	26.6				19.99	19.99	19.99	
4-	-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way	•												
	DID w User Trans		UEPDC	UDTTE		26.6	26.6				19.99	19.99	19.99	
<b>IPOLAR 8 ZE</b>	ERO SUBSTITUTION													П
													1	
Br	88ZS -Superframe Format		UEPDC	CCOSF		0	605				19.99	19.99	19.99	
														T
B	88ZS - Extended Superframe Forma		UEPDC	CCOEF		0	605				19.99	19.99	19.99	
			, , , , ,	1 2 2 2 2								10.00	1.0.00	T
Iternate Mark	k Inversion													
1A	MI -Superframe Format		UEPDC	MCOSF		0	0							
A	MI - Extended SuperFrame Forma		UEPDC	MCOPO		0	0					1		1
	·										1	1	1	Г
												1		
	umber/Trunk Group Establisment Charges													
Te	elephone Number for 2-Way Trunk Grou		UEPDC	UDTGX	0						1			
Te	elephone Number for 1-Way Outward Trunk Grou		UEPDC	UDTGY	0					19.99				П
Te	elephone Number for 1-Way Inward Trunk Group Without DI		UEPDC	UDTGZ	0					19.99				
	DID Numbers for each Group of 20 DID Number		UEPDC	ND4	0					19.99			1	
Dr	DID Numbers, Non- consecutive DID Numbers , Per Numbe		UEPDC	ND5	0					19.99			1	
R/	Reserve Non-Consecutive DID Nos		UEPDC	ND6	0	0	0			19.99			1	
	Reserve DID Numbers		UEPDC	NDV	0	0	0			19.99			+	
			, , , , ,							.,,,,,				T
edicated DS	1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DI	DITS Tr	unk Port											
In	nteroffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination		UEPDC	1LNO1	93.4	140.49	106.69	20	16.34		19.99	19.99	19.99	
In	nteroffice Channel Mileage - Additional rate per mile - 0-8 mil		UEPDC	1LNOA	0.7831	0	0							
In	nteroffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination		UEPDC	1LNO2	0	0	0							
	nteroffice Channel Mileage - Additional rate per mile - 9-25 mil		UEPDC	1LNOB	0.7831	0	0						+	$\mathbf{T}$
	nteroffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination		UEPDC	1LNO3	0	0	0	0					+	$\vdash$
In	nteroffice Channel Mileage - Additional rate per mile - 25+ mil		UEPDC	1LNOC	0.7831	0	0						+	
1.0	ocal Number Portability, per DS0 Activate		UEPDC	LNPCP	3.15	0	0	0				+	+	+
C	Central Office Termininating Poir		UEPDC	CTG	0	-		-					+	
				1	-									Т
												1		T
	LOOP WITH CHANNELIZATION WITH PORT													
	S1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations													
ach System	can have up to 24 combinations of rates depending on type and number of ports u	ısed												
NE DS1 Loop	p													
	-Wire DS1 Loop - UNE Zone 1		1 UEPMG	USLDC 1		0	0							
	-Wire DS1 Loop - UNE Zone 2		2 UEPMG	USLDC 1	28.71	0	0							
4-	-Wire DS1 Loop - UNE Zone 3		3 UEPMG	USLDC 3	43.19	0	0							
	·												1	
NE DSO Cha	annelization Capacities (D4 Channel Bank Configurations)													
	4 DSO Channel Capacity - 1 per DS1		UEPMG	VUM24 1	12.51	0	0						1	
	8 DSO Channel Capacity - 1 per 2 DS1s		UEPMG	VUM48 2	25.02	0	0						1	T
	6 DSO Channel Capacity -1 per 4 DS1s		UEPMG	VUM96 4	50.04	0	0				<del>                                     </del>	+	+	+
	44 DS0 Channel Capacity - 1 per 6 DS1s		UEPMG	VUM14 6		0	0					+	+	+
	92 DS0 Channel Capacity - 1 per 8 DS1s		UEPMG	VUM19 9		-	0				-	+	+	+
		-	UEPMG			0	•				<del> </del>	+	+	+-
	40 DS0 Channel Capacity - 1 per 10 DS1s			VUM20 1		0	0					<del> </del>		₩
	88 DS0 Channel Capacity - 1 per 12 DS1s		UEPMG	VUM28 1	350.12	0	0				<b></b>	4		4
	84 DS0 Channel Capacity - 1 per 16 DS1s		UEPMG	VUM38 1		0	0				1	1		4
	80 DS0 Channel Capacity - 1 per 20 DS1s		UEPMG	VUM40 2	250.2	0	0					1		
	76 DS0 Channel Capacity -1 per 24 DS1s	7	UEPMG	VUM57 2	700.24	0	0					1		⊥ ¯
67	72 DS0 Channel Capacity - 1 per 28 DS1s		UEPMG	VUM67 3	150.28	0	0					1		
												1		
on-Recurring	g Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - C	onvers	ion Charge Based on a	System								1		
Minimum Sy	ystem configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DSO P	orts wit	th Feature Activations.											
lultiples of th	his configuration functioning as one are considered Add'l after the minimum system													
	IRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes		UEPMG	USAC4 0		300.63	16.7				19.99	19.99	19.99	19.
	ions at End User Locations Where 4-Wire DS1 Loop with Channelization with Port C	Combin											1	

Attachment 2 Exhibit C

# Unbundled Network Elements LOUISIANA

	NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature Activation -														
D'	New GA & TN Only		UEPMG	VUMD4	0	715.54	467.54	148.34	17.63			19.99	19.99	19.99	19.99
Bipolar 8 Ze	ero Substitution								1						
	Clear Channel Capability Format, superframe - Subsequent Activity Only		UEPMG	CCOSF	0	0	605					19.99	19.99	19.99	19.99
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only		UEPMG	CCOEF	0	0	605					19.99	19.99	19.99	19.99
Alternate M	Mark Inversion (AMI)														
	Superframe Format		UEPMG	MCOSF											
	Extended Superframe Format		UEPMG	MCOSF		0	0								-
+	Extended Supername Format		UEFINIG	MCOPC	, 0	U	U								
Exchange I	Ports Associated with 4-Wire DS1 Loop with Channelization with Port														
Exchange F															
	Line Side Combination Channelized PBX Trunk Port - Business		UEPPX	UEPCX	1.72	0	0	0	0		19.99				
	Line Side Outward Channelized PBX Trunk Port - Business		UEPPX	UEPOX		0	0	0	0		19.99				
	Line Side Inward Only Channelized PBX Trunk Port without DID		UEPPX	UEP1X	1.72	0	0	0	0		19.99				-
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port		UEPPX	UEPDM	0.52	0	0	0	0		19.99				
Feature Act	tivations - Unbundled Loop Concentration		OLITA	OLI DIVI	9.52	U	U	U	U		19.99				
T catale Act	arvations - Oribunated Ecop Goriochitation														
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX	1PQWM	0.75	25.36	13.4	4.47	4.24			19.99	19.99	19.99	19.99
1															
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		UEPPX	1PQWU	0.75	78.05	18.4	60.03	11.73			19.99	19.99	19.99	19.99
Telephone	Number/ Group Establishment Charges for DID Service														
	DID Trunk Termination (1 per Port)		UEPPX	NDT	0				1						1
	DID Numbers - groups of 20 - Valid all States		UEPPX	ND4	0	0	0				19.99				
	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers		UEPPX UEPPX	ND5	0	0	0				19.99				-
	Reserve DID Numbers		UEPPX	ND6 NDV	0	0	0								
Local Numb	ber Portability		OLITA	INDV	U	U	U								
	Local Number Portability - 1 per port		UEPPX	LNPCP	3.15	0	0								
	6 - Vertical and Optional		OL: 17	2.1. 0.	0.10	ľ	-								
	ching Features Offered with Line Side Ports Only														
	All Features Available		UEPPX	UEPVF	8.28	0	0				19.99				
ED PORT LOC	OP COMBINATIONS - MARKET RATES														
Market Rate	es shall apply where BellSouth is not required to provide unbundled local switching or swit	tch norts no	FCC and/or State (	Ommissi	on rules										
These scen	narios include:	ton ports pc	1 00 ana/or orace c	201111111331	on ruics.										
<ol> <li>Unbundle</li> </ol>	led port/loop combinations that are Not Currently Combined in all of the BellSouth states	except as n	l oted for Georgia and	Tennes	see.										
2. Unbundle	led port/loop combinations that are Not Currently Combined in all of the BellSouth states alled port/loop combinations that are Currently Combined or Not Currently Combined in Zor	ne 1 of the 1	op 8 MSAS in BellS	outh's re	gion for end u										
2. Unbundle	led port/loop combinations that are Not Currently Combined in all of the BellSouth states	ne 1 of the 1	op 8 MSAS in BellS	outh's re	gion for end u				lle).						
2. Unbundle The Top 8 N	led port/loop combinations that are Not Currently Combined in all of the BellSouth states of led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne	ne 1 of the 1 ew Orleans)	op 8 MSAS in BellS NC (Greensboro-W	iouth's re /inston S	gion for end u alem-Highpoin	t/Charlotte-Ga	stonia-Rock Hill	); TN (Nashvi							
2. Unbundle The Top 8 M BellSouth co	led port/loop combinations that are Not Currently Combined in all of the BellSouth states of led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recur	ne 1 of the 1 ew Orleans)	op 8 MSAS in BellS NC (Greensboro-W	iouth's re /inston S	gion for end u alem-Highpoin	t/Charlotte-Ga	stonia-Rock Hill	); TN (Nashvi		ng in lieu of	the Market	Rates and re	serves the rig	ht to true-up	the billing
2. Unbundle The Top 8 M  BellSouth cu The Market	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states; olded port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurt Rate for unbundled ports includes all available features in all states.	ne 1 of the 1 ew Orleans)	Op 8 MSAS in BellS NC (Greensboro-W	n. In the	gion for end u alem-Highpoin interim, BellS	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi	ection preced						
2. Unbundle The Top 8 M  BellSouth cu The Market End Office a	led port/loop combinations that are Not Currently Combined in all of the BellSouth states a led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne surrently is developing the billing capability to mechanically bill the recurring and non-recu- t Rate for unbundled ports includes all available features in all states.	rring Market	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end usalem-Highpoin interim, BellS ations of loop/	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 M  BellSouth cu The Market End Office a For Not Curi	led port/loop combinations that are Not Currently Combined in all of the BellSouth states of led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Nexurrently is developing the billing capability to mechanically bill the recurring and non-recul Rate for unbundled ports includes all available features in all states.  In and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are lister.	rring Market	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end usalem-Highpoin interim, BellS ations of loop/	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 M  BellSouth cu The Market End Office a For Not Curi	led port/loop combinations that are Not Currently Combined in all of the BellSouth states a led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne surrently is developing the billing capability to mechanically bill the recurring and non-recu- t Rate for unbundled ports includes all available features in all states.	rring Market	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end usalem-Highpoin interim, BellS ations of loop/	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N BellSouth ou The Market End Office a For Not Curi	led port/loop combinations that are Not Currently Combined in all of the BellSouth states of led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Nexurrently is developing the billing capability to mechanically bill the recurring and non-recul Rate for unbundled ports includes all available features in all states.  In and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are lister.	rring Market	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end usalem-Highpoin interim, BellS ations of loop/	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N  BellSouth ou The Market End Office a For Not Cur may apply a	lled port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recut Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listeralso and are categorized accordingly.	rring Market	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end usalem-Highpoin interim, BellS ations of loop/	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N  BellSouth ou The Market End Office a For Not Cur may apply a	led port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Sol MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurrently is developing the billing capability to mechanically bill the recurring and non-recurrently and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are lister also and are categorized accordingly.  INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	ne 1 of the 1 ew Orleans)  rring Market this rate exid in the Firs	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end u alem-Highpoin interim, BellS lations of loop/ s for each Por	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N  BellSouth ou The Market End Office a For Not Cur may apply a	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recur Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.	ne 1 of the 1 ew Orleans)  rring Market this rate exid in the Firs	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end u alem-Highpoin interim, BellS ations of loop/ s for each Por	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N  BellSouth ou The Market End Office a For Not Cur may apply a	lled port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recur Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listeralso and are categorized accordingly.  INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	ne 1 of the 1 ew Orleans)  rring Market this rate exid in the Firs	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end u alem-Highpoin interim, BellSi ations of loop/ s for each Por	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N  BellSouth ou The Market End Office a For Not Cur may apply a	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recur Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.	ne 1 of the 1 ew Orleans)  rring Market this rate exid in the Firs	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end u alem-Highpoin interim, BellS ations of loop/ s for each Por	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N BellSouth of The Market End Office a For Not Cur may apply a 2-WIRE VOI UNE Port/Lo	lled port/loop combinations that are Not Currently Combined in all of the BellSouth states.  Iled port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recur Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listeralso and are categorized accordingly.  DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	ne 1 of the 1 ew Orleans)  rring Market this rate exid in the Firs	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end u alem-Highpoin interim, BellSi ations of loop/ s for each Por	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N  BellSouth ou The Market End Office a For Not Cur may apply a	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recult Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listed also and are categorized accordingly.  JICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	ne 1 of the 1 ew Orleans)  rring Market this rate exid in the Firs	op 8 MSAS in BellS NC (Greensboro-W Rates in this section hibit shall apply to al	n. In the	gion for end u alem-Highpoin interim, BellS ations of loop/ s for each Por	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N BellSouth of The Market End Office a For Not Cur may apply a 2-WIRE VOI UNE Port/Lo	lled port/loop combinations that are Not Currently Combined in all of the BellSouth states.  Iled port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recur Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listeralso and are categorized accordingly.  DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	ne 1 of the Tew Orleans) rring Marker this rate ext d in the Firs  1 2 3	op 8 MSAS in BellS NC (Greensboro-W Rates in this section ibit shall apply to all t and Additional NR6	n. In the	gion for end u alem-Highpoin interim, BellSi ations of loop/ s for each Por	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 N BellSouth of The Market End Office a For Not Cur may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recur Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.  INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  Ooop Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	ne 1 of the The Worleans)  rring Market this rate exit d in the Firs  1 2 3	op 8 MSAS in BellSIN NC (Greensboro-Windows)  Rates in this section with the section of the sect	n. In the	gion for end ualem-Highpoin interim, BellSiations of loop/js for each Por 28.05 38.14 63.3	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  Iled port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurrently attention of the programment of the pr	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	op 8 MSAS in BellSI NC (Greensboro-W Rates in this sectio libit shall apply to al t and Additional NR(	n. In the	gion for end ualem-Highpoin interim, BellS ations of loop/s for each Por 28.05 38.14 63.3 14.05 24.14	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states. Ided port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recult Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.  JICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  LOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	Op 8 MSAS in BellSI NC (Greensboro-W Rates in this sectio libit shall apply to al and Additional NRO UEPRX UEPRX UEPRX UEPRX	n. In the combination of the com	gion for end ualem-Highpoin interim, BellS interim,	t/Charlotte-Ga	ne rates in the Cements except urrently Combin	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us in the NRC -	age charge (i	JSOC: UREC	U).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  Iled port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurrently attention of the programment of the pr	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	op 8 MSAS in BellSI NC (Greensboro-W Rates in this sectio libit shall apply to al t and Additional NR(	n. In the	gion for end ualem-Highpoin interim, BellS ations of loop/s for each Por 28.05 38.14 63.3 14.05 24.14	t/Charlotte-Ga	ne rates in the C	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us	age charge (	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	led port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurrently to the property of the property of the currently is developing the billing capability to mechanically bill the recurring and non-recurrently combined ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.    Discord GRADE LOOP WITH 2-WIRE LINE PORT (RES)	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	Op 8 MSAS in BellSI NC (Greensboro-W Rates in this sectio libit shall apply to al t and Additional NRO  UEPRX UEPRX UEPRX UEPRX UEPRX	ueplx	gion for end ualem-Highpoin interim, BellS ations of loop/s for each Por 28.05 38.14 63.3 14.05 24.14 49.3	t/Charlotte-Ga	ne rates in the Cements except urrently Combined in the Cements except urrently Combined in the Cements except urrently Combined in the Cements except urrently Combined in the Cements except urrently Combined in the Cemen	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us in the NRC -	age charge (i Currently Cor	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states. Ided port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recult Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.  JICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  LOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	Op 8 MSAS in BellSI NC (Greensboro-W Rates in this sectio libit shall apply to al and Additional NRO UEPRX UEPRX UEPRX UEPRX	n. In the combination of the com	gion for end ualem-Highpoin interim, BellS interim,	t/Charlotte-Ga	ne rates in the Cements except urrently Combin	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us in the NRC -	age charge (i	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  Ided port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurrently is developing the billing capability to mechanically bill the recurring and non-recurrently combined some switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.  INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  OOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port - residenc	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	Op 8 MSAS in BellS NC (Greensboro-Winchester) Rates in this section with the section of the sect	UEPLX UEPRL UEPRC	gion for end ualem-Highpoin interim, BellS lations of loop/s for each Por 28.05 38.14 63.3 14.05 24.14 49.3 14	t/Charlotte-Ga  puth shall bill ti  port network el  t USOC. For C	erates in the Comments except urrently Combined in the Comments except urrently Combined in the Combined in th	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us in the NRC -	age charge (i Currently Cor	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	led port/loop combinations that are Not Currently Combined in all of the BellSouth states.  led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurrently to the property of the property of the currently is developing the billing capability to mechanically bill the recurring and non-recurrently combined ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.    Discord GRADE LOOP WITH 2-WIRE LINE PORT (RES)	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	OP 8 MSAS in BellSI NC (Greensboro-Wi Rates in this section ibit shall apply to alt and Additional NR( UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ueplx	gion for end ualem-Highpoin interim, BellS ations of loop/s for each Por 28.05 38.14 63.3 14.05 24.14 49.3	t/Charlotte-Ga	ne rates in the Cements except urrently Combined in the Cements except urrently Combined in the Cements except urrently Combined in the Cements except urrently Combined in the Cements except urrently Combined in the Cemen	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us in the NRC -	age charge (i Currently Cor	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states. led port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recut Rate for unbundled ports includes all available features in all states.  and Tandem Switching Usage and Common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.  JICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  LOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	Op 8 MSAS in BellS NC (Greensboro-Winchester) Rates in this section with the section of the sect	UEPLX UEPRL UEPRC	gion for end ualem-Highpoin interim, BellS lations of loop/s for each Por 28.05 38.14 63.3 14.05 24.14 49.3 14	t/Charlotte-Ga  puth shall bill ti  port network el  t USOC. For C	erates in the Comments except urrently Combined in the Comments except urrently Combined in the Combined in th	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us in the NRC -	age charge (i Currently Cor	JSOC: UREC	Ü).
2. Unbundle The Top 8 M BellSouth or The Market End Office a For Not Curr may apply a 2-WIRE VOI UNE Port/Lo	lied port/loop combinations that are Not Currently Combined in all of the BellSouth states.  Ided port/loop combinations that are Currently Combined or Not Currently Combined in Zor MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (Ne currently is developing the billing capability to mechanically bill the recurring and non-recurrently combined song and common Transport Usage rates in the Port section of rrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listerals and are categorized accordingly.  IDCE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	ne 1 of the Tew Orleans) rring Market this rate exid in the First  1 2 3	OP 8 MSAS in BellSI NC (Greensboro-Wi Rates in this section ibit shall apply to alt and Additional NR( UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRC UEPRC	28.05 28.05 38.14 63.3 14.05 24.14 49.3	v/Charlotte-Ga  outh shall bill ti  cort network el  t USOC. For C	erates in the Comments except urrently Combined and the comments except urrently Combined and the components of the comments except urrently Combined and the components except urrently Combined and the components except	); TN (Nashvi Cost-Based se	ection preced	ombinations	which have	a flat rate us in the NRC -	age charge (i Currently Cor	JSOC: UREC	Ψ).

2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (AC			UEPRX	UEPAH	14	90	90			31.92	7.32
2-Wire voice unbundles res, low usage line port with Caller ID (LUI			UEPRX	UEPAP	14	90	90			31.92	7.32
		$\sqcup$					-				
LOCAL NUMBER PORTABILITY				LLUBOY							
Local Number Portability (1 per port			UEPRX	LNPCX	0.35						
FEATURES											
All Features Offered			UEPRX	UEPVF	0	0	0				
7 II 1 Catales Official			OLITOX	OLI VI			·				
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-			UEPRX	USAC2		41.5	41.5				
2-Wire Voice Grade Loop / Line Port Combination - Switch with change			UEPRX	USACC		41.5	41.5				
ADDITIONAL NRCs											
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque			UEPRX	USAS2		0	0				
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)											
2-WIRE VOICE GRADE EOOF WITH 2-WIRE LINE FORT (BOS)											
UNE Port/Loop Combination Rates											
2-Wire VG Loop/Port Combo - Zone		1			28.05						
2-Wire VG Loop/Port Combo - Zone		2			38.14						
2-Wire VG Loop/Port Combo - Zone		3			63.3						
UNE Loop Rates			-								
2-Wire Voice Grade Loop (SL1) - Zone		1	UEPBX	UEPLX	14.05						
2-Wire Voice Grade Loop (SL1) - Zone		2	UEPBX	UEPLX	24.14		1				
2-Wire Voice Grade Loop (SL1) - Zone	1	3	UEPBX	UEPLX	49.3	1	1		+		
O Wire Veice Crede Line Bert (Due)	1	$\vdash$				1	-				
2-Wire Voice Grade Line Port (Bus)  2-Wire voice unbundled port without Caller ID - bu	+		UEPBX	UEPBL	14	90	90		+	31.92	7.00
2-vvire voice unbunaled port without Caller ID - bu	+	$\vdash$	UEPBX	UEPBL	14	90	90		++	31.92	7.32
2-Wire voice unbundled port with Caller + E484 ID - bu			UEPBX	UEPBC	14	90	90			31.92	7.32
2 17110 10106 unbunuleu port with Gallet # £404 lb - bt	1	$\vdash$	OLFDA	OLFBC	14	30	30			31.32	1.32
2-Wire voice unbundled port outgoing only - bu			UEPBX	UEPBO	14	90	90			31.92	7.32
2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Call	ei			12.20						22	
- bus			UEPBX	UEPAX	14	90	90			31.92	7.32
2-Wire voice unbundled Louisiana Bus Area Calling Port with Caller ID (BU			UEPBX	UEPAA	14					31.92	7.32
		$\Box$	-								
LOCAL NUMBER PORTABILITY	1										
Local Number Portability (1 per port		$\sqcup$	UEPBX	LNPCX	0.35		1				
FEATURES	1	$\vdash$		$\rightarrow$		1	-				
real unes	+	$\vdash$				1	<del>                                     </del>		++		
NONRECURRING CHARGES - CURRENTLY COMBINED	+					1	+	<del>                                     </del>	+ +		
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	+	$\vdash$	UEPBX	USAC2		41.5	41.5		+ +		
E Wile voice Orage Loop / Eine i oft Combination - Owiten-as.		$\vdash$	OLF DA	USAUZ		71.0	71.0				
2-Wire Voice Grade Loop / Line Port Combination - Switch with change			UEPBX	USACC		41.5	41.5				
				2		1					
ADDITIONAL NRCs											
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque			UEPBX	USAS2		0	0				
		ш									
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)		$\sqcup$					-				
INF Book and Continue Book	1	$\vdash$					-				
UNE Port/Loop Combination Rates	1	1			20.05	1	-				
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	-	2			28.05 38.14	-	1		+		
2-Wire VG Loop/Port Combo - Zone :	1	3			63.3	1	<u> </u>				
2 THIS TO LOOP/FOR COMBO - ZONE	1	J			05.5		<b>—</b>				
UNE Loop Rates		$\vdash$									
2-Wire Voice Grade Loop (SL1) - Zone	1	1	UEPRG	UEPLX	14.05						
2-Wire Voice Grade Loop (SL1) - Zone		2	UEPRG	UEPLX	24.14						
2-Wire Voice Grade Loop (SL1) - Zone		3	UEPRG	UEPLX	49.3						
2-Wire Voice Grade Line Port Rates (RES - PBX)	1										
				1							
12 Wire VC Unbundled Combination 2 Way DDV Trunk Bort - Dc	1		UEPRG	UEPRD	14	90	90		+	31.92	7.32
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		$\vdash$					-				
			UEPRG	LNDOD	2.45	1	-				
LOCAL NUMBER PORTABILITY				LNPCP	3.15	1	<del>                                     </del>		++		
			UEPRG	1 1		1	1				
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port			UEPRG								
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port			UEPRG								
LOCAL NUMBER PORTABILITY Local Number Portability (1 per port			UEPRG								
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port  FEATURES  NONRECURRING CHARGES - CURRENTLY COMBINED				USAC2		41 5	41 5				
LOCAL NUMBER PORTABILITY Local Number Portability (1 per port			UEPRG	USAC2		41.5	41.5				
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port  FEATURES  NONRECURRING CHARGES - CURRENTLY COMBINED				USAC2 USACC		41.5	41.5				

# Unbundled Network Elements LOUISIANA

2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring						0	0	<u>                                       </u>					
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi						14.64	14.64			19.99	19.99	19.99	1
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)													+
													1
UNE Port/Loop Combination Rates					00.05								+
2-Wire VG Loop/Port Combo - Zone		2			28.05 38.14								+
2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :		3			63.3								+
UNE Loop Rates	1												+
2-Wire Voice Grade Loop (SL1) - Zone	+	1	UEPPX	UEPLX	14.05								$^{+}$
2-Wire Voice Grade Loop (SL1) - Zone		2	UEPPX	UEPLX	24.14								Ť
2-Wire Voice Grade Loop (SL1) - Zone		3	UEPPX	UEPLX	49.3								1
2-Wire Voice Grade Line Port Rates (BUS - PBX)													t
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bi			UEPPX	UEPPC	14	90	90			31.92	7.32		Ī
Line Cide Helevardlad Outward DDV Toyal, Dark Dv			LIEDDY	LIEDDO	44	90	00			24.02	7.00		T
Line Side Unbundled Outward PBX Trunk Port - Bu Line Side Unbundled Incoming PBX Trunk Port - Bu	++	-+	UEPPX UEPPX	UEPPO UEPP1	14 14	90	90 90	<del>                                     </del>	+ +	31.92 31.92	7.32 7.32		+
2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling P	++	-+	UEPPX	UEPL2	14	90	90	<del>                                     </del>	+ +	31.92	7.32		+
2-Wire Voice Unburidled 2-Way Combination PBX Lodisiana Calling P  2-Wire Voice Unbundled PBX LD Terminal Port	+-+		UEPPX	UEPLD	14	90	90	<del>                                     </del>		31.92	7.32		+
2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc			UEPPX	UEPXA	14	90	90			31.92	7.32		t
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por			UEPPX	UEPXB	14	90	90			31.92	7.32		Ť
2-Wire Voice Unbundled PBX LD DDD Terminals Po			UEPPX	UEPXC	14	90	90			31.92	7.32		T
													†
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	<b>↓</b>		UEPPX	UEPXD	14	90	90	1	1	31.92	7.32		4
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc	$\vdash$		UEPPX	UEPXE	14	90	90	<del>                                     </del>	1	31.92	7.32		+
2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling P			UEPPX	UEPXK	14	90	90	<del>                                     </del>	+	31.92	7.32		+
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX	UEPXL	14	90	90			31.92	7.32		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P			UEPPX	UEPXM	14	90	90			31.92	7.32		
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX	UEPXO	14	90	90			31.92	7.32		Ī
													Ť
2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling P 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc			UEPPX UEPPX	UEPXP	14	90	90			31.92	7.32		+
2-wire voice Oriburided 1-way Outgoing PBX Measured PC	+-+		UEPPX	UEPAS	14	90	90			31.92	7.32		$^{+}$
LOCAL NUMBER PORTABILITY													İ
Local Number Portability (1 per port	<b>.</b>		UEPPX	LNPCP	3.15								4
FEATURES	$\vdash$												+
													İ
NONRECURRING CHARGES - CURRENTLY COMBINED													
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-	<u> </u>		UEPPX	USAC2		41.5	41.5						4
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Chan-			UEPPX	USACC		41.5	41.5						
ADDITIONAL NRCs													Ŧ
2-Wire Voice Grade Loop/ Line Port Combination - Subseque			UEPPX	USAS2		0	0						+
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring				$\Box$		0	0		1 T				1
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi						14.64	14.64			19.99	19.99	19.99	#
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT													1
UNE Port/Loop Combination Rates	$\vdash \exists$	$-\mathbb{T}$	-	+	-				+				Ŧ
2-Wire VG Coin Port/Loop Combo – Zone 1	+			+	28.05	<b> </b>		<del>                                     </del>					+
					38.14								†
2-Wire VG Coin Port/Loop Combo – Zone 2					63.3								1
2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3		-				1	1	1					+
2-Wire VG Coin Port/Loop Combo – Zone 3													+
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone			UEPCO	UEPLX	14.05								+
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone			UEPCO	UEPLX	24.14								t
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone													‡
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone			UEPCO	UEPLX	24.14								<u> </u>
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone			UEPCO UEPCO	UEPLX	24.14 49.3	90	90			31.92	7.32		† †
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coice Grade Line Port Rates (Coin)  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL			UEPCO UEPCO	UEPLX UEPLX UEPRF	24.14	90	90			31.92	7.32		  -  -
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Line Port Rates (Coin)  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS, KY, LA, MS, SC)			UEPCO UEPCO UEPCO	UEPLX UEPLX UEPRF	24.14 49.3 14	90	90			31.92	7.32		+
2-Wire VG Coin Port/Loop Combo – Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coice Grade Line Port Rates (Coin)  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL			UEPCO UEPCO	UEPLX UEPLX UEPRF	24.14 49.3								+

2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)	UEPCO	UEPRN	14	90	90		31.92	7.32	1
2-Wire Coin Outward with Operator Screening and 011 Blocking (LA)	UEPCO	UEPLA	14	90	90		31.92	7.32	
2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL,									1
KY, LA, MS)	UEPCO	UEPRH	14	90	90		31.92	7.32	
2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local									
(AL, KY, LA, MS)	UEPCO	UEPCN	14	90	90		31.92	7.32	
									₩
LOCAL NUMBER PORTABILITY									+
Local Number Portability (1 per port	UEPCO	LNPCX	0.35						L
NONRECURRING CHARGES - CURRENTLY COMBINED									+
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-I	UEPCO	USAC2		41.5	41.5				$\blacksquare$
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Chan-	UEPCO	USACC		41.5	41.5				
ADDITIONAL NRCs									
2-Wire Voice Grade Loop/ Line Port Combination - Subseque	UEPCO	USAS2		0	0				
			·						
									ــــــــــــــــــــــــــــــــــــــ
					1				

Page 23 of 23 Version 2Q01: 08/30/01

	NOTES	UNBUNDLED NETWORK ELEMENT Interi	m Zone	BCS	USOC			RATES (\$)	_			1	OSS R	ATES (\$)		
							Nonre	curring			Svc Order Submitted Elec per LSR		Electronic-1st recurring		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs Electronic Disc Add
							_		_				sconnect			
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		shown in the sections for stand-alone loops or loops as part of a combination refers to Geognterconnection.bellsouth.com/become_a_clec/html/interconnection.htm	raphica	lly Deaveraged UNE	Zones. To	view Geog	raphically Deave	Leraged UNE Zo	one Designat	ions by Centra	l Office, re	l fer to Intern	et Website:			
BUNDI FD	FXCHANG	E ACCESS LOOP														+
																_
2	2-WIRE ANA	LOG VOICE GRADE LOOP														
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	1	UEANL	UEAL2	15.58	59.25	43.67	16.35	4.06			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	2	UEANL	UEAL2	20.65	59.25	43.67	16.35	4.06			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	3	UEANL	UEAL2	29.51	59.25	43.67	16.35	4.6			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone	4	UEANL	UEAL2	38.94	59.25	43.67	16.35	4.06			25.52	11.34	16.06	16.0
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zona	1	UEPSR, UEPSB	UEALS	15.58	59.25	43.67	16.35	4.06			25.52	11.34	16.06	16.0
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zon	2		UEALS	20.65	59.25	43.67	16.35	4.06			25.52	11.34	16.06	16.0
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zoni	3		UEALS	29.51	59.25	43.67	16.35	4.6			25.52	11.34	16.06	16.0
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone	4		UEALS	38.94	59.25	43.97	16.35	4.06			25.52	11.34	16.06	16.0
		Engineering Information Document (E	-	UEANL	1		28.72	28.72	1	<del>                                     </del>			1	<del>                                     </del>		+
		Manual Order Coordination for UVL-SL1s (per loop		UEANL	UEAMC		50.29	50.29		1				1	i	
		ivianuai Order Coordination for OVE-SE15 (per loop		UEAINL	CEAIVIC		30.29	30.29	-							+
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR		UEANL	OCOSL		45.27	45.27	1	1				1	i	1
		Order Coordination for Specified Conversion Time for 6 v2-521 (per 25).		OLANL	OCCOL		43.27	43.27								+
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -														+
		Zone 1	1	UEA	UEAL2	18.35	144.01	107.7	40.98	26.95			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -		OLA	OLALE	10.00	144.01	107.7	40.50	20.55			20.02	11.04	10.00	10.
		Zone 2	2	UEA	UEAL2	24.33	144.01	107.7	40.98	26.95			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -		OL/\	O L/ (LL	21.00	111101	101.11	10.00	20.00			20.02	11.01	10.00	10.0
		Zone 3	3	UEA	UEAL2	34.77	144.01	107.7	40.98	26.95			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -														1
		Zone 4	4	UEA	UEAL2	45.88	144.01	107.7	40.98	26.95			25.52	11.34	16.06	
		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		45.27									
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone														
		1	1	UEA	UEAR2	18.35	144.01	107.7	40.98	26.95			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone														
		2	2	UEA	UEAR2	24.33	144.01	107.7	40.98	26.95			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone														
		3	3	UEA	UEAR2	34.77	144.01	107.7	40.98	26.95			25.52	11.34	16.06	16.0
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone														
		4	4	UEA	UEAR2	45.88	144.01	107.7	40.98	26.95			25.52	11.34	16.06	16.0
		Order Coordination for Specified Conversion Time (new LC)		1154	00000		45.07	1						1	i	
		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		45.27	-	-	<del>                                     </del>				<del>                                     </del>		+
		4-Wire Analog Voice Grade Loop - Zone	1	UEA	UFAL4	22.38	289.06	238.19	108.14	57.28			25.52	11.34	16.06	16.
-		4-Wire Analog Voice Grade Loop - Zone 4-Wire Analog Voice Grade Loop - Zone	2	UEA	UEAL4	29.67	289.06	238.19	108.14	57.28			25.52	11.34	16.06	16.0
		4-Wire Analog Voice Grade Loop - Zone 4-Wire Analog Voice Grade Loop - Zone	3	UEA	UEAL4	42.4	289.06	238.19	108.14	57.28			25.52	11.34	16.06	16.0
		4-Wire Analog Voice Grade Loop - Zone 4-Wire Analog Voice Grade Loop - Zone	4	UEA	UEAL4	55.96	289.06	238.19	108.14	57.28			25.52	11.34	16.06	16.0
				02,1	32,127	00.00	200.00	200.10		01.20			20.02			1.0.0
		Order Coordination for Specified Conversion Time (per LS		UEA	OCOSL		45.27	1	1	1				1	i	
																1
2	2-WIRE ISDI	N DIGITAL GRADE LOOP														
		2-Wire ISDN Digital Grade Loop - Zone	1	UDN	U1L2X	21.86	326.38	252	108.14	57.27			25.52	11.34	16.06	16.0
		2-Wire ISDN Digital Grade Loop - Zone	2	UDN	U1L2X	28.97	326.38	252	108.14	57.27			25.52	11.34	16.06	16.
		2-Wire ISDN Digital Grade Loop - Zone	3	UDN	U1L2X	41.4	326.38	252	108.14	57.27			25.52	11.34	16.06	16.
		2-Wire ISDN Digital Grade Loop - Zone	4	UDN	U1L2X	54.64	326.38	252					25.52	11.34	16.06	16.
		0. l 0   F 0   F 0   F 0   F 0   F   O   O   O		LIDAL	0000:		45.07	1		1				1	i	
		Order Coordination For Specified Conversion Time (per LS	-	UDN	OCOSL		45.27	-	1	<del>                                     </del>			1	<del>                                     </del>		+
	2-WIDE III-	versal Digital Channel (UDC) COMPATIBLE LOOP			<del>                                     </del>		-	-	-	<del>                                     </del>				<del>                                     </del>		+
			1	LIDO	LIDCOV	22.40	222.54	158.71	104.00	20.50			25.50	11.04	16.00	10
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	2	UDC	UDC2X UDC2X	32.48 42.06	233.54	158.71 158.71	104.88 104.88	20.59			25.52	11.34 11.34	16.06 16.06	16. 16.
-		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	3	UDC	UDC2X	55.26	233.54 233.54	158.71	104.88	20.59 20.59			25.52 25.52	11.34	16.06	16.
-		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone	4		UDC2X	71.05	233.54	158.71	104.88	20.59			25.52	11.34	16.06	16.0
		z ***** om*orsar bigitar onanner (obo) compatible Loop - Zone	- 4	UDC	JUUZA	71.00	200.04	150./1	104.00	20.08			20.02	11.34	10.00	10.0
1		1	- 1	1			+	<b></b>	1			-				+
	2-WIRF ASV	MMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP			1					1						

						,				 ·			
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 1	1	UAL	UAL2X	10.87	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -												
	Zone 2  2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -	2	UAL	UAL2X	14.4	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
	Zone 3	3	UAL	UAL2X	20.58	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation - Zone 4	4	UAL	UAL2X	27.16	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
	Zone 4	4	UAL	UALZX	27.10	504.82	456.24	105.86	57.25	25.52	11.34	16.06	-
	Order Coordination for Specified Conversion Time (per LS		UAL	OCOSL		45.27							
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1	1	UAL	UAL2W	10.87	204.56	128.86	100.05	15.75	25.52	11.34	16.06	
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -												
	Zone 2  2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -	2	UAL	UAL2W	14.4	204.56	128.86	100.05	15.75	25.52	11.34	16.06	
	Zone 3	3	UAL	UAL2W	20.58	204.56	128.86	100.05	15.75	25.52	11.34	16.06	
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -	4	UAL	1141 3141	27.16	204.56	128.86	400.05	45.75	25.52	44.04	40.00	
	Zone 4	4	UAL	UAL2W	27.10	204.56	120.00	100.05	15.75	25.52	11.34	16.06	H
	Order Coordination for Specified Conversion Time (per LS		UAL	OCOSL		45.27							
2-WIRE HIG	GH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP												
	2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOO  2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -												
	Zone 1	1	UHL	UHL2X	8.5	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 2	2	UHL	UHL2X	11.26	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -		UHL	UHLZX	11.20	504.82	456.24	105.86	57.25	25.52	11.34	16.06	-
	Zone 3	3	UHL	UHL2X	16.1	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 4	4	UHL	UHL2X	21.25	504.82	456.24	105.86	57.25	25.52	11.34	16.06	
					21.20		100.21	100.00	01.20	20.02	11.01	10.00	T
	Order Coordination for Specified Conversion Time (per LS  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation		UHL	OCOSL		45.27							
	Zone 1	1	UHL	UHL2W	8.5	204.56	128.86	100.05	15.75	25.52	11.34	16.06	
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation				44.00	00450	100.00	100.05	45.75	05.50	44.04	10.00	
	Zone 2  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation	2	UHL	UHL2W	11.26	204.56	128.86	100.05	15.75	25.52	11.34	16.06	
	Zone 3	3	UHL	UHL2W	16.1	204.56	128.86	100.05	15.75	25.52	11.34	16.06	
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4	4	UHL	UHL2W	21.25	204.56	128.86	100.05	15.75	25.52	11.34	16.06	
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		45.27							-
4-WIRE HIG	GH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP												
	4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 1	4	UHL	UHL4X	10.36	531.21	482.63	105.86	57.25	25.52	11.34	16.06	
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -	- '	OFIL	UHL4A	10.30	331.21	402.03	103.00	57.25	25.52	11.34	10.00	T
	Zone 2	2	UHL	UHL4X	13.73	531.21	482.63	105.86	57.25	25.52	11.34	16.06	-
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 3	3	UHL	UHL4X	19.62	531.21	482.63	105.86	57.25	25.52	11.34	16.06	
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -												
	Zone 4	4	UHL	UHL4X	25.9	531.21	482.63	105.86	57.25	25.52	11.34	16.06	
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		45.27							
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 1	1	UHL	UHL4W	10.36	221.85	146.16	100.05	15.75	25.52	11.34	16.06	1
1	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation												t
	Zone 2	2	UHL	UHL4W	13.73	221.85	146.16	100.05	15.75	25.52	11.34	16.06	
				1		224.05	146.16	100.05	15.75	25.52	11.34	16.06	
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3	3	UHL	UHL4W	19.62	221.85							1
	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3     Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -												
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3	3	UHL	UHL4W UHL4W	19.62 25.9	221.85	146.16	100.05	15.75	25.52	11.34	16.06	
	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3     Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -							100.05	15.75	25.52	11.34	16.06	
4-WIDE DO	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3     4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4     Order Coordination for Specified Conversion Time (per LS)		UHL	UHL4W		221.85		100.05	15.75	25.52	11.34	16.06	
4-WIRE DS	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3 4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4  Order Coordination for Specified Conversion Time (per LS:  51 DIGITAL LOOP		UHL	UHL4W		221.85		100.05	15.75	25.52	11.34	16.06	
4-WIRE DS	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3     4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4     Order Coordination for Specified Conversion Time (per LS)	4	UHL	UHL4W OCOSL	25.9	221.85 45.27	146.16						
4-WIRE DS	4-Wire Dibuted HDSL Loop without manual service inquiry and facility reservation Zone 3 4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4 Order Coordination for Specified Conversion Time (per LS)  51 DIGITAL LOOP 4-Wire DS1 Digital Loop - Zone 4 4-Wire DS1 Digital Loop - Zone 5 4-Wire DS1 Digital Loop - Zone 5	1	UHL UHL USL	UHL4W OCOSL USLXX USLXX USLXX	25.9 50.99 67.58 96.58	221.85 45.27 599.09	146.16 373.9	133.53 133.53 133.53	56.25	25.52 25.52 25.52	11.34 11.34 11.34	16.06	
4-WIRE DS	4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :	1 2	UHL UHL USL USL	UHL4W OCOSL USLXX USLXX	25.9 50.99 67.58	221.85 45.27 599.09 599.09	146.16 373.9 373.9	133.53 133.53	56.25 56.25	25.52 25.52	11.34 11.34	16.06 16.06	
4-WIRE DS	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4  Order Coordination for Specified Conversion Time (per LS:  S1 DIGITAL LOOP  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :	1 2 3	UHL UHL USL USL USL USL USL	UHL4W OCOSL USLXX USLXX USLXX USLXX	25.9 50.99 67.58 96.58	221.85 45.27 599.09 599.09 599.09 599.09	373.9 373.9 373.9 373.9	133.53 133.53 133.53	56.25 56.25 56.25	25.52 25.52 25.52	11.34 11.34 11.34	16.06 16.06 16.06	
4-WIRE DS	4-Wire Dibuted HDSL Loop without manual service inquiry and facility reservation Zone 3 4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4 Order Coordination for Specified Conversion Time (per LS)  51 DIGITAL LOOP 4-Wire DS1 Digital Loop - Zone 4 4-Wire DS1 Digital Loop - Zone 5 4-Wire DS1 Digital Loop - Zone 5	1 2 3	UHL UHL USL USL USL	UHL4W OCOSL USLXX USLXX USLXX	25.9 50.99 67.58 96.58	221.85 45.27 599.09 599.09 599.09	373.9 373.9 373.9 373.9	133.53 133.53 133.53	56.25 56.25 56.25	25.52 25.52 25.52	11.34 11.34 11.34	16.06 16.06 16.06	
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4  Order Coordination for Specified Conversion Time (per LS:  S1 DIGITAL LOOP  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :	1 2 3	UHL UHL USL USL USL USL USL	UHL4W OCOSL USLXX USLXX USLXX USLXX	25.9 50.99 67.58 96.58	221.85 45.27 599.09 599.09 599.09 599.09	373.9 373.9 373.9 373.9	133.53 133.53 133.53	56.25 56.25 56.25	25.52 25.52 25.52	11.34 11.34 11.34	16.06 16.06 16.06	
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 4  Order Coordination for Specified Conversion Time (per LS)  1 DIGITAL LOOP  4-Wire DS1 Digital Loop - Zone  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :  4-Wire DS1 Digital Loop - Zone :  Order Coordination for Specified Conversion Time (per LS)	1 2 3	UHL UHL USL USL USL USL USL	UHL4W OCOSL USLXX USLXX USLXX USLXX	25.9 50.99 67.58 96.58	221.85 45.27 599.09 599.09 599.09 599.09	373.9 373.9 373.9 373.9	133.53 133.53 133.53	56.25 56.25 56.25	25.52 25.52 25.52	11.34 11.34 11.34	16.06 16.06 16.06	

	4 Wire Unbundled Digital 19.2 Kbps	3	UDL	UDL19	48.51	489	337.93	128.36	64.35		25.52	11.34	16.06	
1	4 Wire Unbundled Digital 19.2 Kbps	4	UDL	UDL19		489	337.93	128.36	64.35		25.52	11.34	16.06	
1	4 Wire Unbundled Digital Loop 56 Kbps - Zone	1	UDL	UDL56		489	337.93	128.36	64.35		25.52	11.34	16.06	+
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	2	UDL	UDL56	33.94	489	337.93	128.36	64.35		25.52	11.34	16.06	1
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	3	UDL	UDL56		489	337.93	128.36	64.35		25.52	11.34	16.06	T
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	4	UDL	UDL56		489	337.93	128.36	64.35		25.52	11.34	16.06	
	Order Coordination for Specified Conversion Time (per LS		UDL	OCOSL		45.27								
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	1	UDL	UDL64		489	337.93	128.36	64.35		25.52	11.34	16.06	
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	2	UDL	UDL64	33.94	489	337.93	128.36	64.35		25.52	11.34	16.06	_
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	3	UDL	UDL64		489	337.93	128.36	64.35		25.52	11.34	16.06	4
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	4	UDL	UDL64	64.02	489	337.93	128.36	64.35		25.52	11.34	16.06	+
	Order Coordination for Specified Conversion Time (per LS		UDL	OCOSL		45.27								
2 MIDE II-	bundled COPPER LOOP													
2-WIRE UNI	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility													+
	reservation - Zone 1	4	UCL	UCLPB	16.85	282.94	163.41	119.58	22.26		19.99	19.99	19.99	
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility		UCL	UCLFB	10.00	202.94	103.41	119.50	22.20		19.99	19.99	19.99	+
	reservation - Zone 2	2	UCL	UCLPB	22.34	282.94	163.41	119.58	22.26		19.99	19.99	19.99	
	2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility		UCL	OCLIB	22.34	202.34	103.41	119.50	22.20		19.95	15.55	13.33	+
	reservation - Zone 3	3	UCL	UCLPB	31.92	282.94	163.41	119.58	22.26		19.99	19.99	19.99	
	2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility	- 3	002	00210	002	202.07					10.00		.0.00	$^{+}$
	reservation - Zone 4	4	UCL	UCLPB	42.13	282.94	163.41	119.58	22.26		19.99	19.99	19.99	
														T
	Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC	:	50.29	50.29							
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility													
	reservation - Zone '	1	UCL	UCLPW	16.85	202.7	127	100.05	15.75		19.99	19.99	19.99	$\perp$
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility													
	reservation - Zone 2	2	UCL	UCLPW	22.34	202.7	127	100.05	15.75		19.99	19.99	19.99	
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility								l I					
	reservation - Zone (	3	UCL	UCLPW	31.92	202.7	127	100.05	15.75		19.99	19.99	19.99	1
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility													
	reservation - Zone 4	4	UCL	UCLPW	42.13	202.7	127	100.05	15.75		19.99	19.99	19.99	1
	Order Coordination for Unbundled Copper Loops (per loop)			1101.110		50.00	50.00					1		
			UCL	UCLMC		50.29	50.29							+
	2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility	1	1101	LICLO	47.74	260.00	150.00	110.50	22.20		10.00	10.00	10.00	
	reservation - Zone '	1	UCL	UCL2L	47.74	269.92	150.39	119.58	22.26		19.99	19.99	19.99	+
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2	2	UCL	UCL2L	70.63	269.92	150.39	119.58	22.26		19.99	19.99	19.99	
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	- 2	UCL	UCLZL	10.03	209.92	100.38	119.00	22.20		19.99	19.99	19.99	+
	reservation - Zone (	3	UCL	UCL2L	104.29	269.92	150.39	119.58	22.26		19.99	19.99	19.99	
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	3	UCL	JOLZL	104.23	203.32	100.00	113.30	LL.LU		10.00	13.33	13.33	+
	reservation - Zone 4	4	UCL	UCL2L	112.55	269.92	150.39	119.58	22.26		19.99	19.99	19.99	
			002	00222	1.12.00	200.02						10.00	.0.00	T
	Order Coordination for Unbundled Copper Loops (per loop)		UCL	UCLMC		50.29	50.29					1		
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		1											
	reservation - Zone 1	1	UCL	UCL2W	47.74	189.68	113.98	100.05	15.75		19.99	19.99	19.99	1
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility													
	reservation - Zone 2	2	UCL	UCL2W	70.63	189.68	113.98	100.05	15.75		19.99	19.99	19.99	
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility						I							
	reservation - Zone (	3	UCL	UCL2W	104.29	189.68	113.98	100.05	15.75		19.99	19.99	19.99	
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility						_			1 7				1
	reservation - Zone 4	4	UCL	UCL2W		189.68	113.98	100.05	15.75		19.99	19.99	19.99	1
	Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC	-	50.29	50.29					1		4
					ļ							1		4
	OMPARTITION IN CONTRACTOR NO. DOC			11=0	44.00	44.00	00.	05.00	7.00	$\rightarrow$	05.50	44.51	40.00	+
	2-Wire Unbundled Copper Loop - Non-Designed Zone	I 1	UEQ	UEQ2X	11.01	44.69	22.4	25.65	7.06		25.52	11.34	16.06	+
	2 Wire Unbundled Copper Loop - Non-Designed - Zone	1 2	UEQ	UEQ2X	12.67	44.69	22.4	25.65	7.06		25.52	11.34	16.06	+
	2 Wire Unbundled Copper Loop - Non-Designed - Zone	1 3	UEQ	UEQ2X UEQ2X	20.22 20.22	44.69 44.69	22.4	25.65 25.65	7.06		25.52 25.52	11.34 11.34	16.06	+
	2 Wire Unbundled Copper Loop - Non-Designed - Zone Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loo	1 4	UEQ UEQ	USBMC	20.22	50.29	22.4 50.29	25.05	7.06		20.02	11.34	16.06	+
	Engineering Information Documer		UEQ	USBIVIC		28.72	28.72					1	1	+
	Loop Testing - Basic 1st Half Hou		UEQ	URET1		78.92	78.92		<del>                                     </del>					+
	Loop Testing - Basic 1st Hall Hot		UEQ	URETA		23.33	23.33							+
														F
4-WIRE CO	PPER LOOP													
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1	1	1101	1101.45	22.04	224.00	244.70	122.00	20.20		10.00	10.00	10.00	
		1	UCL	UCL4S	22.24	331.29	211.76	133.82	28.26		19.99	19.99	19.99	+
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 2	2	UCL	UCL4S	25.82	331.29	211.76	133.82	28.26		19.99	19.99	19.99	
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation -	-   2	UCL	UCL4S	20.02	331.28	211.70	133.02	20.20		19.99	13.33	19.99	+
	Zone 3	3	UCL	UCL4S	28.12	331.29	211.76	133.82	28.26		19.99	19.99	19.99	
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation -	3	UCL	UUL43	20.12	331.23	211.70	100.02	20.20		10.00	13.33	10.00	+
				1		331.29	211.76	133.82	28.26		19.99	19.99	19.99	
	Zone 4	4	UCL	UCL4S	28.12									

	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4W	22.24	251.04	175.34	112.63	21.21		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -			UCL	UCL4VV	22.24	231.04	175.54	112.00	21.21		13.33	13.33	13.33	13.33
	Zone 2		2	UCL	UCL4W	25.82	251.04	175.34	112.63	21.21		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -														
	Zone 3		3	UCL	UCL4W	28.12	251.04	175.34	112.63	21.21		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 4		4	UCL		28.12	251.04	175.34	440.00	24.2		19.99	40.00	19.99	40.00
	Order Coordination for Unbundled Copper Loops (per loc		4	UCL	UCL4W UCLMC	28.12	50.29	50.29	112.63	21.2		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			UCL	UCLIVIC		50.29	50.29							
	reservation - Zone 1		1	UCL	UCL4L	82.53	318.27	198.74	133.82	28.26		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility														
	reservation - Zone 2		2	UCL	UCL4L	127.11	318.27	198.74	133.82	28.26		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		_												
	reservation - Zone (		3	UCL	UCL4L	138.69	318.27	198.74	133.82	28.26		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 4		4	UCL	UCL4L	138.69	318.27	198.74	133.82	28.26		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		50.29	50.29	100.02	20.20		10.00	10.00	10.00	10.00
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility														
	reservation - Zone '		1	UCL	UCL40	82.53	238.02	162.33	112.63	21.2		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility														
	reservation - Zone 2		2	UCL	UCL40	127.11	238.02	162.33	112.63	21.2		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone (		3	UCL	UCL4O	138.69	238.02	162.33	112.63	21.2		19.99	19.99	19.99	19.99
<del></del>	4-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		3	UCL	JUL4U	130.08	230.02	102.33	112.03	21.2		13.33	19.99	19.99	19.99
	reservation - Zone 4		4	UCL	UCL40	138.69	238.02	162.33	112.63	21.2		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		50.29	50.29							
										$\perp$					
OOD MODIFICATION					1					1					
OOP MODIFICATION	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal	to	-	UAL, UHL, UCL,											
	18k ft	ıo		UEQ, ULS	ULM2L		65.09	65.09							
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18			UCL, ULS	ULM2G		341.07	341.07							
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18	K													
	ft			UHL, UCL	ULM4L		65.09	65.09							
	Halanda Halland Marker Barranda (Landonia Alberta de Landonia de L			1101			044.07	044.07							
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18			UCL UAL, UHL, UCL,	ULM4G		341.07	341.07							
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled I			UEQ, UEF, ULS	ULMBT		65.13	65.13							
	<b>3</b>			024,02.,020											
					_										
UB-LOOPS															
	Distribution			HEARI	LIODOA		540.50	540.50				05.50	44.04	40.00	40.00
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L	I		UEANL	USBSA		540.53	540.53				25.52	11.34	16.06	16.06
		I I		UEANL UEANL	USBSA USBSB		540.53 45.21	540.53 45.21				25.52 25.52	11.34 11.34	16.06 16.06	16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L	I I		UEANL			45.21	45.21				25.52	11.34	16.06	16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L	 			USBSB										
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L			UEANL UEANL UEANL	USBSB USBSC USBSD		45.21 379.25 111.97	45.21 379.25 111.97				25.52 25.52 25.52	11.34 11.34 11.34	16.06 16.06	16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		1	UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2	10.75	45.21 379.25 111.97 131.42	45.21 379.25 111.97 61.83	90.07	13.33		25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34	16.06 16.06 16.06	16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		2	UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2 USBN2	14.4	45.21 379.25 111.97 131.42 131.42	45.21 379.25 111.97 61.83 61.83	90.07	13.33		25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		2	UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2 USBN2 USBN2 USBN2	14.4 18.53	45.21 379.25 111.97 131.42 131.42 131.42	45.21 379.25 111.97 61.83 61.83 61.83	90.07 90.07	13.33 13.33		25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L   Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		2	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2 USBN2 USBN2 USBN2 USBN2	14.4	45.21 379.25 111.97 131.42 131.42 131.42 131.42	45.21 379.25 111.97 61.83 61.83 61.83 61.83	90.07	13.33		25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		2	UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2 USBN2 USBN2 USBN2	14.4 18.53	45.21 379.25 111.97 131.42 131.42 131.42	45.21 379.25 111.97 61.83 61.83 61.83	90.07 90.07	13.33 13.33		25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		2 3 4 1 2	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN2 USBN4 USBN4	14.4 18.53 23.19 11.29 19.41	45.21 379.25 111.97 131.42 131.42 131.42 45.27 45.27 157.85 157.85	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26	90.07 90.07 90.07 101.8 101.8	13.33 13.33 13.33 18.57 18.57		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone	1 1 1 1 1 1 1	2 3 4 1 2 3	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2 USBN2 USBN2 USBN2 USBN2 USBN4 USBN4	14.4 18.53 23.19 11.29 19.41 20.9	45.21 379.25 111.97 131.42 131.42 131.42 45.27 157.85 157.85	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26	90.07 90.07 90.07 101.8 101.8 101.8	13.33 13.33 13.33 18.57 18.57 18.57		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L   Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		2 3 4 1 2	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBNC USBN4 USBN4 USBN4	14.4 18.53 23.19 11.29 19.41	45.21 379.25 111.97 131.42 131.42 131.42 45.27 157.85 157.85 157.85 157.85	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88	90.07 90.07 90.07 101.8 101.8	13.33 13.33 13.33 18.57 18.57		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair		2 3 4 1 2 3	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2 USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN5	14.4 18.53 23.19 11.29 19.41 20.9 20.9	45.21 379.25 111.97 131.42 131.42 131.42 131.42 45.27 157.85 157.85 157.85 45.27	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27	90.07 90.07 90.07 101.8 101.8 101.8 90.07	13.33 13.33 13.33 13.57 18.57 18.57 13.33		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK		2 3 4 1 2 3	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN4	14.4 18.53 23.19 11.29 19.41 20.9	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 157.85 45.27 105.88	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29	90.07 90.07 90.07 101.8 101.8 101.8	13.33 13.33 13.33 18.57 18.57 18.57		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L		2 3 4 1 2 3	UEANL UEANL	USBSB USBSC USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN4	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 157.85 45.27 105.88 45.27	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27	90.07 90.07 90.07 101.8 101.8 90.07	13.33 13.33 13.33 13.33 18.57 18.57 18.57 13.33		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK		2 3 4 1 2 3 4	UEANL UEANL	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBMC USBR2 USBN6	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29	90.07 90.07 90.07 101.8 101.8 101.8 90.07	13.33 13.33 13.33 13.57 18.57 18.57 13.33		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair		2 3 4 1 2 3 4	UEANL UEANL	USBSB USBSC USBSC USBSC USBN2 USBN2 USBN2 USBN4	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39	45.21 379.25 111.97 131.42 131.42 131.42 131.42 45.27 157.85 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 48.76 45.27	90.07 90.07 90.07 101.8 101.8 101.8 90.07 90.07	13.33 13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-U   Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-U   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-U   Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-U   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop AWire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone		2 3 4 1 2 3 4	UEANL UEANL	USBSB USBSC USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN6 USBRC	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 157.85 45.27 105.88 45.27 105.88 45.27 118.34 45.27 131.42 131.42	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 61.83 61.83	90.07 90.07 90.07 101.8 101.8 101.8 90.07 90.07 101.8	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L		2 3 4 1 2 3 4	UEANL UEF	USBSB USBSC USBSD USBN2 USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN6 USBR2 USBNC USBR2 USBNC USBR2 USBNC USBR3 USBNC USBR3 USBNC USBR3 USBNC USBR3 USBNC USBR3 USBNC USBR3 USBNC USBR3 USBNC	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 118.34 157.85	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 48.76 48.76 45.27 61.83 61.83 61.83	90.07 90.07 90.07 90.07 101.8 101.8 90.07 90.07 90.07 90.07 90.07	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 18.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-U   Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-U   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-U   Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-U   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone		2 3 4 1 2 3 4	UEANL UEF	USBSB USBSC USBSC USBSC USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN7 USBN7 USBN8	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 157.85 157.85 157.85 157.85 157.85 157.85 157.85 158.84 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131.42 131.42 131.42 131.42	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 46.27 61.83 61.83 61.83 61.83	90.07 90.07 90.07 101.8 101.8 101.8 90.07 90.07 101.8	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33		25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per A-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per A-Wire Analog Voice Grade Loop Distribution - Zone   Zwire Copper Unbundled Sub-Loop Distribution - Zone   Zwire Copper Unbundled Sub-Loop Distribution - Zone   Zwire Copper Unbundled Sub-Loop Distribution - Zone   Zwire Copper Unbundled Sub-Loop Distribution - Zone		2 3 4 1 2 3 4 1 2 3 4	UEANL UEF	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBNC	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 118.34 157.85	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 46.27 48.27 61.83 61.83 61.83 61.83 61.83	90.07 90.07 90.07 90.07 101.8 101.8 90.07 90.07 90.07 90.07 90.07	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 18.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-U   Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-U   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-U   Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-U   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone		2 3 4 1 2 3 4 1 2 3 4	UEANL UEF UEF	USBSB USBSC USBSC USBSC USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN6 USBN7 USBN7 USBN8	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6 12.57	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131.42 131.42 131.42 131.42 131.42 131.42 131.42 131.42 131.42 131.42	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 46.27 61.83 61.83 61.83 61.83	90.07 90.07 90.07 90.07 101.8 101.8 90.07 90.07 101.8 90.07 90.07 90.07 90.07	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33 13.33 13.33 13.33		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06  16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Guiding Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone		1 2 3 4 1 1 2 3 4 1 1 2 3 4 1 1 2 3 4	UEANL UEF UEF	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN6 USBN6 USBR6 USBR6 USBR7 USBR7 USBR7 USBR8 USBR7 USBR8 USBR8 USBR9	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6 12.57 7.46 14.58 18.61	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 105.88 45.27 48.76 45.27 61.83	90.07 90.07 90.07 90.07 101.8 101.8 101.8 90.07 101.8 90.07 90.07 90.07 90.07 101.8 101.8 101.8	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33 13.33 13.33 13.33 13.33 13.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone		1 2 3 4 1 2 3 4 1 2 3 4	UEANL UEF UEF UEF	USBSB  USBSC  USBSD  USBN2  USBN2  USBN2  USBN2  USBN4  UCSL4  UC	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6 12.57 7.46	45.21 379.25 111.97 131.42 131.42 131.42 131.42 45.27 157.85 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131.42 131.42 131.42 131.42 131.42 131.42 157.85 157.85 157.85	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 48.76 45.27 61.83	90.07 90.07 90.07 90.07 101.8 101.8 90.07 90.07 90.07 90.07 90.07 90.07 90.07	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33 13.33 13.33 13.33 13.33 13.57 18.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-L   Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone   Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair		1 2 3 4 1 2 3 4 1 2 3 4 1 2 3 4	UEANL UEF UEF	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN6 USBN6 USBR6 USBR6 USBR7 USBR7 USBR7 USBR8 USBR7 USBR8 USBR8 USBR9	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6 12.57 7.46 14.58 18.61	45.21 379.25 111.97 131.42 131.42 131.42 131.42 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131	45.21 379.25 111.97 61.83 61.83 61.83 45.27 88.26 88.26 105.88 45.27 48.76 45.27 61.83	90.07 90.07 90.07 90.07 101.8 101.8 101.8 90.07 101.8 90.07 90.07 90.07 90.07 101.8 101.8 101.8	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33 13.33 13.33 13.33 13.33 13.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06
Sub-Loop I	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Guiding Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loop per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone Order Coordination for Unbundled Sub-Loop Distribution - Zone		1 2 3 4 1 2 3 4 1 2 3 4 1 2 3 4	UEANL UEF UEF UEF	USBSB  USBSC  USBSD  USBN2  USBN2  USBN2  USBN2  USBN4  UCSL4  UC	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6 12.57 7.46 14.58 18.61	45.21 379.25 111.97 131.42 131.42 131.42 131.42 45.27 157.85 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131.42 131.42 131.42 131.42 131.42 131.42 157.85 157.85 157.85	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 48.76 45.27 61.83	90.07 90.07 90.07 90.07 101.8 101.8 101.8 90.07 101.8 90.07 90.07 90.07 90.07 101.8 101.8 101.8	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33 13.33 13.33 13.33 13.33 13.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Guiding Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loop per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone Order Coordination for Unbundled Sub-Loop Distribution - Zone		1 2 3 4 1 2 3 4 1 2 3 4 1 2 3 4	UEANL UEF UEF UEF	USBSB  USBSC  USBSD  USBN2  USBN2  USBN2  USBN2  USBN4  UCSL4  UC	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6 12.57 7.46 14.58 18.61	45.21 379.25 111.97 131.42 131.42 131.42 131.42 45.27 157.85 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131.42 131.42 131.42 131.42 131.42 131.42 157.85 157.85 157.85	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 48.76 45.27 61.83	90.07 90.07 90.07 90.07 101.8 101.8 101.8 90.07 101.8 90.07 90.07 90.07 90.07 101.8 101.8 101.8	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33 13.33 13.33 13.33 13.33 13.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06
Sub-Loop I	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Guiding Equipment Room - CLEC Feeder Facility Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-L Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loop per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone Order Coordination for Unbundled Sub-Loop Distribution - Zone		1 2 3 4 1 2 3 4 1 2 3 4 1 2 3 4	UEANL UEF UEF UEF	USBSB USBSC USBSC USBN2 USBN2 USBN2 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN4 USBN6 USBSC	14.4 18.53 23.19 11.29 19.41 20.9 20.9 2.79 5.39 8.74 9.31 10.6 12.57 7.46 14.58 18.61	45.21 379.25 111.97 131.42 131.42 131.42 131.42 45.27 157.85 157.85 157.85 157.85 45.27 105.88 45.27 118.34 45.27 118.34 45.27 131.42 131.42 131.42 131.42 131.42 131.42 131.42 157.85 157.85 157.85	45.21 379.25 111.97 61.83 61.83 61.83 61.83 45.27 88.26 88.26 88.26 105.88 45.27 36.29 45.27 48.76 45.27 61.83	90.07 90.07 90.07 90.07 101.8 101.8 101.8 90.07 101.8 90.07 90.07 90.07 90.07 101.8 101.8 101.8	13.33 13.33 13.33 18.57 18.57 18.57 13.33 13.33 13.33 13.33 13.33 13.33 13.33 13.57		25.52 25.52	11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34 11.34	16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06 16.06	16.06  16.06

			UEA,											
	USI Fooder DS0 Set up and Green Day Josephine and 25 painted to		UDN,UCL,UDL,UE C	USBFX		45.21	45.21							
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u USL Feeder DS1 Set-up at DSX location, per DS1 terminatic	_	USL	USBF7		534.46	11.3							+
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zon	1	UEA	USBFA	12.34	185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zoni	2	UEA	USBFA	17.1	185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zon	3	UEA	USBFA	25.55	185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start Loop, Voice Grade - Zon Order Coordination for Specified Conversion Time, per LSR	4	UEA UEA	USBFA	32.36	185.12 45.27	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni	1	UEA	USBFB		45.27 185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni  Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni	2	UEA	USBFB		185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zon	3	UEA	USBFB		185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zoni	4	UEA	USBFB	32.36	185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Order Coordination for Specified Time Conversion, per LSR		UEA	OCOSL		45.27								
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon	1	UEA	USBFC	12.34	185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon	2	UEA	USBFC	17.1	185.12	112.19	108.13	26.82		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zone	3	UEA	LICDEC	25.55	405.40	112.19	400.40	20.02		10.00	40.00	40.00	19.99
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon	4	UEA	USBFC	25.55 32.36	185.12 185.12	112.19	108.13 108.13	26.82 26.82		19.99 19.99	19.99 19.99	19.99 19.99	19.99
	Oribundied Sub-Loop reeder Loop, 2 write Neverse Ballery, Voice Grade - 2011		OLA	USDI C	32.30	100.12	112.13	100.13	20.02		19.55	15.55	13.33	13.33
	Order Coordination For Specified Conversion Time, per LS		UEA	OCOSL		45.27								
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone	1	UEA	USBFD	28.24	213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone	2	UEA	USBFD	32.51	213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zoni	3		USBFD		213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zoni	4	UEA	USBFD	41.5	213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
	Order Consideration For Consideral Consideration Time Book 6		1154	000001		45.07								
	Order Coordination For Specified Conversion Time, Per L\$ Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	1	UEA UEA	OCOSL USBFE	28.24	45.27 213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	2	UEA	USBFE		213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	3	UEA	USBFE		213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Analog Voice Grade Loop-Start Loop - Zoni	4	UEA	USBFE		213.89	139.06	126.45	35.02		19.99	19.99	19.99	19.99
														T
	Order Coordination For Specified Conversion Time, Per LS		UEA	OCOSL		45.27								
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone	1	UDN	USBFF		211.41	136.58	110.37	26.07		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	2	UDN	USBFF		211.41	136.58	110.37	26.07		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	3	UDN	USBFF		211.41 211.41	136.58 136.58	110.37 110.37	26.07 26.07		19.99 19.99	19.99 19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	4	UDN	USBFF	48.23	211.41	136.58	110.37	26.07		19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, Per LS		UDN	OCOSL		45.27								
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	1	UDC	USBFS	22.46	211.41	136.58	110.37	26.07		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	2	UDC	USBFS		211.41	136.58	110.37	26.07		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	3	UDC	USBFS	37.36	211.41	136.58	110.37	26.07		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	4	UDC	USBFS	48.23	211.41	136.58	110.37	26.07		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	1	USL	USBFG	76.62	202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	2	USL	USBFG	178.54	202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	3	USL	USBFG	224.48 538.86	202.5 202.5	127.66 127.66	126.45 126.45	35.02 35.02		19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Oriburidied Sub-Loop Feeder Loop, 4-Wile DST - Zorie	4	USL	USBFG	330.00	202.5	127.00	120.45	33.02		19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, Per LS		USL	OCOSL		45.27								
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone	1	UCL	USBFH	7.07	167.34	92.51	105.53	21.21		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	2	UCL	USBFH		167.34	92.51	105.53	21.21		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	3	UCL	USBFH		167.34	92.51	105.53	21.21		19.99	19.99	19.99	19.99
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone	4	UCL	USBFH	4.13	167.34	92.51	105.53	21.21		19.99	19.99	19.99	19.99
	0.1.0		1101	0000:		45.07					1			
	Order Coordination For Specified Conversion Time, per LS	1	UCL	OCOSL USBFJ	16.34	45.27 201.71	126.88	118.58	27.15		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	2	UCL	USBFJ	16.34	201.71	126.88	118.58	27.15		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	3		USBFJ	11.06	201.71	126.88	118.58	27.15		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	4	UCL	USBFJ	11.06	207.71	126.88	118.58	27.15		19.99	19.99	19.99	19.99
														1
	Order Coordination For Specified Conversion Time, per LS		UCL	OCOSL		45.27								
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	1	UDL	USBFN		202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	2	UDL	USBFN		202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	3	UDL	USBFN	30.57	202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	4		USBFN		202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	1	UDL UDL	USBFO		202.5	127.66 127.66	126.45	35.02		19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	3	UDL	USBFO	30.57	202.5 202.5	127.66	126.45 126.45	35.02 35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	4	UDL	USBFO	28.9	202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	The state of the s		332	000.0	20.0	202.0	121.00	.200	55.52			10.00	.0.00	10.00
	Order Coordination For Specified Time Conversion, per LS		UDL	OCOSL		45.27								
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1	UDL	USBFP		202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	2	UDL	USBFP		202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	3	UDL	USBFP		202.5	127.66	126.45	35.02	19.99	19.99	19.99	19.99	19.99
1 1	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	4	UDL	USBFP	28.9	202.5	127.66	126.45	35.02		19.99	19.99	19.99	19.99

Unbundled	d Sub-Loop Modification												
Silbulated	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W	+ +									+		
	PR	UEF	ULM2X		355.23	12.25				25.52	11.34	16.06	16.0
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W												
	PR	UEF	ULM4X		355.23	12.25				25.52	11.34	16.06	16.
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR												
	unloaded	UEF	ULM4T		559.8	14.28				25.52	11.34	16.06	16.
Unbundled	Network Terminating Wire (UNTW)												
	Unbundled Network Terminating Wire (UNTW) per Pa	UENTW	UENPP	0.37	62.97	62.97				25.52	11.34	16.06	16.
N. c	tudes Device (AUD)												
Network in	nterface Device (NID)	LIENTIN	LINIDAO		07.05	57.00				05.50	44.04	40.00	40
	Network Interface Device (NID) - 1-2 line Network Interface Device (NID) - 1-6 line	UENTW UENTW	UND12 UND16		87.05 129.67	57.38 100				25.52 25.52	11.34 11.34	16.06 16.06	16 16
	Network Interface Device (NID) - 1-6 line  Network Interface Device Cross Connect - 2 V	UENTW	UNDC2		11.79	11.79				25.52	11.34	16.06	16
	Network Interface Device Cross Connect - 4V	UENTW	UNDC4		11.79	11.79				25.52	11.34	16.06	16
	Network interface Device cross connect. 44	OLIVIV	014004		11.75	11.75				20.02	11.04	10.00	10
LED LOOP CO	DICENTRATION												
	Unbundled Loop Concentration - System A (TR00)	ULC	UCT8A	442.98	649.95	649.95				19.99	19.99	19.99	19
	Unbundled Loop Concentration - System B (TR00)	ULC	UCT8B	57.94	270.81	270.81				19.99	19.99	19.99	19
	Unbundled Loop Concentration - System A (TR30)	ULC	UCT3A	484.01	649.95	649.95				19.99	19.99	19.99	19
	Unbundled Loop Concentration - System B (TR30)	ULC	UCT3B	97.64	270.81	270.81				19.99	19.99	19.99	19
	Unbundled Loop Concentration - DS1 Loop Interface Ca	ULC	UCTCO	5.5	126.4	92.02	34.38	9.62		19.99	19.99	19.99	19
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Car	UDN	ULCC1	8.74	21.04	20.93	11.04	10.97		19.99	19.99	19.99	19
	Unbundled Loop Concentration - UDC Loop Interface (Brite Car	UDC	ULCCU	8.74	21.04	20.93	11.04	10.97		19.99	19.99	19.99	19
	Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop		l						1				
-	Interface (POTS Card)	UEA	ULCC2	2.18	21.04	20.93	11.04	10.97	-	19.99	19.99	19.99	19
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface	1154		40.00	04.04	00.00	44.04	40.07		40.00	40.00	40.00	
	(SPOTS Card) Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca	UEA UEA	ULCCR ULCC4	12.99 7.75	21.04 21.04	20.93 20.93	11.04 11.04	10.97 10.97		19.99 19.99	19.99 19.99	19.99 19.99	19
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca	ULC	UCTTC		21.04	20.93	11.04	10.97		19.99	19.99	19.99	19
	Unbundled Loop Concentration - TEST CIRCOTT Call  Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa	UDL	ULCC7	11.47	21.04	20.93	11.04	10.97		19.99	19.99	19.99	19
	Unbundled Loop Concentration - Digital 16.2 Rops Data Loop Interfa	UDL	ULCC5	11.47	21.04	20.93	11.04	10.97		19.99	19.99	19.99	19
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa	UDL	ULCC6	11.47	21.04	20.93	11.04	10.97		19.99	19.99	19.99	19
	Cribarialea 2009 Correctification - Bigital of Tribpe Bata 2009 Interna	UDL				20.00	11.01	10.07		10.00	10.00	10.00	- 10
LED SUB-LOC	DP CONCENTRATION (OUTSIDE CO)												
	P CONCENTRATION (OUTSIDE CO)  DNING ONLY - NO RATE												
		UENTW	UNDBX										
	ONING ONLY - NO RATE	UENTW	UENCE										
	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate	UENTW UEANL,UEF,UEC	UENCE Q,										
	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation	UENTW UEANL,UEF,UEC	UENCE Q, UNECN										
	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE	UENCE Q, UNECN										
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,U	UENCE Q, UNECN DL										
	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE	UENCE Q, UNECN	0	0								
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,U	UENCE Q, UNECN DL JL UNECN	0	0								
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,U UEA,UDN,UCL,U	UENCE Q, UNECN DL UNECN										
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,U	UENCE Q, UNECN DL JL UNECN	0	0								
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE, UDN,UEA,UHL,L C UEA,UDN,UCL,U	UENCE Q, UNECN DL UNECN D USBFQ	0	0								
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,U UEA,UDN,UCL,U	UENCE Q, UNECN DL UNECN D USBFQ										
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC, UC ,UDN,UEA,UHL,L C UEA,UDN,UCL,U C UEA,USL,UCL,UE	UENCE  UNECN  UNECN  UNECN  UNECN  UNBFQ  USBFQ	0	0								
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE, UDN,UEA,UHL,L C UEA,UDN,UCL,U	UENCE Q, UNECN DL UNECN D USBFQ	0	0								
	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC, UC ,UDN,UEA,UHL,L C UEA,UDN,UCL,U C UEA,USL,UCL,UE	UENCE  UNECN  UNECN  UNECN  UNECN  UNBFQ  USBFQ	0	0								
HER, PROVISIO	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE, UDN,UEA,UHL,U C UEA,UDN,UCL,U C UEA,USL,UCL,UC	UENCE UNECN DL UNECN D USBFQ DL USBFR CCOSF	0	0 0 0								
IER, PROVISIO	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE, UDN,UEA,UHL,U C UEA,UDN,UCL,U C UEA,USL,UCL,UC	UENCE UNECN DL UNECN D USBFQ DL USBFR CCOSF	0	0 0 0								
IER, PROVISIO	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,L C UEA,UDN,UCL,U C UEA,USL,UCL,UE USL USL	UENCE UNECN UNECN UNECN UNECN UNECN USBFQ CCOSF	0 0 0 0	0 0 0								
IER, PROVISIO	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  NDLED LOCAL LOOP  In In In In Immunity Immunity Imperiod  I High Capacity Unbundled Local Loop - DS3 - Per Mile per mon	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE, UDN,UEA,UHL,L C UEA,UDN,UCL,U UEA,USL,UCL,UE USL USL	UENCE  UNECN  UNECN  UNECN  UNECN  USBFQ  CCOSF  CCOEF	0 0 0 0	0 0 0 0								
IER, PROVISIO	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  NDLED LOCAL LOOP  NOLED LOCAL LOOP  North minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,L C UEA,UDN,UCL,U C UEA,USL,UCL,UE USL USL USS UES UES UES	UENCE UNECN UNECN UNECN UNECN UNECN UNECN UNECN UNECN UNECN USBFQ CCOSF CCOSF	0 0 0 0 14.16 396.3	0 0 0	527.16	244.7	171.16		31.26	31.26	3.91	3
IER, PROVISIO	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  NDLED LOCAL LOOP  NDLED LOCAL LOOP  Input minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UC ,UDN,UEA,UHL,L C UEA,UDN,UCL,U UEA,USL,UCL,UC USL USL USL USL UES UES UES UES UES UDLSX	UENCE UNECN UNECN UNECN UNECN UNECN UNBFQ USBFQ CCOSF CCOSF 11.5ND UE3PX 11.5ND	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0 901.82								
IER, PROVISIO	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  NDLED LOCAL LOOP  NOLED LOCAL LOOP  North minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,L C UEA,UDN,UCL,U C UEA,USL,UCL,UE USL USL USS UES UES UES	UENCE UNECN UNECN UNECN UNECN UNECN UNECN UNECN UNECN UNECN USBFQ CCOSF CCOSF	0 0 0 0 14.16 396.3	0 0 0 0	527.16 527.16	244.7	171.16 171.16		31.26 31.26	31.26 31.26	3.91	
PACITY UNBU	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  NDLED LOCAL LOOP  NDLED LOCAL LOOP  Input minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UC ,UDN,UEA,UHL,L C UEA,UDN,UCL,U UEA,USL,UCL,UC USL USL USL USL UES UES UES UES UES UDLSX	UENCE UNECN UNECN UNECN UNECN UNECN UNBFQ USBFQ CCOSF CCOSF 11.5ND UE3PX 11.5ND	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0 901.82								
IER, PROVISIO	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  NDLED LOCAL LOOP  Nonth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UC ,UDN,UEA,UHL,L C UEA,UDN,UCL,U UEA,USL,UCL,UC USL USL USL USL UES UES UES UES UES UDLSX	UENCE UNECN UNECN UNECN UNECN UNECN UNBFQ USBFQ CCOSF CCOSF 11.5ND UE3PX 11.5ND	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0 901.82								
PACITY UNBU	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  Unbundled DS1 Loop - STS1 - Per Mile per mon High Capacity Unbundled Local Loop - DS3 - Pacility Termination per mor High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor Loop Makeup - Preordering Without Reservation, per working or spare facility queried	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,L C UEA,UDN,UCL,UE UEA,USL,UCL,UE USL USL USL UES UES UES UDLSX UDLSX	UENCE UNECN UNECN UNECN UNECN UNBFQ USBFQ CCOSF CCOSF CCOEF	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0	527.16							
PACITY UNBU	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  Unbundled DS1 Loop - Expanded	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,L C UEA,UDN,UCL,U UEA,USL,UCL,UE USL USL USL USS UUSS UUSS UUSS UUSS U	UENCE UNECN UNECN UNECN UNECN UNBFQ USBFQ CCOSF CCOSF CCOEF	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0 901.82 901.82	527.16 47.9							
PACITY UNBU	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  NDLED LOCAL LOOP  North minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,L C UEA,UDN,UCL,UE UEA,USL,UCL,UE USL USL USL UES UES UES UDLSX UDLSX	UENCE UNECN UNECN UNECN UNECN UNBFQ USBFQ CCOSF CCOSF CCOEF	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0	527.16							
PACITY UNBU	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No Rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  NDLED LOCAL LOOP  NOLED LOCAL LOOP  Nonth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Manual).	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UC ,UDN,UEA,UHL,L C UEA,UDN,UCL,UC C UEA,USL,UCL,UC USL USL USL USL USS UUSS UUSS UUSS UU	UENCE  UNECN  UNECN  UNECN  UNECN  UNECN  USBFQ  CCOSF  CCOSF  CCOEF  1L5ND  UE3PX  1L5ND  UDLS1  UMKLW  UMKLW	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0 901.82 901.82	527.16 47.9 50.79							3.
PACITY UNBU	NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  NDLED LOCAL LOOP  North minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UE ,UDN,UEA,UHL,L C UEA,UDN,UCL,U UEA,USL,UCL,UE USL USL USL USS UUSS UUSS UUSS UUSS U	UENCE UNECN UNECN UNECN UNECN UNBFQ USBFQ CCOSF CCOSF CCOEF	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0 901.82 901.82	527.16 47.9							
PACITY UNBU	DNING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No Rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  Unbundled DS1 Loop - Expanded Superframe Format option - no rs  NDLED LOCAL LOOP  NOLED LOCAL LOOP  Nonth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).  Loop MakeupWith or Without Reservation, per working or spare facility queried (Manual).	UENTW UEANL,UEF,UEC UENTW UAL,UCL,UDC,UC ,UDN,UEA,UHL,L C UEA,UDN,UCL,UC C UEA,USL,UCL,UC USL USL USL USL USS UUSS UUSS UUSS UU	UENCE  UNECN  UNECN  UNECN  UNECN  UNECN  USBFQ  CCOSF  CCOSF  CCOEF  1L5ND  UE3PX  1L5ND  UDLS1  UMKLW  UMKLW	0 0 0 0 14.16 396.3 14.16	0 0 0 0 0 901.82 901.82	527.16 47.9 50.79							

	l li	ine Sharing Splitter, per System 96 Line Capaci	1	ULS	ULSDA	206.52	377.08	0	354.29	0	0				
		ine Sharing Splitter, per System 35 Eine Capaci	i	ULS	ULSDB		377.08	0	354.29	0	0				
		ine Sharing Splitte, Per System, 8 Line Capaci	ı	ULS	ULSD8	17.21	377.08	0	354.29	0	0				
		ine Sharing - per Line Activatio	l	ULS	ULSDC		36.96	21.17	19.93	9.78		25.52	11.34	16.06	16.06
	Li	ine Sharing - per Subsequent Activity per Line Rearrangeme	ı	ULS	ULSDS		32.73	16.35				25.52	11.34		
	Li	ine Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lines													
		16 pair)	1	ULS	ULSDG		57.62		11.33						
UNBUNDLED	D TRANSPORT	Г													
	COMMON TRA	ANSPORT (Shared)													
		common Transport - Per Mile, Per MOl				0.0000091									
	С	Common Transport - Facilities Termination Per MO				0.0004281									
	NOTE: INTER	OFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS3 =	one montl	h, DS3 and above	four month:	s									
	INTEROFFICE	CHANNEL - DEDICATED TRANSPORT - VOICE GRADE													
		hteroffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per moi		U1TVX	1L5XX	0.0112									
		nteroffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination		01117	120701	0.0112									
	pe	er month		U1TVX	U1TV2	24.75	80.96	54.74	34.27	14.12		31.26	31.26	3.91	3.91
		nteroffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per nonth		U1TVX	1L5XX	0.0112									
		nteroffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination er month		U1TVX	U1TR2		80.96	54.74	34.27	14.12		31.26	31.26	3.91	3.91
							30.30	57.74	J 7.21			31.20	51.20	5.51	5.51
		hteroffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month hteroffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination		U1TVX	1L5XX	0.0112									
		er month		U1TVX	U1TV4	21.75	80.96	54.74	34.27	14.12		31.26	31.26	3.91	3.91
	In	nteroffice Channel - Dedicated Transport - 56 kbps - per mile per mon		U1TDX	1L5XX	0.0112									
	In	nteroffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mor		U1TDX	U1TD5	17.24	80.97	54.74	34.27	14.12		31.26	31.26	3.91	3.91
		nteroffice Channel - Dedicated Transport - 64 kbps - per mile per mon		U1TDX	1L5XX										
	In	nteroffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mor		U1TDX	U1TD6	17.24	80.97	54.74	34.27	14.12		31.26	31.26	3.91	3.91
	INTEROFFICE	CHANNEL - DEDICATED TRANSPORT - DS1													
		nteroffice Channel - Dedicated Channel - DS1 - Per Mile per mor		U1TD1	1L5XX										
	In	nteroffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor		U1TD1	U1TF1	63	178.29	163.4	33.48	29.57		31.26	31.26	3.91	3.91
		CHANNEL - DEDICATED TRANSPORT- DS3													
		nteroffice Channel - Dedicated Transport - DS3 - Per Mile per mor		U1TD3 U1TD3	1L5XX U1TF3		556.75	325.07	123.28	119.71		31.26	31.26	3.91	3.91
	ın	nteroffice Channel - Dedicated Transport - DS3 - Facility Termination per mor		01103	UTIF3	705.42	556.75	325.07	123.28	119.71		31.26	31.26	3.91	3.91
		CHANNEL - DEDICATED TRANSPORT- STS-1													
		nteroffice Channel - Dedicated Transport - STS-1 - Per Mile per mon		U1TS1	1L5XX										
	In	teroffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor		U1TS1	U1TFS	707.97	556.75	325.07	123.28	119.71		31.26	31.26	3.91	3.91
	LOCAL CHAN	NEL - DEDICATED TRANSPORT													
		L CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one mont	h, DS3 an												
		ocal Channel - Dedicated - 2-Wire Voice Grade Per Month ocal Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mon		ULCVX	ULDV2 ULDR2		385.68 385.68	67.24 67.24	75.04 75.04	6.55 6.55		31.26 31.26	31.26 31.26	3.91 3.91	3.91
		ocal Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor ocal Channel - Dedicated - 4-Wire Voice Grade per mon		UNCVX	ULDR2 ULDV4		385.68 386.55	67.24	75.04	6.55 7.51		31.26 31.26	31.26 31.26	3.91	3.91
		ocal Channel - Dedicated - DS1 per month - Zone	1	ULDD1	ULDF1		354.47	307.02	45.45	31.25		31.26	31.26	3.91	3.91
	Lo	ocal Channel - Dedicated - DS1 per month - Zone	2	ULDD1	ULDF1	47.27	354.47	307.02	45.45	31.25		31.26	31.26	3.91	3.91
	Le	ocal Channel - Dedicated - DS1 per month - Zone	3	ULDD1	ULDF1	553.26	354.47	307.02	45.45	30.52		31.26	31.26	3.91	3.91
		ocal Channel - Dedicated - DS3 - Per Mile per mon		ULDD3	1L5NC	11.02	901.82	527.16	244.7	171.16		31.26	31.26	3.91	3.91
		ocal Channel - Dedicated - DS3 - Facility Termination per mon ocal Channel - Dedicated - STS-1- Per Mile per mon		ULDD3 ULDS1	ULDF3 1L5NC	455.69 11.02	901.82	527.10	244.7	171.16		31.20	31.20	3.91	3.91
		ocal Channel - Dedicated - STS-1 - Facility Termination per mon		ULDS1	ULDFS	449.26	901.82	527.16	244.7	171.16		31.26	31.26	3.91	3.91
MULTIPLEXE															
JL I IF LEX		hannelization - DS1 to DS0 Channel Syste		UXTD1	MQ1	125.29	181.84	124.98	21.57	20.05		31.26	31.26	3.91	3.91
	0	CU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)		UDL	1D1DD	1.49	13.13	9.41							
	2-	-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per mont		UDN	UC1CA		13.13	9.41				1			ļ <u></u>
		oice Grade COCI - DS1 to DS0 Channel System - per mon		UEA	1D1VG		13.13	6.41	00.11	65.47		31.26	24.00	3.91	3.91
		IS3 to DS1 Channel System per mont TS1 to DS1 Channel System per mont		UXTD3 UXTS1	MQ3 MQ3	207.87 207.87	355.8	187.69	68.11	65.17		31.26 31.26	31.26 31.26	3.91	3.91
		IS3 Interface Unit (DS1 COCI) used with Loop per mont		USL	UC1D1	15.78	13.13	9.41				31.20	31.20	3.91	3.91
		The state of the s		302			.5.10								
DARK FIBER															
		lark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local		UDF	1L5DC	66.94									
						00.34	1	1	11	1	1			1	1

				1								
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -											
	Interoffice Channe	UDF	1L5DF	32.13								
	NRC Dark Fiber - Interoffice Channe	UDF	UDF14		1276.46	275.36	649.31	404.8	31.26	31.26	3.91	3.91
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local											
	Loop	UDF	1L5DL	66.94								
	NRC Dark Fiber - Local Loop	UDF	UDFL4		1276.46	275.36	649.31	404.8	31.26	31.26	3.91	3.91
TRANSPORT C	DTHER											
Or	ptional Features & Functions:											
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chanı	UNC1X	CCOEF		184.6	23.78	1.96	0.76	29.33	3.93		
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chann	UNC1X	CCOSF		184.6	23.78	1.96	0.76	29.33	3.93		
8XX ACCESS 1	FEN DIGIT SCREENING											
	8XX Access Ten Digit Screening, Per Ca	OHD		0.0005321								
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv	OHD	N8R1X		8.46	0.96			25.52	25.52	16.05	16.05
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation	OHD			17.04	1.93	11.32	0.96	25.52	25.52	16.05	16.05
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation	OHD	N8FTX		17.04	1.93	11.32	0.96	25.52	25.52	16.05	16.05
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb	OHD	N8FCX		5.63	2.81	711.02	0.00	25.52	25.52	16.05	16.05
	8XX Access Ten Digit Screening, Gustoffized Area of Service Fer 6XX North  8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested	OHD	INOI CX		3.03	2.01			23.32	20.02	10.03	10.03
		OUD	NOTAN		0.50	0.77			05.50	05.50	40.05	40.05
	Per 8XX No.	OHD	N8FMX		6.59	3.77			25.52	25.52	16.05	16.05
	8XX Access Ten Digit Screening, Change Charge Per Reque	OHD	N8FAX		9.42	0.96			25.52	25.52	16.05	16.05
	8XX Access Ten Digit Screening, Call Handling and Destination Featur	OHD	N8FDX		5.63	5.63			25.52	25.52	16.05	16.05
LINE INFORMA	ITION DATA BASE ACCESS (LIDB)		_									
LINE INFORMA	LIDB Common Transport Per Quer	OQT		0.0000446								
<del></del>	LIDB Validation Per Quer	OQU		0.0142132		1	<del>                                     </del>				1	
$\vdash$		OQT, OQU	NRPBX	0.0142132	62.62	1	-		25.52	25.52	16.05	16.05
+	LIDB Originating Point Code Establishment or Chanç	JQ1, JQU	INKERX		63.63				25.52	25.52	16.05	16.05
SIGNALING (CO	CS7)											
	CCS7 Signaling Termination, Per STP Por	1DB	PT8SX	161.12					25.52	25.52	16.05	16.05
	CCS7 Signaling Usage, Per TCAP Messag	1DB	1 100%	0.0001115					20.02	20.02	10.00	10.00
	CCS7 Signaling Connection, Per link (A link	1DB	TPP++	21.58	169.72	169.72	134.08	134.08	25.52	25.52	16.05	16.05
	CCS7 Signaling Connection, Per link (A link)  CCS7 Signaling Connection, Per link (B link) (also known as D lin	1DB	TPP++	21.58								
			IPP++		169.72	169.72	134.08	134.08	25.52	25.52	16.05	16.05
	CCS7 Signaling Usage, Per ISUP Messag	1DB 1DB	OTUEO	0.0000456					25.52	25.52	40.05	40.05
	CCS7 Signaling Usage Surrogate, per link per LAT	IDB	STU56	406.53					25.52	25.52	16.05	16.05
	CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per											
	STP affected	1DB	CCAPO		40	40			25.52	25.52	16.05	16.05
	CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per Stp Affected	1DB	CCAPD		8	8			25.52	25.52	16.05	16.05
	Stp Affected	IDB	CCAPD		0	0			25.52	25.52	10.05	10.05
E911 SERVICE												
CALLING NAME	E (CNAM) SERVICE											
CALLING NAME	CNAM for DB Owners, Per Quen	OQV		0.016								
	CNAM for Non DB Owners, Per Quer	OQV		0.016								
	CNAW for Noti DB Owners, Fer Quer	OQV		0.01								
	CNAM (Non-Databs Owner), NRC, applies when using the Character Based User											
	CNAM (Non-Databs Owner), NRC, applies when using the Character Based User Interface (CHUI)	OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
		OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
		OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
LNP QUERY SI	Interface (CHUI)	OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
LNP QUERY SI	Interface (CHUI)	OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
	Interface (CHUI)  ERVICE	OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
	Interface (CHUI)	OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
OF	Interface (CHUI)  ERVICE	OQV	CDDCH		595	595			25.52	25.52	16.05	16.05
OF	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING	OQV	CDDCH	1.2	595	595			25.52	25.52	16.05	16.05
OF	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID	OQV	CDDCH	1.2	595	595			25.52	25.52	16.05	16.05
OF	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC	OQV	CDDCH	1.24	595	595			25.52	25.52	16.05	16.05
OF	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID	OQV	CDDCH	1.24 0.2	595	595			25.52	25.52	16.05	16.05
OF	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC	OQV	CDDCH	1.24	595	595			25.52	25.52	16.05	16.05
OPERATOR CA	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID	OQV	CDDCH	1.24 0.2	595	595			25.52	25.52	16.05	16.05
OPERATOR CA	Interface (CHUI)  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Fully Automated, Per Call - Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC	OQV	CDDCH	1.24 0.2	595	595			25.52	25.52	16.05	16.05
OPERATOR CA	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Fully Automated, Per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  RATOR SERVICES	OQV	CDDCH	1.24 0.2 0.2	595	595			25.52	25.52	16.05	16.05
OPERATOR CA	Interface (CHUI)  ERVICE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  ATOR SERVICES  Inward Operator Services - Verification, Per Minut Inward Operator Services - Verification and Emergency Interrupt - Per Minu	OQV	CDDCH	1.24 0.2 0.2 1.15	595	595			25.52	25.52	16.05	16.05
OPERATOR CA	Interface (CHUI)  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Fully Automated, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  ATOR SERVICES  Inward Operator Services - Verification, Per Minut Inward Operator Services - Verification and Emergency Interrupt - Per Minut  PERATOR CALL PROCESSING	OQV		1.24 0.2 0.2 1.15 1.15								
OPERATOR CA	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  ATOR SERVICES  Inward Operator Services - Verification, Per Minul Inward Operator Services - Verification and Emergency Interrupt - Per Minu  PERATOR CALL PROCESSING  Recording of Custom Branded OA Announcement	OQV	CBAOS	1.24 0.2 0.2 1.15 1.15	7000	7000			19.99	19.99	16.05	16.05
OPERATOR CA	Interface (CHUI)  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Fully Automated, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  ATOR SERVICES  Inward Operator Services - Verification, Per Minul  Inward Operator Services - Verification and Emergency Interrupt - Per Minu	OQV		1.24 0.2 0.2 1.15 1.15								
OPERATOR CA	Interface (CHUI)  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  PERATOR SERVICES  Inward Operator Services - Verification, Per Minut Inward Operator Services - Verification and Emergency Interrupt - Per Minut PERATOR CALL PROCESSING  Recording of Custom Branded OA Announcement Loading of Custom Branded OA Announcement Loading of Custom Branded OA Announcement per shelf/NAV	OQV	CBAOS	1.24 0.2 0.2 1.15 1.15	7000	7000			19.99	19.99		
OPERATOR CA	Interface (CHUI)  ERVICE  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  RATOR SERVICES  Inward Operator Services - Verification, Per Minut Inward Operator Services - Verification and Emergency Interrupt - Per Minu  PERATOR CALL PROCESSING  Recording of Custom Branded OA Announcement Loading of Custom Branded OA Announcement per shelf/NAV  SSISTANCE SERVICES	OQV	CBAOS	1.24 0.2 0.2 1.15 1.15	7000	7000			19.99	19.99		
OPERATOR CA	Interface (CHUI)  PERATOR SERVICES AND DIRECTORY ASSISTANCE  ALL PROCESSING  Oper. Call Processing - Oper. Provided, Per Min Using BST LID  Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using BST LID  Oper. Call Processing - Fully Automated, per Call - Using Foreign LIC  PERATOR SERVICES  Inward Operator Services - Verification, Per Minut Inward Operator Services - Verification and Emergency Interrupt - Per Minut PERATOR CALL PROCESSING  Recording of Custom Branded OA Announcement Loading of Custom Branded OA Announcement Loading of Custom Branded OA Announcement per shelf/NAV	OQV	CBAOS	1.24 0.2 0.2 1.15 1.15	7000	7000			19.99	19.99		

	DIRECTORY	ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)			1									
	DIRECTOR	Directory Assistance Call Completion Access Service (DACC), Per Call Attem	-+			0.1								
		(=====);												
	UNBRANDIN													
		/ TRANSPORT												
		Directory Transport - Local Channel DS				38.91	494.83	435.28	46.85	33.02	59.58	59.58	27.41	27.41
<del></del>		Directory Transport - DS1 Level Interoffice Per Mi			-	0.6598	100.00	147.31	20.50	21.61	25.52	25.52	11.34	44.04
		Directory Transport - DS1 Level Interoffice Per Facility Terminati Switched Common Transport Per DA Access Service Per Ca				74.4 0.0002997	196.28	147.31	26.56	21.61	25.52	25.52	11.34	11.34
		Switched Common Transport Per DA Access Service Per Call Per Mi				0.0002997								
		Access Tandem Switching Per DA Access Service Per Ca	-			0.0023713								
		Directory Transport - Installation NRC, Per Trunk or Signaling Connection				0.0020710	257.73	5.85			171.49	5.85		
		, , , , , , , , , , , , , , , , , , ,						0.00				0.00		
	DIRECTORY	ASSISTANCE DATA BASE SERVICE (DADS)												
		Directory Assistance Data Base Service Charge Per Listir				0.04								
		Directory Assistance Data Base Service, per mont			DBSOF	150								
BRANDING		Y ASSISTANCE												
		Custom Branding Announcement, per Recording to be used with the provision of DA		AMT	CBADA		3000	3000						
		Loading of Custom Branded Announcement per DRAM Card/Switch		AMT	CBADC		690	690						
SELECTIVE	E ROUTING													
		0.1			USRCR		227.99	007.00			43.52	0.00		
		Selective Routing Per Unique Line Class Code Per Request Per Swit	-		USRCR		227.99	227.99			43.52	9.99		
VIDTUAL C	COLLOCATION	ul												
TIK TOAL C	CLLOCATION	1	-+	ueanl,uea,udn,udc,	1									-
		Virtual Collocation - 2-wire Cross Connects (loop		al,uhl,ucl,uec	UEAC2	0.3996	30.93	29.59	12.76	11.43	19.99	19.99	19.99	19.99
		Virtual Collocation - 2-wire Cross Connects (Loop) for Line Splittin	$\top$	UEPSR, UEPSB			30.93	29.59	12.76	11.43	19.99	19.99	19.99	19.99
		Virtual Collocation - 2-wire Cross Connects (por		021 011, 021 02	VE1R2		30.93	29.59	12.76	11.43	19.99	19.99	19.99	19.99
		Virtual Collocation - 4-wire Cross Connects (loop		uea,uhl,ucl,ud	UEAC4		31.17	29.77	12.83	11.43	19.99	19.99	19.99	19.99
		Virtual Collocation - 4-wire Cross Connects (por			VE1R4		31.17	29.77	12.83	11.43	19.99	19.99	19.99	19.99
		Virtual Collocation - 2-Fiber Cross Connect		CLO	CNC2F		41.56	29.82	12.96	10.34	19.99	19.99	19.99	19.99
		Virtual Collocation - 4-Fiber Cross Connects		CLO	CNC4F		50.53	38.78	16.97	14.35	19.99	19.99	19.99	19.99
		Virtual Collocatin - DS1 Cross Connect		USL,ULC,CLO	CNC1X	7.5	155	14						
AIN OF 1 F.O.	TIVE CARRIE	D DOUTING												
AIN SELEC		Regional Service Establishment		SRC	SRCEC		391788				19.99	19.99	19.99	19.99
		End Office Establishment		SRC				220.52						
		Line/Port NRC, per end user		SRC	SRCEO		320.53 2.06	320.53 2.06			19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
		Query NRC, per end user		SRC	SKULP	0.000448	2.06	2.06			19.99	19.99	19.99	19.99
		Query NKC, per query		SKC		0.000446								
AIN - RELL	SOUTH AIN S	MS ACCESS SERVICE												
, 5222														
		AIN SMS Access Service - Service Establishment, Per State, Initial Setup			CAMSE		174.03	174.03	135.96	135.96	25.52	25.52	16.05	16.05
		AIN SMS Access Service - Port Connection - Dial/Shared Access			CAMDP		53.47	53.47	37.7	37.7	25.52	25.52	16.05	16.05
		AIN SMS Access Service - Port Connection - ISDN Access			CAM1P		53.47	53.47	37.7	37.7	25.52	25.52	16.05	16.05
		AIN SMS Access Service - User Identification Codes - Per User ID Code			CAMAU		129.83	129.83	79.91	79.91	25.52	25.52	16.05	16.05
		AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			CAMRC		131.54	131.54	45.77	45.77	25.52	25.52	16.05	16.05
		AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)				0.0029								
		AIN SMS Access Service - Session, Per Minute				0.097565								
		AIN SMS Access Service - Company Performed Session, Per Minute				2.09								
AIN1:	000000	OOL WIT OFFWOR	_	1								1		
AIN - BELLS	SOUTH AIN T	OOLKIT SERVICE		1	1				-	1		1	1	
			1		BAPSC		100.01	100.01	405.00	405.00	05.50	25.50	40.05	40.05
		AIN Toolkit Consider Consider Establishment Character Par Chatal Initial Catal	l l		DAPSC		169.31	169.31 8379	135.96	135.96	25.52 25.52	25.52 25.52	16.05 16.05	16.05 16.05
		AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			DADIV				1					16.05
		AIN Toolkit Service - Training Session, Per Customer	#		BAPVX		8379		27.7	27.7				
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt	$\pm$		BAPTT		39.3	39.3	37.7	37.7	25.52	25.52	16.05	16.05
		AIN Toolkit Service - Training Session, Per Customer							37.7 37.7	37.7 37.7				
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay			BAPTT BAPTD		39.3 39.3	39.3 39.3	37.7	37.7	25.52 25.52	25.52 25.52	16.05 16.05	16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate			BAPTD BAPTM		39.3 39.3 39.3	39.3 39.3 39.3	37.7 37.7	37.7 37.7	25.52 25.52 25.52	25.52 25.52 25.52	16.05 16.05	16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP			BAPTD BAPTM BAPTO		39.3 39.3 39.3 106.9	39.3 39.3 39.3 106.9	37.7 37.7 48.44	37.7 37.7 48.44	25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05	16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP			BAPTD BAPTM BAPTO BAPTC		39.3 39.3 39.3 106.9 106.9	39.3 39.3 39.3 106.9 106.9	37.7 37.7 48.44 48.44	37.7 37.7 48.44 48.44	25.52 25.52 25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05 16.05	16.05 16.05 16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code			BAPTD BAPTM BAPTO		39.3 39.3 39.3 106.9	39.3 39.3 39.3 106.9	37.7 37.7 48.44	37.7 37.7 48.44	25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05	16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Ouery			BAPTD BAPTM BAPTO BAPTC		39.3 39.3 39.3 106.9 106.9	39.3 39.3 39.3 106.9 106.9	37.7 37.7 48.44 48.44	37.7 37.7 48.44 48.44	25.52 25.52 25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05 16.05	16.05 16.05 16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Query AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per			BAPTD BAPTM BAPTO BAPTC	0.0256138	39.3 39.3 39.3 106.9 106.9	39.3 39.3 39.3 106.9 106.9	37.7 37.7 48.44 48.44	37.7 37.7 48.44 48.44	25.52 25.52 25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05 16.05	16.05 16.05 16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Ouery			BAPTD BAPTM BAPTO BAPTC		39.3 39.3 39.3 106.9 106.9	39.3 39.3 39.3 106.9 106.9	37.7 37.7 48.44 48.44	37.7 37.7 48.44 48.44	25.52 25.52 25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05 16.05	16.05 16.05 16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Ouery AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query			BAPTD BAPTM BAPTO BAPTC	0.0256138 0.0065161	39.3 39.3 39.3 106.9 106.9	39.3 39.3 39.3 106.9 106.9	37.7 37.7 48.44 48.44	37.7 37.7 48.44 48.44	25.52 25.52 25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05 16.05	16.05 16.05 16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Query AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per			BAPTD BAPTM BAPTO BAPTC	0.0256138	39.3 39.3 39.3 106.9 106.9	39.3 39.3 39.3 106.9 106.9	37.7 37.7 48.44 48.44	37.7 37.7 48.44 48.44	25.52 25.52 25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05 16.05	16.05 16.05 16.05 16.05 16.05
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code AIN Toolkit Service - Query Charge, Per Ouery AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query			BAPTD BAPTM BAPTO BAPTC	0.0256138 0.0065161 1.79	39.3 39.3 39.3 106.9 106.9	39.3 39.3 39.3 106.9 106.9	37.7 37.7 48.44 48.44	37.7 37.7 48.44 48.44	25.52 25.52 25.52 25.52 25.52 25.52	25.52 25.52 25.52 25.52 25.52 25.52	16.05 16.05 16.05 16.05 16.05	16.05 16.05 16.05 16.05 16.05

	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription				BAPDS	15.93	44.02	44.02	31.28	31.28			25.52	25.52	16.05	16.0
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription					0.0027018	47.21	47.21	01.20	01.20			25.52	25.52	16.05	16.0
OOUF/ADUF/CMI																
DOUF/ADUF/CMI	DS					<del>                                     </del>										
ACCESS DA	AILY USAGE FILE (ADUF)															
	ADUF: Message Processing, per messag					0.004										
	ADUF: Data Transmission (CONNECT:DIRECT), per messag					0.001										
ENHANCED	OPTIONAL DAILY USAGE FILE (EODUF)															
	EODUF: Message Processing, per message					0.004										
ODTIONAL P	DAIL V.LIGAGE EILE (ADUE)															
	DAILY USAGE FILE (ODUF)  ODUF: Recording, per message					0.0001179										
	ODUF: Message Processing, per message					0.0032089										
	ODUF: Message Processing, per Magnetic Tape provisions					54.62										
	ODUF: Data Transmission (CONNECT:DIRECT), per messag					0.0000354										
ED EXTENDED	LINK (EELs)															
	EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; M					; New Orleans	LA;									
	rlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates I states, EEL network elements shown below also apply to currently combined facilities					Δe le Charge a	annlies to curr	ently combined	facilities con	verted to LIN	s (Non-recur	ring rates d	o not apply )			
	eorgia, the EEL network elements apply to ordinarily combined network elements per the						applies to cult	chay combined	Tacilities con	Verted to Oil	23.(14011 10001	ring rates d	/ not apply./			
					J.,											
2-WIRE VOI	ICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (I	EL)	4	LINOVA	HEALA	40.05										-
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone		1	UNCVX	UEAL2	18.35										-
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone		2	UNCVX	UEAL2	24.33				<u> </u>						L
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone		3	UNCVX	UEAL2	34.77 44.77										-
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor		-	UNC1X	1L5XX	0.2293										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor			UNC1X	U1TF1	63										
	DS1 Channelization System Per Mont  Voice Grade COCI - DS1 To Ds0 Interface - Per Mont!			UNC1X UNCVX	MQ1 1D1VG	125.29 0.6988										
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport			UNCVX	TDTVG	0.6988										
	Combination - Zone		1	UNCVX	UEAL2	18.35										
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	04.00										
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport		- 2	UNCVA	UEALZ	24.33										
	Combination - Zone (		3	UNCVX	UEAL2	34.77										
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport															
	Combination - Zone  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon		4	UNCVX	UEAL2 1D1VG	44.77 0.6988										
	Voice Grade COCI - DST to DS0 Charmer System Combination - per mon			UNCVA	IDIVG	0.0966										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNC1X	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3
4-WIRE VOI	ICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (I	EL)														
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	22.38										
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -															
	Zone 2		2	UNCVX	UEAL4	29.67										
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	42.4										
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -		J	ONOVA	OLALA	72.7										
	Zone 4		4	UNCVX	UEAL4	55.96										
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor			UNC1X UNC1X	1L5XX U1TF1	0.2293 63										
	Channelization - Channel System DS1 to DS0 combination Per Mor			UNC1X	MQ1	125.29										_
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mon			UNCVX	1D1VG											
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	Ī	1	LINGVO	LIEA: 4	22.20				1		T				
	Combination - Zone  Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport		- '	UNCVX	UEAL4	22.38										-
	Combination - Zone :		2	UNCVX	UEAL4	29.67			<u> </u>							
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport			1010101		40 :										
	Combination - Zone ( Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport		3	UNCVX	UEAL4	42.4										<del>                                     </del>
	Combination - Zone		4	UNCVX	UEAL4	55.96										
						1										
1	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNC1X	UNCCC	1	11.17	11.17	14.29	14.29			31.26	31.26	3.91	3
	1	T/EEL\														<del>                                     </del>
	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR															+
4-WIRE 56 K	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	· (LLL)											1			
4-WIRE 56 K		(LLL)	1	UNCDX	UDL56	25.61										

Z F	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	48.51								
F														
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -													1
	Zone 4		4	UNCDX	UDL56									Д_
l l	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor			UNC1X	1L5XX	0.2293						+	+	+
1	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mo			UNC1X	U1TF1	63					31.26	31.26	3.91	:
C	Channelization - Channel System DS1 to DS0 combination Per Mor			UNC1X	MQ1	125.29								1
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)			UNCDX	1D1DD	1.49								
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		1		UDL56									
	Combination - Zone   Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		1	UNCDX	UDL56	25.61					31.26	31.26	3.91	
	Combination - Zone 2		2	UNCDX	UDL56	33.94					31.26	31.26	3.91	
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport													
	Combination - Zone :		3	UNCDX	UDL56	48.51					31.26	31.26	3.91	
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		4	LINODY	UDL56	04.00								
	Combination - Zone  OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-		4	UNCDX	UDLS6	64.02						+	+	+
	64kbs)			UNCDX	1D1DD	1.49								
												1		
N	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg	L		UNC1X	UNCCC		11.17	11.17	14.29	14.29	31.26	31.26	3.91	
	BPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR	T (EEL)											+	4
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	25.61								
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -			ONODA	00004	20.01		1				<u> </u>	+	+
Z	Zone 2		2	UNCDX	UDL64	33.94								
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -													
	Zone 3 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -		3	UNCDX	UDL64	48.51							-	-
	First 4-Wire 64kbps Digital Grade Loop in a DST interoffice Transport Combination -		4	UNCDX	UND64	64.02								
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor			UNC1X	1L5XX									+
	·													
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor			UNC1X	U1TF1	63								_
	Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-			UNC1X	MQ1	125.29							+	+
	64kbs)			UNCDX	1D1DD	1.49	0	0						
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport												1	
	Combination - Zone		1	UNCDX	UDL64	25.61								_
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		2	LINODY	LIBLAA	00.04								
	Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		2	UNCDX	UDL64	33.94						+	+	+
lo	Combination - Zone (		3	UNCDX	UDL64	48.51								
А	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport													T
	Combination - Zone		4	UNCDX	UDL64	64.02								_
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.49								
	OFFICES)			ONODA	10100	1.45						+		+
N.	Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			UNC1X	UNCCC		11.17	11.18	14.29	14.29	31.26	31.26	3.91	
														+
	DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (E 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	EL)	1	UNC1X	USLXX	50.99							-	+
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone		2	UNC1X	USLXX							+	<del>                                     </del>	+
4	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone		3	UNC1X	USLXX	96.58								I
4	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone		4	UNC1X	USLXX									+
<u>lr</u>	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		$\vdash$	UNC1X	1L5XX	0.2293			-			+	+	+
١,	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo			UNC1X	U1TF1	63								
														T
N	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNC1X	UNCCC		11.17	11.17	14.29	14.29	31.26	31.26	3.91	
4 WIDE DOLL	DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (E	ELV						-	-					+
	DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (E	EL)	1	UNC1X	USLXX	50.99		<del> </del>				+	<del></del>	+
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		2	UNC1X	USLXX							<u> </u>	$\vdash$	+
F	First DS1Loop in DS3 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	96.58								I
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		4	UNC1X	USLXX									╄
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS3 - Facility Termination per mor		$\vdash$	UNC3X UNC3X	1L5XX U1TF3	5.43 705.42						+	+	+
	DS3 to DS1 Channel System combination per mon			UNC3X	MQ3	207.87		<del>                                     </del>				<u> </u>	+	+
D	DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1		İ					İ		I
Α	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	50.99						4		┵
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	67.58								4
A	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone Additional DS1Loop in DS3 Interoffice Transport Combination - Zone		3	UNC1X UNC1X	USLXX	96.58 127.4						+	+	+
	DS3 Interface Unit (DS1 COCI) combination per montl		7	UNC1X	UC1D1							<u> </u>	$\vdash$	+
_														
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNC3X	UNCCC	1	11.17	11.17	14.29	14.29	31.26	31.26	3.91	

2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	1	UNCVX	UEAL2	18.35								
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	2	UNCVX	UEAL2									
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	3	UNCVX	UEAL2									
2 Wile VO 2000 used with 2 wire VO interesting Transport Combination 2016	J	ONOVA	OLALZ	04.77								
A.1.2 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	4	UNCVX	UEAL2	45.88								
Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor	- 4	UNCVX	1L5XX									
Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility		UNCVA	ILOAA	0.0112								-
Termination per month		UNCVX	U1TV2	24.75					31.26	31.26	3.91	
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCVX	UNCCC		11.17	11.17	14.29	14.29	31.26	31.26	3.91	
4-WIRE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (I												
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	1	UNCVX	UEAL4	22.38								
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	2	UNCVX	UEAL4	29.67								
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	3	UNCVX	UEAL4									
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	4	UNCVX	UEAL4									
Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor		UNCVX	1L5XX	0.0112								
Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility		9.1.9.1.1										
Termination per month		UNCVX	U1TV4	21.75								
Termination per monti		ONOVA	01114	21.70								1
Nonrecurring Currently Combined Network Elements Switch -As-Is Charc		UNCVX	UNCCC		11.17	11.17	14.29	14.29	31.26	31.26	3.91	
Nonrecurring Currently Combined Network Elements Switch -As-is Chart		UNCVA	UNCCC		11.17	11.17	14.29	14.29	31.20	31.20	3.91	-
DS3 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)												-
												_
High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor		UNC3X	1L5ND	14.16								
High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per												
month		UNC3X	UE3PX									
Interoffice Transport - Dedicated - DS3 - Per Mile per mon		UNC3X	1L5XX	5.43								
Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per more		UNC3X	U1TF3	705.42								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC3X	UNCCC		11.17	11.17	14.29	14.29	31.26	31.26	3.91	
							1					
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)												1
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor		UNCSX	1L5ND	14.16								+
High Capacity Unbundled Local Loop - STS1 combination - Fer Mile per mor		UNCSA	ILSIND	14.10								-
		LINICCV	LIDI C4	444.04								
month		UNCSX	UDLS1									-
Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor		UNCSX	1L5XX	5.43								
Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mo		UNCSX	U1TFS	707.97								-
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCSX	UNCCC		11.17	11.17	14.29	14.29	31.26	31.26	3.91	
2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)												
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	1	UNCNX	U1L2X	21.86								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	2	UNCNX	U1L2X									
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	3	UNCNX	U1L2X	41.4								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	4	UNCNX	U1L2X	54.64								
Interoffice Transport - Dedicated - DS1 combination - Per Mi	The state of the s	UNC1X	1L5XX									1
Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per mor		UNC1X	U1TF1	63								1
Channelization - Channel System DS1 to DS0 combination - per mor												+-
Channelization - Channel System DS I to DS0 combination - per mor		UNC1X	MQ1	125.29								-
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon		UNCNX	UC1CA	3.19			1					-
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	1	UNCNX	U1L2X	21.86			<b>_</b>					_
	1											
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	2	UNCNX	U1L2X	28.97								
	1											
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	3	UNCNX	U1L2X	41.4			<u> </u>			<u> </u>	1	Щ.
Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	4	UNCNX	U1L2X	54.64								
		UNCNX	UC1CA	3.19								
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per mon												
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per mon				.]	11.17	11.17	14.29	14.29	31.26	31.26	3.91	
, ,		UNC1X	UNCCC							020	0.0.	1
2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per mon  Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC1X	UNCCC									
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq	EEL)	UNC1X	UNCCC									
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (												
Nonrecurring Currently Combined Network Elements Switch -As-Is Charç  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	50.99								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1 2	UNC1X UNC1X	USLXX USLXX	50.99 67.58								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1 2 3	UNC1X UNC1X UNC1X	USLXX USLXX USLXX	50.99 67.58 96.58								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (  First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1 2	UNC1X UNC1X UNC1X UNC1X	USLXX USLXX USLXX USLXX	50.99 67.58 96.58 127.4								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor	1 2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNCSX	USLXX USLXX USLXX USLXX 1L5XX	50.99 67.58 96.58 127.4 5.43								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati	1 2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNCSX UNCSX	USLXX USLXX USLXX USLXX 1L5XX U1TFS	50.99 67.58 96.58 127.4 5.43 707.97								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System conbination per mon	1 2 3	UNC1X UNC1X UNC1X UNC1X UNCSX UNCSX UNCSX	USLXX USLXX USLXX USLXX 1L5XX U1TFS MQ3	50.99 67.58 96.58 127.4 5.43 707.97 207.87								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati	1 2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNCSX UNCSX	USLXX USLXX USLXX USLXX 1L5XX U1TFS	50.99 67.58 96.58 127.4 5.43 707.97 207.87								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System combination per mont DS3 Interface Unit (DS1 COCI) combination per mont	1 2 3	UNC1X UNC1X UNC1X UNC1X UNCSX UNCSX UNCSX	USLXX USLXX USLXX USLXX 1L5XX U1TFS MQ3	50.99 67.58 96.58 127.4 5.43 707.97 207.87 15.78								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System conbination per mon DS3 Interface Unit (DS1 COCI) combination per mont Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1 2 3 4	UNC1X UNC1X UNC1X UNC1X UNC5X UNCSX UNCSX UNCSX UNC1X UNC1X	USLXX USLXX USLXX USLXX 1L5XX 1L5XX U1TFS MQ3 UC1D1 USLXX	50.99 67.58 96.58 127.4 5.43 707.97 207.87 15.78 50.99								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System conbination per mon DS3 Interface Unit (DS1 COCI) combination per mont Additional DS1Loop in STS1 Interoffice Transport Combination - Zone Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1 2 3 4	UNC1X UNC1X UNC1X UNC1X UNC1X UNC5X UNCSX UNCSX UNCSX UNC1X UNC1X UNC1X	USLXX USLXX USLXX USLXX 1L5XX 1L5XX U1TFS MQ3 UC1D1 USLXX USLXX	50.99 67.58 96.58 127.4 5.43 707.97 207.87 15.78 50.99 67.58								
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT ( First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System conbination per mon DS3 Interface Unit (DS1 COCI) combination per mont Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1 2 3 4	UNC1X UNC1X UNC1X UNC1X UNC5X UNCSX UNCSX UNCSX UNC1X UNC1X	USLXX USLXX USLXX USLXX 1L5XX 1L5XX U1TFS MQ3 UC1D1 USLXX	50.99 67.58 96.58 127.4 5.43 707.97 207.87 15.78 50.99 67.58 96.58								

											T			
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCSX	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3.91
4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)	1													
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		1 UNCDX 2 UNCDX	UDL56 UDL56	25.61 33.94										
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone	1	3 UNCDX	UDL56	48.51							+			
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		4 UNCDX	UDL56	64.02										
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per M		UNCDX	1L5XX	0.0112							+			
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminati		UNCDX	U1TD5	17.24										
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCDX	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3.91
A WIDE ALKEDO DIOTAL EXTENDED LOOP WITH A KEDO DITEROFICE TO ANODODY (FEL)														
4-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)	-	1 UNCDX	LIDLCA	05.04										
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon∉ 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon∉		2 UNCDX	UDL64 UDL64	25.61 33.94							+			
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		3 UNCDX	UDL64	48.51							+			
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		4 UNCDX	UDL64	64.02							+			
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M		UNCDX	1L5XX	0.0112										
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminati		UNCDX	U1TD6	17.24										
Nonrecurring Currently Combined Network Elements Switch -As-Is Charq		UNCDX	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3.91
ADDITIONAL NETWORK ELEMENTS														
			1.											
When used as a part of a currently combined facility, the non-recurring charges do not apply,  When used as ordinarilty combined network elements in Georgia, the non-recurring charges a	put a Sw	the Switch As Is Charge	opiy.						-		+			
when used as ordinariity combined network elements in Georgia, the non-recurring charges a	ppiy and	the Switch As is Charge	does not.											
			<del>                                     </del>								+			
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ea	ch comb	ination)									1			
2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion											+			
Charge	T	UNCVX	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3.91
56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion	1													
Charge		UNCDX	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3.91
DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		UNC1X	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3.91
DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		UNC3X	UNCCC		11.17	11.17	14.29	14.29			31.26	31.26	3.91	3.91
STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge		UNCSX	UNCCC		11.17	11 17	14.29	14.29			31.26	31.26	3.91	3.91
Charge		UNCOA	UNCCC		11.17	11.17	14.29	14.29			31.20	31.20	3.91	3.91
NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month,	DS3 and	above=four months									+			
											+			
OPERATIONAL SUPPORT SYSTEMS														
NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the	state spe	ecific electronic service ord	dering char	rges as order	ed by the State	Commissions								
NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhi	bit is the	BellSouth regional electro	nic service	ordering ch	rge									
NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the			ges, or CLI	EC-1 may ele	ct the regional	electronic servi	ce ordering o	harge.						
NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per	LSR bas	S												
Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces														
(Regional)			SOMEC		3.5									
7		Live Brown Liber	7		L. L. L. L. D. L. L. L. L. L. L. L. L. L. L. L. L. L.		D		.1.0//					
The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to http://www.interconnection.bellsouth.com/become a clec/html/interconnection.htm	o Geogra	phically Deaveraged UNE	Zones. I	o view Geog	raphically Deave	raged UNE Zo	ne Designat	ons by Centi	al Office, re	fer to Intern	et Website:			
http://www.interconnection.beisouth.com/become_a_ciec/ntmi/interconnection.ntm														
UNBUNDLED LOCAL EXCHANGE SWITCHING(PORTS)	1		1 1			1	1					1		
STEELE EVOLUTION OF STATE OF S	-										+			
Exchange Ports	1										+			
NOTE: Although the Port Rate includes all available features in GA & TN, the desired features	will need	to be ordered using retai	I USOCs								1			
an aranapie reaction in error trig the desired features		uomg rotu									+			
2-WIRE VOICE GRADE LINE PORT RATES (RES)	1										1			
Exchange Ports - 2-Wire Analog Line Port- Re:		UEPSR	UEPRL	2.11	22.98	22.98	6.56	6.56			25.52	11.34	16.06	16.06
Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re	<u></u>	UEPSR	UEPRC	2.11	22.98	22.98	6.56	6.56			25.52	11.34	16.06	16.06
			1 7											
Exchange Ports - 2-Wire Analog Line Port outgoing only - Re	1	UEPSR	UEPRO	2.11	22.98	22.98	6.56	6.56			25.52	11.34	16.06	16.06
Exchange Ports - 2-Wire VG unbundled MS extended local dialing parity Port with Ca	a	UEDOS.												
ID - Res.		UEPSR	UEPAT	2.11	22.98	22.98	6.56	6.56			25.52	11.34	16.06	16.06
Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU	1	UEPSR	UEPAP	2.11	22.98	22.98	6.56	6.56			25.52	11.34	16.06	16.06
Subsequent Activity		UEPSR	USASC	0	0	0								
FEATURES Subsequent Activity	+	UEFOR	USASC	U	U	U			-		+			
All Available Vertical Feature	1	UEPSR	UEPVF	6.75	0	0					25.52	11.34	16.06	16.06
, and the state of		JEI GIK	J. VI	5.10	Ŭ	Ŭ					20.02			
2-WIRE VOICE GRADE LINE PORT RATES (BUS)	1										1			
					•									

Ex ID  Ex Ex ID  Ex Su  FEATURES  All  EXCHANGE PC	kchange Ports - 2-Wire Analog Line Port without Caller ID - Bi kchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484  - Bus.  kchange Ports - 2-Wire Analog Line Port outgoing only - Bu kchange Ports - 2-Wire VG unbundled MS extended local dialing parity Port with Ca	UEPSB UEPSB	UEPBC	2.11	22.98	22.98	6.56	6.56			1.34	16.06	十
Ex Ex ID Ex Su FEATURES All EXCHANGE PO	) - Bus:  xchange Ports - 2-Wire Analog Line Port outgoing only - Bu		UEPBC	2.11	22.98	22.98	6.56	6.56	25	- 50	4.04		
Ex Ex ID Ex ID Ex Su FEATURES All	xchange Ports - 2-Wire Analog Line Port outgoing only - Bu		02. 50	2:								16.06	ĺ
EX ID EX Su FEATURES All EXCHANGE PC								0.00	20	7.02	1.04	10.00	+
EX ID EX Su FEATURES All EXCHANGE PO		UEPSB	UEPBO	2.11	22.98	22.98	6.56	6.56	25	5.52	1.34	16.06	
Su Su All										-			T
Su FEATURES All	) - Bus.	UEPSB	UEPAY	2.11	22.98	22.98	6.56	6.56	25	5.52 1	1.34	16.06	
FEATURES All	xhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B	UEPSB	UEPB1	2.11	22.98	22.98	6.56	6.56	25	5.52 1	1.34	16.06	T
FEATURES All EXCHANGE PO										-			T
All EXCHANGE PC	ubsequent Activity	UEPSB	USASC	0	0	0							
EXCHANGE PO													
	I Available Vertical Feature	UEPSB	UEPVF	6.75	0	0			25	5.52 1°	1.34	16.06	T
Fx	ORT RATES (DID & PBX)												Τ
	xchange Ports - 2-Wire DID Port	UEPEX	UEPP2	9.43	238.29	37.43	122.66	7.71	25	5.52 1	1.34	16.06	Τ
													Τ
Ex	xchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabilii	UEPDD	UEPDD	72.96	403.5	191.12	148.66	5.04			9.99	19.99	
	xchange Ports - 2-Wire ISDN Port (See Notes below	UEPTX UEPSX		17.14	145.35	105.83	95.12	21.37	53	3.87 53	3.87	11.34	
	l Features Offered	UEPTX UEPSX			0	0							
	nission/usage charges associated with POTS circuit switched usage will also apply to circuit sw												
	s to B Channel or D Channel Packet capabilities will be available only through BFR/New Busine			ne packet capa	abilities will be	determined via	the Bona Fid	e Request/New Bu	siness Request Process.	<i>i.</i>			
	xchange Ports - 2-Wire ISDN Port Channel Profiles	UEPTX UEPSX		0	0	0							
Ex	xchange Ports - 4-Wire ISDN DS1 Por	UEPEX	UEPEX	105.79	407.08	202.84	162.15	41.07	51	1.03 5	1.03	8.51	
1													1
2-1	Wire VG Unbundled 2-Way PBX Trunk - Re:	UEPSE	UEPRD	2.11	22.98	22.98	6.56	6.56	25	5.52 1	1.34	16.06	1
													1
2-1	Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu	UEPSP	UEPPC	2.11	22.98	22.98	6.56	6.56	25	5.52 1	1.34	16.06	1
													1
	Wire VG Line Side Unbundled Outward PBX Trunk - Bu	UEPSP	UEPPO	2.11	22.98	22.98	6.56	6.56			1.34	16.06	1
2-1	Wire VG Line Side Unbundled Incoming PBX Trunk - Bu	UEPSP	UEPP1	2.11	22.98	22.98	6.56	6.56			1.34	16.06	1
	Wire Analog Long Distance Terminal PBX Trunk - Bu	UEPSP	UEPLD	2.11	22.98	22.98	6.56	6.56			1.34	16.06	4
	Wire Voice Unbundled PBX LD Terminal Port	UEPSP	UEPLD	2.11	22.98	22.98	6.56	6.56	25		1.34	16.06	+
	Wire Vice Unbundled 2-Way PBX Usage Po	UEPSP	UEPXA	2.11	22.98	22.98	6.56	6.56			1.34	16.06	+
2-1	Wire Voice Unbundled PBX Toll Terminal Hotel Por	UEPSP	UEPXB	2.11	22.98	22.98	6.56	6.56	25	5.52 11	1.34	16.06	+
	Was Valid to the English DDD Tarriage D	UEDOS	LIEDVO	0.44	00.00	00.00	0.50	0.50				40.00	1
2-1	Wire Voice Unbundled PBX LD DDD Terminals Po	UEPSP	UEPXC	2.11	22.98	22.98	6.56	6.56	25	5.52 1°	1.34	16.06	+
L.	Wise Vales Habita ded DDV I D Terminal Cuited have 15	LIEDOD	LIEDVE	244	22.22	00.00	0.50	0.50			4.24	40.00	1
	Wire Voice Unbundled PBX LD Terminal Switchboard Pc	UEPSP	UEPXD	2.11	22.98	22.98	6.56	6.56			1.34	16.06	+
	Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt	UEPSP	UEPXE	2.11	22.98	22.98	6.56	6.56	25	5.52 1	1.34	16.06	+
	Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling	LIEDOS	HEDV	2.44	22.00	22.00	0.50	0.50	-	5.50	4.24	40.00	1
Po	UIL	UEPSP	UEPXL	2.11	22.98	22.98	6.56	6.56	25	5.52 1	1.34	16.06	+
3 1	Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P	UEPSP	UEPXM	2.11	22.98	22.98	6.56	6.56	25	5.52	1.34	16.06	1
	Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling	OLF OF	JEF AIVI	2.11	22.30	22.30	0.00	3.30	25	3.02	1.04	10.00	+
Po		UEPSP	UEPXO	2.11	22.98	22.98	6.56	6.56	25	5.52	1.34	16.06	1
		JLF JF	OLF AU	4.11	22.30	22.30	0.00	0.00	25			10.00	+
ر د د	Wire Voice Unbundled 2-Way PBX Mississippi Local Economy Calling Pc	UEPSP	UEPXQ	2.11	22.98	22.98	6.56	6.56	25	5.52	1.34	16.06	1
	Wire Voice Unbundled 2-Way PBX Mississippi Local Optional Calling Po	UEPSP	UEPXR	2.11	22.98	22.98	6.56	6.56			1.34	16.06	t
2.1	Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	UEPSP	UEPXS	2.11	22.98	22.98	6.56	6.56	25	5.52	1.34	16.06	t
	The state of the s		32.7.0	2		22.00	0.00	5.00				. 0.00	t
Si	ubsequent Activity	UEPSP	USASC	0	0	0							1
FEATURES		32. 0.		,		Ť							$^{\dagger}$
	Il Available Vertical Feature	UEPSP UEPSE	UEPVF	6.75	0	0			25	5.52 1	1.34	16.06	$^{\dagger}$
	ORT RATES (COIN)				<del>-</del>								T
	xchange Ports - Coin Por			2.32	22.98	22.98	6.56	6.56	25	5.52 1	1.34	16.06	t
	* '' '												t
NOTE: Transm	nission/usage charges associated with POTS circuit switched usage will also apply to circuit sw	vitched voice and/or circ	uit switcher	d data transmi	ssion by B-Cha	nnels associate	d with 2-wire	ISDN ports.					T
	s to B Channel or D Channel Packet capabilities will be available only through BFR/New Busing								siness Request Process.	i.			T
	·												T
LOCAL SWIT	CHING, PORT USAGE												T
													Τ
End Office Swi	itching (Port Usage)												T
	nd Office Switching Function, Per MOI			0.0023771									T
Er	nd Office Trunk Port - Shared, Per MOL			0.0001927									T
													T
Tandem Switch	hing (Port Usage) (Local or Access Tandem)												T
	andem Switching Function Per MOI			0.0007834									T
	andem Trunk Port - Shared, Per MOl			0.0002834									T
													T
Common Trans	sport												T
	ommon Transport - Per Mile, Per MOl			0.0000091									T
	ommon Transport - Facilities Termination Per MO			0.0004281									T
	'												Ι
PORT/LOOP	COMBINATIONS - COST BASED RATES												⊥
													Τ
	ates are applied where BellSouth is required by FCC and/or State Commission rule to provide l	Unbundled Local Switch	ning or Swi	tch Ports.									Т

WIDE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)		_									
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)											
	oop Combination Rates											
	2-Wire VG Loop/Port Combo - Zone		1			16.71						
	2-Wire VG Loop/Port Combo - Zone		2			21.45						
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone ·	4	3			29.75 38.59						
			-			50.55						
UNE Loop F												
	2-Wire Voice Grade Loop (SL1) - Zone		2	UEPRX UEPRX	UEPLX	14.59 19.33						
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		3	UEPRX	UEPLX	27.63						+
	2-Wire Voice Grade Loop (SL1) - Zone		4	UEPRX	UEPLX	36.47						
0.140 17-1-	Out I I'm Part Part (Part)											
	e Grade Line Port Rates (Res)			UEPRX	UEPRL	2.12				43.52	9.99	
	2-Wire voice unbundled port - residenc			UEPRA	UEPKL	2.12				43.52	9.99	-+
	2-Wire voice unbundled port with Caller ID - re			UEPRX	UEPRC	2.12				43.52	9.99	
	O M/Control of the Lorent and the Control of the Co			HEDDY	LIEBBO	0.40				40.50	0.00	
	2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Mississippi extended local dialing parity port with Caller			UEPRX	UEPRO	2.12				43.52	9.99	-+
	ID - res			UEPRX	UEPAT	2.12				43.52	9.99	
	2-Wire voice unbundles res, low usage line port with Caller ID (LUI		1	UEPRX	UEPAP	2.12				43.52	9.99	
			+								<del>                                     </del>	
FEATURES												
	All Features Offered			UEPRX	UEPVF	6.75	0	0		43.52	9.99	
LOCAL NIII	MBER PORTABILITY		-									-
	Local Number Portability (1 per port			UEPRX	LNPCX	0.35						-
				0		0.00						
	RRING CHARGES (NRCs) - CURRENTLY COMBINED											
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		_	UEPRX	USAC2		5.2	0.41		43.52	9.99	-+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan			UEPRX	USACC		5.2	0.41		43.52	9.99	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa											
	Update						2.87			6.88		
ADDITIONA	L NRCs		_									+
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ			UEPRX	USAS2	0	0	0				
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)		_									
UNE Port/Lo	oop Combination Rates		-									
	2-Wire VG Loop/Port Combo - Zone	1	1			16.71						
	2-Wire VG Loop/Port Combo - Zone		2			21.45						
	2-Wire VG Loop/Port Combo - Zone	3	3			29.75						-
UNE Loop F	Rates		-									-+
•	2-Wire Voice Grade Loop (SL1) - Zone		1	UEPBX	UEPLX	14.59						
	2-Wire Voice Grade Loop (SL1) - Zone		2	UEPBX	UEPLX	19.33						
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		3 4	UEPBX UEPBX	UEPLX	27.63 36.47						-+
	2 TITLE VOICE STAUE LOUP (SET) - ZUITE	- 4	7	ULFBA	OEFLA	30.47						-+
	e Grade Line Port (Bus)											
	2-Wire voice unbundled port without Caller ID - bu		-	UEPBX	UEPBL	2.12				43.52	9.99	
	2-Wire voice unbundled port with Caller + E484 ID - bu			UEPBX	UEPBC	2.12				43.52	9.99	
	2 This fold distributed port with Outlot 1 E-TO-TID DC		$\neg$ †			4.14						
	2-Wire voice unbundled port outgoing only - bu		_	UEPBX	UEPBO	2.12				43.52	9.99	
	2-Wire voice Grade unbundled Mississippi extended local dialing parity port with Caller ID - bus			UEPBX	UEPAY	2.12				43.52	9.99	
	2-Wire voice unbundled incoming only port with Caller ID - Bı		-	UEPBX	UPEB1	2.12				43.52	9.99	-
	MBER PORTABILITY		_   _	LIESSY.	LNDO	0.00						
	Local Number Portability (1 per porl			UEPBX	LNPCX	0.35					<del>                                     </del>	-+
FEATURES			$^{+}$									
	All Features Offered			UEPBX	UEPVF	6.75	0	0		43.52	9.99	
NONDECLIA	DDING CHARGES (NDCs). CURRENTLY COMPINED		_									$-\!\!+\!\!$
NUNKECUK	RRING CHARGES (NRCs) - CURRENTLY COMBINED		- 1		1			1	1 1	1 1	1 1	1

	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa											
	Update					2.87			6.88			
ADDITIONA	I NDCo											
ADDITIONA	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX	USAS2					43.52	9.99		$\vdash$
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBA	U5A52					43.52	9.99		
-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)											-
L-WIIKE VOI	OL ORADE EGG! WITH E THILE ENE ! OR! (REG - 1 DA)											Н
INF Port/L	pop Combination Rates											
DIAL FOICE	2-Wire VG Loop/Port Combo - Zone	1			16.71							-
	2-Wire VG Loop/Port Combo - Zone	2			21.45							-
	2-Wire VG Loop/Port Combo - Zone	3			29.75							$\vdash$
	2-Wire VG Loop/Port Combo - Zone	4			38.59							$\vdash$
	2-Wile VG Loop/Fort Combo - Zone	4			36.39							$\vdash$
JNE Loop F	Pates											$\vdash$
DIAL LOOP I	2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPRG	UEPLX	14.59							-
	2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPRG	UEPLX	19.33							$\vdash$
	2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPRG	UEPLX	27.63							$\vdash$
	2-Wire Voice Grade Loop (SL 1) - Zone	4	UEPRG	UEPLX	36.47							$\vdash$
	2-Wire Voice Grade Loop (SL 1) - Zone	4	UEPRG	UEPLX	36.47							$\vdash$
2-Wire Voic	e Grade Line Port Rates (RES - PBX)											
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - R€		UEPRG	UEPRD	2.12				43.52	9.99		
LOCAL NUM	MBER PORTABILITY											
	Local Number Portability (1 per port		UEPRG	LNPCP	3.5							<u> </u>
												L
FEATURES												_
	All Features Offered		UEPRG	UEPVF	6.75	0	0		43.52	9.99		L
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED											-
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPRG	USAC2		5.2	0.41		43.52	9.99		
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with											
	Change		UEPRG	USACC		5.2	0.41		43.52	9.99		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa											П
	Update					2.87			6.88			
ADDITIONA	NRCs											-
ADDITIONA			LIEDDG	116763	0	0	0					
ADDITIONA	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USAS2	0	0	0		10.00	10.00	10.00	
ADDITIONA			UEPRG	USAS2	0	0 14.64	0 14.64		19.99	19.99	19.99	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi		UEPRG	USAS2	0				19.99	19.99	19.99	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USAS2	0				19.99	19.99	19.99	
2-WIRE VOI	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi		UEPRG	USAS2	0				19.99	19.99	19.99	
2-WIRE VOI	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	1	UEPRG	USAS2	16.71				19.99	19.99	19.99	
2-WIRE VOI	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  Dop Combination Rates 2-Wire VG Loop/Port Combo - Zone	1 2	UEPRG	USAS2	16.71				19.99	19.99	19.99	
2-WIRE VOI	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  oop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2	UEPRG	USAS2	16.71 21.45				19.99	19.99	19.99	
2-WIRE VOI	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  Dop Combination Rates 2-Wire VG Loop/Port Combo - Zone		UEPRG	USAS2	16.71				19.99	19.99	19.99	
2-WIRE VOI JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2	UEPRG	USAS2	16.71 21.45 29.75				19.99	19.99	19.99	
2-WIRE VOI JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2			16.71 21.45 29.75 38.59				19.99	19.99	19.99	
2-WIRE VOI JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2	UEPPX	USAS2	16.71 21.45 29.75				19.99	19.99	19.99	
2-WIRE VOI JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  Dop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Vice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone	2 3 4 1 2	UEPPX UEPPX	UEPLX UEPLX	16.71 21.45 29.75 38.59 14.59 19.33				19.99	19.99	19.99	
2-WIRE VOI JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates  - Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Loop (SL 1) - Zone	3 4	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX	16.71 21.45 29.75 38.59 14.59 19.33 27.63				19.99	19.99	19.99	
2-WIRE VOI JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	2 3 4 1 2	UEPPX UEPPX	UEPLX UEPLX	16.71 21.45 29.75 38.59 14.59 19.33				19.99	19.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX	16.71 21.45 29.75 38.59 14.59 19.33 27.63				19.99	19.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates  - Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Loop (SL 1) - Zone  2-Wire Voice Grade Loop (SL 1) - Zone	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX	16.71 21.45 29.75 38.59 14.59 19.33 27.63				19.99	19.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone 2-Wire Voice Grade Loop (St. 1) - Zone	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX	16.71 21.45 29.75 38.59 14.59 19.33 27.63				19.99	19.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grot  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 4-Grade Line Port Rates (BUS - PBX)  Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47				43.52	9.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPC	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47				43.52	9.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone 1- Wire Voice Grade Loop (SL 1) - Zone	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPPO UEPPO UEPPO UEPPO	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12				43.52 43.52 43.52	9.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  Dop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12				43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99	19.99	
JNE Loop F	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD UEPLD UEPAL	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12				43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99	19.99	
JNE Loop F	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  Dop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12				43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99	19.99	
JNE Loop F	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD UEPLD UEPAL	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12				43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  Dop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Unbundled PBX Trunk Port - Bu 1-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD DDD Terminals Po	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPC UEPPC UEPPA UEPLA UEPXB UEPXB	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12 2.12 2.12				43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99 9.99 9.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPA UEPPA UEPPA UEPX UEPX UEPX UEPX UEPX	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12 2.12 2.12 2.1				43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99 9.99 9.99	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Unbundled Combination 2-Way PBX Trunk Port - Bu 1-Line Side Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD DDD Terminals Po 2-Wire Voice Unbundled PBX LD DDD Terminals Po 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard DD Capable Pc	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPC UEPPC UEPPA UEPLA UEPXB UEPXB	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12 2.12 2.12 2.1				43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99 9.99 9.99	19.99	
JNE Loop F	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPA UEPPA UEPPA UEPXB UEPXB UEPXC	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12 2.12 2.12 2.1				43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99 9.99 9.99 9.99	19.99	
JNE Loop F	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Unbundled Combination 2-Way PBX Trunk Port - Bu 1-Line Side Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD DDD Terminals Po 2-Wire Voice Unbundled PBX LD DDD Terminals Po 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard DD Capable Pc	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPA UEPPA UEPPA UEPX UEPX UEPX UEPX UEPX	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12 2.12 2.12 2.1				43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99 9.99 9.99	19.99	
JNE Loop F	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Gro  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 1-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Grade Loop (SL 1) - Zone 2-Wire Voice Unbundled PBX Trunk Port - Bu Line Side Unbundled Outward PBX Trunk Port - Bu Line Side Unbundled Incoming PBX Trunk Port - Bu Line Side Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPC UEPPC UEPPC UEPX UEPXA UEPXA UEPXE UEPXE UEPXE UEPXE	16.71 21.45 29.75 38.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12 2.12 2.12 2.1				43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99 9.99 9.99 9.99 9.9	19.99	
JNE Port/Lo	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi  CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)  DOP Combination Rates	1 2 3 4 1 2 3	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPPA UEPPA UEPPA UEPXB UEPXB UEPXC	16.71 21.45 29.75 38.59 14.59 19.33 27.63 36.47 2.12 2.12 2.12 2.12 2.12 2.12 2.12 2.1				43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52 43.52	9.99 9.99 9.99 9.99 9.99 9.99 9.99	19.99	

	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Economy Calling Pc	UEPPX	UEPXQ					43.			
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Optional Calling Po	UEPPX	UEPXR					43.			
2	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	UEPPX	UEPXS	2.12				43.	2 9.99		4
LOCAL NILIM	BER PORTABILITY										+
LUCAL NUIVI	Local Number Portability (1 per port	UEPPX	LNPCP	3.15							+
	Local Number Portability (1 per port	UEPPA	LINPUP	3.15							+
FEATURES		-		1							+
	All Features Offered	UEPPX	UEPVF	6.75	0	0		43.	2 9.99		+
					-			1.0			T
NONRECURF	RING CHARGES (NRCs) - CURRENTLY COMBINED										
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As	UEPPX	USAC2	1	5.2	0.41		43.	2 9.99		_
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with	UEPPX	USACC		5.2	0.41		43.	2 9.99		
	Change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa	UEPPA	USACC	<del> </del>	5.2	0.41		43.	2 9.99		+
	Update				2.87				6.88		
		-									
ADDITIONAL											
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ	UEPPX	USAS2	0	0	0					
F	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi			1	14.64	14.64		19.	9 19.99	19.99	-
WIDE VOIC	CE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT										-
2-WIRE VOIC	SE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT			-							
INF Port/Loc	op Combination Rates	+	+							1	+
	2-Wire VG Coin Port/Loop Combo – Zone 1		+	17.06					_		+
	2-Wire VG Coin Port/Loop Combo – Zone 2	_	+	21.8					_		+
	2-Wire VG Coin Port/Loop Combo – Zone 3		_	30.1							
2	2-Wire VG Coin Port/Loop Combo – Zone 4			38.94							
UNE Loop Ra	ates					-					T
2	2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	14.59							
	2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX								
- 2	2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	27.63							_
2	2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	36.47							-
2-Wire Voice	Grade Line Ports (COIN)	-									+
	2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS)		_								+
		UEPCO	UEPRF	2.47				43.	2 9.99		
2	2-Wire Coin 2-Way without Operator Screening and without Blocking; with Dialing Parity										
	(Note 3) (MS)	UEPCO	UEPMC	2.47					43.52 9	99	
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL,										
	KY, LA, MS)	UEPCO	UEPRA	2.47				43.	2 9.99		
	2-Wire Coin 2-W with Operator Screening and Blocking: 011, 900/976, 1+DDD; with	UEPCO	UEPMA	2.47				43.	2 9.99		
	Dialing Parity (MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, MS)	UEPCO	UEPRB					43.			-
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (MS)	UEFCO	UEFRB	2.47				43.	2 9.99		+
ĺ	2 Wile Com 2 Way with Operator Corectning and CTT Blocking, with Blaining 1 artly (way	UEPCO	UEPMB	2.47				43.	2 9.99		
2	2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, &			2.17				10.	2 0.00		
L	Local (AL, KY, LA, MS)	UEPCO	UEPCD	2.47				43.	2 9.99		
	2-Wire Coin 2-W Operator Screening: 900 Block: 900/976, 1+DDD, 011+, Local; with	[ <del></del>					-				
	Dialing Parity (MS)	UEPCO	UEPCJ	2.47				43.	2 9.99		
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)	UEPCO	UEPRN	2.47				43.	2 9.99		-
	2-Wire Coin Outward without Blocking and without Operator Screening; With Dailing	LIEBOO	LIEDAGE	0.47							
	Parity (MS) 2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS)	UEPCO UEPCO	UEPME UEPRJ	2.47 2.47				43.		1	+
4	2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS)  2-Wire Coin Outward with Operator Screening and 011 Blocking; with Dialing Parity	UEFCU	UEFRJ	2.41				43.	2 3.99		+
	(MS)	UEPCO	UEPMD	2.47				43.	2 9.99		
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL,	32.00						40.	0.00		
ŀ	KY, LA, MS)	UEPCO	UEPRH	2.47				43.	2 9.99		
	2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, and Loc										
	(AL, KY, LA, MS)	UEPCO	UEPCN	2.47				43.	2 9.99		
2	2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD, 011+, and Local; with			Ι Τ	T				.		1
	Dialing Parity (MS)	UEPCO	UEPCS	2.47				43.	2 9.99		-
	2-Wire 2-Way Smartline with 900/976 (all states except LA) 2-Wire Coin Outward Smartline with 900/976 (all states except LA)	UEPCO UEPCO	UEPCK	2.47 2.47				43.			-
	2-Wife Coin Outward Smartline with 900/976 (all states except LA)  . UNE COIN PORT/LOOP (RC)	UEPCO	UEPCR	2.47				43.	2 9.99	-	+
- POINT IONAL	. ONE CONTINUOUS (NO)		+						_		+
1	UNE Coin Port/Loop Combo Usage (Flat Rate	UEPCO	URECU	4.62	0	0					
		32100	5.4200	52		<u> </u>					
OCAL NUM	BER PORTABILITY		1								T
	Local Number Portability (1 per port	UEPCO	LNPCX	0.35							
											Т
L											
FEATURES	All Features Offerec	UEPCO	UEPVF	6.75					25.52 11	34 16.06	

2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as			UE	PCO	USAC2		5.2	0.41				43.52	9.99		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan			UE	PCO	USACC		5.2	0.41				43.52	9.99		
ADDITIONAL NRCs															
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ			UE	PCO	USAS2		0	0				43.52	9.99		
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT															
UNE Port/Loop Combination Rates  2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		1				31.12									
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		2				39.6									
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		3				52.14									
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		4				63.91									
UNE Loop Rates															
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone		1		PPX	UECD1	21.71						19.99	19.99	19.99	19.99
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone		2		PPX	UECD1	30.19						19.99	19.99	19.99	19.99
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone		3		PPX PPX	UECD1	42.73 54.5	210.42	135.59	104.08	20.59		19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
2-vviile Alialog voice Grade Loop - (SL2) - Give Zonie		4	- OL	IIIA	OLCDI	34.3	210.42	100.00	104.00	20.55		13.33	19.99	13.33	13.33
UNE Port Rate															
Exchange Ports - 2-Wire DID Por			UE	PPX	UEPD1	9.41						19.99	19.99	19.99	19.99
NONRECURRING CHARGES - CURRENTLY COMBINED															
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-			UE	PPX	USAC1		14.59	3.72				19.99	19.99	19.99	19.99
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowab	le														
Changes			UE	PPX	USA1C		14.59	3.72				19.99	19.99	19.99	19.99
ADDITIONAL NRCs															
2-Wire DID Subsequent Activity - Add Trunks, Per Trun			UE	PPX	USAS1		53.49	53.49				19.99	19.99	19.99	19.99
Talanhana Number/Tumb Cosun Fateblisment Channe															
Telephone Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Porl			115	PPX	NDT	0	0	0				19.99	19.99	19.99	19.99
Additional DID Numbers for each Group of 20 DID Numbe				PPX	ND4	0	0	0				19.99	19.99	19.99	19.99
DID Numbers, Non- consecutive DID Numbers , Per Numbe				PPX	ND5	0	0	0			19.99				
Reserve Non-Consecutive DID number			UE	PPX PPX	ND6	0	0	0			19.99				
Reserve DID Numbers			UE	PPX	NDV	0	0	0			19.99				
LOCAL NUMBER PORTABILITY															
					LNPCP	3.15									
Local Number Portability (1 per port			UE	PPX	LINPUP	3.13									
Local Number Portability (1 per port  2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT			UE	PPX	LNPCP	3.15									
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT			UE	PPX	LNPCP	3.13									
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates			UE	PPB	LNPCP										
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT		1	UE		LNPCP	42.99									
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates		1 2	UE	PPB	LNPCP										
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone			UE UE UEPPB	PPB PPR UEPPR	LNPCP	42.99 53.29									
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		3	UE	PPB PPR	LNPCP	42.99 53.29 67.27									
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone			UE UE UEPPB	PPB PPR UEPPR	LNPCP	42.99 53.29									
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		3	UE UE UEPPB	PPB PPR UEPPR	LNPCP	42.99 53.29 67.27									
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates		3	UEPPB	PPB PPR UEPPR UEPPR		42.99 53.29 67.27 106.55						40.00		40.00	
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		3	UE UE UEPPB	PPB PPR UEPPR		42.99 53.29 67.27						19.99	19.99	19.99	19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates		3	UEPPB	PPB PPR UEPPR UEPPR	USL2X	42.99 53.29 67.27 106.55						19.99	19.99	19.99	19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		3	UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR	USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96						19.99	19.99	19.99	19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		1 2 3	UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94	202.54	450.74	44400	20.50		19.99	19.99 19.99	19.99	19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		1 2	UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96	233.54	158.71	104.88	20.59		19.99	19.99	19.99	19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  UNE Port Rate		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55	233.54	158.71	104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55	233.54	158.71	104.88	20.59		19.99	19.99 19.99	19.99	19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  UNE Port Rate  Exchange Port - 2-Wire ISDN Line Side Po		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55	233.54	158.71	104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE PORT/LOOP Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  UNE Port Rate  Exchange Port - 2-Wire ISDN Line Side Po  NONRECURRING CHARGES - CURRENTLY COMBINED		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55	233.54	158.71	104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  UNE Port Rate  Exchange Port - 2-Wire ISDN Line Side Po		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55	233.54	158.71	104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE PORT/LOOP Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  1-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55			104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE PORTILOOP Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  UNE Port Rate  Exchange Port - 2-Wire ISDN Line Side Po  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion  ADDITIONAL NRCS		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55			104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE PORT/LOOP Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  1-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55			104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE PORTILOOP Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  UNE Port Rate  Exchange Port - 2-Wire ISDN Line Side Po  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion  ADDITIONAL NRCS		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55			104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  UNE Port/Loop Combination Rates  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  UNE Loop Rates  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  UNE Port Rate  Exchange Port - 2-Wire ISDN Line Side Po  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion  ADDITIONAL NRCS		1 2 3	UEPPB UEPPB UEPPB UEPPB UEPPB UEPPB	PPB PPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR UEPPR	USL2X USL2X USL2X USL2X USL2X USL2X	42.99 53.29 67.27 106.55 28.66 38.96 52.94 106.55	76.91	42.99	104.88	20.59		19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99

Attachment 2 Exhibit C

CVCRCSD (MASSFSS)  (UPPS)  (UP																	T
Column	CVS/0	(CSD (DMS/5ESS)		UEPPB	UEPPR	U1UCA	0	0	0								
## MECHANISMS PROFILE CORES: (ALFYLAM SCMS, & TY)    WEIGHT ALFORD   WEIGHT AL				UEPPB	UEPPR		0	0	0								
UPPP   UPPP	CSD			UEPPB	UEPPR	U1UCC	0	0	0								
CONCEST DIASSESS    UEPPR   UEPPR   UEDPR																	
CYS (BV950)																	
UEPPS   UEPP	CVS/0	CSD (DMS/5ESS)		UEPPB	UEPPR	U1UCD	0	0	0								
UEPPS   UEPP																	
SERT TERNANL PROFILE	CVS (	(EWSD)		UEPPB	UEPPR	U1UCE	0	0	0								1
UEPPB   UEPP	000			LIEDDD	HEDDD	HALICE		•									
UEPPB   UEPP	CSD			UEPPB	UEPPR	UTUCF	0	U	0								+
### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT VERTICAL FEATURES  ### AT VERTICAL FEATURES  ### AT VERTICAL FEATUR	USER TERMINAL	PROFILE															T
### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT Vertical Features: One per Channel B User Profile  ### AT VERTICAL FEATURES  ### AT VERTICAL FEATURES  ### AT VERTICAL FEATUR																	T
M Vertical Features - One per Charmel B User Profile	User <sup>-</sup>	Terminal Profile (EWSD only)		UEPPB	UEPPR	U1UMA	0	0	0								
All Vertical Peatures - One per Charmel B User Profile  UEPPB UEPPR   UEPPF   UEPPF   0,75   0   0   0   0   0   0   19.99   1																	
INTEROFFICE CHANNEL MILEAGE Interoffice Channel mileage each, including first mile and facilities termination UEPPB UEPPB MIGNC 20.67 106.72 48.83	VERTICAL FEATU	JRES															+
### NETROPPICE CHANNEL MILEAGE   UEPPB UEP	All Ve	ertical Features - One per Channel B User Profile		LIEPPR	LIEPPR	LIEPVE	6.75	0	0								
Interoffice Channel mileage each, including first mile and facilities termination   UEPPB_UEPPR_MIGNE   UEPPPR_MIGNE   UEPPPR_MIG	7 70	enticum cutation one per chamilion b coor i formo		OLI I D	OL: III	02. 1.	0.70	Ů									+
Interoffice Channel mileage each, additional mile  UEPPB UEPPR MICHAN  0.0323 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	INTEROFFICE CH	IANNEL MILEAGE															T
Interoffice Channel mileage each, additional mile  UEPPB UEPPR MICHAM  0.0323 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0																	
#WIRE DST DIGITAL LOOP WITH 4-WIRE ISBN DST DIGITAL TRUNK PORT  UNE Porticipal Congress (SINN DST Digital Trunk Port - LINE Zone 1 UEPPP 318.6 1	Intero	office Channel mileage each, including first mile and facilities termination		UEPPB	UEPPR	M1GNC	20.67	106.72	48.83					19.99	19.99	19.99	1
#WIRE DST DIGITAL LOOP WITH 4-WIRE ISSN DST DIGITAL TRUMK PORT  WINE PORT DIGITAL COOP WITH 4-WIRE ISSN DST DIGITAL TRUMK PORT  WINE PORT DIGITAL COOP WITH 4-WIRE ISSN DST DIGITAL TRUMK PORT  WINE DST Digital Loop WIRE SAME STAT	Intere	office Channel mileage each, additional mile		HEDDE	HEDDE	M1GNM	0 0222	0	0								
INPERPORT   Combination Rates	intero	ornice Charmer mileage each, additional fille		JEFFB	JEFFR	IVI I GINIVI	0.0323	U	U			+	U				+
INPERPORT   Combination Rates	4-WIRE DS1 DIGIT	TAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT															+
AW DS1 Digital Loop/AW ISDN DS1 Digital Trunk Port - LNE Zone																	l
### DS1 Digital Loop/#W ISDN DS1 Digital Trunk Port - UNE Zone 2 UEPPP 318.5.  ### DS1 Digital Loop/#W ISDN DS1 Digital Trunk Port - UNE Zone 4 UEPPP 572.23   19.99	UNE Port/Loop Co	ombination Rates															
#W DS1 Digital Loop/#W ISON DS1 Digital Trunk Port - UNE Zone #W DS1 Digital Loop/#W ISON DS1 Digital Trunk Port - UNE Zone #W DS1 Digital Loop - UNE Zone																	
#W DS1 Digital Loop 4 Wison DS1 Digital Trunk Port - UNE Zone																	-
A-Wire DS1 Digital Loop - UNE Zone	4W D	OS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	3	UEI	PPP		356.97						40.00				-
4-Wire DST Digital Loop - UNE Zone   1   UEPPP   USLAP   107.5   119.99	4W D	JS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	4	UEI	PPP		672.23						19.99				╁
4-Wire DST Digital Loop - UNE Zone   1   UEPPP   USLAP   107.5   119.99	UNF Loon Rates																+
4-Wire DS1 Digital Loop - UNE Zone		ire DS1 Digital Loop - UNE Zone	1	UEF	PPP	USL4P	107.5							19.99	19.99	19.99	T
4-Wire DS1 Digital Loop - UNE Zone	4-Wir	ire DS1 Digital Loop - UNE Zone	2	UEF	PPP	USL4P	212.71							19.99	19.99	19.99	
4-Wire DS1 Digital Loop - UNE Zone 4 UEPPP USL4P 566.44 504.26 315.65 91.54 23.97 19.99 19	4-Wir	ire DS1 Digital Loop - UNE Zone	3	UEF	PPP	USL4P	251.18							19.99	19.99	19.99	
Exchange Ports - 4-Wire ISDN DS1 Por   UEPPP   UEPPP   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79   UEPPP   U5.79	4-Wir	ire DS1 Digital Loop - UNE Zone	4	UEF	PPP	USL4P	566.44	504.26	315.65	91.54	23.97			19.99	19.99	19.99	
Exchange Ports - 4-Wire ISDN DS1 Por																	
NONRECURING CHARGES - CURRENTLY COMBINED																	-
4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion - Switch-as-it   UEPPP   USACP   0   237.82   156.9   19.99		B ( W															
4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion - Switch-as-is		ange Ports - 4-Wire ISDN DS1 Por		UEF	PPP	UEPPP	105.79							19.99	19.99	19.99	
UEPPP   USACP   0   237.82   156.9   19.99	Excha			UEF	PPP	UEPPP	105.79							19.99	19.99	19.99	
ADDITIONAL NRCS  4-Wire DS1 Loop/4-W ISDN Digit Trk Port - Subset Actvy- Inward/two way tel nos within Std Allowance  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All States except NC)  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowancx  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  4-Wire DS1 Loop / 4-Wire IS	Excha NONRECURRING	CHARGES - CURRENTLY COMBINED		UEF	PPP	UEPPP	105.79							19.99	19.99	19.99	
A-Wire DS1 Loop/4-Wire ISDN DS1 Digital Trunk Port - Subsequent Inward Tel Numbers (All States except NC)	Excha  NONRECURRING  4-Wii	CHARGES - CURRENTLY COMBINED ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -						237.82	156.9								
Std Allowance	Excha  NONRECURRING  4-Wii	CHARGES - CURRENTLY COMBINED ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -						237.82	156.9								
4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All States except NC)	NONRECURRING 4-Wir	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is						237.82	156.9								
States except NC    UEPPP   PR7TO   23.02   23.02   19.99	NONRECURRING 4-Wir Conve	CHARGES - CURRENTLY COMBINED  ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is		UEF	PPP	USACP			156.9					19.99	19.99	19.99	
A-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos   UEPPP   PR7ZT   46.05   46.05   19.99   19.	NONRECURRING 4-Win Convert ADDITIONAL NRC 4-Win Std Al	CHARGES - CURRENTLY COMBINED  ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  Cs  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within		UEF	PPP	USACP			156.9					19.99	19.99	19.99	
Above Std Allowance   UEPPP   PR7ZT   46.05   46.05   19.99	NONRECURRING 4-Win Conve	CHARGES - CURRENTLY COMBINED  ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  Cs  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within killowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All		UEF	PPP	USACP PR7TF		0.9788						19.99	19.99	19.99	
LOCAL NUMBER PORTABILITY	Excha  NONRECURRING  4-Wir  Conve  ADDITIONAL NRC  4-Wir  Std Al  4-Wir  States	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - tersion - Switch-as-is  Ss  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within fullowance re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All se except NC)		UEF	PPP	USACP PR7TF		0.9788						19.99	19.99	19.99	9
Local Number Portability (1 per port   UEPPP   LNPCN   1.75	NONRECURRING 4-Win Conve  ADDITIONAL NRC 4-Win Std Al 4-Wir States 4-Wir 4-Wir	CCHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC); re DS1 Loop / 4-Wire ISDN DS1 Digital Trik Port - Subsequent Inward Tel Nos		UEI	PPP PPP	USACP PR7TF PR7TO		0.9788	23.02					19.99 19.99	19.99 19.99	19.99	99
Local Number Portability (1 per port   UEPPP   LNPCN   1.75	NONRECURRING 4-Win Conve  ADDITIONAL NRC 4-Win Std Al 4-Wir States 4-Wir 4-Wir	CCHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC); re DS1 Loop / 4-Wire ISDN DS1 Digital Trik Port - Subsequent Inward Tel Nos		UEI	PPP PPP	USACP PR7TF PR7TO		0.9788	23.02					19.99 19.99	19.99 19.99	19.99	99
NTERFACE (Provisioning Only)	Exche  NONRECURRING  4-Wir  ADDITIONAL NRC  4-Wir  Std Al  4-Wir  States  4-Wir  Above	CHARGES - CURRENTLY COMBINED  ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  Cs  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within killowance re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC); re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance		UEI	PPP PPP	USACP PR7TF PR7TO		0.9788	23.02					19.99 19.99	19.99 19.99	19.99	99
Voice/Data	Excha  NONRECURRING  4-Wir Conve  ADDITIONAL NRC  4-Wir Std Al 4-Wir States 4-Wir Above	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - version - Switch-as-is  Se  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All se except NC)  re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  re Std Allowance  PORTABILITY		UEI UEI UEI	PPP PPP PPP	USACP PR7TF PR7TO PR7ZT	0	0.9788	23.02					19.99 19.99	19.99 19.99	19.99	99
Voice/Data   UEPPP   PR71V   0   0   0   0   0   0   0   0   0	Excha  NONRECURRING  4-Wir Conve  ADDITIONAL NRC  4-Wir Std Al 4-Wir States 4-Wir Above	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - version - Switch-as-is  Se  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All se except NC)  re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  re Std Allowance  PORTABILITY		UEI UEI UEI	PPP PPP PPP	USACP PR7TF PR7TO PR7ZT	0	0.9788	23.02					19.99 19.99	19.99 19.99	19.99	99
Digital Data   UEPPP   PR71D   0   0   0   0   0   0   0   0   0	Exche  NONRECURRING 4-Wir Conve  ADDITIONAL NRC 4-Wir Std Al 4-Wir States 4-Wir Above  LOCAL NUMBER I Local	CHARGES - CURRENTLY COMBINED  ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within killowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All se except NC)  re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  re Std Allowance  PORTABILITY  I Number Portability (1 per port		UEI UEI UEI	PPP PPP PPP	USACP PR7TF PR7TO PR7ZT	0	0.9788	23.02					19.99 19.99	19.99 19.99	19.99	99
Inward Data	NONRECURRING 4-Wir Conve ADDITIONAL NRC 4-Wir Std Al 4-Wir States 4-Wir Above LOCAL NUMBER I Local INTERFACE (Prov	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - ression - Switch-as-is  Se  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC)  re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos  re Std Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)		UEF UEF UEF	PPP PPP	USACP PR7TF PR7TO PR7ZT	1.75	0.9788 23.02 46.05	23.02 46.05					19.99 19.99	19.99 19.99	19.99	99
New or Additional "B" Channel         UEPPP         PR7BV         0         29.01         19.99         19.99         19.99           New or Additional - Digital Data B Channel         UEPPP         PR7BF         0         29.01         19.99         19.99         19.99           New or Additional Inward Data B Channel         UEPPP         PR7BD         0         29.01         19.99         19.99         19.99	Exche  NONRECURRING  4-Wir  Conve  ADDITIONAL NRC  4-Wir  Std Al  4-Wir  States  4-Wir  LOCAL NUMBER I  LOCAL I  INTERFACE (Prov	CCHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  Fer DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Islowance  Fer DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is sexpet) NC;  Fer DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  FOS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  FORTABILITY  I Number Portability (1 per port  visioning Only)		UEF UEF UEF UEF	PPP PPP PPP	USACP PR7TF PR7TO PR7ZT LNPCN PR71V	1.75	0.9788 23.02 46.05	23.02 46.05					19.99 19.99	19.99 19.99	19.99	99
New or Additional - Voice/Data B Channel         UEPPP         PR7BV         0         29.01         19.99         19.99         19.99           New or Additional - Digital Data B Channel         UEPPP         PR7BF         0         29.01         19.99         19.99         19.99           New or Additional Inward Data B Channel         UEPPP         PR7BD         0         29.01         19.99         19.99         19.99	Excha  NONRECURRING 4-Win Convic ADDITIONAL NRC 4-Win Std Ai A-Wir States 4-Wir Above LOCAL NUMBER I Local INTERFACE (Prov Voice Digita	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Irersion - Switch-as-is  Seresion - Switch-a		UEI UEI UEI UEI UEI	PPP PPP PPP PPP PPP	USACP PR7TF PR7TO PR7ZT LNPCN PR71V PR71D	1.75	0.9788 23.02 46.05	23.02 46.05					19.99 19.99	19.99 19.99	19.99	99
New or Additional - Voice/Data B Channel         UEPPP         PR7BV         0         29.01         19.99         19.99         19.99           New or Additional - Digital Data B Channel         UEPPP         PR7BF         0         29.01         19.99         19.99         19.99           New or Additional Inward Data B Channel         UEPPP         PR7BD         0         29.01         19.99         19.99         19.99	Exche  NONRECURRING 4-Win Conve ADDITIONAL NRC 4-Win Std Ai 4-Wir States 4-Wir Above LOCAL NUMBER I Local INTERFACE (Prov Voice Digita	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Irersion - Switch-as-is  Seresion - Switch-a		UEI UEI UEI UEI UEI	PPP PPP PPP PPP PPP	USACP PR7TF PR7TO PR7ZT LNPCN PR71V PR71D	1.75	0.9788 23.02 46.05	23.02 46.05					19.99 19.99	19.99 19.99	19.99	99
New or Additional - Digital Data B Channel         UEPPP         PR7BF         0         29.01         19.99         19.99         19.99           New or Additional Inward Data B Channel         UEPPP         PR7BD         0         29.01         19.99         19.99         19.99         19.99	Exche  NONRECURRING  4-Wir  4-Wir  Stable  4-Wir  Stable  4-Wir  States  4-Wir  Above  LOCAL NUMBER I  Local  INTERFACE (Prov	CCHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  Te DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance  Te DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC);  Te DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY  Il Number Portability (1 per port  visioning Only)  //Data  al Data  rd Data		UEI UEI UEI UEI UEI	PPP PPP PPP PPP PPP	USACP PR7TF PR7TO PR7ZT LNPCN PR71V PR71D	1.75	0.9788 23.02 46.05	23.02 46.05					19.99 19.99	19.99 19.99	19.99	99
New or Additional Inward Data B Channel         UEPPP         PR7BD         0         29.01         19.99         19.99         19.99	ADDITIONAL NRC ADDITIONAL NRC 4-Wir Std Al 4-Wir States 4-Wir Above  LOCAL NUMBER I Local INTERFACE (Prov Voice, Digita Inwarn	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - resion - Switch-as-is  Se  Ire DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Islowance  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC)  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORT ABILITY  I Number Portability (1 per port  visioning Only)  //Data  I Data  Ir B" Channel		UEF UEF UEF UEF UEF UEF	PPP PPP PPP PPP PPP PPP PPP	USACP PR7TF PR7TO PR7ZT LNPCN PR71V PR71D PR71E	1.75 0 0 0	0.9788 23.02 46.05	23.02 46.05					19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	99
	NONRECURRING 4-Wir Conve ADDITIONAL NRC 4-Wir Std Al 4-Wir States 4-Wir Above LOCAL NUMBER I Local INTERFACE (Prov Voice Digita Inwar Inwar New or Additional	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Irersion - Switch-as-is  Sere DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Islowance  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is execcept NC)  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execcept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execcept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Truk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Truk Port - Subsequent Inward Tel Nos is execept NC.  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Truk Port - Subsequent Inward T		UEI UEI UEI UEI UEI UEI UEI	PPP PPP PPP PPP PPP PPP PPP PPP PPP PP	USACP PR7TF PR7TO PR7ZT LNPCN PR71V PR71D PR71E PR78V	1.75	0.9788 23.02 46.05 0 0 0 29.01	23.02 46.05					19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	99
	Exche  NONRECURRING  4-Wir  Convex  ADDITIONAL NRC  4-Wir  Std Al  4-Wir  States  4-Wir  Above  LOCAL NUMBER I  Local  INTERFACE (Prov  Voice  Digita  Inwara  New or Additional  New or	CCHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC); re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY Il Number Portability (1 per port  visioning Only) //Data al Data rd Data If "B" Channel or Additional - Voice/Data B Channel or Additional - Digital Data B Channel		UEI TF PR7TO PR7ZT LNPCN PR71U PR71D PR71E PR7BV	1.75 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	99		
	Excha  NONRECURRING  4-Wir Conve  ADDITIONAL NRC  4-Wir Std Al 4-Wir States  4-Wir Above  LOCAL NUMBER I Local  INTERFACE (Prov Voice, Digita Inwar  New or Additional New or New or	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - ression - Switch-as-is  28  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC)  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Subsequent Inward Tel Nos es Act Allowance  PORT ABILITY I Number Portability (1 per port  visioning Only)  //Data al Data rd Data  I "B" Channel or Additional - Voice/Data B Channel or Additional - Digital Data B Channel or Additional Inward Data B Channel		UEI TF PR7TO PR7ZT LNPCN PR71V PR71D PR71E PR7BV PR7BF PR7BF	1.75 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	99		
promor national decage content of piguar data de original de la content	NONRECURRING 4-Win Conve ADDITIONAL NRC 4-Win Std Ai 4-Win Std Ai A-Win States 4-Win Above LOCAL NUMBER I Local INTERFACE (Prov Voice Digita Inwan New or Additional New or New or New or New or New or New or	CCHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC); re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY Il Number Portability (1 per port  visioning Only) //Data al Data rd Data If "B" Channel or Additional - Voice/Data B Channel or Additional - Digital Data B Channel		UEI TF PR7TO PR7ZT LNPCN PR71U PR71D PR71E PR7BV	1.75 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	99		
19.99 19.99	NONRECURRING 4-Wir Conve ADDITIONAL NRC 4-Wir Std Ai 4-Wir Std Ai A-Wir States 4-Wir Above LOCAL NUMBER I Local INTERFACE (Prov Voice Digita Inwar New or Additional New or New or New or New or New or New or	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  Es  Fe DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is sexcept NC);  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)  p/Data  al Data rd Data rd Data rd Data  or Additional - Voice/Data B Channel or Additional I Neward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel		UEI TO PR7ZT LNPCN PR71U PR71D PR71E PR7BP PR7BD PR7BD	1.75 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	99		
CALL TYPES	Excha  NONRECURRING  4-Wir Conve  ADDITIONAL NRC  4-Wir Std Al 4-Wir States  4-Wir Above  LOCAL NUMBER I Local  INTERFACE (Prov Voice Digita Inwar  New or Additional New or New or New or New or New or New or	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  Es  Fe DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is sexcept NC);  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)  p/Data  al Data rd Data rd Data rd Data  or Additional - Voice/Data B Channel or Additional I Neward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel		UEI TO PR7ZT  LNPCN PR71D PR71D PR71E PR7BD PR7BS PR7BS PR7BS	1.75 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	99		
CALL TYPES	Excha  NONRECURRING  4-Wir  Converse of the second of the	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  PoS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is execcept NC)  Re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos exected to the sex of		UEI UEI UEI UEI UEI UEI UEI UEI UEI UEI	PPP PPP PPP PPP PPP PPP PPP PPP PPP PP	PR7TF PR7TO PR7ZT LNPCN PR71V PR71D PR71E PR7BV PR7BF PR7BU PR7BU	0 1.75 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	99
CALL TYPES	Excha  NONRECURRING  4-Wir Conve  ADDITIONAL NRC  4-Wirr Std Ai  4-Wirr States  1-Wirr Above  NORE  New or Additional  New or Ne	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - tersion - Switch-as-is  See DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is sexcept NC)  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  POS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)  **JOata  al Data  Ir B** Channel  or Additional - Voice/Data B Channel  or Additional - Digital Data B Channel  or Additional Inward Data B Channel  or Additional Useage Sensitive Voice Data B Channel  or Additional Useage Sensitive Digital Data B Channel		UEI TO PR7ZT LNPCN PR71V PR71D PR71E PR78V PR78F PR78BO PR78S PR7BU PR7C0	0 1.75 0 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	9		
	NONRECURRING 4-Wir 4-Wir Std Al 4-Wir Std Al 4-Wir States 4-Wir Above  LOCAL NUMBER I Local INTERFACE (Prov Voice Digita Inwar New or Additional New or New or	CCHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  CS  re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Allowance  re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC); re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY Il Number Portability (1 per port  visioning Only) //Data al Data rd Data If "B" Channel or Additional - Voice/Data B Channel or Additional - Digital Data B Channel		UEI TF PR7TO PR7ZT LNPCN PR71U PR71D PR71E PR7BV	1.75 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	9		
19.99 19.90 19.90	NONRECURRING 4-Wir Conve ADDITIONAL NRC 4-Wir Std Ai 4-Wir Std Ai A-Wir States 4-Wir Above LOCAL NUMBER I Local INTERFACE (Prov Voice Digita Inwar New or Additional New or New or New or New or New or New or	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  Es  Fe DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is sexcept NC);  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)  p/Data  al Data rd Data rd Data rd Data  or Additional - Voice/Data B Channel or Additional I Neward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel		UEI TO PR7ZT LNPCN PR71U PR71D PR71E PR7BP PR7BD PR7BD	1.75 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	9		
	Excha  NONRECURRING  4-Wir Conve  ADDITIONAL NRC  4-Wir Std Al 4-Wir States  4-Wir Above  LOCAL NUMBER I Local  INTERFACE (Prov Voice Digita Inwar  New or New or New or New or New or New or New or New or New or New or	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  Es  Fe DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is sexcept NC);  Fe DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)  p/Data  al Data rd Data rd Data rd Data  or Additional - Voice/Data B Channel or Additional I Neward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel or Additional Inward Data B Channel		UEI TO PR7ZT LNPCN PR71U PR71D PR71E PR7BP PR7BD PR7BD	1.75 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	99		
CALL TYPES	Excha  NONRECURRING  4-Wir  Converse of the second of the	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - rersion - Switch-as-is  PoS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is execcept NC)  Re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos exected to the sex of		UEI UEI UEI UEI UEI UEI UEI UEI UEI UEI	PPP PPP PPP PPP PPP PPP PPP PPP PPP PP	PR7TF PR7TO PR7ZT  LNPCN PR71D PR71D PR71E PR7BD PR7BS PR7BS PR7BS	0 1.75 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	99
CALL TYPES	NONRECURRING 4-Win Conve ADDITIONAL NRC 4-Win Std Ai 4-Win States 4-Win Above LOCAL NUMBER I Local INTERFACE (Prov Voice Digita Inware New or Additional New oc New	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - tersion - Switch-as-is  See DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is sexcept NC)  Ire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  POS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos re Std Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)  **JOata  al Data  Ir B** Channel  or Additional - Voice/Data B Channel  or Additional - Digital Data B Channel  or Additional Inward Data B Channel  or Additional Useage Sensitive Voice Data B Channel  or Additional Useage Sensitive Digital Data B Channel		UEI TO PR7ZT LNPCN PR71V PR71D PR71E PR7BV PR7BF PR7BU PR7BU	0 1.75 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	99		
ALL TYPE   Inward	Excha  NONRECURING  4-Wir Conve  ADDITIONAL NRC  4-Wirr State 4-Wirr States 4-Wirr Above  Local  NTERFACE (Prov Voice Digital Inwarn New or Additional New or New o	CHARGES - CURRENTLY COMBINED  Ire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - resion - Switch-as-is  28  Re DS1 Loop/4-W ISDN Digit Trk Port - Subsqt Actvy- Inward/two way tel nos within Illowance  Re DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All is except NC)  Re DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos es Extent Allowance  PORTABILITY  I Number Portability (1 per port  visioning Only)  20  20  20  21  28  18  29  Channel  or Additional - Voice/Data B Channel  or Additional - Digital Data B Channel  or Additional Inward Data B Channel  or Additional Inward Data B Channel  or Additional Inward Data B Channel  or Additional Inward Data B Channel  or Additional Useage Sensitive Voice Data B Channel  or Additional Useage Sensitive Digital Data B Channel		UEI TO PR7ZT LNPCN PR71V PR71D PR71E PR78V PR78F PR78BO PR78S PR7BU PR7C0	0 1.75 0 0 0 0 0 0	0.9788 23.02 46.05 0 0 0 29.01 29.01 29.01 29.01 29.01	23.02 46.05					19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	99		

	nnel Mileage ed Each Including First Mile	+	UEPPP	1LN1A	75.0598	196.28	147.31	26.56			19.99	19.99	19.99	+
	ch Airline-Fractional Additional Mil	+-+	UEPPP	1LN1A		130.20	147.31	20.00	<del>                                     </del>		13.33	13.33	13.33	+
Eat	OIT AIRING T TAGGORIA AGGIRIOTIAI IVIII	+	OLFFF	ILIVID	0.0330		<del> </del>							+
4-WIRE DS1 DIG	GITAL LOOP WITH 4-WIRE DDITS TRUNK PORT													Ī
LINE Port/Loop	Combination Rates	+-+												-
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	1	UEPDC		180.01						19.99	19.99	19.99	+
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	2	UEPDC		285.67						19.99	19.99	19.99	+
	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	3	UEPDC		324.14						19.99	19.99	19.99	+
4W	/ DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	4	UEPDC		639.4						19.99	19.99	19.99	╁
			02. 50		000.1						10.00	10.00	10.00	İ
UNE Loop Rate	s Wire DS1 Digital Loop - UNE Zone	1	UEPDC	USLDC	50.99						19.99	19.99	19.99	+
4-1	Wire DS1 Digital Loop - UNE Zone :	2	UEPDC	USLDC	50.99				23.97		19.99	19.99	19.99	+
4-1	Wire DS1 Digital Loop - UNE Zone	3	UEPDC	USLDC	251.18				25.51		19.99	19.99	19.99	+
	Wire DS1 Digital Loop - UNE Zone	4	UEPDC	USLDC	566.44	504.26	315.65	91.54	23.97		19.99	19.99	19.99	$\dagger$
LINE B B														
UNE Port Rate	Vire DDITS Digital Trunk Por	+	UEPDC	UDD1T	72.96						19.99	19.99	19.99	
			02. 50	055.11	72.00						10.00	10.00	10.00	
	IG CHARGES - CURRENTLY COMBINED Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-	+	UEPDC	USAC4		259.07	134.08				19.99	40.00	40.00	
	Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as: Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with	-	UEPDC	USAC4		259.07	134.08				19.99	19.99	19.99	
	1 Changes		UEPDC	USAWA		258.63	133.85				19.99	19.99	19.99	
	Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with													
Cha	ange - Trunk	+	UEPDC	USAWB		258.63	133.85				19.99	19.99	19.99	
ADDITIONAL NE														
	Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel		LIEDDO	LIDTTA		20.04	20.04				40.00	40.00	40.00	
	tivation/Chan - 2-Way Trunl  Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-		UEPDC	UDTTA		28.91	28.91				19.99	19.99	19.99	
Wa	ay Outward Trunk		UEPDC	UDTTB		28.91	28.91				19.99	19.99	19.99	
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan vard Trunk w/out DIC		UEPDC	UDTTC		28.91	28.91				19.99	19.99	19.99	
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -		UEPDC	UDITO		20.91	26.91				19.99	19.99	19.99	+
Inw	vard Trunk with DIC		UEPDC	UDTTD		28.91	28.91				19.99	19.99	19.99	
	Vire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way D w User Trans		UEPDC	UDTTE		28.91	28.91				19.99	19.99	19.99	
BIPOLAR 8 ZER	RO SUBSTITUTION		02. 50	05.12		20.01	20.01				10.00	10.00	10.00	
Be:	ZS -Superframe Format		UEPDC	CCOSF		0	600				19.99	19.99	19.99	
														T
B82	ZS - Extended Superframe Forma	+	UEPDC	CCOEF		0	600				19.99	19.99	19.99	-
Alternate Mark I	Inversion													
0.04	II -Superframe Format		UEPDC	MCOSF		0	0							
AW	ii -Superifame Format	+	UEFDC	WICOSE		U	0							t
AM	II - Extended SuperFrame Forms		UEPDC	МСОРО		0	0							
		_												+
	nber/Trunk Group Establisment Charges													
	ephone Number for 2-Way Trunk Grou	+	UEPDC	UDTGX	0		1						_	+
	ephone Number for 1-Way Outward Trunk Grou	+	UEPDC	UDTGY			+		<b>-</b>	19.99		1	0	+
	ephone Number for 1-Way Inward Trunk Group Without DI	+	UEPDC	UDTGZ ND4	0		1		<b> </b>	19.99 19.99				+
	D Numbers for each Group of 20 DID Number D Numbers, Non- consecutive DID Numbers , Per Numbe	+	UEPDC UEPDC	ND4 ND5	0		1		<del>                                     </del>	19.99		1		+
	serve Non-Consecutive DID Nos	+	UEPDC	ND6	0	0	0		+	19.99		1		+
	serve DID Numbers	+	UEPDC	NDV	0	0	0			19.99				t
	(Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS eroffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination	Trunk Po	UEPDC	1LNO1	74.4	196.28	147.31	26.56	21.61		19.99	19.99	19.99	+
	eroffice Channel Mileage - Additional rate per mile - 0-8 mil	+	UEPDC	1LNOA	0.6598	0	0	20.00	21.01		13.33	13.33	13.33	+
	eroffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination	+	UEPDC	1LNO2	0.0590	0	0							+
	eroffice Channel Mileage - Additional rate per mile - 9-25 mil	+	UEPDC	1LNOB	0.6598	0	0		<del>                                     </del>					+
	eroffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination	+	UEPDC	1LNO3	0.0050	0	0	0						+
Inte	eroffice Channel Mileage - Additional rate per mile - 25+ mil	1 +	UEPDC	1LNOC	0.6598	0	0	, ,						T
Inc	cal Number Portability, per DS0 Activate	+	UEPDC	LNPCP	3.15	0	0	0						+
Cer	ntral Office Termininating Poir	$\bot$	UEPDC	CTG	0		Ŭ	Ŭ						ļ
		+										<del>                                     </del>	1	+
	OOP WITH CHANNELIZATION WITH PORT													İ
Contam in 4 DCs	1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations	$\bot \bot \bot$												Ļ
				1	1		I	i .		1	1	1	1	1
	an have up to 24 combinations of rates depending on type and number of ports used	+												+

	Table Book and Table		LIEBLIO	luoi no luo	T-	1-	-					-	_
	4-Wire DS1 Loop - UNE Zone 1	1	UEPMG UEPMG	USLDC 107.05	0	0							
	4-Wire DS1 Loop - UNE Zone 2	2		USLDC 212.7	0	0							
	4-Wire DS1 Loop - UNE Zone 3	3	UEPMG	USLDC 251.18	0	0							
	4-Wire DS1 Loop - UNE Zone 4	4	UEPMG	USLDC 566.44	0	0				19.99	19.99	19.99	19.99
UNE DSO C	Channelization Capacities (D4 Channel Bank Configurations)				_	_							
	24 DSO Channel Capacity - 1 per DS1		UEPMG	VUM24 115.78	0	0							
	48 DSO Channel Capacity - 1 per 2 DS1s		UEPMG	VUM48 231.56	0	0							
	96 DSO Channel Capacity -1per 4 DS1s		UEPMG	VUM96 463.12	0	0							
	144 DS0 Channel Capacity - 1 per 6 DS1s		UEPMG	VUM14 694.68	0	0							
	192 DS0 Channel Capacity -1 per 8 DS1s		UEPMG	VUM19 926.24	0	0							
	240 DS0 Channel Capacity - 1 per 10 DS1s		UEPMG	VUM20 1157.8	0	0							
	288 DS0 Channel Capacity - 1 per 12 DS1s		UEPMG	VUM28 1389.36	0	0							
	384 DS0 Channel Capacity - 1 per 16 DS1s		UEPMG	VUM38 1852.48	0	0							
	480 DS0 Channel Capacity - 1 per 20 DS1s		UEPMG	VUM40 2315.6	0	0							
	576 DS0 Channel Capacity -1 per 24 DS1s		UEPMG	VUM57 2778.72	0	0							
	672 DS0 Channel Capacity - 1 per 28 DS1s		UEPMG	VUM67 3241.84	0	0							
Non-Recurr	ring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - Conv	ersion C	harge Based on a	System									
A Minimum	System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DSO Ports	with Fe	ature Activations.										
Multiples of	f this configuration functioning as one are considered Add'l after the minimum system co	nfigurat	ion is counted.										
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes	Ī	UEPMG	USAC4 0	300.55	16.7				19.99	19.99	19.99	19.99
System Add	ditions at End User Locations Where 4-Wire DS1 Loop with Channelization with Port Com	bination	Currently Exists	and									
	Currently Combined) In Georgia & Tennessee Only												
,	NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature Activation -												
	New GA & TN Only		UEPMG	VUMD4 0	715.15	327.39	148.05	17.56		19.99	19.99	19.99	19.99
Bipolar 8 Z	Zero Substitution												
	Clear Channel Capability Format, superframe - Subsequent Activity Only		UEPMG	CCOSF 0	0	600		1		19.99	19.99	19.99	19.99
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only		UEPMG	CCOEF 0	0	600		1		19.99	19.99	19.99	19.99
Alternate M	Mark Inversion (AMI)												
	Superframe Format		UEPMG	MCOSF 0	0	0		1					
	Extended Superframe Format		UEPMG	MCOPO 0	0	0							1
			1		T	T .		1					
Exchange F	Ports Associated with 4-Wire DS1 Loop with Channelization with Port												1
Exchange F													1
	10.0	_									_		1
	Line Side Combination Channelized PBX Trunk Port - Business		UEPPX	UEPCX 1.76	0	0	o	0	19.99				
1	The state of the s	_			T	-   -	-	T	10.00		_		1
	Line Side Outward Channelized PBX Trunk Port - Business		UEPPX	UEPOX 1.76	0	0	0	0	19.99				
	Line Side Inward Only Channelized PBX Trunk Port without DID	_	UEPPX	UEP1X 1.76	0	0	0	0	19.99		_		1
	and the state of t	_				-	Ü	<u> </u>	13.33		_		1
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port		UEPPX	UEPDM 9.43	0	0	0	0	19.99				
Feature Act	tivations - Unbundled Loop Concentration	_		22. 2 3.43		-	Ü	<u> </u>	13.33		_		1
. Julius Aci	Singulation Companies Loop Conference on the Con	_									_		1
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX	1PQWM 0.7	25.36	13.39	4.29	4.26		19.99	19.99	19.99	19.99
		_				. 5.55	20					.0.33	
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		UEPPX	1PQWU 0.7	78.03	18.39	60.66	11.85		19.99	19.99	19.99	19.99
Telephone	Number/ Group Establishment Charges for DID Service	_		4.7.5 0.7	. 0.00	10.00	00.00	. 1.00		. 5.55	10.00	10.00	10.00
. CICPITOTIE	DID Trunk Termination (1 per Port)	_	UEPPX	NDT 0		_	_	+		1	+	_	+
	DID TRUIK TOTHINGGOT (1 PCI FOIL)	-			0	0		1	19.99		+		+
	DID Numbers - groups of 20 - Valid all States			IND4 In		Į v		1	19.99	1	+	-	+
	DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number		UEPPX	ND4 0		Λ				10		_	+
	Non-Consecutive DID Numbers - per number		UEPPX	ND5 0	0	0			15.55				
	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers		UEPPX UEPPX	ND5 0 ND6 0	0	0			13.33				
	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers		UEPPX	ND5 0	0	0 0 0			13.33				
Local Numb	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers ber Portability		UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0	0	0 0 0			15.55				
	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers ber Portability Local Number Portability - 1 per port		UEPPX UEPPX	ND5 0 ND6 0	0	0 0 0			10.00				
FEATURES	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Beserve DID Numbers ber Portability Local Number Portability - 1 per port S- Vertical and Optional		UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0	0	0 0 0			10.00				
FEATURES	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Ber Portability Local Number Portability - 1 per port 5 - Vertical and Optional ching Features Offered with Line Side Ports Only		UEPPX UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0 LNPCP 3.15	0	0 0 0							
FEATURES	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Beserve DID Numbers ber Portability Local Number Portability - 1 per port S- Vertical and Optional		UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0	0	0 0 0			19.99				
FEATURES	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Ber Portability Local Number Portability - 1 per port 5 - Vertical and Optional ching Features Offered with Line Side Ports Only		UEPPX UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0 LNPCP 3.15	0	0 0 0							
FEATURES Local Switc	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Beserve DID Numbers ber Portability Local Number Portability - 1 per port 3 - Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available		UEPPX UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0 LNPCP 3.15	0	0 0 0							
FEATURES Local Switc	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Ber Portability Local Number Portability - 1 per port 5 - Vertical and Optional ching Features Offered with Line Side Ports Only		UEPPX UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0 LNPCP 3.15	0	0 0 0 0							
FEATURES Local Switc	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Beserve DID Numbers ber Portability Local Number Portability - 1 per port 3 - Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available		UEPPX UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0 LNPCP 3.15	0	0 0 0 0							
FEATURES Local Switch	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Reserve DID Numbers ber Portability Local Number Portability - 1 per port 5 - Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available  OP COMBINATIONS - MARKET RATES		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0 LNPCP 3.15	0	0 0 0 0							
FEATURES Local Swite D PORT LOC	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Beserve DID Numbers Ber Portability Local Number Portability - 1 per port - Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available  OP COMBINATIONS - MARKET RATES es shall apply where BellSouth is not required to provide unbundled local switching or switch	ports pe	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	ND5 0 ND6 0 NDV 0 LNPCP 3.15	0	0 0 0 0							
FEATURES Local Switch D PORT LOC Market Rate These scen	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Beserve DID Numbers ber Portability Local Number Portability - 1 per port S - Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available OP COMBINATIONS - MARKET RATES es shall apply where BellSouth is not required to provide unbundled local switching or switch varios include:		UEPPX UEPPX UEPPX UEPPX UEPPX  FFCC and/or State	ND5	0	0 0 0 0 0 0							
FEATURES Local Swite  D PORT Loc  Market Rate These scen 1. Unbundl	Non-Consecutive DID Numbers - per number	ept as n	UEPPX UEPPX UEPPX UEPPX UEPPX  FFCC and/or State	ND5	0 0 0	0							
FEATURES Local Swite  D PORT LOC  Market Rate These scen 1. Unbundl 2. Unbundl	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Reserve DID Numbers Local Number Portability Local Number Portability - 1 per port S- Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available OP COMBINATIONS - MARKET RATES  es shall apply where BellSouth is not required to provide unbundled local switching or switch larios include: led port/loop combinations that are Not Currently Combined in all of the BellSouth states excled port/loop combinations that are Currently Combined or Not Currently Combined in Zone 4	ept as n	UEPPX UEPPX UEPPX UEPPX UEPPX  UEPPX  FFCC and/or State loted for Georgia a Fog 8 MSAS in Bel	ND5 0 ND6 0 ND7 0 LNPCP 3.15  UEPVF 6.75  UEPVF 6.75  a Commission rules.  IISouth's region for end	0 0 0 0	0 0 0	uivalent lines.						
FEATURES Local Swite  D PORT LOC  Market Rate These scen 1. Unbundl 2. Unbundl	Non-Consecutive DID Numbers - per number	ept as n	UEPPX UEPPX UEPPX UEPPX UEPPX  UEPPX  FFCC and/or State loted for Georgia a Fog 8 MSAS in Bel	ND5 0 ND6 0 ND7 0 LNPCP 3.15  UEPVF 6.75  UEPVF 6.75  a Commission rules.  IISouth's region for end	0 0 0 0	0 0 0	uivalent lines.	vile).					
PORT LOCAL Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 M	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve DID Numbers ber Portability Local Number Portability - 1 per port 3 - Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available  OP COMBINATIONS - MARKET RATES  es shall apply where BellSouth is not required to provide unbundled local switching or switch harios include: led port/loop combinations that are Not Currently Combined in all of the BellSouth states excled port/loop combinations that are Currently Combined or Not Currently Combined in Zone 1 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (New of MSAs in BellSouth's regi	ept as n l of the Orleans)	UEPPX UEPPX UEPPX UEPPX UEPPX  IEPPX  UEPPX  Output  IFCC and/or State  oted for Georgia a fop 8 MSAS in Bel  is NC (Greensboro-	ND5	0 0 0 0 0 0	0 0 0 0 o o o o o o o o o o o o o o o o	Hill); TN (Nash		19.99				
PORT LOG  Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 M  BellSouth c	Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Reserve DID Numbers Local Number Portability Local Number Portability - 1 per port S- Vertical and Optional ching Features Offered with Line Side Ports Only All Features Available OP COMBINATIONS - MARKET RATES  es shall apply where BellSouth is not required to provide unbundled local switching or switch larios include: led port/loop combinations that are Not Currently Combined in all of the BellSouth states excled port/loop combinations that are Currently Combined or Not Currently Combined in Zone 4	ept as n l of the Orleans)	UEPPX UEPPX UEPPX UEPPX UEPPX  IEPPX  UEPPX  Output  IFCC and/or State  oted for Georgia a fop 8 MSAS in Bel  is NC (Greensboro-	ND5	0 0 0 0 0 0	0 0 0 0 o o o o o o o o o o o o o o o o	Hill); TN (Nash		19.99	t Rates and r	eserves the	right to true-up	the billing

Page 21 of 24

-WIRE VOIL										
,	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)									
UNE Port/Lc	pop Combination Rates									
	2-Wire VG Loop/Port Combo - Zone	1		2	8.59					
	2-Wire VG Loop/Port Combo - Zone	2		3	3.33					
	2-Wire VG Loop/Port Combo - Zone	3			1.63					
	2-Wire VG Loop/Port Combo - Zone	4		5	0.47					
UNE Loop R	Maria de la companya della companya									
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRX	UEPLX 1	4.59					
· ·	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRX		9.33					
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRX		7.63					
	2-Wire Voice Grade Loop (SL1) - Zone	4	UEPRX	UEPLX 3	6.47					
	e Grade Line Port (Res)									
	2-Wire voice unbundled port - residenc		UEPRX	UEPRL	14	90	90		43.52	9.99
	2-Wire voice unbundled port with Caller ID - re		UEPRX	UEPRC	14	90	90		43.52	9.99
	z-wire voice unbundled port with Caller ID - re		UEPKX	UEPRU	14	90	90		43.52	9.99
	2-Wire voice unbundled port outgoing only - re		UEPRX	UEPRO	14	90	90		43.52	9.99
	2-Wire voice unbundles res, low usage line port with Caller ID (LUI		UEPRX		14	90	90		43.52	9.99
	IBER PORTABILITY									
	Local Number Portability (1 per port		UEPRX	LNPCX (	0.35					
FEATURES										
	All Features Offered		UEPRX	UEPVF	0	0	0			
· ·	All Features Offeret		UEFKX	UEFVF	0	U	U			
ADDITIONAL	L NRCs									
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPRX	USAS2		0	0			
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)									
	pop Combination Rates 2-Wire VG Loop/Port Combo - Zone	1		2	28.59					
	2-Wire VG Loop/Port Combo - Zone	2			33.33					
	2-Wire VG Loop/Port Combo - Zone	3			1.63					
	2-Wire VG Loop/Port Combo - Zone	4		5	0.47					
UNE Loop R										
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX		4.59					
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX UEPBX		9.33					
	2-Wire Voice Grade Loop (SL1) - Zone	4	UEPBX		86.47					
	E Wile Voice Clade 2000 (CET) 20116		OL: DX	OE. EX						
2-Wire Voice	e Grade Line Port (Bus)									
	2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	14	90	90		43.52	9.99
	2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBX	UEPBC	14	90	90		43.52	9.99
	2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	14	90	90		43.52	9.99
	2-vviile voice unbunuled port outgoing only - bt		UEFBA	UEFBU	14	90	90		43.32	5.33
LOCAL NUN	MBER PORTABILITY									
	Local Number Portability (1 per port		UEPBX	LNPCX (	0.35					
FEATURES										
NOND-C:	DING CHARGES CHERENTLY COMPLYED									
NUNKECUR	RING CHARGES - CURRENTLY COMBINED									
ADDITIONAL	I NRCs									
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPBX	USAS2		0	0			
			OLI DX	00/102						
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)									
	pop Combination Rates									
	2-Wire VG Loop/Port Combo - Zone	1			8.59					
	2-Wire VG Loop/Port Combo - Zone	2			33.33					
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone ·	3 4		4	1.63				43.52	9.99
Į.	Z-YVIIE VO LOOP/FUIT COIIIDO - ZOIIE	4		5	10.47				43.52	9.99
UNE Loop R	Pates									

	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	3 4	UEPRG UEPRG	UEPLX	27.63 36.47									
	1, ,													
2-Wire Voic	ce Grade Line Port Rates (RES - PBX)													-
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	14	90	90				43.52	9.99		
LOCAL NU	MBER PORTABILITY		UEPRG	LNPCP	3.15									
	Local Number Portability (1 per port		UEPRG	LNPCP	3.15									
FEATURES														
NONDECHE	RRING CHARGES - CURRENTLY COMBINED													-
ADDITIONA														
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring					0	0							
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64				19.99	19.99	19.99	١.
2-WIRE VO	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)													
UNE Port/L	.oop Combination Rates													
	2-Wire VG Loop/Port Combo - Zone	1			28.59									
	2-Wire VG Loop/Port Combo - Zone	3			33.33 41.63									-
1	2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	4			50.47									+
UNE Loop F			HEDDY	LIEDLY	44.50									+
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	2	UEPPX	UEPLX	14.59 19.33				_	+ -				+
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPPX	UEPLX	27.63									ш
	2-Wire Voice Grade Loop (SL1) - Zone	4	UEPPX	UEPLX										
2-Wire Voic	ce Grade Line Port Rates (BUS - PBX)													-
2-Wile Voic	ce Grade Line Fort Nates (DOS -FDA)													
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı		UEPPX	UEPPC	14	90	90				43.52	9.99		
	Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	14	90	90				43.52	9.99		
	Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX	UEPP1	14	90	90				43.52	9.99		
	2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	14	90	90				43.52	9.99		
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc 2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX UEPPX	UEPXA UEPXB	14 14	90 90	90 90				43.52 43.52	9.99		+
	2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	14	90	90				43.52	9.99		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	14	90	90				43.52	9.99		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc		UEPPX	UEPXE		90	90				43.52	9.99		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		LIEDDY	LIEDVI	4.4		00				40.50	0.00		
	Port		UEPPX	UEPXL	14	90	90				43.52	9.99		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX	UEPXM	14	90	90				43.52	9.99		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port		UEPPX	UEPXO	14	90	90				43.52	9.99		
	I OIL		UEFFA	UEFAU	14	90	90				43.32	9.99		+
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Economy Calling Pc		UEPPX	UEPXQ	14	90	90				43.52	9.99		
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Optional Calling Po 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	+	UEPPX UEPPX	UEPXR UEPXS	14 14	90 90	90 90			1	43.52 43.52	9.99		+
1			UEFFA	UEFAS	14	90	90				43.32	3.33		
LOCAL NU	MBER PORTABILITY													
-	Local Number Portability (1 per port	+	UEPPX	LNPCP	3.15									1
FEATURES	s													+
NONDEC	DONNO CUADOFO, CUIDDENTI V COMPINED													
NUNRECUE	RRING CHARGES - CURRENTLY COMBINED	+++		1										
ADDITIONA														
	2-Wire Voice Grade Loop/ Line Port Combination - Subseque		UEPPX	USAS2		0	0							-
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring					0	0							
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64				19.99	19.99	19.99	
a WIDE YO		$\perp$												ļΞ
2-WIRE VO	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT													-
UNE Port/L	.oop Combination Rates													T
	2-Wire VG Coin Port/Loop Combo – Zone 1				28.59									
1	2-Wire VG Coin Port/Loop Combo – Zone 2				33.33									+
	2-Wire VG Coin Port/Loop Combo – Zone 3	1 1			41.63 50.47			1						1

JNE Loop Rates							
2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	14.59				
2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	19.33				
2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	27.63				
2-Wire Voice Grade Loop (SL1) - Zone	UEPCO	UEPLX	36.47			43.52	9.99
2-Wile Voice Grade Loop (SE1) - Zone	OLICO	OLILA	30.47			43.32	5.55
2-Wire Voice Grade Line Port Rates (Coin)							
2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA, MS	UEPCO	UEPRF	14	90	90	43.52	9.99
2-Wire Coin 2-Way without Operator Screening and without Blocking; with Dialing Parity (Note 3) (MS)	UEPCO	UEPMC	14	90	90	43.52	9.99
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS, SC)	UEPCO	UEPRA	14	90	90	43.52	9.99
2-Wire Coin 2-W with Operator Screening and Blocking: 011, 900/976, 1+DDD; with Dialing Parity (MS)	UEPCO	UEPMA	14			43.52	9.99
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, MS)	UEPCO	UEPRB	14	90	90	43.52	9.99
2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (M\$)	UEPCO	UEPMB	14	90	90	43.52	9.99
2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)	UEPCO	UEPCD	14	90	90	43.52	9.99
2-Wire Coin 2-W Oper Screen & Blocking: 900/976, 1+DDD, 011+, and Local; with Dialing Parity (MS)	UEPCO	UEPCJ	14	90	90	43.52	9.99
2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)	UEPCO	UEPRN	14	90	90	43.52	9.99
2-Wire Coin Outward without Blocking and without Operator Screening; with Dialing Parity (MS)	UEPCO	UEPME	14	90	90	43.52	9.99
2-Wire Coin Outward with Operator Screening and 011Blocking (GA, KY, MS)	UEPCO	UEPRJ	14	90	90	43.52	9.99
2-Wire Coin Outward with Operator Screening and 011 Blocking; with Dialing Parity (MS)	UEPCO	UEPMD	14	90	90	43.52	9.99
2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS)	UEPCO	UEPRH	14	90	90	43.52	9.99
2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)	UEPCO	UEPCN	14	90	90	43.52	9.99
2-Wire Coin Out Operator Screen & Blocking: 900/976, 1+DDD, 011+, & Local; with Dialing Parity (MS)	UEPCO	UEPCS	14	90	90	43.52	9.99
LOCAL NUMBER PORTABILITY		+ +					
Local Number Portability (1 per port	UEPCO	LNPCX	0.35				
NONRECURRING CHARGES - CURRENTLY COMBINED		+					
ADDITIONAL NRCs							
2-Wire Voice Grade Loop/ Line Port Combination - Subseque	UEPCO	USAS2		0	0	<del>                                     </del>	

Page 24 of 24 Version 2Q01: 08/30/01

NOTES	S UNBUNDLED NETWORK ELEMENT I	nterim	Zone	BCS	USOC		R	ATES (\$)					OSS RAT	TES (\$)		
											Svc Order Submitted Elec per LSR	LSR	vs. Electronic-1st	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	E
							Nonrecu	rring					nrecurring			
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	т-
						Rec	FIISL	Add I	First	Addi	SOWIEC	SOMAN	SUMAN	SUMAN	SOMAN	+
																t
	ne" shown in the sections for stand-alone loops or loops as part of a combination refers to Ge w.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	ographi	ically [	Deaveraged UNE Z	ones. To view (	Geographically De	eaveraged UNE	Zone Designa	ations by Cent	tral Office, re	fer to Internet	Website:		I		
D EVOLIA	ANGE ACCESS LOOP															+
EN EXCHA	ANGE ACCESS LOOP															+
2 WIDE /	ANALOG VOICE GRADE LOOP															+
Z-VVIRE A	2-Wire Analog Voice Grade Loop - Service Level 1- Statewi		SW	UEANL	UEAL2	15.88	57.99	42.37					26.94	12.76		+
			SW	UEPSR, UEPSB	UEALS	15.88		42.37					26.94			+
	2 Wire Analog Voice Grade Loop -Service Level 1-Statewide- Line Splitt					15.88	57.99	42.37					26.94	12.76		+
-	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zon		3	UEPSR, UEPSB	UEALS UEALS											+
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zon		3		UEALS		00.74	28.74								+
+	Engineering Information Document (E			UEANL UEANL	LIEAMO		28.74		1	<b> </b>	1	+	1	1		+
1	Manual Order Coordination for UVL-SL1s (per loop  Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR			UEANL	UEAMC OCOSL		61.38 45.34	61.38 45.34			1	+	1	1		+
1	Order Coordination for Specified Conversion Filtre for OVE-SET (per ESR			UEMINL	UCUSL		40.04	40.34			1	+	1	1		+
+	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling										1	1	1	1		+
				LIEA	LIEALO	10.5	142.07	100 EC					26.04	10.76		
1	Statewide Order Coordination for Specified Conversion Time (per LS		SW	UEA UEA	UEAL2 OCOSL	19.5	142.97 45.34	106.56	1		+	-	26.94	12.76		+
<del>                                     </del>				UEA	UCUSL		45.34				1	1	1	1		+
1	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling- Statewide		sw	UEA	UEAR2	19.5	142.97	106.56			1		26.94	12.76		
-	Order Coordination for Specified Conversion Time (per LS)		o₩	UEA	OCOSL	15.0	45.34	100.00	1		+	1	20.94	12.70		+
4 WIDE /	ANALOG VOICE GRADE LOOP			UEA	UCUSL		45.54									+
4-WIRE	4-Wire Analog Voice Grade Loop - Statewic		sw	LIEA	UEAL4	27.49	288.47	237.45					26.94	12.76		+
			SW			27.49		237.45					26.94	12.76		+
	Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL		45.34									+
0 140DE 1	INDIA DIGITAL ORANG LOOP															+
2-WIRE I	ISDN DIGITAL GRADE LOOP															_
	2-Wire ISDN Digital Grade Loop - Statewid		SW	UDN	U1L2X	24.98	325.91	251.31					26.94	12.76		4
	Order Coordination For Specified Conversion Time (per LSI			UDN	OCOSL		45.34									+
	11: 18: * 10: 1(180) 001B1TB15100B															-
2-WIRE (	Universal Digital Channel (UDC) COMPATIBLE LOOP															_
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Statewide		SW	UDC	UDC2X	24.98	325.91	251.31					26.94	12.76		_
2-WIRE	ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP															
	2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOF															
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -															
	Statewide		SW	UAL	UAL2X	14.6	504.9	456.17					26.94	12.76		
	Order Coordination for Specified Conversion Time (per LS			UAL	OCOSL		45.34									
	2 Wire Unbundled ADSL Loop without manual service inquiry and facility reservaton -															
	Statewide		SW	UAL	UAL2W	14.6	203.85	128.42					26.94	12.76		
	Order Coordination for Specified Conversion Time (per LS			UAL	OCOSL		45.34									
2-WIRE I	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP										1		1 -			L
											1					
	2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP				<u> </u>				<u> </u>	<u> </u>	1		1	<u> </u>		L
	2 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation		П								1		1			1
	Statewide		sw	UHL	UHL2X	11.98	504.9	456.17					26.94	12.76		┸
	Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		45.34				1		1 -			L
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -										1					
	Statewide		SW	UHL	UHL2W	11.98	221.08	145.65			1		26.94	12.76		
	Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		45.34									Ľ
																┰
4-WIRE I	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	T	T										1			⊥ ¯
	4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation										1					
	Statewide		SW	UHL	UHL4X	13.97	531.35	482.62					26.94	12.76		1
	Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		45.34									1
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -	Ţ	Ţ		Ι 7	T										1
	Statewide		sw	UHL	UHL4W	13.97	277.99	202.56					26.94	12.76		┸
	Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		45.34									Ľ
		I						·			1					L
4-WIRE	DS1 DIGITAL LOOP															Ľ
	4-Wire DS1 Digital Loop - Statewid		SW	USL	USLXX	62.78	714.84	421.47					42.19	12.76		Ι
1	Order Coordination for Specified Conversion Time (per LS		$\Box$	USL	OCOSL		48.31	-								П
																I
4-WIRE 1	19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP		$\Box$					-								П
	4 Wire Unbundled Digital 19.2 Kbps		SW	UDL	UDL19	32.67	489.04	337.51					19.99	19.99	19.99	9
	4 Wire Unbundled Digital Loop 56 Kbp		SW	UDL	UDL56	32.67	489.04	337.51					26.94	12.76		T
	Order Coordination for Specified Conversion Time (per LS			UDL	OCOSL		45.34									Т
								337.51	t	l	t	+	00.04			1
-	4 Wire Unbundled Digital Loop 64 Kbps - Statewic	Į.	SW	UDL	UDL64	32.67	489.04						26.94	12.76		

Page 1 of 18 Version 2Q01: 08/30/01

Description   Description							1			ı	1				
Comparison   Com															
Secretific Zines   1	2-WIRE Unb														
Committee   Comm				1101	LICI DD	40.4	204.05	400.05				40.00	40.00	40.00	19.99
International Control   1999			,	UCL	UCLPB	13.4	281.95	162.85				19.99	19.99	19.99	19.99
The content of the		reservation - Zone 2	2	UCL	UCLPB	21.76	281.95	162.85				19.99	19.99	19.99	19.99
Option Contribution to Memorial of Depart and Depart															
Service Unconsider Congress Congress and control and control in the control of			3			25.01						19.99	19.99	19.99	19.99
Security   Security				UCL	UCLMC		61.38	61.38							
Prince of the control Contro			1	UCL	UCLPW	13.4	250.17	174.74				19.99	19.99	19.99	19.99
2016   100		2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility													
International Content (Conte			2	UCL	UCLPW	21.76	250.17	174.74				19.99	19.99	19.99	19.99
Direct Coordination for Uniformitied Congret League See 1960   1975				LICI	LICI DW	25.01	250.17	174 74				10.00	10.00	10.00	19.99
2000   140,000		Order Coordination for Unbundled Copper Loops (per loop)	-		UCLMC	25.01						19.99	19.99	19.99	19.98
EVEN District Copyres (according for changes and according for changes and control of the cont		2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility													
Proceedings			1	UCL	UCL2L	37.79	268.96	149.86				19.99	19.99	19.99	19.99
2. Wice Debunded Copper LoopCuts products means so, relays and leasily   10.00   10.				LICI	LICI 2I	62.46	269.06	140.00				10.00	10.00	10.00	19.99
Instantion   The Second Continue   The Continue				UCL	UCLZL	63.16	268.96	149.86				19.99	19.99	19.99	19.98
Column		reservation - Zone (	3			73.02	268.96					19.99	19.99	19.99	19.99
Interviolation   Control Computer   Control Computer   Control Computer   Control Control Computer   Control				UCL	UCLMC		61.38	61.38							
SWeet Excluded Copper Loop Loop, without mentural services inquity and facility reservation. Journal of the Completion				1101	1101 014	07.70	400	440.57				40.00	40.00	40.00	40.00
Reservation - Zure   February   1999   199				UCL	UCLZW	31.19	189	113.57			1	19.99	19.99	19.99	19.99
Average   Aver			2	UCL	UCL2W	63.16	189	113.57			1	19.99	19.99	19.99	19.99
Other Coordination for Universided Copper Loops (per too)   February 1999   1		2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility													
27/06 United Condensions of Vitre Uniternated Copper Loop - Non-Designed (per loop - Non-Desig			3			73.02						19.99	19.99	19.99	19.99
Order Coordination Strike Unbrunded Copper Loops Processing (per for Experiment) (Control of Strike Copper Loops) (Control of Control of Strike Copper Loops) (Control of Control of Control of Control of Control of Control (Control of Contro		Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC		61.38	61.38			+			++	
Order Coordination Strike Unbrunded Copper Loops Processing (per for Experiment) (Control of Strike Copper Loops) (Control of Control of Strike Copper Loops) (Control of Control of Control of Control of Control of Control (Control of Contro															
Engineering Information Documer		2-Wire Unbundled Copper Loop Non-Designed - S\	I s			15.88						26.94	26.94	26.94	26.94
Lison Testing - Basic Additional Hid Hou.				UEQ	USBMC										
Logo   Testing - Resex - Additional Half Ho.   Logo   LIBETA   22.33   22.33		Engineering Information Documer			LIDET4										
### AVER COPPEL LOCP   AVER COPPEL LOCP   1-0-000   1-0-000   1-0-000   1-0-000   1-0-000   1-0-000   1-0-000   1-0-000   1-0-000   1-0-00															-
A-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1		2005 Footing Basis Fraultina Flor		024	OILE III		20.00	20.00							
A-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1															
Zone 1															
A-Wire Copper Loop/Short - including manual service inquiry and facility reservation - 2			1	UCI	LICL4S	17.63	330 13	211.02				19 99	19 99	19 99	19.99
4-Wire Copper Loops (per loc   2					00240							10.00		10.00	
Zone 3			2	. UCL	UCL4S	28.89	330.13	211.02				19.99	19.99	19.99	19.99
Order Coordination for Unbundled Copper Loops (per loc   VCL   VCLMC   61.38   61.38							202.42	044.00				40.00	40.00	40.00	40.04
### Copper Loop/Short - without manual service inquiry and facility reservation - Zohe   1			3			33.28						19.99	19.99	19.99	19.99
1   UCL   UCLW    17.83   250.17   174.74     19.99				OOL	OOLIVIO		01.50	01.00							
2   U.C.L   U.C.LW   28.89   250.17   174.74   19.99		1	1	UCL	UCL4W	17.63	250.17	174.74				19.99	19.99	19.99	19.99
4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone   3 UCL UCLMC   61.38   250.17   174.74   19.99		4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone					050.47					40.00	40.00	40.00	40.04
3 UCL UCLAW 33.28 250.17 174.74 19.00 19.99 19.9		2  4 Wire Copper Lean/Short - without manual convice inquiry and facility recordation - Zope	2	. UCL	UCL4W	28.89	250.17	174.74				19.99	19.99	19.99	19.99
Order Coordination for Unbundled Copper Loops (per loc   UCL   UCLMC   61.38   61.38		3	3	UCL	UCL4W	33.28	250.17	174.74				19.99	19.99	19.99	19.99
4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :   1		Order Coordination for Unbundled Copper Loops (per loc													
A-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1   1   UCL   UCLAL   90.07   317.14   198.03   19.99   19.		4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility				50.00	04744	400.00				40.00	40.00	40	46.51
reservation - Zone :   2   UCL   UCL4L   90.07   317.14   198.03   19.99   1			1	UCL	UCL4L	53.68	317.14	198.03		1		19.99	19.99	19.99	19.99
### Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 1  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 2  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 2  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 2  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 2  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 2  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservat - Zone 3  ##Wire Unbundled Loop Modification, Removal of Load Coils - Wire pair less than or equal to 10  ##Wire Unbundled Loop Modification, Removal of Load Coils - Wire less than or equal to 18  ##Wire Unbundled Loop Modification Removal of Load Coils - Wire pair greater than 18  ##Wire Unbundled Loop Modification Removal of Load Coils - Wire pair greater than 18  ##Wire Unbundled Loop Modification Removal of Load Coils - Wire pair greater than 18  ##Wire Unbundled Loop Modification Removal of Load Coils - Wire				UCL	UCL4L	90.07	317.14	198.03				19.99	19.99	19.99	19.99
Order Coordination for Unbundled Copper Loops (per loc  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reserval - Zone 1  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reserval - Zone 2  1 UCL UCL4O  53.68  237.18  161.75  19.99		4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility													
4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reserval			3			104.23						19.99	19.99	19.99	19.99
- Zone 1				UCL	UCLMC		61.38	61.38							
4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reserval 2   UCL   UCL40   90.07   237.18   161.75   19.99   19			1	UCL	UCL40	53.68	237.18	161.75			1	19.99	19.99	19.99	19.99
- Zone 2														15.55	15.5
- Zone 3		- Zone 2	2	UCL	UCL4O	90.07	237.18	161.75				19.99	19.99	19.99	19.99
OCH Coordination for Unbundled Copper Loops (per loc  UCL  UCLMC  61.38  61.38  61.38  61.38  OOP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18  Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18  UHL, UCL  ULM4L  64.85  64.85  64.85  64.85  ULM2G  339.84  339.84  339.84  339.84  339.84			1			40	00= :-	404 ==				46	45		
DOP MODIFICATION  Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18 ULL, ULL ULM2G 339.84 339.84  Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18 UHL, UCL ULM4L 64.85 64.85  Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18 UCL ULM4L 64.85 64.85  Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18 UCL ULM4L 339.84			3			104.23					+	19.99	19.99	19.99	19.9
Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft UEQ, ULS ULM2L 64.85 64.85		Order Socialitation for Oribundied Copper Loops (per loc		UUL	UCLIVIC		01.30	01.30			1				-
Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft UEQ, ULS ULM2L 64.85 64.85															
18k ft	OP MODIFICATION			1141 1000 1141							1				
Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18  UCL, ULS  ULM2G  339.84  339.84  339.84  Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18  UCL  ULM4L  64.85  64.85  339.84  339.84					LILMAN		64.05	64.05			1				
Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18  Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18  UCL ULM4G  UAL, UHL, UCL  ULM4G  339.84  339.84				UEQ, ULS							1			++	
Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18  UCL  ULM4G  339.84  339.84		Chibanalos 2009 Modification, Nemoval of 2000 Colls - 2 wife greater than 10		OOL, OLO	OLIVIZO		333.04	000.04							
UAL, UHL, UCL,		Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18		UHL, UCL	ULM4L		64.85	64.85							
UAL, UHL, UCL,				,											
		Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18	1		ULM4G		339.84	339.84			1			$\longmapsto$	
The state of the s		Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled I			ULMBT		64.9	64.9			1				
		2.25 modification from star of Enaged Tap Nomioval, per unburidied f		JE 4, JE1, JE3	CENTE		5 7.5	0 7.0							

		. '												
		_												
Sub-Loop D		<u> </u>	₩	LIE COL	11055		105	405				45 ==		-
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L	└──'	<u> </u>	UEANL	USBSA		498.09	498.09			26.94	12.76		
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-l	- 1		UEANL	USBSB		45.04	45.04			26.94	12.76		
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I			UEANL	USBSC		313.01	313.01			26.94	12.76		
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I	, , , ,		UEANL	USBSD		108.06	108.06			26.94	12.76		
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone	- 1	1	UEANL	USBN2	7.99	126.03	54.54	71.13	10.16	26.94	12.76	15.12	1
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		2	UEANL	USBN2	12.63	126.03	54.54	71.13	10.16	26.94	12.76	15.12	
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone		3	UEANL	USBN2	14.43	126.03	54.54	71.13	10.16	26.94	12.76	15.12	
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		۰	UEANL	USBMC	14.40	45.34	45.34	71.10	10.10	20.54	12.70	10.12	+
			1			0.00			70.50	40.50	20.01	40.70		+-
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone			UEANL	USBN4	9.23	156.52	79.66	78.56	13.53	26.94	12.76		_
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone		2	UEANL	USBN4	14.63	156.52	79.66	78.56	13.53	26.94	12.76		
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone	'	3	UEANL	USBN4	16.73	156.52	79.66	78.56	13.53	26.94	12.76		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	, '		UEANL	USBMC		45.34	45.34						
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC			UEANL	USBR2	3.5	114.05	37.2	76.58	10.81	26.94	12.76		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		45.34	45.34						
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC	-	+-	LIFANI	USBR4	3.75	127.67	50.82	78.71	10.69	26.94	12.76		+
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		-	UEANL		3.73			70.71	10.05	20.94	12.70		+
			<del>-</del>		USBMC		45.34	45.34						_
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone	'		UEF	UCS2X	7.33	137.1	60.24	76.58	10.81	26.94	12.76		
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone		2	UEF	UCS2X	10.95	137.1	60.24	76.58	10.81	26.94	12.76		
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone		3	UEF	UCS2X	12.36	137.1	60.24	76.58	10.81	26.94	12.76		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		45.34	45.34						
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone	-	1	UFF	UCS4X	7.14	162,24	85.38	78.56	13.53	26.94	12.76		+
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone			UEF	UCS4X	11.09	162.24	85.38	78.56	13.53	26.94	12.76	<b> </b>	+-
													<del>                                     </del>	+
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone		3	UEF	UCS4X	12.63	162.24	85.38	78.56	13.53	26.94	12.76		+
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	'	₩.	UEF	USBMC		45.34	45.34						1
		· '	<u> </u>											
Sub-Loop F	eeger		├	UEA.										-
		,		UDN,UCL,UDL,UD										
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-			C	USBFW		498.09							
		I		UEA,										
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u	I		UDN,UCL,UDL,UD C	USBFX		45.04	45.04						
	USL Feeder - DSU Set-up per Cross Box location - per 25 pair set-t		₩											+
	USL Feeder DS1 Set-up at DSX location, per DS1 termination		Щ.	USL	USBFZ		523.51	11.31						
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zon		1	UEA	USBFA	11.43	122.52	46.61	149.46	59.37	19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zon	, '	2	UEA	USBFA	18.35	122.52	46.61	149.46	59.37	19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zon		3	UEA	USBFA	21.04	122.52	46.61	149.46	59.37	19.99	19.99	19.99	
	Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		45.34							
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zon		1	UEA	USBFB	11.43	122.52	46.61	149.46	59.37	19.99	19.99	19.99	+
			_											+
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zon		2	UEA	USBFB	18.35	122.52	46.61	149.46	59.37	19.99	19.99	19.99	+
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zon		3	UEA	USBFB	21.04	122.52	46.61	149.46	59.37	19.99	19.99	19.99	
	Order Coordination for Specified Time Conversion, per LSR	, '		UEA	OCOSL		45.34							
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon		1	UEA	USBFC	11.43	122.52	46.61	149.46	59.37	19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon		2	UEA	USBFC	18.35	122.52	46.61	149.46	59.37	19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zo	ne												1
	3		3	UEA	USBFC	21.04	122.52	46.61	149.46	59.37	19.99	19.99	19.99	
	Order Coordination For Specified Conversion Time, per LS			UEA	OCOSL	21.04	45.34	40.01	143.40	00.01	15.55	10.00	10.00	+
			1			21.91		144.28			19.99	19.99	19.99	_
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zon			UEA	USBFD		226.36				19.99			
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zon			UEA	USBFD									_
		'	2			35.92	226.36	144.28			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zon		3	UEA	USBFD	35.92 41.37	226.36	144.28 144.28			19.99 19.99		19.99 19.99	ŀ
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per L\$					41.37						19.99		
	Order Coordination For Specified Conversion Time, Per LS			UEA UEA	USBFD OCOSL	41.37	226.36 45.34	144.28			19.99	19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon		3	UEA UEA UEA	USBFD OCOSL USBFE	41.37 21.91	226.36 45.34 226.36	144.28			19.99	19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon		1 2	UEA UEA UEA UEA	USBFD OCOSL USBFE USBFE	41.37 21.91 35.92	226.36 45.34 226.36 226.36	144.28 144.28 144.28			19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon		3	UEA UEA UEA UEA UEA	USBFD OCOSL USBFE USBFE USBFE	41.37 21.91	226.36 45.34 226.36 226.36 226.36	144.28			19.99	19.99 19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS		1 2	UEA UEA UEA UEA UEA UEA	USBFD OCOSL USBFE USBFE USBFE OCOSL	41.37 21.91 35.92 41.37	226.36 45.34 226.36 226.36 226.36 45.34	144.28 144.28 144.28 144.28			19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone		3 1 2 3	UEA UEA UEA UEA UEA UEA UEA UEA UEA UDN	USBFD OCOSL USBFE USBFE USBFE OCOSL USBFF	41.37 21.91 35.92 41.37	226.36 45.34 226.36 226.36 226.36 45.34 202.01	144.28 144.28 144.28 144.28 105.88			19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone		3 1 2 3 1 2	UEA UEA UEA UEA UEA UEA UEA UDN UDN	USBFD OCOSL USBFE USBFE USBFE OCOSL USBFF USBFF	21.91 35.92 41.37 19.63 31.61	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01	144.28 144.28 144.28 144.28 105.88			19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone		3 1 2 3	UEA UEA UEA UEA UEA UEA UEA UEA UEA UDN	USBFD OCOSL USBFE USBFE USBFE OCOSL USBFF	41.37 21.91 35.92 41.37	226.36 45.34 226.36 226.36 226.36 45.34 202.01	144.28 144.28 144.28 144.28 105.88			19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone		3 1 2 3 1 2	UEA UEA UEA UEA UEA UEA UDA UDN UDN	USBFD OCOSL USBFE USBFE USBFE OCOSL USBFF USBFF USBFF	21.91 35.92 41.37 19.63 31.61	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01	144.28 144.28 144.28 144.28 105.88			19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS		3 1 2 3 1 2 3	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN	USBFD OCOSL USBFE USBFE USBFE OCOSL USBFF USBFF USBFF USBFF	41.37 21.91 35.92 41.37 19.63 31.61 36.27	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01 45.34	144.28 144.28 144.28 144.28 105.88 105.88 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		3 1 2 3 1 2 3	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDN UDN	USBFD OCOSL USBFE USBFE USBFE OCOSL USBFF USBFF USBFF USBFF USBFF USBFF	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01 45.34 202.01	144.28 144.28 144.28 144.28 105.88 105.88 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		3 1 2 3 1 2 3 1 2 3	UEA	USBFD OCOSL USBFE USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFF	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 202.01	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		3 1 2 3 1 2 3 1 2 3	UEA	USBFD OCOSL USBFE USBFE USBFE USBFF USBFF OCOSL USBFF USBFF USBFF USBFF USBFS USBFS USBFS	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01 45.34 202.01 202.01 202.01 202.01	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl		3 1 2 3 1 2 3 1 2 3	UEA	USBFD OCOSL USBFE USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFF	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 202.01	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Cope, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatib) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatib) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatib) Unbundled Sub-Loop Feeder, 2-Wire UDC (IDSL compatib) Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		3 1 2 3 1 2 3 1 2 3	UEA	USBFD OCOSL USBFE USBFE USBFE USBFF USBFF OCOSL USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 202.01 393.01	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DST - Zone		3 1 2 3 1 2 3 1 2 3	UEA	USBFD OCOSL USBFE USBFE USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36	226.36 45.34 226.36 226.36 226.36 202.01 202.01 202.01 45.34 202.01 45.34 202.01 202.01 393.01 393.01	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 153.37 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone		3 1 2 3 1 2 3 1 2 3 1 2 3	UEA	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69	226.36 45.34 226.36 226.36 226.36 226.36 20.01 202.01 202.01 45.34 202.01 202.01 393.01 393.01	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Order Coordination For Specified Conversion Time, Per LS		3 1 2 3 1 2 3 1 1 2 3 1 2 3	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC UDC USL USL USL	USBFD OCOSL USBFE USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12	226.36 45.34 226.36 226.36 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 45.34	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 153.37 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19 42.19	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire DSI - Zone		3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3	UEA	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 202.01 393.01 393.01 393.01 45.34 172.89	144.28 144.28 144.28 144.29 105.88 105.88 105.88 105.88 105.88 105.88 105.87 153.37 153.37 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19 42.19 42.19	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Cope, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2-Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone		3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFH USBFH	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 10.66 10.66	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 202.01 393.01 393.01 393.01 45.34 172.89	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.83			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19 42.19 42.19 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL UCL	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG USBFG USBFH USBFH	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 45.34 172.89 172.89	144.28 144.28 144.28 144.29 105.88 105.88 105.88 105.88 105.88 105.88 105.87 153.37 153.37 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19 42.19 42.19	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 3 1 2 3 3 1 2 3 3 3 3	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL UCL	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG USBFH USBFH USBFH	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 19.63 78.161 10.66 10.66 10.66 10.66 10.66	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 393.01 393.01 393.01 393.01 45.34 172.89 172.89 172.89	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.83 153.37 153.37 159.81 159.81			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone		1 2 3 1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 1	UEA UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL UCL UCL UCL	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG USBFH USBFH USBFH USBFH USBFH	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 10.64 18.69	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 172.89 172.89 45.34 207.14	144.28 144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.83 105.83 105.83 105.83 105.81			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19 42.19 42.19 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone Unbundled Sub-Loop Feeder Loop - Zone		3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 3 1 2 3 3 1 2 3 3 3 3	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL UCL	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG USBFH USBFH USBFH	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 19.63 78.161 10.66 10.66 10.66 10.66 10.66	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 393.01 393.01 393.01 393.01 45.34 172.89 172.89 172.89	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.83 153.37 153.37 159.81 159.81			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone		1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 3	UEA	USBFD OCOSL USBFE USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG USBFG USBFG USBFG USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 16.44 18.69 14.68 23.74	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 45.34 172.89 172.89 45.34 207.14	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.88 105.88 105.87 153.37 153.37 153.37 153.37 153.37 153.37 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone		1 2 3 1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 1	UEA	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 10.64 18.69	226.36 45.34 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 45.34 172.89 172.89 172.89 172.89 172.89 207.14	144.28 144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.83 105.83 105.83 105.83 105.81			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 42.19 42.19 42.19 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Order Coordination For Specified Conversion Time, per LS		1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 2 3 3 3 1 1 1 2 3 3 3 1 1 1 2 3 3 3 1 1 1 1	UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL UCL UCL UCL UCL UCL UCL UCL UCL UCL UC	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFH USBFJ USBFJ USBFJ USBFJ USBFJ	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 16.44 18.69 14.68 23.74 27.26	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 393.01 172.89 45.34 207.14 207.14 207.14	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.87 153.37 153.37 153.37 153.37 153.37 153.37 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 12.76 12.76 12.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone		1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 1	UEA UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL UCL UCL UCL UCL UCL UCL UCL UCL UC	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF OCOSL USBFS USBFS USBFS USBFS USBFS USBFS USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 16.44 18.69 14.68 23.74 27.26 26.71	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 202.01 393.01 393.01 393.01 393.01 45.34 172.89 172.89 172.89 45.34 207.14 207.14 207.14 45.34	144.28 144.28 144.28 144.28 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 12.76 12.76 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Order Coordination For Specified Conversion Time, Per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Sup-Sub-Sup-Sup-Sub-Loop Feeder - Per 4-Wire Sup-Sub-Sup-Sup-Sup-Sub-Loop Feeder - Per 4-Wire Sup-Sub-Sub-Sup-Sup-Sub-Sup-Sup-Sub-Sup-Sup-Sub-Sup-Sup-Sub-Sub-Sup-Sup-Sub-Sup-Sub-Sub-Sup-Sup-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub		1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 1	UEA UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL UCL UCL UCL UCL UCL UCL UCL UCL UC	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFS USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 16.44 18.69 14.68 23.74 27.26	226.36 45.34 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 393.01 172.89 45.34 207.14 45.34 207.14	144.28 144.28 144.28 144.28 144.28 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 12.79 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Supper Loop - Zone Sub-Loop Feeder - Per 4-Wire Supper Loop - Zone Sub-Loop Feeder - Per 4-Wire Supper Loop - Zone Sub-Loop Feeder - Per 4-Wire Supper Loop - Zone Sub-Loop Feeder - Per 4-Wire Supper Loop - Zone		1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 3 1 1 2 3 3 1 1 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 3 1 1 1 1	UEA UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL UCL UCL UCL UCL UCL UCL UCL UCL UCL UC	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFG USBFG USBFG USBFG USBFG USBFH	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 19.63 31.61 36.27 19.63 19.69 67.36 78.12 10.66 14.68 23.74 27.26 26.71 44.07 50.83	226.36 45.34 226.36 226.36 226.36 45.34 202.01 202.01 45.34 202.01 202.01 202.01 393.01 393.01 393.01 393.01 172.89 172.89 172.89 172.89 172.89 172.89 172.89 172.89 173.89 174.91 207.14 207.1	144.28 144.28 144.28 144.28 105.88 105.88 105.88 105.88 105.88 105.88 105.88 105.88 105.88 105.88 105.88 105.81 105.87 153.37 153.37 153.37 153.37 153.37 153.37 153.37 153.37 153.37 153.37 153.37 153.37 153.37			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 12.76 12.76 12.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	
	Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone Order Coordination For Specified Conversion Time, Per LS Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 4-Wire DSI - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone Order Coordination For Specified Conversion Time, Per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Sup-Sub-Sup-Sup-Sub-Loop Feeder - Per 4-Wire Sup-Sub-Sup-Sup-Sup-Sub-Loop Feeder - Per 4-Wire Sup-Sub-Sub-Sup-Sup-Sub-Sup-Sup-Sub-Sup-Sup-Sub-Sup-Sup-Sub-Sub-Sup-Sup-Sub-Sup-Sub-Sub-Sup-Sup-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub		1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 1	UEA UEA UEA UEA UEA UEA UEA UEA UDN UDN UDN UDN UDN UDC UDC UDC USL USL USL UCL UCL UCL UCL UCL UCL UCL UCL UCL UC	USBFD OCOSL USBFE USBFE USBFF USBFF USBFF USBFF USBFF USBFF USBFF USBFS USBFS USBFS USBFS USBFS USBFS USBFG	41.37 21.91 35.92 41.37 19.63 31.61 36.27 19.63 31.61 36.27 39.69 67.36 78.12 10.66 16.44 18.69 14.68 23.74 27.26	226.36 45.34 226.36 226.36 45.34 202.01 202.01 202.01 202.01 202.01 202.01 202.01 393.01 393.01 393.01 393.01 172.89 45.34 207.14 45.34 207.14	144.28 144.28 144.28 144.28 144.28 105.88			19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 12.76 12.76 12.76 12.76 12.79 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99	

Page 3 of 18 Version 2Q01: 08/30/01

	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	3	UDL	USBFO	50.83	215	132.92				19.99	19.99	19.99	19.99
	Order Coordination For Specified Time Conversion, per LS	-	UDL	OCOSL		45.34								
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1	UDL	USBFP	26.71	215	132.92				19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	2	UDL	USBFP	44.07	215	132.92				19.99	19.99	19.99	19.99
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	3	UDL	USBFP	50.83	215	132.92				19.99	19.99	19.99	19.99
	Order Coordination For Specified Conversion Time, per LS		UDL	OCOSL		45.34								
Unbundled	Sub-Loop Modification													
	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W PR	'	UEF	ULM2X		353.95	12.2				26.94	12.76		
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR		UEF	ULM4X		353.95	12.2				26.94	12.76		
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR unloaded		UEF	ULM4T		557.78	14.23				26.94	12.76		
Unbundled	Network Terminating Wire (UNTW)													
Unbundled	Unbundled Network Terminating Wire (UNTW) per Pa		UENTW	UENPP	0.44	64.98	64.98				26.94	12.76		
	Onbunded Network Terminating Wife (ONTW) per Fa		OLIVIV	OLIVIT	0.44	04.50	04.50				20.54	12.70		
Network In	terface Device (NID)													
	Network Interface Device (NID) - 1-2 line	1	UENTW	UND12		86.37	56.69				26.94	12.76		
	Network Interface Device (NID) - 1-6 line	1	UENTW	UND16		127.93	98.21				26.94	12.76		
	Network Interface Device Cross Connect - 2 V	1	UENTW	UNDC2		11.68	11.68				26.94	12.76		
	Network Interface Device Cross Connect - 4V	I	UENTW	UNDC4		11.68	11.68				26.94	12.76		
UNBUNDLED LOOP CO											1	1		
	Unbundled Loop Concentration - System A (TR00)		ULC	UCT8A	398.41	652.26	652.26				19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System B (TR00)		ULC	UCT8B	58.36	271.78	271.78				19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System A (TR30:		ULC	UCT3A	439.73	652.25	652.26				19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - System B (TR30:		ULC	UCT3B	98.34	271.78	271.78	00.05	0.40		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - DS1 Loop Interface Ca		ULC	UCTCO	5.52	126.85	92.35	33.65	9.42		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Car		UDN	ULCC1	8.77	21.11	21	10.81	10.74		19.99	19.99	19.99	19.99
<del></del>	Unbundled Loop Concentration - UDC Loop Interface (Brite Car		UDC	ULCCU	8.77	21.11	21	10.81	10.74		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop Interface (POTS Card)		UEA	ULCC2	2.19	21.11	21	10.81	10.74		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface (SPOT	8	ur.	111.000	40.00	21.11	21	10.81	10.74		19.99	19.99	40.00	40.00
	Card)		UEA	ULCCR	13.03								19.99	19.99
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca		UEA	ULCC4	7.77 37.98	21.11 21.11	21 21	10.81	10.74		19.99 19.99	19.99 19.99	19.99 19.99	19.99
	Unbundled Loop Concentration - TEST CIRCUIT Car Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa		ULC UDL	UCTTC ULCC7	11.51	21.11	21	10.81 10.81	10.74		19.99	19.99	19.99	19.99 19.99
					11.51	21.11		10.81	10.74				19.99	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa		UDL	ULCC5	11.51	21.11	21	10.81			19.99	19.99	19.99	19.99
1 1				LILCCS	11.51	21 11		10.91			10.00	10.00	10.00	10.00
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa		UDL	ULCC6	11.51	21.11	21	10.81	10.74		19.99	19.99	19.99	19.99
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa		UDL	ULCC6	11.51	21.11		10.81			19.99	19.99	19.99	19.99
UNBUNDLED SUB-LOC			UDL	ULCC6	11.51	21.11		10.81			19.99	19.99	19.99	19.99
UNBUNDLED SUB-LOC	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa  P CONCENTRATION (OUTSIDE CO)		UDL	ULCC6	11.51	21.11		10.81			19.99	19.99	19.99	19.99
UNBUNDLED SUB-LOC			UDL	ULCC6	11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)		UDL	ULCC6	11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE				11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE   NID - Dispatch and Service Order for NID installation		UENTW	UNDBX	11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE		UENTW UENTW		11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW UENTW UEANL,UEF,UEQ,	UNDBX UENCE	11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE   NID - Dispatch and Service Order for NID installation		UENTW UENTW UEANL,UEF,UEQ, UENTW	UNDBX	11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW UENTW UEATW, UENTW UENTW UALJUC_JUC_JUD	UNDBX UENCE	11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate		UENTW UENTW UEANL,UEF,UEQ, UENTW	UNDBX UENCE	11.51	21.11		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW UENTW UENTW UENTW UENTW ULUCI,UDC,UD LUDN,UEA,UHI,U	UNDBX UENCE UNECN				10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate		UENTW UENTW UENTW UENTW UENTW ULUCI,UDC,UD LUDN,UEA,UHI,U	UNDBX UENCE UNECN				10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate		UENTW UEANL,UEF,UEQ, UENTW UAL,UC,UDC,UD L,UDN,UEA,UHL,U	UNDBX UENCE UNECN				10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no re		UENTW UENTW UENTW UENTW UENTW UAL,UCL,UDC,UD L,UDN,UCA,UHL,U LC UEA,UDN,UCL,UD	UNDBX UENCE UNECN UNECN USBFQ	0	0		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra		UENTW UENTW UEANL,UEF,UEQ,UENTW UAL,UCL,UDC,UD L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD C UEA,USL,UCL,UDL	UNDBX UENCE UNECN UNECN USBFQ USBFR	0 0 0	0		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled DS1 Loop - Superframe Format Option - no rate		UENTW UEANL, UENTW UEANL, UENTW UAL, UCL, UDC, UD L, UDN, UEA, UHL, U C UEA, UDN, UCL, UD UEA, UCL, UDL USL	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF	0 0 0 0 0	0 0 0 0		10.81			19.99	19.99	19.99	19.99
	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra		UENTW UENTW UEANL,UEF,UEQ,UENTW UAL,UCL,UDC,UD L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD C UEA,USL,UCL,UDL	UNDBX UENCE UNECN UNECN USBFQ USBFR	0 0 0	0		10.81			19.99	19.99	19.99	19.99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra		UENTW UEANL, UENTW UEANL, UENTW UAL, UCL, UDC, UD L, UDN, UEA, UHL, U C UEA, UDN, UCL, UD UEA, UCL, UDL USL	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF	0 0 0 0 0	0 0 0 0		10.81			19.99	19.99	19.99	19.99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID1 - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled DS1 Loop - Superframe Format Option - no rate  Unbundled DS1 Loop - Expanded Superframe Format option - no rate  NDLED LOCAL LOOP		UENTW UEANL, UENTW UEANL, UENTW UAL, UCL, UDC, UD L, UDN, UEA, UHL, U C UEA, UDN, UCL, UD UEA, UCL, UDL USL	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF	0 0 0 0 0	0 0 0 0		10.81			19.99	19.99	19.99	19.99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no r.  NDLED LOCAL LOOP  onth minimum billing period		UENTW UENTW UEANL,UEF,UEQ,UD UENTW UAL,UCL,UDC,UD L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD UEA,USL,UCL,UD USL USL	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF	0 0 0 0 0	0 0 0 0		10.81			19.99	19.99	19.99	19.99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled DS1 Loop - Superframe Format Option - no rate  Unbundled DS1 Loop - Expanded Superframe Format option - no rate  NDLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor		UENTW UENTW UENTW UENTW UENTW UALICL,UDC,UD L,UDN,UEA,UHL,U LC  UEA,UDN,UCL,UD UEA,UDN,UCL,UDL USL USL USL	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF	0 0 0 0 0	0 0 0 0 0 0	21	10.81					19.99	19.99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  NDLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor		UENTW UENTW UENTW UENTW UAL,UCL,UDC,UD L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD C UEA,USL,UCL,UDL USL USL USL USL UE3 UE3	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF	0 0 0 0 0 11.12 404.98	0 0 0 0		10.81			19.99	19.99	19.99	19,99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no r.  NOLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - DS3 - Per Saility Termination per mor  High Capacity Unbundled Local Loop - DS3 - Faility Termination per mor		UENTW UENTW UEANL,UEF,UEQ, UENTW UAL,UCL,UDL,UDL,UDN,UCL,UDL C UEA,UDN,UCL,UDL USL USL USL USL UE3 UE3 UE3 UE3 UDSX	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0	21	10.81			53.48	53.48	19.99	19.99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  NDLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor		UENTW UENTW UENTW UENTW UAL,UCL,UDC,UD L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD C UEA,USL,UCL,UDL USL USL USL USL UE3 UE3	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF	0 0 0 0 0 11.12 404.98	0 0 0 0 0 0	21	10.81					19.99	19.99
UNE OTHER, PROVISION HIGH CAPACITY UNBU	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - no rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no r.  NOLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - DS3 - Per Saility Termination per mor  High Capacity Unbundled Local Loop - DS3 - Faility Termination per mor		UENTW UENTW UEANL,UEF,UEQ, UENTW UAL,UCL,UDL,UDL,UDN,UCL,UDL C UEA,UDN,UCL,UDL USL USL USL USL UE3 UE3 UE3 UE3 UDSX	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0	21	10.81			53.48	53.48	19.99	19.99
UNE OTHER, PROVISION	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no re  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no re  Unbundled DS1 Loop - Superframe Format Option - no re  Unbundled DS1 Loop - Expanded Superframe Format option - no re  Unbundled DS1 Loop - Expanded Superframe Format option - no re  Unbundled DS1 Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - DS3 - Fer Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Fer Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Fer Mile per mor		UENTW UENTW UEANL,UEF,UEQ, UENTW UAL,UCL,UDL,UDL,UDN,UCL,UDL C UEA,UDN,UCL,UDL USL USL USL USL UE3 UE3 UE3 UE3 UDSX	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0	21	10.81			53.48	53.48	19.99	19.99
UNE OTHER, PROVISION HIGH CAPACITY UNBU	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No Rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no re  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no re  Unbundled DS1 Loop - Superframe Format Option - no re  Unbundled DS1 Loop - Expanded Superframe Format option - no r.  NDLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor		UENTW UENTW UENTW UENTW UENTW UALICL,UDC,UD L,UDN,UEA,UHL,U LC  UEA,UDN,UCL,UDL USL USL USL USSL UE3 UE3 UE3 UE3 UE3 UB3 UB3 UB3 UB3 UDLSX UDLSX	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0 1124.48 1124.48	699.6 699.6 56.34	10.81			53.48	53.48	19.99	19.99
UNE OTHER, PROVISION HIGH CAPACITY UNBU	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID1 - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled DS1 Loop - Superframe Format Option - no rate  Unbundled DS1 Loop - Expanded Superframe Format option - no rate  NDLED LOCAL LOOP  Onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).		UENTW UENTW UEANL,UEF,UEQ, UENTW UAL,UCL,UDC,UD L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD USL USL USL USL USL USL USS UDLSX UDLSX	UNDBX UENCE UNECN  UNECN  USBFQ  USBFR CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0 1124.48 1124.48	699.6	10.81			53.48	53.48	19.99	19.99
UNE OTHER, PROVISION HIGH CAPACITY UNBU	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no r.  NOLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local L		UENTW UENTW UEANLUEF,UEQ, UENTW UALLOCL,UDC,UDD, L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD USL USL USL USL USL USS UE3 UE3 UE3 UB3 UE3 UDLSX UDLSX UMK UMK	UNDBX UENCE UNECN UNECN USBFQ USBFC CCOSF CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1  UMKLW UMKLP	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0 1124.48 1124.48 56.34 58.56	699.6 699.6 56.34 58.56	10.81			53.48	53.48	19.99	19.99
HIGH CAPACITY UNBU	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID1 - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled DS1 Loop - Superframe Format Option - no rate  Unbundled DS1 Loop - Expanded Superframe Format option - no rate  NDLED LOCAL LOOP  Onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).		UENTW UENTW UENTW UENTW UENTW UALICL,UDC,UD L,UDN,UEA,UHL,U LC  UEA,UDN,UCL,UDL USL USL USL USSL UE3 UE3 UE3 UE3 UE3 UB3 UB3 UB3 UB3 UDLSX UDLSX	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0 1124.48 1124.48	699.6 699.6 56.34	10.81			53.48	53.48	19.99	19.99
HIGH CAPACITY UNBU NOTE: 4 m	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no r.  NOLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local L		UENTW UENTW UEANLUEF,UEQ, UENTW UALLOCL,UDC,UDD, L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD USL USL USL USL USL USS UE3 UE3 UE3 UB3 UE3 UDLSX UDLSX UMK UMK	UNDBX UENCE UNECN UNECN USBFQ USBFC CCOSF CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1  UMKLW UMKLP	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0 1124.48 1124.48 56.34 58.56	699.6 699.6 56.34 58.56	10.81			53.48	53.48	19.99	19.99
UNE OTHER, PROVISION HIGH CAPACITY UNBU	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no r.  NOLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mor  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local L		UENTW UENTW UEANLUEF,UEQ, UENTW UALLOCL,UDC,UDD, L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD USL USL USL USL USL USS UE3 UE3 UE3 UB3 UE3 UDLSX UDLSX UMK UMK	UNDBX UENCE UNECN UNECN USBFQ USBFC CCOSF CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1  UMKLW UMKLP	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0 1124.48 1124.48 56.34 58.56	699.6 699.6 56.34 58.56	10.81			53.48	53.48	19.99	19.99
HIGH CAPACITY UNBU NOTE: 4 m	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled DS1 Loop - Superframe Format Option - no rate  Unbundled DS1 Loop - Expanded Superframe Format option - no rate  NDLED LOCAL LOOP  Onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Mechanized)		UENTW UENTW UENTW UENTW UENTW UAL,UCL,UDC,UD L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UDC UEA,UDN,UCL,UDL USL USL USL USL USS UE3 UE3 UB3 UDLSX UDLSX UDLSX UMK UMK	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1  UMKLW UMKLP PSUMK	0 0 0 0 0 0 11.12 404.98 11.12 417.7	0 0 0 0 0 0 1124.48 1124.48 56.34 58.56 1.04	699.6 699.6 699.6 56.34 58.56 1.04	10.81		0	53.48	53.48	19.99	19.99
HIGH CAPACITY UNBU NOTE: 4 m	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra  Unbundled DS1 Loop - Superframe Format Option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  Unbundled DS1 Loop - Expanded Superframe Format option - no ra  NOLED LOCAL LOOP  onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Per Mile per mon  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mon  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundled Local Loop - STS-1 - Province per mon  High Capacity Unbundl		UENTW UENTW UEANLUEF,UEQ, UENTW UALLOCL,UDC,UDD, L,UDN,UEA,UHL,U LC UEA,UDN,UCL,UD USL USL USL USL USL USS UE3 UE3 UE3 UB3 UE3 UDLSX UDLSX UMK UMK	UNDBX UENCE UNECN UNECN USBFQ USBFC CCOSF CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1  UMKLW UMKLP	0 0 0 0 0 0 11.12 404.98 11.12	0 0 0 0 0 0 1124.48 1124.48 56.34 58.56	699.6 699.6 56.34 58.56	10.81			53.48	53.48	19.99	19.99
HIGH CAPACITY UNBU NOTE: 4 m	P CONCENTRATION (OUTSIDE CO)  NING ONLY - NO RATE  NID - Dispatch and Service Order for NID installation  UNTW Circuit Id Establishment, Provisioning Only - No Rate  Unbundled Contract Name, Provisioning Only - No Rate  Unbundled Contact Name, Provisioning Only - No rate  Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate  Unbundled DS1 Loop - Superframe Format Option - no rate  Unbundled DS1 Loop - Expanded Superframe Format option - no rate  NDLED LOCAL LOOP  Onth minimum billing period  High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor  Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).  Loop Makeup - Preordering With Reservation, per spare facility queried (Mechanized)		UENTW UENTW UEANL,UEP,UEQ, UENTW UAL,UCL,UDC,UD L,UDN,UCA,UHL,U LC UEA,UDN,UCL,UDL USL USL USL USL USS UDLSX UDLSX UDLSX UMK UMK UMK UMK	UNDBX UENCE UNECN UNECN USBFQ USBFR CCOSF CCOEF  1L5ND UE3PX 1L5ND UDLS1  UMKLW UMKLP PSUMK	0 0 0 0 0 0 11.12 404.98 11.12 417.7	0 0 0 0 0 0 1124.48 1124.48 158.56 1.04	699.6 699.6 699.6 1.04	10.81			53.48	53.48	19.99	19.99

Page 4 of 18 Version 2Q01: 08/30/01

	Line Sharing - per Line Activatio		ULS	ULSDC	0.61	56.92	28.59			26.94	12.76	
	Line Sharing - per Subsequent Activity per Line Rearrangeme	i	ULS	ULSDS		35.14	16.29			26.94	12.76	_
UNBUNDLED TRANSPO	RT T								<del>                                     </del>			_
COMMONIT	RANSPORT (Shared)											
COMMON	Common Transport - Per Mile, Per MOI				0.00001							
	Common Transport - Facilities Termination Per MO				0.00034							-
	Common Handport Talamado Tommadon Formo				0.00001							+
NOTE: INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS3	= one month, I	DS3 and above four i	months								
INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE											
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per mo Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination p	0.5	U1TVX	1L5XX	0.0282							
	month	EI	U1TVX	U1TV2	18	175.55	90.65			38.07	38.07	
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile pe	r	OTTVX	01112	10	170.00	30.03			30.07	50.01	
	month		U1TVX	1L5XX	0.0282							
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination p											
	month		U1TVX	U1TR2	18	18	90.65	0 0		38.07	38.07	
				41.500								
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month		U1TVX	1L5XX	0.0282							
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination per month		U1TVX	U1TV4	22.16	128.43	88.27			22.32	22.32	
	per monur		UIIVA	01174	22.10	128.43	00.21		+ + +	22.32	22.32	+
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per mor		U1TDX	1L5XX	0.0282							+
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per more		U1TDX	U1TD5	17.4	175.55	90.65			38.07	38.07	_
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per mor		U1TDX	1L5XX	0.0282							
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mor		U1TDX	U1TD6	17.4	175.55	90.65	0 0		38.07	38.07	
INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT - DS1		U1TD1	41.577	0.5750							+
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mor			1L5XX U1TF1	0.5753	255.24	201.82			38.07	38.07	
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor		U1TD1	UTIFT	71.29	255.24	201.82			38.07	38.07	+
INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT- DS3											
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor		U1TD3	1L5XX	12.98							
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mor		U1TD3	U1TF3	720.38	886.2	670.81			91.26	91.26	
INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT- STS-1											
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mor		U1TS1	1L5XX	6.14	005.74	400.07			50.40	50.40	
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor		U1TS1	U1TFS	790.37	695.71	462.37			53.48	53.48	
												_
	ANNEL - DEDICATED TRANSPORT											
NOTE: LOC	AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one mo	onth, DS3 and a										
	Local Channel - Dedicated - 2-Wire Voice Grade Per Month		ULCVX	ULDV2	40.54	505.07	400.45			42.17	12.76	
	Local Channel - Dedicated - 2-Wire Voice Grade per month - Zone Local Channel - Dedicated - 2-Wire Voice Grade per month - Zone	2	ULCVX	ULDV2 ULDV2	12.51 21.23	595.97	102.45 102.45					
	Local Channel - Dedicated - 2-Wire Voice Grade per month - Zone	3	ULCVX	ULDV2	24.62	595.97 595.97	102.45					+
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor	J	ULCVX	ULDR2	0	0	0	0 0		38.07	38.07	_
	Local Channel - Dedicated - 4-Wire Voice Grade per month - Zone	1	UNCVX	ULDV4	13.4	604.4	105.43					
	Local Channel - Dedicated - 4-Wire Voice Grade per month - Zone	2	UNCVX	ULDV4	22.73	604.4	105.43					
	Local Channel - Dedicated - 4-Wire Voice Grade per month - Zone	3	UNCVX	ULDV4	26.37	604.4	105.43			-		
	Local Channel - Dedicated - 4-Wire Voice Grade per mon		UNCVX	ULDV4						42.17	12.76	
	Local Channel - Dedicated - DS1 per month - Zone	1 2	ULDD1 ULDD1	ULDF1 ULDF1	30.12 51.11	620.63	464.46 464.46					+
	Local Channel - Dedicated - DS1 per month - Zone Local Channel - Dedicated - DS1 per month - Zone	3	ULDD1	ULDF1 ULDF1	51.11	620.63 620.63	464.46 464.46					+
	Local Channel - Dedicated - DS1 per mont	3	ULDD1	ULDF1	33.20	020.03	404.40			86.15	1.77	+
	Local Channel - Dedicated - DS3 - Per Mile per mon		ULDD3	1L5NC	8.66					55.15		_
	Local Channel - Dedicated - DS3 - Facility Termination per mor		ULDD3	ULDF3	496.76	618.5	584.13			56.25	56.25	
	Local Channel - Dedicated - STS-1- Per Mile per mon		ULDS1	1L5NC	8.66							4
	Local Channel - Dedicated - STS-1 - Facility Termination per mor		ULDS1	ULDFS	484.06	1124.48	699.6	0 0		38.07	38.07	
MULTIPLEXERS				-								+
MULTIFLEXERS	Channelization - DS1 to DS0 Channel Syste		UXTD1	MQ1	146.69	222.55	148.22		+ + +	24.85	8.16	+
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb		UDL	1D1DD	146.69	13.09	9.38			۷4.00	0.10	+
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System - per mont		UDN	UC1CA	3.59	13.09	9.38					+
	Voice Grade COCI - DS1 to DS0 Channel System - per mon		UEA	1D1VG	1.27	13.09	9.38					
	DS3 to DS1 Channel System per mont		UXTD3	MQ3	233.1	428.75	241.82			24.78	7.42	
	STS1 to DS1 Channel System per mont		UXTS1	MQ3	233.1	428.75	241.82			38.07	38.07	$\bot$
	DS3 Interface Unit (DS1 COCI) used with Loop per mont		USL	UC1D1	16.07	13.09	9.38					+
DARK FIBER												+
DAKK FIDEK	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local				1	1			+ + +			+
	Channel		UDF	1L5DC	53.86							
	NRC Dark Fiber - Local Channe		UDF	UDFC4		1807	562.96			38.07	38.07	+
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Interoff											
	Channel		UDF	1L5DF	27.71							

Page 5 of 18 Version 2Q01: 08/30/01

		NRC Dark Fiber - Interoffice Channe		UDF	UDF14		1807	562.96	0	0		38.07	38.07		
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local		<u></u>					-						
		Loop		UDF	1L5DL	53.86									
		NRC Dark Fiber - Local Loop		UDF	UDFL4		1807	562.96	0	0		38.07	38.07		
TRANSPOR	T OTHER														
	Optional Fe	atures & Functions:													
		Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chant		UNC1X	CCOEF		184.76	23.6	1.99	0.78		29.33	3.93		
0VV 100E0		Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chani  **SCREENING**		UNC1X	CCOSF		184.76	23.6	1.99	0.78		29.33	3.93		
8XX ACCES			-	OHD		0.0005									
		8XX Access Ten Digit Screening, Per Ca  8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv		OHD	N8R1X	0.0005	7.05	0.96				26.94	26.94		
		8XX Access Ten Digit Screening, Reservation Charge Fel oxx Number Reserv	+ - 1	OHD	NONTA		23.82	2.73				26.94	26.94		
		8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation		OHD	N8FTX		23.82	2.73				26.94	26.94		
		8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb		OHD	N8FCX		5.63	2.82				26.94	26.94		
		8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested													
		Per 8XX No.		OHD	N8FMX		6.59	3.77				26.94	26.94		
		8XX Access Ten Digit Screening, Change Charge Per Reque		OHD	N8FAX		8.01	0.96				26.94	26.94		
		8XX Access Ten Digit Screening, Call Handling and Destination Featur		OHD	N8FDX		5.63					26.94	26.94		
		8XX Access Ten Digit Screening, w/ 8XX No. Delivery, per que		OHD		0.00365									
		8XX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional													
		Complex Features, per quer		OHD	1	0.00431									
		8XX Access Ten Digit Screening, w/ POTS No. Delivery, per que		OHD	1	0.00383									
		8XX Access Ten Digit Screening w/ POTS No. Delivery, with Optional Complex Feature		OUB											
<b>—</b>		per query	+	OHD	1	0.00431		1			1		1		
I INTERNET	MATICHER	FA DAGE ACCECC (LIDD)	$\vdash$		-			-			1				
LINE INFOR	WATION DAT	TA BASE ACCESS (LIDB)	$\vdash$	OQT	-	0.0000		-			1				
	-	LIDB Common Transport Per Quer LIDB Validation Per Quer		OQU	-	0.0003 0.0134		-			1				
		LIDB Validation Per Quer LIDB Originating Point Code Establishment or Chanç		OQT, OQU	NRPBX	0.0134	62.26				62.26	26.94	26.94		
		EIDB Originating Form Code Establishment of Chang	+ - 1	001,000	MINEDA		02.20				02.20	20.54	20.54		
SIGNALING	(CCS7)														
OIOITALIITO	(0001)	CCS7 Signaling Termination, Per STP Por		1DB	PT8SX	132.83						19.99	19.99	19.99	19.99
		CCS7 Signaling Usage, Per TCAP Message		1DB	1 100%	0.00009						10.00	10.00	10.00	10.00
		CCS7 Signaling Connection, Per link (A link		1DB	TPP++	18.22	278.02	278.02				19.99	19.99	19.99	19.99
		CCS7 Signaling Connection, Per link (B link) (also known as D lin		1DB	TPP++	18.22	278.02	278.02				19.99	19.99	19.99	19.99
		CCS7 Signaling Usage, Per ISUP Messag		1DB		0.00004									
		CCS7 Signaling Usage Surrogate, per link per LAT		1DB	STU56	338.98						19.99	19.99	19.99	19.99
		CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per													
		STP affected		1DB	CCAPO		40	40				19.99	19.99	19.99	19.99
		CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per													
		Stp Affected		1DB	CCAPD		8	8				19.99	19.99	19.99	19.99
E911 SERVI	CE														
CALLINGNI	ARE (CHARA)	OFFINIOR	-												
CALLING NA	AIVIE (CNAIVI)		+ -	OQV		0.040									
		CNAM for DB Owners, Per Quer CNAM for Non DB Owners, Per Quer		OQV		0.016 0.01									
		CNAMI IOI NOII DB OWIIEIS, FEI QUEI		OQV		0.01									
		CNAM (Non-Databs Owner), NRC, applies when using the Character Based User													
		Interface (CHUI)		OQV	CDDCH		595	595				26.94	26.94		
LNP QUERY	SERVICE														
				-			-			·					
	OPERATOR	SERVICES AND DIRECTORY ASSISTANCE													
			$\Box$	•											
OPERATOR	CALL PROC														
		Oper. Call Processing - Oper. Provided, Per Min Using BST LID				1.2									
		Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIE				1.24									
<u> </u>		Oper. Call Processing - Fully Automated, per Call - Using BST LID			1	0.11		1					1		
		Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE			1	0.12									
INWARD OP	EDATOR	DVICES	$\vdash$		1			1			1				
INVVARD OP						0.0							1		
-		Inward Operator Services - Verification, Per C: Inward Operator Services - Verification, Per Minu	+		1	0.8 1.15		1					1		
-		Inward Operator Services - Verification, Per Minu  Inward Operator Services - Verification and Emergency Interrupt - Per C	+		1	0.85		1			+		1		
		Inward Operator Services - Verification and Emergency Interrupt - Per C  Inward Operator Services - Verification and Emergency Interrupt - Per Min	+		1	1.15		<u> </u>							
		miwara Operator Octyrices - Verification and Emergency interrupt - Fer Milli				1.10					1				
BRANDING -	- OPERATOR	R CALL PROCESSING	$\vdash$		1			1							
		Recording of Custom Branded OA Announcement			CBAOS		7000	7000				19.99	19.99	19.99	19.99
		Loading of Custom Branded OA Announcement per shelf/NAV			CBAOL		500	500				19.99	19.99	10.00	
		g g g g g g g g g g g g g g g g g g g	$\vdash$		SEAGE		550	500				.5.55	. 5.55		
DIRECTORY	ASSISTANO	CE SERVICES										İ			
		ASSISTANCE ACCESS SERVICE										İ			
		** ** *													

		Directory Assistance Access Service Calls, Charge Per Ca				0.26									
	DIRECTOR	Y ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)				0.000									
		Directory Assistance Call Completion Access Service (DACC), Per Call Attem				0.062									
	UNBRANDI	NG													
		YTRANSPORT													+
		Directory Transport - Local Channel DS				35.68	534.48	462.69				86.15	86.15	1.77	1.77
		Directory Transport - DS1 Level Interoffice Per Mi				0.5753									
		Directory Transport - DS1 Level Interoffice Per Facility Terminati				71.29	217.17	163.75				38.07	38.07	1.77	1.77
		Switched Common Transport Per DA Access Service Per Ca Switched Common Transport Per DA Access Service Per Call Per Mi				0.0002									-
		Access Tandem Switching Per DA Access Service Per Ca				0.0001									+
															+
	DIRECTOR	Y ASSISTANCE DATA BASE SERVICE (DADS)													
		Directory Assistance Data Base Service Charge Per Listir				0.04									
	DIDECTOR	Directory Assistance Data Base Service, per mon			DBSOF	150									
BRANDING	- DIRECTOR	RY ASSISTANCE  Custom Branding Announcement, per Recording to be used with the provision of DA		AMT	CBADA		3000	3000							
		Loading of Custom Branded Announcement per DRAM Card/Switch		AMT	CBADA		690	690							
		Estading of Sustain Branded Announcement per Brand Surgrowten		Zuvii	OBABO		030	030							+
SELECTIVE	E ROUTING														
		Selective Routing Per Unique Line Class Code Per Request Per Swit			USRCR		229.65	229.65				40.18	9.45		
															<u> </u>
VIRTUAL C	OLLOCATIO	N		<b>.</b>		1		1					1	1	<del>                                     </del>
		Virtual Collocation - 2-wire Cross Connects (loop		ueanl,uea,udn,udc,u	UEAC2	0.09	41.78	39.23	4.75	4.75		19.99	19.99	19.99	19.99
	+	Virtual Collocation - 2-wire Cross Connects (loo)  Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splittin	1	al,uhl,ucl,uec UEPSR, UEPSB	VE1LS	0.09	41.78	39.23	4.75	4.75	-	19.99	19.99	19.99	19.99
		Virtual Collocation - 2-wire Cross Connects (por	-	52. 5.1, 62. 6B	VE1R2	0.09	41.78	39.23	4.75	4.75	<del> </del>	19.99	19.99	19.99	19.99
		Virtual Collocation - 4-wire Cross Connects (loop		uea,uhl,ucl,ud	UEAC4	0.18	41.91	39.25	4.73	4.73		19.99	19.99	19.99	19.99
		Virtual Collocation - 4-wire Cross Connects (por			VE1R4	0.18	41.91	39.25	4.73	4.73		19.99	19.99	19.99	19.99
		Virtual Collocation - 2-Fiber Cross Connect		CLO	CNC2F	15.99	67.34	48.55				19.99	19.99	19.99	19.99
		Virtual Collocation - 4-Fiber Cross Connects		CLO	CNC4F	28.74	82.35	63.56				19.99	19.99	19.99	19.99
		Virtual Collocatin - DS1 Cross Connect		USL,ULC,CLO	CNC1X	0.97	71.02	51.08							
AIN SELEC	TIVE CARRIE	ER ROUTING													
		Regional Service Establishment		SRC	SRCEC		391788					19.99	19.99	19.99	19.99
		End Office Establishment		SRC	SRCEO		320.53	320.53				19.99	19.99	19.99	19.99
		Line/Port NRC, per end user		SRC	SRCLP		2.06	2.06				19.99	19.99	19.99	19.99
		Query NRC, per query		SRC		0.000448									
AIN DELL	COLITIL AIN C	BMS ACCESS SERVICE													
AIN - BELL	SOUTH AIN S	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			CAMSE		294.77	294.77				26.94	26.94		
		AIN SMS Access Service - Port Connection - Dial/Shared Access			CAMDP		86.94	86.94				26.94	26.94		
		AIN SMS Access Service - Port Connection - ISDN Access			CAM1P		86.94	86.94				26.94	26.94		+
		AIN SMS Access Service - User Identification Codes - Per User ID Code			CAMAU		200.83	200.83				26.94	26.94		+
		AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			CAMRC		172.05	172.05				26.94	26.94		
		AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)				0.0023									
		AIN SMS Access Service - Session, Per Minute				0.0791									
		AIN SMS Access Service - Company Performed Session, Per Minute				2.08									
AIN DELL	COLITIL AIN T	FOOLKIT SERVICE													-
AIN - BELLS	SOUTH AIN I	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			BAPSC		290.05	290.05				26.94	26.94		
		AIN Toolkit Service - Training Session, Per Customer			BAPVX		8363	8363				26.94	26.94		
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt			BAPTT		72.76	72.76			<del> </del>	26.94	26.94		
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay			BAPTD		72.76	72.76				26.94	26.94	İ	
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate			BAPTM		72.76	72.76				26.94	26.94		<b></b>
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP		1	BAPTO	1	149.95	149.95				26.94	26.94		
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code			BAPTC BAPTF	-	149.95 149.95	149.95 149.95				26.94 26.94	26.94 26.94		<del>                                     </del>
	1	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code  AIN Toolkit Service - Query Charge, Per Query			BAPIF	0.02	149.95	149.95			+	∠0.94	20.94	1	<del>                                     </del>
		AIN Toolkit Service - Query Charge, Per Query  AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per				0.02									<del>                                     </del>
		Query				0.005									
		AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes				1.45									<u> </u>
1		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription			BAPMS	15.98	71.8	71.8				26.94	26.94		<del>                                     </del>
		AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription		1	BAPLS	0.08	47.2	47.2				26.94	26.94		
	-	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription		-	BAPDS BAPES	15.9 0.003	71.8 47.2	71.8 47.2				26.94	26.94 26.94	1	<del>                                     </del>
		AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			DAFES	0.003	41.2	41.2				26.94	20.94		<del>                                     </del>
ODUF/EDO	UF/ADUF/CM	IDS				<b> </b>									<del>                                     </del>
		·				1		1			+				<b>T</b>
	ACCESS DA	AILY USAGE FILE (ADUF)													
		ADUF: Message Processing, per messag				0.004									
		ADUF: Data Transmission (CONNECT:DIRECT), per messagi				0.001		1						1	
	ENHANCES	O OPTIONAL DAILY USAGE FILE (EODUF)				-									<del>                                     </del>
<b>+</b>		EODUF: Message Processing, per messag				0.004		1					1		<del> </del>
	1	LODG: Message Flocessing, per messag		1		0.004	1	1	L	1				1	

Page 7 of 18 Version 2Q01: 08/30/01

OPTION/	AL DAILY USAGE FILE (ODUF)													
	ODUF: Recording, per message				0.0003									
	ODUF: Message Processing, per message				0.0032									
	ODUF: Message Processing, per Magnetic Tape provisions				54.61									
	ODUF: Data Transmission (CONNECT:DIRECT), per message				0.00004									
DEVTEND	ED LINK (EELs)													
LXILIND	ED LINK (LLLS)													
NOTE: N	ew EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami	FI · Ft	Lauderdale ELI:	Nashvilla TN: N	ow Orleans I A									
	harlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates beli				ew Orieans, LA	4								
	all states, EEL network elements shown below also apply to currently combined facilities w				As Is Charge	annlies to curre	ntly combined	facilities con	verted to UN	Es (Non-recui	rring rates do not apply )			
	Georgia, the EEL network elements apply to ordinarily combined network elements per the					applico to carro	inity combined		10.100.10 0.1		ing rates as not apply.)			
11012	Soorgia, the 222 notion of somethic apply to oraniamy combined notions contents por the	0,11.01	o or dorigino o mitori	7.0.10 01.0.90.7										
2-WIRE V	OICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)													
	First 2-Wire VG Loop - Service Level 2/DS1 Interofficed Transport Combination -													
	Statewide	sw	UNCVX	UEAL2	19.5	142.97	106.56				38.07	38.07		
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor		UNC1X	1L5XX	0.5753									
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo		UNC1X	U1TF1	71.29	217.17	163.75				38.07	38.07		
	DS1 Channelization System Per Mont		UNC1X	MQ1	146.69	197.78	140.06							
	Voice Grade COCI - DS1 To Ds0 Interface - Per Montl		UNCVX	1D1VG	1.27	13.09	9.38							
	Each Additional 2-Wire Vg Loop(Sl2) In The Same Ds1 Interoffice Transport Combinati													
	Per Month		UNCVX	UEAL2	19.5	142.97	108.56				21.75	21.75	31.26	
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3	3	UNCVX	UEAL2	1									
+	Voice Grade COCI - DS1 to DS0 Channel System combination - per mor	3	UNCVX	1D1VG	1.27	13.09	9.38					1		$\vdash$
+	Nonrecurring Currently Combined Network Elements Switch -As-Is Char	+	UNC1X	UNCCC	1.21	21.75	21.75	32.28	10.96		38.07	38.07		
1	Transcrang Currently Combined Network Elements Switch "As-18 Chair	+	GNUIA	5,4000		21.73	21.13	02.20	10.50		30.07	55.01		
4-WIRE V	OICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	+	1				1					1		
1	First 4-Wire Analog Voice Grade Loop/DS1 Interoffice Transport Combination -	+	1				1					1		
	Statewide	sw	UNCVX	UEAL4	27.49	288.47	237.45				21.75	21.75	31.26	
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.5753									
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor		UNC1X	U1TF1	71.29	217.17	163.75				38.07	38.07		
	Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	146.69	197.78	140.06							
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mor		UNCVX	1D1VG	1.27	13.09	9.38							
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport													
	Combination - Statewid	SW	UNCVX	UEAL4	27.49	288.47	237.45				38.07	38.07		
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mor  Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCVX UNC1X	1D1VG UNCCC	1.27	13.09 21.75	9.38 21.75	32.28	10.96		38.07	38.07		
	Nonrecurring Currently Combined Network Elements Switch -As-is Char		UNCIA	UNCCC		21.75	21.75	32.20	10.90		36.07	30.07		
4-WIRE 5	6 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL	١												
4-WIKE 3	First 4-Wire 56Kbps Digital Grade Loop/DS1 Interoffice Transport Combination -	.,												
	Statewide	sw	UNCDX	UDL56	37.67	489.04	337.51				21.75	21.75	32.26	
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.5753	100.01	001.01				210	210	02.20	
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Moi		UNC1X	U1TF1	71.29	217.17	163.75				38.07	38.07		
	Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	146.69	197.78	140.06							
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb		UNCDX	1D1DD	2	15.76	11.28							
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport													
	Combination - Statewid	SW	UNCDX	UDL56	37.67	489.04	337.51				21.75	21.75	32.26	
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-				_									
	64kbs)		UNCDX	1D1DD	2	15.76	11.28	32.28	10.96		38.07	38.07		
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC1X	UNCCC		21.75	21.75	32.28	10.96		38.07	38.07		
4-WIDE 6	4 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL	١	+		-		+	1		1		-		-
4-VVIKE 6	First 4-Wire 64Kbps Digital Grade Loop/DS1 Interoffice Transport Combination -	,	+	+	+		1					1		$\vdash$
	Statewide	sw	UNCDX	UDL64	37.67	489.04	337.51				21.75	21.75	32.26	
1	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	348	UNC1X	1L5XX	0.5753	.55.04	557.51				21.73	21.70	52.20	H
	and the second s		2.1017	0,0,1	2.3100									
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Moi		UNC1X	U1TF1	71.29	217.17	163.75				38.07	38.07		
	Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	146.69	197.78	140.06							
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-													
1	64kbs)		UNCDX	1D1DD	2	15.76	11.28							
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	1 -					1					1	T	
-	Combination - Statewid	SW	UNCDX	UDL64	37.67	489.04	337.51				21.75	21.75	12.61	
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		LINODY	40400		45.70	44.00							
1	64kbs)  Nonrecurring Currently Combined Network Elements Switch -As-Is Char	+-	UNCDX UNC1X	1D1DD UNCCC	2	15.76 21.75	11.28 21.75	32.28	10.96	1	38.07	38.07		-
+	inoniecuming Currently Combined Network Elements Switch -As-is Chaft	+	UNCIX	UNCCC	<b>-</b>	21./5	21./5	32.28	10.96		38.07	38.07		
4-WIRF	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	+	+		<b>†</b>		1					1		
L	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Statew	SW	UNC1X	USLXX	62.78	714.84	421.47				21.75	21.75	32.26	
+	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	344	UNC1X	1L5XX	0.5753	7 14.04	741.77				21.70	21.70	52.20	
+	DOGOGOGO DO COMBINACION TO MICE OF MICE		5X		0.0.00		1							
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Moi		UNC1X	U1TF1	71.29	217.17	163.75				38.07	38.07		
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC1X	UNCCC		21.75	21.75	32.28	10.96		38.07	38.07		
4-WIRE D	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)	$oldsymbol{ol}}}}}}}}}}}}}}$				1						1		
	First DOAL and in DOO Interesting Transport Combination Contact	T	LINICAY	USLXX	62.78	714.84	421.47	1	1	1	118.2	104.02	-	
	First DS1Loop in DS3 Interoffice Transport Combination - Statewi	SW	UNC1X			/14.84	421.47				110.2	104.02		
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS3 - Facility Termination per mor	SW	UNC3X UNC3X	1L5XX U1TF3	12.98 720.38	714.84	579.55				118.2	104.02		

Page 8 of 18 Version 2Q01: 08/30/01

	DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	16.07	13.09	9.38	ı	ı				1
	Additional DS1Loop in DS3 Interoffice Transport Combination - Statewi	sw	UNC1X	USLXX	62.78	714.84	421.47			20	8.07	38.07	
	DS3 Interface Unit (DS1 COCI) combination per monti	SW	UNC1X	UC1D1	16.07	13.09	9.38			30	0.07	30.07	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC3X	UNCCC	10.07	21.75	21.75	32.28	10.96	38	8.07	38.07	
	Tremocuring currently combined treatment Elements Circle Tree Criticis		0.100/	0.1000		2	210	02.20	10.00		0.01	00.07	
2-WIRE VO	DICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (	EEL)											
-	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Statewi	sw	UNCVX	UEAL2	19.5	142.97	106.56						
	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor		UNCVX	1L5XX	0.0282								
	Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Terminati												
	per month		UNCVX	U1TV2	18	137.48	52.58			38	8.07	38.07	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCVX	UNCCC		21.75	21.75	32.28	10.96	38	8.07	38.07	
4-WIRE VO	DICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (	EEL)											
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Statew	SW	UNCVX	UEAL4	27.49	288.47	237.45	82.08	12.22				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor		UNCVX	1L5XX	0.0282								
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Terminati												
	per month		UNCVX	U1TV4	22.16	106.11	65.95			22	2.32	22.32	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCVX	UNCCC		21.75	21.75	32.28	10.96	38	8.07	38.07	
DS3 DIGITA	AL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)												
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor		UNC3X	1L5ND	11.12								
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per mo		UNC3X	UE3PX	404.98	1071	646.12						
	Interoffice Transport - Dedicated - DS3 - Per Mile per mon		UNC3X	1L5XX	12.98								
													1
	Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mo		UNC3X	U1TF3	720.38	794.94	679.55				1.26	91.26	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC3X	UNCCC		21.75	21.75	32.28	10.96	38	8.07	38.07	
STS1 DIGIT	TAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)												
	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor		UNCSX	1L5ND	11.12								
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per							1	1				
	month		UNCSX	UDLS1	417.7	1071	646.12						
	Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor		UNCSX	1L5XX	6.14								
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mo		UNCSX	U1TFS	790.37	794.94	679.55				3.48	53.48	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCSX	UNCCC		21.75	21.75	32.28	10.96	38	8.07	38.07	
2-WIRE ISE	DN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)												
	First 2-Wire ISDN Loop/DS1 Interoffice Combination Transport - Statewi	SW	UNCNX	U1L2X	24.98	325.91	251.31			38	8.07	38.07	
	Interoffice Transport - Dedicated - DS1 combination - Per Mi		UNC1X	1L5XX	0.5753								
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per mor		UNC1X	U1TF1	71.29	217.17	163.75			38	8.07	38.07	
	Channelization - Channel System DS1 to DS0 combination - per mor		UNC1X	MQ1	146.69	197.78	140.06						
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon		UNCNX	UC1CA	3.59	15.76	11.28						
	Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Statew	SW	UNCNX	U1L2X	24.98	325.91	251.31			38	8.07	38.07	
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per mon		UNCNX	UC1CA	3.59	15.76	11.28						
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC1X	UNCCC		21.75	21.75	32.28	10.96	38	8.07	38.07	
4-WIRE DS	S1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT												
	First DS1 Loop in STS1 Interoffice Transport Combination - Statewi	SW	UNCIX	USLXX	62.71	714.84	757.03			53	3.48	53.48	
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor		UNCSX	1L5XX	6.14								
	Interoffice Transport - Dedicated - STS1 combination - Facility Terminati		UNCSX	U1TFS	790.37	794.94	679.55			53	3.48	53.48	
	STS1 to DS1 Channel System conbination per mon		UNCSX	MQ3	233.1	403.9	234.4						
	DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	16.07	13.09	9.38						
	Additional DS1Loop in STS1 Interoffice Transport Combination - Statewi	sw	UNC1X	USLXX	62.71	714.84	757.03			38	8.07	38.07	
	DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	16.07	13.09	9.38						
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCSX	UNCCC		21.75	21.75	32.28	10.96	38	8.07	38.07	
						1							
4-WIRE 56	KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)					1							
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Statew	SW	UNCDX	UDL56	37.67	489.04	337.51						
	Hatasaffin Tananant Dadinated Assis FOllows combination Daniel		UNCDX	1L5XX	0.0282								
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per M		UNCDX	U1TD5	17.4	137.48	52.58				8.07	38.07	
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminati		UNCDA		1	21.75	21.75	32.28	10.96	38	8.07	38.07	
			UNCDX	UNCCC		21.73	21.70	02.20					
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch - As-Is Chan		UNCDX	UNCCC		21.73	21.70	32.20					
4-WIRE 64	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Char  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)		UNCDX					32.20					
4-WIRE 64	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Char  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew	SW	UNCDX	UDL64	32.67	489.04	337.51	32.20					
4-WIRE 64	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch - 4s-Is Chan  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew  Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M		UNCDX  UNCDX  UNCDX	UDL64 1L5XX	0.0282	489.04	337.51	32.20					
4-WIRE 64	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Charj Nonrecurring Currently Combined Network Elements Switch -As-Is Charj Nonrecurring Currently Combination - Facility Termination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminat		UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6		489.04 137.48	337.51 52.58				8.07	38.07	
4-WIRE 64	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch - 4s-Is Chan  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew  Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M		UNCDX  UNCDX  UNCDX	UDL64 1L5XX	0.0282	489.04	337.51	32.28	10.96		8.07 8.07	38.07 38.07	
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch - As-Is Chan  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M  Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch - As-Is Chan		UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6	0.0282	489.04 137.48	337.51 52.58						
4-WIRE 64	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch - As-Is Chan  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M  Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination Nonrecurring Currently Combined Network Elements Switch - As-Is Chan		UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6	0.0282	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORI	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan  K ELEMENTS	SW	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6 UNCCC	0.0282	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORK	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Char Nonrecurring Currently Combined Network Elements Switch -As-Is Char Hery State - 1	sw out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6 UNCCC	0.0282	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORK	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan  K ELEMENTS	sw out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6 UNCCC	0.0282	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORK	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Char Nonrecurring Currently Combined Network Elements Switch -As-Is Char Hery State - 1	sw out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6 UNCCC	0.0282	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORK	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Char Nonrecurring Currently Combined Network Elements Switch -As-Is Char Hery State - 1	sw out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6 UNCCC	0.0282	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORK	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Char  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per III Nonrecurring Currently Combined Network Elements Switch - As-Is Char  K ELEMENTS  d as a part of a currently combined facility, the non-recurring charges do not apply, to d as ordinarilty combined network elements in Georgia, the non-recurring charges and	sw out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  S Is charge does a witch As Is Charge	UDL64 1L5XX U1TD6 UNCCC UNCCC	0.0282 17.4	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORI  When used	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Char  KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)  4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per III Nonrecurring Currently Combined Network Elements Switch - As-Is Char  K ELEMENTS  d as a part of a currently combined facility, the non-recurring charges do not apply, to d as ordinarilty combined network elements in Georgia, the non-recurring charges and	sw out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX	UDL64 1L5XX U1TD6 UNCCC	0.0282	489.04 137.48	337.51 52.58						
ADDITIONAL NETWORI  When used	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL) 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan  K ELEMENTS  d as a part of a currently combined facility, the non-recurring charges do not apply, to d as ordinarilty combined network elements in Georgia, the non-recurring charges al chroNet)	sw out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  S Is charge does a witch As Is Charge	UDL64 1L5XX U1TD6 UNCCC UNCCC	0.0282 17.4	489.04 137.48	337.51 52.58						
DDITIONAL NETWORI  When used  When used	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL) 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Statew Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminat Nonrecurring Currently Combined Network Elements Switch -As-Is Chan  K ELEMENTS  d as a part of a currently combined facility, the non-recurring charges do not apply, to d as ordinarilty combined network elements in Georgia, the non-recurring charges al chroNet)	out a Switch A	UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  UNCDX  Witch As is Charge	UDL64 1L5XX U1TD6 UNCCC UNCCC	0.0282 17.4	489.04 137.48	337.51 52.58						

Page 9 of 18 Version 2Q01: 08/30/01

Cha 56/6 Cha DS1 DS3 STS Cha NOTE: Local Ch AL SUPPORT SI OOTE: (1) Electric NOTE: (1) Concluded	84 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion arge  I Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charg  B Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charg  Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charg  Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion arge  Interoffice Or Local Loop used in a COMBINATION - "Switch As Is" Conversion arge  Interoffice Or Local Loop used in a COMBINATION - "Switch As Is" Conversion arge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  Interoffice Channel used in a ComBINATION - "Switch As Is" Conversion C	tate speci	l abov	UNCVX UNCDX UNC1X UNC3X UNCSX uncsx	UNCCC UNCCC UNCCC UNCCC UNCCC		21.75 21.75 21.75 21.75 21.75	21.75 21.75 21.75 21.75 21.75	32.28 32.28 32.28 32.28	10.96 10.96 10.96			38.07 38.07 38.07 38.07	38.07 38.07 38.07 38.07		
DS1 DS3 DS3 STS Cha NOTE: Local Ch L SUPPORT SI OOTE: (1) Electric NOTE: (1) Concluded	Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charges Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charges Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charges Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion arges Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion arges Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion arges Interoffice or Local Loop Used Interoffice or Local Loop Used Interoffice Order: CLEC-1 should contact its contract negotiator if it prefers the st nuclear CLEC-1 may elect either the state specific Commission ordered rates for the	tate speci	l abov	UNC1X UNC3X UNCSX	UNCCC		21.75 21.75	21.75 21.75	32.28 32.28	10.96			38.07	38.07		
DS3 STS Cha ACTE: Local Ch AL SUPPORT SI NOTE: (1) Electre NOTE: (1) Contin NOTE: (1) Condin NOTE: (2) Manu	B Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Chan Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion large  annel - Dedicated Transport - minimum billing period - Below DS3=one month,  YSTEMS  ronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st nued: The electronic service ordering charge currently contained in this rate exhibit inded: CLEC-1 may elect either the state specific Commission ordered rates for the	tate speci	labov	UNC1X UNC3X UNCSX	UNCCC		21.75 21.75	21.75 21.75	32.28 32.28	10.96			38.07	38.07		
DS3 STS Cha ACTE: Local Ch AL SUPPORT SI NOTE: (1) Electre NOTE: (1) Contin NOTE: (1) Condin NOTE: (2) Manu	B Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Chan Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion large  annel - Dedicated Transport - minimum billing period - Below DS3=one month,  YSTEMS  ronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st nued: The electronic service ordering charge currently contained in this rate exhibit inded: CLEC-1 may elect either the state specific Commission ordered rates for the	tate speci	labov	UNC3X UNCSX	UNCCC		21.75	21.75	32.28							
NOTE: Local Ch.  AL SUPPORT SYNOTE: (1) Electre NOTE: (1) Conclin NOTE: (1) Conclin NOTE: (2) Manu	31 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion arge  nannel - Dedicated Transport - minimum billing period - Below DS3=one month,  YSTEMS  ronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st nued: The electronic service ordering charge currently contained in this rate exhibit i luded: CLEC-1 may elect either the state specific Commission ordered rates for the	tate speci	labov	UNCSX						10.96			38.07	38.07		
AL SUPPORT SYNOTE: (1) Electro NOTE: (1) Continuo NOTE: (1) Continuo NOTE: (2) Manu	nannel - Dedicated Transport - minimum billing period - Below DS3=one month,  YSTEMS  ronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st nued: The electronic service ordering charge currently contained in this rate exhibit ituded: CLEC-1 may elect either the state specific Commission ordered rates for the	tate speci	abov		UNCCC		21.75	21.75								
AL SUPPORT SYNOTE: (1) Electro NOTE: (1) Contin NOTE: (1) Conclu NOTE: (2) Manu	YSTEMS onic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st nued: The electronic service ordering charge currently contained in this rate exhibit i luded: CLEC-1 may elect either the state specific Commission ordered rates for the	tate speci	l abov	re=four months					32.28	10.96			38.07	38.07		
AL SUPPORT SYNOTE: (1) Electro NOTE: (1) Contin NOTE: (1) Conclu NOTE: (2) Manu	YSTEMS onic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st nued: The electronic service ordering charge currently contained in this rate exhibit i luded: CLEC-1 may elect either the state specific Commission ordered rates for the	tate speci	abov	e-iour monuis												
NOTE: (1) Electro NOTE: (1) Contin NOTE: (1) Conclu NOTE: (2) Manu	ronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the st nued: The electronic service ordering charge currently contained in this rate exhibit in luded: CLEC-1 may elect either the state specific Commission ordered rates for the															
NOTE: (1) Contin NOTE: (1) Conclu NOTE: (2) Manu	nued: The electronic service ordering charge currently contained in this rate exhibit i luded: CLEC-1 may elect either the state specific Commission ordered rates for the		<u> </u>	I												
NOTE: (1) Conclu NOTE: (2) Manu	luded: CLEC-1 may elect either the state specific Commission ordered rates for the	is the Be'					tate Commissi	ions								
	ual Service Order charge: disconnect in the state of Florida, to be billed on a per IS	electron	nic ser	vice ordering charge:	s, or CLEC-1 m	nay elect the region	nal electronic	service ordering	charge.							
Elec	au sorrios order charge. disconnect, in the state of Florida, to be blied off a per Ed	3R basis	₩													
	ctronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces		$\vdash$													
	gional)		Ь		SOMEC		3.5									
he "Zone" show	un in the sections for stand-alone loops or loops as part of a combination refers to G	Seograph	nically	Deaveraged UNE Zo	ones. To view (	Geographically D	eaveraged UN	IE Zone Designa	ations by Cent	tral Office, ref	er to Internet	Website:				1
ttp://www.interce	connection.bellsouth.com/become_a_clec/html/interconnection.htm		-	-			-	_	-							
LOCAL EXCHA	ANGE SWITCHING(PORTS)															L
xchange Ports			$\perp$							,						
	s n the Port Rate includes all available features in GA & TN, the desired features w	vill need	to be	ordered using retail	USOCs											
	GRADE LINE PORT RATES (RES)		ـــــ	LIEBOD	HEDDI	0.40	24.0	24.0					20.04	40.70		
	change Ports - 2-Wire Analog Line Port- Re: change Ports - 2-Wire Analog Line Port with Caller ID - Re		+	UEPSR UEPSR	UEPRL UEPRC	2.19 2.19	21.6 21.6	21.6 21.6					26.94 26.94	12.76 12.76		ļ —
	hange Ports - 2-Wire Analog Line Port outgoing only - Re			UEPSR	UEPRO	2.19	21.6	21.6					26.94	12.76		
Exch	hange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU			UEPSR	UEPAP	2.19	21.6	21.6					26.94	12.76		
Sub:	sequent Activity		₩	UEPSR	USASC	0	0	0								-
	Available Vertical Feature		+-	UEPSR	UEPVF	3.4	0	0					26.94	12.76		-
	GRADE LINE PORT RATES (BUS)		₩	UEPSB	UEPBL	2.19	21.6	21.6					26.94	12.76		-
	change Ports - 2-Wire Analog Line Port without Caller ID - Bi change Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484		+	UEFSB	UEFBL	2.19	21.0	21.0					20.94	12.76		
ID -	Bus.			UEPSB	UEPBC	2.19	21.6	21.6					26.94	12.76		
	change Ports - 2-Wire Analog Line Port outgoing only - Bu		Ļ—	UEPSB	UEPBO	2.19	21.6	21.6					26.94	12.76		
	nange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B asequent Activity		-	UEPSB UEPSB	UEPB1 USASC	2.19	21.6	21.6					26.94	12.76		
EATURES																
	Available Vertical Feature RT RATES (DID & PBX)		₩	UEPSB	UEPVF	3.4	0	0					26.94	12.76		
	hange Ports - 2-Wire DID Port		+	UEPEX	UEPP2	12.36	108.78	84.6					26.94	12.76		-
Excl	hange Ports - DDITS Port - 4-Wire DS1 Port with DID capabili			UEPDD	UEPDD	123.65	143.53	82.68					19.99	19.99	19.99	
Exch	change Ports - 2-Wire ISDN Port (See Notes below			UEPTX UEPSX	U1PMA	24.5	117.59	117.59					55.3	55.3		
	Features Offerec ssion/usage charges associated with POTS circuit switched usage will also apply to c	circuit sw	vitchec	UEPTX UEPSX d voice and/or circuit	UEPVF switched data to	3.4 ransmission by B-	O Channels assi	0 ociated with 2-w	ire ISDN ports							
NOTE: Access to	to B Channel or D Channel Packet capabilities will be available only through BFR/Ne			equest Process. Rat							Request Pro	cess.				
	hange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0	0	0								
	change Ports - 4-Wire ISDN DS1 Por Vire VG Unbundled 2-Way PBX Trunk - Re		₩	UEPEX UEPSE	UEPEX UEPRD	179.75 2.18	241.63 21.6	241.63 21.6					53.89 26.94	53.89 12.76		-
2-W	/ire VG Unbundled 2-Way PBX Trunk - Re /ire VG Line Side Unbundled 2-Way PBX Trunk - Bu		<del>                                     </del>	UEPSP	UEPPC	2.18	21.6	21.6					26.94	12.76		
2-W	/ire VG Line Side Unbundled Outward PBX Trunk - Bu			UEPSP	UEPPO	2.18	21.6	21.6					26.94	12.76		
	/ire VG Line Side Unbundled Incoming PBX Trunk - Bu		<u> </u>	UEPSP	UEPP1	2.18	21.6	21.6					26.94	12.76		
	/ire Analog Long Distance Terminal PBX Trunk - Bu /ire Voice Unbundled PBX LD Terminal Port		₩	UEPSP UEPSP	UEPLD UEPLD	2.18 2.18	21.6 21.6	21.6 21.6					26.94 26.94	12.76 12.76		-
2-W			<del>                                     </del>	UEPSP	UEPXA	2.18	21.6	21.6					26.94	12.76		
2-W 2-W	/ire Vice Unbundled 2-Way PBX Usage Po			UEPSP	UEPXB	2.18	21.6	21.6					26.94	12.76		
2-W 2-W 2-W 2-W	/ire Vice Unbundled 2-Way PBX Usage Po /ire Voice Unbundled PBX Toll Terminal Hotel Por				UEPXC	2.18	21.6	21.6 21.6					26.94	12.76		L
2-W 2-W 2-W 2-W 2-W	/ire Vice Unbundled 2-Way PBX Usage Po /ire Voice Unbundled PBX Toll Terminal Hotel Por /ire Voice Unbundled PBX LD DDD Terminals Po			UEPSP		0.40			1				26.94	12.76		1
2-W 2-W 2-W 2-W 2-W 2-W	Jire Vice Unbundled 2-Way PBX Usage Po Jire Voice Unbundled PBX Toll Terminal Hotel Por Jire Voice Unbundled PBX LD DDD Terminals Po Jire Voice Unbundled PBX LD Terminal Switchboard Pc			UEPSP	UEPXD	2.18	21.6									<del>                                     </del>
2-W 2-W 2-W 2-W 2-W 2-W	/ire Vice Unbundled 2-Way PBX Usage Po /ire Voice Unbundled PBX Toll Terminal Hotel Por /ire Voice Unbundled PBX LD DDD Terminals Po					2.18 2.18	21.6	21.6					26.94	12.76		
2-W 2-W 2-W 2-W 2-W 2-W	//ire Vice Unbundled 2-Way PBX Usage Po //ire Voice Unbundled PBX FDI Terminal Hotel Por //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD Terminal Switchboard Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled PSX LD Terminal Switchboard IDD Capable Pc			UEPSP UEPSP	UEPXD UEPXE UEPXL	2.18	21.6 21.6	21.6 21.6					26.94 26.94	12.76 12.76		
2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Irie Vice Unbundled 2-Way PBX Usage Po //ire Voice Unbundled PBX Toll Terminal Hotel Por //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD Terminal Switchboard Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P			UEPSP UEPSP	UEPXD UEPXE		21.6	21.6					26.94	12.76		
2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Irie Vice Unbundled 2-Way PBX Usage Po //ire Voice Unbundled PBX Toll Terminal Hotel Por //ire Voice Unbundled PBX Toll Terminals Po //ire Voice Unbundled PBX LD DDT Terminals Po //ire Voice Unbundled PBX LD Terminal Switchboard Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling t			UEPSP UEPSP	UEPXD UEPXE UEPXL	2.18	21.6 21.6	21.6 21.6					26.94 26.94	12.76 12.76		
2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Jire Vice Unbundled 2-Way PBX Usage Po //ire Voice Unbundled PBX Toll Terminal Hotel Por //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD Terminal Switchboard Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pr //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling t //ire Voice Unbundled 1-Way Outgoing PBX Measured Pr			UEPSP UEPSP UEPSP UEPSP UEPSP	UEPXD UEPXE UEPXL UEPXM UEPXO UEPXS	2.18 2.18 2.18 2.18	21.6 21.6 21.6 21.6 21.6	21.6 21.6 21.6 21.6 21.6					26.94 26.94 26.94	12.76 12.76 12.76		
2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Irie Vice Unbundled 2-Way PBX Usage Po //ire Voice Unbundled PBX Toll Terminal Hotel Por //ire Voice Unbundled PBX Toll Terminals Po //ire Voice Unbundled PBX LD DDT Terminals Po //ire Voice Unbundled PBX LD Terminal Switchboard Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling t			UEPSP UEPSP UEPSP UEPSP	UEPXD UEPXE UEPXL UEPXM UEPXO	2.18 2.18 2.18	21.6 21.6 21.6 21.6	21.6 21.6 21.6 21.6					26.94 26.94 26.94 26.94	12.76 12.76 12.76		
2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Irie Vice Unbundled 2-Way PBX Usage Po //ire Voice Unbundled PBX Toll Terminal Hotel Por //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD Terminal Switchboard Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling t //ire Voice Unbundled 1-Way Outgoing PBX Measured Pc //ire Voice Unbundled 1-Way Outgoing PBX Measured Pc //ire Voice Unbundled 1-Way Outgoing PBX Measured Pc			UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP	UEPXD UEPXL UEPXM UEPXO UEPXS USASC	2.18 2.18 2.18 2.18 0	21.6 21.6 21.6 21.6 21.6 0	21.6 21.6 21.6 21.6 21.6 0					26.94 26.94 26.94 26.94 26.94	12.76 12.76 12.76 12.76 12.76		
2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W 2-W	Jire Vice Unbundled 2-Way PBX Usage Po //ire Voice Unbundled PBX Toll Terminal Hotel Por //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD DDD Terminals Po //ire Voice Unbundled PBX LD Terminal Switchboard Pc //ire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pr //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P //ire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P //ire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling t //ire Voice Unbundled 1-Way Outgoing PBX Measured Pr			UEPSP UEPSP UEPSP UEPSP UEPSP	UEPXD UEPXE UEPXL UEPXM UEPXO UEPXS	2.18 2.18 2.18 2.18	21.6 21.6 21.6 21.6 21.6	21.6 21.6 21.6 21.6 21.6					26.94 26.94 26.94 26.94	12.76 12.76 12.76		

	cess to B Channel or D Channel Packet capabilities will be available only through BFR/New	√ Busines	ss Rec	quest Process. Ra	ites for the pack	et capabilities	will be determine	ed via the Bona	Fide Reques	/New Busine	ss Request P	rocess.			
D 1 00 41 014	WITCHING BORT HOLOT														
D LOCAL SW	WITCHING, PORT USAGE	$\longrightarrow$													
End Office (	Switching (Port Usage)	-													
	End Office Switching Function, Per MOI	-				0.0015									
	End Office Trunk Port - Shared, Per MOU	-+				0.00023				-					
	End office Trank For Shared, For Mot	-				0.00020									
Tandem Sw	witching (Port Usage) (Local or Access Tandem)														
	Tandem Switching Function Per MOl					0.0006									
	Tandem Trunk Port - Shared, Per MOI					0.0003									
Common Tr	ransport														
	Common Transport - Per Mile, Per MOI					0.00001									
	Common Transport - Facilities Termination Per MO					0.00034									
D PORT/LOC	OP COMBINATIONS - COST BASED RATES														
	Rates are applied where BellSouth is required by FCC and/or State Commission rule to p							D . E							
End Office a	nall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same and Tandem Switching Usage and Common Transport Usage rates in the Port section of t	manner this rate	as the	ey are applied to the	ne Stand-Alone t	John John Look	ork elements ov	cont for LINE C	oin Port/Loon	Combination	NC.				
												<del></del>			
For Georgia	a and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comb	oined and	d Not (	Currently Combine	ed Combos and	he first and a	ditional Port nor	nrecurring charg	es apply to N	ot Currently	Combined Co	mbos. For 0	Currently Combine	ed Combos in C	SA, TN
and all other	er states, the nonrecurring charges shall be those identified in the Nonrecurring - Currently C	Combine	ed secti	ions.	1		1			1	1		1	1	
2 MIDE VC	ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	$\longrightarrow$			+		+	1	1		1	-	-	+	
z-wike vol	IGE GRADE LOUP WITH Z-WIKE LINE PORT (KES)	$\longrightarrow$			+		+	+			1		-	+	
LINE Dort!	.oop Combination Rates	$\longrightarrow$			+		+	+			1		-	+	
UNE PUICLE	2-Wire VG Loop/Port Combo - Statewid	$\rightarrow$	SW			16.46	-	1		-	+	-		+	
	2-vviile vo Loopir on Collibo - Statewid	-+	οW			10.40					+			+	
UNE Loop R	Rates	-+							1	1	1	_		1	
	2-Wire Voice Grade Loop (SL1) - Statewid	-+	SW	UEPRX	UEPLX	14.18									
	2 Trie Voice Clade 2009 (G21) Clateria			OLI TOX	OL: LX	11.10									
2-Wire Voice	ce Grade Line Port Rates (Res)														
	2-Wire voice unbundled port - residenc	-		UEPRX	UEPRL	2.28							40.18	9.45	
	2-Wire voice unbundled port with Caller ID - re			UEPRX	UEPRC	2.28							40.18	9.45	
	2-Wire voice unbundled port outgoing only - re			UEPRX	UEPRO	2.28							40.18	9.45	
	2-Wire voice unbundles res, low usage line port with Caller ID (LUI			UEPRX	UEPAP	2.28							40.18	9.45	
FEATURES															
	All Features Offered			UEPRX	UEPVF	3.4	0	0					40.18	9.45	
	MARIA DODTARILITY														
LOCAL NUM	MBER PORTABILITY Local Number Portability (1 per por			UEPRX	LNPCX	0.35									
	Local Number Portability (1 per por	-+		UEPRX	LINPUX	0.35					-				
NONDECLID	RRING CHARGES (NRCs) - CURRENTLY COMBINED	-+													
NONKECOK	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	-+		UEPRX	USAC2		2.77	0.4		-			40.18	9.45	
	2 Wile Voice Grade Edop / Eine Fort Goribination - Goriversion - Gwitch as	-+		OLITA	OUNUZ		2.11	0.4					40.10	3.43	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char			UEPRX	USACC		2.77	0.4					40.18	9.45	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database			OL: TO	00/100		2	0.1					10.10	0.10	
	Update						1.42						10.27		
ADDITIONAL															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ			UEPRX	USAS2	0	0	0			1				
														1	
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)										1			1	
											1			1	
	oop Combination Rates										1	_	_	1	
	2-Wire VG Loop/Port Combo - Statewid	$\longrightarrow$	SW		1	16.46	1	+	1			_		+	
LINE L 7	Potes	$\longrightarrow$			+		+	1	1		1	-	-	+	
UNE Loop R		$\longrightarrow$	sw	UEPBX	UEPLX	14.18		+	1		1		+	+	
<del>                                     </del>	2-Wire Voice Grade Loop (SL1) - Statewid	$\rightarrow$	SW	UEPBX	UEPLA	14.18	-	1		-	+	-		+	
	ce Grade Line Port (Bus)	$\rightarrow$	-+		+ -		+	1			+	-	-	+	
2-Wire Valor	2-Wire voice unbundled port without Caller ID - bu	-+	-+	UEPBX	UEPBL	2.28	+	1	1				40.18	9.45	
		-+	-	UEPBX	UEPBC	2.28	1	1			+		40.18	9.45	
	12-Wire voice unbundled port with Caller + E484 III - bi		-+	UEPBX	UEPBO	2.28	+	1	1		1		40.18	9.45	
	2-Wire voice unbundled port with Caller + E484 ID - bt 2-Wire voice unbundled port outgoing only - bt				UPEB1	2.28						-		9.45	
	2-Wire voice unbundled port outgoing only - bu	$\rightarrow$		UEPBX	UPEDI		+						40.18	9.45	
	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B	#		UEPBX	UPEBI								40.18	9.45	
	2-Wire voice unbundled port outgoing only - bt 2-Wire voice unbundled incoming only port with Caller ID - B  MBER PORTABILITY	#											40.18	9.45	
	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B			UEPBX	LNPCX	0.35							40.18	9.45	
LOCAL NUM	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B  MBER PORTABILITY  Local Number Portability (1 per por					0.35							40.18	9.45	
	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B  MBER PORTABILITY  Local Number Portability (1 per por			UEPBX	LNPCX										
LOCAL NUN	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B  MBER PORTABILITY  Local Number Portability (1 per por					0.35	0	0					40.18	9.45	
LOCAL NUM	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B  MBER PORTABILITY  Local Number Portability (1 per por			UEPBX	LNPCX		0	0							
LOCAL NUM	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B  MBER PORTABILITY Local Number Portability (1 per por  All Features Offerec  RRING CHARGES (NRCs) - CURRENTLY COMBINED			UEPBX	LNPCX								40.18	9.45	
LOCAL NUM	2-Wire voice unbundled port outgoing only - bi 2-Wire voice unbundled incoming only port with Caller ID - B  MBER PORTABILITY  Local Number Portability (1 per por			UEPBX	LNPCX		0 2.77	0							

Page 11 of 18 Version 2Q01: 08/30/01

	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database											
	Update				<b></b>	1.42			10.27			
ADDITIONAL	L NRCs			+	i							1
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX	USAS2					40.18	9.45		
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			+		<del>                                     </del>						1
	, ,											
UNE Port/Lo	pop Combination Rates				10.10							
	2-Wire VG Loop/Port Combo - Statewid	SW			16.46							-
UNE Loop R	Rates			+	i							H
	2-Wire Voice Grade Loop (SL 1) - Statewid	SW	UEPRG	UEPLX	14.18							
2-Wire Voice	e Grade Line Port Rates (RES - PBX)											-
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	2.28				40.18	9.45		T
	BER PORTABILITY Local Number Portability (1 per por		UEPRG	LNPCP	3.5							┢
			OLI IKO	EINI OI	0.0							H
FEATURES												
	All Features Offered		UEPRG	UEPVF	3.4	0	0		40.18	9.45		-
NONRECURI	RING CHARGES (NRCs) - CURRENTLY COMBINED			+	i							
										_		
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As     2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with		UEPRG	USAC2		2.77	0.4		40.18	9.45		-
l l	Change		UEPRG	USACC	i I	2.77	0.4		40.18	9.45		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database											
	Update		1	+		1.42		1	10.27			H
ADDITIONAL												Ļ
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USAS2	0	0 14.64	0 14.64		40.00	40.00		
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Gro			+		14.64	14.64		19.99	19.99	19.99	-
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)											
INE Dort/Lo	pop Combination Rates											-
	2-Wire VG Loop/Port Combo - Statewid	sw		-	16.46							1
UNE Loop R	2-Wire Voice Grade Loop (SL 1) - Statewid	SW	UEPPX	UEPLX	14.18							
	2-wire voice Grade Loop (SL 1) - Statewic	SW	UEPPX	UEPLX	14.18							-
	e Grade Line Port Rates (BUS - PBX)											
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bi		UEPPX	UEPPC	2.28				40.18	9.45		
	Line Side Unbundled Outward PBX Trunk Port - Bu Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX UEPPX	UEPPO UEPP1	2.28 2.28				40.18 40.18	9.45 9.45	<b> </b>	-
	2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	2.28				40.18	9.45	<del>                                     </del>	t
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	2.28				40.18	9.45		
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	2.28				40.18	9.45		-
	2-Wire Voice Unbundled PBX LD DDD Terminals Po 2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX UEPPX	UEPXC UEPXD	2.28	<del>                                     </del>			40.18 40.18	9.45 9.45		+
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc		UEPPX	UEPXE	2.28				40.18	9.45		L
	O Miles Veira Habrardad O Marc DDV Hatal/Harrital Farances Administration Co. 11		HEDDY	LIEDVI	0.00		·		40.40	0.45		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX UEPPX	UEPXL UEPXM	2.28 2.28	<del>                                     </del>		1	40.18 40.18	9.45 9.45	$\vdash$	+
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling											Г
	Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX UEPPX	UEPXO UEPXS	2.28 2.28	1			40.18 40.18	9.45		_
ľ	z-write voice Undundled 1-way Outgoing PBX Measured PC	_	UEPPX	UEPXS	2.28				40.18	9.45		H
	IBER PORTABILITY											
	Local Number Portability (1 per por		UEPPX	LNPCP	3.15							F
FEATURES				+		<del>                                     </del>						H
	All Features Offered		UEPPX	UEPVF	3.4	0	0		40.18	9.45		
HONDEST	DING CHARGES (AIDS-), CURRENTLY COMPANIES		1									F
NUNRECURI	RING CHARGES (NRCs) - CURRENTLY COMBINED		+	+								$\vdash$
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is		UEPPX	USAC2	i I	2.77	0.4		40.18	9.45		
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with											
	Change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database	_	UEPPX	USACC		2.77	0.4		40.18	9.45		┢
	2-wire voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update				i I	1.42			10.27	-		
	L NRCs  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPPX	USAS2	0	0	0	<del>                                     </del>				-
ADDITIONAL				USMSZ	U				 		<del></del>	+
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Gro				ļ	14.64	14.64		19.99	19.99	19.99	ð
						14.64	14.64		19.99	19.99	19.99	_

	2-Wire VG Coin Port/Loop Combo – Statewide	+	+	<del>                                     </del>	16.8									+
JNE Loop R		+	+		10.0									+
THE LOOP IT		-	+											t
	2-Wire Voice Grade Loop (SL1) - Statewid		UEPCO	UEPLX	14.18									Ť
														Τ
	e Grade Line Ports (COIN)													
	2-Wire Coin 2-Way without Operator Screening and without Blocking (NC)		UEPCO	UEPND	2.62						40.18	9.45		$\perp$
	2-Wire Coin 2-Way with Operator Screening (NC)		UEPCO	UEPNC	2.62						40.18	9.45	<b> </b>	+
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (NC, TN)		UEPCO	UEPRP							40.40	9.45	'	
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (NC)	-	UEPCO	UEPRP	2.62 2.62						40.18 40.18	9.45	<b></b> '	+
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (NC) 2-Wire Coin 2-Way with Operator Screening: 900 Blocking: 900/976, 1+DDD, 011+, and	$+\!-$	UEPCO	UEPINB	2.02						40.18	9.45	<b> </b>	+
	Local (NC, TN)		UEPCO	UEPCA	2.62						40.18	9.45	'	
	2-Wire Coin Outward with Operator Screening and 011 Blocking (NC)	+	UEPCO	UEPNE	2.62						40.18	9.45	<b>—</b>	+
	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, a	+	02100	OLITIC	2.02						40.10	3.40		+
	Local (NC)		UEPCO	UEPCL	2.62						40.18	9.45	1 '	
	2-Wire 2-Way Smartline with 900/976 (all states except LA)		UEPCO	UEPCK	2.62						40.18	9.45		t
	2-Wire Coin Outward Smartline with 900/976 (all states except LA)		UEPCO	UEPCR	2.62						40.18	9.45		Ť
	L UNE COIN PORT/LOOP (RC)													T
	UNE Coin Port/Loop Combo Usage (Flat Rate		UEPCO	URECU	3.7	0	0							Ť
														Т
OCAL NUM	MBER PORTABILITY													Т
	Local Number Portability (1 per por		UEPCO	LNPCX	0.35	-	1							
														Ţ
EATURES													L	1
	All Features Offered	$-\!$	UEPCO	UEPVF	3.4	(	0				26.94	12.76	<u> </u>	+
		$-\!$			<b></b>								<b></b>	+
	RRING CHARGES - CURRENTLY COMBINED	$-\!$	LIEBOO	1101.05	<b></b>	0.77	<b>—</b>				40.40	0.45	<u> </u>	+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	+	UEPCO	USAC2	<del></del>	2.77	0.4				40.18	9.45	<b></b> '	+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char		UEPCO	USACC	1	2.77	0.4				40.18	9.45	1	1
	z-write voice Grade Loop / Line Port Combination - Conversion - Switch with char	+	UEPCU	USACC		2.11	0.4				40.18	9.45		+
ADDITIONAL	I NPCs	+-	+	<del></del>	<del></del>									+
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ	+	UEPCO	USAS2	$\longrightarrow$	0	0				40.18	9.45	<b>—</b>	+
	2 Wile Voice Grade Edop/Eine Fort Combination Cabacquent Activ	-	OLI GO	OUNUE							40.10	3.40		+
-WIRE VOIC	CE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT	-	+											t
	OL GIOLOGI DOG GILL IIII IL IIII IL DID I II GIIVI GIVI	+	+											+
JNE Port/Lo	pop Combination Rates		+											t
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - Statewid	sw	+		31.07									t
														T
JNE Loop R	Rates													J
	2-Wire Analog Voice Grade Loop - (SL2) - Statewic	sw			19.5	142.97	106.56			19.99				Ι
						-								Γ
JNE Port Ra				oxdot										Ŧ.
	Exchange Ports - 2-Wire DID Por		UEPPX	UEPD1	12.36						19.99	19.99	19.99	1
													<b></b>	+
	RING CHARGES - CURRENTLY COMBINED				<b></b>								<b></b>	+
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as		UEPPX	USAC1	<b></b>	13.26	8.39				19.99	19.99	19.99	+
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable		HEDDY	110440	1	40.00	0.00				40.00	40.00	40.00	
	Changes	-	UEPPX	USA1C		13.26	8.39				19.99	19.99	19.99	+
ADDITIONAL	I NDO-	-	+										<b></b>	+
	2-Wire DID Subsequent Activity - Add Trunks, Per Trun	+-	UEPPX	USAS1		53.49	1				19.99	19.99	19.99	+
	2-vviile DID Subsequent Activity - Add Trunks, Fel Trun	+-	UEFFA	USASI	<del></del>	33.49					19.99	19.99	19.99	+
elenhone N	Number/Trunk Group Establisment Charges	+-	+	<del></del>	+									+
	DID Trunk Termination (One Per Port	+-	UEPPX	NDT	0	0	0				19.99	19.99	19.99	9
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbe	_	UEPPX	NDZ	0	0	0				19.99	19.99	19.99	
	Additional DID Numbers for each Group of 20 DID Numbe	-	UEPPX	ND4	0	0	Ö				19.99	19.99	19.99	
	DID Numbers, Non- consecutive DID Numbers , Per Numbe		UEPPX	ND5	0	0	0			19.99				T
	Reserve Non-Consecutive DID number		UEPPX	ND6	0	0	0			19.99				Ι
	Reserve DID Numbers		UEPPX	NDV	0	0	0			19.99				Ι
						-								ſ
	MBER PORTABILITY													┸
	Local Number Portability (1 per por		UEPPX	LNPCP	3.15									Ţ
				$lue{\Box}$										Ŧ
-WIRE ISDN	N DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT												L	4
													L	4
JNE Port/Lo	pop Combination Rates	$-\!$			<b></b>		1						<u> </u>	+
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - Statewi	sw	UEPPB UEPPR		44.49									L
	Rates	$\pm$												t
JNE Loop R			UEPPB UEPPR	USL2X	20.12	325.91	251.31			19.99				Γ
				. UOLZA	20.12	323.91	201.01			19.99			<b> </b>	+
	2-Wire ISDN Digital Grade Loop - Statewid	SW	OEITE OEITE					l I	1				1 .	
		SW											<del>                                     </del>	+
JNE Port Ra		SW	UEPPB UEPPR		24.37						19.99	19.99	19.99	,

Page 13 of 18 Version 2Q01: 08/30/01

	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Convers	-+	UEPI	B UEPPR	USACB	0	174.35	174.35				19.99	19.99	19.99	19.99
DDITIONAL	L NRCs														
OCAL NUM	MBER PORTABILITY	+													
	Local Number Portability (1 per por		UEPF	B UEPPF	LNPCX	0.35	0	0							
-CHANNEL	USER PROFILE ACCESS:	-													
	CVS/CSD (DMS/5ESS)		UEPF	B UEPPF	U1UCA	0	0	0							
	CVS (EWSD)	-+	UEPF			0	0	0							
	CSD			B UEPPR		0	0	0							
CHANNEL	AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)	_													
-CHANNEL	AREA FLUS USER PROFILE ACCESS. (AL,R1,LA,WS SC,WS, & IN)	_													
SER TERM	MINAL PROFILE														
	User Terminal Profile (EWSD only)		UEPF	B UEPPF	U1UMA	0	0	0							
	FEATURES														
ERTICAL F	FEATURES	-													
	All Vertical Features - One per Channel B User Profile		UEPF	B UEPPF	UEPVF	3.4	0	0							
ITEROFFIC	CE CHANNEL MILEAGE	-+													
	Interoffice Channel mileage each, including first mile and facilities termination	+		B UEPPR		17.42	137.48	52.58				19.99	19.99	19.99	19
	Interoffice Channel mileage each, additional mile		UEPF	B UEPPR	M1GNM	0.0282	0	0			0				
-WIRE DS1	DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT	-+													
NE Port/Lo	pop Combination Rates  4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - Statewic		w	JEPPP		241.72									
				J		211112									
NE Loop R	4-Wire DS1 Digital Loop - UNE Zone	-+	3	JEPPP	USL4P										
				JEITT	COLTI										
NE Port Ra	ate Exchange Ports - 4-Wire ISDN DS1 Por			JEPPP	UEPPP	179.01						19.99	19.99	19.99	19.99
				JEFFF	OLFFF	179.01						15.55	15.55	15.55	15.55
ONRECUR	RING CHARGES - CURRENTLY COMBINED														
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion -Switch-as-it			JEPPP	USACP	0	481.51	481.51				19.99	19.99	19.99	19.99
DDITIONAL	I ADCo														
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Subsequent Inward/2-Way Te	_													
	Nos - NC Only  4-Wire DS1 Loop/4-Wire ISDN Digital Trunk Port - Subsequent Activity Outward tel nos.			JEPPP	PR7TG		1.17	1.17				19.99	19.99	19.99	19.99
	(NC only)			JEPPP	PR7TP		28.17	28.17				19.99	19.99	19.99	19.99
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos			IEDDD.	DD====		50.00								
	Above Std Allowance	-		JEPPP	PR7ZT		56.33	56.33				19.99	19.99	19.99	19.99
	MBER PORTABILITY														
	Local Number Portability (1 per por			JEPPP	LNPCN	1.75									
	Local Number Portability (1 per por			JEPPP	LNPCN	1.75									
	(Provsioning Only)	$\equiv$					0	0							
ITERFACE	(Provsioning Only) Voice/Data			JEPPP	PR71V	0	0	0							
ITERFACE	(Provsioning Only)						0 0 0	0 0 0							
ITERFACE	E (Provsioning Only) Voice/Data Digital Data Inward Data			JEPPP JEPPP	PR71V PR71D	0 0	0	0							
ITERFACE	(Provsioning Only) Voice/Data Digital Data Inward Data  ittional "B" Channel			JEPPP JEPPP JEPPP	PR71V PR71D PR71E	0 0 0	0	0				40.00	40.00		
ITERFACE	(Provsioning Only)  (Voice/Data  Digital Data  Inward Data  Itional "B" Channel  New or Additional - Voice/Data B Channel			JEPPP JEPPP JEPPP	PR71V PR71D PR71E	0 0 0	0 0 36.92	0				19.99	19.99	19.99	
ITERFACE	(Provisioning Only) Voice/Data Digital Data Inward Data Inward Data  Ititional "B" Channel New or Additional - Voice/Data B Channel New or Additional - Digital Data B Channel			JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR7BV PR7BF	0 0 0	0 0 36.92 36.92	0				19.99	19.99	19.99	1
TERFACE	(Provsioning Only) Voice/Data Digital Data Ilmward Data Itional "B" Channel New or Additional - Voice/Data B Channel New or Additional - Digital Data B Channel New or Additional Inward Data B Channel			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR7BV PR7BF PR7BD	0 0 0	0 0 36.92 36.92 36.92	0				19.99 19.99	19.99 19.99	19.99 19.99	1:
ITERFACE	(Provisioning Only) Voice/Data Digital Data Inward Data Inward Data  Ititional "B" Channel New or Additional - Voice/Data B Channel New or Additional - Digital Data B Channel			JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR7BV PR7BF	0 0 0	0 0 36.92 36.92	0				19.99	19.99	19.99	1 1
ITERFACE	(Provisioning Only)  Voice/Data  Digital Data  Inward Data  itional "B" Channel  New or Additional - Voice/Data B Channel  New or Additional - Digital Data B Channel  New or Additional Inward Data B Channel  New or Additional Useage Sensitive Voice Data B Channel  New or Additional Useage Sensitive Digital Data B Channel			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR7BV PR7BF PR7BD PR7BS	0 0 0 0	0 0 36.92 36.92 36.92 36.92	0				19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	1 1
EW OF Addi	(Provsioning Only)  Voice/Data  Digital Data  Inward Data  Ititional "B" Channel  New or Additional - Voice/Data B Channel  New or Additional I- Digital Data B Channel  New or Additional Inward Data B Channel  New or Additional Useage Sensitive Voice Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  New or Additional Useage Sensitive Digital Data B Channel			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR7BV PR7BF PR7BD PR7BS PR7BU	0 0 0 0 0 0 0 0	36.92 36.92 36.92 36.92 36.92 36.92	0				19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	1 1
ew or Addi	(Provisioning Only)  Voice/Data  Digital Data  Inward Data  itional "B" Channel  New or Additional - Voice/Data B Channel  New or Additional - Digital Data B Channel  New or Additional Inward Data B Channel  New or Additional Useage Sensitive Voice Data B Channel  New or Additional Useage Sensitive Digital Data B Channel			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR7BV PR7BF PR7BD PR7BS	0 0 0 0	0 0 36.92 36.92 36.92 36.92	0				19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	1
ew or Addi	(Provsioning Only)  Voice/Data  Digital Data  Ilmward Data  Itional "B" Channel  New or Additional - Voice/Data B Channel New or Additional - Digital Data B Channel New or Additional Inward Data B Channel New or Additional Inward Data B Channel New or Additional Useage Sensitive Voice Data B Channel New or Additional Useage Sensitive Digital Data B Channel New or Additional Useage Sensitive Digital Data B Channel New or Additional Useage Sensitive Digital Data B Channel			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR78V PR78F PR78D PR78S PR78U	0 0 0 0 0 0 0 0	0 0 36.92 36.92 36.92 36.92 36.92	0				19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	1
ew or Addi	(Provsioning Only)  Voice/Data  Digital Data  Inward Data  itional "B" Channel  New or Additional - Voice/Data B Channel New or Additional - Digital Data B Channel New or Additional - Digital Data B Channel New or Additional Inward Data B Channel New or Additional Useage Sensitive Voice Data B Channel New or Additional Useage Sensitive Digital Data B Channel Sensor Additional Useage Sensitive Digital Data B Channel New or Additional Useage Sensitive Digital Data B Channel  Inward Outward  Outward  Two-way			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR78V PR7BF PR7BD PR7BS PR7BU PR7C1 PR7C1	0 0 0 0 0 0 0 0 0	0 0 36.92 36.92 36.92 36.92 36.92	0 0				19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	1
ew or Addi	(Provsioning Only)  Voice/Data  Digital Data  Inward Data  Itional "B" Channel  New or Additional - Voice/Data B Channel  New or Additional - Digital Data B Channel  New or Additional Inward Data B Channel  New or Additional Useage Sensitive Voice Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  S  Inward  Outward  Two-way  hannel Mileage			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR78V PR78F PR78D PR78S PR78U PR761 PR7C1	0 0 0 0 0 0 0 0 0 0	0 0 36.92 36.92 36.92 36.92 0 0	0 0 0				19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	1 1 1 1
ew or Addi	(Provsioning Only)  Voice/Data  Digital Data  Ilmward Data  Itional "B" Channel  New or Additional - Voice/Data B Channel  New or Additional - Digital Data B Channel  New or Additional Data B Channel  New or Additional Inward Data B Channel  New or Additional Inward Data B Channel  New or Additional Useage Sensitive Voice Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  Two-way  Inward  Outward  Two-way  Inannel Mileage  Fixed Each Including First Mile			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR78F PR78F PR78D PR78D PR78D PR78U PR7C1 PR7C0 PR7CC	0 0 0 0 0 0 0 0 0 0 0	0 0 36.92 36.92 36.92 36.92 36.92	0 0	0			19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	1: 1: 1: 1:
ew or Addi	(Provsioning Only)  Voice/Data  Digital Data  Inward Data  Itional "B" Channel  New or Additional - Voice/Data B Channel  New or Additional - Digital Data B Channel  New or Additional Inward Data B Channel  New or Additional Useage Sensitive Voice Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  New or Additional Useage Sensitive Digital Data B Channel  S  Inward  Outward  Two-way  hannel Mileage			JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP JEPPP	PR71V PR71D PR71E PR78V PR78F PR78D PR78S PR78U PR761 PR7C1	0 0 0 0 0 0 0 0 0 0	0 0 36.92 36.92 36.92 36.92 0 0	0 0 0	0			19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	19 19 19 19 19 19

	oop Combination Rates  4W DS1 Digital Loop/4W DDITS Trunk Port - Statewid	- +	SW	UEPDC	1	186.23	1	1					19.99	19.99	19.99	
	The Don Brightan Ecopy TVV DDITO Hunk Fore Foliationia		3**	OLI DO	<del>                                     </del>	100.23						1	15.55	15.55	10.00	
JNE Loop R	Rates				1							1	1			
	4-Wire DS1 Digital Loop - Statewid		SW	UEPDC	USLDC	62.71	714.84	482.62					19.99	19.99	19.99	
JNE Port Ra																
	4-Wire DDITS Digital Trunk Por			UEPDC	UDD1T	123.65	1						19.99	19.99	19.99	
	·															
ONRECUR	RRING CHARGES - CURRENTLY COMBINED															
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-			UEPDC	USAC4		288.86	133.87					19.99	19.99	19.99	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with DS1															
	Changes			UEPDC	USAWA		288.86	133.37					19.99	19.99	19.99	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with															
	Change - Trunk			UEPDC	USAWB		288.86	133.37					19.99	19.99	19.99	
ADDITIONA																
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Service Activity Per Service			LIEDDO	110404		407.00	407.00								
	Order			UEPDC	USAS4		127.63	127.63								
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel Activation/Chan - 2-Way Trunl			UEPDC	UDTTA		28.81	28.81					19.99	19.99	19.99	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-			UEPDC	UDITA		28.81	28.81					19.99	19.99	19.99	
	Way Outward Trunk			UEPDC	UDTTB		28.81	28.81					19.99	19.99	19.99	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan Inward			OLFDC	ODITO		20.01	20.01					15.55	19.55	15.55	
	Trunk Wout DID			UEPDC	UDTTC		28.81	28.81					19.99	19.99	19.99	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -			OLI DO	00110		20.01	20.01				1	13.33	15.55	10.00	
	Inward Trunk with DIE			UEPDC	UDTTD		28.81	28.81					19.99	19.99	19.99	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way			22,00								1	. 5.00	. 5.00	. 5.00	
	DID w User Trans			UEPDC	UDTTE		28.81	28.81					19.99	19.99	19.99	
SIPOLAR 8	ZERO SUBSTITUTION					1										
	B8ZS -Superframe Format			UEPDC	CCOSF		0	615					19.99	19.99	19.99	
	B8ZS - Extended Superframe Forma			UEPDC	CCOEF		0	615					19.99	19.99	19.99	
Alternate Ma	ark Inversion															
	AMI -Superframe Format			UEPDC	MCOSF		0	0				ļ				
	AMI - Extended SuperFrame Forma			UEPDC	MCOPO		0	0				1	1	1		
					1							-	1	1		
	N 1 7 10 5 11 10 10 10 10 10 10 10 10 10 10 10 10				1							-	1	1		
elepnone l	Number/Trunk Group Establisment Charges			UEDDO	UDTOV	-	1	1			-		+	+		
	Telephone Number for 2-Way Trunk Grou			UEPDC UEPDC	UDTGX	0					-	19.99	+	+		
	Telephone Number for 1-Way Outward Trunk Grou			UEPDC	UDTGZ	0				1		19.99	<del>                                     </del>	1		
	Telephone Number for 1-Way Inward Trunk Group Without DI						_	_			-		<del>                                     </del>	1	-	
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbe			UEPDC	NDZ ND4	0	0	0			-	19.99	+	1	-	
	DID Numbers for each Group of 20 DID Number DID Numbers, Non- consecutive DID Numbers , Per Numbe			UEPDC UEPDC	ND4 ND5	0	1	1			-	19.99	+	1	-	
				UEPDC	ND5 ND6	0	0	0			-	19.99	+	+		
	Reserve Non-Consecutive DID Nos Reserve DID Numbers	-		UEPDC	ND6 NDV	0	0	0			1	19.99	+	+		
	TOSSITO DID INUIDOR	+		OLFDO	1404	0	0	U			1	13.33	+	1		
Dedicated F		ITS Tru	ınk P	ort	1	1	1				1		1	1		
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination			UEPDC	1LNO1	71.29	217.17	163.75	0	0			19.99	19.99	19.99	
	Interoffice Channel Mileage - Additional rate per mile - 0-8 mil			UEPDC	1LNOA	0.0783	0	0	-	_		1	,			
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Terminatii			UEPDC	1LNO2	0	0	0				1				
	Interoffice Channel Mileage - Additional rate per mile - 9-25 mil			UEPDC	1LNOB	0.0783	Ö	0								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination			UEPDC	1LNO3	0	0	0	0							
	Interoffice Channel Mileage - Additional rate per mile - 25+ mil			UEPDC	1LNOC	0.0783	0	0								
	Local Number Portability, per DS0 Activate			UEPDC	LNPCP	3.15	0	0	0							
	Central Office Termininating Poir			UEPDC	CTG	0										
													1			
					1											
	1 LOOP WITH CHANNELIZATION WITH PORT				1	1	1							1		
	DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations				1		1	1								
ach Syster	m can have up to 24 combinations of rates depending on type and number of ports us	sed			1											
					1	1	1							1		
JNE DS1 Lo					1							ļ				
	4-wire DS1 Loop UNE - Statewide		SW	UEPMG	USLDC	62.71	1					19.99		1		
					1							ļ				
JNE DSO C	channelization Capacities (D4 Channel Bank Configurations)				1							ļ				
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	123.06	0	0								
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	246.12	0	0								
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	492.24	0	0								
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	738.36	0	0					1	1		
				UEPMG	VUM19	984.48	0	0								
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM20	1230.6	0	0								
	192 DS0 Channel Capacity -1 per 8 DS1s 240 DS0 Channel Capacity - 1 per 10 DS1s					1476.72	0	0				1				
				UEPMG	VUM28								1	+	-	_
	240 DS0 Channel Capacity - 1 per 10 DS1s 288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG			0	0								
	240 DS0 Channel Capacity - 1 per 10 DS1s 288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG UEPMG	VUM38	1968.96	0	0								
	240 DS0 Channel Capacity - 1 per 10 DS1s 288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG UEPMG UEPMG	VUM38 VUM40	1968.96 2461.2	0	0								
	240 DS0 Channel Capacity - 1 per 10 DS1s 288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s 576 DS0 Channel Capacity - 1 per 24 DS1s			UEPMG UEPMG UEPMG UEPMG	VUM38 VUM40 VUM57	1968.96 2461.2 2953.44	0 0	0 0								_
	240 DS0 Channel Capacity - 1 per 10 DS1s 288 DS0 Channel Capacity - 1 per 12 DS1s 384 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG UEPMG UEPMG	VUM38 VUM40	1968.96 2461.2	0 0 0	0 0 0								_

munipies or ans configuration functioning as one are const	sidered Add'l after the minimum system con		re Activations.												
NRC - Conversion (Currently Combined) with or v	without BellSouth Allowed Changes		EPMG	USAC4	0	330.61	16.64					19.99	19.99	19.99	19.99
System Additions at End User Locations Where 4-Wire DS1															
New (Not Currently Combined) In Georgia & Tennessee On			•												
NRC - 1 DS1/D4 Channel Bank - Add NRC for ea	ach Port and Assoc Feature Activation -														
New GA & TN Only		U	EPMG	VUMD4	0	743.74	326.22	149.02	17.68			19.99	19.99	19.99	19.99
Bipolar 8 Zero Substitution															
Clear Channel Capability Format, superframe - S			EPMG	CCOSF	0	0	615					19.99	19.99	19.99	19.99
Clear Channel Capability Format - Extended Sup	perframe - Subsequent Activity Only	U	EPMG	CCOEF	0	0	615					19.99	19.99	19.99	19.99
Alternate Mark Inversion (AMI)															
Superframe Format			EPMG	MCOSF	0	0	0								
Extended Superframe Format		U	EPMG	MCOPO	0	0	0								
Exchange Ports Associated with 4-Wire DS1 Loop with Cha	annelization with Port														
Exchange Ports							_								
Line Side Combination Channelized PBX Trunk P			EPPX EPPX	UEPCX	2.28	0	0	0	0		19.99				
Line Side Outward Channelized PBX Trunk Port		0.		UEPOX	2.28	0	0	0	0		19.99				
Line Side Inward Only Channelized PBX Trunk P			EPPX	UEP1X	2.28	0	0	0	0		19.99				
2-Wire Trunk Side Unbundled Channelized DID T	Trunk Port	UI	EPPX	UEPDM	13.26	0	0	0	0		19.99				
Feature Activations - Unbundled Loop Concentration	2		EDDY	40014/44	0.05	05.07	40.04		4.40			40.00	40.00		40.00
Feature (Service) Activation for each Line Side P			EPPX	1PQWM	0.65	25.27	13.34	4.15	4.12	1	1	19.99	19.99	19.99	19.99
Feature (Service) Activation for each Trunk Side	Port Terminated in D4 Bank	U	EPPX	1PQWU	0.65	77.75	18.33	58.74	11.48		1	19.99	19.99	19.99	19.99
Telephone Number/ Group Establishment Charges for DID	Service		EPPX	NDT	0	1	+			1	1	+		-	_
DID Trunk Termination (1 per Port)	CA NC & CC)			NDT	U	-	-				40.00	1	1	-	
Estab Trk Grp and Provide 1st 20 DID Nos. (FL,G	JA, NO, a 30)		EPPX EPPX	NDZ	U	0	0				19.99	1	1	-	
DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number			EPPX EPPX	ND4	0	0	•			1	19.99	+		-	_
Reserve Non-Consecutive DID Numbers - per number			EPPX EPPX	ND5 ND6	0	0	0			1	19.99	1			
Reserve DID Numbers			EPPX	NDV	0	0	0								
		U	EPPX	NDV	U	U	U								
Local Number Portability  Local Number Portability - 1 per port		110	EPPX	LNPCP	3.15	0	0								
FEATURES - Vertical and Optional		U	EFFA	LINECE	3.15	U	U								
Local Switching Features Offered with Line Side Ports Only	h.														
All Features Available	ly .	110	EPPX	UEPVF	3.4	0	0				19.99				
All realules Available		U	EFFA	UEFVF	3.4	U	U				19.99				
Market Rates shall apply where BellSouth is not required to pr These scenarios include:					3.										
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C	Combined in all of the BellSouth states excep	t as noted f	for Georgia and T	ennessee.		A or more DCC	oquivalent Pr-	20							
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Comb	Combined in all of the BellSouth states excep bined or Not Currently Combined in Zone 1 of	t as noted f	for Georgia and T	ennessee.	end users with	4 or more DS0	equivalent line	es.							
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Comb  The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. La	Combined in all of the BellSouth states excep bined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl	of as noted f f the Top 8 leans); NC	for Georgia and T B MSAS in BellSou (Greensboro-Win	ennessee. hth's region for ston Salem-Hi	end users with	te-Gastonia-Roc	k Hill); TN (Na	ashville).	cedina in lieu of th	ne Market Rat	es and rese	yes the right to	true-up the b	illing differer	nce
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Combinations that are Currently Combinations that are Currently Combinations that are Currently Combinations that are Currently Self-Self-Self-Self-Self-Self-Self-Self-	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orlanically bill the recurring and non-recurring M	of as noted f f the Top 8 leans); NC	for Georgia and T B MSAS in BellSou (Greensboro-Win	ennessee. hth's region for ston Salem-Hi	end users with	te-Gastonia-Roc	k Hill); TN (Na	ashville).	ceding in lieu of the	ne Market Rat	es and rese	ves the right to	true-up the b	illing differer	nce.
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Comb  The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. La	Combined in all of the BellSouth states excep bined or Not Currently Combined in Zone 1 of auderdale, Miami); CA (Atlanta); LA (New Orl lanically bill the recurring and non-recurring Matures in all states.	t as noted f f the Top 8 leans); NC larket Rates	for Georgia and T MSAS in BellSou (Greensboro-Win s in this section.	ennessee. hth's region for ston Salem-Hi In the interim,	end users with ahpoint/Charlot BellSouth shall	te-Gastonia-Roc bill the rates in the	k Hill); TN (Na he Cost-Based	shville). d section pred							nce.
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Comb The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. La BellSouth currently is developing the billing capability to mech The Market Rate for unbundled ports includes all available fea	Combined in all of the BellSouth states excep bined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orlanically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this ra	t as noted to the Top 8 leans); NC larket Rates at exhibit s	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations o	end users with appoint/Charlott BellSouth shall f loop/port netw	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechange the Market Rate for unbundled ports includes all available fea End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates agare categorized accordingly.	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	t as noted to the Top 8 leans); NC larket Rates at exhibit s	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations o	end users with appoint/Charlott BellSouth shall f loop/port netw	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Not Currently Combinations that are Currently Combinations that are Currently Combinations that are Currently Combinations that are Currently is developing the billing capability to make the Note of the Note o	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	t as noted to the Top 8 leans); NC larket Rates at exhibit s	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations o	end users with appoint/Charlott BellSouth shall f loop/port netw	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Comb The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. La BellSouth currently is developing the billing capability to mech The Market Rate for unbundled ports includes all available fea End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates ap are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	t as noted to the Top 8 leans); NC larket Rates at exhibit s	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations o	end users with appoint/Charlott BellSouth shall f loop/port netw	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechange the Combine Market Rate for unbundled ports includes all available fee End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (REUNE PORT/CREULOP)  UNE POrt/Loop Combination Rates	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	of as noted if f the Top 8 leans); NC I larket Rates ate exhibit s ne First and	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations o	end users with ahpoint/Charlot BellSouth shall f loop/port netw ch Port USOC.	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Comb The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. La BellSouth currently is developing the billing capability to mech The Market Rate for unbundled ports includes all available fea End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates ap are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	t as noted to the Top 8 leans); NC larket Rates at exhibit s	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations o	end users with appoint/Charlott BellSouth shall f loop/port netw	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechange the Combined Self-Self-Self-Self-Self-Self-Self-Self-	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	of as noted if f the Top 8 leans); NC I larket Rates ate exhibit s ne First and	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations o	end users with ahpoint/Charlot BellSouth shall f loop/port netw ch Port USOC.	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Comb The Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to me. The Market Rate for unbundled ports includes all available fea End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates ar are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE POrt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.) shall apply to all of Additional NRC of	ennessee.  ith's region for ston Salem-Hi in the interim, I combinations o columns for ea	end users with appoint/Charlott sellSouth shall floop/port network Port USOC.	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechange the Combined Self-Self-Self-Self-Self-Self-Self-Self-	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	of as noted if f the Top 8 leans); NC I larket Rates ate exhibit s ne First and	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.	ennessee.  ith's region for ston Salem-Hillin the interim, combinations of	end users with ahpoint/Charlot BellSouth shall f loop/port netw ch Port USOC.	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, Ft. Least SellSouth Combine The Loop B MSAs in BellSouth currently is developing the billing capability to much The Market Rate for unbundled ports includes all available fee End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT LOOP Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section.) shall apply to all of Additional NRC of	ennessee.  ith's region for ston Salem-Hi in the interim, I combinations o columns for ea	end users with appoint/Charlott sellSouth shall floop/port network Port USOC.	te-Gastonia-Roc bill the rates in the ork elements ex	k Hill); TN (Na he Cost-Based ccept for UNE	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US	OC: URECU)		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C  2. Unbundled port/loop combinations that are Currently Comb The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Le BellSouth currently is developing the billing capability to me. The Market Rate for unbundled ports includes all available fea End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates ap are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE Port/Loop Combination Rates  2-Wire Volce Grade Loop (SL1) - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Line Port (Res)	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSox (Greensboro-Win s in this section. I shall apply to all of Additional NRC of	ennessee. ith's region for ston Salem-Hi In the interim, I combinations o columns for ea	end users with https://doi.org/10.1001	te-Gastonia-Roc bill the rates in the control of th	ik Hill); TN (Nahe Cost-Based	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	SOC: URECU) pined section.		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechange the Combined State of the Combined State of the Combined State of the Combined State of the Combined State of Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE LINE PORT (Description of Combined State of Combined Sta	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	it as noted if the Top 8 (easns); NC (easn	for Georgia and Ti MSAS in BellSou (Greensboro-Win is in this section	ennessee.  th's region for ston Salem-Hi in the interim, in the interim, in the interim, or columns for each of the columns fo	end users with hpoin//Charlot bellSouth shall foop/port network or Port USOC.	te-Gastonia-Roc bill the rates in the control of th	ik Hill); TN (Nahe Cost-Based coept for UNE combined scer	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	SOC: URECU) sined section.		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Not Currently Comb The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Le BellSouth currently is developing the billing capability to me. The Market Rate for unbundled ports includes all available fea End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates ap are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE  UNE POrt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re	Combined in all of the BellSouth states exceptined or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl antically bill the recurring and non-recurring that the sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSot. (Greensboro-Win s in this section.  Additional NRC of UEPRX UEPRX UEPRX UEPRX	ennessee.  th's region for ston Salem-Hi in the interim, in the interim, in the interim, in the interim, in the interim occlumns for each interior occlumns	end users with appoint/Charlot appoint/Charlot apellSouth shall bell south shall floop/port network Port USOC.  28.18  14.18	e-Gastonia-Roc bill the rates in this procedure of the second of the second procedure of the second of the second procedure of the second of the second procedure of the second of the second of the second procedure of the second of the secon	k Hill); TN (Na he Cost-Baser coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechan The Market Rate for unbundled ports includes all available fea Office and Tradem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Loop to treisidenc  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port vith Caller ID - re  2-Wire voice unbundled port outgoing only - re	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win s in this section. I d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  th's region for ston Salem-Hill in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin the interior in the	end users with hpoint/Charlot bell South shall floop/port network or Port USOC.  28.18  14.18  14  14  14  14	e-Gastonia-Roc bill the rates in the bill the rates in the cork elements ex For Currently C	ik Hill); TN (Nahe Cost-Based he Cost-Based coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Comb The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Le BellSouth currently is developing the billing capability to me The Market Rate for unbundled ports includes all available fea End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates ar are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSot. (Greensboro-Win s in this section.  Additional NRC of UEPRX UEPRX UEPRX UEPRX	ennessee.  th's region for ston Salem-Hi in the interim, in the interim, in the interim, in the interim, in the interim occlumns for each interior occlumns	end users with appoint/Charlot appoint/Charlot apellSouth shall bell south shall floop/port network Port USOC.  28.18  14.18	e-Gastonia-Roc bill the rates in this procedure of the second of the second procedure of the second of the second procedure of the second of the second procedure of the second of the second of the second procedure of the second of the secon	k Hill); TN (Na he Cost-Baser coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL La BellSouth currently is developing the billing capability to mechange the Combined of the Market Rate for unbundled ports includes all available fea floor Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates apare categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (R	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win s in this section. I d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  th's region for ston Salem-Hill in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin the interior in the	end users with hpoint/Charlot bell South shall floop/port network or Port USOC.  28.18  14.18  14  14  14  14	e-Gastonia-Roc bill the rates in the bill the rates in the cork elements ex For Currently C	ik Hill); TN (Nahe Cost-Based he Cost-Based coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechan The Market Rate for unbundled ports includes all available fea Office and Tradem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Loop to treisidenc  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port vith Caller ID - re  2-Wire voice unbundled port outgoing only - re	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win s in this section. I d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  th's region for ston Salem-Hill in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin the interior in the	end users with hpoint/Charlot bell South shall floop/port network or Port USOC.  28.18  14.18  14  14  14  14	e-Gastonia-Roc bill the rates in the bill the rates in the cork elements ex For Currently C	ik Hill); TN (Nahe Cost-Based he Cost-Based coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechange the Combined of the More of the Market Rate for unbundled ports includes all available fee End Office and Tandem Switching Usage and Common Tiens For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE LINE	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section. I shall apply to all of d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  ith's region for ston Salem-Hill in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim in the interim in the interim in the interim in the interim in the interim in the interim in the interim in the interim interim in the interim interim in the interim inter	end users with hpoint/Charlot bells out to shall be shall	e-Gastonia-Roc bill the rates in the bill the rates in the cork elements ex For Currently C	ik Hill); TN (Nahe Cost-Based he Cost-Based coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/Loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechan the Market Rate for unbundled ports includes all available fee End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Loop to (SL1) - Statewid  2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled prot outgoing only - re 2-Wire voice unbundled prot outgoing only - re 2-Wire voice unbundled res, low usage line port w LOCAL NUMBER PORTABILITY	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and To MSAS in BellSou (Greensboro-Win is in this section. ) Shall apply to all of 3 Additional NRC ( UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX  UEPRL  UEPRC  UEPRC  UEPRO  UEPAP	end users with hpoint/Charlot bells out to shall be shall	e-Gastonia-Roc bill the rates in the bill the rates in the cork elements ex For Currently C	ik Hill); TN (Nahe Cost-Based he Cost-Based coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechange the Combined of the More of the Market Rate for unbundled ports includes all available fee End Office and Tandem Switching Usage and Common Tiens For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE LINE	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win is in this section. I shall apply to all of d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  ith's region for ston Salem-Hill in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim, in the interim in the interim in the interim in the interim in the interim in the interim in the interim in the interim in the interim interim in the interim interim in the interim inter	end users with hpoint/Charlot bells out to shall be shall	e-Gastonia-Roc bill the rates in the bill the rates in the cork elements ex For Currently C	ik Hill); TN (Nahe Cost-Based he Cost-Based coept for UNE combined scer 90 90	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Not Currently Combinations that are Currently Combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. LebelSouth Currently is developing the billing capability to make the Market Rate for unbundled ports includes all available fee End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-Wire VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORTLOOP Combination Rates   2-Wire VG Loop/Port Combo - Statewid   2-Wire VG Loop/Port Combo - Statewid   2-Wire Voice Grade Loop (SL1) - Statewid   2-Wire Voice Unbundled port with Caller ID - re   2-Wire voice unbundled port untgoing only - ri   2-Wire voice unbundled port outgoing only - ri   2-Wire voice unbundled port   2-Wire voice unbundled   2-Wire voice unbundled   2-Wire voice unbundled   2-Wire voice unbundled   2-Wire voice unbundled   2-Wire voice unbundled   2-Wire voice unbundled   2-Wire voice un	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl anrically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and To MSAS in BellSou (Greensboro-Win is in this section. Is in this section. Is in this section. If the section of the section	UEPLX  UEPLX  UEPRC  UE	end users with hpoint/Charlott sellSouth shall floop/port network or Port USOC.  28.18  14.18  14.18  14.14  14.14  14.14  10.35	e-Gastonia-Roc bill the rates in the little to the little	Is Hill); TN (Nahama) he Cost-Baser  go and t	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Las BellSouth Currently is developing the billing capability to mechan the Market Rate for unbundled ports includes all available fea for Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates apare categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire Voice Grade Loop (SL1) - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundles res, low usage line port w  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per por  FEATURES  All Features Offerex	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 or auderdale, Miamit); GA (Atlanta); LA (New Ori maincially bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined Combin	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win sin this section) shall apply to all of d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  ennessee.  th's region for  ston Salem-Hi in the interni,  nombinations o  columns for ea  UEPLX  UEPRL  UEPRC  UEPRO  UEPRO  UEPAP  LNPCX  UEPVF  UEPAP	end users with hpoint/Charlott sellSouth shall floop/port network or Port USOC.  28.18  14.18  14.18  14.14  14.14  14.14  10.35	e-Gastonia-Roc bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill	90 90 90 41.5	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/Loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Not Currently C 3. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, FL Lease BellSouth currently is developing the billing capability to mechange the Market Rate for unbundled ports includes all available fee End Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates    2-Wire VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates    2-Wire VG Loop/Port Combo - Statewid    UNE Loop Rates    2-Wire Voice Grade Loop (SL1) - Statewid    2-Wire Voice Grade Loop transled port with Caller ID - re    2-Wire voice unbundled port with Caller ID - re    2-Wire voice unbundled port outgoing only - re    2-Wire voice unbundled port outgoing only - re    2-Wire voice unbundleds res, low usage line port w    LOCAL NUMBER PORTABILITY    Local Number Portability (1 per por    FEATURES    All Features Offered	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 or auderdale, Miamit); GA (Atlanta); LA (New Ori maincially bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined Combin	it as noted if the Top 8 (easns); NC (easn	for Georgia and To MSAS in BellSou (Greensboro-Win is in this section. Is in this section. Is in this section. If the section of the section	UEPLX  UEPLX  UEPRC  UE	end users with hpoint/Charlott sellSouth shall floop/port network or Port USOC.  28.18  14.18  14.18  14.14  14.14  14.14  10.35	e-Gastonia-Roc bill the rates in the little to the little	Is Hill); TN (Nahama) he Cost-Baser  go and t	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, FL La BellSouth currently is developing the billing capability to mechan the Market Rate for unbundled ports includes all available fea find Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire Volce Grade Loop (SL1) - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port using only - rr  2-Wire voice unbundled port Unit Caller ID - re  2-Wire voice unbundled port using only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port Unit Caller ID - re  2-Wire voice Unbundled port outgoing only - rr  2-Wire voice Unbundled port outgoing only - rr  2-Wire voice OrabellITY  Local Number Portability (1 per por  FEATURES)  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 or auderdale, Miamit); GA (Atlanta); LA (New Ori maincially bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined Combin	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win sin this section) shall apply to all of d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  ennessee.  th's region for  ston Salem-Hi in the interni,  nombinations o  columns for ea  UEPLX  UEPRL  UEPRC  UEPRO  UEPRO  UEPAP  LNPCX  UEPVF  UEPAP	end users with hpoint/Charlott sellSouth shall floop/port network or Port USOC.  28.18  14.18  14.18  14.14  14.14  14.14  10.35	e-Gastonia-Roc bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill	90 90 90 41.5	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechan The Market Rate for unbundled ports includes all available fea God Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice Unbundled Dort Dort Ontonination Caller View of Caller Combination Caller View of Caller Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combi	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl amicially bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	Tor Georgia and Total MSAS in BellSou (Greensboro-Win Sin this section. It is the section. It is additional NRC of the MSAS of the MSAS of the MSAS of the UEPRX	ennessee.  th's region for ston Salem-Hi in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin in the interin in the interin in the interior in the i	end users with hpoint/Charlott sellSouth shall floop/port network or Port USOC.  28.18  14.18  14.18  14.14  14.14  14.14  10.35	e-Gastonia-Roc bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill	90 90 90 41.5	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top B MSAs in BellSouth's region are: FL (Orlando, FL La BellSouth currently is developing the billing capability to mechan the Market Rate for unbundled ports includes all available fea find Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire Volce Grade Loop (SL1) - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port using only - rr  2-Wire voice unbundled port Unit Caller ID - re  2-Wire voice unbundled port using only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port outgoing only - rr  2-Wire voice unbundled port Unit Caller ID - re  2-Wire voice Unbundled port outgoing only - rr  2-Wire voice Unbundled port outgoing only - rr  2-Wire voice OrabellITY  Local Number Portability (1 per por  FEATURES)  All Features Offerec  2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice Grade Loop / Line Port Combination 2-Wire Voice	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Orl amicially bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Section of the Combined	it as noted if the Top 8 (easns); NC (easn	for Georgia and T MSAS in BellSou (Greensboro-Win sin this section) shall apply to all of d Additional NRC of UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ennessee.  ennessee.  th's region for  ston Salem-Hi in the interni,  nombinations o  columns for ea  UEPLX  UEPRL  UEPRC  UEPRO  UEPRO  UEPAP  LNPCX  UEPVF  UEPAP	end users with hpoint/Charlott sellSouth shall floop/port network or Port USOC.  28.18  14.18  14.18  14.14  14.14  14.14  10.35	e-Gastonia-Roc bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill the rates in the bill	90 90 90 41.5	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		
These scenarios include:  1. Unbundled port/loop combinations that are Not Currently C 2. Unbundled port/loop combinations that are Currently Combine Top 8 MSAs in BellSouth's region are: FL (Orlando, FL Le BellSouth currently is developing the billing capability to mechan The Market Rate for unbundled ports includes all available fea God Office and Tandem Switching Usage and Common Trans For Not Currently Combined scenarios where Market Rates are categorized accordingly.  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RE UNE PORT/Loop Combination Rates  2-Wire VG Loop/Port Combo - Statewid  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire Voice Grade Loop (SL1) - Statewid  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice unbundled port outgoing only - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice unbundled port Loop of Undured Dort With Caller ID - re  2-Wire voice Unbundled Dort Dort Ontonination Caller View of Caller Combination Caller View of Caller Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combination Caller View Order Carde Loop / Line Port Combi	Combined in all of the BellSouth states exceptioned or Not Currently Combined in Zone 1 of auderdale, Miami); GA (Atlanta); LA (New Ori Annically bill the recurring and non-recurring Matures in all states.  Sport Usage rates in the Port section of this repply, the Nonrecurring charges are listed in the Port section of the Institute of the Insti	it as noted if the Top 8 (easns); NC (easn	Tor Georgia and Total MSAS in BellSou (Greensboro-Win Sin this section. It is the section. It is additional NRC of the MSAS of the MSAS of the MSAS of the UEPRX	ennessee.  th's region for ston Salem-Hi in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin, in the interin in the interin in the interin in the interior in the i	end users with hpoint/Charlott sellSouth shall floop/port network or Port USOC.  28.18  14.18  14.18  14.14  14.14  14.14  10.35	e-Gastonia-Roc bill the rates in ti ll the rates in the rates in ti ll the rates in the	90 90 90 41.5	ashville). d section pred Coin Port/Lo	oop Combinations	which have a	flat rate us	age charge (US Currently Comb	9.45 9.45 9.45 9.45		

Attachment 2 Exhibit C

UNE Port/Loop Combination Rates			T								_
2-Wire VG Loop/Port Combo - Statewid	sw		1	28.18							
UNE Loop Rates											
2-Wire Voice Grade Loop (SL1) - Statewid	sw U	UEPBX	UEPLX	14.18							
2-Wire Voice Grade Line Port (Bus)											
2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	14	90	90		40.18	9.45		
2-Wire voice unbundled port with Caller + E484 ID - bι		UEPBX	UEPBC	14	90	90		40.18	9.45		1
2-Wire voice unbundled port outgoing only - bt	l	UEPBX	UEPBO	14	90	90		40.18	9.45		
LOCAL NUMBER PORTABILITY											
Local Number Portability (1 per por	l.	UEPBX	LNPCX	0.35							
FEATURES											
NONRECURRING CHARGES - CURRENTLY COMBINED											
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-	l	UEPBX	USAC2		41.5	41.5					
2-Wire Voice Grade Loop / Line Port Combination - Switch with chan	l	UEPBX	USACC		41.5	41.5					
ADDITIONAL NRCs											
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque	l	UEPBX	USAS2		0	0					
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)											
UNE Port/Loop Combination Rates			1					40.18	9.45	20	
2-Wire VG Loop/Port Combo - Statewid	sw		1	28.18							_
			1								
UNE Loop Rates			1	İ							_
2-Wire Voice Grade Loop (SL1) - Statewid	sw U	JEPRG	UEPLX	14.18				1	1		_
2-Wire Voice Grade Line Port Rates (RES - PBX)			<del>                                     </del>								
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re	- 1	JEPRG	UEPRD	14	90	90		40.18	9.45		
2 Wile Vo Gradinated Combination 2 Way 1 DX Trainer Cit. 18		<u>/L: 110</u>	OE: NO					10.10	0.10		
LOCAL NUMBER PORTABILITY			+								
Local Number Portability (1 per por		JEPRG	LNPCP	3.15							_
Eccar Number 1 Ortability (1 per por		)LI NO	EIVI OI	3.13							
FEATURES			+					+		-	_
PATONEO											_
NONRECURRING CHARGES - CURRENTLY COMBINED		-	+								
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-		JEPRG	USAC2		41.5	41.5		+			_
		JEPRG JEPRG	USACC		41.5	41.5					_
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Chan		JEPRG	USACC		41.5	41.5					
ADDITIONAL NRCs											_
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-					0	0					
Nonrecurring Day of the Company of t								40.00	40.00		_
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Gro					14.64	14.64		19.99	19.99	19.99	_
AND TO SELECT THE SELE											_
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)											-
											-
UNE Port/Loop Combination Rates											
2-Wire VG Loop/Port Combo - Statewid	SW			28.18							
											_
UNE Loop Rates								+ + + + + + + + + + + + + + + + + + + +			
2-Wire Voice Grade Loop (SL1) - Statewid	sw U	UEPPX	UEPLX	14.18				1			
	$\longrightarrow$										,
2-Wire Voice Grade Line Port Rates (BUS - PBX)	$\longrightarrow$		<u> </u>								,
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bi		UEPPX	UEPPC	14	90	90		40.18	9.45		
Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	14	90	90		40.18	9.45		
Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX	UEPP1	14	90	90		40.18	9.45	Ţ	
2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	14	90	90		40.18	9.45		
2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	14	90	90		40.18	9.45		
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	14	90	90		40.18	9.45		
2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	14	90	90		40.18	9.45		
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	14	90	90		40.18	9.45		
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pt	L'	UEPPX	UEPXE	14	90	90		40.18	9.45		
											n
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P		UEPPX	UEPXL	14	90	90		40.18	9.45		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P	L L	UEPPX	UEPXM	14	90	90		40.18	9.45	Ţ	
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling											i .
Port		UEPPX	UEPXO	14	90	90		40.18	9.45		
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	l	UEPPX	UEPXS	14	90	90		40.18	9.45		
LOCAL NUMBER PORTABILITY											
Local Number Portability (1 per por	l	UEPPX	LNPCP	3.15							
			1								
FEATURES							+ + + + + + + + + + + + + + + + + + + +				_
FEATURES			1								
FEATURES  NONRECURRING CHARGES - CURRENTLY COMBINED											
		UEPPX	USAC2		41.5	41.5					_

ADDITIONAL NRCs										i .
2-Wire Voice Grade Loop/ Line Port Combination - Subseque	UEPPX	USAS2		0	0					
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring				0	0					Г
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Gro				14.64	14.64		19.99	19.99	19.99	
										ь
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT										Н
UNE Port/Loop Combination Rates									+-+	Г
2-Wire VG Coin Port/Loop Combo – Statewide			28.18							Ξ
UNE Loop Rates									++	г
2-Wire Voice Grade Loop (SL1) - Statewid	UEPCO	UEPLX	14.18							П
2-Wire Voice Grade Line Port Rates (Coin)									+	Н
2-Wire Coin 2-Way without Operator Screening and without Blocking (NC)	UEPCO	UEPND	14	90	90		40.18	9.45	++	Г
2-Wire Coin 2-Way with Operator Screening (NC)	UEPCO	UEPNC	14	90	90		40.18	9.45	+	Г
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (NC, TN)	UEPCO	UEPRP	14				40.18	9.45		
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (NC)	UEPCO	UEPNB	14	90	90		40.18	9.45	++	Т
2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (NC, TN)	UEPCO	UEPCA	14	90	90		40.18	9.45		
2-Wire Coin Outward with Operator Screening and 011 Blocking (NC)	UEPCO	UEPNE	14	90	90		40.18	9.45	+	
2-Wire Coin Outward with Operator Screening and 81 Sicking: 900/976, 1+DDD, 011+, a Local (NC)	UEPCO	UEPCL	14	90	90		40.18	9.45		
Ecour (10)	52.700	02.02					10.10	0.10		F
LOCAL NUMBER PORTABILITY										Г
Local Number Portability (1 per por	UEPCO	LNPCX	0.35						-	Н
NONRECURRING CHARGES - CURRENTLY COMBINED										Г
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-	UEPCO	USAC2		41.5	41.5					<u> </u>
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Chan	UEPCO	USACC		41.5	41.5					<u> </u>
ADDITIONAL NRCs										
2-Wire Voice Grade Loop/ Line Port Combination - Subsequ€	UEPCO	USAS2		0	0					-
									+	-
										-

Page 18 of 18 Version 2Q01: 08/30/01

RY	NOTES	UNBUNDLED NETWORK ELEMENT Inter	rim Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Electronic-Disc	Orde
							Nonre	ecurring					ecurring			
													connect	т —		_
_						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SO
																4
		n in the sections for stand-alone loops or loops as part of a combination refers to Geogra onnection.bellsouth.com/become_a_clec/html/interconnection.htm	phically D	l eaveraged UNE Zone	es. To view Ge	ographically De	I eaveraged UNE	Zone Designation	ons by Centr	al Office, ref	er to Internet	Website:				
DLED E	XCHANGE ACC	CESS LOOP														+-
2-V		VOICE GRADE LOOP														
	2-Wi	ire Analog Voice Grade Loop - Service Level 1- Zone	1	UEANL	UEAL2	18.48	70.44	44.05					44.22	13.55		
	2-Wi	ire Analog Voice Grade Loop - Service Level 1- Zone	2	UEANL	UEAL2	27.87	70.44	44.05					44.22	13.55		
	2-Wi	ire Analog Voice Grade Loop - Service Level 1- Zone	3	UEANL	UEAL2	36.91	70.44	44.05					44.22	13.55		
		ire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone	1	UEPSR, UEPSB	UEALS	18.48	70.44	44.05					44.22	13.55		T
		ire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone	2	UEPSR, UEPSB	UEALS	27.87	70.44	44.05					44.22	13.55	1	T
		ire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone	3		UEALS	36.91	70.44	44.05					44.22	13.55	1	1
$\dashv$		neering Information Document (E	Ť	UEANL	2_7,1_0	23.01	28.82	28.82								1
		ual Order Coordination for UVL-SL1s (per loop		UEANL	UEAMC	1	62.1	62.1						<b></b>		+
_	Ordo	er Coordination for OVL-SL1s (per loop er Coordination for Specified Conversion Time for UVL-SL1 (per LSR	-	UEANL	OCOSL	1	45.43	45.43						$\vdash$	$\vdash$	+
	Orde	a Coordination for Specified Conversion Time for OVE-SET (per ESR		UEMINL	UCUSL	1	40.40	40.40								+
_	0 14/:	iro Analog Voice Grade Leon - Service Level 2 w/Leon or Cround Stort Si				1										+
	Z-vvi	ire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -	1	UEA	UEAL2	21.57	178.12	128.8					44.42	13.55		
+		ire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -		UEA	ULALZ	16.12	170.12	120.0					77.42	10.00	<del></del>	+
			2	UEA	UEAL2	32.53	178.12	128.8					44.42	12 55		
_	Zone			UEA	UEALZ	32.53	178.12	128.8			1		44.42	13.55	<del> </del>	+-
1		ire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling -	_		UEAL2	43.08	178.12	400.0					44.40	40.55		
	Zone		3	UEA		43.08		128.8					44.42	13.55	+	+-
_	Orde	er Coordination for Specified Conversion Time (per LS		UEA	OCOSL	+	45.43	1			1			-	-	+
	2-Wi	ire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zon	1	UEA	UEAR2	21.57	178.12	128.8					44.42	13.55		
	2-Wi	ire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zon	2	UEA	UEAR2	32.53	178.12	128.8					44.42	13.55		
				<u> </u>												+
	2-Wi	ire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zon	3	UEA	UEAR2	43.08	178.12	128.8					44.42	13.55		
		er Coordination for Specified Conversion Time (per LS		UEA	OCOSL	40.00	45.43	120.0					77.72	10.00	<u> </u>	+
4-V	WIRE ANALOG	VOICE GRADE LOOP		OLA	OOOOL		40.40								1	+
		ire Analog Voice Grade Loop - Zone	1	UEA	UEAL4	29.47	383.39	286.77					44.06	13.55	+	+
-		ire Analog Voice Grade Loop - Zone	2	UEA	UEAL4	44.44	383.39	286.77					44.06	13.55	+	+
				UEA	UEAL4	58.85							44.06			+
		ire Analog Voice Grade Loop - Zone	3		OCOSL	28.82	383.39	286.77					44.06	13.55		+
_	Orde	er Coordination for Specified Conversion Time (per LS	_	UEA	UCUSL		45.43									+-
																+-
2-V		ITAL GRADE LOOP														4
	2-Wi	ire ISDN Digital Grade Loop - Zone	1	UDN	U1L2X	26.68	423.04	301.75					44.42	13.55		
	2-Wi	ire ISDN Digital Grade Loop - Zone	2	UDN	U1L2X	40.24	423.04	301.75					44.42	13.55	1	4
		ire ISDN Digital Grade Loop - Zone	3	UDN	U1L2X	53.85	423.04	301.75					44.42	13.55		
	Orde	er Coordination For Specified Conversion Time (per LS		UDN	OCOSL		45.43			· · · · · · · · · · · · · · · · · · ·						
2-V		I Digital Channel (UDC) COMPATIBLE LOOP				1										
	2-Wi	ire Universal Digital Channel (UDC) Compatible Loop - Zone 1	1	UDC	UDC2X	31.51	235.15	160.05	106.09	21.21			44.42	13.55		
Т		ire Universal Digital Channel (UDC) Compatible Loop - Zone 2	2	UDC	UDC2X	40.95	235.15	160.05	106.09	21.21			44.42	13.55		
$\neg$		ire Universal Digital Channel (UDC) Compatible Loop - Zone 3	3	UDC	UDC2X	47.12	235.15	160.05	106.09	21.21			44.42	13.55		
																I
2-V	WIRE ASYMME	TRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP												<u> </u>	<u> </u>	₩
						1		1								
		IRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP				1								<b></b>	<b></b>	4
		re Unbundled ADSL Loop including manual service inquiry & facility reservation -				1		1								
	Zone	5 1	1	UAL	UAL2X	17.1	600.61	507.33					44.42	13.55		4
		re Unbundled ADSL Loop including manual service inquiry & facility reservation -				1		1								
	Zone		2	UAL	UAL2X	25.79	600.61	507.33					44.42	13.55		
		re Unbundled ADSL Loop including manual service inquiry & facility reservation -								·						
	Zone		3	UAL	UAL2X	34.15	600.61	507.33					44.42	13.55		$\perp$
	Orde	er Coordination for Specified Conversion Time (per LS		UAL	OCOSL		45.43							<u> </u>		╀
	2 Wi	ire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zon	1	UAL	UAL2W	17.1	205.28	129.32	100.74	15.86			44.42	13.55		
		ire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zon	2	UAL	UAL2W	25.79	205.25	129.32	100.74	15.86			44.42	13.55		
-	Z VVI	The onbunding Abolt Loop without manual service inquity & facility reservation - 2011		UAL	UALZVV	23.18	200.20	123.32	100.74	10.00			44.42	13.33	<del>                                     </del>	+
	2 1/1	ire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zon	3	UAL	UAL2W	34.15	205.28	129.32	100.74	15.86			44.42	13.55		
	Z VVI	er Coordination for Specified Conversion Time (per LS	- 3	UAL	OCOSL	J4. IJ	45.43	123.32	100.74	13.00			44.42	10.00		+
_	Ordo						40.40	1			1			1	1	1
$\pm$	Orde	er Coordination for Specified Conversion Time (per LS		O/ IL												-

Page 1 of 22

				!									T
	2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP												
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -	1			40.04	000.04	507.00			44.00	40.55		
	Zone 1  2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -	1	UHL	UHL2X	12.21	600.61	507.33			44.06	13.55		
	Zone 2	2	UHL	UHL2X	18.41	600.61	507.33			44.06	13.55		
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation -			- UNILEX	10.11	000.01	001.00			11.00	10.00		
Z	Zone 3	3	UHL	UHL2X	24.39	600.61	507.33			44.06	13.55		
	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		45.43							
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -								_				
	Zone 1  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -	1	UHL	UHL2W	12.21	222.65	146.68	00.74 15.8	6	44.06	13.55		+
	Zone 2	2	UHL	UHL2W	18.41	222.65	146.68	00.74 15.8	6	44.06	13.55		
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -		OHE	OTILZVV	10.41	222.03	140.00	100.74	0	44.00	13.33		$\vdash$
	Zone 3	3	UHL	UHL2W	24.39	222.65	146.68	00.74 15.8	6	44.06	13.55		
C	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		45.43							
	BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP												+
	4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 1	1	UHL	UHL4X	16.21	625.11	532.78			44.06	13.55		
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -		UNL	UNL4A	10.21	023.11	332.76			44.00	13.33		+
	Zone 2	2	UHL	UHL4X	24.45	625.11	532.78			44.06	13.55		
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation -												Т
Z	Zone 3	3	UHL	UHL4X	32.38	625.11	532.78			44.06	13.55		
c	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		45.43							$\bot$
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -	1	ш	LILL 4W	16.01	270.06	202.00	10.24	_	44.00	12.55		
	Zone 1 4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -		UHL	UHL4W	16.21	279.96	203.99	10.24 20.7	υ	44.06	13.55		╫
	Zone 2	2	UHL	UHL4W	24.45	279.96	203.99	10.24 20.7	5	44.06	13.55		
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation -		O. I.E.		20	2,0.00	200.00	20.	-	11.00	10.00		T
Z	Zone 3	3	UHL	UHL4W	32.38	279.96	203.99	10.24 20.7	5	44.06	13.55		$\perp$
C	Order Coordination for Specified Conversion Time (per LS		UHL	OCOSL		45.43							
	DIGITAL LOOP	1	LICI	HOLVY	50.04	745 77	404.5			40.77	40.55		+
	4-Wire DS1 Digital Loop - Zone : 4-Wire DS1 Digital Loop - Zone :	2	USL	USLXX	59.61 89.9	715.77 715.77	421.5 421.5			43.77 43.77	13.55 13.55		+
1	4-Wire DS1 Digital Loop - Zone :	3	USL	USLXX	119.06	715.77	421.5			43.77	13.55		╫
	Order Coordination for Specified Conversion Time (per LS		USL	OCOSL		48.47							
	56 OR 64 KBPS DIGITAL GRADE LOOP												
	4 Wire Unbundled Digital 19.2 Kbps	1	UDL	UDL19	34.26	602.73	393.5			44.06	13.55		
	4 Wire Unbundled Digital 19.2 Kbps	2	UDL	UDL19	51.67	602.73	393.5			44.06	13.55		+
	4 Wire Unbundled Digital 19.2 Kbps	3	UDL	UDL19	68.43 34.26	602.73	393.5			44.06	13.55		╫
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 4 Wire Unbundled Digital Loop 56 Kbps - Zone	2	UDL UDL	UDL56 UDL56	51.67	602.73 602.73	393.5 393.5			44.06 44.06	13.55 13.55		+
	4 Wire Unbundled Digital Loop 56 Kbps - Zone	3	UDL	UDL56	68.43	602.73	393.5			44.06	13.55		T
	Order Coordination for Specified Conversion Time (per LS		UDL	OCOSL		45.43							
4	4 Wire Unbundled Digital Loop 64 Kbps - Zone	1	UDL	UDL64	34.26	602.73	393.5			44.06	13.55		
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	2	UDL	UDL64	51.67	602.73	393.5			44.06	13.55		+
	4 Wire Unbundled Digital Loop 64 Kbps - Zone	3	UDL	UDL64	68.47	602.73	393.5			44.06	13.55		+
	Order Coordination for Specified Conversion Time (per LS		UDL	OCOSL		45.43							+
													+
2-WIRE Unbu	ındled COPPER LOOP												T
													Т
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility												1
re	reservation - Zone 1	1	UCL	UCLPB	15.24	283.95	163.99	20.42 22.4	2	19.99	19.99	19.99	+
2	reservation - Zone 1 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility												
2 76	reservation - Zone '	2	UCL	UCLPB UCLPB	15.24 17.14	283.95 283.95		20.42 22.4		19.99 19.99	19.99	19.99	
2 10 10 2	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility		UCL	UCLPB	17.14	283.95	163.99	20.42 22.4	2	19.99	19.99	19.99	
2 16 2 16 2	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ;	2	UCL			283.95 283.95	163.99 f		2				
2 16 2 16 2 16 16 16 16 16 16 16 16 16 16 16 16 16	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility	2	UCL	UCLPB UCLPB	17.14	283.95	163.99	20.42 22.4	2	19.99	19.99	19.99	
2 10 2 10 10 10 10 10 10 10 10 10 10 10 10 10	reservation - Zone '	2	UCL	UCLPB UCLPB	17.14	283.95 283.95	163.99 163.99 62.1	20.42 22.4	2	19.99	19.99	19.99	
2 m 2 m C C	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ( 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ( Order Coordination for Unbundled Copper Loops (per loc 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ( 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility	3	UCL UCL UCL	UCLPB UCLPB UCLMC UCLPW	17.14 17.68 15.24	283.95 283.95 62.1 203.42	163.99 163.99 62.1 127.45	20.42   22.4   20.42   22.4   20.74   15.8	2 2 6	19.99 19.99	19.99	19.99 19.99	
2 2 2 2 6 0 0 2 2 2 8	reservation - Zone '  2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone :  2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc  2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :  2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :	2 3	UCL UCL UCL	UCLPB UCLPB UCLMC	17.14 17.68	283.95 283.95 62.1	163.99 163.99 62.1 127.45	20.42 22.4	2 2 6	19.99	19.99	19.99	
22 m 22 m C 2 m 2 2 m 2 2 m	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ( 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ( Order Coordination for Unbundled Copper Loops (per loc 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ( 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ( 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ( 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility	3	UCL UCL UCL UCL	UCLPB UCLPB UCLMC UCLPW UCLPW	17.14 17.68 15.24 17.14	283.95 283.95 62.1 203.42 203.42	163.99 163.99 62.1 127.45 127.45	20.42 22.4 20.42 22.4 100.74 15.4 100.74 15.4	2 2 6 6	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	
2 2 m	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2-Wire Unbundled Copper Loops (per loc - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ' 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone (	3	UCL UCL UCL UCL	UCLPB UCLPB UCLMC UCLPW UCLPW UCLPW	17.14 17.68 15.24	283.95 283.95 62.1 203.42 203.42 203.42	163.99 163.99 62.1 127.45 127.45 127.45	20.42   22.4   20.42   22.4   20.74   15.8	2 2 6 6	19.99 19.99	19.99	19.99 19.99	
2 2 m 2 2 2 m 2 2 2 m 2 2 2 2 m 2	reservation - Zone '	3	UCL UCL UCL UCL	UCLPB UCLPB UCLMC UCLPW UCLPW	17.14 17.68 15.24 17.14	283.95 283.95 62.1 203.42 203.42	163.99 163.99 62.1 127.45 127.45	20.42 22.4 20.42 22.4 100.74 15.4 100.74 15.4	2 2 6 6	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	
2 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2-Wire Unbundled Copper Loops (per loc - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ' 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone (	3	UCL UCL UCL UCL	UCLPB UCLPB UCLMC UCLPW UCLPW UCLPW	17.14 17.68 15.24 17.14	283.95 283.95 62.1 203.42 203.42 203.42	163.99 62.1 127.45 127.45 127.45 62.1	20.42 22.4 20.42 22.4 100.74 15.4 100.74 15.4	2 2 6 6 6 6	19.99 19.99 19.99	19.99 19.99 19.99	19.99 19.99 19.99	
2 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone :  Order Coordination for Unbundled Copper Loops (per loc 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :  2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :  2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :  2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :  2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone :  2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone :  2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :	2 3 1 2 3	UCL LPW UCLPW UCLPW UCLPW UCLPW UCLPW UCLMC	17.14 17.68 15.24 17.14 17.68	283.95 283.95 62.1 203.42 203.42 203.42 62.1 270.89	163.99 163.99 162.1 127.45 127.45 127.45 127.45 120.93 150.93	20.42   22.4   20.42   22.4   100.74   15.1   100.74   15.1   100.74   15.1   100.74   22.4	2 2 6 6 6 6 2 2	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99		
2 2 m C C 2 m C C 2 m C C C 2 m C C C C	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2 Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 0-Zewire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 0-Zewire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ;	3 1 2 3	UCL UCL UCL UCL UCL UCL	UCLPB UCLPB UCLMC UCLPW UCLPW UCLPW UCLPW	17.14 17.68 15.24 17.14 17.68	283.95 283.95 62.1 203.42 203.42 203.42 62.1	163.99 163.99 162.1 127.45 127.45 127.45 127.45 120.93 150.93	20.42   22.4   20.42   22.4   20.74   15.4   20.74   15.4   20.74   15.4	2 2 6 6 6 6 2 2	19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99	
2 2 R 2 R 2 2 R 2 2 R 2 2 R 2 2 R 2 R 2 2 R 2 R 2 2 R 2 R 2 2 R	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :	2 3 1 2 3 1 2	UCL LPW UCLPW UCLPW UCLPW UCLPW UCLMC UCL2L	17.14 17.68 15.24 17.14 17.68 47.77 69.16	283.95 283.95 62.1 203.42 203.42 203.42 62.1 270.89	163.99	20.42   22.42   22.42   22.42   23.42   24.4	2 2 6 6 6 6 2 2 2	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99		
2 2 m 2 2 m	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ; 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone ;	2 3 1 2 3	UCL   UCLPW  UCLMC  UCLPW  UCLMC  UCL2L  UCL2L	17.14 17.68 15.24 17.14 17.68	283.95 283.95 62.1 203.42 203.42 203.42 62.1 270.89 270.89	163.99	20.42   22.4   20.42   22.4   100.74   15.1   100.74   15.1   100.74   15.1   100.74   22.4	2 2 6 6 6 6 2 2 2	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99		
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2 Wire unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :	2 3 1 2 3 1 2	UCL LPW UCLPW UCLPW UCLPW UCLPW UCLMC UCL2L	17.14 17.68 15.24 17.14 17.68 47.77 69.16	283.95 283.95 62.1 203.42 203.42 203.42 62.1 270.89	163.99	20.42   22.42   22.42   22.42   23.42   24.4	2 2 6 6 6 6 2 2 2	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99		
2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 2 m 2	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 2 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3 Corder Coordination for Unbundled Copper Loops (per loc 2 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 5 Corder Coordination for Unbundled Copper Loops (per loc 2 2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone 5	2 3 1 2 3 1 2	UCL UCL UCL UCL UCL UCL UCL UCL UCL UCL	UCLPB  UCLPW  UCLPW  UCLPW  UCLPW  UCLMC  UCLMC  UCL2L  UCL2L  UCLMC	17.14 17.68 15.24 17.14 17.68 47.77 69.16 84.94	283.95 283.95 62.1 203.42 203.42 203.42 62.1 270.89 270.89 62.1	163.99	20.42   22.4	2 2 6 6 6 6 2 2 2 2 2	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	
2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 m 2 2 2 2 m 2	reservation - Zone ' 2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2 Wire unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone : 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone :	2 3 3 1 2 2 3 3	UCL   UCLPW  UCLMC  UCLPW  UCLMC  UCL2L  UCL2L	17.14 17.68 15.24 17.14 17.68 47.77 69.16	283.95 283.95 62.1 203.42 203.42 203.42 62.1 270.89 270.89	163.99	20.42   22.42   22.42   22.42   23.42   24.4	2 2 6 6 6 6 2 2 2 2 2	19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99 19.99	19.99 19.99 19.99 19.99 19.99		

	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		T	1									
	reservation - Zone (	3	UCL	UCL2W	84.94	190.36	114.39 100.74	15.86		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC		62.1	62.1						
												<b></b>	
	2-Wire Unbundled Copper Loop - Non-Designed Zone	1 1	UEQ	UEQ2X	11.01	44.69	22.4 25.65	7.06	+	44.22	13.55	<b></b>	
	2 Wire Unbundled Copper Loop - Non-Designed - Zone	1 2		UEQ2X	12.67	44.69	22.4 25.65			44.22	13.55		+
	2 Wire Unbundled Copper Loop - Non-Designed - Zone	1 3		UEQ2X	20.22	44.69	22.4 25.65			44.22	13.55		
	Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per lo		UEQ	USBMC		62.1	62.1						
	Engineering Information Documer		UEQ			28.82	28.82						
	Loop Testing - Basic 1st Half Hou		UEQ	URET1		78.92	78.92						
	Loop Testing - Basic Additional Half Hou		UEQ	URETA	<u> </u>	23.33	23.33					<b></b>	
		-+	+										+
4-1	WIRE COPPER LOOP		+	+			+		+				+
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zo		+	+	<del>                                     </del>	+	+		+				+
	1	1	UCL	UCL4S	24.55	332.47	212.51 130.98	3 27.68		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zo												
	2	2	UCL	UCL4S	26.13	332.47	212.51 130.98	3 27.68		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zo	3	1101		04.47	000.47	040.54			40.00	40.00	40.00	40.00
	Order Coordination for Unbundled Copper Loops (per loc	3	UCL	UCL4S UCLMC	24.17	332.47 62.1	212.51 130.98 62.1	3 27.68	+	19.99	19.99	19.99	19.99
	Order Coordination for Oribunded Copper Loops (per loc		UCL	UCLIVIC		02.1	02.1		+				+
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zon	1	UCL	UCL4W	24.55	251.94	175.94 110.24	4 20.75		19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zon	2	UCL	UCL4W	26.13	251.94	175.94 110.24	4 20.75		19.99	19.99	19.99	19.99
			1 .				I I -					I	1
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zon	3		UCL4W	24.17	251.94	175.94 110.24	4 20.75		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc	-+	UCL	UCLMC	<del> </del>	62.1	62.1		<del>                                     </del>	+			+
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1	1	UCL	UCL4L	96.61	319.41	199.45 130.98	3 27.66		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		UCL	OCL4L	30.01	319.41	133.43 130.30	27.00	+	13.33	19.99	19.99	13.33
	reservation - Zone 2	2	UCL	UCL4L	148.48	319.41	199.45 130.98	3 27.66		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility												
	reservation - Zone 1	3	UCL	UCL4L	180.12	319.41	199.45 130.98	3 27.66		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC		62.1	62.1					<b></b>	
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservati	1	1101	1101.40	00.04	000.07	4000			40.00	40.00	40.00	40.00
	Zone 1  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservati	1	UCL	UCL4O	96.61	238.87	162.9 110.24	4 20.75	+	19.99	19.99	19.99	19.99
	Zone 2	2	UCL	UCL4O	148.48	238.87	162.9 110.24	4 20.75		19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservati		- 002	00240	140.40	200.01	102.0 110.2-	20.70		13.33	13.55	13.33	10.00
	Zone 3	3	UCL	UCL4O	180.12	238.87	162.9 110.24	4 20.75		19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loc		UCL	UCLMC		62.1	62.1					<b></b>	
												<b></b>	
LOOP MODIFIC	PATION			+	-	-			+			<b></b>	
LOOP MODIFIC	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to		UAL, UHL, UCL,				<del>                                     </del>		+				+
	18k ft		UEQ, ULS	ULM2L		65.32	65.32				ļ	[	
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18		UCL, ULS	ULM2G		342.29	342.29	-					<b>†</b>
	•												
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18		UHL, UCL	ULM4L		65.32	65.32						
											ļ		
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18		UCL	ULM4G		342.29	342.29		+			+	-
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled le		UAL, UHL, UCL, UEQ, UEF, ULS	ULMBT		65.37	65.37				ļ		
	Oribundied Ecop Wodincation Removal of Bridged Tap Removal, per unbundied in		OLQ, OLI , OLS	OLIVIDI	-	03.37	03.37		+				+
SUB-LOOPS			+	1			<del>                                     </del>	_	+ + + + + + + + + + + + + + + + + + + +				
			-	1				_					1
Sı	ub-Loop Distribution		1										
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L	1	UEANL	USBSA		507.75	507.75			44.22	13.55		
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L		UEANL	USBSB		45.37	45.37			44.22	13.55	<b></b>	
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I		UEANL	USBSC		380.6	380.6			44.22	13.55	+	-
<del>                                     </del>	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone	1 1	UEANL UEANL	USBSD USBN2	11.09	111.15 131.88	111.15 62.05 90.69	13.42		44.22 44.22	13.55 13.55		+
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone	1 2		USBN2	15.72	131.88	62.05 90.69			44.22	13.55		+
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone	1 3		USBN2	18.49	131.88	62.05 90.69			44.22	13.55		1
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEANL	USBMC		45.43	45.43						
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zon€	1	O E / II 1 E	USBN4	17.64	158.41	88.58 99.64			44.22	13.55		
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zon€	2		USBN4	24.25	158.41	88.58 99.64			44.22	13.55	<del>                                     </del>	1
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone	3		USBN4	23.63	158.41	88.58 99.64	18.17	++	44.22	13.55		<del></del>
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEANL UEANL	USBMC USBR2	2.01	45.43 106.26	45.43 36.42 90.69	13.42	+	44.22	12.55		+
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEANL	USBRZ	3.01	45.43	36.42 90.69 45.43	13.42	+	44.22	13.55		+
l -	Sub-Loop 4-Wire Intrabuilding Network Cable (INC		UEANL	USBR4	6.7	118.76	48.93 99.64	18.17	+	44.22	13.55		<del>                                     </del>
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEANL	USBMC	T	45.43	45.43	.0.17	<del>                                     </del>			ĺ	1
					8.59	131.88	62.05 90.69	13.42		44.22	13.55		1
		I 1	UEF	UCS2X	8.59	131.00							
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 2	UEF	UCS2X	12.29	131.88	62.05 90.69	13.42		44.22	13.55		
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 2						13.42					

	4 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 1	UEF	UCS4X	9.81	158.41	88.58	99.64	18.17		44.22	13.55		
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 2		UCS4X	17.71	158.41	88.58	99.64	18.17		44.22	13.55		
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone	I 3		UCS4X	15.8	158.41	88.58	99.64	18.17		44.22	13.55		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEF	USBMC		45.43	45.43							
												1		
Sub-Loop I	Feeder													
			UEA,									1		
			UDN,UCL,UDL,UD											
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-		C	USBFW		507.75								
			UEA,									1		
			UDN,UCL,UDL,UD											
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u		C	USBFX		45.37	45.37							
	USL Feeder DS1 Set-up at DSX location, per DS1 terminatio		USL	USBFZ		523.87	11.34					+		
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zon	1	UEA	USBFA	11.16	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zon	2	UEA	USBFA	14.67	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zon	3	UEA	USBFA	18.43	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Order Coordination for Specified Conversion Time, per LSR		UEA	OCOSL	10.43	45.43	113.37	103.30	27.40		13.33	15.55	13.33	
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zon	1	UEA	USBFB	11.16	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zon	2	UEA	USBFB	14.67	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zon	3	UEA	USBFB	18.43	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Order Coordination for Specified Time Conversion, per LSR		UEA	OCOSL		45.43								
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon	1	UEA	USBFC	11.16	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zon	2	UEA	USBFC	14.67	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
										1		1		
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zon	3	UEA	USBFC	18.43	186.56	113.37	109.36	27.48		19.99	19.99	19.99	1
	Order Coordination For Specified Conversion Time, per LS		UEA	OCOSL		45.43								
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zon	1	UEA	USBFD	27.04	215.82	140.72	124.52	35.03		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zon	2	UEA	USBFD	34.46	215.82	140.72	124.52	35.03		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zon	3	UEA	USBFD	32.55	215.82	140.72	124.52	35.03		19.99	19.99	19.99	1
	Order Coordination For Specified Conversion Time, Per LS		UEA	OCOSL		45.43							I	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon	1	UEA	USBFE	27.04	215.82	140.72	124.52	35.03		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon	2	UEA	USBFE	34.46	215.82	140.72	124.52	35.03		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zon	3	UEA	USBFE	32.55	215.82	140.72	124.52	35.03		19.99	19.99	19.99	1
	Order Coordination For Specified Conversion Time, Per LS		UEA	OCOSL		45.43								
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone	1	UDN	USBFF	21.31	212.94	137.84	111.61	26.73		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	2	UDN	USBFF	26.15	212.94	137.84	111.61	26.73		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	3	UDN	USBFF	29.36	212.94	137.84	111.61	26.73		19.99	19.99	19.99	1
	Order Coordination For Specified Conversion Time, Per LS		UDN	OCOSL	20.00	45.43	107.01		20.70		10.00	10.00	10.00	
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	1	UDC	USBFS	21.31	212.94	137.84	111.61	26.73		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	2	UDC	USBFS	26.15	212.94	137.84	111.61	26.73		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible	3	UDC	USBFS	29.36	212.94	137.84	111.61	26.73		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder, 2 Wife ODC (IDSL compatible)  Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	1	USL	USBFG	79.79	204.38	129.38	124.52	35.03		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	2	USL	USBFG	155.94	204.38	129.38	124.52	35.03		19.99	19.99	19.99	1
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	3	USL	USBFG	290.5	204.38	129.38	124.52	35.03		19.99	19.99	19.99	1
	Order Coordination For Specified Conversion Time, Per LS		USL	OCOSL	7.47	45.43	00.04	400.07	04.00		40.00	40.00	40.00	
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone	1	UCL	USBFH	7.47	167.94	92.84	106.27	21.38		19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	2	UCL	USBFH	6	167.94	92.84	106.27	21.38		19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	3	UCL	USBFH	5.74	167.94	92.84	106.27	21.38		19.99	19.99	19.99	
	Order Coordination For Specified Conversion Time, per LS		UCL	OCOSL		45.43								
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	1	UCL	USBFJ	16.51	202.43	127.33	116.06	26.57		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	2	UCL	USBFJ	10.35	202.43	127.33	116.06	26.57		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	3	UCL	USBFJ	10.52	202.43	127.33	116.06	26.57		19.99	19.99	19.99	
	Order Coordination For Specified Conversion Time, per LS		UCL	OCOSL		45.43						<u> </u>		
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Lo	1	UDL	USBFN	26.27	204.38	129.28	124.52	35.03		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Lo	2	UDL	USBFN	26.62	204.38	129.29	124.52	35.03		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Lo	3	UDL	USBFN	25.21	204.38	129.28	124.52	35.03		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	1	UDL	USBFO	26.27	204.38	129.28	124.52	35.03		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zon€	2	UDL	USBFO	26.62	204.38	129.29	124.52	35.03		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	3	UDL	USBFO	25.21	204.38	129.28	124.52	35.03		19.99	19.99	19.99	
	Order Coordination For Specified Time Conversion, per LS		UDL	OCOSL		45.43								
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1	UDL	USBFP	26.27	204.38	129.28	124.52	35.03		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	2	UDL	USBFP	26.62	204.38	129.29	124.52	35.03		19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	3	UDL	USBFP	25.21	204.38	129.28	124.52	35.03		19.99	19.99	19.99	
	Order Coordination For Specified Conversion Time, per LS		UDL	OCOSL	-0.21	45.43	.20.20		55.00		10.00		10.00	
	oraci cooramation for opcomed conversion fillio, per Ec	-	ODL	J000L		70.70						+	+	
Unbundlad	Sub-Loop Modification	-										+	+	
Onbanalea	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W	$\rightarrow$										+	,	
	PR		UEF	ULM2X		356.5	12.29				44.22	13.55		
		-+-	UEF	ULIVIZA	1	330.3	12.29	1			44.22	13.33		
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR		UEF	111 84457		252.5	40.00				****	40.55		
			UEF	ULM4X	1	356.5	12.29	ļ	1		44.22	13.55		
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR											1 '		
	unloaded		UEF	ULM4T		561.8	14.33				44.22	13.55		
												<u> </u>		
					1 —		1	1				1 7	, T	
Unbundled	d Network Terminating Wire (UNTW)													
Unbundled		-	UENTW	UENPP	0.41	62.71	62.71				44.22	13.55		
Unbundled			UENTW	UENPP	0.41	62.71	62.71				44.22	13.55		
			UENTW	UENPP	0.41	62.71	62.71				44.22	13.55		
	Unbundled Network Terminating Wire (UNTW) per Pa		UENTW	UENPP UND12	0.41	62.71 87.36	62.71 57.58				44.22	13.55		
	Unbundled Network Terminating Wire (UNTW) per Pa				0.41									

Page 4 of 22 Version 2Q01: 08/30/01

		Network Interface Device Cross Connect - 4V		UENTW	UNDC4		11.83	11.83					44.22	13.55		
UNDUNDUE	D I OOD COL	ICENTRATION														
UNBUNDLE	D LOOP COR	NCENTRATION		111.0	LICTOA	200.44	050.00	050.00					40.00	40.00	40.00	40.00
		Unbundled Loop Concentration - System A (TR008 Unbundled Loop Concentration - System B (TR008		ULC	UCT8A	398.41 58.36	652.26 271.78	652.26 271.78					19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
		Unbundled Loop Concentration - System B (TR00):		ULC	UCT8B UCT3A	439.73	652.26	652.26					19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - System A (TR30):		ULC	UCT3B	98.34	271.78	271.78					19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - System B (11838) Unbundled Loop Concentration - DS1 Loop Interface Ca		ULC	UCTCO	5.52	126.85	92.35	33.65	9.42			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - ISDN Loop Interface (Brite Car		UDN	ULCC1	8.77	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - UDC Loop Interface (Brite Car		UDC	ULCCU	8.77	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration 2 Wire Voice-Loop Start or Ground Start Loop Interface		050	02000	0	2		10.01	10.7 1			10.00	10.00	10.00	10.00
		(POTS Card)		UEA	ULCC2	2.19	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface (SPOT\$ Card)		UEA	ULCCR	13.03	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca		UEA	ULCC4	7.77	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - TEST CIRCUIT Car		ULC	UCTTC	37.98	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa		UDL	ULCC7	11.51	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfa		UDL	ULCC5	11.51	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa		UDL	ULCC6	11.51	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
UNBUNDLE	D SUB-LOOF	CONCENTRATION (OUTSIDE CO)														
UNE OTHER	, PROVISIO	NING ONLY - NO RATE		UENTW	LINDDY											
		NID - Dispatch and Service Order for NID installation			UNDBX	1			1			4	1			
<b></b>	1	UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW	UENCE	1		1	1	1		1	1	<b>_</b>		
		Understand Contract Name Province Code N. D. C.		UEANL,UEF,UEQ,U									1			
<b></b>		Unbundled Contract Name, Provisioning Only - No Rate		ENTW	UNECN	1		1	-	1			+			
				UAL,UCL,UDC,UDI									1			
		Unboundlad Contest Name Bendalas Color and and		,UDN,UEA,UHL,UL												
		Unbundled Contact Name, Provisioning Only - no rate		С	UNECN	0	0									
				LIEA LIBALLIOL LIB		-										
				UEA,UDN,UCL,UD												
		Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra		С	USBFQ	0	0									
		Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra		UEA,USL,UCL,UDI	USBFR	0	0									
-				USL	CCOSF	0	0									
-		Unbundled DS1 Loop - Superframe Format Option - no ra Unbundled DS1 Loop - Expanded Superframe Format option - no ra		USL	CCOEF	0	0									
		Offibulided DST Loop - Expanded Superfiante Portifiat option - no to		USL	CCOEF	U	U									
HIGH CARA	CITY LINBLIN	DLED LOCAL LOOP														
mon oa a		onth minimum billing period														
-	NOTE. 4 III	High Capacity Unbundled Local Loop - DS3 - Per Mile per mon		UE3	1L5ND	15.33										
		High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor		UE3	UE3PX	382.95	905.04	529.05	239.5	167.53			31.38	31.38	3.94	3.94
		High Capacity Unbundled Local Loop - STS-1 - Per Mile per mor		UDLSX	1L5ND	15.33	000.01	020.00	200.0	107.00			01.00	01.00	0.0 1	0.0 .
		High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor		UDLSX	UDLS1	391.86	905.04	529.05	239.5	167.53			31.38	31.38	3.94	3.94
		ringir dupudity dribunated 200at 200p - 010 1 - 1 dollary formulation per mor		OBLOX	ODEO.	001.00	000.01	020.00	200.0	107.00			01.00	01.00	0.0 .	0.01
LOOP MAKE	-UP															
		Loop Makeup - Preordering Without Reservation, per working or spare facility queried														
		(Manual).		UMK	UMKLW		48.07	48.07								
		Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).		UMK	UMKLP		50.97	50.97								
		Loop MakeupWith or Without Reservation, per working or spare facility queried														
		(Mechanized)		UMK	PSUMK		0.6873	0.6873								
LINE SHARI	NO															
LINE SHAKI	110				1	1		+	1	<del>                                     </del>		1				
		Line Sharing Splitter, per System 96 Line Capaci	L	ULS	ULSDA	216.22	378.42	0	356.76	0		0				
		Line Sharing Splitter, per System 24 Line Capaci	I	ULS	ULSDB	54.05	378.42	0	356.76	0		0				
		Line Sharing Splitte, Per System, 8 Line Capaci	1	ULS	ULSD8	18.02	378.42	0	356.76	0		0				
		Line Sharing - per Line Activatio	_	ULS	ULSDC	0.61	37.09	21.24	20.07	9.85			44.22	13.55		
		Line Sharing - per Subsequent Activity per Line Rearrangeme	1	ULS	ULSDS		32.84	16.41	<del>                                     </del>	-			44.22	13.56		
		Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8 lines (16 pair)	1	ULS	ULSDG		57.83		11.41							
				UL3	OLODG		51.05		11.91							
UNBUNDLE	D TRANSPO	RT														
-	COMPLEX	DANGDORT (Charast)			1				1			1	+			
-	COMMON	RANSPORT (Shared)			+	0.000010:			1		-	+	+			
		Common Transport - Per Mile, Per MOI			+	0.0000121			1		-	+	+			
		Common Transport - Facilities Termination Per MO	-		1	0.0004672		+	-				+	-		
	NOTE: INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS3 = or	ne mont	n, DS3 and above four m	onths											
-	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE		LIATE OF	41.5301	0.040=		1	-	1			+			
<b></b>		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per mo		U1TVX	1L5XX	0.0167			1			1	+	1		
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month		U1TVX	U1TV2	24.3	81.25	54.94	33.54	13.82		1	31.38	31.38	9.8	9.8
	1	monu	l.	UIIVX	UIIVZ	24.3	01.25	54.94	JJ.54	13.82	1	1	31.38	31.38	9.8	9.8

Page 5 of 22

			1	1	r	1	1						
1	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per												
	month	U1TVX	1L5XX	0.0167									
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination pe												
	month	U1TVX	U1TR2	24.3	81.25	54.94	33.54	13.82		31.38	31.38	9.8	9
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month	U1TVX	1L5XX	0.0167									
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination p			04.00	04.05	54.04	00.54	40.00		04.00	04.00		_
	month	U1TVX	U1TV4	21.29	81.25	54.94	33.54	13.82		31.38	31.38	3.94	3.9
		=											
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per moi	U1TDX	1L5XX	0.0167								<del></del>	
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mo	U1TDX	U1TD5	16.76	81.26	54.94	33.54	13.82		31.38	31.38	3.94	3.9
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per moi	U1TDX	1L5XX	0.0282									
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mo	U1TDX	U1TD6	16.76	81.26	54.94	33.54	13.82		31.38	31.38	9.8	9
<b>——</b>													
I	INTEROFFICE CHANNEL - DEDICATED TRANSPORT - DS1												
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mon	U1TD1	1L5XX	0.3415								<del></del>	
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor	U1TD1	U1TF1	77.14	178.93	163.98	32.77	28.95		31.38	31.38	3.94	3.
<u> </u>	INTEROFFICE CHANNEL DEDICATED TRANSPORT DOS												
- 1	INTEROFFICE CHANNEL - DEDICATED TRANSPORT- DS3	LIATES	41.577	0.00									
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mo	U1TD3	1L5XX U1TF3	8.02 880.65	558.74	326.23	120.66	117.17		31.38	31.38	3.94	3.
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mol	U1TD3	U11F3	880.65	558.74	326.23	120.66	117.17		31.38	31.38	3.94	3.5
	INTEROFFICE CHANNEL - DEDICATED TRANSPORT- STS-1												
I	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mor	U1TS1	1L5XX	8.02									
					550.74	200.00	400.00	44747		24.20	24.20	2.04	
$\vdash$	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mo	U1TS1	U1TFS	880.55	558.74	326.26	120.66	117.17		31.38	31.38	3.94	3.
				-							+	++	
+		<del>-      </del>	1			+			-		+		
				-							+	++	
<u> </u>	LOCAL CHANNEL - DEDICATED TRANSPORT	<del>-    </del>	+	1		1					+		
	NOTE: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one mont	DS3 and above—four months											
	Local Channel - Dedicated - Ransport - minimum billing period - below 053=one mont	ULCVX	ULDV2	15.33	387.05	66.48	73.44	6.41	-	31.38	31.38	3.94	3.
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor	ULCVX	ULDR2	15.33	387.05	66.48	73.44	6.41		31.38	31.38	3.94	3
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor  Local Channel - Dedicated - 4-Wire Voice Grade per mon	UNCVX	ULDV4	16.54	387.05	67.35	74.38	7.35		31.38	31.38	3.94	3
	Local Channel - Dedicated - 4-Wire Voice Grade per mon	1 ULDD1	ULDF1	42.62	355.73	308.11	44.48	30.59		31.38	31.38	3.94	3.
	Local Channel - Dedicated - DS1 per month - Zone  Local Channel - Dedicated - DS1 per month - Zone	2 ULDD1	ULDF1	70.32	355.73	308.11	44.48	30.59		31.38	31.38	3.94	3.
	Local Channel - Dedicated - DS1 per month - Zone  Local Channel - Dedicated - DS1 per month - Zone	3 ULDD1	ULDF1	190.68	355.73	308.11	44.48	30.59		31.38	31.38	3.94	3
					355.73	308.11	44.48	30.59		31.38	31.38	3.94	
	Local Channel - Dedicated - DS3 - Per Mile per mon  Local Channel - Dedicated - DS3 - Facility Termination per mon	ULDD3 ULDD3	1L5NC ULDF3	11.93 446	905.04	529.05	239.5	167.53		31.38	31.38	3.94	3.
				11.93	905.04	529.05	239.5	167.53		31.38	31.38	3.94	
	Local Channel - Dedicated - STS-1- Per Mile per mon  Local Channel - Dedicated - STS-1 - Facility Termination per mor	ULDS1 ULDS1	1L5NC ULDFS	435.1	905.04	529.05	239.5	167.53		31.38	31.38	3.94	3.
	Local Channel - Dedicated - 313-1 - Facility Termination per mor	ULDST	ULDF3	433.1	905.04	529.05	239.3	107.55		31.30	31.30	3.94	
MULTIPLEXE	ne e			-							+	++	
WIOLTIFELAL	Channelization - DS1 to DS0 Channel Syster	UXTD1	MQ1	134.46	182.48	125.42	21.12	19.62		31.38	31.38	3.947	3.
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb	UDL	1D1DD	1.49	13.18	9.45	21.12	19.02		31.30	31.30	3.947	
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System - per mont	UDN	UC1CA	3.2	13.18	9.45					+	++	
	Voice Grade COCI - DS1 to DS0 Channel System - per mon	UEA	1D1VG	0.7012	13.18	9.45					+	++	
	DS3 to DS1 Channel System per monti	UXTD3	MQ3	180.03	357.07	188.36	66.66	63.79		31.38	31.38	3.94	3.
	STS1 to DS1 Channel System per monti	UXTS1	MQ3	180.03	337.07	100.30	00.00	03.73		31.38	31.38	3.94	3.
	DS3 Interface Unit (DS1 COCI) used with Loop per mont	USL	UC1D1	10.8	13.18	9.45				01.00	01.00	0.54	
	Doo interface offic (Do r oool) used with Eoop per mone	002	00101	10.0	10.10	3.40					+		
DARK FIBER											+		
D/U(I(T)DEI(	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local										+		
	Channel	UDF	1L5DC	97.65									
	NRC Dark Fiber - Local Channe	UDF	UDFC4	37.00	1281.02	276.34	635.52	396.21		31.26	31.26	3.94	3
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Interoffice		00104		1201.02	270.04	000.02	030.E1		31.20	01.20	0.54	
	Channel	UDF	1L5DF	36.41		1							
<b> </b>	NRC Dark Fiber - Interoffice Channe	UDF	UDF14	50.41	1281.02	276.34	635.52	396.21	1	31.38	31.38	3.94	3
. +	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local	351	ODI 14		1201.02	270.04	000.02	000.21	1	01.00	01.00	5.54	
	Loop	UDF	1L5DL	97.65		1							
	NRC Dark Fiber - Local Loop	UDF	UDFL4	07.00	1281.02	276.34	635.52	396.21		31.38	31.38	3.94	3.
TRANSPORT	OTHER	05.	05.2.		1201.02	270.01	000.02	000.21		01.00	01.00	0.0 .	
						1			1		+	<b>—</b>	
											1		
											1		
(	Optional Features & Functions:										1		
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chann	UNC1X	CCOEF		185.26	23.86	1.99	0.78		29.33	3.93		
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chanr	UNC1X	CCOSF		185.26	23.86	1.99	0.78		29.33	3.93		
	TEN DIGIT SCREENING										1		-
8XX ACCESS	8XX Access Ten Digit Screening, Per Ca	OHD	İ	0.0005227							1		
8XX ACCESS			N8R1X		6.38	0.9583				27.84	27.84		
8XX ACCESS		OHD			22.63	2.73				27.84	27.84		-
8XX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserva	OHD											•
BXX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reservi 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation		N8FTX		22.63	2.73				27.84	27.84		
BXX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv- 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translatio	OHD	N8FTX N8FCX		22.63 5.64	2.73				27.84	27.84 27.84	+	
8XX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv 8XX Access Ten Digit Screening, Per 8XX No. Established WiO POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translatio 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb	OHD OHD											
8XX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv. 8XX Access Ten Digit Screening, Per 8XX No. Established WIO POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translatio 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb. 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested	OHD OHD OHD				2.82					27.84		-
BXX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv- 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translatio 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested Per 8XX No.	OHD OHD OHD	N8FCX N8FMX		5.64 6.6					27.84 27.84	27.84 27.84		
8XX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserve 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested Per 8XX No. 8XX Access Ten Digit Screening, Change Charge Per Reque	OHD OHD OHD	N8FCX		5.64 6.6 7.34	2.82 3.78				27.84	27.84 27.84 27.84		
BXX ACCESS	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv- 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translatio 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested Per 8XX No.	OHD OHD OHD OHD OHD	N8FCX N8FMX N8FAX		5.64 6.6	2.82 3.78				27.84 27.84 27.84	27.84 27.84		

Page 6 of 22

	LIDB Common Transport Per Quer	OQT		0.0000442						T	
	LIDB Validation Per Quer	OQU		0.0145288					+	+	
		OQT, OQU	NRPBX	0.0143200	61.62			27.84	27.84		
	LIDB Originating Point Code Establishment or Chanç	001,000	INCEDA		01.02			21.04	21.04		
CICNIALING	COST									<del></del>	
SIGNALING											
	CCS7 Signaling Termination, Per STP Por	1DB	PT8SX	156.33				19.99	19.99	19.99	19.99
	CCS7 Signaling Usage, Per TCAP Message	1DB		0.0001108							
	CCS7 Signaling Connection, Per link (A link	1DB	TPP++	21.79	277.07	277.07		19.99	19.99	19.99	19.99
	CCS7 Signaling Connection, Per link (B link) (also known as D lin	1DB	TPP++	21.79	277.07	277.07		19.99	19.99	19.99	19.99
	CCS7 Signaling Usage, Per ISUP Messagi	1DB		0.0000452							
	CCS7 Signaling Usage Surrogate, per link per LAT	1DB	STU56	396.55				19.99	19.99	19.99	19.99
	CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per										
	STP affected	1DB	CCAPO		40	40		19.99	19.99	19.99	19.99
	CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per \$										
	Affected	1DB	CCAPD		8	8		19.99	19.99	19.99	19.99
	, modes		00/11/2					10.00	10.00	10.00	10.00
E911 SERV	NCE								+	+	
ESIT SERV	TICE .									<del></del>	
									+	+	
CALLING N	IAME (CNAM) SERVICE										
	CNAM for DB Owners, Per Query	OQV		0.016							
L	CNAM for Non DB Owners, Per Quen	OQV		0.01				1	1		
								1	1		
									1	LT	
	CNAM (Non-Databs Owner), NRC, applies when using the Character Based User										
	Interface (CHUI)	OQV	CDDCH		595	595		27.84	27.84		
								1			
									1		
LNP OUER	Y SERVICE										
				1			<del>                                     </del>	+	+	<del>                                     </del>	
						+		1	+	<del>                                     </del>	
<b>—</b>	OPERATOR SERVICES AND DIRECTORY ASSISTANCE							1	+	<del>                                     </del>	
	OFERATOR SERVICES AND DIRECTORT ASSISTANCE								+	+	
00504705									+	+	
OPERATOR	R CALL PROCESSING										
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIC			1.2							
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIE			1.24							
	Oper. Call Processing - Fully Automated, per Call - Using BST LIE			0.2							
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE			0.2							
INWARD OF	PERATOR SERVICES										
	Inward Operator Services - Verification, Per Minu			1.15							
	Inward Operator Services - Verification and Emergency Interrupt - Per Min			1.15							
BRANDING	- OPERATOR CALL PROCESSING										
	Recording of Custom Branded OA Announcement		CBAOS		7000	7000		19.99	19.99	19.99	19.99
	Loading of Custom Branded OA Announcement per shelf/NAV		CBAOL		500	500		19.99	19.99		
						***					
DIDECTOR'	Y ASSISTANCE SERVICES								+	+	
DIRECTOR	DIRECTORY ASSISTANCE ACCESS SERVICE										
				0.275							
	Directory Assistance Access Service Calls, Charge Per Cε			0.275							
<b></b>	DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)						1	+	<del> </del>		
-	Directory Assistance Call Completion Access Service (DACC), Per Call Attern			0.1				1	4		
L											
	UNBRANDING								1		
	DIRECTORY TRANSPORT										
			1	37.2	534.81	462.81		87.99	87.99	3.11	3.11
	Directory Transport - Local Channel DS										
	Directory Transport - Local Channel DS Directory Transport - DS1 Level Interoffice Per Mi			0.7598		1					
	Directory Transport - DS1 Level Interoffice Per Mil				216.27	162.7		39.63	39.63	3.11	3.11
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Termination			0.7598 94.98	216.27	162.7		39.63	39.63	3.11	3.1
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminati Switched Common Transport Per DA Access Service Per C≀			0.7598 94.98 0.000327	216.27	162.7		39.63	39.63	3.11	3.1
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatic Switched Common Transport Per DA Access Service Per Ca Switched Common Transport Per DA Access Service Per Call Per Mi			0.7598 94.98 0.000327 0.0000303	216.27	162.7		39.63	39.63	3.11	3.11
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per Ca Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Cs			0.7598 94.98 0.000327 0.0000303 0.0024809	216.27	162.7		39.63	39.63	3.11	3.1
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatic Switched Common Transport Per DA Access Service Per Ca Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Ca Directory Transport - DA Interconnection Per DA Service Ca			0.7598 94.98 0.000327 0.0000303							
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per Ca Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Cs			0.7598 94.98 0.000327 0.0000303 0.0024809	216.27	162.7		39.63 87.99	39.63 87.99	3.11	3.11
	Directory Transport - DS1 Level Interoffice Per Mi  Directory Transport - DS1 Level Interoffice Per Facility Terminatit  Switched Common Transport Per DA Access Service Per Ca  Switched Common Transport Per DA Access Service Per Call Per Mi  Access Tandem Switching Per DA Access Service Per Call Per Mi  Directory Transport - DA Interconnection Per DA Service Call  Directory Transport - Installation NRC, Per Trunk or Signaling Connection			0.7598 94.98 0.000327 0.0000303 0.0024809							
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatic Switched Common Transport Per DA Access Service Per Ca Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Ca Directory Transport - DA Interconnection Per DA Service Ca Directory Transport - Installation NRC, Per Trunk or Signaling Connection  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)			0.7598 94.98 0.000327 0.000303 0.0024809 0.000269							
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per Ca Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Ca Directory Transport - DA Interconnection Per DA Service Ca Directory Transport - Installation NRC, Per Trunk or Signaling Connecti  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service Charge Per Listir		DREOF	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269							
DDANINA	Directory Transport - DS1 Level Interoffice Per Mi  Directory Transport - DS1 Level Interoffice Per Facility Terminatit  Switched Common Transport Per DA Access Service Per Ca  Switched Common Transport Per DA Access Service Per Call Per Mi  Access Tandem Switching Per DA Access Service Per Call Per Mi  Access Tandem Switching Per DA Access Service Per Call Per Mi  Directory Transport - DA Interconnection Per DA Service Call Per Mi  Directory Transport - Installation NRC, Per Trunk or Signaling Connection  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)  Directory Assistance Data Base Service Charge Per Listir  Directory Assistance Data Base Service (Data)		DBSOF	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269							
BRANDING	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per C¢ Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Cc Directory Transport - DA Interconnection Per DA Service Cc Directory Transport - Installation NRC, Per Trunk or Signaling Connecti  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)  Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per mon!			0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81	11					
BRANDING	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminativ Switched Common Transport Per DA Access Service Per Cat Switched Common Transport Per DA Access Service Per Cat Recess Tandem Switching Per DA Access Service Per Cat Access Tandem Switching Per DA Access Service Per Cat Directory Transport - DA Interconnection Per DA Service Cat Directory Transport - Installation NRC, Per Trunk or Signaling Connectiv  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per moni - DIRECTORY ASSISTANCE Custom Branding Announcement, per Recording to be used with the provision of DA	AMT	CBADA	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81	11 3000					
BRANDING	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per C¢ Switched Common Transport Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Call Per Mi Access Tandem Switching Per DA Access Service Per Cc Directory Transport - DA Interconnection Per DA Service Cc Directory Transport - Installation NRC, Per Trunk or Signaling Connecti  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)  Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per mon!	AMT AMT		0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81	11					
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per Cat Switched Common Transport Per DA Access Service Per Cat Switched Common Transport Per DA Access Service Per Cat Access Tandem Switching Per DA Access Service Per Cat Directory Transport - DA Interconnection Per DA Service Cat Directory Transport - Installation NRC, Per Trunk or Signaling Connectiv  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per monl DIRECTORY ASSISTANCE  Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch		CBADA	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81	11 3000					
BRANDING	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per Cat Switched Common Transport Per DA Access Service Per Cat Switched Common Transport Per DA Access Service Per Cat Access Tandem Switching Per DA Access Service Per Cat Directory Transport - DA Interconnection Per DA Service Cat Directory Transport - Installation NRC, Per Trunk or Signaling Connectiv  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per monl DIRECTORY ASSISTANCE  Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch		CBADA	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81	11 3000					
	Directory Transport - DS1 Level Interoffice Per Facility Terminativ  Directory Transport - DS1 Level Interoffice Per Facility Terminativ  Switched Common Transport Per DA Access Service Per Cat  Switched Common Transport Per DA Access Service Per Cat  Access Tandem Switching Per DA Access Service Per Cat Per Minder Switching Per DA Access Service Per Cat Per Minder Switching Per DA Access Service Per Cat Per Minder Switching Per DA Service Cat Per Minde		CBADA CBADC	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81 3000 690	3000 690		87.99	87.99		
	Directory Transport - DS1 Level Interoffice Per Mi Directory Transport - DS1 Level Interoffice Per Facility Terminatit Switched Common Transport Per DA Access Service Per Cat Switched Common Transport Per DA Access Service Per Cat Switched Common Transport Per DA Access Service Per Cat Access Tandem Switching Per DA Access Service Per Cat Directory Transport - DA Interconnection Per DA Service Cat Directory Transport - Installation NRC, Per Trunk or Signaling Connectiv  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per monl DIRECTORY ASSISTANCE  Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch		CBADA	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81	11 3000					
SELECTIVE	Directory Transport - DS1 Level Interoffice Per Facility Terminatii  Directory Transport - DS1 Level Interoffice Per Facility Terminatii  Switched Common Transport Per DA Access Service Per Cat  Switched Common Transport Per DA Access Service Per Cat  Switched Common Transport Per DA Access Service Per Cat  Access Tandem Switching Per DA Access Service Per Cat  Directory Transport - DA Interconnection Per DA Service Cat  Directory Transport - Installation NRC, Per Trunk or Signaling Connectii  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)  Directory Assistance Data Base Service Charge Per Listir  Directory Assistance Data Base Service, per moni  - DIRECTORY ASSISTANCE  Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch  E ROUTING  Selective Routing Per Unique Line Class Code Per Request Per Swit		CBADA CBADC	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81 3000 690	3000 690		87.99	87.99		
SELECTIVE	Directory Transport - DS1 Level Interoffice Per Facility Terminativ  Directory Transport - DS1 Level Interoffice Per Facility Terminativ  Switched Common Transport Per DA Access Service Per Cat  Switched Common Transport Per DA Access Service Per Cat  Access Tandem Switching Per DA Access Service Per Cat Per Minder Switching Per DA Access Service Per Cat Per Minder Switching Per DA Access Service Per Cat Per Minder Switching Per DA Service Cat Per Minde	AMT	CBADA CBADC	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81 3000 690	3000 690		87.99	87.99		
SELECTIVE	Directory Transport - DS1 Level Interoffice Per Facility Terminatii  Directory Transport - DS1 Level Interoffice Per Facility Terminatii  Switched Common Transport Per DA Access Service Per Cat  Switched Common Transport Per DA Access Service Per Cat  Switched Common Transport Per DA Access Service Per Cat  Access Tandem Switching Per DA Access Service Per Cat  Directory Transport - DA Interconnection Per DA Service Cat  Directory Transport - Installation NRC, Per Trunk or Signaling Connectii  DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)  Directory Assistance Data Base Service Charge Per Listir  Directory Assistance Data Base Service, per moni  - DIRECTORY ASSISTANCE  Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch  E ROUTING  Selective Routing Per Unique Line Class Code Per Request Per Swit		CBADA CBADC	0.7598 94.98 0.000327 0.000303 0.0024809 0.000269	407.81 3000 690	3000 690		87.99	87.99		

	Virtual Co	allocation 2 Wire Cross Connects (Loop) for Line Splittin		UEPSR, UEPSB	VE1LS	0.3648	41.5	38.94		1	1	19.99	19.99	19.99	19.9
	Virtual Co	bllocation-2 Wire Cross Connects (Loop) for Line Splittin I bllocation - 2-wire Cross Connects (por	_	UEPSK, UEPSB	VE1LS VE1R2	0.3648	41.5	38.94				19.99	19.99	19.99	19.9
				and the second											
		ollocation - 4-wire Cross Connects (loop		uea,uhl,ucl,ud	UEAC4	0.7297	41.56	38.9				19.99	19.99	19.99	19.9
		ollocation - 4-wire Cross Connects (por			VE1R4	0.7297	41.56	38.9				19.99	19.99	19.99	19.9
		ollocation - 2-Fiber Cross Connect		CLO	CNC2F	15.06	69.28	48.89				19.99	19.99	19.99	19.9
	Virtual Co	ollocation - 4-Fiber Cross Connect:		CLO	CNC4F	27.08	84.07	63.68				19.99	19.99	19.99	19.9
	Virtual Co	ollocatin - DS1 Cross Connect:		USL,ULC,CLO	CNC1X	7.5	155	14							
AIN SELEC	TIVE CARRIER ROUTIN	IG													
		Service Establishment		SRC	SRCEC		391788					19.99	19.99	19.99	19.9
		e Establishment		SRC	SRCEO		320.53	320.53				19.99	19.99	19.99	19.9
		NRC, per end user		SRC	SRCLP		2.06	2.06				19.99	19.99	19.99	19.9
	Query NR	RC, per query		SRC		0.000448									
AIN - BELL	SOUTH AIN SMS ACCES	SS SERVICE													
	AIN SMS	Access Service - Service Establishment, Per State, Initial Setup			CAMSE		296.16	296.16				27.84	27.84		
	AIN SMS	Access Service - Port Connection - Dial/Shared Access			CAMDP		87.29	87.29				27.84	27.84		
		Access Service - Port Connection - ISDN Access			CAM1P		87.29	87.29				27.84	27.84		
		Access Service - User Identification Codes - Per User ID Code			CAMAU		202.08	202.08				27.84	27.84		
	AIN SMS	Access Service - Security Card, Per User ID Code, Initial or Replacement			CAMRC		172.26	172.26				27.84	27.84		
	AIN SMS	Access Service - Storage, Per Unit (100 Kilobytes)				0.0028									
	AIN SMS	Access Service - Session, Per Minute				0.0942966									
		Access Service - Company Performed Session, Per Minute				2.07									
	AIN SWIS	Access Service - Company renormed Session, Fer Minute				2.07									
				1	1	1		1		ļ	-	1			
AIN - BELL	SOUTH AIN TOOLKIT SI			1	1					1		1			
	AIN Toolk	kit Service - Service Establishment Charge, Per State, Initial Setup			BAPSC		291.41	291.41				27.84	27.84		
		kit Service - Training Session, Per Customer			BAPVX		8333	8333				27.84	27.84		
		kit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt		1	BAPTT		73.02	73.02		1	1	27.84	27.84		
<b>—</b>			-	1	BAPTD	1	73.02	73.02		1	1	27.84	27.84	-	
L	AIN 100lk	kit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay		1	BAPID	1	13.02	13.02		ļ	-	27.84	21.84		
								I							
		kit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate			BAPTM		73.02	73.02				27.84	27.84		
1		kit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP			BAPTO		150.25	150.25				27.84	27.84		
		kit Service - Trigger Access Charge, Per Trigger, Per DN, CDP			BAPTC		150.25	150.25				27.84	27.84		
	AIN Toolk	kit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code			BAPTF		150.25	150.25				27.84	27.84		
					DAFII	0.0250662	130.23	130.23			1	27.04	27.04		
		kit Service - Query Charge, Per Query				0.0250662									
		kit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per													
	Query					0.0062979									
	AIN Toolk	kit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes				1.73									
	AIN Toolis	kit Service - Monthly report - Per AIN Toolkit Service Subscription			BAPMS	15.93	72.15	72.15				27.84	27.84		
		kit Service - Special Study - Per AIN Toolkit Service Subscription			BAPLS	0.0872769	47.35	47.35				27.84	27.84		
		kit Service - Call Event Report - Per AIN Toolkit Service Subscription			BAPDS	15.84	72.15	72.15				27.84	27.84		
	AIN Toolk	kit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			BAPES	0.0029092	47.35	47.35				27.84	27.84		
ODUF/FDO	UF/ADUF/CMDS														
ODOI7EDO	O TABOT TOMBO														
	400F00 B4II V 11040	DE EU E (ADUE)													
	ACCESS DAILY USAG														
	ADUF: M	Message Processing, per messag				0.004									
	ADUF: D	Oata Transmission (CONNECT:DIRECT), per messag				0.001									
	<b>ENHANCED OPTIONA</b>	AL DAILY USAGE FILE (EODUF)		1	1 -					1					
		Message Processing, per messag		1	i .	0.004				1					
	LODOI .	g	_	1	<b> </b>	5.504		1		t	<b>†</b>	<u> </u>	<b> </b>		
<b></b>	OPTIONAL DAILY USA	ACE EILE (ODLIE)	_	1	1	1		1		1	1			+	
<b></b>				1	1	0.0002862		<del>                                     </del>	<b> </b>	<del>                                     </del>	<del>                                     </del>	1	-		
<b></b>	ODUF: R	Recording, per messag			1				1	1	1			-	
<u> </u>	ODUF: M	Message Processing, per messag				0.0032344				ļ	1				
	ODUF: M	Message Processing, per Magnetic Tape provisions				54.72				ļ					
	ODUF: D	Data Transmission (CONNECT:DIRECT), per messag				0.0000357									
	1 1 -			1				<u> </u>							
									1	1	1	1			-
ENHANCE	D EXTENDED LINK (EEL	_s)													
ENHANCE	EXTENDED LINK (EEL	.s)												1	
ENHANCE	,		Et Laudi	ordale ELI: Nachvilla	TN: Now Orlan	ne I A·									
ENHANCE	NOTE: New EELs avai	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL;			TN; New Orlea	ns, LA;									
ENHANCE	NOTE: New EELs avail	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex	cept Sw	itch As Is Charge.					NE- (N						
ENHANCE	NOTE: New EELs avail NOTE: Charlotte-Gasto NOTE: In all states, EE	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which ar	cept Sw e conve	itch As Is Charge. ted to UNE rates. A S	Switch As Is Cha		currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avail NOTE: Charlotte-Gasto NOTE: In all states, EE	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex	cept Sw e conve	itch As Is Charge. ted to UNE rates. A S	Switch As Is Cha		currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	.)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gaste NOTE: In all states, EE NOTE: In Georgia, the	liable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GAP:	cept Sw e conve	itch As Is Charge. ted to UNE rates. A S	Switch As Is Cha		currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	.)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gaste NOTE: In all states, EE NOTE: In Georgia, the	liable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GAP:	cept Sw e conve	itch As Is Charge. ted to UNE rates. A S	Switch As Is Cha		currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below es EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GA P:  E EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	cept Sw e conve	itch As Is Charge. ted to UNE rates. A S .(No Switch As Is Ch	Switch As Is Charge.)	arge applies to	currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the	liable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GAP:	ccept Sw re conver SC order	itch As Is Charge. ted to UNE rates. A S	Switch As Is Cha		currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gasts NOTE: In all states, EE NOTE: In Georgia, the 2-WIRE VOICE GRADE First 2-Wi	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which at EEL network elements apply to ordinarily combined network elements per the GA P:  EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) lire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	ccept Sw re conver SC order	itch As Is Charge. ted to UNE rates. A S .(No Switch As Is Ch.	Switch As Is Charge.)  UEAL2	arge applies to	currently combi	ned facilities converted to U	NEs.(Non-rei	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gasts NOTE: In all states, EE NOTE: In Georgia, the 2-WIRE VOICE GRADE First 2-Wi	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below es EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GA P:  E EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	ccept Sw re conver SC order	itch As Is Charge. ted to UNE rates. A S .(No Switch As Is Ch	Switch As Is Charge.)	arge applies to	currently combi	ned facilities converted to U	NEs.(Non-re	surring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the 2-WIRE VOICE GRADE First 2-Wi	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below experience of the shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GA Power of the shown of the state of the shown of the same of the shown of the sho	re converse	itch As Is Charge. ted to UNE rates. A \$. (No Switch As Is Ch.  UNCVX  UNCVX	Switch As Is Charge.)  UEAL2  UEAL2	21.57 32.53	currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the  2-WIRE VOICE GRADE First 2-Wi First 2-Wi	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GA P:  EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) irie VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	ccept Sw re conver SC order	itch As Is Charge. ted to UNE rates. A S. (No Switch As Is Ch.  UNCVX  UNCVX  UNCVX  UNCVX	UEAL2 UEAL2 UEAL2	21.57 32.53 43.08	currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the  2-WIRE VOICE GRADE First 2-Wi First 2-Wi	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GA P:  EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) irie VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	re converse	itch As Is Charge. ted to UNE rates. A \$. (No Switch As Is Ch.  UNCVX  UNCVX	Switch As Is Charge.)  UEAL2  UEAL2	21.57 32.53	currently combi	ned facilities converted to U	NEs.(Non-red	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the 2-WIRE VOICE GRADE First 2-Wi First 2-Wi Interoffice	iliable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below es EL network elements shown below also apply to currently combined facilities which are beEL network elements apply to ordinarily combined network elements per the GA P: EEXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) ire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zon ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zon ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zon ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zon ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zon ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zon ire Transport - Dedicated - DS1 combination - Per Mile per moi	re converse	itch As Is Charge. ted to UNE rates. A S. (No Switch As Is Ch.  UNCVX  UNCVX  UNCVX  UNCVX	UEAL2 UEAL2 UEAL2	21.57 32.53 43.08	currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply				
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the  2-WIRE VOICE GRADE First 2-Wi First 2-Wi Interoffice Interoffice	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below es EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GA Piet EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) ire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone are VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone are Transport - Dedicated - DS1 combination - Pacility Termination per mo	re converse	itch As Is Charge. Ted to UNE rates. A 4 (No Switch As Is Ch  UNCVX  UNCVX  UNCVX  UNCVX  UNCVX  UNCIX  UNC1X	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	21.57 32.53 43.08 0.3415 77.14	currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCE	NOTE: New EELs avai NOTE: Charlotte-Gaste NOTE: In all states, EE NOTE: In Georgia, the  2-WIRE VOICE GRADE First 2-Wi First 2-Wi Interoffice Interoffice DS1 Char	iliable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC: Greensboro-Winston Salem-High Point, NC. Use all rates below ex EL network elements shown below also apply to currently combined facilities which are been the selements apply to ordinarily combined network elements per the GA P: EEXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) ire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a	re converse	itch As Is Charge. Ited to UNE rates. A 5 (No Switch As Is Ch.  UNCVX  UNCVX  UNCVX  UNCVX  UNC1X  UNC1X  UNC1X  UNC1X	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 ULEAL2 ULEAL2 ULEAL2 ULEAL2 ULEAL2 ULEAL2 ULEAL2	21.57 21.57 32.53 43.08 0.3415 77.14 134.46	currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			
ENHANCEI	NOTE: New EELs avai NOTE: Charlotte-Gast NOTE: In all states, EE NOTE: In Georgia, the 2-WIRE VOICE GRADE First 2-Wi First 2-Wi Interoffice Interoffice DS1 Char	ilable in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; onia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below es EL network elements shown below also apply to currently combined facilities which are EEL network elements apply to ordinarily combined network elements per the GA Piet EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) ire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone ire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone are VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone are Transport - Dedicated - DS1 combination - Pacility Termination per mo	re converse	itch As Is Charge. Ted to UNE rates. A 4 (No Switch As Is Ch  UNCVX  UNCVX  UNCVX  UNCVX  UNCVX  UNCIX  UNC1X	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	21.57 32.53 43.08 0.3415 77.14	currently combi	ned facilities converted to U	NEs.(Non-re	curring rates	do not apply	)			

Page 8 of 22 Version 2Q01: 08/30/01

	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combinati												
	- Zone 2	2	UNCVX	UEAL2	32.53								
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination	_											
	Zone 3	3	UNCVX	UEAL2	43.08								_
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mor  Nonrecurring Currently Combined Network Elements Switch -As-Is Chan		UNCVX UNC1X	1D1VG UNCCC	0.7012	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
	Nonecuring Currently Combined Network Elements Switch -As-is Chair		UNCIX	UNCCC		11.21	11.21	13.33	13.55	31.30	31.30	3.54	-
4-WIRE	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)												
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone												
	1	1	UNCVX	UEAL4	29.47								
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone												
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zome	2	UNCVX	UEAL4	44.44								+
	3	3	UNCVX	UEAL4	58.85								
_	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.3415								
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Mor		UNC1X	U1TF1	77.14								
	Channelization - Channel System DS1 to DS0 combination Per Mon		UNC1X	MQ1	134.46								
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mor		UNCVX	1D1VG	0.7012								
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1		UNCVX	UEAL4	29.47								
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport		UNCVX	UEAL4	29.47								+
	Combination - Zone 2	2	UNCVX	UEAL4	44.44								
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport												
	Combination - Zone (	3	UNCVX	UEAL4	58.85								1
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC1X	UNCCC		11.21	11.21	13.99	13.99	31.38	31.38	3.94	1
4 WIDE	E 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	-		-	1								1
4-WIKE	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zohe	-											+-
	1	1	UNCDX	UDL56	34.26								
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone												
	2	2	UNCDX	UDL56	51.67								
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone												
	3	3	UNCDX	UDL56	68.43								
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.3415								
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mo		UNC1X	U1TF1	77.14					31.38	31.38	3.94	
_	Channelization - Channel System DS1 to DS0 combination Per Mon		UNC1X	MQ1	134.46					31.30	31.30	3.54	
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb		UNCDX	1D1DD	1.49								
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport												
	Combination - Zone 1	1	UNCDX	UDL56	34.26					31.38	31.38	3.94	1
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	LINODY	LIDI FC	54.07					31.38	24.20	2.04	
	Combination - Zone 2 Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	- 2	UNCDX	UDL56	51.67					31.38	31.38	3.94	-
	Combination - Zone (	3	UNCDX	UDL56	68.43					31.38	31.38	3.94	1
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kt		UNCDX	1D1DD	1.49								
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC1X	UNCCC		11.21	11.21	13.99	13.99	31.38	31.38	3.94	1
4 MIDE	E 64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)												
4-WIKE	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone												+
	1	1	UNCDX	UDL64	34.26								
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone												
	2	2											
			UNCDX	UDL64	51.67								
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone												
	3	3	UNCDX	UDL64	68.43								
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor												
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNCDX UNC1X	UDL64	68.43 0.3415								
	3		UNCDX	UDL64 1L5XX	68.43								
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon		UNCDX UNC1X UNC1X UNC1X	UDL64 1L5XX U1TF1 MQ1	68.43 0.3415 77.14 134.46								
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt		UNCDX UNC1X	UDL64 1L5XX U1TF1	68.43 0.3415 77.14	0	0						
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD	68.43 0.3415 77.14 134.46 1.49	0	0						
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone '		UNCDX UNC1X UNC1X UNC1X	UDL64 1L5XX U1TF1 MQ1	68.43 0.3415 77.14 134.46	0	0						
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	1	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64	68.43 0.3415 77.14 134.46 1.49	0	0						
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :		UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD	68.43 0.3415 77.14 134.46 1.49	0	0						
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	1	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64	68.43 0.3415 77.14 134.46 1.49	0	0						
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :	1 2	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43	0	0						
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt	1 2	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64 1D1DD	68.43 0.3415 77.14 134.46 1.49 34.26 51.67								
	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :	1 2	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43	0	0	13.99	13.99	31.38	31.38	3.94	
4 WIDE	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Nonrecurring Currently Combined Network Elements Switch -As-Is Char	1 2	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64 1D1DD	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43			13.99	13.99	31.38	31.38	3.94	
4-WIRE	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone ' Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Nonrecurring Currently Combined Network Elements Switch - As-Is Char	1 2 3	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64 1D1DD UNCCC	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43			13.99	13.99	31.38	31.38	3.94	
4-WIRE	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo  Channelization - Channel System DS1 to DS0 combination Per Mon  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Nonrecurring Currently Combined Network Elements Switch -As-Is Char  EDS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)  4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zoni	1 2 3	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX	UDL64 11.5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64 UDL64 UDL64 UDL65 UDL64 UDL64 UDL64 UDL64 UDL64 UDL64 UDL64 UDL64	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43 1.49			13.99	13.99	31.38	31.38	3.94	
4-WIRE	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone ' Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Nonrecurring Currently Combined Network Elements Switch - As-Is Char	1 2 3	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64 1L5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64 1D1DD UNCCC	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43			13.99	13.99	31.38	31.38	3.94	
4-WIRE	3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo Channelization - Channel System DS1 to DS0 combination Per Mon  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Nonrecurring Currently Combined Network Elements Switch - As-Is Char  EDS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	1 2 3	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNC1X UNC1X	UDL64 11.5XX  U1TF1 MQ1  1D1DD  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UNCCC	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43 1.49			13.99	13.99	31.38	31.38	3.94	
4-WIRE	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo  Channelization - Channel System DS1 to DS0 combination Per Mon  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone '  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Nonrecurring Currently Combined Network Elements Switch - As-Is Char  E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)  4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone    4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone    1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone    1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone    1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone    1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	1 2 3	UNCDX UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX	UDL64 11.5XX U1TF1 MQ1 1D1DD UDL64 UDL64 UDL64 UDL64 UDL64 UDL64 UNCCC USLXX USLXX USLXX USLXX	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43 1.49 59.61 89.9 119.06 0.3415			13.99	13.99	31.38	31.38	3.94	
4-WIRE	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor  Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mo  Channelization - Channel System DS1 to DS0 combination Per Mon  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :  OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kt Nonrecurring Currently Combined Network Elements Switch -As-Is Char  EDS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)  4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone - Vivie DS1 Digital Loop in Combina	1 2 3	UNCDX UNC1X UNC1X UNC1X UNC1X UNC1X UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNCDX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UDL64 11.5XX  U1TF1 MQ1  1D1DD  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UDL64  UNCCC	68.43 0.3415 77.14 134.46 1.49 34.26 51.67 68.43 1.49			13.99	13.99	31.38	31.38	3.94	

4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)												
First DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	59.61								
First DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	89.9								
First DS1Loop in DS3 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	119.06								
Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Mor		UNC3X	1L5XX	8.02								
Interoffice Transport - Dedicated - DS3 - Facility Termination per mor		UNC3X	U1TF3	880.65								
DS3 to DS1 Channel System combination per mont		UNC3X	MQ3	180.03								
DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	10.8								
Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	59.61								
Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	89.9							1	
Additional DS1Loop in DS3 Interoffice Transport Combination - Zone	3	UNC1X	USLXX	119.06								
DS3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	10.8								
Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNC3X	UNCCC		11.21	11.21	13.99	13.99	31.38	31.38	3.94	
											1	
2-WIRE VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)											1	
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zon	1	UNCVX	UEAL2	21.57								
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zon	2	UNCVX	UEAL2	32.53								
2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zon	3	UNCVX	UEAL2	43.08								
Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Moi		UNCVX	1L5XX	0.0167								
Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination												
per month		UNCVX	U1TV2	24.3					31.38	31.38	3.94	
Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCVX	UNCCC		11.21	11.21	13.99	13.99	31.38	31.38	3.94	
4-WIRE VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)												
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zon	1	UNCVX	UEAL4	29.47								
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zon	2	UNCVX	UEAL4	44.44								
4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zon	3	UNCVX	UEAL4	58.85								
Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Moi		UNCVX	1L5XX	0.0167								
Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination												
per month		UNCVX	U1TV4	21.29							1	
Nonrecurring Currently Combined Network Elements Switch -As-Is Char	-	UNCVX	UNCCC		11.21	11.21	13.99	13.99	31.38	31.38	3.94	
	-							10.00				
DS3 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)	-											
High Capacity Unbundled Local Loop - DS3 combination - Per Mile per moi	-	UNC3X	1L5ND	15.33								
riigii dapadiiy dibandida 2004 200 00 00 00 00 00 00 00 00 00 00 00 00	-	0.1007	120112	10.00								
High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per mo		UNC3X	UE3PX	382.95							1	
Interoffice Transport - Dedicated - DS3 - Per Mile per mon	+	UNC3X	1L5XX	8.02								
interoffice Transport - Dedicated - DSS - Fel Wille per friori	+	UNCOX	ILJAA	0.02								
Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per mo		UNC3X	U1TF3	880.65							1	
Nonrecurring Currently Combined Network Elements Switch -As-Is Char	+	UNC3X	UNCCC	000.03	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
Notifieduring Currently Combined Network Elements Switch -As-is Chair		UNCOA	UNCCC		11.21	11.21	13.99	13.99	31.30	31.30	3.94	
STS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)	+-+									<del></del>		
	-	UNCSX	1L5ND	45.00						+		
High Capacity Unbundled Local Loop - STS1 combination - Per Mile per mor	$-\!\!+\!\!-\!\!\!+$	UNCSX	1L5ND	15.33						<del></del>	<b></b>	
10 1 0 10 10 10 10 10 10 10 10 10 10 10		LINGOV	1101.04	004.00							1	
High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per mo	-	UNCSX	UDLS1	391.86								
Interoffice Transport - Dedicated - STS1 combination - Per Mile per mor	-	UNCSX	1L5XX	8.02								
			===								1	
Interoffice Transport - Dedicated - STS1 combination - Facility Termination per mo	$\longrightarrow$	UNCSX	U1TFS	880.55								
Nonrecurring Currently Combined Network Elements Switch -As-Is Char	$-\!\!+\!\!-\!\!+$	UNCSX	UNCCC		11.21	11.21	13.99	13.99	31.38	31.38	3.94	
	$-\!\!+\!\!-\!\!+$											
2-WIRE ISDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)												
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone												
	1	UNCNX	U1L2X	26.68								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone	2	UNCNX	U1L2X	40.24								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		UNCNX	U1L2X U1L2X	40.24 53.85								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M	2	UNCNX UNCNX UNC1X	U1L2X U1L2X 1L5XX	40.24 53.85 0.3415								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo	2	UNCNX UNCNX UNC1X UNC1X	U1L2X U1L2X 1L5XX U1TF1	40.24 53.85 0.3415 77.14								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor	2	UNCNX UNCNX UNC1X UNC1X UNC1X	U1L2X U1L2X 1L5XX U1TF1 MQ1	40.24 53.85 0.3415 77.14 134.46								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo	2	UNCNX UNCNX UNC1X UNC1X	U1L2X U1L2X 1L5XX U1TF1	40.24 53.85 0.3415 77.14								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per Minimum of Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon	2	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X	U1L2X U1L2X 1L5XX U1TF1 MQ1 UC1CA	40.24 53.85 0.3415 77.14 134.46 3.2								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor	2	UNCNX UNCNX UNC1X UNC1X UNC1X	U1L2X U1L2X 1L5XX U1TF1 MQ1	40.24 53.85 0.3415 77.14 134.46								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per Minderoffice Transport - Dedicated - DS1 combination - Per mor Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	3	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNCNX	U1L2X U1L2X 1L5XX U1TF1 MQ1 UC1CA	40.24 53.85 0.3415 77.14 134.46 3.2 26.68								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon	2	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X	U1L2X U1L2X 1L5XX U1TF1 MQ1 UC1CA	40.24 53.85 0.3415 77.14 134.46 3.2								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	1 2	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX	U1L2X U1L2X 1L5XX 1L5XX U1TF1 MQ1 UC1CA U1L2X	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	3	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX	U1L2X U1L2X 1L5XX U1TF1 MQ1 UC1CA U1L2X U1L2X	40.24 53.85 0.3415 77.14 134.46 3.2 26.68								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon	1 2	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX	U1L2X U1L2X 1L5XX U1TF1 MQ1 UC1CA U1L2X U1L2X U1L2X U1L2X	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24								
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	1 2	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX	U1L2X U1L2X 1L5XX U1TF1 MQ1 UC1CA U1L2X U1L2X	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Nonrecurring Currently Combined Network Elements Switch - As-Is Chan	1 2	UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX	U1L2X U1L2X 1L5XX U1TF1 MQ1 UC1CA U1L2X U1L2X U1L2X U1L2X	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Per mon Nonrecurring Currently Combined Network Elements Switch - As-Is Char  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL)	1 2	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX	U1L2X U1L2X 11.5XX U1TF1 MQ1 UC1CA  U1L2X U1L2X U1L2X U1L2X U1L2X U1CA	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED ST3-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1 2	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX	U1L2X U1L2X U1L2X U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  US1CA UNCCC	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Char	1 2 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX	U1L2X U1L2X 11.5XX U1TF1 MQ1 UC1CA  U1L2X U1L2X U1L2X U1L2X U1L2X U1CA	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination- per mon Nonrecurring Currently Combined Network Elements Switch - As-Is Chan  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1 2 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNC1X UNC1X UNC1X UNC1X	U1L2X U1L2X 11.5XX U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  US1CA UNCCC	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2 59.61 89.9	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone	1 2 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX	U1L2X U1L2X U1L2X U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  US1CA UNCCC	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 1-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 1-wire ISDN Loop in STS1 Interoffice Transport Combination - Zone 1-wire DS1 Loop in STS1 Interoffice Transport Combination - Zone 1-wire ISD1 Loop in STS1 Interoffice Transport Combination - Zone 1-wire ISS1 Loop in STS1 Interoffice Transport Combination - Zone 1-wire ISS1 Loop in STS1 Interoffice Transport Combination - Per Mile Per Mor	1 2 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNC1X	U1L2X U1L2X U1L2X U1TF1 MQ1 UC1CA U1L2X U1L2X U1L2X U1L2X US1CA UNCCC USLXX USLXX USLXX USLXX USLXX	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2 59.61 89.9 119.06	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Fer Mile Per Mor	1 2 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNC1X	U1L2X U1L2X 11.5XX U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  USLXX USLXX USLXX USLXX USLXX USLXX USLXX	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2 59.61 89.9 119.06 8.02 880.55	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon Nonrecurring Currently Combined Network Elements Switch -As-Is Chan 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System conbination per mon	1 2 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNC1X	U1L2X U1L2X U1L2X U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  USLXX USLXX USLXX USLXX UTFS	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2 59.61 89.9 119.06 8.02	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon Nonrecurring Currently Combined Network Elements Switch -As-Is Chan  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System combination per mont DS3 Interface Unit (DS1 CCO) combination per mont	1 2 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNC1X	U1L2X U1L2X U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L5A  UNCCC  USLXX USLXX USLXX USLXX USLXX USLXX USLXX U1TFS MQ3 UC1D1	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2 59.61 89.9 119.06 80.02 880.55 180.03	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System combination per mont DS3 Interface Unit (DS1 COCI) combination per mont Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1 2 3 3 1 1 2 2 3 3 3 1 1 1 2 1 1 1 1 1	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCSX	U1L2X U1L2X U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  UC1CA  UNCCC  USLXX USLXX USLXX USLXX USLXX USLXX UT1FS  MQ3 UC1D1 USLXX	40.24 53.85 0.3415 77.14 134.48 3.2 26.68 40.24 53.85 3.2 59.61 89.9 119.06 8.02 880.55 180.03 10.8	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon  2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon 4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor	1 2 3 3 1 1 2 3 3	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCNX UNCNX UNCNX UNCNX UNCNX UNC1X	U1L2X U1L2X U1TF1 MQ1 UC1CA  U1L2X  USLXX  USLXX  USLXX  USLXX  USLXX  USLXX  USLXX  USLXX  USLXX  USLXX	40.24 53.85 0.3415 77.14 134.46 3.2 26.68 40.24 53.85 3.2 59.61 89.9 119.06 80.02 880.55 10.8 59.61 89.9 10.03	11.21	11.21	13.99	13.99	31.38	31.38	3.94	
First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Per M Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mo Channelization - Channel System DS1 to DS0 combination - per mor 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon  Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - Zon Nonrecurring Currently Combined Network Elements Switch -As-Is Char  4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL) First DS1 Loop in STS1 Interoffice Transport Combination - Zone First DS1 Loop in STS1 Interoffice Transport Combination - Zone Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor Interoffice Transport - Dedicated - STS1 combination - Facility Terminati STS1 to DS1 Channel System combination per mont DS3 Interface Unit (DS1 COCI) combination per mont Additional DS1Loop in STS1 Interoffice Transport Combination - Zone	1 2 3 3 1 1 2 2 3 3 3 1 1 1 2 1 1 1 1 1	UNCNX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCNX UNCSX	U1L2X U1L2X U1TF1 MQ1 UC1CA  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  U1L2X  UC1CA  UNCCC  USLXX USLXX USLXX USLXX USLXX USLXX UT1FS  MQ3 UC1D1 USLXX	40.24 53.85 0.3415 77.14 134.48 3.2 26.68 40.24 53.85 3.2 59.61 89.9 119.06 8.02 880.55 180.03 10.8	11.21	11.21	13.99	13.99	31.38	31.38	3.94	

Page 10 of 22

	4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)													
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zon	1	UNCDX	UDL56	34.26									
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zon	2	UNCDX	UDL56	51.67									
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zon	3	UNCDX	UDL56	68.43									
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per N		UNCDX	1L5XX	0.0167									
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminat		UNCDX	U1TD5	16.76									
	Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCDX	UNCCC		11.21	11.21	13.99	13.99		31.38	31.38	3.94	3.9
	4-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)													
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon	1	UNCDX	UDL64	34.26									
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon	2	UNCDX	UDL64	51.67									
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zon	3	UNCDX	UDL64	68.47									
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per N		UNCDX	1L5XX	0.0167									
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminat  Nonrecurring Currently Combined Network Elements Switch -As-Is Char		UNCDX	U1TD6	16.76	11.21	11.21	40.00	42.00		24.20	24.20	2.04	2.0
	Nonrecurring Currently Combined Network Elements Switch -As-is Char-		UNCDX	UNCCC		11.21	11.21	13.99	13.99		31.38	31.38	3.94	3.9
DDITIONAL	NETWORK ELEMENTS													
DDITIONAL	NETWORK ELEMENTS													
	When used as a part of a currently combined facility, the non-recurrng charges do not apply, but	Switch As Is a	hargo doce annly											
	When used as a part of a currently combined facility, the horriecting charges do not apply, but a When used as ordinarilty combined network elements in Georgia, the non-recurring charges apply	and the Switch	Ae le Chargo doce	not										
	when used as ordinarity combined network elements in Georgia, the non-recurring charges apply	and the Switch	As is cliarge does	not.										
	Node (SynchroNet)										1			
	Node per month		UNCDX	UNCNT	14.55									
			235/1	2		1					1			
	Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to each c	ombination)									1			
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion										1			
	Charge		UNCVX	UNCCC		11.21	11.21	13.99	13.99		31.38	31.38	3.94	3.9
	56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion						1				1			3.0
	Charge		UNCDX	UNCCC	1	11.21	11.21	13.99	13.99		31.38	31.38	3.94	3.9
	DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charg		UNC1X	UNCCC		11.21	11.21	13.99	13.99		31.38	31.38	3.94	3.9
	DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge		UNC3X	UNCCC		11.21	11.21	13.99	13.99		31.38	31.38	3.94	3.9
	STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion													
	Charge		UNCSX	UNCCC		11.21	11.21	13.99	13.99		31.38	31.38	3.94	3.9
	NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, DS3	and above=for	ur months											
		and above=for	ır months											
PERATION	IAL SUPPORT SYSTEMS													
PERATION				g charges as or	rdered by the S	State Commission	ns							
PERATION	IAL SUPPORT SYSTEMS	e specific electr	onic service orderin			State Commission	ns							
PERATION	IAL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat	e specific electr	onic service orderin	ervice ordering	charge			charge.						
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat  NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is	e specific electr	onic service orderin	ervice ordering	charge			charge.						
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e	e specific electr	onic service orderin	ervice ordering	charge			charge.						
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e	e specific electr	onic service orderin	or CLEC-1 may	charge			charge.						
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR	e specific electr	onic service orderin	ervice ordering	charge			charge.						
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR    Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces   (Regional)	e specific electr the BellSouth r lectronic service basis	onic service orderin egional electronic s e ordering charges,	ervice ordering of CLEC-1 may	charge relect the region	onal electronic s	ervice ordering							
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR    Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces   (Regional)	e specific electr the BellSouth r lectronic service basis	onic service orderin egional electronic s e ordering charges,	ervice ordering of CLEC-1 may	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	er to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces	e specific electr the BellSouth r lectronic service basis	onic service orderin egional electronic s e ordering charges,	ervice ordering of CLEC-1 may	charge relect the region	onal electronic s	ervice ordering		al Office, refe	rr to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Ger	e specific electr the BellSouth r lectronic service basis	onic service orderin egional electronic s e ordering charges,	ervice ordering of CLEC-1 may	charge relect the region	onal electronic s	ervice ordering		al Office, refe	rr to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Ger	e specific electr the BellSouth r lectronic service basis	onic service orderin egional electronic s e ordering charges,	ervice ordering of CLEC-1 may	charge relect the region	onal electronic s	ervice ordering		al Office, refe	rr to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Ger	e specific electr the BellSouth r lectronic service basis	onic service orderin egional electronic s e ordering charges,	ervice ordering of CLEC-1 may	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	er to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR    Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	e specific electr the BellSouth r lectronic service basis	onic service orderin egional electronic s e ordering charges,	ervice ordering of CLEC-1 may	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	rr to Internet Website:				
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic si e ordering charges, averaged UNE Zoni	some ordering or CLEC-1 may SOMEC	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	or to Internet Website:				
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic si e ordering charges, averaged UNE Zoni	some ordering or CLEC-1 may SOMEC	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	rr to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will in	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic si e ordering charges, averaged UNE Zoni	some ordering or CLEC-1 may SOMEC	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	er to Internet Website:				
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic si e ordering charges, averaged UNE Zoni	some ordering or CLEC-1 may SOMEC	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	r to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will in	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic si e ordering charges, averaged UNE Zoni	some ordering or CLEC-1 may SOMEC	charge relect the region	onal electronic s	ervice ordering		ral Office, refe	er to Internet Website:	44.42	14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will in 2-wilker VOICE GRADE LINE PORT RATES (RES)	e specific electrithe BellSouth relectronic service basis	onic service orderin egional electronic si e ordering charges, averaged UNE Zoni	some ordering or CLEC-1 may  SOMEC  so. To view Geo	charge elect the regic	3.5	ervice ordering  Zone Designati		al Office, refe	er to Internet Website:	44.42 44.42	14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: Switching Ports - 2-Wire Analog Line Port- Re:	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, averaged UNE Zone ered using retail USC	source ordering or CLEC-1 may  SOMEC  s. To view Ged	charge elect the region ographically De	3.5  averaged UNE	Zone Designati		ral Office, refe	er to Internet Website:				
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentitp://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will not be suchange Ports - 2-Wire Analog Line Port- Re: Exchange Ports - 2-Wire Analog Line Port- Re: Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se ordering charges, averaged UNE Zone red using retail USC UEPSR UEPSR	some ordering or CLEC-1 may  SOMEC  SOMEC  SOMEC  SOMEC  UEPRL  UEPRL  UEPRC	charge relect the region of th	3.5  averaged UNE	Zone Designati		ral Office, refe	rr to Internet Website:	44.42	14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BSTs OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: although the Port Rate includes all available features in GA & TN, the desired features will note: although Ports - 2-Wire Analog Line Port Retex Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port with Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port Wirth Caller ID - Re Exchange Ports - 2-Wire Nalog Line Port Wirth Caller ID - Re Exchange Ports - 2-Wire Nalog Line	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se ordering charges, averaged UNE Zone red using retail USC UEPSR UEPSR	some ordering or CLEC-1 may  SOMEC  SOMEC  SOMEC  SOMEC  UEPRL  UEPRL  UEPRC	charge relect the region of th	3.5  averaged UNE	Zone Designati		al Office, refe	er to Internet Website:	44.42	14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellisouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  Although the Port Rate includes all available features in GA & TN, the desired features will note: The control of the port of the po	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, averaged UNE Zone ered using retail USG UEPSR UEPSR UEPSR	SOMEC  SOMEC  SOMEC  SUBSTITUTE OF THE PROPERTY OF THE PROPERT	charge relect the region or elect the region of the region	3.5 saveraged UNE  24.98 24.98 24.98	Zone Designati  24.98 24.98 24.98		ral Office, refe	er to Internet Website:	44.42 44.42	14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Exchange Ports - 2-Wire Analog Line Port- Re:  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Res (LW8)  Exchange CW8)	e specific electrithe BellSouth relectronic service basis	onic service orderinggional electronic service ordering charges, e ordering charges, averaged UNE Zone ordering charges or the control of the	SOMEC  SO	charge relect the region or elect the region of the region	3.5 saveraged UNE  24.98 24.98 24.98	Zone Designati  24.98 24.98 24.98		ral Office, refe	r to Internet Website:	44.42 44.42 44.42	14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will in Swinger Ports - 2-Wire Analog Line Port - ReExchange Ports - 2-Wire Analog Line Port with Caller ID - ReExchange Ports - 2-Wire Analog Line Port outgoing only - ReExchange Ports - 2-Wire Analog Line Port outgoing only - ReExchange Ports - 2-Wire Vire Vire Vire Vire Vire Vire Vire V	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, averaged UNE Zone electronic se a ordering charges, averaged UNE Zone electronic se a ordering charges, averaged UNE Zone electronic services de la constant de la cons	SOMEC SOMEC	charge velect the region velect the region velect the region ographically De velect the region ographically De velect the region of vel	3.5  24.98 24.98 24.98	Zone Designati  24.98 24.98 24.98 24.98		ral Office, refe	er to Internet Website:	44.42 44.42 44.42	14.63 14.63 14.63		
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: Sechange Ports - 2-Wire Analog Line Port - Re: Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re: Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re: Exchange Ports - 2-Wire Vig unbundled SC extended local dialing parity Port with Caller ID - Res: Exchange Ports - 2-Wire Vig unbundled SC extended local dialing port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled South Carolina Area Calling port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled res, low usage line port with Caller ID (LL Subsequent Activity)	e specific electrithe BellSouth relectronic service basis	onic service orderinggional electronic service ordering charges, e ordering charges, averaged UNE Zone electronic services de using retail USG UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR	SOMEC  SO	charge relect the region of re	3.5  averaged UNE  24.98  24.98  24.98  24.98	Zone Designati  Zone Designati  24.98 24.98 24.98 24.98		ral Office, refe	rr to Internet Website:	44.42 44.42 44.42	14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will in 2-WIRE VOICE GRADE LINE PORT RATES (RES)  Exchange Ports - 2-Wire Analog Line Port Ree Exchange Ports - 2-Wire Analog Line Port with Coller ID - Re Exchange Ports - 2-Wire Analog Line Port outgoing only - Re Exchange Ports - 2-Wire Virg Unbundled SC extended local dialing parity Port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Virg Unbundled Sc extended local dialing port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU Subsequent Activity	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic sie ordering charges, e ordering charges, averaged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged VEPSR UNE VEPSR VEPSR UNE VEPSR VE	SOMEC SOMEC SOMEC SOMEC UEPRL UEPRL UEPRO UEPAJ UEPAJ UEPAJ USASC	2.35 2.35 2.35 2.35 2.35 0	3.5  averaged UNE  24.98 24.98 24.98 24.98 24.98 24.98 0	Zone Designati  24.98 24.98 24.98 24.98 24.98 0		ral Office, refe	er to Internet Website:	44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63		
PERATION	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: Sechange Ports - 2-Wire Analog Line Port - Re: Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re: Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re: Exchange Ports - 2-Wire Vig unbundled SC extended local dialing parity Port with Caller ID - Res: Exchange Ports - 2-Wire Vig unbundled SC extended local dialing port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled South Carolina Area Calling port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire Vig unbundled res, low usage line port with Caller ID (LL Subsequent Activity)	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, averaged UNE Zone electronic se a ordering charges, averaged UNE Zone electronic se a ordering charges, averaged UNE Zone electronic services de la constant de la cons	SOMEC SOMEC	charge of elect the region	3.5  24.98 24.98 24.98 24.98 24.98 24.98	Zone Designati  24.98 24.98 24.98 24.98 24.98 24.98		ral Office, refe	rr to Internet Website:	44.42 44.42 44.42	14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DIOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Ports rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Ports rate includes all available features in GA & TN, the desired features will note: Although the Ports rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will note: Although the Port Rate includes all available features in GA & TN, the desired features will no	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic sie ordering charges, e ordering charges, averaged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged VEPSR UNE VEPSR VEPSR UNE VEPSR VE	SOMEC SOMEC SOMEC SOMEC UEPRL UEPRL UEPRO UEPAJ UEPAJ UEPAJ USASC	2.35 2.35 2.35 2.35 2.35 0	3.5  averaged UNE  24.98 24.98 24.98 24.98 24.98 24.98 0	Zone Designati  24.98 24.98 24.98 24.98 24.98 0		al Office, refe	or to Internet Website:	44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gen http://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DIOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note through the Port Rate includes all available features in GA & TN, the desired features will note through the Port Rate includes all available features in GA & TN, the desired features will note	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, a ordering charges, averaged UNE Zone using retail USC UEPSR	SOMEC  SO	2.35 2.35 2.35 2.35 2.35 2.35 2.35 2.35	3.5  24.98 24.98 24.98 24.98 24.98 0 0	Zone Designati  24.98 24.98 24.98 24.98 24.98 0 0		al Office, refe	er to Internet Website:	44.42 44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentite;//www.interconnection.bellisouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: The state of Florida, to be billed on a per LSR (RES)  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Gunbundled SC extended local dialing parity Port with Caller ID - Res.  Exchange Ports - 2-Wire VG unbundled Scuth Carolina Area Calling port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Scuth Carolina Area Calling port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU Subsequent Activity  FEATURES  All Available Vertical Feature  2-WIRE VOICE GRADE LINE PORT RATES (BUS)  Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bi	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic sie ordering charges, e ordering charges, averaged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged UNE Zone veraged VEPSR UNE VEPSR VEPSR UNE VEPSR VE	SOMEC SOMEC SOMEC SOMEC UEPRL UEPRL UEPRO UEPAJ UEPAJ UEPAJ USASC	2.35 2.35 2.35 2.35 2.35 0	3.5  averaged UNE  24.98 24.98 24.98 24.98 24.98 24.98 0	Zone Designati  24.98 24.98 24.98 24.98 24.98 0		ral Office, refe	er to Internet Website:	44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: although the Port Rate includes all available features in GA & TN, the desired features will note: although the Port Rate includes all ine Port Ree  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port outgoing only - Re  Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled SC extended local dialing port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU Subsequent Activity  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU Subsequent Activity  Exchange Ports - 2-Wire VG unbundled Line Port without Caller ID - Bi  Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller + Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller + Exchange Ports - 2-Wire VG unbundled Line Port without Caller ID - Bi	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic service ordering epinal electronic se ordering charges, averaged UNE Zone verdering charges.  UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR	SOMEC  SOMEC  SOMEC  SOMEC  SOMEC  UEPRL UEPRC UEPRO UEPAJ UEPAJ UEPAJ UEPAF USASC  UEPVF	2.35 2.35 2.35 2.35 2.35 2.35 2.35 2.35	3.5  24.98 24.98 24.98 24.98 0 0 24.98	Zone Designati  24.98 24.98 24.98 24.98 24.98 0 0 24.98		ral Office, refe	r to Internet Website:	44.42 44.42 44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  D LOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports Although the Port Rate includes all available features in GA & TN, the desired features will in NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will in Exchange Ports - 2-Wire Analog Line Port -Re Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Analog Line Port outgoing only - Re Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire Analog Line Port w	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, a ordering charges, averaged UNE Zone using retail USC UEPSR UEPSB UEPSB UEPSB	SOMEC  SO	2.35 2.35 2.35 2.35 2.35 2.35 2.35 2.35	3.5  24.98 24.98 24.98 24.98 24.98 24.98 24.98 24.98	Zone Designati  Zone Designati  24.98 24.98 24.98 24.98 0 0 0 24.98 24.98		ral Office, refe	or to Internet Website:	44.42 44.42 44.42 44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BSTs OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: although the Port Rate includes all available features in GA & TN, the desired features will note: although the Port Rate includes all in Port-Re  Exchange Ports - 2-Wire Analog Line Port-Re  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Mg unbundled SC extended local dialing parity Port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LL Subsequent Activity  FEATURES  All Available Vertical Feature  2-WIRE VOICE GRADE LINE PORT RATES (BUS)  Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484  - Bus.  Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bi  Exchange Ports - 2-Wire Analog Line Port wit	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic service ordering epinal electronic se ordering charges, averaged UNE Zone verdering charges.  UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR UEPSR	SOMEC SOMEC SOMEC SOMEC UEPRL UEPRL UEPRO UEPAJ UEPAJ UEPAJ UEPAF USASC UEPVF	2.35 2.35 2.35 2.35 2.35 2.35 2.35 2.35	3.5  24.98 24.98 24.98 24.98 0 0 24.98	Zone Designati  24.98 24.98 24.98 24.98 24.98 0 0 24.98		al Office, refe	or to Internet Website:	44.42 44.42 44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR (Regional)  Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: through the Port Rate includes all available features in GA & TN, the desired features will note: Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re Exchange Ports - 2-Wire Analog Line Port outgoing only - Re Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Sc extended local dialing port with Caller ID - Res (LW8)  Exchange Ports - 2-Wire VG unbundled Fent without Caller ID - Bl Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484  Bus.  Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484  Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, a ordering charges, averaged UNE Zone using retail USC UEPSR UEPSB UEPSB UEPSB	SOMEC  SO	2.35 2.35 2.35 2.35 2.35 2.35 2.35 2.35	3.5  24.98 24.98 24.98 24.98 24.98 24.98 24.98 24.98 24.98	Zone Designati  24.98 24.98 24.98 24.98 0 0 24.98 24.98 24.98 24.98 24.98 24.98		al Office, refe	er to Internet Website:	44.42 44.42 44.42 44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63 14.63 14.63 14.63		
PERATION	AL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the stat NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhibit is NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the e NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per LSR Electronic OSS Charge, per LSR, submitted via BSTs OSS interactive interfaces (Regional)  The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Gentity://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm  DLOCAL EXCHANGE SWITCHING(PORTS)  Exchange Ports  NOTE: Although the Port Rate includes all available features in GA & TN, the desired features will note: although the Port Rate includes all available features in GA & TN, the desired features will note: although the Port Rate includes all in Port-Re  Exchange Ports - 2-Wire Analog Line Port-Re  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re  Exchange Ports - 2-Wire Mg unbundled SC extended local dialing parity Port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID - Res (LWB)  Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LL Subsequent Activity  FEATURES  All Available Vertical Feature  2-WIRE VOICE GRADE LINE PORT RATES (BUS)  Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484  - Bus.  Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bi  Exchange Ports - 2-Wire Analog Line Port wit	e specific electrithe BellSouth relectronic service basis	onic service ordering egional electronic se a ordering charges, a ordering charges, averaged UNE Zone using retail USC UEPSR UEPSB UEPSB UEPSB	SOMEC  SO	2.35 2.35 2.35 2.35 2.35 2.35 2.35 2.35	3.5  24.98 24.98 24.98 24.98 24.98 24.98 24.98 24.98	Zone Designati  Zone Designati  24.98 24.98 24.98 24.98 0 0 0 24.98 24.98		ral Office, refe	er to Internet Website:	44.42 44.42 44.42 44.42 44.42 44.42 44.42 44.42	14.63 14.63 14.63 14.63 14.63 14.63		

Page 11 of 22

1		Exchange Ports - 2-Wire VG unbundled South Carolina Bus Area Calling Port with Calle	r													
		ID - Bus (LMB) Subsequent Activity			UEPSB UEPSB	UEPAB	2.35	24.98	24.98				44.42	14.63		
	ATURES	Subsequent Activity			UEPSB	USASC	0	0	0							
FEA		All Available Vertical Feature			UEPSB	UEPVF	6.29	0	0				44.42	14.63		
EVC		PORT RATES (DID & PBX)			UEPSB	UEPVF	6.29	U	U				44.42	14.63		
EVC		Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.86	239.14	37.56	120.05	7.54		67.52	67.52		
-		Exchange Ports - 2-Wire DID Port  Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabili			UEPDD	UEPDD	73.62	404.94	191.8	145.5	4.93		19.99	19.99	19.99	1:
		Exchange Ports - DDIT'S Port - 4-Wire DST Port with DID capabili Exchange Ports - 2-Wire ISDN Port (See Notes below			UEPTX UEPSX	U1PMA		145.86			21.52		67.52		19.99	- 13
		All Features Offered			UEPTX UEPSX	UEPVF	13.38 6.29	0	106.21	95.79	21.52		67.52	67.52		
1101									0	IODNI .						
		smission/usage charges associated with POTS circuit switched usage will also apply to c														
NOT	TE: Acce	ss to B Channel or D Channel Packet capabilities will be available only through BFR/Nev	v Busines	ss Req						de Request/N	New Busines:	s Request Process.				
		Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0	0	0							
		Exchange Ports - 4-Wire ISDN DS1 Por			UEPEX	UEPEX	107.44	408.53	203.56	158.7	21.52		65.48	65.48		
		2-Wire VG Unbundled 2-Way PBX Trunk - Re:			UEPSE	UEPRD	2.35	24.36	24.36				41.86	14.46		
		2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu			UEPSP	UEPPC	2.35	24.36	24.36				41.86	14.46		
		2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu			UEPSP	UEPPO	2.35	24.36	24.36				41.86	14.46		
		2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu			UEPSP	UEPP1	2.35	24.36	24.36				41.86	14.46		
		2-Wire Analog Long Distance Terminal PBX Trunk - Bu			UEPSP	UEPLD	2.35	24.36	24.36				41.86	14.46		
_		2-Wire Voice Unbundled PBX LD Terminal Port			UEPSP	UEPLD	2.35	24.36	24.36				41.86	14.46		
		2-Wire Vice Unbundled 2-Way PBX Usage Po			UEPSP	UEPXA	2.35	24.36	24.36				41.86	14.46		
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Por			UEPSP	UEPXB	2.35	24.36	24.36				41.86	14.46		
		2-Wire Voice Unbundled PBX LD DDD Terminals Po			UEPSP	UEPXC	2.35	24.36	24.36				41.86	14.46		
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc			UEPSP	UEPXD	2.35	24.36	24.36				41.86	14.46		
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Po			UEPSP	UEPXE	2.35	24.36	24.36				41.86	14.46		
	ļ	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P		1	UEPSP	UEPXL	2.35	24.36	24.36		1		41.86	14.46	1 1	
-		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		1	UEPSP	UEPXM	2.35	24.36	24.36				41.86	14.46	+	
-+-		2 17110 10100 Onbunulou 2-11ay F DA Hotel/Hospital Economy Room Calling Pi		+	ULITOF	OLFAIVI	۵.۵۵	24.30	24.30		<b> </b>	<del>                                     </del>	41.00	14.40	+ +	
		0.000		1	LIEBOD	LIEDVC	0.05	04.00	04.00		1		44.00	44.40	1 1	
-		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling P		1	UEPSP	UEPXO	2.35	24.36	24.36				41.86	14.46	+ +	
		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		1	UEPSP	UEPXS	2.35	24.36	24.36				41.86	14.46	1	
		2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus Calling Po			UEPSP	UEPXT	2.35	24.36	24.36				41.86	14.46		
		Subsequent Activity			UEPSP	USASC	0	0	0							
FEA	ATURES															
		All Available Vertical Feature			UEPSP UEPSE	UEPVF	6.29	0	0				41.86	14.46		
FXC		PORT RATES (COIN)														
		Exchange Ports - Coin Por					2.77	24.75	24.75				43,48	14.57		
-		Excitating 1 of to Control		+			2.11	24.10	24.70				40.40	14.07		
		Exchange port - 4-wire ISDN trunk port -all available features includ  Exchange Port - 2-wire ISDN digital line side port with three features include				UEPEX U1PMA	251 36.01	311.73 70.32	311.73 70.32				65.48 67.52	65.48 67.52		
_		Exchange Fort - 2-wire ISDN digital line side port with three readures inclut				UTFINA	30.01	70.32	70.32				67.52	67.32		
DLED LO	OCAL SWI	TCHING, PORT USAGE														
End	d Office S	witching (Port Usage)														
		End Office Switching Function, Per MOI														
							0.0019295									
		End Office Trunk Port - Shared, Per MOI														
		End Office Trunk Port - Shared, Per MOl					0.0019295 0.0002581									
Tan		End Office Trunk Port - Shared, Per MOl														
Tan	ndem Swit	End Office Trunk Port - Shared, Per MOL tching (Port Usage) (Local or Access Tandem)					0.0002581									
Tan	ndem Swit	End Office Trunk Port - Shared, Per MOl  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOl					0.0002581									
Tan	ndem Swit	End Office Trunk Port - Shared, Per MOL tching (Port Usage) (Local or Access Tandem)					0.0002581									
	ndem Swit	End Office Trunk Port - Shared, Per MOI tching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOI Tandem Trunk Port - Shared, Per MOI					0.0002581									
	ndem Swit	End Office Trunk Port - Shared, Per MOl tching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOl Tandem Trunk Port - Shared, Per MOl insport					0.0002581 0.0006843 0.0004034									
	ndem Swit	End Office Trunk Port - Shared, Per MOI tching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOI Tandem Trunk Port - Shared, Per MOI unsport Common Transport - Per Mile, Per MOI					0.0002581 0.0006843 0.0004034 0.0000121									
	ndem Swit	End Office Trunk Port - Shared, Per MOl tching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOl Tandem Trunk Port - Shared, Per MOl insport					0.0002581 0.0006843 0.0004034									
Com	ndem Swit	End Office Trunk Port - Shared, Per MOI tching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOI Tandem Trunk Port - Shared, Per MOI unsport Common Transport - Per Mile, Per MOI Common Transport - Facilities Termination Per MO					0.0002581 0.0006843 0.0004034 0.0000121									
Com	ndem Swit	End Office Trunk Port - Shared, Per MOI tching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOI Tandem Trunk Port - Shared, Per MOI unsport Common Transport - Per Mile, Per MOI					0.0002581 0.0006843 0.0004034 0.0000121									
Com	ndem Swit	End Office Trunk Port - Shared, Per MOI tching (Port Usage) (Local or Access Tandem) Tandem Switching Function Per MOI Tandem Trunk Port - Shared, Per MOI unsport Common Transport - Per Mile, Per MOI Common Transport - Facilities Termination Per MO					0.0002581 0.0006843 0.0004034 0.0000121									
Com	ndem Swit	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  sinsport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES	provide III	Inhund	ad Local Switching o	r Switch Ports	0.0002581 0.0006843 0.0004034 0.0000121 0.0004672									
Com	ndem Swit	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  unsport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to p					0.0002581 0.0006843 0.0004034 0.0000121 0.0004672	nation of this D	oto Evhibit							
Com  Cost  Feat	ndem Swit	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  msport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to:  Il apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same	manner	as the	y are applied to the S	Stand-Alone Un	0.0002581 0.0006843 0.0004034 0.0000121 0.0004672	ection of this R	ate Exhibit.							
Com Cost Feat End	ondem Swit	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  sinsport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  Common Transport - Stacilities Termination Per MO  COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to p  Il apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same  at Tandem Switching Usage and Common Transport Usage rates in the Port section of	manner this rate	as the exhibit	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  blundled Port se	elements exce	ept for UNE Coi							
Com  Cost Feat End For	ondem Swith and on Training of the State of	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to p  Il apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same and Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Com	manner this rate	as the exhibit	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  blundled Port se	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN anc	d all oth
Com  Cost Feat End For	ondem Swith and on Training of the State of	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  sinsport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  Common Transport - Stacilities Termination Per MO  COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to p  Il apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same  at Tandem Switching Usage and Common Transport Usage rates in the Port section of	manner this rate	as the exhibit	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Cost Feat End For	or Georgia an recurring	End Office Trunk Port - Shared, Per MOI  tohing (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Tandem Trunk Port - Shared, Per MOI  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to plia poly to the Unbundled Port/Loop Combination - Cost Based Rate section in the same drandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.	manner this rate	as the exhibit	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Cost Feat End For	or Georgia an recurring	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to p  Il apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same and Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Com	manner this rate	as the exhibit	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Cost Feat End For	or Georgia an recurring	End Office Trunk Port - Shared, Per MOI  tohing (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Tandem Trunk Port - Shared, Per MOI  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to plia poly to the Unbundled Port/Loop Combination - Cost Based Rate section in the same drandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.	manner this rate	as the exhibit	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Cost Feat End For nonr	mmon Tra  ORT/LOOP  St Based F atures sha d Office ar r Georgia a nrecurring	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pill apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same dradender witching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.	manner this rate	as the exhibit	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Cost Feat End For nonr	ort/Loop st Based F atures sha d Office ar r Georgia a nrecurring	End Office Trunk Port - Shared, Per MOI  tohing (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Tandem Trunk Port - Shared, Per MOI  Unsport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pil apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same and Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.  E GRADE LOOP WITH 2-WIRE LINE PORT (RES)	manner this rate	as the exhibit id Not	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port seop/port network e first and addit	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Cost Feat End For nonr	ort/Loop st Based Fatures sha do Office ar r Georgia a nrecurring WIRE VOIC	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Tandem Trunk Port - Shared, Per MOI  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to 7  Il apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same of Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.  EE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  Top Combination Rates  2-Witre VG Loop/Port Combo - Zone	manner this rate	as the exhibit id Not	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port se op/port network e first and addit	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	to lle b
Cost Feat End For nonr	ondem Swith Manager State Stat	End Office Trunk Port - Shared, Per MOI  tohing (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pil apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same drandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.  EGRADE LOOP WITH 2-WIRE LINE PORT (RES)  DO Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	manner this rate	as the exhibit ad Not of the last of the l	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port se op/port networke e first and addit	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	to lle b
Cost Feat End For nonr	ondem Swith Manager State Stat	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Tandem Trunk Port - Shared, Per MOI  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to 7  Il apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same of Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.  EE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  Top Combination Rates  2-Witre VG Loop/Port Combo - Zone	manner this rate	as the exhibit id Not	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port se op/port network e first and addit	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d dill oth
Control Cost Feat End For I nonn	orr/Loop st Based F atures sha d Office ar r Georgia a nrecurring	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Switching Function Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pill apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same drandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined Sections.  E GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DO Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	manner this rate	as the exhibit ad Not of the last of the l	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port se op/port networke e first and addit	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	the black that the bl
Com  Cost Feat End For (  nonr  2-Wi	ondem Swith Manager State Stat	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Switching Function Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pill apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same drandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined Sections.  E GRADE LOOP WITH 2-WIRE LINE PORT (RES)  DO Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	manner this rate	as the exhibit ad Not of the last of the l	y are applied to the S shall apply to all con	Stand-Alone Un nbinations of lo	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port se op/port networke e first and addit	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Com  Cost Feat End For (  nonr  2-Wi	mmon Tra  ORT/LOOF st Based F atures sha d Office an r Georgia a r Georgia a r recurring WIRE VOIC	End Office Trunk Port - Shared, Per MOI  tohing (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Tandem Trunk Port - Shared, Per MOI  tomsport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pill apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same drandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.  EGRADE LOOP WITH 2-WIRE LINE PORT (RES)  OP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	manner this rate	as the exhibit ad Not of the last of the l	y are applied to the S shall apply to all con Currently Combined (	Stand-Alone Un nbinations of lo Combos and th	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port seop/port network e first and addit  20.71 29.35 37.68	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Com  Cost Feat End For (  nonr  2-Wi	mmon Tra  mmon Tra  ora table and the second of the second	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pli apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same and Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges Islaed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.  E GRADE LOOP WITH 2-WIRE LINE PORT (RES)  Do Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	manner this rate	as the exhibited Not of the second se	y are applied to the S shall apply to all con Currently Combined (	Stand-Alone Un nbinations of lo Combos and th	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port se op/port network e first and addit  20.71 29.35 37.68	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Control Cost Feat End For I nonn	mmon Tra  ORT/LOOP  St Based Fatures sha do Office ar r Georgia annecurring  WIRE VOIC  IE Port/Loo  IE Loop Ra	End Office Trunk Port - Shared, Per MOI  tohing (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  unsport  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pill apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same for Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Combination State Commission of and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Combination Cost Based Rate section in the same for and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Combination Cost Commission Commission Rates  EGRADE LOOP WITH 2-WIRE LINE PORT (RES)  DO Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	manner this rate	as the exhibited Not 0	y are applied to the S shall apply to all con Currently Combined (	Stand-Alone Un hbinations of lo Combos and th	0.0002581  0.0006843 0.0004034  0.0004072  0.0004672  bundled Port se op/port network e first and addit 22.71 29.35 37.68	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth
Control Cost Feat End For I nonn	mmon Tra  ORT/LOOP  St Based Fatures sha do Office ar r Georgia annecurring  WIRE VOIC  IE Port/Loo  IE Loop Ra	End Office Trunk Port - Shared, Per MOI  tching (Port Usage) (Local or Access Tandem)  Tandem Switching Function Per MOI  Tandem Trunk Port - Shared, Per MOI  Insport  Common Transport - Per Mile, Per MOI  Common Transport - Per Mile, Per MOI  Common Transport - Facilities Termination Per MO  P COMBINATIONS - COST BASED RATES  Rates are applied where BellSouth is required by FCC and/or State Commission rule to pli apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the same and Tandem Switching Usage and Common Transport Usage rates in the Port section of and Tennessee, the recurring UNE Port and Loop charges Islaed apply to Currently Comcharges shall be those identified in the Nonrecurring - Currently Combined sections.  E GRADE LOOP WITH 2-WIRE LINE PORT (RES)  Do Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	manner this rate	as the exhibited Not of the second se	y are applied to the S shall apply to all con Currently Combined (	Stand-Alone Un nbinations of lo Combos and th	0.0002581  0.0006843 0.0004034  0.0000121 0.0004672  bundled Port se op/port network e first and addit  20.71 29.35 37.68	elements exce	ept for UNE Coi				Currently Com	bined Combos	s in GA, TN and	d all oth

Page 12 of 22

	2-Wire voice unbundled port - residenc	$\neg$	UEPRX	UEPRL	3.69			43.19	9.91
	2-Wire voice unbundled port - resident.  2-Wire voice unbundled port with Caller ID - re		UEPRX	UEPRC	3.69			43.19	9.91
1			UEPRX	UEPRO	3.69			43.19	9.91
	2-Wire voice unbundled port outgoing only - re	-+-+	UEFRA	UEFRU	3.08			43.19	3.31
	2-Wire voice Grade unbundled South Carolina extended local dialing parity port with Ca ID - res		UEPRX	UEPAU	2.00			43.19	9.91
		$\longrightarrow$			3.69				
	2-Wire voice unbundled South Carolina Area Calling port with Caller ID - res (LV	$\longrightarrow$	UEPRX	UEPAJ	3.69			43.19	9.91
	2-Wire voice unbundles res, low usage line port with Caller ID (LU	$\longrightarrow$	UEPRX	UEPAP	3.69			43.19	9.91
FEATURES									
	All Features Offered		UEPRX	UEPVF	6.29	0	0	43.19	9.91
LOCAL NUM	MBER PORTABILITY								
	Local Number Portability (1 per por		UEPRX	LNPCX	0.35				
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED								
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as	-	UEPRX	USAC2		1.59	0.4	43.19	9.91
	2 Wile Video Clade 2007 2 me i cit combination Conversion Cwitch at	-	OLI TOT	COMOL		1.00	U	10.10	0.01
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char		UEPRX	USACC		1.59	0.4	43.19	9.91
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with Chair  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database		OLFIX	USACC		1.55	0.4	45.19	9.91
						0.71		8.91	
	Update					0.71		8.91	
		$\longrightarrow$							
ADDITIONAL		$\longrightarrow$							
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ	$\longrightarrow$	UEPRX	USAS2	0	0	0		
		$\longrightarrow$							
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	$\longrightarrow$							
UNE Port/Lo	oop Combination Rates			1					
	2-Wire VG Loop/Port Combo - Zone	1			20.71				
	2-Wire VG Loop/Port Combo - Zone	2			29.35				
	2-Wire VG Loop/Port Combo - Zone	3			37.68				
UNE Loop R	Paras	-							
ONE Ecop II	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	17.02				
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPBX	UEPLX	25.66				
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	33.99				
	2-Wife Voice Glade Loop (SE1) - Zolle		OLFBA	OLFLX	33.33				
0 140 1/		$\rightarrow$							
2-Wire Voice	e Grade Line Port (Bus)								
	2-Wire voice unbundled port without Caller ID - bt	-	UEPBX	UEPBL	3.69			43.19	9.91
	2-Wire voice unbundled port with Caller + E484 ID - bi		UEPBX	UEPBC	3.69			43.19	9.91
	2-Wire voice unbundled port outgoing only - bt		UEPBX	UEPBO	3.69			43.19	9.91
	2-Wire voice Grade unbundled South Carolina extended local dialing parity port with Ca								
	ID - bus		UEPBX	UEPAZ	3.69			43.19	9.91
	2-Wire voice unbundled incoming only port with Caller ID - B		UEPBX	UPEB1	3.69			43.19	9.91
	2-Wire voice unbundled South Carolina Bus Area Calling Port with Caller ID (LM		UEPBX	UEPAB	3.69			43.19	9.91
LOCAL NUM	MBER PORTABILITY								
	Local Number Portability (1 per por		UEPBX	LNPCX	0.35				
	Zoda Namosi i Stabiniy (1 por por	-	OE! D/t	Liti Ox	0.00				
FEATURES		-							
	All Factoria Office.	-+-+	LIEDDY	LIEDVE	0.00			42.40	0.04
	All Features Offered	=	UEPBX	UEPVF	6.29	0	0	43.19	9.91
			UEPBX	UEPVF	6.29	0	0	43.19	9.91
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED				6.29				
NONRECUR			UEPBX	UEPVF USAC2	6.29	1.59	0.4	43.19	9.91
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPBX	USAC2	6.29	1.59	0.4		
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char				6.29				
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database		UEPBX	USAC2	6.29	1.59	0.4	43.19	
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char		UEPBX	USAC2	6.29	1.59	0.4		
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update		UEPBX	USAC2	6.29	1.59	0.4	43.19	
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs		UEPBX UEPBX	USAC2 USACC	6.29	1.59	0.4	43.19	9.91
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update		UEPBX	USAC2	6.29	1.59	0.4	43.19	
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX UEPBX	USAC2 USACC	6.29	1.59	0.4	43.19	9.91
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs		UEPBX UEPBX	USAC2 USACC	6.29	1.59	0.4	43.19	9.91
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX UEPBX	USAC2 USACC	6.29	1.59	0.4	43.19	9.91
ADDITIONAL 2-WIRE VOICE	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)		UEPBX UEPBX	USAC2 USACC	6.29	1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOIC	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  oop Combination Rates		UEPBX UEPBX	USAC2 USACC		1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOIC	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  L Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  oop Combination Rates  2-Wire VG Loop/Port Combo - Zone	1	UEPBX UEPBX	USAC2 USACC	20.71	1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOIC	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  oop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	2	UEPBX UEPBX	USAC2 USACC	20.71 29.35	1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOIC	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  L Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  oop Combination Rates  2-Wire VG Loop/Port Combo - Zone		UEPBX UEPBX	USAC2 USACC	20.71	1.59	0.4	43.19	9.91
ADDITIONAL 2-WIRE VOIC	RINIG CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  coop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	2	UEPBX UEPBX	USAC2 USACC	20.71 29.35	1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOIC	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)	2 3	UEPBX UEPBX UEPBX	USAC2 USACC USAS2	20.71 29.35 37.68	1.59	0.4	43.19	9.91
ADDITIONAL 2-WIRE VOIC	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  OOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	3	UEPBX UEPBX UEPBX UEPBX	USAC2 USACC  USAS2	20.71 29.35 37.68	1.59	0.4	43.19	9.91
ADDITIONAL 2-WIRE VOIC	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  oop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loof (Rate Loop (SL 1) - Zone  2-Wire Voice Grade Loop (SL 1) - Zone	1 2	UEPBX  UEPBX  UEPBX  UEPBX  UEPRG  UEPRG	USAC2 USACC USAS2 USAS2 USAS2	20.71 29.35 37.68 17.02 25.66	1.59	0.4	43.19	9.91
ADDITIONAL 2-WIRE VOIC	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  OOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	3	UEPBX UEPBX UEPBX UEPBX	USAC2 USACC  USAS2	20.71 29.35 37.68	1.59	0.4	43.19	9.91
ADDITIONAL 2-WIRE VOIC	RRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  oop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loof (Rate Loop (SL 1) - Zone  2-Wire Voice Grade Loop (SL 1) - Zone	1 2	UEPBX  UEPBX  UEPBX  UEPBX  UEPRG  UEPRG	USAC2 USACC USAS2 USAS2 USAS2	20.71 29.35 37.68 17.02 25.66	1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOIC  UNE Port/Lo	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  COOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	1 2	UEPBX  UEPBX  UEPBX  UEPBX  UEPRG  UEPRG	USAC2 USACC USAS2 USAS2 USAS2	20.71 29.35 37.68 17.02 25.66	1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOIC  UNE Port/Lo	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)	1 2	UEPBX  UEPBX  UEPBX  UEPBX  UEPBX  UEPRG UEPRG UEPRG UEPRG	USAC2 USACC  USAS2  USAS2  UEPLX UEPLX UEPLX UEPLX	20.71 29.35 37.68 17.02 25.66 33.99	1.59	0.4	43.19 8.91 43.19	9.91
ADDITIONAL  2-WIRE VOIC  UNE Port/Lo	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)  COOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone	1 2	UEPBX  UEPBX  UEPBX  UEPBX  UEPRG  UEPRG	USAC2 USACC USAS2 USAS2 USAS2	20.71 29.35 37.68 17.02 25.66	1.59	0.4	43.19	9.91
ADDITIONAL  2-WIRE VOICE  UNE Loop R  2-Wire Voice	RING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update  L NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ  ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)	1 2	UEPBX  UEPBX  UEPBX  UEPBX  UEPBX  UEPRG UEPRG UEPRG UEPRG	USAC2 USACC  USAS2  USAS2  UEPLX UEPLX UEPLX UEPLX	20.71 29.35 37.68 17.02 25.66 33.99	1.59	0.4	43.19 8.91 43.19	9.91

FEATURES					1				1	
	All Features Offered		UEPRG	UEPVF	6.29	0	0	43.19	9.91	
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED									
									1	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As	_	UEPRG	USAC2		1.59	0.4	43.19	9.91	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change		UEPRG	USACC		1.59	0.4	43.19	9.91	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database	+	OLFING	USACC		1.55	0.4	45.19	5.51	
	Update					0.71		8.91	1	
	- Space	$\neg$				0		0.01		
ADDITIONAL	L NRCs									
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USAS2	0	0	0			
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou					14.64	14.64	19.99	19.99	19.99
o Mark	OF OR ARE LOOP WITH A WIRE LINE PORT (RUO, PRV)	_								
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	$-\!$							<b>——</b>	
LINE Bort/Lo	con Combination Pates	+							$\vdash$	
	pop Combination Rates    2-Wire VG Loop/Port Combo - Zone	1			20.71					
	2-Wire VG Loop/Port Combo - Zone	2			29.35					
	2-Wire VG Loop/Port Combo - Zone	3			37.68					
UNE Loop Ra										
	2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPPX	UEPLX	17.02					
	2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPPX	UEPLX	25.66					
	2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPPX	UEPLX	33.99					
2 Wire V-	e Grade Line Port Rates (BUS - PBX)	+		-	1					
	Eine Side Unbundled Combination 2-Way PBX Trunk Port - Bt	+	UEPPX	UEPPC	3.69			43.19	9.91	
	Line Side Unbundled Cumbination 2-way PBX Trunk Port - Bt	$-\!\!\!+\!\!\!-\!\!\!\!-\!\!\!\!-$	UEPPX	UEPPO	3.69			43.19	9.91	
	Line Side Unbundled Incoming PBX Trunk Port - Bu	-	UEPPX	UEPP1	3.69		+	43.19	9.91	
	2-Wire Voice Unbundled PBX LD Terminal Port	+	UEPPX	UEPLD	3.69			43.19	9.91	
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	3.69			43.19	9.91	
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	3.69			43.19	9.91	
	2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	3.69			43.19	9.91	
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	3.69			43.19	9.91	
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pr		UEPPX	UEPXE	3.69			43.19	9.91	
	2 Wiss Valor Habitandad 2 Watt BBY Habal/Harristal Forestern Administrative Collins B		UEPPX	UEPXL	2.00			43.19	9.91	
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P	-+-	UEPPX	UEPXL	3.69 3.69			43.19	9.91	
	2-vviile voice officialitied 2-vvay FBX Hotel/Hospital Economy Room Calling Fi	+	UEFFX	UEFAIVI	3.09			45.19	9.91	
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling P		UEPPX	UEPXO	3.69			43.19	9.91	
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	$\neg$	UEPPX	UEPXS	3.69			43.19	9.91	
	2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus Calling Pc		UEPPX	UEPXT	3.69			43.19	9.91	
	MBER PORTABILITY									
	Local Number Portability (1 per por		UEPPX	LNPCP	3.15					
FEATURES										
	All Features Offered	$-\!\!+\!\!-\!\!\!-\!\!\!\!-$	UEPPX	UEPVF	6.29	0	0	43.19	9.91	
	All Features Offerec	$-\!\!\!+\!\!\!-\!\!\!\!-\!\!\!\!-$	UEPPX	UEPVF	6.29	U	U	43.19	9.91	
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED	-							-	
	Tanto di anto de la trata del la trata de									
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As		UEPPX	USAC2		1.59	0.4	43.19	9.91	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with									
	Change		UEPPX	USACC		1.59	0.4	43.19	9.91	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database	17							ı [ —	T
	Update	!				0.71		8.91		
ADDITION	L NDC-	+		-	1				<del></del>	
ADDITIONAL		+	LIEDDY	LICACC	0	0	0		<del></del>	$\longrightarrow$
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou	+	UEPPX	USAS2	U	14.64	14.64	19.99	19.99	19.99
	F DA Gubsequent Activity - Change/Realiange Multilline Hunt Glot	+				14.04	14.04	19.99	19.99	19.99
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT	-								
UNE Port/Lo	pop Combination Rates									-
	2-Wire VG Coin Port/Loop Combo – Zone				21.06					
	2-Wire VG Coin Port/Loop Combo – Zone		-		21.06 29.7					
	2-Wire VG Coin Port/Loop Combo – Zone	$\perp \!\!\! \perp \!\!\! \perp$			28.03				$\perp$	
	tates									
UNE Loop Ra			LIEDOO	UEPLX	17.02					
UNE Loop Ra	2-Wire Voice Grade Loop (SL1) - Zone	-	UEPCO							
UNE Loop Ra	2-Wire Voice Grade Loop (SL1) - Zone	$\pm$	UEPCO	UEPLX	25.66					
UNE Loop Ra										
UNE Loop Ra	2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	25.66					_

2	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)										I	
-	0.1111 0 1 0.111 1 0 1 0 1 1 0 1 1 1 1 1		UEPCO	UEPSA	4.04				43.19	9.91		_
2	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (SI 2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)	-	UEPCO	UEPSH	4.04				43.19	9.91		+
2	2-wire Coin 2-way with Operator Screening and 011 Blocking; with Dialing Parity (SC)		UEPCO	UEPSC	4.04				43.19	9.91	ı	
-	2-Wire Coin 2-Way with Operator Screening and: 900 Blocking: 900/976, 1+DDD, 011+,		UEFCO	UEFSC	4.04				43.19	9.91		+
	and Local (SC)		UEPCO	UEPCC	4.04				43.19	9.91	ı	
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local; Enhanced		UEFCO	UEFCC	4.04				43.19	9.91		+
	Call OPT 3YV (SC)		UEPCO	UEPCE	4.04				43.19	9.91	İ	
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local; Enhanced		OLI OO	OLI OL	7.07				40.10	5.51		+
	Call OPT AP7 (SC)		UEPCO	UEPCF	4.04				43.19	9.91	İ	
	2-Wire Coin Outward without Blocking and without Operator Screening (S		UEPCO	UEPSG	4.04				43.19	9.91		+
5	2-Wire Coin Outward with Operator Screening and 011 Blocking (S		UEPCO	UEPSF	4.04				43.19	9.91		t
2	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)										i	T
	3 . , , ( ,		UEPCO	UEPSJ	4.04				43.19	9.91	i	
2	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and										i	т
	Local (SC)		UEPCO	UEPCM	4.04				43.19	9.91	İ	
	2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD, 011+, Local; Enhanced										i	Ī
	Calling OPT 3YW (SC)		UEPCO	UEPCP	4.04				43.19	9.91	İ	
2	2-Wire 2-Way Smartline with 900/976 (all states except L		UEPCO	UEPCK	4.04				43.19	9.91		П
	2-Wire Coin Outward Smartline with 900/976 (all states except L		UEPCO	UEPCR	4.04				43.19	9.91		
ADDITIONAL 7	UNE COIN PORT/LOOP (RC)										i	П
l	UNE Coin Port/Loop Combo Usage (Flat Rate		UEPCO	URECU	4.05	0	0				ı	
												┖
	BER PORTABILITY											
L	Local Number Portability (1 per por		UEPCO	LNPCX	0.35						ı	1
<b></b>												1
FEATURES											<b></b>	1
P	All Features Offered	_	UEPCO	UEPVF	6.29	0	0		41.86	14.46		+
NONDECHE	DING CHARGES CURRENTLY COMPINED	-			-							+
	RING CHARGES - CURRENTLY COMBINED	-	LIEDOO	110400	-	1.50	0.4		42.40	0.04		+
2	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPCO	USAC2	-	1.59	0.4		43.19	9.91		+
-	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with char		UEPCO	USACC		1.59	0.4		43.19	9.91	ı	1
<del>     </del>	2-vviile voice Grade Loop / Line Fort Combination - Conversion - Switch with char		UEFCO	USACC	1	1.09	0.4		43.18	9.91		+
ADDITIONAL I	NRCs				<del>                                     </del>							+
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPCO	USAS2		0	0		43.19	9.91		+
<del></del>	2 This Voice Stade EcopyEme Fort Combination - Subsequent Activ	-	0L1-00	OUNUZ	<b>†</b>	U	<u> </u>		70.10	5.51		+
2-WIRE VOIC	E GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT										í	
											i	Т
UNE Port/Loc	op Combination Rates										i	
2	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	1			29.68						i	Т
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	2			37.74						i	
2	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone	3			44.4							
UNE Loop Rat	ates										i	
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	1	UEPPX	UECD1	20.85				19.99	19.99	19.99	
2	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	2	UEPPX	UECD1	28.91				19.99	19.99	19.99	
2	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone	3	UEPPX	UECD1	35.57				19.99	19.99	19.99	
UNE Port Rate												
E	Exchange Ports - 2-Wire DID Por		UEPPX	UEPD1	8.83				19.99	19.99	19.99	9
<u> </u>					1							1
	RING CHARGES - CURRENTLY COMBINED		( Imper)	110.5.	1	44	0.70		10.77	40.77	40.55	1
<del></del>	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as	_	UEPPX	USAC1	+	14.62	3.73		19.99	19.99	19.99	+
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable		HEDDY	110440	1	44.00	0.70		40.00	40.00	40.00	1
- C	Changes		UEPPX	USA1C	-	14.62	3.73		19.99	19.99	19.99	+
ADDITIONAL I	NPCe				1							+
	2-Wire DID Subsequent Activity - Add Trunks, Per Trun		UEPPX	USAS1	1	53.68			19.99	19.99	19.99	+
<del></del>	2 This Sis cassequent richity rich frame, i et ffull	-	OLITA	OUAUI	<del>                                     </del>	55.00			15.55	13.33	10.00	+
Telephone No	umber/Trunk Group Establisment Charges	-			<del>                                     </del>							+
	DID Trunk Group Establishent Charges	-	UEPPX	NDT	0	0	0		19.99	19.99	19.99	+
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbe		UEPPX	NDZ	Ŭ	0	0		19.99	19.99	19.99	$^{\dagger}$
	Additional DID Numbers for each Group of 20 DID Numbe		UEPPX	ND4	0	0	0		19.99	19.99	19.99	T
	DID Numbers, Non- consecutive DID Numbers , Per Numbe		UEPPX	ND5	_	0	0	19.99				T
R	Reserve Non-Consecutive DID number		UEPPX	ND6	0	0	0	 19.99				l
	Reserve DID Numbers		UEPPX	NDV	0	0	0	19.99			i	
					L							
LOCAL NUME	BER PORTABILITY											
	Local Number Portability (1 per por		UEPPX	LNPCP	3.15							
	· · · · · · · · · · · · · · · · · · ·											Ľ
L						1						L
L	DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT											
2-WIRE ISDN											l	+
2-WIRE ISDN	DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT  op Combination Rates											
2-WIRE ISDN		1	UEPPB UEPPR		38.58							

Page 15 of 22

	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone	3	UEPPB	UEPPR	,	55.29								İ	
UNE Loop R	ates														
	2-Wire ISDN Digital Grade Loop - UNE Zone	1	UEPPB	UEPPR	USL2X	27.38						19.99	19.99	19.99	
	2-Wire ISDN Digital Grade Loop - UNE Zone	2	UEPPB	UEPPR	USL2X							19.99	19.99	19.99	
	2-Wire ISDN Digital Grade Loop - UNE Zone	3	UEPPB	UEPPR	USL2X	44.09						19.99	19.99	19.99	
UNE Port Ra		Ŭ	OZ. I D	<u> </u>	UOLEX	11.00						10.00	10.00	10.00	Ŧ
ONL FOIL K	Exchange Port - 2-Wire ISDN Line Side Po		UEPPB	UEPPR	UEPPB	11.2						19.99	19.99	19.99	İ
NONRECUR	RING CHARGES - CURRENTLY COMBINED														
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Convers		UEPPB	UEPPR	USACB	0	77.18	54.15				19.99	19.99	19.99	
ADDITIONAL	L NRCs														+
LOCAL NUM	MBER PORTABILITY														Ŧ
LOCAL HOI			LIEDDD	UEPPR	LNPCX	0.35	0	0							+
	Local Number Portability (1 per por		UEPPB	UEPPR	LNPCX	0.35	0	0							+
B-CHANNEL	USER PROFILE ACCESS:														+
	CVS/CSD (DMS/5ESS)  CVS (EWSD)		UEPPB UEPPB		U1UCA U1UCB	0	0	0					<b>  </b>		4
	CSD CSD		UEPPB		U1UCC	0	0	0							$\pm$
R-CHANNEI	AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)												$\vdash$		+
D-OHARREL	CVS/CSD (DMS/5ESS)		UEPPB	UEPPR	U1UCD	0	0	0							1
	CVS (EWSD)		UEPPB	UEPPR	U1UCE	0	0	0						i	
	CSD		UEPPB	UEPPR	U1UCF	0	0	0							
LISED TEDM	MINAL PROFILE														I
OOLK TEKIN	User Terminal Profile (EWSD only)		LIEPPR	UEPPR	U1UMA	0	0	0						 	1
VERTICAL F			OLITE	OLITIK	OTOMIX	0	Ŭ	0							İ
VERTICAL															+
	All Vertical Features - One per Channel B User Profile		UEPPB	UEPPR	UEPVF	6.29	0	0							+
INTEROFFIC	CE CHANNEL MILEAGE Interoffice Channel mileage each, including first mile and facilities termination		LIEDDE	UEPPR	M1GNC	20.74	136.44	51.37				19.99	19.99	19.99	10
	Interoffice Channel mileage each, including first mile and facilities termination  Interoffice Channel mileage each, additional mile		UEPPB		M1GNM	0.0373	0	0			0	19.99	19.99	19.99	9
4-WIRE DS1	DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT														
UNE Port/Lo	pop Combination Rates														+
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	1		PPP		221.03									I
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	3		PPP PPP		301.73 434.8									
UNE Loop R	Rates												<del>                                     </del>		+
	4-Wire DS1 Digital Loop - UNE Zone	1		PPP	USL4P	113.59						19.99	19.99	19.99	I
	4-Wire DS1 Digital Loop - UNE Zone : 4-Wire DS1 Digital Loop - UNE Zone :	3		PPP PPP	USL4P USL4P	194.29 327.36						19.99 19.99	19.99 19.99	19.99 19.99	+
UNE Port Ra	ate														+
<b></b>	Exchange Ports - 4-Wire ISDN DS1 Por		UE	PPP	UEPPP	107.44						19.99	19.99	19.99	#
NONRECUR	RING CHARGES - CURRENTLY COMBINED														Ŧ
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion - Switch-as-is		110	EPPP	USACP	0	238.67	157.46				19.99	19.99	19.99	
			- OL	FFF	USACI	0	230.07	137.40				15.55	19.99	19.99	İ
ADDITIONAL	L NRCs  4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within										1		$\vdash$		+
	Std Allowance  4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All State		UE	EPPP	PR7TF		0.9822				1	19.99	19.99	19.99	1
	except NC)		UE	EPPP	PR7TO		23.02	23.02			1	19.99	19.99	19.99	1
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowance		UE	PPP	PR7ZT		46.05	46.05				19.99	19.99	19.99	1
	1	- 1	1	l.	1			1	1	1	1	1			
					i i								1	ı	

	Local Number Portability (1 per por		UEPPP	LNPCN	1.75								
	Voice/Data		UEPPP	PR71V	0	0	0						I
	Digital Data		UEPPP	PR71D	0	0	0						I
	Inward Data		UEPPP	PR71E	0	0	0						
New or Add	ditional "B" Channel		LIEDDD	DD7D1		00.14				40.00	40.00		_
	New or Additional - Voice/Data B Channel		UEPPP UEPPP	PR7BV PR7BF	0	29.11				19.99	19.99 19.99	19.99	
	New or Additional - Digital Data B Channel		UEPPP	PR7BD	-	29.11				19.99		19.99	
	New or Additional Inward Data B Channel  New or Additional Useage Sensitive Voice Data B Channel		UEPPP	PR7BS	0	29.11 29.11				19.99 19.99	19.99 19.99	19.99 19.99	
	New or Additional Useage Sensitive Voice Data B Channel		UEPPP	PR7BU	0	29.11				19.99	19.99	19.99	
	New of Additional Oseage Sensitive Digital Data & Channel		OLFFF	FICTO	U	29.11				19.99	19.99	19.99	,
CALL TYPE	-S												1
OALL TITL	Inward		UEPPP	PR7C1	0	0	0						1
	Outward		UEPPP	PR7C0	0	0	0						1
	Two-way		UEPPP	PR7CC	0	0	0						1
													Ī
Interoffice C	Channel Mileage		•										I
	Fixed Each Including First Mile		UEPPP	1LN1A	95.7398	216.27	162.7	0		19.99	19.99	19.99	)
	Each Airline-Fractional Additional Mil		UEPPP	1LN1B	0.7598								4
4-WIRE DS1	1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT												
UNE Port/Le	oop Combination Rates		LIEDDO		107.01		ļ				10.05		-
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	1	UEPDC	-	187.21		-	<del>                                     </del>		19.99	19.99	19.99	
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone : 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone :	3	UEPDC UEPDC		267.91 400.98			<del>                                     </del>		19.99 19.99	19.99 19.99	19.99 19.99	
	+** DOT DIGITAL LOOP/+** DOTTO THANK FOR - ONE ZONE	3	UEFDC		400.90					19.99	19.99	19.99	1
UNE Loop F	Rates												1
	4-Wire DS1 Digital Loop - UNE Zone	1	UEPDC	USLDC	113.59					19.99	19.99	19.99	1
	4-Wire DS1 Digital Loop - UNE Zone :	2	UEPDC	USLDC				2	3.46	19.99	19.99	19.99	1
	4-Wire DS1 Digital Loop - UNE Zone	3	UEPDC	USLDC	327.36					19.99	19.99	19.99	
UNE Port R			LIEBBO	UDDAT	70.00					40.00	40.00		
	4-Wire DDITS Digital Trunk Por		UEPDC	UDD1T	73.62					19.99	19.99	19.99	9
NONRECUE	RRING CHARGES - CURRENTLY COMBINED												-
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as		UEPDC	USAC4		259.56	134.33			19.99	19.99	19.99	9
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with DS1												Ì
	Changes		UEPDC	USAWA		259.56	134.33			19.99	19.99	19.99	9
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with												
	Change - Trunk		UEPDC	USAWB		259.56	134.33			19.99	19.99	19.99	9
ADDITIONA	AL NRCs												
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-												1
	Way Outward Trunk		UEPDC	UDTTB		29.01	29.01			19.99	19.99	19.99	9
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan Inward												۱
	Trunk w/out DID		UEPDC	UDTTC		29.01	29.01			19.99	19.99	19.99	9
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan - Inward												۱
	Trunk with DID  4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way		UEPDC	UDTTD		29.01	29.01	<del>                                     </del>		19.99	19.99	19.99	3
	DID w User Trans		UEPDC	UDTTE		29.01	29.01			19.99	19.99	19.99	
BIPOI AR S	E ZERO SUBSTITUTION		ULFUC	ODITE	1	25.01	23.01	<del>                                     </del>		19.99	19.99	19.99	1
JEAN 0	B8ZS -Superframe Format		UEPDC	CCOSF		0	605			19.99	19.99	19.99	1
	B8ZS - Extended Superframe Forma		UEPDC	CCOEF		0	605			19.99	19.99	19.99	1
Alternate M	lark Inversion												1
	AMI -Superframe Format		UEPDC	MCOSF	1	0	0	<b></b>					1
	AMI - Extended SuperFrame Forms		UEPDC	МСОРО		0	0						1
													+
Telephone	Number/Trunk Group Establisment Charges												
	Telephone Number for 2-Way Trunk Grou		UEPDC	UDTGX	0								-
	Telephone Number for 1-Way Outward Trunk Grou	_	UEPDC	UDTGY	0		+			19.99	1		-
	Telephone Number for 1-Way Inward Trunk Group Without D		UEPDC	UDTGZ	0	0	0	<del>                                     </del>		19.99 19.99	_		1
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbe DID Numbers for each Group of 20 DID Number		UEPDC UEPDC	NDZ ND4	0	U	U			19.99			+
	DID Numbers for each Group of 20 DID Number  DID Numbers, Non- consecutive DID Numbers , Per Number		UEPDC	ND5	0		<b>†</b>	<del>                                     </del>		19.99	1		+
	Reserve Non-Consecutive DID Nos		UEPDC	ND6	0	0	0			19.99			1
	Reserve DID Numbers		UEPDC	NDV	0	Ö	0			19.99			
Dedicated [	DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Tr	unk Port											1
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination		UEPDC	1LNO1	94.98	216.27	162.7	0	0	19.99	19.99	19.99	
	Interoffice Channel Mileage - Additional rate per mile - 0-8 mil		UEPDC	1LNOA	0.7598	0	0						1
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination		UEPDC	1LNO2 1LNOB	0.7598	0	0	<del>                                     </del>					-
					0.7508		0		1	1 1		1	
	Interoffice Channel Mileage - Additional rate per mile - 9-25 mil Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination		UEPDC UEPDC	1LNO3	0.7550	0	0	0					+

	Local Number Portability, per DS0 Activate		UEPDC	LNPCP	3.15	0	0	0							
	Central Office Termininating Poir		UEPDC	CTG	0										
4-WIRE DS1	LOOP WITH CHANNELIZATION WITH PORT														
	DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations														
	m can have up to 24 combinations of rates depending on type and number of ports used														_
Each System	m can have up to 24 combinations of rates depending on type and number of ports used	_													_
															_
UNE DS1 Loc															
	4-Wire DS1 Loop - UNE Zone 1	1	UEPMG	USLDC	113.59	0	0								
	4-Wire DS1 Loop - UNE Zone 2	2	UEPMG	USLDC	194.29	0	0								
	4-Wire DS1 Loop - UNE Zone 3	3	UEPMG	USLDC	327.36	0	0								
_						-	-								
LINE DOO CH	hannelization Capacities (D4 Channel Bank Configurations)														_
			UEPMG	VUM24	100 17		-			-					_
	24 DSO Channel Capacity - 1 per DS1				103.47	0	0								
	48 DSO Channel Capacity - 1 per 2 DS1s		UEPMG	VUM48	206.94	0	0								
	96 DSO Channel Capacity -1per 4 DS1s		UEPMG	VUM96	413.88	0	0								
	144 DS0 Channel Capacity - 1 per 6 DS1s		UEPMG	VUM14	620.82	0	0								
	192 DS0 Channel Capacity -1 per 8 DS1s		UEPMG	VUM19	827.76	0	0								
	240 DS0 Channel Capacity - 1 per 10 DS1s	-	UEPMG	VUM20	1034.7	0	0			1					+
		+				0	0	-	+	1	+	+	+	+	+
	288 DS0 Channel Capacity - 1 per 12 DS1s	-	UEPMG	VUM28	1241.64	0	U		1	1	1		1		_
	384 DS0 Channel Capacity - 1 per 16 DS1s		UEPMG	VUM38	1655.52	0	0								
T	480 DS0 Channel Capacity - 1 per 20 DS1s		UEPMG	VUM40	2069.4	0	0								
	576 DS0 Channel Capacity -1 per 24 DS1s		UEPMG	VUM57	2483.28	0	0								T
	672 DS0 Channel Capacity - 1 per 28 DS1s		UEPMG	VUM67	2897.16	0	0			1	1				
+	The second secon	-					1		1		1	1			_
Non Provi	ing Charges (NBC) Associated with 4 Wire DC4 I with Charge-linites with D 1 2	n C'	no Boood on - Co		1	1			+	+	+	+	+	+	+
	ing Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - Conversio			m	1	1			+	1	1	-	1	+	_
	System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DSO Ports with					_									
	this configuration functioning as one are considered Add'l after the minimum system configu	ration i	is counted.		<u> </u>										
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes		UEPMG	USAC4	0	301.62	16.76					19.99	19.99	19.99	19
	litions at End User Locations Where 4-Wire DS1 Loop with Channelization with Port Combinat	ion Cu	rrently Exists and												
	urrently Combined) In Georgia & Tennessee Only		I ZNOLO UNU							1					
	NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature Activation -	-								1					-
			LIEDIAO				405.04	4 40 00	47.00			40.00	40.00		
	New GA & TN Only		UEPMG	VUMD4	U	717.71	425.81	149.08	17.69			19.99	19.99	19.99	19
	ero Substitution														
	Clear Channel Capability Format, superframe - Subsequent Activity Only		UEPMG	CCOSF	0	0	605					19.99	19.99	19.99	19
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only		UEPMG	CCOEF	0	0	605					19.99	19.99	19.99	19
Alternate Ma	ark Inversion (AMI)														
7 atomato ma	Superframe Format	_	UEPMG	MCOSF	0	0	0			1					
	Extended Superframe Format		UEPMG		0	0	0			-					_
	Extended Superirame Format		UEPING	MCOPC	U	0	0								_
Exchange Po	Ports Associated with 4-Wire DS1 Loop with Channelization with Port														
Exchange Po	Ports														
	Line Side Combination Channelized PBX Trunk Port - Business		UEPPX	UEPCX	1.65	0	0	0	0		19.99				
	Line Side Outward Channelized PBX Trunk Port - Business		UEPPX	UEPOX	1.65	0	0	0	0		19.99				-1-
	Line Side Inward Only Channelized PBX Trunk Port without DID		UEPPX	UEP1X	1.65	0	0	0	0		19.99				-
			UEPPX	UEPDM		0	0	0	0	-					_
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port	-	UEPPA	UEPUN	8.86	U	U	U	U	1	19.99	-	1	-	
	ivations - Unbundled Loop Concentration														
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX	1PQWM	0.7	25.45	13.44	4.2	4.17			19.99	19.99	19.99	19
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		UEPPX	1PQWU	0.7	78.31	18.46	59.37	11.6			19.99	19.99	19.99	19
	Number/ Group Establishment Charges for DID Service														
	DID Trunk Termination (1 per Port)	+	UEPPX	NDT	0					1	1				-
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)	+	UEPPX	NDZ	0	0	0	-1	+	1	19.99	1	1	+	+
		+			-	U	U	-	+	1		+	+	+	+
	DID Numbers - groups of 20 - Valid all States	_	UEPPX	ND4	0	U	U		1		19.99		1		_
1 1	Non-Consecutive DID Numbers - per number		UEPPX	ND5	0	0	0				19.99				
	Reserve Non-Consecutive DID Numbers		UEPPX	ND6	0	0	0				1		1		
		T	UEPPX	NDV	0	0	0			1					T
	Reserve DID Numbers				1										
					3.15	0	0		1		1	1			_
Local Numbe	er Portability		LIEPPY			U	v			1	1	+	+	+	+
Local Numbe	er Portability Local Number Portability - 1 per port		UEPPX	LNPCP	0.10									1	
Local Numbe	er Portability Local Number Portability - 1 per port - Vertical and Optional		UEPPX	LNPCP	0.10										
Local Numbe	er Portability   Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only														
Local Numbe	er Portability Local Number Portability - 1 per port - Vertical and Optional		UEPPX	UEPVF	6.29	0	0				19.99				-
Local Numbe	er Portability   Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only					0	0				19.99				
Local Numbe	er Portability   Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only					0	0				19.99				
Local Numbe	er Portability   Local Number Portability - 1 per port   Vertical and Optional     hing Features Offered with Line Side Ports Only     All Features Available					0	0				19.99				
Local Numbe	er Portability   Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only					0	0				19.99				
Local Numbe	er Portability   Local Number Portability - 1 per port   Vertical and Optional     hing Features Offered with Line Side Ports Only     All Features Available					0	0				19.99				
Local Numbe FEATURES - Local Switch	er Portability   Local Number Portability - 1 per port   - Vertical and Optional   hing Features Offered with Line Side Ports Only   All Features Available		UEPPX	UEPVF		0	0				19.99				
Local Number FEATURES - Local Switch FORT LOOF Market Rates	er Portability  Local Number Portability - 1 per port  - Vertical and Optional hing Features Offered with Line Side Ports Only  All Features Available  - P COMBINATIONS - MARKET RATES  s shall apply where BellSouth is not required to provide unbundled local switching or switch port	s per F	UEPPX	UEPVF		0	0				19.99				
FEATURES - Local Switch  ED PORT LOOF  Market Rates These scenar	er Portability   Local Number Portability - 1 per port   - Vertical and Optional   hing Features Offered with Line Side Ports Only   All Features Available		UEPPX  CC and/or State Com	UEPVF mission rules.		0	0				19.99				
Local Number FEATURES - Local Switch    Department    ED PORT LOOF    Market Rates   These scenar	er Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available - P COMBINATIONS - MARKET RATES - sshall apply where BellSouth is not required to provide unbundled local switching or switch port arios include: - de port/loop combinations that are Not Currently Combined in all of the BellSouth states except a	as note	UEPPX  CC and/or State Com d for Georgia and Tei	UEPVF mission rules. nnessee.	6.29	0	0				19.99				
Local Number FEATURES - Local Switch    Department    ED PORT LOOF    Market Rates   These scenar	er Portability   Local Number Portability - 1 per port   - Vertical and Optional   hing Features Offered with Line Side Ports Only   All Features Available	as note	UEPPX  CC and/or State Com d for Georgia and Tei	UEPVF mission rules. nnessee.	6.29	0 or more DS0 e	0				19.99				
FEATURES - Local Switch  Department of the second of the s	er Portability   Local Number Portability - 1 per port  - Vertical and Optional  hing Features Offered with Line Side Ports Only   All Features Available   Proceedings	as note	UEPPX  CC and/or State Com d for Georgia and Te 8 MSAS in BellSouth	mission rules.	6.29						19.99				
Local Number FEATURES - Local Switch  D PORT LOOP  Market Rates These scenar 1. Unbundler 2. Unbundler The Top 8 M	er Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available - P COMBINATIONS - MARKET RATES - sshall apply where BellSouth is not required to provide unbundled local switching or switch port arios include: - de port/loop combinations that are Not Currently Combined in all of the BellSouth states except a	as note he Top ans); No	UEPPX  CC and/or State Com d for Georgia and Te 8 MSAS in BellSouth	mission rules.	6.29 d users with 4 oint/Charlott	e-Gastonia-Roc	k Hill); TN (Na:	shville).	ding in lieu of t	he Market R		serves the rinh	t to true-up th	ne billina differ	rence

Page 18 of 22 Version 2Q01: 08/30/01

-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)								_	
	op Combination Rates									
	2-Wire VG Loop/Port Combo - Zone	1			31.02					
	2-Wire VG Loop/Port Combo - Zone	2			39.66					
	2-Wire VG Loop/Port Combo - Zone	3			47.99					+
JNE Loop Ra	atos									+
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRX	UEPLX	17.02					+
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRX	UEPLX	25.66					_
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRX	UEPLX	33.99					
	Grade Line Port (Res)									
	2-Wire voice unbundled port - residenc		UEPRX	UEPRL	14	90	90	43.		
	2-Wire voice unbundled port with Caller ID - re		UEPRX	UEPRC	14	90	90	43.		
	2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundles res, low usage line port with Caller ID (LU		UEPRX	UEPRO UEPAP	14 14	90 90	90	43. 43.		+
	2-Wife voice unburidles res, low usage line port with Caller ID (LO		UEFRA	UEFAF	14	90	90	43.	9 9.91	+
OCAL NUM	BER PORTABILITY									1
	Local Number Portability (1 per por		UEPRX	LNPCX	0.35					1
FEATURES			-							
	All Features Offered		UEPRX	UEPVF	0	0	0			
				1	1				-	
ADDITIONAL	NPCo				1		+ + + + + + + + + + + + + + + + + + + +			+
	NRCs - 2-Wire Voice Grade Loop/Line Port Combination - Subseque	-	UEPRX	USAS2	1	0	0		-	+
	INING - 2-VALIE VOICE GLADE LOOP/LINE FOR COMBINATION - Subseque	-	UEPKA	USASZ	1	U	0		-	+
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)						<del>                                     </del>			1
2 .010										1
UNE Port/Loc	op Combination Rates									
	2-Wire VG Loop/Port Combo - Zone	1			31.02					
	2-Wire VG Loop/Port Combo - Zone	2			39.66					
	2-Wire VG Loop/Port Combo - Zone	3			47.99					
UNE Loop Ra	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	17.02					
	2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone	2	UEPBX	UEPLX	25.66					+
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	33.99					+
										_
2-Wire Voice	Grade Line Port (Bus)									
	2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	14	90	90	43.		
	2-Wire voice unbundled port with Caller + E484 ID - bi		UEPBX	UEPBC	14	90	90	43.	9 9.91	
	2-Wire voice unbundled port outgoing only - bt		UEPBX	UEPBO	14	90	90	43.	9 9.91	
	2-Wire voice Grade unbundled South Carolina extended local dialing parity port with Ca ID - bus		UEPBX	UEPAZ	14	90	90	43.	9 9.91	
	2-Wire voice unbundled South Carolina Bus Area Calling Port with Caller ID (LM		UEPBX	UEPAB	14	90	90	43.		+
	2-Wife voice dribuildied South Calonila bus Alea Caning Fort with Caner ID (Livi		OLFBA	OLFAD	14	30	90	45.	3.31	+
LOCAL NUM	BER PORTABILITY									
	Local Number Portability (1 per por		UEPBX	LNPCX	0.35					
			-							
FEATURES										
IONIDES	DING CHARGES CURRENTLY COMPINED								-	
NONRECURE	RING CHARGES - CURRENTLY COMBINED						+ + + + + + + + + + + + + + + + + + + +		-	
ADDITIONAL	NDCe			+	1				-	+
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque	-	UEPBX	USAS2	1	0	0		-	+
	111.0 2 11110 VOICE Grade Loop/Line Fort Combination - Subseque		OLFDA	00/102		Ü	<u> </u>			1
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)									
	op Combination Rates		-							
	2-Wire VG Loop/Port Combo - Zone	1			31.02					
	2-Wire VG Loop/Port Combo - Zone	2		1	39.66		+		-	
	2-Wire VG Loop/Port Combo - Zone	3			47.99		+ + + + + + + + + + + + + + + + + + + +			+
JNE Loop Ra	atne	-		+	1		<del>                                     </del>		-	+
DINE LOOP RE	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRG	UEPLX	17.02		+ + + + + + + + + + + + + + + + + + + +			+
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRG	UEPLX	25.66		<del>                                     </del>			+
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRG	UEPLX	33.99					1
	Grade Line Port Rates (RES - PBX)		-							
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	14	90	90	43.	9 9.91	
					1					
OCAL NUM	BER PORTABILITY	1		1	1	1		1 1		
	Local Number Portability (1 per por		UEPRG	LNPCP	3.15					

NONRECUR	RRING CHARGES - CURRENTLY COMBINED									
		$\Box$								
ADDITIONA	AL NRCs	$\rightarrow$		1	1		1			
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurri				1	0	0			
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou	$\overline{}$				14.64	14.64		19.99	19.99 19.99
	1 BX GabSequent Activity Grange/Realitange Malanine Frant Grot	-				14.04	14.04		10.00	15.55
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)									
UNE Port/Lo	oop Combination Rates									
	2-Wire VG Loop/Port Combo - Zone	1			31.02					
	2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone	3			39.66 47.99					
	2-Wile VG Loop/Fort Combo - Zone				47.99					
UNE Loop F	Rates									
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPPX	UEPLX	17.02					
	2-Wire Voice Grade Loop (SL1) - Zone	2	UEPPX	UEPLX	25.66					
	2-Wire Voice Grade Loop (SL1) - Zone	3	UEPPX	UEPLX	33.99					
		$\longrightarrow$		1	1		1			
2-Wire Voic	te Grade Line Port Rates (BUS - PBX)	$\rightarrow$	LIEDDY	LIEDDO	4.		00		10.10	0.04
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bu	$\rightarrow$	UEPPX	UEPPC	14	90	90		43.19	9.91
	Line Side Unbundled Outward PBX Trunk Port - Bu Line Side Unbundled Incoming PBX Trunk Port - Bu	+	UEPPX UEPPX	UEPPO UEPP1	14 14	90	90 90	+	43.19 43.19	9.91 9.91
	2-Wire Voice Unbundled PBX LD Terminal Port	+	UEPPX	UEPLD	14	90	90	+	43.19	9.91
	2-Wire Voice Unbundled PBX ED Terminal Port  2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc	-	UEPPX	UEPXA	14	90	90		43.19	9.91
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	14	90	90		43.19	9.91
	2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	14	90	90		43.19	9.91
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	$\perp$	UEPPX	UEPXD	14	90	90		43.19	9.91
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc	$\longrightarrow$	UEPPX	UEPXE	14	90	90		43.19	9.91
	2 Wire Voice Unbundled 2 Way PRV Hetel/Hearth-I Fare and Administrative C. "		HEDDY	HEBY	14	90	00		40.40	0.01
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling P 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Pi	+	UEPPX UEPPX	UEPXL	14	90	90 90		43.19 43.19	9.91 9.91
	2 *****C **Oloc Onbundled 2-***ay F DA Flote#HoSpital Economy Room Calling P	+	ULFFA	GEFAIVI	14	30	9U		45.19	3.31
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling P		UEPPX	UEPXO	14	90	90		43.19	9.91
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc	$\rightarrow$	UEPPX	UEPXS	14	90	90		43.19	9.91
LOCAL NUM	MBER PORTABILITY									
	Local Number Portability (1 per por		UEPPX	LNPCP	3.15					
		$\longrightarrow$								
FEATURES		$\longrightarrow$								
NONRECUR	RRING CHARGES - CURRENTLY COMBINED									
ADDITIONA	IL NRCs									
	2-Wire Voice Grade Loop/ Line Port Combination - Subseque		UEPPX	USAS2		0	0			
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurri	$\longrightarrow$				0	0		19.99	
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou						14.64			
i e						14.64			13.33	19.99 19.99
2-WIDE VOI	ICE CRADE LOOP WITH 2-WIPE ANALOG LINE COIN BORT					14.64			13.33	19.99 19.99
2-WIRE VOI	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT					14.04			10.00	19.99
						14.04			13.33	19.99
	oop Combination Rates				31.02	14.04			10.00	19.99
	oop Combination Rates				39.66	14.04			10.00	19.99
	oop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone					14.04			10.00	19.99
UNE Port/Lo	oop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone				39.66	14.04			10.00	19.99
	oop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  Rates		LIPPO C		39.66 47.99	14.04			10.00	19.99
UNE Port/Lo	Ooop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  Rates  2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	39.66 47.99 17.02	14.04			10.00	19.99
UNE Port/Lo	oop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	39.66 47.99 17.02 25.66	14.04			10.00	19.99
UNE Port/Lo	Ooop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  Rates  2-Wire Voice Grade Loop (SL1) - Zone				39.66 47.99 17.02	14,04			10.00	19.99
UNE Port/Lo	oop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone    Compared Co		UEPCO	UEPLX	39.66 47.99 17.02 25.66	14.04			10.00	19.99
UNE Port/Lo	oop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	39.66 47.99 17.02 25.66	90	90		43.19	19.99
UNE Port/Lo	cop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K		UEPCO UEPCO	UEPLX UEPLX UEPSD	39.66 47.99 17.02 25.66 33.99	90			43.19	9.91
UNE Port/Lo	Coop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K  LA, MS, SC)		UEPCO UEPCO	UEPLX UEPLX	39.66 47.99 17.02 25.66 33.99		90			
UNE Port/Lo	cop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K		UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA	39.66 47.99 17.02 25.66 33.99	90	90		43.19 43.19	9.91
UNE Port/Lo	cop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K  LA, MS, SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)		UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA UEPSA	39.66 47.99 17.02 25.66 33.99	90 90 90	90		43.19 43.19 43.19	9.91 9.91 9.91
UNE Port/Lo	coop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K  LA, MS, SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)		UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA	39.66 47.99 17.02 25.66 33.99	90	90		43.19 43.19	9.91
UNE Port/Lo	cop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K  LA, MS, SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA UEPSA UEPSH	39.66 47.99 17.02 25.66 33.99 14 14 14	90 90 90 90	90 90 90		43.19 43.19 43.19 43.19	9.91 9.91 9.91 9.91
UNE Port/Lo	coop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K  LA, MS, SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)  2-Wire Coin 2-Way with Operator Screening and O11 Blocking (S)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)		UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA UEPSA	39.66 47.99 17.02 25.66 33.99	90 90 90	90		43.19 43.19 43.19	9.91 9.91 9.91
UNE Port/Lo	cop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way without Operator Screening and without Blocking (S 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K LA, MS, SC) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (S) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA UEPSA UEPSH UEPSC	39.66 47.99 17.02 25.66 33.99 14 14 14 14	90 90 90 90 90	90 90 90 90		43.19 43.19 43.19 43.19 43.19	9.91 9.91 9.91 9.91 9.91
UNE Port/Lo	cop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way with Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K  LA, MS, SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (SC)		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA UEPSA UEPSH	39.66 47.99 17.02 25.66 33.99 14 14 14	90 90 90 90	90 90 90		43.19 43.19 43.19 43.19	9.91 9.91 9.91 9.91
UNE Port/Lo	cop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone 2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Coin 2-Way without Operator Screening and without Blocking (S 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K LA, MS, SC) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (S) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC) 2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and		UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO UEPCO	UEPLX UEPLX UEPSD UEPRA UEPSA UEPSH UEPSC	39.66 47.99 17.02 25.66 33.99 14 14 14 14	90 90 90 90 90	90 90 90 90		43.19 43.19 43.19 43.19 43.19	9.91 9.91 9.91 9.91 9.91
UNE Port/Lo	coop Combination Rates  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire VG Coin Port/Loop Combo – Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Coin 2-Way without Operator Screening and without Blocking (S  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, K  LA, MS, SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking (S)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)  2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (SC)  2-Wire Coin 2-W Oper Screen & Blocking: 900/976, 1+DDD, 011+ & Local; Enhanced		UEPCO C UEPCC	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90 90 90 90	90 90 90 90 90		43.19 43.19 43.19 43.19 43.19	9.91 9.91 9.91 9.91 9.91 9.91	

2-Wire Coin Outward with Operator Screening and 011 Blocking (S	UEPCO	UEPSF	14	90	90		43.19	9.91	
2-Wire Coin Outward with Operator Screening and 011 Blocking (S 2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)	UEFCO	UEFSF	14	90	90		43.19	9.91	<del>                                     </del>
2-Wile Colli Outward with Operator Screening and Blocking, 011, 900/976, 1+DDD (SC)	UEPCO	LIEDCI	4.4	00	00		42.40	0.04	
	UEPCO	UEPSJ	14	90	90		43.19	9.91	
2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and									
Local (SC)	UEPCO	UEPCM	14	90	90		43.19	9.91	
2-Wire Coin Out Oper Screen & Block: 900/976, 1+DDD, 011+, & Local ; w/ Enhanced									
Call OPT 3YW (SC)	UEPCO	UEPCP	14	90	90		43.19	9.91	
LOCAL NUMBER PORTABILITY									
Local Number Portability (1 per por	UEPCO	LNPCX	0.35						
NONRECURRING CHARGES - CURRENTLY COMBINED									-
ADDITIONAL NRCs									
2-Wire Voice Grade Loop/ Line Port Combination - Subseque	UEPCO	USAS2		0	0				

Page 21 of 22 Version 2Q01: 08/30/01

RY NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	RATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic- t Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increme Charg Manual Order Electro Disc A
			$\vdash$				Nonre	curring					ecurring			
			$\vdash$								,		connect			
			$\vdash \vdash$			Rec	First	Add'l	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
			لــــا													
The "Zone" s http://www.in	shown in the sections for stand-alone loops or loops as part of a combination refers to terconnection.bellsouth.com/become_a_clec/html/interconnection.htm	Geograph	nically	Deaveraged UNE Z	ones. To v	iew Geograph	nically Deaverag	ed UNE Zone [	Designations	by Central O	ffice, refer t	o Internet W	ebsite:			
DLED EXCHANGE	E ACCESS LOOP															+
			$\vdash$									<b>_</b>	<b>_</b>	<b></b>	<u> </u>	
2-WIRE ANA	LOG VOICE GRADE LOOP		لــــا													
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone		1	UEANL	UEAL2	13.99	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone		2	UEANL	UEAL2	17.23	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone		3	UEANL	UEAL2	22.53	31.99	20.02	10.65	1.41			20.35	10.54	13.32	13
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone		1	UEPSR, UEPSB	UEALS	15.92	78.93	50.98	10.65	1.41			20.35	10.54	13.32	13
	2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone	-	2	UEPSR, UEPSB	UEALS	20.79	78.93	50.98	10.65	1.41			20.35	10.54	13.32	13
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zoni		3	UEPSR, UEPSB	UEALS	27.18	78.93	50.98	10.65	1.41		+	20.35	10.54	13.32	13
_	Engineering Information Document (E		J	UEANL	ULALO	41.10	28.8	28.8	10.00	1.41			20.33	10.04	10.02	+ -
-	Engineering information Document (E	$\longrightarrow$	$\vdash$	UEANL	+		20.0	20.0	1			+	+	+	<del> </del>	+-
	Manual Order Coordination for UVL-SL1s (per loop			UEANL	UEAMC		36.46	36.46								
	Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR			UEANL	OCOSL		36.52	36.52								
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling Zone 1	-	1	UEA	UEAL2	16.56	75.06	48.2	28.7	17.64			20.35	10.54	13.32	1:
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling Zone 2	-	2	UEA	UEAL2	21.63	75.06	48.2	28.7	17.64			20.35	10.54	13.32	1:
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling	-	$\neg$	-												
	Zone 3		3	UEA	UEAL2	28.28	75.06	48.2	28.7	17.64			20.35	10.54	13.32	1
	2010 0			OLA	O E / KEE	20.20	70.00	.0.2	20.1	11.01		+	20.00	10.01	10.02	+
	Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL		34.29									
			$\vdash$	UEA	OCOSE		34.29					+	+	+	<del>                                     </del>	+
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zor	ie	ایا	UEA	UEAR2	40.50	75.00	48.2	00.7	47.04			00.05	40.54	40.00	
	1 0 W/- A - I - W O- I - I O - i - I I O - /D D - W - O I 7		ᆲ	UEA	UEAR2	16.56	75.06	48.2	28.7	17.64		+	20.35	10.54	13.32	13
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zor	ie	ا م ا	UEA	UEAR2	21.63	75.06	48.2	28.7	17.64			20.35	10.54	13.32	1:
	2 Wire Apples Veins Conda Lang. Consider Lang 12 m/Devens Better: Circuling. Zen			UEA	UEARZ	21.03	75.00	40.2	20.7	17.04			20.33	10.54	13.32	- 1,
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zo 3	ie	3	UEA	UEAR2	28.28	75.06	48.2	28.7	17.64			20.35	10.54	13.32	1:
4-WIRE ANA	Order Coordination for Specified Conversion Time (per LS) LOG VOICE GRADE LOOP			UEA	OCOSL		34.29									
	4-Wire Analog Voice Grade Loop - Zone		1	UEA	UEAL4	24.7	122.76	85.57	76.35	39.16			20.35	10.54	13.32	13
	4-Wire Analog Voice Grade Loop - Zone		2	UEA	UEAL4	32.25	122.76	85.57	76.35	39.16			20.35	10.54	13.32	1:
	4-Wire Analog Voice Grade Loop - Zone		3	UEA	UEAL4	42.17	122.76	85.57	76.35	39.16		<del></del>	20.35	10.54	13.32	1
	Order Coordination for Specified Conversion Time (per LS			UEA	OCOSL	12.11	34.29	55.57	7 0.00	00.10			20.00	10.01	10.02	T
	N DIGITAL GRADE LOOP			UEA	OCOSE		34.29									
	2-Wire ISDN Digital Grade Loop - Zone		1	UDN	U1L2X	22	142.76	88.88	76.35	39.16		+	20.35	10.54	13.32	1
				UDN								+				
	2-Wire ISDN Digital Grade Loop - Zone		2		U1L2X	29.02	142.76	88.88	76.35	39.16		<del>                                     </del>	20.35	10.54	13.32	1
	2-Wire ISDN Digital Grade Loop - Zone	-	3	UDN	U1L2X	37.95	142.76	88.88	76.35	39.16			20.35	10.54	13.32	1
	Order Coordination For Specified Conversion Time (per LS			UDN	OCOSL		34.29									+
	versal Digital Channel (UDC) COMPATIBLE LOOP		ا										1			
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1		1	UDC	UDC2X	21.15	228.92	152.42	110.01	21.63		<u> </u>	20.35	10.54	13.32	1
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2	-	2	UDC	UDC2X	27.62	228.92	152.42	110.01	21.63			20.35	10.54	13.32	1
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 3		3	UDC	UDC2X	36.12	228.92	152.42	110.01	21.63		1	20.35	10.54	13.32	1
	MMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP					* *						<u> </u>				F
	2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOO											<u> </u>				_
1	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -		ا . ا													1 .
	Zone 1		1	UAL	UAL2X	13.82	270.01	234.63	74.54	39.14			20.35	10.54	13.32	1
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -		, 1					1								1
	Zone 2		2	UAL	UAL2X	18.05	270.01	234.63	74.54	39.14			20.35	10.54	13.32	1
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation -		!			23.6	270.01	234.63	74.54	20.11			20.35	10.54	40.00	1 .
						23.6	1 270.01	234.63	1 7/15/	39.14	ı	1	1 20 35	1054	13.32	1
	Zone 3		3	UAL	UAL2X	23.0		204.00	74.54	00.11			20.00	10.54		T
			3	UAL	OCOSL OCOSL	23.0	34.29	204.00	74.54	00.11			20.55	10.34		

2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -	I	ı									I		1
Zone 2	- 1	2	UAL	UAL2W	18.05	31.99	20.02	10.65	1.41	20.35	10.54	13.32	13
2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3	1	3	UAL	UAL2W	23.6	31.99	20.02	10.65	1.41	20.35	10.54	13.32	13
Order Coordination for Specified Conversion Time (per LS			UAL	OCOSL		34.29							
			OAL	COCOL		04.20							
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP													
2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOO													
2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation Zone 1	1	1	UHL	UHL2X	10.83	270.01	234.63	74.54	39.14	20.35	10.54	13.32	1
2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation	-												
Zone 2  2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation		2	UHL	UHL2X	14.15	270.01	234.63	74.54	39.14	20.35	10.54	13.32	
Zone 3		3	UHL	UHL2X	18.5	270.01	234.63	74.54	39.14	20.35	10.54	13.32	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		34.29							
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation	1												
Zone 1  2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation	1	1	UHL	UHL2W	10.83	31.99	20.02	10.65	1.41	20.35	10.54	13.32	
Zone 2	1	2	UHL	UHL2W	14.15	31.99	20.02	10.65	1.41	20.35	10.54	13.32	
2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3	1	3	UHL	UHL2W	18.5	31.99	20.02	10.65	1.41	20.35	10.54	13.32	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		34.29							
			OTILE	00002		01.20							
4-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP  4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation.	nn -												
Zone 1		1	UHL	UHL4X	13.93	279.6	244.22	74.54	39.14	20.35	10.54	13.32	
4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation Zone 2	on -	2	UHL	UHL4X	18.2	279.6	244.22	74.54	39.14	20.35	10.54	13.32	
4-Wire Unbundled HDSL Loop including manual service inquiry and facility reservation Zone 3	on -	3	UHL	UHL4X	23.8	279.6	244.22	74.54	39.14	20.35	10.54	13.32	
		J			20.0		244.22	74.04	00.14	20.00	10.04	10.02	
Order Coordination for Specified Conversion Time (per LSI  4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation			UHL	OCOSL		34.29							
Zone 1	1	1	UHL	UHL4W	13.93	31.99	20.02	10.65	1.41	20.35	10.54	13.32	
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 2	1	2	UHL	UHL4W	18.2	31.99	20.02	10.65	1.41	20.35	10.54	13.32	
4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation Zone 3	1 -	3	UHL	UHL4W	23.8	31.99	20.02	10.65	1.41	20.35	10.54	13.32	
Order Coordination for Specified Conversion Time (per LS			UHL	OCOSL		34.29							
			OTILE	00002		01.20							
4-WIRE DS1 DIGITAL LOOP  4-Wire DS1 Digital Loop - Zone		1	USL	USLXX	57.73	313.08	219.72	96.86	40.45	18.98	8.43	11.95	
4-Wire DS1 Digital Loop - Zone :		2	USL	USLXX	75.4	313.08	219.72	96.86	40.45	18.98	8.43	11.95	
4-Wire DS1 Digital Loop - Zone		3	USL	USLXX	98.59	313.08	219.72	96.86	40.45	18.98	8.43	11.95	
Order Coordination for Specified Conversion Time (per LS			USL	OCOSL		34.59							
4-WIRE 19.2. 56 OR 64 KBPS DIGITAL GRADE LOOP	1									-			
4 Wire Unbundled Digital 19.2 Kbps	1	1	UDL	UDL19	31.1	207.01	141.38	90.7	+	20.35	10.54	13.32	
4 Wire Unbundled Digital 19.2 Kbps	1	2	UDL	UDL19	40.61	207.01	141.38	90.7	44.18	20.35	10.54	13.32	
4 Wire Unbundled Digital 19.2 Kbps		3	UDL	UDL19	53.11	207.01	141.38	90.7	44.18	20.35	10.54	13.32	
4 Wire Unbundled Digital Loop 56 Kbps - Zone		1	UDL	UDL56	31.1	207.01	141.38	90.7	44.18	20.35	10.54	13.32	
4 Wire Unbundled Digital Loop 56 Kbps - Zone		2	UDL	UDL56	40.61	207.01	141.38	90.7	44.18	20.35	10.54	13.32	
4 Wire Unbundled Digital Loop 56 Kbps - Zone		3	UDL	UDL56	53.11	207.01	141.38	90.7	44.18	20.35	10.54	13.32	
													1
Order Coordination for Specified Conversion Time (per LS	1	١.	UDL	OCOSL	01.1	34.29	444.00	00.7	44.40	00.00	10.51	40.00	
4 Wire Unbundled Digital Loop 64 Kbps - Zone	-	1 2	UDL UDL	UDL64	31.1 40.61	207.01 207.01	141.38 141.38	90.7 90.7	44.18 44.18	20.35 20.35	10.54 10.54	13.32 13.32	
4 Wire Unbundled Digital Loop 64 Kbps - Zone 4 Wire Unbundled Digital Loop 64 Kbps - Zone	-	3	UDL	UDL64 UDL64	53.11	207.01	141.38	90.7	44.18	20.35	10.54	13.32	
	+	3			JJ. I I		191.30	30.1	77.10	20.33	10.04	13.32	
				OCOSL		34.29							
4 Wire Unbundled Digital Loop 64 Kbps - Zone  Order Coordination for Specified Conversion Time (per LS			UDL									i l	1
			UDL										
Order Coordination for Specified Conversion Time (per LS   2-WIRE Unbundled COPPER LOOP  2 Wire Unbundled Copper Loop/Short including manual service inquiry & fac.													
Order Coordination for Specified Conversion Time (per LS)  2-WIRE Unbundled COPPER LOOP	ı	SW	UCL	UCLPB	12.16	131.99	120.02	10.65	1.41	20.35	10.54	13.32	
Order Coordination for Specified Conversion Time (per LS)  2-WIRE Unbundled COPPER LOOP  2 Wire Unbundled Copper Loop/Short including manual service inquiry & fac. reservation - Statewide  Order Coordination for Unbundled Copper Loops (per loc	1	SW			12.16	131.99 36.52	120.02 36.52	10.65	1.41	20.35	10.54	13.32	
Order Coordination for Specified Conversion Time (per LS)  2-WIRE Unbundled COPPER LOOP  2 Wire Unbundled Copper Loop/Short including manual service inquiry & fac. reservation - Statewide	1	sw	UCL	UCLPB	12.16			10.65	1.41	20.35	10.54	13.32	

	2-Wire Unbundled Copper Loop/Long - includes manual svc inquiry and facility		$\overline{}$								$\neg$		i	
	reservation - Statewide	1	sw	UCL	UCL2L	12.16	131.99	120.02	10.65	1.41	20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		36.52	36.52					ı	
	2-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility										1			
	reservation - Statewid Order Coordination for Unbundled Copper Loops (per loc		SW	UCL	UCL2W UCLMC	12.16	31.99 36.52	20.02 36.52	10.65	1.41	20.35	10.54	13.32	13.32
	Order Coordination for Oribundied Copper Loops (per loc		+	UCL	UCLIVIC		30.32	30.32			+			
	OMES Hale all a Constant No. Doctor 17.	<del></del>	1	LIEO	LIEONY	10.10	04.00	00.00	40.05		40.00	40.00	10.00	40.00
	2-Wire Unbundled Copper Loop - Non-Designed Zone 2 Wire Unbundled Copper Loop - Non-Designed - Zone		2	UEQ UEQ	UEQ2X UEQ2X	13.19 17.23	31.99 31.99	20.02	10.65 10.65	1.41	19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
	2 Wire Unbundled Copper Loop - Non-Designed - Zone	i	3	UEQ	UEQ2X	22.53	31.99	20.02	10.65	1.41	19.99	19.99	19.99	19.99
	Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loc			UEQ	USBMC		36.52	36.52						
	Engineering Information Documer			UEQ			28.8	28.8				1		
	Loop Testing - Basic 1st Half Hou Loop Testing - Basic Additional Half Hoι	+	+	UEQ UEQ	URET1 URETA		78.92 23.33	78.92 23.33			+	<del></del>		
	Loop resultg - basic Additional Hall Flot		+	OLQ	UKLIA		25.55	23.33			+			
4	4-WIRE COPPER LOOP	<del></del>	₩											
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Statewide	1 .	sw	UCL	UCL4S	12.16	131.99	120.02	10.65	1.41	20.35	10.54	13.32	13.32
-	Order Coordination for Unbundled Copper Loops (per loc		SW	UCL	UCLMC	12.10	36.52	36.52	10.05	1.41	20.33	10.54	13.32	13.32
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -	ī .	1	002	2 320		00.02	00.02			+			
	Statewide	1	sw	UCL	UCL4W	12.16	31.99	20.02	10.65	1.41	20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loc	<del></del>	₩	UCL	UCLMC		36.52	36.52			 +	<b></b>		
	4-Wire Unbundled Copper Loop/Long - includes manual svc inquiry and facility reservation - Statewidi	1 .	sw	UCL	UCL4L	12.15	131 99	120.02	10.65	1 41	20.35	10.54	13.32	13.32
-	Order Coordination for Unbundled Copper Loops (per loc		SW	UCL	UCLMC	12.15	36.52	36.52	10.05	1.41	20.35	10.54	13.32	13.32
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility		<b>†</b>	002	2 320		00.02	00.02			+			
	reservation - Statewide		SW	UCL	UCL40	12.16	31.99	20.02	10.65	1.41	20.35	10.54	13.32	13.32
	Order Coordination for Unbundled Copper Loops (per loc			UCL	UCLMC		36.52	36.52				1		
			+		+-+						+			
OOP MODIF	FICATION		+								+			
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal	to		UAL, UHL, UCL,							_			
	18k ft			UEQ, ULS	ULM2L		65.4	65.4						
-	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18	<u> </u>	+	UCL, ULS	ULM2G		710.71	23.77			+	<del></del>		
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18	i^		UHL, UCL	ULM4L		65.4	65.4					ı	
	it is a second of the second o		_	OTIE, OOE	OLIVIAL		00.4	00.4			+			
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18			UCL	ULM4G		710.71	23.77						
		Ι.		UAL, UHL, UCL,									ı	
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled I	$\vdash$	+	UEQ, UEF, ULS	ULMBT		65.44	65.44			+			
SUB-LOOPS			+								+			
			1								_			
5	Sub-Loop Distribution													
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-L Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-L	1 1			USBSA		517.25	517.25						13.32
				UEANL			10.00				20.35	10.54	13.32	10.02
1	Sub-Loop - Fel Closs Box Editation - Fel 25 Fall Fallet Set-C		-	UEANL	USBSB		42.68	42.68			20.35 20.35	10.54 10.54	13.32	13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I	I	$\vdash$				42.68 313.01	42.68 313.01			20.35 20.35 20.35		13.32 13.32 13.32	13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-l	I	<del>                                     </del>	UEANL	USBSB		42.68 313.01	313.01			20.35	10.54	13.32	13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I	1		UEANL UEANL UEANL	USBSB USBSC USBSD		42.68 313.01 108.06	313.01 108.06			20.35 20.35 20.35	10.54 10.54	13.32 13.32	13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-l Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-l Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi		SW	UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2	10.02	42.68 313.01 108.06 148.84	313.01 108.06 112.34	73.14	36.65	20.35	10.54	13.32	13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair	1	sw 1	UEANL UEANL UEANL	USBSB USBSC USBSD		42.68 313.01 108.06	313.01 108.06	73.14	36.65	20.35 20.35 20.35 20.35	10.54 10.54	13.32 13.32	13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-l Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-l Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone	1	1 2	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL	USBSB USBSC USBSD USBN2 USBMC USBMC USBN4 USBN4	7.3 9.54	42.68 313.01 108.06 148.84 34.29 147.93	313.01 108.06 112.34 34.29 75.11 75.11		16.98 16.98	20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone	1	1	UEANL	USBSB USBSC USBSD USBN2 USBMC USBN4 USBN4 USBN4	7.3	42.68 313.01 108.06 148.84 34.29 147.93 147.93	313.01 108.06 112.34 34.29 75.11 75.11	99.96	16.98	20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair	1	1 2	UEANL	USBSB  USBSC  USBSD  USBN2  USBMC  USBN4  USBN4  USBN4  USBN4  USBMC	7.3 9.54 12.47	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29	99.96 99.96 99.96	16.98 16.98 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INK	1	1 2	UEANL	USBSB  USBSC  USBSD  USBN2  USBMC  USBN4  USBN4  USBN4  USBN4  USBN6  USBR2	7.3 9.54	42.68 313.01 108.06 148.84 34.29 147.93 147.93 147.93 34.29 94.56	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35	99.96 99.96	16.98 16.98	20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INC Order Coordination for Unbundled Sub-Loops, per sub-loop pair	1	1 2	UEANL	USBSB  USBSC  USBSD  USBN2  USBMC  USBN4  USBN4  USBN4  USBN4  USBN6  USBN6  USBN6  USBN6  USBN6  USBN6  USBN6  USBN6  USBN6	7.3 9.54 12.47 1.35	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29	99.96 99.96 99.96 94.41	16.98 16.98 16.98 13.09	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INK	1	1 2	UEANL	USBSB  USBSC  USBSD  USBN2  USBN4  USBN4  USBN4  USBN4  USBN4  USBN6  USBR2  USBRC  USBRC  USBRC  USBRC  USBRC	7.3 9.54 12.47	42.68 313.01 108.06 148.84 34.29 147.93 147.93 147.93 34.29 94.56	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35	99.96 99.96 99.96	16.98 16.98 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INC Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INC Order Coordination for Unbundled Sub-Loops, per sub-loop pair		1 2 3	UEANL UEANL UEANL UEANL	USBSB  USBSC  USBSD  USBN2  USBN4  USBN4  USBN4  USBN4  USBN6  USBN6  USBR6  USBR7  USBR6  USBR6  USBR7  USBR7  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8	7.3 9.54 12.47 1.35 2.26 5.16	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29 37.1 34.29 37.1 34.29 37.89	99.96 99.96 99.96 94.41 99.96	16.98 16.98 16.98 13.09 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 4-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone	I I	1 2 3	UEANL UEANL UEANL UEANL	USBSB  USBSC  USBSD  USBN2  USBMC  USBN4  USBN4  USBN4  USBR0  USBR1  USBR2  USBR6  USBR6  USBR6  USBR7  USBR7  USBR7  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8  USBR8	7.3 9.54 12.47 1.35 2.26 5.16 6.74	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71	313.01 108.06 112.34 34.29 75.11 75.11 34.29 29.35 34.29 37.1 34.29 37.1 34.29 37.1	99.96 99.96 99.96 94.41 99.96 94.41 94.41	16.98 16.98 16.98 13.09 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone		1 2 3	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEF	USBSB  USBSC  USBSC  USBSC  USBN2 USBMC USBN4 USBN4 USBN4 USBN4 USBRC	7.3 9.54 12.47 1.35 2.26 5.16	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 37.1 34.29 37.89 37.89 37.89	99.96 99.96 99.96 94.41 99.96	16.98 16.98 16.98 13.09 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 4-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair	I I	1 2 3	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEF	USBSB  USBSC  USBSC  USBN2  USBMC  USBN4  USBN4  USBN4  USBN6  USBR6  USBR7  US	7.3 9.54 12.47 1.35 2.26 5.16 6.74 8.81	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71 110.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29 37.1 34.29 37.89 37.89 37.89 37.89 34.29	99.96 99.96 99.96 94.41 99.96 94.41 94.41 94.41	16.98 16.98 16.98 13.09 13.09 13.09 13.09 13.09	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INK Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone	I I	1 2 3	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEF	USBSB  USBSC  USBSC  USBSC  USBN2 USBMC USBN4 USBN4 USBN4 USBN4 USBRC	7.3 9.54 12.47 1.35 2.26 5.16 6.74	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 37.1 34.29 37.89 37.89 37.89	99.96 99.96 99.96 94.41 99.96 94.41 94.41	16.98 16.98 16.98 13.09 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INC  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 4-Wire Intrabuilding Network Cable (INC  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 1 1	1 2 3 1 1 2 3 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 1	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEF UEF	USBSB  USBSC  USBSC  USBN2  USBN4  USBN4  USBN4  USBN4  USBN4  USBN4  USBN5  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR4  USBMC  USBR4  USBMC  UCS2X  UCS2X  UCS2X  UCS2X  UCS4X  UCS4X  UCS4X	7.3 9.54 12.47 1.35 2.26 5.16 6.74 8.81	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71 110.71 117.71 117.71 117.71 117.71 117.71 117.71 117.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29 37.89 37.89 37.89 37.89 34.29 44.3 44.3	99.96 99.96 99.96 94.41 99.96 94.41 94.41 94.41	16.98 16.98 16.98 13.09 13.09 13.09 13.09 13.09 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 2-Wire Intrabuilding Network Cable (INC Order Coordination for Unbundled Sub-Loops, per sub-loop pair Sub-Loop 4-Wire Intrabuilding Network Cable (INC Order Coordination for Unbundled Sub-Loops, per sub-loop pair 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 Wire Copper Unbundled Sub-Loop Distribution - Zone Order Coordination for Unbundled Sub-Loop pair Sub-loop pair 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 1 1	1 2 3 1 1 2 3 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 1	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEF UEF	USBSB USBSC USBSC USBSC USBMC USBMC USBMC USBMM USBMM USBMM USBMS USBMC USBMC USBMC USBMC USBMC USBMC USBMC USBMC USBMC USBMC USBMC UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX UCSEX	7.3 9.54 12.47 1.35 2.26 5.16 6.74 8.81 6.52 8.52	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71 110.71 34.29 117.71 110.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29 37.1 34.29 37.89 37.89 37.89 34.29 44.3	99.96 99.96 99.96 94.41 99.96 94.41 94.41 99.96 99.96	16.98 16.98 16.98 13.09 16.98 13.09 13.09 13.09 13.09 16.98 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 4-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 1 1	1 2 3 1 1 2 3 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 1	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEF UEF	USBSB  USBSC  USBSC  USBN2  USBN4  USBN4  USBN4  USBN4  USBN4  USBN4  USBN5  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR4  USBMC  USBR4  USBMC  UCS2X  UCS2X  UCS2X  UCS2X  UCS4X  UCS4X  UCS4X	7.3 9.54 12.47 1.35 2.26 5.16 6.74 8.81 6.52 8.52	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71 110.71 117.71 117.71 117.71 117.71 117.71 117.71 117.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29 37.89 37.89 37.89 37.89 34.29 44.3 44.3	99.96 99.96 99.96 94.41 99.96 94.41 94.41 99.96 99.96	16.98 16.98 16.98 13.09 16.98 13.09 13.09 13.09 13.09 16.98 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
\$	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INC  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 4-Wire Intrabuilding Network Cable (INC  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 1 1	1 2 3 1 1 2 3 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 1	UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEANL UEF UEF	USBSB  USBSC  USBSC  USBN2  USBN4  USBN4  USBN4  USBN4  USBN4  USBN4  USBN5  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR4  USBMC  USBR4  USBMC  UCS2X  UCS2X  UCS2X  UCS2X  UCS4X  UCS4X  UCS4X	7.3 9.54 12.47 1.35 2.26 5.16 6.74 8.81 6.52 8.52	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71 110.71 117.71 117.71 117.71 117.71 117.71 117.71 117.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29 37.89 37.89 37.89 37.89 34.29 44.3 44.3	99.96 99.96 99.96 94.41 99.96 94.41 94.41 99.96 99.96	16.98 16.98 16.98 13.09 16.98 13.09 13.09 13.09 13.09 16.98 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32
\$	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-I  Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-I  Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Statewi  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 2-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  Sub-Loop 4-Wire Intrabuilding Network Cable (INK  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  2 Wire Copper Unbundled Sub-Loop Distribution - Zone  Order Coordination for Unbundled Sub-Loops, per sub-loop pair  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone  4 Wire Copper Unbundled Sub-Loop Distribution - Zone	1 1 1	1 2 3 3 1 2 3 3 1 2 3 3 1 2 3 3 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 3 3 1 1 1 1	UEANL UEF UEF	USBSB  USBSC  USBSC  USBN2  USBN4  USBN4  USBN4  USBN4  USBN4  USBN4  USBN5  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR2  USBMC  USBR4  USBMC  USBR4  USBMC  UCS2X  UCS2X  UCS2X  UCS2X  UCS4X  UCS4X  UCS4X	7.3 9.54 12.47 1.35 2.26 5.16 6.74 8.81 6.52 8.52	42.68 313.01 108.06 148.84 34.29 147.93 147.93 34.29 94.56 34.29 116.14 34.29 110.71 110.71 110.71 117.71 117.71 117.71 117.71 117.71 117.71 117.71	313.01 108.06 112.34 34.29 75.11 75.11 75.11 34.29 29.35 34.29 37.89 37.89 37.89 37.89 34.29 44.3 44.3	99.96 99.96 99.96 94.41 99.96 94.41 94.41 99.96 99.96	16.98 16.98 16.98 13.09 16.98 13.09 13.09 13.09 13.09 16.98 16.98	20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35 20.35	10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54 10.54	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32	13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32 13.32

Attachment 2 Exhibit C

		UEA,											
HOLE THE POOL OF T		UDN,UCL,U			40.00	42.68							
USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-u USL Feeder DS1 Set-up at DSX location, per DS1 termination		C	USBFX		42.68 531.04	42.68 11.34							
Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade- Statew	S		USBFA	12.05	122.24	85.05	76.35	39.16	19.99	20.35	10.54	13.32	13.32
Order Coordination for Specified Conversion Time, per LS	3	UEA	OCOSL	12.00	34.29	03.03	70.55	39.10	13.33	20.33	10.54	13.32	13.32
Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Statew	S		USBFB	12.05	122.24	85.05	76.35	39.16		20.35	10.54	13.32	13.32
Order Coordination for Specified Time Conversion, per LS		UEA	OCOSL		34.29								
Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade Loop -													
Statewide	S	w UEA	USBFC	12.05	122.24	85.05	76.35	39.16		20.35	10.54	13.32	13.32
												'	
Order Coordination For Specified Conversion Time, per LS		UEA	OCOSL		34.29								
Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zoni	1		USBFD		137.31	61.93	118.04	30.13		20.35	10.54	13.32	13.32
Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zon	2		USBFD		137.31	61.93	118.04	30.13		20.35	10.54	13.32	13.32
Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zoni	3	B UEA	USBFD	36.76	137.31	61.93	118.04	30.13		20.35	10.54	13.32	13.32
Out of Control Fundamental Control Con			00001		04.00							'	
Order Coordination For Specified Conversion Time, Per LS		UEA	OCOSL	04.50	34.29 137.31	04.00	118.04	00.40		20.35	10.54	10.00	40.00
Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	2		USBFE USBFE	21.52 28.11	137.31	61.93 61.93	118.04	30.13 30.13		20.35	10.54	13.32 13.32	13.32 13.32
Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zoni	3		USBFE		137.31	61.93	118.04	30.13		20.35	10.54	13.32	13.32
Onburidied Sub-Loop Feeder Loop, 4 Wife Loop-Start, Voice Grade - Zoni	<del> '</del>	DEA	USBFE	30.76	137.31	01.93	110.04	30.13		20.55	10.54	13.32	13.32
Order Coordination For Specified Conversion Time, Per LS		UEA	OCOSL		34.29							1 '	
Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone	1		USBFF	16.11	142.83	67.45	104.67	18.53		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	1 2		USBFF	21.04	142.83	67.45	104.67	18.53		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone	3		USBFF	27.51	142.83	67.45	104.64	18.53		19.99	19.99	19.99	19.99
The state of the s		3314	00011		2.00	27.10							. 3.00
Order Coordination For Specified Conversion Time, Per LS		UDN	OCOSL		34.29							1	
Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	1		USBFS	16.11	142.83	67.45	104.67	18.53		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	2		USBFS	21.04	142.83	67.45	104.67	18.53		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatibl	3	UDC UDC	USBFS	27.51	142.83	67.45	104.64	18.53		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	1	USL	USBFG		116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	2		USBFG		116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone	3	USL USL	USBFG	67.86	116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
												1 '	
Order Coordination For Specified Conversion Time, Per LS		USL	OCOSL		34.29								
Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone	1		USBFH		114.27	38.89	104.64	18.53		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	1 2		USBFH		114.27	38.89	104.64	18.53		19.99	19.99	19.99	19.99
Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	3	B UCL	USBFH	16.26	114.27	38.89	104.64	18.53		19.99	19.99	19.99	19.99
0.1.00			00001		04.00							'	
Order Coordination For Specified Conversion Time, per LS Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	1	UCL	OCOSL USBFJ	14.37	34.29 123.41	48.03	110.44	22.53		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	2		USBFJ	18.76	123.41	48.03	110.44	22.53		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	3		USBFJ	24.53	123.41	48.03	110.44	22.53		19.99	19.99	19.99	19.99
out 2007 recueir 1 of 4 Wile Copper 2006	<del></del>	, 002	OODI 0	24.00	120.41	40.00	110.44	22.00		10.00	10.00	10.00	10.00
Order Coordination For Specified Conversion Time, per LS		UCL	OCOSL		34.29							1 '	
Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	1		USBFN	26.06	116	40.62	106.82	18.91	19.99	19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loc	2	2 UDL	USBFN	34.03	116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loα	3		USBFN	44.5	116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	1		USBFO		116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	2		USBFO		116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zon∈	3	UDL.	USBFO	44.5	116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
												1 '	
Order Coordination For Specified Time Conversion, per LS		UDL	OCOSL		34.29								
Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1		USBFP		116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1 2		USBFP	34.03	116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	3	B UDL	USBFP	44.5	116	40.62	106.82	18.91		19.99	19.99	19.99	19.99
0.1.00		LIDI	OCOSL		34.29							'	
Order Coordination For Specified Conversion Time, per LS		UDL	UCUSL		34.29								
Unbundled Sub-Loop Modification													
Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-	M												
PR	) v	UEF	ULM2X		335.35	7.82				20.34	10.54	13.32	13.32
Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-	\h/	OLI	OLIVIZA		333.33	7.02				20.54	10.54	10.02	13.32
PR	1	UEF	ULM4X		335.36	7.82				20.35	10.54	13.32	13.32
Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per P	<del> </del>	UEF	OLIVI4A	1	555.50	1.02	1			20.00	10.54	10.02	10.02
unloaded		UEF	ULM4T		528.48	9.74				20.35	10.54	13.32	13.32
	+	021	0241		220.10		1						. 5.02
Unbundled Network Terminating Wire (UNTW)													
Unbundled Network Terminating Wire (UNTW) per Pa	1	UENT\	V UENPP	0.45	2.48	2.48				20.35	10.54	13.32	13.32
Network Interface Device (NID)													
Network Interface Device (NID) - 1-2 line		UENT\			89.69	54.56				20.35	10.54	13.32	13.32
Network Interface Device (NID) - 1-6 line		UENT\	V UND16		129.65	94.51				20.35	10.54	13.32	13.32
											40.54	13.32	13.32
Network Interface Device Cross Connect - 2 V		UENT\			0.74	0.74				20.35	10.54		
					0.74 0.74	0.74 0.74				20.35	10.54	13.32	13.32
Network Interface Device Cross Connect - 2 V Network Interface Device Cross Connect - 4V		UENT\											
Network Interface Device Cross Connect - 2 V		UENT\		307.07			4.18						

	CO Channel Interface - 2-Wire Voice Grade		ULC	ULCC2	1.2	9.57	9.52	8.66	8.6		20.35	10.54	13.32	13
	Unbundled Loop Concentration - System A (TR00)		ULC	UCT8A	500.18	613.6	613.6	0.00	0.0		20.35	10.54	13.32	13
	Unbundled Loop Concentration - System B (TR00)		ULC	UCT8B	54.82	255.67	255.67				20.35	10.54	13.32	13
	Unbundled Loop Concentration - System A (TR30):		ULC	UCT3A	539	613.6	613.6				20.35	10.54	13.32	13
	Unbundled Loop Concentration - System A (1730):  Unbundled Loop Concentration - System B (TR30):		ULC	UCT3B	92.37	255.67	255.67				20.35	10.54	13.32	13
	Unbundied Edop Concentration Cystem B (11000)		OLO	OOTOD	32.01	200.01	200.07				20.00	10.04	10.02	- 10
	Unbundled Loop Concentration - DS1 Loop Interface Ca		ULC	UCTCO	6.23	74.39	53.07	30.23	8.46		20.35	10.54	13.32	13
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Car		UDN	ULCC1	8.46	8.69	8.65	9.71	9.65		20.35	10.54	13.32	13
	Unbundled Loop Concentration - UDC Loop Interface (Brite Car		UDC	ULCCU	8.46	8.69	8.65	9.71	9.65		20.35	10.54	13.32	1:
	Unbundled Loop Concentration - 2 Wire Voice-Loop Start or Ground Start Loop		ODC	ULCCU	0.40	0.03	0.03	3.71	3.03		20.55	10.54	13.32	- 1
	Interface (POTS Card)		UEA	ULCC2	2.32	8.69	8.65	9.71	9.65		20.35	10.54	13.32	1
_	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface		OLA	ULCCZ	2.02	0.03	0.00	3.71	3.03		20.55	10.54	13.32	
	(SPOTS Card)		UEA	ULCCR	12.45	8.69	8.65	9.71	9.65		20.35	10.54	13.32	1
_	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Ca		UEA	ULCC4	7.53	8.69	8.65	9.71	9.65		20.35	10.54	13.32	13
	Unbundled Loop Concentration - 14 Wife Voice Loop Interface (Specials Ca		ULC	UCTTC	35.77	8.69	8.65	9.71	9.65		20.35	10.54	13.32	1
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfa		UDL	ULCC7	11.03	8.069	8.65	9.71	9.65		20.35	10.54	13.32	1
	Unbundled Loop Concentration - Digital 16.2 Kbps Data Loop Interfa		UDL	ULCC5	11.03	8.69	8.65	9.71	9.65		20.35	10.54	13.32	1
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfa		UDL	ULCC6	11.03	8.69	8.65	9.71	9.65		20.35	10.54	13.32	1
	Onbundled Loop Concentration - Digital 64 Rbps Data Loop Interna		ODL	OLCCO	11.03	0.05	0.00	3.71	9.00		20.33	10.54	13.32	- 1
OLED SUB-LOC	DP CONCENTRATION (OUTSIDE CO)													
DEED 30B-E00	CONCENTRATION (COTSIDE CO)													
HER. PROVISION	ONING ONLY - NO RATE													
			UENTW	UNDBX										
	NID - Dispatch and Service Order for NID installation									+				
	UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW UEANL,UEF,UEQ	UENCE				-		+				
	Unbundled Contract Name, Provisioning Only - No Rate		UENTW	UNECN										
_	Oribundled Contract Name, Frovisioning Only - No Nate		UAL,UCL,UDC,UD											
			,UDN,UEA,UHL,U											
	Habitadlad Castast Nama Benjisiasias Calif		C C		0	0								
	Unbundled Contact Name, Provisioning Only - no rate		U	UNECN	U	0								
			UEA,UDN,UCL,UI	,										
	Unbundled Sub Loop Fooder 2 Wire Cross Barr Israel		UEA,UDN,UCL,UI	USBFQ	0	0					1			1
-	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no ra		U	USBLQ	U	U		1			1			
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no ra		UEA,USL,UCL,UD	USBER	0	0					1			1
	Unbundled DS1 Loop - Superframe Format Option - no ra		USL	CCOSF	0	0								
	Unbundled DS1 Loop - Expanded Superframe Format option - no re		USL	CCOEF	0	0								
APACITY UNBU	INDLED LOCAL LOOP													
NOTE: 4 m	nonth minimum billing period													
	High Capacity Unbundled Local Loop - DS3 - Per Mile per mon		UE3	1L5ND	9.19									
	High Capacity Unbundled Local Loop - DS3 - Facility Termination per mor		UE3	UE3PX	374.24	595.67	304.5	234.83	170.16		36.84	36.84	19.01	1
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per mon		UDLSX	1L5ND	9.19									
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per mor		UDLSX	UDLS1	389.35	595.37	304.5	215.82	151.15		36.84	36.84	19.01	1
MAKE-UP														
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried													
	(Manual).	1	UMK	UMKLW		100	100	<u></u>			<u> </u>			L
	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).	1	UMK	UMKLP		100	100							
			OWIN					1						
	Loop MakeupWith or Without Reservation, per working or spare facility queried									1				
	Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)	1	UMK	PSUMK		0.6888	0.6888							
ARING		ı		PSUMK		0.6888	0.6888							
ARING	(Mechanized)	1	UMK		400			450						
ARING	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci	1	UMK	ULSDA	100	150	0	150	0	0				
ARING	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci	1	UMK ULS ULS	ULSDA ULSDB	25	150 150	0 0	150	0	0				
ARING	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci	1	UMK  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8	25 8.33	150 150 150	0 0 0	150 150	0		20.25	10.54	12.22	
ARING	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	UMK  ULS  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8 ULSDC	25	150 150 150 40	0 0 0 0 21.39	150	0	0	20.35	10.54	13.32	1
ARING	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	UMK  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8	25 8.33	150 150 150	0 0 0	150 150	0	0	20.35 20.35	10.54 10.54	13.32	1
	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme		UMK  ULS  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8 ULSDC	25 8.33	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1
DLED TRANSPO	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme		UMK  ULS  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8 ULSDC	25 8.33	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1
DLED TRANSPO	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  ORT  TRANSPORT (Shared)	1 1 1 1 1 1 1	UMK  ULS  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8 ULSDC	25 8.33 0.61	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1
DLED TRANSPO	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per System - Activatio Line Sharing - per System - Activatio  Line Sharing - per Subsequent Activity per Line Rearrangeme  ORT  TRANSPORT (Shared) Common Transport - Per Mile, Per MOL		UMK  ULS  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8 ULSDC	25 8.33 0.61	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1
DLED TRANSPO	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  ORT  TRANSPORT (Shared)	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	UMK  ULS  ULS  ULS  ULS  ULS	ULSDA ULSDB ULSD8 ULSDC	25 8.33 0.61	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1
DLED TRANSPO COMMON	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitter, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  ORT  TRANSPORT (Shared) Common Transport - Per Mile, Per MOL Common Transport - Facilities Termination Per MO		ULS ULS ULS ULS ULS	ULSDA ULSDB ULSDB ULSDC ULSDS	25 8.33 0.61	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1
DLED TRANSPO COMMON :	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitte, Per System 24 Line Capaci Line Sharing Splitte, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  ORT  TRANSPORT (Shared) Common Transport - Per Mile, Per MOL Common Transport - Facilities Termination Per MO  TEROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS3 =		ULS ULS ULS ULS ULS	ULSDA ULSDB ULSDB ULSDC ULSDS	25 8.33 0.61	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1
DLED TRANSPO COMMON :	(Mechanized)  Line Sharing Splitter, per System 96 Line Capaci Line Sharing Splitter, per System 24 Line Capaci Line Sharing Splitter, Per System, 8 Line Capaci Line Sharing - per Line Activatio Line Sharing - per Line Activatio Line Sharing - per Subsequent Activity per Line Rearrangeme  ORT  TRANSPORT (Shared) Common Transport - Per Mile, Per MOL Common Transport - Facilities Termination Per MO		ULS ULS ULS ULS ULS	ULSDA ULSDB ULSDB ULSDC ULSDS	25 8.33 0.61	150 150 150 40	0 0 0 0 21.39	150 150	0	0			13.32	1

			1						,				
1	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination		U1TVX	U1TV2	40.50	55.00	47.07	27.96	0.54	00.00	04.00	9.8	40.54
-	per month Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile p		UTIVX	U11V2	18.58	55.39	17.37	27.96	3.51	20.35	21.09	9.8	10.54
	month	ei	U1TVX	1L5XX	0.0174								
-	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination		OTIVA	TESAA	0.0174								
	per month	'	U1TVX	U1TR2	18.58	55.39	17.37	27.96	3.51	20.35	21.09	9.8	10.54
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month		U1TVX	1L5XX	0.0054								
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination	n	11477.07	1147774	04.00	07.07	00.00	00.70	40.07	45.00	45.00	0.00	0.00
	per month		U1TVX	U1TV4	24.09	37.87	26.02	30.78	13.07	15.08	3 15.08	8.66	8.66
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per mon		U1TDX	1L5XX	0.0174								
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per mor		U1TDX	U1TD5	17.98	55.39	17.37	27.96	3.51	20.35	21.09	9.8	10.54
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per mor		U1TDX	1L5XX	0.0174	00.00	17.07	21.50	0.01	20.00	21.03	3.0	10.04
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per mor		U1TDX	U1TD6	17.98	55.39	17.37	27.96	3.51	20.35	21.09	9.8	10.54
	INTEROFFICE CHANNEL - DEDICATED TRANSPORT - DS1												
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per mor		U1TD1	1L5XX	0.3525								
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per mor		U1TD1	U1TF1	77.86	112.4	76.27	19.55	14.99	20.3	21.09	9.8	10.54
	INTERPRETION CHANNEL DEDICATED TRANSPORT DOS												
	INTEROFFICE CHANNEL - DEDICATED TRANSPORT- DS3		U1TD3	1L5XX	2.34								
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per mor Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per mor		U1TD3	U1TF3	848.99	395.29	176.56	109.04	105.91	36.84	36.84	19.01	19.01
-	interoffice Charmer - Dedicated Transport - DSS - Lacinty Termination per mor		01103	01113	040.55	393.29	170.50	103.04	100.01	30.0	30.04	13.01	13.01
	INTEROFFICE CHANNEL - DEDICATED TRANSPORT- STS-1		1										
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per mon		U1TS1	1L5XX	2.34								1
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per mor		U1TS1	U1TFS	849.3	395.29	176.56	109.04	105.91	36.84	36.84	19.01	19.01
	LOCAL CHANNEL - DEDICATED TRANSPORT												
	NOTE: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one m	nonth, DS3 ar		hs									
	Local Channel - Dedicated - 2-Wire Voice Grade Per Month		ULCVX	ULDV2	19.43	199.33	24.16	54.81	4.8	20.3		13.32	13.32
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per mor		ULCVX	ULDR2	19.43	199.33	24.16	54.81	4.8	20.3		9.8	10.54
	Local Channel - Dedicated - 4-Wire Voice Grade per mon		UNCVX	ULDV4	20.56	201.53	24.83	55.52	5.51	20.35		13.32	13.32
	Local Channel - Dedicated - DS1 per mont		ULDD1	ULDF1	40.99	277.35	233.26	33.18	22.3	45.68	1.76	21.75	1.76
	Local Channel - Dedicated - DS3 - Per Mile per mon  Local Channel - Dedicated - DS3 - Facility Termination per mon		ULDD3 ULDD3	1L5NC ULDF3	7.15 611.3	595.37	304.5	215.82	151.15	36.84	36.84	19.01	19.01
	Local Channel - Dedicated - STS-1- Per Mile per mon		ULDS1	1L5NC	7.15	000.01	304.5	210.02	101.10	50.0-	7 30.04	13.01	13.01
	Local Channel - Dedicated - STS-1 - Facility Termination per mon		ULDS1	ULDFS	599.59	588.07	297.2	215.82	151.15	20.35	21.09	9.8	10.54
MULTIPLE													
	Channelization - DS1 to DS0 Channel System		UXTD1	MQ1	80.77	141.67	77.11	44.47	42.62	20.3	9.8	11.49	1.18
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)		UDL	1D1DD	1.82	6.07	4.66						
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per mont		UDN UEA	UC1CA 1D1VG	3.1 0.91	6.07 6.07	4.66 4.66						
<b>_</b>	Voice Grade COCI - DS1 to DS0 Channel System - per mon DS3 to DS1 Channel System per mont		UXTD3	MQ3	222.98	308.03	108.47	6.34	4.23	20.35	9.8	11.49	1.18
	STS1 to DS1 Channel System per mont		UXTS1	MQ3	222.98	308.03	108.47	6.34	4.23	20.35		9.8	9.8
	DS3 Interface Unit (DS1 COCI) used with Loop per monti		USL	UC1D1	17.58	6.07	4.66	0.04	4.20	20.00	21.00	3.0	3.0
DARK FIBI	ER												
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local												
	Channe		UDF	1L5DC	53.23								
L	NRC Dark Fiber - Local Channe		UDF	UDFC4		1219.22	169.75	453.22	339.34	20.3	21.09	9.8	10.54
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -		UDF	1L5DF	50.00								
	Interoffice Channe  NRC Dark Fiber - Interoffice Channe		UDF	UDF14	53.23	1219.22	169.75	453.22	339.34	20.35	21.09	9.8	10.54
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local		UDF	UDF 14		1219.22	109.75	403.22	339.34	20.3	21.09	9.0	10.34
	Dark i iber, i our i iber Stranus, i er route wile or i raction i mereor per monti i - Locar												
1	Loop		UDF	11.5DI	53 23								
	Loop NRC Dark Fiber - Local Loop		UDF UDF	1L5DL UDFL4	53.23	1219.22	169.75	453.22	339.34	20.35	21.09	9.8	10.54
TRANSPO	Loop NRC Dark Fiber - Local Loop RT OTHER		UDF UDF	1L5DL UDFL4	53.23	1219.22	169.75	453.22	339.34	20.38	21.09	9.8	10.54
TRANSPO	NRC Dark Fiber - Local Loop		UDF UDF		53.23	1219.22	169.75	453.22	339.34	20.35	5 21.09	9.8	10.54
TRANSPO	NRC Dark Fiber - Local Loop		UDF UDF		53.23	1219.22	169.75	453.22	339.34	20.39	5 21.09	9.8	10.54
TRANSPO	NRC Dark Fiber - Local Loop		UDF UDF		53.23	1219.22	169.75	453.22	339.34	20.35	5 21.09	9.8	10.54
TRANSPO	NRC Dark Fiber - Local Loop  RT OTHER		UDF	UDFL4	53.23								
TRANSPO	NRC Dark Fiber - Local Loog  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chans		UDF UNC1X	UDFL4		185.16	23.85	2.03	0.79	20.38	5 21.09	9.8	10.54
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chans Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chans		UDF	UDFL4							5 21.09		
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chans Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chans ESS TEN DIGIT SCREENING		UNC1X UNC1X	UDFL4		185.16	23.85	2.03	0.79	20.38	5 21.09	9.8	10.54
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chans Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chans Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chans CREENING BXX Access Ten Digit Screening, Per Ca		UDF UNC1X	CCOEF CCOSF		185.16	23.85	2.03	0.79	20.38	5 21.09 5 21.09	9.8	10.54
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chanz Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chanz Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chanz ESS TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca 8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio		UNC1X UNC1X OHD	CCOEF CCOSF		185.16 185.16 5.21 11.47	23.85 23.85	2.03	0.79	20.34	5 21.09 5 21.09 5 20.35	9.8	10.54 10.54
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chanz Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chanz Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chanz ESS TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca 8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio		UNC1X UNC1X UNC1X OHD OHD	CCOEF CCOSF		185.16 185.16 5.21	23.85 23.85 0.76	2.03 2.03	0.79	20.3(20.3)	5 21.09 5 21.09 6 20.35 5 20.35	9.8 9.8 13.28	10.54 10.54 13.28
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant ESS TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb		UNC1X UNC1X UNC1X OHD OHD OHD	CCOEF CCOSF		185.16 185.16 5.21 11.47	23.85 23.85 0.76 1.46	2.03 2.03 7.34	0.79 0.79	20.3 20.3: 20.3:	5 21.09 5 21.09 5 20.35 5 20.35 5 20.35	9.8 9.8 13.28 13.28	10.54 10.54 13.28 13.28
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chans Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chans Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chans ESS TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca 8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserv 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb 8XX Access Ten Digit Screening, Mattiple InterLATA CXR Routing Per CXR Requester	3	UNC1X UNC1X UNC1X OHD OHD OHD OHD	CCOEF CCOSF N8R1X N8FTX N8FCX		185.16 185.16 5.21 11.47 11.47 4.47	23.85 23.85 0.76 1.46 1.46	2.03 2.03 7.34	0.79 0.79	20.3 20.3 20.3 20.3 20.3 20.3 20.3	5 21.09 5 21.09 5 20.35 5 20.35 6 20.35 5 20.35	9.8 9.8 13.28 13.28 13.28 13.28	10.54 10.54 13.28 13.28 13.28
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant ESS TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requeste- Per 8XX No.	1	UNC1X UNC1X UNC1X OHD OHD OHD OHD OHD	CCOEF CCOSF N8R1X N8FTX N8FCX N8FMX		185.16 185.16 5.21 11.47 11.47 4.47 5.23	23.85 23.85 0.76 1.46 1.46 2.24	2.03 2.03 7.34	0.79 0.79	20.3i 20.3i 20.3i 20.3i 20.3i 20.3i 20.3i 20.3i	5 21.09 5 21.09 5 20.35 5 20.35 5 20.35 5 20.35 5 20.35	9.8 9.8 13.28 13.28 13.28 13.28	10.54 10.54 13.28 13.28 13.28 13.28
	NRC Dark Fiber - Local Loop  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant ESS TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation 8XX Access Ten Digit Screening, Costomized Area of Service Per 8XX No. 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requester Per 8XX No. 8XX Access Ten Digit Screening, Change Charge Per Reque	3	UNC1X UNC1X UNC1X OHD OHD OHD OHD OHD OHD OHD	CCOEF CCOSF N8R1X N8FTX N8FCX N8FMX N8FAX		185.16 185.16 5.21 11.47 11.47 4.47 5.23 5.97	23.85 23.85 0.76 1.46 1.46	2.03 2.03 7.34	0.79 0.79	20.34 20.34 20.33 20.33 20.33 20.33 20.33	5 21.09 5 21.09 5 20.35 5 20.35 5 20.35 5 20.35 5 20.35 5 20.35	9.8 9.8 13.28 13.28 13.28 13.28 13.28 13.28	10.54 10.54 13.28 13.28 13.28 13.28 13.28
	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant ESS TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translatio 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numb 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requeste- Per 8XX No.	3	UNC1X UNC1X UNC1X OHD OHD OHD OHD OHD	CCOEF CCOSF N8R1X N8FTX N8FCX N8FMX		185.16 185.16 5.21 11.47 11.47 4.47 5.23	23.85 23.85 0.76 1.46 1.46 2.24	2.03 2.03 7.34	0.79 0.79	20.3i 20.3i 20.3i 20.3i 20.3i 20.3i 20.3i 20.3i	5 21.09 5 21.09 5 20.35 5 20.35 5 20.35 5 20.35 5 20.35 5 20.35	9.8 9.8 13.28 13.28 13.28 13.28	10.54 10.54 13.28 13.28 13.28 13.28
8XX ACCE	NRC Dark Fiber - Local Loop  RT OTHER  Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Chant SES TEN DIGIT SCREENING  8XX Access Ten Digit Screening, Per Ca  8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation 8XX Access Ten Digit Screening, Castomized Area of Service Per 8XX No.  8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requester Per 8XX No.  8XX Access Ten Digit Screening, Change Charge Per Reque	i	UNC1X UNC1X UNC1X OHD OHD OHD OHD OHD OHD OHD	CCOEF CCOSF N8R1X N8FTX N8FCX N8FMX N8FAX		185.16 185.16 5.21 11.47 11.47 4.47 5.23 5.97	23.85 23.85 0.76 1.46 1.46 2.24	2.03 2.03 7.34	0.79 0.79	20.34 20.34 20.33 20.33 20.33 20.33 20.33	5 21.09 5 21.09 5 20.35 5 20.35 5 20.35 5 20.35 5 20.35 5 20.35	9.8 9.8 13.28 13.28 13.28 13.28 13.28 13.28	10.54 10.54 13.28 13.28 13.28 13.28 13.28

		LIDB Validation Per Quer	OQU		0.0117403								
		LIDB Originating Point Code Establishment or Chanç	OQT, OQU	NRPBX		49.03	-			20.35	20.35	13.28	13.28
OLONIAL INIO	(0007)						-	-					
SIGNALING		CCS7 Signaling Termination, Per STP Por	1DB	PT8SX	138.41			-		20.35	20.35	13.32	13.32
		CCS7 Signaling Termination, Per STP Por CCS7 Signaling Usage, Per TCAP Messag	1DB	PIBSA	0.0000916		+			20.35	20.35	13.32	13.32
		CCS7 Signaling Osage, Per TCAP Messag CCS7 Signaling Connection, Per link (A link	1DB	TPP++	17.84	130.84	130.84			20.35	20.35	13.32	13.32
-		CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D lin	1DB	TPP++	17.84	130.84	130.84			20.35	20.35	13.32	13.32
		CCS7 Signaling Connection, Per link (B link) (also known as B lin	1DB	IFF++	0.0000373	130.04	130.04	+		20.33	20.33	13.32	13.32
		CCS7 Signaling Usage Surrogate, per link per LAT	1DB	STU56	352.3		+	+		20.35	20.35	13.32	13.32
		CCS7 Signaling Osage Surrogate, per link per EAT  CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per	100	31030	332.3		+	+		20.33	20.55	13.32	13.32
		STP affected	1DB	CCAPO		40	40			20.35	20.35	13.32	13.32
		CCS7 Signaling Point Code, per Destination Point Code Establishment or Change, Per	100	00/11 0		10				20.00	20.00	10.02	10.02
		Stp Affected	1DB	CCAPD		8	8			20.35	20.35	13.32	13.32
E911 SERVIO	CE												
CALLING NA	AME (CNAM)												
		CNAM for DB Owners, Per Query	OQV		0.016								
		CNAM for Non DB Owners, Per Query	OQV		0.01								
							1					1	
		CNAM (Non-Databs Owner), NRC, applies when using the Character Based User		00									46.55
		Interface (CHUI)	OQV	CDDCH		595	595			20.35	20.35	13.28	13.28
$\vdash$			_				+					+	
$\vdash$							<del> </del>	<del> </del>			+	+	
LNDOUEDV	CEDVICE							-					
LNP QUERY	SERVICE						<b>+</b>	<del></del>			+	+	
$\vdash$			<del>-  </del>	_			+				+	1	
							+						
OPERATOR	CALL PROC	CESSING					+	+					
OI ERATOR		Oper. Call Processing - Oper. Provided, Per Min Using BST LID			1.2		+				-		
		Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIC			1.24		+	1			-		
		Oper. Call Processing - Fully Automated, per Call - Using BST LID			0.2		+				_		
		Oper. Call Processing - Fully Automated, per Call - Using Foreign LIE			0.2								
							1				-		
INWARD OP	ERATOR SE	RVICES											
		Inward Operator Services - Verification, Per Ca			1								
		Inward Operator Services - Verification and Emergency Interrupt - Per C			1.95								
BRANDING -		R CALL PROCESSING											
		Recording of Custom Branded OA Announcement		CBAOS		7000	7000			19.99	19.99	19.99	19.99
		Loading of Custom Branded OA Announcement per shelf/NAV		CBAOL		500	500			19.99	19.99		
		CE SERVICES											
		Y ASSISTANCE ACCESS SERVICE											
		Directory Assistance Access Service Calls, Charge Per Ca			0.275								
	DIRECTORY	Y ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)											
		Directory Assistance Call Completion Access Service (DACC), Per Call Attem			0.1								
	UNBRANDIN						ļ					1	
	DIRECTORY	Y TRANSPORT										1	
		Directory Transport - Local Channel DS			133.81	868.97	486.83			20.35	20.35	13.28	13.28
$\vdash$		Directory Transport - DS1 Level Interoffice Per Mi			23	100.10	100.15			00.05		40.00	10.00
$\vdash$		Directory Transport - DS1 Level Interoffice Per Facility Termination			90	100.49	100.49			20.35	20.35	13.28	13.28
-	DIDECTORY	Y ASSISTANCE DATA BASE SERVICE (DADS)					<del> </del>	<del></del>			+	+	
$\vdash$	PIKECIORY				0.04		<del> </del>	<del></del>			+	+	
$\vdash$		Directory Assistance Data Base Service Charge Per Listir Directory Assistance Data Base Service, per mont		DBSOF	0.04 150		<del> </del>	<del></del>			+	+	
BEANDING		Y ASSISTANCE	_	DBSOF	150		+				+	+	
PLANDING -	- PIKECI OK	Custom Branding Announcement, per Recording to be used with the provision of DA	AMT	CBADA		3000	3000	+			+	+	
		Loading of Custom Branded Announcement per DRAM Card/Switch	AMT	CBADA		690	690	+			+	+	
$\vdash$		Loading of Oddion Branded Announcement per DRAW Cald/OWIGH	AIVII	CDADC		090	080	+			+	+	
SELECTIVE	ROLITING		<del>-  </del>	_			+				+	1	
SELECTIVE	KOUTING		_	-			+	+			+	+	
		Selective Routing Per Unique Line Class Code Per Request Per Swit		USRCR		179.6	179.6			30.89	7.03	1	
-		Ociocaro rodanigir di onique Line olassi odde nel request nel swit	<del>-  </del>	OUNCR		113.0	119.0			30.89	1.03	1	
	OLI OCATION	N .	_	+			+	+			+	+	
VIRTUAL CO	PELOCATION	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splittin	UEPSR, UEPSE	VE1LS	0.57	11.62	9.9	10.38	8.66	19.99	19.99	19.99	19.99
VIRTUAL CO				CNC2F	15.64	41.56	29.82	10.30	0.00	19.99	13.33	13.33	13.33
VIRTUAL CO		Virtual Collocation - 2-Fiber Cross Connect											
VIRTUAL CO		Virtual Collocation - 2-Fiber Cross Connect Virtual Collocation - 4-Fiber Cross Connects	CLO										
VIRTUAL CO		Virtual Collocation - 4-Fiber Cross Connects	CLO	CNC4F	28.11	50.53	38.78	10.46	8.75				
VIRTUAL CO								10.46	8.75				

	Regional Service Establishment	SRC	SRCEC		391788					19.99	19.99	19.99	19.9
	End Office Establishment	SRC	SRCEC		320.53	320.53				19.99	19.99	19.99	19.9
	Line/Port NRC, per end user	SRC	SRCLP	)	2.06	2.06				19.99	19.99	19.99	19.9
	Query NRC, per query	SRC		0.000448									
I - BELLSOUT	TH AIN SMS ACCESS SERVICE										<u> </u>		
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup		CAMSE	_	135.56	135.56			1 1	20.35	20.35	13.28	13.2
	Ain Sins Access Service - Service Establishment, Fer State, Initial Setup		CAIVISE	-	135.56	135.56				20.35	20.35	13.20	13.20
	AIN SMS Access Service - Port Connection - Dial/Shared Access		CAMDE		41.75	41.75			1 1	20.35	20.35	13.28	13.2
	7414 GMG 7400033 GCTVICC T GR GOTINGCROTT Blaw Griated 7400035	<del> </del>	Ortivibi		41.73	41.73				20.33	20.33	13.20	13.20
	AIN SMS Access Service - Port Connection - ISDN Access		CAM1F	•	41.75	41.75			1 1	20.35	20.35	13.28	13.2
	AIN SMS Access Service - User Identification Codes - Per User ID Code		CAMAL	J	96.63	96.63			1 1	20.35	20.35	13.28	13.2
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement		CAMRO		113.67	113.67				20.35	20.35	13.28	13.2
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)			0.0024									
	AIN SMS Access Service - Session, Per Minute			0.0820123									
	AIN SMS Access Service - Company Performed Session, Per Minute			2.27					+ +		<b>_</b>		
DELL COLIT	TH AIN TOOLKIT SERVICE										-		+
- DELLSOUI	ITI AIN TOULNIT SERVICE	+ +			<del>                                     </del>	1	1		1		<del>                                     </del>		+
	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup		BAPSC		132.04	132.04				20.35	20.35	13.28	13.2
	AIN Toolkit Service - Training Session, Per Customer	++	BAPVX		7915	7915			+ +	20.35	20.35	13.28	13.2
	AIN Toolkit Service - Training Session, Fer Customer  AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt	++	BAPTT		31.21	31.21		<del>                                     </del>	+	20.35	20.35	13.28	13.2
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay	1 1	BAPTD		31.21	31.21			1	20.35	20.35	13.28	13.2
							İ						1
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate		BAPTM		31.21	31.21				20.35	20.35	13.28	13.
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP		BAPTO		85.24	85.24				20.35	20.35	13.28	13.
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP		BAPTC		85.24	85.24				20.35	20.35	13.28	13.2
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code		BAPTF		85.24	85.24				20.35	20.35	13.28	13.2
	AIN Toolkit Service - Query Charge, Per Query			0.0211882									
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query			0.0054774									
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes			1.5						-			
	And Toolkit Service - Sor Storage Charge, Fer Sivis Access Account, Fer Too Kilobytes	<del>                                     </del>		1.5									+
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription		BAPMS	17.43	33.52	33.52			1 1	20.35	20.35	13.28	13.2
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription		BAPLS		36.23	36.23				20.35	20.35	13.28	13.2
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription		BAPDS		33.52	33.52				20.35	20.35	13.28	13.2
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription		BAPES	0.0511435	36.23	36.23				20.35	20.35	13.28	13.2
UF/EDOUF/AI	DUF/CMDS	+ +											+
0.72200.774	5017011150												1
	ADUF: Message Processing, per messag			0.004									
	ADUF: Data Transmission (CONNECT:DIRECT), per messag			0.001									
	EODUF: Message Processing, per message			0.004									
0.07	TIONAL DAILY USAGE FILE (ODUF)										ļ		-
OPI	ODUF: Recording, per message	+		0.0000044							<b>_</b>		+
	ODUF: Message Processing, per message	+ +		0.0027366							<del>                                     </del>		+
	ODUF: Message Processing, per Magnetic Tape provisions	++		52.75					1				1
	ODUF: Data Transmission (CONNECT:DIRECT), per messag			0.0000339									1
IANCED EXT	TENDED LINK (EELs)								1		<u> </u>		4
	TE No. FEL	FI FO L	N	1	l	1	1		1		<del>                                     </del>		1
	TE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami,			N; New Orleans,	LA;	1		<del>                                     </del>	1		<u> </u>		+
	TE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates belo TE: In all states, EEL network elements shown below also apply to currently combined facilities wh			witch As Is Ch	no annlice to	reportly	ad faciliti	converted to UNITS ALS	n roourei	too de est	onnly'		+
	TE: In Georgia, the EEL network elements snown below also apply to currently combined facilities white in Georgia, the EEL network elements apply to ordinarily combined network elements per the G				ge applies to cu	irrently combin	led facilities	converted to UNES.(No	n-recurring ra	tes do not	арріу.)		+
1401		30 Oraci (NO SWILL	A3 13 Gridi	9~./	<b> </b>	1	1		+		<del>                                     </del>		+
2-W	IRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	++							1				1
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	1 UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone	2 UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.
					l								1
			UEAL2		108.76	35.47	72.94	10.86	1	20.35	21.09	9.8	10
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zonu	3 UNCVX					1	1	1			1	1
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor	UNC1X	1L5XX		.=				+				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor	UNC1X UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.9		20.35	21.09	9.8	10.
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor DS1 Channelization System Per Mont	UNC1X UNC1X UNC1X	U1TF1 MQ1	77.86 80.77	214.52	49.95	70.07 75.98	30.9 13.6		20.35	21.09	9.8	10.
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per mor Interoffice Transport - Dedicated - DS1 combination - Facility Termination per mor	UNC1X UNC1X	U1TF1	77.86 80.77						20.35	21.09	9.8	10.

Page 8 of 22

	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone:	2	UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86		20.35	21.09	9.8	
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport													Γ
	Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon	3	UNCVX	UEAL2 1D1VG	28.28 0.91	108.76 5.7	35.47 4.42	72.94	10.86		20.35	21.09	9.8	ļ
	Voice Grade COCI - DST to DS0 Channer System combination - per mon		UNCVA	IDIVG	0.91	5.7	4.42							t
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC1X	UNCCC		52.73	24.62	9.12	9.12		20.35	21.09	9.8	
4-WIRE VO	DICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)													ł
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -													t
	Zone 1	1	UNCVX	UEAL4	24.7	108.76	35.47	72.94	10.86		20.35	21.09	9.8	
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	2	UNCVX	UEAL4	32.25	108.76	35.47	72.94	10.86		20.35	21.09	9.8	
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination -													t
	Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	3	UNCVX UNC1X	UEAL4 1L5XX	42.17 0.3525	108.76	35.47	72.94	10.86		20.35	21.09	9.8	-
	Interoffice Transport - Dedicated - DS1 - Combination - Fer Mile Fer Mor		UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.9		20.35	21.09	9.8	t
	Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	80.77	214.52	49.95	75.98	13.6					T
	Voice Grade COCI - DS1 to DS0 Channel System combination - per mon		UNCVX	1D1VG	0.91	5.7	4.42							
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport													Г
	Combination - Zone	1	UNCVX	UEAL4	24.7	108.76	35.47	72.94	10.86		20.35	21.09	9.8	
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 2	2	UNCVX	UEAL4	32.25	108.76	35.47	72.94	10.86		20.35	21.09	9.8	
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport													t
	Combination - Zone :  Voice Grade COCI - DS1 to DS0 Channel System combination - per mon	3	UNCVX	UEAL4 1D1VG	42.17 0.91	108.76 5.7	35.47 4.42	72.94	10.86		20.35	21.09	9.8	H
	voice Grade COOL - DOT to DOC Granner System Combination - per mon		UNCVA	וטועט	0.91	ა./	4.42							t
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charç	$\vdash$	UNC1X	UNCCC		52.73	24.62	9.12	9.12		20.35	21.09	9.8	L
4-WIRE 56	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)													t
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -		LINODY	LIDI 50	24.4	400.70	25 47	70.04	40.00		20.05	04.00		Γ
-	Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	1	UNCDX	UDL56	31.1	108.76	35.47	72.94	10.86		20.35	21.09	9.8	H
	Zone 2	2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86		20.35	21.09	9.8	L
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3	3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86		20.35	21.09	9.8	
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mon	J	UNC1X	1L5XX	0.3525	100.70	00.47	72.54	10.00		20.00	21.00	3.0	
	Liver West Towns of Deliver Land 1994		UNC1X	U1TF1	77.00	474.04	440.40	70.07	30.9		00.05	04.00	0.0	
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Mor				77.86	171.24	113.12				20.35	21.09	9.8	+
	Channelization - Channel System DS1 to DS0 combination Per Mor  OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kb)		UNC1X UNCDX	MQ1 1D1DD	80.77 1.82	214.52 5.7	49.95 4.42	75.98	13.6	<del>                                     </del>			<del>                                     </del>	t
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone	1	UNCDX	UDL56	31.1	108.76	35.47	72.94	10.86		20.35	21.09	9.8	t
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport													T
	Combination - Zone : Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86		20.35	21.09	9.8	H
	Combination - Zone :	3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86		20.35	21.09	9.8	L
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-		UNCDX	1D1DD	1.00	E 7	4.40							l
	64kbs)		UNCDX	טטוטו	1.82	5.7	4.42						<del>                                     </del>	t
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charg	$\vdash$	UNC1X	UNCCC		52.73	24.62	9.12	9.12		20.35	21.09	9.8	L
4-WIRE 64	I KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)													t
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -		LINODY	LIBI C :	04.4	100.76	05.47	70.04	40.00		00.05	04.00		Γ
	Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination -	1	UNCDX	UDL64	31.1	108.76	35.47	72.94	10.86	<del>                                     </del>	20.35	21.09	9.8	H
	Zone 2	2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86		20.35	21.09	9.8	L
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3	3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86		20.35	21.09	9.8	
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Mor	Ĭ	UNC1X	1L5XX	0.3525	. 50.70	55.47	, 2.04	10.00		20.00	200	5.0	t
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Mor		UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.9		20.35	21.09	9.8	
	Channelization - Channel System DS1 to DS0 combination Per Mor		UNC1X	MQ1	80.77	214.52	49.95	75.98	13.6		20.33	21.08	3.0	t
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-													Γ
	64kbs)		UNCDX	1D1DD	1.82	5.7	4.42						-	H
	Additional 4-vvice 64Kpps Digital Grade Loopin same DS1 Interoffice Transport	1	UNCDX	UDL64	31.1	108.76	35.47	72.94	10.86		20.35	21.09	9.8	L
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone			1			35.47	72.94	10.86		20.35	21.09	9.8	
	Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	LINCDY	LIDL 64	40.61			1 /2.54	10.00		۷.۵۵ ک		9.0	+
	Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :	2	UNCDX	UDL64	40.61	108.76	35.47			1		21.00		
	Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone :	3	UNCDX	UDL64	40.61 53.11	108.76	35.47	72.94	10.86		20.35	21.09	9.8	L
	Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport								10.86		20.35		9.8	
	Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)		UNCDX	UDL64	53.11	108.76 5.7	35.47 4.42	72.94				21.09		
	Combination - Zone Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone : OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		UNCDX	UDL64	53.11	108.76	35.47		9.12		20.35		9.8	

	Vire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone	2	UNC1X	USLXX	75.4	228.4	161.74	79.87	24.88	20.35	21.09	9.8	_
	Vire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone		UNC1X	USLXX	98.59	228.4	161.74	79.87	24.88	20.35	21.09	9.8	_
Inte	eroffice Transport - Dedicated - DS1 combination - Per Mile Per Mor		UNC1X	1L5XX	0.3525								-
Inte	eroffice Transport - Dedicated - DS1 combination - Facility Termination Per Moi		UNC1X	U1TF1	77.86	171.24	113.12	70.07	30.9	20.35	21.09	9.8	
Nor	nrecurring Currently Combined Network Elements Switch -As-Is Charç		UNC1X	UNCCC		52.73	24.62	9.12	9.12	20.35	21.09	9.8	
4-WIRE DS1 DIG	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)												-
	st DS1Loop in DS3 Interoffice Transport Combination - Zone	1	UNC1X	USLXX	57.73	228.4	161.74	79.87	24.88	20.35	21.09	9.8	$\dashv$
	st DS1Loop in DS3 Interoffice Transport Combination - Zone	2	UNC1X	USLXX	75.4	228.4	161.74	79.87	24.88	20.35	21.09	9.8	$\top$
Firs	st DS1Loop in DS3 Interoffice Transport Combination - Zone		UNC1X	USLXX	98.59	228.4	161.74	79.87	24.88	20.35	21.09	9.8	1
Inte	eroffice Transport - Dedicated - DS3 combination - Per Mile Per Mor		UNC3X	1L5XX	2.34				=				T
	eroffice Transport - Dedicated - DS3 - Facility Termination per mor		UNC3X	U1TF3	848.99	428.1	153.81	64.43	35.43	20.35	21.09	9.8	T
DS:	3 to DS1 Channel System combination per mon		UNC3X	MQ3	222.98	319.48	126.63	45.53	17.05				
	3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	17.58	6.52	2.58						
	ditional DS1Loop in DS3 Interoffice Transport Combination - Zone		UNC1X	USLXX	57.73	228.4	161.74	79.87	24.88	20.35	21.09	9.8	
Add	ditional DS1Loop in DS3 Interoffice Transport Combination - Zone		UNC1X	USLXX	75.4	228.4	161.74	79.87	24.88	20.35	21.09	9.8	
Ado	ditional DS1Loop in DS3 Interoffice Transport Combination - Zone		UNC1X	USLXX	98.59	228.4	161.74	79.87	24.88	20.35	21.09	9.8	
DS:	3 Interface Unit (DS1 COCI) combination per montl		UNC1X	UC1D1	17.58	6.52	2.58						+
Nor	nrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC3X	UNCCC		52.73	24.62	9.12	9.12	20.35	21.09	9.8	
2-WIRE VOICE	GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)												
	WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone	1	UNCVX	UEAL2	16.56	108.76	35.47	72.94	10.86	20.35	21.09	9.8	1
	WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone		UNCVX	UEAL2	21.63	108.76	35.47	72.94	10.86	20.35	21.09	9.8	1
	VireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone		UNCVX	UEAL2	28.28	108.76	35.47	72.94	10.86	20.35	21.09	9.8	
Inte	eroffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Mor		UNCVX	1L5XX	0.0174								I
	eroffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility		11110177	1147710	10.50	70.00	44.00	00.00		00.55	24.00	0.0	Ī
Ter	mination per month		UNCVX	U1TV2	18.58	79.86	44.06	69.32	31	20.35	21.09	9.8	+
Nor	nrecurring Currently Combined Network Elements Switch -As-Is Charg		UNCVX	UNCCC		52.73	24.62	9.12	9.12	20.35	21.09	9.8	+
4-WIRE VOICE	GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)												+
	VireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	1	UNCVX	UEAL4	24.7	108.75	35.47	72.94	10.85	20.35	21.09	9.8	
4-W	VireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone	2	UNCVX	UEAL4	32.25	108.75	35.47	72.94	10.85	20.35	21.09	9.8	T
	VireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone		UNCVX	UEAL4	42.17	108.75	35.47	72.94	10.85	20.35	21.09	9.8	
Inte	eroffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Mor		UNCVX	1L5XX	0.0054								
	eroffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility												
Ter	mination per month		UNCVX	U1TV4	24.09	79.83	44.08	69.32	31	20.35	21.09	9.8	
Nor	nrecurring Currently Combined Network Elements Switch -As-Is Charç		UNCVX	UNCCC		52.73	24.62	9.12	9.12	20.35	21.09	9.8	
DS3 DIGITAL EX	XTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)												
Hig	h Capacity Unbundled Local Loop - DS3 combination - Per Mile per mor		UNC3X	1L5ND	9.19								
	h Capacity Unbundled Local Loop - DS3 combination - Facility Termination per												
	nth		UNC3X	UE3PX	374.24	240.23	180.87	106.78	45.24				
Inte	eroffice Transport - Dedicated - DS3 - Per Mile per mon		UNC3X	1L5XX	2.34								+
Inte	eroffice Transport - Dedicated - DS3 combination - Facility Termination per per mo		UNC3X	U1TF3	848.99	428.01	153.81	64.43	35.43	20.35	21.09	9.8	
Nor	nrecurring Currently Combined Network Elements Switch -As-Is Charg		UNC3X	UNCCC		52.73	24.62	9.12	9.12	20.35	21.09	9.8	
				2.1000		52.170				20.00		3.0	1
	EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)  th Capacity Unbundled Local Loop - STS1 combination - Per Mile per more		UNCSX	1L5ND	9.19								+
	h Capacity Unbundled Local Loop - STS1 combination - Per Mile per moi	+	OINCOV	UNCLI	9.19								+
moi			UNCSX	UDLS1	389.35	240.23	180.87	106.78	45.24				
	eroffice Transport - Dedicated - STS1 combination - Per Mile per mor		UNCSX	1L5XX	2.34	270.20	100.07	100.70	70.27				İ
Inte	eroffice Transport - Dedicated - STS1 combination - Facility Termination per mo		UNCSX	U1TFS	849.3	428.01	153.81	64.43	35.43	20.35	21.09	9.8	T
	nrecurring Currently Combined Network Elements Switch -As-Is Charc		UNCSX	UNCCC	2.0.0	52.73	24.62	9.12	9.12	20.35	21.09	9.8	T
	· ·		UNCOX	UNCCC		52.73	24.02	9.12	9.12	20.35	21.09	9.8	1
	XTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)												4
	st 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		UNCNX	U1L2X	22	108.76	35.47	72.94	10.86	20.35	21.09	9.8	_
	st 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		UNCNX	U1L2X	29.02	108.76	35.47	72.94	10.86	20.35	21.09	9.8	+
	st 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone		UNCNX	U1L2X	37.95	108.76	35.47	72.94	10.86	20.35	21.09	9.8	+
	eroffice Transport - Dedicated - DS1 combination - Per Mi eroffice Transport - Dedicated - DS1 combintion - Facility Termination per mor		UNC1X UNC1X	1L5XX U1TF1	0.3525 77.86	171.24	113.12	70.07	30.9	20.35	21.09	9.8	+
	annelization - Channel System DS1 to DS0 combination - per mor		UNC1X	MQ1	80.77	49.95	75.98	13.6	50.5	20.33	21.03	3.0	+
2-w	vire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per mon		UNCNX	UC1CA	3.1	6.16	0.6	13.0					1
Ado	ditional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	1	UNCNX	U1L2X	22	108.76	35.47	72.94	10.86	20.35	21.09	9.8	
													İ
Add	ditional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	2	UNCNX	U1L2X	29.02	108.76	35.47	72.94	10.86	20.35	21.09	9.8	
		1 1		1	07.05	108.76	35.47	72.94	10.86	20.35	21.09	9.8	
Add	ditional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zon	3	UNCNX	U1L2X	37.95	108.76	35.47	72.94	10.00	20.33	21.09	3.0	

Nonrecurring Currently Combined Network Elements Switch -As-Is Charq			UNC1X	UNCCC		52.73	24.62	9.12	9.12		20.35	21.09	9.8	10.54
4-WIRE DS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT	(EEL)													
First DS1 Loop in STS1 Interoffice Transport Combination - Zone	(CCL)	1	UNC1X	USLXX	57.73	228.4	161.74	79.87	24.88	<del>                                     </del>	20.35	21.09	9.8	10.54
First DS1 Loop in STS1 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	75.4	228.4	161.74	79.87	24.88	<del>                                     </del>	20.35	21.09	9.8	10.54
First DS1 Loop in S1S1 Interoffice Transport Combination - Zone  First DS1 Loop in STS1 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	98.59	228.4	161.74	79.87	24.88	<del>                                     </del>	20.35	21.09	9.8	10.54
Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mor		3	UNCSX	1L5XX	2.34	220.4	101.74	19.01	24.00		20.33	21.09	9.0	10.54
						100.01	450.04	04.40	05.40		00.05	04.00	0.0	40.54
Interoffice Transport - Dedicated - STS1 combination - Facility Terminati			UNCSX	U1TFS	849.3	426.01	153.61	64.43	35.43		20.35	21.09	9.8	10.54
STS1 to DS1 Channel System conbination per mon			UNCSX	MQ3	222.98	428.01	153.81	64.43	25.43					
DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1	17.58	5.7	4.42							
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	57.73	228.4	161.74	79.87	24.88		20.35	21.09	9.8	10.54
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		2	UNC1X	USLXX	75.4	228.4	161.74	79.87	24.88		20.35	21.09	9.8	10.54
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone		3	UNC1X	USLXX	98.59	228.4	161.74	79.87	24.88		20.35	21.09	9.8	10.54
DS3 Interface Unit (DS1 COCI) combination per montl			UNC1X	UC1D1	17.58	5.7	4.42							
Nonrecurring Currently Combined Network Elements Switch -As-Is Charç			UNCSX	UNCCC		52.73	24.62	9.12	9.12		20.35	21.09	9.8	10.54
4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)														
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		1	UNCDX	UDL56	31.1	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.54
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		2	UNCDX	UDL56	40.61	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.54
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone		3	UNCDX	UDL56	53.11	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.54
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per M			UNCDX	1L5XX	0.174									
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Terminati			UNCDX	U1TD5	22.1	58.54	38.32	13.98	8.59		20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNCDX	UNCCC		52.73	24.62	9.12	9.12		20.35	21.09	9.8	10.54
4-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)														
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		1	UNCDX	UDL64	31.1	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.54
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		2	UNCDX	UDL64	40.61	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.54
		3	UNCDX	UDL64	53.11	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.54
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone		3	UNCDX	1L5XX	0.174	108.76	35.47	72.94	10.86		20.35	21.09	9.8	10.54
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per M														
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Terminati			UNCDX	U1TD6	22.1	58.54	38.32	13.98	8.59	+	20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements Switch -As-Is Charg			UNCDX	UNCCC		52.73	24.62	9.12	9.12		20.35	21.09	9.8	10.54
TIONAL NETWORK ELEMENTS														
		ا ــــــا												
When used as a part of a currently combined facility, the non-recurring charges do not apply, to														
When used as ordinarilty combined network elements in Georgia, the non-recurring charges a	pply and the	he Swi	tch As Is Charge	does not.										
Node (SynchroNet)														
Node (SynchroNet) Node per month			UNCDX	UNCNT	17.11									
Node per month			UNCDX	UNCNT	17.11									
Node per month  Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear		nation)	UNCDX	UNCNT	17.11									
Node per month		nation)			17.11									
Node per month		nation)	UNCDX	UNCNT	17.11	52.73	24.62	9.12	9.12		20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear  2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversio Charge  56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion		nation)	UNCVX	UNCCC	17.11									
Node per month		nation)			17.11	52.73 52.73	24.62	9.12	9.12 9.12		20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge		nation)	UNCVX	UNCCC	17.11	52.73	24.62	9.12	9.12		20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear  2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversio Charge  56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion		nation)	UNCVX	UNCCC	17.11									
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		nation)	UNCVX UNCDX UNC1X	UNCCC	17.11	52.73 52.73	24.62 24.62	9.12	9.12		20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		nation)	UNCVX	UNCCC	17.11	52.73	24.62	9.12	9.12		20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear   2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion   Charge		nation)	UNCVX UNCDX UNC1X UNC3X	UNCCC UNCCC UNCCC	17.11	52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char		nation)	UNCVX UNCDX UNC1X	UNCCC	17.11	52.73 52.73	24.62 24.62	9.12	9.12		20.35	21.09	9.8	10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char	h		UNCVX UNCDX UNC1X UNC3X UNC3X	UNCCC UNCCC UNCCC	17.11	52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear   2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion   Charge	h		UNCVX UNCDX UNC1X UNC3X UNC3X	UNCCC UNCCC UNCCC	17.11	52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char   DS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   ST51 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month,	h		UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX	UNCCC UNCCC UNCCC UNCCC	17.11	52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon	h		UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX sfour months UNCXV	UNCCC UNCCC UNCCC	19.43	52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear   2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion   Charge	h		UNCVX UNCDX UNC1X UNC3X UNCSX Sefour months UNCXV UNCXV	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC		52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon	h		UNCVX UNCDX UNC1X UNC3X UNC3X UNCSX sfour months UNCXV	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC	19.43 20.56	52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  NOTE: Local Channel - Dedicated Transport - minimum billing period - Below DS3=one month, Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - 4-Wire Voice Grade per mon Local Channel - Dedicated - DS1 Per Mont	h		UNCVX UNCDX UNC1X UNC3X UNCSX Sefour months UNCXV UNCXV	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC	19.43 20.56	52.73 52.73 52.73	24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Node per month	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX Sefour months UNCXV UNCXV UNCXV UNCXV	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC	19.43 20.56 40	52.73 52.73 52.73 52.73	24.62 24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Node per month	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX UNCSX UNCSX UNCXV UNCXV UNCXV UNCXV UNC1X	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV2 ULDV4 ULDF1	19.43 20.56 40	52.73 52.73 52.73 52.73 52.73	24.62 24.62 24.62 24.62	9.12 9.12 9.12	9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nortecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX Efour months UNCXV UNCXV UNC1X UNC1X Ctronic service of th regional electr	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV4 ULDF1	19.43 20.56 40 ges as ordered ordering charge	52.73 52.73 52.73 52.73 52.73	24.62 24.62 24.62 24.62 24.62	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char   STS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated - 2-Wire Voice Grade per mon   Local Channel - Dedicated - 2-Wire Voice Grade per mon   Local Channel - Dedicated - 4-Wire Voice Grade per mon   Local Channel - Dedicated - DS1 Per Mont   RATIONAL SUPPORT SYSTEMS   NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhit NOTE: (1) Continued: The electronic service ordering charge currently contained in this rate exhit NOTE: (1) Continued: CLEC-1 may elect either the state specific Commission ordered rates for the contract of the commission ordered rates for the contract of the commission ordered rates for the contract of the commission ordered rates for the contract of the commission ordered rates for the contract of the commission ordered rates for the contract of the commission ordered rates for the contract of the commission ordered rates for the contract of	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX Efour months UNCXV UNCXV UNC1X UNC1X Ctronic service of th regional electr	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV4 ULDF1	19.43 20.56 40 ges as ordered ordering charge	52.73 52.73 52.73 52.73 52.73	24.62 24.62 24.62 24.62 24.62	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nortecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX Efour months UNCXV UNCXV UNC1X UNC1X Ctronic service of th regional electr	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV4 ULDF1	19.43 20.56 40 ges as ordered ordering charge	52.73 52.73 52.73 52.73 52.73	24.62 24.62 24.62 24.62 24.62	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  NOTE: Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - DS1 Per Mont  RATIONAL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX Efour months UNCXV UNCXV UNC1X UNC1X Ctronic service of th regional electr	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV4 ULDF1	19.43 20.56 40 ges as ordered ordering charge	52.73 52.73 52.73 52.73 52.73	24.62 24.62 24.62 24.62 24.62	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX Efour months UNCXV UNCXV UNC1X UNC1X Ctronic service of th regional electr	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV4 ULDV4 ULDF1 rdering char onic service rges, or CLE	19.43 20.56 40 ges as ordered ordering charge	52.73 52.73 52.73 52.73 by the State Cone	24.62 24.62 24.62 24.62 24.62	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge  DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char  DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge  NOTE: Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - 2-Wire Voice Grade per mon Local Channel - Dedicated - DS1 Per Mont  RATIONAL SUPPORT SYSTEMS  NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it prefers the NOTE: (1) Concluded: CLEC-1 may elect either the state specific Commission ordered rates for the NOTE: (2) Manual Service Order charge: disconnect, in the state of Florida, to be billed on a per	DS3 and a	above=	UNCVX UNCDX UNC1X UNC3X UNCSX Efour months UNCXV UNCXV UNC1X UNC1X Ctronic service of th regional electr	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC ULDV2 ULDV4 ULDF1	19.43 20.56 40 ges as ordered ordering charge	52.73 52.73 52.73 52.73 52.73	24.62 24.62 24.62 24.62 24.62	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12		20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char   STS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated - 2-Wire Voice Grade per mon   Local Channel - Dedicated - 2-Wire Voice Grade per mon   Local Channel - Dedicated - 4-Wire Voice Grade per mon   Local Channel - Dedicated - DS1 Per Mont   State Contract   State C	DS3 and a	above=	UNCVX  UNCDX  UNC1X  UNC3X  UNCSX  -four months  UNCXV  UNC1X  UNC1X  UNC1X  ctronic service of the regional electrosice ordering characteristics.	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SULDV2 ULDV2 ULDV4 ULDF1 rdering char onic service gges, or CLE	19.43 20.56 40 ges as ordered ordering charge C-1 may elect t	52.73 52.73 52.73 52.73 52.73 by the State Conel her regional elect	24.62 24.62 24.62 24.62  missions  ronic service of	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12		20.35 20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX  UNCDX  UNC1X  UNC3X  UNCSX  -four months  UNCXV  UNC1X  UNC1X  UNC1X  ctronic service of the regional electrosice ordering characteristics.	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SULDV2 ULDV2 ULDV4 ULDF1 rdering char onic service gges, or CLE	19.43 20.56 40 ges as ordered ordering charge C-1 may elect t	52.73 52.73 52.73 52.73 52.73 by the State Conel her regional elect	24.62 24.62 24.62 24.62  missions  ronic service of	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12	ffice, refer to Internet Wet	20.35 20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Char   STS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge   NOTE: Local Channel - Dedicated - 2-Wire Voice Grade per mon   Local Channel - Dedicated - 2-Wire Voice Grade per mon   Local Channel - Dedicated - 4-Wire Voice Grade per mon   Local Channel - Dedicated - DS1 Per Mont   State Contract   State C	DS3 and a	above=	UNCVX  UNCDX  UNC1X  UNC3X  UNCSX  -four months  UNCXV  UNC1X  UNC1X  UNC1X  ctronic service of the regional electrosice ordering characteristics.	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SULDV2 ULDV2 ULDV4 ULDF1 rdering char onic service gges, or CLE	19.43 20.56 40 ges as ordered ordering charge C-1 may elect t	52.73 52.73 52.73 52.73 52.73 by the State Conel her regional elect	24.62 24.62 24.62 24.62  missions  ronic service of	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12	ffice, refer to Internet Wet	20.35 20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   S6/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX  UNCDX  UNC1X  UNC3X  UNCSX  -four months  UNCXV  UNC1X  UNC1X  UNC1X  ctronic service of the regional electrosice ordering characteristics.	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SULDV2 ULDV2 ULDV4 ULDF1 rdering char onic service gges, or CLE	19.43 20.56 40 ges as ordered ordering charge C-1 may elect t	52.73 52.73 52.73 52.73 52.73 by the State Conel her regional elect	24.62 24.62 24.62 24.62  missions  ronic service of	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12	ffice, refer to Internet Wet	20.35 20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   S6/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX  UNCDX  UNC1X  UNC3X  UNCSX  -four months  UNCXV  UNC1X  UNC1X  UNC1X  ctronic service of the regional electrosice ordering characteristics.	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SULDV2 ULDV2 ULDV4 ULDF1 rdering char onic service gges, or CLE	19.43 20.56 40 ges as ordered ordering charge C-1 may elect t	52.73 52.73 52.73 52.73 52.73 by the State Conel her regional elect	24.62 24.62 24.62 24.62  missions  ronic service of	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12	ffice, refer to Internet Wet	20.35 20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   S6/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX  UNCDX  UNC1X  UNC3X  UNCSX  -four months  UNCXV  UNC1X  UNC1X  UNC1X  ctronic service of the regional electrosice ordering characteristics.	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SULDV2 ULDV2 ULDV4 ULDF1 rdering char onic service gges, or CLE	19.43 20.56 40 ges as ordered ordering charge C-1 may elect t	52.73 52.73 52.73 52.73 52.73 by the State Conel her regional elect	24.62 24.62 24.62 24.62  missions  ronic service of	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12	ffice, refer to Internet Wet	20.35 20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54
Nonrecurring Currently Combined Network Elements "Switch As Is" Charge (One applies to ear 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge   S6/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	DS3 and a	above=	UNCVX  UNCDX  UNC1X  UNC3X  UNCSX  -four months  UNCXV  UNC1X  UNC1X  UNC1X  ctronic service of the regional electrosice ordering characteristics.	UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC UNCCC SULDV2 ULDV2 ULDV4 ULDF1 rdering char onic service gges, or CLE	19.43 20.56 40 ges as ordered ordering charge C-1 may elect t	52.73 52.73 52.73 52.73 52.73 by the State Conel her regional elect	24.62 24.62 24.62 24.62  missions  ronic service of	9.12 9.12 9.12 9.12	9.12 9.12 9.12 9.12	ffice, refer to Internet Wet	20.35 20.35 20.35 20.35	21.09 21.09 21.09	9.8 9.8 9.8	10.54 10.54 10.54

NOTE: Alth	ough the Port Rate includes all available features in GA & TN, the desired features will need	to be ordered using reta	ail USOCs										
2-WIRE VO	ICE GRADE LINE PORT RATES (RES)									-			
	Exchange Ports - 2-Wire Analog Line Port- Re:	UEPSR	UEPRL	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	_
	Fusheres Date 2 Miss Assles Line Destroit Collins D.	LIEBOE	UEPRC	4.00	0.00	0.10	2.00	2.02		20.25	40.54	40.00	
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Re	UEPSR	UEPRC	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Re	UEPSR	UEPRO	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled TN extended local dialing parity Port with Caller	OLI OIL	OLI IXO	1.00	3.30	5.15	5.00	2.02		20.00	10.04	10.02	
	ID - Res.	UEPSR	UEPAQ	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Plus with Caller ID - Res (AC	UEPSR	UEPAH	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID -												
	Res (F2R)  Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID -	UEPSR	UEPAK	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Res (TACER)	UEPSR	UEPAL	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID -	OZ. OK	OLI AL	1.00	0.00	0.10	0.00	2.02		20.00	10.01	10.02	
	Res (TACSR)	UEPSR	UEPAM	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID -												
	Res (1MF2X)	UEPSR	UEPAN	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled Tennessee Area Calling port with Caller ID - Res (2MR)	UEPSR	UEPAO	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LU	UEPSR	UEPAO	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exonange 1 on 2 mile vo unbunuleu res, low usage line port with odiler ID (EU	OLISK	OLF AF	1.00	5.55	3.13	3.00	2.32		20.00	10.54	10.02	
	Subsequent Activity	UEPSR	USASC	0	0	0							
FEATURES													Ξ
	All Available Vertical Feature	UEPSR	UEPVF	0	0	0				20.35	10.54	13.32	_
2-WIRE VO	ICE GRADE LINE PORT RATES (BUS)												_
L WINL VO	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bi	UEPSB	UEPBL	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484	52.55	J 2.1. DE		2.00	27.0	2.00				. 5.5 .	. 3.02	
	ID - Bus.	UEPSB	UEPBC	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Subsect Date Of Was Audio Live Bud and in 1997			4	0	0 :-		0.00		00.55	46 - 1	40.00	
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bu	UEPSB	UEPBO	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled TN extended local dialing parity Port with Caller ID - Bus.	UEPSB	UEPAV	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - B	UEPSB	UEPB1	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area Calling Port Economy												
	Option - Bus (TACC1)	UEPSB	UEPAC	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Exchange Ports - 2-Wire VG unbundled TN Bus 2-Way Area Calling Port Standard			4	0	0 :-		0.00		00.55	46 - 1	40.00	
	Option - Bus (TACC2)  Exchange Ports - 2-W VG unbundled TN Bus 2-Way Collierville & Memphis Local	UEPSB	UEPAD	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Calling Port - Bus (B2F)	UEPSB	UEPAE	1.89	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	Subsequent Activity	UEPSB	USASC	0	0	0							
FEATURES		LIEDOS	HED) (E							20.25	10.51	40.00	
EXCHANGE	All Available Vertical Feature E PORT RATES (DID & PBX)	UEPSB	UEPVF	0	0	0				20.35	10.54	13.32	
	Exchange Ports - 2-Wire DID Port	UEPEX	UEPP2	8.97	47.75	47.01	9.21	8.47		20.35	10.54	13.32	
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capabilii	UEPDD	UEPDD	35.74	75.93	38.15	8.77	8.04		19.99	19.99	19.99	_
	Exchange Ports - 2-Wire ISDN Port (See Notes below	UEPTX UEPS)		16.26	30.23	29.49	4.1	4.1		41.43	42.17	9.8	
	nsmission/usage charges associated with POTS circuit switched usage will also apply to circuit s								Des :				
NUIE: ACC	ess to B Channel or D Channel Packet capabilities will be available only through BFR/New Busin Exchange Ports - 2-Wire ISDN Port Channel Profiles	ness Request Process. UEPTX UEPS		e packet capabil 0	ities will be dete	rmined via the	⊳ona ⊦ide R	equest/New Business Re	equest Proces	55.			
	Exchange Ports - 2-Wire ISDN Port Channel Profiles  Exchange Ports - 4-Wire ISDN DS1 Por	UEPTX UEPS	UEPEX	75.04	148.66	147.18	38.46	36.98		40.69	42.17	9.07	
	Endings Fore Finite Iobit Do FF of	JEI EX	OLI LX	70.04	140.00	147.10	55.45	55.50		40.00	74.11	3.01	
	2-Wire VG Unbundled 2-Way PBX Trunk - Re:	UEPSE	UEPRD	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
													Ī
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bu	UEPSP	UEPPC	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu	UEPSP	UEPPO	1.79	9.93	9.19	3.66	2 92		20.35	10.54	13.32	
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bu 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bu	UEPSP	UEPP0	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	2-Wire Analog Long Distance Terminal PBX Trunk - Bu	UEPSP	UEPLD	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	2-Wire Analog TN 2-Way Calling Plan PBX Trunk - Bu	UEPSP	UEPT2	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	2-Wire TN Outward Calling Plan PBX Trunk - Bu	UEPSP	UEPTO	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	2-Wire Voice Unbundled PBX LD Terminal Port	UEPSP	UEPLD	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	2-Wire Voice Unbundled 2-Way PBX Tennessee Calling Pc	UEPSP	UEPT2 UEPTO	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
	2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee Calling Pr 2-Wire Vice Unbundled 2-Way PBX Usage Po	UEPSP UEPSP	UEPXA	1.79 1.79	9.93 9.93	9.19 9.19	3.66 3.66	2.92 2.92	1	20.35	10.54 10.54	13.32 13.32	
B.1.7	2-Wire Voice Unbundled PBX Toll Terminal Hotel Por	UEPSP	UEPXB	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
B.1.7	2-Wire Voice Unbundled PBX LD DDD Terminals Po	UEPSP	UEPXC	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	
D. I./								1	1			1	
B.1.7	2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc	UEPSP	UEPXD	1.79	9.93	9.19	3.66	2.92		20.35	10.54	13.32	

2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling B.1.7 Port			UEPSP	UEPXL	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.4
B.1.7 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P			UEPSP	UEPXM	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.4
2-W Voice Unbundled 1-Way Out PBX Hotel/Hospital Economy Administrative Calling	9														
B.1.7 Port TN Calling Por  2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling			UEPSP	UEPXN	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.4
B.1.7 Port			UEPSP	UEPXO	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.4
B.1.7 2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc			UEPSP	UEPXS	1.79	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.4
B.1.7 2-Wire Voice Unbundled PBX Collierville and Memphis Calling Pt B.1.7 2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Calling Pt			UEPSP UEPSP	UEPXV	1.79 1.79	9.93 9.93	9.19 9.19	3.66 3.66	2.92 2.92			20.35	10.54 10.54	13.32 13.32	1.4
,								3.00	2.92			20.33	10.54	13.32	1.4
Subsequent Activity FEATURES			UEPSP	USASC	0	0	0								
All Available Vertical Feature  EXCHANGE PORT RATES (COIN)			UEPSP UEPSE	UEPVF	0	0	0					20.35	10.54	13.32	1.4
Exchange Ports - Coin Por					2.11	9.93	9.19	3.66	2.92			20.35	10.54	13.32	1.4
NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to	o circuit s	witched	l voice and/or circu	it switched	data transmissio	on by B-Channe	ls associated v	vith 2-wire ISI	ON ports.						
NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/I	New Busi	ness Re	equest Process. R	ates for the	packet capabil	ties will be dete	ermined via the	Bona Fide R	equest/New	Business Re	quest Proce	SS.			
DLED LOCAL SWITCHING, PORT USAGE															
End Office Switching (Port Usage)  End Office Switching Function, Per MOI		+		+ -	0.0008041					1					
End Office Switching Function, Fel MOL					0.0008041					1					
Tandem Switching (Port Usage) (Local or Access Tandem)															
Tandem Switching Function Per MOI					0.0009778			1		1					
Common Transport	-	+						1		1					
Common Transport - Per Mile, Per MOL					0.0000064										
Common Transport - Facilities Termination Per MO					0.0003871										
N ED BORTH COR COMPINATIONS COOT BACER BATES															
DLED PORT/LOOP COMBINATIONS - COST BASED RATES															
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission rule to	o provide	Unbun	dled Local Switchi	og or Switc	h Ports										
Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sa End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section	me manr	ier as tr	ney are applied to t	ne Stand-A	Alone Unbundle	Port section of	i this Rate Exh	ipit.							
For Georgia and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently C states, the nonrecurring charges shall be those identified in the Nonrecurring - Currently Combined	sections		Continuity Combin	1	o and the met a		T. T. G. T. G. G. G. G. G. G. G. G. G. G. G. G. G.	, onargoo app	,, 10 1101 00	T					
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
UNIT Death and Combination Dates															
UNE Port/Loop Combination Rates															
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone		1			14.18										
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone		2			18.01										
2-Wire VG Loop/Port Combo - Zone															
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone		2			18.01										
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone		3	UEPRX	UEPLX	18.01 23.02										
2-Wire VG Loop/Port Combo - Zone		3 1 2	UEPRX	UEPLX	18.01 23.02 12.48 16.31										
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone		3			18.01 23.02										
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		3 1 2	UEPRX UEPRX	UEPLX UEPLX	18.01 23.02 12.48 16.31										
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone		3 1 2	UEPRX	UEPLX	18.01 23.02 12.48 16.31	22.14	15.25	8.45	3.91			30.89	7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone:  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone  2-Wire Voice Grade Line Port Rates (Res)		3 1 2	UEPRX UEPRX	UEPLX UEPLX	18.01 23.02 12.48 16.31 21.32	22.14	15.25 15.25	8.45 8.45	3.91			30.89	7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone:  UNE Loop Rates 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop - (SL1) - Zone 2-Wire voice unbundled Loop - residenc 2-Wire voice unbundled port - residenc		3 1 2	UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	18.01 23.02 12.48 16.31 21.32 1.7	22.14	15.25	8.45	3.91			30.89	7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller Loop - In the Cal	er	3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO	18.01 23.02 12.48 16.31 21.32 1.7 1.7	22.14	15.25 15.25	8.45 8.45	3.91 3.91			30.89	7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call ID - re	er	3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO	18.01 23.02 12.48 16.31 21.32 1.7 1.7	22.14 22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91 3.91			30.89 30.89 30.89	7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone:  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller Loop - Incompany - Incom	ər	3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO	18.01 23.02 12.48 16.31 21.32 1.7 1.7	22.14	15.25 15.25	8.45 8.45	3.91 3.91			30.89	7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC	ər	3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRO UEPAQ UEPAH	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7	22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE	ər	3 1 2	UEPRX	UEPRC UEPRO UEPAQ UEPAH UEPAK UEPAL	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2	ər	3 1 2	UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAQ UEPAH UEPAK	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone:  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee Area Plus with Caller ID - res (AC 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE)	ər	3 1 2	UEPRX UEPRX	UEPLX UEPRC UEPRO UEPAO UEPAO UEPAH UEPAK UEPAL UEPAM UEPAN	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE	er	3 1 2	UEPRX UEPRX	UEPLX UEPRC UEPRO UEPAQ UEPAH UEPAK UEPAL UEPAM	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TMF2) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TMF2)	er	3 1 2	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRC UEPRC UEPRO UEPAQ UEPAH UEPAK UEPAL UEPAM UEPAN UEPAN UEPAO	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc  2-Wire voice unbundled port with Caller ID - re  2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS)	ər	3 1 2	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRC UEPRC UEPRO UEPAQ UEPAH UEPAK UEPAL UEPAM UEPAN UEPAN UEPAO	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc  2-Wire voice unbundled port - residenc  2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS)	ər	3 1 2	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRC UEPRC UEPRO UEPAQ UEPAH UEPAK UEPAL UEPAM UEPAN UEPAN UEPAO	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 3-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 3-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 3-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS 3-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 3-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 3-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS)	er	3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAQ UEPAH UEPAK UEPAH UEPAM UEPAN UEPAN UEPAO UEPAP	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03		
2-Wire VG Loop/Port Combo - Zone 2-Wire VG Loop/Port Combo - Zone: 2-Wire VG Loop/Port Combo - Zone: 2-Wire Voice Orade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone 2-Wire voice Grade Loop (SL1) - Zone 2-Wire voice unbundled port - residenc 2-Wire voice unbundled port with Caller ID - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice unbundled port outgoing only - re 2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call ID - res 2-Wire voice unbundled Tennessee Area Plus with Caller ID - res (AC 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS) 2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS)	er	3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPRL UEPRC UEPRO UEPAQ UEPAH UEPAK UEPAH UEPAM UEPAN UEPAN UEPAO UEPAP	18.01 23.02 12.48 16.31 21.32 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7	22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14 22.14	15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25 15.25	8.45 8.45 8.45 8.45 8.45 8.45 8.45 8.45	3.91 3.91 3.91 3.91 3.91 3.91 3.91 3.91			30.89 30.89 30.89 30.89 30.89 30.89 30.89 30.89	7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03 7.03		

	RING CHARGES (NRCs) - CURRENTLY COMBINED		LIEDDY	110400		4.00	0.00				22.22	7.00	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPRX	USAC2		1.03	0.29				30.89	7.03	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		UEPRX	USACC		1.03	0.29				30.89	7.03	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa												
	Update					0.76					7.97		
ADDITIONAL	L NRCs												
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPRX	USAS2	0	0	0						
0 MIDE VOI	OF ORARE LOOP WITH A WIRE LINE PORT (RUG)												
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)												
UNE Port/Lo	pop Combination Rates												
	2-Wire VG Loop/Port Combo - Zone	1			14.18								
	2-Wire VG Loop/Port Combo - Zone : 2-Wire VG Loop/Port Combo - Zone :	3			18.01 23.02								
	2-wire vG Loop/Port Combo - Zone	3			23.02								
UNE Loop R	lates												
	2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	12.48								
	2-Wire Voice Grade Loop (SL1) - Zone 2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX UEPBX	UEPLX	16.31 21.32								
2-Wire Voice	e Grade Line Port (Bus)	3	OLFDX	OLITEX	21.02								
	2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	1.7	22.14	15.25	8.45	3.91		30.89	7.03	
	2-Wire voice uphundled port with Caller + E484 ID by		UEPBX	UEPBC	1.7	22.14	15.25	8.45	3 01		30.89	7.03	
	2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBA	UEPBC	1.7	22.14	15.25	6.45	3.91		30.89	7.03	
	2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	1.7	22.14	15.25	8.45	3.91		30.89	7.03	
	2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Caller		HEDDY	HEDAY	1.7	22.11	45.05	0.45	2.04		20.22	7.00	T
	ID - bus 2-Wire voice unbundled incoming only port with Caller ID - Bı		UEPBX	UEPAV UPEB1	1.7	22.14 22.14	15.25 15.25	8.45 8.45	3.91 3.91		30.89 30.89	7.03 7.03	
	2-Wire voice unbundled Trennessee Bus 2-Way Area Calling Port Economy Option		OLI DX	OI LD1	1.7	22.14	10.20	0.40	0.01		50.05	7.00	
	(TACC1)		UEPBX	UEPAC	1.7	22.14	15.25	8.45	3.91		30.89	7.03	
	2-Wire voice unbundled Tennessee Bus 2-Way Area Calling Port Standard Option (TACC2)		UEPBX	UEPAD	1.7	22.14	15.25	8.45	3.91		30.89	7.03	
	2-Wire voice unbundled Tennessee Bus 2-Way Collierville and Memphis Local Calling		UEFBA	UEFAD	1.7	22.14	15.25	0.40	3.91		30.69	7.03	
	Port (B2F)		UEPBX	UEPAE	1.7	22.14	15.25	8.45	3.91		30.89	7.03	
LOCAL NUM	IBER PORTABILITY												
	Local Number Portability (1 per port		UEPBX	LNPCX	0.35								
FEATURES			(IEDD)/			_							
	All Features Offered		UEPBX	UEPVF	0	0	0				30.89	7.03	
NONRECUR	RING CHARGES (NRCs) - CURRENTLY COMBINED												-
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		UEPBX	USAC2		1.03	0.29				30.89	7.03	
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan		UEPBX	USACC		1.03	0.29						
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa		UEFBA	USACC		1.03	0.29						
	Update					0.76					7.97		
ADDITIONAL	LADO												
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		UEPBX	USAS2							30.89	7.03	
				2 37 102									
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)												
UNE Port/I o	pop Combination Rates												
	2-Wire VG Loop/Port Combo - Zone	1			14.18								
	2-Wire VG Loop/Port Combo - Zone :	2			18.01								
	2-Wire VG Loop/Port Combo - Zone	3			23.02								
	2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPRG	UEPLX	12.48								
	2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPRG	UEPLX	16.31								
	2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPRG	UEPLX	21.32								
Z-WIFE VOICE	e Grade Line Port Rates (RES - PBX)						1						
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	1.7	22.14	15.25	8.45	3.91		30.89	7.03	
	IBER PORTABILITY  Local Number Portability (1 per port		UEPRG	LNPCP	3.5								$\rightarrow$
	Ecoal Hambol Foliability (1 por por		OLI ING	LITTOF	5.5								-
FEATURES													
	All Features Offered		UEPRG	UEPVF	0	0	0				30.89	7.03	
NONRECUR	RING CHARGES (NRCs) - CURRENTLY COMBINED												
				_			1						
				1									

2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with												T
Change		UEPRG	USACC		1.03	0.29			30.89	7.03		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa												T
Update					0.76				7.97			<u> </u>
												₩
ADDITIONAL NRCs  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPRG	USAS2	0	•							+-
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi		UEPRG	USAS2	0	0 14.64	0 14.64			19.99	19.99	19.99	
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	-				14.64	14.64			19.99	19.99	19.99	+
Z-MINE FOIGE GRADE EGG! MINIZ-MINE EINE FORT (BGG - FBA)												+
UNE Port/Loop Combination Rates												+
2-Wire VG Loop/Port Combo - Zone	1			14.18								+
2-Wire VG Loop/Port Combo - Zone	2			18.01								T
2-Wire VG Loop/Port Combo - Zone	3			23.02								
UNE Loop Rates												4
2-Wire Voice Grade Loop (SL 1) - Zone	1	UEPPX	UEPLX	12.48								
2-Wire Voice Grade Loop (SL 1) - Zone	2	UEPPX	UEPLX	16.31								+-
2-Wire Voice Grade Loop (SL 1) - Zone	3	UEPPX	UEPLX	21.32								+
2-Wire Voice Grade Line Port Rates (BUS - PBX)												+
2-Wile Voice Glade Lille Fort Rates (DOS - FDX)												+
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı		UEPPX	UEPPC	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
Line Side Oribundled Outward PBX Trunk Port - Bu  Line Side Unbundled Incoming PBX Trunk Port - Bu	-	UEPPX	UEPP0	1.7	22.14	15.25	8.45	3.91	30.89	7.03		+
2-Wire Voice Unbundled PBX LD Terminal Port	-+	UEPPX	UEPLD	1.7	22.14	15.25	8.45	3.91	30.89	7.03		+
2-Wire Voice Unbundled 2-Way Combination PBX Tennessee Calling P	-	UEPPX	UEPT2	1.7	22.14	15.25	8.45	3.91	30.89	7.03		$^{+}$
2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee Calling Po		UEPPX	UEPTO	1.7	22.14	15.25	8.45	3.91	30.89	7.03		T
2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	1.7	22.14	15.25	8.45	3.91	30.89	7.03		T
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
		-										+
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	1.7	22.14	15.25	8.45	3.91	30.89	7.03		4
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Pc		UEPPX	UEPXE	1.7	22.14	15.25	8.45	3.91	30.89	7.03		+-
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		UEPPX	UEPXL	1.7	22.14	15.25	8.45	2.04	30.89	7.03		
Port		UEPPX	UEPXL	1.7	22.14	15.25	8.45	3.91	30.89	7.03		+-
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX	UEPXM	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
2-Wire Voice Unbundled 1W Out PBX Hotel/Hospital Economy Administrative Calling		OLITA	OLI XIVI	1.7	22.14	13.23	0.45	5.51	30.03	7.03		+
Port TN Calling Por		UEPPX	UEPXN	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling												T
Port		UEPPX	UEPXO	1.7	22.14	15.25	8.45		30.89	7.03		
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX	UEPXS	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
2-Wire Voice Unbundled PBX Collierville and Memphis Calling Po		UEPPX	UEPXU	1.7	22.14	15.25	8.45	3.91	30.89	7.03		4
2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Callling Pi		UEPPX	UEPXV	1.7	22.14	15.25	8.45	3.91	30.89	7.03		+-
LOCAL NUMBER PORTABILITY												┿
Local Number Portability (1 per port		UEPPX	LNPCP	3.15								+
Local Number Fortability (1 per pon		UEFFX	LINECE	3.13								+
FEATURES												+
All Features Offered	-	UEPPX	UEPVF	0	0	0			30.89	7.03		$^{+}$
						_			20.00			T
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED												
O Wiss Visits Conda Lang/Line Bod Conditions (BDV)		HEDDY	USAC2		4.00	0.00			20.55	7.00	-	
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with		UEPPX	USAC2		1.03	0.29	1		30.89	7.03		+
Change		UEPPX	USACC		1.03	0.29			30.89	7.03		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Databa												
Update					0.76				7.97			4
ADDITIONAL NRCs							-		+ +			+
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activ		UEPPX	USAS2	0	0	0						I
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64	l		19.99	19.99	19.99	¥Ξ
UNE Port/Loop Combination Rates												+
				14.18								T
2-Wire VG Coin Port/Loop Combo – Zone 1				18.01								T
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2				00.00								I
2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3				23.02					1 1 -	1		1 -
2-Wire VG Coin Port/Loop Combo – Zone 1   2-Wire VG Coin Port/Loop Combo – Zone 2   2-Wire VG Coin Port/Loop Combo – Zone 3   UNE Loop Rates												-
2-Wire VG Coin Port/Loop Combo – Zone 1   2-Wire VG Coin Port/Loop Combo – Zone 2   2-Wire VG Coin Port/Loop Combo – Zone 3   UNE Loop Rates   2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	12.48								I
2-Wire VG Coin Port/Loop Combo – Zone 1     2-Wire VG Coin Port/Loop Combo – Zone 2     2-Wire VG Coin Port/Loop Combo – Zone 3     2-Wire VG Coin Port/Loop Combo – Zone 3     2-Wire Voice Grade Loop (SL1) - Zone     2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	12.48 16.31								F
2-Wire VG Coin Port/Loop Combo - Zone 1				12.48								
2-Wire VG Coin Port/Loop Combo – Zone 1		UEPCO UEPCO	UEPLX	12.48 16.31 21.32	22.44	45.05	0.45	2.04	20.00	7.00		
2-Wire VG Coin Port/Loop Combo – Zone 1		UEPCO	UEPLX	12.48 16.31	22.14	15.25	8.45	3.91	30.89	7.03		

		-	UEPCO	) UEPTA	1.7	22.14	15.25	8.45	3.91	30.89	7.03		$\overline{}$
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (TN) 2-Wire Coin 2-Way with Operator Screening: 900 Blocking: 900/976, 1+DDD, 011+, and	d	$\rightarrow$	UEFCO	UEPTA	1.7	22.14	10.20	0.40	3.81	30.89	1.03		+
Local (NC, TN)			UEPCO	UEPCA	1.7	22.14	15.25	8.45	3.91	30.89	7.03		
2-Wire Coin Outward with Operator Screening and 011 Blocking (TN)			UEPCO	UEPTC	1.7	22.14	15.25	8.45	3.91	30.89	7.03		Т
2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+,												[	
and Local (TN)		$\rightarrow$	UEPCO		1.7	22.14	15.25	8.45	3.91	30.89	7.03	<b></b>	4
2-Wire 2-Way Smartline with 900/976 (all states except LA)		$\rightarrow$	UEPCO							30.89	7.03	<del>                                     </del>	+
2-Wire Coin Outward Smartline with 900/976 (all states except LA)  ADDITIONAL UNE COIN PORT/LOOP (RC)		$\rightarrow$	UEPCO	) UEPCR	1.88					30.89	7.03		+
ADDITIONAL ONE COIN FORT/LOOF (NO)		$\dashv$		-									+
UNE Coin Port/Loop Combo Usage (Flat Rate			UEPCO	URECU	3.45	0	0					[	
Local Number Portability (1 per port			UEPCO		0.35								T
													T
FEATURES													
All Features Offerec			UEPCO	) UEPVF	0	C	0			20.35	10.54	13.32	
		$\rightarrow$										<b></b>	4
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as		$\rightarrow$	UEPCO	USAC2		1.03	0.29			30.89	7.03	<b></b>	+
2 Wire Voice Crade Lean / Line Bort Combination Conversion Switch with abort			UEPCO	USACC		1.03	0.20			20.00	7.02		
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with chan 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activ		$\dashv$	UEPCO	USAS2		0	0.29			30.89 30.89	7.03 7.03		+
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT		$\rightarrow$	02100	00/102			Ü			50.03	7.00		+
													+
UNE Port/Loop Combination Rates		$\neg$											$\top$
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		1			18.38								$\top$
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		2			19.87								I
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone		3			24.78								┸
												<b></b>	
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone		1	UEPPX									<del></del>	+
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone		2	UEPPX UEPPX										+
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone Exchange Ports - 2-Wire DID Por	-+	3	UEPPX		16 8.78	45.44	29.94	8.45	3.91	19.99	19.99	19.99	+
NONRECURRING CHARGES - CURRENTLY COMBINED		$\dashv$	UEFFX	UEFDI	0.70	45.44	29.94	0.40	3.91	19.99	19.99	19.99	+
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-		$\rightarrow$	UEPPX	USAC1		8.76	5.75			19.99	19.99	19.99	+
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable	e	-+	OLITA	- 00/101		0.70	3.73			10.00	15.55	15.55	+
Changes			UEPPX	USA1C		8.76	5.75			19.99	19.99	19.99	_
Telephone Number/Trunk Group Establisment Charges		_										<u> </u>	Ŧ
DID Trunk Termination (One Per Port		$\neg$	UEPPX	NDT	0	0	0			19.99	19.99	19.99	+
Additional DID Numbers for each Group of 20 DID Numbe		$\neg$	UEPPX		0	0	0			19.99	19.99	19.99	+
DID Numbers, Non- consecutive DID Numbers , Per Numbe			UEPPX	ND5	0	0	0						T
Reserve Non-Consecutive DID number			UEPPX	ND6	0	0	0						
Reserve DID Numbers		$\rightarrow$	UEPPX	( NDV	0	0	0					<b></b>	4
LOCAL NUMBER PORTARILITY		$\rightarrow$										<del>                                     </del>	+
LOCAL NUMBER PORTABILITY		$\rightarrow$	LIEDDY	LNDCD	2.45							+	+
Local Number Portability (1 per port		$\rightarrow$	UEPPX	LNPCP	3.15								+
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT		$\dashv$											+
2-WIKE ISDN DIGITAL GRADE LOOF WITH 2-WIKE ISDN DIGITAL LINE SIDE FORT		$\dashv$											+
UNE Port/Loop Combination Rates		-+											+
	-	$\dashv$	UEPPB							1			+
												Į.	
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		1 1	UEPPR		32.27								
	-+	1		2									-
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		1 2 l			32.27 34.78								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone			UEPPB U	JEPPR	34.78								+
			UEPPB U	2									_
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone			UEPPB U	JEPPR	34.78								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		3 (	UEPPB U	JEPPR JEPPR	34.78 44.32								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		3 (	UEPPB U	JEPPR	34.78								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		3 L	UEPPB UI	JEPPR JEPPR	34.78 44.32 16.2								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone		3 L	UEPPB UI	JEPPR JEPPR	34.78 44.32								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone		3 L 1 L 2 L 3 L	UEPPB UI UEPPB UI UEPPB UI UEPPB UI	JEPPR JEPPR USL2X JEPPR USL2X JEPPR USL2X	34.78 44.32 16.2 18.71 28.25								
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  Exchange Port - 2-Wire ISDN Line Side Po		3 L 1 L 2 L 3 L	UEPPB UI UEPPB UI UEPPB UI UEPPB UI	JEPPR JEPPR	34.78 44.32 16.2 18.71 28.25	141.75	118.37	49.2	43.26	19.99	19.99	19.99	
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  Exchange Port - 2-Wire ISDN Line Side Po NONRECURRING CHARGES - CURRENTLY COMBINED		3 L 1 L 2 L 3 L	UEPPB UI UEPPB UI UEPPB UI UEPPB UI	JEPPR JEPPR USL2X JEPPR USL2X JEPPR USL2X	34.78 44.32 16.2 18.71 28.25	141.75	118.37	49.2	43.26	19.99	19.99	19.99	
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  Exchange Port - 2-Wire ISDN Line Side Po  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination -		3 L 1 L 2 L 3 L	UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UE	JEPPR USL2X JEPPR USL2X JEPPR USL2X JEPPR USL2X EPPR USL2X	34.78 44.32 16.2 18.71 28.25 16.07			49.2	43.26				
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  Exchange Port - 2-Wire ISDN Line Side Po NONRECURRING CHARGES - CURRENTLY COMBINED		3 L 1 L 2 L 3 L	UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UE	JEPPR JEPPR USL2X JEPPR USL2X JEPPR USL2X	34.78 44.32 16.2 18.71 28.25 16.07	141.75	118.37	49.2	43.26	19.99	19.99	19.99	
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone Exchange Port - 2-Wire ISDN Line Side Port Combination - Conversion		3 L 1 L 2 L 3 L	UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UE	JEPPR USL2X JEPPR USL2X JEPPR USL2X JEPPR USL2X EPPR USL2X	34.78 44.32 16.2 18.71 28.25 16.07			49.2	43.26				
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  Exchange Port - 2-Wire ISDN Line Side Po  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion  ADDITIONAL NRCS		3 L 1 L 2 L 3 L	UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UE	JEPPR USL2X JEPPR USL2X JEPPR USL2X JEPPR USL2X EPPR USL2X	34.78 44.32 16.2 18.71 28.25 16.07			49.2	43.26				
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone Exchange Port - 2-Wire ISDN Line Side Port Combination - Conversion		3 1 2 1 3 1	UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UE	JEPPR USL2X JEPPR USL2X JEPPR USL2X JEPPR USL2X EPPR USL2X	34.78 44.32 16.2 18.71 28.25 16.07			49.2	43.26				
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  Exchange Port - 2-Wire ISDN Line Side Po  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion  ADDITIONAL NRCs  2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy - Non Feature/Add Trunk		3 1 2 1 3 1	UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UE	EPPR USL2X JEPPR USL2X JEPPR USL2X JEPPR USL2X EPPR USPPB EPPR USACB	34.78 44.32 16.2 18.71 28.25 16.07	117.23		49.2	43.26	19.99	19.99	19.99	
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  2-Wire ISDN Digital Grade Loop - UNE Zone  Exchange Port - 2-Wire ISDN Line Side Po  NORECURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion  ADDITIONAL NRCs  [2-Wire ISDN Loop / 2-Wire ISDN Port Combination - Sub Actvy - Non Feature/Add]		3 1 2 1 3 1	UEPPB UI UEPPB UI UEPPB UI UEPPB UI UEPPB UE UEPPB UE	EPPR USL2X JEPPR USL2X JEPPR USL2X JEPPR USL2X EPPR USPPB EPPR USACB	34.78 44.32 16.2 18.71 28.25 16.07	117.23		49.2	43.26	19.99	19.99	19.99	

							1	1	I					i .	T
CVS/CSD (DMS/5ESS)			UEPPR U		0	0	0			İ				I	
CVS (EWSD)			UEPPR U			0	0			I					T
CSD		UEPPB	UEPPR L	U1UCC	0	0	0			i					
										ı					
B-CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)								T							
CVS/CSD (DMS/5ESS)		UEPPB	UEPPR U	U1UCD	0	0	0								
					I					I		1			
CVS (EWSD)		UEPPB	UEPPR L	U1UCE	0	0	0			<b>.</b>					
					I					İ				I	
CSD		UEPPB	UEPPR I	U1UCF	0	0	0	+	$\vdash$			<b> </b>	<del></del>		+
USER TERMINAL PROFILE															
User Terminal Profile (EWSD only)		UEPPB	UEPPR L	U1UMA	0	0	0							<u> </u>	
VERTICAL FEATURES		┼			<del>                                     </del>		<del> </del>	+	$\vdash$			<b> </b>	<del> </del>		
All Vertical Features - One per Channel B User Profile		UEPPB	UEPPR L	JEPVF	0	0	0								+
Interoffice Channel mileage each, including first mile and facilities termination		UEPPB	UEPPR I	M1GNC	17.91	53.99	17.37					19.99	19.99	19.99	9
Interoffice Channel mileage each, additional mile		UEPPB	UEPPR N	M1GNM	0.173	0	0				0				
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT															
INF Port/Loop Combination Pates	-+	+	$\longrightarrow$				<del> </del>	+	$\vdash$		<del>                                     </del>	<b> </b>	$\vdash$	<u>'</u>	+
UNE Port/Loop Combination Rates  4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	1	UEPF	DD	$\longrightarrow$	132.58		+	+			+	<b></b>			+
4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone  4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	2	UEPF			150.25	-	+	+	<b>—</b>		+	<u> </u>	$\vdash$		+
4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone	3				173.44	<del>                                     </del>	+	+					$\overline{}$		+
		1	-	-			+	+		i	<u> </u>			I	+
4-Wire DS1 Digital Loop - UNE Zone	1	UEPF	PP	USL4P	57.73										
4-Wire DS1 Digital Loop - UNE Zone	2	UEPF	PP I	USL4P	75.4					i				i	
4-Wire DS1 Digital Loop - UNE Zone	3	UEPF	PP I	USL4P	98.59										
Exchange Ports - 4-Wire ISDN DS1 Por		UEPF	PP I	UEPPP	74.85	415.53	366.9	89.28	77.43	<b>.</b>		19.99	19.99	19.99	
NONRECURRING CHARGES - CURRENTLY COMBINED					<b></b>					<b></b>					4
4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -		LIED			0	000 50	000 50			I		40.00	40.00	10.00	
Conversion -Switch-as-is		UEPF	PP (	USACP		328.53	328.53	+				19.99	19.99	19.99	+
ADDITIONAL NRCs		+					+	+			+	<b>—</b>			+
4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within Std Allowance		UEPF	PP	PR7TF		0.94		-				19.99	19.99	19.99	
4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All															
States except NC		UEPF	PP F	PR7TO	<b>+</b>	22.36	22.36					19.99	19.99	19.99	
4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowance		UEPF	'PP	PR7ZT	1	44.71	44.7			l	1	19.99	19.99	19.99	
LOCAL NUMBER PORTABILITY		+					+	+			+	<b>—</b>			+
Local Number Portability (1 per port		UEPF	PP	LNPCN	1.75		+	+							+
Local Hamber Fortability (1 per por		- OLI I		-141 014	1.70		+	+			<del>                                     </del>				+
INTERFACE (Provsioning Only)					1		1	1							T
Voice/Data		UEPF	PP	PR71V	0	0	0			i				i	
Digital Data		UEPF		PR71D		0	0								1
Inward Data		UEPF	PP !	PR71E	0	0	0								
					<b></b>	<u> </u>	<b>_</b>		oxdot	<b></b>	<b> </b>	ļ	<b></b>	·	4
New or Additional "B" Channel	_		100	DDTS		00.00	<b>_</b>	4			<b></b>	40.00	10.55	· · · · · · · · ·	_
New or Additional - Voice/Data B Channel		UEPF		PR7BV	0	28.39	+	4			<b></b> '	19.99	19.99	19.99	
New or Additional - Digital Data B Channel	$-\!+\!-$	UEPF		PR7BF	0	29.11	+	+		<b></b>	<del></del> '	19.99	19.99	19.99	
New or Additional Inward Data B Channel	-+	UEPF		PR7BD		29.39	+	+			<del>                                     </del>	19.99	19.99	19.99	
New or Additional Useage Sensitive Voice Data B Channel	-+	UEPF		PR7BS PR7BU		28.39 28.39	-	+			+	19.99 19.99	19.99	19.99	
New or Additional Useage Sensitive Digital Data B Channel	-+	UEPI		- KIBU	0	20.39	+	+	$\vdash$		+	19.99	19.99	19.99	2
CALL TYPES	-+	+	+				+	+			+				+
Inward		UEPF	PP	PR7C1	0	0	0	1			<b>†</b>				+
Outward	-	UEPF	PP	PR7C0	0	0	0	+			$\vdash$				+
Two-way		UEPF	PP I	PR7CC	0	0	0								I
Interoffice Channel Mileage		+	$\longrightarrow$			<del>                                     </del>	+	+	<b></b>		<del>                                     </del>	<u> </u>			+
Fixed Each Including First Mile		UEPF	'PP	1LN1A	76.1825	145.98	109.85	19.55				19.99	19.99	19.99	9
Each Airline-Fractional Additional Mil		UEPF		1LN1B	0.3525	5.55		1.5.55		i				10.00	1
															T
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT	$-\!$		-	-				+	$\vdash$		<del></del>		$\vdash$		F
		+		$\longrightarrow$		+	+	+			+				+
UNE Port/Loop Combination Rates					I					ļ.				'	

	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone		UEPDC	-	110.95		+	l			19.99	19.99	19.99	_
4	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone	3 L	UEPDC		134.14						19.99	19.99	19.99	
	4-Wire DS1 Digital Loop - UNE Zone		UEPDC	USLDC	57.53									
	4-Wire DS1 Digital Loop - UNE Zone		UEPDC	USLDC	75.4				40.45					
	4-Wire DS1 Digital Loop - UNE Zone		UEPDC	USLDC	98.59									
-	4-Wire DDITS Digital Trunk Por	L	UEPDC	UDD1T	35.55	342.8	257.87	61.41	48.49		19.99	19.99	19.99	
	RING CHARGES - CURRENTLY COMBINED													
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-	L	UEPDC	USAC4		312.91	312.91				19.99	19.99	19.99	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with													
l lr	DS1 Changes	L	UEPDC	USAWA		312.91	312.91				19.99	19.99	19.99	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with													
(	Change - Trunk	L	UEPDC	USAWB		312.91	312.91				19.99	19.99	19.99	
ADDITIONAL	- NRCs													
á	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Service Activity Per Service													
(	Order	ι	UEPDC	USAS4		94.88	94.88							
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel													
	Activation/Chan - 2-Way Trunl	ı	UEPDC	UDTTA		108.67	108.67				19.99	19.99	19.99	
,	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Chan - 1-													
	Way Outward Trunk	1	UEPDC	UDTTB		108.67	108.67				19.99	19.99	19.99	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan										. 3.00		. 5.00	7
	Inward Trunk w/out DIC	1	UEPDC	UDTTC		108.67	108.67				19.99	19.99	19.99	
L L	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -			55110		. 55.01	. 50.07			1	. 5.55	.0.00	. 5.55	7
	Inward Trunk with DIC		UEPDC	UDTTD		108.67	108.67				19.99	19.99	19.99	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-Way		021 00	ODITO		100.07	100.07				10.00	10.00	10.00	-
	DID w User Trans		UEPDC	UDTTE		108.67	108.67				19.99	19.99	19.99	
RIPOLAR 9 7	ZERO SUBSTITUTION		OLI DO	ODITE		100.07	100.07			1	13.33	13.33	13.33	-
DIF OLAR 02	zero sobstitution													$\dashv$
ļ ,	B8ZS -Superframe Format		UEPDC	CCOSF		0	590				19.99	19.99	19.99	
	DOZO -Oupername i onfldt		OLFDC	CCOSF		U	390				19.99	19.99	19.99	-
	B8ZS - Extended Superframe Forma		UEPDC	CCOEF		0	590				19.99	19.99	19.99	
	DOZO - Extended Superifatile Follifa		OLFDC	CCOEF		U	390				19.99	19.99	19.99	-
Alternate Mai	ark Inversion													
Alternate Mai	III III III III III III III III III II													-
	AMI Considered Format		UEPDC	MCOSF		0	0							
/	AMI -Superframe Format		UEPDC	WCOSF		U	U							
	AND Francisco Francisco		LIEBBO	140000		0								
	AMI - Extended SuperFrame Forma		UEPDC	МСОРО		0	0							
<u> </u>														
	lumber/Trunk Group Establisment Charges				_									
	Telephone Number for 2-Way Trunk Grou		UEPDC	UDTGX	0									
	Telephone Number for 1-Way Outward Trunk Grou		UEPDC	UDTGY	0					19.99				_
	Telephone Number for 1-Way Inward Trunk Group Without DI		UEPDC	UDTGZ	0					19.99				_
	DID Numbers for each Group of 20 DID Number		UEPDC	ND4	0					19.99				_
	DID Numbers, Non- consecutive DID Numbers , Per Numbe		UEPDC	ND5	0					19.99				_
	Reserve Non-Consecutive DID Nos		UEPDC	ND6	0	0	0							_
	Reserve DID Numbers	ι	UEPDC	NDV	0	0	0			19.99				_
		L												_
	S1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDITS Tru													
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination		UEPDC	1LNO1	75.83	145.98	109.85	19.66	14.99		19.99	19.99	19.99	_
الحصي	Interoffice Channel Mileage - Additional rate per mile - 0-8 mil	L	UEPDC	1LNOA	0.3525	0	0				1			Ц
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination		UEPDC	1LNO2	0	0	0				ļ		1	_
	Interoffice Channel Mileage - Additional rate per mile - 9-25 mil		UEPDC	1LNOB	0.3525	0	0			_				
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination		UEPDC	1LNO3	0	0	0	0			1			_
	Interoffice Channel Mileage - Additional rate per mile - 25+ mil		UEPDC	1LNOC	0.3525	0	0				ļ		1	_
	Local Number Portability, per DS0 Activate		UEPDC	LNPCP	3.15	0	0	0		_				
			UEPDC	CTG	0					_				
	Central Office Termininating Poir	L	UEFDC	010	•			l		_				
	Central Office Termininating Poir		UEFDC	010	Ŭ				1	1			1	
			OEFDC	010									1	П
4-WIRE DS1	LOOP WITH CHANNELIZATION WITH PORT		OEFDC	010										_
4-WIRE DS1 System is 1 [	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations		UEFDC											_
4-WIRE DS1 System is 1 I	LOOP WITH CHANNELIZATION WITH PORT		UEFDC											
4-WIRE DS1 System is 1 [	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations		OEFDC	010										
4-WIRE DS1 System is 1 I Each System	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations n can have up to 24 combinations of rates depending on type and number of ports used		OEFDC											
4-WIRE DS1 System is 1 I Each System UNE DS1 Loc	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations n can have up to 24 combinations of rates depending on type and number of ports used op	1 UEPMO				0	0							
4-WIRE DS1 System is 1 I Each System UNE DS1 Loc	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations in can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1	1 UEPMO	lG	USLDC	57.73	0	0							
4-WIRE DS1 System is 1 I Each System UNE DS1 Loc	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations In can have up to 24 combinations of rates depending on type and number of ports used  op 4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2	1 UEPMO	IG IG	USLDC	57.73 75.4	0	0 0							
4-WIRE DS1 System is 1 I Each System UNE DS1 Loc	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations in can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1	1 UEPMO	IG IG	USLDC	57.73 75.4	0 0 0	0 0 0							
4-WIRE DS1 System is 1 Each System UNE DS1 Loc	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations n can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3	1 UEPMO	IG IG	USLDC	57.73 75.4	0 0 0	0 0 0							
4-WIRE DS1 System is 1 I Each System UNE DS1 Loc	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations In can have up to 24 combinations of rates depending on type and number of ports used  op  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  nannelization Capacities (D4 Channel Bank Configurations)	1 UEPMC 2 UEPMC 3 UEPMC	IG G G	USLDC USLDC USLDC	57.73 75.4 98.59		0 0 0							
4-WIRE DS1 System is 1 IE Each System UNE DS1 Loc  UNE DS0 Ch	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations In can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3 Inannelization Capacities (D4 Channel Bank Configurations) 24 DSO Channel Capacity - 1 per DS1	1 UEPMC 2 UEPMC 3 UEPMC	IG IG IG	USLDC USLDC USLDC	57.73 75.4 98.59	0	0							
4-WIRE DS1 System is 1 IE Each System UNE DS1 Loc  UNE DS0 Ch	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations n can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3 nannelization Capacities (D4 Channel Bank Configurations) 24 DS0 Channel Capacity - 1 per DS1 48 DS0 Channel Capacity - 1 per DS1 48 DS0 Channel Capacity - 1 per DS1 48 DS0 Channel Capacity - 1 per DS1	1 UEPMC 2 UEPMC 3 UEPMC UEPMC	IG G G	USLDC USLDC USLDC USLDC VUM24 VUM48	57.73 75.4 98.59 131.87 263.74	0	0 0 0							
4-WIRE DS1 System is 1 I Each System UNE DS1 Loc  UNE DS0 Ch	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations In can have up to 24 combinations of rates depending on type and number of ports used  op  4-Wire DS1 Loop - UNE Zone 1  4-Wire DS1 Loop - UNE Zone 2  4-Wire DS1 Loop - UNE Zone 3  nannelization Capacities (D4 Channel Bank Configurations) 24 DS0 Channel Capacity - 1 per DS1 48 DS0 Channel Capacity - 1 per 2 DS1s 96 DSO Channel Capacity - 1 per 4 DS1s	1 UEPMC 2 UEPMC 3 UEPMC UEPMC UEPMC	IG G IG IG	USLDC USLDC USLDC VUM24 VUM48 VUM96	57.73 75.4 98.59 131.87 263.74 527.48	0	0							
4-WIRE DS1 System is 1 I Each System UNE DS1 Loo	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations n can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3 Annnelization Capacities (D4 Channel Bank Configurations) 24 DSO Channel Capacity - 1 per DS1 48 DSO Channel Capacity - 1 per 2 DS1s 96 DSO Channel Capacity - 1 per 4 DS1s 44 DSO Channel Capacity - 1 per 5 DS1s	1 UEPMC 2 UEPMC 3 UEPMC UEPMC UEPMC UEPMC UEPMC	G G G G G G	USLDC USLDC USLDC VUM24 VUM48 VUM96 VUM14	57.73 75.4 98.59 131.87 263.74 527.48 791.42	0	0							
4-WIRE DS1 System is 1 IE Each System UNE DS1 Loc  UNE DS0 Ch	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations n can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3 nannelization Capacities (D4 Channel Bank Configurations) 24 DS0 Channel Capacity - 1 per DS1 48 DS0 Channel Capacity - 1 per SD1s 96 DS0 Channel Capacity - 1 per 4 DS1s 144 DS0 Channel Capacity - 1 per 6 DS1s 149 DS0 Channel Capacity - 1 per 6 DS1s 149 DS0 Channel Capacity - 1 per 6 DS1s	1 UEPMC 2 UEPMC 3 UEPMC UEPMC UEPMC UEPMC UEPMC	G G G G G G G	USLDC USLDC USLDC VUM24 VUM48 VUM96 VUM914 VUM19	57.73 75.4 98.59 131.87 263.74 527.48 791.42 827.76	0	0							
4-WIRE DS1 System is 1 IE Each System UNE DS1 Loc  UNE DS0 Ch	LOOP WITH CHANNELIZATION WITH PORT DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations n can have up to 24 combinations of rates depending on type and number of ports used op 4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2 4-Wire DS1 Loop - UNE Zone 3 Annnelization Capacities (D4 Channel Bank Configurations) 24 DSO Channel Capacity - 1 per DS1 48 DSO Channel Capacity - 1 per 2 DS1s 96 DSO Channel Capacity - 1 per 4 DS1s 44 DSO Channel Capacity - 1 per 5 DS1s	1 UEPMC 2 UEPMC 3 UEPMC UEPMC UEPMC UEPMC UEPMC	G G G G G G G	USLDC USLDC USLDC VUM24 VUM48 VUM96 VUM14	57.73 75.4 98.59 131.87 263.74 527.48 791.42 827.76	0 0 0 0 0	0 0 0 0 0							

	100,000		LIEBLIO	1.0.0.00	1.	1.							
	384 DS0 Channel Capacity - 1 per 16 DS1s		UEPMG	VUM38 2109.92	0	0							
	480 DS0 Channel Capacity - 1 per 20 DS1s		UEPMG	VUM40 2637.4	0	0							
	576 DS0 Channel Capacity -1 per 24 DS1s		UEPMG	VUM57 3164.88	0	0							
_	672 DS0 Channel Capacity - 1 per 28 DS1s	1	UEPMG	VUM67 3692.36	0	0					_	1	
	or 2 Boo Granner Gapacity 1 per 20 Bors	+	OEI MO	V 0 1 1 0 0 3 2 . 3 0	U	0	-			-	+	+	
		Ļ	<del></del>										
	ırring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port -			stem									
A Minimum	m System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DSO	Ports with	n Feature Activations.										
	of this configuration functioning as one are considered Add'l after the minimum syste			1								_	
munipies	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes		UEPMG	USAC4 0	303.61	15.74				40.00	40.00	40.00	40.00
					303.61	15.74				19.99	19.99	19.99	19.99
System Ad	dditions at End User Locations Where 4-Wire DS1 Loop with Channelization with Port	ι Combina	ation Currently Exists and	i									
New (Not C	Currently Combined) In Georgia & Tennessee Only												
	NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature Activation	n -		1							_	_	
	New GA & TN Only	T	UEPMG	VUMD4 0	704.68	441.48	138.36	16.41		19.99	19.99	19.99	19.99
			UEFING	VOIVID4 0	704.00	441.40	130.30	10.41		19.99	19.99	19.99	19.99
Bipolar 8 Z	Zero Substitution												
	Clear Channel Capability Format, superframe - Subsequent Activity Only		UEPMG	CCOSF 0	0	590				19.99	19.99	19.99	19.99
	ordar ordanio oupubliky romat, oupomano oupubliky only	+	020	0000.	0	000				13.33	10.00	10.00	13.33
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only		UEPMG	CCOEF 0	0	590				19.99	19.99	19.99	19.99
Alternate N	Mark Inversion (AMI)												
		_		1							_	_	
	Considerate Format		LIEDMO	MCOCE									
	Superframe Format		UEPMG	MCOSF 0	U	U						+	
	Extended Superframe Format		UEPMG	MCOPO 0	0	0	1			1		1	
		T										1	
Evolures	e Ports Associated with 4-Wire DS1 Loop with Channelization with Port	+	<del>                                     </del>	+ +	1	1				1	-	+	_
		+	+	+	-	+				+	-	+	_
Exchange	PORTS	4	<del>                                     </del>				_			1			
		1	T							1			
	Line Side Combination Channelized PBX Trunk Port - Business		UEPPX	UEPCX 1.79	0	0	0	0	19.99	1		1	
-		+	+ + + · · · · · · · · · · · · · · ·	1 2 2 1 1 1 7 9	1	+-		-	. 5.55	1	-	+	_
	List Oile Oile Oile and Ole and Free L Brown Brown		UEPPX	UEDOV 1						1			
	Line Side Outward Channelized PBX Trunk Port - Business	<del></del>		UEPOX 1.79	U	U	0	U	19.99				
	Line Side Inward Only Channelized PBX Trunk Port without DID		UEPPX	UEP1X 1.79	0	0	0	0	19.99				
	·	1	<del>                                     </del>	1								_	
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port		UEPPX	UEPDM 8.97	0	0	0	0	19.99				
			UEFFX	UEFDINI 8.97	U	U	U	U	19.99				
Feature Ac	ctivations - Unbundled Loop Concentration												
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX	1PQWM 0.66	23.94	12.64	3.82	3.8		19.99	19.99	19.99	19.99
		+	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		20.01	12.01	0.02	0.0		10.00	-10.00	10.00	10.00
			UEDDV										
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		UEPPX	1PQWU 0.66	73.67	17.37	54.09	10.57		19.99	19.99	19.99	19.99
Telephone	e Number/ Group Establishment Charges for DID Service												
	DID Trunk Termination (1 per Port)	+	UEPPX	NDT 0							_	-	
	DID Numbers - groups of 20 - Valid all States	+	UEPPX		0						-	+	
				110		0							
	Non-Consecutive DID Numbers - per number		UEPPX	ND5 0	0	0			19.99				
	Reserve Non-Consecutive DID Numbers	T	UEPPX	ND6 0	0	0						1	
	Reserve DID Numbers	+	UEPPX	NDV 0	0	0					_	+	
<del></del>		+	OLITA	INDV 0	U	U					-		
Local Num	mber Portability												
	Local Number Portability - 1 per port		UEPPX	LNPCP 3.15	0	0							
FEATURES	S - Vertical and Optional	1	<del>                                     </del>	1								_	
		+	+ +	+								+	-
Local Switch	itching Features Offered with Line Side Ports Only												
	All Features Available		UEPPX	UEPVF 0	0	0			19.99				
				T									
		<del>                                     </del>		+ + +		+						+	
		+	<b>+</b>								_		
)LED PORT LO	OOP COMBINATIONS - MARKET RATES											$\pm$	
OLED PORT LO	OOP COMBINATIONS - MARKET RATES											+	
DLED PORT LO	OOP COMBINATIONS - MARKET RATES												
		witch port	s per ECC and/or State Co	ymmission rules									
Market Rate	ates shall apply where BellSouth is not required to provide unbundled local switching or s	witch port	s per FCC and/or State Co	ommission rules.									
Market Rate These scen	ates shall apply where BellSouth is not required to provide unbundled local switching or senaios include:												
Market Rate These scen 1. Unbund	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state	es except	as noted for Georgia and	Tennessee.									
Market Rate These scen 1. Unbund	ates shall apply where BellSouth is not required to provide unbundled local switching or senaios include:	es except	as noted for Georgia and	Tennessee.	s with 4 or more	DS0 equivale	ent lines.						
Market Rate These scen 1. Unbund 2. Unbund	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include: died port/loop combinations that are Not Currently Combined in all of the BellSouth state died port/loop combinations that are Currently Combined or Not Currently Combined in 2	es except Zone 1 of	as noted for Georgia and The Top 8 MSAS in BellSo	Tennessee. buth's region for end user									
Market Rate These scen 1. Unbund 2. Unbund The Top 8 I	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include: died porviloop combinations that are Not Currently Combined in all of the BellSouth state died porviloop combinations that are Currently Combined or Not Currently Combined in 2 B MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (	es except Zone 1 of (New Orle	as noted for Georgia and the Top 8 MSAS in BellSorans); NC (Greensboro-Wir	Tennessee. buth's region for end user nston Salem-Highpoint/C	harlotte-Gastoni	a-Rock Hill);	ΓN (Nashville)						
Market Rate These scen 1. Unbund 2. Unbund The Top 8 I	ates shall apply where BellSouth is not required to provide unbundled local switching or separators include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth stated died port/loop combinations that are Currently Combined or Not Currently Combined in 2 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-re	es except Zone 1 of (New Orle	as noted for Georgia and the Top 8 MSAS in BellSorans); NC (Greensboro-Wir	Tennessee. buth's region for end user nston Salem-Highpoint/C	harlotte-Gastoni	a-Rock Hill);	ΓN (Nashville)		of the Market Ra	Ites and res	serves the righ	t to true-up	the billing
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference.	ates shall apply where BellSouth is not required to provide unbundled local switching or sanarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2  MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA ( currently is developing the billing capability to mechanically bill the recurring and non-re by.	es except Zone 1 of (New Orle	as noted for Georgia and the Top 8 MSAS in BellSorans); NC (Greensboro-Wir	Tennessee. buth's region for end user nston Salem-Highpoint/C	harlotte-Gastoni	a-Rock Hill);	ΓN (Nashville)		of the Market Ra	ites and res	serves the righ	t to true-up	the billing
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference.	ates shall apply where BellSouth is not required to provide unbundled local switching or sanarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2  MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA ( currently is developing the billing capability to mechanically bill the recurring and non-re by.	es except Zone 1 of (New Orle	as noted for Georgia and the Top 8 MSAS in BellSorans); NC (Greensboro-Wir	Tennessee. buth's region for end user nston Salem-Highpoint/C	harlotte-Gastoni	a-Rock Hill);	ΓN (Nashville)		of the Market Ra	ites and res	serves the righ	it to true-up	the billing
Market Rate These scen 1. Unbundi 2. Unbundi The Top 8 I BellSouth c difference. The Market	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include: died porvloop combinations that are Not Currently Combined in all of the BellSouth state died porvloop combinations that are Currently Combined or Not Currently Combined in 28 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-resected are the set Rate for unbundled ports includes all available features in all states.	es except Zone 1 of (New Orle ecurring Ma	as noted for Georgia and the Top 8 MSAS in BellSovans); NC (Greensboro-Wir arket Rates in this section.	Tennessep. puth's region for end user nston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville st-Based sect	ion preceding in lieu o					
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-rest.  et Rate for unbundled ports includes all available features in all states.  e and Tandem Switching Usage and Common Transport Usage rates in the Port section.	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbund 2. Unbund The Top 8 I BellSouth c difference. The Market End Office For Not Cui	ates shall apply where BellSouth is not required to provide unbundled local switching or sanarios include: dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA ( currently is developing the billing capability to mechanically bill the recurring and non-re b. et Rate for unbundled ports includes all available features in all states. e and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are lis	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbund 2. Unbund The Top 8 I BellSouth c difference. The Market End Office For Not Cui	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-rest.  et Rate for unbundled ports includes all available features in all states.  e and Tandem Switching Usage and Common Transport Usage rates in the Port section.	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cu may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include: dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 in MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA ( currently is developing the billing capability to mechanically bill the recurring and non-re- beth Rate for unbundled ports includes all available features in all states.  e and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are list valso and are categorized accordingly.	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cu may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include: dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 in MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA ( currently is developing the billing capability to mechanically bill the recurring and non-re- beth Rate for unbundled ports includes all available features in all states.  e and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are list valso and are categorized accordingly.	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cu may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or sanarios include: dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA ( currently is developing the billing capability to mechanically bill the recurring and non-re b. et Rate for unbundled ports includes all available features in all states. e and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are lis	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  diled port/loop combinations that are Not Currently Combined in all of the BellSouth state diled port/loop combinations that are Currently Combined or Not Currently Combined in 28 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-regent Rate for unbundled ports includes all available features in all states.  Be and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listy also and are categorized accordingly.	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include: dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-restance of the combined ports includes all available features in all states.  The Rate for unbundled ports includes all available features in all states.  The and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listy also and are categorized accordingly.  OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and the Top 8 MSAS in BellSolans); NC (Greensboro-Wir arket Rates in this section.  The second of the section of the sec	Tennessee, uth's region for end user nston Salem-Highpoint/C . In the interim, BellSout combinations of loop/por columns for each Port U	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  diled port/loop combinations that are Not Currently Combined in all of the BellSouth state diled port/loop combinations that are Currently Combined or Not Currently Combined in 28 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-regent Rate for unbundled ports includes all available features in all states.  Be and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listy also and are categorized accordingly.	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and as noted for Georgia and the Top 8 MSAS in BellSotans); NC (Greensboro-Wir arket Rates in this section.	Tennesseb.  juth's region for end user inston Salem-Highpoint/C . In the interim, BellSout	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  died port/loop combinations that are Not Currently Combined in all of the BellSouth state died port/loop combinations that are Currently Combined or Not Currently Combined in 2 of MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-reseate Rate for unbundled ports includes all available features in all states.  The state for unbundled ports includes all available features in all states.  The and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are list also and are categorized accordingly.  The Combination Rates  Loop Combination Rates  12-Wire VG Loop/Port Combo - Zone	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and the Top 8 MSAS in BeilSonans); NC (Greensboro-Wir arket Rates in this section.  Le exhibit shall apply to all of First and Additional NRC of the NRC of	Tennessee, uth's region for end user nston Salem-Highpoint/C . In the interim, BellSout combinations of loop/por columns for each Port U	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-reset Rate for unbundled ports includes all available features in all states.  et Rate for unbundled ports includes all available features in all states. et and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listy also and are categorized accordingly.  OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  //Loop Combination Rates //Loop Combination Rates	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and the Top 8 MSAS in BellSoi and; the Top 8 MSAS in BellSoi ans); NC (Greensboro-Wir arket Rates in this section.  Le exhibit shall apply to all to First and Additional NRC of the MSAS in Bell and Additional NRC of the MSAS in Bell and Additional NRC of the MSAS in Bell and Additional NRC of the MSAS in Bell and Additional NRC of the MSAS in Bell and Additional NRC of the MSAS in Bell and Additional NRC of the MSAS in Bell and the MSAS in Bell a	Tennesse, uth's region for end user inston Salem-Highpoint/C . In the interim, BellSout combinations of loop/por columns for each Port U:  26.48 30.31	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundi 2. Unbundi The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or sanarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-restance of the combined ports includes all available features in all states.  et and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listy also and are categorized accordingly.  OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  1. Loop Combination Rates 2Wire VG Loop/Port Combo - Zone 2Wire VG Loop/Port Combo - Zone 2Wire VG Loop/Port Combo - Zone 2Wire VG Loop/Port Combo - Zone	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and the Top 8 MSAS in BeilSonans); NC (Greensboro-Wir arket Rates in this section.  Le exhibit shall apply to all of First and Additional NRC of the NRC of	Tennesse, uth's region for end user nston Salem-Highpoint/C . In the interim, BellSout combinations of loop/por columns for each Port U:	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundl 2. Unbundl The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  diled port/loop combinations that are Not Currently Combined in all of the BellSouth state diled port/loop combinations that are Currently Combined or Not Currently Combined in 2 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-regative to the combined ports includes all available features in all states.  Be and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listy also and are categorized accordingly.  OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CLOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  Rates	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and the Top 8 MSAS in BellSorans); NC (Greensboro-Wir arket Rates in this section.  Le exhibit shall apply to all of First and Additional NRC of the MSAS	Tennesseb. uth's region for end user inston Salem-Highpoint/C. In the interim, BellSout combinations of loop/por columns for each Port U:  26.48 30.31 35.32	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundi 2. Unbundi The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  dled port/loop combinations that are Not Currently Combined in all of the BellSouth state dled port/loop combinations that are Currently Combined or Not Currently Combined in 2 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-reset Rate for unbundled ports includes all available features in all states.  et Rate for unbundled ports includes all available features in all states.  et and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are list also and are categorized accordingly.  OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:  2-Wire VG Loop/Port Combo - Zone:	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and the Top 8 MSAS in BellSolans); NC (Greensboro-Wir arket Rates in this section.  Telephone is the section of the sect	Tennessee, uth's region for end user nston Salem-Highpoint/C . In the interim, BellSout combinations of loop/por columns for each Port U:  26.48 30.31 35.32  UEPLX 12.48	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).
Market Rate These scen 1. Unbundi 2. Unbundi The Top 8 I BellSouth c difference. The Market End Office For Not Cur may apply a	ates shall apply where BellSouth is not required to provide unbundled local switching or senarios include:  diled port/loop combinations that are Not Currently Combined in all of the BellSouth state diled port/loop combinations that are Currently Combined or Not Currently Combined in 2 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA (currently is developing the billing capability to mechanically bill the recurring and non-regative to the combined ports includes all available features in all states.  Be and Tandem Switching Usage and Common Transport Usage rates in the Port section urrently Combined scenarios where Market Rates apply, the Nonrecurring charges are listy also and are categorized accordingly.  OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  CLOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  2-Wire VG Loop/Port Combo - Zone  Rates	es except Zone 1 of (New Orle ecurring Management) of this rat	as noted for Georgia and the Top 8 MSAS in BellSorans); NC (Greensboro-Wir arket Rates in this section.  Le exhibit shall apply to all of First and Additional NRC of the MSAS	Tennesseb. uth's region for end user inston Salem-Highpoint/C. In the interim, BellSout combinations of loop/por columns for each Port U:  26.48 30.31 35.32	harlotte-Gastoni h shall bill the ra	a-Rock Hill); tes in the Cos	(Nashville) st-Based sect	ion preceding in lieu o	ns which have a	flat rate us	sage charge (U	JSOC: UREO	CU).

OWEN Ville Containing Book (Bus)					1		1		
2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residenc		UEPRX	UEPRL	14	90	90		30.89	7.03
z-wire voice unbuildied port - residenc		UEFRA	UEFKL	14	90	90		30.09	7.03
2-Wire voice unbundled port with Caller ID - re		UEPRX	UEPRC	14	90	90		30.89	7.03
2 Will voice distributed por will called 15 Te		OLI TOX	OL: NO			00		00.00	1.00
2-Wire voice unbundled port outgoing only - re		UEPRX	UEPRO	14	90	90		30.89	7.03
2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call	er .	<u>, , , , , , , , , , , , , , , , , , , </u>							
ID - res		UEPRX	UEPAQ	14	90	90		30.89	7.03
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (F2		UEPRX	UEPAK	14	90	90		30.89	7.03
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACE		UEPRX	UEPAL	14	90	90		30.89	7.03
		<u>, , , , , , , , , , , , , , , , , , , </u>							
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (TACS		UEPRX	UEPAM	14	90	90		30.89	7.03
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (1MF2		UEPRX	UEPAN	14	90	90		30.89	7.03
2-Wire voice unbundled Tennessee Area Calling port with Caller ID - res (2M		UEPRX	UEPAO	14	90	90		30.89	7.03
2-Wire voice unbundles res, low usage line port with Caller ID (LUI		UEPRX	UEPAP	14	90	90		30.89	7.03
LOCAL NUMBER PORTABILITY									
Local Number Portability (1 per port		UEPRX	LNPCX	0.35					
FEATURES									
All Features Offerec		UEPRX	UEPVF	0	0	0			
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-		UEPRX	USAC2		41.5	41.5			
2-Wire Voice Grade Loop / Line Port Combination - Switch with change		UEPRX	USACC		41.5	41.5			
ADDITIONAL NRCs									
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPRX	USAS2		0	0			
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)									
UNE Port/Loop Combination Rates									
2-Wire VG Loop/Port Combo - Zone	1			26.48					
2-Wire VG Loop/Port Combo - Zone	2			30.31					
2-Wire VG Loop/Port Combo - Zone	3			35.32					
UNE Loop Rates									
2-Wire Voice Grade Loop (SL1) - Zone	1	UEPBX	UEPLX	12.48					
2-Wire Voice Grade Loop (SL1) - Zone	2	UEPBX	UEPLX	16.31					
2-Wire Voice Grade Loop (SL1) - Zone	3	UEPBX	UEPLX	21.32					
2-Wire Voice Grade Line Port (Bus)									
2-Wire voice unbundled port without Caller ID - bu		UEPBX	UEPBL	14	90	90		30.89	7.03
2-Wire voice unbundled port with Caller + E484 ID - bu		UEPBX	UEPBC	14	90	90		30.89	7.03
2-Wire voice unbundled port outgoing only - bu		UEPBX	UEPBO	14	90	90		30.89	7.03
2-Wire voice Grade unbundled Tennessee extended local dialing parity port with Call	er								
ID - bus		UEPBX	UEPAV	14	90	90		30.89	7.03
2-Wire voice unbundled Tennessee Bus 2-Way Area Calling Port Economy Option									
(TACC1)		UEPBX	UEPAC	14				30.89	7.03
2-Wire voice unbundled Tennessee Bus 2-Way Area Calling Port Standard Option									
(TACC2)		UEPBX	UEPAD	14	90	90		30.89	7.03
2-Wire voice unbundled Tennessee Bus 2-Way Collierville and Memphis Local Calling	)								
Port (B2F)		UEPBX	UEPAE	14				30.89	7.03
LOCAL NUMBER PORTABILITY									
Local Number Portability (1 per port		UEPBX	LNPCX	0.35					
FEATURES									
NONRECURRING CHARGES - CURRENTLY COMBINED									
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-		UEPBX	USAC2		41.5	41.5			
2-Wire Voice Grade Loop / Line Port Combination - Switch with change		UEPBX	USACC		41.5	41.5			
ADDITIONAL NRCs									
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subseque		UEPBX	USAS2		0	0			
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)									
UNE Port/Loop Combination Rates									
2-Wire VG Loop/Port Combo - Zone	1			26.48					
2-Wire VG Loop/Port Combo - Zone	2			30.31					
2-Wire VG Loop/Port Combo - Zone	3			35.32					
UNE Loop Rates									
2-Wire Voice Grade Loop (SL1) - Zone	1	UEPRG	UEPLX	12.48					
2-Wire Voice Grade Loop (SL1) - Zone	2	UEPRG	UEPLX	16.31					
2-Wire Voice Grade Loop (SL1) - Zone	3	UEPRG	UEPLX	21.32					
2-Wire Voice Grade Line Port Rates (RES - PBX)									
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Re		UEPRG	UEPRD	14	90	90		30.89	7.03
LOCAL NUMBER PORTABILITY									
Local Number Portability (1 per port		UEPRG	LNPCP	3.15					
FEATURES									
NONRECURRING CHARGES - CURRENTLY COMBINED									
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-		UEPRG	USAC2		41.5	41.5			
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Chan-		UEPRG	USACC		41.5	41.5			

Attachment 2 Exhibit C

Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt Grou					0 14.64	14.64	19.99	19.99	19.99	
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)					1		10.00	10.00	10.00	Г
UNE Port/Loop Combination Rates										Γ
2-Wire VG Loop/Port Combo - Zone	1			26.48						٢
2-Wire VG Loop/Port Combo - Zone :	2			30.31						Ē
2-Wire VG Loop/Port Combo - Zone	3			35.32						L
UNE Loop Rates										L
2-Wire Voice Grade Loop (SL1) - Zone	1	UEPPX	UEPLX	12.48						₽
2-Wire Voice Grade Loop (SL1) - Zone	2	UEPPX	UEPLX	16.31						₽
2-Wire Voice Grade Loop (SL1) - Zone	3	UEPPX	UEPLX	21.32						+
2-Wire Voice Grade Line Port Rates (BUS - PBX)										+
Line Side Hebundled Combination 3 Way DBY Trunk Bort - Bu		UEPPX	UEPPC	14	90	90	30.89	7.03		
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bı		UEPPX	UEPPC	14	90	90	30.89	7.03		H
Line Side Unbundled Outward PBX Trunk Port - Bu		UEPPX	UEPPO	14	90	90	30.89	7.03		L
Line Side Unbundled Incoming PBX Trunk Port - Bu		UEPPX	UEPP1	14	90	90	30.89	7.03		H
2-Wire Voice Unbundled PBX LD Terminal Port		UEPPX	UEPLD	14	90	90	30.89	7.03		Н
2-Wire Voice Unbundled 2-Way Combination PBX Tennessee Calling P		UEPPX	UEPT2	14	30	30	30.89	7.03		Н
2-Wire Voice Unbundled 1-Way Outgoing PBX Tennessee Calling Pr		UEPPX	UEPTO	14			30.89	7.03		Г
2-Wire Voice Unbundled 2-Way Combination PBX Usage Pc		UEPPX	UEPXA	14	90	90	30.89	7.03		Γ
2-Wire Voice Unbundled PBX Toll Terminal Hotel Por		UEPPX	UEPXB	14	90	90	30.89	7.03		Г
										Г
2-Wire Voice Unbundled PBX LD DDD Terminals Po		UEPPX	UEPXC	14	90	90	30.89	7.03		H
2-Wire Voice Unbundled PBX LD Terminal Switchboard Pc		UEPPX	UEPXD	14	90	90	30.89	7.03		l
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Po		UEPPX	UEPXE	14	90	90	30.89	7.03		Γ
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling										Ī
Port		UEPPX	UEPXL	14	90	90	30.89	7.03		L
										l
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling P		UEPPX	UEPXM	14	90	90	30.89	7.03		⊬
2-Wire Voice Unbundled 1-W Out PBX Hotel/Hospital Economy Administrative Calling		UEPPX	HEDVA	1.4	90	00	20.00	7.00		l
Port TN  2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling		UEPPX	UEPXN	14	90	90	30.89	7.03		+
Port		UEPPX	UEPXO	14	90	90	30.89	7.03		ı
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Pc		UEPPX	UEPXS	14	90	90	30.89	7.03		H
2-Wire Voice Unbundled PBX Collierville and Memphis Calling Pc		UEPPX	UEPXU	14	90	90	30.89	7.03		H
2-Wire Voice Unbundled 2-Way PBX Tennessee RegionServ Callling P		UEPPX	UEPXV	14	90	90	30.89	7.03		Г
LOCAL NUMBER PORTABILITY		OLITA	OL: XV				00.00	7.00		Г
Local Number Portability (1 per port		UEPPX	LNPCP	3.15						Г
FEATURES										Г
NONRECURRING CHARGES - CURRENTLY COMBINED										Г
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-		UEPPX	USAC2		41.5	41.5				Г
										Г
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change		UEPPX	USACC		41.5	41.5				L
ADDITIONAL NRCs									20	L
2-Wire Voice Grade Loop/ Line Port Combination - Subseque		UEPPX	USAS2		0	0				L
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-										1
Nonrecurring					0	0				╄
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Groi					14.64	14.64	19.99	19.99	19.99	⊢
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT										⊢
UNE Port/Loop Combination Rates				00.40				1		+
2-Wire VG Coin Port/Loop Combo – Zone 1			-	26.48	-			<del>                                     </del>		$\vdash$
2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3			-	30.31 35.32	-			1		$\vdash$
UNE Loop Rates			+ +	33.32				+		$\vdash$
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	12.48	<b> </b>			<del>                                     </del>		Н
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	16.31	<b>†</b>					Н
2-Wire Voice Grade Loop (SL1) - Zone		UEPCO	UEPLX	21.32						Г
2-Wire Voice Grade Line Port Rates (Coin)										Г
2-Wire Coin 2-Way without Operator Screening and without Blocking (TN)		UEPCO	UEPTB	14	90	90	30.89	7.03		Г
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (NC,										Г
TN)		UEPCO	UEPRP	14	1		30.89	7.03		l
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (TN)		UEPCO	UEPTA	14	90	90	30.89	7.03		Γ
2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and										ī
Local (NC, TN)		UEPCO	UEPCA	14	90	90	30.89	7.03		L
2-Wire Coin Outward with Operator Screening and 011 Blocking (TN)		UEPCO	UEPTC	14	90	90	30.89	7.03		₽
2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+,										1
and Local (TN)		UEPCO	UEPOT	14	90	90	30.89	7.03		+
LOCAL NUMBER PORTABILITY		LIEBOO	LNDOV	0.05	-			1		+
Local Number Portability (1 per port		UEPCO	LNPCX	0.35	-					$\vdash$
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-I		UEPCO	USAC2		41.5	41.5				$\vdash$
z-wire voice Grade Loop/ Line Port Combination - Switch-As-i		UEPCO	USAC2		41.5	41.5				$\vdash$
O Miles Vision Conde Lang / Line Dark Combination Coulteb with Chan		UEPCO	USACC		41.5	41.5		1		1
		ULTUU	UUMUU		91.0	41.0		_		+
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Chan-							J.			

# AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications, Corp. ("NewSouth") a Delaware corporation.

WHEREAS, The Parties desire to amend that certain Interconnection Agreement between BellSouth and NewSouth dated May 18, 2001 (the "Interconnection Agreement") in order to incorporate rates established by the Louisiana Public Service Commission ("PSC") in Docket Number U-24717-A, on September 21, 2001;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, BellSouth and NewSouth hereby convenant and agree as follows:

- 1. Those permanent rates established by the Louisiana PSC in Docket No. U-24717-A for certain Unbundled Network Elements and Local Interconnection in Louisiana are as set forth in Exhibit 1 to this Amendment attached hereto and incorporated herein by this reference.
- 2. All rate elements and rates in Attachments 1, 2, 3, 5 and 7 of the Interconnection Agreement for Louisiana are hereby deleted and replaced in their entirety with the corresponding rates and rate elements in Exhibit 1.
- 3. The Parties hereby agree to delete Sections 4.2.3, 4.4, 4.6.1, 4.6.1.2, 5.3.2.1, 5.3.2.3 of Attachment 2 of the Interconnection Agreement and replace them with the following:
  - 4.2.3 BellSouth shall provide EEL combinations to NewSouth in the states of Georgia, Kentucky, Louisiana, and Tennessee regardless of whether or not such EELs are Already Combined. In all other states, BellSouth shall make available to NewSouth those EEL combinations described in Section 4.3 below only to the extent such combinations are Already Combined.
  - 4.4 Other Network Element Combinations

In the states of Georgia, Kentucky, Louisiana, and Tennessee BellSouth shall make available to NewSouth, in accordance with Section 4.6 below: (1) combinations of network elements other than EELs that are Already Combined; and (2) combinations of network elements other than EELs that are not Already Combined but that BellSouth ordinarily combines in its network. In all other states, BellSouth shall make available to NewSouth, in accordance with Section 4.5 below, combinations of network elements other than EELs only to the extent such combinations are Already Combined.

- 4.6.1 Georgia, Kentucky, Louisiana, and Tennessee
- 4.6.1.2 For combinations of loop and transport network elements not set forth in Section 4.3, where the elements are not Already Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the

stand-alone non-recurring and recurring charges of the network elements which make up the combination.

- 5.3.2.1 In Georgia, Kentucky, Louisiana, and Tennessee, BellSouth shall provide to NewSouth combinations of port and loop network elements to NewSouth on an unbundled basis regardless of whether or not such combinations are Currently Combined except in those locations where BellSouth is not required to provide circuit switching, as set forth in Section 5.2.2 above. The rates for such combinations shall be the cost based rates set forth in Exhibit C of this Attachment.
- 5.3.2.3 In all states other than Georgia, Kentucky, Louisiana, and Tennessee, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 5.2.1 and 5.2.2, BellSouth shall provide to NewSouth combinations of port and loop network elements that are not Currently Combined. The rates for such combinations shall be negotiated by the Parties.
- 4. The Parties agree that all of the other provisions of the Interconnection Agreement, dated May 18, 2001, shall remain in full force and effect.
- 5. The Parties further agree that either or both of the Parties is authorized to submit this Amendment to the Louisiana Public Service Commission or other regulatory body having jurisdiction over the subject matter of this Amendment, for approval subject to Section 252(e) of the federal Telecommunications Act of 1996.

This Amendment is made effective upon the date that it is signed by both Parties.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the data indicated below.

BellSouth Telecommunications, Inc.	NewSouth Communications, Corp.
On File	On File
Signature	Signature
Chris Boltz	Jake Jennings
Name	Name
Managing Director	Vice President – Regulatory Affairs
Title	Title
November 14, 2001	November 8, 2001
Date	Date

## **EXHIBIT 1**

### Attachment 1 Page 1 Exhibit E

# RESALE DISCOUNTS AND RATES

		LOUISIANA
APPLICABLE DISCOUNTS		
RESIDENCE		20.72%
BUSINESS		20.72%
CSAs*		9.05%
* Unless noted in this row, the discount for	or Business will be the appl	icable discount rate for CSA
OPERATIONAL SUPPORT SYSTE	CMS (OSS) RATES	
<u>ELEMEN</u> T	<u>USOC</u>	
Electronic LSR	SOMEC	\$3.50
Manual LSR	SOMAN	\$19.99
ODUF/EODUF/CMDS RATES		
ENHANCED OPTION DAILY USAGE FIL	E (EODUF)	
EODUF: Message Processing, per	message	\$0.250015
OPTIONAL DAILY USAGE FILE (ODUF)		
ODUF: Recording, per message		\$0.0000117
ODUF: Message Processing, per message	e	\$0.004641
ODUF: Message Processing, per Magnet	ic Tape provisioned	\$48.45
ODUF: Data Transmission (CONNECT:	DIRECT), per msg	\$0.00010568

Version 3Q01: 10/18/01

## **EXHIBIT 1**

## **RESALE DISCOUNTS AND RATES**

Attachment 1 Page 2 Exhibit E

		LOUISIANA
CUSTOM BRANDING ANNOUNCE	MENT (CBA)	
DIRECTORY ASSISTANCE (DA) CBA via O	LNS SOFTWARE	
Recording of DA CBA		\$3,000.00
Loading of DA CBA per DRAM Card/Swi	tch per OCN	\$1,700.00
DIRECTORY ASSISTANCE (DA) UNBRANI	DING via OLNS SOFTWAR	E
Loading of DA per OCN (1 OCN p	er Order)	\$420.00
Loading of DA per Switch, per OCN		\$16.00
OPERATOR ASSISTANCE (OA) CBA via Ol	LNS SOFTWARE	
<u>ELEMEN</u> T	<u>USOC</u>	
Recording of OA CBA	CBAOS	\$7,000.00
Loading of OA CBA per shelf/ NAV per		
OCN	CBAOL	\$500.00
Loading of DA CBA per DRAM Card/Swi	tch per OCN	\$1,170.00
OPERATOR ASSISTANCE (OA) UNBRAND		
Loading of OA per OCN - Regional		\$1,200.00

Version 3Q01: 10/18/01

							RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonre	curring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'I
					Rec	First	Add'l	First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	" shown in the sections for stand-alone loops or loops as part of a combination refers v.interconnection.bellsouth.com/become a clec/html/interconnection.htm	to Geo	graphical	lly Deaver	aged UNE Zone	es. To view Ge	ographically Dea	averaged UN	NE Zone Desi	gnations by	Central Offic	e, refer to Inte	ernet Website:		
nttp.//www	w.interconnection.beilsouth.com/become_a_clec/ntmi/interconnection.ntm	1	ı											1	
INBUNDLED EXCHA	NGE ACCESS LOOP														
2-WIRE A	NALOG VOICE GRADE LOOP			115410	10.00	20.54	10.07				45.00				
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone		UEANL		12.90 23.33	36.54 36.54	16.87 16.87				15.20 15.20				
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3	3	UEANL		48.43	36.54	16.87				15.20				
	Loop Testing - Basic 1st Half Hour		UEANL		40.40	33.17	33.17				10.20				
	Loop Testing - Basic Additional Half Hour		UEANL			19.28	19.28								
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone	1	UEPSR, UEPSB	UEALS	12.90	36.54	16.87				15.20				
			UEPSR,												
	2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone 2	2	UEPSB	UEALS	23.33	36.54	16.87	0.00	0.00		15.20				
			UEPSR,												
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone :	3	UEPSB	UEALS	48.43	36.54	16.87	0.00	0.00		15.20				
	Engineering Information Document (EI)		UEANL			13.04	13.04								
	Manual Order Coordination for UVL-SL1s (per loop)		LIEANII	UEAMC		7.92	7.92								
	Ivianual Order Coordination for OVE-SETS (per 100p)		UEAINL	ULAIVIC		1.52	1.52								
	Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR)		UEANL	OCOSL		17.56	17.56								
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start			115410	44.00	400.40	05.70				45.00				
	Signaling - Zone 1  2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start	1	UEA	UEAL2	14.93	102.10	65.72				15.20				
	Signaling - Zone 2	2	UEA	UEAL2	25.35	102.10	65.72				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start	-	OLA	OLMEZ	20.00	102.10	00.72				10.20				
	Signaling - Zone 3	3	UEA	UEAL2	50.46	102.10	65.72				15.20				
	Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		17.56									
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling · Zone 1	1	UEA	UEAR2	14.93	102.10	65.72				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling ·		ULA	ULAKZ	14.93	102.10	05.72				13.20				
	Zone 2	2	UEA	UEAR2	25.35	102.10	65.72				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling ·														
	Zone 3	3	UEA	UEAR2	50.46	102.10	65.72				15.20				
				00001		17.50									
1-WIDE A	Order Coordination for Specified Conversion Time (per LSR)  NALOG VOICE GRADE LOOP		UEA	OCOSL		17.56									
7-WILL 7	4-Wire Analog Voice Grade Loop - Zone 1	1	UEA	UEAL4	30.81	127.40	91.02				15.20				
	4-Wire Analog Voice Grade Loop - Zone 2	2		UEAL4	38.32	127.40					15.20				
	4-Wire Analog Voice Grade Loop - Zone 3	3	UEA	UEAL4	60.39	127.40					15.20				
	Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		17.56									
2 MIDE 10	SDN DIGITAL GRADE LOOP														
Z-VVIKE IS	2-Wire ISDN Digital Grade Loop - Zone 1	1	UDN	U1L2X	22.09	113.34	76.96				15.20		1	1	
	2-Wire ISDN Digital Grade Loop - Zone 2	2		U1L2X	35.28	113.34	76.96				15.20				
	2-Wire ISDN Digital Grade Loop - Zone 3	3		U1L2X	65.18	113.34	76.96				15.20		<u> </u>		
	Order Coordination For Specified Conversion Time (per LSR		UDN	OCOSL		17.56									
2 WIDE I	Iniversal Digital Channel (UDC) COMPATIBLE LOOP												1	1	
	ANY EL SON DIGITAL CHANNEL LUDGI GUNTA LIDLE LUUF	11	1	1			1		1	i .	1	1	1	1	1

						I	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	urring	Nonrecurring I	Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st		Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'l
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2	2		UDC2X	35.28 65.18	113.34 113.34	76.96				15.20 15.20				+
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 3	3	UDC	UDC2X	65.18	113.34	76.96				15.20				
2-WIRE A	SYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP														
	2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE														
	LOOP														<b></b>
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation			1141 61/	40.00	447.00	00.00				45.00				ĺ
	<ul> <li>Zone 1</li> <li>Wire Unbundled ADSL Loop including manual service inquiry &amp; facility reservation</li> </ul>	1	UAL	UAL2X	12.29	117.08	68.36				15.20				<b>—</b>
	- Zone 2	2	UAL	UAL2X	14.09	117.08	68.36				15.20				ĺ.
	2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation		O/IL	OTTLET	14.00	117.00	00.00				10.20				
	- Zone 3	3	UAL	UAL2X	15.75	117.08	68.36				15.20				I
															l
	Order Coordination for Specified Conversion Time (per LSR)		UAL	OCOSL		17.56									
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -														I
	Zone 1  2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservator -	1	UAL	UAL2W	12.29	92.83	56.02				15.20				+
	Zone 2	2	UAL	UAL2W	14.09	92.83	56.02				15.20				ĺ
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -		UAL	UALZVV	14.03	92.03	30.02				13.20				
	Zone 3	3	UAL	UAL2W	15.75	92.83	56.02				15.20				ĺ
	Order Coordination for Specified Conversion Time (per LSR)		UAL	OCOSL		17.56									
															<b></b>
2-WIRE F	IIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP  2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP														
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation														
	- Zone 1	1	UHL	UHL2X	9.79	125.50	76.77				15.20				
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation														ĺ
	<ul> <li>Zone 2</li> <li>Wire Unbundled HDSL Loop including manual service inquiry &amp; facility reservation</li> </ul>	2	UHL	UHL2X	11.52	125.50	76.77				15.20				<del></del>
	- Zone 3	3	UHL	UHL2X	12.74	125.50	76.77				15.20				1
	2010 0		OTTE	OTILEX	12.77	120.00	70.77				10.20				
	Order Coordination for Specified Conversion Time (per LSR)		UHL	OCOSL		17.56									ĺ
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility														ĺ
	reservation - Zone 1	1	UHL	UHL2W	9.79	101.24	64.43				15.20				<b></b>
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility	2		UHL2W	11.52	404.04	04.40				45.00				ĺ.
	reservation - Zone 2  2 Wire Unbundled HDSL Loop without manual service inquiry and facility		UHL	UHLZW	11.52	101.24	64.43				15.20				
	reservation - Zone 3	3	UHL	UHL2W	12.74	101.24	64.43				15.20				I
	1000 Validit 2010 C		0	O. ILL II	12		01.10				10.20				
	Order Coordination for Specified Conversion Time (per LSR)		UHL	OCOSL		17.56									l
															<u> </u>
4-WIRE I	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP														<b></b>
	4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 1	4	UHL	UHL4X	16.24	153.26	104.54				15.20				ĺ
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility		UNL	UNL4X	10.24	155.26	104.54				15.20				
	reservation - Zone 2	2	UHL	UHL4X	16.65	153.26	104.54				15.20				ĺ
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility														ĺ
	reservation - Zone 3	3	UHL	UHL4X	17.34	153.26	104.54				15.20				L
				l <sup>-</sup>											1
	Order Coordination for Specified Conversion Time (per LSR)		UHL	OCOSL		17.56									l
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1	1	UHL	UHL4W	16.24	129.00	92.20				15.20				I
+	4-Wire Unbundled HDSL Loop without manual service inquiry and facility		OFIL	OI IL4VV	10.24	129.00	32.20	-			10.20				
	reservation - Zone 2	2	UHL	UHL4W	16.65	129.00	92.20				15.20				I
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility														
	reservation - Zone 3	3	UHL	UHL4W	17.34	129.00	92.20				15.20				1
	Order Coordination for Specified Conversion Time (per LSR)		UHL	OCOSL		17.56									
															<b></b>
4-WIRE [	DS1 DIGITAL LOOP														

						F	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec		Nonrecurring Dis		Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Add'l
4 Wire DC4	Digital Loop Zone 1	1	USL	USLXX	Rec 85.70	First 245.16	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Digital Loop - Zone 1 Digital Loop - Zone 2	2		USLXX	194.96	245.16	152.98 152.98				15.20 15.20				-
4-Wire DS1	Digital Loop - Zone 3	3		USLXX	491.94	245.16	152.98				15.20				
	dination for Specified Conversion Time (per LSR)		USL	OCOSL	101.01	17.56	102.00				10.20				
4-WIRE 19.2, 56 OR 64 K	KBPS DIGITAL GRADE LOOP														
	undled Digital 19.2 Kbps	1	UDL	UDL19	30.99	121.86	85.48				15.20				
4 Wire Unbu	undled Digital 19.2 Kbps	2	UDL	UDL19	36.78	121.86	85.48				15.20				
	undled Digital 19.2 Kbps	3	UDL	UDL19	38.92	121.86	85.48				15.20				
	undled Digital Loop 56 Kbps - Zone 1	1	UDL	UDL56	30.99	121.86	85.48				15.20				
4 Wire Unbu	undled Digital Loop 56 Kbps - Zone 2	2	UDL	UDL56	36.78	121.86	85.48				15.20				
4 Wire Unbu	undled Digital Loop 56 Kbps - Zone 3	3	UDL	UDL56	38.92	121.86	85.48				15.20				
	dination for Specified Conversion Time (per LSR)	<b>!</b>	UDL	OCOSL		17.56									<u> </u>
	undled Digital Loop 64 Kbps - Zone 1	1	UDL	UDL64	30.99	121.86	85.48				15.20				<b>_</b>
4 Wire Unbu	undled Digital Loop 64 Kbps - Zone 2 undled Digital Loop 64 Kbps - Zone 3	3	UDL	UDL64 UDL64	36.78 38.92	121.86	85.48 85.48				15.20 15.20				-
4 Wire Unbu	undled Digital Loop 64 Kbps - Zone 3	3	UDL	UDL64	38.92	121.86	85.48				15.20				-
Order Coord	dination for Specified Conversion Time (per LSR)		UDL	OCOSL		17.56									
2-WIRE Unbundled COP															
	undled Copper Loop/Short including manual service inquiry & facility	l .													
reservation - 2-Wire Unbureservation -	undled Copper Loop/Short including manual service inquiry & facility	2	UCL	UCLPB	12.29	116.18 116.18	67.46 67.46				15.20 15.20				
	undled Copper Loop/Short including manual service inquiry & facility	-	UCL	UCLPB	14.09	110.10	67.46				15.20				
reservation		3	UCL	UCLPB	15.75	116.18	67.46				15.20				
Teservation	2010 0		OOL	OOLI D	10.70	110.10	07.40				10.20				
Order Coord	dination for Unbundled Copper Loops (per loop		UCL	UCLMC		7.92	7.92								
2-Wire Unbureservation	undled Copper Loop/Short without manual service inquiry and facility - Zone 1	1		UCLPW	12.29	91.92	55.12				15.20				
	undled Copper Loop/Short without manual service inquiry and facility														
reservation		2	UCL	UCLPW	14.09	91.92	55.12				15.20				
	undled Copper Loop/Short without manual service inquiry and facility			LIGI DIA	45.75	04.00	55.40				45.00				
reservation -	- Zone 3	3	UCL	UCLPW	15.75	91.92	55.12				15.20				
Order Coord	dination for Unbundled Copper Loops (per loop)		UCL	UCLMC		7.92	7.92								
reservation -		1	UCL	UCL2L	17.21	116.18	67.46				15.20				
2-Wire Unbureservation	undled Copper Loop/Long - includes manual svc. inquiry and facility	2	UCL	UCL2L	24.98	116.18	67.46				15.20				
	undled Copper Loop/Long - includes manual svc. inquiry and facility	3	UCL	UCL2L	39.57	116.18	67.46				15.20				
	dination for Unbundled Copper Loops (per loop)		UCL	UCLMC		7.92	7.92								
reservation -		1	UCL	UCL2W	17.21	91.92	55.12				15.20				
reservation -	undled Copper Loop/Long - without manual service inquiry and facility - Zone 2 undled Copper Loop/Long - without manual service inquiry and facility	2	UCL	UCL2W	24.98	91.92	55.12				15.20				
reservation -	- Zone 3	3	UCL	UCL2W	39.57	91.92	55.12				15.20				
Order Coord	dination for Unbundled Copper Loops (per loop	1	UCL	UCLMC		7.92	7.92			-					
2-Wire Unh	undled Copper Loop - Non-Designed Zone 1	1	UEQ	UEQ2X	12.40	35.27	15.60				15.20			<u> </u>	
	undled Copper Loop - Non-Designed 25/16 1	2		UEQ2X	14.32	35.27	15.60				15.20				
	undled Copper Loop - Non-Designed - Zone 3	3		UEQ2X	16.87	35.27	15.60				15.20				
	dination 2 Wire Unbundled Copper Loop - Non-Designed (per loop)			USBMC		7.92	7.92								
Engineering	Information Document		UEQ			13.04	13.04								
Loop Testing	g - Basic 1st Half Hour		UEQ	URET1	<u> </u>	33.17	33.17								

						1	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	eurring	Noncouri	a Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'l
					Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Loop Testing - Basic Additional Half Hour		UEQ	URETA		19.28	19.28								
4-WIRE	COPPER LOOP														
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1	1	UCL	UCL4S	22.27	139.69	90.96				15.20				
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation		OOL	0CL43	22.21	139.09	90.90				13.20				
	- Zone 2	2	UCL	UCL4S	18.95	139.69	90.96				15.20				
İ	<ul> <li>4-Wire Copper Loop/Short - including manual service inquiry and facility reservation</li> <li>- Zone 3</li> </ul>	3	UCL	UCL4S	10.99	139.69	90.96				15.20				
	Order Coordination for Unbundled Copper Loops (per loop	,	UCL	UCLMC	10.55	7.92	7.92				10.20				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -				22.55						45.55				
<del>                                     </del>	Zone 1  4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -	1	UCL	UCL4W	22.27	115.43	78.63				15.20			<del>                                     </del>	
	Zone 2	2	UCL	UCL4W	18.95	115.43	78.63				15.20				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -														
	Zone 3 Order Coordination for Unbundled Copper Loops (per loop	3	UCL	UCL4W UCLMC	10.99	115.43 7.92	78.63 7.92				15.20				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		OOL	OCLIVIC		7.92	7.92								
	reservation - Zone 1	1	UCL	UCL4L	26.17	139.69	90.96				15.20				
Í	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2	2	UCL	UCL4L	28.47	139.69	90.96				15.20				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		UCL	UCL4L	20.47	139.09	90.90				15.20				
	reservation - Zone 3	3	UCL	UCL4L	62.93	139.69	90.96				15.20				
	Order Coordination for Unbundled Copper Loops (per loop  4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility		UCL	UCLMC		7.92	7.92								
	reservation - Zone 1	1	UCL	UCL4O	26.17	115.43	78.63				15.20				
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility														
	reservation - Zone 2	2	UCL	UCL40	28.47	115.43	78.63				15.20				
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 3	3	UCL	UCL4O	62.93	115.43	78.63				15.20				
	Order Coordination for Unbundled Copper Loops (per loop	,		UCLMC	3-133	7.92	7.92								
LOOP MODIFICATION	M.														
LOOP MODIFICATIO	N .		UAL,												
	Unbounded Lore Medification Deposits of Lord Coils, O.Wissonia Lore than to		UHL, UCL,												
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft		UEQ, ULS	ULM2L		0.00	0.00								
			UCL,												
<del></del>	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k f Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to		ULS UHL.	ULM2G		0.00	0.00				-			-	<del>                                     </del>
	18K ft		UCL	ULM4L		0.00	0.00								
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18k														
	ft		UCL UAL,	ULM4G		0.00	0.00								
			UHL, UCL, UEQ,												
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled		UEF,	LUMBT		40.45	40.45								
	loop		ULS	ULMBT		12.15	12.15				<del>                                     </del>			<del>                                     </del>	<del>                                     </del>
SUB-LOOPS							ı	1	1	1	1	l .			1
	on Distribution														
	pp Distribution Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Ur		UEANL	USBSA		144.09	144.09				15.20				
	pp Distribution Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-U Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-U \$\text{f}\$			USBSA USBSB		144.09 10.99	144.09 10.99				15.20 15.20				
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Uş Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Uş		UEANL	USBSB		10.99	10.99				15.20				
SUB-LOOPS Sub-Loc	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up			USBSB											

						I	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	surring	Nonrecurri	ng Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st		Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I
					Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1			USBN2	7.57	63.89	30.06				15.20				<del> </del>
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2 Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	2		USBN2 USBN2	12.75 21.45	63.89 63.89	30.06 30.06				15.20 15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pai	3		USBMC	21.45	7.92	7.92				15.20				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1	1		USBN4	11.76	76.75	42.92				15.20				<del>                                     </del>
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2	2		USBN4	16.84	76.75	42.92				15.20				
-	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3	3		USBN4	19.27	76.75	42.92				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pai			USBMC		7.92	7.92								
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)		UEANL	USBR2	2.91	51.48	17.65				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pai			USBMC		7.92	7.92								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)			USBR4	6.58	57.54	23.71				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pai			USBMC		7.92	7.92								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1	UEF	UCS2X	6.26	63.89	30.06				15.20				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	3	UEF	UCS2X UCS2X	10.07 12.70	63.89	30.06 30.06				15.20 15.20		1		<del></del>
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone : Order Coordination for Unbundled Sub-Loops, per sub-loop pai	3	UEF	USBMC	12.70	63.89 7.92					15.20				<del> </del>
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1	UEF	UCS4X	8.03	7.92	7.92 42.92				15.20				<del></del>
	4 Wire Copper Unburidled Sub-Loop Distribution - Zone 2		UEF	UCS4X	10.71	76.75	42.92				15.20				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		UEF	UCS4X	6.08	76.75	42.92				15.20				
-	Order Coordination for Unbundled Sub-Loops, per sub-loop pai	Ü	UEF	USBMC	0.00	7.92	7.92				10.20				
							-								
Sub-Loc	op Feeder		UEA,												
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-uş		UDN,U CL,UDL UDC UEA, UDN,U	USBFW		144.09									
	UOL Fd DC0 C-t C D Iti 05it		CL,UDL			40.00	40.00								1
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 terminatior		UDC	USBFX		10.99 568.98	10.99 11.30								+
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone	1	UEA	USBFA	8.71	89.81	54.35				15.20				<del>                                     </del>
-	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone:	2	UEA	USBFA	13.64	89.81	54.35				15.20				
							0.100								
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone	3	UEA	USBFA	30.21	89.81	54.35				15.20				1
	Order Coordination for Specified Conversion Time, per LSF			OCOSL		17.56									
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone	1	UEA	USBFB	8.71	89.81	54.35				15.20				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone	2	UEA	USBFB	13.64	89.81	54.35				15.20				<del></del>
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone Order Coordination for Specified Time Conversion, per LSF	3	UEA	USBFB OCOSL	30.21	89.81 17.56	54.35				15.20				<del> </del>
	Order Coordination for Specified Time Conversion, per ESF		ULA	UCUSL		17.56									+
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone	1	UEA	USBFC	8.71	89.81	54.35				15.20				1
	Sinsanaisa sas 200p i codoi 200p, 2 vino novolas sanoty, voice Grade - 2011e	<u> </u>	OLA	300.0	5.71	00.01	04.00				10.20				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone	2	UEA	USBFC	13.64	89.81	54.35				15.20				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade														
	Zone 3	3	UEA	USBFC	30.21	89.81	54.35				15.20		1		
															1
1	Order Coordination For Specified Conversion Time, per LSF Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone	_	UEA	OCOSL	04.44	17.56	07.01				45.00			-	
		1		USBFD	21.44 24.66	103.69 103.69	67.31 67.31				15.20 15.20		<del>                                     </del>		
		2			24.pb	103.69	07.37			1			1	1	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone	2				103 60	67 31								
		3		USBFD	42.84	103.69	67.31				15.20				<del> </del>
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone : Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone :			USBFD			67.31				15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone : Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone Order Coordination For Specified Conversion Time, Per LSF Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone	1	UEA UEA	USBFD OCOSL USBFE	42.84 21.44	17.56 103.69	67.31				15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone: Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone  Order Coordination For Specified Conversion Time, Per LSF  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone	1 2	UEA UEA UEA	OCOSL USBFE USBFE	42.84 21.44 24.66	17.56 103.69 103.69	67.31 67.31				15.20 15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone : Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone Order Coordination For Specified Conversion Time, Per LSF Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone	1	UEA UEA UEA	USBFD OCOSL USBFE	42.84 21.44	17.56 103.69	67.31				15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone : Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone Order Coordination For Specified Conversion Time, Per LSF Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone	1 2	UEA UEA UEA UEA	OCOSL USBFE USBFE USBFE	42.84 21.44 24.66	17.56 103.69 103.69 103.69	67.31 67.31				15.20 15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone: Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone  Order Coordination For Specified Conversion Time, Per LSF Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Order Coordination For Specified Conversion Time, Per LSF	1 2 3	UEA UEA UEA UEA UEA	OCOSL USBFE USBFE USBFE OCOSL	21.44 24.66 42.84	17.56 103.69 103.69 103.69	67.31 67.31 67.31				15.20 15.20 15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone : Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone Order Coordination For Specified Conversion Time, Per LSF Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone	1 2	UEA UEA UEA UEA UEA UEA	OCOSL USBFE USBFE USBFE	42.84 21.44 24.66	17.56 103.69 103.69 103.69	67.31 67.31				15.20 15.20				

							RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec			ig Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'I		Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'I
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination For Specified Conversion Time, Per LSF		UDN	OCOSL		17.56									
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible	1		USBFS	15.44	102.58	66.20				15.20				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible	2		USBFS	23.32	102.58	66.20				15.20				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible	3		USBFS	44.57	102.58	66.20				15.20				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone ' Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone '	1	USL	USBFG	55.38 167.83	98.15 98.15	61.77 61.77				15.20 15.20				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone :	3	USL	USBFG	469.87	98.15	61.77				15.20				-
	Official and Cab Ecop 1 code: Ecop; 4 Wile Bot 25/16 C	Ū	OOL	OOD! C	400.01	30.10	01.77				10.20				
	Order Coordination For Specified Conversion Time, Per LSF		USL	OCOSL		17.56									
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone	1	UCL	USBFH	6.96	81.36	44.98				15.20				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	2	UCL	USBFH	4.97	81.36	44.98				15.20				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	3	UCL	USBFH	3.99	81.36	44.98				15.20				
	Order Coordination For Specified Conversion Time par LSE		LICI	00000		47.50									
	Order Coordination For Specified Conversion Time, per LSF Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone	1	UCL	OCOSL USBFJ	15.68	17.56 98.07	61.69				15.20				<del>                                     </del>
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2	2	UCL	USBFJ	9.68	98.07	61.69				15.20				-
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone :	3		USBFJ	6.39	98.07	61.69				15.20				
	Order Coordination For Specified Conversion Time, per LSF		UCL	OCOSL		17.56									
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	1	UDL	USBFN	22.61	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	2	UDL	USBFN	22.87	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	3		USBFN	24.25	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone	1 2	UDL	USBFO	22.61 22.87	98.15 98.15	61.77 61.77				15.20 15.20				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone : Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone :	3		USBFO	24.25	98.15	61.77				15.20				+
	Oub 200p 1 code: 1 cl 4 Wile 00 Napo Bigital Grade 200p 201e (	Ū	ODL	OOD! O	24.20	30.10	01.77				10.20				
	Order Coordination For Specified Time Conversion, per LSF		UDL	OCOSL		17.56									
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone	1	UDL	USBFP	22.61	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2	2		USBFP	22.87	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone (	3	UDL	USBFP	24.25	98.15	61.77				15.20				
	Order Coordination For Specified Conversion Time, per LSF		UDL	OCOSL		17.56									
	Sub Loop Feeder - DS3 - Per Mile Per Montf		UE3	1L5SL	17.00										
	Sub Loop Feeder - DS3 - Facility Termination Per Month			USBF1	368.44	3,381.00	406.56	158.98	90.12		15.20				
	Sub Loop Feeder – STS-1 – Per Mile Per Montf			1L5SL	17.00										
	Sub Loop Feeder - STS-1 - Facility Termination Per Month			USBF7	395.92	3,381.00	406.56	158.98	90.12		15.20				
	Sub Loop Feeder – OC-3 – Per Mile Per Montf			1L5SL	12.90										
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per Month Sub Loop Feeder - OC-3 - Facility Termination Per Month			USBF5 USBF2	60.45 594.77	3,381.00	406.56	158.98	90.12		15.20				-
	Sub Loop Feeder - OC-12 - Per Mile Per Month			1L5SL	15.87	3,301.00	400.56	150.98	90.12		15.20				-
	Sub Loop Feeder - OC-12 - Fer Mile Fer Month  Sub Loop Feeder - OC-12 - Facility Termination Protection Per Month			USBF6	683.03										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month			USBF3	1,922.00	3,381.00	406.56	158.98	90.12		15.20				
	Sub Loop Feeder - OC-48 - Per Mile Per Month			1L5SL	52.07	-									
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per Month			USBF9	341.64										1
	Sub Loop Feeder - OC-48 - Facility Termination Per Month			USBF4	1,663.00	3,566.00	406.56	158.98	90.12		15.20				
	Sub Loop Feeder - OC-12 Interface On OC-48		UDL48	USBF8	385.45	787.24	406.56	158.98	90.12		15.20				+
Unbundl	led Sub-Loop Modification														
Jiibailai	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per														<b>†</b>
	2-W PR		UEF	ULM2X		0.00	0.00				15.20				
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4 W PR		UEF	ULM4X		0.00	0.00				15.20				
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR unloaded		UEF	ULM4T		224.55	4.29				15.20				
l laborII	lad Naturals Tarminating Wire (HNTW)			-											-
Unbundi	led Network Terminating Wire (UNTW)		LIENTA	LIENDO	0.2454	44.70	44.70				45.00				
Pr	Unbundled Network Terminating Wire (UNTW) per Pair		UENIW	UENPP	0.3454	14.72	14.72				15.20				
Network	Interface Device (NID)			<u> </u>				l	l	L	1	l	L	l	

						F	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc	- Rec	Nonrec First	rurring Add'l	Nonrecurrin First	g Disconnect Add'l	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1 st		Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. : Electronic-Disc Add'I
	Network Interface Device (NID) - 1-2 lines		UENTW	UND12	Kec	42.26	27.83	First	Addi	SOMEC	15.20	SUMAN	SOMAN	SUMAN	SUMAN
	Network Interface Device (NID) - 1-6 lines			UND16		62.86	48.43				15.20				
	Network Interface Device Cross Connect - 2 W			UNDC2		5.73	5.73				15.20				
	Network Interface Device Cross Connect - 4W			UNDC4		5.73	5.73				15.20				
UNBUNDLED LOOP			111.0	UCT8A	374.26	316.00	316.00				15.20				<u> </u>
	Unbundled Loop Concentration - System A (TR008) Unbundled Loop Concentration - System B (TR008)			UCT8B	53.40	131.67	131.67				15.20				
	Unbundled Loop Concentration - System A (TR303)			UCT3A	412.08	316.00	316.00				15.20				
	Unbundled Loop Concentration - System B (TR303)			UCT3B	89.98	131.67	131.67				15.20				
	Unbundled Loop Concentration - DS1 Loop Interface Card	1	ULC	истсо	5.12	61.46	44.74				15.20				
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Card			ULCC1	8.12	10.23	10.18				15.20				
	Unbundled Loop Concentration - UDC Loop Interface (Brite Card			ULCCU	8.12	10.23	10.18								
	Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop				0.00	40.00	40.40								
	Interface (POTS Card) Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface		UEA	ULCC2	2.03	10.23	10.18								
	(SPOTS Card)		UEA	ULCCR	12.07	10.23	10.18								
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Card			ULCC4	7.20	10.23	10.18								
	Unbundled Loop Concentration - TEST CIRCUIT Carc			UCTTC	35.19	10.23	10.18				15.20				<u> </u>
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interfact Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interfact			ULCC7 ULCC5	10.67 10.67	10.23 10.23	10.18 10.18								
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interfact			ULCC6	10.67	10.23	10.18								
	<b>J</b>														
	Unbundled Loop Concentration - Loop Interface For Digital 19.2 Kbps Data				10.63	10.23	10.18								
					10.00	10.20	10110								
UNE OTHER, PROVIS	SIONING ONLY - NO RATE														
	NID - Dispatch and Service Order for NID installation		UENTW	UNDBX											
	UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW	UENCE											
	, , , , , , , , , , , , , , , , , , ,		UEANL,												
			UEF,UE												
	Unbundled Contract Name, Provisioning Only - No Rate		Q,UENT W	UNECN											
	Cristinated Contract Hame, Frovisioning Only 140 Nate		UAL,UC												
			L,UDC,												
			UDL,UD N,UEA,												
			UHL,UL												
	Unbundled Contact Name, Provisioning Only - no rate			UNECN	0.00	0.00									
			UEA,UD N.UCL.												
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate			USBFQ	0.00	0.00									
			UEA,US		0.00	3.00									
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate		L,UCL,U DL	USBFR	0.00	0.00									
	Onbundied Sub-Loop reedel-4 wile Closs box Jumper - no fatt	1	DL	USBER	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate		USL	CCOSF	0.00	0.00									
	Unbundled DS1 Loop - Expanded Superframe Format option - no rate		USL	CCOEF	0.00	0.00									
	BUNDLED LOCAL LOOP	L								<u> </u>					
NOTE: 4	4 month minimum billing period		1												
		_													
	High Capacity Unbundled Local Loop - DS3 - Per Mile per month		UE3		10.04	420.40	056.00				15.00				<del> </del> 1
				UE3PX	10.04 362.34 10.04	438.46	256.30				15.20				

						F	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC	-	Nonrecu	urring	Nonrecurrin	g Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Electronic-Disc	Incremental Charge - Manual Svc Order vs. c Electronic-Disc Add'l
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month		UDLSX	UDLS1	Rec 374.56	First 438.46	Add'I 256.30	First	Add'I	SOMEC	SOMAN 15.20	SOMAN	SOMAN	SOMAN	SOMAN
LOOP MAKE-UP					-										
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).		UMK	UMKLW		23.29	23.29								
	queneu (manuar).		OWIN	OWINLY		23.29	23.29								
	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).		UMK	UMKLP		24.70	24.70								
	Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)		UMK	PSUMK		0.19	0.19								
LINE SHARING															
	Line Sharing Splitter, per System 96 Line Capacity		ULS	ULSDA	187.17	183.33	0.00	0.00	0.00		0.00		<del>                                     </del>	<del>                                     </del>	+
	Line Sharing Splitter, per System 24 Line Capacity		ULS	ULSDB	46.79	183.33	0.00	0.00	0.00		0.00				
	Line Sharing Splitter, Per System, 8 Line Capacity Line Sharing - per Line Activation			ULSD8 ULSDC	15.59 0.61	183.33 17.97	0.00 10.29	0.00	0.00		0.00 15.20			<u> </u>	
	Line Sharing - per Subsequent Activity per Line Rearrangemen			ULSDS	0.01	15.91	7.95	0.00	0.00		15.20				
	Line Sharing-DLEC Owned Splitter in CO-CFA activaton-deactivation (per LSOD)		ULS	ULSDG		83.98		0.00							
UNBUNDLED TRANSP	PORT														
			<u> </u>										<del> </del>	<u> </u>	
NOTE: IN	TEROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below D	) S3 = 0	ne month	n, DS3 and	above four mc	nths									
INTEROE	FICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE		<u> </u>										<del> </del>	<u> </u>	
INTERO	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per				-										
	month Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility		U1TVX	1L5XX	0.013										
	Termination per month		U1TVX	U1TV2	22.60	39.36	26.62				15.20				
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per month		U1TVX	1L5XX	0.013										
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination per month		U1TVX	U1TR2	22.60	39.36	26.62	0.00	0.00		15.20				
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month		U1TVX	1L5XX	0.013	39.30	20.02	0.00	0.00		15.20				
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility														
	Termination per month		U1TVX	U1TV4	19.81	39.36	26.62				15.20		<del> </del>		
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month		U1TDX	1L5XX	0.013										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month		U1TDX	U1TD5	15.61	39.37	26.62				15.20				
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per montr		U1TDX		0.013	00.07	20.02				10.20				
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month		U1TDX	U1TD6	15.61	39.37	26.62	0.00	0.00		15.20				
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT - DS1														<del>                                     </del>
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			1L5XX	0.2652	20.05	70.41		-		15.00				
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month		ומווט ווט	U1TF1	70.47	86.69	79.44				15.20		<u> </u>		<u> </u>
INTEROF	FICE CHANNEL - DEDICATED TRANSPORT- DS3  Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month		LIATES	1L5XX	6.04				-				<u> </u>	<u> </u>	
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per montr  Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per montr			U1TF3	850.45	270.69	158.05				15.20		<del>                                     </del>		+
INITEDOS	FICE CHANNEL - DEDICATED TRANSPORT- STS-1														
INTEROF	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month		U1TS1	1L5XX	6.04										+
		1													1

						I	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc	-	Nonrec	urring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'l
					Rec	First	Add'l	Nonrecurrin First	g Disconnect Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CHANNEL - DEDICATED TRANSPORT														
NOTE: L	OCAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one	month													<b></b>
	Local Channel - Dedicated - 2-Wire Voice Grade Per Month			ULDV2	18.32	187.51	32.21	0.00	0.00		15.20				
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per montt  Local Channel - Dedicated - 4-Wire Voice Grade per montt		UNDVX	ULDR2	18.32 19.41	187.51 187.94	32.21 32.63	0.00	0.00		15.20 15.20				
	Local Channel - Dedicated - 4-Wife Voice Grade per month  Local Channel - Dedicated - DS1 per month - Zone 1	1	ULDD1		39.18	172.34	149.27				15.20				
	Local Channel - Dedicated - DS1 per month - Zone 2	2	ULDD1		121.58	172.34	149.27				15.20				
	Local Channel - Dedicated - DS1 per month - Zone 3		ULDD1		70.02	172.34	149.27				15.20				
	Local Channel - Dedicated - DS3 - Per Mile per month		ULDD3	1L5NC	7.82										
	Local Channel - Dedicated - DS3 - Facility Termination per month		ULDD3	ULDF3	469.44	438.46	256.30				15.20				
	Local Channel - Dedicated - STS-1- Per Mile per month		ULDS1	1L5NC	7.82										
	Local Channel - Dedicated - STS-1 - Facility Termination per month		ULDS1	ULDFS	457.22	438.46	256.30	0.00	0.00		15.20				
MULTIPLEXERS		-													
MULTIPLEXERS	Channelization - DS1 to DS0 Channel System		UXTD1	MO1	105.09	88.41	60.76				15.20				<del></del>
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)		UDL	1D1DD	1.38	6.39	4.58				15.20				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month			UC1CA	2.96	6.39	4.58								
	Voice Grade COCI - DS1 to DS0 Channel System - per month		UEA	1D1VG	0.6497	6.39	4.58								
	DS3 to DS1 Channel System per month		UXTD3		201.48	172.99	91.25				15.20				
	STS1 to DS1 Channel System per month		UXTS1		201.48	172.99	91.25				15.20				
	DS3 Interface Unit (DS1 COCI) used with Loop per month		USL	UC1D1	11.78	6.39	4.58								
DARK FIBER															
DARKTIBER	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local Channel		UDF	1L5DC	52.23										
	NRC Dark Fiber - Local Channel		UDF	UDFC4		620.60	133.88				15.20				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -														
	Interoffice Channel NRC Dark Fiber - Interoffice Channe		UDF UDF	1L5DF UDF14	25.28	620.60	133.88	0.00	0.00		15.20				
	Dark Fiber - Interoffice Channe Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month - Local Loop		UDF	1L5DL	52.23	620.60	133.88	0.00	0.00		15.20				
	NRC Dark Fiber - Local Loop		UDF	UDFL4		620.60	133.88	0.00	0.00		15.20				
TRANSPORT OTHER															
Optional	Features & Functions:														
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channe		LINC1X	CCOEF		184.65	23.70	1.97	0.77			29.20	3.92		
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channe			CCOSF		184.65	23.70	1.97	0.77			29.20			i
8XX ACCESS TEN DI															
	8XX Access Ten Digit Screening, Per Cal		OHD		0.0006387										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserve		OHD	N8R1X		2.51	0.43				15.20				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translation:		OHD			5.77	0.78				15.20				
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translation:		OHD	N8FTX		5.77	0.78				15.20				l
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Numbe		OHD	N8FCX		2.51	1.26				15.20				
	8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR Requested Per 8XX No		OHD	N8FMX N8FAX		2.93 2.93	1.68 0.43				15.20				
	8XX Access Ten Digit Screening, Change Charge Per Reques 8XX Access Ten Digit Screening, Call Handling and Destination Feature:			N8FAX N8FDX		2.93	0.43				15.20 15.20				<del></del>
	8XX Access Ten Digit Screening, Call Handling and Destination Feature:		OHD	HOLDY	0.0006387	2.51					10.20				
	8XX Access Ten Digit Screening, w/ POTS No. Delivery, per quen		OHD		0.0006387										
LINE INFORMATION	DATA BASE ACCESS (LIDB)														
	LIDB Common Transport Per Query		OQT		0.0000221										
	LIDB Validation Per Query	1	OQU		0.0135077						1			l	

						ı	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	всѕ	usoc		Nonrec	urring	Nonrooveri	ng Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'l
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	LIDB Originating Point Code Establishment or Change		OQT, OQU	NRPBX		33.33					15.20				
SIGNALING (CCS7)															
(000)	CCS7 Signaling Termination, Per STP Port		UDB	PT8SX	147.60						15.20				
	CCS7 Signaling Usage, Per TCAP Message		UDB		0.000064										
	CCS7 Signaling Connection, Per link (A link)		UDB	TPP++	15.77	34.50					15.20				
	CCS7 Signaling Connection, Per link (B link) (also known as D link)		UDB	TPP++	15.77	34.50	34.50				15.20				
	CCS7 Signaling Usage, Per ISUP Messag€		UDB		0.000016										
	CCS7 Signaling Usage Surrogate, per link per LATA		UDB	STU56	732.10						15.20				
	CCS7 Signaling Point Code, per Originating Point Code Establishment or Change,														ι ¬
	per STP affected		UDB	CCAPO		28.17	28.17				15.20				
	CCS7 Signaling Point Code, per Destination Point Code Establishment or Change,	l T		7	$\neg$										ι 7
	Per Stp Affected		UDB	CCAPD		28.17	28.17				15.20				
															<b></b>
E911 SERVICE									1						<b></b>
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 1				18.32	187.51	32.21		1		15.20				<b></b>
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 2				18.32	187.51	32.21				15.20				
	Local Channel - Dedicated - 2-wr Voice Grade - Zone (				18.32	187.51	32.21				15.20				
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile				0.013	20.20	00.00				45.00				<b>——</b>
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Terminatior Local Channel - Dedicated - DS1 - Zone 1				22.60 39.18	39.36 172.34	26.62 149.27				15.20				
	Local Channel - Dedicated - DS1 - Zone 1  Local Channel - Dedicated - DS1 - Zone 2				121.58	172.34	149.27				15.20 15.20				
	Local Channel - Dedicated - DS1 - Zone 2  Local Channel - Dedicated - DS1 - Zone 3				70.02	172.34	149.27				15.20				
	Interoffice Transport - Dedicated - DS1 - Zone 3				0.2652	172.34	149.21				15.20				
	Interoffice Transport - Dedicated - DS1 Per Facility Termination				70.47	86.69	79.44				15.20				
	interentice transport Dedicated Deliter admity reminiation				10.41	00.00	70.44				10.20				
CALLING NAME (CNA	M) SERVICE														
,	CNAM for DB Owners, Per Query		OQV		0.0010217										
	CNAM for Non DB Owners, Per Query		OQV		0.0010217										
	CNAM For DB Owners - Service Establishment		OQV			22.29					15.20				
	CNAM For Non DB Owners - Service Establishmen		OQV			22.29					15.20				
	CNAM For DB Owners - Service Provisioning With Point Code Establishmen		OQV			962.22	711.64				15.20				
	CNAM For Non DB Owners - Service Provisioning With Point Code Establishmen		OQV			332.43	238.05				15.20				
	CNAM (Non-Databs Owner), NRC, applies when using the Character Based User		001/	000011		505.00	505.00				45.00				
	Interface (CHUI)		OQV	CDDCH		595.00	595.00				15.20				<b>——</b>
LNP QUERY SERVICE															
LIVE WOLK I SERVICE	-														<u> </u>
	LNP Charge Per query				0.0008559				1		1				$\overline{}$
	LNP Service Establishment Manual				3.0000009	12.16			1		1				
	LNP Service Provisioning with Point Code Establishment					576.33	294.43								$\overline{}$
						2.2.00									
OPERAT	OR SERVICES AND DIRECTORY ASSISTANCE														
OPERATOR CALL PR	OCESSING				-										<del>                                     </del>
OF ERATOR CALL PR					1.20				-		1				<del></del>
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDE Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDE				1.20				1	1	1				<del></del>
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDE  Oper. Call Processing - Fully Automated, per Call - Using BST LIDE			1	0.20				+		1		1		
	Oper. Call Processing - Fully Automated, per Call - Using Bot LIDE				0.20				1		1				<del></del>
	open. Can't lococomy if any natomatou, per Can't Comy religing LIDE				0.20										$\overline{}$
INWARD OPERATOR	SERVICES								1		1				
3. 2 OK	Inward Operator Services - Verification, Per Minute				1.15				1		1				
	Inward Operator Services - Verification, 1 et Minute  Inward Operator Services - Verification and Emergency Interrupt - Per Minute				1.15										$\overline{}$
	The second secon				5										
BRANDING - OPERAT	OR CALL PROCESSING														
	Recording of Custom Branded OA Announcemen			CBAOS		7,000.00	7,000.00				15.20				
	Loading of Custom Branded OA Announcement per shelf/NA\			CBAOL		500.00	500.00				15.20				
						,,,,,,	,			1		1	1	1	

DIRECTORY ASSISTANCE ACCESS SERVICE (DAC)						ļ	RATES (\$)					OSS R	ATES (\$)		
DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE CALL COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPLETION ACCESS SERVICE (CALL)  DIRECTORY ASSET NATE COMPL	CATEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec	urring	Nonrecurrin	g Disconnect	Submitted Elec per LSR	Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-1st	Charge - Manual Svc Order vs. Electronic-Add'l	Charge - Manual Svc Order vs. Electronic-Disc 1st	Charge - Manual Svc Order vs. Electronic-Disc Add'I
Loading of OA yee OCH (Regoral)	Linhranding via OLNS for LINER CLEC				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
DRECTORY ASSETTANCE ACCRESS REWICE  DRECTORY ASSETTANCE ACCRESS REWICE  DRECTORY ASSETTANCE ACCRESS REWICE (DAC)  DRECTORY ASSETTANCE ACCRESS REWICE (DAC)  DRECTORY ASSETTANCE ACCRESS REWICE (DAC)  DRECTORY ASSETTANCE ACCRESS REWICE (DAC)  DRECTORY ASSETTANCE ACCRESS REWICE (DAC)  DRECTORY ASSETTANCE ACCRESS REWICE (DAC)  SIN A Common Transport per Directory Assettance Access Service Cell Mile  DRECTORY ASSETTANCE ACCRESS REWICE (DAC)  SIN A Common Transport per Directory Assettance Access Service Cell Mile  DRECTORY ASSETTANCE DATA BASE SERVICE (DAC)  DRECTORY ASSETTANCE DATA BASE SERVICE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE DATA BASE						1 200 00	1 200 00								
DIRECTORY ASSISTANCE ACCESS SERVICE (DAC)	25dding of 571 per 5511 (Hogishal)					1,200.00	1,200.00								
Directory Assistance Access Service (DACC), Per Call Attemp	DIRECTORY ASSISTANCE SERVICES														
### DIRECTION ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)															<b>├</b>
DIRECTORY TRANSPORT	Directory Assistance Access Service Calls, Charge Per Cal				0.25										
DIRECTORY TRANSPORT	DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)														
SVIA Common transport per Directory Assistance Access Service Call   0.0003					0.10										[
SVIA Common transport per Directory Assistance Access Service Call   0.0003															
SVIA Common transport per Directory Assistance Access Service Call   0.0003	DIDECTORY TRANSPORT														<u> </u>
SWA Common Transport part Directory Assistance Access Service Call   0.00004   0.000					0.0003										
Access Tandem Switching per Directory Assistance Access Service Cal   0.000055   0.000   0.000055															
DRECTORY ASSISTANCE   DRECTORY ASSISTANCE															
DRECTORY ASSISTANCE   DRECTORY ASSISTANCE	3. 1														
DIRECTOR   ASSISTANCE DATA BASE SERVICE (DADS)   Directory Assistance Data Base Service (Drazpe Per Litting   Discours   Discours (Drazpe Per Litting   Discours   Discours (Drazpe Per Litting   Discours   Discours   Discours (Drazpe Per Litting   Discours   Disc															ļl
Directory Assistance Data Base Service Change Per Listing   Descory Assistance Data Base Service, per mort   Descory Assistance Data Base Service, per per per per per per per per per per	DS3 to DS1 Multiplexer per DA Access Service Cal				0.00018										
Directory Assistance Data Base Service Change Per Listing   Descory Assistance Data Base Service, per mort   Descory Assistance Data Base Service, per per per per per per per per per per	DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS)														
DRANDING - IDRICOTORY ASSIStance Data Base Service, per mont!   DRSOF   150.00					0.04										
Facility Based CLEC   Recording and Provisioning of DA Custom Branded Announcemen   AMT   CBADA   6,000.00   6,000.00	Directory Assistance Data Base Service, per month			DBSOF	150.00										
Recording and Provisioning of DA Custom Branded Announcement per DRAM Card/Switcl   AMT   BADA   6,000.00   6,000.00															ļ!
Loading of Custom Branded Announcement per DRAM Card/Switch   AMT   CBADC   1,170,00   1,170,00			ANAT	CDADA		0.000.00	0.000.00								<del>                                     </del>
UNEF CLEC Recording of DA Custom Branded Announcemen   3,000,00   1,170,00															
Recording of DA Custom Branded Announcement   3,000,000   1,170,			AIVII	CBADC		1,170.00	1,170.00								
Loading of DA Custom Branded Announcement per DRAM Card/Switch per OCN						3.000.00	3.000.00								
Unbranding via OLNS for UNEP CLEC   Loading of DA per CN(1 CON per Order)   420.00						-,									
Loading of DA per OCN (1 OCN per Order)						1,170.00	1,170.00								
Loading of DA per Switch per OCN   16.00   16.00   16.00						100.00	400.00								
Selective Routing   Selective Routing Per Unique Line Class Code Per Request Per Switcl   USRCR   82.25   82.25   15.20															
Selective Routing Per Unique Line Class Code Per Request Per Switc	Loading of DA per dwitch per don't					10.00	10.00								
Selective Routing Per Unique Line Class Code Per Request Per Switc															
Virtual Collocation - 2-wire Cross Connects (loop)   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   UEPSR   VETLS   UEPSR	SELECTIVE ROUTING														
Virtual Collocation - 2-wire Cross Connects (loop)   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   VETLS   UEPSR   UEPSR   UEPSR   VETLS   UEPSR	Colorativa Boutina Bookhala Clara Coda Bookhala Contain			LICDOD		00.05	00.05				45.00				[
Ueanl,   U	Selective Routing Per Unique Line Class Code Per Request Per Switch			USRCR		82.25	82.25				15.20				
Virtual Collocation - 2-wire Cross Connects (loop)   UEAC2	VIRTUAL COLLOCATION														
Virtual Collocation - 2-wire Cross Connects (loop)   UEAC2															
Virtual Collocation - 2-wire Cross Connects (loop)   Virtual Collocation - 2-wire Cross Connects (Loop) for Line Splitting   UEPSR   UEPSR   VETLS   0.0296   11.94   11.46   0.00   0.00   15.20     VIrtual Collocation - 2-wire Cross Connect, Exchange Port 2-Wire Analog - Re:   UEPSR   VETLS   0.26   23.04   22.11   11.46   0.00   0.00   15.20     VIrtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade Res   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire Line Side PBX   Trunk - Bus   UEPSR   VETLS   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX   Trunk - Res   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26   23.04   22.11   VIRTUAL COLLOCATION 2-Wire Cross Connect, Exchange Port 2-Wire ISDN   UEPSR   VETR2   0.26															l l
Virtual Collocation - 2-wire Cross Connects (Loop) for Line Splitting															l l
Virtual Collocation - 2 Wire Cross Connects (Loop) for Line Splitting   UEPSR   VE1LS   0.0296   11.94   11.46   0.00   0.00   15.20	Virtual Collocation - 2-wire Cross Connects (loop)			UEAC2	0.0296	11.94	11.46								l l
Virtual Collocation - 2 Wire Cross Connects (Loop) for Line Splitting					1.12.30										
Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-Wire Analog - Res   UEPSR   VER2   0.26   23.04   22.11   19.99															
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Line Side PBX   UEPRX   PE1R2   0.26   23.04   22.11   19.99								0.00	0.00		15.20	40.00	40.00	40.00	10.00
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Line Side PBX   UEPSP   VE1R2   0.26   23.04   22.11   19.99	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-Wire Analog - Res		UEPSK	vEIR2	0.26	23.04	22.11					19.99	19.99	19.99	19.99
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Line Side PBX   UEPSP   VE1R2   0.26   23.04   22.11   19.99	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade Res		UEPRX	PE1R2	0.26	23.04	22.11					19.99	19.99	19.99	19.99
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX   UEPSE   VE1R2   0.26   23.04   22.11   19.99   19.9	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Line Side PBX														
Trunk - Res		+	UEPSP	VE1R2	0.26	23.04	22.11					19.99	19.99	19.99	19.99
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus   UEPSB VETR2   0.26   23.04   22.11   19.99   19.			HEPSE	VF1R2	0.26	23 04	22 11					19 99	19 99	19 99	19 99
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN         UEPSX VE1R2         0.26         23.04         22.11         19.99         1															
Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN         UEPTX         VER2         0.26         23.04         22.11         19.99 <td>Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN</td> <td></td> <td>UEPSX</td> <td>VE1R2</td> <td>0.26</td> <td>23.04</td> <td>22.11</td> <td></td> <td></td> <td></td> <td></td> <td>19.99</td> <td>19.99</td> <td>19.99</td> <td>19.99</td>	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN		UEPSX	VE1R2	0.26	23.04	22.11					19.99	19.99	19.99	19.99
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN		UEPTX	VE1R2		23.04	22.11					19.99	19.99	19.99	19.99
	Virtual Collocation A-Wire Cross Connect, Evahones Bort DDITS 4 Wire DS4		HEDDO	\/E1D4	0.50	າວ າວ	22.24					10.00	10.00	10.00	10.00
	Virtual Collocation 4-Wire Cross Connect, Exchange Port DDITS 4-Wire DS1  Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1				0.52	23.23	22.24					19.99		19.99	19.99

						ı	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec	urring	Nonrecurri	ng Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'l
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation - 4-wire Cross Connects (loop)		uea,uhl, ucl,udl	UEAC4	0.0591	12.04	11.53								
	Virtual Collocation - 2-Fiber Cross Connects			CNC2F	2.65	20.29	14.76								
	Virtual Collocation - 4-Fiber Cross Connects			CNC4F	5.31	24.81	19.29								
	Villadi Collocation - 1 ibol Clock Collingue		USL,UL	0.10	0.01	2	10.20								
	Virtual Collocatin - DS1 Cross Connects			CNC1X	1.04	21.39	15.47								
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per	r													
	linear foot		AMTFS	PE1ES	0.0024										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support														
	Structure, per linear ft		AMTFS	PE1DS	0.0036									1	
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per														
	cable		AMTFS			534.79									
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support					-				1					
	Structure, per cable		AMTFS			534.79									
AIN SELECTIVE CAR															
	Regional Service Establishment			SRCEC		100,209.33					15.20				
	End Office Establishment			SRCEO		164.29	164.29				15.20				
	Query NRC, per query		SRC		0.0030293										
AIN - BELLSOUTH A	N SMS ACCESS SERVICE														
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup			CAMSE		38.30	38.30				15.20				
	AIN SMS Access Service - Port Connection - Dial/Shared Access			CAMDP		7.60	7.60				15.20				
	AIN SMS Access Service - Port Connection - ISDN Access			CAM1P		7.60	7.60				15.20				
	AIN SMS Access Service - User Identification Codes - Per User ID Code			CAMAU		33.99	33.99				15.20				
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			CAMRC		41.39	41.39				15.20				
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)				0.0022										
	AIN SMS Access Service - Session, Per Minute				0.5795										
	AIN SMS Access Service - Company Performed Session, Per Minute				0.8104										
AIN - BELLSOUTH A	N TOOLKIT SERVICE														
	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			BAPSC		38.30	38.30				15.20				
	AIN Toolkit Service - Training Session, Per Customer			BAPVX		4,175.10	4,175.10		1		15.20			1	
														1	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt	<u> </u>		BAPTT		7.60	7.60				15.20				
			1						1					1	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay	1		BAPTD		7.60	7.60		1		15.20			1	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook			DART										1	
	Immediate			BAPTM		7.60	7.60		1		15.20			1	
	AINI Taglicit Carriago Trigger Assess Charact Day Trianger Day 2000			DARTO		20 4-	00.4-				45.00			1	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP	<u> </u>		BAPTO		33.47	33.47		1		15.20			-	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDF	<u> </u>		BAPTC		33.47	33.47		1		15.20			-	
	AIN Toulist Consider Triangue Assess Observe Dou Triangue De 201 5 / O 1		1	DADTE					1		45.55			1	
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code	-		BAPTF	0.0500446	33.47	33.47		1		15.20			1	-
	AIN Toolkit Service - Query Charge, Per Query	ļ			0.0536446				1		1			1	
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node,				0.000500										
	Per Query	1			0.006569				1		1			1	
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100				0.55										
	Kilobytes	-			0.06				1		1			1	-
	AIN Tabilit Carries Monthly raport Des AIN Tabilit Carries Culture C		1	BAPMS	10.00	7.00	7.0-		1		45.00			1	
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription		-		10.90	7.60	7.60		-	-	15.20				
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription	1	-	BAPLS	2.80	8.41	8.41		-		15.20			-	
	AIN Toolkit Sonice Call Event Penert Per AIN Toolkit Sonice Subscription			BAPDS	8.20	7.60	7.60				15.20				
1	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription	1	1	DAPUS	8.∠0	7.00	00.1			1	15.∠0		1	1	

					ı	RATES (\$)				ı	OSS R	ATES (\$)	ı	
CATEGORY UNBUNDLED NETWORK ELEMENT	Zc	ne BCS	usoc		Nonrec	urring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increme Charge Manual S Order v Electronic Add'I
					First		Nonrecurrin First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	
AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Sul	bscription		BAPES	Rec 0.09	8.41	Add'I 8.41	FIFSt	Addi	SOMEC	15.20	SOMAN	SOMAN	SOMAN	SOMA
DOUF/ADUF/CMDS														
ACCESS DAILY USAGE FILE (ADUF)														
ADUF: Message Processing, per message				0.007983										
ADUF: Data Transmission (CONNECT:DIRECT), per message				0.00012681										
ENHANCED ORTIONAL DAILY HOLDE EILE (FORHE)														
ENHANCED OPTIONAL DAILY USAGE FILE (EODUF)  EODUF: Message Processing, per message				0.250015										
LODOI . Message i focessilly, per message				0.230015										
OPTIONAL DAILY USAGE FILE (ODUF)														
ODUF: Recording, per message				0.0000117										
ODUF: Message Processing, per message ODUF: Message Processing, per Magnetic Tape provisioned			-	0.004641 48.45										
ODUF: Data Transmission (CONNECT:DIRECT), per message				0.00010568										
				0.000.000										
CED EXTENDED LINK (EELs)														
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: O NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently comannly.)	. Use all rate	s below e	xcept Swi	tch As Is Charg	e.		lies to curre	ntly combine	ed facilities	converted to	O UNEs.(Non-r	ecurring rate	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC	. Use all rate	es below e	are conve	tch As Is Charg	e.		lies to curre	ntly combine	ed facilities	converted to	O UNEs.(Non-r	ecurring rate	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently comapply.)	. Use all rate	es below e	are conve	tch As Is Charg	e.		lies to curre	ntly combine	ed facilities	converted to	) UNEs.(Non-r	ecurring rate	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC  NOTE: In all states, EEL network elements shown below also apply to currently comapply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/  First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z	d network el	es below e dies which ements.(N	are conve	tch As Is Charg	e.		lies to curre	ntly combine	ed facilities	converted to	o UNEs.(Non-r	ecurring rate	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC  NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/  First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2	d network el	es below entires which ements.(N	are conve	tch As Is Charge orted to UNE rate As Is Charge.)	e. es. A Switch As	Is Charge app	lies to curre	ntly combine	ed facilities		D UNEs.(Non-r	ecurring rate:	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC  NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined apply.  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/  First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina	d network el  ANSPORT (I	es below et ies which ements.(N	are converse some some some some some some some so	tch As Is Charge orted to UNE rate As Is Charge.)	e. es. A Switch As 94.21	45.09	0.00	0.00	ed facilities	15.20 15.20	D UNEs.(Non-r	ecurring rates	s do not	
NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined apply.)  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT STATES AND ADS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 2 Dne 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 3 Dne 3	d network el	es below esties which ements.(No. 1971) UNCV UNCV UNCV	are converse of Switch A  X UEAL2  X UEAL2  X UEAL2	tch As Is Charge orted to UNE rate As Is Charge.) 14.93 25.35 50.46	e. es. A Switch As	Is Charge app			ed facilities	15.20	o UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC  NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined apply.  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/  First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina	d network el  ANSPORT (I	es below esties which ements.(No. 1971) UNCV UNCV UNCV	xcept Swi are conve to Switch / X UEAL2 X UEAL2 X UEAL2 X UEAL2 X 1L5XX	tch As Is Charge orted to UNE rate As Is Charge.)	e. es. A Switch As 94.21	45.09	0.00	0.00	ed facilities	15.20 15.20	o UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Interoffice Transport - Dedicated - DS1 combination - Facility Termination p	LUSE All rate nbined facility depends on the second state of the second	ements.(N UNCV UNCV UNCV UNCV UNCV UNCV	xcept Swi are conve  o Switch A  X UEAL2  X UEAL2  X UEAL2  X UEAL2  X UEAL2  X UITF1	tch As Is Charge erted to UNE rate As Is Charge.) 14.93 25.35 50.46 0.2652 70.47	94.21 94.21 94.21	45.09 45.09 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20	o UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per montf Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Montf	LUSE All rate nbined facility depends on the second state of the second	ements.(N	are converse values of Switch A VIEAL2 VIEAL	tch As Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09	94.21 94.21 94.21 143.58 59.97	45.09 45.09 45.09 103.88 12.96	0.00	0.00	ed facilities	15.20 15.20	o UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC  NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined apply.  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/  First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 2 Zone 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month  Interoffice Transport - Dedicated - DS1 combination - Facility Termination p  DS1 Channelization System Per Month  Voice Grade COCI - DS1 To Ds0 Interface - Per Month	c. Use all rational r	ements.(N	xcept Swi are conve  o Switch A  X UEAL2  X UEAL2  X UEAL2  X UEAL2  X UEAL2  X UITF1	tch As Is Charge erted to UNE rate As Is Charge.) 14.93 25.35 50.46 0.2652 70.47	94.21 94.21 94.21	45.09 45.09 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20	o UNEs.(Non-r	ecurring rate	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per montf Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Montf	c. Use all rational r	es below es which ements.(N UNCV UNCV UNCV UNCV UNCV UNCV UNCL UNCL UNCL UNCL UNCL UNCL UNCL UNCL	are converse values of Switch A VIEAL2 VIEAL	tch As Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09	94.21 94.21 94.21 143.58 59.97	45.09 45.09 45.09 103.88 12.96	0.00	0.00	ed facilities	15.20 15.20 15.20	D UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont!  Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Mont!  Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport	c. Use all rational r	ements.(N EEL) UNCV UNCV UNC1. UNC1. UNC1. UNC1. UNC1. UNCV	xcept Switch // x UEAL2 X UEAL2 X UEAL2 X UEAL2 X 1L5XX X U1TF1 X MQ1 X 1D1VG	tch As Is Charge,  as Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09  0.6497	94.21 94.21 94.21 143.58 59.97 5.91	45.09 45.09 45.09 103.88 12.96 4.26	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20	o UNEs.(Non-r	ecurring rate	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC  NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/  First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile per montf  Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Montf  Voice Grade COCI - DS1 To Ds0 Interface - Per Month  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2	c. Use all rational r	ements.(N EEL) UNCV UNCV UNC1 UNC1 UNC1 UNCV	are converse of Switch //	tch As Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09  0.6497  14.93	94.21 94.21 94.21 143.58 59.97 5.91 94.21	45.09 45.09 45.09 103.88 12.96 4.26 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20	D UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Mont/ Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2	d network el  ANSPORT (I  cone 1  dition -  di	ements.(No. 1) UNCV UNCV UNCV UNCV UNCV UNCV UNCV UNCV	xcept Switch // x UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UITF1 X MQ1 X 1D1VG X UEAL2 X UEAL2 X UEAL2	tch As Is Charge, reted to UNE rate As Is Charge,  14.93 25.35 50.46 0.2652 70.47 105.09 0.6497 14.93 25.35 50.46	94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21	45.09 45.09 45.09 103.88 12.96 4.26 45.09 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20	D UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per montf Interoffice Transport - Dedicated - DS1 combination - Per Mile per montf Voice Grade COCI - DS1 To DS0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2	d network el  ANSPORT (I  cone 1  dition -  di	ements.(No. 1) UNCV UNCV UNCV UNCV UNCV UNCV UNCV UNCV	are converse of Switch //	tch As Is Charge, reted to UNE rate As Is Charge,  14.93 25.35 50.46 0.2652 70.47 105.09 0.6497 14.93 25.35 50.46	94.21 94.21 94.21 143.58 59.97 5.91 94.21	45.09 45.09 45.09 103.88 12.96 4.26 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20	D UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined  2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Mont/ Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2	C. Use all rational r	ements.(N  EEL)  UNCV  UNCV  UNC1  UNC1  UNC1  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV	xcept Switch // x UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UITF1 X MQ1 X 1D1VG X UEAL2 X UEAL2 X UEAL2	tch As Is Charge, reted to UNE rate As Is Charge,  14.93 25.35 50.46 0.2652 70.47 105.09 0.6497 14.93 25.35 50.46	94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21	45.09 45.09 45.09 103.88 12.96 4.26 45.09 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20	D UNEs.(Non-r	recurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per montf Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Montf Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per montf	c. Use all rate his	ements.(N  EEL)  UNCV	xcept Switch // o Switch // x UEAL2 X UEAL2 X UEAL2 X 1L5XX X 1D1VG X UEAL2 X UEAL2 X UEAL2 X 1D1VG	tch As Is Charge, reted to UNE rate As Is Charge,  14.93 25.35 50.46 0.2652 70.47 105.09 0.6497 14.93 25.35 50.46	94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21 94.21	45.09 45.09 45.09 103.88 12.96 4.26 45.09 45.09 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20 15.20	D UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WiRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per montf Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	d network el  ANSPORT (I  continued facili  d network el  ANSPORT	ements.(N EEL)  UNCV UNCV UNCV UNCV UNCV UNCV UNCV UNC	xcept Switch // x UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X IL5XX X U1TF1 X MQ1 X 1D1VG X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2	tch As Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09  0.6497  14.93  25.35	94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21 94.21 5.91	45.09 45.09 45.09 103.88 12.96 4.26 45.09 45.09 45.09 45.09 5.43	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20 15.20	D UNES.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3  Interoffice Transport - Dedicated - DS1 combination - Per Mile per montl  Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Montl  Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1  Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3  Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transpor Comb	ANSPORT (I	ements.(N EEL)  UNCV UNCV UNCV UNCV UNCV UNCV UNCV UNC	xcept Switch // o Switch // x UEAL2 X UEAL2 X UEAL2 X 1L5XX X 1D1VG X UEAL2 X UEAL2 X UEAL2 X 1D1VG	tch As Is Charge, reted to UNE rate As Is Charge,  14.93 25.35 50.46 0.2652 70.47 105.09 0.6497 14.93 25.35 50.46	94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21 94.21	45.09 45.09 45.09 103.88 12.96 4.26 45.09 45.09 45.09	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20 15.20	D UNES.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Voice Grade COC1 - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COC1 - DS1 to DS0 Channel System combination - per mont/ Nonrecurring Currently Combined Network Elements Switch -As-Is Charge  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Comb Zone 1 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Comb	ANSPORT (I	es below es below es below es which the sements.(No. 1997)  EEL)  UNCV  UNCV  UNCV  UNC1  UNC1  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV  UNCV	xcept Switch // x UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X IL5XX X U1TF1 X MQ1 X 1D1VG X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2	tch As Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09  0.6497  14.93  25.35	94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21 94.21 5.91	45.09 45.09 45.09 103.88 12.96 4.26 45.09 45.09 45.09 45.09 5.43	0.00	0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20 15.20	D UNES.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 2 Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per montly Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Montl Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Comb Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Comb	C. Use all rate  Indicate the second	ements.(N EEL)  UNCV UNCV UNCV UNCV UNCV UNCV UNCV UNC	xcept Swi are conve  o Switch /  X UEAL2 X UEAL2 X UEAL2 X 1L5XX X U1TF1 X MQ1 X 1D1VG X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL4 X UEAL4	tch As Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09  0.6497  14.93  25.35  30.81  30.81	94.21 94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21 5.91 5.43	45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09	0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20	D UNEs.(Non-r	ecurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combina Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Interoffice Transport - Dedicated - DS1 combination - Per Mile per mont/ Voice Grade COCI - DS1 To DS0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per mont/ Nonrecurring Currently Combined Network Elements Switch -As-Is Charge  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Comb Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Comb Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Comb	d network el ANSPORT (I cont ort ort ort ort ort ort ort ort ort or	ements.(No. 1) UNCV UNCV UNCV UNCV UNCV UNCV UNCV UNCV	xcept Switch // x UEAL2 x UEAL2 x UEAL2 x UEAL2 x UEAL2 x UEAL2 x UEAL2 x UEAL4 x UEAL4 x UEAL4	tch As Is Charge,  14.93 25.35 50.46 0.2652 70.47 105.09 0.6497 14.93 25.35 50.46 0.6497 30.81 38.32	94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21 94.21 94.21	45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09	0.00	0.00 0.00 0.00 0.00 0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20	D UNEs.(Non-r	recurring rates	s do not	
NOTE: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC NOTE: In all states, EEL network elements shown below also apply to currently com apply.)  NOTE: In GA, TN, KY, & LA, the EEL network elements apply to ordinarily combined 2-WiRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Z First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 2 Zone 2  First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per montly  Interoffice Transport - Dedicated - DS1 combination - Facility Termination p DS1 Channelization System Per Montly Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per montly Nonrecurring Currently Combined Network Elements Switch -As-Is Charge  4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TR/ First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	C. Use all rate  Indicate the second	ements.(No. 1) EEL)  UNCV UNCV UNCV UNCV UNCV UNCV UNCV UNC	xcept Swi are conve  o Switch /  X UEAL2 X UEAL2 X UEAL2 X 1L5XX X U1TF1 X MQ1 X 1D1VG X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UEAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL2 X UPAL4 X UEAL4	tch As Is Charge.)  14.93  25.35  50.46  0.2652  70.47  105.09  0.6497  14.93  25.35  30.81  30.81	94.21 94.21 94.21 94.21 143.58 59.97 5.91 94.21 94.21 5.91 5.43	45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09 45.09	0.00 0.00 0.00 0.00	0.00 0.00 0.00 0.00	ed facilities	15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20	D UNES.(Non-r	ecurring rates	s do not	

						I	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec	curring	Nonrecurring	g Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic-Di: Add'I
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month		UNCVX	1D1VG	0.6497	5.91	4.26								
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1	4	UNCVX	LIEAL 4	30.81	94.21	45.09	0.00	0.00		15.20				
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	_	UNCVX	UEAL4	30.61	94.21	45.09	0.00	0.00		15.20				
	Combination - Zone 2	2	UNCVX	UEAL4	38.32	94.21	45.09	0.00			15.20				
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport														
	Combination - Zone 3	3	UNCVX		60.39	94.21	45.09	0.00	0.00		15.20				
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month		UNCVX	1D1VG	0.6497	5.91	4.26								
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		5.43	5.43	0.00	0.00		15.20				
4-WIRE 56	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSP	ORT (I	EEL)												
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination	Ì													
	Zone 1	1	UNCDX	UDL56	30.99	94.21	45.09				15.20				
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	2	UNCDX	LIDLES	36.78	94.21	45.09	0.00	0.00		15.20				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination		UNCDX	UDL56	36.78	94.21	45.09	0.00	0.00		15.20				
	Zone 3	3	UNCDX	UDL56	38.92	94.21	45.09	0.00	0.00		15.20				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		UNC1X		0.2652										
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per														
	Month		UNC1X		70.47	143.58	103.88	0.00			15.20				
	Channelization - Channel System DS1 to DS0 combination Per Month OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)		UNC1X UNCDX		105.09 1.38	59.97 5.91	12.96 4.26	0.00	0.00						
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		ONCDA	10100	1.50	3.91	4.20								
	Combination - Zone 1	1	UNCDX	UDL56	30.99	94.21	45.09	0.00	0.00		15.20				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport														
	Combination - Zone 2	2	UNCDX	UDL56	36.78	193.82	92.77	82.08	12.22		15.20				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3	3	UNCDX	UDL56	38.92	193.82	92.77	82.08	12.22		15.20				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-	3	UNCDA	UDLS6	30.92	193.02	92.77	02.00	12.22		15.20				
	64kbs)		UNCDX	1D1DD	1.38	5.91	4.26								
	· ·														
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		5.43	5.43	0.00	0.00		15.20				
1-WIDE 64	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSP	OPT (	EEI \												
4-WIKE 04	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination	UK1 (I	LLL)												
	Zone 1	1	UNCDX	UDL64	30.99	94.21	45.09	0.00	0.00		15.20				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination														
	Zone 2	2	UNCDX	UDL64	36.78	94.21	45.09	0.00	0.00		15.20				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination Zone 3	3	UNCDX	UDL64	38.92	94.21	45.09	0.00	0.00		15.20				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	3	UNC1X		0.2652	34.21	45.09	0.00	0.00		15.20				
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per														
	Month		UNC1X		70.47	143.58	103.88	0.00	0.00		15.20				
	Channelization - Channel System DS1 to DS0 combination Per Month		UNC1X	MQ1	105.09	59.97	12.96	0.00	0.00		15.20				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)		LINCDY	1D1DD	1.38	5.91	4.26								
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDA	טטוטו	1.30	5.91	4.20								
	Combination - Zone 1	1	UNCDX	UDL64	30.99	94.21	45.09	0.00	0.00		15.20				<u></u>
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport														
	Combination - Zone 2  Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	UNCDX	UDL64	36.78	94.21	45.09	0.00	0.00		15.20				
	Combination - Zone 3	3	UNCDX	UDL64	38.92	94.21	45.09	0.00	0.00		15.20				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-	3	CINODA	JDL04	30.92	34.21	45.09	0.00	0.00		10.20				
	64kbs)		UNCDX	1D1DD	1.38	5.91	4.26								
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		LINC1Y	UNCCC		5.43	5.43	0.00	0.00		15.20				
	Informeduring Currently Combined Network Elements Switch -AS-IS Charge		ONCIX	UNCCC		5.43	5.43	0.00	0.00		15.20				
4-WIRE DS	51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT	(EEL)	)												
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1	1	UNC1X		85.70	169.22	100.89				15.20				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2	2	UNC1X		194.96	169.22	100.89	0.00	0.00		15.20				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3	3	UNC1X	USLXX	491.94	169.22	100.89	0.00	0.00		15.20				

						ı	RATES (\$)				1	OSS R	ATES (\$)	Ī	
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	urring	Nonrecurring	g Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I
					Rec	First	Add'I	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Montl Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per		UNC1X	1L5XX	0.2652										
	Month		UNC1X	U1TF1	70.47	143.58	103.88				15.20				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		5.43	5.43	0.00	13.91		15.20				
4-WIRE D	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT	(EEL)	)												
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 1	1	UNC1X		85.70	169.22	100.89				15.20				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 2	2	UNC1X		194.96 491.94	169.22	100.89				15.20				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month	3	UNC1X UNC3X		6.04	169.22	100.89				15.20				
	Interoffice Transport - Dedicated - DS3 - Facility Termination per month		UNC3X		850.45	296.68	121.16				15.20				
	DS3 to DS1 Channel System combination per month		UNC3X	MQ3	201.48	107.05	48.07								
	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1	1	UNC1X UNC1X		11.78 85.70	5.91 169.22	4.26 100.89	0.00	0.00		15.20				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1  Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2	2	UNC1X		194.96	169.22	100.89	0.00	0.00		15.20				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone (	3	UNC1X		491.94	169.22	100.89	0.00	0.00		15.20				
	DS3 Interface Unit (DS1 COCI) combination per month		UNC1X		11.78	5.91	4.26								
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC3X	UNCCC		5.43	5.43	0.00	0.00		15.20				
2-WIRE V	OICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPOR	T (EEL	L)												
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1	1	UNCVX	UEAL2	14.93	94.21	45.09				15.20				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2	2	UNCVX	UEAL2	25.35	94.21	45.09	0.00	0.00		15.20				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3	3	UNCVX		50.46	94.21	45.09	0.00	0.00		15.20				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month		UNCVX	1L5XX	0.013										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility Termination per month		UNCVX	11471/2	22.60	72.60	41.75				15.20				
	Termination per montin		UNCVX	UTIVZ	22.00	72.00	41.73				15.20				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCVX	UNCCC		5.43	5.43	0.00	0.00		15.20				
4-WIRE V	OICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPOR														
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2	2	UNCVX		30.81 38.32	94.21 94.21	45.09 45.09	0.00	0.00		15.20 15.20				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3	3	UNCVX		60.39	94.21	45.09	0.00	0.00		15.20				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Month	_	UNCVX		0.013	Ş			*****						
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility														
	Termination per month		UNCVX	U1IV4	19.81	72.60	41.75				15.20				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCVX	UNCCC		5.43	5.43	0.00	0.00		15.20				
DS3 DIGI	TAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)														<u></u>
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per		UNC3X		10.04										
+	month Interoffice Transport - Dedicated - DS3 - Per Mile per month		UNC3X		362.34 6.04	188.45	125.51								
	Interoffice Transport - Dedicated - DS3 - Per Mile per monti		UNUSA	ILOAA	0.04										
	month		UNC3X	U1TF3	850.45	296.68	121.16				15.20				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC3X	UNCCC		5.43	5.43	0.00	0.00		15.20				
STS1 DIG	SITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL	)													
	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month		UNCSX	1L5ND	10.04										
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per														
	month		UNCSX		374.56 6.04	188.45	125.51								<u> </u>
	Interoffice Transport - Dedicated - STS1 combination - Per Mile per month  Interoffice Transport - Dedicated - STS1 combination - Facility Termination per		UNCOX	YYCTI	6.04										<del>                                     </del>
	month		LINICOV	U1TFS	830.19	296.68	121.16	0.00	0.00	1	15.20			1	1

CATEGORY  UNBUNDLED NETWORK ELEMENT  Zone  BCS  USOC  BCS  USOC  BCS  USOC  Svc Order Submitted Electronic-Disc Electronic-1st  Nonrecurring  Nonrecurring Disconnect  Nonrecurring Disconnect  Nonrecurring Disconnect  Nonrecurring Disconnect  Nonrecurring Disconnect  Nonrecurring Disconnect  Nonrecurring Disconnect  Svc Order Svc Order Svc Order Vs. Electronic-Disc Electronic-Add'l 1st Add							ı	RATES (\$)				1	OSS R	ATES (\$)	ı	
Noncounting Currently Contributed National Statemant S	CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	curring			Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'l
Average   Company   Comp						Rec	First	Add'l			SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
First ZWINE ISSN Logs in a DST Interoffice Contribution Transport 2-700 2   1,0400		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCSX	UNCCC		5.43	5.43	0.00	0.00		15.20				
First 24/Wer ISSN Loop in a DSI Interdiffic Combination Transport Carelles   1,000 NN, VILLEX   12,000 N	2-WIRE IS	SDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)														
First 2-Year   SDN Loop in a SDR   Interesting Transport - Zeroes   3   MNCNN   U12X   65.19   4.21   45.09   0.00   15.20			1													
Interesting Transport Decideal DSI contribution - Per Mills   MINCIN   ILBAX   0.2562																
Interestinate Transport: Declarated : DST contribution: Facility Termination per month   Declaration: Chaired System DST to DSt contribution: per month   Additional 2-wine (DSM Loop in some DST interestination contribution c			3				94.21	45.09	0.00	0.00		15.20				<del>                                     </del>
Channelstean - Channel System DS1 to DS3 continued for personal Continued on the Continued Con		interestines transport bedicated be combination i et with		ONOTA	TEOXIX	0.2002										
2-wise ISDN COCT (BRITE) - DSI to DS0 Channel System combination - per month   Additional Zerus (DSN Loop in same DSI Interdiffice Transport Combination - Zeru   Additional Zerus (DSN Loop in same DSI Interdiffice Transport Combination - Zeru   Zerus (DSN Loop in same DSI Interdiffice Transport Combination - Zeru   Zerus (DSN Loop in same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in Same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in Same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in Same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in Same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in Same DSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Combination - Zerus   Zerus (DSN Loop in SSI Interdiffice Transport Comb												15.20				<u> </u>
Additional 2-wire IDSN Loop in same DS1 Interdifice Transport Combination - Zone   1 UNCNX U1L2X   22.09   94.21   45.09   0.00   0.00   15.20		Channelization - Channel System DS1 to DS0 combination - per month		UNC1X	MQ1	105.09	59.97	12.96	0.00	0.00						
Additional 2-wire IDSN Loop in same DS1 Interdifice Transport Combination - Zone   1 UNCNX U1L2X   22.09   94.21   45.09   0.00   0.00   15.20		2 wire ISDN COCI (PRITE) DS1 to DS0 Channel System combination		LINCAL	110404	2.00	E 04	4.00								
Additional 2-wire IDSN Loop in same DS Interdifice Transport Combination - Zone   2 UNCNX UTL2X   35.28   94.21   45.09   0.00   0.00   15.20   3 UNCNX UTL2X   35.28   94.21   45.09   0.00   0.00   15.20   3 UNCNX UTL2X   35.28   94.21   45.09   0.00   0.00   15.20   3 UNCNX UTL2X   35.28			1						0.00	0.00		15 20				
2		Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone	'	ONCINA	UTLZX	22.03	34.21	43.03	0.00	0.00		13.20				
3		2	2	UNCNX	U1L2X	35.28	94.21	45.09	0.00	0.00		15.20				1
Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone 3	3	UNCNX	U1L2X	65.18	94.21	45.09	0.00	0.00		15.20				
### Control   Co		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month		UNCNX	UC1CA	2.96	5.91	4.26								
First DS1 Loop in STS1 InterOffice Transport Combination - Zone 1		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		5.43	5.43	0.00	0.00		15.20				<u> </u>
First DS1 Logo in STS1 Interoffice Transport Combination - Zone 2   UNCYX USLXX 194,66   169.22   100.89   0.00   0.00   15.20	4-WIRE D	S1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPOR	II RT (EE	L)												
First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1 2 UNCIX USLXX 491.94 169.22 100.89 0.00 0.00 15.20					USLXX	85.70	169.22	100.89	0.00	0.00		15.20				
Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Mont! UNCSX   1.5XX   6.04				UNC1X	USLXX											
Interoffice Transport - Dedicated - STS1 combination - Facility Termination   UNCSX UTFS   83.0   9   296.68   121.16     15.20			3	UNC1X	USLXX		169.22	100.89	0.00	0.00		15.20				
STS1 to DS1 Channel System combination per month							222.22	101.10				45.00				<b></b>
DS3 Interface Unit (DS1 COCI) combination per month												15.20				<del></del>
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 UNCIX USLXX 85.70 169.22 100.89 0.00 0.00 15.20 15.20 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2 UNCIX USLXX 194.96 169.22 100.89 0.00 0.00 15.20 15.																
Additional DS1Loop in STS1 Interoffice Transport Combination - Zone : 2 UNCIX USLXX 194.96 169.92 100.98 0.00 0.00 15.20   Additional DS1Loop in STS1 Interoffice Transport Combination - Zone : 3 UNCIX USLXX 491.94 169.22 100.98 0.00 0.00 15.20   DS3 Interface Unit (DS1 COCI) combination per month UNCIX UCID1 11.78 5.91 4.26   Nonrecurring Currently Combined Network Elements Switch -As-Is Charge UNCSX UNCCC 5.43 5.43 0.00 0.00 15.20    4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEU)   4-WIRE 56 KBPS LOOpl4-wire 56 kbps Interoffice Transport Combination - Zone : 1 UNCDX UDL56 30.99 94.21 45.09 0.00 0.00 15.20   4-WIRE 56 kbps Loopl4-wire 56 kbps Interoffice Transport Combination - Zone : 2 UNCDX UDL56 38.92 94.21 45.09 0.00 0.00 15.20   Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination UNCDX UTL55 15.61 72.60 41.75   Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination UNCDX UTL55 15.61 72.60 41.75   Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination UNCDX UTL55 15.61 72.60 41.75    4-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 kBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 kBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 kBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 kBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 kBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EXTENDED LOOP WITH 64 kBPS INTEROFFICE TRANSPORT (EEU)    4-WIRE 64 kBPS DIGITAL EX			1						0.00	0.00		15.20				
DS3 Interface Unit (DS1 COCI) combination per month			2							0.00						
Nonrecurring Currently Combined Network Elements Switch -As-Is Charge   UNCSX UNCCC   5.43   5.43   0.00   0.00   15.20			3						0.00	0.00		15.20				
### 4-WIRE 56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)    4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1						11.78										
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1   1   UNCDX   UDL56   30.99   94.21   45.09   0.00   0.00   15.20		· · · · · · · · · · · · · · · · · · ·		UNCSX	UNCCC		5.43	5.43	0.00	0.00		15.20				
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2   2 UNCDX UDL56   38.78   94.21   45.09   0.00   0.00   15.20	4-WIRE 5		L)													
4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone : 3 UNCDX   UNCD			1						0.00	0.00						<del>                                     </del>
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile UNCDX 1L5XX 0.0130																
Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination   UNCDX   U1TD5   15.61   72.60   41.75   15.20			3				34.21	43.03	0.00	0.00		13.20				
4-WiRE 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)   4-Wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1   1 UNCDX UDL64   30.99   94.21   45.09   0.00   0.00   15.20     4-Wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2   2 UNCDX UDL64   36.78   94.21   45.09   0.00   0.00   15.20     4-Wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2   2 UNCDX UDL64   38.92   94.21   45.09   0.00   0.00   15.20     4-Wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport - Dedicated - 4-Wire 64 kbps combination - Per Mile   UNCDX UDL64   38.92   94.21   45.09   0.00   0.00   15.20     Interoffice Transport - Dedicated - 4-Wire 64 kbps combination - Facility Terminatior   UNCDX U1TD6   15.61   72.60   41.75   15.20     Nonrecurring Currently Combined Network Elements Switch -As-Is Charge   UNCDX UNCCC   5.43   5.43   0.00   0.00   0.00   15.20     VICOX U1TD6   15.61   72.60   41.75   15.20							72.60	41.75				15.20				
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1   1 UNCDX UDL64   30.99   94.21   45.09   15.20     4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2   2 UNCDX UDL64   36.78   94.21   45.09   0.00   0.00   15.20     4-wire 64 kbps Interoffice Transport Combination - Zone 3   UNCDX UDL64   36.78   94.21   45.09   0.00   0.00   15.20     4-wire 64 kbps Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Milk   UNCDX   1.5XX   0.0130   UNCDX UDL64   36.78   94.21   45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCDX	UNCCC		5.43	5.43	0.00	0.00		15.20				
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1   1 UNCDX UDL64   30.99   94.21   45.09   15.20     4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2   2 UNCDX UDL64   36.78   94.21   45.09   0.00   0.00   15.20     4-wire 64 kbps Interoffice Transport Combination - Zone 3   UNCDX UDL64   36.78   94.21   45.09   0.00   0.00   15.20     4-wire 64 kbps Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Milk   UNCDX   1.5XX   0.0130   UNCDX UDL64   36.78   94.21   45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00   0.00   0.00   15.20     45.09   0.00	4-WIRF 6	 4 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (FF	L)													<del> </del>
4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2 UNCDX UDL64 36.78 94.21 45.09 0.00 0.00 15.20				UNCDX	UDL64	30.99	94.21	45.09				15.20				
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile UNCDX 1L5XX 0.0130  Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination UNCDX U1TD6 15.61 72.60 41.75 15.20  Nonrecurring Currently Combined Network Elements Switch -As-Is Charge UNCDX UNCCC 5.43 5.43 0.00 0.00 15.20		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2		UNCDX	UDL64		94.21	45.09				15.20				
Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility Termination UNCDX U1TD6 15.61 72.60 41.75 15.20  Nonrecurring Currently Combined Network Elements Switch -As-Is Charge UNCDX UNCCC 5.43 5.43 0.00 0.00 15.20			3				94.21	45.09	0.00	0.00		15.20				<del> </del>
Nonrecurring Currently Combined Network Elements Switch -As-Is Charge UNCDX UNCCC 5.43 5.43 0.00 0.00 15.20							72,60	41,75				15,20				
ADDITIONAL NETWORK ELEMENTS									0.00	0.00						
	ADDITIONAL NETWOR	RK ELEMENTS														

						l	RATES (\$)					OSS R	ATES (\$)	ī	
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc	-	Nonrec	surring		g Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic-Di Add'l
					Rec	First	Add'I	First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	t of a currently combined facility, the non-recurrng charges do not app														
When used as ordina	arilty combined network elements in Georgia, the non-recurring charge	s apply	and the	Switch A	s Is Charge do	es not.									
Access to DCS - Cus	stomer Reconfiguration (FlexServ)														
Node (SynchroNet)															
Treat (cymentory															
Node pe	er month		UNCDX	UNCNT	15.43										
2/4-Wire	ntly Combined Network Elements "Switch As Is" Charge (One applies to B VG Interoffice Channel used in a COMBINATION - "Switch As Is"	each o													
Conversi	sion Charge ops Interoffice Channel used in a COMBINATION - "Switch As Is"		UNCVX	UNCCC		5.43	5.43	0.00	0.00		15.20				
Conversi	sion Charge		UNCDX	UNCCC		5.43	5.43	0.00	0.00		15.20				
DS1 Inte Charge	eroffice Channel used in a COMBINATION - "Switch As Is" Conversion		UNC1X	UNCCC		5.43	5.43	0.00	0.00		15.20				
	eroffice Channel used in a COMBINATION - "Switch As Is" Conversion			UNCCC		5.43	5.43	0.00	0.00		15.20				
STS1 Int	teroffice or Local Loop used in a COMBINATION - "Switch As Is"														
Convers	sion Charge		UNCSX	LINCCC		5.43	5.43	0.00	0.00		15.20				
Conversi				011000			2								
	el - Dedicated Transport - minimum billing period - Below DS3=one mo	nth, DS	3 and ab		months										
NOTE: Local Channe	el - Dedicated Transport - minimum billing period - Below DS3=one mo	nth, DS	3 and ab		months										
NOTE: Local Channe	el - Dedicated Transport - minimum billing period - Below DS3=one mo			ove=four					ions						
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Continued:	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the transport of the service ordering charge currently contained in this rate ex	ne state	specific	ove=four electronic outh regio	service orderin	g charges as ordering of	dered by the Sta	ate Commiss							
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Continued: NOTE: (1) Concluded	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate extended to the contract of the contract negotiator in this rate extended to the contract of the contract	ne state hibit is t	specific he BellSo	ove=four electronic outh regio	service orderin	g charges as ordering of	dered by the Sta	ate Commiss		ing charge.					
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Continued: NOTE: (1) Concluded	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the transport of the service ordering charge currently contained in this rate ex	ne state hibit is t	specific he BellSo	ove=four electronic outh regio	service orderin	g charges as ordering of	dered by the Sta	ate Commiss		ing charge.					
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Continued: NOTE: (1) Concluded NOTE: (2) Manual Se	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates to ervice Order charge: disconnect, in the state of Florida, to be billed on a prefer of the contract of	ne state hibit is t	specific he BellSo	ove=four electronic outh regio	service orderin	g charges as ordering of	dered by the Sta	ate Commiss		ing charge.					
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic : NOTE: (1) Concluded NOTE: (2) Manual Se	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces.	ne state hibit is t	specific he BellSo ectronic s basis	electronic outh regio	service orderin	g charges as or ervice ordering o or CLEC-1 may	dered by the Sta	ate Commiss		ing charge.					
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Continued: NOTE: (1) Concluded NOTE: (2) Manual Se	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces.	ne state hibit is t	specific he BellSo ectronic s basis	ove=four electronic outh regio	service orderin	g charges as ordering of	dered by the Sta	ate Commiss		ing charge.					
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic : NOTE: (1) Concluded NOTE: (2) Manual Se Electronic (Regions The "Zone" shown in	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates to revice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers	ne state hibit is t r the ele er LSR	specific he BellSo ectronic s basis	electronic outh regio ervice ord	service orderin nal electronic se ering charges,	g charges as orr ervice ordering o or CLEC-1 may	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	met Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic : NOTE: (1) Concluded NOTE: (2) Manual Se Electronic (Regions The "Zone" shown in http://www.interconne	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates to revice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refersaction.bellsouth.com/become_a_clec/html/interconnection.htm	ne state hibit is t r the ele er LSR	specific he BellSo ectronic s basis	electronic outh regio ervice ord	service orderin nal electronic se ering charges,	g charges as orr ervice ordering o or CLEC-1 may	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	met Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic : NOTE: (1) Concluded NOTE: (2) Manual Se Electronic (Regions The "Zone" shown in	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates to revice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refersaction.bellsouth.com/become_a_clec/html/interconnection.htm	ne state hibit is t r the ele er LSR	specific he BellSo ectronic s basis	electronic outh regio ervice ord	service orderin nal electronic se ering charges,	g charges as orr ervice ordering o or CLEC-1 may	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Concluded NOTE: (2) Manual Se Electronic (Regional The "Zone" shown in 1 http://www.interconne	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates to revice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refersaction.bellsouth.com/become_a_clec/html/interconnection.htm	ne state hibit is t r the ele er LSR	specific he BellSo ectronic s basis	electronic outh regio ervice ord	service orderin nal electronic se ering charges,	g charges as orr ervice ordering o or CLEC-1 may	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Concluded NOTE: (2) Manual Se Electronic (Regione The "Zone" shown in http://www.interconne UNDLED LOCAL EXCHANGE Exchange Ports	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if its prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates to revice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refersaction.bellsouth.com/become_a_clec/html/interconnection.htm	to Geo	specific sectronic sebasis	electronic outh region ervice ord SOMEC	service orderin nal electronic se ering charges, o	g charges as or ervice ordering of or CLEC-1 may 3.50 es. To view Geo	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe  RATIONAL SUPPORT SYSTE  NOTE: (1) Electronic:  NOTE: (1) Concluded  NOTE: (2) Manual Se  Electronic  (Regional  The "Zone" shown in thttp://www.interconne  INDLED LOCAL EXCHANGE  Exchange Ports  NOTE: Although the	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the telectronic service ordering charge currently contained in this rate exid: CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  E SWITCHING(PORTS)	to Geo	specific sectronic sebasis	electronic outh region ervice ord SOMEC	service orderin nal electronic se ering charges, o	g charges as or ervice ordering of or CLEC-1 may 3.50 es. To view Geo	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe  RATIONAL SUPPORT SYSTE  NOTE: (1) Electronic:  NOTE: (1) Concluded  NOTE: (2) Manual Se  Electronic  (Regional  The "Zone" shown in thttp://www.interconne  INDLED LOCAL EXCHANGE  Exchange Ports  NOTE: Although the	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator if it prefers the contract negotiator in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a profice OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination referse action.bellsouth.com/become_a_clec/html/interconnection.htm  ESWITCHING(PORTS)	to Geo	specific sectronic sebasis	electronic outh region ervice ord SOMEC	service orderin nal electronic se ering charges, o	g charges as or ervice ordering of or CLEC-1 may 3.50 es. To view Geo	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe  RATIONAL SUPPORT SYSTE  NOTE: (1) Electronic:  NOTE: (1) Concluded  NOTE: (2) Manual Se  Electronic  (Regional  The "Zone" shown in thttp://www.interconne  INDLED LOCAL EXCHANGE  Exchange Ports  NOTE: Although the	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the telectronic service ordering charge currently contained in this rate exid: CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  E SWITCHING(PORTS)	to Geo	specific sectronic sebasis	electronic outh region ervice ord SOMEC	service orderin nal electronic se ering charges, o	g charges as or ervice ordering of or CLEC-1 may 3.50 es. To view Geo	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe  RATIONAL SUPPORT SYSTE  NOTE: (1) Electronic:  NOTE: (1) Concluded  NOTE: (2) Manual Se  Electronic  (Regional  The "Zone" shown in thttp://www.interconne  INDLED LOCAL EXCHANGE  Exchange Ports  NOTE: Although the	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the telectronic service ordering charge currently contained in this rate exid: CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  E SWITCHING(PORTS)	to Geo	specific sectronic sebasis	electronic outh region ervice ord SOMEC	service orderin nal electronic se ering charges, o	g charges as or ervice ordering of or CLEC-1 may 3.50 es. To view Geo	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Continued: NOTE: (2) Manual Se NOTE: (2) Manual Se Electronic (Regione: The "Zone" shown in http://www.interconne UNDLED LOCAL EXCHANGE Exchange Ports NOTE: Although the 2-WIRE VOICE GRAE	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the selectronic service ordering charge currently contained in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  E SWITCHING(PORTS)  Port Rate includes all available features in GA, KY, LA & TN, the desired DE LINE PORT RATES (RES)	to Geo	specific he BellScotronic sibasis graphical	some some some some some some some some	service orderin nal electronic se ering charges, aged UNE Zone	g charges as orr ervice ordering or or CLEC-1 may 3.50 es. To view Geo	dered by the Sta charge elect the region ographically Dea	ate Commiss	service order			e, refer to Inte	rnet Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Continued: NOTE: (2) Manual Se NOTE: (2) Manual Se Electronic (Regione: The "Zone" shown in http://www.interconne UNDLED LOCAL EXCHANGE Exchange Ports NOTE: Although the 2-WIRE VOICE GRAE	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the telectronic service ordering charge currently contained in this rate exid: CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  E SWITCHING(PORTS)	to Geo	specific sectronic sebasis	some some some some some some some some	service orderin nal electronic se ering charges, o	g charges as or ervice ordering of or CLEC-1 may 3.50 es. To view Geo	dered by the Sta sharge elect the region	ate Commiss	service order		Central Office	e, refer to Inte	rnet Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Conditued: NOTE: (2) Manual Se NOTE: (2) Manual Se Electronic (Regional The "Zone" shown in http://www.interconne INDLED LOCAL EXCHANGE Exchange Ports NOTE: Although the 2-WIRE VOICE GRAD	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the selectronic service ordering charge currently contained in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  E SWITCHING(PORTS)  Port Rate includes all available features in GA, KY, LA & TN, the desired DE LINE PORT RATES (RES)	to Geo	specific he BellScotronic subasis  graphical  res will n	some some some some some some some some	service orderin nal electronic se ering charges, aged UNE Zone	g charges as orr ervice ordering or or CLEC-1 may 3.50 es. To view Geo	dered by the Sta charge elect the region ographically Dea	ate Commiss	service order			e, refer to Inte	rnet Website:		
NOTE: Local Channe  RATIONAL SUPPORT SYSTE  NOTE: (1) Electronic:  NOTE: (1) Continued:  NOTE: (2) Manual Se  Electronic  (Regiona:  The "Zone" shown in http://www.interconne  UNDLED LOCAL EXCHANGE  Exchange Ports  NOTE: Although the  2-WIRE VOICE GRAD  Exchange	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the structure of the str	to Geo	specific he BellSocitronic subasis  graphical  res will n  UEPSR	some or some of the some of th	service orderin nal electronic se ering charges, aged UNE Zone e ordered using	g charges as or ervice ordering cor CLEC-1 may 3.50 es. To view Geo	dered by the Statharge elect the region organization organization of the state of t	ate Commiss	service order		15.20	e, refer to inte	rnet Website:		
NOTE: Local Channe  RATIONAL SUPPORT SYSTE  NOTE: (1) Electronic:  NOTE: (1) Concluded  NOTE: (2) Manual Se  Electronic  (Regione:  The "Zone" shown in http://www.interconne  INDLED LOCAL EXCHANGE  Exchange Ports  NOTE: Although the  2-WIRE VOICE GRAD  Exchange	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the service Order: CLEC-1 should contact its contract negotiator if it prefers the service Order charge or service ordering charge currently contained in this rate exists of the service Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  ESWITCHING(PORTS)  Port Rate includes all available features in GA, KY, LA & TN, the desired DE LINE PORT RATES (RES)  ge Ports - 2-Wire Analog Line Port with Caller ID - Res ge Ports - 2-Wire Analog Line Port outgoing only - Res ge Ports - 2-Wire VG unbundled LA extended local dialing parity Port with	to Geo	specific he BellScotronic subasis  graphical  res will n  UEPSR  UEPSR	somec ord	service orderin nal electronic se ering charges, aged UNE Zone ordered using 1.52 1.52 1.52	g charges as orrervice ordering or CLEC-1 may  3.50 es. To view Geo  g retail USOCs  2.31  2.31	dered by the Statharge elect the regions ographically Dea	ate Commiss	service order		15.20 15.20 15.20	e, refer to Inte	met Website:		
NOTE: Local Channe  RATIONAL SUPPORT SYSTE  NOTE: (1) Electronic:  NOTE: (1) Concluded  NOTE: (2) Manual Se  Electronic  (Regional  The "Zone" shown in http://www.interconne  INDLED LOCAL EXCHANGE  Exchange Ports  NOTE: Although the  2-WIRE VOICE GRAE  Exchange	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the time the electronic service ordering charge currently contained in this rate exist. CLEC-1 may elect either the state specific Commission ordered rates for ervice Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers action.bellsouth.com/become_a_clec/html/interconnection.htm  E SWITCHING(PORTS)  Port Rate includes all available features in GA, KY, LA & TN, the desired DE LINE PORT RATES (RES)  ge Ports - 2-Wire Analog Line Port- Res ge Ports - 2-Wire Analog Line Port outgoing only - Res ge Ports - 2-Wire Analog Line Port outgoing only - Res ge Ports - 2-Wire VG unbundled LA extended local dialing parity Port with	to Geo	specific he BellScotronic subasis  graphical  res will n  UEPSR  UEPSR	some or some of the some of th	service orderin nal electronic se ering charges, aged UNE Zone e ordered using	g charges as or ervice ordering cor CLEC-1 may 3.50 es. To view Geo	dered by the Statharge elect the region organization organization of the state of t	ate Commiss	service order		15.20	e, refer to Inte	met Website:		
NOTE: Local Channe RATIONAL SUPPORT SYSTE NOTE: (1) Electronic: NOTE: (1) Concluded NOTE: (2) Manual Se  Electronic (Regione: The "Zone" shown in http://www.interconne UNDLED LOCAL EXCHANGE Exchange Ports NOTE: Although the  2-WIRE VOICE GRAD  Exchange Ex	el - Dedicated Transport - minimum billing period - Below DS3=one more  EMS  Service Order: CLEC-1 should contact its contract negotiator if it prefers the service Order: CLEC-1 should contact its contract negotiator if it prefers the service Order charge or service ordering charge currently contained in this rate exists of the service Order charge: disconnect, in the state of Florida, to be billed on a public OSS Charge, per LSR, submitted via BST's OSS interactive interfaces all)  the sections for stand-alone loops or loops as part of a combination refers section.bellsouth.com/become_a_clec/html/interconnection.htm  ESWITCHING(PORTS)  Port Rate includes all available features in GA, KY, LA & TN, the desired DE LINE PORT RATES (RES)  ge Ports - 2-Wire Analog Line Port with Caller ID - Res ge Ports - 2-Wire Analog Line Port outgoing only - Res ge Ports - 2-Wire VG unbundled LA extended local dialing parity Port with	to Geo	specific he BellScotronic subasis  graphical  graphical  UEPSR  UEPSR  UEPSR	somec ord	service orderin nal electronic se ering charges, aged UNE Zone ordered using 1.52 1.52 1.52	g charges as orrervice ordering or CLEC-1 may  3.50 es. To view Geo  g retail USOCs  2.31  2.31	dered by the Statharge elect the regions ographically Dea	ate Commiss	service order		15.20 15.20 15.20	e, refer to Inte	met Website:		

Page 17 of 32

						ı	RATES (\$)					OSS R	ATES (\$)		ı
ATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	urring		ıg Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increme Charge Manual order of Electronic Add'
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	Subsequent Activity		UEPSR	USASC	0.00	0.00	0.00								
FEATURE															
	All Available Vertical Features		UEPSR	UEPVF	0.00	0.00	0.00				15.20				
2-WIDE V	OICE GRADE LINE PORT RATES (BUS)														
Z-VVIIXL V	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus		UEPSB	UEPBL	1.52	2.31	2.21				15.20				
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with														
	Caller+E484 ID - Bus.		UEPSB	UEPBC	1.52	2.31	2.21				15.20				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus		UEPSB	UEPBO	1.52	2.31	2.21				15.20				
	Exchange Ports - 2-Wire VG unbundled LA extended local dialing parity Port with Caller ID - Bus.		UEPSB	LIEDAY	1.52	2.31	2.21				15.20				
	Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus		UEPSB		1.52	2.31	2.21				15.20				
	Exchange Ports - 2-Wire VG unbundled Louisiana Bus Area Calling Port with Caller ID - Bus (BUC)		UEPSB		1.52	2.31	2.21				15.20				
	Subsequent Activity	i	UEPSB		0.00	0.00	0.00								
FEATURE															
	All Available Vertical Features		UEPSB	UEPVF	0.00	0.00	0.00				15.20				
EXCHANG	GE PORT RATES (DID & PBX)														
	Exchange Ports - 2-Wire DID Port		UEPEX	UEPP2	8.29	115.85	18.20				15.20				
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability		UEPDD	UEPDD	68.47	196.18	92.92				15.20				
	Exchange Ports - 2-Wire ISDN Port (See Notes below.		UEPTX UEPSX	U1PMA	10.07	70.76	51.46				15.20				
	All Features Offered		UEPTX UEPSX	UEPVF	0.00	0.00	0.00								
NOTE: T	ransmission/usage charges associated with POTS circuit switched usage will also apply	to circ	uit switch	ed voice a	nd/or circuit sw	itched data tran	smission by B-0	Channels ass	sociated with	2-wire ISDN	ports.				
NOTE: A	ccess to B Channel or D Channel Packet capabilities will be available only through BFF	₹/New !	Business	Request	Process. Rates	for the packet	capabilities will	be determin	ed via the Bo	na Fide Req	uest/New Bu	siness Reque	st Process.		
			UEPTX												
	Exchange Ports - 2-Wire ISDN Port Channel Profiles  Exchange Ports - 4-Wire ISDN DS1 Port		UEPSX	U1UMA											
	Exchange Ports - 4-Wire ISDN DST Port	$\vdash$		LIEDEV	0.00	0.00	0.00				15.00				
		4 1	UEPEX	UEPEX	94.82	0.00 197.92	0.00 98.62				15.20				
	2-Wire VG Unbundled 2-Way PBX Trunk - Res		UEPSE								15.20 15.20				
	2-Wire VG Unbundled 2-Way PBX Trunk - Res 2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPRD	94.82	197.92	98.62								
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus		UEPSE	UEPRD	94.82 1.52	197.92 30.37 30.37	98.62 14.42 14.42				15.20 15.20				
			UEPSE	UEPPC UEPPO	94.82	197.92 30.37	98.62 14.42				15.20				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Analog Long Distance Terminal PBX Trunk - Bus		UEPSP UEPSP UEPSP UEPSP	UEPPC UEPPO UEPP1 UEPLD	94.82 1.52 1.52 1.52 1.52 1.52	30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus  2-Wire Analog Long Distance Terminal PBX Trunk - Bus  2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por		UEPSP UEPSP UEPSP UEPSP UEPSP	UEPRD UEPPC UEPPO UEPP1 UEPLD UEPL2	94.82 1.52 1.52 1.52 1.52 1.52 1.52	30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus  2-Wire Analog Long Distance Terminal PBX Trunk - Bus  2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por  2-Wire Voice Unbundled PBX LD Terminal Ports		UEPSP UEPSP UEPSP UEPSP UEPSP	UEPRD UEPPC UEPPO UEPP1 UEPLD UEPLD UEPL2 UEPLD	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52	30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Analog Long Distance Terminal PBX Trunk - Bus 2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Vice Unbundled 2-Way PBX Usage Por		UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP	UEPRD UEPPC UEPPO UEPP1 UEPLD UEPL2 UEPLD UEPXA	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52	30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus  2-Wire Analog Long Distance Terminal PBX Trunk - Bus  2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por  2-Wire Voice Unbundled PBX LD Terminal Ports  2-Wire Vice Unbundled 2-Way PBX Usage Por  2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports		UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP	UEPPO UEPPO UEPP1 UEPLD UEPLD UEPL2 UEPLD UEPXA UEPXB	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus  2-Wire Nalog Long Distance Terminal PBX Trunk - Bus  2-Wire Nalog Long Distance Terminal PBX Trunk - Bus  2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por  2-Wire Voice Unbundled PBX LD Terminal Ports  2-Wire Voice Unbundled 2-Way PBX Usage Por  2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports  2-Wire Voice Unbundled PBX LD DDD Terminals Por		UEPSP C UEPPO UEPP1 UEPLD UEPL2 UEPLD UEPXA UEPXB	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20					
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus  2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus  2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus  2-Wire Analog Long Distance Terminal PBX Trunk - Bus  2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por  2-Wire Voice Unbundled PBX LD Terminal Ports  2-Wire Vice Unbundled 2-Way PBX Usage Por  2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports  2-Wire Voice Unbundled PBX LD DDD Terminals Por  2-Wire Voice Unbundled PBX LD DDD Terminals Por		UEPSP C UEPPO UEPP1 UEPLD UEPLD UEPLD UEPXA UEPXA UEPXC UEPXC	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20					
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Analog Long Distance Terminal PBX Trunk - Bus 2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Vice Unbundled 2-Way PBX Usage Por 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por		UEPSP D UEPPC UEPPO UEPP1 UEPL2 UEPL2 UEPLD UEPXA UEPXB UEPXC UEPXD UEPXD	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.52	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20					
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled 2-Way PBX Usage Por 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Por  2-Wire Voice Unbundled PBX LD Terminal Switchboard Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por		UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP UEPSP	UEPRD UEPPC UEPPO UEPP1 UEPL2 UEPL2 UEPLA UEPXA UEPXB UEPXB UEPXC UEPXD UEPXB	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.5	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire Na Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Nanlog Long Distance Terminal PBX Trunk - Bus 2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calliing Por 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port		UEPSP D UEPPC UEPPO UEPLD UEPL2 UEPLD UEPXB UEPXB UEPXC UEPXC UEPXD UEPXC UEPXD UEPXE UEPXL UEPXL	94.82  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20					
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Analog Long Distance Terminal PBX Trunk - Bus 2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Vice Unbundled 2-Way PBX Usage Por 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling Por 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Por 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Por		UEPSP D UEPPC UEPPO UEPLD UEPL2 UEPLD UEPXB UEPXB UEPXC UEPXC UEPXD UEPXC UEPXD UEPXE UEPXL UEPXL	94.82 1.52 1.52 1.52 1.52 1.52 1.52 1.52 1.5	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20					
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus 2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire Na Line Side Unbundled Incoming PBX Trunk - Bus 2-Wire Nanlog Long Distance Terminal PBX Trunk - Bus 2-Wire Voice Unbundled 2-Way PBX Louisiana Calling Por 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por 2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calliing Por 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port		UEPSE UEPSP D UEPPC UEPPO UEPLD UEPL2 UEPLD UEPXB UEPXB UEPXC UEPXC UEPXD UEPXC UEPXD UEPXE UEPXL UEPXL	94.82  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52  1.52	197.92 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37 30.37	98.62 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42 14.42				15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20					

					- 1	RATES (\$)					OSS R	ATES (\$)		
CATEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec	urring	Nonrecurrin	ng Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremen Charge Manual S Order vs Electronic- Add'l
				Rec	First	Add'l	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Subsequent Activity		UEPSP	USASC	0.00	0.00	0.00								
FEATURES		-												
All Available Vertical Features		UEPSE	UEPVF	0.00	0.00	0.00				15.20				
EXCHANGE PORT RATES (COIN)														
Exchange Ports - Coin Port				1.52	2.31	2.21				15.20				
NOTE: Transmission/usage charges associated with POTS circuit switched usage will also app	ly to circ	cuit switch	ned voice	and/or circuit sv	vitched data tran	smission by B-0	Channels as:	sociated with	2-wire ISDN	ports.				
NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BF	R/New	Business	Request	Process. Rate	s for the packet	capabilities will	be determin	ed via the Bo	na Fide Req	uest/New Bu	isiness Reque	st Process.		
DLED LOCAL SWITCHING, PORT USAGE														
End Office Switching (Port Usage)														
End Office Switching Function, Per MOU				0.001868										
End Office Trunk Port - Shared, Per MOU				0.00018										
Tandem Switching (Port Usage) (Local or Access Tandem)				0.0004007										
Tandem Switching Function Per MOU Tandem Trunk Port - Shared, Per MOU				0.0001067 0.000222										
Tandem Trunk Port - Shared, Per MOC				0.000222										
Common Transport														
Common Transport - Per Mile, Per MOU				0.0000032										
Common Transport - Facilities Termination Per MOU				0.0003748										
Common Transport - Facilities Termination Per MOU  DLED PORT/LOOP COMBINATIONS - COST BASED RATES				0.0003748										
· · · · · · · · · · · · · · · · · · ·				ocal Switching c		oundled Port se	ction of this	Rate Exhibit.						
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru	same n	nanner as	s they are	ocal Switching o	Stand-Alone Unb				Coin Port/Lo	pop Combina	tions.			
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the	same non of this	nanner as s rate ext	s they are nibit shall rently Com	ocal Switching of applied to the sapply to all combined and Not	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	nbined	
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrecu	same non of this	nanner as s rate ext	s they are nibit shall rently Com	ocal Switching of applied to the sapply to all combined and Not	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	nbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges lise	same non of this	nanner as s rate ext	s they are nibit shall rently Com	ocal Switching of applied to the sapply to all combined and Not	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	nbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port sectifor Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	same non of this	nanner as s rate ext	s they are nibit shall rently Com	ocal Switching of applied to the sapply to all combined and Not	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1	same mon of this	nanner as s rate ext	s they are nibit shall rently Com	applied to the sapply to all combined and Not e identified in the sapply to all combined and not expect the sapply to all combined and not expect to all combine	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonreculation of the Composition of Combosition Combosit	same mon of this sted appurring ch	nanner as s rate ext	s they are nibit shall rently Com	applied to the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply the sapply to all the sapply to all the sapply to all the sapply to all the sapply the s	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1	same mon of this	nanner as s rate ext	s they are nibit shall rently Com	applied to the sapply to all combined and Not e identified in the sapply to all combined and not expect the sapply to all combined and not expect to all combine	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	nbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port sectic  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3	same mon of this sted appurring ch	nanner as s rate ext	s they are nibit shall rently Com	applied to the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply to all the sapply the sapply to all the sapply to all the sapply to all the sapply to all the sapply the s	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	nbined	
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port secti  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges lis  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  1-Wire Voice Grade Loop (SL1) - Zone 1	same mon of this sted appurring ch	s rate exh ly to Curr arges sha	s they are nibit shall ently Com all be thos	applied to the sapply to all combined and Not e identified in the	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2	same non of this sted appurring ch	s rate ext ly to Curr arges sha	s they are nibit shall ently Com all be thos	applied to the sapply to all combined and Not e identified in the sapply to all combined and sapply to	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	mbined	
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port secti  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges lis  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  1-Wire Voice Grade Loop (SL1) - Zone 1	same non of this sted appurring ch	s rate exh ly to Curr arges sha	s they are nibit shall ently Com all be thos	applied to the sapply to all combined and Not e identified in the	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	nbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port sectifor Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	same non of this sted appurring ch	s rate ext ly to Curr arges sha	s they are nibit shall ently Com all be thos	applied to the sapply to all combined and Not e identified in the sapply to all combined and sapply to	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrecustry of the Composition of Composition Composition (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3	same non of this sted appurring ch	s rate extended to the control of th	s they are nibit shall ently Cornall be thos	applied to the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply sa	Stand-Alone Unt bitand-Alone (Included Included	o/port network e ned Combos ai - Currently Con	elements exc nd the first a	cept for UNE		urring charge		Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port sectifor Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	same non of this sted appurring ch	s rate ext ly to Curr arges sha	s they are nibit shall ently Cornall be thos	applied to the sapply to all combined and Not e identified in the sapply to all combined and sapply to	Stand-Alone Unb binations of loop Currently Comb	p/port network e	elements exc nd the first a	cept for UNE				Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrecustry of the Composition of Composition Composition (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3	same non of this sted appurring ch	ueprx Ueprx Ueprx Ueprx Ueprx Ueprx Ueprx	s they are nibit shall ently Cornall be thos	applied to the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply to all combined and Not e identified in the sapply sa	Stand-Alone Unt bitand-Alone (Included Included	o/port network e ned Combos ai - Currently Con	elements exc nd the first a	cept for UNE		urring charge		Currently Con	mbined	
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port sectic  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res	same non of this sted appurring ch	ueprx Ueprx Ueprx Ueprx Ueprx Ueprx Ueprx Ueprx	s they are nibit shall ently Corrall be those unibit shall be those unibit shall be those unibit shall be those unibits used to the shall be those unibits used to the shall be those unibits used to the shall be th	applied to the sapply to all combined and Not e identified in the	Stand-Alone Unit binations of loop Currently Combine Nonrecurring	o/port network e ned Combos ai - Currently Con	elements exc nd the first a	cept for UNE		urring charge		Currently Con	mbined	
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res	same non of this sted appurring ch	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	s they are nibit shall ently Comall be those used to be those used to be those used to be those used to be those used to be those used to be those used to be those used to be those used to be the th	applied to the sapply to all combined and Not e identified in the	Stand-Alone Untroduced to the control of the contro	o/port network e ned Combos ai - Currently Con 19.08	elements exc nd the first a	cept for UNE		15.20		Currently Con	mbined	
DLED PORT/LOOP COMBINATIONS - COST BASED RATES  Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port sectic  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res  2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res	same non of this sted appurring ch	UEPRX	ueplx Ueplx Ueprc Ueprc Ueprc Ueprc Ueprc	applied to the sapply to all combined and Not e identified in the	Stand-Alone Unit binations of loop Currently Combine Nonrecurring 38.85 38.85 38.85	19.08	elements exc nd the first a	cept for UNE		15.20 15.20 15.20		Currently Con	mbined	
Cost Based Rates are applied where BellSouth is required by FCC and/or State Commission ru  Features shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the  End Office and Tandem Switching Usage and Common Transport Usage rates in the Port section  For Georgia, Kentucky, Louisiana and Tennessee, the recurring UNE Port and Loop charges list  Combos. For Currently Combined Combos in GA, KY, LA, TN and all other states, the nonrect  2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Louisiana extended local dialing parity port with	same n on of this total appropriate the approp	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	ueplx Ueplx Ueprx Ueprx Ueprx Ueprx Uepra Uepra Uepra Uepra	applied to the sapplied to the sapplied to the sapply to all combined and Not e identified in table 13.75 49.62 11.77 22.39 48.26 1.36 1.36	Stand-Alone Untroduced in the Indian Stand-Alone Untroduced in the Indian Standard Indian Ind	19.08	elements exc nd the first a	cept for UNE		15.20 15.20		Currently Con	mbined	

						RATES (\$)					OSS R	ATES (\$)		
CATEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	eurring		ng Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'I
All Features Offered		UEPRX	UEPVF	Rec 0.00	First 0.00	Add'I 0.00	First	Add'l	SOMEC	SOMAN 15.20	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)		UEPRX	LNPCX	0.35										<del> </del>
		OZ. TO	Liti Ox	0.00										
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is		UEPRX	LICACO		0.10	0.10				15.20				<u> </u>
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-s 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			USACC		0.10	0.10				15.20				
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update					0.00					15.20				
ADDITIONAL NRCs		UEPRX	110400	0.00	0.00	0.00				15.20				
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity  2-Wire VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)		UEPKX	USASZ	0.00	0.00	0.00				15.20				
UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1	1			13.13										
2-Wire VG Loop/Port Combo - Zone 1	2			23.75									20.00	
2-Wire VG Loop/Port Combo - Zone 3	3			49.62										
UNE Loop Rates														
2-Wire Voice Grade Loop (SL1) - Zone 1	1	UEPBX		11.77										
2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	3	UEPBX UEPBX		22.39 48.26										
				.00										
2-Wire Voice Grade Line Port (Bus)  2-Wire voice unbundled port without Caller ID - bus		UEPBX	HEDRI	1.36	38.85	19.08				15.20				
2-Wire voice unbundled port with Caller + E484 ID - bus		UEPBX		1.36	38.85	19.08				15.20				
2-Wire voice unbundled port outgoing only - bus			UEPBO	1.36	38.85	19.08				15.20				
2-Wise voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - bus			UEPAX	1.36	38.85	19.08				15.20				
2-Wire voice unbundled incoming only port with Caller ID - Bus		UEPBX	UPEB1	1.36	38.85	19.08				15.20				
2-Wire voice unbundled Louisiana Bus Area Calling Port with Caller ID (BUC		UEPBX	UEPAA	1.36	38.85	19.08				15.20				
LOCAL NUMBER PORTABILITY														
Local Number Portability (1 per port)		UEPBX	LNPCX	0.35										
FEATURES														
All Features Offered		UEPBX	UEPVF	0.00	0.00	0.00				15.20				
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED														
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with		UEPBX			0.10	0.10				15.20				
change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent		UEPBX	USACC		0.10	0.10					F 40			
Database Update					0.00						5.12			
ADDITIONAL NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity		UEPBX	USAS2								31.92	7.32		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)														
UNE Port/Loop Combination Rates														
2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2	1 2			13.13 23.75										-
2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	3			49.62										1

							RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	curring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'l
					Rec	First	Add'I	First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNE Loo	pp Rates														
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	UEPRG	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2	2	UEPRG	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL 1) - Zone 3	3	UEPRG		48.26										
2-Wire V	oice Grade Line Port Rates (RES - PBX)														<del>                                     </del>
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res		UEPRG	UEPRD	1.36	66.91	31.29				15.20				
LOCAL	NUMBER PORTABILITY														
LOUAL	Local Number Portability (1 per port)		UEPRG	LNPCP	3.50										
					0.00										
FEATUR	ES														-
	All Features Offered		UEPRG	UEPVF	0.00	0.00	0.00				15.20				
NONREC	CURRING CHARGES (NRCs) - CURRENTLY COMBINED														
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As- Is		UEPRG	USAC2		7.68	1.85				15.20				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with		LIEBBO	110400		7.00	4.05					04.00			
	Change  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent		UEPRG	USACC		7.68	1.85					31.92	7.32		<del>                                     </del>
	Database Update					0.00						5.12			-
ADDITIO	NAL NRCs														
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity  PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group		UEPRG	USAS2	0.00	0.00 7.11	0.00 7.11					31.92 19.99	7.32 19.99	19.99	19.9
2-WIRE \	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)														
UNE Por	t/Loop Combination Rates														
	2-Wire VG Loop/Port Combo - Zone 1	1			13.13										
	2-Wire VG Loop/Port Combo - Zone 2	2			23.75										
	2-Wire VG Loop/Port Combo - Zone 3	3			49.62										<del></del>
UNE Loo	pp Rates														
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	UEPPX		11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3	2	UEPPX UEPPX		22.39 48.26										
	z-vviile voice Grade Loop (SL 1) - Zorie :	3	UEPPA	UEPLX	40.20										
2-Wire V	oice Grade Line Port Rates (BUS - PBX)														
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus		UEPPX	UEPPC	1.36	66.91	31.29				15.20				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPO	1.36	66.91	31.29				15.20				
	Line Side Unbundled Incoming PBX Trunk Port - Bus		UEPPX		1.36	66.91	31.29				15.20	·			
	2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling Por		UEPPX		1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled 2-Way Combination PBX Usage Por	-		UEPLD UEPXA	1.36 1.36	66.91 66.91	31.29 31.29				15.20 15.20				-
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Por 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPXA	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled PBX LD DDD Terminals Por			UEPXC	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Por			UEPXD	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por			UEPXE	1.36	66.91	31.29				15.20				
<del></del>	2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling Por		UEPPX		1.36	66.91	31.29				15.20				

						RATES (\$)					OSS R	ATES (\$)		
CATEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	curring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1 st	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'I
				Rec	First	Add'l	Nonrecurring First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Cal Port	ling	UEPPX	LIEDVI	1.36	66.91	31.29				15.20				
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Por 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room		UEPPX	UEPXM	1.36	66.91	31.29				15.20				-
Calling Port		UEPPX	UEPXO	1.36	66.91	31.29				15.20				
2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling		LIEDDY	HEDYD	4.00	66.04	24.00				45.00				
Port  2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Por		UEPPX	UEPXP	1.36 1.36	66.91 66.91	31.29 31.29				15.20 15.20	31.92	7.32		
		OLITA	OLI XO	1.00	00.01	01.20				10.20	01.02	1.02		
LOCAL NUMBER PORTABILITY														
Local Number Portability (1 per port)		UEPPX	LNPCP	3.15					-	-				-
FEATURES														
All Features Offered		UEPPX	UEPVF	0.00	0.00	0.00				15.20				
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED														
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-A	ıs.													
ls .		UEPPX	USAC2		7.68	1.85				15.20				
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch w	rith	LIEDDY	116466		7.60	1.05					24.02	7.00		
Change   2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent		UEPPX	USACC		7.68	1.85					31.92	7.32		
Database Update					0.00						5.12			
ADDITIONAL NICO-														
ADDITIONAL NRCs  2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity		UEPPX	115452	0.00	0.00	0.00					31.92	7.32		-
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group		OLITA	UUAUZ	0.00	7.11	7.11					19.99	19.99	19.99	19.9
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT														
UNE Port/Loop Combination Rates														
2-Wire VG Coin Port/Loop Combo – Zone 1				13.13										
2-Wire VG Coin Port/Loop Combo – Zone 2				23.75										
2-Wire VG Coin Port/Loop Combo – Zone 3 UNE Loop Rates				49.62										-
2-Wire Voice Grade Loop (SL1) - Zone 1		UEPCO	UEPLX	11.77										
2-Wire Voice Grade Loop (SL1) - Zone 2		UEPCO	UEPLX	22.39										
2-Wire Voice Grade Loop (SL1) - Zone 3		UEPCO	UEPLX	48.26										
ONE W. O. L. I. D. (2000)														
2-Wire Voice Grade Line Ports (COIN)  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, L/	Δ													
MS)	٠,	UEPCO	UEPRF	1.36	38.85	19.08				15.20				
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DD (AL, KY, LA, MS)	D		UEPRA	1.36	38.85	19.08				15.20				
2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, MS)			UEPRB	1.36	38.85	19.08				15.20				
2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+	, &				38.85	19.08								
Local (AL, KY, LA, MS)  2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA,		UEPCO	UEPCD	1.36	38.85	19.08				15.20				+
MS)		UEPCO	UEPRN	1.36	38.85	19.08				15.20				
2-Wire Coin Outward with Operator Screening and 011 Blocking (LA)		UEPCO	UEPLA	1.36	38.85	19.08				15.20				
2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DI (AL, KY, LA, MS)	DD	LIEPCO	UEPRH	1.36	38.85	19.08				15.20				
2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, at	nd		UEPCN	1.36	38.85	19.08				15.20				
Local (AL, KY, LA, MS)														1

				1	ı	RATES (\$)				ı	OSS R	ATES (\$)		
NTEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	urring	Nonrecurri	ng Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Electronic-Disc 1st	Add'l
2-Wire Coin Outward Smartline with 900/976 (Louisiana only)				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ADDITIONAL UNE COIN PORT/LOOP (RC)		UEPCO	UEPCB	1.36	38.85	19.08				15.20				-
7.55.110.10.12.00.11.00.11.00.11.00.1														<b>—</b>
UNE Coin Port/Loop Combo Usage (Flat Rate)		UEPCO	URECU	1.81	0.00	0.00							<u> </u>	+
LOCAL NUMBER PORTABILITY														
Local Number Portability (1 per port)		UEPCO	LNPCX	0.35										
FEATURES														
NONRECURRING CHARGES - CURRENTLY COMBINED														-
INCINICOMAING CHARGES - CORRENTET COMDINED										<del>                                     </del>				<del>                                     </del>
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with		UEPCO	USAC2		0.10	0.10				15.20				-
change		UEPCO	USACC		0.10	0.10					31.92	7.32		
ADDITIONAL NRCs													<u> </u>	-
ADDITIONAL NRCS														+
2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity		UEPCO	USAS2		0.00	0.00					31.92	7.32	<u> </u>	
2-WIRE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT														
UNE Port/Loop Combination Rates														
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1	1 2			23.20 33.62										
2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3	3			58.73										
UNE Loop Rates														-
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1	1	UEPPX	UECD1	14.93	102.10	65.72				15.20				1
2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2 2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		UEPPX		25.35 50.46	102.10 102.10	65.72 65.72				15.20 15.20			ļ	
	3	ULFFX	OECDI	30.40	102.10	05.72				13.20				
UNE Port Rate		UEPPX	LIEDD4	8.27	445.05	18.20				15.20				
Exchange Ports - 2-Wire DID Port		UEPPX	UEPDT	8.27	115.85	18.20				15.20				_
NONRECURRING CHARGES - CURRENTLY COMBINED		LIEDDY	110404		7.40	4.04				45.00				
2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-is 2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth		UEPPX	USACT		7.10	1.81				15.20				
Allowable Changes		UEPPX	USA1C		7.10	1.81				15.20				
ADDITIONAL NRCs														+
2-Wire DID Subsequent Activity - Add Trunks, Per Trunk		UEPPX	USAS1		26.01	26.01				15.20				
														+
Telephone Number/Trunk Group Establisment Charges														
DID Trunk Termination (One Per Port)  Additional DID Numbers for each Group of 20 DID Numbers		UEPPX UEPPX		0.00	0.00	0.00				15.20 15.20				
DID Numbers, Non- consecutive DID Numbers , Per Number		UEPPX	ND5	0.00	0.00	0.00				15.20				
Reserve Non-Consecutive DID numbers		UEPPX	ND6	0.00	0.00	0.00				15.20			<u> </u>	
Reserve DID Numbers		UEPPX	NDV	0.00	0.00	0.00				15.20				+
LOCAL NUMBER PORTABILITY		HESS	LNESS											1
Local Number Portability (1 per port)		UEPPX	LNPCP	3.15						<del>                                     </del>		-	-	+
2-WIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT														1
UNE Port/Loop Combination Rates														
		UEPPB UEPPR		27.48					1					

					I	RATES (\$)				Ī	OSS R	ATES (\$)	Ī	
CATEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec	urring	Nonrecurrin	g Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st		Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'l
		UEPPB		Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2	2	UEPPR UEPPB		40.34										<del>                                     </del>
2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3	3	UEPPR		70.99										<u> </u>
UNE Loop Rates														
2-Wire ISDN Digital Grade Loop - UNE Zone 1	1	UEPPB UEPPR	USL2X	19.09	113.34	76.96				15.20				
2-Wire ISDN Digital Grade Loop - UNE Zone 2	2	UEPPB UEPPR	USL2X	31.95	113.34	76.96				15.20				
2-Wire ISDN Digital Grade Loop - UNE Zone 3	3	UEPPB UEPPR	USL2X	62.60	113.34	76.96				15.20				
UNE Port Rate														
Exchange Port - 2-Wire ISDN Line Side Por		UEPPB UEPPR	UEPPB	8.39	70.76	51.46				15.20				
NONRECURRING CHARGES - CURRENTLY COMBINED		LIEDDD												
2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion		UEPPB UEPPR	USACB	0.00	37.40	26.23				15.20				
ADDITIONAL NRCs														
LOCAL NUMBER PORTABILITY														
Local Number Portability (1 per port)		UEPPB UEPPR	LNPCX	0.35	0.00	0.00								
B-CHANNEL USER PROFILE ACCESS:														
CVS/CSD (DMS/5ESS)		UEPPB UEPPR	U1UCA	0.00	0.00	0.00								<u> </u>
CVS (EWSD)		UEPPB UEPPR	U1UCB	0.00	0.00	0.00								
CSD		UEPPB UEPPR	U1UCC	0.00	0.00	0.00								<u></u>
B-CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)														
CVS/CSD (DMS/5ESS)		UEPPB UEPPB	U1UCD	0.00	0.00	0.00								<u> </u>
CVS (EWSD)		UEPPR	U1UCE	0.00	0.00	0.00								
CSD		UEPPB UEPPR	U1UCF	0.00	0.00	0.00								
USER TERMINAL PROFILE														
User Terminal Profile (EWSD only)		UEPPR	U1UMA	0.00	0.00	0.00								
VERTICAL FEATURES		UEPPB												
All Vertical Features - One per Channel B User Profile		UEPPR	UEPVF	0.00	0.00	0.00				15.20				
INTEROFFICE CHANNEL MILEAGE														
Interoffice Channel mileage each, including first mile and facilities termination		UEPPB UEPPR		22.613	39.36	26.62				15.20				
Interoffice Channel mileage each, additional mile		UEPPB UEPPR		0.013	0.00	0.00				15.20				
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT														
UNE Port/Loop Combination Rates														

Second   Second Secon						ı	RATES (\$)				OSS R	ATES (\$)		
### 1995 Spella Long-Will SEND-SC Depail Train Part - MR Zene 1   1 USPP   128.0   129	CATEGORY UNBUNDLED NETWORK ELEMENT	Zone	всѕ	usoc	_	Nonrec	curring		Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic-Disc	Manual Svc Order vs. Electronic-Dis
AVY DST Deptat Looped VSD (DS) Depta Trush Port - VINE Zone 1   1   UEPPP   19,649   199,551					Rec	First	Add'I		SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
WRI Loop Base	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1				180.52									
Net Logs Rates	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2													
4-WYR DST Digital Logo - LINE Zone 1	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3	3	UEPPP		586.76									
4-WYR DST Digital Logo - LINE Zone 1	UNF Loon Rates													
4-Wite DSE Digital Logo - UNB Zone 2   2 USEPP USLAP 148 10 1298   15.20		1	UEPPP	USL4P	85.70	245.16	152.98			15.20				
A-Wite DST Digital Logo - LINE Zono 2   3 USPPP USL-P 481.94   240.16   10.298   15.20		2												
Containing Ports - 4-Wire ISON DST Port   UEPPP   UEPPP   94.82   197.92   94.82   15.50		3	UEPPP	USL4P	491.94	245.16	152.98			15.20				
Containing Ports - 4-Wire ISON DST Port   UEPPP   UEPPP   94.82   197.92   94.82   15.50														
NOMECURENC CAMERGE - CURRENT - COMBINED   No.														
A-Wire DST Digital Loop / 4-Wire ISSN DST Digital Trunk Port - Curbaration - Conversion - Switch-self-self-self-self-self-self-self-self	Exchange Ports - 4-Wire ISDN DS1 Port	1	UEPPP	UEPPP	94.82	197.92	98.62			15.20				
A-Wire DST Digital Loop / 4-Wire ISSN DST Digital Trunk Port - Curbaration - Conversion - Switch-self-self-self-self-self-self-self-self	NONDECLIDDING CHARGES CLIDDENTLY COMPINED	1												
Conversion - Switch-saries		1												
ADDITIONAL NRC6  4-Wire DS1 Loop4-W ISDN Digit Tirk Port - Subseq Actry- Invaridne way tell nos white ISS Advances and the ISSN DS1 Digital Trunk Port - Outward Tell Numbers (AI States except NC).  4-Wire DS1 Loop 4-Wire ISDN DS1 Digital Tirk Port - Subsequent Inward Tell Nos Above Stat Allowance.  10-CAL NUMBER PORTABILITY  Local Number Portability (1 per port)  Local Number Portability (1 per portability (1 per port)  Local Number Portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 per portability (1 pe			LIEPPP	LISACE	0.00	115.63	76 20			15 20				
A-Wire DST Loop+A-WISDN Digit Tix Port - Subsequent Invand Tel Numbers (All A-Wire ISDN DST Digital Trush Port - Outward Tel Numbers (All A-Wire ISDN DST Digital Trush Port - Outward Tel Numbers (All A-Wire ISDN DST Digital Trush Port - Subsequent Invand Tel No. Above Std Allowance   UEPPP   PR7TT	OUTVETSIOTT "SWILLIF'AS"IS	1	JEFFF	USAUP	0.00	110.03	10.29			10.20				
4-Wire DST Loop A-W ISON Digit Tix Port - Subsequent inward Tel Numbers (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Numbers (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Numbers (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Numbers (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Number (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Number (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Number (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Number (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Number (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Number (AI 4-Wire ISON DST Digital Trunk Port - Outward Tel Number (AI 4-Wire ISON DST Digital Trunk Port - United Number (AI 4-Wire ISON DST Digital Trunk Port - United Number (AI 4-Wire ISON DST Digital Loop/4W DDITS Trunk Port - UNIX Zone 2	ADDITIONAL NRCs	1												
Within Std Allowance   UEPPP   PRTF														
A-Wire DST Loop / 4-Wire ISDN DST Digital Trunk Port - Outward Tel Numbers (All States except NC)			LIFPPP	PR7TF		0.48				15.20				
UEPPP   PR7TO			02			0.10				10.20				
Above St Allowance			UEPPP	PR7TO		11.18	11.18			15.20				
Local Number Portability (1 per port)		:												
Local Number Portability (1 per port)	Above Std Allowance		UEPPP	PR7ZT		22.35	22.35			15.20				
Local Number Portability (1 per port)														
Local Number Portability (1 per port)	LOCAL NUMBER PORTABILITY													
NTERFACE (Provisioning Only)			LIEDDD	LNDCN	1 75									
Voice/Data	Local Number Portability (1 per port)		UEPPP	LINPCIN	1.75									
Voice/Data	INTERFACE (Provisioning Only)													
Digital Data   UEPPP   R7710   0.00			UEPPP	PR71V	0.00	0.00	0.00							
New or Additional 'B' Channel			UEPPP	PR71D										
New or Additional - Voice/Data B Channe	Inward Data		UEPPP	PR71E	0.00	0.00	0.00							
New or Additional - VoiceData B Channe														
New or Additional Digital Data B Channe														
New or Additional University Data B Channe   UEPPP   PR7BD   0.00   14.11   15.20   15.20   14.11   15.20   15.20   14.11   15.20   15.20   14.11   15.20   15.20   14.11   15.20   15.20   14.11   15.20   15.20   14.11   15.20   15.20   15.20   14.11   15.20   15.20   15.20   14.11   15.20   15.20   15.20   15.20   14.11   15.20														
New or Additional Useage Sensitive Voice Data B Channe														
New or Additional Useage Sensitive Digital Data B Channe														
CALL TYPES		-												
Inward	New or Additional Useage Sensitive Digital Data B Channe	1	UEPPP	PK/BU	0.00	14.11				15.20				
Inward	CALL TYPES	1												
Outward		1	HEDDD	PR7C1	0.00	0.00	0.00							
Two-way		1												
Interoffice Channel Mileage		1												
Fixed Each Including First Mile		1	32	50	0.00	0.00	0.00							
Fixed Each Including First Mile	Interoffice Channel Mileage	1												
4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT  UNE Port/Loop Combination Rates  UNE Port/Loop Combination Rates  UNE DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1 1 UEPDC 154.17 15.20  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2 2 UEPDC 263.43 15.20  4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3 3 UEPDC 560.41 15.20	Fixed Each Including First Mile					86.69	79.44			15.20				
UNE Port/Loop Combination Rates         1         UEPDC         154.17         15.20           4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2         2         UEPDC         263.43         15.20           4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2         2         UEPDC         263.43         15.20           4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3         3         UEPDC         560.41         15.20	Each Airline-Fractional Additional Mile		UEPPP	1LN1B	0.2652									
UNE Port/Loop Combination Rates         1         UEPDC         154.17         15.20           4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1         1         UEPDC         263.43         15.20           4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2         2         UEPDC         263.43         15.20           4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3         3         UEPDC         560.41         15.20														
4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1       1 UEPDC       154.17       15.20         4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2       2 UEPDC       263.43       15.20         4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3       3 UEPDC       560.41       15.20	4-WIRE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT	-												
4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1 1 UEPDC 154.17 15.20 15.20 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2 2 UEPDC 263.43 15.20 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3 3 UEPDC 560.41 15.20	UNE Port/Loop Combination Rates													
4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2       2 UEPDC       263.43       15.20         4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3       3 UEPDC       560.41       15.20		1	UEPDC		154.17					15.20				
4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3 3 UEPDC 560.41 15.20		2												
UNE Loop Rates		3	UEPDC		560.41					15.20				

						I	RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec	eurring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increme Charg Manual Order c Electroni Add
					Rec	First	Add'I	Nonrecurrin First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
	4-Wire DS1 Digital Loop - UNE Zone 1	1	UEPDC	USLDC	85.70	245.16	152.98	11130	Addi	JOINEO	15.20	JOHNA	JOHNA	JOHNA	3011
	4-Wire DS1 Digital Loop - UNE Zone 2	2	UEPDC	USLDC	194.96	245.16	152.98				15.20				
	4-Wire DS1 Digital Loop - UNE Zone 3	3	UEPDC	USLDC	491.94	245.16	152.98				15.20				1
UNE Port	Rate														+
	4-Wire DDITS Digital Trunk Port		UEPDC	UDD1T	68.47	196.18	92.92				15.20				
NONRECU	JRRING CHARGES - CURRENTLY COMBINED														
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is		UEPDC	USAC4		125.75	65.08				15.20				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with DS1 Changes		UEPDC	USAWA		125.75	65.08				15.20				
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk			USAWB		125.75	65.08				15.20				
ADDITION	AL NRCs														-
ADDITION	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel														_
	Activation/Chan - 2-Way Trunk 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel Activation/Char	1		UDTTA		14.06	14.06				15.20				-
	- 1-Way Outward Trunk 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan			UDTTB		14.06	14.06				15.20				-
	Inward Trunk wout DID  4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan - Inward Trunk with DID			UDTTD		14.06 14.06	14.06				15.20 15.20				
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2- Way DID w User Trans			UDTTE		14.06	14.06				15.20				+
BIPOLAR	8 ZERO SUBSTITUTION														
	B8ZS -Superframe Format		UEPDC	CCOSF		0.00	605.00				15.20				_
	B8ZS - Extended Superframe Format		UEPDC	CCOEF		0.00	605.00				15.20				
Alternate	Mark Inversion														+
	AMI -Superframe Format		UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Formal		UEPDC	мсоро		0.00	0.00								
Telephone	e Number/Trunk Group Establisment Charges														1
	Telephone Number for 2-Way Trunk Group		UEPDC	UDTGX	0.00						15.20				+
	Telephone Number for 1-Way Outward Trunk Group		UEPDC	UDTGY	0.00						15.20				
	Telephone Number for 1-Way Inward Trunk Group Without DIC		UEPDC	UDTGZ	0.00						15.20				-
	DID Numbers for each Group of 20 DID Numbers		UEPDC	ND4	0.00						15.20		<u> </u>		1
	DID Numbers, Non- consecutive DID Numbers , Per Number		UEPDC	ND5	0.00						15.20				
	Reserve Non-Consecutive DID Nos		UEPDC	ND6	0.00	0.00	0.00				15.20			<u> </u>	-
	Reserve DID Numbers		UEPDC	NDV	0.00	0.00	0.00				15.20				

							RATES (\$)					OSS R	ATES (\$)		
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	urring	Nonrecurrin	g Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st		Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Interoffice Channel Mile	eage - Fixed rate 0-8 miles (Facilities Termination		UEPDC	1LNO1	70.47	86.69	79.44				15.20				
Interoffice Channel Mile	eage - Additional rate per mile - 0-8 miles		UEPDC	1LNOA	0.2652	0.00	0.00								
	eage - Fixed rate 9-25 miles (Facilities Termination			1LNO2	0.00	0.00	0.00								
	-														
Interoffice Channel Mile	eage - Additional rate per mile - 9-25 miles		UEPDC	1LNOB	0.2652	0.00	0.00								<del>                                     </del>
Interoffice Channel Mile	eage - Fixed rate 25+ miles (Facilities Termination		UEPDC	1LNO3	0.00	0.00	0.00	0.00							<u> </u>
Interoffice Channel Mile	eage - Additional rate per mile - 25+ miles		UEPDC	1LNOC	0.2652	0.00	0.00								
Local Number Portabilit	ty, per DS0 Activated		UEPDC	LNPCP	3.15	0.00	0.00	0.00							
Central Office Terminina	nating Point		UEPDC	CTG	0.00										<u> </u>
															<del>                                     </del>
4-WIRE DS1 LOOP WITH CHANNEL	I IZATION WITH PORT														<del></del>
	nel Bank, and up to 24 Feature Activations														
	ombinations of rates depending on type and number of po	rts use	ed												
UNE DS1 Loop															
4-Wire DS1 Loop - UNE		1	UEPMG		85.70	0.00	0.00				15.20				
4-Wire DS1 Loop - UNE		2		USLDC	194.96	0.00	0.00				15.20				
4-Wire DS1 Loop - UNE	E Zone 3	3	UEPMG	USLDC	491.94	0.00	0.00				15.20				
LINE DSO Channelization Canacitic	es (D4 Channel Bank Configurations)														<del>                                     </del>
24 DSO Channel Capacition			UEPMG	VUM24	97.35	0.00	0.00				15.20				
48 DSO Channel Capac	, ,			VUM48	194.70	0.00	0.00				15.20				
96 DSO Channel Capac				VUM96	389.40	0.00	0.00				15.20				
144 DS0 Channel Capa				VUM14	584.10	0.00	0.00				15.20				
192 DS0 Channel Capa	acity -1 per 8 DS1s		UEPMG	VUM19	778.80	0.00	0.00				15.20				
240 DS0 Channel Capa	acity - 1 per 10 DS1s		UEPMG	VUM20	973.50	0.00	0.00				15.20				
288 DS0 Channel Capa	acity - 1 per 12 DS1s		UEPMG	VUM28	1,168.20	0.00	0.00				15.20				
384 DS0 Channel Capa	acity - 1 per 16 DS1s		UEPMG	VUM38	1,557.60	0.00	0.00				15.20				
480 DS0 Channel Capa				VUM40	1,947.00	0.00	0.00				15.20				
576 DS0 Channel Capa	· ·			VUM57	2,336.40	0.00	0.00				15.20				
672 DS0 Channel Capa	acity - 1 per 28 DS1s		UEPMG	VUM67	2,725.80	0.00	0.00				15.20				<del>                                     </del>
Non Beauting Charges (NBC) A	perioted with A Wire DC4 I can with Channellating with D			Charas 7	and an a C···	la m									<del> </del>
	sociated with 4-Wire DS1 Loop with Channeliztion with Por is One (1) DS1, One (1) D4 Channel Bank, and Up To 24 DS					terfi									<del>                                     </del>
	is One (1) DS1, One (1) D4 Channel Bank, and Up 10 24 DS ctioning as one are considered Add'I after the minimum sy														<b></b>
	and the minimum ay		- Jingara												
NRC - Conversion (Cur	rrently Combined) with or without BellSouth Allowed Changes		UEPMG	USAC4	0.00	146.13	8.12				15.20				
· · · · · · · · · · · · · · · · · · ·	ations Where 4-Wire DS1 Loop with Channelization with P		mbinatio	n Curren	ly Exists and										
New (Not Currently Combined) In G	Georgia & Tennessee Only														
1 DS1/D4 Channel Ban GA, LA, KY &TN Only	nk - Add NRC for each Port and Assoc Fea Activation - New		UEPMG	VUMD4	0.00	715.54	467.54				15.20				
Bipolar 8 Zero Substitution			,	7	0.00	7 10.04	407.04				10.20				
·	ity Format, superframe - Subsequent Activity Only		UEPMG	CCOSF	0.00	0.00	605.00				15.20				

							RATES (\$)				1	OSS R	ATES (\$)	Т	
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc	_	Nonrec	urring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st		Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incrementa Charge - Manual Sw Order vs. Electronic-D Add'l
					Rec	First		Nonrecurrin First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
					Kec	First	Add'l	FIFSt	Addi	SOMEC	SOMAN	SUMAN	SUMAN	SOMAN	SOMAN
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only		UEPMG	CCOEF	0.00	0.00	605.00				15.20				
Alternate	Mark Inversion (AMI)														
	Superframe Format		UEPMG		0.00	0.00	0.00								
	Extended Superframe Format		UEPMG	МСОРО	0.00	0.00	0.00								
	Ports Associated with 4-Wire DS1 Loop with Channelization with Port														
Exchange	Ports														
	Line Side Combination Channelized PBX Trunk Port - Business		UEPPX	UEPCX	1.52	0.00	0.00	0.00	0.00		15.20				
	V. 614 0 4 4 401 V. 488V.T. 4.8 4.8 5.4		HEDDY	LIEDOV	4.50	2.22	2.22		2.22		45.00				
	Line Side Outward Channelized PBX Trunk Port - Business		UEPPX	UEPOX	1.52	0.00	0.00	0.00	0.00		15.20				
	Line Side Inward Only Channelized PBX Trunk Port without DID		UEPPX	UEP1X	1.52	0.00	0.00	0.00	0.00		15.20				
	·														
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port		UEPPX	UEPDM	8.29	0.00	0.00	0.00	0.00		15.20				
Feature A	ctivations - Unbundled Loop Concentration														
	5		HEDDY	40014/44											
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX	1PQWM	0.6497	25.36	13.40				15.20				
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		UEPPX	1PQWU	0.6497	78.05	18.40				15.20				
Telephon	e Number/ Group Establishment Charges for DID Service														
·	DID Trunk Termination (1 per Port)		UEPPX	NDT	0.00						15.20				
	DID Numbers - groups of 20 - Valid all States		UEPPX	ND4	0.00	0.00	0.00				15.20				
	Non-Consecutive DID Numbers - per number		UEPPX	ND5	0.00	0.00	0.00				15.20				
	Reserve Non-Consecutive DID Numbers		UEPPX	ND6	0.00	0.00	0.00				15.20				
	Reserve DID Numbers		UEPPX	NDV	0.00	0.00	0.00				15.20				
Local Nur	mber Portability														
	Local Number Portability - 1 per port		UEPPX	LNPCP	3.15	0.00	0.00								
FEATURE	S - Vertical and Optional														
Local Swi	itching Features Offered with Line Side Ports Only														
	All Features Available		UEPPX	UEPVF	0.00	0.00	0.00				15.20				
NDLED PORT LO	OOP COMBINATIONS - MARKET RATES														
Market Ra	ates shall apply where BellSouth is not required to provide unbundled local switching or	switch	ports per	FCC and/	or State Comm	ission rules.									
These sce	enarios include:														
1. Unbun	dled port/loop combinations that are Not Currently Combined in all of the BellSouth stat	tes exc	cept as no	ted for Ge	orgia, Kentuck	y, Louisiana, ar	d Tennessee.								
2. Unbun	dled port/loop combinations that are Currently Combined or Not Currently Combined in	Zone	1 of the T	op 8 MSA	S in BellSouth's	region for end	users with 4 or	more DS0 e	quivalent lines	<b>s</b> .					
The Top 8	MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta); LA	(New	Orleans):	NC (Green	nsboro-Winston	Salem-Highno	int/Charlotte-Ga	stonia-Rock	Hill); TN (Nas	shville).					
BellSouth	currently is developing the billing capability to mechanically bill the recurring and non-re he right to true-up the billing difference.										eceding in lie	u of the Marke	et Rates and		
	et Rate for unbundled ports includes all available features in all states.														
	and Tandem Switching Usage and Common Transport Usage rates in the Port section arge (USOC: URECU).	n of thi	s rate exh	nibit shall a	pply to all comi	oinations of loop	o/port network e	elements exc	ept for UNE	Coin Port/Lo	op Combina	tions which ha	ave a flat rate		

						!	RATES (\$)				1	OSS R	ATES (\$)	ı	
CATEGORY UNBUNDLED NETWO	RK ELEMENT Z	Cone	BCS	usoc		Nonrec	urring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increme Charg Manual Order Electronic Add
					Rec	First	Add'l	Nonrecurrin First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT	(RES)														+
UNE Port/Loop Combination Rates															
2-Wire VG Loop/Port Combo - Zone 1		1			25.77										+
2-Wire VG Loop/Port Combo - Zone 1		2			36.39										†
2-Wire VG Loop/Port Combo - Zone 3		3			62.26										
UNE Loop Rates															
2-Wire Voice Grade Loop (SL1) - Zone 1		1 L	JEPRX	LIEPI X	11.77										+
2-Wire Voice Grade Loop (SL1) - Zone 2	III		JEPRX		22.39										t
2-Wire Voice Grade Loop (SL1) - Zone 3			JEPRX		48.26										1
2-Wire Voice Grade Line Port (Res)							-						-		
2-Wire voice unbundled port - residence		l	JEPRX	UEPRL	14.00	90.00	90.00					31.92	7.32		
2-Wire voice unbundled port with Caller ID -	res	ι	JEPRX	UEPRC	14.00	90.00	90.00					31.92	7.32		
2-Wire voice unbundled port outgoing only -		ι	JEPRX	UEPRO	14.00	90.00	90.00					31.92	7.32		
2-Wire voice Grade unbundled Louisiana ext Caller ID - res	ended local dialling parity port with	ι	JEPRX	UEPAS	14.00	90.00	90.00					31.92	7.32		-
2-Wire voice unbundled Louisiana Area Plus	with Caller ID - res (RUI)	ı	JEPRX	UFPAG	14.00	90.00	90.00					31.92	7.32		
2-Wire voice unburidled Louisiana Area Plus				UEPAH	14.00	90.00	90.00					31.92	7.32		+
2-Wire voice unbundles res, low usage line p				UEPAP	14.00	90.00	90.00					31.92	7.32		
LOCAL NUMBER PORTABILITY															
Local Number Portability (1 per port)		ı	JEPRX	LNPCX	0.35										-
FEATURES AND ON A DESCRIPTION OF THE PERSON			IEDDY	LIED) (E	2.22		2.22								
All Features Offered		L	JEPKX	UEPVF	0.00	0.00	0.00								+
2-Wire Voice Grade Loop / Line Port Combin	ation - Switch-as-is	l	JEPRX	USAC2		41.50	41.50					31.92	7.32		
2-Wire Voice Grade Loop / Line Port Combin	ation - Switch with change	ι	JEPRX	USACC		41.50	41.50								
ADDITIONAL NRCs															₩
NRC - 2-Wire Voice Grade Loop/Line Port C	ombination - Subsequen	ι	JEPRX	USAS2		0.00	0.00					31.92	7.32		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT	(BUS)														+
	( )														
UNE Port/Loop Combination Rates															
2-Wire VG Loop/Port Combo - Zone 1	III	1			25.77										
2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3		3			36.39 62.26										+
2-vviie vo Loop/Fort Combo - Zorie 3		J			02.20										<b>†</b>
UNE Loop Rates															
2-Wire Voice Grade Loop (SL1) - Zone 1			JEPBX		11.77										1
2-Wire Voice Grade Loop (SL1) - Zone 2		2 L	JEPBX	UEPLX	22.39										
2-Wire Voice Grade Loop (SL1) - Zone 3		3 L	JEPBX	UEPLX	48.26										+
2-Wire Voice Grade Line Port (Bus)		_													<b>†</b>
2-Wire voice unbundled port without Caller II	) - bus	ι	JEPBX	UEPBL	14.00	90.00	90.00					31.92	7.32		
2-Wire voice unbundled port with Caller + E4				UEPBC	14.00	90.00	90.00					31.92	7.32		
2-Wire voice unbundled port outgoing only -				UEPBO	14.00	90.00	90.00					31.92	7.32		
2-Wire voice Grade unbundled Louisiana ext Caller ID - bus	ended local dialing parity port with		JEPBX		14.00	90.00	90.00					31.92	7.32		
2-Wire voice unbundled Louisiana Bus Area	Calling Port with Caller ID (BLIC			UEPAA	14.00	90.00	90.00				1	31.92	7.32		+
2-vviie voice dilbuilded Louisialia Dus Alea	Canning I OIL WILLI CAREL ID (DOC	L	,LI DA	OLI AA	14.00	30.00	30.00			<del>                                     </del>	<del>                                     </del>	31.32	1.32		+

					I	RATES (\$)					OSS R	ATES (\$)		
TATEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc		Nonrec	urring	Nonrocurrin	ig Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'l
				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)		UEPBX	LNDCV	0.35										
FEATURES		UEPBA	LINPUX	0.35										
VOLUETOUR DIVIDATE AUREFUT V COMPINED														
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is		UEPBX	USAC2		41.50	41.50					31.92	7.32		
2-Wire Voice Grade Loop / Line Port Combination - Switch with change			USACC		41.50	41.50					01.02	1.02		
ADDITIONAL NRCs														
NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequen		UEPBX	USAS2		0.00	0.00					31.92	7.32	<del>                                     </del>	
					2.30	2.20								
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)														
UNE Port/Loop Combination Rates														
2-Wire VG Loop/Port Combo - Zone 1	1			25.77										
2-Wire VG Loop/Port Combo - Zone 2	2			36.39										
2-Wire VG Loop/Port Combo - Zone 3	3			62.26										
INIT I D														
UNE Loop Rates	1	UEPRG	LIEDLY	11.77										
2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2		UEPRG		22.39										
2-wire voice Grade Loop (SL1) - Zone z		UEPRG	UEPLX	22.39										
2-Wire Voice Grade Loop (SL1) - Zone 3	3	UEPRG	UEPLX	48.26										
2-Wire Voice Grade Line Port Rates (RES - PBX)														
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res		UEPRG	UEPRD	14.00	90.00	90.00					31.92	7.32		
LOCAL NUMBER PORTABILITY														
Local Number Portability (1 per port)		UEPRG	LNPCP	3.15										
FEATURES														
NONRECURRING CHARGES - CURRENTLY COMBINED														
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is		UEPRG	USAC2		41.50	41.50					31.92	7.32		
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change		UEPRG	USACC		41.50	41.50								
ADDITIONAL NRCs														
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-														
Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group					0.00 14.64	0.00 14.64					19.99	19.99	19.99	19.9
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)														
UNE Port/Loop Combination Rates														
2-Wire VG Loop/Port Combo - Zone 1	1			25.77										
2-Wire VG Loop/Port Combo - Zone 2	2			36.39										
2-Wire VG Loop/Port Combo - Zone 3	3			62.26										
UNE Loop Rates														-
2-Wire Voice Grade Loop (SL1) - Zone 1	1	UEPPX	UEPLX	11.77										
2-Wire Voice Grade Loop (SL1) - Zone 2	2	UEPPX	UEPLX	22.39										
2-Wire Voice Grade Loop (SL1) - Zone 3	3	UEPPX	UEPLX	48.26										
	II .	1	1					l	1	1	1	I	1	1

					I	RATES (\$)					OSS R	ATES (\$)		
CATEGORY UNBUNDLED NETWORK ELEMENT	Zone	BCS	USOC		Nonrec	curring			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'l
				Rec	First	Add'I	Nonrecurrin First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus		UEPPX	UEPPC	14.00	90.00	90.00					31.92	7.32		
Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPO	14.00	90.00	90.00					31.92	7.32		
Line Side Unbundled Incoming PBX Trunk Port - Bus 2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling Por		UEPPX		14.00	90.00	90.00					31.92	7.32		<del></del>
2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Cailing Por  2-Wire Voice Unbundled PBX LD Terminal Ports		UEPPX		14.00 14.00	90.00	90.00					31.92 31.92	7.32 7.32		
2-Wire Voice Unbundled 1-BX LB Terminary one 2-Wire Voice Unbundled 2-Way Combination PBX Usage Por		UEPPX		14.00	90.00	90.00					31.92	7.32		
2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPXB	14.00	90.00	90.00					31.92	7.32		
2-Wire Voice Unbundled PBX LD DDD Terminals Por		UEPPX	UEPXC	14.00	90.00	90.00					31.92	7.32		
2-Wire Voice Unbundled PBX LD Terminal Switchboard Por		LIEDDY	UEPXD	14.00	90.00	90.00					31.92	7.32		
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Por		UEPPX		14.00	90.00	90.00					31.92			<del>                                     </del>
2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling Por		UEPPX		14.00	90.00	90.00					31.92			
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling	g	<u> </u>												
Port		UEPPX		14.00	90.00	90.00					31.92	7.32		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Por		UEPPX	UEPXM	14.00	90.00	90.00					31.92	7.32		
Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room     Calling Port     Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling		UEPPX	UEPXO	14.00	90.00	90.00					31.92	7.32		
Port   Port		HEDDY	UEPXP	14.00	90.00	90.00					31.92	7.32		
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Por		UEPPX		14.00	90.00	90.00					31.92	7.32		<del>                                     </del>
2 Trillo Tolog Chibanaloa T Tray Galgoing T Britingadarda Tol		OL: 17	02.70	1 1.00	00.00	00.00					01.02	7.02		
LOCAL NUMBER PORTABILITY														
Local Number Portability (1 per port)		UEPPX	LNPCP	3.15										
FEATURES														
NONRECURRING CHARGES - CURRENTLY COMBINED														
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is		UEPPX	USAC2		41.50	41.50					31.92	7.32		
2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change			USACC		41.50	41.50								
ADDITIONAL NRCs														
2-Wire Voice Grade Loop/ Line Port Combination - Subsequen		UEPPX	USAS2		0.00	0.00					31.92	7.32		
2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-														
Nonrecurring PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group					0.00 14.64	0.00 14.64					19.99	19.99	19.99	19.
PBA Subsequent Activity - Change/Reamange Multilline Hunt Group					14.04	14.04					19.99	19.99	19.99	19.
2-WIRE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT														
UNE Port/Loop Combination Rates	1													
2-Wire VG Coin Port/Loop Combo – Zone 1				25.77										
2-Wire VG Coin Port/Loop Combo – Zone 2				36.39		_								
2-Wire VG Coin Port/Loop Combo – Zone 3	1			62.26										1
UNE Loop Rates	1-		<del>                                     </del>						-					<del></del>
2-Wire Voice Grade Loop (SL1) - Zone 1	1	UEPCO	HED! V	11.77								1		<del>                                     </del>
2-Wire Voice Grade Loop (SL1) - Zone 1	1	UEPCO		22.39										<b>—</b>
2-Wire Voice Grade Loop (SL1) - Zone 3		UEPCO		48.26										
2 Wire Voice Grade Line Bort Bates (Coin)					-		-							<del>                                     </del>
2-Wire Voice Grade Line Port Rates (Coin)  2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA,	1-		<del>                                     </del>						-					-
2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, K1, LA, MS) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD		UEPCO	UEPRF	14.00	90.00	90.00					31.92	7.32		
2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD  (AL, KY, LA, MS, SC)  2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, MS)		UEPCO	UEPRA	14.00	90.00	90.00					31.92	7.32		
2-vviile Colii 2-vvay with Operator Screening and OTT Blocking (AL, LA, MS)		UEPCO	UEPRB	14.00	90.00	90.00					31.92	7.32		

						ı	RATES (\$)				ı	OSS R	ATES (\$)		1
CATEGORY	UNBUNDLED NETWORK ELEMENT	Zone	BCS	usoc	_	Nonrec	urring	Nonroqueric	ng Disconnect	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic-D Add'I
					Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, &														
	Local (AL, KY, LA, MS)		UEPCO	UEPCD	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA, MS)		UEPCO	UEPRN	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward with Operator Screening and 011 Blocking (LA)		UEPCO	UEPLA	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL. KY, LA, MS)		UEPCO	UEPRH	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)		UEPCO	UEPCN	14.00	90.00	90.00					31.92	7.32		
LOCAL	NUMBER PORTABILITY														
	Local Number Portability (1 per port)		UEPCO	LNPCX	0.35										
NONRE	CURRING CHARGES - CURRENTLY COMBINED														
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is		UEPCO	USAC2		41.50	41.50					31.92	7.32		
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change		UEPCO	USACC		41.50	41.50								
ADDITIO	DNAL NRCs														<b></b>
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequen		UEPCO	USAS2		0.00	0.00					31.92	7.32		
NOTE:	If no rate is identified in the contract, the rates for the specific service or function will be	ac cot f	orth in or	policable P	all South toriff o	r as pagatiated	by the Parties	inon roquios	t by oither Do	rh.					
NOTE:	in no rate is identified in the contract, the rates for the specific service of function will be	as set 1	orun in ap	phicanie Di	onoouth tailli 0	as negonaled	by the Faitles t	apon reques	L by eilliei Pa	ity.					<del>                                     </del>

Page 32 of 32 Version 3Q01: 10/18/01

#### LOCAL INTERCONNECTION Louisiana

						1			RATES (\$)					OSS R	ATES (\$)		
														1	1	Incremental	Incremental
												Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
		LOCAL INTERCONNECTION	Interim	Zone	BCS	USOC				Nonre	curring	Submitted	Submitted		Charge - Manual	Order vs.	Order vs.
												Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-	Electronic-Disc
									curring		onnect	per LSR	LSR	Electronic-1st		Disc 1st	Add'l
CATEGORY	NOTES						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
													+				
LOCAL INTER	CONNECTION (CA	L ALL TRANSPORT AND TERMINATION)											+				
LOCAL INTERC		ide a rate indicates that the Parties have agreed	ta bill and	Iraan			h the element:	uill be sees		ait and MTA	tueffie end u			NTA traffia			
	NOTE. DK DESI	lue a rate indicates that the Farties have agreed	to bill allu	кеер	on usay	a. As suc	n, the element v	viii be asses	Section trans	I and wit A	traine, and n	101 1011-1	Tansii anu n	I	<del>.</del>		
																	1
	TANDEM SWITC	HING															1
	TARDEMOVITO	Tandem Switching Function Per MOU		1	OHD		0.0005507bk						1				<del>                                     </del>
		Multiple Tandem Switching, per MOU (applies to			ONE		0.000000751										
		intial tandem only)			OHD		0.0005507bk										
		,,,															
	TRUNK CHARGE																
		Installation Trunk Side Service - per DS0			OHD	TPP++		334.94	56.98								
		Dedicated End Office Trunk Port Service-per DS0	**		OHD	TDE0P	0.00										
					0H1												
		Dedicated End Office Trunk Port Service-per DS1	**		OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
					OH1												
		Dedicated Tandem Trunk Port Service-per DS1**		L		TDW1P	0.00										
LOCAL INTER		ent is recovered on a per MOU basis and is include	d in the Er	nd Offic	ce Switch	ing and I	andem Switchin	g, per MOU	rate elements				+				-
LOCAL INTERC	CONNECTION (TR	(ANSPORT)															
	COMMON TRAN	l SPORT (Shared)				-							-				
	COMMON TRAN	Common Transport - Per Mile, Per MOU			OHD	-	0.0000032bk						-				
		Common Transport - Per Mile, Per MOU  Common Transport - Facilities Termination Per			OHD		0.0000032bk										<del>                                     </del>
		MOU			OHD		0.0003748bk										
		IWOO IWOO		1	OHD		0.0003740DK						1				+
	INTEROFFICE C	HANNEL - DEDICATED TRANSPORT - VOICE GR	RADE														
		Interoffice Channel - Dedicated Transport - 2															
		Wire Voice Grade - Per Mile per month			DHL. OH	1L5NF	0.013										
		Interoffice Channel - Dedicated Transport- 2-			,												
		Wire Voice Grade - Facility Termination per															
		month			DHL, OH	1L5NF	22.60	39.36	26.62								
	INTEROFFICE C	HANNEL - DEDICATED TRANSPORT - 56/64 KB	PS														
		Interoffice Channel - Dedicated Transport - 56		l .	S									1	1		
		kbps - per mile per month			JHL, OH	1L5NK	0.013						1		<b>.</b>		<b></b>
ĺ		Interoffice Channel - Dedicated Transport - 56			ייוס וווכ	1L5NK	45.04	20.27	00.00					1	1		
		kbps - Facility Termination per month Interoffice Channel - Dedicated Transport - 64			JHL, OHI	V TL5NK	15.61	39.37	26.62				-				-
		kbps - per mile per month			THI OH	1L5NK	0.013								1		
-		Interoffice Channel - Dedicated Transport - 64		Η,	-ι ι∟, ΟΙ <b>Π</b> Ι	I LOININ	0.013					<del>                                     </del>	<del>†                                      </del>	<del> </del>	<del>                                     </del>		<b>-</b>
		kbps - Facility Termination per month			оні оні	1L5NK	15.61	39.37	26.62	0.00	0.00						
				$\vdash$	,		10.01	30.01	20.02	0.00	0.00			1	1		
	INTEROFFICE C	HANNEL - DEDICATED TRANSPORT - DS1		1									1		1		
	, , , , , , , , , , , , , , , , , , , ,	Interoffice Channel - Dedicated Channel - DS1 -											1				
		Per Mile per month		0	H1 OH1	1L5NL	0.2652						1	]	I		
		Interoffice Channel - Dedicated Tranport - DS1 -															
		Facility Termination per month		0	H1 OH1	1L5NL	70.47	86.69	79.44			<u> </u>		<u> </u>			
	INTEROFFICE C	HANNEL - DEDICATED TRANSPORT- DS3															
		Interoffice Channel - Dedicated Transport - DS3 -															
		Per Mile per month		0	H3 OH3	1L5NM	6.04					ļ	1		ļ		ļ
		Interoffice Channel - Dedicated Transport - DS3 -													1		
1		Facility Termination per month		0	H3 OH3	1L5NM	850.45	270.69	158.05								

#### LOCAL INTERCONNECTION Louisiana

									RATES (\$)					OSS R	ATES (\$)		
																Incremental Charge -	Incremental Charge -
		LOCAL INTERCONNECTION	Interim	Zone	BCS	usoc						Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental	Manual Svc Order vs.	Manual Svc Order vs.
										Nonre	ecurring	Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-	Electronic-Disc
							_		curring		onnect	per LSR	LSR		Electronic-Add'l	Disc 1st	Add'l
CATEGORY	NOTES						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
																	+
																	+
	LOCAL CHANNE	L - DEDICATED TRANSPORT															
		Local Channel - Dedicated - 2-Wire Voice Grade															
		per month		(	OHL OH	TEFV2	18.32	187.51	32.21								
		Local Channel - Dedicated - 4-Wire Voice Grade															
		per month		(		TEFV4	19.41	187.94	32.63								
		Local Channel - Dedicated - DS1 per month			OH1	TEFHG	39.18	172.34	149.27								
		Local Channel - Dedicated - DS3 Facility															
		Termination per month			OH3	TEFHJ	469.44	438.46	256.30								
	LOCAL INTERC	ONNECTION MID-SPAN MEET										1					+
		s service ride Mid-Span Meet, one-half the tariffed	service I	ocal C	hannel r	ate is an	olicable.										+
				1		1											1
	MULTIPLEXERS																
		Channelization - DS1 to DS0 Channel System		0	H1 OH1N	SATN1	105.09	88.41	60.76								
		D00 / D04 01			OH3	0.4.7110	224.42	470.00	04.05								
	-	DS3 to DS1 Channel System per month			OH3MS	SATNS	201.48	172.99	91.25								+
					OH1												
		DS3 Interface Unit (DS1 COCI) per month				SATCO	11.78	6.39	4.58								
		, , ,															1
	Notes: If no rate	e is identified in the contract, the rates, terms and co	anditions fo	or the s	necific s	ervice or											
		s set forth in applicable BellSouth tariff.			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	0 0 01											
											1	1	1				1

Page 2 of 3 Version 3Q01: 10/18/01

## SERVICE PROVIDER NUMBER PORTABILITY Louisiana

Line   Line										RATES (\$)					OSS R	ATES (\$)		
Note   Note				Interim						.,,			Svc Order	Svc Order	Incremental	Incremental	Charge -	Incremental Charge - Manual Svc
Notes   Not			UNBUNDLED NETWORK ELEMENT		Zone	BCS	USOC				Nonre	curring	Submitted	Submitted	Charge - Manual	Charge - Manual	Order vs.	Order vs.
CATEGORY   NOTES   NOTES   Rec   First   Add'1   First   Add'1   First   Add'1   SOMEC   SOMAN   SOM									Nonro	urring	Disco	nnoot						Electronic-Dis Add'l
INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF	CATEGORY	NOTES						Rec										SOMAN
RCF, per number ported (Business Line   TNPBL   2.91   0.25								1.00	101	/ luu i	1 0.	, au	0020	Compar	Compar	00	Compar	00
RCF, per number ported (Business Line   TNPBL   2.91   0.25																		
RCF, per number ported (Residence Line   TNPRL   2.91   0.25	INTERIM SERVICI	E PROVIDER NUMBER	PORTABILITY - RCF															
RCF, add*  capacity for simultaneous call forwarding, pe additional path   1,24     1,24     2,02   2,02   2,01   2,01   3,50   19,99   19,9			RCF, per number ported (Business Line				TNPBL	2.91	0.25									
Additional path			RCF, per number ported (Residence Line				TNPRL	2.91	0.25									
RCF, per service order, per location (Residence   TNPRD   2.02   2.01   2.01   3.50   19.99								1.24										
NTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID			RCF, per service order, per location (Business				TNPBD		2.02	2.02	2.01	2.01	3.50		19.99	19.99	19.99	19.99
DID per number ported (Residence)			RCF, per service order, per location (Residence				TNPRD		2.02	2.02	2.01	2.01	3.50		19.99	19.99	19.99	19.99
DID per number ported (Residence)   TNPDR   0.42	INTERIM SERVIC	E PROVIDER NUMBER	PORTABILITY - DID															<del>                                     </del>
DID per service order, per location (Residence   TNPRD   2.02   2.02   2.01   3.50   19.99   19.99   19.99   19.99   19.99   19.90							TNPDR		0.42									
DID per service order, per location (Business   TNPBD   2.02   2.02   2.01   3.50   19.99   19.99   19.99   19.99   19.90			DID per number ported (Business				TNPDB		0.42									
DID, per trunk termination, Initis   TNPT2   68.47   185.13   3.50   19.99   19.99   19.99   19.90			DID per service order, per location (Residence				TNPRD		2.02	2.02	2.01	2.01	3.50		19.99	19.99	19.99	19.99
DID, per trunk termination, Subsequer         TNPT2         68.47         68.79         3.50         19.99         19.99         19.99         19.99         19.99										2.02	2.01	2.01						19.99
																		19.99
SERVICE PROVIDER NUMBER PORTABILITY (RIPH)			DID, per trunk termination, Subsequer				TNPT2	68.47	68.79				3.50		19.99	19.99	19.99	19.99
	SERVICE PROVID	DER NUMBER PORTA	I BILITY (RIPH)															
																		L

Page 1 of 1 Version 3Q01: 10/18/01

#### ODUF/ADUF/CMDS Louisiana

								RATES (\$)					OSS R	ATES (\$)		
		UNBUNDLED NETWORK ELEMENT Interio	m Zon	e BCS	USOC				Nonre	curring	Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.		Incremental Charge - Manual Svc Order vs. Electronic-Disc	Svc Order vs
							Nonre	curring	Disco	onnect	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
CATEGORY	NOTES					Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	HE/CMDC															
ODUF/EDOUF/ADI	UF/CIVIDS															
	ACCESS DAILY US	AGE FILE (ADUF)														
		ADUF: Message Processing, per message			N/A	0.007983										
		ADUF: Data Transmission (CONNECT:DIRECT), per message			N/A	0.00012681										
	OPTIONAL DAILY U	JSAGE FILE (ODUF)														
		ODUF: Recording, per message			N/A	0.0000117										
		ODUF: Message Processing, per message			N/A	0.004641										
		ODUF: Message Processing, per Magnetic Tape provisioned			N/A	48.45										
		ODUF: Data Transmission (CONNECT:DIRECT), per message			N/A	0.00010568										
	CENTRALIZED MES	SSAGE DISTRIBUTION SERVICE (CMDS)														
		CMDS: Message Processing, per message			N/A	0.004										
		CMDS: Data Transmission (CONNECT:DIRECT), per message			N/A	0.001										
		identified in the contract, the rate for the specific service or function w tariff or as negotiated by the Parties upon request by either Party.	vill be as	set forth in	1											

Page 1 of 1 Version 3Q01: 10/18/01

# AMENDMENT TO INTERCONNECTION AGREEMENT BETWEEN BELLSOUTH TELECOMMUNICATIONS, INC. AND NEWSOUTH COMMUNICATIONS CORP. DATED MAY 18, 2001

This Agreement (the "Amendment") is made and entered into between BellSouth Telecommunications, Inc. ("BellSouth") a Georgia corporation, and NewSouth Communications, Corp. ("NewSouth") a Delaware corporation. This Amendment shall be deemed effective ten business days following the date of the last signature of both Parties ("Effective Date")

WHEREAS, The Parties desire to amend the Agreement between BellSouth and NewSouth dated May 18, 2001 in order to incorporate rates established by the Florida Public Service Commission ("PSC") in Docket Number 990649-TP, on May 25, 2001 and subsequently amended by the PSC on October 18, 2001

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. The Parties agree that all the rates and rate elements in Attachments 1, 2, 3, 5 and 7 of the Agreement for Florida are hereby deleted and replaced in their entirety with the corresponding rates and rate elements in Exhibit 1, which is attached hereto and incorporated herein by this reference.
- 2. All of the other provisions of the Interconnection Agreement shall remain unchanged and in full force and effect.
- 3. Either or both of the Parties are authorized to submit this Amendment to the appropriate State Public Service Commissions or other Regulatory Agencies for approval subject to Section 252 (e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

NewSouth Communications Corp.	BellSouth Telecommunications, Inc.
Original Signature on File	Original Signature on File
Signature	Signature
Jake E. Jennings Name	C. W. Boltz Name
Vice President – Regulatory Affairs Title	Managing Director Title
February 22, 2002 Date	February 25, 2002 Date

## Exhibit 1

### RESALE DISCOUNTS AND RATES

PPLICABLE DISCOUNTS  RESIDENCE  BUSINESS  CSAs*  Unless noted in this row, the discount for Business will be the applicable discount rate for CSAs.  PERATIONAL SUPPORT SYSTEMS (OSS) RATES  ELEMENT  LUSOC  Ctronic LSR  SOMEC  mual LSR  DUF/EODUF/CMDS RATES  NHANCED OPTION DAILY USAGE FILE (EODUF)  DUF: Message Processing, per message  PTIONAL DAILY USAGE FILE (ODUF)  DUF: Recording, per message  OUF: Message Processing, per message  OUF: Message Processing, per message  OUF: Message Processing, per message		FLORIDA
APPLICABLE DISCOUNTS		
SIDENCE  SINESS  As*  as noted in this row, the discount for Business will be the applicable discount rate for CSAs.  RATIONAL SUPPORT SYSTEMS (OSS) RATES  ELEMENT  Inic LSR  SILSR  SILSR  SILSR  SINCED OPTION DAILY USAGE FILE (EODUF)  F: Message Processing, per message  DNAL DAILY USAGE FILE (ODUF)  Recording, per message  Message Processing, per message		21.83%
BUSINESS		16.81%
ESIDENCE  USINESS  SAs*  less noted in this row, the discount for Business will be the applicable discount rate for CSAs.  ERATIONAL SUPPORT SYSTEMS (OSS) RATES  ELEMENT  ELEMENT  SOMEC  TONIC LSR  SOMEC  TONIC LSR  SOMAN  UF/FODUF/CMDS RATES  IANCED OPTION DAILY USAGE FILE (EODUF)  UF: Message Processing, per message  IONAL DAILY USAGE FILE (ODUF)  F: Recording, per message  F: Message Processing, per message  F: Message Processing, per message		
* Unless noted in this row, the discount for Business will be the applicable discount	unt rate for CSAs.	
OPERATIONAL SUPPORT SYSTEMS (OSS) RATES		
<u>ELEMENT</u>	<u>USOC</u>	
Electronic LSR	SOMEC	\$3.50
Manual LSR	SOMAN	\$19.99
ODUF/EODUF/CMDS RATES		
ENHANCED OPTION DAILY USAGE FILE (EODUF)		
EODUF: Message Processing, per message		0.229109
OPTIONAL DAILY USAGE FILE (ODUF)		
ODUF: Recording, per message		0.0000071
ODUF: Message Processing, per message		0.006835
ODUF: Message Processing, per Magnetic Tape provisioned		48.96
ODUF: Data Transmission (CONNECT:DIRECT), per msg		0.00010811

Version 4Q01: 12/01/01

														ľ			
UNBL	JNDLED	NETWORK ELEMENTS - Florida			I		1						1	Attachment:	2		Exhibit: B
САТ	EGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order		Incremental Charge - Manual Svc	Incremental Charge - Manual Svc	Incremental Charge - Manual Svo
			m									Submitted Elec	Submitted Manually		Order vs. Electronic-	Order vs. Electronic-	Order vs. Electronic-
	1									ı		per LSR	per LSR	1st	Add'l	Disc 1st	Disc Add'l
							Rec		curring	Nonrecurring					RATES (\$)		T
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	The "Zo	l ne" shown in the sections for stand-alone loops or loops as p	part of a	comb	nation refers to Geo	graphically [	Deaveraged UN	E Zones. To v	iew Geograph	ically Deaverag	ed UNE Zone	l Designatio	ns by Centra	al Office, refer	to Internet W	/ebsite:	
		ww.interconnection.bellsouth.com/become_a_clec/html/interc	onnecti	on.htn	1				1			1	-	ı	ı		1
OPER/	ATIONAL	SUPPORT SYSTEMS					<u> </u>		<u> </u>	ļļ			ļ				
l	NOTE: (	Electronic Service Order: NewSouth should contact its con	tract ne	gotiate	or if it prefers the sta	te specific el	lectronic servic	e ordering ch	arges as order	ed by the State	Commissions	. The elect	ronic servic	e ordering ch	arge currently	y contained in	n this rate
ļ	exhibit i	s the BellSouth regional electronic service ordering charge. I	NewSou	th may	elect either the stat	e specific Co	mmission orde	ered rates for t	he electronic s	service ordering	charges, or N	lewSouth n	nay elect the	e regional elec	ctronic servic	e ordering ch	arge.
		2) Any element that can be ordered electronically will be bille															
		s that cannot be ordered electronically at present per the BBR SOMAN, will be applied to a CLECs bill when it submits an LS				category refi	ects the charg	e illai Would D	e billed to a Ci	LEC Once electi	onic ordering	capabilitie	s come on-l	me for that el	ement. Other	wise, the mai	iuai ordering
	J	Manual Service Order Charge, Disconnect Only (FL)				SOMAN		1.83									
		Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces (Regional)				SOMEC		3.50									
UNBU	NDLED EX	(CHANGE ACCESS LOOP				SOIVIEC		3.50									
	2-WIRE	ANALOG VOICE GRADE LOOP															
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		2	UEANL UEANL	UEAL2 UEAL2	12.79 17.27	49.57 49.57	22.83 22.83	25.62	6.57 6.57		11.90 11.90				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	33.36	49.57	22.83	25.62 25.62	6.57		11.90				
		Loop Testing - Basic 1st Half Hour		Ţ	UEANL	URET1	00.00	77.09									
		Loop Testing - Basic Additional Half Hour Engineering Information Document (EI)			UEANL UEANL	URETA		33.12 12.28	12.28								
		Manual Order Coordination for UVL-SL1s (per loop)*			UEANL	UEAMC		9.00	9.00								+
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR) *			UEANL	OCOSL		23.02	23.02								
	2-WIRE	Unbundled COPPER LOOP			-												
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1	-	1	UEQ UEQ	UEQ2X UEQ2X	13.83 15.29	41.64 41.64	19.02 19.02	19.65 19.65	5.09 5.09		11.90 11.90				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2 2 Wire Unbundled Copper Loop - Non-Designed - Zone 3		3	UEQ	UEQ2X	20.29	41.64	19.02	19.65	5.09		11.90				+
		Order Coordination 2 Wire Unbundled Copper Loop - Non- Designed (per loop)			UEQ	USBMC		9.00	9.00								
		Engineering Information Document			UEQ	OODIVIO		12.28	12.28								+
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		77.09									
UNBUI	NDLED EX	Loop Testing - Basic Additional Half Hour  CCHANGE ACCESS LOOP			UEQ	URETA		33.12								<del>                                     </del>	<del>                                     </del>
		ANALOG VOICE GRADE LOOP															
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1	1	1	UEPSR UEPSB	UEALS	12.79	49.57	22.83	25.62	6.57		10.73				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1	ı		UEPSR UEPSB	UEABS	12.79	49.57	22.83	25.62	6.57		10.73				
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting- Zone 2	1	2	UEPSR UEPSB	UEALS	17.27	49.57	22.83	25.62	6.57		10.73				
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting- Zone 2	ı		UEPSR UEPSB	UEABS	17.27	49.57	22.83	25.62	6.57		10.73				
		Z Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 3	ı	3	UEPSR UEPSB	UEALS	33.36	49.57	22.83	25.62	6.57		10.73				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 3	ı		UEPSR UEPSB	UEABS	33.36	49.57	22.83	25.62	6.57		10.73				
UNBU		CHANGE ACCESS LOOP				_	22.20				2.3.						
	2-WIRE	ANALOG VOICE GRADE LOOP															
		CLEC to CLEC Conversion Charge without outside dispatch (UVL-SL1)			UEANL	UREWO		48.11	22.01				11.90				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1		1	UEA	UEAL2	14.50	135.75	82.47	63.53	12.01		11.90				

JNBUNDLED	NETWORK ELEMENTS - Florida				<del></del>	<u> </u>			<u> </u>	· · · · · · · · · · · · · · · · · · ·			Attachment:	2	1	Exhibit:
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increment Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or				115410	40.57	105.75	00.47	00.50	40.04		44.00				
	Ground Start Signaling - Zone 2  2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or		2	UEA	UEAL2	19.57	135.75	82.47	63.53	12.01		11.90	-			
	Ground Start Signaling - Zone 3		3	UEA	UEAL2	37.82	135.75	82.47	63.53	12.01		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		23.02									
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	Battery Signaling - Zone 1		1	UEA	UEAR2	14.50	135.75	82.47	63.53	12.01		11.90				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 2		2	UEA	UEAR2	19.57	135.75	82.47	63.53	12.01		11.90				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		<u> </u>	0271	02/11/2	.0.0.	100.70	02	00.00	12.01		11.00				
	Battery Signaling - Zone 3		3	UEA	UEAR2	37.82	135.75	82.47	63.53	12.01		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		23.02									
	CLEC to CLEC Conversion Charge without outside dispatch			UEA	UREWO		131.83	38.27				11.90				
4-WIRE	ANALOG VOICE GRADE LOOP				<b></b>											
	4-Wire Analog Voice Grade Loop - Zone 1 4-Wire Analog Voice Grade Loop - Zone 2		2	UEA UEA	UEAL4 UEAL4	23.02 31.07	167.86 167.86	115.15 115.15	67.08 67.08	15.56 15.56		11.90 11.90				
	4-Wire Analog Voice Grade Loop - Zone 2  4-Wire Analog Voice Grade Loop - Zone 3			UEA	UEAL4	60.02	167.86	115.15	67.08	15.56		11.90	-			
	Order Coordination for Specified Conversion Time (per LSR)		3	UEA	OCOSL	00.02	23.02	113.13	07.00	13.30		11.90				
2-WIRE	ISDN DIGITAL GRADE LOOP			OLA	00002		20.02									
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	21.76	147.69	94.41	62.23	10.71		11.90	1			
	2-Wire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	29.38	147.69	94.41	62.23	10.71		11.90				
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	56.76	147.69	94.41	62.23	10.71		11.90				
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		23.02									
	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		121.17	33.09				11.90				
2-WIRE	Universal Digital Channel (UDC) COMPATIBLE LOOP															
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1		1	UDC	UDC2X	21.76	147.69	94.41	62.23	10.71		11.90				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		2	UDC	UDC2X	29.38	147.69	94.41	62.23	10.71		11.90				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		3	UDC	UDC2X	56.76		94.41	62.23	10.71						
	CLEC to CLEC Conversion Charge without outside dispatch		3	UDC	UREWO	56.76	147.69 121.17	33.09	62.23	10.71		11.90 11.90				
2-WIRE	ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPA	TIBLE	OOP	ODC	UKLVVO		121.17	33.09				11.50				
	2 Wire Unbundled ADSL Loop including manual service inquiry		1													
	& facility reservation - Zone 1		1	UAL	UAL2X	12.65	149.53	103.85	75.05	15.63		11.90				
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 2		2	UAL	UAL2X	17.08	149.53	103.85	75.05	15.63		11.90				
	2 Wire Unbundled ADSL Loop including manual service inquiry		_		LIALOY	00.00	440.50	100.0=	75.0-	45.00		44.00				
	& facility reservation - Zone 3  Order Coordination for Specified Conversion Time (per LSR)		3	UAL	UAL2X OCOSL	33.00	149.53 23.02	103.85	75.05	15.63	}	11.90	1		<del>                                     </del>	1
-	2 Wire Unbundled ADSL Loop without manual service inquiry &		-	UAL	UCUSL		23.02						+			1
	facility reservaton - Zone 1		1	UAL	UAL2W	12.65	124.83	71.12	60.64	9.12		11.90	1			
	2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservator - Zone 2			UAL	UAL2W	17.08	124.83	71.12	60.64	9.12		11.90				
	2 Wire Unbundled ADSL Loop without manual service inquiry &		<del>                                     </del>	O/ IL	UNLZVV	17.00	124.03	11.12	00.04	5.12	1	11.50	t		<del> </del>	
	facility reservaton - Zone 3		3	UAL	UAL2W	33.00	124.83	71.12	60.64	9.12		11.90	1			
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		23.02									
	CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		124.83	29.33				11.90				
2-WIRE	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPAT	IBLE L	OOP													
	2 Wire Unbundled HDSL Loop including manual service inquiry						,=						1			
$\rightarrow$	& facility reservation - Zone 1		1	UHL	UHL2X	9.97	159.09	113.41	75.05	15.63	1	11.90				
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation - Zone 2		2	UHL	UHL2X	13.46	159.09	113.41	75.05	15.63		11.90				
-	2 Wire Unbundled HDSL Loop including manual service inquiry			OLIE	JIILZA	13.40	155.05	113.41	75.05	13.03		11.90	<del> </del>			
1	& facility reservation - Zone 3		3	UHL	UHL2X	26.00	159.09	113.41	75.05	15.63		11.90	I		1	
1	Order Coordination for Specified Conversion Time (per LSR)		T	UHL	OCOSL		23.02				İ		İ	İ	İ	

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL2W	9.97	134.40	80.69	60.64	9.12		11.90				
	Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2		2	UHL	UHL2W	13.46	134.40	80.69	60.64	9.12		11.90				
	2 Wire Unbundled HDSL Loop without manual service inquiry		_													
	and facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UHL UHL	UHL2W OCOSL	26.00	134.40	80.69	60.64	9.12		11.90				
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		23.02 134.40	29.33				11.90				
4-WIRE	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPAT	IBI F I (	OOP	OTIL	UKLWO	†	134.40	29.33				11.50				
- WINE	4 Wire Unbundled HDSL Loop including manual service inquiry	IDEE E	<u> </u>													
	and facility reservation - Zone 1  4-Wire Unbundled HDSL Loop including manual service inquiry		1	UHL	UHL4X	15.69	193.31	138.98	77.15	12.61		11.90				
	and facility reservation - Zone 2  4-Wire Unbundled HDSL Loop including manual service inquiry		2	UHL	UHL4X	21.17	193.31	138.98	77.15	12.61		11.90				
	and facility reservation - Zone 3		3	UHL	UHL4X	40.90	193.31	138.98	77.15	12.61		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL	-	23.02									
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL4W	15.69	168.62	115.47	62.74	11.22		11.90				
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2		2	UHL	UHL4W	21.17	168.62	115.47	62.74	11.22		11.90				
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		3	UHL	UHL4W	40.90	168.62	115.47	62.74	11.22		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		23.02									
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		134.40	29.33				11.90				
	DS1 DIGITAL LOOP			1101	USLXX	70.44	040.75	101.10	04.00	10.50		44.00				
	4-Wire DS1 Digital Loop - Zone 1 4-Wire DS1 Digital Loop - Zone 2			USL USL	USLXX	73.44 99.13	313.75 313.75	181.48 181.48	61.22 61.22	13.53 13.53		11.90 11.90				ļ
	4-Wire DS1 Digital Loop - Zone 3			USL	USLXX	191.51	313.75	181.48	61.22	13.53		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL	101.01	23.02		01.22	.0.00		11.00				
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		130.25	40.04				11.90				
4-WIRE 1	19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	26.39	161.56	108.85	67.08	15.56		11.90				
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	35.62	161.56	108.85	67.08	15.56		11.90				
	4 Wire Unbundled Digital 19.2 Kbps			UDL UDL	UDL19	68.82	161.56	108.85	67.08	15.56		11.90 11.90				ļ
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1 4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56 UDL56	26.39 35.62	161.56 161.56	108.85 108.85	67.08 67.08	15.56 15.56		11.90				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3			UDL	UDL56	68.82	161.56	108.85	67.08	15.56		11.90				
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		23.02									
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1			UDL	UDL64	26.39	161.56	108.85	67.08	15.56		11.90				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2			UDL	UDL64	35.62	161.56	108.85	67.08	15.56		11.90				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL	UDL64	68.82	161.56	108.85	67.08	15.56	<b> </b>	11.90				
	Order Coordination for Specified Conversion Time (per LSR)  CLEC to CLEC Conversion Charge without outside dispatch			UDL UDL	OCOSL UREWO		23.02 131.67	38.68			<u> </u>	11.90				
2-WIDE I	Unbundled COPPER LOOP			UDL	UKEWU	<del>                                     </del>	131.07	38.88			<b>_</b>	11.90	1	-		<del>                                     </del>
Z-WINE (	2-Wire Unbundled Copper Loop/Short including manual service				+											<del> </del>
	inquiry & facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Short including manual service		1	UCL	UCLPB	12.65	148.50	102.82	75.05	15.63		11.90				
	inquiry & facility reservation - Zone 2		2	UCL	UCLPB	17.08	148.50	102.82	75.05	15.63		11.90				
	2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 3		3	UCL	UCLPB	33.00	148.50	102.82	75.05	15.63		11.90	1			
	Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLPB	33.00	9.00	9.00	75.05	15.63		11.90				+
	2-Wire Unbundled Copper Loop/Short without manual service			001	COLIVIO		3.00	3.00			1					
	inquiry and facility reservation - Zone 1 2-Wire Unbundled Copper Loop/Short without manual service		1	UCL	UCLPW	12.65	123.81	70.09	60.64	9.12		11.90				
	inquiry and facility reservation - Zone 2		2	UCL	UCLPW	17.08	123.81	70.09	60.64	9.12		11.90				
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 3		3	UCL	UCLPW	33.00	123.81	70.09	60.64	9.12		11.90				

	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)	Ι				Incremental Charge - Manual Svc	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss i	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								<b>↓</b>
	2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility reservation - Zone 1		1	UCL	UCL2L	37.07	148.50	102.82	75.05	15.63		11.90				
	2-Wire Unbundled Copper Loop/Long - includes manual svc.			002	OOLEL	07.07	140.00	102.02	70.00	10.00		11.00				<del>                                     </del>
<u> </u>	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	50.04	148.50	102.82	75.05	15.63		11.90				
1   '	2-Wire Unbundled Copper Loop/Long - includes manual svc.									4= 00						
$\vdash$	inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL UCL	UCL2L UCLMC	96.67	148.50 9.00	102.82 9.00	75.05	15.63		11.90				<b> </b>
	2-Wire Unbundled Copper Loop/Long - without manual service			OCL	OCLIVIC		3.00	3.00								<del> </del>
	inquiry and facility reservation - Zone 1		1	UCL	UCL2W	37.07	123.81	70.09	60.64	9.12		11.90				
i I 🗔	2-Wire Unbundled Copper Loop/Long - without manual service		2		1101 014	F0.01	400.07	70.00	00.51	0.10		44.60				
$\vdash$	inquiry and facility reservation - Zone 2 2-Wire Unbundled Copper Loop/Long - without manual service		2	UCL	UCL2W	50.04	123.81	70.09	60.64	9.12		11.90				<b> </b>
1   '	inquiry and facility reservation - Zone 3		3	UCL	UCL2W	96.67	123.81	70.09	60.64	9.12		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00		-						
	CLEC to CLEC Conversion Charge without outside dispatch															
$\vdash$	(UCL -Des) CLEC to CLEC Conversion Charge without outside dispatch			UCL	UREWO		123.81	31.41				11.90				<b></b>
1   '	(UCL-ND)			UEQ	UREWO		44.69	22.01				11.90				
4-WIRE	COPPER LOOP			024	CILLITO			22.01				11.00				1
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4S	18.03	177.87	132.76	77.15	17.73		11.90				
	4-Wire Copper Loop/Short - including manual service inquiry															
	and facility reservation - Zone 2  4-Wire Copper Loop/Short - including manual service inquiry		2	UCL	UCL4S	24.34	177.87	132.76	77.15	17.73		11.90				
<u> </u>	and facility reservation - Zone 3		3	UCL	UCL4S	47.02	177.87	132.76	77.15	17.73		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00								<u> </u>
1   '	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4W	18.03	153.18	100.03	62.74	11.22		11.90				
	4-Wire Copper Loop/Short - without manual service inquiry and			OCL	OCL4VV	10.03	133.16	100.03	02.74	11.22		11.90				<del> </del>
	facility reservation - Zone 2		2	UCL	UCL4W	24.34	153.18	100.03	62.74	11.22		11.90				
	4-Wire Copper Loop/Short - without manual service inquiry and															
$\vdash$	facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL UCL	UCL4W UCLMC	47.02	153.18 9.00	100.03 9.00	62.74	11.22		11.90				<b> </b>
	4-Wire Unbundled Copper Loop/Long - includes manual svc.			UUL	JOLIVIC		9.00	9.00				<del>                                     </del>				<del>                                     </del>
	inquiry and facility reservation - Zone 1		1	UCL	UCL4L	64.52	177.87	132.76	77.15	17.73		11.90				
	4-Wire Unbundled Copper Loop/Long - includes manual svc.							100 ==		4=		44.65				
$\vdash\!\!\!-\!\!\!\!\!-\!$	inquiry and facility reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - includes manual svc.		2	UCL	UCL4L	87.09	177.87	132.76	77.15	17.73		11.90				<del> </del>
1 1 '	inquiry and facility reservation - Zone 3		3	UCL	UCL4L	168.25	177.87	132.76	77.15	17.73		11.90				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		9.00	9.00				50				
	4-Wire Unbundled Copper Loop/Long - without manual svc.						,== :			-						
$\vdash$	inquiry and facility reservation - Zone 1  4-Wire Unbundled Copper Loop/Long - without manual svc.		1	UCL	UCL4O	64.52	153.18	100.03	62.74	11.22		11.90				<del>                                     </del>
1 1 '	inquiry and facility reservation - Zone 2		2	UCL	UCL4O	87.09	153.18	100.03	62.74	11.22		11.90				
	4-Wire Unbundled Copper Loop/Long - without manual svc.		_			500			0E.14	22		50				<b>†</b>
	inquiry and facility reservation - Zone 3		3	UCL	UCL4O	168.25	153.18	100.03	62.74	11.22		11.90				<u> </u>
	Order Coordination for Unbundled Copper Loops (per loop)			UCL UCL	UCLMC		9.00 123.81	9.00 31.41			-	11.90				<b></b>
LOOP MODIFICA	CLEC to CLEC Conversion Charge without outside dispatch			UUL	UKEWU		123.81	31.41				11.90				<del>                                     </del>
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire			UAL, UHL, UCL,	1							t				<del>                                     </del>
	pair less than or equal to 18k ft			UEQ, ULS	ULM2L		0.00	0.00								<u> </u>
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft			UCL, ULS	ULM2G		343.12	343.12								
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18K ft			UHL, UCL	ULM4L		0.00	0.00								

UNBUNDLE'	D NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: E
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge -
						Rec	Nonrec		Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18k ft			UCL	ULM4G		343.12	343.12								
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop			UAL, UHL, UCL, UEQ, UEF, ULS	ULMBT		10.52	10.52								
SUB-LOOPS																
Sub-Lo	oop Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set- Up	I		UEANL	USBSA		487.23	487.23				11.90				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	I		UEANL	USBSB		6.25	6.25				11.90				
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up	1		UEANL	USBSC		169.25	169.25				11.90				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up	- 1		UEANL	USBSD		38.65	38.65				11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN2	7.61	60.19	21.78	47.50	5.26		11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN2	10.27	60.19	21.78	47.50	5.26		11.90				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN2	19.85	60.19	21.78	47.50	5.26		11.90				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC	19.03	9.00	9.00	47.50	3.20		11.90				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -					0.40			40.74	0.00		44.00				
	Zone 1 Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -		1	UEANL	USBN4	8.12	68.83	30.42	49.71	6.60		11.90				
	Zone 2 Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -		2	UEANL	USBN4	10.96	68.83	30.42	49.71 49.71	6.60		11.90				
	Zone 3		3	UEANL	USBN4	21.18	68.83	30.42	49.71	6.60		11.90				
$\vdash$	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC	0.50	9.00	9.00	47.70			44.00				
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)			UEANL	USBR2	3.50	51.84	13.44	47.50	5.26		11.90				
<u> </u>	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00								
$\vdash \vdash \vdash$	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	Ī		UEANL	USBR4	6.68	55.91	17.51	49.71	6.60		11.90				
i 1	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		9.00	9.00								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	ı	1	UEF	UCS2X	6.25	60.19	21.78	47.50	5.26		11.90				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	I	2	UEF	UCS2X	8.44	60.19	21.78	47.50	5.26		11.90				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	I	3	UEF	UCS2X	16.30	60.19	21.78	47.50	5.26		11.90				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		<u>L</u>	UEF	USBMC		9.00	9.00			<u></u>					
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1			UEF	UCS4X	5.20	68.83	30.42	49.71	6.60		11.90				
$\longleftarrow \longleftarrow$	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	!		UEF	UCS4X	7.02	68.83	30.42	49.71	6.60		11.90				ļ
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	1	3	UEF	UCS4X	13.55	68.83	30.42	49.71	6.60		11.90				
11	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		<u> </u>	UEF	USBMC		9.00	9.00								ļ
Unbun	dled Sub-Loop Modification Unbundled Sub-Loop Modification - 2-W Copper Dist Load		1		+						-					
	Coil/Equip Removal per 2-W PR			UEF	ULM2X		10.11	10.11				11.90				
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR			UEF	ULM4X		10.11	10.11				11.90				
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR unloaded			UEF	ULM4T		15.58	15.58				11.90				
	dled Network Terminating Wire (UNTW)															
Unbun-																
Unbun	Unbundled Network Terminating Wire (UNTW) per Pair Set-Up Work: Site Visit Survey, per MDU			UENTW UENTW	UENPP UENVS	0.2286	18.02 120.11	18.02 120.11				11.90				

ATT ELEMENTS	UNBUNDLE	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
Section   Sect				Zone	BCS	usoc			RATES(\$)			Submitted Elec	Svc Order Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Manual Svc Order vs.
Silve total Sea Lip, the Temman Austronous Comment (1997)   Silve							Rec	Nonrec			Disconnect						
Access Termonal Productionage, per Termonal Applicational   USPNW USPNT   100.26   100.25										First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Access Terminal Provisioning, per Terminal Auditorial   Centre																	<b></b>
Terrinals					UENTW	UEN1T		101.09	101.09								+
DATTY Par Proviousing, par Fast for 1st Terrinary   DENTY   SENPT   4.46   4.65   4.					LIENTA/	LIENOT		100.05	400.05								i
ENTITY Per Procession   Perit of Assistant Hermate   USPITY   USPIPA   3.64   3.64   3.64   3.64   3.64   3.64   3.64   3.64   3.64   3.64   3.64   3.65																	$\vdash$
New Name Therefore Device (NBD)							1										<del></del>
Network Interface Device (1967) - 12 intes	Networ			1	OLIVIV	OLINFA		3.04	3.04								<del> </del>
Network Interface Device (NDI) - 1-6 into   OENTW UND16   11-048   85.20   11-50	Herwei				UENTW	UND12		68.08	42.80				11.90				
Netrock Interface Device Cross Contends - 2 W							1										
Security   Security																	ſ
Sile-Loop Feeder   Sile-Loop F									7.63								
USL-Freader, DSS Set-up per Cross Box location - CLEC   UEA   USSPTW   487.23   11.90   USL-Freader, DSS Set-up per Cross Box location - Per 25 pair   UPA   USL-Freader, DSS Set-up per Cross Box location - Per 25 pair   USL-VILLE, USSPTW   487.23   11.90   USL-Freader, DSS Set-up per Cross Box location - Per 25 pair   USL-VILLE, USSPTW   6.26   6.26   11.90   USSPTW   6.26   6.26   11.90   USSPTW   6.26   6.26   11.90   USSPTW   6.26   6	SUB-LOOPS				_					_							
Distribution Facility Serving   UDNUCLUDILUDC USBFYV   487.23   11.90	Sub-Lo																<u> </u>
USE, Fender DSO Set up per Clores Box location - per 25 pair   UFA, USE PRIOR   USE, Fender DSI Set-up at DSX location, per DSI termination   USE, USEPX   52.54   11.30   11.90						l								1	]		1
Set-up   Set-up at DSX location, per DS1 termination   USL   USBF2   5.244   11.32   11.90				1		USBFW		487.23					11.90				<b></b>
USL   Seeder DSI Setup at DSX location, per DSI termination   USL   USBFZ   S2241   11.32     11.90						HODEN		0.05	0.05				44.00				i
Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice   1 UEA USBFA   8.05   92.75   51.24   58.45   13.07   11.90   11												1					<del></del>
Grade - Zone 1	-				USL	USBFZ		522.41	11.32				11.90				<del></del>
Grade - Zone 2		Grade - Zone 1		1	UEA	USBFA	8.05	92.75	51.24	58.45	13.07		11.90				
Unbundled Sub-Loop Feeder Loop, Pare 2 Wire Ground-Start, Voice Grade - Zone 1				2	LIEA	LICDEV	10.97	02.75	51.24	50 A5	12.07		11.00				
Voice Grade - Zone 3					ULA	USBI A	10.07	92.73	31.24	36.43	13.07		11.90				<del> </del>
Order Coordination for Specified Conversion Time, per LSR   UEA   USBFB   8.05   92.75   51.24   58.45   13.07   11.90				3	UEA	USBFA	21.00	92.75	51.24	58.45	13.07		11.90				i
Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 UEA USBFB 8.05 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 2 UEA USBFB 10.87 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 2 UEA USBFB 21.00 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 UEA USBFC 8.05 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 UEA USBFC 8.05 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 UEA USBFC 10.87 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 UEA USBFC 10.87 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 UEA USBFC 10.87 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 UEA USBFC 10.87 92.75 51.24 58.45 13.07 11.90 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 106.92 64.46 63.54 14.83 11.90 URA USBFC 17.26 URA USBFC 17									-								
Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Slart, Voice   2 UEA																	
Grade - Zone 2				1	UEA	USBFB	8.05	92.75	51.24	58.45	13.07		11.90				L
Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Viole Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 4 Wire Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub-Sub-							40.00		=	=0.4=							ĺ
Grade - Zone 3   3   UEA   USBFB   21.00   92.75   51.24   58.45   13.07   11.90				2	UEA	USBFB	10.87	92.75	51.24	58.45	13.07		11.90				+
Order Coordination for Specified Time Conversion, per LSR   UEA   USBFC   8.05   92.75   51.24   58.45   13.07   11.90   11.				2	LIEA	LICDED	21.00	02.75	E1 24	E0 4E	12.07		11 00				1
Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Viole Grade - Zone 1  Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Viole Grade - Zone 2  Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Viole Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Viole Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Viole Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 2  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 2  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3  Unbundled	-			3			21.00		51.24	58.45	13.07		11.90				<del>                                     </del>
Voice Grade - Zone 1				1	ULA	OCOSL		25.02									<del> </del>
Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,   2 UEA   USBFC   10.87   92.75   51.24   58.45   13.07   11.90				1	UEA	USBFC	8.05	92.75	51.24	58.45	13.07		11.90				1
Voice Grade - Zone 2			1	r			3.55	323	J2-	550	.0.07			1	1		
Battery, Voice Grade - Zone 3				2	UEA	USBFC	10.87	92.75	51.24	58.45	13.07		11.90				1
Order Coordination For Specified Conversion Time, per LSR																	
Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1				3			21.00		51.24	58.45	13.07	ļ	11.90				<b></b>
Grade - Zone 1					UEA	OCOSL		23.02						ļ	ļ		<del></del>
Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2				١.,		HODED	47.00	400.00	04.10	00 = 1	44.00		44.00				1
Grade - Zone 2				1	UEA	OSRLD	17.26	106.92	64.46	63.54	14.83	<b> </b>	11.90	<del>                                     </del>	<del>                                     </del>		<del></del>
Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 3				2	LIEΔ	LISBED	22.20	106.02	64.46	62.54	14.00		11 00	1	1		1
Grade - Zone 3	<del>    </del>		1		OLA	טיום פט	23.29	100.92	04.40	03.54	14.83	1	11.90	1	1		
Order Coordination For Specified Conversion Time, Per LSR				3	UEA	USBFD	45.00	106.92	64 46	63 54	14 83		11.90	1	1		1
Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice   Grade - Zone 1				Ť			.5.00		010	33.04							
Grade - Zone 1							1										ſ
Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice   2 UEA USBFE   23.29   106.92   64.46   63.54   14.83   11.90				1	UEA	USBFE	17.26	106.92	64.46	63.54	14.83		11.90				1
Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3		Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
Grade - Zone 3				2	UEA	USBFE	23.29	106.92	64.46	63.54	14.83		11.90	ļ	ļ		<b></b>
Order Coordination For Specified Conversion Time, Per LSR				_				, = = =									1
Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1 1 UDN USBFF 17.04 109.71 66.68 60.21 12.49 11.90 11				3			45.00		64.46	63.54	14.83		11.90				<del></del>
Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2				1			17.04		66.60	60.24	12 40	<b> </b>	11.00	<del>                                     </del>	<del>                                     </del>		<del></del>
Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3 3 UDN USBFF 44.43 109.71 66.68 60.21 12.49 11.90	<del>                                     </del>											<b>_</b>		-	1		<del>                                     </del>
	<del>                                     </del>											<b> </b>		1	1		
		Order Coordination For Specified Conversion Time, Per LSR	1		UDN	OCOSL	77.73	23.02	00.00	00.21	12.43		11.50	<b> </b>	<b> </b>		<u> </u>

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					RATES (\$)		
			<u> </u>	LIBO		.=	First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible) Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC UDC	USBFS USBFS	17.04 23.00	109.71 109.71	66.68 66.68	60.21 60.21	12.49 12.49		11.90 11.90	-			$\vdash$
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	44.43	109.71	66.68	60.21	12.49		11.90	1			$\vdash \vdash \vdash$
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1			USL	USBFG	46.27	133.77	78.02	85.16	21.21		11.90				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2			USL	USBFG	62.45	133.77	78.02	85.16	21.21		11.90				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	USL	USBFG	120.65	133.77	78.02	85.16	21.21		11.90				
	Order Coordination For Specified Conversion Time, Per LSR			USL	OCOSL		23.02									Ļ'
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1		1	UCL	USBFH	7.25	85.27	42.24	58.54	10.82		11.90				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 2		2	UCL	USBFH	9.79	85.27	42.24	58.54	10.82		11.90				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 3		3	UCL UCL	USBFH	18.92	85.27 23.02	42.24	58.54	10.82		11.90				
	Order Coordination For Specified Conversion Time, per LSR Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	14.22	99.66	57.20	60.98	12.28		11.90				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		_	UCL	USBFJ	19.20	99.66	57.20	60.98	12.28		11.90				$\vdash$
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	37.09	99.66	57.20	60.98	12.28		11.90				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		23.02									
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	18.68	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2		USBFN	25.21	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	48.71	100.62	58.16	63.54	14.83		11.90				$\longleftarrow$
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 1 Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		1	UDL	USBFO	18.68	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		2	UDL	USBFO	25.21	100.62	58.16	63.54	14.83		11.90				
	Zone 3 Order Coordination For Specified Time Conversion, per LSR		3	UDL UDL	USBFO OCOSL	48.71	100.62 23.02	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1		1	UDL	USBFP	18.68	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2		2	UDL	USBFP	25.21	100.62	58.16	63.54	14.83		11.90				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3		3	UDL	USBFP	48.71	100.62	58.16	63.54	14.83		11.90				
	Order Coordination For Specified Conversion Time, per LSR		Ť	UDL	OCOSL		23.02	00.10	00.01	1 1100		11.00				
SUB-LOOPS																
Sub-Loo								· · · · ·								
	Sub Loop Feeder - DS3 - Per Mile Per Month			UE3	1L5SL	15.69	2 222 22	107.1-	400.00	04.50		11.00				<del></del>
	Sub Loop Feeder - DS3 - Facility Termination Per Month Sub Loop Feeder - STS-1 - Per Mile Per Month	<u> </u>	-	UE3 UDLSX	USBF1 1L5SL	347.59 15.69	3,386.00	407.15	166.83	94.58	-	11.90	<del>                                     </del>			<del>                                     </del>
	Sub Loop Feeder - STS-1 - Fer Mille Fer Month			UDLSX	USBF7	402.09	3,386.00	407.15	166.83	94.58		11.90				<del>                                     </del>
	Sub Loop Feeder - OC-3 - Per Mile Per Month Sub Loop Feeder - OC-3 - Facility Termination Protection Per			UDLO3	1L5SL	11.90	0,000.00	107.110	100.00	0 1.00		1.100				
	Month			UDLO3	USBF5	62.98										1 '
	Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3	USBF2	547.22	3,386.00	407.15	166.83	94.58		11.90	1	İ		
	Sub Loop Feeder - OC-12 - Per Mile Per Month			UDL12	1L5SL	14.65										
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per Month			UDL12	USBF6	502.47										1
	Sub Loop Feeder - OC-12 - Facility Termination Per Month		1	UDL12	USBF3	1,577.00	3,386.00	407.15	166.83	94.58		11.90	<del> </del>			<del>                                     </del>
	Sub Loop Feeder - OC-48 - Per Mile Per Month			UDL48	1L5SL	48.06	2,222.00			200			1	İ		
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per Month			UDL48	USBF9	251.80										
	Sub Loop Feeder - OC-48 - Facility Termination Per Month			UDL48	USBF4	1,589.00	3,572.00	407.15	168.35	95.43		11.90				
	Sub Loop Feeder - OC-12 Interface On OC-48			UDL48	USBF8	331.15	788.39	407.15	168.35	95.43		11.90				
UNBUNDLED LC	OP CONCENTRATION		ļ		LIOTC		0=0.4-	0=0.1-								<b>├</b>
	Unbundled Loop Concentration - System A (TR008) Unbundled Loop Concentration - System B (TR008)			ULC ULC	UCT8A UCT8B	449.49 53.44	359.42 149.76	359.42 149.76				11.90 11.90	<del>                                     </del>			<del>                                     </del>
	Unbundled Loop Concentration - System B (TR008) Unbundled Loop Concentration - System A (TR303)		<b>-</b>	ULC	UCT3A	487.33	359.42	359.42			1	11.90	<del> </del>	1		$\vdash$
	onbunuled Loop Concentration - System A (1K303)	<u> </u>	<u> </u>	ULU	JUC 13A	487.33	359.42	309.42			<u> </u>	11.90	1	<u> </u>		

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Loop Concentration - System B (TR303)			ULC	UCT3B	90.05	149.76	149.76	10.10	4.00		11.90				
	Unbundled Loop Concentration - DS1 Loop Interface Card Unbundled Loop Concentration - ISDN Loop Interface (Brite			ULC	UCTCO	5.04	71.70	51.52	18.49	4.82		11.90				
	Card)			UDN	ULCC1	8.00	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - UDC Loop Interface (Brite			05.1	02001	0.00	10.00	10.00	0	0.7.0		11.00				
	Card)			UDC	ULCCU	8.00	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop Interface (POTS Card)			UEA	ULCC2	2.00	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery						40.50	40.50		. =-						
<del>                                     </del>	Loop Interface (SPOTS Card) Unbundled Loop Concentration - 4 Wire Voice Loop Interface			UEA	ULCCR	11.90	16.59	16.50	6.77	6.73		11.90				
	(Specials Card)			UEA	ULCC4	7.10	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	34.68	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop															
	Interface			UDL	ULCC7	10.51	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface			UDL	ULCC5	10.51	16.59	16.50	6.77	6.73		11.90				
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop			UDL	ULCC6	10.51	16 50	16.50	6.77	6.73		11.00				
LINE OTHER DE	Interface OVISIONING ONLY - NO RATE			UDL	ULCC6	10.51	16.59	16.50	6.77	6.73		11.90				
ONE OTHER, FF	NID - Dispatch and Service Order for NID installation			UENTW	UNDBX											
	UNTW Circuit Id Establishment, Provisioning Only - No Rate			UENTW	UENCE											
				UEANL,UEF,UEQ,U												
LINE OTHER RE	Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN											
UNE OTHER, PE	OVISIONING ONLY - NO RATE															
	Unbundled Contact Name, Provisioning Only - no rate			UAL,UCL,UDC,UDL, UDN,UEA,UHL,ULC	UNECN	0.00	0.00									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no															
	rate			UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			UEA,USL,UCL,UDL	USBFR	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
	Unbundled DS1 Loop - Expanded Superframe Format option -			002		0.00	0.00									
	no rate			USL	CCOEF	0.00	0.00									
	UNBUNDLED LOCAL LOOP															
NOTE: 4	month minimum billing period High Capacity Unbundled Local Loop - DS3 - Per Mile per															
	month			UE3	1L5ND	10.92										
	High Capacity Unbundled Local Loop - DS3 - Facility Termination per month			UE3	UE3PX	386.88	556.37	343.01	139.13	96.84		11.90				
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per month			UDLSX	1L5ND	10.92										
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month			UDLSX	UDLS1	426.60	556.37	343.01	139.13	96.84		11.90				
LOOP MAKE-UP				UDLOX	UDLST	426.60	556.57	343.01	139.13	90.04		11.90				
LOGI MARLE GI	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).			UMK	UMKLW		52.17	52.17								
	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).			UMK	UMKLP		55.07	55.07								
	Loop MakeupWith or Without Reservation, per working or spare facility queried (Mechanized)			UMK	PSUMK		0.6784	0.6784								
HIGH FREQUEN					. Joint	†	0.0704	0.0704								
	RS-CENTRAL OFFICE BASED															
	Line Sharing Splitter, per System 96 Line Capacity - True up pending approval by PSC	I		ULS	ULSDA	119.72	379.13	0.00	347.90	0.00		0.00				

	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: E
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Submitted Elec	Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Sharing Splitter, per System 24 Line Capacity - True up pending approval by PSC			ULS	ULSDB	29.93	379.13	0.00	347.90	0.00		0.00				
-+	Line Sharing Splitter, Per System, 8 Line Capacity	H	i i	ULS	ULSD8	8.33	150.00	0.00	150.00	0.00		0.00				
	Line Sharing-DLEC Owned Splitter in CO-CFA activaton-	<u> </u>	<u> </u>	020	02000	0.00	100.00	0.00	100.00	0.00		0.00				
	deactivation (per LSOD) - True up pending approval by PSC		- 1	ULS	ULSDG		115.72		86.29							
	Line Sharing-DLEC Owned Splitter in CO-CFA activation- deactivation (per occurance of each group of 24 lines) - True up pending approval by PSC			ULS	ULSDG		57.94		11.13							
END US	ER ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY	SPECT	RUM A													
	Line Sharing - per Line Activation - True up pending approval			_												
	by PSC			ULS	ULSDC	0.00	29.68	21.28	19.57	9.61	1	10.73				
	Line Sharing - per Subsequent Activity per Line Rearrangement - True up pending approval by PSC	١,		ULS	ULSDS		21.68	16.44				10.73				
	Line Splitting - per line activation DLEC owned splitter	'	<u> </u>	UEPSR UEPSB	UREOS	0.61	21.08	16.44				10.73				
	Line Splitting - per line activation BST owned - physical			UEPSR UEPSB	UREBP	0.638	29.68	21.28	19.57	9.61						
-	Line Splitting - per line activation BST owned - virtual			UEPSR UEPSB	UREBV	1.134	29.68	21.28	19.57	9.61						
UNBUNDLED TI																
INTERO	FFICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -								+							
	Per Mile per month			U1TVX	1L5XX	0.0091										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -															
	Facility Termination per month			U1TVX	U1TV2	25.32	47.35	31.78	18.31	7.03		11.90				
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per Mile per month			U1TVX	1L5XX	0.0091										
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat															
	Facility Termination per month Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade -			U1TVX	U1TR2	25.32	47.35	31.78	18.31	7.03		11.90				
	Per Mile per month	1		U1TVX	1L5XX	0.0091										
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade			01117	120701	0.0001	İ		İ							
	- Facility Termination per month			U1TVX	U1TV4	22.58	47.35	31.78	18.31	7.03		11.90				
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			U1TDX	1L5XX	0.0091										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
	Termination per month			U1TDX	U1TD5	18.44	47.35	31.78	18.31	7.03		11.90				
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			U1TDX	1L5XX	0.0091										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility															
	Termination per month			U1TDX	U1TD6	18.44	47.35	31.78	18.31	7.03	1	11.90				
INTERO	FFICE CHANNEL - DEDICATED TRANSPORT - DS1 Interoffice Channel - Dedicated Channel - DS1 - Per Mile per		-								<del>                                     </del>	-				
	month			U1TD1	1L5XX	0.1856										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			U1TD1	U1TF1	88.44	105.54	98.47	21.47	19.05		11.90			<u>.</u>	
INTERC	FFICE CHANNEL - DEDICATED TRANSPORT- DS3			ועווט	UTIFI	00.44	105.54	90.47	21.47	19.05		11.90				
- ATTENO	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per										<b>†</b>	<b>†</b>				
	month			U1TD3	1L5XX	3.87										
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			U1TD3	U1TF3	1,071.00	335.46	219.28	72.03	70.56		11.90				
INTERC	FFICE CHANNEL - DEDICATED TRANSPORT- STS-1			01103	01113	1,071.00	333.46	213.20	12.03	10.30		11.90				
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per						t		t							
	month		ļ	U1TS1	1L5XX	3.87										
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month			U1TS1	U1TFS	1,056.00	335.46	219.28	72.03	70.56		11.90				
			1	01101	01113	1,050.00	333.40	213.20	12.03	70.56	<u> </u>	11.90				
LOCAL	CHANNEL - DEDICATED TRANSPORT															

UNBUNDLE	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					RATES (\$)		
	Local Channel - Dedicated - 2-Wire Voice Grade per month -						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Zone 1		1	ULDVX	ULDV2	21.94	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 2-Wire Voice Grade per month -															
	Zone 2		2	ULDVX	ULDV2	29.62	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 2-Wire Voice Grade per month - Zone 3		3	UNDVX	ULDV2	57.22	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat. Per		Ü	CNDVX	OLDVZ	07.22	200.04	40.01	07.00	4.00		11.50				
	month - Zone 1		1	ULDVX	ULDR2	21.94	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat. Per		2	LII DVA	LII DDO	29.62	2005.04	40.07	27.02	4.00		44.00				
	Month - Zone 2 Local Channel - Dedicated - 2-Wire Voice Grade Rev. Bat. Per			ULDVX	ULDR2	29.62	265.84	46.97	37.63	4.00		11.90				
	Month - Zone 3		3	ULDVX	ULDR2	57.22	265.84	46.97	37.63	4.00		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade per month -															
	Zone 1		1	UNDVX	ULDV4	22.81	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade per month - Zone 2		2	UNDVX	ULDV4	30.79	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - 4-Wire Voice Grade per month -		_	O. I.D. I.Y.	02271	00.70	200.01		11122	0.00		11100				
	Zone 3		3	UNDVX	ULDV4	59.48	266.54	47.67	44.22	5.33		11.90				
	Local Channel - Dedicated - DS1 per month - Zone 1		1	ULDD1	ULDF1	35.28	216.65	183.54	24.30	16.95		11.90				
	Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 3		3	ULDD1 ULDD1	ULDF1 ULDF1	47.63 92.01	216.65 216.65	183.54 183.54	24.30 24.30	16.95 16.95		11.90 11.90				
	Local Channel - Dedicated - DS3 - Per Mile per month		J	ULDD3	1L5NC	8.50	210.03	103.54	24.30	10.33		11.30				
	Local Channel - Dedicated - DS3 - Facility Termination per															
	month			ULDD3	ULDF3	531.91	556.37	343.01	139.13	96.84		11.90				
	Local Channel - Dedicated - STS-1- Per Mile per month  Local Channel - Dedicated - STS-1 - Facility Termination per			ULDS1	1L5NC	8.50										
	month			ULDS1	ULDFS	540.69	556.37	343.01	139.13	96.84		11.90				
MULTIPLEXER	S					5.5.55										
	Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	146.77	101.42	71.62	11.09	10.49		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UDL	1D1DD	2.10	10.07	7.08				11.90				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per			UDL	טטוטו	2.10	10.07	7.08				11.90				
	month			UDN	UC1CA	3.66	10.07	7.08				11.90				
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	1.38	10.07	7.08				11.90				
	DS3 to DS1 Channel System per month			UXTD3	MQ3	211.19	199.28	118.64	40.34	39.07		11.90				
<del>                                     </del>	STS1 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) used with Loop per month			UXTS1 USL	MQ3 UC1D1	211.19 13.76	199.28 10.07	118.64 7.08	40.34	39.07	1	11.90 11.90				
DARK FIBER	255				30101	13.70	10.07	1.00			<u> </u>	11.00				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Channel			UDF	1L5DC	55.04	75101	400.00	252.21	000 **		44.00				
<del>                                     </del>	NRC Dark Fiber - Local Channel Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction			UDF	UDFC4		751.34	193.88	356.21	230.11	1	11.90				
	Thereof per month - Interoffice Channel			UDF	1L5DF	26.85										
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		751.34	193.88	356.21	230.11		11.90				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction			LIDE	41.50:	0:										
<del>                                     </del>	Thereof per month - Local Loop  NRC Dark Fiber - Local Loop			UDF UDF	1L5DL UDFL4	55.04	751.34	193.88	356.21	230.11		11.90				
TRANSPORT O				וטטו	ODI L4		751.34	193.08	330.21	230.11	1	11.90				
	al Features & Functions:															
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent -															
$\vdash$	per DS1 Channel		<b>—</b>	UNC1X	CCOEF		184.92	23.82	2.07	0.80	<b> </b>	11.90				
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel			UNC1X	CCOSF		184.92	23.82	2.07	0.80		11.90				
8XX ACCESS T	EN DIGIT SCREENING				33301		104.02	20.02	2.07	0.00	<u> </u>	11.50				
	8XX Access Ten Digit Screening, Per Call			OHD		0.0006252										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX			OLID	NODAY			0.70				44.00				
	Number Reserved	l		OHD	N8R1X	l	4.15	0.70			L	11.90	1			l

CATEGORY	NETWORK ELEMENTS - Florida										1	1	Attachment:	_		Exhibit: E
1	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
											per LSR	per LSR	1St	Addi	DISC 1St	DISC Add1
						Dee .	Names		Naa	D:			222	RATES (\$)		
$\vdash$						Rec	Nonrec First	urring Add'l	Nonrecurring First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O						11130	Addi	11130	Addi	JOHLE	JOHAN	JOHIAN	JONIAN	JOWAN	JOHIAN
	POTS Translations			OHD			8.78	1.18	5.77	0.70		11.90				
'	8XX Access Ten Digit Screening, Per 8XX No. Established With			OLID	NOETY		0.70	4.40	F 77	0.70		44.00				
	POTS Translations  8XX Access Ten Digit Screening, Customized Area of Service			OHD	N8FTX	-	8.78	1.18	5.77	0.70		11.90				
'	Per 8XX Number			OHD	N8FCX		4.15	2.07				11.90				
	8XX Access Ten Digit Screening, Multiple InterLATA CXR															
	Routing Per CXR Requested Per 8XX No.			OHD	N8FMX		4.85	2.78				11.90				
	8XX Access Ten Digit Screening, Change Charge Per Request 8XX Access Ten Digit Screening, Call Handling and Destination			OHD	N8FAX		4.85	0.70				11.90				<del></del>
	Features			OHD	N8FDX		4.15	4.15				11.90				
<u> </u>	8XX Access Ten Digit Screening, w/ 8XX No. Delivery, per query			OHD		0.0006252										
'	8XX Access Ten Digit Screening, w/ POTS No. Delivery, per			OHD		0.0006252										
LINE INFORMAT	TON DATA BASE ACCESS (LIDB)			OHD		0.0006252			+							<del></del>
	LIDB Common Transport Per Query			OQT		0.0000203										
	LIDB Validation Per Query			OQU		0.0136959										
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		55.13	55.13	55.13	55.13		11.90				
SIGNALING (CC				LIDD	DTOCY	425.05										<b>├</b>
	CCS7 Signaling Termination, Per STP Port CCS7 Signaling Usage, Per TCAP Message			UDB UDB	PT8SX	135.05 0.0000607										<del></del>
	CCS7 Signaling Connection, Per link (A link)			UDB	TPP++	17.93	43.57	43.57	18.31	18.31		11.90				
	CCS7 Signaling Connection, Per link (B link) (also known as D															
	link)			UDB	TPP++	17.93	43.57	43.57	18.31	18.31		11.90				
$\vdash$	CCS7 Signaling Usage, Per ISUP Message CCS7 Signaling Usage Surrogate, per link per LATA			UDB UDB	STU56	0.0000152 694.32										<del></del>
$\vdash$	CCS7 Signaling Osage Surrogate, per link per LATA  CCS7 Signaling Point Code, per Originating Point Code			UDB	31036	094.32										<del></del>
	Establishment or Change, per STP affected			UDB	CCAPO		46.03	46.03	46.03	46.03		11.90				İ
E911 SERVICE																
	Local Channel - Dedicated - 2-wr Voice Grade - Zone 1					21.94	265.84	46.97	37.63	4.00		11.90				
$\vdash$	Local Channel - Dedicated - 2-wr Voice Grade - Zone 2 Local Channel - Dedicated - 2-wr Voice Grade - Zone 3				+	29.62 57.22	265.84 265.84	46.97 46.97	37.63 37.63	4.00		11.90 11.90				<b>├</b> ──
$\vdash$	Interoffice Transport - Dedicated - 2-wr Voice Grade - 2-wr Wile					0.0091	203.04	40.57	37.03	4.00		11.90				<del>                                     </del>
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility															
	Termination					25.32	47.35	31.78	18.31	7.03		11.90				
igwdot	Local Channel - Dedicated - DS1 - Zone 1 Local Channel - Dedicated - DS1 - Zone 2					35.28 47.63	216.65 216.65	183.54 183.54	21.47 21.47	19.05 19.05		11.90 11.90				<b>├</b>
	Local Channel - Dedicated - DS1 - Zone 2  Local Channel - Dedicated - DS1 - Zone 3					92.01	216.65	183.54	21.47	19.05		11.90				<del></del>
	Interoffice Transport - Dedicated - DS1 Per Mile					0.1856	210.00	100.04	21.47	10.00		11.50				
	·															
	Interoffice Transport - Dedicated - DS1 Per Facility Termination					88.44	105.54	98.47	21.47	19.05		11.90				ــــــ
	(CNAM) SERVICE CNAM for DB Owners, Per Query			OQV		0.001024										——
	CNAM for Non DB Owners, Per Query			OQV		0.001024			1							$\vdash$
	CNAM For DB Owners - Service Establishment			OQV		0.001024	25.35	25.35	19.01	19.01		11.90				
	CNAM For Non DB Owners - Service Establishment			OQV			25.35	25.35	19.01	19.01		11.90				
1 1 7	CNAM For DB Owners - Service Provisioning With Point Code					Ι Τ	4 500 60	4 477 00	050.00	050.00		44.65				
$\vdash$	Establishment  CNAM For Non DB Owners - Service Provisioning With Point			OQV			1,592.00	1,177.00	352.36	259.09	1	11.90				<del></del>
1 1 '	Code Establishment			OQV			546.51	393.82	358.06	259.09		11.90				
LNP Query Servi						<del> </del>	3.3.31	555.02	555.00	200.00		50				
	LNP Charge Per query			OQV		0.000852										
	LNP Service Establishment Manual			·		1	13.83	13.83	12.71	12.71	1	11.90			l —	1
	LNP Service Provisioning with Point Code Establishment						655.50	334.88	297.03	218.40	1	11.90				

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					RATES (\$)		
-	Oper. Call Processing - Oper. Provided, Per Min Using BST						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	LIDB					1.20										
	Oper. Call Processing - Oper. Provided, Per Min Using															
	Foreign LIDB  Oper. Call Processing - Fully Automated, per Call - Using BST					1.24										$\longleftarrow$
	LIDB					0.20										1
	Oper. Call Processing - Fully Automated, per Call - Using															
INWARD ORED	Foreign LIDB ATOR SERVICES					0.20										$\vdash$
INWARD OFER	Inward Operator Services - Verification, Per Call					1.00										<del>                                     </del>
	Inward Operator Services - Verification and Emergency Interrupt					1.00										
	- Per Call					1.95										
BRANDING - OF	PERATOR CALL PROCESSING				CDAGC		7 000 00	7 000 00				44.00				
	Recording of Custom Branded OA Announcement Loading of Custom Branded OA Announcement per shelf/NAV				CBAOS CBAOL		7,000.00 500.00	7,000.00 500.00				11.90 11.90				<del></del>
Unbrand	ling via OLNS for UNEP CLEC				CBAOL		300.00	300.00				11.90				
	Loading of OA per OCN (Regional)						1,200.00	1,200.00				11.90				
	SISTANCE SERVICES															
DIRECT	ORY ASSISTANCE ACCESS SERVICE															
DIRECT	Directory Assistance Access Service Calls, Charge Per Call ORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DA	ACC)			-	0.271744										├──
DIRECT	Directory Assistance Call Completion Access Service (DACC),	ACC)	1													<del>                                     </del>
	Per Call Attempt					0.10										1
DIRECT	ORY TRANSPORT															
	SWA Common transport per Directory Assistance Access Service Call					0.0003										
	SWA Common Transport per Directory Assistance Access Service Call Mile					0.00004										
	Access Tandem Switching per Directory Assistance Access Service Call					0.00055										
	Directory Assistance Interconnection per Directory Assistance Access Service Call					0.00										
DIRECTORY AS	DS3 to DS1 Multiplexer per DA Access Service Call SISTANCE SERVICES					0.00018										<del></del>
	ORY ASSISTANCE DATA BASE SERVICE (DADS)										<b> </b>	1				$\vdash$
	Directory Assistance Data Base Service Charge Per Listing					0.04										
	Directory Assistance Data Base Service, per month				DBSOF	150.00										
	RECTORY ASSISTANCE Based CLEC															$\vdash$
Facility	Recording and Provisioning of DA Custom Branded		$\vdash$													$\vdash$
	Announcement Loading of Custom Branded Announcement per DRAM			AMT	CBADA		6,000.00	6,000.00								
UNEP C	Card/Switch			AMT	CBADC		1,170.00	1,170.00								
UNEPC	Recording of DA Custom Branded Announcement		$\vdash$				3,000.00	3,000.00								$\vdash$
	Loading of DA Custom Branded Announcement per DRAM						3,300.00	5,500.00								
	Card/Switch per OCN						1,170.00	1,170.00								
	ling via OLNS for UNEP CLEC															$ldsymbol{oxed}$
	Loading of DA per OCN (1 OCN per Order)		$\vdash$				420.00	420.00				1				$\longmapsto$
SELECTIVE RO	Loading of DA per Switch per OCN		$\vdash$		1		16.00	16.00								$\vdash$
JEELSHIVE RO	Selective Routing Per Unique Line Class Code Per Request Per		H													$\vdash$
	Switch				USRCR		93.55	93.55	12.71	12.71		11.90				<u>1</u> J
VIRTUAL COLL						_										lacksquare
	Virtual Collocation - Application Cost		$\vdash$	CLO	EAF		4,122.00	2,848.30				1				<b></b>
	Virtual Collocation - Cable Installation Cost, per cable Virtual Collocation - Floor Space, per sq. ft.		$\vdash$	CLO CLO	ESPCX ESPVX	4.25	965.00	2,750.00			-	}		1		$\vdash$
	virtual Collocation - Floor Space, per sq. ft.	L		CLU	ESHAY	4.25			l .		<u> </u>	I	1	1		

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Virtual Collocation - Power, per breaker amp			CLO	ESPAX	6.95										<b></b>
	Virtual Collocation - Cable Support Structure, per entrance															1 '
	cable			CLO	ESPSX	13.35										<b></b>
	Virtual Collocation - 2-wire Cross Connects (loop), per 100 ckts			ueanl,uea,udn,udc, ual,uhl,ucl,ueq	UEAC2	5.02	1,157.00	1,157.00				11.90				
	Nintered Collegetion Assists Constants (Icon) and 400 olds			المند الماد الماد	UEAC4	5.00	4 457 00	1,157.00				44.00				1 '
$\vdash$	Virtual Collocation - 4-wire Cross Connects (loop), per 100 ckts Virtual Collocation - 2-Fiber Cross Connects	<b>!</b>	<del>                                     </del>	uea,uhl,ucl,udl CLO	CNC2F	5.02 6.71	1,157.00 2,431.00	1,157.00		-	<b> </b>	11.90 11.90	-	-		<del>                                     </del>
	Virtual Collocation - 4-Fiber Cross Connects			CLO	CNC2F CNC4F	6.71	2,431.00					11.90				<del></del>
<del></del>	Virtual Collocation - 4-Fiber Cross Connects  Virtual Collocatin - DS1 Cross Connects	<b>-</b>	<del>                                     </del>	USL,ULC,CLO	CNC4F CNC1X	7.50	155.00	14.00		-	<b>-</b>	11.90	<b> </b>	-		<del>                                     </del>
$\vdash$	Virtual Collocatin - DS3 Cross Connects  Virtual Collocatin - DS3 Cross Connects	1		USL,ULC,CLO	CNC1X CND3X	7.50 56.25	155.00	11.83		1	<b> </b>	11.90	1	1		<del></del>
<del>                                     </del>	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable	1	<del>                                     </del>	OOL,OLO,OLO	CINDOX	50.25	131.90	11.03		<del> </del>		11.90	<del> </del>	<del> </del>		
	Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			AMTFS	PE1ES	0.0028										
	Cable Support Structure, per linear ft			AMTFS	PE1DS	0.0041										1 '
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable			AWITTO	LIDO	0.0041										<b>——</b>
	Support Structure, per cable			AMTFS			535.54									1 '
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax			741111			000.01									
	Cable Support Structure, per cable			AMTFS			535.54									1 '
	Virtual Collocatin - Security Escort - Basic, per quarter hour			CLO	SPTBQ		10.89									
	,															
	Virtual Collocatin - Security Escort - Overtime, per quarter hour			CLO	SPTOQ		13.64									
	Virtual Collocatin - Security Escort - Premium, per quarter hour			CLO	SPTPQ		16.40									
	Virtual Collocation - 2-wire Cross Connects (loop), per 100 ckts			CLO		5.02	1,157.00									
	Virtual Collocation - 4-wire Cross Connects (loop), per 100 ckts			CLO		5.02	1,157.00									i
<del> </del>	Virtual Collocation - DS-1/DCS, PER 28 CKTS		1	CLO	VE11S	226.39	1,950.00				1	1				<del></del>
<b>—</b>	Virtual Collocation - DS-1.DSX, PER 28 CKTS			CLO	VE11X	11.51	1,950.00				1					
	Virtual Collocation - DS-3/DCS, PER CKT			CLO	VE13S	56.97	528.00									
	Virtual Collocation - DS-3/DSC, PER CKT			CLO	VE13X	10.06	528.00									
	Virtual Collocation - Virtual to Virtual connection, per fiber, per															
-	cable Virtual Collocation - Virtual to Virtual connection - DS1/DS-3, per			CLO		0.19	526.17									
	cable			CLO		0.17	134.46									
	Virtual Collocatin - Maintenance in CO - Basic, per quarter hour	l		CLO	SPTRE		10.89									1
	Virtual Collocatin - Maintenance in CO - Overtime, per quarter	1				t	10.00						<b> </b>	<b> </b>		<u> </u>
	hour Virtual Collocatin - Maintenance in CO - Premium per quarter			CLO	SPTOE		13.64									
VIDTUAL COLL	hour			CLO	SPTPE		16.40									<u> </u>
VIRTUAL COLL	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-	l	-								<b> </b>	<u> </u>	ļ	-		<del></del>
	Wire Analog - Res			UEPSR	VE1R2	0.524	11.57	11.57				11.90				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire			LIEDBY	DE4D0	0.504	44.57	44 57				44.00				1
<del></del>	Voice Grade Res Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-	<del>                                     </del>	-	UEPRX	PE1R2	0.524	11.57	11.57		<b> </b>	<b> </b>	11.90	<del>                                     </del>	<del>                                     </del>		<del></del>
	Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.524	11.57	11.57				11.90				ļ
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX Trunk - Res			UEPSE	VE1R2	0.524	11.57	11.57				11.90				
1 1	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire	1								]			]	]		1
	Analog Bus Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire		-	UEPSB	VE1R2	0.524	11.57	11.57				11.90				1
	ISDN			UEPSX	VE1R2	0.524	11.57	11.57				11.90				<u> </u>
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			UEPTX	VE1R2	0.524	11.57	11.57				11.90				

UNBU	INDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATI	EGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
							Rec	Nonrec			g Disconnect	COMEC	SOMAN		RATES (\$)	SOMAN	SOMAN
		Virtual Collocation 4-Wire Cross Connect, Exchange Port DDITS 4-Wire DS1			UEPDD	VE1R4	0.524	First 11.57	Add'I 11.57	First	Add'l	SOMEC	11.90	SOMAN	SOMAN	SOMAN	SOMAN
		Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			UEPEX	VE1R4	0.524	11.57	11.57				11.90				
VIRTU	AL COLLO	OCATION			UEPEX	VE IK4	0.524	11.57	11.57				11.90	1			<del>                                     </del>
VIII (10)	l GOLL	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.0297	33.86	31.95				10.73				
AIN SE	LECTIVE	CARRIER ROUTING			OLFSK, OLFSB	VLILO	0.0291	33.00	31.93				10.73				<b>—</b>
		Regional Service Establishment			SRC	SRCEC		193,444.00		7,737.00			11.90				
		End Office Establishment			SRC	SRCEO		187.36	187.36	0.69	0.69		11.90				
AIN D	ELLEON	Query NRC, per query TH AIN SMS ACCESS SERVICE			SRC		0.0031868					-	-				<del></del>
AIN - B	ELLSOU	AIN SMS ACCESS SERVICE AIN SMS Access Service - Service Establishment, Per State,										<del>                                     </del>	<del>                                     </del>				<del>                                     </del>
		Initial Setup			A1N	CAMSE		43.56	43.56	44.93	44.93		11.90				<del>                                     </del>
		AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		8.64	8.64	10.03	10.03		11.90				
		AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		8.64	8.64	10.03	10.03		11.90				
		AIN SMS Access Service - User Identification Codes - Per User ID Code			A1N	CAMAU		38.66	38.66	29.88	29.88		11.90				
		AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			A1N	CAMRC		75.10	75.10	12.93	12.93		11.90				
		AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)			, , , ,	07 4711 10	0.0028	70.10	70.10	12.00	12.00		11.00				
		AIN SMS Access Service - Session, Per Minute					0.7809										
		AIN SMS Access Service - Company Performed Session, Per Minute					0.4609										
AIN - B	ELLSOU	TH AIN TOOLKIT SERVICE					0.4609										
		AIN Toolkit Service - Service Establishment Charge, Per State,															
		Initial Setup			CAM	BAPSC		43.56	43.56	44.93	44.93		11.90				
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAPVX		8,439.00	8,439.00				11.90				<del> </del>
		DN, Term. Attempt				BAPTT		8.64	8.64	10.03	10.03		11.90				
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		8.64	8.64	10.03	10.03		11.90				
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		8.64	8.64	10.03	10.03		11.90				<u> </u>
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				ВАРТО		38.06	38.06	15.86	15.86		11.90				
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP				ВАРТС		38.06	38.06	15.86	15.86		11.90				
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code				BAPTF		38.06	38.06	15.86	15.86		11.90				
		AIN Toolkit Service - Query Charge, Per Query					0.0535927	55.55	55.56	10.00	10.00		11.50				
		AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.0063698										
		AllN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100 Kilobytes					0.06										
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service			CAM	DADMO		9.64	9.64	6.00	6.00		11.00				
		Subscription AIN Toolkit Service - Special Study - Per AIN Toolkit Service			CAM	BAPMS	8.34	8.64	8.64	6.08	6.08		11.90				
		Subscription AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service			CAM	BAPLS	3.73	9.56	9.56				11.90				
		Subscription AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit			CAM	BAPDS	4.73	8.64	8.64	6.08	6.08		11.90				
ΕΝΗΔΝ	ICED FYT	Service Subscription FENDED LINK (EELs)			CAM	BAPES	0.12	9.56	9.56			<del>                                     </del>	11.90	1			<del>                                     </del>
		lew EELs available in State of Georgia, density zone 1 of follow	wing Si	MAs: O	rlando, FL; Miami, F	L; Ft. Laude	rdale, FLI; Nasl	nville, TN; New	Orleans, LA;					1			
		harlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-H															

UNBUNDLE	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec		curring		g Disconnect				RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NOTE: I	n all states, EEL network elements shown below also apply to	curren	ly con	bined facilities whi	ch are conve	rted to UNE rate	es. A Switch A	s Is Charge ap	plies to curren	tly combined	acilities co	nverted to U	JNEs.(Non-rec	urring rates of	lo not apply.)	
NOTE: I	n GA, TN, KY, LA & MS, the EEL network elements apply to or	dinarily	comb	ned network elemei												
2-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTE First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport	ROFFIC	ETRA	NSPORT (EEL)	1											<del>                                     </del>
	Combination - Zone 1		1	UNCVX	UEAL2	14.50	127.59	60.54	48.00	6.31		11.90				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
	Transport Combination - Zone 2		2	UNCVX	UEAL2	19.57	127.59	60.54	48.00	6.31		11.90				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3		3	UNCVX	UEAL2	37.82	127.59	60.54	48.00	6.31		11.90				ĺ
	Interoffice Transport - Dedicated - DS1 combination - Per Mile				41 = 204											
	per month Interoffice Transport - Dedicated - DS1 combination - Facility			UNC1X	1L5XX	0.1856					-	-				<del>                                     </del>
	Termination per month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	DS1 Channelization System Per Month			UNC1X	MQ1	146.77	57.28	14.74	1.50	1.34						
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	1.38	6.71	4.84								<b>.</b>
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	14.50	127.59	60.54	48.00	6.31		11.90				ĺ
	Each Additional 2-Wire VG Loop(SL2) in the same DS1															
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	19.57	127.59	60.54	48.00	6.31		11.90				<b></b>
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	37.82	127.59	60.54	48.00	6.31		11.90				
	Voice Grade COCI - DS1 to DS0 Channel System combination -		- 3		OLALZ	37.02	127.00	00.34	40.00	0.51		11.30				
	per month			UNCVX	1D1VG	1.38	6.71	4.84								
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIRE	VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTE	ROFFIC	E TRA		ONOCC		0.30	0.50	0.30	0.90		11.30				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1		1	UNCVX	UEAL4	23.02	127.59	60.54	48.00	6.31		11.90				<b></b>
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	31.07	127.59	60.54	48.00	6.31		11.90				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice			LINOVA	UE AL 4	00.00	407.50	00.54	40.00	0.04		44.00				
	Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCVX	UEAL4	60.02	127.59	60.54	48.00	6.31		11.90				
	Per Month			UNC1X	1L5XX	0.1856										1
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	Channelization - Channel System DS1 to DS0 combination Per			ONCIX	UTIT	00.44	174.40	122.40	45.01	17.55		11.50				
	Month			UNC1X	MQ1	146.77	57.28	14.74	1.50	1.34						<b></b>
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	1.38	6.71	4.84								
	Additional 4-Wire Analog Voice Grade Loop in same DS1															
	Interoffice Transport Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1		1	UNCVX	UEAL4	23.02	127.59	60.54	48.00	6.31		11.90				<u> </u>
	Interoffice Transport Combination - Zone 2	<u> </u>	2	UNCVX	UEAL4	31.07	127.59	60.54	48.00	6.31		11.90				<u>i                                     </u>
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	60.02	127.59	60.54	48.00	6.31		11.90				
	Voice Grade COCI - DS1 to DS0 Channel System combination -		3		UEAL4	60.02	127.59	60.54	48.00	0.31		11.90				
	per month			UNCVX	1D1VG	1.38	6.71	4.84								<b> </b>
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				İ
4-WIRE	56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 IN	NTEROF	FICE 1	RANSPORT (EEL)			-									
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	26.39	127.59	60.54	48.00	6.31		11.90				1
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice										t					
	Transport Combination - Zone 2		2	UNCDX	UDL56	35.62	127.59	60.54	48.00	6.31	1	11.90				<u> </u>

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Submitted Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	68.82	127.59	60.54	48.00	6.31		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	146.77	57.28	14.74	1.50	1.34						
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	2.10	6.71	4.84								
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	26.39	127.59	60.54	48.00	6.31		11.90				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	35.62	127.59	60.54	48.00	6.31		11.90				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	68.82	127.59	60.54	48.00	6.31		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)			UNCDX	1D1DD	2.10	6.71	4.84								
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIRE	64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 IN	NTERO	FICE 1	RANSPORT (EEL)												
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	26.39	127.59	60.54	48.00	6.31		11.90				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	35.62	127.59	60.54	48.00	6.31		11.90				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	68.82	127.59	60.54	48.00	6.31		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	146.77	57.28	14.74	1.50	1.34						
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)			UNCDX	1D1DD	2.10	6.71	4.84								
	Combination - per month (2.4-64kbs) Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	26.39	127.59	60.54	48.00	6.31		11.90				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	35.62	127.59	60.54	48.00	6.31		11.90				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	68.82	127.59	60.54	48.00	6.31		11.90				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)		3	UNCDX	1D1DD	2.10	6.71	4.84	40.00	0.31		11.50				
	Combination - per month (2.4-64kbs)  Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC	2.10	8.98	8.98	8.98	8.98		11.90				
4-WIRE	IS Charge DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTER	ROFFIC	F TRAI		UNCCC		0.98	0.98	0.98	0.98	1	11.90				
7 7777	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1		1	UNC1X	USLXX	73.44	217.75	121.62	51.44	14.45		11.90				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	99.13	217.75	121.62	51.44	14.45		11.90				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3		3	UNC1X	USLXX	191.51	217.75	121.62	51.44	14.45		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				

UNBUNDLE	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)	I			Submitted Manually		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					RATES (\$)		
	Name of the Control Name of Figure 2 (1)						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIRE	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTE	ROFFIC	E TRAI				0.00		3.00	9.00						
	First DS1Loop in DS3 Interoffice Transport Combination - Zone						0.17.77	404.00								
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	73.44	217.75	121.62	51.44	14.45		11.90				<del></del>
	2		2	UNC1X	USLXX	99.13	217.75	121.62	51.44	14.45		11.90				
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		3	LINICAV	USLXX	191.51	217.75	121.62	E1 11	14.45		11 00				
	Interoffice Transport - Dedicated - DS3 combination - Per Mile		3	UNC1X	USLXX	191.51	217.75	121.02	51.44	14.45		11.90				<del>                                     </del>
	Per Month			UNC3X	1L5XX	3.87										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per			LINICOV	LIATEO	4 074 00	220.00	400.00	20.00	40.04		44.00				
-	month DS3 to DS1 Channel System combination per month			UNC3X UNC3X	U1TF3 MQ3	1,071.00 211.19	320.00 115.50	138.20 56.54	38.60 12.16	18.81 4.26		11.90				-
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.76	6.71	4.84	12.10	4.20						
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 1		1	UNC1X	USLXX	73.44	217.75	121.62	51.44	14.45		11.90				
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	99.13	217.75	121.62	51.44	14.45		11.90				
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 3 DS3 Interface Unit (DS1 COCI) combination per month		3	UNC1X	USLXX UC1D1	191.51	217.75	121.62	51.44	14.45		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNC1X	UCIDI	13.76	6.71	4.84								
	Is Charge			UNC3X	UNCCC		8.98	8.98	8.98	8.98		11.90				
2-WIRE	VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTE	ROFFI	CE TRA	NSPORT (EEL)												
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	14.50	127.59	60.54	48.00	6.31		11.90				ĺ
	2-WireVG Loop used with 2-wire VG Interoffice Transport		-	UNCVA	UEALZ	14.50	127.59	60.54	46.00	0.31		11.90				<del> </del>
	Combination - Zone 2		2	UNCVX	UEAL2	19.57	127.59	60.54	48.00	6.31		11.90				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	37.82	407.50	60.54	48.00	6.31		11.90				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per		3	UNCVX	UEAL2	31.82	127.59	60.54	48.00	6.31		11.90				
	Mile Per Month			UNCVX	1L5XX	0.0091										
	Interoffice Transport - Dedicated - 2- Wire Voice Grade							=====	45.00							ĺ
	combination - Facility Termination per month  Nonrecurring Currently Combined Network Elements Switch -As-			UNCVX	U1TV2	25.32	94.70	52.59	45.28	18.03		11.90				<del></del>
	Is Charge			UNCVX	UNCCC		8.98	8.98	8.98	8.98		11.90				
4-WIRE	VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTE	ROFFI	CE TRA	NSPORT (EEL)												
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1		4	UNCVX	UEAL4	23.02	127.59	60.54	48.00	6.31		11.90				ĺ
	4-WireVG Loop used with 4-wire VG Interoffice Transport		<u>'</u>	ONOVA	ULAL4	23.02	127.39	00.54	40.00	0.31		11.90				
	Combination - Zone 2		2	UNCVX	UEAL4	31.07	127.59	60.54	48.00	6.31		11.90				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	60.02	127.59	60.54	48.00	6.31		11.90				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per		3	ONCVX	OLAL4	00.02	127.59	00.34	48.00	0.31		11.90				<del></del>
	Mile Per Month			UNCVX	1L5XX	0.0091										
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility Termination per month			UNCVX	U1TV4	22.58	94.70	52.59	45.28	18.03		11.90				1
	Nonrecurring Currently Combined Network Elements Switch -As-			OINCVA	01174	22.58	94.70	5∠.59	45.28	18.03	-	11.90				<del>                                     </del>
	Is Charge			UNCVX	UNCCC		8.98	8.98	8.98	8.98		11.90				
DS3 DI	GITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE	TRAN	SPORT	(EEL)							ļ	ļ				
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month			UNC3X	1L5ND	10.92										İ
	High Capacity Unbundled Local Loop - DS3 combination -										1					
	Facility Termination per month			UNC3X	UE3PX	386.88	226.42	154.73	67.10	26.27						<b></b>
LL	Interoffice Transport - Dedicated - DS3 - Per Mile per month	1		UNC3X	1L5XX	3.87			l	I.	<u> </u>	<u> </u>	<u> </u>	<u> </u>		<u> </u>

UNBUNDLED !	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)			Submitted Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
-	Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per month			UNC3X	U1TF3	1,071.00	320.00	138.20	38.60	18.81		11.90				
	Nonrecurring Currently Combined Network Elements Switch -As- ls Charge			UNC3X	UNCCC		8.98	8.98	8.98	8.98		11.90				
	GITAL EXTENDED LOOP WITH DEDICATED STS1 INTEROFF	CE TRA	NSPO		011000		0.50	0.00	0.00	0.00		11.00				
	High Capacity Unbundled Local Loop - STS1 combination - Per			<u> </u>												
	Mile per month			UNCSX	1L5ND	10.92										
l l	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month			UNCSX	UDLS1	426.60	226.42	154.73	67.10	26.27						
	Interoffice Transport - Dedicated - STS1 combination - Per Mile per month			UNCSX	1L5XX	3.87										
i	Interoffice Transport - Dedicated - STS1 combination - Facility															
	Termination per month  Nonrecurring Currently Combined Network Elements Switch -As-			UNCSX	U1TFS	1,056.00	320.00	138.20	38.60	18.81		11.90				
	Is Charge			UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				
2-WIRE IS	SDN EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT	(EEL)														
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1		1	UNCNX	U1L2X	21.76	127.59	60.54	48.00	6.31		11.90				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination															
	Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination		2	UNCNX	U1L2X	29.38	127.59	60.54	48.00	6.31		11.90				
	Transport - Zone 3		3	UNCNX	U1L2X	56.76	127.59	60.54	48.00	6.31		11.90				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.1856										
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per month			UNC1X	U1TF1	88.44	174.46	122.46	45.61	17.95		11.90				
	Channelization - Channel System DS1 to DS0 combination - per month			UNC1X	MQ1	146.77	57.28	14.74	1.50	1.34						
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System			0.1017			01.20		1.00							
	combination - per month			UNCNX	UC1CA	3.66	6.71	4.84								
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 1		1	UNCNX	U1L2X	21.76	127.59	60.54	48.00	6.31		11.90				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 2		2	UNCNX	U1L2X	29.38	127.59	60.54	48.00	6.31		11.90				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone 3		3	UNCNX	U1L2X	56.76	127.59	60.54	48.00	6.31		11.90				
1	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System		3						40.00	0.31		11.90				
	combintaion- per month  Nonrecurring Currently Combined Network Elements Switch -As-			UNCNX	UC1CA	3.66	6.71	4.84								<del>                                     </del>
	Is Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
	OS1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INT	EROFF	ICE TR	ANSPORT (EEL)												
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	73.44	217.75	121.62	51.44	14.45		11.90				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	99.13	217.75	121.62	51.44	14.45		11.90				
i i	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	191.51	217.75	121.62	51.44	14.45		11.90				
ı	Per Month		Ŭ	UNCSX	1L5XX	3.87	20	.202	0.1.44	10		50				
l l	Termination - Facility Termination			UNCSX	U1TFS	1,056.00	320.00	138.20	38.60	18.81		11.90				
	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	211.19	320.00	130.20	30.00	10.01		11.30				
I	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	13.76	6.71	4.84								
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	73.44	217.75	121.62	51.44	14.45		11.90				<u> </u>
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	99.13	217.75	121.62	51.44	14.45		11.90				

UNBUNDI FE	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually	Incremental Charge - Manual Svc	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonre	curring	Nonrecurring	g Disconnect			oss i	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Additional DS1Loop in STS1 Interoffice Transport Combination -		_													ł
	Zone 3		3	UNC1X	USLXX	191.51 13.76	217.75	121.62 4.84	51.44	14.45		11.90				<b></b>
	DS3 Interface Unit (DS1 COCI) combination per month  Nonrecurring Currently Combined Network Elements Switch -As-			UNC1X	UC1D1	13.76	6.71	4.84			1					<b></b>
	Is Charge	1		UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				ł
4-WIRF	56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFI	FICE TE	ANSP		UNCCC		0.90	0.90	0.90	0.50		11.90				
4 111112	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport	10 <u>2 11</u>		OITT (LLL)												
	Combination - Zone 1		1	UNCDX	UDL56	26.39	127.59	60.54	48.00	6.31		11.90				ł
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
	Combination - Zone 2		2	UNCDX	UDL56	35.62	127.59	60.54	48.00	6.31		11.90				<u> </u>
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
	Combination - Zone 3		3	UNCDX	UDL56	68.82	127.59	60.54	48.00	6.31		11.90				<b></b>
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination -			LINODY	41.500/	0.0004										ł
+	Per Mile			UNCDX	1L5XX	0.0091					1					<del></del>
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Termination			UNCDX	U1TD5	18.44	94.70	52.59	45.28	18.03		11.90				ł
	Nonrecurring Currently Combined Network Elements Switch -As-			UNCDA	01103	10.44	34.70	32.39	45.20	10.03		11.90				
	Is Charge			UNCDX	UNCCC		8.98	8.98	8.98	8.98		11.90				ł
4-WIRE	64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROF	FICE TE	RANSP													
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport			` '												1
	Combination - Zone 1		1	UNCDX	UDL64	26.39	127.59	60.54	48.00	6.31		11.90				<u> </u>
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport															ĺ
	Combination - Zone 2		2	UNCDX	UDL64	35.62	127.59	60.54	48.00	6.31		11.90				<u> </u>
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	68.82	127.59	60.54	48.00	6.31		11.90				
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile			UNCDX	1L5XX	0.0091										
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination -			LINCDY	LIATEC	40.44	04.70	50.50	45.00	40.00		44.00				ł
+	Facility Termination			UNCDX	U1TD6	18.44	94.70	52.59	45.28	18.03	1	11.90				<del></del>
	Nonrecurring Currently Combined Network Elements Switch -As- Is Charge	1		UNCDX	UNCCC		8.98	8.98	8.98	8.98		11.90				ł
ADDITIONAL N	ETWORK ELEMENTS			UNCDA	UNCCC		0.90	0.90	0.90	0.50		11.90				
	sed as a part of a currently combined facility, the non-recurrn	a charc	ies do	not apply, but a Swi	tch As Is ch	arge does app	v.									
	sed as ordinarilty combined network elements in Georgia, the															i
Node (S	SynchroNet)															i
Nonrecu	urring Currently Combined Network Elements "Switch As Is" C	harge (	One ap	plies to each combi	nation)			-							·	
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge			UNCVX	UNCCC		8.98	8.98	8.98	8.98		11.90				<u></u>
	56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge			UNCDX	UNCCC		8.98	8.98	8.98	8.98		11.90				
	DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge			UNC1X	UNCCC		8.98	8.98	8.98	8.98		11.90				
	DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge			UNC3X	UNCCC		8.98	8.98	8.98	8.98		11.90				
	STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is" Conversion Charge			UNCSX	UNCCC		8.98	8.98	8.98	8.98		11.90				
NOTE: L	ocal Channel - Dedicated Transport - minimum billing period	- Below	DS3=0			months										i
UNBUNDLED L	OCAL EXCHANGE SWITCHING(PORTS)															
	ge Ports															ļ
	Although the Port Rate includes all available features in GA, K	Y, LA &	TN, the	e desired features wi	Il need to be	ordered using	retail USOCs									<del></del>
2-WIRE	VOICE GRADE LINE PORT RATES (RES)	-	-	LIEDOD	HEDDI	4.40	0.74	2.00	4.00	4.00		44.00				<del></del>
$\vdash$	Exchange Ports - 2-Wire Analog Line Port- Res.	<u> </u>	-	UEPSR	UEPRL	1.40	3.74	3.63	1.88	1.80	-	11.90				
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.40	3.74	3.63	1.88	1.80		11.90				<b> </b>
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.40	3.74	3.63	1.88	1.80		11.90				l

NBUNDLED	NETWORK ELEMENTS - Florida												Attachment: Incremental	2 Incremental	Incremental	Exhibit:
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge
						Rec	Nonrec			g Disconnect				RATES (\$)		
	Fush on a Posto O Wire VC unburglad Florida area celling with						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAI
	Exchange Ports - 2-Wire VG unbundled Florida area calling with Caller ID - Res.			UEPSR	UEPAF	1.40	3.74	3.63	1.88	1.80		11.90				
	Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID (LUM)			UEPSR	UEPAP	1.40	3.74	3.63	1.88	1.80		11.90				
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00								
FEATUR																
	All Available Vertical Features			UEPSR	UEPVF	2.26	0.00	0.00				11.90				
2-WIRE \	VOICE GRADE LINE PORT RATES (BUS)															
	Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus			UEPSB	UEPBL	1.40	3.74	3.63	1.88	1.80		11.90				
	Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484 ID - Bus.			UEPSB	UEPBC	1.40	3.74	3.63	1.88	1.80		11.90				
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.40	3.74	3.63	1.88	1.80		11.90				
	Exhange Ports - 2-Wire VG unbundled incoming only port with															
	Caller ID - Bus Subsequent Activity			UEPSB UEPSB	UEPB1	1.40 0.00	3.74 0.00	3.63 0.00	1.88	1.80		11.90				
FEATUR				UEPSB	USASC	0.00	0.00	0.00								
	All Available Vertical Features			UEPSB	UEPVF	2.26	0.00	0.00				11.90				
	GE PORT RATES (DID & PBX)			OLI OD	OLI VI	2.20	0.00	0.00				11.50				
	2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.40	39.06	18.18	12.35	0.7187		11.90				
_	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPSP UEPSP	UEPXD UEPXE	1.40	39.06 39.06	18.18 18.18	12.35 12.35	0.7187		11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPSP	UEPXL	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			ULFSF	OLFAL	1.40	39.00	10.10	12.33	0.7107		11.50				
	Room Calling Port			UEPSP	UEPXM	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPSP	UEPXO	1.40	39.06	18.18	12.35	0.7187		11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.40	39.06	18.18	12.35	0.7187		11.90				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00								
FEATUR				LIEDOD LIEDOE	LIEDVE	0.00	0.00	0.00				44.00				
	All Available Vertical Features GE PORT RATES (COIN)			UEPSP UEPSE	UEPVF	2.26	0.00	0.00				11.90				
EXCHAIN	Exchange Ports - Coin Port					1.40	3.74	3.63	1.88	1.80		11.90				
NOTE: T	ransmission/usage charges associated with POTS circuit swi	itched ι	usage v	will also apply to c	ircuit switched								orts.		I	1
	Access to B Channel or D Channel Packet capabilities will be													Request Prod	ess.	
BUNDLED LO	OCAL EXCHANGE SWITCHING(PORTS)		Ĺ												<u> </u>	
	GE PORT RATES (DID & PBX)															
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.73	78.41	15.82	41.94	4.26		11.90			1.83	
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability			UEPDD	UEPDD	54.95	151.11	77.75	48.81	3.10		11.90			1.83	
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)		i –	UEPTX UEPSX	U1PMA	8.83	46.83	50.68	27.64	11.93		11.90		İ	1.83	1
			•	UEPTX UEPSX	UEPVF	2.26						11.90			1.83	

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
													lu anamantal	In anom and al	la casa a satal	
													Incremental		Incremental	Incremental
													Charge -	Charge -	Charge -	Charge -
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)			Svc Order	Svc Order	Manual Svc	Manual Svc	Manual Svc	Manual Svc
	= ==== •	m						- (.,			Submitted	Submitted	Order vs.	Order vs.	Order vs.	Order vs.
											Elec	Manually	Electronic-	Electronic-	Electronic-	Electronic-
											per LSR	per LSR	1st	Add'l	Disc 1st	Disc Add'l
					-						per Lore	per Lore	100	Addi	D100 100	DISC Add I
						Rec	Nonrec	urring	Nonrecurrin	g Disconnect			OSS F	RATES (\$)		
			1			1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN		SOMAN	SOMAN
						l l	11100	Addi	11100	Addi	COME	COMPAR	COMPAR	COMPAN	OOMAN	COMPAR
NOTE:	Access to B Channel or D Channel Packet capabilities will be	availabl	o only	through BED/Now	Bueinose Doa	unet Process	Datas for the n	acket canabili	tion will be det	orminad via th	o Bona Eida	Doguest/N	ow Bueinoee I	Poguest Proc	000	
NOTE. 7	Exchange Ports - 2-Wire ISDN Port Channel Profiles	availabi	Unity	UEPTX UEPSX	U1UMA	0.00	0.00	0.00		l	e Bolla Flue	Requestri	ew Dusiliess	Request Floci	coo.	
	Exchange Ports - 4-Wire ISDN Port			UEPEX	UEPEX	82.74	174.61	95.17		18.23	1	11.90			1.83	
LINDLINDI ED I	OCAL SWITCHING, PORT USAGE			UEPEX	UEPEA	02.74	174.01	95.17	49.00	10.23		11.90			1.03	
					-											
End Offi	fice Switching (Port Usage)					0.0007000										
	End Office Switching Function, Per MOU					0.0007662					ļ					
<b>—</b>	End Office Trunk Port - Shared, Per MOU		<u> </u>		1	0.000164			1	1	1					
Tandem	n Switching (Port Usage) (Local or Access Tandem)		1		-	0.000101-					ļ					
$\vdash$	Tandem Switching Function Per MOU		1		-	0.0001319					ļ					
<u> </u>	Tandem Trunk Port - Shared, Per MOU		1		-	0.000235					ļ					
Commo	on Transport		ļ													
$\vdash$	Common Transport - Per Mile, Per MOU		<u> </u>		1	0.0000035										
	Common Transport - Facilities Termination Per MOU					0.0004372					1					
	ORT/LOOP COMBINATIONS - COST BASED RATES															
	ased Rates are applied where BellSouth is required by FCC and															
Features	s shall apply to the Unbundled Port/Loop Combination - Cost	Based	Rate se	ection in the same r	nanner as the	y are applied to	the Stand-Alo	ne Unbundled	Port section	of this Rate Ex	hibit.					
1 1	fice and Tandem Switching Usage and Common Transport Usa	ge rate	s in the	Port section of the	is rate exhibit	shall apply to a	all combination	ns of loop/por	t network elem	ents except fo	r UNE Coin	Port/Loop	Combinations	S.		
End Offi																
End Offi	Turio and Turio															
				P-4- I b-4												
For Geo	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Por															
For Geo	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a	re com	missior	n ordered cost base	ed rates and in											
For Geo for all st other sta	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Por tates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the	re com	missior	n ordered cost base	ed rates and in											
For Geo for all st other sta 2-WIRE	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Por tates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)	re com	missior	n ordered cost base	ed rates and in											
For Geo for all st other sta 2-WIRE	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a lates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates	re com	mission curring	n ordered cost base	ed rates and in	n AL, FL, NC an										
For Geo for all st other sta 2-WIRE	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or VIVIO Combination Rates  2-Wire VG Loop/Port Combo - Zone 1	re com	mission curring	n ordered cost base	ed rates and in	n AL, FL, NC an										
For Geo for all st other sta 2-WIRE	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  [2-Wire VG Loop/Port Combo - Zone 1  [2-Wire VG Loop/Port Combo - Zone 2	re com	mission curring	n ordered cost base	ed rates and in	14.11 18.23										
For Geo for all st other st: 2-WIRE UNE Poi	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3	re com	mission curring	n ordered cost base	ed rates and in	n AL, FL, NC an										
For Geo for all st other st: 2-WIRE UNE Poi	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a lates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates	re com	mission curring	n ordered cost base - Currently Combin	ed rates and in ned sections.	14.11 18.23 33.04										
For Geo for all st other st: 2-WIRE UNE Poi	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a lates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  opp Rates  2-Wire Voice Grade Loop (SL1) - Zone 1	re com	mission curring	ordered cost base - Currently Combin	ed rates and in med sections.	14.11 18.23 33.04										
For Geo for all st other st: 2-WIRE UNE Poi	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates    2-Wire VG Loop/Port Combo - Zone 1   2-Wire VG Loop/Port Combo - Zone 2   2-Wire VG Loop/Port Combo - Zone 3   2-Wire VG Loop/Port Combo - Zone 3   2-Wire Voice Grade Loop (SL1) - Zone 1   2-Wire Voice Grade Loop (SL1) - Zone 2	re com	mission curring	ordered cost base - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin	ded rates and in med sections.	14.11 18.23 33.04 12.94 17.06										
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a lates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  TOP Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2	re com	mission curring	ordered cost base - Currently Combin	ed rates and in med sections.	14.11 18.23 33.04										
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a lates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3	re com	mission curring	ordered cost base - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin	ued rates and in	14.11 18.23 33.04 12.94 17.06 31.87	d SC these no	nrecurring ch				he Market R				
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a lates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  TOP Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2	re com	mission curring	ordered cost base - Currently Combin - Currently Combin - Currently Combin - Currently Combin - Currently Combin	ded rates and in med sections.	14.11 18.23 33.04 12.94 17.06										
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res	re com	mission curring	UEPRX	ueplx	14.11 18.23 33.04 17.06 31.87 1.17	90.00 90.00	90.00 90.00				11.90 11.90				
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  100 Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence	re com	mission curring	ordered cost base - Currently Combin  UEPRX UEPRX UEPRX UEPRX UEPRX	ued rates and in the description of the description	14.11 18.23 33.04 12.94 17.06 31.87	d SC these no	nrecurring cha				he Market F				
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res	re com	mission curring	UEPRX	ueplx	14.11 18.23 33.04 17.06 31.87 1.17	90.00 90.00	90.00 90.00				11.90 11.90				
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res	re com	mission curring	UEPRX	ueplx	14.11 18.23 33.04 17.06 31.87 1.17	90.00 90.00	90.00 90.00				11.90 11.90				
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res	re com	mission curring	UEPRX	UEPLX	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17	90.00 90.00	90.00 90.00 90.00				11.90 11.90 11.90				
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or LOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res	re com	mission curring	UEPRX	UEPLX	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17	90.00 90.00	90.00 90.00 90.00				11.90 11.90 11.90				
For Geo for all st other st 2-WIRE UNE POI	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPAF	14.11 18.23 33.04 17.06 31.87 1.17 1.17	90.00 90.00	90.00 90.00 90.00				11.90 11.90				
For Geo for all st other st. 2-WIRE UNE Pol  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  RES	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00	arges are Mark			11.90 11.90 11.90				
For Geo for all st other st 2-WIRE UNE POI  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  RES  All Features Offered	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPAF	14.11 18.23 33.04 17.06 31.87 1.17 1.17	90.00 90.00	90.00 90.00 90.00	arges are Mark			11.90 11.90				
For Geo for all st other sts 2-WIRE UNE Poi  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  3-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  3-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00	arges are Mark			11.90 11.90 11.90				
For Geo for all st other st: 2-WIRE UNE POI  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges and test, the nonrecurring charges and test, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled res, low usage line port with Caller ID (LUM)  RES  All Features Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00	arges are Mark			11.90 11.90 11.90				
For Geo for all st other st: 2-WIRE UNE POI  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or I/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  RES  All Features Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)  CURRING CHARGES (NRCs) - CURRENTLY COMBINED	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00	arges are Mark			11.90 11.90 11.90				
For Geo for all st other st: 2-WIRE UNE POI  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID (LUM)  RES  [All Features Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)  CURRING CHARGES (NRCs) - CURRENTLY COMBINED	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAF UEPAP	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00	arges are Mark			11.90 11.90 11.90 11.90				
For Geo for all st other st: 2-WIRE UNE POI  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  10- Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port vith Caller ID - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  3-Wire voice unbundled Florida Area Calling with Caller ID - res  3-Wire voice unbundled Florida Area Calling With Caller ID - res  3-Wire voice Unbundled Florida Area Calling With Caller ID - res  3-Wire voice Unbundled Florida Area Calling With Caller ID - res  3-Wire voice Unbundled Florida Area Calling With Caller ID - res  3-Wire voice Unbundled Florida Area Calling With Caller ID - res  4-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00	arges are Mark			11.90 11.90 11.90				
For Geo for all st other st: 2-WIRE UNE POI  UNE Loc  2-Wire V	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges as tates, the nonrecurring charges as tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundles res, low usage line port with Caller ID  (LUM)  RES  All Features Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)  CURRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is	re com	mission curring	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00	arges are Mark			11.90 11.90 11.90 11.90				
For Geo for all st 2-WIRE UNE POI  UNE Loc  2-Wire V  FEATUR  LOCAL I  NONREC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID (LUM)  RES  All Features Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)  CURRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change	re com	mission curring	UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAF UEPAP	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00 90.00	arges are Mark			11.90 11.90 11.90 11.90				
For Geo for all st 2-WIRE UNE POI  UNE Loc  2-Wire V  FEATUR  LOCAL I  NONREC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Provice Calling with Caller ID - res  2-Wire voice unbundled Florida Provice Calling with Caller ID - res  2-Wire voice unbundled Florida Provice Calling with Caller ID - res  2-Wire voice Unbundled Florida Provice Calling With Caller ID - res  2-Wire voice Unbundled Florida Provice Calling With Caller ID - res  2-Wire voice Unbundled Florida Provice Calling With Caller ID - res  2-Wire voice Unbundled Florida Provice Calling With Caller ID - res  2-Wire voice Unbundled Florida Provice Calling With Caller ID - res  2-Wire voice Unbundled Florida Provice Calling With Caller ID - res  2-Wire voice Unbundled Florida Provice Calling With Caller ID - res  2-Wire voice Caller Voice Ca	re com	mission curring	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00	arges are Mark			11.90 11.90 11.90 11.90				
For Geo for all st 2-WIRE UNE POI  UNE Loc  2-Wire V  FEATUR  LOCAL I  NONREC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges as tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) ort/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID res  2-Wire voice unbundled Florida Area Calling with Caller ID res  2-Wire voice unbundled Florida Orea Calling with Caller ID res  3-Wire voice unbundled Florida Orea Calling with Caller ID res  2-Wire voice Unbundled Florida Orea Calling with Caller ID res  2-Wire voice Unbundled Florida Orea Calling over the Caller ID res  2-Wire voice Unbundled Florida Orea Calling over the Caller ID res  2-Wire voice Unbundled Florida Orea Calling over the Caller ID res  2-Wire voice Unbundled Florida Orea Calling over the Caller ID res  2-Wire voice Unbundled Florida Orea Calling over the Caller ID res  2-Wire voice Unbundled Florida Orea Calling over the Caller ID res  2-Wire voice Unbundled Florida Orea Calling over the Caller ID res  2-Wire voice Orea Caller ID res  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change  0-NAL NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent	re com	mission curring	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO UEPAF UEPAF UEPAP UEPAF UEPAP UEPAF UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17 2.26	90.00 90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00 0.102	arges are Mark			11.90 11.90 11.90 11.90 11.90				
For Geo for all st 2-WIRE UNE POI  UNE Lot  2-Wire V  LOCAL I  NONREC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges and the those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or I/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  Voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID res  2-Wire voice unbundled Florida Area Calling with Caller ID res  2-Wire voice unbundled Florida Area Calling with Caller ID res  2-Wire voice Unbundled Florida Area Calling with Caller ID res  2-Wire voice Unbundled Florida Area Calling with Caller ID res  2-Wire voice Unbundled Florida Area Calling with Caller ID res  2-Wire voice Unbundled Florida Area Calling with Caller ID res  2-Wire voice Unbundled Florida Area Calling with Caller ID res  2-Wire voice Unbundled Florida Area Calling with Caller ID res  2-Wire voice Unbundled Florida Area Calling with Caller ID res  2-Wire voice Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)  CURRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change  ONAL NRCs  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity	re com	mission curring	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17	90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00	arges are Mark			11.90 11.90 11.90 11.90				
For Geo for all st other st. 2-WIRE UNE POI  UNE Loc  2-Wire V  FEATUR  LOCAL I  NONREC  ADDITIC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) nt/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Proving Port with Caller ID - res  2-Wire voice unbundled Florida Proving Port Vith Caller ID - res  2-Wire voice Unbundled Florida Proving Port Vith Caller ID - res  2-Wire voice Unbundled Florida Proving Port Vith Caller ID - res  2-Wire voice Unbundled Florida Proving Port Vith Caller ID - res  2-Wire voice Unbundled Florida Proving Port Vith Caller ID - res  2-Wire voice Unbundled Florida Proving Port Combination - Conversion - Switch-as-is  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch With change  DNAL NRCS  2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	re com	mission curring	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO UEPAF UEPAF UEPAP UEPAF UEPAP UEPAF UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17 2.26	90.00 90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00 0.102	arges are Mark			11.90 11.90 11.90 11.90 11.90				
For Geo for all st other st. 2-WIRE UNE POI  UNE Loc  2-Wire V  FEATUR  LOCAL I  NONREC  ADDITIC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or VLoop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling over the Caller ID (LUM)  RES  [All Features Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)  CURRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change  ONAL NRCs  2-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	re com	mission purring	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO UEPAF UEPAF UEPAP UEPAF UEPAP UEPAF UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17 2.26	90.00 90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00 0.102	arges are Mark			11.90 11.90 11.90 11.90 11.90				
For Geo for all st other st. 2-WIRE UNE POI  UNE LOC  2-WIRE V  FEATUR  LOCAL  NONREC  ADDITIC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges and test, the nonrecurring charges and test, the nonrecurring charges and test. The portage of the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or VILoop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  1	re com	mission curring	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO UEPAF UEPAF UEPAP UEPAF UEPAP UEPAF UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC	14.11 18.23 33.04 17.06 31.87 1.17 1.17 1.17 2.26 0.35	90.00 90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00 0.102	arges are Mark			11.90 11.90 11.90 11.90 11.90				
For Geo for all st other st. 2-WIRE UNE POI  UNE Loc  2-Wire V  FEATUR  LOCAL I  NONREC  ADDITIC	orgia, Kentucky, Louisiana, Tennessee, the recurring UNE Portates. In GA, KY, LA, MS and TN these nonrecurring charges a tates, the nonrecurring charges shall be those identified in the VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) or VLoop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  op Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  voice Grade Line Port Rates (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling over the Caller ID (LUM)  RES  [All Features Offered  NUMBER PORTABILITY  Local Number Portability (1 per port)  CURRING CHARGES (NRCs) - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change  ONAL NRCs  2-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity  VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	re com	mission purring	UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO UEPAF UEPAF UEPAP UEPAF UEPAP UEPAF UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC	14.11 18.23 33.04 12.94 17.06 31.87 1.17 1.17 1.17 2.26	90.00 90.00 90.00 90.00 90.00 90.00 0.00	90.00 90.00 90.00 90.00 90.00 0.00 0.102	arges are Mark			11.90 11.90 11.90 11.90 11.90				

ATEGORY   RATE BLEMENTS   Interface   ACT   AC	Ex		Exhibit
UNEL Loop Rates	ntal Incremental Incre e - Charge - Ch Svc Manual Svc Man vs. Order vs. Order ic- Electronic- Elec	ncremental Inc Charge - C Ianual Svc Ma Order vs. C Electronic- El	Charge Manual S Order vi
UPER Loop Rates			
EVENT VIOLE Grade Loop (SL1) - Zone 2	N SOMAN SC	SOMAN	SOMAI
2-Wine Votes Grade Long (SL1) - Zone 2   2   UFPRX   UFPX   VFPX   37.87     2-Wine Votes Grade Long (SL1) - Zone 3   UFPX   UFPX   UFPX   37.87     2-Wine Votes Canded Long (SL1) - Zone 3   UFPX			
2-Wive Vote Grade Long (St.1) - Zanna 3			+
2-Wire Votes Grade Lue Port (Bus)		+	+
2-Wire votes unfunded port will caller 10 - bus   UEPBX   UEPBC   11.77   90.00   90.00   11.90   2-Wire votes unfunded port will caller 2-Bus   UEPBX   UEPBC   11.77   90.00   90.00   11.90   11.90   2-Wire votes unfunded port output of the port output of t			+
2-Vivir voice unbunded port with Caller 1 = 1641   D - Use			_
2.Wire votor unbrundled morning only port with Callor ID - Bus   UPER			
UCAL NUMBER PORTABILITY   UEPBX   UE			
Coad Number Portability (1 per port)			
FEATURES	$\bot$		
All Features Offered   UEPBX   UEPVF   2.26   0.00   0.00   11.90   11.90			
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED		<del></del>	+
2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is - S		<del></del>	+
2-Wire Voice Grade Loop (Line Port Combination - Conversion - Switch with change   UEPBX			+
Switch with change			+
2-Wire Voice Grade Loop/Line Port Combination - Subsequent   Activity   UEPBX   USAS2   0.00   0.00   11.90   11.90			
Activity   USAS2			
2-Wire Voice Grade LoOP WITH 2-Wire LINE PORT (RES - PBX)			
UNE Port/Loop Combination Rates			
2-Wire VG Loop/Port Combo - Zone 1			
2-Wire VG Loop/Port Combo - Zone 2   2   18.23   3.30			
2-Wire VG Logo/Port Combo - Zone 3   3   3   33.04			+
UNE Loop Rates		+	+
2-Wire Voice Grade Loop (St. 1) - Zone 1		<del></del>	+
2-Wire Voice Grade Loop (St. 1) - Zone 2			_
2-Wire Voice Grade Line Port Rates (RES - PBX)			
E-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			
Res			
Local Number Portability (1 per port)			
FEATURES	$\bot$		
All Features Offered			
NONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED		<del></del>	+
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is	<del>-   -   -</del>	+	+
2-Wire Voice Grade Loop/ Line Port Combination (PBX) -   UEPRG   USACC   8.45   1.91   11.90			+
ADDITIONAL NRCs     2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			+
2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity - Change/Rearrange Multiline Hunt Group   7.09   7.09   11.90   1	+ +		+
PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group   7.09   7.09   7.09   11.90			
UNE Port/Loop Combination Rates         1         14.11         1         14.11         1			
2-Wire VG Loop/Port Combo - Zone 1 1 14.11			
1 10 10 10 10 10 10 10 10 10 10 10 10 10			
2-Wire VG Loop/Port Combo - Zone 2 2 18.23			4
2-Wire VG Loop/Port Combo - Zone 3   3   33.04	-		+
UNE Loop Rates         2-Wire Voice Grade Loop (SL 1) - Zone 1         1         UEPPX         UEPLX         12.94         1         UEPX         U	<del></del>	<del></del>	+
Z-Write Votice Grade Loop (St. 1) - Zone 2   1   UEPPX   UEPLX   17.06	+	+	+
Z-Write Voice Grade Loop (St. 1) - Zone 3	+	+	+
2-Wire Voice Grade Line Port Rates (BUS - PBX)	<del>-   -   -   -   -   -   -   -   -   -  </del>		+

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge - Manual Svc Order vs.
						Rec	Nonrec	urring	Nonrecurring	g Disconnect			ossi	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX UEPPX	UEPPC UEPPO	1.17 1.17	90.00 90.00	90.00				11.90 11.90				
	Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPPO UEPP1	1.17	90.00	90.00				11.90				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.17	90.00	90.00				11.90				+
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.17	90.00	90.00				11.90				<b>†</b>
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.17	90.00	90.00				11.90				
2	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.17	90.00	90.00				11.90				1
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.17	90.00	90.00				11.90				
(	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPPX	UEPXE	1.17	90.00	90.00				11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX	UEPXL	1.17	90.00	90.00				11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPPX	UEPXM	1.17	90.00	90.00				11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX	UEPXO	1.17	90.00	90.00				11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UMBER PORTABILITY			UEPPX	UEPXS	1.17	90.00	90.00				11.90				
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								
FEATURE																
	All Features Offered			UEPPX	UEPVF	2.26	0.00	0.00				11.90				
	URRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-Is			UEPPX	USAC2		8.45	1.91				11.90				
(	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPPX	USACC		8.45	1.91				11.90				
	NAL NRCs															4
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			LIEDDY	110400	0.00	0.00	0.00				44.00				
	Subsequent Activity PBX Subsequent Activity - Change/Rearrange Multiline Hunt		<u> </u>	UEPPX	USAS2	0.00	0.00	0.00				11.90				+
	Group	<u> </u>					7.86	7.86				11.90				<u> </u>
	OICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT															<del> </del>
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			14.11										+
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			18.23										1
2	2-Wire VG Coin Port/Loop Combo – Zone 3		3			33.04				<u> </u>						
UNE Loop								· · · · · ·								
	2-Wire Voice Grade Loop (SL1) - Zone 1		_	UEPCO	UEPLX	12.94				ļ			ļ	ļ		
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	17.06				1	<u> </u>	1	<b> </b>	<del> </del>		<del>                                     </del>
	2-Wire Voice Grade Loop (SL1) - Zone 3 pice Grade Line Ports (COIN)	1	3	UEPCO	UEPLX	31.87					<del>                                     </del>	1	-			<del> </del>
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,				+					1	<del>                                     </del>	1		1		<del>                                     </del>
9	900/976, 1+DDD (FL)			UEPCO	UEP2F	1.17	90.00	90.00				11.90				<u> </u>
(	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (FL)			UEPCO	UEPFA	1.17	90.00	90.00				11.90				<u> </u>
9	2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL)			UEPCO	UEPCG	1.17	90.00	90.00				11.90				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (AL, FL)			UEPCO	UEPRK	1.17	90.00	90.00				11.90				
2	CAL: 1 L)  Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+ (FL)			UEPCO	UEPOF	1.17	90.00	90.00				11.90				
2	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	1.17	90.00	90.00				11.90				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)	1		UEPCO	UEPCK	1.17	90.00	90.00		1	1	11.90	1	1		†
1	2-Wire Coin Outward Smartline with 900/976 (all states except LA)			UEPCO	UEPCR	1.17	90.00	90.00				11.90				

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ADDITIO	DNAL UNE COIN PORT/LOOP (RC) UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	1.86	90.00	90.00								
LOCAL	NUMBER PORTABILITY			UEPCO	UKECU	1.00	90.00	90.00								
LOGAL	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
FEATUR	RES															
NONRE	CURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPCO	USAC2		0.102	0.102				11.90				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPCO	USACC		0.102	0.102				11.90				
ADDITIO	DNAL NRCs			OLFOO	USAUU	+	0.102	0.102	+		<del>                                     </del>	11.90				
ADDITIO	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0.00	0.00				11.90				
	ORT/LOOP COMBINATIONS - COST BASED RATES															
	VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK F	PORT														
UNE Po	rt/Loop Combination Rates															
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			23.21										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			28.28 46.53										
UNELO	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3  op Rates		3			46.53			-							
UNE LO	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	14.50						11.90			1.83	
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		2	UEPPX	UECD1	19.57						11.90			1.83	
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3			UEPPX	UECD1	37.82						11.90			1.83	
UNE Po																
	Exchange Ports - 2-Wire DID Port			UEPPX	UEPD1	8.71						11.90			1.83	
NONRE	CURRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-is			UEPPX	USAC1		7.85	1.87				11.90				
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes			UEPPX	USA1C		7.85	1.87				11.90				
ADDITIO	DNAL NRCs			UEPPA	USAIC		7.00	1.07				11.90				
ADDITIO	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX	USAS1		32.26	32.26				11.90				
Telepho	ne Number/Trunk Group Establisment Charges															
	DID Trunk Termination (One Per Port)			UEPPX	NDT	0.00	0.00	0.00				11.90			1.83	
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers			UEPPX	NDZ	0.00	0.00	0.00				11.90			1.83	
	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX	ND4	0.00	0.00	0.00				11.90			1.83	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX	ND5	0.00	0.00	0.00				11.90			1.83	
	Reserve Non-Consecutive DID numbers			UEPPX	ND6	0.00	0.00	0.00				11.90			1.83	
1 0041	Reserve DID Numbers NUMBER PORTABILITY	1		UEPPX	NDV	0.00	0.00	0.00	1			11.90	-		1.83	
LUCAL	Local Number Portability (1 per port)	1		UEPPX	LNPCP	3.15	0.00	0.00	+							
2-WIRF	ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINI	E SIDF	PORT	OLI I X	L141 OI	5.15	0.00	0.00	+							
	rt/Loop Combination Rates					1										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB UEPPR		32.09										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB UEPPR		38.15										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3		3	UEPPB UEPPR		59.94										
UNE Lo	op Rates												İ			
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB UEPPR	USL2X	24.71						11.90			1.83	
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB UEPPR	USL2X	30.77						11.90			1.83	
<del></del>	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB UEPPR	USL2X	52.56	, and the second					11.90			1.83	
UNE Po		ļ		HEDDD HEDDS	HEDDS	7.00			1			44.00			1.00	
<u> </u>	Exchange Port - 2-Wire ISDN Line Side Port	<u> </u>	l l	UEPPB UEPPR	UEPPB	7.38						11.09	l .	l	1.83	

UNBUNDLED	NETWORK ELEMENTS - Florida													Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	В	cs	USOC			RATES(\$)				Submitted Manually		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
							Rec	Nonrec		Nonrecurring					RATES (\$)		
NONDE	NURRING GUARGES OURRENTLY COMPINED							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NONREC	CURRING CHARGES - CURRENTLY COMBINED  2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port																<b>├</b>
	Combination - Conversion			UEPPB	UEPPR	USACB	0.00	25.22	17.00				11.90			1.83	ĺ
	NAL NRCs							-									
	NUMBER PORTABILITY																
	Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
	NEL USER PROFILE ACCESS:  CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00				-				<del> </del>
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00				1				<del>                                     </del>
	CSD		<b>1</b>	UEPPB	UEPPR	U1UCC	0.00	0.00	0.00				1				
	NEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC	,MS, &	TN)														
USER TI	ERMINAL PROFILE																
VEDTIC	User Terminal Profile (EWSD only) AL FEATURES			UEPPB	UEPPR	U1UMA	0.00	0.00	0.00				<u> </u>				<del></del>
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR	LIED\/E	2.26	0.00	0.00				11.90				<del></del>
	FFICE CHANNEL MILEAGE			OLFFB	ULFFR	OLFVI	2.20	0.00	0.00				11.50				<b>—</b>
	Interoffice Channel mileage each, including first mile and																
	facilities termination			UEPPB		M1GNC	18.4491	47.35	31.78	18.31	7.03		11.90			1.83	<u> </u>
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.0091	0.00	0.00				11.90			1.83	
	DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK	PORT															<b></b>
UNE Poi	t/Loop Combination Rates  4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE																<del></del>
	Zone 1   4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		1	UEPPP			156.18										<u> </u>
	Zone 2		2	UEPPP			181.87										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		3	UEPPP			274.25										
UNE Loc	pp Rates																
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP		USL4P	73.44						11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 2 4-Wire DS1 Digital Loop - UNE Zone 3		2	UEPPP UEPPP		USL4P USL4P	99.13 191.51						11.90 11.90			1.83 1.83	<del>                                     </del>
UNE Poi			3	UEPPP		USL4P	191.51						11.90			1.83	<del>                                     </del>
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP		UEPPP	82.74						11.90			1.83	
	CURRING CHARGES - CURRENTLY COMBINED			02		02	02.7 .						11.00			1.00	
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination - Conversion -Switch-as-is			UEPPP		USACP	0.00	84.17	61.38				11.90			1.83	
ADDITIO	NAL NRCs						0.00										
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within Std Allowance			UEPPP		PR7TF		0.5412					11.90			1.83	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All States except NC)			UEPPP		PR7TO		12.71	12.71				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos Above Std Allowance			UEPPP		PR7ZT		25.42	25.42				11.90			1.83	
LOCAL	NUMBER PORTABILITY			32111				20.42	20.42				11.30			1.00	$\vdash$
	Local Number Portability (1 per port)			UEPPP		LNPCN	1.75										
	ACE (Provsioning Only)																
	Voice/Data			UEPPP		PR71V	0.00	0.00	0.00				1				
ļ <u> </u>	Digital Data			UEPPP		PR71D	0.00	0.00	0.00								<del></del>
Nower	Inward Data Additional "B" Channel			UEPPP		PR71E	0.00	0.00	0.00				<del>                                     </del>				<del></del>
New or A	New or Additional - Voice/Data B Channel			UEPPP		PR7BV	0.00	15.48					11.90			1.83	<del></del>
	New or Additional - Digital Data B Channel		<b>1</b>	UEPPP		PR7BF	0.00	15.48					11.90			1.83	
	New or Additional Inward Data B Channel			UEPPP		PR7BD	0.00	15.48					11.90			1.83	
	New or Additional Useage Sensitive Voice Data B Channel			UEPPP		PR7BS	0.00	15.48					11.90			1.83	
	New or Additional Useage Sensitive Digital Data B Channel			UEPPP		PR7BU	0.00	15.48					11.90			1.83	
CALL T			ļ	LIEBSS		DD70:											
	Inward	1	Ļ	UEPPP		PR7C1	0.00	0.00	0.00			1	1	l			

IDOIIDE	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit:
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add
						Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS F	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Interoffic	ce Channel Mileage			UEPPP	1LN1A	00.0050	405.54	00.47	04.47	10.05		44.00			1.93	
-+-	Fixed Each Including First Mile  Each Airline-Fractional Additional Mile			UEPPP	1LN1B	88.6256 0.1856	105.54	98.47	21.47	19.05		11.90			1.93	
4-WIRE	DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT			UEPPP	ILINID	0.1656										
	rt/Loop Combination Rates															
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		128.39						11.90			1.83	
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2		2	UEPDC		154.08						11.90			1.83	
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		246.46				•		11.90			1.83	
UNE Lor	op Rates															
	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	73.44						11.90			1.83	
	4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	99.13						11.90			1.83	
UNE Por	4-Wire DS1 Digital Loop - UNE Zone 3	<del>                                     </del>	3	UEPDC	USLDC	191.51					-	11.90			1.83	
UNE POI	4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	54.95						11.90			1.83	
NONRE(	CURRING CHARGES - CURRENTLY COMBINED			OLI DO	ODDII	34.33						11.30			1.00	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Switch-as-is			UEPDC	USAC4		95.31	46.71				11.90			1.83	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
	- Conversion with DS1 Changes			UEPDC	USAWA		95.31	46.71				11.90			1.83	
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination															
ADDITIO	- Conversion with Change - Trunk			UEPDC	USAWB		95.31	46.71				11.90			1.83	
ADDITIO	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC -															
	Subsequent Channel Activation/Chan - 2-Way Trunk			UEPDC	UDTTA		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent			OLI DO	OBTIN		10.00	10.00				11.50			1.00	
	Channel Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel															
	Activation/Chan Inward Trunk w/out DID			UEPDC	UDTTC		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan															
	Activation Per Chan - Inward Trunk with DID			UEPDC	UDTTD		15.69	15.69				11.90			1.83	
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan			LIEDDO	LIDTTE		45.00	45.00				44.00			4.00	
BIDOL A	Activation / Chan - 2-Way DID w User Trans  R 8 ZERO SUBSTITUTION	-		UEPDC	UDTTE		15.69	15.69			-	11.90			1.83	-
DIFULA	B8ZS -Superframe Format	<del>                                     </del>		UEPDC	CCOSF		0.00	655.00				11.90			1.83	
_	B8ZS - Extended Superframe Format	1		UEPDC	CCOEF		0.00	655.00				11.90			1.83	
Alternat	e Mark Inversion			-	1			,,,,,,,								
	AMI -Superframe Format			UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format			UEPDC	MCOPO		0.00	0.00								
Telepho	ne Number/Trunk Group Establisment Charges															
$-\!\!\!\!+\!\!\!\!-\!\!\!\!\!-$	Telephone Number for 2-Way Trunk Group	ļ		UEPDC	UDTGX	0.00						11.90			1.83	
	Telephone Number for 1-Way Outward Trunk Group			UEPDC UEPDC	UDTGY	0.00						11.90 11.90			1.83 1.83	
-	Telephone Number for 1-Way Inward Trunk Group Without DID DID Numbers, Establish Trunk Group and Provide First Group	<del>                                     </del>		UEPUC	UDIGZ	0.00					-	11.90			1.83	
	of 20 DID Numbers	l		UEPDC	NDZ	0.00	0.00	0.00				11.90			1.83	
	DID Numbers for each Group of 20 DID Numbers	1		UEPDC	ND4	0.00	0.00	0.00				11.90			1.83	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00						11.90			1.83	
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				11.90			1.83	
	Reserve DID Numbers			UEPDC	NDV	0.00	0.00	0.00				11.90			1.83	
Dedicate	ed DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1	Digital I	oop w	ith 4-Wire DDITS T	runk Port											
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities	l		LIEDDO	41.110.1		,		a							
$-\!$	Termination)	<b> </b>		UEPDC	1LNO1	88.44	105.54	98.47	21.47	19.05		11.90			1.83	<b> </b>
		l	1	UEPDC	1LNOA	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			ULFDC												

UNBUNDLEI	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: E
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec		Nonrecurring	Disconnect				RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.1856	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities															
	Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.1856	0.00	0.00								İ
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00							<b>—</b>
	Central Office Termininating Point			UEPDC	CTG	0.00	0.00	0.00	0.00							-
4-WIRE	DS1 LOOP WITH CHANNELIZATION WITH PORT			02. 50	0.0	0.00										
	is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activ	ations			1						1			İ		
	ystem can have up to 24 combinations of rates depending on t		numb	er of ports used									<u> </u>	<u> </u>		
	S1 Loop															
	4-Wire DS1 Loop - UNE Zone 1			UEPMG	USLDC	73.44	0.00	0.00		· · · · · · · · · · · · · · · · · · ·						
	4-Wire DS1 Loop - UNE Zone 2			UEPMG	USLDC	99.13	0.00	0.00								
<u> </u>	4-Wire DS1 Loop - UNE Zone 3	l	3	UEPMG	USLDC	191.51	0.00	0.00								
UNE DS	60 Channelization Capacities (D4 Channel Bank Configuration	s)														
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	118.06	0.00	0.00				11.90			1.83	
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	236.12	0.00	0.00				11.90			1.83	
-	96 DSO Channel Capacity -1per 4 DS1s			UEPMG UEPMG	VUM96	472.24	0.00	0.00				11.90			1.83	
	144 DS0 Channel Capacity - 1 per 6 DS1s 192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM14 VUM19	708.36 944.48	0.00	0.00				11.90 11.90			1.83 1.83	
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1,180.60	0.00	0.00			1	11.90			1.83	
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG	VUM28	1,416.72	0.00	0.00				11.90			1.83	
	384 DS0 Channel Capacity - 1 per 16 DS1s			UEPMG	VUM38	1,888.96	0.00	0.00				11.90			1.83	
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2,361.20	0.00	0.00			1	11.90			1.83	
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2.833.44	0.00	0.00				11.90			1.83	
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	3,305.68	0.00	0.00				11.90			1.83	
Non-Re	curring Charges (NRC) Associated with 4-Wire DS1 Loop with	Channe	liztion				tem									
	num System configuration is One (1) DS1, One (1) D4 Channel															
Multiple	es of this configuration functioning as one are considered Add	d'I after	the mir	nimum system confi	guration is c	ounted.										
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes			UEPMG	USAC4	0.00	96.77	4.24				11.90				
System	Additions at End User Locations Where 4-Wire DS1 Loop with	Chann	elizatio	on with Port Combin	ation Currer	tly Exists and										
	ot Currently Combined) In GA, KY, LA, & TN Only															
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc															
	Fea Activation - New GA, LA, KY, MS, &TN Only			UEPMG	VUMD4	0.00	726.11	468.21	145.32	17.24	ļ	11.90	ļ	ļ		
Bipolar	8 Zero Substitution				<u> </u>						<u> </u>	11.90	ļ	ļ		
	Clear Channel Capability Format, superframe - Subsequent Activity Only			UEPMG	CCOSF	0.00	0.00	655.00				11.90				
	Clear Channel Capability Format - Extended Superframe -															
<b>—</b>	Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	655.00				11.90				
Alterna	te Mark Inversion (AMI)			LIEDMO	MCOCE	0.00	0.00	0.00			ļ		<del> </del>	<del> </del>		
<del>                                     </del>	Superframe Format  Extended Superframe Format			UEPMG UEPMG	MCOSF MCOPO	0.00	0.00	0.00			<b> </b>		<b> </b>	-		
Evokon	ge Ports Associated with 4-Wire DS1 Loop with Channelization	n with F	ort	ULFIVIG	IVICUPU	0.00	0.00	0.00			<b> </b>					<del>                                     </del>
	ge Ports	WILII F	UI L													
	Ĭ													İ		
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.38	0.00	0.00	0.00	0.00		11.90	1	]	1.83	1
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.38	0.00	0.00	0.00	0.00		11.90			1.83	
	Line Cide Inward Only Channel and DDV Tarrel, Dark William DD			UEPPX	UEP1X	1.38	0.00	0.00	0.00	0.00		44.00			4.00	
<del>                                     </del>	Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEP1X UEPDM	1.38 8.71	0.00	0.00	0.00	0.00	-	11.90 11.90	-		1.83 1.83	<del></del>
Foature	Activations - Unbundled Loop Concentration			ULFFA	OEPDIVI	0.71	0.00	0.00	0.00	0.00	<u> </u>	11.90			1.83	
reature	Feature (Service) Activation for each Line Side Port Terminated	1			<del>                                     </del>						<del>                                     </del>	1	<del> </del>	<del> </del>		<del></del>
	in D4 Bank			UEPPX	1PQWM	0.66	25.40	13.41	3.96	3.93		11.90			1.83	<b></b>
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			UEPPX	1PQWU	0.66	78.16	18.42	56.03	10.95		11.90			1.83	1

UNBUNDLED NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
										per Loix	per Lorc			Diac 1at	DISC Add I
					Rec	Nonred First	curring Add'l	Nonrecurring First	g Disconnect Add'l	SOMEC	SOMAN		RATES (\$) SOMAN	SOMAN	SOMAN
Telephone Number/ Group Establishment Charges for DID Service						FIRST	Add I	FIRST	Addi	SOMEC	SUMAN	SOWAN	SOWAN	SOMAN	SUMAN
DID Trunk Termination (1 per Port)		1	UEPPX	NDT	0.00	0.00	0.00				11.90				
Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00				11.90				
DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				11.90				
Non-Consecutive DID Numbers - per number  Reserve Non-Consecutive DID Numbers	ļ	-	UEPPX UEPPX	ND5 ND6	0.00	0.00	0.00				11.90 11.90				
Reserve Non-Consecutive DID Numbers  Reserve DID Numbers		1	UEPPX	NDV	0.00	0.00	0.00				11.90				
Local Number Portability		1	OLITA	1101	0.00	0.00	0.00				11.00				
Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
FEATURES - Vertical and Optional					ļ										
Local Switching Features Offered with Line Side Ports Only	<del>                                     </del>		LIEDDY	HED) (E	0.00	0.00	0.00				44.00			4.00	
All Features Available UNBUNDLED PORT LOOP COMBINATIONS - MARKET RATES	1	-	UEPPX	UEPVF	2.26	0.00	0.00			<del>                                     </del>	11.90			1.83	
Market Rates shall apply where BellSouth is not required to provide	unbundi	led loca	l al switching or switc	h ports per	FCC and/or Sta	te Commission	n rules.				<del>                                     </del>				
These scenarios include:		1		ponto por		30									
1. Unbundled port/loop combinations that are Not Currently Combin	ed in Al	abama,	Florida, North Caro	lina and Sou	th Carolina.										
2. Unbundled port/loop combinations that are Currently Combined of	r Not Cu	urrently	Combined in Zone	1 of the Top	8 MSAS in Bel	ISouth's regio	n for end users	with 4 or more	e DS0 equivale	ent lines.					
The Top 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderda															
BellSouth currently is developing the billing capability to mechanica	llv bill th	he recu	rring and non-recurr	ing Market I	Rates in this se	ction except fo	or nonrecurring	charges for n	ot currently co	mbined in	AL. FL. NC a	and SC. In th	e interim whe	re BellSouth	cannot bill
Market Rates, BellSouth shall bill the rates in the Cost-Based section											,,,o .		·		
The Market Rate for unbundled ports includes all available features in	n all stat	tes.													
End Office and Tandem Switching Usage and Common Transport Us				rate exhibit	shall apply to	•			ents except fo	or UNE Coin	Port/Loop	Combinations	s which have a	a flat rate usa	ge charge
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).	age rate	s in the	Port section of this			all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the	age rate Nonrec	s in the	Port section of this			all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).	age rate Nonrec	s in the	Port section of this			all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates	age rate Nonrec	s in the	Port section of this		d Additional N	all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1	age rate Nonrec	curring cording	Port section of this		d Additional N	all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2	age rate Nonrec	curring cording	Port section of this		26.79 31.27	all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3	age rate Nonrec	curring cording	Port section of this		d Additional N	all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  [2-Wire VG Loop/Port Combo - Zone 1	age rate Nonrec	es in the curring cording	Port section of this charges are listed in ly.	the First ar	26.79 31.27 47.36	all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3	age rate Nonrec	courring cordings 1 1 2 3 1 1	Port section of this		26.79 31.27	all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates    2-Wire VG Loop/Port Combo - Zone 1   2-Wire VG Loop/Port Combo - Zone 2   2-Wire VG Loop/Port Combo - Zone 3    UNE Loop Rates    2-Wire Voice Grade Loop (SL1) - Zone 1	age rate Nonrec	cordings 1 2 3 1 1 2	Port section of this charges are listed in by.	the First ar	26.79 31.27 47.36	all combinatio	ns of loop/por	t network elem							
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)	age rate Nonrec	cordings 1 2 3 1 1 2	P Port section of this charges are listed in ly.  UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX	26.79 31.27 47.36 12.79 17.27 33.36	all combinatio	or each Port US	t network elem			, the Nonrec				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residence	age rate Nonrec	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPLX	26.79 31.27 47.36 12.79 17.27 33.36	all combinatio	or each Port US	t network elem			the Nonrec				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residence  2-Wire voice unbundled port with Caller ID - res	age rate Nonrec	cordings 1 2 3 1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL	26.79 31.27 47.36 12.79 17.27 33.36	all combinatio	preach Port US	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residence	age rate Nonrec	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPLX	26.79 31.27 47.36 12.79 17.27 33.36	all combinatio	or each Port US	t network eleme			the Nonrec				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res	age rate Nonrec	cordings 1 2 3 1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRL	26.79 31.27 47.36 12.79 17.27 33.36	all combinatio	preach Port US	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE POYLOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundled Florida Area Calling with Caller ID	age rate Nonrec	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00	90.00 90.00	90.00 90.00	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundled sers, low usage line port with Caller ID (LUM)	age rate Nonrec	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00	90.00 90.00	90.00 90.00	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  LOCAL NUMBER PORTABILITY	age rate Nonrec	cordings 1 2 3 1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00	90.00 90.00 90.00	90.00 90.00	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE POYLOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)	age rate Nonrec	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00	90.00 90.00 90.00	90.00 90.00	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)	age rate Nonrec	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAP	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE POYLOOP Combination Rates  2-Wire VG Loop/Port Combo - Zone 1  2-Wire VG Loop/Port Combo - Zone 2  2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1  2-Wire Voice Grade Loop (SL1) - Zone 2  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res  2-Wire voice unbundled Florida Area Calling with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)	age rate Nonrec	cordings 1 2 3 1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00	90.00 90.00 90.00	90.00 90.00	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Line Port (Res)  2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)  FEATURES  All Features Offered  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is	age rate Nonrec Zed acc	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAP	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00	90.00 90.00 90.00 90.00	t network eleme			11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port (Res)  2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)  FEATURES  All Features Offered  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is	age rate Nonrec Zed acc	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAP LNPCX UEPVF	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00 90.00 41.50	90.00 90.00 90.00 90.00 41.50	t network eleme			11.90 11.90 11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  LOCAL NUMBER PORTABILITY Local Number Portability (1 per port)  FEATURES  All Features Offered  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is	age rate Nonrec Zed acc	cordings 1 2 3 1 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAP LNPCX UEPVF	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00	90.00 90.00 90.00	t network eleme			11.90 11.90 11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundled Florida Area Calling with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)  FEATURES  All Features Offered  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Switch with change  ADDITIONAL NRCS  NRC - 2-Wire Voice Grade Loop/Line Port Combination -	age rate Nonrec Zed acc	cordings 1 2 3 1 1 2	Department of this charges are listed in the charges are listed in the charges are listed in the charges are listed in the charges are listed in the charges are listed in the charge of	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAF UEPAF UEPAP LNPCX UEPVF USAC2 USACC	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00 90.00 41.50	90.00 90.00 90.00 90.00 90.00 41.50	t network eleme			11.90 11.90 11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundled Florida Area Calling with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)  FEATURES  All Features Offered  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Switch with change  ADDITIONAL NRCS  NRC - 2-Wire Voice Grade Loop/Line Port Combination - Switch with Subsequent	age rate Nonrec Zed acc	cordings 1 2 3 1 1 2	Department of this charges are listed in the	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAF UEPAP LNPCX UEPVF	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00 90.00 41.50	90.00 90.00 90.00 90.00 41.50	t network eleme			11.90 11.90 11.90 11.90				
End Office and Tandem Switching Usage and Common Transport Us (USOC: URECU).  For Not Currently Combined scenarios where Market Rates apply, the Combined section. Additional NRCs may apply also and are categor 2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)  UNE Port/Loop Combination Rates  2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3  UNE Loop Rates  2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire Voice Grade Loop (SL1) - Zone 3  2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res  2-Wire voice unbundled Florida Area Calling with Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM)  LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)  FEATURES  All Features Offered  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Switch with change  ADDITIONAL NRCS  NRC - 2-Wire Voice Grade Loop/Line Port Combination -	age rate Nonrec Zed acc	cordings 1 2 3 1 1 2	Department of this charges are listed in the charges are listed in the charges are listed in the charges are listed in the charges are listed in the charges are listed in the charge of	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAF UEPAF UEPAP LNPCX UEPVF USAC2 USACC	26.79 31.27 47.36 12.79 17.27 33.36 14.00 14.00 14.00 14.00	90.00 90.00 90.00 90.00 90.00 41.50	90.00 90.00 90.00 90.00 90.00 41.50	t network eleme			11.90 11.90 11.90 11.90				

CATEGORY RATE ELEMENTS  RATE ELEMENTS  Description of the content	ncremental Charge - Charge - Manual Svc Order vs. Electronic-Add'l Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
2-Wire VG Loop/Port Combo - Zone 1		SOMAN
2-Wire VG Loop/Port Combo - Zone 1	SOMAN SOMAN	SOMAN
2-Wire VG Loop/Port Combo - Zone 2   2   31.27		
2-Wire VG Loop/Port Combo - Zone 3   3   47.36		
UNE Loop Rates		
2-Wire Voice Grade Loop (SL1) - Zone 1		
2-Wire Voice Grade Loop (SL1) - Zone 2   2 UEPBX UEPLX   17.27		
2-Wire Voice Grade Loop (SL1) - Zone 3   3 UEPBX UEPLX   33.36		
2-Wire Voice Grade Line Port (Bus)		
2-Wire voice unbundled port with Caller + E484 ID - bus   UEPBX   UEPBC   14.00   90.00   90.00   90.00   11.90		
2-Wire voice unbundled port outgoing only - bus   UEPBX   UEPBO   14.00   90.00   90.00   90.00   11.90		
LOCAL NUMBER PORTABILITY  Local Number Portability (1 per port)  FEATURES  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is LOCAL NUMBER PORTABILITY  UEPBX USAC2  41.50  41.50  41.50  ADDITIONAL NRCS		
Local Number Portability (1 per port)		
FEATURES  NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is		
NONRECURRING CHARGES - CURRENTLY COMBINED  2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is  2-Wire Voice Grade Loop / Line Port Combination - Switch with change  UEPBX  USAC2  41.50  41.50  11.90  ADDITIONAL NRCS		
2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is UEPBX USAC2 41.50 41.50 11.90 2-Wire Voice Grade Loop / Line Port Combination - Switch with change UEPBX USACC 41.50 41.50 41.50		
2-Wire Voice Grade Loop / Line Port Combination - Switch with change UEPBX USACC 41.50 41.50 41.50 41.50		
2-Wire Voice Grade Loop / Line Port Combination - Switch with change UEPBX USACC 41.50 41.50 41.50 41.50	1	
ADDITIONAL NRCs		
NRC - 2-Wire Voice Grade Loop/Line Port Combination -		
Subsequent   UEPBX USAS2   0.00   0.00   11.90		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)		
UNE Port/Loop Combination Rates		
2-Wire VG Loop/Port Combo - Zone 1 1 26.79		
2-Wire VG Loop/Port Combo - Zone 2 2 31.27		
2-Wire VG Loop/Port Combo - Zone 3   3   47.36		
UNE Loop Rates		
2-Wire Voice Grade Loop (SL1) - Zone 1		
Z-Write Voice Grade Loop (St.1) - Zone 3   Z   UEPRG   UEPLX   17.27	<del></del>	
2-Wire Voice Grade Line Port Rates (RES - PBX)	<del></del>	
2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res UEPRG UEPRD 14.00 90.00 90.00 11.90		
LOCAL NUMBER PORTABILITY		
Local Number Portability (1 per port)  UEPRG LNPCP 3.15	<del>-                                     </del>	
FEATURES SET TO		
NONRECURRING CHARGES - CURRENTLY COMBINED		
2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is UEPRG USAC2 41.50 41.50 11.90		
2-Wire Voice Grade Loop/ Line Port Combination - Switch with		
Change		
ADDITIONAL NRCS  2 Wire Loop/Line Side Port Combination - Non feature -	$\longrightarrow$	
2 Wile Lough Line Side Port Combination - Non readure - Subsequent Activity - Nonrecurring 0.00 0.00		
PBX Subsequent Activity - Change/Rearrange Multiline Hunt		
Group 7.09 7.09 11.90		
2-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)		
UNE Port/Loop Combination Rates		
2-Wire VG Loop/Port Combo - Zone 1 1 26.79	$\longrightarrow$	
2-Wire VG Loop/Port Combo - Zone 2     2     31.27        2-Wire VG Loop/Port Combo - Zone 3     3     47.36		
UNE Loop Rates	<del>-   -  </del>	
ONE LOUP Rates	<del>-   -  </del>	
2-Wrie Voice Grade Loop (St.1) - Zone 2   1 UEPPX   UEPLX   17.27   17	<del></del>	
2-Wire Voice Grade Loop (St.1) - Zone 3		
2-Wire Voice Grade Line Port Rates (BUS - PBX)		

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonred	urring	Nonrecurring	Disconnect			ossı	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14.00	90.00	90.00				11.90				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14.00	90.00	90.00				11.90				<del> </del>
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX UEPPX	UEPXC UEPXD	14.00 14.00	90.00 90.00	90.00				11.90 11.90				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPPA	UEPAD	14.00	90.00	90.00				11.90				
	Capable Port			UEPPX	UEPXE	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPPA	UEFAL	14.00	90.00	90.00				11.90				
	Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX	UEPXO	14.00	90.00	90.00				11.90				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00				11.90				
	IUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15										
FEATURI																
NONREC	URRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPPX	USAC2		41.50	41.50				11.90				
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with			02.17.	00/102		11.00	11.00				11100				
	Change			UEPPX	USACC		41.50	41.50								
ADDITIO	NAL NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2		0.00	0.00				11.90				
	2 Wire Loop/Line Side Port Combination - Non feature -			02.17.	00/102		0.00	0.00				11100				
	Subsequent Activity- Nonrecurring						0.00	0.00								
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt						7.00	7.00				44.00				
	Group /OICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT	-					7.09	7.09				11.90				-
	t/Loop Combination Rates		<b>†</b>		+	<del>                                     </del>					<del>                                     </del>	<b> </b>				<del>                                     </del>
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			26.79										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			31.27										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			47.36										
UNE Loo			<u> </u>	LIEBOO	HEDIX	10.77					<u> </u>	<u> </u>				
	2-Wire Voice Grade Loop (SL1) - Zone 1		2	UEPCO	UEPLX	12.79 17.27					<del>                                     </del>	<del>                                     </del>				
	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO UEPCO	UEPLX	33.36					<del>                                     </del>	<del>                                     </del>		1		
	pice Grade Line Port Rates (Coin)	1		52, 55	OLI LA	33.30					1	1	1	1		
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011,															
	900/976, 1+DDD (FL) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking		1	UEPCO	UEP2F	14.00	90.00	90.00			1	11.90				1
	(FL)			UEPCO	UEPFA	14.00	90.00	90.00				11.90				
	2-Wire Coin 2-Way with Operator Screening and Blocking:															
	900/976, 1+DDD, 011+, and Local (FL) 2-Wire Coin Outward with Operator Screening and 011 Blocking		1	UEPCO	UEPCG	14.00	90.00	90.00			-	11.90				
	(AL, FL)			UEPCO	UEPRK	14.00	90.00	90.00				11.90				
	2-Wire Coin Outward with Operator Screening and Blocking:			LIEDOO	UEDCE											
	900/976, 1+DDD, 011+ (FL) 2-Wire Coin Outward with Operator Screening and Blocking:		<del>                                     </del>	UEPCO	UEPOF	14.00	90.00	90.00				11.90				
	900/976, 1+DDD, 011+, and Local (FL, GA)			UEPCO	UEPCQ	14.00	90.00	90.00				11.90				
LOCAL N	IUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										

CATEGORY   RATE ELEMENTS   In the case   Section   Sec	UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
NONTICUTERING CHARGES - CURRENTLY COMBINED   NONTICUTERING CHARGES - CURRENTLY COMBINED   NONTICUTERING CHARGES - CURRENTLY COMBINED   NONTICUTERING CHARGES - CURRENTLY COMBINED   NONTICUTERING CHARGES - CURRENTLY COMBINED   NONTICUTERING CHARGES - CURRENTLY COMBINED   NONTICUTERING CHARGES - CURRENTLY COMBINED   NONTICUTERING CHARGES - CURRENTLY COMBINED CHAR				Zone	BCS	USOC			RATES(\$)			Submitted Elec	Svc Order Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
NOMERCHARD CHARGES - CURRENTY COMMENTS							Rec										
Description								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
SAMP   VICE of Gride Long / Line Print Combination - Switch with   LEPCO   USASCC   4150   4150	NONREC	CURRING CHARGES - CURRENTLY COMBINED															
SAMP   VICE of Gride Long / Line Print Combination - Switch with   LEPCO   USASCC   4150   4150		2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			LIEDCO	LISAC2		41 50	41 50				11 00				
Change		2-Wire Voice Grade Loop/ Line Port Combination - Switch with			OLI CO	OOAOZ		41.50	41.50				11.30				
Description   Description					UEPCO	USACC		41.50	41.50								
UNNUMED CENTREX PORTALOP COMBINATIONS																	
UNNUMED CENTREX PORTALOP COMBINATIONS																	
UNBUNDLE PORTLOP COMBINATIONS - COST BASED BATES	LINIDI NIC: ED C				UEPCO	USAS2		0.00	0.00				11.90				
UNE PORTREX - 1 ARSS - (Valid in ALF, EAKT, CLANK XTA nohy)   2-Wire Vol Loop/XWire Vole Carde Fort (Centracy Fort Combo				1		+					-	1					
2-Wire Vot Loop2-Wire Votes Grade Port (Centres) Port Combo   1				<u> </u>		+					-	1		1			
Non-Design   Service   S						+											
2.Wire Vot Conf2x Pert Votes Grade Port (Centrex) Port Combo   1 UEP91   14,111			1			1					1						
Non-Design   1 UEP91   14.11   1.11		2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				†					İ						
Non-Design		Non-Design		1	UEP91		14.11										
2-Wire Vol Loop/2-Wire Volos Grade Port (Centrex)Port Combo   Non-Design   Non-De				2	LIED01		10.22										
Non-Design   3 UEP91   33.04					ULF91		10.23										
UNE Port/Loop Combination Rates (Design)				3	UEP91		33.04										
2-Wire Vox Loop/2-Wire Voxee Grade Port (Centrex)Port Combo   1 UEP91   16.53   1																	
Design																	
Design   2 UPE91   21.60				1	UEP91		16.53										
Design   State   Comparation   State   Comparation   State																	
Design   3   UPP1   37.85				2	UEP91		21.60										
UNE Loop Rate																	
2-Wire Voice Grade Loop (St. 1) - Zone 1				3	UEP91		37.85										
2-Wire Voice Grade Loop (St. 1) - Zone 2				1	LIED04	LIECC1	12.04										
2-Wire Voice Grade Loop (St. 1) - Zone 3   3 UEP91   UECS1   31.87													-				
2-Wire Voice Grade Loop (SL 2) - Zone 1																	
2-Wire Voice Grade Loop (SL 2) - Zone 2   2   UEP91   UECS2   20.43																	
2-Wire Voice Grade Dopt (St. 2) - Zone 3   3   UEP91   UECS2   36.68																	
All States (Except North Carolina and Sout Carolina)  2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area  2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area  2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area  2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area  2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area  2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area  11.90  11.83  2-Wire Voice Grade Port (Serving Wire Center - 800 Service Term - Basic Local Area  2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area  11.90  11.83  2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area  11.90  11.90  1.83  2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area  11.90  11.90  1.83  2-Wire Voice Grade Port (Centrex Vibro Gende Port Terminated on 800 Service Term - Basic Local Area  11.90  11.90  1.83  2-Wire Voice Grade Port Centrex Vibro Gend				3	UEP91	UECS2	36.68										
2-Wire Voice Grade Port (Centrex ) Basic Local Area   UEP91   UEPYA   1.17     11.90   1.83																	
2-Wire Voice Grade Port (Centrex 800 termination)Basic Local   UEP91   UEPYB   1.17						ļ					ļ						
Area				1	UEP91	UEPYA	1.17						11.90			1.83	
2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local   UEP91   UEPYH   1.17					LIED01	LIEDVB	1 17						11.00			1 02	
Area   UEP91   UEPYH   1.17   11.90   1.83				1	OL1 31	OLFID	1.17				1		11.90			1.03	
2-Wire Voice Grade Port (Centrex from diff Serving Wire Center - 800 Service   UEP91   UEPYM   1.17   UEPYM   1.190   1.83					UEP91	UEPYH	1.17						11.90			1.83	
Center/2 Basic Local Area   UEP91   UEPYM   1.17     11.90   1.83						1	,				İ						
Term - Basic Local Area		Center)2 Basic Local Area			UEP91	UEPYM	1.17				<u> </u>		11.90	<u> </u>		1.83	
2-Wire Voice Grade Port terminated in on Megalink or equivalent   UEP91																	
- Basic Local Area					UEP91	UEPYZ	1.17					<u> </u>	11.90			1.83	
2-Wire Voice Grade Port Terminated on 800 Service Term -   UEP91					LIEDO4	LIED) (2											
Basic Local Area					UEP91	UEPY9	1.17						11.90			1.83	
Georgia and Florida Only					LIFP91	LIEPY2	1 17						11 00			1 92	
2-Wire Voice Grade Port (Centrex )			1	1	OE1 31	JLI 12	1.17				1	1	11.50			1.03	
2-Wire Voice Grade Port (Centrex 800 termination)			1		UEP91	UEPHA	1.17				1		11.90			1.83	
2-Wire Voice Grade Port (Centrex with Caller ID)1														1			
2-Wire Voice Grade Port (Centrex from diff Serving Wire		2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPHH	1.17						11.90			1.83	
		2-Wire Voice Grade Port (Centrex from diff Serving Wire															
Center)2					UEP91	UEPHM	1.17						11.90			1.83	
2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service					LIEDO4	LIEDLIZ							44.00			4.00	
Term   UEP91 UEPHZ   1.17     11.90   1.83		Ierm			UEP91	UEPHZ	1.17				l		11.90	l		1.83	

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec			g Disconnect				RATES (\$)		
-							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPH9	1.17						11.90			1.83	
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPH2	1.17						11.90			1.83	
Local S	witching Centrex Intercom Funtionality, per port			UEP91	URECS	0.7384										
Local N	umber Portability			OLF91	UKLCS	0.7364										
	Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Feature				LIEBO.	LIEDVE											
-	All Standard Features Offered, per port  All Select Features Offered, per port			UEP91 UEP91	UEPVF UEPVS	2.26 0.00	370.70									
<del>                                     </del>	All Centrex Control Features Offered, per port			UEP91	UEPVS	2.26	310.10									
NARS																
	Unbundled Network Access Register - Combination			UEP91	UARCX	0.00	0.00	0.00								
$\vdash$	Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial			UEP91 UEP91	UAR1X UAROX	0.00	0.00	0.00								
Miscella	aneous Terminations			UEF91	UARUX	0.00	0.00	0.00								
	Trunk Side															
	Trunk Side Terminations, each			UEP91	CENA6	8.81										
Interoffi	ce Channel Mileage - 2-Wire			LIEDO4	MODO	05.00										
<b>—</b>	Interoffice Channel Facilities Termination - Voice Grade Interoffice Channel mileage, per mile or fraction of mile			UEP91 UEP91	MIGBC MIGBM	25.32 0.0091										
Feature	Activations (DS0) Centrex Loops on Channelized DS1 Service			OLI 01	WIIODWI	0.0001										
D4 Char	nnel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP91	1PQW7	0.66										
	Different Wire Center			UEP91	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP91	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.66										
Non-Re	Curring Charges (NRC) Associated with UNE-P Centrex		<b>—</b>		1							ļ				
	Conversion - Currently Combined Switch-As-Is with allowed changes, per port			UEP91	USAC2		21.50	8.42								
	Conversion of Existing Centrex Common Block			UEP91	USACN		5.17	8.32								
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	618.82									
$\vdash$	New Centrex Customized Common Block		<b>—</b>	UEP91	M1ACC	0.00	618.82					ļ				
	Secondary Block, per Block NAR Establishment Charge, Per Occasion			UEP91 UEP91	M2CC1 URECA	0.00	71.31 66.48									
	CENTREX - 5ESS (Valid in All States)					0.00	33.13									
	/G Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE Po	rt/Loop Combination Rates (Non-Design)		1		<u> </u>											
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo- Non-Design  2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo		1	UEP95		14.11										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95		18.23										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP95		33.04										
UNE Po	rt/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP95		16.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP95		21.60										

	NETWORK ELEMENTS - Florida		1										Attachment:			Exhibit: E
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted Manually		Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec			g Disconnect				RATES (\$)		
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Design		3	UEP95		37.85										
UNE Lo																
	2-Wire Voice Grade Loop (SL 1) - Zone 1			UEP95	UECS1	12.94										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	17.06										<b></b>
$\longrightarrow$	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95 UEP95	UECS1 UECS2	31.87 15.36										<b></b>
	2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2		1 2	UEP95 UEP95	UECS2	20.43						-				<del>                                     </del>
	2-Wire Voice Grade Loop (SL 2) - Zone 2  2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	36.68										
UNE Po			J	02.00	32002	55.00	-									
All State	es				İ		İ									<b>†</b>
	2-Wire Voice Grade Port (Centrex ) Basic Local Area			UEP95	UEPYA	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95	UEPYH	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP95	UEPYM	1.17						11.90			1.83	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP95	UEPYZ	1.17						11.90			1.83	
	Wire Voice Grade Port terminated in on Megalink or equivalent     Basic Local Area			UEP95	UEPY9	1.17						11.90			1.83	
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP95	UEPY2	1.17						11.90			1.83	
AL, KY,	LA, MS, SC, & TN Only															
FL & GA																
	2-Wire Voice Grade Port (Centrex )			UEP95	UEPHA	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPHB	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPHH	1.17						11.90			1.83	<b></b>
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPHM	1.17						11.90			1.83	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPHZ	1.17						11.90			1.83	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPH9	1.17						11.90			1.83	
<del> </del>	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPH2	1.17				ļ		11.90			1.83	
Local S	Witching  Control Intercom Funtionality, per part		-	LIEDOE	LIBECS	0.7204	+									
l ocal M	Centrex Intercom Funtionality, per port umber Portability			UEP95	URECS	0.7384										
Local N	Local Number Portability (1 per port)			UEP95	LNPCC	0.35	ł				1	-				<del></del>
Feature						0.00	+			1	1					
	All Standard Features Offered, per port			UEP95	UEPVF	2.26										
	All Select Features Offered, per port			UEP95	UEPVS	0.00	370.70									
	All Centrex Control Features Offered, per port			UEP95	UEPVC	2.26										
NARS					<u> </u>											<u> </u>
$\longrightarrow$	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00								<del>                                     </del>
	Unbundled Network Access Register - Indial Unbundled Network Access Register - Outdial			UEP95 UEP95	UAR1X UAROX	0.00	0.00	0.00		1	1	-				
Miscoll	Unbundled Network Access Register - Outdial aneous Terminations			UEP95	UARUX	0.00	0.00	0.00			-	-				<del> </del>
	Trunk Side				+		ł				1	-				
	Trunk Side Terminations, each			UEP95	CEND6	8.81										
4-Wire I	Digital (1.544 Megabits)					2.01										
	DS1 Circuit Terminations, each			UEP95	M1HD1	54.95										
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	15.69									
Interoff	ice Channel Mileage - 2-Wire															
	Hataraffica Channal Facilities Tanaination	Ì	i l	UEP95	MIGBC	25.32						L				<u> </u>
	Interoffice Channel Facilities Termination Interoffice Channel mileage, per mile or fraction of mile		1	UEP95	MIGBM	0.0091	- 1									

INBUNDLE	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: I
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -
						Rec	Nonrec	urring	Nonrecurring	g Disconnect			oss	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
D4 Cha	nnel Bank Feature Activations			LIEBAE	1001110											
	Feature Activation on D-4 Channel Bank Centrex Loop Slot		1	UEP95	1PQWS	0.66										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.66										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop			021 00	11 Q110	0.00										-
	Slot			UEP95	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP95	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot		1	UEP95	1PQWV	0.66										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP95	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot		1	UEP95	1PQWA	0.66										
Non-Re	curring Charges (NRC) Associated with UNE-P Centrex	1		00	~	0.00	-			1	1			1		t
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP95	USAC2	0.00	21.50	8.42								
	Conversion of Existing Centrex Common Block, each			UEP95	USACN		5.17	8.32								
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	618.82									
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	618.82									
UNE D	NAR Establishment Charge, Per Occasion		1	UEP95	URECA	0.00	66.48									
	CENTREX - DMS100 (Valid in All States) VG Loop/2-Wire Voice Grade Port (Centrex) Combo		1													
	ort/Loop Combination Rates (Non-Design)				+											-
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				1		1									
	Non-Design		1	UEP9D		14.11										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP9D		18.23										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			LIEDOD		00.04										
LINE Do	Non-Design ort/Loop Combination Rates (Design)		3	UEP9D	+	33.04										
UNE PO	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -		1													
	Design		1	UEP9D		16.53										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -					70.00										
	Design		2	UEP9D		21.60										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP9D		37.85								ļ		
UNE Lo	pop Rate		1	UEP9D	LIECC4	12.94	-				1	1		-		-
	2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D UEP9D	UECS1 UECS1	12.94							-			
	2-Wire Voice Grade Loop (SL 1) - Zone 2  2-Wire Voice Grade Loop (SL 1) - Zone 3	1	3	UEP9D	UECS1	31.87	ł				1	1		1		<del>                                     </del>
	2-Wire Voice Grade Loop (SL 1) - Zone 3		1	UEP9D	UECS2	15.36										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	20.43								<u> </u>		
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	36.68										
UNE Po																
ALL ST			1	LIEDOD	LIEDYA					-	<u> </u>	44.00		-	1.00	
	2-Wire Voice Grade Port (Centrex ) Basic Local Area 2-Wire Voice Grade Port (Centrex 800 termination)Basic Local	1	1	UEP9D	UEPYA	1.17				1	1	11.90			1.83	<del>                                     </del>
	Area			UEP9D	UEPYB	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local		1	J_1 JD	02.10	1.17	-				1	11.30			1.00	
	Area			UEP9D	UEPYC	1.17						11.90		1	1.83	
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local						İ									
	Area			UEP9D	UEPYD	1.17						11.90			1.83	
			1		1				ı	1			l			
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local			LIEBAR												
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area      2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local			UEP9D	UEPYE	1.17						11.90			1.83	

I CHADOMATED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)	ı			Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec		curring		g Disconnect				RATES (\$)		
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Area			UEP9D	UEPYG	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local															
	Area			UEP9D	UEPYT	1.17						11.90			1.83	<b></b>
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local			OLI 3D	OLI 10	1.17						11.30			1.00	
	Area			UEP9D	UEPYV	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local															ĺ
	Area 2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local			UEP9D	UEPY3	1.17						11.90			1.83	<del> </del>
	Area			UEP9D	UEPYH	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp				Ì											
	Indication))3 Basic Local Area			UEP9D	UEPYW	1.17						11.90			1.83	<b></b>
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			OLF9D	OLF 13	1.17						11.90			1.03	<u> </u>
	2 Basic Local Area			UEP9D	UEPYM	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3															ĺ
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPYO	1.17						11.90			1.83	<del>                                     </del>
	Basic Local Area			UEP9D	UEPYP	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3															
	Basic Local Area			UEP9D	UEPYQ	1.17						11.90			1.83	<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEFTK	1.17						11.90			1.03	<del>                                     </del>
	Basic Local Area			UEP9D	UEPYS	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3															ĺ
	Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPY4	1.17						11.90			1.83	<del> </del>
	Basic Local Area			UEP9D	UEPY5	1.17						11.90			1.83	ĺ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3															
	Basic Local Area			UEP9D	UEPY6	1.17						11.90			1.83	<b></b>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			LIEDOD	UEPY7	1.17						11.00			1 02	
	Basic Local Area 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service			UEP9D	UEF1/	1.17			<b> </b>			11.90			1.83	<b>—</b>
	Term			UEP9D	UEPYZ	1.17			<u> </u>			11.90			1.83	<u>i</u>
	2-Wire Voice Grade Port terminated in on Megalink or equivalent															1
$\vdash$	Basic Local Area 2-Wire Voice Grade Port Terminated on 800 Service Term Basic			UEP9D	UEPY9	1.17			<del>                                     </del>		1	11.90			1.83	<del>                                     </del>
	Local Area			UEP9D	UEPY2	1.17						11.90			1.83	İ
FL & GA	Only															
	2-Wire Voice Grade Port (Centrex)			UEP9D	UEPHA	1.17						11.90			1.83	L
$\vdash$	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D UEP9D	UEPHB UEPHC	1.17 1.17			<b>.</b>			11.90 11.90			1.83 1.83	<del>                                     </del>
<del>                                     </del>	2-Wire Voice Grade Port (Centrex / EBS-PSE1)3 2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D UEP9D	UEPHD	1.17			<del>                                     </del>	1		11.90			1.83	<del>                                     </del>
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPHE	1.17			<b>†</b>	1	1	11.90	1		1.83	<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPHF	1.17			1			11.90			1.83	
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPHG	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPHT	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPHU	1.17						11.90			1.83	
$\longrightarrow$	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPHV	1.17			ļ			11.90			1.83	<b> </b>
$\vdash$	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPH3	1.17						11.90			1.83	<del>                                     </del>
$\vdash$	2-Wire Voice Grade Port (Centrex with Caller ID) 2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp			UEP9D	UEPHH	1.17		-	<del>                                     </del>	1	-	11.90			1.83	<del>                                     </del>
	Indication)3			UEP9D	UEPHW	1.17			1			11.90			1.83	İ

BUNDLED	NETWORK ELEMENTS - Florida											Attachment	: 2		Exhibit:
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC	ı		RATES(\$)	I	Subr E	nitted Submi	Incrementa Charge - der Manual Svo ted Order vs. Ily Electronic-	I Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec		Nonrecurring Disc				RATES (\$)		
				LIEBAR			First	Add'l	First A	dd'I SOI	IEC SOM		SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)			UEP9D	UEPHJ	1.17					11	90		1.83	
	2-wire voice Grade Fort (Certifex from diri Serving Wire Certier)			UEP9D	UEPHM	1.17					11	90		1.83	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPHO	1.17						90		1.83	
	· ·														
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPHP	1.17						90		1.83	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPHQ	1.17					11	90		1.83	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPHR	1.17					11	90		1.83	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPHS	1.17					11	90		1.83	
	2 VIII VOICE CIAGE I ON (CONTROVAMEN CIVE / EBC MOOTE)2, C			OLI OD	CELTIC	1.17					· · ·	50		1.00	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPH4	1.17					11	90		1.83	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPH5	1.17					11	90		1.83	
	O Miles Males October Deat (October 1111 or OMO (EDO MESA))			LIEDOD	LIEBLIA	1.17						00		4.00	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPH6	1.17					11	90		1.83	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPH7	1.17					11	90		1.83	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service											-			
	Term			UEP9D	UEPHZ	1.17					11	90		1.83	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPH9	1.17						90		1.83 1.83	
Local Sv	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPH2	1.17					- 11	90		1.83	
LUCAI 3V	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.7384									
Local Nu	Imber Portability			02. 03	0.1200	0.7001									
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35									
Features															
	All Standard Features Offered, per port			UEP9D	UEPVF	2.26									
	All Select Features Offered, per port			UEP9D	UEPVS	0.00 2.26	370.70								
NARS	All Centrex Control Features Offered, per port			UEP9D	UEPVC	2.20									
IVANO	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00							
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00							
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00							
	neous Terminations														
2-Wire T	runk Side														
4 10/: D	Trunk Side Terminations, each			UEP9D	CEND6	8.81									
4-Wire D	igital (1.544 Megabits) DS1 Circuit Terminations, each			UEP9D	M1HD1	54.95									
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	15.69								
	ce Channel Mileage - 2-Wire			02.05		0.00	10.00								
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	25.32									
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0091		•							
	Activations (DS0) Centrex Loops on Channelized DS1 Service														
D4 Chan	nel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.66				<del></del>			+		
	reacure Activation on D-4 Channel Bank Centrex Loop Slot			UELAD	IPQWS	0.66							+		
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.66									
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop			LIEDOD	100/4/7	0.00									
	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP9D	1PQW7	0.66						+			
	Different Wire Center			UEP9D	1PQWP	0.66									
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.66									

UNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit: B
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurrin	g Disconnect			oss i	RATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.66										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.66										
Non-Red	curring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP9D	USAC2		21.50	8.42								
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		5.17	8.32								
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	618.82									
-+-	New Centrex Customized Common Block NAR Establishment Charge, Per Occasion	<u> </u>	<del>                                     </del>	UEP9D	M1ACC	0.00	618.82			<b> </b>	ļ	1	-			
LINE D (	INAR Establishment Charge, Per Occasion  CENTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)	<b>!</b>	<del>                                     </del>	UEP9D	URECA	0.00	66.48			-	<b> </b>	1				-
	G Loop/2-Wire Voice Grade Port (Centrex) Combo	1	<del> </del>		1	<b>H</b>			1	1	<b> </b>	<b> </b>	1			1
	rt/Loop Combination Rates (Non-Design)				-											
OIL I O	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				+	1					1					
	Non-Design		1	UEP9E		14.11										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Non-Design		2	UEP9E		18.23										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo															
	Non-Design		3	UEP9E		33.04										
UNE Po	rt/Loop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -	1	1	UEP9E		16.53										
	Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		1	UEP9E	_	16.53										
	Design		2	UEP9E		21.60										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -			UEF9E	_	21.00										
	Design		3	UEP9E		37.85										
UNE Loc				OLI 3L		37.03										
OIL LO	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	12.94					1					
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP9E	UECS1	17.06										
	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP9E	UECS1	31.87										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	15.36										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9E	UECS2	20.43										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9E	UECS2	36.68										
UNE Poi																
AL, FL,	KY, LA, MS, & TN only							<u> </u>								
	2-Wire Voice Grade Port (Centrex ) Basic Local Area			UEP9E	UEPYA	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local	l		LIEDOE	LIED. C											
	Area	<u> </u>	<u> </u>	UEP9E	UEPYB	1.17				-	<u> </u>	11.90	ļ		1.83	<b></b>
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local		1	UEP9E	UEPYH	1.17						11.90			1.83	
-+-	Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire	<b>!</b>	<del>                                     </del>	UEP9E	UEPYH	1.17				-	<b> </b>	11.90			1.83	-
	Center)2 Basic Local Area	l		UEP9E	UEPYM	1.17						11.90			1.83	
<del></del>	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	1		OLI JL	OLF IIVI	1.17			<del> </del>	<del> </del>		11.90			1.03	<del> </del>
	Term - Basic Local Area	l		UEP9E	UEPYZ	1.17						11.90			1.83	
<del>-  </del>	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1	<b>!</b>		<u> </u>	/				1		11.50			1.00	<b> </b>
	- Basic Local Area	l		UEP9E	UEPY9	1.17						11.90			1.83	
	2-Wire Voice Grade Port Terminated on 800 Service Term -		i –	-	1	'''			İ	1		1	İ			İ
1	Basic Local Area	l	1	UEP9E	UEPY2	1.17						11.90			1.83	1
Florida					1											
	2-Wire Voice Grade Port (Centrex )			UEP9E	UEPHA	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPHB	1.17						11.90			1.83	
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPHH	1.17						11.90			1.83	L
	2-Wire Voice Grade Port (Centrex from diff Serving Wire	l	1	l	l	I						1				1
	Center)2	<u> </u>	<u> </u>	UEP9E	UEPHM	1.17					ļ	11.90			1.83	ļ
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service	l		LIEDOE	LIEDUZ							44.00			4.00	
	Term	l	Ĺ	UEP9E	UEPHZ	1.17			1	l	l .	11.90	l		1.83	<u> </u>

JNBUNDLED	NETWORK ELEMENTS - Florida												Attachment:	2		Exhibit:
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Increments Charge - Manual Sv Order vs. Electronic Disc Add
						B	<b>N</b> 1		<b>N</b> 1				000	DATEO (A)		
						Rec	Nonrec First	urring Add'l	First	g Disconnect Add'l	SOMEC	SOMAN	SOMAN	RATES (\$) SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E UEP9E	UEPH9 UEPH2	1.17 1.17						11.90			1.83	
Local S	witching			OLI OL	OLITIZ	,						11.00			1.00	
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.7384										
Local N	umber Portability															
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35					1	ļ				
Feature	All Standard Features Offered, per port			UEP9E	UEPVF	2.26				<del>                                     </del>	1	-				<del>                                     </del>
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	370.70			-		-				<u> </u>
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	2.26	310.10			<b>†</b>						
NARS	2227 Control 1 Catalog Cholod, por port					2.20				1		l –				
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00								
	neous Terminations															
2-Wire 1	runk Side				051100	2.21										
4 10/: [	Trunk Side Terminations, each			UEP9E	CEND6	8.81										
4-wire L	Digital (1.544 Megabits) DS1 Circuit Terminations, each		1	UEP9E	M1HD1	54.95						-				
	DS0 Channel Activated Per Channel			UEP9E	M1HDO	0.00	15.69									
Interoffi	ce Channel Mileage - 2-Wire			OLI 3L	WITIDO	0.00	15.05									
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	25.32				İ						
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0091										
	Activations (DS0) Centrex Loops on Channelized DS1 Service															
D4 Char	nel Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.66										<u> </u>
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot Feature Activation on D-4 Channel Bank FX Trunk Side Loop			UEP9E	1PQW6	0.66										
	Slot			UEP9E	1PQW7	0.66										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9E	1PQWP	0.66										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop			UEP9E	1PQWV	0.66										
	Slot			UEP9E	1PQWQ	0.66				1						1
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.66										
Non-Re	curring Charges (NRC) Associated with UNE-P Centrex				1	2.00				1			İ			
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP9E	USAC2		21.50	8.42								
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		5.17	8.32								
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	618.82	·								
	New Centrex Customized Common Block			UEP9E	M1ACC	0.00	618.82									
No.	NAR Establishment Charge, Per Occasion		<b>.</b>	UEP9E	URECA	0.00	66.48				<u> </u>		ļ	ļ		
	Required Port for Centrex Control in 1AESS, 5ESS & EWSD									<del>                                     </del>	1	1	<del>                                     </del>			<del>                                     </del>
	Requires Interoffice Channel Mileage Requires Specific Customer Premises Equipment				+					+						<del>                                     </del>
14016.2 -	requires openine oustomer i remises Equipment				+					<del>                                     </del>	1	-	<del> </del>			<del></del>
										1						
					1					1			İ			

LOCA	INTE	RCONNECTION - Florida								I				Attachment:	2		Exhibit: A
LOCA	L 1141 L	INCOMMECTION - Florida								<u>l</u>							
														Incremental			Incremental
														Charge -	Charge -	Charge -	Charge -
CATE	GORY	RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)					Manual Svc	Manual Svc		Manual Svc
			m						***			Submitted	Submitted		Order vs.	Order vs.	Order vs.
												Elec	Manually	Electronic-	Electronic-	Electronic-	Electronic-
												per LSR	per LSR	1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonred			Disconnect				RATES (\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			<u> </u>	<u> </u>													
LOCAL	INTER	CONNECTION (CALL TRANSPORT AND TERMINATION)															
		"bk" beside a rate indicates that the Parties have agreed to bi	ll and k	eep for	that element pursua	nt to the ter	ms and conditi	ons in Attachr	nent 3.								
		M SWITCHING															
		Tandem Switching Function Per MOU			OHD		0.0006019bk										
		Multiple Tandem Switching, per MOU (applies to intial tandem															
		only)	<u> </u>		OHD		0.0006019bk										
	TRUNK	CHARGE	ļ	<u> </u>	O. I.D.												
<u> </u>		Installation Trunk Side Service - per DS0	<b> </b>	<u> </u>	OHD	TPP++	0.00	336.43bk	57.38bk						-		
-	-	Dedicated End Office Trunk Port Service-per DS0** Dedicated End Office Trunk Port Service-per DS1**	<u> </u>		OHD 0H1 OH1MS	TDE0P TDE1P	0.00										
-	-	Dedicated Tandem Trunk Port Service-per DS1**	<del>                                     </del>	-	OHD	TDW0P	0.00			-					1		
		Dedicated Tandem Trunk Port Service-per DS0**			OH1 OH1MS	TDW1P	0.00										
		rate element is recovered on a per MOU basis and is included	in the	Fnd Of				I rate elements									
	COMM	ON TRANSPORT (Shared)					, per										
		Common Transport - Per Mile, Per MOU			OHD		0.0000035bk										
		Common Transport - Facilities Termination Per MOU			OHD		0.0004372bk										
LOCAL		CONNECTION (TRANSPORT)															
	INTERC	OFFICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADI	<u> </u>														
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -			OLIL OLIM	41 ENE	0.000451										
		Per Mile per month Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			OHL, OHM	1L5NF	0.0091bk										
		Facility Termination per month			OHL, OHM	1L5NF	25.32bk	31.78bk		7.03bk							
	INTER	DFFICE CHANNEL - DEDICATED TRANSPORT - 56/64 KBPS			OTIL, OTIVI	TESIVI	25.52bk	31.700K		7.000K							
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
		per month			OHL, OHM	1L5NK	0.0091bk										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility															
		Termination per month			OHL, OHM	1L5NK	18.44bk	31.78bk		7.03bk							
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile															
		per month			OHL, OHM	1L5NK	0.0091bk										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month			OHL, OHM	1L5NK	18.44bk	31.78bk		7.03bk							
-	INTER	DEFICE CHANNEL - DEDICATED TRANSPORT - DS1	<del>                                     </del>	-	Oi IL, UNIVI	TLOINT	18.440K	31.78DK		7.USDK					1		
-	ERC	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per	1	<del>                                     </del>													
		month			OH1. OH1MS	1L5NL	0.1856bk										
		Interoffice Channel - Dedicated Tranport - DS1 - Facility													1		
L	<u></u>	Termination per month	<u></u>		OH1, OH1MS	1L5NL	88.44bk	98.47bk		19.05bk		<u> </u>					
	INTERC	OFFICE CHANNEL - DEDICATED TRANSPORT- DS3															
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per															
		month	ļ		OH3, OH3MS	1L5NM	3.87bk										
		Interoffice Channel - Dedicated Transport - DS3 - Facility			OH3. OH3MS	1L5NM	1074 0011	240.00		70 501 1							
-	LOCAL	Termination per month  CHANNEL - DEDICATED TRANSPORT	<u> </u>		UNS, UNSIVIS	ILDINICAL	1071.00bk	219.28bk		70.56bk							
-	LOCAL	Local Channel - Dedicated - 2-Wire Voice Grade per month	1	<b>-</b>	OHL, OHM	TEFV2	21.94bk	265.84bk	46.97bk	37.63bk	4.00bkbk	1			1		
<b>-</b>		Local Channel - Dedicated - 2-Wire Voice Grade per month	<b> </b>		OHL, OHM	TEFV4	22.81bk	266.54bk	47.67bk		5.33bk	1					
		Local Channel - Dedicated - DS1 per month			OH1	TEFHG	35.28bk	216.65bk	183.54bk		16.95bk						
		·			-										1		
		Local Channel - Dedicated - DS3 Facility Termination per month	<u> </u>		OH3	TEFHJ	531.91bk	556.37bk	343.01bk	139.13bk	96.84bk				<u> </u>		
		INTERCONNECTION MID-SPAN MEET							•								
		If Access service ride Mid-Span Meet, one-half the tariffed ser	rvice Lo	cal Ch	annel rate is applical	ole.											
<u> </u>	MULTI	PLEXERS	<u> </u>	ļ	OLIA OLIANA	CATNI	440 771 1	404 407	74.00::	44.00**	40.40::				ļ		
-	-	Channelization - DS1 to DS0 Channel System DS3 to DS1 Channel System per month	1		OH1, OH1MS OH3, OH3MS	SATN1 SATNS	146.77bk 211.19bk	101.42bk 199.28bk	71.62bk 118.64bk	11.09bk 40.34bk	10.49bk 39.07bk						
-		DS3 Interface Unit (DS1 COCI) per month	<del>                                     </del>	1	OH3, OH3MS	SATCO	13.76bk	199.280k	7.08bk	40.34DK	39.U/DK					-	
	l	DOO INTENACE ONLY (DO FOOD) PER MONTH	<u> </u>	<u> </u>	OTTI, OTTINO	UNICO	13.700K	10.070K	/ .UoDK	l		1	1		·	·	1

LOCAL INTE	RCONNECTION - Florida												Attachment:	3		Exhibit: A
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)			Submitted Elec	Submitted	Charge - Manual Svc Order vs. Electronic-	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
Natan	If no rate is identified in the contract, the rates, terms, and co		- for th		franction wi	Rec	Nonred First	Add'l	Nonrecurring First	Add'l	SOMEC		SOMAN	RATES (\$) SOMAN	SOMAN	SOMAN

SERVICE PROVIDER NUMBER PORTABILITY - Florida								Attachment:	5		Exhibit: A				
	lut and				NATEO(Ψ)							Charge -	Charge -	Incremental Charge -	Charge -
CATEGORY RATE ELEMENTS	Interi	Zone	BCS	USOC			RATES(\$)							Manual Svc	
	m									Submitted	Submitted	Order vs.	Order vs.	Order vs.	Order vs.
										Elec	Manually	Electronic-	Electronic-	Electronic-	Electronic-
										per LSR	per LSR	1st	Add'l	Disc 1st	Disc Add'l
					Rec	Nonro	curring	Nonrecurring	n Dissennest			000	RATES (\$)		
		-			Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		-			Filst Add 1 113t Add 1						SUMAN	SUMAN	SOMAN	SOWAN	SUMAN
		1								1					
		1													
		1													
NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO,		EC rate	reflects the charge	that would b	e billed to a CL	EC once elect	ronic ordering	capabilities co	ome on-line fo	r that eleme	nt. Otherw	ise, the manu	al ordering ch	arge, SOMAN	, will be
		EC rate	reflects the charge	that would b		EC once elect	ronic ordering	capabilities co	ome on-line fo	r that eleme	nt. Otherw	ise, the manu	al ordering ch	narge, SOMAN	, will be
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth		EC rate		TNPBL	e billed to a CL	LEC once elect	ronic ordering 0.4145	0.0415	ome on-line fo 0.0415	3.50	nt. Otherwi	Ī	al ordering ch	narge, SOMAN	, will be
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF		EC rate											al ordering ch		, will be
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF  RCF, per number ported (Business Line)  RCF, per number ported (Residence Line)  RCF, Per Additional Path	1.			TNPBL TNPRL	2.05 2.05 0.7179	0.4145 0.4145	0.4145 0.4145	0.0415 0.0415	0.0415 0.0415	3.50 3.50	11.90 11.90			1.83	
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, Per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL	2.05 2.05 0.7179 er to BellSouth	0.4145 0.4145	0.4145 0.4145 ules for Local (	0.0415 0.0415 Ordering (BBR-	0.0415 0.0415 -LO) to determ	3.50 3.50	11.90 11.90	ordered elect	ronically. Fo	1.83 1.83	nts that
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL I. Please refithat would b	2.05 2.05 0.7179 er to BellSouth	0.4145 0.4145 's Business Ri .EC once elect	0.4145 0.4145 ules for Local C	0.0415 0.0415 Ordering (BBR- capabilities co	0.0415 0.0415 -LO) to determ ome on-line fo	3.50 3.50 3.60 Ine if a proor that eleme	11.90 11.90 uct can be nt. Otherw	ordered electise, the manu.	ronically. Fo	1.83 1.83 r those elemei	nts that
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID     DID per number ported (Residence)	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL  I. Please refethat would b	2.05 2.05 0.7179 er to BellSouth	0.4145 0.4145 's Business Ri EC once elect	0.4145 0.4145 ules for Local C ronic ordering 0.6923	0.0415 0.0415 Drdering (BBR- capabilities co	0.0415 0.0415 -LO) to determ ome on-line fo 0.6923	3.50 3.50 3.60 Ine if a proc r that eleme	11.90 11.90 uct can be nt. Otherw	ordered electise, the manu.	ronically. Fo	1.83 1.83 1.83 1.83 1.83	nts that
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID     DID per number ported (Residence)     DID per number ported (Business)	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL I. Please refthat would b TNPDR TNPDR TNPDB	2.05 2.05 0.7179 er to BellSouth e billed to a CL	0.4145 0.4145 ***********************************	0.4145 0.4145 Ules for Local Gronic ordering 0.6923 0.6923	0.0415 0.0415 O.0415 Ordering (BBR- capabilities co	0.0415 0.0415 -LO) to determ ome on-line fo 0.6923 0.6923	3.50 3.50 3.50 ine if a proc r that eleme 3.50 3.50	11.90 11.90 uct can be nt. Otherw	ordered electise, the manu.	ronically. Fo	1.83 1.83 1.83 1.83 1.83	nts that
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID     DID per number ported (Residence)     DID per number ported (Business)     DID, per trunk termination, Initial	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL  I. Please refethat would b	2.05 2.05 0.7179 er to BellSouth	0.4145 0.4145 's Business Ri EC once elect	0.4145 0.4145 ules for Local C ronic ordering 0.6923	0.0415 0.0415 Drdering (BBR- capabilities co	0.0415 0.0415 -LO) to determ ome on-line fo 0.6923	3.50 3.50 3.60 Ine if a proc r that eleme	11.90 11.90 uct can be nt. Otherw	ordered electise, the manu.	ronically. Fo	1.83 1.83 1.83 1.83 1.83	nts that
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, per number ported (Residence Line)     RCF, Per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth     INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID     DID per number ported (Residence)     DID per number ported (Business)     DID, per trunk termination, Initial     SERVICE PROVIDER NUMBER PORTABILITY (RIPH)	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL I. Please refthat would b TNPDR TNPDR TNPDB	2.05 2.05 0.7179 er to BellSouth e billed to a CL	0.4145 0.4145 'S Business Ri EC once elect 0.6923 0.6923 161.29	0.4145 0.4145 0.04145 Ules for Local Cronic ordering 0.6923 0.6923 80.58	0.0415 0.0415 O.0415 Ordering (BBR- capabilities co	0.0415 0.0415 -LO) to determ ome on-line fo 0.6923 0.6923	3.50 3.50 3.50 Ine if a proc r that eleme 3.50 3.50 3.50	11.90 11.90 luct can be nt. Otherwi	ordered elect ise, the manu	ronically. Fo	1.83 1.83 1.83 1.83 1.83 1.83	nts that
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, per number ported (Residence Line)     RCF, per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID     DID per number ported (Residence)     DID per number ported (Business)     DID, per trunk termination, Initial     SERVICE PROVIDER NUMBER PORTABILITY (RIPH)     RIPH, Functionality, Per Rearrangement	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL I. Please refthat would b TNPDR TNPDR TNPDB	2.05 2.05 0.7179 er to BellSouth be billed to a CL	0.4145 0.4145 S Business Ri EC once elect 0.6923 0.6923 161.29	0.4145 0.4145 ules for Local tronic ordering 0.6923 0.6923 80.58	0.0415 0.0415 Drdering (BBR: capabilities co 0.6923 0.6923 32.73	0.0415 0.0415 -LO) to determ me on-line fo 0.6923 0.6923 32.73	3.50 3.50 Ine if a proor that eleme 3.50 3.50 3.50	11.90 11.90 11.90 11.90 11.90 11.90	ordered elect ise, the manu	ronically. Fo	1.83 1.83 1.83 1.83 1.83 1.83 1.83	nts that
cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth INTERIM SERVICE PROVIDER NUMBER PORTABILITY - RCF    RCF, per number ported (Business Line)     RCF, per number ported (Residence Line)     RCF, per number ported (Residence Line)     RCF, Per Additional Path     NOTE: Any element that can be ordered electronically will be cannot be ordered electronically at present per the BBR-LO, applied to a CLEC's bill when it submits an LSR to BellSouth     INTERIM SERVICE PROVIDER NUMBER PORTABILITY - DID     DID per number ported (Residence)     DID per number ported (Business)     DID, per trunk termination, Initial     SERVICE PROVIDER NUMBER PORTABILITY (RIPH)	n.  billed according the listed SOM	ng to tr	ne SOMEC rate listed	TNPBL TNPRL I. Please refthat would b TNPDR TNPDR TNPDB	2.05 2.05 0.7179 er to BellSouth e billed to a CL	0.4145 0.4145 'S Business Ri EC once elect 0.6923 0.6923 161.29	0.4145 0.4145 0.04145 Ules for Local Cronic ordering 0.6923 0.6923 80.58	0.0415 0.0415 O.0415 Ordering (BBR- capabilities co	0.0415 0.0415 -LO) to determ ome on-line fo 0.6923 0.6923	3.50 3.50 3.50 Ine if a proc r that eleme 3.50 3.50 3.50	11.90 11.90 luct can be nt. Otherwi	ordered elect ise, the manu	ronically. Fo	1.83 1.83 1.83 1.83 1.83 1.83	nts that

DUF/ADUF/CMDS	6 - Florida												Attachment:	7		Exhibit:
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic-	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	g Disconnect			oss	RATES (\$)					
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			1			-										
DUF/ADUF/CMDS			1													
	/ USAGE FILE (ADUF)		1			1									1	
	Message Processing, per message				N/A	0.014391										
	Data Transmission (CONNECT:DIRECT), per message				N/A	0.00012973										
	LY USAGE FILE (ODUF)															
	Recording, per message				N/A	0.0000071										
	Message Processing, per message Message Processing, per Magnetic Tape provisioned				N/A N/A	0.006835 48.96										
	Data Transmission (CONNECT:DIRECT), per message				N/A	0.00010811										
	MESSAGE DISTRIBUTION SERVICE (CMDS)						•									
CMDS:	Message Processing, per message		<b> </b>		N/A	0.004										
CMDS:	Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
CMDS:		service	e or fund	ction will be as set f			h tariff or as r	egotiated by th	ne Parties upor	n request by	r ei	either Party.	r either Party.	r either Party.	r either Party.	r either Party.

4Q01:12/01/01

## Amendment to the Interconnection Agreement By and Between BellSouth Telecommunications, Inc. And NewSouth Communications, Corp. Dated May 18, 2001

Pursuant to this Agreement, (the "Amendment"), NewSouth Communications, Corp., ("NewSouth"), and BellSouth Telecommunications, Inc. ("BellSouth"), hereinafter referred to collectively as the "Parties," hereby agree to amend that certain Interconnection Agreement between the Parties dated May 18, 2001 ("Agreement").

WHEREAS, BellSouth and NewSouth entered into an Interconnection Agreement on May 18, 2001, and;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. The Parties hereby agree to delete in entirety and replace the Trunk Charge rates contained in Exhibit A of Attachment 3 for all states attached hereto as Exhibit 1.
- 2. All of the other provisions of the Agreement, dated May 18, 2001, shall remain in full force and effect.
- 3. Either or both of the Parties is authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

NewSouth Communications, Corp.	BellSouth Telecommunications, Inc
Original Signature on File	Original Signature on File
Signature	Signature
Jake E. Jennings	C. W. Boltz
Name	Name
Vice President, Regulatory	Managing Director
Title	Title
5/6/02	5/7/02
Date	Date

LOCA	L INTE	RCONNECTION - Alabama												Attachment:	3	Exhibit: A	
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted		Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATE	ORY	RATE ELEMENTS	m	Zone	BCS	USOC		RAT	'ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		<u> </u>
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		CONNECTION (CALL TRANSPORT AND TERMINATION)															
		'bk" beside a rate indicates that the Parties have agreed to bi							nent 3.								
		harge is applicable only to transit traffic and is applied in ad-	dition to	applic	cable switching and/	or interconn	ection charges										
	TRUNK	CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		333.69bk	56.91bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
		Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00										
	** This	rate element is recovered on a per MOU basis and is included	l in the	End Of	fice Switching and T	andem Swit	ching, per MOl	J rate elements	1		•						

04/12/02 Page 1 of 9

LOCA	INTE	RCONNECTION - Florida												Attachment:	3	Exhibit: A	
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	DRY	RATE ELEMENTS	m	Zone	BCS	USOC		RAT	ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													_	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							D	Rec Nonrecurring			Disconnect		1	OSS	Rates(\$)		I.
							Rec	Rec Nonrecurring N			Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL	INTER	CONNECTION (CALL TRANSPORT AND TERMINATION)															
	NOTE:	"bk" beside a rate indicates that the Parties have agreed to bi	ll and k	eep for	that element pursua	ant to the ter	ms and conditi	ons in Attachn	nent 3.								
		CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		336.43bk	57.38bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
		Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00		•		•						
	** This	rate element is recovered on a per MOU basis and is included	in the	End Of	fice Switching and 1	andem Swit	ching, per MOL	J rate elements									

04/12/02 Page 2 of 9

LOCA	LOCAL INTERCONNECTION - Georgia Attachment: 3 Exh														Exhibit: A		
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ATEGORY RATE ELEMENTS		m	Zone	BCS	USOC		RAT	ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												1	_	Electronic-	Electronic-	Electronic-	Electronic-
										1st	Add'l	Disc 1st	Disc Add'l				
							D	Nonrec	urring	Nonrecurring	Disconnect		1	OSS	Rates(\$)		1
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL	INTER	CONNECTION (CALL TRANSPORT AND TERMINATION)															
	NOTE:	"bk" beside a rate indicates that the Parties have agreed to bi	ll and k	eep for	that element pursua	ant to the ter	ms and conditi	ons in Attachn	nent 3.								
		CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		333.28bk	56.84bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
Dedicated Tandem Trunk Port Service-per DS1** OH1 OH1MS TDW1P 0.00																	
	** This	rate element is recovered on a per MOU basis and is included	in the	End Of	fice Switching and 1	andem Swit	ching, per MOL	J rate elements									

04/12/02 Page 3 of 9

LOCA	L INTE	RCONNECTION - Kentucky												Attachment:	3	Exhibit: A	
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC		RAT	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect			OSS Rates(\$)			
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		CONNECTION (CALL TRANSPORT AND TERMINATION)															
	NOTE:	"bk" beside a rate indicates that the Parties have agreed to bi	ill and k	eep for	that element pursua	ant to the ter	ms and conditi	ons in Attachn	nent 3.								
	TRUNK	CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		334.09bk	57.12bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
		Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00										
	** This	rate element is recovered on a per MOU basis and is included	d in the	End Of	fice Switching and	Tandem Swit	ching, per MOl	J rate elements	3		•						

04/12/02 Page 4 of 9

LOCAL	LOCAL INTERCONNECTION - Louisiana Attachment: 3 Exh														Exhibit: A		
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGO	ATEGORY RATE ELEMENTS		m	Zone	BCS	USOC		RAT	ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							B	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		I.
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL I	NTER	CONNECTION (CALL TRANSPORT AND TERMINATION)															
N	NOTE:	"bk" beside a rate indicates that the Parties have agreed to bi	ll and k	eep for	that element pursua	ant to the ter	ms and conditi	ons in Attachn	nent 3.								
T		CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		334.94bk	56.98bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00		•		•						
Dedicated Tandem Trunk Port Service-per DS1**  OH1 OH1MS TDW1P 0.00																	
*	* This	rate element is recovered on a per MOU basis and is included	in the	End Of	fice Switching and 1	andem Swit	ching, per MOL	J rate elements									

04/12/02 Page 5 of 9

LOCAL	LOCAL INTERCONNECTION - Mississippi Attachment: 3 Exh														Exhibit: A		
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ATEGORY RATE ELEMENTS		m	Zone	BCS	USOC		RAT	ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												1		Electronic-	Electronic-	Electronic-	Electronic-
											1st	Add'l	Disc 1st	Disc Add'l			
							D	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		I.
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL	INTER	CONNECTION (CALL TRANSPORT AND TERMINATION)															
	NOTE:	"bk" beside a rate indicates that the Parties have agreed to bi	ll and k	eep for	that element pursua	ant to the ter	ms and conditi	ons in Attachn	nent 3.								
		CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		334.11bk	56.98bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
Dedicated Tandem Trunk Port Service-per DS1**  OH1 OH1MS TDW1P 0.00																	
	** This	rate element is recovered on a per MOU basis and is included	l in the	End Of	fice Switching and 1	andem Swit	ching, per MOL	J rate elements									

04/12/02 Page 6 of 9

LOCAL	LOCAL INTERCONNECTION - North Carolina Attachment: 3 Exhi														Exhibit: A		
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGO	ATEGORY RATE ELEMENTS		m	Zone	BCS	USOC		RAT	ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			'''									1		Electronic-	Electronic-	Electronic-	Electronic-
											1st	Add'l	Disc 1st	Disc Add'l			
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		I.
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL II	NTER	CONNECTION (CALL TRANSPORT AND TERMINATION)															
N	IOTE:	"bk" beside a rate indicates that the Parties have agreed to bi	ll and k	eep for	that element pursua	ant to the ter	ms and conditi	ons in Attachn	nent 3.								
T		CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		333.54bk	56.88bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00		·		·						
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
Dedicated Tandem Trunk Port Service-per DS1** OH1 OH1MS TDW1P 0.00																	
*1	* This	rate element is recovered on a per MOU basis and is included	l in the	End Of	fice Switching and T	andem Swit	ching, per MOl	J rate elements	i		•						

04/12/02 Page 7 of 9

LOCA	L INTE	RCONNECTION - South Carolina												Attachment:	3	Exhibit: A	
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	m	Zone	BCS	USOC		RAT	ES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
														Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		CONNECTION (CALL TRANSPORT AND TERMINATION)															
	NOTE: '	"bk" beside a rate indicates that the Parties have agreed to bi	ill and k	eep for	that element pursua	ant to the ter	ms and conditi	ons in Attachn	nent 3.								
	TRUNK	CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		335.14bk	57.16bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
		Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P	0.00										
	** This	rate element is recovered on a per MOU basis and is included	d in the	End Of	fice Switching and 1	andem Swit	ching, per MOL	J rate elements									

04/12/02 Page 8 of 9

LOCAL	LOCAL INTERCONNECTION - Tennessee Attachment: 3 Exh														Exhibit: A		
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
			Interi									Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ATEGORY RATE ELEMENTS		m	Zone	BCS	USOC		RAT	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
												1		Electronic-	Electronic-	Electronic-	Electronic-
										1st	Add'l	Disc 1st	Disc Add'l				
							Rec	Nonrecurring		Nonrecurring	Disconnect			oss	Rates(\$)		I.
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOCAL	INTER	CONNECTION (CALL TRANSPORT AND TERMINATION)															
	NOTE:	"bk" beside a rate indicates that the Parties have agreed to bi	ll and k	eep for	that element pursua	ant to the ter	ms and condit	ions in Attachn	nent 3.								
		CHARGE															
		Installation Trunk Side Service - per DS0			OHD	TPP++		334.29bk	57.01bk								
		Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00										
		Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
		Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
Dedicated Tandem Trunk Port Service-per DS1** OH1 OH1MS TDW1P 0.00																	
	* This	rate element is recovered on a per MOU basis and is included	l in the	End Of	fice Switching and T	andem Swit	ching, per MO	U rate elements	3		•						

04/12/02 Page 9 of 9